Supplement A

A review of the effectiveness and cost-effectiveness of personal, social and health education in primary schools focusing on sex and relationships and alcohol education for young people aged 5 to 11 years

FINAL REPORT

Lisa Jones, Geoff Bates, Jennifer Downing, Harry Sumnall, Mark A Bellis Centre for Public Health, Liverpool John Moores University



June 2009

Contribution of authors (alphabetical order)

Geoff Bates	Study selection, data extraction and quality assessment. Contributed to the writing of all sections of the report.
Mark Bellis	Overall management responsibility for the project. Commented on various drafts of the review.
Jennifer Downing	Study selection, data extraction and quality assessment. Contributed to the writing of all sections of the report.
Lisa Jones	Management responsibility for the project and responsible for overall content of the final report. Study selection, data extraction and quality assessment. Contributed to the writing of all sections of the report
Harry Sumnall	Management responsibility for the project. Contributed to study selection and commented on various drafts of the review.

Acknowledgements

In addition to the authors of the report we would like to acknowledge the contribution of Claire Shaw from the Centre for Public Health, LJMU for her assistance in full text screening.



Table of Contents

G	los	ssary	/		vii
Li	st	of al	obrev	viations	.ix
Е	xeo	cutiv	e Su	immary	.xi
1		Intro	oduc	tion	. 1
	1.	1	Aim	s and objectives	. 1
	1.:	2	Res	earch question	. 1
2		Bac	kgro	und	2
	2.	1	Fac	tors influencing alcohol use and sexual health behaviours	.4
	2.	2	Per	sonal, social and health education	.7
	2.3	3	Gov	vernment policy	.7
3		Met	hodc	ology	11
	3.	1	Sea	rch strategy	11
	3.2	2	Incl	usion and exclusion criteria	12
		3.2.	1	Population	12
3.2.2 3.2.3		2	Interventions	12	
		3	Comparator(s)	12	
		3.2.	4	Outcomes	12
		3.2.	5	Study design	13
	3.	3	Stu	dy selection and data extraction strategy	13
	3.4	4	Qua	ality assessment strategy	13
	3.	5	Met	hods of analysis/synthesis	14
		3.5.	1	Effectiveness studies	14
		3.5.	2	Published economic evaluations	14
	3.	6	Der	iving evidence statements and assessing applicability	14
4		Sun	nmar	y of study identification	15
	4.	1	Rev	iew of effectiveness and cost-effectiveness	15
		4.1.	1	Included studies	16
	4.1.2			Excluded studies	17

5	Re	eview	iew of effectiveness			
	5.1	.1 Systematic reviews and meta-analyses		18		
	5.1	1.1	Overview of evidence identified	18		
	5.1	1.2	Summary and evidence statements	19		
	5.2	Alc	cohol education	21		
	5.2	2.1	Overview of evidence identified	21		
	5.2	2.2	Classroom-based programmes led by teachers or external contributors	21		
	5.2	2.3	Single session interventions	27		
	5.2	2.4	Summary and evidence statements	30		
	5.3	Dru	ug education (including alcohol) programmes	34		
	5.3	3.1	Overview of evidence identified	34		
	5.3	3.2	Classroom-based programmes led by teachers	34		
	5.3	3.3	Classroom-based programmes led by external contributors	41		
	5.3	3.4	Multicomponent programmes	48		
	5.3	3.5	Other in school approaches	50		
	5.3	3.6	Summary and evidence statements	56		
	5.4	Se	ex and relationships education	65		
	5.4	4.1	Overview of evidence identified	65		
	5.4	1.2	Abstinence-based programmes	65		
	5.4	4.3	AIDS education programmes	67		
	5.4	1.4	Other in-school approaches	70		
	5.4	1.5	Summary and evidence statements	72		
	5.5	Ge	eneral health education programmes	77		
	5.5	5.1	Overview of evidence identified	77		
	5.5	5.2	General health education programmes	77		
	5.5	5.3	Summary and evidence statements	79		
	5.6	So	cial development programmes	82		
	5.6	5.1	Overview of evidence identified	82		
	5.6	6.2	Programmes with a school-based component	82		

	5.6	.3	Programmes combining school and family-based components	
	5.6	5.4 Summary and evidence statements		91
6	Dis	cuss	ion	97
(6.1	Su	mmary of the review of effectiveness	97
	6.1	.1	Systematic reviews and meta-analyses	97
	6.1	.2	Alcohol education	97
	6.1	.3	Drug education (including alcohol)	97
	6.1	.4	Sex and relationships education	
	6.1	.5	General health education programmes	
	6.1	.6	Social development programmes	
(6.2	Su	mmary of review of economic evaluations	
(6.3	Str	engths and limitations	
	6.3	.1	Quality of the included studies	
	6.3	.2	Applicability	100
	6.3	.3	How and why programmes worked	100
	6.3	.4	Targeting wider risk behaviours	100
(6.4	Re	search recommendations	101
7	Co	nclus	sions	102
8	Re	ferer	ICES	103

Appendices

Appendix 1. Non-statutory framework for PSHE	106
Appendix 2. References to included studies	109
Appendix 3. References to excluded studies	116
Appendix 4. References to studies not available for assessment	153
Appendix 5. Results of the quality assessment	159
Appendix 6. Results of quality assessment for studies included in Jones and collea	agues
(2007)	170
Appendix 7. Conversion table for English key stages and US grade equivalents	175



Tables of figures

Figure 4.1	Summary of study	entification1	5
------------	------------------	---------------	---

Table of tables

Table 2.1. Risk and protective factors for adolescent sexual behaviour, use of contraception,
pregnancy and childbearing4
Table 4.1. Summary of study designs identified for inclusion 16
Table 5.1. Summary table for systematic reviews and meta-analyses
Table 5.2. Summary of programme content: Alcohol education programmes led by teachers
or external contributors21
Table 5.3. AMPS: Alcohol use in 7/8 th grade (Shope et al., 1992; RCT -)
Table 5.4. AMPS: Alcohol misuse in 7/8 th grade (Shope et al., 1992; RCT -)
Table 5.5. AMPS: Curriculum index scores in 7/8 th grade (Shope et al., 1992; RCT -) 23
Table 5.6. PY/PM: intervention effects (Padget et al., 2006; NRCT +)
Table 5.7. Summary of programme content: Single session alcohol education interventions
Table 5.8. Alcohol education: classroom-based programmes
Table 5.9. Alcohol education: single session interventions 33
Table 5.10. Programme content: Drug education programmes led by teachers
Table 5.11. LST: Intervention effects (Botvin et al., 2003; RCT -)
Table 5.12. ALF: Intervention effects (Bühler et al., 2008; RCT +)
Table 5.13. Project Charlie: Intervention effects - dichotomous (Hurry & McGurk, 1997;
Hurry et al., 2000; RCT +)
Table 5.14. Project Charlie: Intervention effects - continuous (Hurry & McGurk, 1997; Hurry
et al., 2000; RCT +)
Table 5.15. Programme content: Drug education programmes led by external contributors 42
Table 5.16. BABES: Intervention effects (Abbey et al., 1990; RCT +)
Table 5.17. BABES Plus: Intervention effects (Hahn et al., 2007; NRCT +)
Table 5.18. Programme content: Multicomponent drug education programmes
Table 5.19. Programme content: Other in-school drug education approaches 50
Table 5.20. Refusal skills intervention: Intervention effects (Corbin et al., 1993; RCT -) 53
Table 5.21. Life Education Centres: Intervention effects (school-level analysis) (Hawthorne
1995; CBA –)
Table 5.22. Drug (including alcohol) education: classroom-based programmes led by
teachers



Table 5.23. Drug (including alcohol) education: classroom-based programmes led	by
external contributors	. 62
Table 5.24. Drug (including alcohol) education: multicomponent programmes	. 63
Table 5.25. Drug (including alcohol) education: other in-school approaches	. 64
Table 5.26. Summary of programme content: abstinence-based programmes	. 65
Table 5.27. SCW: Intervention effects at 18-month follow-up (Denny & Young, 2006; NF	٢СТ
–)	. 67
Table 5.28. Programme components: AIDS education programmes	. 67
Table 5.29. Programme components: other in-school approaches	. 70
Table 5.30. Learning about Parenting: Intervention effects (Masterpasqua et al., 1992; NF	٢СТ
–)	71
Table 5.31. Sex and relationships education: Abstinence-based programmes	. 75
Table 5.32. Sex and relationships education: AIDS/HIV education programmes	. 75
Table 5.33. Sex and relationships education: Other in-school approaches	. 76
Table 5.34. Programme components: General health education programmes	. 77
Table 5.35. General health education programmes	. 81
Table 5.36. Summary of programme content: Single component social developn	nent
programmes	. 82
Table 5.37. Summary of programme content: Multi-component social developn	nent
programmes	. 83
Table 5.38. SSDP: Intervention effects at age 18 (Hawkins et al., 1999; NRCT +)	. 85
Table 5.39. SSDP: Intervention effects at age 21 (Hawkins et al., 2005; NRCT +)	. 86
Table 5.40. SSDP: Intervention effects at age 21 - dichotomous (Lonczak et al., 2002; NF	۲CY
+)	. 87
Table 5.41. SSDP: Intervention effects at age 21 - continuous (Lonczak et al., 2002; NRC);T +)
	. 87
Table 5.42. Developmental drug prevention programme: Unsupervised alcohol use (F	urr-
Holden et al., 2004)	. 90
5.43. Social development programmes	. 95
Table 8.1. Quality assessment: Systematic reviews and meta-analyses	159
Table 8.2. Quality assessment: Randomised controlled trials	160
Table 8.3. Quality assessment: Randomised controlled trials continued	162
Table 8.4. Quality assessment: Non-randomised controlled trials	164
Table 8.5. Quality assessment: Controlled before and after studies	166
Table 8.6. Quality assessment: Uncontrolled before and after studies	168
Table 8.7. Quality assessment for RCTs and NRCTs	170
Table 8.8. Quality assessment for controlled before and after studies	172



Glossary

American school grades	Education is divided into 3 levels: elementary school, junior high (or middle) school and high school (see Appendix 7)
Bias	Deviation of results or inferences from the truth, or processes leading to such deviation. Any trend in the collection, analysis, interpretation, publication or review of data that can lead to conclusions that are systematically different from the truth.
Binge drinking	Consuming large quantities of alcohol over a short period of time. Often associated with drinking to become intoxicated.
Cluster randomisation	A trial where the unit of randomisation is a cluster of participants (e.g. a school).
Controlled Before and After study (CBA)	Intervention groups are tested and data collected before and after the intervention has been administered. Differ from controlled non-randomised trials in that participants are not allocated to intervention or control groups, but rather a 'convenience' control sample is used.
Drug education programmes	Programmes that include a focus on illegal drugs or tobacco in addition to alcohol.
Effect size	Effect size is a term used for a family of indices that measure the magnitude of the relationship between variables or treatment effect. Effect sizes are commonly used in meta-analyses as unlike significance tests these indices are independent of sample size.
General health education programmes	Programmes that are health based but include aspects and outcomes relating to alcohol or sex and relationships
Intention to treat analysis	A method of data analysis in which all participants are analysed in the group they were assigned to at randomisation regardless of treatment adherence.
Internal validity	How well the study has minimised sources of bias and how likely it is that the intervention caused the observed outcomes.
Key stage	Pupils' progress through school is measured in key stages. Each key stage covers a number of school years. Starting at key stage 1 and finishing at key stage 4 (see Appendix 7).
Meta-analysis	The combination of quantitative evidence from a number of studies.
Non-Randomised Controlled Trial	These are trials where participants or clusters are allocated between intervention and control groups but the allocation is not randomised or quasi-randomised (e.g. alternate allocation).
Odds ratio	The odds of the event occurring in one group (e.g.



	intervention) divided by the odds of the event occurring in the other group (e.g. control).
Randomised Controlled Trial	Individuals or, defined groups of individuals (clusters) are randomised to either an intervention or a control group. If well implemented, randomisation should ensure that intervention and control groups only differ in their exposure to treatment.
Risk ratio	The risk of the event in the one group (e.g. intervention) divided by the risk of the event in the other group (e.g. control).
Social development programmes	Programmes that aim to impact upon alcohol use or sex and relationships through social development education
Solomon group four design	Population assigned to one of four groups including two experimental and two control groups. Two groups receive a pre-test and post-test, two groups receive only a post-test
Standardised mean difference	Expresses the size of the intervention effect in each study relative to the variability observed in that study.
Systematic review	A method of locating, appraising and synthesising evidence from primary studies, which adheres to a scientific methodology.
Uncontrolled Before and After Study	Intervention groups are tested and data collected before and after the intervention has been administered. No control group is used for comparison purposes.

List of abbreviations

AAPT	Adolescent Alcohol Prevention Trial
ALF	Allgemeine Lebenskompetenzen und Fertigkeiten
AMPS	Alcohol Misuse Prevention Study
BABES	Beginning Alcohol and Addictions Basic Education Studies
CBA	Controlled Before and After study
CMO	Chief Medical Officers
CWPT	Class wide peer tutoring
CSS	Case-control study
DARE	Drug Abuse Resistance Education
DAW	Drugs at Work
DfES	Department for Education And Skills
DH	Department of Health
ES	Effect Size
FAME	Family Action Model for Empowerment
GBG	Good Behaviour Game
GI	General information
GRAT	Get Real About Tobacco
HLAY	Here's Looking At You
ITT	Intention to treat
LEC	Life Education Centre
LIFT	Linking the Interests of Families and Teachers
LST	Life Skills Training
NICE	National Institute for Health and Clinical Excellence
NR	Not reported
NRCT	Non-Randomised Controlled Trial
OR	Odds Ratio
PADAPE	Preventing Alcohol and Drug Abuse Through Primary Education
PSHE	Personal Social and Health Education
PT	Post-test
PY/PM	Protecting You/Protecting Me
QCA	Qualifications and Curriculum Authority
R+	Rehearsal plus
RCT	Randomised Controlled Trial
RHC	Raising Healthy Children
SCW	Sex Can Wait
SD	Standard Deviation
SE	Standard Error
SR	Systematic Review



- SRE Sex and relationships education
- SSDP Seattle Social Development Project
- STI Sexually transmitted infection
- SYF Say Yes First
- TITH Tuning into Health



Executive Summary

Objectives

This review sought to identify effective and cost-effective interventions and programmes that focus on health literacy and personal skills in relation to alcohol and sex and relationships for primary school aged children.

Methods

The methods for the review followed NICE protocols for the development of NICE public health guidance. Eighteen databases were searched for effectiveness and cost-effectiveness studies published since 1990. Two reviewers independently screened all titles and abstracts. All data extraction and quality assessment was undertaken by one reviewer and checked for accuracy by a second reviewer. Each study was also graded (++, + or -) based on the extent to which the design and execution of the study minimised the potential sources of bias. Results of the data extraction and quality assessment for each study were presented in structured tables and as a narrative summary. Where possible, effect sizes were presented for individual studies, but heterogeneity across the included studies precluded meta-analysis.

Review of effectiveness

A total of two systematic reviews and meta-analyses, and 73 primary studies were included in the review of effectiveness. A broad range of programme approaches were identified including 14 studies which examined alcohol education approaches and nine studies of SRE approaches. Of the remaining studies, 33 studies examined drug education programmes that also focused on alcohol, three studies examined general health education programmes and 16 studies examined interventions which targeted developmental risk factors for later alcohol use and sexual behaviour.

Systematic reviews and meta-analyses

Two reviews were identified that evaluated school-based interventions for primary school aged children. One review focused specifically on the prevention of alcohol use and the other focused on substance use prevention. Both reviews concluded that there was limited evidence to determine which programme approaches were most effective for primary school aged children. In addition, one review suggested that school-based interventions targeting young adolescents may be more effective. Intervention with younger children was identified as most effective when it took place across multiple domains, most typically combining school and family-based intervention.



Evidence statement 1

There is strong evidence from two systematic reviews, which focused on the prevention of alcohol use, to suggest that interventions targeting primary school aged children may be less effective than those that target young adolescents. Interventions targeting alcohol use in primary school aged children may be more effective if they take place in more than one domain, for example by combining school and family components.

Alcohol education

Thirteen studies were identified that examined alcohol education programmes; nine studies were classroom-based curriculums led by teachers or external contributors, and four studies examined one-off intervention sessions. The Protecting You/Protecting Me programme was shown to have significant effects on knowledge about the effects of alcohol on development and the brain, and vehicle safety in relation to the prevention of exposure to drink driving. The long term effects on alcohol use behaviours was examined for three programmes (Alcohol Misuse Prevention Study, Adolescent Alcohol Prevention Trial and Protecting You/Protecting Me) but none of these programme had consistent long term effects on alcohol use.

Evidence statement 2

- 2(a) There is moderate evidence from two RCTs, one NRCT and one CBA study of a classroom-based programme to suggest that an intervention focused on alcohol prevention and vehicle safety can improve knowledge of the effects of alcohol on development and the brain, and vehicle safety in relation to drink driving. This evidence may be only partially applicable to the UK because the programme's emphasis on the prevention of injury through drink driving is only partially relevant to PSHE delivery in primary schools focusing on SRE and alcohol education.
- 2(b) There is insufficient and inconsistent evidence from three RCTs and one NRCT to determine the effects of alcohol education programmes on alcohol use in later years.

Drug education (including alcohol)

Thirty-two studies were identified that examined drug education programmes that included a focus on illegal drugs (and tobacco) in addition to alcohol. Twenty studies reported on 18 classroom-based programmes, led by teachers or external contributors. In addition, four studies reported on programmes that combined in-school approaches with parent education and eight studies reported on a range of other in-school approaches including theatre in

education and a programme based on a retreat format. There was a lack of evidence to determine the effects of drug education approaches on knowledge and attitudes in relation to alcohol use, and no consistent evidence for the effects of these programme on personal and social skills. There was evidence from one study of a culturally tailored programme for Native American students that the programme had long-term, positive effects on alcohol use.

Evidence statement 3

3(a) There is moderate evidence from one RCT to suggest that a culturally tailored skills training intervention for Native American students may have long-term effects on alcohol use. However, this evidence is not applicable to the UK given the cultural specificity of this programme. There is insufficient and inconsistent evidence from four RCTs, four CBA studies and one UBA study to determine the effects of other drug education approaches on alcohol use in later years.

Sex and relationships education

Nine studies were identified that examined seven programmes focusing on different approaches to sex and relationships education. Young people aged 10-12 years who participated in the Sex Can Wait programme reported long-term (>12 months) improvements in knowledge relating to the abstinence-based curriculum. The curriculum addressed self-esteem, reproductive anatomy and physiology, changes associated with puberty, values and decision-making skills. In addition, at the 18 month follow-up, participants in the Sex Can Wait programme were less likely to report that they had been involved in sexual activity in the past 30 days compared to a control group. Two programmes, an abstinence-orientated empowerment programme (FAME) and an HIV/AIDS prevention programme designed to promote communication, were shown to have effects on communication with parents. Participants in the FAME programme reported short term (<6 months) improvements in their communication with their parents, and the effects of the HIV/AIDS prevention programme resulted in medium term (up to 12 months) increases in communication approaches on attitudes and values relating to sexual health.



Evidence statement 4

- 4(a) There is weak evidence from two NRCTs to suggest that an abstinence education programme that targeted children aged 10-12 years can improve sexual health knowledge, but the long term impact on sexual behaviours is less clear. This evidence may be directly applicable to the UK because the curriculum topic and content of this programme is relevant to PSHE delivery in primary schools focusing on SRE and alcohol education.
- 4(b) There is moderate evidence from one RCT to suggest that SRE programmes targeting communication, such as I Want to, I Can...Prevent HIV/AIDS, can improve parent and child communication about sexual health. This evidence may be directly applicable to the UK because the curriculum topic and content of these programmes are relevant to PSHE delivery in primary schools focusing on SRE and alcohol education.
- 4(c) There is inconsistent and insufficient evidence from two NRCTs, one CBA study and two UBA studies to determine the effectiveness of SRE programmes on attitudes and values relating to sexual health.

General health education

Three studies examined general health education programmes that included modules or curriculum topics related to alcohol education or SRE. All three studies were rated poorly in terms of their design and conduct and there was insufficient evidence to determine the effects of these programmes on outcomes relating to alcohol use and sexual health.

Evidence statement 5

5(a) There is insufficient and inconsistent evidence from two CBA studies and one UBA study to determine the effects of general health education programmes that targeted primary school age children on outcomes related to alcohol use and sexual health.

Social development

Sixteen studies were identified that examined seven programmes focused on social development interventions designed to positively influence behaviour in later life including alcohol use and sexual health behaviour. The majority of these programmes combined school- and family-based intervention components and were shown to have long term (>12 months) positive impacts on attachment to school, academic performance and problem behaviour, and improve social skills. At age 18, participants who received the Seattle Social Development Programme in grades 1-6 reported drinking on fewer occasions than control students and were less likely to have had sex, sex with multiple partners, or to have been pregnant. At age 21 although there was no effect of the intervention on alcohol use,



compared to control students, participants reported an older age of first sexual experience, fewer lifetime sexual partners, were less likely to have been pregnant and were more likely to use condoms. A Dutch study of the Good Behavior Game, which targeted in classroom behaviours, found that although participation did not have a significant effect on past year or past month alcohol use among 10-13 year olds, there was a reduced rate in the growth of alcohol use between these ages for students who participated in the programme. Follow up of students who participated in the Baltimore-based study of the programme as young adults, showed that participation in the Good Behavior Game was associated with lower rates of lifetime alcohol use and dependence.

Evidence statement 6

- 6(a) There is moderate evidence from one RCT, three NRCTs and one CSS study to suggest that programmes, which target social development and combine school and family-based components, may positively impact on attachment to school and academic performance. This evidence may only be partially applicable to the UK because these programmes were developed and evaluated in the USA, and the findings may not be generalisible beyond the populations studied.
- 6(b) There is moderate evidence from three RCTs, one NRCT and one CSS study to suggest that programmes, which target social development and combine school and family-based components, may have a positive impact on problem behaviours and social skills. This evidence may only be partially applicable to the UK because these programmes were developed and evaluated in the USA, and the findings may not be generalisible beyond the populations studied.
- 6(c) There is moderate evidence from two NRCTs to suggest that a social development programme, which combined school and family-based components, may have long term impacts on alcohol use and sexual behaviour in young adulthood. This evidence may only be partially applicable to the UK because these programmes were developed and evaluated in the USA, and the findings may not be generalisible beyond the populations studied.
- 6(d) There is strong evidence from three RCTs to suggest that the Good Behavior Game, which targeted behaviours in the classroom, may impact on alcohol abuse and dependence in adulthood and slow the rate of alcohol use in adolescence. This evidence may be directly applicable to the UK because although the programme was developed and evaluated in the USA, it has been replicated in populations outside of the USA.



Review of published economic evaluations

No published economic evaluation studies were identified for inclusion in the review.

Discussion and conclusions

Overall, this review of the effectiveness and cost-effectiveness of PSHE in primary schools focusing on SRE and alcohol education has highlighted a number of weaknesses in the evidence base. There is evidence that social development programmes, which combine school- and family-based components, may have long term impacts on school attachment, social skills, alcohol use and sexual health. However, the applicability of these programmes warrants further study in a UK context before widespread implementation can be supported. There is a lack of clear, long-term evidence for the effectiveness and cost-effectiveness of other approaches to SRE and alcohol education, and further good quality, UK-based research is needed.



1 Introduction

1.1 Aims and objectives

This review was undertaken to support the development of guidance by the National Institute for Health and Clinical Excellence (NICE) aimed at promoting school, college and community-based personal, social and health education (PSHE),^{1,2} with particular reference to sexual health behaviour and alcohol. As such, the review sought to identify effective and cost-effective interventions and programmes that focus on health literacy and personal skills in relation to alcohol use and sexual health.

1.2 Research question

The following four research questions were addressed:

- 1 What services, interventions, programmes, policies or strategies for children aged 5-11 years old are effective and cost-effective in contributing to the achievement of the "Every child matters" outcomes for PSHE, related to sexual health and alcohol?
- 2 What elements/components of those services, interventions, programmes, policies or strategies for children aged 5-11 years old are effective and cost-effective in contributing to the achievement of the "Every child matters" outcomes for PSHE, related to sexual health and alcohol?
- 3 How can schools, governors, parents and carers, families and communities contribute to the effective and cost-effective delivery of PSHE in particular, sex and relationship and alcohol education to achieve health-related "Every child matters" outcomes in children aged 5-11 years old?
- 4 In what ways can professionals, practitioners, peers, volunteers and services in education and health settings provide effective and cost-effective support for the delivery of PSHE – in particular sex and relationship and alcohol education – in schools and communities?

² Since September 2008, PSHE has been referred to as Personal, Social, Health and Economic education (MacDonald, 2009) and this subject is covered under the umbrella term PSHE.



¹ The term personal, social and health education is used in its broadest sense to refer to schoolsbased programmes and curriculum approaches that are intended to promote personal development and wellbeing.

2 Background

Early onset sexual activity has been related to sexual risk-taking behaviour. The second National Attitudes and Lifestyles Survey (NATSAL, 2000) found that 30% of men and 26% of women in Britain reported having sex before the age of 16 (Wellings et al., 2001). Furthermore, Tripp and Viner (2005) found that 18% of boys and 15% of girls had experienced full sexual intercourse before age 15 years, with similar proportions having engaged in oral sex (Tripp & Viner, 2005). Unprotected sex at first intercourse is a risk factor for poor sexual health outcomes, for example teenage pregnancy, and data suggest that the use of contraception at first sex is declining (Wellings et al., 2001). Having unsafe sex at first intercourse is, in part through lack of knowledge, lack of access to contraception, lack of skills and self efficacy to negotiate contraception, having sex whilst under the influence of drugs or alcohol, or inadequate self efficacy to resist pressure. Around 10% of boys in the United Kingdom report that they were under the influence of drugs or alcohol when they first had sex, and 11% of girls report being pressurised by their partner when they first had sex. Of those under 16 years who have ever had sex, about a third to a half of both sexes report ever having had unsafe sex (Tripp & Viner, 2005). The result of this can be seen through the 19% increase in diagnoses of the top five sexually transmitted infections (STIs) for young people under age 16 years (Chlamydia, gonorrhoea, syphilis, warts, herpes) (Health Protection Agency, 2008).

Furthermore, the rate of teenage pregnancy in the UK has been one of the highest in Western Europe for the past ten years (Unicef, 2001), with 7.1% of all live births in 2008 attributed to mothers aged under 20 years (Department of Health, 2009). Although there are examples of progress in local areas (Department for Education and Skills, 2006), nationally initiatives to reduce the UK's under-18 teenage pregnancy rate by 50% in line with the 2010 targets (Social Exclusion Unit, 1999) have been largely unsuccessful. With current under 18 conception rates for England at 41.7 per 1000 young women aged 15-17 years, compared to 43.6 in 2000 (Office for National Statistics and Teenage Pregnancy Unit, 2009). The impact of teenage pregnancy on the mother can be lower educational achievement, a greater likelihood of raising a child as a single parent and fewer opportunities to gain good employment resulting in a low socio-economic status. Further, the child of a teenage parent is more likely to experience poor health, be a low academic achiever, be involved in crime, misuse drugs and alcohol and be more likely to become a teenage parent themselves, thereby perpetuating the cycle (Department of Health, 2009; Unicef, 2001). Wellings and colleagues (2001) showed that young people who leave school later after gaining qualifications are more likely to be sexually competent; more likely to use contraception at



first sexual intercourse and are less likely to experience pregnancy. Long Acting Reversible Contraception (LARC) and Emergency Hormonal Contraceptives (EHC) may be useful methods to prevent pregnancy yet there is a greater need for effective prevention interventions to address this issue. NATSAL participants aged 16-19 years reported that lessons at school were their most common source of sexual health information (Wellings et al., 2001), which highlights the impact of school-based prevention interventions.

The relationship between sexual risk-taking, substance misuse and anti-social behaviour has been previously highlighted (Independent Advisory Group on Sexual Health and HIV, 2007). A quarter of young people who binge drink become involved in anti-social behaviour and young people are most likely to begin drinking alcohol between age 11 to 15 years (Clemens et al., 2008). Over 30% of young people (aged ≤15 years) in the UK report ever being drunk two or more times compared to an average of 15% across other OECD (Organisation for Economic Cooperation and Development) nations (UNICEF, 2007). In England in 2007, 20% of 11 year olds reported that they had ever drank alcohol, with this figure increasing to 54% of 11-15 year olds reported ever drinking alcohol. Furthermore, the quantity of alcohol consumed weekly by 11 to 13 year old boys has increased by 6.4 units per week to 11.9 in 2006 compared to the 2001 figure. Girls consumed 8.4 units, up from 2.7 in 2001 (Department of Health, 2007). These data support the view that children who begin drinking young (typically below the age of 13) drink more often and in greater quantities and are more likely to drink to intoxication than those who delay drinking.

As with alcohol dependence and abuse; vulnerability to alcohol misuse in later adolescence appears greatest among those who begin drinking prior to age 13 (CMO, 2009). Alcohol dependence has also been associated with other psychopathological diagnoses such as nicotine dependence, drug dependence, and anti-social personality disorder (McGue & lacono, 2008). Evidence suggests that early aggressive behaviours, including direct aggression, fighting, and hitting and those defined as conduct disorders have consistently are related to early initiation of underage drinking (Spoth et al., 2008). A further study by Malone and colleagues (2004) revealed a relationship between symptoms of alcohol dependence at age 17 and antisocial behaviour by age 20 in males, concluding that alcohol use may be the reason for antisocial behaviour persisting into adulthood. Regular alcohol use behaviour is already established in a minority of school children aged 11 to 12 years (around 1%) and this increases with age (Clemens et al., 2008). This highlights the fact that prior to beginning secondary school some young people will have already experienced a situation requiring self-efficacy, negotiation skills and knowledge of the negative aspects associated with risk activities, which could be influential in their decision not to participate (or to defer participation) in such activities until they are older. Recent draft guidance on alcohol



consumption for young people from the Chief Medical Officers recognises that there are no safe drinking limits for young people and recommends that those under the age of 15 years should not consume any alcohol (CMO, 2009).

2.1 Factors influencing alcohol use and sexual health behaviours

Socio-economic status and in particular high levels of deprivation are associated with increased alcohol use and poor sexual health outcomes (Bellis et al., 2009). In addition, evidence supports the view that early risk-taking behaviour such as sexual behaviour and alcohol misuse may be influenced by early onset puberty which can influence social decision-making (Bellis et al., 2006; Costello et al., 2007; Goodson et al., 1997). Furthermore, young people who mature earlier are more likely to socialise with older peers and subsequently participate in high-risk behaviours (Zimmer-Gembeck et al., 2008). Young people's sexual and drinking behaviour can also be a result of young people attempting to demonstrate their developing maturity, as a means of experimenting with new found curiosities and in an attempt to mimic perceived adult behaviours (see Table 2.1). Young people's drinking behaviour can also be strongly influenced by parental drinking patterns, for example, children of problem drinkers are more likely to develop drinking problems (van Der Vost et al., 2009). Positive social norms regarding drinking, either through family, peers or the media can influence young people's attitudes and behaviours. This can be the case, particularly if parents do not discourage their child's alcohol use.

Table	2.1.	Risk	and	protective	factors	for	adolescent	sexual	behaviour,	use	of
contraception, pregnancy and childbearing											

Community	Community disadvantage and disorganisation				
	+ High level of education				
	- High unemployment rate				
	+ High income level				
	- High crime rate				
Family	Structure and economic advantage				
	+ Two (vs. one) parent families				
	- Changes in marital status				
	+ High level of parent's education				
	+ High parental income				



	Positive family dynamics and attachment							
	+ Parental support and family connectedness							
	+ Sufficient parental supervision and monitoring							
	Family attitudes about and modelling of sexual risk taking and early childbearing							
	- Mother's early age at first sex and first birth							
	- Single mother's dating and cohabitation behaviours							
	+ Conservative parental attitudes about premarital or teen sex							
	+ Positive parental attitudes about contraception							
	- Older sibling's early sexual behaviour and age of first birth							
Peer	Peer attitudes and behaviours							
	+ High grades amongst friends							
	- Peer's substance use and delinquent and non-normative behaviour							
	- Sexually active peers (or perception thereof)							
	+ Positive peer norms or support for condom or contraceptive use							
Partner	Partner attitudes							
	+ Partner support for contraception							
Teen	Biological antecedents							
	- Older age and greater physical maturity							
	- Higher hormone levels							
	Ethnicity							
	+ Being White (vs. Black or Hispanic)							
	Attachment to and success in school							
	+ Good school performance							
	+ Educational aspirations and plans for the future							
	Attachment to religious institutions							
	+ Frequent religious attendance							



Problem or risk-taking behaviours				
- Tobacco, alcohol or drug use				
- Problems behaviours or delinquency				
- Other risk behaviours				
Emotional distress				
- Higher level of stress				
- Depression				
- Suicide ideation				
Characteristics of relationship with partners				
- Early and frequent dating				
- Going steady, having a close relationship				
- Greater number of romantic partners				
- Having a partner 3 or more years older				
Sexual abuse				
- History of prior sexual coercion or abuse				
Sexual beliefs, attitudes and skills				
+ Conservative attitudes towards premarital sex				
+ Greater perceived susceptibility to pregnancy, STDs/HIV				
+ Importance of avoiding pregnancy, childbearing and STDs				
+ Greater knowledge about contraception				
+ More positive attitudes about contraception				
+ Greater perceived self-efficacy in using condoms or contraception				

Key: + protective factor; - risk factor

Source: Kirby D (2001) Emerging Answers: Research Findings on Programs to Reduce Teen Pregnancy. Washington: National Campaign to Prevent Teenage Pregnancy



2.2 Personal, social and health education

From 2011, PSHE education is expected to become a statutory requirement for both primary and secondary schools; however, currently PSHE education consists of a non-statutory framework (with citizenship). At Key Stages 1 and 2 there are four broad themes to the curriculum:

- Developing confidence and responsibility and making the most of pupils' abilities;
- Preparing to play an active role as citizens;
- Developing a healthier, safer lifestyle; and
- Developing good relationships and respecting differences between people.

Sex and relationships education curriculum and standards guidance (Department for Education and Employment, 2000) states that at primary school level, sex and relationship education (SRE) should contribute to PSHE education by ensuring that all children:

- Develop confidence in talking, listening, and thinking about feelings and relationships;
- Are able to name parts of the body and describe how their bodies work;
- Can protect themselves and ask for help and support;
- Are prepared for puberty

Alcohol education is also located within wider provision for PSHE, and according to curriculum guidance (Qualifications and Curriculum Authority, 2003) should enable pupils to increase knowledge and understanding, explore attitudes, and develop skills for making healthy, informed choices, including choices about drugs, alcohol and tobacco. The non-statutory framework for PSHE is presented is Appendix 1.

2.3 Government policy

The public health White paper *Choosing Health: making healthy choices easier* (2004) acknowledges and supports the need for comprehensive PSHE. The White paper promotes the *healthy schools programme* which encourages a whole school approach to health that includes comprehensive PSHE that incorporates sex, relationships and alcohol education (Department of Health, 2005). Furthermore, schools must comply with this element if they are to achieve their Healthy School status (Department of Health, 2004). Additional guidance for schools recommends that all young people, starting from early key stages, receive age and ability appropriate drug (including alcohol) education that helps to develop their knowledge, skills, attitudes and understanding of drugs in order to build an appreciation of healthy lifestyles (Department for Education and Skills, 2004a). Currently, compulsory

education on sex and alcohol is via the national science curriculum and addresses the effects of alcohol on mental and physical health of a person. Additionally, this curriculum addresses the human reproductive cycle, including conception, adolescence and sexually transmitted infections. This level of education is reserved for key stage three curricula designed for secondary school pupils (Qualifications and Curriculum Authority, 2007). However, at key stage 2 pupils are introduced to the topic of alcohol and its effects on health and human growth is addressed from key stage 1 (Qualifications and Curriculum Authority, 1999). Sex and relationship and alcohol education in schools, delivered through PSHE, supports the healthy living blueprint for schools and the extended schools guidance to improve access to general health as well as sexual health advice and services (Department for Education and Skills, 2004b; 2006a; 2007). A curriculum that adopts a prevention approach to sex and substance use also supports the accelerated teenage pregnancy strategy which recognises that those local authorities (LA) and primary care trusts that are expected to reach the 2010 teenage pregnancy targets have given PSHE a high priority within schools and have received LA support to develop comprehensive SRE in all schools. Furthermore, LAs successfully reducing the rates of teenage pregnancy were those that provided SRE training packages that were taken up by teachers (Department for Education and Skills, 2006b). The focus on school and parental responsibility remains a key issue in the education of young people. Work has been carried out as part of the Extended Schools Programme to support parents and aims to tackle early emotional and behavioural problems. Evidence shows that parenting programmes are effective at tackling conduct disorders and improving parenting (Lindsay et al., 2008). The Parenting Early Intervention Programme (PEIP) targets parents of children age 8 to 13 years at risk of negative outcomes and will be rolled out across all local authorities from April 2009 (Lindsay et al., 2008).

The latest review of the National Strategy for Sexual Health and HIV (Christophers et al., 2008) also supports statutory and comprehensive SRE and furthermore it calls for the introduction of SRE for young people not in education. In addressing the wider remit for sexual health the aims of the strategy are to:

- Reduce transmission of HIV and STIs
- Reduce prevalence of undiagnosed HIV and STIs
- Reduce unintended pregnancy
- Improve health and social care for people living with HIV
- Reduce the stigma associated with HIV and STIs



8

Following on from this strategy the review aims to focus on those issues related to the strategy that are likely to accelerate its implementation with five key strategic areas aiming to:

- Prioritise sexual health as a key public health issue and sustain high-level leadership at local, regional and national levels
- Build strategic partnerships
- Commission for improved sexual health
- Investing more in prevention, and
- Deliver modern sexual health services

The Alcohol Harm Reduction Strategy for England was published in 2004. Its four key aims were to:

- Improve the information available to individuals and to start the process of change in the culture of drinking to get drunk
- To better identify and treat alcohol misuse
- To prevent and tackle alcohol-related crime and disorder and deliver improved services to victims and witnesses
- To work with the alcohol industry to tackle the harms caused by alcohol.

Subsequently the next steps in the national alcohol strategy (Department of Health, 2007) aim to further reduce the harm associated with alcohol misuse by working to:

- Ensure that laws and licensing powers protect young people and successfully address the issues relating to irresponsibly managed premises.
- Focus on the minority of drinkers who cause or experience the most harm, such as: young people under age 18 years who drink; those 20-24 years who binge drink; and harmful drinkers.
- Shape the environment so that it encourages sensible drinking.

Following from this guidance reducing alcohol-related hospital admissions is now a measure of performance. The alcohol Improvement Programme is supporting Primary Care Trusts (PCTs) to address the needs of patients who consume harmful or hazardous levels of alcohol. Improve the health messages, and in particular alcohol unit information, on labels and work with the alcohol industry to encourage responsible promotions. Further work is currently being carried out in order to address alcohol-related influences on young people. The Home Office is in the process of consulting on the forthcoming mandatory licensing

code of practice which aims to tackle irresponsible sales of alcohol and associated crime. This incorporates a commitment to examine the impact of alcohol advertising and the influence of displaying alcohol (Home Office, 2009).

The Department for Children, Schools and Families (formerly the Department for Education and Skills) recommends a joint approach between Every Child Matters and the drug strategy in order to prevent drug harm (including alcohol). They propose three main objectives, which are to:

- Reform delivery and reforming delivery through closer links between the Updated National Drug Strategy and the Every Child Matters: Change for Children locally, regionally and nationally
- Ensuring provision is built around the needs of vulnerable children and young people. Including more focus on prevention and early intervention with those most at risk, with drug misuse considered as part of assessments, care planning and intervention by all agencies providing services for children, including schools
- Building services and workforce capacity. Developing a range of universal, targeted and specialist provision to meet local needs and ensure delivery of workforce training to support it.

3 Methodology

3.1 Search strategy

Systematic searches of electronic databases and websites were undertaken to identify studies that examined the effectiveness and/or cost-effectiveness of alcohol education and/or SRE delivered in isolation or as part of a wider programme of study such as PSHE or its equivalents. Searches were conducted across a range of health, education and social care databases as shown in Box 3.1.

Box 3.1. Health, education and social care databases

- ASSIA (Applied Social Science Index and Abstracts)
- CINAHL (Cumulative Index of Nursing and Allied Health Literature)
- Database of Abstracts of Reviews of Effectiveness (DARE)
- The Cochrane Library
- EMBASE
- ERIC
- British Education Index
- Australian Education Index
- HMIC (or Kings Fund catalogue and DH data)
- MEDLINE
- PsycINFO
- Sociological Abstracts
- Social Science Citation Index
- EPPI Centre databases
- The Campbell Collaboration
- C2-SPECTR & C2-PROT Campbell Collaboration

Economic evaluation studies were identified by searching the following major health economics databases:

- NHS Economic Evaluations Database (NHS EED),
- EconLit

3.2 Inclusion and exclusion criteria

3.2.1 Population

Studies were eligible for inclusion if they included children aged 5 to 11 years old in full time education. This included children in primary schools and those receiving education outside of a mainstream school setting including:

- children receiving home education,
- children receiving education in pupil referral units.

3.2.2 Interventions

Studies were eligible for inclusion if they examined interventions that focused on SRE and/or alcohol education. Relevant intervention approaches included:

- Interventions and programmes agreed, planned or delivered by teachers or other professionals.
- Interventions and programmes planned and/or delivered by external agencies and individuals.
- Intervention involving the 'informal' and extended school curriculum.
- Peer led education

3.2.3 Comparator(s)

Studies were eligible for inclusion if they compared the intervention of interest against a no intervention control or against another intervention approach. Studies that did not include a control group for comparison were also eligible for inclusion.

3.2.4 Outcomes

Studies were eligible for inclusion if they examined the following key outcomes in relation to alcohol education and sex and relationships education:

- Knowledge and understanding,
- Personal and social skills,
- Attitudes and values,
- Health and social outcomes related to alcohol use and sexual health.



3.2.5 Study design

Systematic reviews, randomised controlled trials, controlled non-randomised studies and, controlled and uncontrolled before and after studies that compared a school-based intervention against no intervention or another type of intervention were eligible for inclusion in the assessment of effectiveness.

Studies were eligible for inclusion in the assessment of cost-effectiveness if they were economic evaluations conducted alongside trials, modelling studies and analyses of administrative databases. Only full economic evaluations that compared two or more options and considered both costs and consequences (including cost-effectiveness, cost-utility and cost-benefit analyses) were included.

3.3 Study selection and data extraction strategy

All titles and abstracts retrieved were screened independently by two reviewers (LJ, GB, JD, HS) according to the inclusion/exclusion criteria described above. Disagreements were resolved through consensus and where necessary a third reviewer was consulted. Relevant articles were retrieved in full and full text screening was undertaken independently by two reviewers (LJ, GB, JD, CS).

One reviewer (LJ, GB, JD) independently extracted and assessed the quality of the individual studies into an Access database. All data extraction and quality assessment were independently checked for accuracy by a second reviewer. The results of the data extraction are presented in an addendum to this report.

3.4 Quality assessment strategy

The quality of the studies was assessed according to criteria set out in the NICE Centre for Public Health Excellence Methods Manual (2009). Each of the effectiveness and cost-effectiveness studies was graded using a code, ++, + or – based on the extent to which the potential sources of bias had been minimised:

- ++ All or most of the criteria have been fulfilled. Where they have not been fulfilled the conclusions are thought very unlikely to alter.
- + Some of the criteria have been fulfilled. These criteria that have not been fulfilled or not adequately described are thought unlikely to alter the conclusions.
- Few or no criteria have been fulfilled. The conclusions of the study are thought likely or very likely to alter.



3.5 Methods of analysis/synthesis

3.5.1 Effectiveness studies

The results of the data extraction and quality assessment for each study of effectiveness are presented in structured tables and as a narrative summary. The possible effects of study quality on the effectiveness data and review findings are also discussed within the text of the review.

Studies are grouped according to intervention approach (e.g. teacher or external contributor, curriculum or whole school approach) and the outcomes examined. Where sufficient data were available, intervention effect sizes have been calculated and presented as risk ratios (RR) for dichotomous data and as mean differences for continuous data. Where study authors reported intervention effect sizes, these have been extracted directly as RRs or odds ratios (OR) as reported in the original publication. Forest plots were generated for single studies using RevMan (version 5) and are presented in an addendum to this report.

Heterogeneity between the included studies was assessed by considering differences in (a) the study population, (b) intervention approach, (c) outcome measures, and (d) study quality. However, given the anticipated heterogeneity between the included studies it was judged to be unlikely that pooling would be appropriate or feasible.

3.5.2 Published economic evaluations

No published economic evaluations were identified.

3.6 Deriving evidence statements and assessing applicability

Evidence statements were derived based on the strength of the evidence in relation to intervention approach and the outcomes examined. The strength of evidence was determined by considering the quality, quantity and consistency of the evidence presented in the included studies.

Each evidence statement was assessed to judge applicability based on the similarity of the population, setting, intervention and outcomes of the included studies to the criteria outlined in the review question.



4 Summary of study identification

4.1 Review of effectiveness and cost-effectiveness

A total of 7,629 references were identified from the literature searches. Following screening of titles and abstracts, 572 studies were identified as potentially relevant. Of these, 17 references were for foreign language articles, 10 were for conference abstracts and 44 were not available in time for assessment. References for these articles are presented in Appendix 4. A total of 501 full text articles were screened for inclusion. The process of study identification is summarised in Figure 4.1.



*Includes a further 14 studies included in Jones et al (2007)

**Not identified in the literature searches conducted for this review

Figure 4.1. Summary of study identification



4.1.1 Included studies

A total of 68 articles met the criteria for inclusion in the review of effectiveness, including 14 articles that were included in a systematic review of the effectiveness and cost-effectiveness of interventions in primary and secondary schools to prevent and/or reduce alcohol use, previously conducted by the lead author and colleagues (Jones et al., 2007). In addition, six articles included in the previous review but not identified in the searches for this review met the criteria for inclusion and one article was identified through reference checking. Therefore, a total of 75 articles were included in the review of the effectiveness. No articles met the criteria for inclusion in the review of cost-effectiveness.

Studies were initially grouped according to whether they focused on SRE or alcohol education. However, an additional group of studies (n=16) examined interventions which targeted developmental risk factors for later alcohol use and sexual behaviour and these were grouped together as social development programmes. Initially the majority of the articles identified were grouped together as alcohol education (n=50), however these articles were further subdivided into those programmes which focused specifically on alcohol (n=14), substance use including alcohol (n=33), or general health education (n=3).

Of the 75 articles identified for inclusion, two articles were systematic reviews/meta-analyses, 33 articles reported on randomised controlled trials (RCTs) and 15 articles reported on non-randomised controlled trials (NRCTs). In addition, 25 articles reported on observational studies, 11 of which did not include a control group. The range of study designs identified for inclusion in the review is presented in Table 4.1.

Section	Total	Study design					
		SR/MA	RCT	NRCT	СВА	UBA	
Alcohol education	14	1	10	1	1	1	
SRE	9	0	2	3	1	3	
Drug education	33	1	12	5	9	6	
Health education	3	0	0	0	2	1	
Social development	16	0	9	6	1	0	
Total	75	2	33	15	14	11	
SR/MA – systematic review or meta-analyses; RCT – randomised controlled trial; NRCT – nonrandomised controlled trial; CBA – controlled before and after study; UBA – uncontrolled before and after study							

 Table 4.1. Summary of study designs identified for inclusion

Over 80% of the articles identified reported on studies conducted in the USA (n=59). Six articles reported on studies conducted in the UK, and the remaining eight articles reported



on studies conducted in Canada (n=3), Australia (n=2), Germany, Mexico, and The Netherlands (all n=1).

4.1.2 Excluded studies

A total of 433 studies did not meet the criteria for inclusion in the review. References for the excluded studies are presented in Appendix 3. Excluded studies were grouped according to the reason for exclusion, studies were excluded because: (1) the intervention examined did not focus on SRE or alcohol education (n=184); (2) the intervention targeted groups of atrisk or high-risk young people (n=49); (3) the study did not meet the design criteria for inclusion (n=137); (4) the intervention was not school-based (n=13); or (5) the population targeted was not relevant to the review question (n=50).



5 Review of effectiveness

5.1 Systematic reviews and meta-analyses

5.1.1 Overview of evidence identified

Two reviews were identified that evaluated school-based interventions for primary school aged children (Spoth et al., 2008; Gottfredson & Wilson, 2003). Spoth and colleagues (2008) specified interventions that addressed alcohol use and Gottfredson and Wilson (2003) reviewed studies aiming to prevent substance use, including alcohol.

Quality Assessment

Both reviews (Gottfredson and Wilson 2003; Spoth et al., 2008) were coded 'SR ++' for quality. Both were well conducted reviews, which stated an appropriate research question, fully described the literature search process and methodologies applied and assessed the quality of included studies.

Findings

Gottfredson and Wilson (2003; SR ++) reviewed 94 studies of interventions to reduce problem behaviours in children with the long-term aim of reducing substance abuse. They reported that interventions in primary-school aged children were less effective than programmes delivered to adolescents and that only programmes targeted at middle-school aged children produced evidence clearly indicating reductions in alcohol or any other drug use. A key focus of the review was on high risk students and interventions were reported to be more effective when delivered to higher risk children. However, this finding was based on the results of only five studies and the difference in the effect sizes calculated was not significant. Briefer programmes of 4.5 months or less were found to be generally as effective as longer term interventions and peer only delivered interventions were the most effective form of intervention.

Likewise, Spoth and colleagues (2008; SR ++) concluded that there was only limited research evidence on interventions that targeted emerging alcohol use amongst primary school-aged children. Programme effects on risk behaviours for alcohol use, such as aggressive or problem behaviour, rather than alcohol use *per se* were reported in studies targeting younger children and students were rarely followed up into middle or high school. The authors reported that interventions in primary school-aged children were more effective if they took place in more than one domain, and studies typically included school and family based components.


5.1.2 Summary and evidence statements

Two systematic reviews were identified for inclusion. The review by Spoth and colleagues (2008; SR ++) focused specifically on the prevention of alcohol use, whereas Gottfredson and Wilson (2003; SR ++) focused on substance use prevention.

Both reviews concluded that there was limited evidence to determine which programme approaches were most effective for primary school aged children, and Gottfredson and Wilson (2003; SR ++) suggested that school-based interventions targeting young adolescents may be more effective. Spoth and colleagues (2008; SR ++) suggest that intervention with younger children may be most effective when it takes place across multiple domains, most typically combining school and family-based intervention.

Evidence statement 1

There is strong evidence from two systematic reviews¹, which focused on the prevention of alcohol use, to suggest that interventions targeting primary school aged children may be less effective than those that target young adolescents. Interventions targeting alcohol use in primary school aged children may be more effective if they take place in more than one domain, for example by combining school and family components.

Gottfredson & Wilson, 2003 (SR ++); Spoth et al., 2008 (SR ++)



Author (Year)	Design	Inclusion/Exclusion	Number of studies	Findings
Gottfredson & Wilson, 2003	SR ++	Interventions to reduce problem behaviours among children/youth that measured alcohol or other drug use	94 studies included	Cognitive-behaviourally based prevention programmes were more effective at reducing substance use when delivered to high-risk than general school population studies, but this was based on just 5 studies and the difference was non-significant. Peer alone delivered interventions were most effective.
Spoth et al., 2008	SR ++	Intervention studies that reduce problem behaviours in children and include outcomes related to substance use	41 studies included	Few elementary school interventions were followed up long enough to test their effect on alcohol use. A number of interventions showed significant reductions in aggression and disruption.

5.2 Alcohol education

5.2.1 Overview of evidence identified

A total of 13 primary studies were identified that examined alcohol education programmes targeting children aged 11 years and under. Nine studies were classroom-based curriculums led by teachers or external contributors, and four studies examined one-off intervention sessions.

5.2.2 Classroom-based programmes led by teachers or external contributors

Nine studies were identified that examined four classroom-based alcohol education programmes. Shope and colleagues (1992) reported on the Alcohol Misuse Prevention Study (AMPS) which examined a social pressures resistance-training curriculum for children in fifth and sixth grade. Gamble and Burgess (1994) reported on an alcohol awareness programme for fifth grade students that emphasised the negative effects of alcohol and Donaldson and colleagues (1995; 2000) reported on the Adolescent Alcohol Prevention Trial (AAPT), a social influence-based programme for students in fifth or seventh grade. Five studies (Bell et al., 2005a, 2005b, 2007; Bohman et al., 2004; Padget et al., 2006) reported on the Protecting You/Protecting Me (PY/PM) programme which taught children in the first to fifth grade about alcohol and vehicle safety. A summary of the content for each of these programmes is presented in Table 5.2.

Table 5.2.	Summary	of programme	content:	Alcohol	education	programmes	led by	y
teachers o	or external o	contributors						

Programme	Reference(s)	Programme components
AMPS	Shope et al., 1992	 Teacher-led social pressures resistance training curriculum 4 sessions delivered over 4 weeks in first year (5th and 6th grade classes) 3 booster sessions in second year (5th grade classes only)
Alcohol awareness programme	Gamble & Burgess, 1994	 8 lessons delivered over 10 weeks Lessons covered the effects of alcohol on mind and body, decision making skills, media influence and problems associated with alcohol Delivered by classroom teachers
AAPT	Donaldson et al., 1995; 2000	 Delivered by trained project staff Four conditions 8 lessons of resistance skills training 8 lessons of normative education 10 lessons of combined resistance skills training and normative education 4 lessons information only control
Protecting You/Protecting Me	Bell et al., 2005a; 2005b; 2007; Bohman et al., 2004; Padget et al., 2006	 Teacher or peer led (high school students) alcohol prevention and vehicle safety programme Programme taught in grades 1-5 8 lessons per grade; 1 hour per lesson



5.2.2.1 Alcohol Misuse Prevention Study (AMPS)

The AMPS curriculum was designed as a social pressures resistance-training curriculum with the aim of teaching students about alcohol use and misuse. Shope and colleagues (1992) examined a version of the curriculum delivered over two years with four sessions delivered over four weeks in the first year and three additional "booster" sessions delivered one week apart in the second year. Schools (fifth and sixth grade classes) were assigned to receive the curriculum plus booster (fifth grade classes only), curriculum only or control.

Quality assessment

Shope and colleagues (1992) randomly assigned 49 schools to a pretest or no pretest condition, and then to intervention or control conditions. It was difficult to judge whether the study had been well conducted because few details were reported about the study methodology. In addition, the authors reported little information on the pretest equivalence of the sample and attrition was relatively large over the two and half year duration of the study (28% were lost to follow-up). The study was therefore rated 'RCT –'.

Findings

Shope and colleagues (1992; RCT -) found that there were no significant differences in levels of alcohol use or misuse³ between fifth and sixth grade students who received the AMPS curriculum (with or without booster sessions) and students in the control group at any follow-up. Shope and colleagues (1992; RCT -) also measured understanding of the AMPS curriculum material using a curriculum index score. Intervention students scored significantly higher on the curriculum index than comparison students at all three post-test assessments (p<0.001). Effect sizes were calculated for the follow-up when students where in the seventh and eighth grade and are shown in Table 5.3, Table 5.4 and Table 5.5.

Comparison	Intervention				Control	Standardised mean	
Comparison	Mean	SD	Total	Mean	SD	Total	difference (95% CI)
Fifth grade							
Curriculum + booster vs. control	0.62	1.04	406	0.68	1.25	458	-0.05 (-0.19, 0.08)
Curriculum vs. control	0.59	1.03	541	0.68	1.25	458	-0.08 (-0.20, 0.05)
Sixth grade							
Curriculum vs. control	0.89	1.28	922	0.94	1.45	421	-0.04 (-0.15, 0.08)

Table 5.3. AMPS: Alcohol use in 7/8th grade (Shope et al., 1992; RCT -)

³ Alcohol misuse was measured by 10 items reflecting overindulgence, trouble with peers and trouble with adults experienced as a result of alcohol use.



Comparison	Intervention			Control			Standardised mean	
Comparison	Mean	SD	Total	Mean	SD	Total	difference (95% CI)	
Fifth grade								
Curriculum + booster vs. control	0.85	1.5	416	0.8	1.52	462	0.03 (-0.10, 0.17)	
Curriculum vs. control	0.79	1.47	564	0.8	1.52	462	-0.01 (-0.13, 0.12)	
Sixth grade								
Curriculum vs. control	1.17	1.71	974	1.27	1.87	449	-0.06 (-0.17, 0.06)	

Table 5.4. AMPS: Alcohol misuse in 7/8th grade (Shope et al., 1992; RCT -)

Table 5.5. AMPS: Curriculum index scores in 7/8th grade (Shope et al., 1992; RCT -)

Comparison	Intervention				Control	Standardised mean	
Comparison	Mean	SD	Total	Mean	SD	Total	difference (95% CI)
Fifth grade							
Curriculum + booster vs. control	10.31	3.26	429	8.55	2.9	484	0.57 (0.44, 0.70)
Curriculum vs. control	9.62	2.98	590	8.55	2.9	484	0.36 (0.24, 0.48)
Sixth grade							
Curriculum vs. control	10.48	3.01	1009	9.68	2.83	462	0.27 (0.16, 0.38)

5.2.2.2 Alcohol awareness programme

Gamble and Burgess (1994) developed an awareness programme for fifth grade elementary school students (mean age 10.9 years), which emphasised the negative effects that alcohol could have both on their lives and the lives of others. The programme consisted of eight lessons that covered the effects of alcohol on the mind and body, decision making skills, media influence and problems associated with alcohol. The programme was implemented by a student teacher at a local college.

Quality assessment

The curriculum was examined using an uncontrolled before and after design. Sixty-five children participated in the study and follow-up was based on immediate post-test. Overall due to the weak study design utilised the study was rated '-'.

Findings

Gamble and Burgess (1994; UBA –) examined the effects of the alcohol awareness curriculum on knowledge. Several of the items tested indicated that there had been no change in knowledge between pre and post-test scores. However, students showed improvements of 50% or more on four items: the concept that alcohol is a drug, all alcoholic beverages have equivalent amounts of alcohol, the effects of alcohol on the body and that



alcoholics can be anyone. The authors noted that girls showed more of an overall improvement in knowledge than boys.

5.2.2.3 Adolescent Alcohol Prevention Trial (AAPT)

The AAPT curriculum, taught entirely by trained project staff, was based on social influence theory and aimed at the prevention of alcohol misuse (Donaldson et al., 1995; 2000). Four different intervention approaches were examined: resistance skills training; normative education; a combination of resistance skills training and normative education; and information provision only. Students either received the main programme in fifth grade and a booster programme in seventh grade, or received the main programme in seventh grade only.

Quality assessment

The AAPT was based on an RCT design with schools as the unit of assignment (RCT –). Few methodological details were reported regarding the method of randomisation and details were not reported on the baseline comparability of participants. Donaldson and colleagues (1995) did not clearly report how many participants or clusters were randomised to each condition, and details on attrition were also lacking.

Findings

Donaldson and colleagues (1995; RCT -) found that among students who believed it was not acceptable to drink alcohol there was a significant positive relationship between seventh grade refusal skills and eighth grade alcohol use (fifth grade students: p<0.05; seventh grade students: p<0.01). The same analysis for adolescents who believed that it was acceptable to drink revealed a positive, but non-significant relationship between seventh grade refusal skills and eighth grade alcohol use. Donaldson and colleagues (2000; RCT -) analysed additional data from AAPT students using both self-report and reciprocal best friend reports of alcohol (and other substance) use. Results were separately analysed for students attending public and private schools. For the sample of students attending public schools, those who received normative education reported significantly lower scores on the alcohol index measure and significantly lower rates of lifetime alcohol use in the eighth (p<0.01), ninth (p<0.01) and tenth grades (p<0.05; 1-, 2- and 3-year follow-ups, respectively) compared to students receiving comparison interventions. In addition, students who received normative education reported lower rates of 30-day alcohol use at the 1- and 3-year followups (p<0.001 and p<0.05, respectively) and drunkenness at the 1- and 2-year follow-ups (both p<0.01). For the majority of outcomes, students who received resistance skills training reported using alcohol significantly more often than students who had not received resistance skills training. For the sample of students in private schools, there were no effects



of normative education on alcohol use behaviours at any follow-up. However, private school students who received resistance skills training reported significantly lower rates of 30-day alcohol use at the 1- and 3-year follow-ups (both p<0.01) and a lower prevalence of drunkenness at the 3-year follow-up (p<0.05).

5.2.2.4 Protecting You/Protecting Me (PY/PM)

Five studies (Bohman et al., 2004; Bell et al., 2005a; 2005b; 2007; Padget et al., 2006) reported on evaluations of the PY/PM programme. The programme was delivered over 5 years from the first to fifth grade and consisted of 8 lessons a year (40 lessons in total), which focused on teaching students about the adverse effects of alcohol on the brain and vehicle safety skills. The programme was taught either by elementary school teachers or high school students enrolled in a peer-helping course. Two studies (Bohman et al., 2004; Bell et al., 2005a) examined the effects of the peer led programme for third to fifth grade students and one study examined effects on first and second grade students (Bell et al., 2007). Bell and colleagues (2005b) examined the effects of a teacher led version of the programme on fourth and fifth grade students who had received the programme in previous school years and Padget and colleagues (2006) examined the effects among fifth grade students who had received consecutive years of PY/PM programming (peer or teacher led).

Quality assessment

Three evaluations of the PY/PM programme were based on RCT cluster designs. All three RCTs generally followed the same methodology with elementary schools at each intervention site (based on proximity to a high school with students enrolled in a peer-helping course) randomised to either the intervention or control condition. Study methodology was fairly well reported across the three RCTs, with the exception of Bell and colleagues (2005a), which did not report adequate information on participants lost to follow up and outcomes were reported at post-test only. The study of the teacher led version of PY/PM was reported to be based on a 'quasi-experimental design' (Bell et al., 2005b), however as the authors did not describe how participants were allocated to the intervention and control groups, it was labelled a CBA. Other aspects of the methodology were adequately reported and the study was rated '+' for internal consistency. Padget and colleagues (2006) also used a quasiexperimental design (NRCT) to examine the effects of PY/PM. The intervention group consisted of fifth grade classrooms from five schools that had begun implementing the PY/PM programme five years previously. Intervention and comparison schools were matched on size, racial/ethnic composition, and percentage of students eligible for free lunches. However, intervention and control schools did not appear to be well matched in terms of size or 'economic disadvantage'. On the whole the study appeared to have been



adequately conducted and attrition across the study was relatively low. This study was therefore coded 'NRCT +'.

Findings

a) First and second grade students

Bell and colleagues (2007; RCT +) reported that there were significant programme effects for one measure of knowledge about the brain (p<0.05), but not on a second that examined knowledge of brain importance. There were also significant programme effects for three out of four measures relating to vehicle safety (all p<0.05), for one out of two media awareness questions (p<0.001), and on attitudes towards the harm of teenage drinking (p<0.05). The programme did not have any effect on decision making or use of rules in first and second grade students.

b) Third to fifth grade students

The first year evaluation of the programme with third, fourth and fifth grade students (Bohman et al., 2004; RCT +) found that the programme had significant effects on media literacy (p<0.05) and knowledge of brain development (p<0.001) at post-test and follow up. Significant gains were also made in the intervention group in terms of vehicle safety skills; intervention students reported a greater increase in vehicle safety skills and fewer intentions to ride with an alcohol impaired driver relative to the control group at post-test and follow up (p<0.05). There was no significant effect of the programme on the following measures: knowledge of brain importance, attitudes to drinking and driving, underage drinking, rules, stress management, social skills and decision making.

In the second year evaluation of the programme (Bell et al., 2005a; RCT +), significant gains were made in media literacy and vehicle safety skills in the intervention group compared to the control group at post-test (p<0.05 and p<0.01, respectively), but there were no other significant changes in mean knowledge scores in the tested areas. There were no significant changes at post-test in mean skill scores on the measures of decision making, stress management or social skills or in attitudes to drinking.

c) Students with consecutive years of programming

Bell and colleagues (2005b; CBA +) examined the effects of a teacher led version of the programme on fourth and fifth grade students, who had been taught the PY/PM programme in previous school years. The authors reported that a significant impact of PY/PM was found for six out of eight outcome measures. Students in the intervention group gained stress management and decision-making skills (p<0.05), and vehicle safety skills (p<0.001), increased their knowledge of development and reported positive changes in terms of the



perceived harm of alcohol (p<0.001) and attitudes to underage drinking (p<0.05). The PY/PM programme did not have an impact on media literacy or on drinking and safety intentions. On the measures for which the programme demonstrated an impact, student's scores improved with increased exposure to the programme.

Padget and colleagues (2006; NRCT +) reported that the PY/PM programme had a small, but non-significant effect on drinking in the past 30 days at immediate post-test (mean difference -0.06; 95% CI -0.13, 0.01). The PY/PM intervention had significant positive effects on knowledge about the brain and alcohol (p<0.01); the perceived harm of and attitudes towards underage alcohol use (both p<0.05), and alcohol use intentions (p<0.01). In addition, the programme had significant effects on increasing vehicle safety skills (p<0.01) and reducing riding with a drinking driver (p<0.05). Effect sizes were calculated for these outcomes and are presented in Table 5.6.

	In	terventic	on		Control	Standardised	
Outcome	Mean	SD	Total	Mean	SD	Total	mean difference (95% Cl)
Knowledge about brain and alcohol	3.74	0.35	283	3.38	0.55	151	0.84 (0.63, 1.04)
Perceived harm of underage alcohol use	3.00	0.77	283	2.72	0.75	151	0.37 (0.17, 0.57)
Alcohol intentions	2.40	0.58	283	2.35	0.58	151	0.09 (-0.11, 0.28)
Media literacy	3.44	0.50	283	3.22	0.54	151	0.43 (0.23, 0.63)
Vehicle safety skills	3.62	0.56	283	3.07	0.72	151	0.89 (0.68, 1.09)
Attitudes towards underage drinking	3.80	0.54	283	3.61	0.82	151	0.29 (0.09, 0.49)
Drank past 30 days	0.09	0.28	283	0.15	0.36	151	-0.19 (-0.39, 0.00)
Riding with a drinking driver	0.13	0.34	283	0.23	0.42	151	-0.27 (-0.47, -0.07)

Table 5.6. PY/PM: intervention effects (Padget et al., 2006; NRCT +)

5.2.3 Single session interventions

Four studies were identified that examined single session alcohol education interventions. Three studies examined videotaped presentations which focused on media literacy training (Austin & Johnson, 1995), normative education (Godbold, 1999) and expectancy modification (Kraus et al., 1994). Cruz and Dunn (2003) also examined an expectancy modification intervention but the session was based around a presentation, quiz and discussion. A summary of content for these single session interventions is presented in Table 5.7.



Programme	Reference(s)	Programme components			
Media literacy training	Austin & Johnson, 1995	 Media literacy training One session of video, adverts, handouts and discussion Session led by researcher 			
Normative education	Godbold, 1999	 Session delivered by teachers Normative vs. information video Immediate or delayed attack video 			
Expectancy modification	Cruz & Dunn, 2003	 Taught by researchers who developed the intervention One-off session 			
	Kraus et al., 1994	 Two video interventions (expectancy based) Control video intervention presenting facts concerning alcohol's harmful effects 			

Table 5.7. Summary of programme content: Single session alcohol education interventions

5.2.3.1 Media literacy training

Austin and Johnson (1995) evaluated the effectiveness of a media literacy lesson for third grade students. The intervention was a one off session and aimed to enhance resistance skills towards media portrayals of alcohol. The session included a video that discussed techniques used by advertisers to sell products, a critique of alcohol advertising and discussion, and a handout.

Quality assessment

Evaluation of the media literacy lesson was based on an RCT Solomon group four design. A convenience sample of third grade students were randomised to the intervention or control condition and pre-test or no pre-test condition. The study methodology was not well reported and it was unclear how many students had been assigned to each condition and also if the groups were comparable at baseline. In addition only 44% of the original sample completed the follow-up three months later. The study was therefore rated '--' for internal consistency.

Findings

The media literacy lesson was evaluated in terms of its effects on children's perception of alcohol norms, alcohol portrayals in advertising, and alcohol-related behaviour (Austin & Johnson, 1995; RCT –). At immediate post-test, there were significant differences between treatment and non-treatment groups in understanding of persuasive intent (p<0.001), perceptions of realism (p<0.001), and perceptions of social norms for alcohol use (p<0.01), however at the three month follow-up, only perceptions of realism remained significant (p<0.01). Children who received the lesson were also less likely than control students to choose an alcohol-related toy⁴, when offered the choice (p<0.001).

⁴ Children were asked to view two toys that looked either like a can of fizzy drink or like a can of beer.



5.2.3.2 Normative education

Godbold (1999) conducted a test of inoculation theory applied to the context of adolescent alcohol use. Two groups of sixth grade students viewed a video which focused on the normative components of alcohol use and two groups viewed a video focusing on information related to alcohol use. Students also viewed an additional video based around two adverts for beer either at the time of the first video, or two weeks later, which were used to represent peer pressure to drink alcohol ('attack video').

Quality assessment

The evaluation was based on an RCT design with individual students randomised to one of six groups following pretesting. Further details of the methodology were not well reported, for example it was not clear how many students were allocated to each condition and therefore whether groups were comparable at baseline. The study was rated '-' for internal consistency.

Findings

At the first post-test, when all students had viewed the video but only half had viewed the additional 'attack' video, there was a significant difference between the normative (intervention) group and the information and control group (p<0.05) on the measure of peer acceptance of alcohol use (Godbold, 1999; RCT –). Students who viewed the normative messages had the lowest estimations of peer acceptance of alcohol use. At a second posttest, when all groups had viewed the 'attack' video there was no effect by message type on peer acceptance. In addition, there was no effect of message type on attitudes/behavioural intentions at either post-test.

5.2.3.3 Expectancy modification

Two studies examined expectancy modification interventions for fourth grade students (Cruz & Dunn, 2003; Kraus et al., 1994). Cruz and Dunn (2003) examined the impact of an expectancy challenge compared to traditional alcohol information. The expectancy modification intervention examined involved a presentation by one of the researchers, which examined students' beliefs about the effects of alcohol and why they thought people drank, and discussion regarding the pharmacological effects of excessive alcohol consumption. Students also participated in a quiz. Classrooms assigned to receive the traditional alcohol information intervention were given a presentation that emphasised the negative and harmful effects of alcohol consumption. The expectancy modification intervention examined by Kraus and colleagues (1994) was based on a 10-minute video presentation. One group viewed a tape based on an adult-model of expectancies and a second group viewed a tape featuring puppets (puppet-model expectancies). The control group received a facts only model.



Quality assessment

Both studies (Cruz & Dunn 2003; Kraus et al., 1994) used an RCT cluster design to examine the effects of the expectancy modification interventions. Cruz and Dunn (2003) did not report full details of the method of randomisation and the analyses only included participants who had completed both pre- and post-test measures. Follow-up measures were taken one week after intervention. Overall the internal consistency of the study was rated '-'. The study by Kraus and colleagues (1994) was adequately reported in terms of methodology, but it was not clear how many students were assigned to each group and therefore the study was rated '-' for internal consistency.

Findings

Cruz and Dunn (2003; RCT –) reported that the expectancy modification group exhibited the greatest amount of change in alcohol expectancies. Children participating in the expectancy modification intervention were less likely than the information only or control students to express positive or arousing expectancies at post-test (statistical significance not reported). Children who received the traditional alcohol information intervention were also less likely to express positive expectancies but this represented a smaller amount of change than occurred in the expectancy challenge group. Kraus and colleagues (1994; RCT –) found that children's alcohol expectancies were influenced by the intervention (p<0.001). The puppet-model video reduced expectancy endorsement but the adult-model video had the opposite effect and increased expectancy endorsement.

5.2.4 Summary and evidence statements

Thirteen primary studies were identified that examined alcohol education programmes, of which nine studies were evaluations of classroom-based curriculums and four studies were evaluations of single session interventions.

5.2.4.1 Knowledge and understanding

Impact on knowledge was examined for two programmes. The PY/PM programme was shown to have significant effects on knowledge about the effects of alcohol on development and the brain, vehicle safety and media literacy. This effect was demonstrated with third, fourth and fifth grade students (Bohman et al., 2004; RCT +; Bell et al., 2005a; RCT +, 2005b; CBA +) and students in first and second grade (Bell et al., 2007; RCT +). An alcohol awareness programme (Gamble & Burgess, 1994; UBA –) found some short term effects on knowledge related to alcohol awareness including the effects of alcohol on the body.



5.2.4.2 Attitudes and values

Impact on attitudes was examined for four single session interventions. Two studies that examined expectancy modification interventions (Cruz & Dunn, 2003; Kraus et al., 1994; both RCT –), one presentation-based the other video-based, both found that the interventions examined modified expectancies in the short term. A normative education approach delivered via video had positive effects on attitudes to alcohol at immediate posttest, but this finding was not maintained at follow up two weeks later. Media literacy training (Austin & Johnson, 1995; RCT –) had only very short term effects on children's perceptions of alcohol portrayals in advertising, children's perceptions of social norms or alcohol-related behaviours.

5.2.4.3 Personal and social skills

One study (Bell et al., 2000b; CBA +) reported that the PY/PM programme had significant effects on stress management and decision making for fourth and fifth grade students, but this finding was not replicated in other PY/PM studies of this, or younger, age groups.

5.2.4.4 Alcohol use and/or sexual health

The long term effects on alcohol use behaviours was examined for three programmes, AMPS (Shope et al., 1992; RCT –), AAPT (Donaldson et al., 1995; 2000; both RCT –) and PY/PM (Padget et al., 2006; NRCT +). None of these programme had consistent long term effects on alcohol use.

Evidence statement 2

- 2(a) There is moderate evidence from two RCTs, one NRCT and one CBA study¹ of a classroom-based programme to suggest that an intervention focused on alcohol prevention and vehicle safety can improve knowledge of the effects of alcohol on development and the brain, and vehicle safety in relation to drink driving. This evidence may only partially applicable to the UK because the programme's emphasis on the prevention of injury through drink driving is only partially relevant to PSHE delivery in primary schools focusing on SRE and alcohol education.
- 2(b) There is insufficient and inconsistent evidence from three RCTs and one NRCT² to determine the effects of alcohol education programmes on alcohol use in later years.
- ¹ Bohman et al., 2004 (RCT +); Bell et al., 2005b (CBA +); Padget et al., 2006 (NRCT +); Bell et al., 2007 (RCT +)
- ² Shope et al., 1992 (RCT -); Donaldson et al., 1995, 2000 (both RCT -); Padget et al., 2006 (NRCT +)



Table 5.8. Alcohol education: classroom-based programmes

Author (Year)	Design	Population	Follow-up	Analysed	Findings
Gamble & Burgess, 1994 Alcohol awareness programme	UBA -	USA Mean 10.9 years N=65	PT	NR	No change in knowledge for some items; improvements of 50% or more on four items: the concept that alcohol is a drug, all alcoholic beverages have equivalent amounts of alcohol, the effects of alcohol on the body and that alcoholics can be anyone.
Shope et al., 1992 Alcohol Misuse Prevention Study (AMPS)	RCT -	USA 5 th to 6 th grade n= 5,356	26 months	72% followed up	Partial effectiveness demonstrated in subgroup of students with prior drinking experience. Intervention students scored higher on curriculum index.
Adolescent Alcohol Preve	ntion Trial (AAF	<u>(T°</u>			
Donaldson et al., 1995	RCT -	USA 5 th and 7 th grade N= 11,995	PT at 1 year	80% completed the PT questionnaire	Resistance training delayed the onset of alcohol use, but only when adolescents believed it is not acceptable to drink
Donaldson et al., 2000	RCT -	USA 5 th and 7 th grade N= 11,995	8th, 9th and 10th grades	NR	Significant effect on alcohol use of normative education in public school students. Negative effects of resistance skills training. No effects seen in private school students.
Protecting You/Protecting	Ме				
Bohman et al., 2004	RCT +	USA $3^{rd} - 5^{th}$ grade N=321	PT, 5 weeks (intervention only)	81% at PT	Significant effect of the programme on media literacy, knowledge of brain development and vehicle safety. Also lower intentions to ride with an alcohol impaired driver. No effects on knowledge of brain importance, stress management, social skills or decision making.
Bell et al., 2005a	RCT +	USA 3 rd – 5 th grade N=717	PT, 6 weeks (intervention only)	85% completed study	Significant gains in media literacy and vehicle safety skills. No other significant changes in knowledge or attitudes to drinking.
Bell et al., 2005b (Teacher led)	CBA +	USA 4 th – 5 th grade N=848	PT, 12 months	85% completed the study	Significant impact on stress-management skills, decision-making skills, vehicle safety skills, perceived harm of alcohol, development and underage drinking attitudes. No effect on drinking and safety intentions or media literacy.
Padget et al., 2006 (Teacher or peer)	NRCT +	USA 5 th grade N=493	PT	88% completed study	Small, but non-significant effect on past 30-day drinking. Significant positive effects on knowledge about the brain and alcohol; the perceived harm of and attitudes towards underage alcohol use, and alcohol use intentions. Also significant effects on increasing vehicle safety skills and reducing riding with a drinking driver.

Author (Year)	Design	Population	Follow-up	Analysed	Findings
Bell et al., (2007)	RCT +	USA 1 st – 2 nd grade N=858	PT	87% completed study	Significant programme effects for one measure of knowledge about the brain, but not a second. Also for measures relating to vehicle safety, and media awareness, attitudes towards the harm of teenage drinking. No effect on attitudes towards drinking in non-driving teenagers, rules or decision making.

Table 5.9. Alcohol education: single session interventions

Author (Year)	Design	Population	Follow-up	Analysed	Findings
Austin & Johnson, 1995 Media literacy training	RCT -	USA 3 rd grade N=246	Immediate post-test; 3 months	44% completed 3 month follow up	Significant effect on persuasive intent, perceptions of realism, perceptions of social norms for alcohol use and predrinking behaviour at immediate post-test. However, at 3 months follow-up, only perceptions of realism remained significant.
Cruz & Dunn (2003) Expectancy modification intervention	RCT -	USA 4 th grade N=216	1 week	87% completed the study	Intervention group exhibited the greatest amount of change in alcohol expectancies; less likely to express positive or arousing expectancies.
Godbold (1999) Normative education	RCT -	USA 11 years N=417	PT, 2 weeks	NR	Students who received the additional advert immediately after the initial advert showed significantly less favourable attitudes/behavioural intentions towards alcohol use at PT but only approaching significance at the 2 week follow-up.
Kraus et al., (1994) Expectancy modification intervention	RCT -	USA 2 nd -4 th grade N=292	PT, 4 weeks	92% completed study	Children's alcohol expectancies were influenced by the intervention; puppet- model video reduced expectancy endorsement but adult-model videos increased expectancy endorsement.

5.3 Drug education (including alcohol) programmes

5.3.1 Overview of evidence identified

A total of 32 primary studies were identified that examined drug education programmes that included a focus on illegal drugs (and tobacco) in addition to alcohol. A total of 20 studies reported on 18 classroom-based programmes, led by teachers (n=11 studies) or external contributors (n=9 studies). Four studies reported on programmes which combined in-school approaches with parent education and eight studies reported on a range of other in-school approaches including theatre in education and a programme based on a retreat format.

5.3.2 Classroom-based programmes led by teachers

Eleven studies examined nine drug education programmes led by teachers. Programme approaches examined included those based on life skills training (LST) in five studies (Botvin et al., 2003; Kreutter & Gewirtz, 1991; Bühler et al., 2008; Hurry & McGurk, 1997; Hurry et al., 2000) and science-based drug education in two studies (Holtz & Twombly, 2007; Sigelman et al., 2004). A summary of programme content is presented in Table 5.10.

Programme	Reference(s)	Programme components
Tuning In To Health: Alcohol and Other Drug Decisions	Ambtman et al., 1990	 Curriculum covered drug effects, decision- making, alternatives to drug use, healthy lifestyle promotion Length/intensity varied between schools
Life Skills Training	Botvin et al., 2003	 Main purpose of developing personal and social skills 24 classes taught to elementary school children in grades 3 to 6.
	Kreutter & Gewirtz, 1991	 Botvin's life skills training 18 sessions for grade 6 Taught by external trainer
Allgemeine Lebenskompetenzen und Fertigkeiten	Bühler et al., 2008	 General life skills training 12 lessons taught to fifth grade students
keepin' it REAL	Hecht et al., 2008	 Culturally grounded programme adapted for elementary school students 2 lessons per week over 8 weeks 30-45 minutes each lesson
Brain Power!	Holtz & Twombly, 2007	 Science-based drug education programme Tailored programme of education on legal and illegal drugs for each age group 1 lesson per week for 6 weeks
Project Charlie	Hurry & McGurk, 1997; Hurry et al., 2000	 Life skills curriculum Weekly 30 minute lessons over 1 year
Million Dollar Machine	Schinke & Tepavac, 1995	 Knowledge and resistance skills training 8 week programme Assembly and classroom lessons
Drug and alcohol curriculum	Sigelman et al., 2004	 Science-based alcohol and drug curriculum Audiotape curriculum delivery 1 hour sessions on 3 consecutive days
Here's Looking at You 2000	Stevens et al., 1996	No details reported

Table 5.10. Programme content: Drug education programmes led by teachers



5.3.2.1 Tuning into Health

Ambtman and colleagues (1990) examined the Tuning into Health (TITH) programme, which targeted students in second to sixth grade and was aimed at reducing the future incidence of problems associated with drugs including alcohol. The programme focused on helping students to understand what drugs are, their effects on the body, the factors that influence people to use or not use drugs, alternatives to drug use and on using decision making as a way to deflect influences that promote drug use. The duration of the programme was not reported and the length and intensity of the units taught varied between schools.

Quality assessment

Evaluation of TITH was based on an NRCT design. Participation in the programme or control group was decided by the schools themselves, and within the pool of schools a random sample of classrooms were drawn for assessment. This resulted in a large number of participating students (n=2,406) and at post-test, 87% of the sample was retained. Overall the study was rated '+' for internal consistency.

Findings

Ambtman and colleagues (1990; NRCT +) examined the impact of the TITH programme on knowledge gains. Intervention schools improved more than control schools on knowledge of the essential elements of the programme, and there were significant differences between the intervention and control groups in urban schools in all grades (p<0.01 or better). In rural schools only grades 3,4 and 5 showed significant differences in effect of the programme on knowledge scores between the intervention and control groups the intervention and control groups (p<0.01 or better).

5.3.2.2 Life Skills Training (LST)

Three studies (Botvin et al., 2003; Bühler et al., 2008; Kreutter & Gewirtz, 1991) examined the effects of LST. Botvin's LST programme was originally designed to be delivered to students in the seventh grade with the main purpose of developing personal and social skills. Botvin and colleagues (2003) examined the effectiveness of LST with elementary school children in third to sixth grade. The intervention consisted of 24 classes taught over three years. The "Allgemeine Lebenskompetenzen und Fertigkeiten" (ALF) programme examined by Bühler and colleagues (2008) was based around general LST. The programme targeted fifth grade students and consisted of eight sessions on LST and four sessions on substance-related issues. Kreutter & Gewirtz (1991) examined the effects of Botvin's LST with a sample of sixth grade students.

Quality assessment

Botvin and colleagues (2003) used an RCT design to examine the effects of LST with elementary school students. The RCT was coded '-' because although the study appeared to



have been adequately conducted, 44% of participants were lost to follow-up at the post-test 3 months from baseline. The evaluation of the programme by Bühler and colleagues (2008) was based on an RCT design. Classrooms in participating schools were randomly assigned to intervention or control condition and to a pre-test or no pre-test condition (Solomon-four group design). Although the study methodology was well reported the follow-up evaluation was conducted at the end of the school year only (i.e. immediate post-test) and 30% of students were lost to follow-up. The study was rated '+' for internal consistency. Kreutter & Gewirtz (1991) based their evaluation on a CBA design and the comparison group was selected from a population "thought to be roughly equivalent" to the intervention group. Overall the study methodology was only briefly reported and it was therefore difficult to judge the internal validity of the design and it was rated 'CBA -'.

Findings

Following delivery of LST, results collected at post-test indicated that there were no differences between intervention and control students in terms of drinking frequency or the proportion drinking in the past year (Botvin et al., 2003; RCT –). However, analysis at the school level found that intervention schools had lower drinking prevalence compared to control schools, and this difference approached significance (p=0.054). LST students reported significantly more anti-drinking attitudes (p<0.05; also approached significance at the school level analyses, p=0.051) and increased substance use knowledge (p<0.05) relative to control students. Intervention students also reported lower normative expectations for peer alcohol use (p<0.001) as well as marginally higher levels of self-esteem (p=0.06; significant at the school level analyses, p<0.05) than control students. No significant differences were observed on the other measures reported (advertising knowledge, social skills knowledge, refusal skills knowledge, teen or adult drinking norms, and risk-taking). Effect sizes were calculated and are presented in Table 5.11

Outcomes	Interv	ention sc	hools	Cor	ntrol scho	Standardised mean	
Outcomes	Mean	SD	Total	Mean	SD	Total	(95% CI)
Drinking frequency	0.436	0.038	9	0.520	0.037	11	-0.47 [-1.37, 0.42]
Drink in past year	0.134	0.007	9	0.178	0.007	11	-0.75 [-1.66, 0.17]
Anti-drinking attitudes	2.792	0.022	9	2.762	0.018	11	0.77 [-0.15, 1.69]
Substance use knowledge	5.327	0.245	9	5.280	0.218	11	0.14 [-0.74, 1.02]
Peer drink norms	1.292	0.073	9	1.418	0.069	11	-0.63 [-1.54, 0.28]
Teen drink norms	0.550	0.042	9	0.586	0.039	11	-0.20 [-1.09, 0.68]
Adult drink norms	1.595	0.042	9	1.625	0.038	11	-0.29 [-1.18, 0.60]
Risk-taking	1.716	0.046	9	1.764	0.043	11	-0.44 [-1.33, 0.46]
Self-esteem	2.678	0.043	9	2.593	0.037	11	1.06 [0.10, 2.01]

Table 5.11. LST: Intervention effects (Botvin et al., 2003; RCT -)



The ALF programme did not have any effects on the measure of 'alcohol abuse', which the authors classified as students who "reported any use beyond trying" (Bühler et al., 2008; RCT +). At baseline 13.1% of intervention students reported alcohol abuse compared to 10.4% of control students. At post-test, 10.8% and 10.5% of intervention and control students respectively, reported alcohol abuse (RR 1.01; 95% CI 0.59, 1.75). There were significant programme effects on knowledge about skilled behaviour and life skill resources, and students in the intervention group reported greater gains than control students on these measures (p<0.001). However, there were no programme effects on knowledge about unskilled behaviour and life skills deficits. On the measure of alcohol affinity, students in the intervention group reported a more critical view against alcohol consumption than students in the control group (p<0.001).

Outcomes	Intervention schools			Control schools			Standardised
Outcomes	Mean	SD	Total	Mean	SD	Total	(95% CI)
Alcohol affinity	1.98	0.74	256	1.81	0.7	192	0.23 (0.05, 0.42)
Knowledge skilled behaviour	0.93	0.16	256	0.88	0.21	192	0.27 (0.08, 0.46)
Knowledge unskilled behaviour	0.59	0.3	256	0.59	0.28	192	0.00 (-0.19, 0.19)
Resources	37.7	12.8	256	34.9	11.6	192	0.23 (0.04, 0.41)
Deficits	15.5	5.4	256	16	5.2	192	-0.09 (-0.28, 0.09)

Table 5.12. ALF: Inter	vention effects	(Bühler et al.,	, 2008; RCT +)
------------------------	-----------------	-----------------	----------------

Kreutter and Gewirtz (1991; CBA –) found that compared to the comparison group, intervention students reported a significantly greater gain in scores in terms of knowledge (p<0.001), self-concept (p<0.01) and passivity (p<0.01). However, there was no difference between the intervention and comparison group on the measure of locus of control.

5.3.2.3 Project Charlie

Two UK-based studies (Hurry et al., 2000; Hurry and McGurk, 1997) examined the effectiveness of Project Charlie, a 'life skills' drug education programme developed in the USA for primary school aged children. Children aged 9-10 years received the programme over one school year and were followed up three years later at the age of 14.

Quality assessment

The evaluation of Project Charlie (Hurry and McGurk, 1997; Hurry et al., 2000) appeared to have been adequately conducted and was coded 'RCT +'. However, there were few details reported about the method of randomisation and the number of students included in the evaluation was relatively small.



Findings

Following delivery of Project Charlie, Hurry and McGurk (1997; RCT +) found that there was no difference in lifetime alcohol use between intervention and control students (RR 0.80; 95% CI 0.46, 1.40). Three years later when participants were aged 14, Hurry and colleagues (2000; RCT +) again found that there was no difference between intervention and control groups in terms of their alcohol use (RR 1.02; 95% CI 0.72, 1.45). At immediate post-test, Hurry and McGurk (1997; RCT +) reported that Project Charlie students had significantly higher decision-making skills than control students (p<0.05). However, there were no significant differences between groups in terms of self-esteem, intention to drink alcohol or peer pressure. Three years later at age 14 (Hurry et al., 2000; RCT +), there were no differences between intervention and control students in terms of decision-making skills, peer pressure resistance or drug knowledge. However, children who received the Project Charlie programme expressed more negative attitudes towards drugs than control children (p=0.05). Effect sizes were calculated and are presented in Table 5.13 and Table 5.14.

Table 5.13. Project Charlie: Intervention effects – dichotomous (Hurry & McGurk, 1997; Hurry et al., 2000; RCT +)

Outcomo	Interve	ention	Con	trol			
Outcome	Events	Total	Events	Total	KK (95% CI)		
Hurry & McGurk, 1997							
Lifetime alcohol use	17	65	18	55	0.80 (0.46, 1.40)		
Intention to use alcohol	33	65	30	55	0.93 (0.66, 1.31)		
Hurry et al., 2000							
Lifetime alcohol use	16	20	11	14	1.02 (0.72, 1.45)		
Peer pressure resistance	15	20	7	14	1.50 (0.84, 2.68)		

Table 5.14. Project Charlie: Intervention effects - continuous (Hurry & McGurk, 1997;Hurry et al., 2000; RCT +)

Outcomos	Intervention schools			Control schools			Standardised
Outcomes	Mean	SD	Total	Mean	SD	Total	(95% CI)
Hurry & McGurk, 1997							
Knowledge of drugs	13.3	3.5	48	10.7	3.2	37	0.76 (0.32, 1.21)
Decision making skills	15.9	4.3	65	14.3	4.9	55	0.35 (-0.02, 0.71)
Hurry et al., 2000							
Knowledge of drugs	13.0	1.6	20	12.7	1.4	14	0.19 (-0.49, 0.88)
Attitudes towards drugs	3.8	0.4	20	3.5	0.4	14	0.76 (0.05, 1.47)
Decision making skills	18.7	4.0	20	21.1	3.5	14	-0.62 (-1.32, 0.08)



5.3.2.4 keepin it REAL

Hecht and colleagues (2008) examined the fifth grade version of the keepin it REAL programme, which was originally designed for seventh and eighth grade students. The fifth grade version included 12 sessions and was based on the same basic curriculum content as the standard seventh grade multicultural version, which focused on enhancing anti-drug expectancies, normative beliefs, and refusal self-efficacy, and facilitating the development of decision making and resistance skills.

Quality assessment

Hecht and colleagues (2008) used an RCT design to examine the effectiveness of the programme. Twenty-three schools were randomly assigned to the intervention or control condition. Overall the study was well reported but details were lacking regarding the method of randomisation and a fairly large proportion of the sample were lost to follow-up (28%). The study was rated '+'.

Findings

At the 12 month follow-up assessment, there were no differences in lifetime or recent subtance use between intervention and control students (Hecht et al., 2008; RCT +). In addition, there was no difference between intervention and control students in terms of refusal efficacy, use of active decision making or the likelihood of using hypothetical alcohol resistance strategies. However, compared to control students, students who received the intervention reported greater increases in the number of resistance strategies used (p<0.001). At the follow-up assessment, intervention students perceived that relatively more of his or her peers were using substances than control students (p<0.001). On the measures of student's subtance use intentions, parents' and friends' anti-drug injunctive norms, personal antidrug norms and substance use expectancies there was no difference between intervention and control groups.

5.3.2.5 Million Dollar Machine

One study (Schinke and Tepavac, 1995) examined the effectiveness of an 8-week substance abuse prevention curriculum, the Million Dollar Machine, which focused on knowledge and resistance skills training.

Quality assessment

The evaluation of the Million Dollar Machine (Schinke and Tepavac, 1995) used an NRCT design. The study methodology was not well reported and consequently the study was rated '-'. Although the authors reported that intervention and control students were well matched, they were not permitted to collect demographic details from study participants. It was



therefore not possible to judge whether participants were matched on factors such as age and sex.

Findings

Fourth grade students who participated in the Million Dollar Machine substance abuse prevention programme (Schinke & Tepavac, 1995; NRCT -) reported significantly less actual and potential time drinking compared to fourth graders in the control group (p<0.05).

5.3.2.6 Science-based drug education

Two studies (Sigelman et al., 2004; Holtz & Twombly, 2007) examined science-based drug education curriculums. Sigelman and colleagues (2004) examined two versions of a drug and alcohol curriculum for elementary school children explaining how substances affect behaviour and health, consisting of a causally coherent version and a less coherent version, compared to a disease control curriculum. The 'coherent' curriculum was designed to teach the elements of a scientific, brain-mediated theory of drug effects in a causally coherent sequence. The 'less coherent' curriculum included the same content as the coherent curriculum, but was reordered so that the consequences of drug use on health and behaviour were discussed before the drug's effects on the body and brain. Holtz and Twombly (2007) evaluated the effects of a science education curriculum (Brain Power!) on drug knowledge and attitudes. The curriculum involved a tailored programme of education on legal and illegal drugs for each age group targeted (fourth and fifth grade). The programme lasted for six weeks with one lesson delivered each week.

Quality assessment

Sigelman and colleagues (2004) randomly assigned participants to intervention or control groups. The study appeared to have been adequately conducted but few details were reported regarding the methods of randomisation or baseline comparability and the study was coded 'RCT +'. Brain Power! was evaluated using an NRCT design (Holtz & Twombly 2007). Students from two schools were assigned by classroom to the intervention or control group. Allocation to the intervention and control groups resulted in an imbalance between the two groups in terms of the racial and grade composition of the groups, although these were adjusted for in subsequent analyses. In addition, follow-up was limited to immediate post-test only and details of participants lost to follow up were not reported. The study was rated '-' for internal consistency.

Findings

Sigelman and colleagues (2004; RCT +) found that two drug and alcohol programmes explaining how substances affect behaviour and health had no significant effects on alcohol



use. There was no difference between intervention and control students in terms of alcohol use in the previous month at immediate post-test or 1-year follow-up. The programme also had no significant effects on attitudes to alcohol use or intentions to use alcohol. Holtz and Twombly (2007; NRCT –) examined the effects of the Brain Power! programme on knowledge. At post-test, the intervention group showed statistically significant greater improvements in knowledge about drugs compared to the control group (p<0.01). The authors did not examine the effects of the programme on other outcomes.

5.3.2.7 Here's Looking at You 2000

Stevens and colleagues (1996) reported on the New Hampshire Substance Abuse Prevention Study. Two approaches were examined in the study, a school curriculum for grades 1 to 12 (Here's Looking at You 2000) and a parent communication course combined with a community task force. Further information about the programme components and duration were not reported.

Quality assessment

The study was examined using a CBA design and the two intervention approaches were compared to a delayed intervention control. The study methodology was poorly reported and the study was consequently rated '-'.

Findings

There was no effect of the curriculum on the initiation of or drinking for students in grades 4-6 (Stevens et al., 1996; CBA -).

5.3.3 Classroom-based programmes led by external contributors

Nine studies examined seven drug education programmes led by external contributors. Three studies (Abbey et al., 1990; Witt & Witt, 1995; Hahn et al., 2007) examined the Beginning Alcohol and Addictions Basic Education Studies (BABES) programme which focused on psychosocial skills training for young children. The remaining six studies (Baker, 2004; Hall-Long & Dishop, 1999; Peterson & Woodward, 1993; Schinke et al., 2000; Welham, 2007; Wright, 2007) examined different approaches to drug education programmes delivered by a range of different types of external contributors. A summary of programme content is presented in Table 5.15.



Programme	Reference(s)	Programme components
BABES	Abbey et al., 1990; Witt & Witt, 1995	 Psychosocial skills training for young children 1 lesson per week over 7 weeks; 1 hour per lesson Taught by trained presenter/facilitator Key components included storytelling, group discussion and role play
BABES Plus	Hahn et al., 2007	 7 lessons (40–50 minutes) BABES Plus included a parent-child interaction component Taught by school counsellor
Preventing Alcohol and Drug Abuse Through Primary Education (PADAPE)	Baker, 2004	 Incorporated lessons from two programmes (Here's Looking At You and Get Real About Tobacco) Taught by trained instructors 6 lessons in second grade 8 lessons in third to fifth grade
Drug education programme	Hall-Long & Dishop, 1999	 First and third grade drug education programme Senior nursing students delivered the intervention Two lessons per week over 8 weeks Each session lasted 30-45 minutes
CHOICE programme	Peterson & Woodward, 1993	 Designed to teach children specific things they can learn to feel good, without using drugs or alcohol Delivered by counsellors trained in the CHOICE programme 1 lesson per week for 45 minutes over one semester
Curriculum for Native American students	Schinke et al., 2000	 Based on LST Sessions incorporated cultural content Community involvement component based on community mobilisation
Enrichment programme	Welham, 2007	 Planned curriculum for preschool to year 6 Five themes: (1) knowledge of body and body functions ;(2) taking care of the body;(3) medicines and drugs;(4) identifying and dealing with danger; and (5) identifying and managing emotions. Specialist 'visitors' delivered the programme
Drug At Work (DAW)	Wright, 2007	 Programme emphasised the indirect effects of drug use on non-users 7 sessions in fifth grade 1 follow up session in sixth grade Taught by undergraduate students

Table 5.15. Programme content: Drug education programmes led by external contributors

5.3.3.1 Beginning Alcohol and Addictions Basic Education Studies (BABES)

Three studies examined the BABES programme, a social competency programme designed to teach young children about the consequences of alcohol and drug use. Two studies (Abbey et al., 1990; Witt & Witt, 1995) examined the effects of the programme with second grade students and one study (Hahn et al., 2007) examined an expanded version of the programme which incorporated a home-based component (BABES Plus) selected from a



population of elementary schools in which more than 40% of the students received free or reduced lunch.

Quality assessment

Abbey and colleagues (1990) evaluated the BABES programme using an RCT design. Three second grade classrooms from one school were randomly assigned to the intervention or control group. Overall the study was adequately reported and although the follow-up was only one month, a high proportion of the sample was followed up. The study was rated '+' for internal consistency. Witt & Witt (1995) evaluated the effect of the BABES programme on knowledge and attitudes using a UBA design. Due to the weak study design utilised and the poor reporting of methods the study was rated as 'UBA –'. Hahn and colleagues (2007) reported that a 'quasi-experimental design' was used to examine the effects of an expanded version of the BABES programme. Three schools, classified as high risk (>40% of students received a free or reduced price lunch), were selected and randomly assigned to one of three groups, BABES only, BABES Plus or control. Overall the study was well reported but because details were lacking regarding allocation concealment the study was rated 'NRCT +' for internal consistency.

Findings

Abbey and colleagues (1990; RCT +) found that at follow-up (one month after receiving the programme) the intervention group scored significantly higher than the control group on the knowledge test based directly on BABES material (p<0.01). However, there was no significant difference between groups on the knowledge test which applied BABES material to different situations. The intervention group reported significantly more negative attitudes towards the effects of alcohol than the control group (p<0.05). Effect sizes were calculated for these measures and are presented in Table 5.16. Abbey and colleagues (1990; RCT +) also examined the effects of the BABES programme on measures of coping, decision making, help seeking, peer pressure resistance, responsibility taking or self esteem, but there were no significant differences at post-test between the intervention and control groups on any of these measures. However, at follow-up, control group members demonstrated more active coping skills on one of the three coping scenarios than intervention students (p<0.05).

Outcomos	Intervention schools			Control schools			Standardised
Outcomes	Mean	SD	Total	Mean	SD	Total	(95% CI)
Attitudes about alcohol	0.31	0.16	31	0.23	0.22	24	0.42 (-0.12, 0.96)
Self-esteem	0.8	0.18	31	0.7	0.17	24	0.56 (0.02, 1.10)



Application of course material	14.52	3.62	31	14.24	3.52	24	0.08 (-0.46, 0.61)
BABES picture test	18.87	3.09	31	15.84	3.96	24	0.85 (0.30, 1.41)

Based on a UBA study, Witt & Witt (1995; UBA –) found that after receiving the BABES programme, students consistently scored higher at post-test on the concepts set in each of the lessons, indicating significant gains in knowledge (with the exception of knowledge relating to the lesson on self-image and feelings) (statistical significance not reported).

Hahn and colleagues (2007; NRCT +) examined the effects of an enhanced version of the BABES programme on child adjustment. Following intervention, parents whose children had received the BABES Plus programme rated their children as having less anxiety/withdrawal than did parents who children received the BABES only programme (p<0.05). Parents whose children received BABES Plus also rated their children as more socially competent than did parents whose children were assigned to BABES only group (p<0.05) or the control group (p<0.05). There was no difference between any of the group on the anger/aggression scale. Effect sizes were calculated and are presented in Table 5.17.

Outcome	Comparison	Standardised mean difference (95% CI)
Home environment	BABES Plus vs. control	0.01 (-0.49, 0.51)
	BABES vs. control	-0.12 (-0.64, 0.40)
Parant depressive symptoms	BABES Plus vs. control	-0.31 (-0.81, 0.20)
	BABES vs. control	0.05 (-0.47, 0.56)
Porent invelvement	BABES Plus vs. control	-0.19 (-0.69, 0.32)
Falent involvement	BABES vs. control	0.03 (-0.49, 0.54)
A pyint (Mithdrowol	BABES Plus vs. control	0.20 (-0.30, 0.70)
Anxiety/withdrawai	BABES vs. control	-0.13 (-0.65, 0.38)
Capiel confidence	BABES Plus vs. control	0.06 (-0.44, 0.57)
Social confidence	BABES vs. control	0.17 (-0.34, 0.69)
Aggregation	BABES Plus vs. control	0.06 (-0.44, 0.57)
Aggression	BABES vs. control	0.17 (-0.35, 0.69)

Table 5.17. BABES Plus: Intervention effects (Hahn et al., 2007; NRCT +)

5.3.3.2 Preventing Alcohol and Drug Abuse through Primary Education (PADAPE)

Baker (2004) evaluated the PADAPE programme which incorporated lessons from two other drug education programmes, the drugs and alcohol component of the programme was adapted from Here's Looking At You (HLAY) and the components that addressed tobacco were adapted from Get Real About Tobacco (GRAT). The content of the programme focused on knowledge about drugs, skills for refusing drugs and social skills. The PADAPE



programme consisted of six lessons in second grade, and eight lessons each in the third, fourth and fifth grades. Lessons were taught by educators trained to teach HLAY and GRAT.

Quality assessment

The study was conducted in two phases, phase one was based on a UBA design and phase two included a control group for comparison (CBA design). The study methodology was not well reported, it was not clear whether students who participated in the study were comparable and it was not clear how many students completed the study. Overall both phases of the study were rated '-' for internal consistency.

Findings

In the uncontrolled phase of the study the author noted that there were significant overall increases in performance from pre-test to post-test at all grades (p<0.001), and that these findings were maintained at follow up (Baker, 2004; CBA -). However, in second and fifth grade, the number of correct responses related to alcohol did not significantly improve. In the second phase of the study, which included a comparison group, students in second, third, and fourth grades who had not previously received the intervention programme performed better than students who had received the PADAPE programme, with the exception of one group. A comparison of fifth grade students revealed no significant differences between groups.

5.3.3.3 Hall-Long and Dishop's drug education programme

Hall-Long and Dishop (1999) conducted a pilot study of a first and third grade drug education programme, which was designed to increase knowledge about medicines, alcohol, tobacco and illegal drugs. The programme was taught by senior nursing students and consisted of two, 30-45 minute lessons each week over eight weeks.

Quality assessment

The study conducted by Hall-Long and Dishop (1999) was a pilot study, and the study methodology was based on an UBA design. In addition to lacking a control group for comparison, it was unclear how long students had been followed-up and the reliability of the outcomes measures used was not discussed. The study was rated '-' for internal consistency.

Findings

Hall-Long and Dishop (1999; UBA -) reported that there was an average increase in student's knowledge test scores by 30% compared to pre-test scores and increases on every area of knowledge examined.



5.3.3.4 CHOICE programme

Peterson and Woodward (1993) evaluated the CHOICE drug education programme for sixth grade students. This programme was designed to teach children alternative ways to feel good without using drugs or alcohol. The programme incorporated a video and cooperative learning techniques. Sessions were conducted once a week for 45 minutes over one school year.

Quality assessment

The study design was quasi-experimental but it was not reported how schools were assigned to the intervention and control groups. Other aspects of the methodology were also poorly reported and the study was rated '-' for internal consistency.

Findings

Peterson and Woodward (1993; NRCT -) examined the effects of the CHOICE programme on the self concept and locus of control. The authors reported that although a consistent trend in the direction of increased levels of self-concept and greater internal locus of control was found for the intervention school compared to the control school, the only marginally statistically significant difference was found on a self-concept semantic differential scale (p=0.05). That is, compared to students in the control group, students who received the CHOICE programme had a significantly higher self-concept as measured by the semantic differential.

5.3.3.5 Curriculum for Native American students

Schinke and colleagues (2000) examined a culturally tailored school-based substance abuse prevention programme for third, fourth and fifth grade Native American students. The school-based prevention programme was based on life skills training and incorporated Native American values, legends and stories. Students in one intervention arm also participated in a community involvement component that involved community activities and media programming.

Quality assessment

On the whole the study methodology used to evaluate the programme for Native American students was adequately reported (Schinke et al., 2000). However, details were lacking regarding the methods used to randomly assign schools between the intervention and control arms. The study was coded 'RCT +'.

Findings

Schinke and colleagues (2000; RCT +) found that significantly fewer Native American students who participated in the school curriculum only, and school curriculum plus

community groups reported alcohol consumption at 30- and 42-month follow-up compared to control (p<0.01). In addition, significantly fewer participants in the school curriculum only group reported alcohol use at 30- and 42- month follow-ups compared to both control and participants who received the school curriculum plus community components (p<0.01).

5.3.3.6 Drug education enrichment programme

Welham (2007) reported on a study of the efficacy of drugs education delivered to children aged 7 to 11 years (enrichment programme). The main component features of the programme were a comprehensive written curriculum and use of a mobile classroom. The programme was delivered by specialist external providers and the themes covered by the programme were: knowledge of body and body functions; taking care of the body; medicines and drugs; identifying and dealing with danger; and, identifying and managing emotions.

Quality assessment

Evaluation of the enrichment programme was based on an uncontrolled before and after study design and cross-sectional pre- and post-testing. Details of the study methodology were largely unreported and the study was therefore rated 'UBA –' for internal consistency.

Findings

Following delivery of the intervention, and with subsequent teacher support in-class, outcomes indicated that children's knowledge of how to stay healthy and the likely impact of drugs, alcohol and smoking on the maintenance of health and wellbeing had improved (statistical significance not reported). Welham (2007; UBA -) noted that pupils were overtly conscious of the likely future impact of older pupils on their ability to stay drug free on transferring to secondary school.

5.3.3.7 Drugs at Work (DAW)

Wright (2007) examined the effectiveness of a normative drug education programme, Drugs at Work, which emphasised the indirect impacts of drug use. Students participated in interactive, hands on activities, which included the DAW simulation exercise, media literacy training, and resistance skills training. The programme included seven sessions in fifth grade and one follow up session in sixth grade and was taught by undergraduate students.

Quality assessment

Evaluation of the DAW programme was based on a CBA design. However, the programme was implemented in comparison schools part way through the evaluation and both intervention and comparison schools also began implementing the DARE programme. The evaluation was therefore divided into three phases, for example in phase one, a quasi-experimental design was used to compare data from a baseline sample of sixth grade

students who did not receive the programme with students who received the programme in later years and their peers. Due to the complications in the study design the study was rated 'CBA –' for internal consistency.

Findings

Compared to baseline data collected in 1990, participants (or their classmates) who received DAW (1991-1992) were significantly more likely to have negative attitudes to drinking alcohol (p<0.01) and smoking (p<0.01), but not towards illegal drug use (Wright, 2007; CBA -). DAW participants (or their classmates) were significantly less likely to smoke (p<0.01), have been drunk (p<0.05) or used illicit drugs (p<0.05), and were significantly more likely to report having friends who would stop them from getting drunk (p<0.01). In an analysis of all seven years of data collected, the author reported results which indicated that the programme had less consistent effects on the use of alcohol than it did on smoking or illegal drug use.

5.3.4 Multicomponent programmes

Four studies (Rollin et al., 1992, 1995; Zavela et al., 1997, 2004) examined multicomponent programmes. Project KICK (Rollin et al., 1992; 1995) combined peer modelling for fifth grade students with a parent education component and Say Yes First (Zavela et al., 1997; 2004) included a curriculum combined with case management for high risk youth and a parent education programme. A summary of programme content is presented in Table 5.18.

Programme	Reference(s)	Programme components
Project KICK	Rollin et al., 1992; 1995	 Peer modelling and parent education Seventh grade peer leaders served as a 'buddy' for two twenty minute sessions each week Delivered over two school years
Say Yes First	Zavela et al., 1997; 2004	 Multicomponent programme; universal curriculum and case management of high- risk youth 5 year programme Included parent education component

Table 5.18. Programme content: Multicomponent drug education programmes

5.3.4.1 Project KICK

Two studies (Rollin et al., 1992; 1995) examined Project KICK which combined a peer modelling intervention with parent education. Seventh grade students from a middle school served as the positive peer models and taught drug awareness, drug refusal skills, and self-esteem building activities to groups of third grade students in two 20-minute sessions per week. Peer leaders were trained by KICK staff in two 20-minute sessions per week utilising role playing, lecture, discussion, video tape presentation, and small group activities. Parent sessions were held approximately once every two months, and included educational



seminars on behaviour management, stress management, parent/child communication, drug education and family management.

Quality assessment

Project KICK was evaluated using an RCT design; two classes within the same school were randomly assigned to the intervention or control group. Further details regarding the method of randomisation were not reported and allocation resulted in imbalances in the intervention and control groups, although these were adjusted for in subsequent analyses. No details were reported on the number of students followed up and follow-up time was inadequate. The study was rated 'RCT -' for internal consistency.

Findings

Rollin and colleagues (1992; 1995; RCT -) reported a significant effect of the intervention on scores on the Drug Knowledge Survey (p<0.01). Students who received the intervention had improved more than the control group on this measure by the second post-test (end of phase two). The intervention also had a significant effect on scores on the measures of life management, decision making and drug refusal skills, and self-concept (all p<0.01). Students in the intervention group had a greater improvement on these measures compared to the control group at the second post-test.

5.3.4.2 Say Yes First

Two studies (Zavela et al., 1997; 2004) examined the effectiveness of Say Yes First (SYF), which aimed to promote resiliency and protective factors in young people. The programme was implemented over five years, between grades four and eight. The intervention was educationally based but also employed case management techniques for high-risk youths and their families. The programme included parent education programmes, alternative youth and family activities, SYF councils and youth leadership training.

Quality assessment

Both studies of SYF used CBA designs and were given a 'CBA –' rating. This was because of poor reporting of key elements, a lack of equivalence between conditions at baseline, and the use of historical control in the study reported on by Zavela and colleagues (1997).

Findings

Zavela and colleagues (1997; CBA -) followed students who had participated in the SYF programme from the fourth to the eighth grade. They found that students who received the SYF programme reported lower prevalence of 'ever' use of alcohol than comparison students in the 1993-1994 and 1994-1995 cohorts (p<0.05). For past 30-day use of alcohol, students in the intervention cohort reported lower use than students in the 1994-1995

comparison cohort (p<0.05). Three year follow up data reported by Zavela and colleagues (2004) showed lower scores on measures of lifetime alcohol use, 30 day alcohol use and amount of alcohol use in SYF students compared to control students. However these differences were not significant.

5.3.5 Other in school approaches

Seven studies were identified that examined other in-school approaches to drug education. Two studies (Allison et al., 1990; Paxton et al., 1998) examined teachers training and support programmes. Five studies (Corbin et al., 1993; Hawthorne et al., 1995; Hawthorne, 1996; Tudor-Smith et al., 1995; Starkey & Orme, 2001) examined short term or single session interventions and one study (Raybuck & Hicks, 1994) examined an intervention delivered in a retreat format away from the classroom. A summary of programme content is presented in Table 5.19.

Programme	Reference(s)	Programme components
DAPPER	Allison et al., 1990	 Intensive staff development programme and in-service training 5 sessions of 3 hours
Drug education programme	Paxton et al., 1998	 Half day of teacher training and support in drug education delivery Programme delivered in 4 hourly periods, one per day over one week
Refusal skills intervention	Corbin et al., 1993	 Refusal skills training and rehearsal Trained by Psychology Majors Delivered over 3 days Sessions lasted for 45 minutes per day
Life Education Centres	Hawthorne et al., 1995; Hawthorne, 1996	 Mobile Life Education Centre presentation Preparatory and follow-up classroom work by classroom teachers
	Tudor-Smith et al., 1995	 'Decisions' programme for 10-11 year olds Substance use prevention and peer pressure resistance training One-off lesson, lasting 30 mins-2 hours Taught by trained educator
Theatre in Education	Starkey & Orme, 2001	 One day interactive drama production and workshop Facilitated by actors with teaching/workshop experience
KIDS CARE	Raybuck & Hicks, 1994	 Retreat delivered away from classroom Once every school year; half day for kindergarten – 2nd grade, and a full day for 3th – 6th grade External facilitator

Table 5.19. Programme content: Other in-school drug education approaches

5.3.5.1 Teacher training and support

Two studies (Allison et al., 1990; Paxton et al., 1998) examined teachers training and support programmes. Allison and colleagues (1990) examined D.A.P.P.E.R., which was developed in the USA and largely based on knowledge and resistance skills training. The D.A.P.P.E.R. curriculum was developed from the Life Skills Training model and was



supported by provision of intensive staff development focusing on knowledge, attitudes and implementing skills and in-service training. The training schedule included five, three-hour sessions, and recipients were encouraged to hold one to two hour workshops in their school following training. One study examined this programme and compared students' drinking at the end of the school year in schools where teachers had received intensive D.A.P.P.E.R training compared to those that only received in-service training and those that received curriculum material but no staff development (Allison et al., 1990). Paxton and colleagues (1998) reported on a project conducted in Northumberland, UK that aimed to help teachers provide effective drug education through training and support. Eighteen schools participated in the project. Following assessment of the needs and concerns of year 5 pupils, their teachers and parents, a half day of teacher training was provided on methods of good practice. All teachers received training in first aid and theatre in education, in addition to choosing two other specific drug education topics (drug specific card game, alcohol module, photograph module, smoking, peer pressure, solvents, why use drugs, or cartoon modules). Teachers then delivered the programme in four hourly periods, one per day over one week and a parents evening was held.

Quality assessment

Allison and colleagues (1990) did not match the unit of allocation (school) and analysis (individual), and groups were not matched on their intentions to use alcohol in the future. Furthermore, there was little detail on curriculum implementation; this study was therefore given a rating of RCT –.The study conducted by Paxton and colleagues (1998) was largely descriptive and mostly focused on process evaluation. Although data were collected before and after programme delivery a control group was not utilised and the internal consistency of the study was rated 'UBA –'.

Findings

Allison and colleagues (1990; RCT –) reported that there were no differences between intervention students taught by teachers who were intensively trained or who received inservice training and control students, on any of the measures of alcohol use at post-test (lifetime use; drinking with parents; unsupervised drinking; intentions to drink), or on measures of alcohol related knowledge, problem solving, coping, and decision-making.

The only outcome of the programme examined by Paxton and colleagues (1998; UBA –) focused on seeking help or advice from appropriate individuals if they found or were offered a bag containing drugs. Following the drug education programme, more children said that they would take the drugs to the police (p<0.001) and more also said that they would take the drugs to their parents (p<0.001). Fewer children said they would throw them away



(p<0.001) or say no and run away (p<0.01). There was no significant change in the number of children who reported that they would give them to their mother or to 'Other'. Fewer pupils reported that they would intend to talk to the police if offered or found drugs (p<0.001), and more would intend to talk to their family (p<0.001), teacher (p<0.001) or friends (p<0.01). There was no significant change in the number that would talk to their mother.

5.3.5.2 Refusal skills intervention

Corbin and colleagues (1993) examined the impact of two intervention strategies on children's drug refusal skills and drug-related knowledge in third grade. Both intervention strategies (Rehearsal Plus [R+] and general information [GI]) taught children drug knowledge, assertiveness skills, decision making skills and specific drug refusal skills. However, students in the R+ group received an additional component that involved rehearsal of the behavioural training whilst students in the GI group were taught general knowledge. Training for both interventions was conducted over three days with 45 minutes per session per day.

Quality assessment

Effectiveness of the two treatment strategies was examined in an RCT. Children in one school were individually randomised to one of the two treatment strategies or a no intervention control. The sample size for the study was small, with a high rate of attrition (30%) and only students who completed the pre- and post-test were included in the analyses. The study was rated 'RCT –' for internal consistency.

Findings

Corbin and colleagues (1993; RCT –) examined the impact of the two intervention strategies on general knowledge scores. Participants' scores in the R+ group did not change between post-test and follow-up, but participant's scores in the GI group decreased (p<0.05). For drug knowledge, both the R+ and GI groups had higher means than the control group (p<0.001), but there was no difference between the two intervention groups. For the measure of rationale⁵, the R+ group had higher post-test mean than the GI and control groups (p<0.001), although there was no difference between the GI and control groups. There were no differences between the intervention and control groups on the measure of assertiveness. However, students in the R+ group had higher mean scores on the measure of decision-making at post-test than the GI group and the control group). There were no difference between the GI group and the control group. There were no difference between the GI group and the control group. There were no difference between the GI group and the control group. There were no difference between the GI group and the control group. There were no difference between the GI group and the control group. There were no difference between the GI group and the control group. There were no differences between the GI groups for the measure of refusal behaviours. However,

⁵ Participants were asked to provide a brief rationale to justify their responses on the measure of drug knowledge e.g. when asked to differentiate between "good" and bad" drugs.



the group differences were significant between the R+ and GI groups when averaged across post-test and follow-up times, with the R+ group performing better. The R+ group showed significantly more refusal behaviours (in or out of sequence) than either the GI or control groups (p<0.001). There was no difference between GI and control groups. Effect sizes were calculated and are presented in Table 5.20.

Table 5.20. Refusal skills intervention: Intervention effects	6 (Corbin et al., 1993; RCT –
---------------------------------------------------------------	-------------------------------

Outcome	Comparison	Standardised mean difference (95% Cl)
Sequence behaviour	GI vs. control	0.01 (-0.65, 0.68)
	R+ vs. control	1.68 (0.96, 2.40)
	GI vs. control	0.12 (-0.54, 0.79)
	R+ vs. control	1.71 (0.99, 2.44)
	GI vs. control	1.10 (0.38, 1.82)
General knowledge	R+ vs. control	-0.14 (-0.75, 0.48)
Drug knowledge	GI vs. control	1.11 (0.39, 1.83)
Drug knowledge	R+ vs. control	1.70 (0.98, 2.43)
Desision molting	GI vs. control	-0.26 (-0.93, 0.41)
Decision making	R+ vs. control	1.31 (0.62, 1.99)
Deficiencia	GI vs. control	-0.12 (-0.78, 0.55)
Rationale	R+ vs. control	1.24 (0.57, 1.92)
Acceptivences	GI vs. control	-0.11 (-0.78, 0.56)
ASSEI117611655	R+ vs. control	0.03 (-0.58, 0.65)

5.3.5.3 Life Education Centres

The Life Education programme evaluated by Hawthorne and colleagues (1995, 1996) consisted of three parts, preparatory classroom work, the Life Education presentation and follow-up work conducted by a classroom teacher. The Life Education presentation took place in a mobile classroom with an emphasis on learning how the body worked and identifying drug use pressures. Tudor-Smith and colleagues (1995) examined the short term impact of the Decisions programme of Life Education Centres (LECs) in Wales. Children were intended to receive one LEC programme per year in infant, primary and middle school. The Decisions programme targeted 10-11 year olds and using audio-visual aids, games, films, and role play taught children about substance use prevention and peer pressure resistance.

Quality assessment

Hawthorne and colleagues (1995) used a CBA design to examine the effects of the Life Education curriculum. The study compared students who had been exposed to the Life Education programme over five consecutive years with students not exposed to the programme but who had received conventional school-based drug education. Few details

about the study methodology were reported and it was therefore difficult to judge how well the study had been conducted. The study was consequently coded 'CBA -'. The design of the study by Tudor-Smith and colleagues (1995) was based on a UBA design and was rated '-' because of the weak design utilised.

Findings

Hawthorne and colleagues (1995; CBA -) found that students who received the Life Education programme were significantly more likely to report having drunk alcohol than non-Life Education students (OR 1.3; 95% Cl 1.0, 1.6). The authors reported that these findings were largely due to boys who received Life Education being significantly more likely to have drunk alcohol than non-Life Education boys (OR 1.3; 95% Cl 1.1, 2.1) as there was no difference between girls. There was no difference between students who had received the Life Education programme and those who had not in terms of drinking in the previous month. However, boys who received the Life Education programme were more likely to have drunk in the previous month than non-Life Education boys (OR 1.7; 95%Cl 1.1, 2.4). Again, there was no difference between girls. Boys who received the Life Education programme were also significantly more likely than non-Life Education boys to report having drunk two or more drinks per occasion (OR 1.4; 95% Cl 1.0, 1.9). This finding was significant across the whole group at the student level analysis but not the school level. Effect sizes are presented for the school-level analysis in Table 5.21.

Table 5.21. Life Education Centres: Intervention effects (school-level analysis)(Hawthorne 1995; CBA -)

Outcomes	OR (95% CI)	
Lifetime alcohol use		
All students	1.30 (1.06, 1.60)	
Boys	1.50 (1.07, 2.10)	
Girls	1.10 (0.64, 1.90)	
Past month alcohol use		
All students	1.20 (0.90, 1.60)	
Boys	1.70 (1.20, 2.40)	
Girls	1.10 (0.64, 1.90)	
Alcohol misuse (2+ glasses)		
All students	1.20 (0.90, 1.60)	
Boys	1.40 (1.03, 1.90)	
Girls	1.10 (0.58, 2.10)	

Tudor-Smith and colleagues (1995; UBA -) reported a significant improvement after two months in children's ability to recognise substances such as heroin, pharmaceuticals,


cigarettes, and alcoholic drinks as drugs (p<0.05). The majority of the children's beliefs about drugs did not change after two months, with the exception of views on advertising, perceptions of smokers and drinkers (p<0.05), with children becoming better informed on these matters. There were no statistically significant changes in substance use behaviours, although the authors reported that the proportions of students using alcohol and cigarettes had increased by follow-up (+4% and +9%, respectively, at the two-month follow-up). Furthermore, there was a statistically significant increase in the proportion of young people who reported that friends had smoked or drank alcohol (p<0.05).

5.3.5.4 KIDS CARE

Raybuck and Hicks (1994) examined the effects of the KIDS CARE programme that was designed to increase self esteem and reduce drug and alcohol use. The programme was administered in a retreat format away from the school, parents, and teachers. Children engaged in activities, discussions and games designed to teach concepts and skills. The modules were age appropriate and focused on developing prosocial ways of bonding and building self esteem. Classes engaged in informal follow-up activities and discussion following the retreat. The retreat was facilitated by an external facilitator and was repeated every school year between kindergarten and sixth grade. The retreat was a half day for students in kindergarten, first and second grade, and a full day for students in third, fourth, fifth and sixth grade.

Quality assessment

The study conducted by Raybuck and Hicks (1994) was observational, but included a comparison group from a school which had not implemented the KIDS CARE programme. The study methodology was not clearly reported and it was not clear, for example, how many students were in the intervention and comparison groups and whether they were balanced at baseline. The study was rated 'CBA -' for internal consistency.

Findings

Raybuck and Hicks (1994; CBA -) found that the intervention did not have significant effects on self-esteem on a standardised measure. However, on "Circle Words" (an adjective checklist self-esteem measure), children who participated in KIDS CARE had significantly improved scores on the positive scale compared to the comparison group (p<0.01), but there was no significant change on the negative scale. The retreat programme had a significant effect on the sociometric status of previously rejected or neglected children with high risk children in the intervention group more likely receive positive peer nominations (as children they like the most or whom they wish to spend more time with) than children in the comparison group.



5.3.5.5 Theatre in Education

Starkey and Orme (2001) conducted an evaluation of a primary school drug education drama project. The project involved an interactive drama production and workshop day for 10-11 year olds taught by actors with teaching or workshop experience.

Quality assessment

The impact of the drama project was assessed using a UBA design. The study was not well reported and was rated 'UBA -'.

Findings

Starkey and Orme (2001; UBA -) reported that significant increases were seen in children's ability to name specific drugs (e.g. cigarettes, alcohol, heroin) between pre- and post-test, increasing from 53% at pretest to 71% at post-test (p<0.001). The authors also reported a significant improvement in young people's response to seeking help when presented with a lost bag (potentially containing illegal drugs) with 9% saying that they would phone or tell the police (p<0.05 compared with pretest). Following participation in the programme, the children demonstrated change in their attitudes towards drugs with an increase in the percentage agreeing that some drugs could be good for you for medical reasons (p<0.01) and if you take the right amount (p<0.01).

5.3.6 Summary and evidence statements

A total of 32 primary studies examined drug education programmes that included a focus on illegal drugs in addition to alcohol. A range of programme approaches were identified; classroom-based programmes led by teachers or external contributors, programmes that combined in-school approaches with parent education, and other in-school approaches including theatre in education and a retreat-based programme.

5.3.6.1 Knowledge and understanding

The impact on knowledge and understanding was examined in 14 programmes, including nine classroom-based programmes led by teachers or external contributors, one multicomponent programme and four studies of other in-school approaches to drug education.

Participation in TITH in grades 2 to 6 was associated with improvements in knowledge related to the curriculum, although effects were more consistent in urban schools than rural schools (Ambtman et al., 1990; NRCT +). Four studies reported on the effects of life skills training on knowledge. Three studies reported that LST had an effect on knowledge at posttest (Botvin et al., 2003; RCT -; Bühler et al., 2008; RCT +; Kreutter & Gewirtz, 1991; CBA –). However, follow up of the students, who participated in Project Charlie, which was based



on the LST model, demonstrated that the intervention did not have long term effects on drug knowledge (Hurry et al., 2000; RCT +). Two studies (Abbey et al., 1990; RCT +; Witt & Witt, 1995; UBA -) that examined the BABES programme provided evidence that participation in the programme resulted in short term increases in knowledge relating to the curriculum, which focused on teaching young children about the consequences of drug and alcohol use. The results of the PADAPE programme (Baker, 2004; CBA -) were unclear; in phase one of the study participants reported increase in knowledge and skills, but in phase two students who had received the programme in the previous year reported worse outcomes than those who had not. A drug education programme for first and third grade students delivered by nursing students (Hall-Long & Dishop, 1999; UBA -) resulted in increases in knowledge across the curriculum areas of the programme. A UK study of an enrichment programme found that the intervention had a positive effect on children's knowledge (Welham, 2007; UBA -). Two science-based programmes (Sigelman et al., 2004; RCT +; Holtz & Twombly, 2007; NRCT -) were found to have improvements on knowledge about drugs immediately following curriculum delivery, but not in the longer term in one study (Sigelman et al., 2004; RCT +). Project KICK, a multicomponent programme that combined peer modelling for third grade students with parent education, had significant short term effects on drug knowledge (Rollin et al., 1992; 1995; RCT -).

One study (Allison et al., 1990; RCT –) that examined the effects of a teacher training intervention (intensive training vs. in-service training), D.A.P.P.E.R., found that the intensive intervention had no effects on students' knowledge in comparison to the in-service training approach. A second study that examined a project designed to support drug education delivery (Paxton et al., 1998; UBA –) reported that there were improvements in responses to seeking help with a lost bag. Following participation in a theatre in education programme (Starkey & Orme, 2001; UBA –), significant increases were seen in children's ability to name drugs, and in responses to seeking help with a lost bag belp with a lost bag of drugs. Children aged 10-11 years who participated in the 'Decisions' programme (Life Education Centres) also reported short term improvements in their ability to recognise substances (Tudor-Smith et al., 1995; UBA –).

5.3.6.2 Attitudes and values

Participants in the DAW programme (Wright, 2007; CBA –), which included interactive sessions, were more likely to have negative attitudes to alcohol. Botvin and colleagues (2003; RCT –) reported that LST had a significant effect on anti-drinking attitudes, and this finding was supported by the German study of LST by Bühler and colleagues (2008; RCT +). A science-based programme (Sigelman et al., 2004; RCT +) had no effects on attitudes or intentions towards alcohol. There were also no effects of the D.A.P.P.E.R. teacher training



programme on attitudes (Allison et al., 1990; RCT –) and The Decisions programme of the Life Education Centres was not found to have had any impact on substance-related beliefs (Tudor-Smith et al., 1995; UBA –).

5.3.6.3 Personal and social skills

There were no clear effects of the BABES programme on psychosocial skills. However, one study of the programme conducted with a high risk population (Hahn et al., 2007; NRCT +), which included an additional parent education component, had some impact on anxiety/withdrawal and social competence among children who received the enhanced intervention. Three studies examined the effects of LST for elementary students on personal and social skills. Botvin and colleagues (2003; RCT -) reported that the programme had effects on normative expectations and self esteem. A replication study by Kreutter & Gewirtz (1995; CBA -) also found that the programme had effects on self-esteem. There were also immediate effects of the LST-based Project Charlie on decision-making skills (Hurry & McGurk, 1997; RCT +), but not on self-esteem. Long term follow-up of the participants in Project Charlie found that the intervention did not have long term effects on personal or social skills (Hurry et al., 2000; RCT +). There were immediate effects of Project KICK (Rollin et al., 1992; 1995; RCT -) on participants' self esteem, decision making and refusal skills, with participants in the programme having improved outcomes on these measures. A rehearsal plus strategy that focused on rehearsal following refusal skills training had significant effects on refusal skills and decision making among participants. There were no effects of D.A.P.P.E.R., a teacher training programme on problem solving and coping (Allison et al., 1990; RCT -), and the results of the KIDS CARE programme, which although focused on improving self esteem provided no evidence of an intervention effect (Raybuck & Hicks, 1994; CBA -).

5.3.6.4 Alcohol use or sexual health

The longer term effects on alcohol use behaviours were examined for nine programmes. Positive effects were reported for two programmes, DAW and a curriculum for Native American students, and were inconsistent for two further programmes SYF and LST. One programme provided by the Australian Life Education Centre, had potentially negative effects on student's drinking in later years. No effects were reported for four programmes, keepin it REAL, Million Dollar Machine, a science-based programme, and a teacher training programme DAPPER.

The curriculum for Native American students (Schinke & Tepavac, 2000; RCT +) had long term effects on alcohol use, with students who participated in the programme less likely to report consuming alcohol, 30 and 42 months after receiving the programme. The effects of



the DAW programme were less robust, due to the weak study design employed (Wright, 2007; CBA –). However, compared to students in preceding years who did not receive the programme, DAW participants were less likely to have been drunk. However, overall the programme had a less consistent effect of the use of alcohol than tobacco or illegal drugs. The study of LST conducted by Botvin and colleagues (2003; RCT –) found that the intervention had effects on drinking frequency and drinking in the past year at post-test. However these findings were not replicated in a German study of LST (Bühler et al., 2008; RCT +) or in the evaluation of Project Charlie (Hurry & McGurk, 1997; Hurry et al., 2000; both RCT +). The effects of Say Yes First, a multicomponent programme that combined a universal curriculum with parent education and case management for high risk students, were also inconsistent (Zavela et al., 1997; 2004; CBA –). Alcohol use was found to be lower in eighth grade amongst students who had received the programme in fourth grade, but by the eleventh grade there were no significant difference between intervention and comparison students.

An evaluation of the Australian Life Education Centre curriculum (Hawthorne, 1995; CBA –) found that the curriculum had no short-term effects on alcohol use and had potentially increased alcohol use in boys who participated in the programme. One evaluation of the UK Life Education curriculum for 10-11 year olds (Decisions programme) found that the intervention had no effects on alcohol use (Tudor-Smith et al., 1995; UBA –).

Evidence statement 3

- 3(a) There is moderate evidence from one RCT¹ to suggest that a culturally tailored skills training intervention for Native American students may have long-term effects on alcohol use. However, this evidence is not applicable to the UK given the cultural specificity of this programme. There is insufficient and inconsistent evidence from four RCTs, four CBA studies and one UBA study² to determine the effects of other drug education approaches on alcohol use in later years.
- ¹ Schinke et al., 2000 (RCT +)
- ² Botvin et al., 2003 (RCT -); Bühler et al., 2008 (RCT +); Hurry & McGurk, 1997; Hurry et al., 2000 (both RCT+); Zavela et al., 1997, 2004 (both CBA -); Hawthorne, 1995 (CBA -); Wright 2007 (CBA -); Tudor-Smith et al., 1995 (UBA -)



Author (Year)	Design	Population	Follow-up	Analysed	Findings
Ambtman et al., 1990 Tuning In To Health: Alcohol and Other Drug Decisions	NRCT +	Canada 2 nd – 6 th grade N=2,406	Not clear	87% completed study	Intervention schools improved more than control schools on knowledge of essential elements of the programme.
Botvin et al., 2003 Life Skills Training	RCT -	USA 4 th to 6 th grade N= 1,954	PT	56%	Prevalence of drinking was significantly lower in intervention schools. Significant effects on attitudes (anti-drinking), knowledge, normative expectations and self-esteem. No effects on advertising knowledge, social skills knowledge, refusal skills knowledge, teen or adult drinking norms, and risk-taking.
Bühler et al., 2008 Allgemeine Lebenskompetenzen und Fertigkeiten	RCT +	Germany Mean 10.8 years N=643	PT	70% completed study	Significant programme effects on knowledge about skilled behaviour and life skill resources, but no programme effects were found concerning knowledge about unskilled behaviour and life skills deficits. Intervention students developed a more critical view against alcohol consumption, but intervention had no effects on alcohol use.
Hecht et al., 2008 keepin' it REAL	RCT +	USA Mean 10.4 years N=1,566	PT, 12 months	72% at 12 months	Intervention students reported greater increases in their quantity of resistance strategies but perceived that relatively more of his or her peers were using substances than control students. No difference in student's substance use intentions, parents' and friends' anti-drug injunctive norms, personal anti drug norms and substance use expectancies. No difference in lifetime or recent substance use.
Holtz & Twombly, 2007 Brain Power!	NRCT -	USA 4 th and 5 th grade N=112	PT	NR	Intervention group showed statistically significant improvements in knowledge about drugs.
Hurry & McGurk, 1997; Hurry et al., (2000) Project Charlie	RCT +	UK 10 years N= 120	PT, 3 years	77% at 3 years	No significant differences between intervention and control students in alcohol use at 13-14 years. Intervention students had significantly higher decision- making skills than control students at PT. No significant difference in self- esteem, intention to drink alcohol or peer pressure.
Kreutter & Gewirtz, 1991	CBA -	USA 6 th grade N=216	PT	NR	Intervention students had greater gains in knowledge and on measures of self- esteem.
Schinke & Tepavac, 1995 Million Dollar Machine	NRCT -	USA 3 rd - 6 th grade N= 2,475	PT, 6 months	NR	Fourth grade students in the intervention group reported significantly less actual and potential time spent drinking.

Table 5.22. Drug (including alcohol) education: classroom-based programmes led by teachers

Author (Year)	Design	Population	Follow-up	Analysed	Findings
Sigelman et al., 2004 Drug and alcohol curriculum	RCT +	USA Grades 3-6 N= 327	PT, 1 year	82% completed the study	Significant intervention effect on knowledge at PT, but not 1 year. No difference between groups in attitudes or intentions. Programme did not have any significant effects on alcohol use.
Stevens et al., 1996 Here's Looking at You 2000	CBA -	USA 4 th -6 th grade N= NR	36 months	NR	No effects on initiation or drinking for students in 4-6 grades at baseline.

Table 5.23. Drug (including alcohol) education: classroom-based programmes led by external contributors

Author (Year)	Design	Population	Follow-up	Analysed	Findings
Baker, 2004 Preventing Alcohol and Drug Abuse Through Primary Education (PADAPE)	CBA -	USA 2 nd -5 th grade N=1,521	1 year	NR	In phase one, significant overall increase in knowledge and skills at PT. In phase two, students who had not received the intervention programme in the previous year performed better than students who had received the PADAPE programme,
Hall-Long & Dishop, 1999 Drug education programme	UBA -	USA 1 st and 3 rd grade N=263	РТ	NR	There was an average increase in knowledge test scores by 30% compared to pre-test scores and increases on every area of knowledge examined
Peterson & Woodward, 1993 CHOICE programme	NRCT -	USA 6 th grade N=116	РТ	NR	Compared to students in the control group, students who received the CHOICE programme had a significantly higher self-concept as measured by semantic differential.
Schinke et al., 2000 Curriculum for Native American students	RCT +	USA Mean 10.3 years N= 1,396	6, 18, 30 and 42 months	86% completed study	Significantly smaller percentage of participants in skills, and skills + community conditions reported alcohol consumption at 30 and 42 months. Fewer participants in the skills only condition reported alcohol use at 30 and 42 months (vs. control).
Welham, 2007 Enrichment programme	UBA -	UK Year 8 N=240	PT	NA	Intervention with subsequent teacher support in-class affected positively children's knowledge of how to stay healthy and the likely impact of drugs, alcohol and smoking on the maintenance of health and wellbeing.

Author (Year)	Design	Population	Follow-up	Analysed	Findings
Wright, 2007 Drug At Work (DAW)	CBA -	USA 5 th and 6 th grade N=2,691	6 th and 7 th grade	NR	DAW participants (or their classmates) were significantly more likely to have negative attitudes to drinking alcohol, less likely to have been drunk and were significantly more likely to report havingfriends who would stop them from getting drunk.
Beginning Alcohol and Ad	dictions Basic	Education Stud	ies (BABES)	•	
Abbey et al., 1990	RCT +	USA 6-8 years N=55	1 month	NR	Significant effects on knowledge based directly on BABES material but not on a knowledge test applying BABES material to different situations. Intervention group reported significantly more negative attitudes about alcohol's effects than the control group. No effects on measures of coping, decision making, help seeking, peer pressure resistance, responsibility taking or self esteem. However, control group members demonstrated more active coping skills on one of the three coping scenarios.
Hahn et al., 2007 BABES Plus	NRCT +	USA Mean 5.8 years N=126	1 and 6 months post- intervention	NR	BABES Plus parents rated their children as having less anxiety/withdrawal than did the BABES Only parents, and as more socially competent than did parents in both the BABES Only and control groups. For the Aggression scale, neither of the main effects nor their interaction was significant.
Witt &Witt 1995	UBA -	USA 7-9 years N=132	PT	NR	Significant gains in knowledge (with the exception of knowledge relating to the lesson on self-image and feelings)

Table 5.24. Drug (including alcohol) education: multicomponent programmes

Author (Year)	Design	Population	Follow-up	Analysed	Findings
Rollin et al., 1992; 1995	RCT -	USA 3 rd grade	PT, 6-7 months	NR	Significant intervention effect on drug knowledge, and life management, decision making skills and refusal skills, and also self concept.
Project KICK		N=62			
Say Yes First					
Zavela et al., 1997	CBA -	USA 4 th grade N= 430	5 years	Not reported	Alcohol use lower than in preceding cohorts.
Zavela et al., 2004	CBA -	USA 4 th grade N= 156	8 years	Not reported	No significant difference between intervention and control students. No difference in attitudes to school or drugs, ability to resist peer pressure, social competence, or school performance and attendance.

Table 5.25. Drug (including alcohol) education: other in-school approaches

Author (Year)	Design	Population	Follow-up	Analysed	Findings
Allison et al., 1990 DAPPER	RCT -	Canada 5 th grade N=266	End of school year	82% completed study	No difference between groups on any of the measures of alcohol use at PT, or on knowledge, problem solving, coping and attitudes, or decision making.
Corbin et al., 1993 Refusal skills intervention	RCT -	USA 8-10 years N=74	PT, 4 weeks (intervention only)	77% completed study	GI participants had higher general knowledge scores, both R+ and GI groups had higher drug knowledge than the control group at PT (no difference between two intervention groups). No differences on assertiveness refusal behaviours between intervention groups, but R+ intervention had effects on decision making skills and demonstrated more refusal behaviours.
Paxton et al., 1998 Drug education programme	UBA -	UK 9-10 years N=1,428	Not clear	NR	Following the teacher training and delivery, more children said that they would take the drugs to the police or to their parents. Fewer pupils reported that they would intend to talk to the police if offered or found drugs and more would intend to talk to their family, teacher or friends.
Raybuck & Hicks, 1994 KIDS CARE	CBA -	USA 3 rd and 5 th grade N=132	1-2 weeks	NR	No significant effect of intervention on self-esteem on standardised measures, intervention group significantly improved on the positive scale of the "Circle Words" measure of self-esteem compared to the control group (no significant change on the negative scale). Significant effect on the sociometric status of previously rejected or neglected children with high risk children in the intervention group more likely to receive positive peer nominations than control children.
Starkey & Orme, 2001 Theatre in Education	UBA -	UK 10-11 years N= 6 schools	4 weeks	85-98%	Significant increases were seen in children's ability to name specific drugs and response to seeking help when presented with a lost bag. Children also showed change in attitudes towards drugs with the realisation that some drugs could be good for you for medical reasons.
Life Education Centres			-	•	
Hawthorne et al., 1995; Hawthorne, 1996	CBA -	Australia Year 6 N= 3,019	PT	Not reported	Intervention had significant negative effects on alcohol use, particularly in boys. No preventive effects of the programme at the school or population level. Indication that programme was harmful.
Tudor-Smith et al., 1995	UBA -	UK 10-11 years N=509	1 week, 2 months	67% completed FU	Significant improvement after two months in ability to recognise substances such as: heroin, pharmaceuticals, cigarettes, and alcoholic drinks as drugs. The majority of beliefs about drugs did not change after two months and there was no statistically significant changes in substance use behaviours.

5.4 Sex and relationships education

5.4.1 Overview of evidence identified

Overall nine studies were identified that examined seven programmes, which focused on different approaches to SRE. Two programmes focused on abstinence approaches; three programmes were HIV/AIDS prevention approaches; one programme employed a parenting and care-giving approach; and another aimed to improve young people's sexual health knowledge, personal insight and motivation. All seven programmes were curriculum based and delivered in schools.

5.4.2 Abstinence-based programmes

Four studies were identified that examined two abstinence-based approaches to SRE. One study (Abel & Greco, 2008) examined the Family Action Model for Empowerment (FAME) and three studies (Denny et al., 1999; Denny & Young, 2006; Spear et al., 1997) examined the Sex Can Wait (SCW) Programme. A summary of programme content is presented in Table 5.26.

Programme	Reference(s)	Programme content
FAME (Family Action Model for Empowerment)	Abel & Greco, 2008	 8-week curriculum administered within the public school system After-school programme continued for eight sessions; employed a psychoeducational model and art; music, dance, group activities, role-play, audiovisual aids and written materials.
Sex Can Wait	Denny et al., 1999; Denny & Young, 2006; Spear et al., 1997	 Curriculum series taught at upper elementary school, middle school and high school. Teachers trained at a workshop over 3.5 days. Delivered in 23 lessons over 5 weeks.

Table 5.26. Summary of programme content: abstinence-based programmes

5.4.2.1 FAME (Family Action Model for Empowerment)

One study (Abel & Greco, 2008) evaluated an abstinence-orientated empowerment programme to prevent teenage pregnancy (FAME). This was a multi-dimensional school and community-based intervention which aimed to strengthen healthy family functioning and highlight sexual abstinence as a positive choice for young people. Delivered over an eightweek period, this programme focussed upon asset building, parent-teen communication and relationships, self-esteem, healthy relationships dealing with peer pressure and the benefits of valuing abstinence approaches.

Quality assessment

This study employed a UBA design and was given a UBA – rating. This rating reflected the poor study design and the limitations of the study findings as a result of immediate post-test results only and substantial levels of attrition (>20%).



Findings

Post-test results showed that the programme had significant positive effects on the following outcomes (all p<0.05): attitudes towards parents communication with their child or children, overall communication with parents, ability to resist peer pressure, self-esteem and perceived ability to abstain from sex (Abel & Greco, 2008; UBA –).

5.4.2.2 Sex Can Wait

Three studies reported on the Sex Can Wait (SCW) programme (Spear et al., 1997; Denny et al., 1999; Denny & Young, 2006). The SCW programme was an abstinence-based curriculum delivered across 23 lessons over a five week period, addressing self-esteem, reproductive anatomy and physiology, changes associated with puberty, values and decision-making skills. The programme included upper elementary, middle school and high school components, and only the results of the upper elementary curriculum are reported here.

Quality assessment

One study (Spear et al., 1997) of the SCW programmes was based on a CBA design, whilst the second study (Denny et al., 1999; Denny & Young, 2006) employed an NRCT design. Study quality was difficult to determine as the authors reported limited details of their methodology. Furthermore, as control and intervention students were located in the same schools there are concerns regarding contamination. In addition, there was either a high rate of attrition (Denny & Young, 2006) or attrition was not reported (Spear et al., 1997), and poor reporting of baseline comparisons. Both studies were rated as '-' for internal consistency.

Findings

Spear and colleagues (1997; CBA –) field tested the SCW programme and reported that at post-test, participants in the intervention group showed significantly higher knowledge scores (p<0.001), expressed more positive attitudes towards abstinence (p<0.01) and more desirable attitudes (p<0.05). However, no significant differences were found for parental communication factors or intent to remain abstinent.

Denny and colleagues (1999; NRCT –) also reported significant post-test differences between the intervention and control groups for knowledge scores (p<0.001), greater levels of self-efficacy (p<0.05), and attitudes towards abstinence (p<0.05). Intervention students were also more hopeful about the future than control students (p<0.05), although they showed no significant differences between pre- and post-test for sexual behaviour or abstinent intentions. At 18 months follow-up (Denny & Young, 2006; NRCT -), results showed that students in the intervention group had significantly higher knowledge scores than control students (p<0.05). There was no significant difference between the intervention



and control groups for self-efficacy, decision-making, attitudes, hopefulness, intention to remain abstinent (see Table 5.27). For two sexual behaviour outcomes, there was no significant difference between intervention and control students for whether they had ever had sexual intercourse (OR 0.59; 95% CI 0.21, 1.64), but students in the intervention groups were less likely than control students to report that they had had sexual intercourse in the past 30 days (OR 0.08; 95% CI 0.03, 0.23). Effect sizes were calculated and are presented in Table 5.27.

Table 5.27. SCV	V: Intervention	effects at	18-month	follow-up	(Denny a	& Young,	2006;
NRCT -)							

Quitaomo	Intervention			Control			Standardised
Outcome	Mean	SD	Total	Mean	SD	Total	(95% CI)
Knowledge	62	15	158	50	15	38	0.80 (0.43, 1.16)
Attitude	3.49	0.51	158	3.47	0.59	38	0.04 (-0.32, 0.39)
Hopelessness	73	19	158	73	19	38	0.00 (-0.35, 0.35)
Self-efficacy	3.87	0.74	158	3.64	0.71	38	0.31 (-0.04, 0.67)
Decision making	3.33	0.51	155	3.34	0.51	37	-0.02 (-0.38,0.34)
Intention to remain abstinent	3.3	1.27	156	2.7	1.28	35	0.47 (0.10, 0.84)

5.4.3 AIDS education programmes

Three studies were identified that examined three AIDS education programmes (Gaskins et al., 2002; Pick et al., 2007; Schonfeld et al., 1995). A summary of programme content is presented in Table 5.28.

	mine components. ADS	education programmes
Programme	Reference(s)	Programme content
An HIV/AIDS Awareness Education Programme	Gaskins et al., 2002	 Month-long curriculum-based education programme 1-2 hours for kindergarten-1st grade students; 3-4 hours for 2nd-3rd grade students; 4-5 hours for 4th-5th grade students.
I Want to, I CanPrevent HIV/AIDS	Pick et al., 2007	 Curriculum-based programme implemented over 15-20 weeks. Teacher training workshops over 40 hours. Incorporating colourful, interactive workbooks, group work, brain-storming, role-playing, storytelling, debating, discussions and audiovisual activities.
AIDS Education Programme	Schonfeld et al., 1995	 Programme delivered to kindergarten, 2nd grade and 4th grade students. Six 45-60 minute lessons over a three week period. Developmentally based programme that included demonstration, drawing exercises and interactive activities.

Table 5.28.	Programme	components	AIDS educa	ation program	nmes
10010 0.20.	riogramme	components		alion program	11103



5.4.3.1 HIV/AIDS Awareness Education Programme

One study reported the results of an HIV/AIDS awareness education programme delivered to students from kindergarten to fifth grade (Gaskins et al., 2002). The programme was delivered in 1-5 hours over one month dependent upon the age of the target population. Sessions are delivered to students in the classroom and utilised art, puppets, films, books and lectures.

Quality assessment

Gaskins and colleagues (2002) used an UBA design to evaluate the effects of the programme. The lack of a control group limits the reliability of the study results. In addition, limited detail of the intervention was provided and only immediate post-test analysis was carried out. The study was assessed as 'UBA –' for internal consistency.

Findings

Post-test results from Gaskins and colleagues (2002; UBA –) for fourth and fifth grade students showed significant improvements in knowledge scores relating to HIV transmission, with a decline in only one question relating to HIV symptoms. Second and third grade students reported similarly significant improvements in knowledge scores (p<0.001), with all questions showing an increase in the percentage of correct responses post-test. In comparison, kindergarten and first grade students reported significantly lower knowledge scores at post-test (p<0.001). However, all grades showed significantly increased levels of comfort in proposed social scenarios with an HIV positive person (p<0.001).

5.4.3.2 I Want to, I Can... Prevent HIV/AIDS

Pick and colleagues (2007) reported on an HIV prevention programme I Want to, I Can...Prevent HIV/AIDS, a life skills programme which was designed to promote communication as a protective factor against high-risk sexual behaviour. This study incorporated a variety of creative classroom strategies (e.g. storytelling, role-playing) with fourth grade students over a 15 to 20 week period throughout the school year.

Quality Assessment

The study by Pick and colleagues (2007) was based on an RCT cluster design and appeared to have been well designed; however the method of allocation to intervention and control groups was not explained. Other aspects of the study were well reported; the authors reported controlling for baseline measures and provided further details to show that there was no contamination between groups. The study was coded as 'RCT +'.



Findings

Pick and colleagues (2007; RCT +) reported significant post-test results in the intervention group for improved attitude towards communication (p<0.001), improved scores for self-efficacy (p<0.001), intentions regarding communication (p<0.001), communication behaviour (p<0.001), and perceived norms about communication (p<0.001). At post-test the intervention group was also more likely to discuss behaviours on taboo (p<0.001), romantic (p<0.001) and threatening or unpleasant topics (p<0.05). Girls were more likely than boys to report positive attitudes, self-efficacy and intentions towards discussing difficult topics. However, boys were significantly more likely to discuss threatening or unpleasant topics (p<0.001).

5.4.3.3 AIDS Education Programme

Schonfeld and colleagues (1995) reported on an AIDS education programme delivered to kindergarten, second grade and fourth grade students over a three-week period. This developmentally based education programme aimed to teach children to differentiate between communicable and non-communicable illnesses with specific reference to HIV/AIDS.

Quality Assessment

This study by Schonfeld and colleagues (1995) employed an appropriate methodology with study population being representative of the eligible population, there were no significant differences at baseline between the intervention and control groups, and researchers were blind to the control/treatment condition during testing and analysis. However although allocation was reported as randomised, the authors provided limited information about the allocation of classrooms to the intervention and control groups, hence this study was assessed as 'RCT +' for internal consistency.

Findings

Results of the study by Schonfeld and colleagues (1995; RCT +) showed that at two and a half months follow-up, the intervention group had significantly higher scores for knowledge of the following: causality and prevention of AIDS (p<0.001); the causality and prevention of colds (p<0.01); and causality of cancer (p<0.05). However, there was no difference between groups for knowledge of cancer prevention at follow-up. Students in the intervention group mentioned significantly more accurate causes of HIV/AIDS (e.g. sexual transmission, blood transmission, mother-to child transmission). However, significantly fewer children in the intervention group mentioned injecting drug use as a route of HIV/AIDS transmission (p<0.01). Overall, the intervention was equally effective across all grades.



5.4.4 Other in-school approaches

Two studies (Masterpasqua et al., 1992; Wackett & Evans, 2000) were identified that examined other approaches to SRE. Masterpasqua and colleagues (1992) examined a parenting programme for fifth and sixth grade students and Wackett and Evans (2000) examined an intervention designed to improve young people's sexual health knowledge, motivation, personal insight and skills. A summary of programme content is presented in Table 5.29.

Programme	Reference(s)	Programme content
Learning about Parenting/Learning to Care	Masterpasqua et al., 1992	 Two cohorts of 5th-6th graders received the programme over two years. The programme focussed on developmental milestones, individual differences, and parental care for children.
Choices and Changes	Wackett & Evans, 2000	 Programme aims to improve sexual health knowledge, motivation and personal insight and skills. Delivered in eight 1 hour sessions over four weeks.

Table 5.29. Programme components: other in-school approaches

5.4.4.1 Learning about Parenting/Learning to Care

One study (Masterpasqua et al., 1992) reported findings from the school-based Parenting and Care-giving programme, which was aimed at improving children's understanding of care in order to prevent the long-term impact of negative early childrearing experiences. The authors reported on data from two cohorts of fifth and six grade students.

Quality Assessment

Masterpasqua and colleagues (1992) failed to report details of how classrooms were allocated to intervention or control groups. They also did not provide sufficient information about baseline measurements, pre-test results or follow up times. Therefore, this study was coded as a 'NRCT –'.

Findings

The post-test results from year one (without pre-test results) showed a significant increase in the number of parenting (p<0.01) and nurturing solutions (p<0.05) that intervention students were able to provide to common parent-child problems, compared to control (Masterpasqua et al., 1992; NRCT –). However, there was no difference between intervention and control groups in parenting knowledge or the number of physically punishing solutions provided. The results for year two (where pre-test results were also available) showed that intervention students had significant increases in parenting knowledge (p<0.001) and in the number of total and caring solutions provided in response to parent-child problems (p<0.001 and p<0.01, respectively), compared to control students. Intervention students also reported a



reduction in the number of physically punishing solutions (p<0.001), compared to the control group. Effect sizes were calculated for these outcomes and are presented in Table 5.30.

Table 5.30. Learning	about Parenting:	Intervention	effects (M	lasterpasqua e	et al., '	1992;
NRCT -)						

Quitaama	Intervention			Control			Standardised mean	
Outcome	Mean	SD	Total	Mean	SD	Total	(95% CI)	
Year 1								
Parenting knowledge	33.99	3.73	67	33.05	4.17	71	0.24 (-0.10, 0.57)	
Total solutions	18.91	2.85	67	17.29	3.17	71	0.53 (0.19, 0.87)	
Care solutions	5.2	1.81	67	4.43	1.87	71	0.42 (0.08, 0.75)	
Punishing solutions	0.52	0.56	67	0.59	0.84	71	-0.10 (-0.43, 0.24)	
Year 2								
Parenting knowledge	34.59	3.63	108	32.97	4.08	109	0.42 (0.15, 0.69)	
Total solutions	18.85	3.43	108	17.52	3.53	109	0.38 (0.11, 0.65)	
Care solutions	5.59	1.8	108	5.03	1.82	109	0.31 (0.04, 0.58)	
Punishing solutions	0.8	0.7	108	0.91	0.88	109	-0.14 (-0.40, 0.13)	

5.4.4.2 Choices and Changes programme

One study (Wackett & Evans, 2000) examined the effects of the Choices and Changes programme, which aimed to improve young people's sexual health knowledge, motivation, personal insight and skills.

Quality Assessment

Wackett and Evans (2000) used a UBA design to examine the effects of the Choices and Changes programme. The authors reported limited information on the intervention and control groups, and no details of baseline measurements were reported. Overall the design of the study was limited and was assessed as a 'UBA –' for internal consistency.

Findings

Few analyses were conducted on the data, however proportions responding positively to the intervention were tabulated and pooled results examined (Wackett & Evans, 2000; UBA –). Findings showed that participants' responses for assertiveness skills were high at pre-test (70.9%) and remained similar at follow-up (between 70.7% and 66.3%). Views of supporting environments (43.8%) improved slightly at post-test and follow-up times (47.2%, 50.0% and 56.3% respectively). Pooled results showed an improvement in knowledge regarding, for example, fertility and anatomy (from 58.8% pre-test to 65.6% at final follow-up). Similarly, young people showed stronger positive views regarding the importance of the life programme for themselves (from 69.9% to 77.4%) and their peers (from 68.4% to 77.1%).



Participants also reported stronger positive views on the importance of setting boundaries (from 78.7% to 88.2%); the influence of the media on young people's body image and sporting performance (from 44.8% to 51.2%); and (limited to grade 7 only) their views that it is important for young people to discuss and set sexual limits when dating (from 52.2% to 69.8%). Further results indicated a small decrease in self efficacy at final follow-up (from 70.9% to 66.3%).

5.4.5 Summary and evidence statements

A total of seven programmes were identified that examined primary school-based sex and relationship interventions. The programmes identified included a variety of intervention approaches including abstinence-based approaches, HIV/AIDS prevention, parenting and care-giving, and sexual health education.

5.4.5.1 Knowledge and understanding

Five programmes reported on the impact on participants' knowledge and understanding. Young people participating in the SCW programme (Spear et al., 1997; CBA -; Denny et al., 1999; Denny & Young, 2006; both NRCT -) reported improved knowledge relating to the abstinence-based curriculum at immediate post-test and at 18 months follow-up. Two HIV prevention programmes (Gaskins et al., 2002; UBA -; Schonfeld et al., 1995; RCT +) reported improvements in knowledge acquisition, however the findings of the study carried out by Gaskins and colleagues (2002; UBA -) were inconsistent and insufficient to evaluate in light of the poor quality of the study. Masterpasqua and colleagues (1992; NRCT -) reported improvements in the intervention group for parenting knowledge, whilst Wackett & Evans (2000; UBA -) reported marginal improvements in sexual health knowledge. However, both these studies were poorly designed.

5.4.5.2 Attitudes and values

Overall, four programmes reported outcomes relating to attitudes and values. In a programme designed to prevent teenage pregnancy (Abel & Greco, 2008; UBA –), participants reported an increase in feelings that they mattered to their parents, and increased self-efficacy and behavioural intentions towards abstinence. Participants in the SCW programme reported no difference in attitudes or values at post-test (Spear et al., 1997; CBA –; Denny et al., 1999; NRCT –) or at follow-up (Denny & Young, 2006; NRCT –). Pick and colleagues (2007; NRCT +) reported that the effects of their programme resulted in improved attitudes towards communication at one year follow-up. Wackett and Evans, (2000; UBA –) showed that participants reported increased attitudes and values on the importance of the programme, views of the media and communicating about sex. These results were maintained at follow-up.



5.4.5.3 Personal and social skills

Personal and social skills outcomes were reported in five programmes. Participants in the FAME programme (Abel & Greco, 2008; UBA –) reported significant increases in personal and social skills including improved communication with their parents. Gaskins and colleagues (2002; UBA –) reported that participants in an HIV/AIDS awareness education programme showed significant increases in perceived ease of socialising with HIV positive people. Pooled results from the Choices and Changes programme (Wackett & Evans, 2000; UBA –) showed no difference in assertiveness skills at follow-up. Participants in the SCW programme reported greater levels of self-efficacy, but no difference in levels of decision-making at follow-up (Denny et al., 1999; Denny & Young, 2006; both NRCT –). Pick and colleagues (2007; RCT +) reported that the effects of their programme resulted in increased communication with parents at one year follow-up.

5.4.5.4 Health and social outcomes related to alcohol and sexual health

Health and social outcomes were examined for the SCW programme (Denny et al., 1999; Denny & Young, 2006; both NRCT –). Findings showed that there was no difference between the intervention and control groups for behavioural intentions. However, at 18 months follow-up, the intervention group was less likely to report sexual activity in the past 30 days in comparison with the control group (Denny & Young, 2006; NRCT –).



Evidence statement 4

- 4(a) There is weak evidence from two NRCTs¹ to suggest that an abstinence education programme that targeted children aged 10-12 years can improve sexual health knowledge, but the long term impact on sexual behaviours is less clear. This evidence may be directly applicable to PSHE delivery in primary schools focusing on SRE and alcohol education because the curriculum topic and content of this programme are relevant.
- 4(b) There is moderate evidence from one RCT² to suggest that SRE programmes targeting communication, such as I Want to, I Can...Prevent HIV/AIDS, can improve parent and child communication about sexual health. This evidence may be directly applicable to the UK because the curriculum topic and content of these programmes are relevant to PSHE delivery in primary schools focusing on SRE and alcohol education.
- 4(c) There is inconsistent and insufficient evidence from two NRCTs, one CBA study and two UBA studies³ to determine the effectiveness of SRE programmes on attitudes and values relating to sexual health.
- ¹ Denny et al., 1999; Denny and Young 2006 (both NRCT -)
- ² Pick et al., 2007 (RCT +)
- ³ Abel and Greco 2008 (UBA -); Spear et al., 1997 (CBA -); Denny et al., 1999; Denny and Young 2006 (both NRCT -); Wackett and Evans 2000 (UBA -)



Author (Year)	Design	Population	Follow-up	Analysed	Findings
Abel & Greco, 2008 FAME (Family Action Model for Empowerment)	UBA -	USA 5 th -9 th grade N=123 (intervention)	PT	>20% loss to follow- up	Significant changes in attitudes at PT.
Sex Can Wait					
Spear et al., 1997	CBA –	USA 5 th -6 th grade N=287	РТ	Not reported	Intervention group had significantly higher knowledge scores, attitudes towards abstinence and more life skill factors.
Denny et al., 1999	NRCT -	USA 5 th -6 th grade N=301	РТ	Not reported	The intervention group reported significantly higher knowledge scores, desirable attitudes and attitudes towards abstinence.
Denny et al., 2006	NRCT -	USA 5 th -6 th grade N=376	18 months	196 (52%) at 18 month follow up	Intervention students had significantly higher knowledge scores than control students, and were significantly less likely to have had sexual intercourse in the past 30 days. No significant difference between groups for self-efficacy, decision-making, attitudes, hopefulness, or behavioural outcomes: intention to remain abstinent, or whether ever had sexual intercourse.

Table 5.31. Sex and relationships education: Abstinence-based programmes

Table 5.32. Sex and relationships education: AIDS/HIV education programmes

Author (Year)	Design	Population	Follow-up	Analysed	Findings
Gaskins et al., 2002 An HIV/AIDS Awareness Education Programme	UBA –	USA Kindergarten-5 th grade N=339	PT	358 (one class was not available at pre- test)	Kindergarten and 1 st grade students reported a significant decrease in knowledge scores, but, 2 nd -5 th grade students reported significant increases. Increased reports of perceived comfort at being around an HIV positive person for all grades.
Pick et al., 2007 I Want to, I CanPrevent HIV/AIDS	RCT +	Mexico 4 th grade N=1581	1 year	Only those completing pre- and post-test were reported	Intervention group showed significant improvements in communication attitudes, self-efficacy, intentions, behaviour, and perceived socio-cultural norms relating to communication. At PT the intervention group were also more likely to discuss behaviours on taboo, romantic and unpleasant topics.

Author (Year)	Design	Population	Follow-up	Analysed	Findings
Schonfeld et al., 1995 AIDS Education Programme	RCT +	USA Kindergarten-6 th grade N=189	2.5 months	166 (88%)	Intervention group had significantly higher scores for the causality and prevention of AIDS and colds and causality of cancer, and mentioned significantly more correct causes of AIDS. No difference in knowledge of cancer prevention was identified after the intervention.

Table 5.33. Sex and relationships education: Other in-school approaches

Author (Year)	Design	Population	Follow-up	Analysed	Findings
Masterpasqua et al., 1992 Learning about Parenting/Learning to Care	NRCT -	USA 5 th -6 th grade N= 217	PT	Not reported	Significant improvements in the intervention group's parenting knowledge, including the number of total solutions and positive solutions to parent-child problems they were able to provide in comparison to the control group.
Wackett & Evans, 2000 Choices and Changes	UBA -	Canada 4 th -7 th grade N=938	PT, 1 and 3-4 months	10-15% loss to follow- up	Increases in sexual health knowledge, but no change in assertiveness skills or supportive environments. The importance of when dating students to discuss sexual limits at the outset of a relationship reported a high response at follow-up.

5.5 General health education programmes

5.5.1 Overview of evidence identified

Three studies (Andrews, 1992; Utley et al., 2001; Young et al., 1997) were identified that examined general health education programmes which included modules or topics related to alcohol education or SRE.

Programme	Reference(s)	Programme components		
Growing Healthy Curriculum	Andrews, 1992	 School health curriculum for kindergarten to sixth grade Length/intensity not reported Delivered by teachers 		
Peer tutoring	Utley et al., 2001	 Health education curriculum based on peer tutoring programme for students with developmental disabilities 3 sessions a week for 3 weeks Taught by teachers and peers 		
Contemporary Health Series	Young et al., 1997	 Life skills modules from a health education curriculum 16 modules in total; 3 compulsory life skills modules and 13 optional modules Taught by teachers and counsellors 		

Table 5.34. Programme components: General health education programmes

5.5.2 General health education programmes

5.5.2.1 Growing Healthy Curriculum

Andrews (1992) examined the impact of the Growing Healthy Curriculum. The programme focused on improving students' attitudes towards good health practices and behaviour. The programme was delivered in kindergarten to sixth grade but no further details about the programme were reported.

Quality assessment

Although a control group was utilised in the evaluation of the Growing Healthy Curriculum (Andrews, 1992), the study methodology was poorly reported. There were few details describing the intervention and the number of participants in the study was not reported although it was reported that the study included students from five school districts. Overall the study was rated 'CBA –' for internal consistency.

Findings

There was little evidence to suggest that the Growing Healthy Curriculum programme (Andrews, 1992; CBA –) had an impact on attitudes to drinking, although students who received the programme from kindergarten to sixth grade were significantly less likely to think they would drink as adults in the fifth and ninth grades (p<0.05). There were no differences between groups for the percentage of students who had tried alcohol. However,

students who received the Growing Healthy curriculum from kindergarten to sixth grade were significantly more likely to drink on a regular basis in third, fifth, sixth and ninth grades (p <0.05) than control students.

5.5.2.2 Peer tutoring for students with developmental disabilities

Utley and colleagues (2001) examined the effectiveness of class wide peer tutoring (CWPT) in a health education curriculum for children aged 7-9 years with developmental disabilities. Further information was not reported about the nature of the students' disabilities but intelligence test scores in the sample ranged from 52 to 57. The units of the curriculum covered the following topics: body parts and their functions; poisons; drugs and their effects; and dangerous situations. Prior to implementation, students and teachers participated in training sessions for CWPT, after which, CWPT was delivered for 30-minutes for three days per week. After five weeks, peer tutoring was withdrawn for two weeks and traditional teaching methods used followed by three more weeks of peer tutoring.

Quality assessment

The study by Utley and colleagues (2001) was the only study identified in the literature searches that conducted research in a sample with disabilities. Disability research has historically relied on small samples and uncontrolled designs. Researchers may face difficulties in finding research participants and it may not be possible to find an adequately matched comparison groups (Odom et al., 2007). However this study was based on a particularly small sample size and consequently tests of significance were not reported. The authors described the study as 'a BAB reversal design' (i.e. intervention was delivered [B], then withdrawn [A], and then reinstated [B]), and a control group was not used for comparison, rather students' outcomes were compared over the period when CWPT was and was not in place. The study was rated 'UBA –' for internal consistency.

Findings

Utley and colleagues (2001; UBA -) reported that students had an increase in knowledge on all of the areas covered by the curriculum.

5.5.2.3 Contemporary Health Series

Young and colleagues (1997) examined the effects of selected modules from the Contemporary Health Series for sixth grade students, which consisted of 16 modules focusing on health and life skills. Following training, teachers implemented three life skills modules which focused on skills development (the remaining 13 modules were optional).



Quality assessment

The design of the study used to evaluate the Contemporary Health Series (Young et al., 1997) was not clearly described but was judged to be a CBA, as allocation to intervention and control groups was not described as experimental. As the study methodology was poorly reported the study was rated '-' for internal consistency.

Findings

Compared with the control group, students who received the Contemporary Health Series (Young et al., 1997; CBA –) reported a greater positive change in their school and home self-esteem (p<0.05 and p<0.001, respectively), relationship/communication characteristics (p<0.001) and decision-making skills (p<0.05). Intervention students also reported less positive attitudes toward the use of alcohol than control students, and showed positive changes on the measures of practices and perceptions of peers' and parents' norms regarding drug use and other illegal drugs (all p<0.001).

5.5.3 Summary and evidence statements

Three studies examined general health education programmes that included modules or topics related to alcohol education and SRE. One study examined a programme for children with developmental disabilities.

5.5.3.1 Knowledge and understandings

One study of an intervention that targeted general health behaviours examined the impact on knowledge. A peer tutoring intervention for children with developmental disabilities (Utley et al., 2001; UBA –) resulted in increases in knowledge about the body and its functions, and the effects of drugs. However, these findings were based on a small sample size (n=5) so should be interpreted with caution.

5.5.3.2 Attitudes and values

The impact on attitudes and values was examined in two general health education approaches. There were no effects of the Growing Healthy Curriculum for kindergarten to sixth grade students on attitudes towards alcohol use (Andrews, 1992; CBA –). The life skills components of a general health curriculum were found to have had an impact on attitudes to alcohol at post-test, with intervention students reporting less positive attitudes (Young et al., 1997; CBA –).

5.5.3.3 Personal and social skills

One study (Young et al., 1997; CBA –) examined programme impact on personal and social skills. The life skills components of a general health curriculum were found to have resulted



in positive changes in self esteem and in decision making skills among intervention students at post-test.

5.5.3.4 Alcohol and sexual health

No reported studies reported this outcome

Evidence statement 5

1

5(b) There is insufficient and inconsistent evidence from two CBA studies and one UBA study¹ to determine the effects of general health education programmes that targeted primary school age children on outcomes related to alcohol use and sexual health.

Andrews 1992 (CBA -); Young et al., 1997 (CBA -); Utley et al., 2001 (UBA -)



Table 5.35	General	health	education	programmes
------------	---------	--------	-----------	------------

Author (Year)	Design	Population	Follow-up	Analysed	Findings
Andrews, 1992 Growing Healthy Curriculum	CBA -	USA Kindergarten 6 th grade N= NR	Tested every year 6 th -12 th grade	NR	No evidence that the programme had any impact on alcohol use or behaviours.
Utley et al., 2001 Peer tutoring	UBA -	USA 7-9 years N=5	PT	100%	Results showed an increase in knowledge for all areas covered in the intervention: body parts, body functions, poisons, drugs and their effects, dangerous situations.
Young et al., 1997 Contemporary Health Series	CBA -	USA 6 th grade N=328	PT	NR	Intervention resulted in a positive change compared with control group in pupils' school and home self-esteem, relationship/communication characteristics, decision-making skills, and attitudes toward the use of alcohol.

5.6 Social development programmes

5.6.1 Overview of evidence identified

A total of 16 studies were identified that examined seven programmes focused on social development interventions designed to positively influence later behaviour. Six programmes combined school and family-based components while one programme was school-based only. Six programmes looked at outcomes related to alcohol or drugs and one programme investigated outcomes related to both substance misuse and sexual health.

5.6.2 Programmes with a school-based component

Three studies (Kellam et al., 2008; Poduska et al., 2008; van Lier et al., 2009) were identified that examined one social development programme, the Good Behavior Game, which included school-based components only.

Table 5.36. Summary of programme content: Single component social developmentprogrammes

Programme	Reference(s)	Programme content
Good Behavior Game	Kellam et al., 2008; Poduska et al., 2008; van Lier et al., 2009	 2 year programme Classroom based game; children rewarded for adhering to class rules Children assigned to groups and encouraged to manage their own and team mates behaviour.

5.6.2.1 Good Behavior Game

Three articles (Kellam et al., 2008; Poduska et al., 2008; van Lier et al., 2009) examined the effects of the Good Behavior Game (GBG), which aimed to promote pro social behaviours whilst reducing disruptive and aggressive behaviour in the classroom over two years in primary school aged children. Two articles (Kellam et al., 2008; Poduska et al., 2008) reported on the effects of the programme at age 19-21, in a cohort of students from schools in Baltimore, USA, who received the programme in the first and second grades (6-8 years old). van Lier and colleagues (2009) examined the impact of the programme on young adolescent outcomes in a sample of 7-year old children from elementary schools in Rotterdam and Amsterdam, the Netherlands. The GBG involved implementing teacher and student chosen rules and rewarding children who did not violate the rules.

Quality assessment

Both studies of the GBG were cluster RCTs and were rated '++' for quality. Both studies detailed well-described interventions, significant follow up times, intention to treat analysis and well reported outcomes. However although well reported, neither study fully detailed the methods used to randomly assign clusters and only the Baltimore-based study (Kellam et al., 2008; Poduska et al., 2008) adequately described the source population.



Findings

Kellam and colleagues (2008; RCT ++) reported a reduction in lifetime alcohol use/dependence disorders in those who participated in the Good Behavior Game; intervention participants reported marginally significantly fewer alcohol use/dependence disorders in young adulthood compared to internal control classes (p=0.08) and significantly fewer compared to external classes (p<0.05). A marginally significant reduction in the odds of an alcohol diagnosis (OR 0.50; 95% CI 0.25, 0.99) was also reported, implying a 50% reduction in the odds of a lifetime alcohol abuse disorder for those in the intervention group. In young adulthood (Poduska et al., 2008; RCT ++), rates of any service use were significantly lower for males in the intervention group compared to internal controls in both cohorts (p<0.05). Rates of drug treatment service use did not significantly differ between groups. In cohort two significantly less males in the intervention group accessed mental or medical health services than internal controls (p<0.01).

In the Dutch study, van Lier and colleagues (2009; RCT ++) found no significant effect of participation in the Good Behavior Game on past month or past year alcohol use at three or six year follow up. Over half (54%) of children reported consuming alcohol between the ages of 10-15. However, the authors found that compared to the control group, the rate of growth of alcohol use between three and six years following participation in the intervention was significantly reduced (p<0.05).

5.6.3 Programmes combining school and family-based components

Twelve studies were identified that examined six social development programmes, which combined school and family-based components.

Programme	Reference(s)	Programme content
Seattle Social Development Project	Hawkins et al., 1999; 2005 ; Lonczak et al.,. 2002 ; O'Donnell et al., 1995	 5 or 2 year programme versions Classroom instruction and management Child skill development Parent intervention
Raising Healthy Children	Brown et al., 2005; Catalano et al., 2003	 Teacher and staff development workshops After-school tutoring sessions and study clubs (Grades 4-6), Parenting workshops and in-home services for selected families (Grades 1-8).
Linking the Interests of Families and Teachers	Eddy et al., 2003; Reid et al., 1999	 Classroom-based programme (20 lessons) Playground behaviour intervention Parent management training programme and weekly newsletters Ongoing access to a classroom-based telephone answering machine

Table	5.37.	Summary	of	programme	content:	Multi-component	social	development
progra	amme	S						



Programme	Reference(s)	Programme content
Developmental drug prevention programmes	Furr-Holden et al., 2004; Ialongo et al., 1999	 Classroom centred intervention: curriculum enhancements, classroom behaviour management practices. Family School Partnership: staff trained in parent- school communication, weekly home-school activities, parent workshops
Child Development Project	Battistich et al., 2000; 2004	 3 year classroom curriculum in the upper 3 grades of elementary school School wide activities Family involvement activities at home
Positive Action Program	Flay et al., 2003	 Classroom curriculum (140 lessons per grade) School wide climate programme Parental and community involvement

5.6.3.1 Seattle Social Development Project

Four articles reported on evaluations of the Seattle Social Development Project (SSDP) (Hawkins et al., 1999, 2005; Lonczak et al., 2002; O'Donnell et al., 1995). The SSDP was delivered to students in the first to sixth grades (full intervention) or fifth and sixth grade only (late intervention), and included modified teaching practices, child social skills training, and developmentally appropriate parent training (O'Donnell et al., 1995). In the fifth and sixth grades, parents were offered participation in Preparing for the Drug Free Years, a five session programme designed to reduce a child's risk for drug use (Hawkins et al., 1999). Both studies were part of a larger ongoing longitudinal study. O'Donnell and colleagues (1995) reported outcomes for students who had received the full intervention programme at the beginning of fifth grade and at the end of sixth grade. Hawkins and colleagues (1999) reported 6-year follow-up data for all fifth grade students assigned to the full and late intervention or control groups. Hawkins and colleagues (2005) reported follow-up data relating to mental health, crime and substance use at age 21 for those who received the full intervention group.

Quality assessment

Evaluation of the SSDP was based on a quasi-experimental design (NRCT), with participants non-randomly assigned to intervention or control groups. The study by O'Donnell and colleagues (1995) appeared to have been adequately conducted. The intervention and comparison conditions were well described and the authors tested for attrition biases within the sample. The study was coded 'NRCT +'. The study by Hawkins and colleagues (1999) was also well reported and judged to have been adequately conducted. The study experienced a low rate of attrition, with 93% of participants followed up at 6 years, and was rated 'NRCT +'. Two studies (Lonczak et al., 2002; Hawkins et al., 2005) were follow-up studies of Hawkins and colleagues (1999) so their quality assessment



rating was based on this study. In these two studies, with follow-up of nine years, over 90% of the original sample was retained in the analyses.

Findings

O'Donnell and colleagues (1995; NRCT +) found that at the end of sixth grade there were no differences between intervention and control students on a measure of lifetime alcohol use in a subsample of low income participants (Males: mean difference 0.01 95% CI -0.25, 0.27; Females: mean difference -0.21 95% CI -0.45, 0.03).

At the 6-year follow-up when students were aged 18, Hawkins and colleagues (1999; NRCT +) reported that significant differences were found between control and full SSDP intervention students on alcohol use measures. Although there was no differences in lifetime alcohol use, fewer full intervention students than control students reported having drunk alcohol 10 or more times in the past year (RR 0.61 95% CI 0.39, 0.95; p<0.05). Full intervention students were also significantly less likely than control students to have engaged in sexual intercourse (p<0.05) and were less likely to have had multiple partners by the age of 18 (p<0.05). More control students had been pregnant or gotten someone pregnant, although this finding only approached statistical significance (p=0.06), and there was no difference between the full intervention and control in the number of participants that had fathered or had had a baby. At age 18, students who had received the full SSDP intervention reported significantly stronger commitment (p<0.01) and attachment to school (p<0.05) compared to control students. The authors did not find any significant effects of the late intervention programme. Effect sizes are presented in Table 5.38,

Outcome	Comparison	Prevalence difference (95% CI)
Lifetime alashal usa	Full intervention vs. control	-1.00 (-10.45, 8.45)
	Late intervention vs. control	-0.80 (-9.05, 7.45)
Lifetime equally active	Full intervention vs. control	-10.90 (-19.80, -2.00)
	Late intervention vs. control	-6.90 (-14.41, 0.61)
Lifetime multiple cox portpore	Full intervention vs. control	-11.80 (-22.31, -1.29)
Lifeume multiple sex partiers	Late intervention vs. control	-2.40 (-11.55, 6.75)
Lifetime been pregnant or gotten a woman	Full intervention vs. control	-9.30 (-17.94, -0.66)
pregnant	Late intervention vs. control	1.00 (-7.35, 9.35)
Lifetime had as fathered a haby	Full intervention vs. control	-5.20 (-12.00, 1.60)
	Late intervention vs. control	-0.40 (-7.01, 6.21)

	Table 5.38. SSDP:	Intervention	effects at age	18 (Hawkins	et al.,	1999; NRCT	+)
--	-------------------	--------------	----------------	-------------	---------	------------	----

At aged 21 (Hawkins et al., 2005; NRCT +), there were no significant differences between the full and late intervention and control groups for past month alcohol use. Significant outcomes reported for the full-intervention group in comparison to the control groups



included better regulation of emotions (p<0.01), fewer symptoms of social phobias (p<0.05), fewer suicidal thoughts (p<0.01), lower likelihood of being involved in a wide variety of crime in the past year or having a lifetime court record (p<0.05) and a higher likelihood of graduating high school (p<0.01). The vast majority of outcomes for the late-intervention group were non-significant compared to the control group. Effect sizes were calculated and are presented in Table 5.39.

Outcome	Comparison	Standardised mean difference (95% Cl)
	Full intervention vs. control	0.10 (0.02, 0.18)
High school graduate	Late intervention vs. control	0.04 (-0.04, 0.12)
Poor omotional regulation	Full intervention vs. control	-0.15 (-0.25, -0.05)
	Late intervention vs. control	-0.03 (-0.11, 0.05)
	Full intervention vs. control	-0.35 (-0.80, 0.10)
Anxiety symptom count	Late intervention vs. control	-0.06 (-0.45, 0.33)
Social phabia symptom sount	Full intervention vs. control	-0.30 (-0.54, -0.06)
Social phobla symptom count	Late intervention vs. control	-0.06 (-0.28, 0.16)
Dennesius summtem count	Full intervention vs. control	-0.63 (-1.34, 0.08)
Depressive symptom count	Late intervention vs. control	-0.35 (-0.96, 0.26)
Quiside the use to	Full intervention vs. control	-0.30 (-0.48, -0.12)
	Late intervention vs. control	-0.25 (-0.41, -0.09)
	Full intervention vs. control	-0.02 (-0.08, 0.04)
Anxiety diagnostic criteria met	Late intervention vs. control	-0.01 (-0.05, 0.03)
	Full intervention vs. control	-0.06 (-0.14, 0.02)
Social phobla diagnostic criteria met	Late intervention vs. control	-0.01 (-0.09, 0.07)
	Full intervention vs. control	-0.08 (-0.18, 0.02)
Depressive diagnostic criteria met	Late intervention vs. control	-0.08 (-0.16, -0.00)
	Full intervention vs. control	-0.08 (-0.16, -0.00)
Any substance use in past month	Late intervention vs. control	-0.06 (-0.14, 0.02)

Table 5.39. SSDP:	Intervention	effects at	age 21	(Hawkins e	et al.,	2005; NRC	;T +))
-------------------	--------------	------------	--------	------------	---------	-----------	-------	---

Lonczak and colleagues (2002; NRCT +) also examined the effects of the SSDP at age 21 years. Compared to control participants, participants in the full intervention group had, on average, their first sexual experience significantly later (p<0.05) and significantly fewer lifetime sexual partners (p<0.05). Participants in the full intervention group were significantly more likely to report condom use during last intercourse than those in the control group (p<0.05). There were no effects of the intervention on past-year condom use frequency among single participants, condom use during first intercourse or STD diagnosis. Females in the full intervention group were significantly less likely to have become pregnant (p<0.05) or to have had a baby (p<0.05) by age 21 years than females in the control group, but there were no significant differences between the number of intervention and control males who



reported causing a pregnancy or birth. Effect sizes were calculated for these outcomes are presented in Table 5.40 and Table 5.41.

Table 5.40.	SSDP: Intervention	effects at	age 21 -	dichotomous	(Lonczak et	: al., 20	02;
NRCT +)							

Outcome	Interve	ention	Cor	trol		
Outcome	Events	Total	Events	Total	OR (95% CI)	
Condom use during first intercourse	96	131	127	192	1.42 (0.87, 2.3)	
Condom use during last intercourse	53	89	68	154	1.88 (1.11, 3.19)	
Lifetime STD	19	144	37	205	0.67 (0.38, 1.27)	
Lifetime pregnancy	27	71	55	99	0.50 (0.27, 0.93)	
Lifetime birth	16	71	40	99	0.42 (0.21, 0.84)	
Causing pregnancy	25	73	38	106	0.95 (0.51, 1.78)	
Fathering a child	17	73	21	106	1.22 (0.59, 2.53)	

Table 5.41. SSDP:	Intervention	effects	at age	21 -	continuous	(Lonczak	et al.,	2002;
NRCT +)								

Outcome	Intervention				Control	Standardised		
Outcome	Mean	SD	Total	Mean	SD Tota		(95% CI)	
Age at first sexual experience	16.32	2.34	131	15.75	2.35	188	0.24 (0.02, 0.47)	
Frequency of condom use	3.28	1.37	81	3.12	1.45	142	0.11 (-0.16, 0.39)	
No of lifetime sexual partners	3.58	2.2	144	4.13	2.05	205	-0.26 (-0.47, -0.05)	

5.6.3.2 Raising Healthy Children

Two articles (Brown et al., 2005; Catalano et al., 2003) evaluated the effectiveness of the Raising Healthy Children (RHC) programme, a multicomponent programme aimed at reducing adolescent alcohol, cannabis and cigarette use. The programme consisted of: (1) school intervention strategies (including a series of teacher and staff development workshops teaching proactive classroom management techniques; cooperative learning methods and strategies to promote student motivation, participation, reading, and interpersonal problem solving skills); (2) student intervention strategies delivered during grades 4-6, with booster sessions throughout middle and high school years; (3) family intervention strategies (including parenting workshops and in home services, delivered in grades 1-8). Data were reported for students who participated in the programme as first and second grade students and were followed in the first and second grades (Catalano et al., 2003) and from sixth through to tenth grade (Brown et al., 2005).



Quality assessment

Whilst the RCT of the RHC programme (Brown et al., 2005; Catalano et al., 2003) generally appeared to have been well conducted, full details of the methods of randomisation and of the source population were not reported. As a result of these limitations to the study the RCT was rated '+'.

Findings

There was no significant difference between students who received the RHC programme and control students in terms of change in alcohol use over 5 years (Brown et al., 2005). However, there was a significant intervention effect on alcohol use frequency. There was a significantly greater rate of linear decline in alcohol frequency in the intervention group during Grades 8-10 relative to the control group (adjusted mean frequency ES = 0.40; p< 0.05).

Catalano and colleagues (2003) reported the effects of the intervention on teacher and parent rated academic performance and social behaviour. In comparison to the control group, teachers rated intervention students as having significantly higher academic performance and commitment to school (p<0.05) and significantly higher social competency (p<0.01) with increasing growth rate (p<0.01) and significantly lower levels of anti social behaviour (p<0.05) with decreasing growth rate. When controlled for gender, income and baseline scores, parent reported data indicated intervention students had higher academic performance and school commitment (p<0.05) than control students.

5.6.3.3 Linking the Interests of Families and Teachers (LIFT)

Two articles (Eddy et al., 2003; Reid et al., 1999) evaluated the effectiveness of the Linking the Interests of Families and Teachers (LIFT) prevention programme. The aim of the programme was to tackle conduct problems including the use of alcohol and other substances. The programme consisted of a 10-week classroom-based programme delivered to first and fifth grade students, a playground behaviour intervention, a behaviour management programme for parents and weekly newsletters, and ongoing access to a classroom-based telephone answering machine. Reid and colleagues (1999) evaluated the study following its completion and Eddy and colleages (2003) evaluated the programme's effects on fifth graders four years on.

Quality assessment

The study was not well reported (RCT -). Few details were reported regarding the method of randomisation. In addition, intervention and control participants were not matched at



baseline. Intervention students were significantly younger and less likely to be from an ethnic minority.

Findings

Eddy and colleagues (2003; RCT -) reported that significant differences were found in hazard rates between the LIFT intervention schools and control schools. Self-reports of patterned alcohol use during middle school (alcohol use at least once every 2 or 3 months) indicated that youth in the control group were 1.49 times more likely to report patterned alcohol use during middle school than youth in the intervention group. Reid and colleagues (1999), reporting on immediate outcomes from the intervention, found that teacher rated social skills were significantly higher for children who received the intervention group than in the control group.

5.6.3.4 Developmental drug prevention programme

Two studies (Furr-Holden et al., 2004, lalongo et al., 1999) examined the effectiveness of a programme that incorporated two developmental drug prevention programmes, which targeted problem behaviours in primary school aged children. The two programmes examined were a classroom-centred intervention that combined a classroom-based curriculum with teacher training and a family-school partnership intervention that emphasised parent-school communication and partnership building through workshops and communication activities. Ialongo and colleagues (1999) followed up students at one year and Furr-Holden and colleagues (2004) followed up students five, six and seven years after the intervention between the ages of 11 and 14 years.

Quality assessment

First grade classrooms in nine schools were randomly assigned using computer generated methods to the intervention or control group. Other study methodology aspects were clearly reported such as the level of attrition, which was relatively low over the long follow-up duration of the study, and details of the control group. The study by lalongo and colleagues (1999) was rated 'RCT ++' for internal consistency. However in the study by Furr-Holden and colleagues (2004), it was clear whether the intervention and control groups were comparable at baseline and the study was coded 'RCT +'.

Findings

At one year follow up, it was reported that those who received the family-school partnership intervention demonstrated significantly fewer problem behaviours than the control group and those who received the classroom curriculum were rated as displaying fewer problem behaviours than boys in the control group (lalongo et al., 1999; RCT ++). Improvements



were reported for boys only for reading in the classroom curriculum and family-school partnership groups and for maths in the classroom curriculum group.

Furr-Holden and colleagues (2004; RCT +) reported that over three years of assessment in grades 6-8 (age 11-14 years), the percentage of students reporting unsupervised alcohol use did not differ significantly across the intervention and control groups as shown in Table 5.42, and was lowest in the control group (29% of control participants vs. 34% and 37% of classroom-centred intervention participants and family-school partnership participants, respectively).

Table 5.42.	Developmental	drug	prevention	programme:	Unsupervised	alcohol	use
(Furr-Holde	n et al., 2004)						

Comparison	Intervention		Cor	ntrol	Pick ratio (05% CI)*			
Companson	Events	Total	Event	Total	KISK 1810 (95 % CI)			
Classroom-centred intervention vs. control	65	192	52	178	0.95 (0.58, 1.54)			
Family-school partnership vs. control	73	196	52	178	1.07 (0.67, 1.71)			
*Adjusted for age, sex, family type, teacher-rated total problems, parent management, and family history of drug, alcohol, or tobacco use								

5.6.3.5 Positive Action Program

One study reported on the long-term effectiveness of the Positive Action Program (Flay et al., 2003) that aimed to reduce problem behaviours and enhance school performance through a detailed curriculum consisting of over 140 lessons per grade through kindergarten to sixth grade, a school-climate programme and family involvement components. The study retrospectively examined the effects of the programme through the differences in middle and high schools classes by numbers of students who had or had not received the Positive Action Program in previous years.

Quality assessment

This case-control study (CSS) was rated 'CSS +' for quality. The study was reasonably well reported with follow up at four years. The overall sample was not matched on baselines, but data was provided for a matched sample. Little description was provided of the source population and selection.

Findings

Flay and colleagues (2003; CCS +) reported significantly fewer problem behaviours in middle and high schools with higher numbers of Positive Action graduates. Schools with over 80% Positive Action graduates were significantly less likely to misuse substances in middle schools (p<0.01) and high schools (p<0.05). There appeared to be educational benefits of the programme with students at elementary schools who received Positive Action scoring 45% better on the Florida Reading Test and 4.5% better on the Florida
Comprehensive Aptitude Test. Middle schools with higher number of Positive Action graduates scored better for reading and maths.

5.6.3.6 Child Development Project

Two studies reported on the effects of the Child Development Project (Battistich et al., 2000; 2004); a programme that aimed to reduce drug use and other problem behaviours through a three year curriculum combined with school-wide activities involving families and activities at home involving both student and parents. The programme was carried out during the final three years of elementary school, and students were followed up post-intervention (Battistich et al., 2000) and during middle school (Battistich et al., 2004).

Quality assessment

The study adequately described allocation of schools, the intervention and details around outcomes from the programme. However, the programme was not implemented equally in schools and the study involved a different cohort of students in each year. Details around participant numbers and attrition rate were not adequately covered. Overall the study was rated 'NRCT +' for quality.

Findings

Following intervention (Battistich et al., 2000; NRCT +), alcohol use among programme students declined from baseline, whereas comparison students showed a small, but non-significant increase (mean difference 0.15; p<0.10). Intervention students' alcohol use in 'high change' schools declined over time, whereas control students increased their use (mean difference 0.18; p<0.05). In a follow-up study, Battistich and colleagues (2004; NRCT +) reported that intervention and comparison students did not differ with respect to alcohol use in middle school. Intervention students scored significantly higher than comparison students in self-efficacy (p<0.01) but not in terms of global self-esteem. Intervention students scored higher than comparison students for positive teacher relations (p<0.05) and liking for school (p<0.05). For sense of school community and task orientation towards learning, differences approached significance, but no differences were found for academic performance.

5.6.4 Summary and evidence statements

In total, eight programmes that examined primary school-based social development interventions with outcomes relating to substance use or sex and relationships were identified. Interventions focused on the students' behaviour, social skills and values and outcomes for six studies included alcohol and sexual behaviour in adolescence or young adulthood.



5.6.4.1 Knowledge and understanding

None of the included studies reported this outcome.

5.6.4.2 Attitudes and values

Four programmes reported on outcomes relating to attitudes and values. Hawkins and colleagues (2005; NRCT +) reported a positive relationship between students who received the full SSDP intervention and high school graduation, and at follow-up aged 18, intervention students were more committed and attached to their school (Hawkins et al., 1999; NRCT +). Participation in the Raising Healthy Children programme was positively associated with teacher-rated academic performance and school commitment (Catalano et al., 2003; RCT +) while Battistich and colleagues (2004; NRCT +) reported that the Child Development Program effectively improved teacher relations and liking for school in comparison to controls, but found no effect on academic performance. The Positive Action Program was associated with positively impacting upon reading and maths performance at middle school (Flay et al., 2003; CCS +).

5.6.4.3 Personal and social skills

Of the seven programmes detailed here, six included outcomes relating to personal and social skills. In comparison to control students, at the age of 21 full-intervention group participants in the SSDP reported better regulation of emotions and fewer symptoms of social phobias (Hawkins et al., 2005; NRCT +). Teacher-rated levels of anti-social behaviour were found to be lower in students who had received the Raising Healthy Children intervention (Catalano et al., 2003; RCT +) and teacher-rated social skills were deemed to be higher in students who had been exposed to LIFT (Reid et al., 1999; RCT -) when compared to control groups. Ialongo and colleagues (1999; RCT +) found that one year following the Developmental Drug Prevention Programmes those who received the family-school partnership intervention or classroom curriculum demonstrated fewer problem behaviours than controls. Fewer problem behaviours were also reported by Flay and colleagues (2003; CSS +) in their evaluation of the Positive Action Program in middle and high schools with higher numbers of Positive Action graduates. Intervention students in the Child Development Project scored higher than controls in sense of efficacy and global self-esteem and had more positive teacher relations (Battistich et al., 2004; NRCT +).

5.6.4.4 Health and social outcomes related to alcohol and sex and relationships

Evaluation of six programmes included results relating to alcohol use and one of these six also included outcomes in relation to sexual health behaviour.

Following participation in the SSDP, at age 18 full-intervention students reported drinking on fewer occasions than control students (Hawkins et al., 1999; NRCT +) but at age 21 there

was no effect of the intervention on past month alcohol use (Hawkins et al., 2005; NRCT +). Two studies of the SSDP examined sexual health outcomes. At age 18 (Hawkins et al., 1999; NRCT +), the full intervention had positive impacts on sexual health outcomes including number of sexual partners and pregnancy. Lonczak and colleagues (2002; NRCT +) reported that at age 21, the same intervention impacted positively on age of first sexual experience, number of sexual partners, pregnancy, and condom use during last intercourse. A Dutch study of the Good Behaviour Game programme found that it positively impacted upon the rate of growth of alcohol use between 3 and 6 year follow up, but not on past year alcohol use at either time (van Lier et al., 2009; RCT ++). However, Kellam and colleagues (2008: RCT ++) reported that the Good Behaviour Game reduced lifetime alcohol use and dependence in intervention students in young adulthood. LIFT positively impacted upon frequency of alcohol use; participants in the control group were more likely to use alcohol once every 2 or 3 months during middle school than the intervention group (Eddy et al., 2003; RCT -). However, Furr-Holden and colleagues (2004; RCT +) found that in grades 6-8, the percentage of students reporting unsupervised alcohol use was not affected by exposure to the developmental drug prevention programmes. Raising Healthy Children did not reduce prevalence of alcohol use (Brown et al., 2005; RCT +), but did reduce frequency of use. Flay and colleagues (2003; CCS +) reported that participation in the Positive Action Program was associated with less drug use in middle school and high school, but did not specify any programme effects on alcohol that were independent of other drug use.



Evidence statement 6

- 6(a) There is moderate evidence from one RCT, three NRCTs and one CSS study¹ to suggest that programmes, which target social development and combine school and family-based components, may positively impact on attachment to school and academic performance. This evidence may only be partially applicable to the UK because these programmes were developed and evaluated in the USA, and the findings may not be generalisible beyond the populations studied.
- 6(b) There is moderate evidence from three RCTs, one NRCT and one CSS study² to suggest that programmes, which target social development and combine school and family-based components, may have a positive impact on problem behaviours and social skills. This evidence may only be partially applicable to the UK because these programmes were developed and evaluated in the USA, and the findings may not be generalisible beyond the populations studied.
- 6(c) There is moderate evidence from three NRCTs³ to suggest that a social development programme, which combined school and family-based components, may have long term impacts on alcohol use and sexual behaviour in young adulthood. This evidence may only be partially applicable to the UK because these programmes were developed and evaluated in the USA, and the findings may not be generalisible beyond the populations studied.
- 6(d) There is strong evidence from three RCTs⁴ to suggest that the Good Behavior Game, which targeted behaviours in the classroom, may impact on alcohol abuse and dependence in adulthood and slow the rate of alcohol use in adolescence. This evidence may be directly applicable to the UK because although the programme was developed and evaluated in the USA, it has been replicated in populations outside of the USA.
- ¹ Hawkins et al., 1999, 2005 (both NRCT +); Catalano et al., 2003 (RCT +); Battistich et al., 2004 (NRCT +); Flay et al., 2003 (CSS+);
- ² Catalano et al., 2003 (RCT +); Reid et al., 1999 (RCT -); Ialongo et al., 1999 (RCT +); Flay et al., 2003 (CSS+); Battistich et al., 2004 (NRCT +);
- ³ Hawkins et al., 1999, 2005; Lonczak et al., 2002 (all NRCT +)
- ⁴ Kellam et al., 2008; Poduska et al., 2008; van Lier et al., 2009 (all RCT ++)



5.43. Social development programmes

Author (Year)	Design	Population	Follow-up	Analysed	Findings				
Seattle Social Developme	ent Programme	·							
O'Donnell et al., 1995	NRCT +	USA 1 st grade N = 177	PT (6 years)	60%	No significant differences between intervention and control students.				
Hawkins et al., 1999	NRCT +	USA 5 th grade N = 643	6 years	93%	Fewer students receiving the full intervention reported heavy drinking (compared to control students). No difference in lifetime alcohol use between groups. Students in the full intervention condition reported significantly stronger commitment and attachment to school.				
Hawkins et al., 2005	NRCT +	USA $1^{st} - 6^{th}$ grade & $5^{th} - 6^{th}$ grade N = 643	9 years	94%	No significant differences between the full and late intervention and control groups for past month alcohol use were reported.				
Lonczak et al., 2002	NRCT +	1 st - 6 th grade N= 376	9 years	93%	Students in the intervention condition reported later age of first sexual experience, lower number of sexual partners, higher rates of condom use in last sexual encounter and lower pregnancy rates amongst females. No significant differences between groups for condom use during the past year or during first intercourse or rate of STD diagnosis.				
Raising Healthy Children									
Brown et al., 2005	RCT +	USA Mean 7.7 (SD 0.6) years N= 1,040	6, 7, 8 years	88%	No significant difference between groups in terms of change in alcohol use over 5 years. However, significant intervention effect on alcohol use frequency (greater rate of linear decline in alcohol frequency in the intervention group)				
Catalano et al., 2003	RCT +	USA Mean 7.4 (SD 0.6) years N = 938	1, 2 years	98%	Teachers rated intervention students as having higher academic performance, commitment to school and social competency and lower levels of anti social behaviour. Parent reported data indicated intervention students had higher academic performance and school commitment.				
Developmental drug prevention programmes									
Furr Holden et al., 2004	RCT +	USA Mean 6.2 (SD 0.3) years N = 653	4, 5 and 6 years	84%	Little impact of either intervention on alcohol use				
lalongo et al., 1999	RCT +	USA Mean 6.2 (SD 0.3) years N = 653	1 year	91%	Students receiving the family-school partnership intervention demonstrated fewer problem behaviours than the control group. Those receiving the classroom curriculum displayed fewer problem behaviours than boys in the control group.				

Author (Year)	Design	Population	Follow-up	Analysed	Findings				
Good Behaviour Game									
Kellam et al., 2008; Poduska et al., 2008	RCT ++	USA 1 st grade N = 922	At age 19-21 years	75%	Intervention students had lower lifetime alcohol use/ dependence disorder than controls and a significant reduction in the odds of a lifetime alcohol abuse disorder. No significant differences between intervention and control groups for access to drug treatment services. Rates of all service use were lower for males against controls in both cohorts.				
van Lier et al., 2009	RCT ++	The Netherlands Mean 6.9 (SD 0.6) years N = 666	3 - 6 years	72%	No significant effect of participation in the Good Behavior Game on past month or past year alcohol use at three or six year follow up. The rate of growth of alcohol use between three and six years following participation in the Game was significantly reduced				
Child Development Project									
Battistich et al., 2000	NRCT +	USA Grades 3-5 or 4-6 N = 24 schools	4 years	Not reported	Alcohol use among programme students declined, whilst alcohol use in comparison students increased. Intervention students' alcohol use in "high change" schools declined over time, whereas control students' alcohol use increased.				
Battistich et al., 2004	NRCT +	USA Grades 3-5 or 4-6 N = 1246	5-7 years	Not reported	Intervention and comparison students did not differ significantly with respect to their use of alcohol in middle school				
Linking the Interests of Families and Teachers (LIFT)									
Eddy et al., 2003	RCT -	USA Mean 10.4 years N = 361	4 years	2.8% dropped out	Youth in the control group more likely to report patterned alcohol use during middle school.				
Reid et al., 1999	RCT -	USA 1 st and 5 th grades N = 671	1 year (PT)	Not reported	Teacher rated social skills of children in the intervention group were significantly higher than in the control group. Mothers of children in the intervention group with higher levels of aversive verbal behaviour improved the most.				
Other programmes									
Flay et al., 2003 Positive Action Program	CCS +	USA K-6 grade N = 36 schools	4 years	Not applicable	Schools with higher numbers of intervention students were reported to have significantly fewer problem behaviours in middle and high schools. Schools with over 80% Positive Action graduates were significantly less				
5	1		1	1	l likely to use drugs in middle and high schools.				

6 Discussion

6.1 Summary of the review of effectiveness

The review of effectiveness included a total of two systematic reviews and meta-analyses, and 73 primary studies. Studies were grouped according to the intervention focus, resulting in 14 studies of alcohol education programmes, 33 studies of drug (including alcohol) education programmes, nine SRE programmes, three general health education programmes and 16 social development programmes.

6.1.1 Systematic reviews and meta-analyses

Two systematic reviews were identified for inclusion. The review by Spoth and colleagues (2008) focused specifically on the prevention of alcohol use, whereas Gottfredson and Wilson (2008) focused on substance use prevention. Both reviews concluded that there was limited evidence to determine which programme approaches were most effective for primary school aged children, and Gottfredson and Wilson (2008) suggested that intervention with young adolescents may be more effective. Spoth and colleagues (2008) suggested that intervention with younger children may be most effective when it takes place across multiple domains, most typically combining school and family-based intervention.

6.1.2 Alcohol education

A total of 13 studies were identified that examined alcohol education programmes targeting children aged 11 years and under. Nine studies were classroom-based curriculums led by teachers or external contributors, and four studies examined one-off intervention sessions. Of the classroom-based programmes, the PY/PM programme that focused on teaching students about the adverse effects of alcohol on the brain and vehicle safety skills, was shown to have significant effects on knowledge about the effects of alcohol on development and the brain, vehicle safety and media literacy. This effect was demonstrated with third, fourth and fifth grade students (Bohman et al., 2004; Bell et al., 2005a, 2005b) and students in first and second grade (Bell et al., 2007). The long term effects on alcohol consumption were examined for three programmes (AMPS, APPT and PY/PM) but none showed consistent effects. Of the single session interventions, there was evidence that they modified their intended target in the short term (i.e. the expectancy modification intervention modified expectancies) but these effects appeared to be short lived.

6.1.3 Drug education (including alcohol)

A total of 32 studies were identified that examined drug education programmes, which focused on illegal drugs (and tobacco) in addition to alcohol. A total of 20 studies reported on



18 classroom-based programmes, led by teachers (n=11 studies) or external contributors (n=9 studies). Four studies reported on programmes which combined in-school approaches with parent education and eight studies reported on a range of other in-school approaches including theatre in education and a programme based on a retreat format. The effects of these programmes on knowledge and attitudes in relation to alcohol use were not clear. Although LST has been shown to be effective with young adolescents, only very short term outcomes were presented for the study conducted with primary school aged children, and although the programme appeared to have positive effects on normative expectations and self-esteem, these findings were not supported by other studies of the LST approach in this age group. Only two programmes demonstrated effects on alcohol use, a curriculum for Native American students and a normative education programme, DAW. However, the evaluation of the DAW programme was rated poorly and therefore the results of this programme should be interpreted with caution.

6.1.4 Sex and relationships education

Eight studies were identified that examined seven programmes focusing on different approaches to SRE. Two programmes focused on abstinence approaches; three programmes were HIV/AIDS prevention approaches; one programme employed a parenting and care-giving approach; and another aimed to improve young people's sexual health knowledge, personal insight and motivation. All seven programmes were curriculum based and delivered in schools. Programmes based on an abstinence approach or HIV/AIDS prevention appeared to improve knowledge and there were positive effects of two programmes, which specifically targeted communication (FAME and I Want to, I Can...Prevent HIV/AIDS), on parental communication. One abstinence-based programme, Sex Can Wait, was shown to have some long term effects on sexual behaviour, although there was no difference in whether participants had ever had sex, participants who received the intervention were less likely than control participants to report sexual intercourse in the last 30 days, 18 months after intervention.

6.1.5 General health education programmes

Three studies (Andrews, 1992; Utley et al., 2001; Young et al., 1997) were identified that examined general health education programmes which included modules or topics related to alcohol education or SRE. There was no clear evidence of the effectiveness of these programmes in terms of alcohol use or sexual health or related behaviours. Although, one study that examined the impact of the life skills components of a larger health curriculum found post-test effects on attitudes to alcohol, self esteem and decision making skills.



6.1.6 Social development programmes

A total of 16 studies were identified that examined seven programmes focused on social development interventions designed to positively influence later behaviour. Six programmes combined school and family-based components while one programme was school-based only. Programmes that combined classroom-based intervention with components targeting parental participation, and focusing on wider problem behaviours had long term effects on attachment to school, social skills and alcohol use, and the SSDP also had long term effects on sexual health behaviours.

6.2 Summary of review of economic evaluations

No studies of economic evaluations were identified for inclusion in the review.

6.3 Strengths and limitations

This review of the effectiveness and cost-effectiveness of PSHE in primary schools focusing on SRE and alcohol education was based on a comprehensive and systematic literature review. Over 7,500 titles and abstracts were screened for inclusion in the review, and 501 full text articles were reviewed. In addition, the review has been conducted using a standardised and transparent approach, adhering to NICE protocols for the development of NICE public health guidance.

6.3.1 Quality of the included studies

The studies identified for inclusion in the review were based on a range of study designs, with the majority of studies based on an RCT cluster design. Overall only five studies, two SRs and three RCTs, were rated high quality (++). Of the remaining studies, approximately a third were rated moderate quality and two-thirds were rated poor quality. Across the majority of studies, whether participants and/or investigators were blind to group assignment, and whether allocation was concealed was poorly reported on; with most studies not reporting on these aspects. In addition, it was difficult to state with confidence that contamination was acceptability low. In terms of analyses, very few studies reported than an intention-to-treat (ITT) analysis had been conducted, few studies were reported to be sufficiently powered or presented power calculations, and effect size estimates were rarely reported. In addition, not all studies provided sufficient data to calculate effect sizes, with a lack of detail regarding numbers allocated to intervention and control groups. In relation to the external validity of the included studies, generalisiblity was on the whole difficult to judge. Very few studies reported details about the source population or whether the selection of participants resulted in a representative sample.

Approximately 60% of studies only reported the short-term effects (<6 months) of the interventions and programmes examined, with a large proportion of these studies only



reporting immediate post-test effects. There was, therefore, limited evidence on the long term effects of the programmes included in the review on alcohol use and sexual behaviour. The exception to this was the studies that examined social development programmes, the majority of which examined the long term effects of these programmes into young adulthood. Because of the short length of follow-up, outcomes reported tended to be limited to knowledge, attitudes and skills. The choice of outcomes measured across studies varied greatly, and there was little consistency in how outcomes were assessed. This meant that it was not possible to combine studies in a meta-analysis.

6.3.2 Applicability

Six studies (Hurry & McGurk, 1997; Hurry et al., 2000; Paxton et al., 1998; Starkey & Orme, 2001; Tudor-Smith et al., 1995; Welham et al., 2007) reported on evaluations conducted in the UK, all of which evaluated drug education programmes. Four studies were based on UBA designs with short term (<6 months) follow-up and two studies were of an RCT which followed up participants three years later. As with previous reviews in the field of alcohol and drug prevention (Jones et al., 2006, 2007) the vast majority of studies were conducted in the USA. This therefore limits the applicability of a substantial proportion of the evidence identified. As reported previously, the generalisibility of the included studies was difficult to judge and this further limits the applicability of the evidence identified.

6.3.3 How and why programmes worked

Due to the short timescales available, it was beyond the scope of the review to undertake a full examination of how and why the programmes that demonstrated effectiveness worked. In addition, no studies were identified that sought to address this question or that reported mediation or component analysis. It was also not possible to examine how different programme providers or the way in which an intervention was delivered, impacted on programme effectiveness.

6.3.4 Targeting wider risk behaviours

The studies included in this review that examined social development programmes demonstrate that programmes targeting developmental outcomes, such as conduct disorders and social skills, may have long term impacts on alcohol use and sexual health (Hawkins et al., 1999; Lonczak et al., 2002). It was beyond the scope of the review to examine programmes that did not focus on SRE and/or alcohol education. However, the literature searches identified a large body of literature which examined interventions that targeted key risk factors that may predict later alcohol use, including programmes targeting conduct problems and social skills. Two reviews (Adi et al., 2007a, 2007b) undertaken to support the development of NICE guidance on promoting the mental wellbeing of children in



primary education highlighted that there was strong evidence to support the effectiveness of multi-component programmes, which typically combined social skills development curricula and programmes for parents.

6.4 Research recommendations

This review had identified a number of gaps in the evidence in relation to the effectiveness of PSHE in primary schools focusing on SRE and alcohol education. In particular, a lack of research arising from the UK limits the conclusions that can be drawn from the studies identified for inclusion in this review. The following are listed as key research recommendations:

- There needs to be further evaluation of the effectiveness and cost-effectiveness of PSHE approaches in primary school focusing on alcohol education and SRE, which are currently being delivered or planned in the UK. In addition, full economic evaluations are required that consider both the costs and consequences of implementing these types of interventions and programmes.
- Improvements in study design and quality of reporting are needed with respect to all types of study designs. For RCTs, improvements are required with respect to the methods used to randomise participants or clusters and for quasi-experimental study designs, authors should ensure that methods used to allocate intervention and control participants or clusters are clearly reported.



7 Conclusions

Overall, this review of the effectiveness and cost-effectiveness of PSHE in primary schools focusing on SRE and alcohol education has highlighted a number of weaknesses in the evidence base. There is evidence that social development programmes, which combine school- and family-based components, may have long term impacts on school attachment, social skills, alcohol use and sexual health. However, the applicability of these programmes warrants further study in a UK context before widespread implementation can be supported. There is a lack of clear, long-term evidence for the effectiveness of other approaches to SRE and alcohol education, and further good quality, UK-based research is needed.

8 References

Adi Y, Kiloran A, Janmohamed K, Stewart-Brown S (2007a) Systematic review of the effectiveness of interventions to promote mental wellbeing in primary schools. Report 1: Universal approaches which do not focus on violence or bullying. Warwick, University of Warwick.

Adi Y, McMillan AS, Kiloran A, Stewart-Brown S (2007b) Systematic review of the effectiveness of interventions to promote mental wellbeing in primary schools. Report 3: Universal approaches with a focus on prevention of violence and bullying. Warwick, University of Warwick.

Bellis MA, Morleo M, Tocque K, Dedman D, Phillips-Howard PA, Perkins C, et al., (2009) Contributions of alcohol use to teenage pregnancy: an examination of geographical and evidence based associations. North West Public Health Observatory, Centre for Public Health, Liverpool John Moores University.

Bellis MA, Downing J, Ashton JA. (2006) Adults at 12? Trends in puberty and their public health consequences. Journal of Epidemiology and Community Health 60:910–911.

Christophers H, Mann S, Lowbury R. (2008) Progress and priorities - working together for high quality sexual health: Review of the national strategy for sexual health and HIV. London, Medical Foundation for AIDS & Sexual Health.

Clemens S, Jotangia D, Lynch S, Nicholson S, Pigott S. (2008) Drug use, smoking and drinking among young people in England in 2007. London, NHS Information Centre.

Costello EJ, Sung M, Worthman C, Angold A. (2007) Pubertal maturation and the development of alcohol use and abuse. Drug and Alcohol Dependence 88S:S50–S59.

CMO. (2009) Draft guidance on the consumption of alcohol by children and young people, from the Chief Medical Officers for England, Wales and Northern Ireland. London, Department of Health.

Department for Education and Employment. (2000) Sex and relationship education guidance. London, Department for Education and Employment.

Department for Education and Skills. (2004) Drugs: Guidance for schools. London, DfES.

Department for Education and Skills. (2006) Teenage pregnancy: accelerating the strategy to 2010. London, DfES.

Department of Health. (2009) Health profiles for England 2008. London, Department of Health.



Department of Health (2007) Safe. Sensible. Social. The next steps in the National Alcohol Strategy. London: Department of Health.

Department of Health (2004). Choosing Health: making healthy choices easier. London, Department of Health.

Goodson P, Evans A, Edmundson E. (1997) Female adolescents and onset of sexual intercourse: a theory-based review of research from 1984 to 1994. Journal of Adolescent Health. 21:147-156.

Health Protection Agency (2008). Selected STI diagnoses and diagnosis rates from GUM clinics in the UK: 2003-2007. London, Health Protection Agency [http://www.hpa.org.uk: Accessed 18th May 2009].

Home Office (2009). Safe. Sensible. Social. Selling alcohol responsibly: A consultation on the new code of practice for alcohol retailers. London, Home Office.

Independent Advisory Group on Sexual Health and HIV (2007). Sex, drugs, alcohol and young people. London, Department of Health.

Jones L, Sumnall H, Witty K, Wareing M, McVeigh J, Bellis M (2006) A review of communitybased interventions to reduce substance misuse among vulnerable and disadvantaged young people. Liverpool, National Collaborating Centre for Drug Prevention.

Jones L, James M, Jefferson T, Lushey C, Morleo M, Stokes E, Sumnall H, Bellis M (2007) A review of the effectiveness and cost-effectiveness of interventions delivered in primary and secondary schools to prevent and/or reduce alcohol use by young people under 18 years old. Liverpool, National Collaborating Centre for Drug Prevention.

Kirby D (2001) Emerging Answers: Research Findings on Programs to Reduce Teen Pregnancy. Washington, National Campaign to Prevent Teenage Pregnancy.

Lindsay G, Davies H, Band S, Cullen MA, Cullen S, Strand S, Hasluck C, Evans R, Stewart-Brown S (2008) Parenting early pathfinder evaluation. London, Department for Children, Schools and Families.

MacDonald A (2009) Independent review of the proposal to make Personal, Social, Health and Economic (PSHE) education statutory. London, Department for Children, Schools and Families.

Malone SM, Taylor J, Marmorstein NR, McGue M, Iacono WG (2004) Genetic and environmental influences on antisocial behavior and alcohol dependence from adolescence to early adulthood. Development and Psychopathology 16:943–966.

McGue M, Iacono WG (2008) The adolescent origins of substance use disorders. International Journal of Methods in Psychiatric Research 17(S1):S30–S38.



Odom SL, Horner RH, Snell ME, Blacher J (2007) Handbook of Developmental Disabilities. New York, Guildford Press

Office for National Statistics and Teenage Pregnancy Unit (2009) Under-18 conception statistice1998-2007. Office for National Statistics and Teenage Pregnancy Unit.

Qualifications and Curriculum Authority (2007) The National Curriculum 2007. London, Qualifications and Curriculum Authority.

Qualifications and Curriculum Authority (1999) Science: The national curriculum for England. London, Qualifications and Curriculum Authority.

Social Exclusion Unit (1999) Teenage Pregnancy. London, Stationary Office.

Spoth R, Greenberg M, Turrisi R (2008) Preventive interventions addressing underage drinking: State of the evidence and steps toward public health impact. Pediatrics 121:S311-S336.

Tripp J, Viner R (2005) Sexual health, contraception, and teenage pregnancy. BMJ 330:590-593.

Unicef (2001). A league table of teenage births in rich nations. Innocenti report card 3. Florence, Unicef Innocenti Research Centre.

Unicef (2007). Child poverty in perspective: An overview of child well-being in rich countries. Innocenti report card 7. Florence, Unicef Innocenti Research Centre.

Wellings K, Nanchahal K, Macdowall W, McManus S, Erens B, Mercer CH, Johnson AM, Copas AJ, Korovessis C, Fenton KA, Field J. (2001) Sexual behaviour in Britain: early heterosexual experience. Lancet 358:1843–50.

Van Der Vorst H, Vermulst AA, Meeus WHJ, Deković M, Engels RCME. (2009) Identification and Prediction of Drinking Trajectories in Early and Mid-Adolescence. Journal of Clinical Child and Adolescent Psychology 38(3):329–341.

Zimmer-Gembeck MJ, Collins WA (2008) Gender, mature appearance, alcohol use, and dating as correlates of sexual partner accumulation from ages 16–26 years. Journal of Adolescent Health 42:564–572.



Appendix 1. Non-statutory framework for PSHE

Key stage 1

- 1. Developing confidence and responsibility and making the most of their abilities
 - a) to recognise what they like and dislike, what is fair and unfair, and what is right and wrong
 - b) to share their opinions on things that matter to them and explain their views
 - c) to recognise, name and deal with their feelings in a positive way
 - d) to think about themselves, learn from their experiences and recognise what they are good at
 - e) how to set simple goals.

2. Preparing to play an active role at citizens

- a) to take part in discussions with one other person and the whole class
- b) to take part in a simple debate about topical issues
- c) to recognise choices they can make, and recognise the difference between right and wrong
- d) to agree and follow rules for their group and classroom, and understand how rules help them
- e) to realise that people and other living things have needs, and that they have responsibilities to meet them
- f) that they belong to various groups and communities, such as family and school
- g) what improves and harms their local, natural and built environments and about some of the ways people look after them
- h) to contribute to the life of the class and school
- i) to realise that money comes from different sources and can be used for different purposes.

3. Developing a healthy, safer lifestyle

- a) how to make simple choices that improve their health and wellbeing
- b) to maintain personal hygiene
- c) how some diseases spread and can be controlled
- d) about the process of growing from young to old and how people's needs change
- e) the names of the main parts of the body
- f) that all household products, including medicines, can be harmful if not used properly
- g) rules for, and ways of, keeping safe, including basic road safety, and about people who can help them to stay safe.
- 4. Developing good relationships and respecting differences between people
 - a) to recognise how their behaviour affects other people
 - b) to listen to other people, and play and work cooperatively



- c) to identify and respect the differences and similarities between people
- d) that family and friends should care for each other
- e) that there are different types of teasing and bullying, that bullying is wrong, and how to get help to deal with bullying.

Key stage 2

Developing confidence and responsibility and making the most of their abilities

- a) to talk and write about their opinions, and explain their views, on issues that affect themselves and society
- b) to recognise their worth as individuals by identifying positive things about themselves and their achievements, seeing their mistakes, making amends and setting personal goals
- c) to face new challenges positively by collecting information, looking for help, making responsible choices, and taking action
- d) to recognise, as they approach puberty, how people's emotions change at that time and how to deal with their feelings towards themselves, their family and others in a positive way
- e) about the range of jobs carried out by people they know, and to understand how they can develop skills to make their own contribution in the future
- f) to look after their money and realise that future wants and needs may be met through saving.

Preparing to play an active role as citizens

- a) to research, discuss and debate topical issues, problems and events
- b) why and how rules and laws are made and enforced, why different rules are needed in different situations and how to take part in making and changing rules
- c) to realise the consequences of anti-social and aggressive behaviours, such as bullying and racism, on individuals and communities
- d) that there are different kinds of responsibilities, rights and duties at home, at school and in the community, and that these can sometimes conflict with each other
- e) to reflect on spiritual, moral, social, and cultural issues, using imagination to understand other people's experiences
- f) to resolve differences by looking at alternatives, making decisions and explaining choices
- g) what democracy is, and about the basic institutions that support it locally and nationally
- h) to recognise the role of voluntary, community and pressure groups



- i) to appreciate the range of national, regional, religious and ethnic identities in the United Kingdom
- j) that resources can be allocated in different ways and that these economic choices affect individuals, communities and the sustainability of the environment
- k) to explore how the media present information.

Developing a healthy, safer lifestyle

- a) what makes a healthy lifestyle, including the benefits of exercise and healthy eating, what affects mental health, and how to make informed choices
- b) that bacteria and viruses can affect health and that following simple, safe routines can reduce their spread
- c) about how the body changes as they approach puberty
- d) which commonly available substances and drugs are legal and illegal, their effects and risks
- e) to recognise the different risks in different situations and then decide how to behave responsibly, including sensible road use, and judging what kind of physical contact is acceptable or unacceptable
- f) that pressure to behave in an unacceptable or risky way can come from a variety of sources, including people they know, and how to ask for help and use basic techniques for resisting pressure to do wrong
- g) school rules about health and safety, basic emergency aid procedures and where to get help.

Developing good relationships and respecting the differences between people

a) that their actions affect themselves and others, to care about other people's feelings and to try to see things from their points of view



Appendix 2. References to included studies

Abbey A, Oliansky D, Stilianos K, Hohlstein LA, Kaczynski R (1990) Substance abuse prevention for second graders: are they too young to benefit? Journal of Applied Developmental Psychology 11(2):149-162.

Abel EM, Greco M (2008) A preliminary evaluation of an abstinence-oriented empowerment program for public school youth. Research on Social Work Practice 18(3):223-231.

Allison KR, Silverman G, Dignam C (1990) Effects on students of teacher training in use of a drug education curriculum. Journal of Drug Education 20(1):31-46.

Ambtman R, Madak P, Koss D, Strople MJ (1990) Evaluation of a comprehensive elementary school curriculum-based drug education program. Journal of Drug Education 20(3):199-225.

Andrews RL (1992) The effects of school health curricula on knowledge, attitudes, and the onset of substance abuse from kindergarten to grade 12. Laramie, Wyoming University College of Education.

Austin EW, Johnson KK (1995) Direct and indirect effects of media literacy training on third graders' decision-making for alcohol. Paper presented at the Annual Meeting of the International Communication Association (May 25-29, Albuquerque).

Baker V (2004) Preventing alcohol and drug abuse through primary education: An evaluation of a school-based substance abuse prevention program. Dissertation Abstracts International: Section B: The Sciences and Engineering 64(7-B).

Battistich V, Schaps E, Watson M, Solomon D, Lewis C (2000) Effects of the Child Development Project on students' drug use and other problem behaviors. Journal of Primary Prevention 21(1):75-99.

Battistich V, Schaps E, Wilson N (2004) Effects of an elementary school intervention on students' "connectedness" to school and social adjustment during middle school. Journal of Primary Prevention 24(3):243-262.

Bell ML, Baker TK, Falb T (2005a) Protecting you/protecting me: evaluation of a student-led alcohol prevention and traffic safety program for elementary students. Journal of Alcohol & Drug Education 49(1):33-54.

Bell ML, Kelley-Baker T, Rider R, Ringwalt C (2005b) Protecting you/protecting me: effects of an alcohol prevention and vehicle safety program on elementary students. Journal of School Health 75(5):171-7.



Bell ML, Padget A, Kelley BT, Rider R (2007) Can first and second grade students benefit from an alcohol use prevention program? Journal of Child and Adolescent Substance Abuse 16:89-107.

Bohman TM, Barker ED, Bell ML, Lewis CM, Holleran L, Pomeroy E (2004) Early intervention for alcohol use prevention and vehicle safety skills: evaluating the Protecting You/Protecting Me curriculum. Journal of Child & Adolescent Substance Abuse 14(1):17-41.

Botvin GJ, Griffin KW, Paul E, Macaulay AP (2003) Preventing tobacco and alcohol use among elementary school students through life skills training. Journal of Child & Adolescent Substance Abuse 12(4):1-18.

Brown EC, Catalano RF, Fleming CB, Haggerty KP, Abbott RD (2005) Adolescent substance use outcomes in the Raising Healthy Children project: A two-part latent growth curve analysis. Journal of Consulting and Clinical Psychology 73(4):699-710.

Bühler A, Schroder E, Silbereisen RK (2008) The role of life skills promotion in substance abuse prevention: a mediation analysis. Health Education Research 23(4):621-32.

Catalano RF, Mazza JJ, Harachi TW, Abbott RD, Haggerty KP, Fleming CB (2003) Raising healthy children through enhancing social development in elementary school: Results after 1.5 years. Journal of School Psychology 41(2):143-164.

Corbin SK, Jones RT, Schulman RS (1993) Drug refusal behavior: the relative efficacy of skills-based and information-based treatment. Journal of Pediatric Psychology 18(6):769-84.

Cruz IY, Dunn ME (2003) Lowering risk for early alcohol use by challenging alcohol expectancies in elementary school children. Journal of Consulting & Clinical Psychology 71(3):493-503.

Denny G, Young M (2006) An evaluation of an abstinence-only sex education curriculum: an 18-month follow-up. Journal of School Health 76(8):414-22.

Denny G, Young M, Spear CE (1999) An evaluation of the Sex Can Wait abstinence education curriculum series. American Journal of Health Behavior 23(2):134-143.

Donaldson SI, Graham JW, Piccinin AM, Hansen WB (1995) Resistance-skills training and onset of alcohol use: evidence for beneficial and potentially harmful effects in public schools and in private catholic schools. Health Psychology 14(4):291-300.

Donaldson SI, Thomas CW, Graham JW, Au JG, Hansen WB (2000) Verifying drug abuse prevention program effects using reciprocal best friend reports. Journal of Behavioral Medicine 23(6):585-601.



Eddy JM, Reid JB, Stoolmiller M, Fetrow RA (2003) Outcomes during middle school for an elementary school-based preventive intervention for conduct problems: Follow-up results from a randomized trial. Behavior Therapy 34(4):535-552.

Flay BR, Allred CG (2003) Long-term effects of the Positive Action (R) program. American Journal of Health Behavior 27:S6-S21.

Furr-Holden CDM, Ialongo NS, Anthony JC, Petras H, Kellam SG (2004) Developmentally inspired drug prevention: middle school outcomes in a school-based randomized prevention trial. Drug & Alcohol Dependence 73(2):149-58.

Gamble RJ, Burgess S (1994) The D'Youville College campus school: alcohol collaboration or how to provide alcohol awareness at the elementary school level. Journal of Drug Education 24(4):363-8.

Gaskins SW, Beard SR, Wang MQ (2002) An HIV/AIDS education program for children in grades K-5. Journal of HIV/AIDS Prevention & Education for Adolescents & Children 5(1/2):31-44.

Godbold LC (1999) Conferring resistance to peer pressure among adolescents: Using the inoculation paradigm to discourage alcohol use. Dissertation Abstracts International Section A: Humanities and Social Sciences 59(7-A).

Gottfredson DC, Wilson DB (2003) Characteristics of effective school-based substance abuse prevention. Prevention Science 4(1):27-38.

Hahn EJ, Hall LA, Rayens MK, Myers AV, Bonnel G (2007) School- and home-based drug prevention: Environmental, parent, and child risk reduction. Drugs-Education Prevention and Policy 14(4):319-331.

Hall-Long BA, Dishop ML (1999) Never too soon: a pilot first and third grade drug education program. Journal of School Nursing 15(3):34-9.

Hawkins J, Catalano R, Kosterman R, Abbott R, Hill K (1999) Preventing Adolescent healthrisk behaviours by strengthening protection during childhood. Archives of Pediatric and Adolescent Medicine 153(3):226-34.

Hawkins JD, Kosterman R, Catalano RF, Hill KG, Abbott RD (2005) Promoting positive adult functioning through social development intervention in childhood: long-term effects from the Seattle Social Development Project. Archives of Pediatrics & Adolescent Medicine 159(1):25-31.



Hawthorne G (1996) The social impact of Life Education: estimating drug use prevalence among Victorian primary school students and the statewide effect of the Life Education programme. Addiction 91(8):1151-9.

Hawthorne G, Garrard J, Dunt D (1995) Does Life Education's drug education programme have a public health benefit?. Addiction 90(2):205-15.

Hecht ML, Elek E, Wagstaff DA, Kam JA, Marsiglia F, Dustman P, Reeves L, Harthun M (2008) Immediate and short-term effects of the 5th grade version of the keepin' it REAL substance use prevention intervention. Journal of Drug Education 38(3):225-251.

Holtz KD, Twombly EC (2007) A preliminary evaluation of the effects of a science education curriculum on changes in knowledge of drugs in youth. Journal of Drug Education 37(3):317-33.

Hurry J, Lloyd C, McGurk H (2000) Long-term effects of drugs education in primary school. Addiction Research 8(2):183-202.

Hurry J, McGurk H (1997) An evaluation of a primary prevention programme for schools. Addiction Research 5(1):23-38.

lalongo NS, Werthamer L, Kellam SG (1999) Proximal impact of two first-grade preventive interventions on the early risk behaviors for later substance abuse, depression, and antisocial behavior. American Journal of Community Psychology 27(5):599-641.

Kellam SG, Brown CH, Poduska JM, Ialongo NS, Wang W, Toyinbo P, Petras H, Ford C, Windham A, Wilcox HC (2008) Effects of a universal classroom behavior management program in first and second grades on young adult behavioral, psychiatric, and social outcomes. Drug and Alcohol Dependence 95:S5-S28.

Kraus D, Smith GT, Ratner HH (1994) Modifying alcohol-related expectancies in gradeschool children. Journal of Studies on Alcohol 55(5):535-542.

Kreutter KJ, Gewirtz H, Davenny JE, Love C (1991) Drug and alcohol prevention project for sixth graders: first-year findings. Adolescence 26(102):287-93.

Lonczak HS, Abbott RD, Hawkins JD, Kosterman R, Catalano RF (2002) Effects of the Seattle Social Development Project on sexual behavior, pregnancy, birth, and sexually transmitted disease outcomes by age 21 years. Archives of Pediatrics & Adolescent Medicine 156(5):438-47.

Masterpasqua F, Meyer S, Brooks K, Miller MJ (1992) Teaching Children about Parenting. Paper presented at the Annual Convention of the American Psychological Association (August 16, Washington, DC).



O'Donnell J, Hawkins JD, Catalano RF, Abbott RD, Day LE (1995) Preventing school failure, drug use, and delinquency among low-income children: long-term intervention in elementary schools. American Journal of Orthopsychiatry 65(1):87-100.

Padget A, Bell ML, Shamblen SR, Ringwalt CL (2006) Does learning about the effects of alcohol on the developing brain affect children's alcohol use? Prevention Science 7(3):293-302.

Paxton R, Finnigan S, Haddow M, Allott R, Leonard R (1998) Drug education in primary schools: putting what we know into practice. Health Education Journal 57(2):117-28.

Peterson AV, Woodward GD (1993) Quantitative analysis of the CHOICE drug education program for sixth grade students. Journal of Reality Therapy 13(1):40-45.

Pick S, Givaudan M, Sirkin J, Ortega I (2007) Communication as a protective factor: evaluation of a life skills HIV/AIDS prevention program for Mexican elementary-school students. AIDS Education & Prevention 19(5):408-21.

Poduska JM, Kellam SG, Wang W, Brown CH, Ialongo NS, Toyinbo P (2008) Impact of the Good Behavior Game, a universal classroom-based behavior intervention, on young adult service use for problems with emotions, behavior, or drugs or alcohol. Drug and Alcohol Dependence 95:S29-S44.

Raybuck CS, Hicks GF (1994) KIDS CARE: Building Resilience in Children's Environments. Journal of Alcohol and Drug Education 39(3):34-45.

Reid JB, Eddy JM, Fetrow RA, Stoolmiller M (1999) Description and immediate impacts of a preventive intervention for conduct problems. American Journal of Community Psychology 27(4):483-517.

Rollin SA, And O (1992) A school-based drug education health promotion research project. Presented at the Annual Meeting of the American Psychological Association (August 14-18, Washington, DC).

Rollin SA, Rubin R, Marcil R, Ferullo U, Buncher R (1995) Project KICK: a school-based drug education health promotion research project. Counselling Psychology Quarterly 8(4):345-359.

Schinke SP, Tepavac L (1995) Substance abuse prevention among elementary school students. Drugs and Society 8(3-4):15-27.

Schinke SP, Tepavac L, Cole KC (2000) Preventing substance use among Native American youth: three-year results. Addictive Behaviors 25(3):387-97.



Schonfeld DJ, O'Hare LL, Perrin EC, Quackenbush M, Showalter DR, Cicchetti DV (1995) A randomized, controlled trial of a school-based, multi-faceted AIDS education program in the elementary grades: the impact on comprehension, knowledge and fears. Pediatrics 95(4):480-6.

Shope JT, Dielman TE, Butchart AT, Campanelli PC, Kloska DD (1992) An elementary school-based alcohol misuse prevention program: a follow-up evaluation. Journal of Studies on Alcohol 53(2):106-21.

Sigelman CK, Rinehart CS, Sorongon AG, Bridges LJ, Wirtz PW (2004) Teaching a coherent theory of drug action to elementary school children. Health Education Research 19(5):501-13.

Spear C, Young M, Denny G (1997) Field testing of an abstinence-based sexuality education program for upper elementary school students. Journal of Health Education 28(6):335-345.

Spoth R, Greenberg M, Turrisi R (2008) Preventive interventions addressing underage drinking: State of the evidence and steps toward public health impact. Pediatrics 121:S311-S336.

Starkey F, Orme J (2001) Evaluation of a primary school drug drama project: methodological issues and key findings. Health Education Research 16(5):609-22.

Stevens MM, Mott LA, Youells F (1996) Rural adolescent drinking behavior: three year follow-up in the New Hampshire Substance Abuse Prevention Study. Adolescence 31(121):159-166.

Tudor-Smith C, Frankland J, Playle R, Moore L (1995) Life education centres: an evaluation of a mobile health education resource in Wales for children. Health Education Journal 54(4):393-404.

Utley CA, Reddy SS, Delquadri JC, Greenwood CR, Mortweet SL, Bowman V (2001) ClassWide peer tutoring: An effective teaching procedure for facilitating the acquisition of health education and safety facts with students with developmental disabilities. Education & Treatment of Children 24(1):1-27.

van Lier PA, Huizink A, Crijnen A (2009) Impact of a preventive intervention targeting childhood disruptive behavior problems on tobacco and alcohol initiation from age 10 to 13 years. Drug & Alcohol Dependence 100(3):228-234.

Wackett J, Evans L (2000) An evaluation of the choices and changes student program: A grade four to seven sexual health education program based on the Canadian Guidelines for Sexual Health Education. Canadian Journal of Human Sexuality 9(4):265-273.

Welham CA (2007) A Study of the Effectiveness of a Healthy Lifestyles Approach to Drugs Education with Children between 7+ and 11 years of Age. International Journal of Adolescence and Youth 13(3):149-173.

Witt DD, Witt SD (1995) Primary prevention - life skills for young children: an evaluation of the BABES programme. Education 3-13 23(2):45-50.

Wright LS (2007) A norm changing approach to drug prevention. Journal of Drug Education 37(2):191-215.

Young M, Kelley RM, Denny G (1997) Evaluation of selected life-skill modules from the contemporary health series with students in grade. Perceptual & Motor Skills 84(3 Pt 1):811-8.

Zavela KJ, Battistich V, Dean BJ, Flores R, Barton R, Delaney RJ (1997) Say Yes First: a longitudinal, school-based alcohol and drug prevention project for rural youth and families. Journal of Early Adolescence 17(1):67-96.

Zavela KJ, Battistich V, Gosselink CA, Dean BJ (2004) Say Yes First: follow-up of a five year rural drug prevention programme. Journal of Drug Education 34(1):73-88.



Appendix 3. References to excluded studies

• Intervention targeted 'at risk' or high risk population (n=49)

Abell ML, Fraser MW, Galinsky MJ (2001) Early Intervention for Aggressive Behavior in Childhood: A Pilot Study of a Multi-Component Intervention with Elementary School Children and Their Families. Journal of Family Social Work 6(4):19-37.

Adedapo VI (1999) The effects of social skills training on the socially appropriate and antisocial behaviors of elementary students with serious emotional disorders and at-risk behaviors in the general education setting. Dissertation Abstracts International Section A: Humanities and Social Sciences 59(8-A).

Barnstable R, Cargill L, Gehlbach S, Workman H (1997) Advancing the Pro-Social Skills in At-Risk Elementary Students through Curricular Interventions. Master's Action Research Project, Saint Xavier University.

Belgrave FZ (2002) Relational theory and cultural enhancement interventions for African American adolescent girls. Public Health Reports 117 Suppl 1:S76-81.

Bierman KL, Coie JD, Dodge KA, Greenberg MT, Lochman JE, McMahon RJ, Pinderhughes EE, Conduct Problems Prevention Res G (1999) Initial impact of the fast track prevention trial for conduct problems: I. The high-risk sample. Journal of Consulting and Clinical Psychology 67(5):631-647.

Bierman KL, Coie JD, Dodge KA, Greenberg MT, Lochman JE, McMahon RJ, Pinderhughes EE, Conduct Problems Prevention Res G (2002a) Evaluation of the first 3 years of the Fast Track prevention trial with children at high risk for adolescent conduct problems. Journal of Abnormal Child Psychology 30(1):19-35.

Bierman KL, Coie JD, Dodge KA, Greenberg MT, Lochman JE, McMahon RJ, Pinderhughes EE, Conduct Problems Prevention Research Group (2002b) Predictor variables associated with positive Fast Track outcomes at the end of third grade. Journal of Abnormal Child Psychology 30(1):37-52.

Bierman KL, Coie JD, Dodge KA, Greenberg MT, Lochman JE, McMahon RJ, Pinderhughes EE, Conduct Problems Prevention Research Group (2002c) Using the Fast Track randomized prevention trial to test the early-starter model of the development of serious conduct problems. Development and Psychopathology 14(4):925-943.

Bierman KL, Greenberg MT, Conduct Problems Prevention Research G (1996) Social skills training in the Fast Track Program. In: *Preventing childhood disorders, substance abuse, and delinquency*. Peters RD, McMahon RJ eds. Sage Publications, Thousand Oaks, CA.

Bierman KL, Lavallee KL, Nix RL, The Conduct Problems Prevention Research G (2005) The impact of first-grade "friendship group" experiences on child social outcomes in the fast track program. Journal of Abnormal Child Psychology 33(3):307-324.

Burlew K, Neely D, Johnson C, Hucks TC, Purnell B, Butler J, Lovett M, Burlew R (2000) Drug attitudes, racial identity, and alcohol use among African American adolescents. Journal of Black Psychology 26(4):402-20.

Cherry VR, Belgrave FZ, Jones W, Kennon DK, Gray FS, Phillips F (1998) NTU: An africentric approach to substance abuse prevention among African American youth, in Journal of Primary Prevention 18(3):319-339

DeMar J (1997) A school-based group intervention to strengthen personal and social competencies in latency-age children. Social Work in Education 19(4):219-30.

Denham A, Hatfield S, Smethurst N, Tan E, Tribe C (2006) The Effect of Social Skills Interventions in the Primary School. Educational Psychology in Practice 22:33-51.

DeRosier ME (2004) Building relationships and combating bullying: effectiveness of a school-based social skills group intervention. Journal of Clinical Child & Adolescent Psychology 33(1):196-201.

DeRosier ME, Marcus SR (2005) Building friendships and combating bullying: effectiveness of S.S.GRIN at one-year follow-up. Journal of Clinical Child & Adolescent Psychology 34(1):140-50.

DeWit DJ, Steep B, Silverman G, Stevens-Lavigne A, Ellis K, Smythe C, Rye BJ, Braun K, Wood E (2000) Evaluating an in-school drug prevention program for at-risk youth. Alberta Journal of Educational Research 46(2):117-133.

Everhart KD (2001) Promoting resiliency in at-risk children through an integrated adult-child mentoring and character development program: An analysis of mentoring strategies with regard to outcomes. Dissertation Abstracts International: Section B: The Sciences and Engineering 62(2-B).

Flem A, Thygesen R, Valas H, Magnes E (1998) A social skills intervention programme for kindergarten children at risk of developing behavioural problems. European Journal of Special Needs Education 13(2):208-15.

FloresFahs PJ, Lorion RP, Jakob D (1997) Impact of home-school liaisons on the reduction of risk factors for ATOD use among preadolescents. Journal of Community Psychology 25(5):487-503.



Glanz NL (1993) Self-Esteem and Achievement: Case Study of Success with Elementary At-Risk Students. Department of Teacher Education, Augusta College.

Hernandez LP, Hernandez A, Lopez ME, Kreider H, Coffman J (2000) Local and National Implementation of the Families And Schools Together (FAST) Program. The School Community Journal 10(1):85-110.

Imao RK (1990) Stress and coping skills for elementary school children: A comparative study on the relative effectiveness of a social skills training program and a problem-solving skills training program. Dissertation Abstracts International 51(2-A):415-416.

Kaminski RA, Stormshak EA, Good RH, Goodman MR (2002) Prevention of substance abuse with rural head start children and families: results of project STAR. Psychology of Addictive Behaviors 16(4 Suppl):S11-26.

Kilgore TL (1994) Effects of a direct instruction social skills program on the behavior, selfesteem, and peer acceptance of elementary students with limited social skills development. Dissertation Abstracts International: Section B: The Sciences and Engineering 55(4-B).

King KA, Vidourek RA, Davis B, McClellan W (2002) Increasing self-esteem and school connectedness through a multidimensional mentoring program. Journal of School Health 72(7):294-299.

Kjobli J, Sorlie M-A (2008) School outcomes of a community-wide intervention model aimed at preventing problem behavior. Scandinavian Journal of Psychology 49(4):365-75.

Kraizer S, Witte S, Fryer GE, Miyoshi T (1993) Reach & Challenge: Evaluating the Effectiveness of Programs for At- Risk Youth. Published in *Creating the Quality School: Selected Readings*, University of Oklahoma.

Lane KL (1999) Young students at risk for anti-social behavior: the utility of academic and social skills interventions. Journal of Emotional and Behavioral Disorders 7(4):211-23.

Lavallee KL, Bierman KL, Nix RL, Conduct Problems Prevention Research G (2005) The impact of first-grade "friendship group" experiences on child social outcomes in the fast track program. Journal of Abnormal Child Psychology 33(3):307-24.

Lien-Thorne S, Kamps D (2005) Replication study of the first step to success early intervention program. Behavioral Disorders 31(1):18-32.

Lopez MF (2000) A comparison of self-modeling and peer-modeling as interventions to improve the teacher-preferred social skills of at-risk and special education students. Dissertation Abstracts International Section A: Humanities and Social Sciences 60(12-A).

LoSciuto L, Hilbert SM, Fox MM, Porcellini L, Lanphear A (1999) A two-year evaluation of the Woodrock Youth Development Project. The Journal of Early Adolescence 19 (4): 488-507.

Lupton C, Sheppard C (2000) A contradiction in terms? An evaluation of a single agency home-school support project. British Journal of Special Education 27(4):186-90.

Maddern L, Franey J, McLaughlin V, Cox S (2004) An Evaluation of the Impact of an Interagency Intervention Programme to Promote Social Skills in Primary School Children. Educational Psychology in Practice 20:135-155.

McConaughy SH, Kay PJ, Fitzgerald M (1999) The Achieving, Behaving, Caring project for preventing EBD: two-year outcomes. Journal of Emotional and Behavioral Disorders 7(4):224-39.

McMahon CM, Wacker DP, Sasso GM, Melloy KJ (1994) Evaluation of the multiple effects of a social skill intervention. Behavioral Disorders 20(1):35-50.

Middleton MB (1995) The effects of social skills instruction and parent participation on aggressive behaviors, antisocial behaviors, and prosocial skills exhibited by primary-age students. Dissertation Abstracts International Section A: Humanities and Social Sciences 55(10-A).

Miller SC (1998) A program to promote the sociomoral growth of at-risk youth. Dissertation Abstracts International Section A: Humanities and Social Sciences 59(3-A).

Morgan V, Pearson S (1994) Social skills training in a junior school setting. Educational Psychology in Practice 10(2):99-103.

Niles MD, Reynolds AJ, Roe-Sepowitz D (2008) Early childhood intervention and early adolescent social and emotional competence: second-generation evaluation evidence from the Chicago Longitudinal Study. Educational Research 50(1):55-73.

Reid MJ, Webster-Stratton C, Hammond M (2007) Enhancing a classroom social competence and problem-solving curriculum by offering parent training to families of moderate- to high-risk elementary school children. Journal of Clinical Child and Adolescent Psychology 36(4):605-620.

Rentschler DM (1997) A longitudinal study to determine the long-term effects of prevention and intervention substance abuse strategies on at-risk fourth grade students and their families. Dissertation Abstracts International Section A: Humanities and Social Sciences 57(8-A).



Rose DA (1996) The effects of a social skills incentive program on the behavior of students in the regular classroom and in special area classes. Dissertation Abstracts International Section A: Humanities and Social Sciences 57(3-A).

Shapiro MS (1998) An integrative group approach to enhance social competence in a rural elementary school. Dissertation Abstracts International: Section B: The Sciences and Engineering 59(6-B).

St Pierre TL, Mark MM, Kaltreider DL, Campbell B (2001) Boys & Girls Clubs and school collaborations: A longitudinal study of a multicomponent substance abuse prevention program for high-risk elementary school children. Journal of Community Psychology 29(2):87-106.

Warner CM (2007) Examining the effectiveness of social skills training and parenting skills training in the rural schools. Dissertation Abstracts International: Section B: The Sciences and Engineering 67(12-B).

Watson M (2006) Long-term effects of moral/character education in elementary school: In pursuit of mechanisms. Journal of Research in Character Education 4:1-18.

Wayland LAL (2003) Treatment of children with problem behaviours: The efficacy of conjoint behavioural consultation versus videotape therapy and the impact on parent-teacher collaboration. Dissertation Abstracts International: Section B: The Sciences and Engineering 64(1-B).

• Study did not meet design criteria for inclusion (n=137)

Anderson SW, Moore PA (2009) The impact of education and school-based counseling on children's and adolescents' views of substance abuse. Journal of Child & Adolescent Substance Abuse 18(1):16-24.

Anon (1991) Alcohol: The Gateway Drug. OREA Evaluation Report. New York, City Board of Education.

Anon (2006a) An Ethics Curriculum for Children. Rockville, MD, What Works Clearinghouse.

Anon (2006b) Positive Action. Rockville, MD, What Works Clearinghouse.

Anon (2006c) Too Good for Violence. Rockville, MD, What Works Clearinghouse.

Anon (2007) Caring School Community[™] (Formerly, The Child Development Project). Revised. Rockville, MD, What Works Clearinghouse.



Barth RP (1993) Promoting self-protection and self-control through life skill training. Children and Youth Services Review 15(4):281-293.

Beaudoin M-N (2001) Promoting respect and tolerance in schools: Addressing bullying with the 'Bugging Bug' Project. Journal of Systemic Therapies 20(3):10-24.

Beets MW (2007) Factors associated with the implementation fidelity of a school-based social and character development program: Findings from the positive action program, Hawaii. Dissertation Abstracts International: Section B: The Sciences and Engineering 68(6-B).

Bierman KL, Coie JD, Dodge KA, Greenberg MT, Lochman JE, McMahon RJ, Pinderhughes EE (2000) Merging universal and indicated prevention programs: The fast track model. Addictive Behaviors 25(6):913-927.

Binford V, Robertson R (1996) Fairfield Court Elementary: Teamwork through home, school, and community. Journal of Education for Students Placed at Risk 1(3):219-232.

Botvin GJ, Kantor LW (2000) Preventing alcohol and tobacco use through life skills training. Alcohol Research & Health 24(4):250-7.

Brown MD (1991) House Bill No. 2531 passes community based reduction of teenage pregnancy programs. Kansas Nurse 66(8):26.

Bundy ML, White PN (1990) Parents as Sexuality Educators: A Parent Training Program. Journal of Counseling and Development 68(3):321-23.

Bush PJ, Iannotti RJ (1992) Elementary Schoolchildren's Use of Alcohol, Cigarettes and Marijuana and Classmates' Attribution of Socialization. Drug and Alcohol Dependence 30(3):275-287.

Carlson CE (1990) HIPP: A comprehensive school-based substance abuse program with cooperative community involvement. Journal of Primary Prevention 10(4):289-302.

Carroll LA, Miltenberger RG, O'Neill HK (1992) A review and critique of research evaluating child sexual abuse prevention programs. Education & Treatment of Children 15(4):335-354.

Chatterji P, Caffray CM, Jones AS, Lillie-Blanton M, Werthamer L (2001) Applying cost analysis methods to school-based prevention programs. Prevention Science 2(1):45-55.

Churney AH (2001) Promoting children's social and emotional development: A follow-up evaluation of an elementary school-based program in social decision-making/social problemsolving. Dissertation Abstracts International Section A: Humanities and Social Sciences 62(1-A).



Clark EM, Farley JW (1990) Sex Education for Young Children with Special Needs. Preventing School Failure 34(2):21-22.

Cross D, Pintabona Y, Hall M, Hamilton G, Erceg E, Roberts C (2003) The Friendly Schools Project: an empirically grounded school-based bullying prevention program. Australian Journal of Guidance and Counselling 13:36-46.

Crosswaite C, Tooby J, Cyster R (2004) SPICED: Evaluation of a drug education project in Kirklees primary schools. Health Education Journal 63(1):61-69.

Dielman TE, Kloska DD, Leech SL, Schulenberg JE, Shope JT (1992) Susceptibility to peer pressure as an explanatory variable for the differential effectiveness of an alcohol misuse prevention program in elementary schools. Journal of School Health 62(6):233-7.

Donaldson SI, Graham JW, Hansen WB (1994) Testing the generalizability of intervening mechanism theories: understanding the effects of adolescent drug use prevention interventions. Journal of Behavioral Medicine 17(2):195-216.

Donovan P (1993) Project offers counseling and condoms to Philadelphia students. Family Planning Perspectives 25(4):180.

Dryfoos JG (1995) Full service schools: revolution or fad? Journal of Research on Adolescence 5(2):147-72.

Dykeman C, Nelson JR (1996) Students' Evaluations of the Effectiveness of Substance Abuse Education: The Impact of Different Delivery Modes. Journal of Child & Adolescent Substance Abuse 5(2):43-61.

Dzeletovic A, Popovic R (1999) Yugoslavia: preventing the spread of HIV and STDs. Entre Nous 42:15-6.

Eddy JM, Reid JB, Fetrow RA (2000) An elementary school-based prevention program targeting modifiable antecedents of youth delinquency and violence: Linking the Interests of Families and Teachers (LIFT). Journal of Emotional and Behavioral Disorders 8(3):165-176.

English J (1994) Innovative practices in comprehensive health education programs for elementary schools. Journal of School Health 64(5):188-91.

Epanya A, Delude BR (1996) Country watch: Cameroon. AIDS/STD Health Promotion Exchange (3):9-10.

Faith MA, Malcolm KT, Newgent RA (2008) Reducing potential mental health issues and alcohol abuse through an early prevention model for victims of peer harassment. Work - a Journal of Prevention Assessment & Rehabilitation 31(3):327-335.



Farquhar C (1990) Safer teaching or safer sex? Primary teaching in the age of HIV /AIDS. Children and Society 4(90):293-303.

Farrell AD, Meyer AL, Dahlberg LL (1996) Richmond youth against violence: a school-based program for urban adolescents. American Journal of Preventive Medicine 12(5 Suppl):13-21.

Faulk D, Mancuso FM (1998) A collaborative effort for sex education in rural school settings. Nursing & Health Care Perspectives 19(6):271-3.

Femia EE, Zarit SH, Blair C, Jarrott SE, Bruno K (2008) Intergenerational preschool experiences and the young child: Potential benefits to development. Early Childhood Research Quarterly 23(2):272-287.

Fife BL (1994) An Assessment of the Drug Abuse Resistance Education (DARE) Program in Fort Wayne, Indiana. Muncie, IN, Ball State University.

Fox R (2000) Tanzanian AIDS project works towards "good things for young people". Lancet 355(9216):1703.

Frey KS, Hirschstein MK, Guzzo BA (2000) Second Step: preventing aggression by promoting social competence. Journal of Emotional and Behavioral Disorders 8(2):102-12.

Fuller HA, Damico AM, Rodgers S (2004) Impact of a Health and Media Literacy Curriculum on 4th-Grade Girls: A Qualitative Study. Journal of Research in Childhood Education 19:66-78.

Gabhainn SN, Kelleher CC (2000) School health education and gender: an interactive effect? Health Education Research 15(5):591-602.

Gadin KG, Hammarstrom A (2002) Can school-related factors predict future health behaviour among young adolescents? Public Health 116:22-29.

Gale McKee L, Forehand R, Miller KS, Whitaker DJ, Long N, Armistead L (2007) Are parental gender role beliefs a predictor of change in sexual communication in a prevention program? Behavior Modification 31(4):435-53.

Garmiene A, Zemaitiene N, Zaborskis A (2006) Family time, parental behaviour model and the initiation of smoking and alcohol use by ten-year-old children: an epidemiological study in Kaunas, Lithuania. BMC Public Health 6:287.

Gerouki M (2007) Sexuality and relationships education in the Greek primary schools--See no evil, hear no evil, speak no evil. Sex Education 7:81-100.

Giuliano JD (1994) A peer education program to promote the use of conflict resolution skills among at-risk school age males. Public Health Reports 109(2):158-61.



Gorman-Smith D, Tolan P, Henry DB, Quintana E, Lutovsky K, Leventhal A (2007) Schools and Families Educating Children: A Preventive Intervention for Early Elementary School Children. In *Preventing youth substance abuse: Science-based programs for children and adolescents*. Peters RD, McMahon RJ eds. Thousand Oaks, CA, Sage Publications.

Greenspan A (1992) Effects of education on reproductive behavior: lessons from Pakistan. Asia-Pacific Population & Policy (23):1-4.

Griffith JS (1999) D.A.R.E.-ing to be different? A drug prevention and life skills education programme for primary schools. Early Child Development and Care 158:95-105.

Gropper M, Liraz Z, Portowicz D, Schindler M (1995) Computer integrated drug prevention: a new approach to teach lower socioeconomic Fifth and Sixth grade Israeli children to say no to drugs. Social Work in Health Care 22(2):87-103.

Haffner DW (1997) What's wrong with abstinence-only sexuality education programs? Siecus Report 25(4):9-13.

Halstead JM (2005) Teaching about love. British Journal of Educational Studies 53(3):290-305.

Hansen WB, Graham JW, Wolkenstein BH, Rohrbach LA (1991) Program integrity as a moderator of prevention program effectiveness: results for fifth-grade students in the adolescent alcohol prevention trial. Journal of Studies on Alcohol 52(6):568-79.

Harrington-lueker D (1991) Kids and condoms. American School Board Journal 178(5):18-22.

Haydock A (1998) Alcohol education in primary schools. Nursing Standard 13(6):43-6.

Hayes RJ, Changalucha J, Ross DA, Gavyole A, Todd J, Obasi AIN, Plummer ML, Wight D, Mabey DC, Grosskurth H (2005) The MEMA kwa Vijana project: design of a community randomised trial of an innovative adolescent sexual health intervention in rural Tanzania. Contemporary Clinical Trials 26(4):430-42.

Jalongo M (2004) Stories that Teach Life Lessons. Early Childhood Today 19(2):8-43.

Jordan DW, LeMetais J (1997) Social skilling through cooperative learning. Educational Research 39(1):3-21.

Joubert G, Cronje HS (2008) An evaluative study to determine the effectiveness of a primary school-based HIV/AIDS learning intervention on the knowledge of the learners. South African Family Practice 50(3).

Kim H (2004) The effects of sex education programs on knowledge and attitude related to sex of elementary school high grade students. Sexuality and Disability 22(4):377.



Kinsman J, Harrison S, Kengeya-Kayondo J, Kanyesigye E, Musoke S, Whitworth J (1999) Implications of a Comprehensive AIDS Education Programme for Schools in Masaka District, Uganda. AIDS Care 11(5):591-601.

Kirby D, Miller BC (2002) Interventions designed to promote parent-teen communication about sexuality. In: *Talking sexuality: Parent-adolescent communication*. Feldman SS, Rosenthal DA eds. pp 93-110. San Francisco, Jossey-Bass.

Krasnow JH, And O (1992) The Social Competency Program of the Reach Out to Schools Project. Project Report, 1990-91. Wellesley, MA, Stone Center for Developmental Services and Studies.

Lammers JW (1996) The effects of curriculum on student health behaviors: a case study of the growing healthy curriculum on health behaviors of eighth grade students. Journal of Health Education 27(5):278-286.

Larsen T, Samdal O (2007) Implementing second step: Balancing fidelity and program adaptation. Journal of Educational and Psychological Consultation 17(1):1-29.

Lawton Smith S (2007) Catching them early. Mental Health Today 25-27.

Lee A, Cheng FFK, Fung Y, St Leger L (2006) Can Health Promoting Schools contribute to the better health and wellbeing of young people? The Hong Kong experience. Journal of Epidemiology & Community Health 60(6):530-6.

Lee A, Siu DCH, Au SKL, Chen RCI, Cheng KW, Yau FT, Tong LCT, Chan SY, Tsang WW, Ho M (2004) What are the needs of students? An experience from a district based health promoting schools project in Hong Kong. Asia-Pacific Journal of Public Health 16 Suppl:S17-21.

Lenderyou G (1994) Sex education: a school-based perspective. Sexual and Marital Therapy 9(2):127-44.

Lewis C, Battistich V, Schaps E (1990) School-based primary prevention: what is an effective program? New Directions for Child Development (50):35-59.

Ling DC, Tremblay CH (2005) AIDS education, condom demand, and the sexual activity of American youth. Health Economics 14(8):851-867.

Liong B, Tracy A, Kenny M, Brogan D (2008) Gender differences in the relational health of youth participating in a Social Competency Program. Journal of Community Psychology 36(4):499-514.



Lister-Sharp D, Chapman S, Stewart-Brown S, Sowden A (1999) Health promoting schools and health promotion in schools: two systematic reviews. Health Technology Assessment 3(22):1-207.

Lloyd C, Joyce R, Hurry J, Ashton M (2000) The effectiveness of primary school drug education. Drugs-Education Prevention and Policy 7(2):109-126.

Lloyd J (1996) Alcohol and Young People: A Case for Supporting Education about Alcohol in Primary and Secondary Schools. Educational Review 48(2):153-161.

Long JK, Soble L (1999) Report: An art-based violence prevention project for sixth grade students. The Arts in Psychotherapy 26(5):329-344.

Lowden K, Davidson P (2002) Evaluation of the Scotland Against Drugs "Stepping Stones" Board Game. Research Report. Edinburgh, Scottish Council for Research in Education.

Lynch N (2004) The Effectiveness of the Hazelden "No Bullying Program: Preventing Bullying at School" in High-Risk Neighborhoods and Communities. Dissertation Abstracts International, A: The Humanities and Social Sciences 65(6):2392-a.

Maccoby N (1990) Communication and health education research: potential sources for education for prevention of drug use. NIDA Research Monograph 93:1-23.

Males M (1993) School-age pregnancy: why hasn't prevention worked? Journal of School Health 63(10):429-32.

Martin TC, Josiah-Martin JA, Roberts CW, Henry HP (2008) Toward Effective School-based Substance Abuse Prevention "Breaking the Cycle" Programme in Antigua and Barbuda. West Indian Medical Journal 57(4):360-363.

Martorella AM, Portugues AM (1998) Prevention of sexual abuse in children with learning disabilities. Child Abuse Review 7(5):355-9.

Matsumoto S (1995) Sex education and sexual behaviour of adolescents in Japan. Annals of the Academy of Medicine, Singapore 24(5):696-9.

Mattey EA (1996) Teach us body sense (TUBS): a health education program for primary students. Pediatric Nursing 22(6):545-51.

McElearney A, Roosmale-Cocq S, Scott J, Stephenson P (2008) Exploring the anti-bullying role of a befriending peer support programme: A case study within the primary school setting in Northern Ireland. Child Care in Practice 14:109-130.

Meers BW, Werch CE, Hedrick B, Lepper J (1995) Prevention training for teachers: A contemporary challenge. Journal of Alcohol and Drug Education 40(3):47-51.
Meyer AL, Allison KW, Reese LRE, Gay FN, Multisite Violence Prevention P (2004) Choosing to be violence free in middle school - The student component of the GREAT schools and families universal program. American Journal of Preventive Medicine 26(1):20-28.

Meyer AL, Farrell AD (1998) Social Skills Training To Promote Resilience in Urban Sixth-Grade Students: One Product of an Action Research Strategy To Prevent Youth Violence in High-Risk Environments. Education and Treatment of Children 21(4):461-88.

Moody EE (1994) Current trends and issues in childhood sexual abuse prevention programs. Elementary School Guidance & Counseling 28(4):251-256.

Morton JL (2008) "I Feel Good!" a weekly wellness radio broadcast for elementary school children. Journal of School Nursing 24(2):83-7.

Novakova D (1999) Prevention of drug misuse for first grade primary schools pupils. Drugs: Education, Prevention and Policy 6(3):367-71.

Oates J (1992) Julie and the kids club. Health Visitor 65(92):56-7.

Odom SL, McConnell SR, Chandler LK (1994) Acceptability and feasibility of classroombased social interaction interventions for young children with disabilities. Exceptional Children 60(3):226-236.

Oldfield D, Hays BJ, Megel ME (1996) Evaluation of the effectiveness of Project Trust: an elementary school-based victimization prevention strategy. Child Abuse & Neglect 20(9):821-32.

Pankratz MM, Jackson NJ, Giles SM, Ringwalt CL, Bliss K, Bell ML (2006) Implementation Fidelity in a Teacher-Led Alcohol Use Prevention Curriculum. Journal of Drug Education 36:17-333.

Pappas DM, Werch CE, Carlson JM (1998) Recruitment and retention in an alcohol prevention program at two inner-city middle schools. Journal of School Health 68(6):231-6.

Passy R (2005) Family Values and Primary Schools: An Investigation into Family- Related Education. British Educational Research Journal 31:723-736.

Peng HC, Chou CC, Wang WY (2005) The implementation and effectiveness of an initiative game activities program on social interaction in fifth- and sixth-grade students. Research Quarterly for Exercise and Sport 76(1):A107-A107.

Powell K, Simpson T (2000) A plus for puberty: multimedia sexuality education for upper primary school students. Youth Studies Australia 19(4):45-47.



Prinz RJ, Smith EP, Dumas JE, Laughlin JE, White DW, Barron R (2001) Recruitment and retention of participants in prevention trials involving family-based interventions. American Journal of Preventive Medicine 20(Suppl1):31-37.

Quine S, Pierce JP (1992) A role for drug awareness and prevention programs external to the school? Health Education Research 7(92):259-267.

Reichstetter RA (2001) Wake County Safe Schools/Health Students Project. Year Two-Mid-Year Performance Report. October 1, 2000-March 31, 2001. E&R Report. Raleigh, Wake County Public School System Department of Evaluation and Research.

Resnicow K, Cross D, Wynder E (1993) The Know Your Body program: a review of evaluation studies. Bulletin of the New York Academy of Medicine 70(3):188-207.

Reynolds AJ, Temple JA (2008) Cost-effective early childhood development programs from preschool to third grade. Annual Review of Clinical Psychology 4:109-139.

Riggs NR, Greenberg MT, Kusche CA, Pentz MA (2006) The mediational role of neurocognition in the behavioral outcomes of a social-emotional prevention program in elementary school students: effects of the PATHS Curriculum. Prevention Science 7(1):91-102.

Roffey S (2006) 'Respect' in practice: the challenge of emotional literacy in education. Sydney, University of Western Sydney.

Roffman DM, Tyksinski K (1998) Body openings: A professional development program in health and human sexuality education for early childhood in an independent school setting. Journal of Sex Education and Therapy 23(1):73-82.

Rohrbach LA, Graham JW, Hansen WB (1993) Diffusion of a school-based substance abuse prevention program: predictors of program implementation. Preventive Medicine 22(2):237-60.

Russo T, Emmett J, Lundeberg M, Monsour F, et al., (1996) Open meetings in the elementary school: Facilitating the development of social interest. Journal of Reality Therapy 15(2):82-89.

Salisbury CL, Gallucci C, Palombaro MM, Peck CA (1995) Strategies that promote social relations among elementary students with and without severe disabilities in inclusive schools. Exceptional Children 62(2):125-37.

Schall E (1994) School-based health education: what works? American Journal of Preventive Medicine 10(3 Suppl):30-2.



Schonfeld DJ (2000) Teaching young children about HIV and AIDS. Child and Adolescent Psychiatric Clinics of North America 9(2):375-387.

Schonfeld DJ, Quackenbush M (2000) Teaching Young Children about AIDS. Principal 79(5):33-35.

Schwarz J, Michaely U, Keppler C (1991) 'Healthy living is fun': a creative way of health education. Health Promotion International (3):229-32.

Shucksmith J, Wood S (1998) Keep a cool head: drug education in primary schools. Health Education 98:191-199.

Silvia ES, Thorne J, Tashjian CA (1997) School-Based Drug Prevention Programs: A Longitudinal Study in Selected School Districts. Final Report. Research Triangle Park, Research Triangle Institute.

Simpkins SD, Fredricks JA, Davis-Kean PE, Eccles JS (2006) Healthy Mind, Healthy Habits: The Influence of Activity Involvement in Middle Childhood. In: Developmental contexts in middle childhood: Bridges to adolescence and adulthood. Huston AC, Ripke MN eds. Cambridge, Cambridge University Press.

Simpson KJ, Young TL (1998) The Harrison Healthy Kids Center: a comprehensive elementary school-based health program. Journal of School Health 68(3):116-9.

Smith EP, Dumas JE, Prinz R (2006) Prevention Approaches to Improve Child and Adolescent Behavior and Reduce Deviant Peer Influence. In: Deviant peer influences in programs for youth: Problems and solutions. Dodge KA, Dishion TJ, Lansford JE eds. New York, Guildford Press.

Solomon GB (1997) Fair Play in the Gymnasium: Improving Social Skills among Elementary School Students. Journal of Physical Education Recreation and Dance 68(5):22-25.

St Leger LH (1999) The opportunities and effectiveness of the health promoting primary school in improving child health--a review of the claims and evidence. Health Education Research 14(1):51-69.

Stevens V, Van Oost P, De Bourdeaudhuij I (2001) Implementation process of the Flemish antibullying intervention and relation with program effectiveness. Journal of School Psychology 39(4):303-317.

Stipek D, de la Sota A, Weishaupt L (1999) Life lessons: An embedded classroom approach to preventing high-risk behaviors among preadolescents. Elementary School Journal 99(5):433-451.



Swisher JD, Ashby JS (1993) Review of process and outcome evaluations of team training. Journal of Alcohol and Drug Education 39(1):66-77.

Tatchell T, Waite P, Tatchell R, Kaderavek J, Strobel S, Jordan T (2004) Substance abuse in sixth grade: results of a prevention program on adolescents' self-reported drug-related attitudes and behavior. Psychological Reports 94(1):225-6.

Terzian MA, Fraser MW (2005) Preventing aggressive behavior and drug use in elementary school: Six family-oriented programs. Aggression and Violent Behavior 10(4):407-435.

Valois RF, Hoyle TB (2000) Formative evaluation results from the Mariner Project: a coordinated school health pilot program. Journal of School Health 70(3):95-103.

Wasow E, Ruhf R (1992) Safe Spaces: Drug and Alcohol Prevention Education for Special Needs and Drug Exposed K-2 Children [and] Videotape. Guide to the Staff Development Video Program for K-2 Special Needs Educators. New York, NY, Bank Street College of Education.

Webster-Stratton C, Reid MJ (2007) Incredible Years Parents and Teachers Training Series: A Head Start Partnership to Promote Social Competence and Prevent Conduct Problems. In: *Preventing youth substance abuse: Science-based programs for children and adolescents.* Tolan P, Szapocznik J, Sambrano S eds. Washington DC, American Psychological Association.

Weissberg RP, Shriver TP (1996) School-Based Prevention Programs: A Comprehensive Strategy. Spotlight on Student Success. No. 113. Philadelphia, PA, Mid-Atlantic Laboratory for Student Success.

Werch CE, Owen DM (2002) latrogenic effects of alcohol and drug prevention programs. Journal of Studies on Alcohol 63(5):581-590.

Wilcox C (2005) Needs assessment and development of a bullying prevention program for elementary school students. Dissertation Abstracts International: Section B: The Sciences and Engineering 66(6-B).

William T. Grant Foundation, Consortium on the School-Based Promotion of Social Competence New York (1992) Drug and alcohol prevention curricula. In: *Communities that care: Action for drug abuse prevention.* Hawkins DJ, Catalano RF eds. San Francisco, Jossey Bass.

Woolley CCM, Gabriels TCM (1999) Children's conceptualization of some child sexual abuse prevention concepts as taught by "Keeping Ourselves Safe," a New Zealand prevention programme. Australasian Journal of Disaster and Trauma Studies 3(1).



Wyvill B, Ives R (2000) Finding out about young people's ideas on drugs and drug use - methodology. Drugs: Education, Prevention and Policy 7(2):127-37.

Young M, Rausch S (1991) Be a winner: Arkansas' approach to involving law enforcement officers in drug education. Journal of Drug Education 21(2):183-189.

Zins J, Elias M, Greenberg M (2003) Facilitating success in school and in life through social and emotional learning. Perspectives in Education 21(4):55-67.

• Intervention examined was not alcohol education and/SRE related (n=184)

Aber JL, Brown JL, Jones SM (2003) Developmental trajectories toward violence in middle childhood: course, demographic differences, and response to school-based intervention. Developmental Psychology 39(2):324-48.

Aber JL, Jones SM, Brown JL, Chaudry N, Samples F (1998) Resolving conflict creatively: evaluating the developmental effects of a school-based violence prevention program in neighborhood and classroom context. Development & Psychopathology 10(2):187-213.

Adalbjarnardottir S (1993) Promoting Children's Social Growth in the Schools: An Intervention Study. Journal of Applied Developmental Psychology 14(4):461-484.

Adi Y, Killoran A, Janmohamed K, Stewart-Brown S (2007) Systematic review of the effectiveness of interventions to promote mental wellbeing in children in primary education. Report 1: Universal approaches: non-violence related outcomes. Warwick, University of Warwick.

Albers EC (1992) An evaluation measuring the acquisition and retention of child sexual assault prevention concepts by second grade school children. School social work journal 17(1):10-16.

Amatruda M-J (2006) Conflict Resolution and Social Skill Development With Children. Journal of Group Psychotherapy, Psychodrama & Sociometry 58:168-181.

Anderson-Butcher D, Nay S, Newsome WS (2003) Social Skills Intervention during Elementary School Recess: A Visual Analysis. Children and Schools 25(3):135-146.

Andreou E, Didaskalou E, Vlachou A (2008) Outcomes of a curriculum-based anti-bullying intervention program on students' attitudes and behavior. Emotional & Behavioural Difficulties 13(4):235-249.

Araji SK, Fenton R, Straugh T (1995) Child Sexual Abuse: Description and Evaluation of a K-6 Prevention Curriculum. The Journal of Primary Prevention 16(2):149-164.

Ary DV, Biglan A, Glasgow R, Zoref L, Black C, Ochs L, Severson H, Kelly R, Weissman W, Lichtenstein E, et al (1990) The efficacy of social-influence prevention programs versus "standard care": are new initiatives needed? Journal of Behavioral Medicine 13(3):281-96.

Baker JB (2008) Using computer-based muves to develop social skills in elementary children: An exploratory study. Dissertation Abstracts International Section A: Humanities and Social Sciences 69(5-A).

Bektas M, Ozturk C (2008) Effect of Health Promotion Education on Presence of Positive Health Behaviors, Level of Anxiety and Self-Concept. Social Behavior and Personality 36(5):681-690.

Beran TN, Tutty L, Steinrath G (2004) An Evaluation of a Bullying Prevention Program for Elementary Schools. Canadian Journal of School Psychology 19:99-116.

Bergeson T, Daybell MS, Riggers ML, Mueller MT, Johnson MM (2004) Addressing Adolescent Substance Abuse: An Evaluation of Washington's Prevention and Intervention Services Program. 2001-03 Final Report. Olympia, Washington Office of the State Superintendent of Public Instruction.

Berrick JD, Barth RP (1992) Child sexual abuse prevention: Research review and recommendations. Social Work Research & Abstracts 28(4):6-15.

Blumberg EJ, Chadwick MW, Fogarty LA, Speth TW, et al (1991) The touch discrimination component of sexual abuse prevention training: Unanticipated positive consequences. Journal of Interpersonal Violence 6(1):12-28.

Briggs F, Hawkins RM (1994) Follow-up data on the effectiveness of New Zealand's national school based child protection program. Child Abuse & Neglect 18(8):635-43.

Cappella E (2005) The prevention of social aggression among girls. Dissertation Abstracts International: Section B: The Sciences and Engineering 65(9-B).

Cappella E, Weinstein R (2006) The Prevention of Social Aggression among Girls. Social Development 15(3):434-462.

Cassell JR (1995) Improving Self-Control in Upper Elementary Students through a Program of Character, Civic, and Social Education.

Chaux E (2007) Aulas en Paz: A multicomponent program for the promotion of peaceful relationships and citizenship competencies. Conflict Resolution Quarterly 25:79-86.

Choi HS, Heckenlaible-Gotto MJ (1998) Classroom-based social skills training: Impact on peer acceptance of first-grade students. Journal of Educational Research 91(4):209-214.



Cooper WO, Lutenbacher M, Faccia K (2000) Components of effective youth violence prevention programs for 7-to 14-year-olds. Archives of Pediatrics & Adolescent Medicine 154(11):1134-1139.

Counts MA (2004) Smart kids/safe kids: Evaluation of a child sexual abuse prevention program. Dissertation Abstracts International Section A: Humanities and Social Sciences 64(10-A).

Curtis C, Norgate R (2007) An evaluation of the promoting alternative thinking strategies curriculum at key stage 1. Educational Psychology in Practice 23:33-44.

Davidson ML (2000) Using moral maxims to promote character development in sixth-grade students: A collaborative action researchapproach for planning, implementing, and evaluating comprehensive character education strategies. Dissertation Abstracts International Section A: Humanities and Social Sciences 61(6-A).

Desocio J, Stember L, Schrinsky J (2006) Teaching children about mental health and illness: a school nurse health education program. Journal of School Nursing 22(2):81-6.

Devargas RC (1999) A study of "lessons in character": The effect of moral development curriculum upon moral judgment. (character education, fifth-grade). Dissertation Abstracts International Section A: Humanities and Social Sciences 59(11-A).

Dinger MK, Ogletree RJ, Johnson D (2000) Assessment of health knowledge after "a healthy adventure". Journal of School Health 70(3):104-6.

Dion E, Fuchs D, Fuchs LS (2005) Differential effects of peer-assisted learning strategies on students' social preference and friendship making. Behavioral Disorders 30(4):421-429.

DuRant RH, Barkin S, Krowchuk DP (2001) Evaluation of a peaceful conflict resolution and violence prevention curriculum for sixth-grade students. Journal of Adolescent Health 28(5):386-393.

Durlak JA, Wells AM (1997) Primary prevention mental health programs for children and adolescents: A meta-analytic review. American Journal of Community Psychology 25(2):115-152.

Edmondson L, Hoover J (2008) Process Evaluation of a Bullying Prevention Program: A Public School- County Health Partnership. Reclaiming Children and Youth. The Journal of Strength based Interventions 16:9-33.

El Hassan K, Kahil R (2005) The Effect of "Living Values: An Educational Program" on Behaviors and Attitudes of Elementary Students in a Private School in Lebanon. Early Childhood Education Journal 33:81-90.



Elinoff MJ (2006) Friends against bullying: Evaluation of a peer-mentoring program to reduce bullying in schools. Dissertation Abstracts International Section A: Humanities and Social Sciences 67(2-A).

Embry DD, Flannery DJ, Vazsonyi AT, Powell KE, Atha H (1996) PeaceBuilders: A theoretically driven, school based model for early violence prevention. American Journal of Preventive Medicine 12(5):91-100.

Erwin PG, Ruane GE (1993) The effects of a short-term social problem solving programme with children. Counselling Psychology Quarterly 6(4):317-323.

Fekkes M, Pijpers FIM, Verloove-Vanhorick SP (2006) Effects of antibullying school program on bullying and health complaints. Archives of Pediatrics & Adolescent Medicine 160(6):638-44.

Fixler B (2000) A Caring and Sharing Environment Helps Teach Values in Kindergarten Students. Chicago, IL, Saint Xavier University.

Flannery DJ, Vazsonyi AT, Liau AK, Guo SY, Powell KE, Atha H, Vesterdal W, Embry D (2003) Initial behavior outcomes for the PeaceBuilders universal school-based violence prevention program. Developmental Psychology 39(2):292-308.

Fonagy P, Twemlow SW, Vernberg E, Sacco FC, Little TD (2005) Creating a peaceful school learning environment: the impact of an antibullying program on educational attainment in elementary schools. Medical Science Monitor 11(7):CR317-25.

Fraser MW, Galinsky MJ, Smokowski PR, Day SH, Terzian MA, Rose RA, Guo S (2005) Social information-processing skills training to promote social competence and prevent aggressive behavior in the third grades. Journal of Consulting & Clinical Psychology 73(6):1045-55.

Fuqua DS (2008) Safe @ Last: The evaluation of a child sexual abuse prevention program for elementary students. Dissertation Abstracts International Section A: Humanities and Social Sciences 69(4-A).

Germaine RW (2001) Values education influence on elementary students' self-esteem. Dissertation Abstracts International Section A: Humanities and Social Sciences 62(3-A).

Gibbons SL, Ebbeck V (1997) The effect of different teaching strategies on the moral development of physical education students. Journal of Teaching in Physical Education 17(1):85-98.



Gibbons SL, Ebbeck V, Weiss MR (1995) Fair Play for Kids: effects on the moral development of children in physical education. Research Quarterly for Exercise & Sport 66(3):247-55.

Giles MM, Cogan D, Cox C (1991) A music and art program to promote emotional health in elementary-school-children. Journal of Music Therapy 28(3):135-148.

Ginsburg-Block MD, Rohrbeck CA, Fantuzzo JW (2006) A meta-analytic review of social, self-concept, and behavioral outcomes of peer-assisted learning. Journal of Educational Psychology 98(4):732-749.

Green J, Howes F, Waters E, Maher E, Oberklaid F (2005) Promoting the social and emotional health of primary school-aged children: reviewing the evidence base for school-based interventions. International Journal of Mental Health Promotion 7(3):30-36.

Hahn R, Fuqua-Whitley D, Wethington H, Lowy J, Crosby A, Fullilove M, Johnson R, Liberman A, Moscicki E, Price L, Snyder S, Tuma F, Cory S, Stone G, Mukhopadhaya K, Chattopadhyay S, Dahlberg L (2007) Effectiveness of universal school-based programs to prevent violent and aggressive behavior: a systematic review. American Journal of Preventive Medicine 33(2 Suppl):S114-S129.

Hallford A, Borntrager C, Davis JL (2006) Evaluation of a Bullying Prevention Program. Journal of Research in Childhood Education 21:11.

Hayward KS, Pehrsson DE (2000) Interdisciplinary action supporting sexual assault prevention efforts in rural elementary schools. Journal of Community Health Nursing 17(3):141-50.

Hazzard A, Webb C, Kleemeier C, Angert L, Pohl J (1991) Child sexual abuse prevention: evaluation and one-year follow-up. Child Abuse & Neglect 15(1-2):123-38.

Hebert M, Lavoie F, Piche C, Poitras M (2001) Proximate effects of a child sexual abuse prevention program in elementary school children. Child Abuse & Neglect 25(4):505-22.

Hennessey BA (2007) Promoting social competence in school-aged children: The effects of the open circle program. Journal of School Psychology 45(3):349-360.

Hepler JB (1994) Evaluating the effectiveness of a social skills program for preadolescents. Research on Social Work Practice 4(4):411-435.

Hepler JB (1997) Evaluating a social skills program for children with learning disabilities. Social Work with Groups 20(3):21-36.

Heydenberk RA, Heydenberk WR (2005) Increasing meta-cognitive competence through conflict resolution. Education and Urban Society 37(4):431-452.

Holsen I, Smith BH, Frey KS (2008) Outcomes of the Social Competence Program Second Step in Norwegian Elementary Schools. School Psychology International 29(1):71-88.

Houlette MA, Gaertner SL, Johnson KM, Banker BS, Riek BM, Dovidio JF (2004) Developing a More Inclusive Social Identity: An Elementary School Intervention. Journal of Social Issues 60(1):35-55.

Howard KA, Flora J, Griffin M (1999) Violence-prevention programs in schools: State of the science and implications for future research. Applied & Preventive Psychology 8(3):197-215.

Hunter L (1999) Preventing violence through the promotion of social competence and positive interethnic contact: An evaluation of three elementary school-based violence prevention instructional approaches. Dissertation Abstracts International Section A: Humanities and Social Sciences 59(8-A).

Hutchings J, Lane E, Owen RE, Gwyn R (2004) The introduction of the Webster-Stratton Incredible Years Classroom Dinosaur School Programme in Gwynedd, North Wales: A pilot study. Educational and Child Psychology 21:4-15.

lalongo N, Poduska J, Werthamer L, Kellam S (2001) The distal impact of two first-grade preventive interventions on conduct problems and disorder in early adolescence. Journal of Emotional and Behavioral Disorders 9(3):146-160.

Jensen P, And O (1994) Improving Social Behavior of Fifth through Eighth Grade Students through Curriculum Intervention and Teaching Practices. Saint Xavier University

Jenson JM, Dieterich WA (2007) Effects of a skills-based prevention program on bullying and bully victimization among elementary school children. Prevention Science 8(4):285-96.

Johnson DW, Johnson R, Dudley B, Acikgoz K (1994) Effects of conflict-resolution training on elementary-school students. Journal of Social Psychology 134(6):803-817.

Jones CA, Gower D (1994) Effects of values instruction on third and fourth grade students.

Jones RT, McDonald DW, Fiore MF, Arrington T, Randall J (1990) A primary preventive approach to children's drug refusal behavior: the impact of rehearsal-plus. Journal of Pediatric Psychology 15(2):211-23.

Josephson HL-E (2006) Fostering the social development of shy students: A preventative intervention study. Dissertation Abstracts International Section A: Humanities and Social Sciences 66(11-A).

Kam C, Greenberg MT, Kusche CA (2004) Sustained effects of the PATHS curriculum on the social and psychological adjustment of children in special education. Journal of Emotional & Behavioral Disorders 12(2):66-79.



Kiemel KB (1997) Assessing the fears and anxieties of second and third graders after participating in a sexual abuse prevention program. Dissertation Abstracts International: Section B: The Sciences and Engineering 58(1-B).

Kilian JM, Fish MC, Maniago EB (2006) Making Schools Safe: A System-Wide School Intervention to Increase Student Prosocial Behaviors and Enhance School Climate. Journal of Applied School Psychology 23(1):1-30.

Kimball KA (1998) Reducing Negative Behaviors of Elementary School Students through a Program Which Honors Values Discussions, the Arts, and Satisfies Children's Basic Needs. Fort Lauderdale, Nova Southeastern University.

King DR, Jr. (2001) Classroom-based social skills training as primary prevention in kindergarten: Teacher ratings of social functioning. Dissertation Abstracts International Section A: Humanities and Social Sciences 61(8-A).

Krasnow JH, And O (1992) The Social Competency Program of the Reach Out to Schools Project. Project Report, 1991-92. No.3. Wellesley, MA. Stone Center for Developmental Services and Studies

Kulaksizoglu A, Dilmac B (2003) Imparting education in human values to primary education students and testing it by implementing a moral maturity scale. Studia Psychologica 45(1):43-50.

Lakin L, Littledyke M (2008) Health promoting schools: Integrated practices to develop critical thinking and healthy lifestyles through farming, growing and healthy eating. International Journal of Consumer Studies 32:253-259.

Leadbeater D, Hoglund W, Woods T (2003) Changing contexts? The effects of a primary prevention program on classroom levels of peer relational and physical victimization. Journal of Community Psychology 31(4):397-418.

Lee A, Wong MCS, Keung VMW, Yuen HSK, Cheng F, Mok JSY (2008) Can the concept of Health Promoting Schools help to improve students' health knowledge and practices to combat the challenge of communicable diseases: Case study in Hong Kong? BMC Public Health 8:42.

Levy SR, Perhats C, Nash-Johnson M, Welter JF (1992) Reducing the risks in pregnant teens who are very young and those with mild mental retardation. Mental Retardation 30(4):195-203.



Lewis TJ, Sugai G, Colvin G (1998) Reducing problem behavior through a school-wide system of effective behavioral support: Investigation of a school-wide social skills training program and contextual interventions. School Psychology Review 27(3):446-459.

Lillenstein JA (2002) Efficacy of a social skills training curriculum with early elementary students in four parochial schools. Dissertation Abstracts International Section A: Humanities and Social Sciences 62(9-A).

Linares LO, Rosbruch N, Stern MB, Edwards ME, Walker G, Abikoff HB, Alvir JMJ (2005) Developing Cognitive-Social-Emotional Competencies to Enhance Academic Learning. Psychology in the Schools 42:405-417.

Lindberg LC, Stahle A, Ryden L (2006) Long-term influence of a health education programme on knowledge and health behaviour in children. European Journal of Cardiovascular Prevention & Rehabilitation 13(1):91-7.

Liu H-CM (2008) The effect of a learning-related social skills (LRSS) physical education training program on kindergarten children. Dissertation Abstracts International Section A: Humanities and Social Sciences 69(5-A).

MacIntyre D, Carr A (1999) Evaluation of the effectiveness of the Stay Safe Primary Prevention Programme for child sexual abuse. Child Abuse and Neglect 23(12):1307-25.

Mackey JP (2007) The effects of pathways youth development program curriculum in enhancing the emotional and behavioral functioning of fifth and eighth grade urban youth. Dissertation Abstracts International: Section B: The Sciences and Engineering 67(8-B).

MacMillan HL, MacMillan JH, Offord DR, Griffith L, MacMillan A (1994) Primary prevention of child sexual abuse: a critical review. Part II. Journal of Child Psychology & Psychiatry & Allied Disciplines 35(5):857-76.

Madak PR, Berg DH (1992) The prevention of sexual abuse: An evaluation of "Talking About Touching.". Canadian Journal of Counselling 26(1):29-40.

Manees AM (2008) Student attitudes and shame dynamics before and after a bullying prevention program. Dissertation Abstracts International: Section B: The Sciences and Engineering 69(4-B).

Martin D, Martin M (2007) Implementing A Family/School Partnership In An Urban Elementary School To Reduce Negative Behavior And Increase Academic Achievement. Family Therapy 34(3):141-152.

McIntyre L, And O (1996) The Dartmouth Health Promotion Study: A Failed Quest for Synergy in School Health Promotion. Journal of School Health 66(4):132-37.



McKee L (1997) The effectiveness of a personal safety program for first grade children. Dissertation Abstracts International: Section B: The Sciences and Engineering 58(4-B).

Meleskie LK (1994) Improving the Awareness of Personal and Oral Hygiene in Second Graders. Fort Lauderdale, Nova Southeastern University.

Meraviglia MG, Becker H, Rosenbluth B, Sanchez E, Robertson T (2003) The Expect Respect Project: creating a positive elementary school climate. Journal of Interpersonal Violence 18(11):1347-1361.

Merrell KW, Juskelis MP, Tran OK, Buchanan R (2008) Social and Emotional Learning in the Classroom: Evaluation of Strong Kids and Strong Teens or Students' Social-Emotional Knowledge and Symptoms. Journal of Applied School Psychology 24(2):209-224.

Minton SJ, O' Moore AM (2008) The effectiveness of a nationwide intervention programme to prevent and counter school bullying in Ireland. International Journal of Psychology & Psychological Therapy 8:1-12.

Mishara BL, Ystgaard M (2006) Effectiveness of a mental health promotion program to improve coping skills in young children: Zippy's Friends. Early Childhood Research Quarterly 21(1):110-123.

Mokrue K (2003) Using a social and emotional skills curriculum to decipher the role family environment plays in social competence among urban elementary school children. Dissertation Abstracts International: Section B: The Sciences and Engineering 64(6-B).

Mokrue K, Elias MJ, Bry BH (2005) Dosage Effect and the Efficacy of a Video-Based Teamwork-Building Series with Urban Elementary School Children. Journal of Applied School Psychology 21:67-97.

Monasch LB (2004) A program evaluation: Implementation of an abbreviated social skills program using folk literature in the regular classroom. Dissertation Abstracts International: Section B: The Sciences and Engineering 64(8-B).

Mukoma W, Flisher AJ (2004) Evaluations of health promoting schools: a review of nine studies. Health Promotion International 19(3):357-368.

Mytton JA, DiGuiseppi C, Gough DA, Taylor RS, Logan S (2002) School-based violence prevention programs: systematic review of sedondary prevention trials. Archives of Pediatrics and Adolescent Medicine 156(8):752-762.

Norris JA (1998) Promoting social competence and reducing violence and negative social interactions in a multicultural school setting. Dissertation Abstracts International Section A: Humanities and Social Sciences 59(5-A).



Oganowski JL, And O (1996) The Wisconsin Elementary Health Education Pilot Project: On-Site Interviews of Student Learning and Curricular Integration. Journal of Health Education 27(4):235-41.

Omizo MM, Omizo SA, D'Andrea MJ (1992) Promoting wellness among elementary school children. Journal of Counseling & Development 71(2):194-198.

O'Moore AM, Minton SJ (2005) Evaluation of the effectiveness of an anti-bullying programme in primary schools. Aggressive Behavior 31:609-622.

Orpinas P, Horne AM, Staniszewski D (2003) School bullying: Changing the problem by changing the school. School Psychology Review 32(3):431-444.

Orpinas P, Kelder S, Frankowski R, Murray N, Zhang Q, McAlister A (2000) Outcome evaluation of a multi-component violence-prevention program for middle schools: the Students for Peace project. Health Education Research 15(1):45-58.

Petermann F, Natzke H (2008) Preliminary Results of a Comprehensive Approach to Prevent Antisocial Behaviour in Preschool and Primary School Pupils in Luxembourg. School Psychology International 29(5):606-626.

Ramseier CA, Leiggener I, Lang NP, Bagramian RA, Inglehart MR (2007) Short-term effects of hygiene education for preschool (kindergarten) children: a clinical study. Oral Health & Preventive Dentistry 5(1):19-24.

Rattigan PJC (1997) A study of the effects of cooperative, competitive and individualistic goal structures on skill development, affect, and social integration in physical education classes. Dissertation Abstracts International Section A: Humanities and Social Sciences 58(5-A).

Reed VA, Jernstedt GC (2000) Teaching children about health: an example of secondary gain in an academic-community partnership. Education for Health 13(3):357-65.

Resnicow K, Cohn L, Reinhardt J, Cross D, Futterman R, Kirschner E, Wynder EL, Allegrante JP (1992) A three-year evaluation of the know your body program in inner-city schoolchildren. Health Education Quarterly 19(4):463-80.

Resnicow K, Cross D, Wynder E (1991) The role of comprehensive school-based interventions: The results of four Know Your Body studies. Annals of the New York Academy of Sciences 623:285-298.

Resnicow K, Davis M, Smith M, Baranowski T, Lin LS, Baranowski J, Doyle C, Wang DQT (1998) Results of the TeachWell worksite wellness program. American Journal of Public Health 88(2):250-257.



Reynolds AJ, Temple JA, Ou S-R, Robertson DL, Mersky JP, Topitzes JW, Niles MD (2007) Effects of a school-based, early childhood intervention on adult health and well-being: a 19-year follow-up of low-income families. Archives of Pediatrics & Adolescent Medicine 161(8):730-9.

Richman RD (1998) Preventing disordered eating; promoting healthy attitudes and behaviors: A school-based program. Dissertation Abstracts International: Section B: The Sciences and Engineering 59(1-B).

Rispens J, Aleman A, Goudena PP (1997) Prevention of child sexual abuse victimization: a meta-analysis of school programs. Child Abuse and Neglect 21(10):975-987.

Robinson CW, Zajicek JM (2005) Growing minds: The effects of a one-year school garden program on six constructs of life skills of elementary school children. Horttechnology 15(3):453-457.

Rock EA, Hammond M, Rasmussen S (2004) School-Wide Bullying Prevention Program for Elementary Students. Journal of Emotional Abuse 4(3-4):225-239.

Romano JL, Miller JP, Nordness A (1996) Stress and well-being in the elementary school: A classroom curriculum. School Counselor 43(4):268-276.

Rooney R, Roberts C, Kane R, Pike L, Winsor A, White J, Brown A (2006) The Prevention of Depression in 8- to 9-Year-Old Children: A Pilot Study. Australian Journal of Guidance & Counselling 16:76-90.

Roseberry LL (1997) An applied experimental evaluation of conflict resolution curriculum and social skills development. Dissertation Abstracts International Section A: Humanities and Social Sciences 58(3-A).

Royal CW (2000) The effect of a moral education program on parents of elementary school students. Dissertation Abstracts International Section A: Humanities and Social Sciences 61(2-A).

Russell-Mayhew S, Arthur N, Ewashen C (2007) Targeting students, teachers and parents in a wellness-based prevention program in schools. Brunner-Mazel Eating Disorders Monograph Series 15(2):159-81.

Salvatore AJ (2006) An anti-bullying strategy: Action research in a 5/6 intermediate school. Dissertation Abstracts International Section A: Humanities and Social Sciences 67(3-A).

Salzman M, D'Andrea M (2001) Assessing the impact of a prejudice prevention project. Journal of Counseling & Development 79(3):341-346.



Sanchez E, Robertson TR, Lewis CM, Rosenbluth B, Bohman T, Casey DM (2001) Preventing bullying and sexual harassment in elementary schools: The Expect Respect Model. Journal of Emotional Abuse 2(2-3):157-180.

Sawyer MG, MacMullin C, Graetz B, Said JA, Clark JJ, Baghurst P (1997) Social skills training for primary school children: a 1-year follow-up study. Journal of Paediatrics & Child Health 33(5):378-83.

Schaber PM (2008) The effectiveness of a classroom-based intervention for social aggression. Dissertation Abstracts International Section A: Humanities and Social Sciences 68(9-A).

Schellenberg RC (2005) School violence: Evaluation of an elementary school peer mediation program. Dissertation Abstracts International Section A: Humanities and Social Sciences 66(6-A).

Schindler J, Detert RA, Pretasky B, Oganowski JL (1996) The Wisconsin Elementary Health Education Pilot Project: year 2 written assessment of knowledge, attitudes, and problem solving skill changes. Journal of Health Education 27(5):286-294.

Sefer J (1995) The effects of play oriented curriculum on creativity in elementary school children. Gifted Education International 11(1):4-17.

Sharma M (1998) Evaluation of a coping intervention developing problem solving skills among upper elementary school children based on social cognitive theory. Dissertation Abstracts International: Section B: The Sciences and Engineering 58(7-B).

Sharma M, Petosa R, Heaney CA (1999) Evaluation of a brief intervention based on social cognitive theory to develop problem-solving skills among sixth-grade children. Health Education & Behavior 26(4):465-77.

Sharpe T, Brown M, Crider K (1995) The effects of a sportsmanship curriculum intervention on generalized positive social behavior of urban elementary school students. Journal of Applied Behavior Analysis 28(4):401-416.

Shechtman Z (1997) Enhancing social relationships and adjusting behavior in the Israeli classroom. Journal of Educational Research 91(2):99-107.

Silvestri L, Dantonio M, Eason S (1994) Enhancement of self-esteem in at-risk elementary students. Journal of Health Education 25(1):30-41.

Simon TR, Ikeda RM, Smith EP, Reese LE, Rabiner DL, Miller-Johnson S, Winn DM, Dodge KA, Asher SR, Horne AM, Orpinas P, Martin R, Quinn WH, Tolan PH, Gorman-Smith D, Henry DB, Gay FN, Schoeny M, Farrell AD, Meyer AL, Sullivan TN, Allison KW (2008) The



Multisite Violence Prevention Project: Impact of a Universal School-Based Violence Prevention Program on Social-Cognitive Outcomes. Prevention Science 9(4):231-244.

Slee PT, Mohyla J (2007) The PEACE Pack: an evaluation of interventions to reduce bullying in four Australian primary schools. Educational Research 49(2):103-114.

Smith BT (1997) The effect of a cooperative learning intervention on the social skills enhancement of a third grade physical education class. Dissertation Abstracts International Section A: Humanities and Social Sciences 57(10-A).

Soffer AG (2003) School-based social skills training to reduce children's depressive symptomatology. Dissertation Abstracts International Section A: Humanities and Social Sciences 63(12-A).

Sontag AM (2006) Prevention of bullying: Evaluation of steps to respect. Dissertation Abstracts International Section A: Humanities and Social Sciences 67(1-A).

Sorlie M-A, Ogden T (2007) Immediate Impacts of PALS: A School-Wide Multi-Level Programme Targeting Behaviour Problems in Elementary School. Scandinavian Journal of Educational Research 51(5):471-492.

Sterling RM, Barbetta PM, Heward WL, Heron TE (1997) A comparison of active student response and on-task instruction on the acquisition and maintenance of health facts by fourth grade special education students. Journal of Behavioral Education 7(2):151-165.

Stevens V, De Bourdeaudhuij I, Van Oost P (2000a) Bullying in Flemish schools: an evaluation of anti-bullying intervention in primary and secondary schools. British Journal of Educational Psychology 70(2):195-210.

Stevens V, Van Oost P, De Bourdeaudhuij I (2000b) The effects of an anti-bullying intervention programme on peers' attitudes and behaviour. Journal of Adolescence 23(1):21-34.

Stewart D, Sun J, Patterson C, Lemerle K, Hardie M (2004) Promoting and building resilience in primary school communities: evidence from a comprehensive 'health promoting school' approach. International Journal of Mental Health Promotion 6:26-33.

Sthair VL (1995) A Behavior Change Program To De-Escalate Fighting in Intermediate Grades of an Elementary School.

Suarez M (2000) Promoting social competence in deaf students: The effect of an intervention program. Journal of Deaf Studies and Deaf Education 5(4):323-336.

Taub J (2002) Evaluation of the Second Step Violence Prevention Program at a rural elementary school. School Psychology Review 31(2):186-200.

Taylor CA, Liang B, Tracy AJ, Williams LM, Seigle P (2002) Gender differences in middle school adjustment, physical fighting, and social skills: Evaluation of a social competency program. Journal of Primary Prevention 23(2):259-272.

Teglasi H, Rothman L (2001) A classroom-based program to reduce aggressive behavior. Journal of School Psychology 39(1):71-94.

Telljohann SK, Everett SA, Price JH (1997) Evaluation of a third grade sexual abuse curriculum. Journal of School Health 67(4):149-53.

Terzian MA (2007) Preventing aggressive behavior by promoting social informationprocessing skills: A theory-based evaluation of the making choices program. Dissertation Abstracts International Section A: Humanities and Social Sciences 68(4-A).

Thompson KL, Bundy KA, Broncheau C (1995) Social skills training for young adolescents: symbolic and behavioral components. Adolescence 30(119):724-34.

Thompson KL, Bundy KA, Wolfe WR (1996) Social skills training for young adolescents: cognitive and performance components. Adolescence 31(123):505-21.

Tillman YR (1995) Improving Social Skills in Second Graders through the Implementation of a Peace Education/Conflict Resolution Skills Curriculum.

Tissen I, Hergovich A, Spiel C (2007) School-based social training with and without dogs: Evaluation of their effectiveness. Anthrozoos 20(4):365-373.

Tran OTK (2008) Promoting social and emotional learning in schools: An investigation of massed versus distributed practice schedules and social validity of the strong kids curriculum in late elementary aged students. Dissertation Abstracts International Section A: Humanities and Social Sciences 68(8-A).

Truitt JC, Kalbfleisch PJ (1995) The Communication of Help-Seeking Skills to Children.

Tutty LM (1992) The ability of elementary school children to learn child sexual abuse prevention concepts. Child Abuse & Neglect 16(3):369-84.

Tutty LM (1997) Child sexual abuse prevention programs: evaluating Who Do You Tell. Child Abuse & Neglect 21(9):869-81.

Tutty LM (2000) What children learn from sexual abuse prevention programs: Difficult concepts and developmental issues. Research on Social Work Practice 10(3):275-300.

Twemlow SW, Biggs BK, Nelson TD, Vernberg EM, Fonagy P (2008) Effects of participation in a martial arts-based antibullying program in elementary schools. Psychology in the Schools 45(10):947-959.



Tynes-Jones JM (2007) A social skills program in third grade classrooms. Dissertation Abstracts International: Section B: The Sciences and Engineering 67(9-B).

Uchida H, Fujiwara Y, Nishigaki T, Sakuda H, Shinkai S (2008) The educational support program in a Japanese elementary school offered by the senior volunteers and its influence on children's school life and human relations. Gerontologist 48:276-276.

Ulutas I, Oemeroglu E (2007) The effects of an emotional intelligence education program on the emotional intelligence of children. Social Behavior and Personality 35(10):1365-1372.

Van Schoiack L (2000) Promoting social-emotional competence: Effects of a socialemotional learning program and corresponding teaching practices in the schools. Dissertation Abstracts International Section A: Humanities and Social Sciences 61(6-A).

Vincent DS (2004) The evaluation of a social-emotional intelligence program: Effects on fifth graders' prosocial and problem behaviors. Dissertation Abstracts International Section A: Humanities and Social Sciences 64(10-A).

Vogel J, Bowers C, Meehan C, Hoeft R, Bradley K (2004) Virtual reality for life skills education: program evaluation. Deafness & Education International 6(1):39-51.

Vreeman RC, Carroll AE (2007) A systematic review of school-based interventions to prevent bullying. Archives of Pediatrics and Adolescent Medicine 161(1):78-88.

Walker CE (2007) Teaching students to be peacemakers: Implementing a conflict resolution and peer mediation training in a Minneapolis K-6 charter school (Minnesota). Dissertation Abstracts International Section A: Humanities and Social Sciences 68(2-A).

Walsh-Bowers RT (1992) A creative drama prevention program for easing early adolescents' adjustment to school transitions. Journal of Primary Prevention 13(2):131-147.

Webster-Stratton C, Reid MJ (2008) Strengthening social and emotional competence in young children who are socioeconomically disadvantaged: Preschool and kindergarten school-based curricula. In: *Social competence of young children*: Risk, disability, and intervention. Brown WH, Odom SL, McConnell SR eds. Baltimore, MD, Paul H Brookes Publishing.

Webster-Stratton C, Reid MJ, Stoolmiller M (2008) Preventing conduct problems and improving school readiness: evaluation of the Incredible Years Teacher and Child Training Programs in high-risk schools. Journal of Child Psychology and Psychiatry 49(5):471-488.

Wiist WH, Jackson RH, Jackson KW (1996) Peer and community leader education to prevent youth violence. American Journal of Preventive Medicine 12(5 Suppl):56-64.



Williams CA (2006) The effects of social skills and media on student achievement in elementary school students. Dissertation Abstracts International Section A: Humanities and Social Sciences 67(1-A).

Wise KL, Bundy KA, Bundy EA, Wise LA (1991) Social skills training for young adolescents. Adolescence 26(101):233-41.

Yoshikawa H (1995) Long-Term Effects of Early Childhood Programs on Social Outcomes and Delinquency. Future of Children 5(3):51-75.

Zuckerman GA (1994) A pilot-study of a 10-day course in cooperative learning for beginning Russian first-graders. Elementary School Journal 94(4):405-420.

Zukauskas JA (1998) Improving cooperative behavior through the use of social skills instruction. Chicago, IL, Saint Xavier University.

• Intervention(s) examined was not based in a school setting (n=13)

Fiscian VS, Obeng EK, Goldstein K, Shea JA, Turner BJ (2009) Adapting a multifaceted U.S. HIV prevention education program for girls in Ghana. AIDS Education and Prevention 21(1):67-79.

Fraser MW, Day SH, Galinsky MJ, Hodges VG, Smokowski PR (2004) Conduct problems and peer rejection in childhood: a randomized trial of the making choices and strong families programs. Research on Social Work Practice 14(5):313-324.

Gifford SA (2001) Effects of after-school programs on the relationships among emotional regulation, behavior regulation, and social competence. Dissertation Abstracts International: Section B: The Sciences and Engineering 62(6-B).

Godley MD, Velasquez R (1998) Effectiveness of the Logan Square prevention project: Interim results. Drugs and Society 12(1-2):87-103.

Kumpfer KL, Molgaard V, Spoth R (1996) The Strengthening Families Program for the prevention of delinquency and drug use. In: Preventing childhood disorders, substance abuse, and delinquency. Peters RD, McMahon RJ. Thousand Oaks, CA, Sage Publications.

McCormick A, McKay MM, Marla, Wilson, McKinney L, Paikoff R, Bell C, Baptiste D, Coleman D, Gillming G, Madison S, Scott R (2000) Involving families in an urban HIV preventive intervention: How community collaboration addresses barriers to participation. AIDS Education and Prevention 12(4):299-307.



O'Donnell L, Stueve A, Agronick G, Wilson-Simmons R, Duran R, Jeanbaptiste V (2005) Saving Sex for Later: an evaluation of a parent education intervention. Perspectives on Sexual & Reproductive Health 37(4):166-73.

Park J, Kosterman R, Hawkins JD, Haggerty KP, Duncan TE, Duncan SC, Spoth R (2000) Effects of the "Preparing for the Drug Free Years" curriculum on growth in alcohol use and risk for alcohol use in early adolescence. Prevention Science 1(3):125-38.

Ruchross HS (1992) The Child and Family Options program - primary drug and alcohol prevention for young-children. Journal of Community Psychology:39-54.

Shuey DA, Babishangire BB, Omiat S, Bagarukayo H (1999) Increased sexual abstinence among in-school adolescents as a result of school health education in Soroti district, Uganda. Health Education Research 14(3):411-9.

Spoth RL, Redmond C, Shin C (2000) Reducing adolescents' aggressive and hostile behaviors: randomized trial effects of a brief family intervention 4 years past baseline. Archives of Pediatrics & Adolescent Medicine 154(12):1248-57.

Werch CE, Young M, Clark M, Garrett C, Hooks S, Kersten C (1991) Effects of a take-home drug prevention program on drug-related communication and beliefs of parents and children. Journal of School Health 61(8):346-50.

Wilkerson JM (1995) Training parents to teach their children about sexual abuse prevention: Parent and child outcomes. Dissertation Abstracts International Section A: Humanities and Social Sciences 56(2-A).

• Population targeted by the intervention(s) did not meet review criteria (n=50)

Allamani A, Forni E, Ammannati P, Basetti Sani I, Centurioni A (2000) Alcohol carousel and children's school drawings as part of a community educational strategy. Substance Use and Misuse 35(1 and 2):125-39.

Bastien S, Sango W, Mnyika KS, Masatu MC, Klepp KI (2008) Changes in exposure to information, communication and knowledge about AIDS among school children in Northern Tanzania, 1992-2005. AIDS Care 20(3):382-387.

Brown SM (1997) Design, implementation and evaluation of a pilot sexuality education curriculum for middle school special education students. Dissertation Abstracts International: Section B: The Sciences and Engineering 58(4-B).



Chandarana PC, Conlon P, Noh S, Field VA (1990) The impact of AIDS education among elementary school students. Canadian Journal of Public Health Revue Canadienne de Sante Publique. 81(4):285-9.

Coyle KK, Kirby DB, Marin BV, Gomez CA, Gregorich SE (2004) Draw the line/respect the line: a randomized trial of a middle school intervention to reduce sexual risk behaviors. American Journal of Public Health 94(5):843-51.

Deveaux L, Stanton B, Lunn S, Cottrell L, Yu S, Brathwaite N, Li X, Liu H, Marshall S, Harris C (2007) Reduction in human immunodeficiency virus risk among youth in developing countries. Archives of Pediatrics & Adolescent Medicine 161(12):1130-9.

Dukes RL, Ullman JB, Stein JA (1995) An evaluation of D.A.R.E. (Drug Abuse Resistance Education), using a Solomon Four-Group design with latent variables. Evaluation Review 19(4):409-435.

Eisen M, Zellman GL, Murray DM (2003) Evaluating the Lions-Quest "Skills for Adolescence" drug education program. Second-year behavior outcomes. Addictive Behaviors 28(5):883-97.

Gosin M, Hecht ML, Marsiglia FF (2003) Keepin' it R.E.A.L.: a drug resistance curriculum tailored to the strengths and needs of pre-adolescents of the southwest. Journal of Drug Education 33(2):119-142.

Hamalainen S, Keinanen-Kiukaanniemi S (1992) A controlled study of the effect of one lesson on the knowledge and attitudes of schoolchildren concerning HIV and Aids. Health Education Journal 51(3):135-8.

Howard NM, Horne AM, Jolliff D (2001) Self-efficacy in a new training model for the prevention of bullying in schools. Journal of Emotional Abuse 2(2-3):181-191.

Hunt C (2007) The effect of an education program on attitudes and beliefs about bullying and bullying behaviour in junior secondary school students. Child and Adolescent Mental Health 12 (1): 21-6.

Joffe V, Poulton H (2001) The joint delivery of personal and social education (PSE) in a special school. International Journal of Language & Communication Disorders 36 (Suppl):228-33.

Kinsler J, Sneed CD, Morisky DE, Ang A (2004) Evaluation of a school-based intervention for HIV/AIDS prevention among Belizean adolescents. Health Education Research 19(6):730-8.



Klepp KI, Ndeki SS, Seha AM, Hannan P, Lyimo BA, Msuya MH, Irema MN, Schreiner A (1994) AIDS education for primary school children in Tanzania: an evaluation study. AIDS 8(8):1157-62.

Komro KA, Perry CL, Murray DM, VeblenMortenson S, Williams CL, Anstine PS (1996) Peer-planned social activities for preventing alcohol use among young adolescents. Journal of School Health 66(9):328-334.

Komro KA, Perry CL, Veblen-Mortenson S, Farbakhsh K, Kugler KC, Alfano KA, Dudovitz BS, Williams CL, Jones-Webb R (2006) Cross-cultural adaptation and evaluation of a homebased program for alcohol use prevention among urban youth: the "Slick Tracy Home Team Program". Journal of Primary Prevention 27(2):135-54.

Komro KA, Perry CL, Veblen-Mortenson S, Williams CL, Roel JP (1999) Peer leadership in school and community alcohol use prevention activities. Journal of Health Education 30(4):202-211.

Lokanc-Diluzio W, Cobb H, Harrison R, Nelson A (2007) Building Capacity to Talk, Teach, and Tackle Sexual Health. The Canadian Journal of Human Sexuality 16(3-4):135-143.

Loveland-Cherry CJ (2005) Alcohol, children, and adolescents. Annual Review of Nursing Research 23:135-77.

Maticka-Tyndale E, Wildish J, Gichuru M (2007) Quasi-experimental evaluation of a national primary school HIV intervention in Kenya. Evaluation & Program Planning 30(2):172-86.

Moberg DP, Piper DL (1998) The Healthy for Life project: sexual risk behavior outcomes. AIDS Education & Prevention 10(2):128-48.

Obasi AI, Cleophas B, Ross DA, Chima KL, Mmassy G, Gavyole A, Plummer ML, Makokha M, Mujaya B, Todd J, Wight D, Grosskurth H, Mabey DC, Hayes RJ (2006) Rationale and design of the MEMA kwa Vijana adolescent sexual and reproductive health intervention in Mwanza Region, Tanzania. AIDS Care 18(4):311-22.

Olweus D (1993) Bully/victim problems among schoolchildren: Long-term consequences and an effective intervention program. In: *Mental disorder and crime*. Hodgins S ed. Thousand Oaks, CA, Sage Publications.

Perry CL, Williams CL, Forster JL, Wolfson M, Wagenaar AC, Finnegan JR, McGovern PG, Veblen-Mortenson S, Komro KA, Anstine PS (1993) Background, conceptualization and design of a community-wide research program on adolescent alcohol use: Project Northland. Health Education Research 8(1):125-36.



Perry CL, Williams CL, Veblen-Mortenson S, Toomey TL, Komro KA, Anstine PS, McGovern PG, Finnegan JR, Forster JL, Wagenaar AC, Wolfson M (1996) Project Northland: outcomes of a communitywide alcohol use prevention program during early adolescence. American Journal of Public Health 86(7):956-65.

Phuphaibul R, Thanooruk R, Leucha Y, Sirapo-ngam Y, Kanobdee C (2005) The impacts of the "Immune of Life" for teens module application on the coping behaviors and mental health of early adolescents. Journal of Pediatric Nursing 20(6):461-8.

Reddy KS, Arora M, Perry CL, Nair B, Kohli A, Lytle LA, Stigler M, Prabhakaran D (2002) Tobacco and alcohol use outcomes of a school-based intervention in New Delhi. American Journal of Health Behavior 26(3):173-81.

Ringwalt CL, Greene JM, Ennett ST, Iachan R, Clayton RR, Leukefeld CG (1994) Past and Future Directions of the D.A.R.E. Program: An Evaluation Review. Draft Final Report. Durham, NC, Research Triangle Institute.

Ross DA, Changalucha J, Obasi AI, Todd J, Plummer ML, Cleophas-Mazige B, Anemona A, Everett D, Weiss HA, Mabey DC, Grosskurth H, Hayes RJ (2007) Biological and behavioural impact of an adolescent sexual health intervention in Tanzania: a community-randomized trial. AIDS 21(14):1943-55.

Shope JT, Copeland LA, Kamp ME, Lang SW (1998) Twelfth grade follow-up of the effectiveness of a middle school-based substance abuse prevention program. Journal of Drug Education 28(3):185-97.

Shope JT, Copeland LA, Marcoux BC, Kamp ME (1996) Effectiveness of a school-based substance abuse prevention program, in Journal of Drug Education, 26(4):323-37.

Shope JT, Kloska DD, Dielman TE, Maharg R (1994) Longitudinal evaluation of an enhanced alcohol misuse prevention study (AMPS) curriculum for grades six-eight. Journal of School Health 64(4):160-6.

Snow DL, Tebes JK, Arthur MW, Tapasak RC (1992) Two-year follow-up of a social-cognitive intervention to prevent substance use. Journal of Drug Education 22(2):101-14.

Stigler MH, Kugler KC, Komro KA, Leshabari MT, Klepp KI (2006) AIDS education for Tanzanian youth: a mediation analysis. Health Education Research 21(4):441-451.

Van Schoiack-Edstrom L, Frey KS, Beland K (2002) Changing adolescents' attitudes about relational and physical aggression: An early evaluation of a school-based intervention. School Psychology Review 31(2):201-216.



Vicary JR, Henry KL, Bechtel LJ, Swisher JD, Smith EA, Wylie R, Hopkins AM (2004) Life skills training effects for high and low risk Rural Junior High School Females. Journal of Primary Prevention 25(4):399-416.

Webster-Stratton C, Reid MJ, Hammond M (2001) Preventing conduct problems, promoting social competence: a parent and teacher training partnership in head start. Journal of Clinical Child Psychology 30(3):283-302.

Werch CE, Carlson JM, Owen DM, DiClemente CC, Carbonari JP (2001) Effects of a stagebased alcohol preventive intervention for inner-city youth. Journal of Drug Education 31(2):123-38.

Werch CE, Carlson JM, Pappas DM, DiClemente CC (1996) Brief nurse consultations for preventing alcohol use among urban school youth. Journal of School Health 66(9):335-339.

Werch CE, Owen DM, Carlson JM, DiClemente CC, Edgemon P, Moore M (2003) One-year follow-up results of the STARS for Families alcohol prevention program. Health Education Research 18(1):74-87.

Werch CE, Pappas DM, Carlson JM, DiClemente CC (1998) Short and long-term effects of a pilot prevention program to reduce alcohol consumption. Substance Use and Misuse 33(11):2303-21.

Werch CE, Pappas DM, Carlson JM, Edgemon P, Sinder JA, DiClemente CC (2000) Evaluation of a brief alcohol prevention program for urban school youth. American Journal of Health Behavior 24(2):120-132.

Williams C, Griffin KW, Macaulay AP, West TL, Gronewold E (2005) Efficacy of a drug prevention CD-ROM intervention for adolescents. Substance Use & Misuse 40(6):869-78.

Williams CL, Perry CL, Dudovitz B, Veblen-Mortenson S, Anstine PS, Komro KA, Toomey TL (1995) A home-based prevention program for sixth-grade alcohol use: Results from project Northland. The Journal of Primary Prevention 16(2):125-147.

Williams CL, Perry CL, Farbakhsh K, Veblen-Mortenson S (1999) Project Northland: comprehensive alcohol use prevention for young adolescents, their parents, schools, peers and communities. Journal of Studies on Alcohol - Supplement 13:112-24.

Wurtele SK, Gillispie EI, Currier LL, Franklin CF (1992a) A comparison of teachers vs. parents as instructors of a personal safety program for preschoolers. Child Abuse & Neglect 16(1):127-37.



Wurtele SK, Kast LC, Melzer AM (1992b) Sexual abuse prevention education for young children: a comparison of teachers and parents as instructors. Child Abuse & Neglect 16(6):865-76.

Yampolskaya S, Brown EC, Vargo AC (2004) Assessment of Teen Pregnancy Prevention Interventions Among Middle School Youth. Child & Adolescent Social Work Journal 21:69-83.

Zhai F (2008) Effects of Head Start on the outcomes of participants. Dissertation Abstracts International Section A: Humanities and Social Sciences 69(1-A).



Appendix 4. References to studies not available for assessment

Not available for full text screening (n=44)

Angelone LA (2008) Increasing empathy in middle school youth: Effects of social skills training on sixth grade students. Dissertation Abstracts International Section A: Humanities and Social Sciences 68(11-A).

Anon (1995) Initiatives. Mauritius: a right to know. MFPA launches sex education in primary schools. Africa Link: a Publication of the Africa Region:13.

Anon (1999) Harvard Education Letter, 1999. Harvard Education Letter 15(1-6):50.

Battistich V (2003) Effects of a school-based program to enhance prosocial development on children's peer relations and social adjustment. Journal of Research in Character Education 1:1-17.

Battistich V, Watson M, Solomon D, Schaps E, Solomon J (1991) The Child Development Project: A comprehensive program for the development of prosocial character. In: *Handbook of moral behavior and development, Vol. 1: Theory; Vol. 2: Research; Vol. 3: Application.* Kurtines WM, Gewirtz JL eds. Hillsdale, NJ, L Erlbaum.

Blau GM (1990) Drug and alcohol use prevention: Utility and effectiveness with rural fifthgrade students. Dissertation Abstracts International 50(8-A).

Cox SE (1996) Family life education and child's perception of self, others, and family. Dissertation Abstracts International Section A: Humanities and Social Sciences 56(8-A).

Dorpat N (1994) PRIDE: Substance abuse education intervention program. American Indian and Alaska Native Mental Health Research 4(Mono):122-133.

Egypto AC, Pinto MC, Bock SD (1996) Brazilian organization develops "sexual guidance" programs defined by long-term communication. Siecus Report 24(3):16-7.

Finger WR (1993) Seeking better ways to teach youth about AIDS. Network International Communications in Library Automation 14(2):16-9.

Flannery DJ, Torquati J (1993) An elementary school substance abuse prevention program: teacher and administrator perspectives. Journal of Drug Education 23(4):387-97.

Gamache D, Snapp S (1995) Teach your children well: Elementary schools and violence prevention. In: *Ending the cycle of violence: Community responses to children of battered women.* Peled E, Jaffe PG, Edleson JL eds. Thousand Oaks, CA, Sage Publications, Inc.

Green J (1997) A survey of sex education in primary schools in the Northern and Yorkshire regions. International Journal of Health Education, Manchester 35:81-86.

Gruber EA (2008) Effects of social and emotional skills training on sixth grade students' knowledge of prosocial skills and their attitude toward violence. Dissertation Abstracts International Section A: Humanities and Social Sciences 68(11-A).

Hahn EJ, Hall LA, Simpson MR (1998) Drug prevention with high risk families and young children. Journal of Drug Education 28(4):327-45.

Hollander C (1997) A lesson program for schoolchildren about a clean and healthy life-style: a pilot study. Vibro (90):1-7.

Hudson DL (1991) Develop and Implement a Peace Education Curriculum for Elementary School Students through a Planned Program of Instruction. Fort Lauderdale, FL, Nova Southeastern University.

Hughes D, Titman C (2003) Healthy Living Program: whole school approach at North Shore Primary School. Health Education Australia 3(2):33-36.

Hunt J (1996) The access to and success in presenting mainstream curriculum to intellectually disabled adolescents. Parkville-Victoria, University of Melbourne.

Hurry J, Lloyd C (1997) A follow up evaluation of Project Charlie: a life skills drug education programme for primary schools. London, Home Office Central Drugs Prevention Unit.

Ilika AL, Obionu CO (2002) Personal hygiene practice and school-based health education of children in Anambra State, Nigeria. Nigerian Postgraduate Medical Journal 9(2):79-82.

Jamison J, Ashby P, Hamilton K, Lewis G, MacDonald A, Saunders L (1998) The health promoting school: final report of the ENHPS evaluation project in England. London, Health Education Authority.

Jenkins NF (1995) The effectiveness of Body Basics Training on the socialization component of adaptive behavior of mildly intellectually disabled students. Dissertation Abstracts International: Section B: The Sciences and Engineering 56(3-B).

Kearney DS (1990) A comparative study of two child sexual abuse prevention programs for five-year-old kindergarten children. Dissertation Abstracts International 50(12-A, Pt 1).

Keshishian AI (1993) The effects of a relationship-oriented intervention on elementary school children. Dissertation Abstracts International 53(7-B).

Kim S, McLeod JH, Shantzis C (1990) A short-term outcome evaluation of the "I'm Special" drug abuse prevention program: a revisit using SCAT inventory. Journal of Drug Education 20(2):127-38.

Kirby D (1999) Sexuality and sex education at home and school. Adolescent Medicine State of the Art Reviews 10(2):195-209.



La Londe SK (2004) The effects of a school-based violence prevention program on children's social and emotional behavior. Dissertation Abstracts International: Section B: The Sciences and Engineering 65(5-B).

Larsen L (2004) Not in my classroom: a response to the Values Education Study. Critical and Creative Thinking 12:43-52.

Mackay MA (1995) An evaluation of a school-based drug prevention program at the elementary level. Dissertation Abstracts International: Section B: The Sciences and Engineering 56(1-B).

Mackereth C, Scott H, Ellins A, Brazier C (1998) Sexual health messages: working with young people. Community Practitioner 71(12):412-414.

Meyer AL (1995) The effectiveness of a peer-led positive youth development program for sixth graders. Dissertation Abstracts International: Section B: The Sciences and Engineering 55(9-B).

Oroge SA (1993) NERDC: implementing population family life education. Nigeria's Population:22-3.

Raynal ME, Chen WW (1997) Evaluation of a drug prevention program for young high risk students. International Quarterly of Community Health Education 16(2):187-195.

Rosenbluth B, Whitaker DJ, Sanchez E, Valle LA (2004) The Expect Respect project: preventing bullying and sexual harassment in US elementary schools. In: *Bullying in schools: How successful can interventions be?* Smith PK, Pepler DJ, Rigby K eds. New York, Cambridge University Press.

Sames JA (1993) Events and relationships: A case study of the process in a social skills group to promote social competency in elementary school children. Dissertation Abstracts International 54(2-B):1152-1153.

Schmidt JJ, Shields CW (1998) Integration of Guidance Lessons Using Invitational Concepts in a Friendship Curriculum. Journal of Invitational Theory and Practice 5(2):107-22.

Scott AL, Brabham LH (1993) Wellness on a Shoestring. Principal 73(2):21-23.

Shulman DM (2001) Using psychological theories to teach children sexual abuse prevention skills: The safety through education and practice program. Dissertation Abstracts International: Section B: The Sciences and Engineering 61(10-B).

Silbert LB, Silbert AJ (1991) S+T+R+O+N+G Kids Life Skills Program: Levels K-5. Publication source unknown.



Simons CW (1991) Effects of a sexual assault prevention program on the behavior of elementary school children. Dissertation Abstracts International 51(8-B).

Tait C, McMahon B, Montgomery K (2004) Stick and stones: queer sexualities in schools. Teacher Learning Network 11:18-19.

Werch CE, Pappas DM, Carlson JM, DiClemente CC (1999) Six-month outcomes of an alcohol prevention program for inner-city youth. American Journal of Health Promotion 13(4):237-40.

Young M, Kersten C, Werch C (1996) Evaluation of a parent child drug education program. Journal of Drug Education 26(1):57-68.

• Foreign language papers (n=17)

Ayotte V, Laurendeau MC (1998) [Effectiveness of an educational intervention in improving the self concept of second to sixth grade students]. Canadian Journal of Public Health Revue Canadienne de Sante Publique. 89(3):176-80.

Bowi U, Ott G, Tress W (2008) [Without fists--violence prevention in primary school]. Praxis der Kinderpsychologie und Kinderpsychiatrie 57(7):509-20.

Felizari GM (1990) [School nursing and sex education for adolescents]. Revista Gaucha de Enfermagem 11(2):12-9.

Hansen ML, Bulow HH, Naess B (1990) [Condoms and adolescents. A prospective study of a new method of teaching]. Ugeskrift for Laeger 152(35):2478-80.

Hausser D (1993) [The AIDS problem in Burundi and its prevention among young people]. Sozial- und Praventivmedizin 38(6):398-400.

Hollederer A, Bolcskei PL (2001) [Care and preventive experts lecture in the elementary school programme "Klasse2000" - co-operative teaching]. Gesundheitswesen 63(10):619-24.

Krannich S, Sanders M, Ratzke K, Diepold B, Cierpka M (1997) [FAUSTLOS--a curriculum for promoting social competence and preventing aggressive and violent behavior in children]. Praxis der Kinderpsychologie und Kinderpsychiatrie 46(3):236-47.

Morstadt E, Pfaff G, Schwarz J, Potschke-Langer M (1991) [Design of and involvement in an interventional study on health education in basic schools: the Pforzheim Study]. Sozial- und Praventivmedizin 36(3):147-53.

Moscicka M, Przewlocka T (1995) [Evaluation of "the ever changing" educational program]. Roczniki Panstwowego Zakladu Higieny 46(4):417-26.



No TS, Park JS (1999) [The effect of health promotion education on the health behavior performance of elementary school students]. Korean Nurse 38:87-99.

Pellai A, Castelli B, Scyslowska G, Bassoli L, Lanzi M, Pazardjiklian I, Sacchetti A (2003) [Child sexual abuse primary prevention: outcome evaluation of a health education project implemented in Milan's elementary schools]. Annali di Igiene 15(5):529-39.

Saito M, Yamagishi Y, Saito S, Edo K (2001) [Medical education for students in compulsory education: the conception of the preparation of three graded textbooks and preliminary evaluation of lectures using these textbooks]. Yakugaku Zasshi - Journal of the Pharmaceutical Society of Japan 121(3):247-52.

Schick A, Cierpka M (2006) [Evaluation of the Faustlos-Curriculum for Kindergartens]. Praxis der Kinderpsychologie und Kinderpsychiatrie 55(6):459-74.

Sprober N, Schlottke PF, Hautzinger M (2006) [ProACT + E: A programme to prevent bullying in schools and to increase the positive development of students. Evaluation of a school-based, universal, primary preventive programme for secondary schools that includes teachers, students and parents]. Zeitschrift fur Klinische Psychologie und Psychotherapie 35(2):140-50.

Verban Z, Belovic B, Benkovic A (1999) [Quiz about AIDS -- an interesting form of health education]. Obzornik Zdravstvene Nege 33(3/4):175-179.

Villalbi JR, Auba J, Garcia Gonzalez A (1993) [Results of a school addictive substances abuse prevention program: the Barcelona PASE pilot project]. Gaceta Sanitaria 7(35):70-7.

Zivkovic M, Bjegovic V, Vukovic D, Marinkovic J (1998) [Evaluation of the effect of the health education intervention project "Healthy School"]. Srpski Arhiv Za Celokupno Lekarstvo 126(5-6):164-70.

• Conference abstracts (full text not available) (n=10)

Bierman KL, Coie JD, Dodge KA, Greenberg MT, Lochman JE, McMahon RJ, Pinderhughes EE (2000) Merging universal and indicated prevention programs: The Fast Track model. Paper presented at the International Symposium on Addictions 2000, Hyannis, Massachusetts.

Dedobbeleer N, Desjardins S (1998) Outcomes of an ecological and participatory approach to prevent alcohol and other drug "abuse" among multiethnic adolescents. Paper presented at the Meeting of the Canadian Public Health Association, Montreal, Canada.

Enah C (2004) An intervention to increase intentions and behavioral skills to postpone sexual activity in Cameroonian preadolescent female students. Paper presented at the

Midwest Nursing Research Society 28th Annual Research Conference, St Louis, Missouri, USA.

Frey KS, Nolen SB, Edstrom LVS, Hirschsten MK (2001) Effects of a school-based socialemotional competence program: Linking children's goals, attributions, and behavior. Paper presented at the 9th Annual Meeting of the Society for Prevention Research, Washington, DC.

Furr-Holden CDM, Ialongo NS, Anthony JC, Petras H, Kellam SG (2003) The distal impact of two 1st grade randomized preventive interventions on drug sampling by youths: recent evidence from the Baltimore prevention program. Paper presented at the 65th Annual Scientific Meeting of the College on Problems of Drug Dependence, Bal Harbour, Florida, USA.

Grossman DC, Neckerman HJ, Koepsell TD, Liu PY, Asher KN, Beland K, Frey K, Rivara FP (1995) Effectiveness of a violence prevention curriculum among children in elementary school - A randomized controlled trial. Paper presented at the Annual Meeting of the Ambulatory Pediatric Association, Washington, DC.

Hawkins JD, Catalano RF, Kosterman R, Abbott R, Hill KG (1996) Preventing adolescent health-risk behaviors by strengthening protection during childhood. Paper presented at the Meeting of the Society for Life History Research, London, England.

Leff SS, Costigan T, Power TJ (2001) Using participatory research to develop a playgroundbased prevention program. Paper presented at the 109th Annual Convention of the American Psychological Association, San Francisco, California.

Linares LO, Rosbruch N, Stern MB, Edwards ME, Walker G, Abikoff HB, Alvir JMJ (2003) Developing cognitive-social-emotional competencies to enhance academic learning. Paper presented at the Biennial Meeting of the Society for Research in Child Development, Tampa, Fl.

Tutty LM (1998) What children learn from sexual abuse prevention programs: Difficult concepts and developmental issues. Paper presented at the International Conference on Research for Social Work Practice, Miami, Florida.



Appendix 5. Results of the quality assessment

Table 8.1. Quality assessment: Systematic reviews and meta-analyses

- 1.1 The study addresses an appropriate and clearly focused question
- 1.2 A description of the methodology used is included
- 1.3 The literature search was sufficiently rigorous to identify all relevant studies
- 1.4 Study quality is assessed and taken into account
- 1.5 There are enough similarities between the studies selected to make combining them reasonable

Key: ++ Well covered +Adequately addressed - Poorly addressed × Not addressed N/A Not applicable

Reference(s)		Coding				
	1.1	1.2	1.3	1.4	1.5	County
Gottfredson & Wilson, 2003	++	++	+	++	+	++
Spoth et al., 2008	++	++	++	++	+	++

Table 8.2. Quality assessment: Randomised controlled trials

	Abbey, 1990	Austin & Johnson, 1995 ^a	Bell et al., 2005a	Bell et al., 2007	Bohman et al., 2004	Bühler et al., 2008	Catalano et al., 2003	Corbin et al., 1993 ^ª	Cruz & Dunn, 2003
Section 1: Population									
1.1 Is the source population or source area well described?	NR	NR	NR	NR	NR	+	NR	+	NR
1.2 Is the eligible population or area representative of the source population or area?	NR	NR	NR	+	NR	-	NR	NR	NR
1.3 Do the selected participants or areas represent the eligible population or area?	+	NR	NR	NR	NR	++	NR	-	NR
Section 2: Method of allocation									
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	+	+	+	+	+	+	NR	+	+
2.2 Were interventions (and comparisons) well described and appropriate?	++	++	+	+	++	++	+	+	++
2.3 Was the allocation concealed?	NR	NR	NR	NR	NR	NR	NR	NR	NR
2.4 Were participants and/or investigators blind to exposure and comparison?	NR	NR	NR	NR	NR	NR	NA	NR	NR
2.5 Was the exposure to the intervention and comparison adequate?	NR	NA	++	NR	NR	NR	++	-	-
2.6 Was contamination acceptably low?	NR	NA	NR	NR	NR	NR	NA	NR	NR
2.7 Were other interventions similar in both groups?	NR	NR	NA	NR	NR	NR	NA	-	NA
2.8 Were all participants accounted for at study conclusion?	++	-	NR	++	++	-	++	-	NR
2.9 Did the setting reflect usual UK practice?	NR	NR	NR	NR	NR	NR	NR	NR	NR
2.10 Did the intervention or control comparison reflect usual UK practice?	NR	NR	NR	NR	NR	NR	NR	NR	NR
Section 3: Outcomes									
3.1 Were outcome measures reliable?	+	+	+	+	NR	++	++	++	++
3.2 Were all outcome measurements complete?	NR	NR	NR	NR	NR	++	NA	++	NR
3.3 Were all important outcomes assessed?	++	++	+	+	++	++	++	++	+

	Abbey, 1990	Austin & Johnson, 1995 ^a	Bell et al., 2005a	Bell et al., 2007	Bohman et al., 2004	Bühler et al., 2008	Catalano et al., 2003	Corbin et al., 1993 ^ª	Cruz & Dunn, 2003
3.4 Were outcomes relevant?	++	-	++	++	++	++	++	++	++
3.5 Were there similar follow-up times in exposure and comparison groups?	+	++	++	NR	+	NR	++	NR	NR
3.6 Was follow-up time meaningful?	+	-	-	-	+	-	++	-	-
Section 4: Analyses			•						
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	++	NR	+	+	++	+	+	NR	++
4.2 Was Intention to treat (ITT) analysis conducted?	NA	NR	NA	NA	NR	NA	NA	NR	NR
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	NR	++	NR	NR	NR	++	NR	-	++
4.4 Were the estimates of effect size given or calculable?	NR	NR	NR	++	NR	++	NR	-	++
4.5 Were the analytical methods appropriate?	++	+	+	+	+	++	++	+	++
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	++	+	+	+	+	++	+	+	++
Section 5: Summary	Section 5: Summary								
5.1 Are the study results internally valid (i.e. unbiased)?	+	+	+	-	+	+	+	-	-
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	-	-	-	-	-	+	-	-	-
^a RCT based on randomisation at the individual level NR – not reported; NA – not applicable									

Table 8.3. Quality assessment: Randomised controlled trials continued

	Godbold, 1999	Hecht et al., 2008	lalongo et al., 1999	Kellam et al., 2008	Kraus et al., 1994	Pick et al., 2007	Rollin et al., 1992; 1995	Schonfeld et al., 1995	van Lier et al., 2009
Section 1: Population					• •		•		
1.1 Is the source population or source area well described?	+	NR	NR	+	NR	+	NR	+	NR
1.2 Is the eligible population or area representative of the source population or area?	++	NR	NR	++	NR	NR	NR	+	NR
1.3 Do the selected participants or areas represent the eligible population or area?	NR	+	NR	++	+	NR	NR	+	NR
Section 2: Method of allocation									
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	+	+	++	+	+	+	-	+	+
2.2 Were interventions (and comparisons) well described and appropriate?	+	++	++	++	++	+	+	++	++
2.3 Was the allocation concealed?	NR	NR	++	+	NR	NR	-	NR	+
2.4 Were participants and/or investigators blind to exposure and comparison?	NR	NA	NA	NA	NR	NR	NA	+	NA
2.5 Was the exposure to the intervention and comparison adequate?	-	++	++	NR	++	+	+	+	++
2.6 Was contamination acceptably low?	NR	NR	NA	++	NR	++	NR	NA	++
2.7 Were other interventions similar in both groups?	-	++	NA	++	NR	NR	NR	NA	NR
2.8 Were all participants accounted for at study conclusion?	NR	+	+	+	+	-	-	+	++
2.9 Did the setting reflect usual UK practice?	NR	NR	NR	NR	NR	NR	NR	NR	NR
2.10 Did the intervention or control comparison reflect usual UK practice?	NR	NR	NR	NR	NA	NR	NR	NR	NR
Section 3: Outcomes									
3.1 Were outcome measures reliable?	+	+	++	++	+	++	+	++	NR
3.2 Were all outcome measurements complete?	NR	+	++	++	NA	++	+	NR	++
3.3 Were all important outcomes assessed?	+	+	++	++	++	++	+	++	++
3.4 Were outcomes relevant?	+	++	++	++	++	++	+	++	++
	Godbold, 1999	Hecht et al., 2008	lalongo et al., 1999	Kellam et al., 2008	Kraus et al., 1994	Pick et al., 2007	Rollin et al., 1992; 1995	Schonfeld et al., 1995	van Lier et al., 2009
-------------------------------------------------------------------------------------------	------------------	-----------------------	-------------------------	------------------------	-----------------------	----------------------	------------------------------	------------------------	--------------------------
3.5 Were there similar follow-up times in exposure and comparison groups?	++	++	++	++	++	NR	+	+	++
3.6 Was follow-up time meaningful?	-	+	+	++	+	++	-	+	++
Section 4: Analyses									
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	NR	+	+	++	++	NR	-	++	NR
4.2 Was Intention to treat (ITT) analysis conducted?	NA	NR	++	++	NA	NA	NR	NA	++
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	+	NR	NR	NR	NR	NR	NR	NR	NR
4.4 Were the estimates of effect size given or calculable?	NR	+	_	++	NR	++	-	NR	+
4.5 Were the analytical methods appropriate?	+	++	++	++	+	++	+	++	++
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	+	++	-	++	+	++	-	+	+
Section 5: Summary									
5.1 Are the study results internally valid (i.e. unbiased)?	-	+	++	++	+	+	-	+	++
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	+	-	-	+	+	-	-	+	-
NR – not reported; NA – not applicable									

Table 8.4. Quality assessment: Non-randomised controlled trials

	Ambtman et al., 1990	Battistich et al., 2000	Denny & Young, 2006	Hahn et al., 2007	Hawkins et al., 1999	Holtz & Twombly, 2007	Masterpaqua et al., 1992	Peterson & Woodward, 1993
Section 1: Population								
1.1 Is the source population or source area well described?	NR	NR	NR	NR	+	+	NR	NR
1.2 Is the eligible population or area representative of the source population or area?	++	NR	NR	+	+	NR	NR	NR
1.3 Do the selected participants or areas represent the eligible population or area?	++	NR	NR	+	+	NR	NR	NR
Section 2: Method of allocation								
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	+	+	-	+	-	-	-	NR
2.2 Were interventions (and comparisons) well described and appropriate?	+	+	+	+	++	+	+	-
2.3 Was the allocation concealed?	NA	-	-	NR	-	NR	NR	NR
2.4 Were participants and/or investigators blind to exposure and comparison?	NR	NA	NA	NA	NR	NR	NR	NA
2.5 Was the exposure to the intervention and comparison adequate?	-	-	NR	NR	++	++	++	NR
2.6 Was contamination acceptably low?	NA	NR	+	++	NR	NR	NA	NR
2.7 Were other interventions similar in both groups?	NR	NR	NR	++	NR	NR	NR	NR
2.8 Were all participants accounted for at study conclusion?	++	NR	-	-	++	NR	NR	NR
2.9 Did the setting reflect usual UK practice?	NR	NR	NR	NR	NR	NR	NR	NR
2.10 Did the intervention or control comparison reflect usual UK practice?	NR	NR	NR	NR	NR	NR	NR	NR
Section 3: Outcomes								
3.1 Were outcome measures reliable?	-	NR	+	++	NR	+	+	++
3.2 Were all outcome measurements complete?	+	NR	+	++	++	NR	NR	+

	Ambtman et al., 1990	Battistich et al., 2000	Denny & Young, 2006	Hahn et al., 2007	Hawkins et al., 1999	Holtz & Twombly, 2007	Masterpaqua et al., 1992	Peterson & Woodward, 1993
3.3 Were all important outcomes assessed?	+	++	+	++	++	+	+	-
3.4 Were outcomes relevant?	+	++	+	++	++	+	++	+
3.5 Were there similar follow-up times in exposure and comparison groups?	-	+	+	++	++	+	NR	NR
3.6 Was follow-up time meaningful?	-	-	+	+	++	-	NR	-
Section 4: Analyses			•		•	•		
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	NR	+	NR	+	++	+	-	NR
4.2 Was Intention to treat (ITT) analysis conducted?	NA	NA	NA	NR	NR	NA	NA	NR
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	NR	NR	NR	NR	NR	NR	NR	NR
4.4 Were the estimates of effect size given or calculable?	NR	+	NR	+	++	NR	NR	-
4.5 Were the analytical methods appropriate?	+	+	+	++	++	+	+	+
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	+	+	NR	++	++	+	+	-
Section 5: Summary					-			
5.1 Are the study results internally valid (i.e. unbiased)?	+	+	-	+	+	-	-	-
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	+	_	-	-	+	-	_	_
NR – not reported; NA – not applicable								

Table 8.5. Quality assessment: Controlled before and after studies

	Andrews, 1992	Baker, 2004	Bell et al., 2005b	Flay et al., 2003	Kreutter & Gewirtz, 1991	Raybuck & Hicks, 1994	Spear et al., 1997	Stevens et al., 1996	Wright, 2007	Young et al., 1997
Section 1: Population				•		•		•		
1.1 Is the source population or source area well described?	NR	++	+	NR	NR	NR	NR	+	+	NR
1.2 Is the eligible population or area representative of the source population or area?	NR	+	NR	NR	NR	NR	NR	+	NR	NR
1.3 Do the selected participants or areas represent the eligible population or area?	NR	-	NR	NR	NR	-	NR	NR	NR	NR
Section 2: Method of allocation										
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	+	NR	+	NA	+	+	NR	NA	NA	NR
2.2 Were interventions (and comparisons) well described and appropriate?	-	++	NR	+	-	-	+	-	++	-
2.3 Was the allocation concealed?	NA	NR	NR	NA	NA	NA	NR	+	NA	NR
2.4 Were participants and/or investigators blind to exposure and comparison?	NR	NR	NR	NA	NR	NA	NA	+		NR
2.5 Was the exposure to the intervention and comparison adequate?	NR	+	++	NR	NR	NR	NR	NR	NR	+
2.6 Was contamination acceptably low?	NA	NR	NR	NA	NR	NA	NR	NA	-	NA
2.7 Were other interventions similar in both groups?	NR	NR	NR	-	NR	NR	NR	NA	-	NA
2.8 Were all participants accounted for at study conclusion?	-	NR	NR	NA	NR	-	-	++	-	NR
2.9 Did the setting reflect usual UK practice?	NR	NR	NR	NR	NR	NR	NR	NR	NR	NR
2.10 Did the intervention or control comparison reflect usual UK practice?	NR	NR	NR	NR	NR	NR	NR	NR	NR	NR
Section 3: Outcomes										
3.1 Were outcome measures reliable?	NR	+	+	+	NR	+	++	NR	NR	++
3.2 Were all outcome measurements complete?	NR	NR	NR	+	NA	NR	NR	NA	+	++
3.3 Were all important outcomes assessed?	++	-	+	+		-	+	++	+	++

3.4 Were outcomes relevant?	++	+	+	+	++	+	++	++	++	++
3.5 Were there similar follow-up times in exposure and comparison groups?	NR	NA	+	+	+	NR	NR	++	NR	++
3.6 Was follow-up time meaningful?	-	+	++	+	-	-	-	++	++	++
Section 4: Analyses										
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	NR	-	++	+	NR	-	NR	NR	NR	NR
4.2 Was Intention to treat (ITT) analysis conducted?	NA	NA	NA	NA	NA	NA	NR	NA	NA	NA
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	NR	+	NR							
4.4 Were the estimates of effect size given or calculable?	NR	NR	NR	+	NR	NR	NR	NR	NR	NR
4.5 Were the analytical methods appropriate?	-	++	++	+	+	+	+	++	+	+
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	+	+	++	+	++	+	+	++	+	++
Section 5: Summary										
5.1 Are the study results internally valid (i.e. unbiased)?	-	-	-	+	-	-	-	-	-	-
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	_	-	-	_	_	_	-	+	_	+
NR – not reported; NA – not applicable										

Table 8.6. Quality assessment: Uncontrolled before and after studies

	Abel & Greco, 2008	Gamble & Burgess, 1994	Gaskins et al., 2002	Hall-Long & Dishop, 1999	Paxton et al., 1998	Starkey & Orme, 2001	Tudor- Smith et al., 1995	Utley et al., 2001	Wackett et al., 2000	Welham, 2007	Witt & Witt, 1995
Section 1: Population				·		•	•				
1.1 Is the source population or source area well described?	NR	NR	+	+	-	NR	-	++	NR	NR	NR
1.2 Is the eligible population or area representative of the source population or area?	NR	NR	NR	NR	NR	NR	NR	NR	NR	NR	NR
1.3 Do the selected participants or areas represent the eligible population or area?	NR	NR	NR	NR	NR	-	NR	-	NR	NR	NR
Section 2: Method of allocation				-							
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	NA	NA	NA	NA	NA	NA	NA	NA	NR	NA	NA
2.2 Were interventions (and comparisons) well described and appropriate?	+	+	_	++	-	NA	-	+	+	+	+
2.3 Was the allocation concealed?	NA	NA	NA	NA	NA	NA	NA	NA	NR	NA	NA
2.4 Were participants and/or investigators blind to exposure and comparison?	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2.5 Was the exposure to the intervention and comparison adequate?	NA	NA	NR	NR	NR	NA	-	+	++	NR	NR
2.6 Was contamination acceptably low?	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2.7 Were other interventions similar in both groups?	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2.8 Were all participants accounted for at study conclusion?	1	NR	-	NR	+	+	-	++	++	NA	_
2.9 Did the setting reflect usual UK practice?	NR	NR	NR	NR	NR	NR	NR	NR	NR	NR	NR
2.10 Did the intervention or control comparison reflect usual UK practice?	NR	NR	NR	NR	NR	NR	NR	NR	NR	NR	NR
Section 3: Outcomes		-		•		•					
3.1 Were outcome measures reliable?	+	NR	++	NR	NR	-	+	+	-	NR	NR
3.2 Were all outcome measurements complete?	NA	NR	+	NR	NR	+	NR	++	++	NA	NR
3.3 Were all important outcomes assessed?	NA	-	+	+	-	-	+	++	-	NA	+

3.4 Were outcomes relevant?	NA	+	+	++	-	+	+	++	+	NA	+
3.5 Were there similar follow-up times in exposure and comparison groups?	NA	++	NA	NA							
3.6 Was follow-up time meaningful?	-	-	-	NR	-	-	+	-	+	NR	-
Section 4: Analyses											
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	NA	NR	NA	NA							
4.2 Was Intention to treat (ITT) analysis conducted?	NA										
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	NR	NA	NR	NR	NR	++	NR	-	NR	NR	NR
4.4 Were the estimates of effect size given or calculable?	+	-	+	NR	-	+	NR	NA	NR	NA	NA
4.5 Were the analytical methods appropriate?	+	+	+	-	-	++	++	-	-	NA	-
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	+	-	+	NA	NA	+	NR	NR	NR	NA	NA
Section 5: Summary											
5.1 Are the study results internally valid (i.e. unbiased)?	-	-	-	-	-	-	-	-	-	-	-
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	-	_	_	-	_	-	-	-	_	_	-
NR – not reported; NA – not applicable											

Appendix 6. Results of quality assessment for studies included in Jones and colleagues (2007)

Table 8.7. Quality assessment for RCTs and NRCTs

- 1.1 The study addressed an appropriate and clearly focused question
- 1.2 The assignment of participants to intervention groups is randomised
- 1.3 An adequate concealment method is used
- 1.4 Participants and investigators are kept 'blind' about intervention allocation
- 1.5 The intervention and control groups are similar at the start of the trial
- 1.6 The only difference between groups is the intervention under investigation
- 1.7 All relevant outcomes are measured in a standard, valid and reliable way
- 1.8 What percentage of the participants or clusters recruited into each intervention arm of the study dropped out before the study was completed?
- 1.9 All participants are analysed in the groups to which they were allocated? (ITT)
- 1.10 Where the study is carried out at more than one site, results are comparable for all sites

Key: ++ Well covered + Adequately covered - Poorly covered × Not addressed NR Not reported N/A Not applicable

Poforonoo(c)	Question										Dating
Reference(s)	1.1	1.2	1.3	1.4	1.5	1.6	1.7	1.8	1.9	1.10	Rating
Allison et al., 1990	++	-	×	N/A	-	+	++	8%	×	×	-
Botvin et al., 2003	++	-	×	N/A	++	×	+	Matched data not available for 44%	×	×	-
Brown et al., 2005	++	+	NR	N/A	+	-	++	12%	+	×	+
Donaldson et al., 1995; 2000	+	-	NR	N/A	NR	NR	++	NR	NR	N/A	-

Poforonoo(c)	Question										Doting
Reference(s)	1.1	1.2	1.3	1.4	1.5	1.6	1.7	1.8	1.9	1.10	Rating
Eddy et al., 2003	++	+	×	×	-	-	+	3%	+	N/A	-
Furr-Holden, 2004	++	++	N/A	N/A	-	-	+	16%	+	NR	+
Hurry and McGurk 1997; Hurry et al., 2000	++	+	×	NR	-	+	+	NR	NR	N/A	+
O'Donnell et al., 1995	+	N/A	N/A	N/A	+	+	+	40%	NR	N/A	+
Padget et al., 2006	++	N/A	N/A	N/A	+	++	++	Intervention 12% and control 12%	×	NR	+
Schinke and Tepavac, 1995	++	N/A	N/A	N/A	×	++	+	NR	×	NR	-
Schinke et al., 2000	++	+	+	×	++	-	++	14.11% total	++	×	+
Shope et al., 1992	+	×	×	N/A	-	×	-	28% at 2.5 yr	×	×	-
Sigelman et al., 2004	++	-	×	N/A	+	-	++	NR	×	NR	+

Table 8.8. Quality assessment for controlled before and after studies

- 1.1 Contemporaneous data collection
- Score DONE pre and post intervention periods for study and control sites are the same.
- Score NOT CLEAR if it is not clear in the paper, e.g. dates of collection are not mentioned in the text.
- Score NOT DONE if data collection was not conducted contemporaneously during pre and post intervention periods for study and control sites.
- 1.2 Appropriate choice of control site

Studies using second site as controls:

• Score DONE if study and control sites are comparable with respect to dominant reimbursement system, level of care, setting of care and academic status.

• Score NOT CLEAR if not clear from paper whether study and control sites are comparable.

• Score NOT DONE if study and control sites are not comparable.

1.3 Baseline measurement

• Score DONE if performance or patient outcomes were measured prior to the intervention, and no substantial differences were present across study groups (e.g. where multiple pre intervention measures describe similar trends in intervention and control groups);

• Score NOT CLEAR if baseline measures are not reported, or if it is unclear whether baseline measures are substantially different across study groups;

• Score NOT DONE if there are differences at baseline in main outcome measures likely to undermine the post intervention differences (e.g. are differences between the groups before the intervention similar to those found post intervention).

1.4 Characteristics for studies using second site as control

• Score DONE if the authors state explicitly that the primary outcome variables were assessed blindly OR the outcome variables are objective e.g. length of hospital stay, drug levels as assessed by a standardised test;

- Score NOT CLEAR if not specified in the paper;
- Score NOT DONE if the outcomes were not assessed blindly.
- 1.5 Blinded assessment of primary outcome(s)

• Score DONE if the authors state explicitly that the primary outcome variables were assessed blindly OR the outcome variables are objective e.g. length of hospital stay, drug levels as assessed by a standardised test;

- Score NOT CLEAR if not specified in the paper;
- Score NOT DONE if the outcomes were not assessed blindly.
- 1.6 Protection against contamination
- Studies using second site as control

• Score DONE if allocation was by community, institution, or practice and is unlikely that the control group received the intervention;

• Score NOT CLEAR if providers were allocated within a clinic or practice and communication between experimental and group providers was likely to occur;

• Score NOT DONE if it is likely that the control group received the intervention (e.g. cross-over studies or if individuals rather than providers were randomised).

1.7 Reliable primary outcome measure(s)

• Score DONE if two or more raters with at least 90% agreement or kappa greater than or equal to 0.8 OR the outcome is obtained from some automated system e.g. length of hospital stay, drug levels as assessed by a standardised test;

- Score NOT CLEAR if reliability is not reported for outcome measures that are obtained by chart extraction or collected by an individual;
- Score NOT DONE if agreement is less than 90% or kappa is less than 0.8.
- 1.8 Follow up of professionals (protection against exclusion bias)
- Score DONE if outcome measures obtained 80-100% subjects allocated to groups. (Do not assume 100% follow-up unless stated explicitly.);
- Score NOT CLEAR if not specified in the paper;
- Score NOT DONE if outcome measures obtained for less than 80% of individuals allocated to groups.

1.9 Follow up of individuals

• Score DONE if outcome measures obtained 80-100% of individuals allocated to groups or for individuals who entered the study. (Do not assume 100% follow-up unless stated explicitly.);

• Score NOT CLEAR if not specified in the paper;

• Score NOT DONE if outcome measures obtained for less than 80% of individuals allocated to groups or for less than 80% of individuals who entered the study.

Poforonoo(c)	Question									Coding		
Kelerence(S)	1.1	1.2	1.3	1.4	1.5	1.6	1.7	1.8	1.9	coung		
Hawthorne et al., 1995; Hawthorne, 1996	Not clear	Done	Not clear	Done	Done	Not clear	Not done	Not clear	Not clear	-		
Zavela et al., 1997	Not done	Not done	Not done	Not clear	Not done	Done	Done	Not clear	Done	-		
Zavela et al., 2004	Done	Not clear	Not done	Not done	Not done	Not clear	Done	Not clear	Not clear	-		

Age	Eng	land	U	SA
		Year		Grade
0-4		-		-
4-5	Pre-School	-	Pre Kindergarten	-
5-6	Primary School	1	Kindergarten	-
6-7	(Key Stage 1)	2		1
7-8		3		2
8-9	Junior School	4	Elementary School	3
9-10	(Key stage 2)	5		4
10-11		6		5
11-12	Lower	7		6
12-13	Secondary	8	Middle School	7
13-14	(Rey stage 3)	9		8
14-15	Upper	10		9
15-16	(Key stage 4)	11	High School	10
16-17	6th Form	-	nigh School	11
17-18	College	-		12

Appendix 7. Conversion table for English key stages and US grade equivalents



Supplement B

A review of the effectiveness and cost-effectiveness of personal, social and health education in secondary school and further education focusing on sex and relationships and alcohol education for young people aged 11 to 19 years

FINAL REPORT

Lisa Jones, Geoff Bates, Jennifer Downing, Harry Sumnall, Mark A Bellis Centre for Public Health, Liverpool John Moores University



October 2009

Acknowledgements

In addition to the authors of the report we would like to acknowledge the contribution of Katrina Stredder and Olivia Wooding from the Centre for Public Health, LJMU for their assistance in full text screening and data extraction.

Table of Contents

1	Int	roduction	. 30
	1.1	Aims and objectives	. 30
	1.2	Research question	. 30
2	Ba	ckground	. 31
	2.1	Personal, Social Health education (PSHE)	. 32
	2.2	Government policy	. 33
3	Me	ethodology	. 35
	3.1	Search strategy	. 35
	3.2	Inclusion and exclusion criteria	. 36
	3.3	Data extraction strategy	. 37
	3.4	Quality assessment strategy	. 37
	3.5	Methods of analysis/synthesis	. 37
4	Su	mmary of study identification	. 39
	4.1	Review of effectiveness and cost-effectiveness	. 39
5	Alc	cohol and drug education programmes	. 41
	5.1	Systematic reviews and meta-analyses	. 41
	5.2	Classroom-based programmes: alcohol specific	. 48
	5.3	Classroom-based: substances including alcohol	. 66
	5.4	Brief or single session interventions	. 99
	5.5	Multicomponent school- and community-based programmes	115
	5.6	Peer support and counselling programmes	123
	5.7	Review of published economic evaluations	130
6	Se	x and relationships education	135
	6.1	Systematic reviews and meta-analyses	135
	6.2	UK-based studies	140
	6.3	Abstinence-only programmes	156
	6.4	Abstinence-plus programmes	171
	6.5	HIV and sexual risk-reduction programmes	200
	6.6	Other school-based approaches	219
	6.7	Review of published economic evaluations	229

7 General health education programmes		233	
	7.1	General health education programmes	233
8 Dis		scussion	246
	8.1	Alcohol and drug education programmes	246
	8.2	Sex and relationships education	250
	8.3	General health education programmes	254
	8.4	Strengths and limitations	255
	8.5	Research recommendations	257
9	Co	onclusions	258
	9.1	Alcohol and drug education programmes	258
	9.2	Sex and relationships education programmes	258
	9.3	General health education programmes	258
10	10 References		

Appendices

Appendix 1. References to included studies	262
Appendix 2. References to excluded studies	275
Appendix 3. Summary of quality assessment	316
Appendix 4. Summary of quality assessment for studies included in Jones et al (2007)	341
Appendix 5. Conversion table for English key stages and US grade equivalents	352

Tables of figures

Figure 4.1. Process of study selection	39
----------------------------------------	----

Table of tables

Table 4.1. Summary of study designs identified for inclusion	40
Table 5.1. Alcohol education: systematic reviews and meta-analyses	46
Table 5.2. Summary of programme content: classroom-based (alcohol specific)	19
Table 5.3. Classroom-based alcohol education programmes: short-term programme effects of	on
knowledge, attitudes and skills	30
Table 5.4. Classroom-based alcohol education programmes: short-term programme effects on heal	th
and social outcomes Error! Bookmark not define	d.
Table 5.5. Classroom-based alcohol education programmes: medium-term programme effects of	on
knowledge, attitudes and skills define	d.

Table 5.6. Classroom-based alcohol education programmes: medium-term programme effects on
health and social outcomes Error! Bookmark not defined.
Table 5.7. Classroom-based alcohol education programmes: long-term programme effects on
knowledge, attitudes and skills Error! Bookmark not defined.
Table 5.8. Classroom-based alcohol education programmes: long-term programme effects on health
and social outcomes Error! Bookmark not defined.
Table 5.9. Summary of programme content: classroom-based (substance use including alcohol) 67
Table 5.10. Classroom-based substance use prevention programmes: short-term programme effects
on knowledge, skills and attitudes Error! Bookmark not defined.
Table 5.11. Classroom-based substance use prevention programmes: short-term programme effects
on health and social outcomes Error! Bookmark not defined.
Table 5.12. Classroom-based substance use prevention programmes: medium-term programme
effects on knowledge, attitudes and skills Error! Bookmark not defined.
Table 5.13. Classroom-based substance use prevention programmes: medium-term programme
effects on health and social outcomes Error! Bookmark not defined.
Table 5.14. Classroom-based substance use prevention programmes: long-term programme effects
on knowledge, attitudes and skills Error! Bookmark not defined.
Table 5.15. Classroom-based substance use prevention programmes: long-term programme effects
on health and social outcomes Error! Bookmark not defined.
Table 5.16. Alcohol education: brief or single session interventions
Table 5.17. Brief and single session interventions: short-term programme effects on knowledge skills
and attitudes
Table 5.18. Brief and single session interventions: short-term programme effects on health and social
outcomes Error! Bookmark not defined.
Table 5.19. Brief and single session interventions: medium-term programme effects on knowledge,
attitudes and skills Error! Bookmark not defined.
Table 5.20. Brief and single session interventions: medium-term programme effects on health and
social outcomes Error! Bookmark not defined.
Table 5.21. Alcohol education: multicomponent school and community programmes 116
Table 5.22. Multicomponent programmes: short-term programme effects on knowledge, attitudes and
skills
Table 5.23. Multicomponent programmes: short-term programme effects on health and social
outcomes
Table 5.24. Alcohol education: peer support and counselling programmes
Table 5.25. Peer support and counselling programmes: short-term programme effects on knowledge,
attitudes and skills
Table 5.26. Peer support and counselling programmes: short-term programme effects on health and
social outcomes
Table 5.27. Peer support and counselling programmes: medium-term programme effects on health
and social outcomes

Table 5.28. Cost elements (reproduced from Swisher et al., 2004)
Table 5.29. Approximate direct costs of the MPP by component (in thousands of dollars) 132
Table 5.30. Results from the cost-effectiveness analysis of the MPP 133
Table 6.1. Sex and relationships education: systematic reviews and meta-analyses
Table 6.2. Sex and relationships education: UK studies
Table 6.3. UK SRE studies: short-term programme effects on knowledge, attitudes and skills
Table 6.4. UK SRE studies: short-term programme effects on health and social outcomes
Table 6.5. UK SRE studies: medium-term programme effects on knowledge, attitudes and skills 153
Table 6.6. UK SRE studies: medium-term programme effects on health and social outcomes 154
Table 6.7. UK SRE studies: long term programme effects on knowledge, attitudes and skills 155
Table 6.8. UK SRE studies: long-term programme effects on health and social outcomes
Table 6.9. Sex and relationships education: abstinence-only programmes 157
Table 6.10. Abstinence-only programmes: short-term programme effects on knowledge, attitudes and
skills
Table 6.11. Abstinence-only programmes: short-term programme effects on health and social
outcomes
Table 6.12. Abstinence-only programmes: medium-term programme effects on knowledge, attitudes
and skills
Table 6.13. Abstinence-only programmes: medium-term programme effects on health and social
outcomes
Table 6.14. Abstinence-only programmes: long-term programme effects on knowledge, attitudes and
skills
Table 6.15. Abstinence-only programmes: long-term programme effects on health and social
outcomes 170
Table 6.16. Sex and relationships education: abstinence-plus programmes 172
Table 6.17. Abstinence-plus programmes: short-term programme effects on knowledge, attitudes and
skills
Table 6.18. Abstinence-plus programmes: short-term programme effects on health and social
outcomes
Table 6.19. Abstinence-plus programmes: medium-term programme effects on knowledge, attitudes
and skills
Table 6.20. Abstinence-plus programmes: medium-term programme effects on health and social
outcomes
Table 6.21. Abstinence-plus programmes: long-term programme effects on knowledge, attitudes and
skills
Table 6.22. Abstinence-plus programmes: Long term programme effects on health and social
outcomes
Table 6.23. Sex and relationships education: HIV and sexual risk-reduction programmes
Table 6.24. HIV and sexual risk-reduction programmes: short-term programme effects on knowledge,
attitudes and skills

Table 6.25. HIV and sexual risk-reduction programmes: short-term programme effects on health and			
social outcomes			
Table 6.26. HIV and sexual risk-reduction programmes: medium-term programme effects on			
knowledge, attitudes and skills			
Table 6.27. HIV and sexual risk-reduction programmes: medium-term programme effects on health			
and social outcomes			
Table 6.28. HIV and sexual risk-reduction programmes: long term programme effects on knowledge,			
attitudes and skills			
Table 6.29. HIV and sexual risk-reduction programmes: long term programme effects on health and			
social outcomes			
Table 6.30. Sex and relationship education: other school-based approaches 220			
Table 6.31. Other school-based SRE approaches: short-term programme effects on knowledge,			
attitudes and skills			
Table 6.32. Other school-based SRE approaches: short-term programme effects on health and social			
outcomes			
Table 6.33. Other school-based SRE approaches: long-term programme effects on knowledge,			
attitudes and skills			
Table 6.34. Other school-based SRE approaches: long-term programme effects on health and social			
outcomes			
Table 6.35. Programme implementation costs, Safer Choices			
6.36. Sex and relationships education: summary of economic evaluation studies			
Table 7.1. General health education programmes 234			
Table 7.2. General health education: short-term programme effects on knowledge, attitudes and skills			
Table 7.3. General health education: short-term programme effects on health and social outcomes242			
Table 7.4. General health education: medium-term programme effects on knowledge, attitudes and			
skills			
Table 7.5. General health education: medium-term programme effects on health and social outcomes			
Table 7.6. General health education: long-term programme effects on knowledge, attitudes and skills			
Table 7.7. General health education: long-term programme effects on health and social outcomes 244			

Glossary

Abstinence-only programmes	Programmes that encourage and promote abstinence as the best and only way to prevent pregnancy, HIV and other STIs.
Abstinence-plus programmes	Programmes that emphasise abstinence as the safest way to prevent pregnancy, HIV and other STIs, but also promote safer sex through the use of contraceptives.
American school grades	Education is divided into 3 levels: elementary school, junior high (or middle) school and high school.
Bias	Deviation of results or inferences from the truth, or processes leading to such deviation. Any trend in the collection, analysis, interpretation, publication or review of data that can lead to conclusions that are systematically different from the truth.
Binge drinking	Consuming large quantities of alcohol over a short period of time. Often associated with drinking to become intoxicated.
Cluster randomisation	A trial where the unit of randomisation is a cluster of participants (e.g. a school).
Controlled before and after study (CBA)	Intervention groups are tested and data collected before and after the intervention has been administered. Differ from controlled non-randomised trials in that participants are not allocated to intervention or control groups, but rather a 'convenience' control sample is used.
Drug education programmes	Programmes that include a focus on illegal drugs or tobacco in addition to alcohol.
Effect size	Effect size is a term used for a family of indices that measure the magnitude of the relationship between variables or treatment effect. Effect sizes are commonly used in meta- analyses as unlike significance tests these indices are independent of sample size.
General health education programmes	Programmes that are health based but include aspects and outcomes relating to alcohol or sex and relationships
HIV and sexual risk-reduction programmes	Programmes that focus on HIV prevention and HIV risk- behaviour, sexual risk-behaviour or a combination of both.
Intention to treat analysis	A method of data analysis in which all participants are analysed in the group they were assigned to at randomisation regardless of treatment adherence.
Internal validity	How well the study has minimised sources of bias and how likely it is that the intervention caused the observed outcomes.
Key stage	Pupils' progress through school is measured in key stages. Each key stage covers a number of school years. Starting at key stage 1 and finishing at key stage 4.
Long-term outcome	Study outcomes as evaluated at over one year post- intervention.
Medium-term outcome	Study outcomes as evaluated at six months to one year post- intervention.
Mean difference	The difference between two means divided by an estimate of the within group standard deviation.
Meta-analysis	The combination of quantitative evidence from a number of studies.
Non-Randomised Controlled Trial	These are trials where participants or clusters are allocated between intervention and control groups but the allocation is

	not randomised or quasi-randomised (e.g. alternate allocation).
Odds ratio	The odds of the event occurring in one group (e.g. intervention) divided by the odds of the event occurring in the other group (e.g. control).
Randomised Controlled Trial	Individuals or, defined groups of individuals (clusters) are randomised to either an intervention or a control group. If well implemented, randomisation should ensure that intervention and control groups only differ in their exposure to treatment.
Risk ratio	The risk of the event in the one group (e.g. intervention) divided by the risk of the event in the other group (e.g. control).
Short-term outcomes	Study outcomes evaluated at less than six months post- intervention.
Social development programmes	Programmes that aim to impact upon alcohol use or sex and relationships through social development education
Systematic review	A method of locating, appraising and synthesising evidence from primary studies, which adheres to a scientific methodology.
Uncontrolled before and after study	Intervention groups are tested and data collected before and after the intervention has been administered. No control group is used for comparison purposes.

Abbreviations

AAPT	Adolescent Alcohol Prevention Trial
AMPS	Alcohol Misuse Prevention Study
BPBR	Be Proud! Be Responsible
CBA	Controlled before and after study
CARE	Community Awareness and Relationship Education
CBT	Cognitive Behavioural Therapy
CEA	Cost-effectiveness analysis
DARE	Drug Abuse Resistance Education
DfES	Department for Education And Skills
DH	Department of Health
HBM	Health Belief Model
HIV	Human immunodeficiency virus
I-LST	Infused Life Skills Training
ICER	Incremental cost-effectiveness ratio
ICU	Information about Consequences of Use
ITS	Interrupted times series
ІТТ	Intention to treat
LST	Life Skills Training
MMP	Midwest Prevention Project
MPM	Managing the Pressures before Marriage
MCMF	My Choice, My Future!
NICE	National Institute for Health and Clinical Excellence
NR	Not reported
NRCT	Non-Randomised Controlled Trial
OR	Odds Ratio
PID	Pelvic inflammatory disease
PSHE	Personal Social and Health Education
PT	Post-test
PY/PM	Protecting You/Protecting Me
QCA	Qualifications and Curriculum Authority
RAPP	Rochester AIDS Prevention Project for Youth
RCT	Randomised Controlled Trial
RCV	Re-capturing the Vision
RHC	Raising Healthy Children
SHARE	Sexual Health and Relationships: Safe, Happy and Responsible
SLT	Social Learning Theory
SR	Systematic Review
SRE	Sex and relationships education
STARS	Start Taking Alcohol Risks Seriously

STI	Sexually transmitted infection
SWAAT	Students Working Against AIDS Together
UBA	Uncontrolled before and after study
YTCs	Youth Training Centres
YAPP	Youth AIDS Prevention Project

EXECUTIVE SUMMARY

OBJECTIVES

This review sought to identify effective and cost-effective interventions and programmes that focus on alcohol education and sex and relationships education for secondary school aged children.

METHODS

The methods for the review followed NICE protocols for the development of NICE public health guidance. Sixteen databases were searched for effectiveness and cost-effectiveness studies published since 1990. Two reviewers independently screened all titles and abstracts. Data extraction and quality assessment were undertaken by one reviewer and checked for accuracy by a second reviewer. Each study was also graded (++, + or -) based on the extent to which the design and execution of the study minimised the potential sources of bias. Results of the data extraction and quality assessment for each study of effectiveness and cost-effectiveness were presented in structured tables and as a narrative summary.

ALCOHOL AND DRUG EDUCATION PROGRAMMES

A total of 119 articles met the criteria for inclusion in the review of alcohol and drug education programmes. Fourteen articles were systematic reviews and meta-analyses, 103 articles reported on the evaluation of an alcohol or substance use education programme, and two articles were economic evaluation studies. Of the 103 articles, 74 reported on evaluations of classroom-based programmes; 20 of which were alcohol specific, and 54 of which focused on substance use including alcohol. Also identified were 15 articles that reported on evaluations of brief behavioural or single session interventions, nine articles reporting on two multicomponent school- and community-based programmes and five articles reporting on evaluations of peer support and/or counselling programmes.

Systematic reviews and meta-analyses

A total of 14 systematic reviews and meta-analyses were identified for inclusion. The majority of the reviews identified examined the effectiveness of programmes targeting substance use including alcohol, and only three reviews focused specifically on the prevention of alcohol use. A good quality review found that there was no consistent evidence to determine which programmes were effective over the short to medium-term, but highlighted three programmes which were effective over the longer term. These included the family-based, Strengthening Families programme, and two school-based programmes, Botvin's LST and a culturally-focused curriculum for Native American students. A second review highlighted promising evidence from six additional programmes, Keepin it REAL, the Midwest Prevention Project, Project Northland, Healthy School and Drugs, Project ALERT, and the SHAHRP. Two reviews identified evidence to suggest that peer leaders strengthened the effects of school-based interventions, although another review found that any beneficial effects of peer involvement were lost when they were combined with teacher-led activities. One review did not find any evidence to suggest that any particular agency or external contributor was more effective than another.

Evidence statement 1

- 1 (a) There is strong evidence from two systematic reviews to suggest that a secondary-level schoolbased programme, Botvin's LST, can produce long-term reductions (greater than 3 years) in alcohol use. Other promising intervention approaches include: Keepin it REAL, the Midwest Prevention Project, Project Northland, Healthy School and Drugs, Project ALERT, and SHAHRP.
- 1 (b) There is moderate evidence from two systematic reviews to suggest that programmes delivered by peer leaders may be more beneficial than programmes delivered by teachers or other contributors.

Classroom-based programmes

A total of 74 articles were identified for inclusion that reported on evaluations of classroom-based programmes. Overall, 20 articles were identified for inclusion that reported on the evaluation of 12 alcohol education programmes across 15 studies and 54 articles were identified that examined 22 classroom-based substance use (including alcohol) prevention programmes across 34 studies.

Of the 15 studies identified for inclusion that examined alcohol education programmes, nine studies were RCTs, three were NRCTs and three were CBA studies. The 12 alcohol education programmes were primarily classroom-based curriculums, but two programmes incorporated additional materials and activities for parents. The programmes identified targeted students across a range of age groups: eight programmes targeted students aged 14 or younger and four programmes were targeted at older adolescents. Across eight studies that examined intervention effects on knowledge related to alcohol use there were indications that alcohol-specific education programmes generally increased alcohol or curriculum knowledge over the short-term. However, effects on medium- and long-term knowledge acquisition were weaker. Eight studies examined young people's alcohol-related attitudes and values, finding non-significant programme effects across the majority of programmes. However, the SHAHRP programme, which was based on a harm reduction approach, had positive short- and long-term effects on students' alcohol-related attitudes. Short-term increases in safer alcohol-related attitudes were also reported in a study that examined a highly-role specific programme compared to a less-role specific alcohol programme. Few studies examined intervention effects on personal and social skills. Intervention impacts on a range of alcohol-related measures were examined across the included studies. The SHAHRP programme appeared to have the most consistent effects on short-term alcohol use, and additionally had effects on hazardous/harmful drinking. Medium- to long-term effects on alcohol consumption were found to be limited. Studies were either methodologically poor, as in the case of the AAPT programme, or reported conflicting or diminished effects. For example, 17-months after delivery of the SHAHRP programme the positive short-term effects appeared to be declining. Although intervention effects favoured SHAHRP, differences between intervention and control students in terms of their alcohol consumption and other measures of alcohol use including harmful/hazardous drinking were no longer significant. There were no long-term effects of a longer term version of the AMPS programme on alcohol consumption, but there did appear to be intervention effects on alcohol misuse.

Evidence statement 2

- 2 (a) There is strong evidence from four RCTs, two NRCTs and two CBA studies to suggest that classroom-based alcohol specific programmes are effective at increasing alcohol-related knowledge in the short-term, but have inconsistent or mixed effects on alcohol-related knowledge in the medium- to long-term. Findings may only be partially applicable to the UK as studies were implemented within Australia, Germany and the USA and may not be generalisable beyond the populations studied.
- 2 (b) Overall, there is inconsistent evidence from four RCTS, three NRCTs and one CBA study to determine the effects of alcohol specific education programmes on attitudes and values relating to alcohol. However, there is moderate evidence from one NRCT to suggest that programmes based on a harm reduction approach may have positive short- to long-term effects on students' alcohol-related attitudes. In addition, there is weak evidence from one NRCT to suggest that programmes with a high level of role-specification for providers may have short-term positive impacts on attitudes and values.
- 2 (c) There is inconsistent evidence from one RCT and one CBA study to determine the effects of programmes focusing on reducing the harm from drinking and driving on drink driving measures.
- 2 (d) There is moderate evidence from five RCTs, three NRCTs and two CBA studies to suggest that alcohol-specific education programmes may have mixed short-term effects on health outcomes relating to alcohol use. One NRCT of a programme focusing on harm reduction through skillsbased activities (SHAHRP), showed short-term reductions in alcohol use. In particular effects were seen on risky drinking behaviours such as drunkenness and binge drinking. Findings may only be partially applicable to the UK as this study was conducted in Australia and may not be generalisable beyond the populations studied.
- 2 (e) There is moderate evidence from eight RCTS, one NRCT and one CBA study to suggest that alcohol-specific education programmes have limited medium- to long-term effects on health outcomes related to alcohol use, such as frequency of alcohol consumption and drunkenness. Findings may only be partially applicable to the UK as studies were implemented outside the UK and may not be generalisable beyond the populations studied.

Of the 34 studies, which examined 22 classroom-based substance use (including alcohol) prevention programmes, 23 were based on RCT designs, seven were NRCTs and four were CBA studies. Although all of the programmes were primarily classroom-based, five programmes combined school components with family- and/or community-based components. Two studies combined two originally school-based only programmes (DARE and LST, respectively) with components targeting parents. The majority of programmes targeted students aged 14 or younger. Four studies, including evaluations of LST, AMPS, DARE, and the Healthy School and Drugs Project, respectively, examined programme effects on alcohol-related knowledge. Overall both short and medium-term increases in alcohol knowledge were reported but these were not sustained long-term. Twenty-one studies

reported outcomes relating to alcohol or substance use attitudes and behavioural intentions. There were inconsistent effects on attitudes towards alcohol use and peer norms, but eight studies, which examined behavioural intentions indicated generally positive programme effects on intentions to drink or get drunk. Intervention effects on personal and social skills were examined across a small number of programmes but found to be inconsistent. Four programmes, the Positive Youth Development Programme, the Unplugged programme, a revised version of Project Alert, and the Healthy School and Drug Project had positive short-term programme effects on alcohol use. However, the findings of Positive Youth Development Programme were limited by the poor quality of the study. Positive longer term effects were demonstrated for two programmes, Keepin It REAL and Be Under Your Own Influence/All Stars, which combined school and media intervention components. The strongest evidence of effectiveness came from a series of studies which examined Botvin's LST. Two studies found positive short- and medium-term effects on drinking frequency and binge-drinking, and these were sustained long-term. However, replication of the programme by other research groups suggests that there may be issues with the transferability of LST to other settings.

Evidence statement 3

- 3 (a) There is moderate evidence from two RCTs, one NRCT and one CBA study to suggest that classroom-based substance use programmes are effective at improving knowledge relating to substance use and its effects in the short- to medium-term, but that these effects are not sustained in the long-term. Findings may only be partially applicable to the UK as studies were implemented outside the UK and may not be generalisable beyond the populations studied.
- 3 (b) There is moderate evidence from 12 RCTs, seven NRCTs and two CBA studies to suggest that classroom-based substance use programmes may have mixed effects on student's substance use-related attitudes and values. There is moderate evidence from seven RCTs, four NRCTs and two CBA studies to suggest that these programmes may impact on attitudes to substance use in the short- to medium-term and further evidence from five RCTs and three NRCTs to suggest that they may have a positive impact on long-term behavioural intentions. There is weak evidence from six RCTs, one NRCT and one CBA study to suggest that classroom-based substance use programmes have no medium- to long-term effects on peer norms. Findings may only be partially applicable to the UK as studies were implemented in the USA and may not be generalisable beyond the populations studied.
- 3 (c) There is weak and inconsistent evidence from one RCT, three NRCTs and one CBA study to determine the effects of classroom-based substance use programmes on young peoples' personal and social skills.
- 3 (d) There is moderate evidence from 14 RCTs, two NRCTs and two CBA studies to suggest that the following classroom-based substance use programmes may have inconsistent or no effects on alcohol use: DARE, Going Places, Lion's Quest SFA, All Stars Senior, Project Alert, Project SMART, Project TND, NARCONON drug education curriculum. In addition, there is weak

evidence from one CBA study to suggest that the Adolescent Decision Making Programme may have potentially harmful long-term effects on alcohol consumption.

- 3 (e) There is weak evidence from two RCTs and one NRCT to suggest that programmes that combine school-based curriculums with additional components may have positive effects on alcohol consumption. The Healthy School and Drugs Project, a three-year programme, which included a nine lesson teacher-led curriculum, formulation of school policy on substance use and involvement of parents, had short-term effects on alcohol use, but longer term effects of the programme have not been examined. Positive longer term effects were demonstrated for two programmes, Keepin It REAL and Be Under Your Own Influence/All Stars, which combined school-based curriculums with media intervention components. Findings may only be partially applicable to the UK as studies were implemented in the Netherlands and the USA and may not be generalisable beyond the populations studied.
- 3 (f) There is moderate evidence from three RCTs to suggest that LST has positive short-, mediumand long-term effects on drinking frequency and binge drinking. However, there is moderate evidence from three RCTs and one NRCT to suggest that there may be issues with the transferability of LST to other settings. Findings may only be partially applicable to the UK as studies were implemented in Spain and the USA and may not be generalisable beyond the populations studied.

Brief behavioural or single session interventions

A total of 15 articles were identified that reported on evaluations of seven brief behavioural or single session intervention approaches across 13 studies. Of the 13 studies, 11 studies were RCTs, one was an NRCT and one was a CBA study. All thirteen studies were primarily school-based but six studies examined interventions which incorporated materials targeting parents. A range of providers were utilised including school nurses, physicians, teachers, fitness professionals, consultants, trained research staff and motivational speakers. Two studies were based on mailed intervention materials and therefore did not involve a provider in the delivery.

None of the studies examined intervention effects on knowledge or understanding but eleven studies examined intervention effects on students' attitudes and values. Across these studies, there was an indication of positive intervention effects in the short-term, resulting in increases in negative views of alcohol and/or its consequences and a decrease in alcohol-related expectancies. Brief behavioural or single session intervention approaches appeared to have inconsistent short- and medium-term effects on student's intentions to drink. Few studies examined intervention effects on personal and social skills, and for two studies that examined impacts on self-control and parent-child relationships intervention approaches based on brief nurse consultations, indicated mixed, but generally positive effects of this programme approach on alcohol consumption and heavy drinking in the short-to medium-term. An alcohol tailored beverage programme and an intervention founded on the Behaviour-Image Model had inconsistent effects on alcohol use, and two further programmes, a

single session on the dangers of binge drinking, and the one week long Programme "Kickoff", had no effects on alcohol use.

Evidence statement 4

- 4 (a) There is moderate evidence from nine RCTs, one NRCT and one CBA study to suggest that brief behavioural or single session intervention approaches relating to alcohol use may have mixed effects on attitudes and values relating to alcohol use. There is moderate evidence from four RCT, one NRCT and one CBA study to suggest that these programmes may have positive short-term effects on how student's view alcohol use and its consequences and further evidence from five RCTs and one NRCT to suggest that brief behavioural interventions have mixed or inconsistent effects on intentions to drink. Findings may only be partially applicable to the UK as the majority of studies were implemented in the USA and may not be generalisable beyond the populations studied. In addition, the emphasis of the STARS for Families and Project SPORT interventions on abstinence may be of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- 4 (b) There is inconsistent evidence from two studies to determine the effects of brief behavioural and single session intervention approaches on personal and social skills.
- 4 (c) There is moderate evidence from five RCTs to suggest that brief behavioural interventions based on nurse-led consultations, such as the STARS for Families and Project SPORT programmes, can produce short-term reductions in alcohol use, but further moderate evidence from three RCTs to suggest that these effects may not be sustained in the medium-term. There is weak evidence from two RCTs, one NRCT and one CBA study to suggest that other brief behavioural and single session intervention approaches may have a limited impact on alcohol consumption. These findings may only be partially applicable to the UK as the majority of studies were implemented in the USA and may not be generalisable beyond the populations studied. In addition, the emphasis of the STARS programme on abstinence may be of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.

Multicomponent school- and community-based programmes

Nine studies were identified that examined two multicomponent, school- and community-based programmes: Project Northland and the Midwest Prevention Project. Both programmes were based in communities in the USA and combined comprehensive school-based curriculums, with community-based activities and parental involvement components. All nine studies identified for inclusion were based on an RCT design.

None of the studies examined intervention effects on knowledge or understanding, or personal and social skills. There no were effects of Project Northland on attitudes and values and neither of the studies of the MPP examined intervention effects on these outcomes. Project Northland significantly reduced growth in binge drinking and tendency to use alcohol during Phase I and II of the programme, however, during the interim phase of the programme growth in alcohol use was greater among intervention students than control students. The three-year MPP did not have significant effects on

alcohol use in one cohort of ninth/tenth grade students, but a short-term secondary prevention effect was reported in a second cohort.

Evidence statement 5

- 5 (a) There is no evidence from eight RCTs to determine the impact of multicomponent, school- and community-based programmes on knowledge, or personal and social skills.
- 5 (b) There is moderate evidence from three RCTs to suggest that Project Northland, a long-term multicomponent, school- and community-based programme, has no effects on attitudes and values related to alcohol consumption.
- 5 (c) There is moderate evidence from two RCTs to suggest that the Midwest Prevention Project has no effects on alcohol consumption and inconsistent evidence from five RCTs to suggest that Project Northland may have mixed effects on alcohol consumption. Two RCTs showed reductions in alcohol consumption, particularly among younger adolescents, but replication of the programme among an urban sample showed that the programme was not effective. Findings may only be partially applicable to the UK as studies were implemented in the USA and may not be generalisable beyond the populations studied.

Peer support and counselling programmes

Five studies examined five peer support and counselling programmes. Of the five studies identified for inclusion, one study was based on an RCT design; two studies were NRCTs; and two studies were based on CBA designs. All five programmes were school-based only and the provider for three peer support programmes was peers alone. The two counselling programmes were delivered by health counsellor or educational psychology students, respectively.

None of the studies examined intervention effects on knowledge or understanding, or personal and social skills. Three studies examined short-term intervention effects on attitudes and values. For two peer leadership programmes there appeared to be modest impacts on attitudes to alcohol and one study of a counselling programme found that the programme had a positive impact on the number of psychological problems that students' attributed to their alcohol use. Neither of the counselling programmes was shown to be effective in reducing alcohol consumption, and one programme had potentially harmful effects on high school students' alcohol consumption. There were inconsistent effects of peer support programmes on alcohol use.

Evidence statement 6

There is inconsistent evidence from one RCT, two NRCTs and two CBA studies to determine the effectiveness of counselling and peer support on attitudinal and behavioural outcomes related to alcohol use. Findings may only be partially applicable to the UK as studies were implemented in the USA and may not be generalisable beyond the populations studied.

Review of published economic evaluations

Two studies were identified that met the criteria for inclusion in the review of published economic evaluations. One study assessed the cost-effectiveness of standard and infused LST, and a second

study assessed the costs, benefits and cost-effectiveness of the MPP. The standard LST programme was found to be more cost effective than I-LST by \$33.46 per student after 1 year of intervention delivery. In the second year, however, standard LST had no effects and the authors concluded that I-LST was more cost-effective. The 3-year total costs of the two programmes were estimated at \$109,429.04 and \$93,088.17, respectively. The results of the cost-benefit analysis (CBA) of the MPP demonstrated a \$700 net saving per family per year resulting from a reduction in the incidence of monthly drunkenness. Cost benefits ratios were also shown to be favourable (ratio to \$1 spent on prevention to saving is \$1:1.69). Compared to "usual" drug education the ICER of the MPP was reported to be equal to the ratio of its incremental cost per incremental effects, equivalent to \$10 per net reduction in the incidence of monthly drunkenness.

Evidence statement 7

There is inconsistent evidence from two economic evaluation studies to determine the costeffectiveness of school-based interventions that aim to prevent or reduce alcohol use in young people under 18 years old. This evidence may be of limited applicability to a UK context because cost and benefit estimates were based on data from studies conducted in the USA.

SEX AND RELATIONSHIPS EDUCATION

A total of 75 articles met the criteria for inclusion in the review of sex and relationships education programme. Nine articles were systematic reviews and meta-analyses, 65 articles reported on evaluations of sex and relationships education interventions, and one article was an economic evaluation study.

Systematic reviews and meta-analyses

Nine systematic reviews evaluated abstinence only and abstinence plus safer-sex promotion programmes (Underhill et al., 2007, 2008; Bennett & Assefi, 2005), safer-sex promotion (Oakley et al., 1995; Franklin et al., 2007; Kirby et al., 1994; Pedlow and Carey, 2003; Robin et al., 2004) and sexuality-focused interventions (Sales et al., 2006). Findings from three reviews that examined abstinence-only and abstinence-plus programmes (Underhill et al., 2007, 2008; Bennett and Assefi, 2005) indicated that abstinence-only programmes have limited effects or are ineffective for preventing or reducing sexual risk behaviours. In addition, Oakley et al (1995) found evidence to suggest that abstinence only education may have an adverse effect and actually increase sexual experimentation among students. For programmes that incorporated information on safe sex and use of contraception, there was evidence from five reviews (Underhill et al., 2008; Pedlow & Carey, 2003; Franklin et al., 1997; Kirby et al., 1994; Oakley et al., 1995) to suggest that interventions may have effects on preventing sexual risk behaviours, but that these effects tend to be modest. There was no evidence that sexuality and AIDS education increased sexual activity.

Evidence statement 8

- 8 (a) There is strong evidence from three systematic reviews to suggest that abstinence-only programmes have limited effects or are ineffective for preventing or reducing sexual risk behaviours.
- 8 (b) There is moderate evidence from five systematic reviews to suggest that interventions incorporating information on safer sex and contraceptive use may have positive, but limited effects on preventing sexual risk behaviours. There is no evidence that such programmes increase the occurrence of sexual activity among young people.
- 8 (c) There is moderate evidence from four systematic reviews to suggest that effective characteristics of sexual risk reduction interventions include: (1) a theoretical basis; (2) use of trained adult health educators as providers; and (3) provision of highly specific content focusing on sexual risk reduction.

UK-based studies

Twelve UK studies, evaluating seven programmes, were identified that could be defined as predominantly sex and relationships education. Of the 12 studies, six were RCTs, two studies were based on an NRCT design, and five were CBA studies. Seven studies (Henderson et al., 2007; Mellanby et al., 1995; 2001; Stephenson et al., 2004; 2008; Tucker et al., 2007; Wight et al., 2002) reported on evaluations of three comprehensive school-based programmes, A PAUSE, RIPPLE and SHARE, respectively. Two studies (Gillies et al., 1990; Bellingham et al., 1993) reported on evaluations of the Streetwize UK, AIDS education comic, Denman et al (1995) examined a theatre in education programme and two studies (Graham et al., 2002; Magnusson et al., 2004) examined single lessons on emergency contraception and contraceptive services, respectively.

Across four studies (Tucker et al., 2007, Wight et al., 2002; Stephenson et al., 2004; Mellanby et al., 2001) that examined three comprehensive school-based programmes, there were indications that these programmes had significant effects on knowledge about STIs. Three studies (Denman et al., 1995; Bellingham and Gillies, 1993; Gillies et al., 1990) that examined a theatre in AIDS/HIV education programme and the Streetwize UK comic, respectively, had positive impacts on knowledge about HIV, and a teacher-led intervention about emergency contraception (Graham et al., 2002) increased knowledge relating to contraception. The effects of three comprehensive SRE programmes on attitudes were mixed. There were inconsistent or no effects of the peer-led RIPPLE programme and A PAUSE, but there were positive programme effects of the SHARE programme (Tucker et al., 2007) on attitudes concerning condom use and STI prevention, and on self-efficacy to use condoms. Other intervention approaches focusing on HIV prevention (Denman et al., 1995; Bellingham and Gillies, 1993; Gillies et al., 1990) and emergency contraception (Graham et al., 2002) had limited impacts on attitudes and values. Results for skills outcomes were limited with few studies reporting on these outcomes. In addition, few of the studies found significant programme effects on health outcomes related to sexual health. There were positive medium- and long-term effects of the A PAUSE (Mellanby et al., 1995) and RIPPLE (Stephenson et al., 2004) programmes on the number of students who reported ever having had sex, although for RIPPLE this effect was only apparent among females at the 18-month follow-up. There were no effects of either the RIPPLE (Stephenson et al., 2004; 2008) or SHARE (Wight et al., 2002) programmes on use of condoms or other forms of contraception. There was mixed evidence on the effects of these programmes on pregnancy. There were no medium-term effects of the SHARE (Wight et al., 2002) or RIPPLE (Stephenson et al., 2004) programmes on rates of unintended pregnancies, however, Stephenson et al (2008) found that at age 20, students who participated in the peer-led RIPPLE programme were less likely to have been pregnant. There were no long-term effects of the RIPPLE (Stephenson et al., 2008) or SHARE (Henderson et al., 2006) programmes on abortion rates. Other intervention approaches focusing on HIV prevention (Denman et al., 1995; Bellingham and Gillies, 1993), and single lessons regarding emergency contraception (Graham et al., 2002) and contraceptive services (Magnusson et al., 2004), respectively, had no impact on sexual behaviours.

Evidence statement 9

- 9 (a) There is moderate evidence from two RCTs and two CBA studies to suggest that comprehensive sex education programmes may be effective at increasing students' knowledge about STIs in the short- to long-term. In addition, there is weak evidence from one RCT and two CBA studies to suggest that brief interventions focusing on HIV prevention, such as theatre in education or a comic-based intervention, may have short-term positive effects on knowledge about HIV/AIDs. This evidence is directly applicable as these studies were conducted in the UK.
- 9 (b) Overall, there is inconsistent evidence from two RCTs and two CBA studies to determine the effects of comprehensive sex education programmes on attitudes and values relating to sexual health. However, there is weak evidence from one CBA study to suggest that a two-year sex education programme aimed at reducing unsafe sexual behaviour and unwanted pregnancy (SHARE) may have positive effects on long-term attitudes and intentions regarding condoms. There is moderate evidence from two RCTs, one NRCT and two CBA studies to suggest that brief or single session interventions focusing on HIV prevention or contraceptives and contraceptive services may have a limited impact on students' attitudes and values. This evidence is directly applicable as the study was conducted in the UK.
- 9 (c) There is inconsistent evidence from two RCTs and one CBA study to determine the effects of UK-based SRE approaches on personal and social skills.
- 9 (d) There is moderate evidence from one RCT and one NRCT to suggest that comprehensive SRE programmes, that include peer-led sessions such as RIPPLE and A PAUSE, may delay sexual initiation, but strong evidence from three RCTs and one NRCT to suggest that SRE programmes and single session interventions focusing on contraceptives and contraceptive services may have no impacts on condom or contraceptive use. This evidence is directly applicable as the studies were conducted in the UK.
- 9 (e) There is mixed evidence from three RCTs on the effects of comprehensive SRE programmes on outcomes relating to pregnancy. There is moderate evidence from one RCT of the peer-led

RIPPLE programme to suggest that this programme may reduce teenage pregnancy, but not abortion, rates in the long-term and strong evidence from one RCT of the teacher-led SHARE programme to suggest that this programme has no long-term effects on conceptions or terminations. This evidence is directly applicable as the studies were conducted in the UK.

Abstinence-only programmes

Ten articles were identified that evaluated eight programmes defined as abstinence-only programmes across nine studies. These programmes encouraged and promoted abstinence as the best and only way to prevent pregnancy, HIV and other STIs. Of the nine studies, two were RCTs and seven were NRCTs. Four programmes were teacher-led and one was peer-led. The provider for three programmes was not reported. Abstinence-only programmes were generally targeted at students younger than 14 years, with the exception of the Sex Can Wait programme, which consists of upper elementary, middle and high school components.

Overall, abstinence-only programmes appeared to be effective at increasing knowledge related to STIs in the short- to medium-term. Evidence was lacking on the longer term effects of abstinence-only programmes on knowledge. Eight studies examined effects on attitudes and values, and abstinence-only programmes were generally found to have had a positive effect on participants' beliefs and attitudes towards abstinence, and no programmes reported an adverse programme effect on attitudes towards abstinence or intentions to have sex. Results from the studies that examined intervention effects on parent-child communication indicated that abstinence-only programmes had non-significant or inconsistent effects on the initiation of sexual activity. Two studies, which reported outcomes relating to contraception, found no impact on contraception use. One abstinence-only programme, Success Express, had a negative effect; at follow-up intervention students reported greater lifetime sexual experience than controls. However, in a replication study of this programme there was no difference between intervention and control students on this measure.

Evidence statement 10

- 10 (a) There is moderate evidence from one RCT and two NRCTs to suggest that abstinence only programmes may have positive short- to long-term effects on knowledge relating to STIs. This evidence may only be partially applicable because the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- 10 (b) There is weak evidence from four NRCTs to suggest that abstinence only programmes may have short-term positive effects on attitudes and intentions towards sexual activity. There is weak evidence from one NRCT to judge the impact of abstinence programs on long-term intentions. This evidence may only be partially applicable because the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- 10 (c) There is moderate evidence from two RCTs and four NRCTs to suggest that abstinence only programmes may have no impact on the initiation of sexual behaviours or the maintenance of sexual abstinence. In addition, there is moderate evidence from one RCT and three NRCTs to suggest that abstinence only programmes may have no impact on or increase sexual activity. This evidence may only be partially applicable because the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- 10 (d) There is weak evidence from one RCT to suggest that abstinence only education programmes may have no impact on long-term pregnancy rates and contraception use. This evidence may only be partially applicable because the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.

Abstinence-plus programmes

A total of 24 articles were identified that reported on the evaluation of 15 abstinence plus programmes across 18 studies. Abstinence plus programmes were defined as those that reported an emphasis on abstinence as the safest way to avoid HIV/STI infection and pregnancy, but also promoted safer sex through the use of contraceptives. Of the 18 studies identified, ten were RCTs and eight were NRCTs. Two programmes, Safer Choices and YAPP, incorporated activities for parents and one programme, Protection Express, was exclusively peer-led. The programmes identified tended to target older adolescents (>14 years) or students across a range of ages, from the age of 13 upwards.

Thirteen studies examined intervention effects on sexual health knowledge, finding that abstinenceplus programmes were generally effective in improving sexual health knowledge in the short- and medium-term. In addition, three studies reported sustained long-term increases in knowledge among students exposed to abstinence plus programmes compared to controls. The results of 14 studies demonstrated that intervention effects on behavioural intentions, attitudes to sexual behaviour, and self-efficacy were inconsistent and there was no clear indication of the direction of effects. In addition, some programmes, such as Draw the Line/Respect the Line, had differing effects on male and female students. Across five programmes that incorporated skills building activities there were positive shortto medium-term effects on skills relating to sexual risk prevention. There was no indication of shortterm intervention effects on the initiation of sexual involvement and at the medium-term follow-up, programme effects were largely inconsistent with four studies reporting no effects on initiation of sexual intercourse. In addition, intervention effects on frequency of sexual intercourse, number of sexual partners and contraceptive use were found to be inconsistent or non-existent in the short-, medium- and long-term. Two studies examined the medium- to long-term effects of a HLM-SLT curriculum and Reducing the Risk, respectively, on pregnancy. Neither study identified significant programme effects on this outcome.

Evidence statement 11

- 11 (a) There is moderate evidence from three RCTs and three NRCTs to suggest that programmes that emphasise abstinence but that also promote safer sex, may produce short- to long-term term improvements in sexual health-related knowledge. This evidence may only be partially applicable because the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- 11 (b) There is inconsistent evidence from four RCTs and four NRCTs to determine the effects of abstinence-plus programmes on attitudes, behavioural intentions and self-efficacy relating to sexual behaviour.
- 11 (c) There is inconsistent evidence from one RCT and two NRCT to determine the long-term effects of abstinence-plus programmes on communication with parents. There is moderate evidence from one RCT and one NRCT to suggest abstinence-plus programmes that incorporate skills building activities may have positive, short and medium- to long-term effects on skills relevant to prevention. This evidence may only be partially applicable because the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- 11 (d) There is moderate evidence from five RCTs and five NRCTs to suggest that abstinence-plus programmes may not have a consistent short-, medium- or long-term impact on the initiation of sexual activity or the maintenance of abstinence. In addition, there is moderate evidence from four RCTs and two NRCTs to suggest that abstinence plus programmes may not have an impact on frequency of sexual activity and risky sexual behaviours.
- 11 (e) There is moderate evidence from six RCTs and three NRCTs to suggest that abstinence-plus programmes may not have a consistent impact on condom and other contraceptive use, and moderate evidence from one RCT and one NRCT to suggest that abstinence-plus programmes have no medium- to long-term impact on pregnancy rates.

HIV and sexual risk-reduction programmes

Overall, 11 studies were identified that examined HIV and sexual risk-reduction programmes. Studies were defined by their specific focus on HIV prevention and HIV risk-behaviour, sexual risk-behaviour or a combination of both. Of the 11 included studies, eight were RCTs, two were NRCT and one was based on a CBA study design. All programmes were delivered in school time however one programme included six computer-based activities to be completed outside of school time. The included studies focused on different ages and school years. One programme targeted students aged 12-14 years, but in general, programmes were targeted at older adolescents.

Seven studies examined intervention effects on general HIV and sexual health knowledge, finding significant effects on knowledge over the short-term. Two studies, which explored medium- to long-term effects on knowledge of HIV and contraception, reported significant effects. Outcomes relating to attitudes were reported by seven studies. Condom self-efficacy, perception of social norms and

condom use/prevention intentions were the outcomes most commonly reported across these studies but intervention effects were found to be inconsistent with no clear direction of effect. Short-term programme effects on personal and social skills were predominantly positive and included positive programme effects on behavioural prevention skills and condom negotiation skills. However, none of the included studies examined medium- to long-term impacts on skills. Intervention impacts on sexual initiation were explored in four studies, which overall indicated inconsistent effects on this outcome. Further studies indicated no impact on sexual activity or the numbers of sexual partners. Six studies showed positive short-term programme effects on condom use and protected intercourse, but longer term programme effects on contraception use appeared to be limited. Limited outcomes were presented on HIV/STI testing, alcohol or drug use and pregnancy.

Evidence statement 12

- 12 (a) There is moderate evidence from five RCTs, two NRCTs and one CBA study to suggest that HIV and sexual risk-reduction programmes can improve sexual health and HIV knowledge in the short-, medium-and long-term. This evidence may be only partially applicable to the UK as five of the studies were conducted in the USA, one in Italy and one in the Netherlands and may not be generalisable beyond the populations studied.
- 12 (b) There is mixed evidence from four RCTs and one NRCT that examined the effects of HIV and sexual risk-reduction programmes on young people's≥1(4 years) attitudes and values towards sexual health and alcohol. This evidence may be only partially applicable to the UK as studies were carried out in the USA, Italy and the Netherlands and may not be generalisable beyond the populations studied.
- 12 (c) There is moderate evidence from two RCTs, one NRCT and one CBA study to suggest that HIV and sexual risk-reduction programmes may improve personal and social skills including behavioural prevention skills and condom negotiation skills in the short-term. There was no evidence to determine the effects of HIV and sexual risk-reduction on personal and social skills in the medium- to long-term. This evidence may be only partially applicable to the UK as studies were carried out in the USA and Italy and may not be generalisable beyond the populations studied.
- 12 (d) There is moderate evidence from seven RCTs to suggest that HIV and sexual risk-reduction programmes may have no effects on sexual initiation, frequency of sexual activity or number of sexual partners. This evidence may be only partially applicable to the UK as studies were carried out in the USA, Norway, Italy and Netherlands and may not be generalisable beyond the populations studied.
- 12 (e) There is strong evidence from three RCTs, two NRCTs and one CBA study to suggest that HIV and sexual risk-reduction programmes can increase condom use or protected intercourse in the short- to medium-term. However, there was moderate evidence from two RCTs to suggest that the long-term effects of HIV and sexual risk-reduction programmes on contraceptive use may be limited. This evidence may be only partially applicable to the UK as

studies were carried out in the USA, Norway, and Sweden and may not be generalisable beyond the populations studied.

12 (f) There is moderate evidence from two RCTs to suggest that HIV and sexual risk-reduction programmes have no medium- to long-term effect on sexually transmitted infections, alcohol and drug use or on conceptions. As both studies were implemented in the USA findings may be only partially applicable to the UK and may not be generalisable beyond the populations studied.

Other school-based approaches

Seven studies were identified that reported on six different programme approaches relating to sexual health; one was an NRCT, five were CBA studies and one study was based on an interrupted time series design. Two studies examined school-based clinic programmes, two studies reported findings from one programme using baby simulators (Baby Think it Over), and three studies examined the effects of combined community and school-based programmes.

None of the studies examined intervention effects on knowledge or understanding. Three studies, which examined school-based health clinics and an infant simulation intervention, respectively, reported inconsistent effects on attitudes towards sexual health and only one study examined programme effects on personal and social skills. There were no effects of the Baby Think it Over infant simulation programme on any of the sexual behaviour measures examined. Three studies that examined a school and community partnership approach to tackling teenage pregnancy, generally found that although there were reductions in pregnancy rates among 14-17 year olds at the intervention sites, these reductions were not found to be significant compared to non-intervention sites. However, one study that examined pregnancy rates over 20 years concluded that the intervention had had a positive effect on teenage pregnancies. Two studies, which examined the long-term effects of school-based health centres, found that these programmes did not have consistent effects on participant's sexual behaviour.

Evidence statement 13

- 13 (a) There is inconsistent evidence from one NRCT and two CBAs to determine the effects of school-based clinics or an infant simulation intervention on knowledge, attitudes, values and personal and social skills.
- 13 (b) There is weak evidence from one NRCT and one CBA study to suggest that infant simulation programmes have no effect on health outcomes related to sexual health. This evidence may be only partially applicable to the UK as studies were carried out in the USA and may not be generalisable beyond the populations studied.
- 13 (c) There is weak evidence from one CBA studies and two ITS to suggest that a comprehensive school- and community-based approach to teenage pregnancy may produce modest reductions in teenage pregnancy rates. This evidence may be only partially applicable to the UK as studies were carried out in the USA and may not be generalisable beyond the populations studied.

13 (d) There is inconsistent evidence from two CBAs to determine the effects of school-based clinics on health outcomes related to sexual health.

Review of published economic evaluations

One study was identified that met the criteria for inclusion in the review of published economic evaluations. The study evaluated the cost-effectiveness and cost benefits of a school-based sex and relationships education programme, Safer Choices. Overall the net benefit of the Safer Choices programmes was \$174,276 and the benefit-cost ratio was 2.65, indicating that for every \$1 spent on the programme, \$2.65 were saved in medical and societal costs. The generalisbility of the study to a UK context was unclear as the data used in the evaluation was based on studies conducted in the USA, and utilised other US population estimates. However, the authors state that the methods and data used were conservative and it is possible that the intervention may be cost saving in a UK context.

Evidence statement 14

There is moderate evidence from one economic evaluation study to suggest that a sex and relationships education programme, Safer Choices, may be cost-effective and cost saving. This evidence may be of limited applicability to a UK context because cost and benefit estimates were based on data from studies conducted in the USA.

GENERAL HEALTH EDUCATION PROGRAMMES

Systematic reviews and meta-analyses

No systematic reviews or meta-analyses were identified for inclusion in the review of general health education programmes.

General health education programmes

Overall, nine studies reported on the evaluation of six general health education programmes that reported relevant alcohol and sexual education outcomes. Of the nine studies, eight were RCTs and one was a CBA study. All six programmes were primarily delivered in schools. Four programmes were solely school-based and two programmes incorporated both school and community elements. All six programmes targeted young adolescents aged less than 14 years.

One study examined programme effects on knowledge, finding no significant effects of Project Model Health on curriculum knowledge. However, there were positive medium-term effects of this programme on attitudes towards postponing sex and using contraceptives, and attitudes and intentions towards alcohol and other drugs. A second study found a medium-term impact on students' perceptions of peer attitudes towards alcohol and other drugs in those receiving the intensive condition of the Healthy for Life Programme. However, there was no longer term effect of this programme. None of the studies examined intervention effects on personal and social skills. Two programmes, the Reach for Health curriculum and the Aban Aya project, which incorporated community components, had positive effects on sexual behaviour in the medium- to long-term. Four

school-based programmes had either no effect or harmful effects on sexual behaviours and alcohol use.

Evidence statement 15

- 15 (a) There is inconsistent evidence from one RCT and one CBA study to determine the effect of general health education programmes on knowledge, attitudes and values relating to sexual health and alcohol use.
- 15 (b) There is moderate evidence from three RCTs to suggest that general health education programmes incorporating an intensive community intervention element in conjunction with a curriculum base may have a positive effect on sexual behaviour and substance use. The evidence may only be partially applicable to the UK as programmes were implemented in the USA and focused primarily on black and minority ethnic groups. As such their generalisbility may be limited to the populations studied.
- 15 (c) There is moderate evidence from five RCTs to suggest that curriculum-based, general health education programmes have no impact on, and in some cases may have a negative impact on, sexual behaviours and alcohol use. Studies were based on two programmes conducted in the USA and Australia and therefore the evidence may not be generalisable beyond the populations studied.

Review of published economic evaluations

No published economic evaluation studies were identified for inclusion in the review of general health education programmes.

CONCLUSIONS

Alcohol and drug education programmes

The evidence suggests that classroom-based programmes, regardless of whether the focus is on alcohol alone or as one of a number of substances, may have beneficial effects on alcohol-related knowledge, particularly in the short- to medium-term. Programme effects on attitudes and values were mixed and inconsistent across a range of intervention approaches and the evidence was insufficient to draw conclusions about the impact of these programmes on personal and social skills. Overall, the findings of the review of alcohol and drug education programmes highlight that there is a lack of clear, medium- to long-term evidence for the effectiveness of school-based alcohol education programmes on health and social outcomes relating to alcohol use. In addition, the applicability to a UK context of those programmes that have demonstrated effectiveness, such as LST, STARS for Families and Project SPORT, is limited. There is lack of evidence on which to draw conclusions about the cost-effectiveness of alcohol and drug education programmes. Further good quality UK-based research of promising or novel intervention approaches, including assessment of cost-effectiveness, is required in order to improve the evidence base on which to make UK-based policy and practice recommendations for PSHE focusing on alcohol education.

Sex and relationships education programmes

There were consistently positive programme effects on acquisition of sexual health knowledge, across the SRE education programmes that examined this outcome, regardless of whether the programme emphasised abstinence or not. A range of outcomes were reported with regards to attitudes and values and programme effects were mixed or inconsistent across these measures. It was therefore not possible to draw unequivocal conclusions about the impact of SRE programmes on attitudes and values relating to sexual health. The evidence suggests that while abstinence-only programmes have no effects on health and social outcomes related to sexual health, programmes that incorporate information on safer sex and contraceptive use may have positive, but limited effects on the prevention of sexual risk behaviours, in particular limited effects on contraceptive use. Although the applicability of some of these programmes to a UK context is limited, these conclusions are supported by the evidence drawn from studies conducted in the UK. There is lack of evidence on which to draw conclusions about the cost-effectiveness of SRE programmes. Further good quality UK-based research of promising or novel intervention approaches, including assessment of cost-effectiveness, is required in order to build on the evidence base on which to make UK-based policy and practice recommendations for PSHE focusing on SRE.

General health education programmes

There was a lack of evidence on which to draw clear conclusions about the effects of the general health education programmes on knowledge, and attitudes and values relating to alcohol use and sexual health. The evidence suggests that general health education programmes, which incorporate an intensive community intervention element in conjunction with a curriculum base may have a positive effect on sexual behaviour and substance use. There were no effects, or in some cases harmful effects on sexual health and alcohol use outcomes for curriculum-based, general health education programmes. In addition, the applicability of these programmes to a UK context was limited.

1 Introduction

1.1 Aims and objectives

This review was undertaken to support the development of guidance by the National Institute for Health and Clinical Excellence (NICE) aimed at promoting school, college and community-based personal, social and health education (PSHE), with particular reference to sexual health behaviours and alcohol use behaviours. As such, the review sought to identify effective and cost-effective interventions and programmes that focus on modifying attitudinal and behavioural outcomes in relation to alcohol use and sexual health.

1.2 Research question

The following key questions were addressed:

- 1. What services, interventions, programmes, policies or strategies for children and young people aged 11 years and above are effective and cost effective in contributing to the achievement of the "Every child matters" outcomes for PSHE, related to sexual health and alcohol?
- 2. What elements/components of those services, interventions, programmes, policies or strategies for children and young people aged 11 years and above are effective and cost effective in contributing to the achievement of the "Every child matters" outcomes for PSHE, related to sexual health and alcohol?

2 Background

Alcohol use and sexual health in addition to being independent issues of public health concern (see section 2 in Jones et al., 2009), are also interrelated public health issues. Research findings suggest that generally alcohol use per se, and binge drinking in particular, is associated with being sexually active (Miller et al., 2007; Ramisetty-Mikler et al., 2004). Specifically, research has indicated that early regular alcohol consumption (usually before age 16 years) is associated with early onset sexual activity (Choquet et al., 1992; Robertson and Plant, 1988). Further research implies that early alcohol consumption and a high level of alcohol consumption results in an increased likelihood of a higher number of sexual partners (Lowry et al., 1994; Ramisetty-Mikler et al., 2004). Alcohol use at first sexual intercourse has also been associated with greater risk-taking behaviours such as lower levels of condom use (Dye and Upchurch, 2006); and studies have found lower levels of condom use in those who binge drink or who have alcohol-related problems (Kim-Godwin et al., 2007). Furthermore, alcohol use has been reported as a substance used to facilitate sexual encounters (Bellis et al., 2008). Thus, the misuse of alcohol has been linked to unprotected sex (Hibell et al., 2004) and to some extent to regretted sex (Hibell et al., 2004; Bellis et al., 2008) and forced sex (Miller et al., 2007) also. In addition, alcohol consumption, particularly increased levels of alcohol consumption, has been associated with females becoming pregnant or males making a female pregnant (Miller et al., 2007). Recent work has further highlighted the relationship between teenage pregnancy and alcohol consumption (Bellis et al., 2009), with findings indicating a link between alcohol-related hospital admissions and conception rates. Even after accounting for deprivation data showed the highest conception rates to be in areas with the highest alcohol-related hospital admissions. This emphasises the importance of SRE and alcohol education, particularly in secondary schools, prior to the onset of sexual activity and alcohol consumption and the uptake of risky behaviours.

Many young people under age 18 are aware that they consume more alcohol now than same age people would have ten years ago (Department of Health, 2007). Furthermore, alcohol is associated with crime and anti-social behaviour which is estimated to cost £7.3 billion; with an annual cost of £6.4 billion due to alcohol-related loss of productivity in the workplace; and the human and emotional impact of alcohol-related crime estimated to cost £4.7 billion (The Strategy Unit, 2004). It is important to measure the health costs in addition to the social costs of alcohol use. In 2008 research was carried out to measure the impact of alcohol on health. Findings revealed that males aged 16-24 years have the highest proportion of deaths attributable to alcohol (26.6%). Furthermore, hospital admission for mental and behavioural disorders due to alcohol use were the most common cause of admission in both males and females aged under 45 years in England (Jones et al., 2008).

In addition to the long-term negative health outcomes related to alcohol there is also the contribution of poor sexual health which disproportionally affects young people. Rates of sexually transmitted infections for the top five infections (Chlamydia, gonorrhoea, syphilis, herpes and warts) are increasing. Infections in females aged 16-19 years have increased 15% from 2004 to 2008 and in young males aged 16-19 years numbers of infections have increased by 23% (Health Protection

Agency, 2009). Limited information exists on the wider financial and health costs of poor sexual health. However, they are likely to be considerable. Thus separately, preventing alcohol use and poor sexual health can have social and health benefits and addressing the two together, beginning in PSHE classes, could compound the benefits.

2.1 Personal, Social Health education (PSHE)

(For details of PSHE in primary schools please see section 2.2 in Jones et al., 2009).

Progress is currently being made towards making PSHE statutory within UK schools. Currently Key Stages 3 and 4 there are themes to the curriculum that ought to incorporate points as follows:

National Curriculum Science

Key stage 3

- That fertilisation in humans is the fusion of a male and female cell.
- About the physical and emotional changes that take place during adolescence.
- About the human reproductive system, including the menstrual cycle and fertilisation.
- How the foetus develops in the uterus how the growth of bacteria and the replication of viruses can affect health.

Key stage 4

- The way in which hormonal control occurs, including the effects of sex hormones.
- Some medical uses of hormones, including the control and promotion of fertility.
- The defence mechanisms of the body.
- How sex is determined in humans.

Sex and relationship education curriculum and standards guidance (Department for Education and Employment, 2000) states that at secondary school level, sex and relationship education (SRE) should contribute to PSHE education by preparing young people for and adult life and ensuring that they can:

- Develop positive values and a moral framework that will guide their decisions, judgements and behaviour;
- Be aware of their sexuality and understand human sexuality;
- Understand the arguments for delaying sexual activity;
- Understand the reasons for having protected sex;
- Understand the consequences of their actions and behave responsibly within sexual and pastoral relationships;
- Have the confidence and self-esteem to value themselves and others and respect for individual conscience and the skills to judge what kind of relationships they want;
- Communicate effectively;

- Have sufficient information and skills to protect themselves and, where they have one, their partner from unintended/unwanted conceptions, and sexually transmitted infections including HIV;
- Avoid being exploited or exploiting others;
- Avoid being pressured into unwanted or unprotected sex;
- Access confidential sexual health advice, support and if necessary treatment; and
- Know how the law applies to sexual relationships.

Alcohol education is also included in the wider provision for PSHE and according to curriculum guidance (Qualifications and Curriculum Authority, 2007) it ought to teach the facts and laws about drug, alcohol and tobacco use and misuse, and the personal and social consequences of misuse for themselves and for others. The Department for Children Schools and Families (2008) recognises the need to address the issue of high quality and consistent sex and relationship education within further education institutions such as colleges, particularly for those post-16 years and those aged 14-19 years in non-school education. It also recognised that, since further education colleges are the main provider of post-16 education for those with learning difficulties and disabilities it is important to address inequalities in service provision as their sexual health needs may have previously been overlooked (Department for Education and Skills, 2007).

2.2 Government policy

(For further details of government policies related to PSHE please see section 2.3 in Jones et al., 2009).

Sex and relationship and alcohol education are intrinsic to the whole school approach promoted in the healthy schools programme (Department of Health, 2005). In order to gain healthy schools status for PSHE the Department of Health (2005) states that schools must demonstrate that they have met the criteria in the following:

- uses the PSHE framework to deliver a planned programme of PSHE, in line with DfES/Qualifications and Curriculum Authority (QCA) guidance;
- monitors and evaluates PSHE provision to ensure the quality of teaching and learning;
- assesses pupils' progress and achievement in line with QCA guidance;
- has a named member of staff responsible for PSHE provision with status, training and appropriate senior management support within the school;
- has up to date policies in place developed through wide consultation, implemented, and monitored and evaluated for impact – covering sex and relationship education, drug education and incidents, child protection, and confidentiality;
- has an implemented non-smoking policy, or is working towards being smoke free by September 2007;

- involves professionals from appropriate external agencies to create specialist teams to support PSHE delivery and to improve skills and knowledge, such as a school nurse, sexual health outreach workers and drug education advisers;
- has arrangements in place to refer pupils to specialist services who can give professional advice on matters such as contraception, sexual health and drugs;
- uses local data and information to inform activities and support important national priorities such as reducing teenage pregnancies, sexually transmitted infections and drug/alcohol misuse;
- ensures provision of appropriate PSHE professional development opportunities for staff such as the Certification Programmes for teachers and nurses offered by DH/DfES; and
- has mechanisms in place to ensure all pupils' views are reflected in curriculum planning, teaching and learning, and the whole school environment, including those with special educational needs and specific health conditions, as well as disaffected pupils, young carers and teenage parents.

An extended schools approach to education promotes the general, sexual and substance-related health of adolescents aged 11 to 19 years through improving access to contraceptive and sexual health advice and targeted specialist substance use prevention services (Department for Educational and Skills, 2006). Sexual health services in education settings that are most successful are those that bring together a range of partners including those in the key areas of health, education, social services, and youth support services (Department for Education and Skills, 2007). Sexual health services includes a wide variety of people involved in setting up services from PCT and local authorities, further education, and the surrounding community.

The Department of Health (2009) provided additional funding of £26.8m in 2008 to support improved access to contraceptive services. Progress has been made relating to the provision of young people's sexual health clinics linked to educational institutions. However, the Sex Education forum (Emmerson, 2008 and 2007) carried out a national mapping survey of sexual health services in educational settings. Findings showed that 29% of secondary schools have on-site sexual health services, with pupil referral units providing a greater level of on-site services (34.4%). However, a much larger proportion (72%) of further education settings, including sixth-forms provided some sexual health service for their students.

3 Methodology

3.1 Search strategy

Systematic searches of electronic databases and websites were undertaken to identify studies that examined the effectiveness and/or cost-effectiveness of alcohol education and/or SRE delivered in isolation or as part of a wider programme of study such as PSHE or its equivalents. Searches were conducted across a range of health, education and social care databases as shown in Box 3.1.

Box 3.1. Health, education and social care databases

- ASSIA (Applied Social Science Index and Abstracts)
- CINAHL (Cumulative Index of Nursing and Allied Health Literature)
- Database of Abstracts of Reviews of Effectiveness (DARE)
- The Cochrane Library
- EMBASE
- ERIC
- British Education Index
- Australian Education Index
- HMIC (or Kings Fund catalogue and DH data)
- MEDLINE
- PsycINFO
- Sociological Abstracts
- Social Science Citation Index
- EPPI Centre databases
- The Campbell Collaboration
- C2-SPECTR & C2-PROT Campbell Collaboration

Economic evaluation studies were identified by searching the following major health economics databases:

- NHS Economic Evaluations Database (NHS EED)
- EconLit

3.2 Inclusion and exclusion criteria

3.2.1 Population

Studies were eligible for inclusion if they included children aged 11 to 19 years old in full time education. This included children in secondary schools, sixth form and further education colleges and those receiving education outside of a mainstream school setting including:

- Children receiving home education
- Children receiving education in pupil referral units

Studies were eligible for inclusion if they were undertaken in the UK, Western Europe, Australia, New Zealand, Canada and the USA.

3.2.2 Interventions

Studies were eligible for inclusion if they examined interventions that focused on SRE and/or alcohol education. Relevant intervention approaches included:

- Interventions and programmes agreed, planned or delivered by teachers or other professionals
- Interventions and programmes planned and/or delivered by external agencies and individuals
- Intervention involving the 'informal' and extended school curriculum
- Peer led education

3.2.3 Comparator(s)

Studies were eligible for inclusion if they compared the intervention of interest against a no intervention control or against another intervention approach.

3.2.4 Outcomes

Studies from outside of the UK were eligible for inclusion only if they examined the primary outcome of interest:

• Health and social outcomes relating to alcohol use and sexual health

The following secondary outcomes were assessed but, unless the study was conducted in the UK, only where a study reported primary outcomes of interest:

- Knowledge and understanding
- Personal and social skills
- Attitudes and values

3.2.5 Study design

Systematic reviews, randomised controlled trials, controlled non-randomised studies and, controlled before and after studies that compared a school-based intervention against no intervention or another type of intervention were eligible for inclusion in the assessment of effectiveness.

Studies were eligible for inclusion in the assessment of cost-effectiveness if they were economic evaluations conducted alongside trials, modelling studies and analyses of administrative databases. Only full economic evaluations that compared two or more options and considered both costs and consequences (including cost-effectiveness, cost utility and cost-benefit analyses) were included.

3.3 Data extraction strategy

All titles and abstracts retrieved were screened independently by two reviewers (LJ, GB, JD, HS, KS) according to the inclusion/exclusion criteria described above. Disagreements were resolved through consensus and where necessary a third reviewer was consulted. Relevant articles were retrieved in full and full text screening was undertaken independently by two reviewers (LJ, GB, HS, OW).

One reviewer (LJ, GB, HS, OW) independently extracted and assessed the quality of the individual studies into an Access database. All data extraction and quality assessment were independently checked for accuracy by a second reviewer. The results of the data extraction are presented in an addendum to this report.

3.4 Quality assessment strategy

The quality of the studies was assessed according to criteria set out in the NICE Centre for Public Health Excellence Methods Manual (2009). Each of the effectiveness and cost-effectiveness studies was graded using a code, ++, + or – based on the extent to which the potential sources of bias had been minimised:

- ++ All or most of the criteria have been fulfilled. Where they have not been fulfilled the conclusions are thought very unlikely to alter.
- + Some of the criteria have been fulfilled. These criteria that have not been fulfilled or not adequately described are thought unlikely to alter the conclusions.
- Few or no criteria have been fulfilled. The conclusions of the study are thought likely or very likely to alter.

Results of the quality assessment are presented in Appendix 4 and 5.

3.5 Methods of analysis/synthesis

3.5.1 Effectiveness studies

The results of the data extraction and quality assessment for each study of effectiveness are presented in structured tables and as a narrative summary. The possible effects of study quality on the effectiveness data and review findings are also discussed within the text of the review.

Studies are grouped according to intervention focus (e.g. alcohol, substance use, abstinence) and for alcohol studies by approach (e.g. single session, classroom-based) and the outcomes examined. Where sufficient data were available, intervention effect sizes have been calculated and presented as odds ratios (OR) for dichotomous data and as mean differences for continuous data. Where study authors reported intervention effect sizes, these have been extracted directly as risk ratios (RR) or OR

as reported in the original publication. Forest plots were generated for single studies using RevMan (version 5) and are presented in an addendum to this report. Heterogeneity between the included studies was assessed by considering differences in (a) the study population, (b) intervention approach, (c) outcome measures, and (d) study quality. However, given the anticipated heterogeneity between the included studies it was judged to be unlikely that pooling would be appropriate or feasible.

3.5.2 Published economic evaluations

Details of each identified published economic evaluation, together with a critical appraisal of its quality were presented in structured tables and as a narrative summary. For economic studies conducted alongside trials, the validity of the included studies was assessed by considering the source of the resource use and effectiveness data, the methods used to measure and calculate costs, the methods of analysis used and the generalisbility of the results to the UK population.

4 Summary of study identification

4.1 Review of effectiveness and cost-effectiveness

A total of 11,621 references were identified from the literature searches. Following screening of titles and abstracts, 936 articles were identified as potentially relevant and attempts were made to source the full text articles. Of these articles, 163 were not available, seven were conference abstracts and 12 were foreign language articles. These studies were therefore not subject to further screening and a total of 724 full-text articles were screened against the inclusion criteria for the study. The process of study selection is summarised in Figure 1.



*Includes a further 49 studies included in Jones et al (2007)

**Not identified in the literature searches conducted for this review

Figure 4.1. Process of study selection

4.1.1 Included studies

A total of 143 articles met the criteria for inclusion. A further 60 articles that were included in a systematic review of the effectiveness and cost-effectiveness of interventions in primary and secondary schools to prevent and/or reduce alcohol use, previously conducted by the lead author and colleagues (Jones et al., 2007), but not identified in the searches conducted for this review, also met the criteria for inclusion. Therefore a total of 203 articles were identified for inclusion in the review of effectiveness and cost-effectiveness. Of these, 119 articles focused on alcohol and drug education programmes, 75 articles focused on sex and relationships education and the remaining nine articles examined general education programmes which targeted both alcohol, and sex and relationships. Twenty-three articles were systematic reviews and/or meta-analyses, 152 studies were based on experimental designs of which 111 used random assignment. Twenty-five observational studies were identified for inclusion including 22 controlled before and after (CBA) studies, two interrupted time series (ITS) and one uncontrolled before and after study¹ (UBA). Three economic evaluation studies were also identified.

Section	Total	SR/MA	RCT	NRCT	СВА	Other	Economic evaluation	
Alcohol	119	14	72	19	12	0	2	
SRE	75	9	31	22	9	3*	1	
General	9	-	8	-	1	0	-	
Total 203 23 111 41 22 3 3								
*one UBA study and two ITS								

Table 4.1. Summary of study designs identified for inclusion

4.1.2 Excluded studies

A total of 581 articles did not meet the criteria for inclusion in the review and were excluded for the following reasons:

- Study design did not meet design criteria for inclusion in the review, n=320
- Population targeted by the intervention(s) did not meet the review criteria, n=202
- Intervention examined was not based in a school setting, n=38
- Intervention or intervention was not alcohol education or SRE related, n=17
- Intervention targeted at risk or high risk population, n=4

References for the excluded studies are presented in Appendix 2.

¹ This study was included because it was related to two UK-based RCTs of the RIPPLE programme.

5 Alcohol and drug education programmes

A total of 119 articles met the criteria for inclusion in the review of alcohol and drug education programmes. Fourteen articles were systematic reviews and meta-analyses, 103 articles reported on the evaluation of an alcohol or substance use education programme, and two articles were economic evaluation studies. Of the 103 articles, 74 reported on evaluations of classroom-based programmes; 20 of which were alcohol specific, and 54 of which focused on substance use including alcohol. Also identified were 15 articles that reported on evaluations of brief or single session interventions, nine articles reporting on two multicomponent school- and community-based programmes and five articles reporting on evaluations of peer support and/or counselling programmes.

5.1 Systematic reviews and meta-analyses

5.1.1 Overview of evidence identified

Fourteen systematic reviews and meta-analyses were identified for inclusion, which examined interventions targeting alcohol and other substance use.

5.1.2 Quality assessment

Four reviews (Foxcroft et al., 2002; White et al., 2004; Spoth et al., 2008; Gottfredson & Wilson, 2003) were rated good quality (++ rating), eight reviews (Coggans et al., 2003; Dusenbury et al., 1997; Loveland-Cherry, 2003; Skara & Sussman, 2003; Tobler, 1993; Tobler et al., 1997, 2000; Werch & Owen, 2002) were rated moderate quality (+ rating) and the remaining two reviews (Bruvold, 1990; Cuijpers, 2002) were rated poor quality (- rating). All 14 reviews addressed an appropriate and clearly focused question, but descriptions of the methodology used were poorly reported in four reviews (Bruvold, 1990; Coggans et al., 2003; Cuijpers, 2002; Dusenbury et al., 1997) as details about data extraction and quality assessment were not reported. In one review (Bruvold, 1990), details were also lacking on how studies were located for inclusion and so it was not possible to determine whether the literature review was sufficiently rigorous. For four further reviews (Cuijpers, 2002; Tobler, 1993; Tobler et al., 1997, 2000) details of the literature search were poorly reported. Five reviews (Foxcroft et al., 2002; Loveland-Cherry, 2003; Tobler et al., 1997, 2000; Gottfredson and Wilson, 2003) fully assessed and took study quality into account when reporting findings and this was judged to have been adequately done in four further reviews (Dusenbury et al., 1997; Tobler, 1993; White et al., 2004; Spoth et al., 2008). The remaining reviews (Bruvold, 1990; Coggans et al., 2003; Cuijpers, 2002; Skara & Sussman, 2003; Werch & Owen, 2002) did not adequately assess study quality. The majority of the reviews presented findings in a narrative synthesis, and of the studies that statistically combined studies the methods used appeared to have been suitable.

5.1.3 Findings

5.1.3.1 Classroom-based programmes

Seven reviews (Loveland-Cherry, 2003; Tobler, 1993; Tobler et al., 1997, 2000; Cuijpers, 2002; Gottfredson and Wilson, 2003) examined the effectiveness of school-based programmes delivered within classroom-based curriculums. Four articles were identified that reported on a series of meta-analyses undertaken by Tobler (1993; Tobler et al., 1997; Tobler et al., 2000; Black et al., 1998).

Loveland-Cherry (2003) reported that among programmes that lasted for three months or less, decreases in potential drinking were reported. However, overall the author reported that the effects of the programmes examined were inconsistent, relatively small and short-lived, and that few studies demonstrated long-term results. The most recent meta-analysis undertaken by Tobler and colleagues (2000; SR +) reported that for programmes targeting alcohol use, there was a small, but nonsignificant difference between non-interactive and interactive programmes, in favour of interactive programmes. In a subset of high-quality programmes, the difference between non-interactive and interactive programmes targeting alcohol was found to be significantly in favour of interactive programmes. Based on the evidence reviewed, Cuijpers (2002) proposed evidence-based quality criteria for drug prevention programmes. They stated that programmes should be based on welldesigned scientific research demonstrating effectiveness, use interactive delivery methods, be based on the social influence model and focus on normative education, commitment not to use substances, and intentions not to use. In addition, the authors reported that adding community interventions, lifeskills training and/or the use of peer leaders strengthened the effects of school-based interventions. Gottfredson and Wilson (2003) combined data from 94 studies of school-based prevention activities. Assessment of the relationship between programme effectiveness and age, suggested an advantage for programmes delivered to middle/junior high school students (equivalent to ages 11-15 years) compared to programmes targeting elementary or senior school age students (although the difference was not statistically significant). Neither programme length nor duration was found to be predictive of student outcomes. Examination of providers revealed a positive benefit of the involvement of peers as leaders but this benefit disappeared in studies which combined peer and teacher involvement.

5.1.3.2 Multicomponent programmes

Five reviews (Dusenbury et al., 1997; Foxcroft et al., 2002; Gottfredson & Wilson, 2003; Skara & Sussman, 2003; Spoth et al., 2008) examined the effects of a range of substance use prevention programmes. Two reviews (Foxcroft et al., 2002; Spoth et al., 2008) specifically focused on interventions aimed at the prevention or reduction of alcohol use.

Foxcroft et al (2002) reported that it was difficult to draw firm conclusions about the effectiveness of interventions delivered over the short- and medium-term because of mixed findings. Three programmes were found to have long-term effects on alcohol use, Botvin's Life Skills Training (LST), a culturally tailored curriculum for Native Americans and the family-based Strengthening Families Programme. The Strengthening Families Programme was highlighted as showing particular promise as an effective intervention. Based on a reanalysis of data to account for participants lost to follow-up,

the authors calculated a number needed to treat (NNT) of 9 for three alcohol use behaviours: ever used alcohol (95% CI: 5 to ∞), ever used alcohol without permission (95% CI: 5 to 160), and ever been drunk (95% CI: 5 to 327). This indicated that for every nine individuals who received the intervention, 4 years later there would be one fewer young person reporting that they had ever used alcohol, used alcohol without permission or ever been drunk. Ten drug prevention curricula were included in the review by Dusenbury and colleagues (1997). The authors reported that the substance use curricula reviewed had been shown to effectively reduce substance use, and that in particular LST has been shown to have effects into young adulthood. Six curricula (Alcohol Misuse Prevention Project, Growing Healthy, Know Your Body, LST, Project Northland and STAR) were shown to have intervention effects lasting for at least two years after the pretest. The authors report that two programmes (Project Alert and DARE) did not appear to have sustained effects on drug use, although they had variable success at reducing substance use in the short-term. Skara and Sussman (2003) reported very little data that was specific to alcohol use, but two studies that reported on the Healthy School and Drugs Project and the Midwest Prevention Project, respectively, provided sufficient information for the calculation of the percentage reduction in weekly alcohol use rates from baseline to follow-up (for experimental conditions relative to control conditions). Long-term reductions were 6.9% and 11.7% for the two programmes, respectively. Of six studies assessing alcohol or cannabis use, the authors reported that five had maintained long-term reductions in alcohol use at the end of the study period. Spoth et al (2008) examined the literature on preventive interventions for underage drinking. Forty-one interventions met the criteria for inclusion in the review; 12 met the criteria for "most promising" evidence and 29 met the criteria for "mixed or emerging" evidence. For young people aged 10-15 years eight school-based or multicomponent interventions were highlighted as "most promising": Keepin it REAL, Midwest Prevention Project, Project Northland, BiCultural Competence Skills programme, Healthy School and Drugs, LST, Project ALERT, School Health and Harm Reduction Project.

5.1.3.3 Specific intervention approaches

Bruvold (1990) and Coggans et al (2003) reviewed the effectiveness of specific intervention approaches, the California school-based risk reduction programme and Botvin's Life Skills Training (LST) programme, respectively.

Coggan et al (2003) mainly focused on the effects of LST on illicit drug use outcomes. The authors commented that in terms of alcohol use, effects of the LST programme "can be positive if relatively modest in scale" but that completeness of delivery and fidelity were important in maximising the effects of LST. The authors also reported that there was some evidence that the positive impact of LST on alcohol and smoking could reduce likelihood of progression to illicit drug use, but that the evidence was not conclusive. The authors also reported that a well-implemented LST programme could positively affect knowledge, attitudes and behaviour with respect to alcohol use. Bruvold (1990) calculated study effect sizes for alcohol behavioural outcomes. Programmes based on rational theory were found to have a smaller effect on alcohol use behaviours than programmes based on developmental theory (effect size [ES] 0.02 vs. 0.20, respectively). However, given the small number

of developmental-based programmes included in the meta-analysis (n=2) the authors stated that this result should be interpreted with caution.

5.1.3.4 Other reviews

Two reviews examined other aspects of the effectiveness of alcohol and drug education programmes. Werch and Owen (2002) undertook a systematic analysis of published studies to determine whether iatrogenic effects occurred, and where harmful effects occurred, under what circumstances. White et al (2004) undertook a review of the role of external contributors in delivering substance use education.

Werch and Owen (2002) identified 17 studies for inclusion in their review of iatrogenic programme effects. Of these studies, 47% with negative substance outcomes were focused on the prevention of alcohol use. The alcohol prevention programmes reviewed (n=8) resulted in 19 harmful effects, or an average of 2.4 negative outcomes for every programme. The majority were non-behavioural measures (58%); examples provided by the authors included increased estimates of alcohol offers, pro-alcohol attitudes and increased expectations about drinking in the future. For all programmes overall, the greatest number of negative programme effects was associated with social-influence based programmes (59%), and the next largest harmful were programmes associated with knowledge/attitudes/values models (23%). Werch and Owen (2002) suggested that the social influence-based strategy of teaching students' resistance skills training may be the element of these studies that results in negative effects.

White et al (2004) identified evaluations on the use of 16 different types of external contributors. The authors found that there was no evidence to suggest that any particular agency or external contributor was more effective (in terms of being well received by pupils and teachers and/or leading to knowledge, attitude or behaviour change in the pupils) than any other in providing drug education. There was some evidence that the DARE programme could achieve short-term changes in knowledge, attitudes and behaviour, but that these effects were found to decay rapidly. However, the authors report that police officers could provide a valuable contribution (e.g. bringing specialist knowledge) to drug education when used in a supplementary role. Peer-delivered education, theatre in education and Life Education Centre programmes were evaluated but there was insufficient evidence to judge their effects on behaviour change. The authors reported that programmes delivered by nurses were shown to produce short-term knowledge gains, and to have effects on knowledge and alcohol use for up to 6 months.

5.1.4 Summary and evidence statements

A total of 14 systematic reviews and meta-analyses were identified for inclusion. The majority of the reviews identified examined the effectiveness of programmes targeting substance use including alcohol, and only three reviews (Foxcroft et al., 2002; Loveland-Cherry, 2003; Spoth et al., 2008) focused specifically on the prevention of alcohol use.

Foxcroft et al (2002) found that there was no consistent evidence to determine which programmes were effective over the short to medium-term, but highlighted three programmes which were effective over the longer term. These included the family-based, Strengthening Families programme, and two

school-based programmes, Botvin's LST and a culturally-focused curriculum for Native American students. Spoth et al (2008) highlighted promising evidence from six additional programmes, Keepin it REAL, the Midwest Prevention Project, Project Northland, Healthy School and Drugs, Project ALERT, and the School Health and Harm Reduction Project. Two reviews (Cuijpers, 2002; Gottfredson and Wilson, 2003) identified evidence to suggest that peer leaders strengthened the effects of school-based interventions, although Gottfredson and Wilson (2003) found that any beneficial effects of peer involvement were lost when they were combined with teacher-led activities. White et al (2004) did not find any evidence to suggest that any particular agency or external contributor was more effective than another.

Evidence statement 1

- 1 (c) There is strong evidence from two systematic reviews¹ to suggest that a secondary-level school-based programme, Botvin's LST, can produce long-term reductions (greater than 3 years) in alcohol use. Other promising intervention approaches include: Keepin it REAL, the Midwest Prevention Project, Project Northland, Healthy School and Drugs, Project ALERT, and the School Health and Harm Reduction Project.
- 1 (d) There is moderate evidence from two systematic reviews² to suggest that programmes delivered by peer leaders may be more beneficial than programmes delivered by teachers or other contributors. However, there is strong evidence from one systematic review³ to suggest that the positive benefits of peer involvement may disappear if teachers were also involved in delivery.
- ¹ Foxcroft et al., 2002; 2003 (SR ++); Spoth et al., 2008 (SR ++)
- ² Gottfredson and Wilson, 2003 (SR ++); Cuijpers, 2002 (SR -)
- ³ Gottfredson and Wilson, 2003 (SR ++)

Author (Year)	Design	Inclusion/exclusion	Number of studies	Findings
Bruvold 1990	SR -	Outcome evaluations of Californian programmes delivered to 4th to 8th graders	8 studies	Programmes based on rational model had more impact on knowledge (ES = 0.61) than developmental programmes (ES = 0.26); less effect on attitudinal outcomes (ES = -0.01; 0.04). Rational based programmes had less of an effect on alcohol use behaviours than developmental programmes (ES = 0.02 vs. 0.20).
Coggan et al., 2003	SR +	Evaluations of Botvin's LST programme	45 reports	The authors reported that there was some evidence that positive impact of LST on alcohol and smoking could reduce likelihood of progression, but the data were not conclusive. A well-implemented LST programme can positively affect knowledge, attitudes and behaviour with respect to smoking and alcohol use.
Cuijpers 2002	SR -	Universal school-based drug prevention aimed at tobacco, alcohol and illegal drugs	30 studies	Evidence-based quality criteria for programmes proposed, see evidence table for full details.
Dusenbury et al., 1997	SR +	Primary prevention of alcohol and/or drug use; classroom-based curricula	47 programmes	Eight programmes were shown to be effective at reducing tobacco or drug use in at least some studies. LST has been shown to have effects into young adulthood. In addition, six curricula (Alcohol Misuse Prevention Project, Growing Healthy, Know Your Body, LST, Project Northland and STAR) were shown to have intervention effects lasting for at least two years after the pretest. Project Alert and DARE did not appear to have sustained effects on drug use
Foxcroft et al., 2002, 2003	SR ++	Psychosocial and educational interventions aimed at the primary prevention of alcohol misuse by young people aged up to 25 years	56 studies	Twenty studies demonstrated evidence of ineffectiveness. No firm conclusions about the effectiveness of prevention in the short and medium-term were possible. But over the longer term (>3 years), the Strengthening Families Programme showed more promise as an effective prevention intervention.
Gottfredson & Wilson, 2003	SR ++	Interventions to reduce problem behaviours that measured effects on alcohol and other drug use	94 studies	Reported an advantage for programmes delivered to middle/junior high school students (equivalent to ages 11-15 years) compared to programmes targeting elementary or senior school age students (although the difference was not statistically significant). Neither programme length nor duration was found to be predictive of student outcomes. Positive benefit of peer involvement, but this disappeared when combined with teacher involvement.
Loveland-Cherry, 2003	SR +	Interventions that prevent alcohol use in children and adolescents.	73 studies	Overall the effects of the programmes were inconsistent, relatively small and short lived. Few studies have demonstrated long-term results. An increase in alcohol knowledge was reported by several studies.

	Table 5.1. Alcohol	education:	systematic	reviews	and	meta-anal	yses
--	--------------------	------------	------------	---------	-----	-----------	------

Author (Year)	Design	Inclusion/exclusion	Number of studies	Findings
Skara and Sussman 2003	SR +	Controlled evaluations of school- and community- based prevention programmes providing at least a 2-years follow-up.	25 studies	Very little data specific to alcohol use. Two studies provided sufficient information for calculation of the % reduction in weekly alcohol use rates from baseline to follow-up; long-term reductions were 6.9 and 11.7%, respectively.
Spoth et al., 2008	poth et al., 2008 SR ++ Interventions that reduce problem behaviours in children and include outcomes related to substance use 41 studies		Eight school-based or multicomponent interventions were highlighted as "most promising": Keepin it REAL, Midwest Prevention Project, Project Northland, BiCultural Competence Skills programme, Healthy School and Drugs, LST, Project ALERT, School Health and Harm Reduction Project.	
Tobler 1993; Tobler et al. 1997, 2000, Black et al., 1998	SR +	School-based drug prevention programmes	207 programmes	Small, but nonsignificant difference between non-interactive and interactive programmes, in favour of interactive programmes. In a subset of high-quality programmes, the difference between non-interactive and interactive programs targeting alcohol was found to be significantly in favour of interactive programmes.
Werch and Owen 2002	SR +	Programmes aimed at either slowing onset of substances, or reducing use;	17 studies	Alcohol prevention programmes (n=8) resulted in 19 harmful effects, or an average of 2.4 negative outcomes for every programme. The majority were non-behavioural measures (58%).
White et al., 2004	SR ++	Studies that evaluated the contribution of external contributors to classroom- based interventions or targeted groups within the school context/curriculum	114 reports	No evidence to suggest that any particular agency or external contributor was more effective (in terms of being well received by pupils and teachers and/or leading to knowledge, attitude or behaviour change in the pupils) than any other.

5.2 Classroom-based programmes: alcohol specific

5.2.1 Overview of evidence identified

Twenty articles were identified for inclusion that reported on the evaluation of 12 alcohol education programmes across 15 studies. Three articles (McBride et al., 2000, 2003, 2004) examined the effectiveness of the School Health and Alcohol Harm Reduction Project (SHAHRP) across the same sample of students and were grouped together as a single study. In addition, five articles (Hansen & Graham, 1991; Donaldson et al., 1995; Donaldson et al., 2000; Palmer et al., 1998; Kreft, 1998) examined the effectiveness of the Adolescent Alcohol Prevention Trial (AAPT) curriculum across the same sample of 11,995 students. Hansen and Graham (1991) examined data from students who received the main programme in seventh grade, and two further articles (Kreft 1998; Palmer et al., 1998) reanalysed data for this sample. These three articles were grouped as one study and two articles by Donaldson et al (1995, 2000), which examined data for the whole sample of students, were grouped as a second study of the AAPT. A summary of the programme content across the 15 studies identified is presented in Table 5.2.

Eight studies were conducted in the USA, four in Australia, and one each in the UK, Germany and Norway. All 12 programmes were primarily classroom-based curriculums; two programmes, Resilient Families (Shortt et al., 2007) and an alcohol education programme (Morgenstern et al., 2009) incorporated materials and activities for parents. Teachers were the sole provider for nine programmes (Alcohol Education Package [Bagnall, 1990]; Project SAAV [Baumann, 2006]; SHAHRP [McBride et al., 2000, 2003, 2004]; two unnamed alcohol education curriculums [Morgenstern et al., 2009; Schnepf, 2002]; Resisting Pressures to Drink and Drive [Newman et al., 1992]; CLIMATE schools [Newton et al., 2009; Vogl et al., 2009]; AMPS [Shope et al., 1994; Shope et al., 1996a]; Resilient Families [Shortt et al., 2007]) and two programmes (Students Against Drink Driving [Klitzner et al., 1994]; and an unnamed alcohol education curriculum [Wilhelmsen et al., 1994]) combined teacher- and peer-led activities. One programme (AAPT [Donaldson et al., 1995, 2000; Hansen & Graham, 1991; Kreft, 1998; Palmer et al., 1998]) was delivered by external project staff.

The theoretical basis for the programme or intervention was not reported for five programmes. Two programmes, Project SAAV (Baumann, 2006) and Resisting Pressures to Drink and Drive (Newman et al., 1992) were based on a combination of theories. The most commonly applied theory across the 12 programmes was the social influence theory (n=3 programmes). Other theories applied included social learning theory and social cognitive theory.

The number of students recruited for participation across the 15 studies ranged from 45 to over 10,000 students. Evaluation of eight programmes was based on sample sizes of greater than 1,500 students: Alcohol Education Package (Bagnall, 1990); an unnamed alcohol education curriculum (Morgenstern et al., 2009); AMPS (Shope et al., 1994, 1996a); Resilient Families (Shortt et al., 2007); SHAHRP (McBride et al., 2000; 2003; 2004); Resisting Pressures to Drink and Drive (Newman et al., 1992); Students Against Drink Driving (Klitzner et al., 1994); and the AAPT (Donaldson et al., 1995, 2000; Hansen & Graham, 1991; Kreft, 1998; Palmer et al., 1998).

Table 5.2. Summary of programme content: classroom-bas	ed (alcohol specific)
--------------------------------------------------------	-----------------------

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Hansen & Graham, 1991; Kreft, 1998; Palmer et al., 1998	RCT -	USA n=3,011 ^ª 7th grade	School	AAPT : One year programme; Resistance Skills Training + ICU (8 lessons); Normative Education + ICU (8 lessons); Resistance Skills Training + Normative Education (10 lessons)	Social influence theory	Project staff
Donaldson et al., 1995; 2000	RCT -	USA n=11,995 ^b 8th grade	School	AAPT : One year programme; see Hansen & Graham, 1991 for details	Social influence theory	Project staff
Bagnall, 1990	CBA -	UK n=1,560 12-13 years	School	Alcohol Education Package: 4 or 5 social education lessons including group work, optional role play exercises	Social influence theory	Seconded specialist teachers or regular classroom teachers
Shope et al., 1996a	CBA +	USA n=2,031 ^c 10th-11th grade	School	AMPS: 4 sessions delivered over 4 weeks in first year (5th and 6th grade classes); 3 booster sessions in second year; 5 sessions in 10th grade	Social learning theory	Teachers
Newton et al., 2009	RCT +	Australia n=944 mean 13.1 years	School	CLIMATE Schools : Six lessons (40 minutes each); 15-20 min computer-based lesson and classroom based activities	NR	Teachers
Vogl et al., 2009	RCT +	Australia N=1,466 mean 13 years	School	CLIMATE Schools : Six-lesson harm minimization course; 15–20-minute computer-based lesson and class activities.	NR	Teachers
Shope et al., 1994	RCT -	USA N=3,704 6th-8th grade	School	Enhanced AMPS : 8 sessions in 6th grade; 5 sessions in 7th grade; 4 sessions in 8th grade	Social learning theory	Teachers
Baumann, 2006	RCT -	USA n=256 mean 17.1 vears	School	Project SAAV : Three sessions (50-minutes each); information and discussion, coping skills, homework assignment	Problem-behaviour theory, self-regulation theory, transtheoretical model of change	Teachers

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)		
Shortt et al., 2007	RCT +	Australia n=2,315 mean 12.3 years	School, family	Resilient Families : One year programme; relationship problem solving, communication, emotional awareness, peer resistance skills, conflict resolution, quiz for parents	NR	Teachers		
Newman et al., 1992	RCT -	USA n=~3,500 14-15 years	School	Resisting Pressures to Drink and Drive: Two year programme, 10 lessons; resistance skills training, video based drama, student workbook	Problem behaviour theory, social cognitive theory, role theory, educational immunization	Teachers		
McBride et al., 2000; 2003; 2004	NRCT +	Australia n=2,343 12-13 years	School	SHAHRP : 17 consecutive skills-based activities; 12 activities including skill rehearsal, and group decision-making and discussion in second year	Social inoculation	Teachers		
Klitzner et al., 1994	CBA +	USA n=4,174 9th-12th grade	School	Students Against Drink Driving: Assembly, student committee, 15 session curriculum (10th grade), 'contract for life'	NR	Teachers, Peers		
Morgenstern et al., 2009	RCT ++	Germany n=1,686 mean 13 years	School, family	Four class units, a student booklet and a parent booklet; addressed social influences, enhancing motivation to avoid substances, consequences of alcohol use, media/advertising literacy, resistance skills and normative beliefs	NR	Teachers		
Schnepf, 2002	NRCT -	USA n=45 mean 15.2 years	School	7 sessions (40 minutes each); peer- vs. teacher-led alcohol education curriculum	NR	Teachers		
Wilhelmsen et al., 1994	NRCT -	Norway 7th grade n=915	School	10 lessons over two month; highly role specific vs. less role specific; alcohol use and social traditions, norms for alcohol use, managing drinking pressures, attitudes to alcohol use	Social cognitive theory	Teachers, Peers		
^a Examined data from AAPT students who received the main programme in 7 th grade; ^b Examined data for the whole sample of AAPT students; ^c Followed up a sample of students who had received the AMPS curriculum as 6 th -8 th graders and an additional five sessions in 10 th grade.								

Eight programmes targeted students aged 14 or younger and of these seven were taught to students aged 12-13 years. Four programmes, Resisting Pressures to Drink and Drive, Students Against Drink Driving, an unnamed alcohol education curriculum (Schnepf, 2002), and Project SAAV were targeted at older adolescents (aged >14 years) and one programme, AMPS, targeted students in the 6th to the 10th grade (11-15 years).

Four programmes were based on immediate post-test follow-up only: Alcohol Education Package, two unnamed alcohol education curriculums (Schnepf, 2002; Wilhelmsen et al., 1994) and the enhanced AMPS (Shope et al., 1994). Five studies (Shortt et al., 2007; McBride et al., 2000, 2003, 2004; Shope et al., 1996a; Hansen & Graham, 1991; Kreft, 1998; Palmer et al., 1998 Donaldson et al., 1995; 2000) reported on follow-ups greater than 12 months.

5.2.2 Quality assessment

Of the 15 studies identified, nine studies were RCTs, three were non-randomised controlled trials and three were controlled before and after studies. All nine RCTs were based on cluster randomisation. Study quality was mixed across the studies. One study (Morgenstern et al., 2009), an RCT was rated good quality, six studies (Klitzner et al., 1994; McBride et al., 2004; Newton et al., 2009; Shope et al., 1996a; Shortt et al., 2007; Vogl et al., 2009) were rated moderate quality, and eight studies (Bagnall, 1990; Baumann, 2006; Donaldson et al., 2000; Hansen & Graham, 1991; Newman et al., 1992; Schnepf, 2002; Shope et al., 1994; Wilhelmsen et al., 1994) were rated poor quality. All nine RCTs were reported as randomised but only one study (Morgenstern et al., 2009) reported that allocation was concealed appropriately. Outcomes measures were reliable across most of the RCTs and NRCTs, but three studies (Hansen & Graham, 1991; Klitzner et al., 1994; Shope et al., 1996a) either did not report on the validity or reliability of the outcome measures used or this aspect of study design was poorly reported. Important and relevant outcomes were reported across all of the included studies. The baseline comparability of the intervention and control groups was poorly or not reported in eight studies (Shortt et al., 2007; Baumann, 2006; Donaldson et al., 1995, 2000; Newman et al., 1992; Shope et al., 1994; McBride et al., 2000, 2003, 2004; Wilhelmsen et al., 1994; Bagnall, 2003) and in three studies (Bagnall, 2003; Donaldson et al., 1995, 2000; Newman et al., 1992) it was not clear how many participants were assigned to intervention and control groups. Only one study (Morgenstern et al., 2009) reported that an ITT analysis had been undertaken.

5.2.3 Findings

5.2.3.1 Short-term results (<6 months)

Eleven studies (Bagnall, 1990; Donaldson et al., 1995; Baumann, 2006; Klitzner et al., 1994; McBride et al., 2000, 2003, 2004; Morgenstern et al., 2009; Newton et al., 2009; Schnepf, 2002; Shope et al., 1994; Vogl et al., 2009; Wilhelmsen et al., 1994) reported short-term follow-up data for 10 programmes: Alcohol Education Package, AAPT, Project SAAV, Students Against Drink Driving, SHAHRP, CLIMATE, the AMPS curriculum and three unnamed curricula.

Knowledge and understanding

Seven studies (Bagnall, 1990; McBride et al., 2000, 2003, 2004; Morgenstern et al., 2009; Newton et al., 2009; Schnepf, 2002; Shope et al., 1994; Vogl et al., 2009) examined short-term intervention effects on knowledge related to alcohol use. Bagnall (1990) reported that the Alcohol Education Package curriculum had no impact on knowledge, but the remaining six studies (McBride et al., 2000, 2003, 2004; Morgenstern et al., 2009; Newton et al., 2009; Schnepf, 2002; Shope et al., 1994; Vogl et al., 2009) reported that intervention students demonstrated significantly greater alcohol-related knowledge at follow-up than control students.

Attitudes and values

Eight studies (Bagnall, 1990; Baumann, 2006; McBride et al., 2000, 2003, 2004; Morgenstern et al., 2009; Newton et al., 2009; Schnepf, 2002; Vogl et al., 2009; Wilhelmsen et al., 1994) examined shortterm intervention effects on students' attitudes and values. Of five studies (Bagnall, 1990; Morgenstern et al., 2009; McBride et al., 2000, 2003, 2004; Schnepf, 2002; Wilhelmsen et al., 1994) that examined programme effects on attitudes, three studies (Bagnall, 1990; Morgenstern et al., 2009; Schnepf, 2002) reported no effects on alcohol-related attitudes. McBride et al (2000; 2003; 2004) reported that students who participated in SHAHRP showed significantly safer alcohol-related attitudes and Wilhelmsen et al (1994) found a highly-role specific prevention programme had positive impacts on students' attitudes to abstention. Two studies (Morgenstern et al., 2009; Wilhelmsen et al., 1994) examined intervention effects on students' intentions towards alcohol. Morgenstern et al (2009) reported no intervention impact, but Wilhelmsen et al (1994) found that students who participated in a highly-role specific prevention programme were more likely to report an intention to abstain from alcohol than control students. They also reported high norms to abstain. Three studies (Baumann, 2006; Newton et al., 2009; Vogl et al., 2009) examined changes in alcohol-related expectancies. Baumann (2006) reported that there were no effects of the Project SAAV programme on alcoholrelated expectancies. Two studies (Newton et al., 2009; Vogl et al., 2009) which examined the Clinical Management and Treatment Education (CLIMATE) schools model, based around a computerised harm minimisation course, reported conflicting results. Vogl et al (2009) found that the increase in positive alcohol-related expectancies was significantly less for females in CLIMATE schools compared to control females (p<0.001), and that boys in CLIMATE schools reported a decrease on this measure compared to an increase among control males (p<0.01). Newton et al (2009), who reported on a cross-validation of the study by Vogl et al (2009) found no difference in alcohol-related expectancies between CLIMATE schools and control schools.

Personal and social skills

One study (Klitzner et al., 1994) examined the effects of a programme designed to address drinking and driving, on drinking and driving behaviour. At post-test, Klitzner et al (1994) found that the intervention had no impact on an index of driving while intoxicated or riding with impaired drivers.

Health and social outcomes relating to alcohol use and sexual health

Eleven studies (Bagnall, 1990; Donaldson et al., 1995; Baumann, 2006; Klitzner et al., 1994; McBride et al., 2000, 2003, 2004; Morgenstern et al., 2009; Newton et al., 2009; Schnepf, 2002; Shope et al.,

1994; Vogl et al., 2009; Wilhelmsen et al., 1994) reported short-term follow-up data related to alcohol use for 10 programmes.

Six studies (Bagnall, 1990; Baumann, 2006; Donaldson et al., 1995; McBride et al., 2000, 2003, 2004; Newton et al., 2009; Wilhelmsen et al., 1994) reported on programmes that demonstrated evidence of reducing alcohol use. Students who participated in the first year of the SHAHRP programme (McBride et al., 2001, 2003, 2004) consumed significantly less alcohol overall and per occasion than students in the control groups. Intervention students who received the Alcohol Education Package (Bagnall, 1990) were significantly less likely than students in the control group to have drunk alcohol in the last 7 days, but the intervention had no impact on frequency of use. Students participating in Project SAAV (Baumann, 2006) reported drinking less frequently over time than controls. Donaldson et al (1995) reported a positive relationship between participation in a seventh grade refusal skills programme, but the relationship was only significant among students who believed it was not acceptable to drink at baseline. Among students in CLIMATE schools who participated in a computerised harm minimisation programme (Newton et al., 2009), average alcohol consumption decreased over the short-term compared to an increase among control students. However, another study of the CLIMATE schools programme by Vogl et al (2009) found no short-term effects of the programme on alcohol consumption. Students who received a highly-role specific programme (Wilhelmsen et al., 1994) reported drinking less than both control students and students who received a less-role specific programme. There were no differences observed between students who received the less-role specific programme and control students. Four studies (Klitzner et al., 1994; Morgenstern et al., 2009; Shope et al., 1994; Schnepf, 2002) reported on programmes that had no effects on alcohol use in young people.

Eight studies examined intervention effects on measures of excessive alcohol consumption. Five studies (Bagnall, 1990; Newton et al., 2009; Schnepf, 2002; Shope et al., 1994; Vogl et al., 2009) found no intervention effects of excessive and problem drinking. Baumann (2006) reported that compared to students in the control groups, students participating in Project SAAV reported binge drinking less frequently over time and SHAHRP students (McBride et al., 2000, 2003, 2004) were significantly less likely than comparison students to report consuming alcohol to harmful or hazardous levels² once a month or more. Morgenstern et al (2009) found that there was no statistically significant effect of the intervention on lifetime drunkenness, but that intervention students were significantly less likely than composite for the intervention students to report life-time binge drinking.

Five studies (Bagnall, 1990; Baumann, 2006; McBride et al., 2000, 2003, 2004; Newton et al., 2009; Vogl et al., 2009) examined harms resulting from alcohol use. Two studies (Newton et al., 2009; Vogl et al., 2009) that examined the impact of the CLIMATE schools programme found no intervention effects on harm associated with students' own use of alcohol. SHAHRP students reported less harm associated with their own use of alcohol over the previous 12 months compared to the comparison group (McBride et al., 2000, 2003, 2004). Bagnall (1990) found that there was no difference in the number of intervention and control students reporting that they had ever had a hangover or an

² Defined as consuming more than 2 (for females) or 4 (for males) standard drinks (10g alcohol) per occasion.

alcohol-induced stomach upset. Baumann (2006) reported that students participating in Project SAAV had fewer alcohol-related consequences than control students.

5.2.3.2 Medium-term results (up to 12 months)

Seven studies (Donaldson et al., 2000; Hansen & Graham, 1991; Klitzner et al., 1994; Morgenstern et al., 2009; Newman et al., 1992; Newton et al., 2009; Vogl et al., 2009) reported medium-term follow data for five programmes: AAPT, Students Against Drink Driving, an alcohol education intervention, Resisting Pressures to Drink and Drive, and CLIMATE schools.

Knowledge and understanding

Four studies (Morgenstern et al., 2009; Newman et al., 1992; Newton et al., 2009; Vogl et al., 2009) examined medium-term intervention effects on alcohol-related knowledge. Two studies (Morgenstern et al., 2009; Newman et al., 1992) reported positive intervention effects on knowledge, and two studies (Newton et al., 2009; Vogl et al., 2009) found that despite finding short-term effects, there were no medium-term effects on knowledge for the CLIMATE schools programme.

Attitudes and values

Three studies (Morgenstern et al., 2009; Newton et al., 2009; Vogl et al., 2009) examined mediumterm intervention effects of alcohol-related attitudes and values. Morgenstern et al (2009) found that an alcohol education intervention did not impact on alcohol attitudes or intentions. Two studies (Newton et al., 2009; Vogl et al., 2009) that examined a computerised harm minimisation intervention (CLIMATE schools programme) reported conflicting medium-term results on alcohol-related expectancies. Vogl et al (2009) reported that although expectancies increased over time in both the intervention and control group, the increase among intervention females was smaller. At 12-month follow-up, both males and females in the intervention group did not report as great an increase in positive alcohol-related expectancies as did students in the control group. At 6-month follow-up, in a cross-validation of this study, Newton et al (2009) reported no intervention effects on alcohol-related expectancies.

Personal and social skills

Two studies (Klitzner et al., 1994; Newman et al., 1992) of two drink driving programmes examined intervention effects on outcomes related to drinking and driving. At the 1-year follow-up, there were no effects of the Students Against Drink Driving programme (Klitzner et al., 1994) on an index of driving while intoxicated or riding with impaired drivers. The second study, of the Resisting the Pressures to Drink and Drive curriculum (Newman et al., 1992), found that although the number of times students had ridden with a drinking driver in the last 30 days increased in both the intervention and control groups, the increase among intervention students was significantly less.

Health and social outcomes relating to alcohol use and sexual health

Seven studies (Donaldson et al., 2000; Hansen & Graham, 1991; Klitzner et al., 1994; Morgenstern et al., 2009; Newman et al., 1992; Newton et al., 2009; Vogl et al., 2009) examined medium-term intervention effects on health outcomes related to alcohol use. Three studies (Hansen & Graham

1991; Donaldson et al., 2000; Kreft 1998) reported 1-year follow-up data for the AAPT. One study (Hansen & Graham, 1991) reported that students who received normative education had lower rates of alcohol consumption, although based on analyses conducted at the classroom level (Kreft, 1998) the programme (including the normative education component) was shown to not be effective. In a second study of the AAPT (Donaldson et al., 2000), lower rates of lifetime and 30-day alcohol consumption were only found among a subgroup of public school students who received the normative education component. Three further programmes were shown to have no medium-term effects on alcohol use: Students Against Drink Driving (Klitzner et al., 1994); an alcohol education intervention (Morgenstern et al., 2009); and Resisting Pressures to Drink and Drive (Newman et al., 1992).

Inconsistent results were found for the harm minimisation intervention examined in the CLIMATE school programme (Newton et al., 2009; Vogl et al., 2009). Vogl et al (2009) reported that among females, average alcohol consumption remained relatively constant between baseline and 6-month follow-up for the intervention group and increased in the control group (p<0.001 for comparison). At 12 months, control group students reported a greater increase in average alcohol consumption compared to the intervention group (p<0.05 for comparison). For boys there were no significant differences between intervention groups. Lack of significant effects was also noted in the cross-validation study by Newton et al (2009); finding no significant difference in average weekly alcohol consumption between intervention and control groups at the 6-month follow-up.

Four studies (Donaldson et al., 2000; Morgenstern et al., 2009; Newton et al., 2009; Vogl et al., 2009) examined medium-term intervention effects on excessive alcohol consumption. Morgenstern et al (2009) found that although the alcohol education curriculum examined did not impact on lifetime drunkenness (adjusted OR 0.77 95% CI: 0.52, 1.12), intervention students were significantly less likely than control students to report lifetime binge drinking (adjusted OR 0.74; 95% CI 0.57, 0.97). Donaldson et al (2000) reported an impact of the AAPT on drunkenness; students who attended public schools and who received the normative education component of the intervention reported lower rates of drunkenness. Two studies (Vogl et al., 2009; Newton et al., 2009) of the CLIMATE schools alcohol programme examined intervention impact on harms arising from students' alcohol use. At 6-month follow-up (Vogl et al., 2009; Newton et al., 2009) there were no differences in the number of harms experienced between intervention and control students, however, at the 12-months follow-up (Vogl et al., 2009) females in the intervention group had significantly less of an increase in harm than the control students (p<0.05).

5.2.3.3 Long-term (>12 months)

Long-term follow-up data were reported for four programmes: AAPT (Donaldson et al., 2000; Hansen & Graham, 1991; Kreft, 1998; Palmer et al., 1998); SHAHRP (McBride et al., 2000, 2003, 2004); AMPS (Shope et al., 1996a); and Resilient Families (Shortt et al., 2007).

Knowledge and understanding

Two studies (McBride et al., 2004; Shope et al., 1996a) reported on long-term intervention effects on knowledge. McBride et al (2004) found that impacts on alcohol-related knowledge among students who participated in SHAHRP were not maintained 17-months later. A study based on data from a sample of students who had received the AMPS curriculum as sixth through eighth graders and additional sessions in 10th grade (Shope et al., 1996a) found that intervention students had significantly more alcohol-related knowledge than comparison students at the end of 12th grade.

Attitudes and values

Positive effects of the SHAHRP programme on alcohol-related attitudes were shown to have been maintained to the end of the study, 17 months after the final phase of the intervention (McBride et al., 2004).

Personal and social skills

None of the included studies examined long-term programme effects on personal and social skills.

Health and social outcomes relating to alcohol use and sexual health

Shope et al (1996a) followed up high school-aged students who had received the AMPS curriculum in sixth grade. Delivery of the sixth grade curriculum had no long-term effects on alcohol use in high school, and following delivery of a tenth grade curriculum, there were no differences in alcohol use between intervention and comparison students at follow-up two years later in 12th grade. At the final follow-up of the SHAHRP programme, 17 months after delivery, McBride et al (2004) reported that the total amount of alcohol consumed by intervention and comparison students was beginning to converge (significance not reported) and the short-term effects of the intervention on alcohol use were not maintained. Palmer et al (1998) examined the long-term effects of the AAPT curriculum. Normative education was found to be more effective than information only control in terms of reducing alcohol use, two years after delivery of the programme. A second study that examined the AAPT curriculum (Donaldson et al., 2000) found that, for a sample of students attending public schools, those who received normative education reported significantly lower scores on the alcohol index measure and significantly lower rates of lifetime alcohol use in the ninth and tenth grades (2- and 3year follow-ups, respectively) and lower rates of 30-day alcohol use in the tenth grade compared to students receiving comparison interventions. Shortt et al (2007) found that the Resilient Families intervention had no effects on alcohol use status. Three studies (Donaldson et al., 2000; McBride et al., 2004; Shope et al., 1996a) reported long-term intervention effects on measures of excessive drinking. Donaldson et al (2000) found that the normative education component of the AAPT programme was associated with lower rates of drunkenness among students attending public schools at the 2-year but not the 3-year follow-up, when students were in 10th grade. Following delivery of the AMPS tenth grade curriculum (Shope et al., 1996a), students who received the intervention reported significantly less alcohol misuse than comparison students at the end of twelfth grade. There were no long-term intervention effects of the SHAHRP programme on hazardous or harmful drinking, but intervention students reported experiencing less harm from their drinking than control students.

5.2.4 Summary and evidence statements

Overall, 20 articles were identified for inclusion that reported on the evaluation of 12 alcohol education programmes across 15 studies. The majority of the included studies were conducted in the USA, but four studies were conducted in Australia and one each in the UK, Germany and Norway. All 12 programmes were primarily classroom-based curriculums, but two programmes incorporated additional materials and activities for parents. Programmes were largely teacher-led. However, two programmes, Students Against Drink Driving (Klitzner et al., 1994) and an unnamed alcohol education curriculum (Wilhelmsen et al., 1994) combined teacher- and peer-led activities, and one programme, the AAPT, (Donaldson et al., 1995, 2000; Hansen & Graham, 1991; Kreft, 1998; Palmer et al., 1998) was delivered exclusively by external project staff. The programmes identified targeted students across a range of age groups; eight programmes targeted students aged 14 or younger and four programmes were targeted at older adolescents.

5.2.4.1 Knowledge and understanding

Eight studies (Bagnall, 1990; McBride et al., 2004; Morgenstern et al., 2009; Newton et al., 2009; Schnepf, 2002; Vogl et al., 2009; Newman et al., 1992; Shope et al., 1996a) examined intervention effects on knowledge related to alcohol use. These studies demonstrated that alcohol-specific education programmes generally increased alcohol or curriculum knowledge over the short-term. Effects on medium-term knowledge were weaker across four studies (Morgenstern et al., 2009; Newman et al., 1992; Newton et al., 2009; Vogl et al., 2009) and for two studies that examined programme effects on the long-term acquisition of knowledge, for SHAHRP (McBride et al., 2004) and AMPS (Shope et al., 1996b), results were limited or inconsistent.

5.2.4.2 Attitudes and values

Eight studies (Bagnall, 1990; Baumann, 2006; McBride et al., 2004; Morgenstern et al., 2009; Newton et al., 2009; Schnepf, 2002; Vogl et al., 2009; Wilhelmsen et al., 1994) examined young people's alcohol-related attitudes and values. The majority of studies reported no significant programme effects on alcohol-related attitudes. Short-term increases in safer alcohol-related attitudes were reported by Wilhelmsen et al (1994) who examined a highly-role specific programme compared to a less-role specific alcohol programme. Improvements were also found among intervention students in the highly-role specific programme in attitudes towards alcohol abstinence and intentions to abstain from alcohol. The SHAHRP programme (McBride et al., 2004) also had positive short- and long-term effects on students' alcohol-related attitudes. Medium-term effects on attitudes and skills were also mixed. Two studies (Newton et al., 2009; Vogl et al., 2009) of the CLIMATE schools programmes reported conflicting effects on alcohol-related expectancies and another alcohol education intervention (Morgenstern et al., 2009) had no impact on attitudes and values.

5.2.4.3 Personal and social skills

Two studies (Klitzner et al., 1994; Newman et al., 1992) examined short- and medium-term effects of two programmes, which primarily focused on reducing the harm from drinking and driving. There were no short- or medium-term effects of the Students Against Drink Driving programme (Klitzner et al.,

1994) on drink driving measures but a second programme, Resisting the Pressures to Drink and Drive (Newman et al., 1992) reduced the number of students who reported riding with a drinking driver at the medium-term follow-up.

5.2.4.4 Health and social outcomes relating to alcohol use and sexual health

Programme effects on alcohol consumption were examined across all studies. However a range of measures of alcohol consumption were reported across the included studies. Ten studies (Bagnall, 1990; Donaldson et al., 1995; Baumann, 2006; Klitzner et al., 1994; McBride et al., 2000; Morgenstern et al., 2009; Newton et al., 2009; Schnepf, 2002; Vogl et al., 2009; Wilhelmsen et al., 1994) that examined short-term effects on alcohol consumption, in general, reported inconsistent findings. The SHAHRP programme (McBride et al., 2004) appeared to have the most consistent effects on short-term alcohol use, and additionally had effects on hazardous/harmful drinking. Seven further studies (Bagnall, 1990; Baumann, 2006; Morgenstern et al., 2009; Newton et al., 2009; Schnepf, 2002; Shope et al., 1994; Vogl et al., 2009) examined programme impacts on heavy drinking, with two studies (Baumann, 2006; Morgenstern et al., 2009) reporting beneficial effects of Project SAAV and an unnamed alcohol education programme, respectively, on binge drinking. Medium- to long-term effects on alcohol consumption were found to be limited across nine alcohol education programmes (Donaldson et al., 2000; Palmer et al., 1998; Klitzner et al., 1994; McBride et al., 2004; Morgenstern et al., 2009; Newman et al., 1992; Newton et al., 2009; Shope et al., 1994; Shortt et al., 2007; Vogl et al., 2009). There appeared to be long-term effects of the normative education component of the AAPT programme (Donaldson et al., 2000; Palmer et al., 1998), but these findings were limited by the poor quality of the studies that examined this programme. In addition, based on additional analyses conducted using the classroom as unit of analysis, Kreft (1998) concluded the whole programme to be ineffective. Two studies of the CLIMATE schools programme (Newton et al., 2009; Vogl et al., 2009) reported conflicting medium-term effects and 17-months after delivery of the SHAHRP programme (McBride et al., 2004) the positive short-term effects appeared to be declining. Although intervention effects favoured SHAHRP, differences between intervention and control students in terms of their alcohol consumption and other measures of alcohol use including harmful/hazardous drinking were no longer significant. There were no long-term effects of a long-term version of the AMPS programme (Shope et al., 1996a) on alcohol consumption, but there did appear to be intervention effects on alcohol misuse.

Evidence statement 2

- 2 (f) There is strong evidence from four RCTs, two NRCTs and two CBA studies¹ to suggest that classroom-based alcohol specific programmes are effective at increasing alcohol-related knowledge in the short-term, but have inconsistent or mixed effects on alcohol-related knowledge in the medium- to long-term. Findings may only be partially applicable to the UK as studies were implemented within Australia, Germany and the USA and may not be generalisable beyond the populations studied.
- 2 (g) Overall, there is inconsistent evidence from four RCTs, three NRCTs and one CBA study² to determine the effects of alcohol specific education programmes on attitudes and values relating
to alcohol. However, there is moderate evidence from one NRCT³ to suggest that programmes based on a harm reduction approach may have positive short- to long-term effects on students' alcohol-related attitudes. In addition, there is weak evidence from one NRCT⁴ to suggest that programmes with a high level of role-specification for providers may have short-term positive impacts on attitudes and values.

- 2 (h) There is inconsistent evidence from one RCT and one CBA study⁵ to determine the effects of programmes focusing on reducing the harm from drinking and driving on drink driving measures.
- 2 (i) There is moderate evidence from five RCTs, three NRCTs and two CBA studies⁶ to suggest that alcohol-specific education programmes may have mixed short-term effects on health outcomes relating to alcohol use. One NRCT⁷ of a programme focusing on harm reduction through skills-based activities (SHAHRP), showed short-term reductions in alcohol use. In particular effects were seen on risky drinking behaviours such as drunkenness and binge drinking. Findings may only be partially applicable to the UK as this study was conducted in Australia and may not be generalisable beyond the populations studied.
- 2 (j) There is moderate evidence from eight RCTs, one NRCT and one CBA study⁸ to suggest that alcohol-specific education programmes have limited medium- to long-term effects on health outcomes related to alcohol use, such as frequency of alcohol consumption and drunkenness. Findings may only be partially applicable to the UK as studies were implemented outside the UK and may not be generalisable beyond the populations studied.
- ¹ Bagnall, 1990 (CBA -); McBride et al., 2004 (NRCT +); Morgenstern et al., 2009 (RCT ++); Newton et al., 2009 (RCT +); Schnepf, 2002 (NRCT -); Vogl et al., 2009 (RCT +); Newman et al., 1992 (RCT -); Shope et al., 1996a (CBA +)
- ² Bagnall, 1990 (CBA -); Baumann, 2006 (RCT -); McBride et al., 2004 (NRCT +); Morgenstern et al., 2009 (RCT ++); Newton et al., 2009 (RCT +); Schnepf, 2002 (NRCT -); Vogl et al., 2009 (RCT +); Wilhelmsen et al., 1994 (NRCT -)
- ³ McBride et al., 2000; 2003; 2004 (NRCT +)
- ⁴ Wilhelmsen et al., 1994 (NRCT -)
- ⁵ Klitzner et al., 1994 (CBA +); Newman et al., 1992 (RCT -)
- ⁶ Bagnall, 1990 (CBA -); Donaldson et al., 1995 (RCT -); Baumann, 2006 (RCT -); Klitzner et al., 1994 (CBA +); McBride et al., 2000 (NRCT +); Morgenstern et al., 2009 (RCT ++); Newton et al., 2009 (RCT +); Schnepf, 2002 (NRCT -); Vogl et al., 2009 (RCT +); Wilhelmsen et al., 1994 (NRCT -)
- ⁷ McBride et al., 2000; 2003; 2004 (NRCT +)
- ⁸ Donaldson et al., 2000 (RCT -); Palmer et al., 1998 (RCT -); Klitzner et al., 1994 (CBA +); McBride et al., 2004 (NRCT +); Morgenstern et al., 2009 (RCT ++); Newman et al., 1992 (RCT -); Newton et al., 2009 (RCT +); Shope et al., 1994 (RCT -); Shortt et al., 2007 (RCT +); Vogl et al., 2009 (RCT +)

Ctudy.	Deting	Intervention	Comparator	Follow-up		Outcomes	
Study	Rating	Intervention	Comparator	Follow-up	Knowledge	Attitudes	Skills
Bagnall, 1990	CBA -	Alcohol Education Package n=NR	No intervention n=NR	PT (10 months) n=NR	NS knowledge	NS negative attitudes towards alcohol	-
Baumann, 2006	RCT -	Project SAAV n=144	No intervention n=112	3 months (58%)	-	NS alcohol-related expectancies	-
Klitzner et al., 1994	CBA +	Students Against Drink Driving n=1,900	No intervention N=2,074	PT n=NR	-	-	NS index of driving while intoxicated or riding with impaired driver
McBride et al., 2000; 2003; 2004	NRCT +	SHAHRP n=1,111	No intervention N=1,232	PT n=NR	↑ knowledge***	↑ safer alcohol-related attitudes**	-
Morgenstern et al., 2009	RCT ++	Alcohol education intervention n=911	Usual curriculum N=964	PT (96%)	↑ alcohol knowledge**	NS alcohol attitudes and intentions	-
Newton et al., 2009	RCT +	CLIMATE Schools n=513	Usual curriculum N=431	PT (69%)	↑ knowledge scale**	NS alcohol-related expectancies	-
Schnepf, 2002	NRCT -	Unnamed programmes Peer led, n=13 Teacher led, n=19	Religious education n=13	PT n=NR	$m{\Lambda}$ alcohol knowledge***	NS negative attitudes towards alcohol	-
Shope et al., 1994	RCT -	Enhanced AMPS n=840	NR n=885	6 th -8 th grade NR	↑ curriculum knowledge*	-	-
Vogl et al., 2009	RCT +	CLIMATE Schools n=611	Usual curriculum N=855	PT (77.2%)	↑ knowledge***		-
Wilhelmsen et al., 1994	NRCT -	Highly-role specific, n=279 Less-role specific, n=314	Usual curriculum n=262	PT (95%)	-	 ↑ attitudes to abstention (highly role specific only)** ↑ Intention to abstain (highly role specific only)** ↑ Norms to abstain (highly role specific only)* 	-
*p≤0.05; **p≤0.01; ***	ʻp≤0.001; ↑	increase relative to co	omparator; 🗸 decrease	relative to com	nparator; NS not significant;	 outcome not reported 	

Table 5.3. Classroom-based alcohol education programmes: short-term programme effects on knowledge, attitudes and skills

Ctudy	Dating	Intervention	Comparator	Follow-up		Health outcomes			
Study	Rating	Intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other		
Bagnall, 1990	CBA -	Alcohol Education Package n=NR	No intervention n=NR	PT (10 months) n=NR	✓ drunk alcohol in last week* NS frequency of consumption	NS consumed >3 units alcohol	NS ever had a hangover NS alcohol-induced stomach upset		
Donaldson et al., 1995	RCT -	AAPT n=NR	ICU only n=NR	PT (1 year) n=NR	✓ onset of alcohol use (believed it was not acceptable to drink; resistance skills training only*)	-	-		
Baumann, 2006	RCT -	Project SAAV n=144	No intervention n=112	3 months (58%)	ullet drinking frequency**				
Klitzner et al., 1994	CBA +	Students Against Drink Driving n=1,900	No intervention n=2,074	PT n=NR	NS drinking quantity	-	-		
McBride et al., 2000; 2003; 2004	NRCT +	SHAHRP n=1,111	No intervention n=1,232	PT n=NR	 ✓ alcohol consumption*** ✓ alcohol per occasion* 		↓ harm from own use* NS harm from others use		
Morgenstern et al., 2009	RCT ++	Alcohol education intervention n=911	Usual curriculum n=964	PT (96%)	NS past alcohol month use NS lifetime alcohol use	NS lifetime drunkenness ↓ lifetime binge drinking*	-		
Newton et al., 2009	RCT +	CLIMATE Schools n=513	Usual curriculum n=431	PT (69%)		NS frequency of drinking to excess on a single occasion, past 3 months	NS harm associated with own use of alcohol		
Schnepf, 2002	NRCT -	Peer led, n=13 Teacher led, n=19	Religious education n=13	PT n=NR	NS alcohol consumption	NS problem drinking	-		
Shope et al., 1994	RCT -	Enhanced AMPS n=840	NR n=885	6 th -8 th grade NR	-	NS alcohol misuse	-		
Vogl et al., 2009	RCT +	CLIMATE Schools n=611	Usual curriculum n=855	PT (77.2%)	NS average alcohol consumption	NS frequency of drinking to excess on a single occasion in the past 3 months	NS harms from own use of alcohol		
Wilhelmsen et al., 1994	NRCT -	Highly-role specific, n=279 Less-role specific, n=314	Usual curriculum n=262	PT (95%)	✓ alcohol consumption (highly role specific only)**	-	-		
*p≤0.05; **p≤0.01; ***	′p≤0.001; ↑	increase relative to con	mparator; 🗸 decrease	relative to con	nparator; NS not significant;	- outcome not reported			

Table 5.4. Classroom-based alcohol education programmes: short-term programme effects on health outcomes relating to alcohol use

						Outeennee		
Study	Pating	Intervention	Comparator	Follow-up		Outcomes		
Study	Rating		comparator	r onow-up	Knowledge	Attitudes	Skills	
Klitzner et al., 1994	CBA +	Students Against Drink Driving n=1,900	No intervention n=2,074	1 year (74%)	-	-	NS index of driving while intoxicated or riding with impaired driver	
Morgenstern et al., 2009	RCT ++	Alcohol education intervention n=911	Usual curriculum n=964	1 year (85%)	↑ alcohol knowledge*	NS alcohol attitudes and intentions	-	
Newman et al., 1992	RCT -	Resisting Pressures to Drink and Drive n=51 classes	Usual curriculum n=36 classes	1 year n=NR	↑ knowledge***	-	✓ ridden with a drinking driver*	
Newton et al., 2009	RCT +	CLIMATE Schools n=513	Usual curriculum n=431	6 months (62%)	NS knowledge scale	NS alcohol-related expectancies	-	
Vogl et al., 2009	PCT -	CLIMATE Schools n=611	Usual curriculum n=855	6 month (71.5%)	NS knowledge	↓ positive alcohol- related expectancies (females only**)	-	
	KUT +			12 months (70.9%)	NS knowledge	↓ positive alcohol- related expectancies (females***/boys**)	-	
*p≤0.05; **p≤0.01; **	*p≤0.0 <mark>01; ↑</mark>	increase relative to comp	arator; 🗸 decrease	relative to con	nparator; NS not significant;	- outcome not reported		

Table 5.5. Classroom-based alcohol education programmes: medium-term programme effects on knowledge, attitudes and skills

Study	Poting	Intervention	Comparator	Follow-up		Outcomes			
Sludy	кашу	Intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other		
Donaldson et al., 2000	RCT -	AAPT n=NR	ICU only n=NR	8 th grade (1 year)	 Public schools students receiving normative education only^a: ↓ scores on alcohol index measure^{**} ↓ lifetime alcohol use^{***} ↓ 30-day alcohol use^{***} 	Public schools students receiving normative education only ^a : ↓ drunkenness**	-		
Hansen & Graham, 1991; Kreft, 1998; Palmer et al., 1998	RCT -	AAPT Resistance skills training, n=33 classes Normative education, n=27 classes Combined, n=26 classes	Information only n=32 classes	1 year N=2,416 (80%)	NS composite measure of alcohol use	-	-		
Klitzner et al., 1994	CBA +	Students Against Drink Driving n=1,900	No intervention n=2,074	1 year (74%)	NS drinking quantity	-	-		
Morgenstern et al., 2009	RCT ++	Alcohol education intervention n=911	Usual curriculum n=964	1 year (85%)	NS past alcohol month use NS lifetime alcohol use	NS lifetime drunkenness ↓ lifetime binge drinking*	-		
Newman et al., 1992	RCT -	Resisting Pressures to Drink and Drive n=51 classes	Usual curriculum n=36 classes	1 year n=NR	NS drinking behaviours	-	-		
Newton et al., 2009	RCT +	CLIMATE Schools n=513	Usual curriculum n=431	6 months (62%)	NS average consumption	NS frequency of drinking to excess on a single occasion, past 3 months	NS harm associated with own use of alcohol		
Vogl et al., 2009	RCT +	CLIMATE Schools n=611	Usual curriculum n=855	6 month (71.5%)			NS harms from own use of alcohol		
	KUT +			12 months (70.9%)	✓ average alcohol consumption (females only*)		✓ harms from own use of alcohol (females only*)		
*p≤0.05; **p≤0.01; *** ^a For students in priva	*p≤0.001; ↑ ate schools. ;	increase relative to comp and those assigned to oth	erator; V decrease	relative to con ditions all findir	nparator; NS not significant; ngs were NS	- outcome not reported			

Table 5.6. Classroom-based alcohol education programmes: medium-term programme effects on health outcomes relating to alcohol use

Study	Deting	Intervention	Comparator	Follow-up	Outcomes					
	Raung				Knowledge	Attitudes	Skills			
McBride et al., 2000; 2003; 2004	NRCT +	SHAHRP n=1,111	No intervention n=1,232	17 months n=1,778 (76%)	NS knowledge	↑ safer alcohol-related attitudes*	-			
Shope et al., 1996a	CBA +	AMPS n=507	NR n=530	2 years NR	↑ knowledge*	-	-			
*p≤0.05; **p≤0.01; ***	*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported									

Table 5.7. Classroom-based alcohol education programmes: long-term programme effects on knowledge, attitudes and skills

Table 5.8. Classroom-based alcohol education programmes: long-term programme effects on health outcomes

Study	Rating	Intervention	Comparator	Follow-up	Outcomes		
Sludy	Rating	intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other
Donaldson et al., 2000	DOT	AAPT n=NR	ICU only n=NR	9 th grade (2 years)	Public schools students receiving normative education only ^a : ↓ scores on alcohol index measure** ↓ lifetime alcohol use** NS 30-day alcohol use	For public schools students receiving normative education only ^a : ✔ drunkenness**	-
	KUT-			10 th grade (3 years)	 Public schools students receiving normative education only^a: ↓ scores on alcohol index measure* ↓ lifetime alcohol use* ↓ 30-day alcohol use* 	NS drunkenness	-
Palmer et al., 1998	RCT -	AAPT Resistance skills training, n=33 classes Normative education, n=27 classes Combined, n=26 classes	Information only n=32 classes	2 years (46%)	↓ composite measure of alcohol use (normative education only**)	-	-

Study	Pating	Intervention	Comparator	Follow-up	Outcomes					
Study	Кашіў	intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other			
McBride et al., 2000; 2003; 2004	NRCT +	SHAHRP n=1,111	No intervention n=1,232	17 months n=1,778 (76%)	NS alcohol consumption NS alcohol per occasion	NS hazardous/harmful drinking	 ↓ harm from own use ^b NS harm from others use 			
Shope et al., 1996a	CBA +	AMPS n=507	NR n=530	2 years NR	NS alcohol use	ullet alcohol misuse*	-			
Shortt et al., 2007	RCT +	Resilient Families n=NR	Usual curriculum n=NR	14 months n=NR	NS alcohol use	-	-			
*p≤0.05; **p≤0.01; *** ^a For students in priva	*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported ^a For students in private schools, and those assigned to other intervention conditions all findings were NS, ^b p value not reported									

5.3 Classroom-based: substances including alcohol

5.3.1 Overview of evidence identified

A total of 54 articles were identified that reported on evaluations of 22 classroom-based programmes targeting substance use (including alcohol) across 34 studies. Eleven articles reported on evaluations of the Drug Abuse Resistance Education (DARE) programme. Four articles by Clayton et al (1991; 1996), Bennett (1995) and Lynam et al (1999) examined the effectiveness of DARE in a cohort of sixth grade students in schools in Kentucky, USA and were grouped as a single study. Three articles reported on a cohort of students in Illinois, USA (Rosenbaum et al., 1994; Ennett et al., 1994; Rosenbaum & Hanson 1998), following students from sixth through twelfth grade. Dukes et al (1996; 1997) examined the long-term effectiveness of DARE, following a cohort of students in Colorado Springs, USA over six years and two additional studies (Ringwalt et al., 1991; Harmon 1993) examined the short-term effectiveness of DARE in two different student cohorts. Nine articles reported on evaluations of Botvin's Life Skills Training (LST) programme. Seven articles by Botvin et al (1990a, 1990b, 1995a, 1995b, 1997, 2001a, 2001b) reported on five studies of the programme. Two studies (Botvin et al., 1990a; 1990b; 1995a) focused on delivery of the LST programme among predominantly White, middle-class seventh grade students and three studies (Botvin et al., 1997, 2001a, 2001b; Griffin et al., 2003) focused on delivery of LST among populations of inner-city, minority students. Fraguela et al (2003) examined the effects of the LST programme in a Spanish school.

Thirty studies were conducted in the USA, and one each in Canada, Spain and the Netherlands. The EU-Dap study of the Unplugged programme (Faggiano et al., 2008) was conducted across seven European countries: Austria, Belgium, Germany, Greece, Italy, Spain and Sweden. Although all of the programmes were primarily classroom-based, five programmes (Unplugged, Project Alert, Coalition for Youth Quality of Life, Keepin it REAL and Be Under Your Own Influence/All Stars) combined school components with family- and/or community-based components. Two studies (Perry et al., 2003; Spoth et al., 2002, 2005, 2008) combined two originally school-based only programmes (DARE and LST, respectively) with components targeting parents. Perry et al (2003) examined the effectiveness of DARE Plus, which combined the core components of the DARE programme with a 4-session classroom-based peer-led, parental involvement programme, extracurricular activities and the formation of neighbourhood action teams. Spoth et al (2002; 2005; 2008) examined the combination of the standard LST programme (15 sessions in seventh grade) with the family-based Strengthening Families programme.

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Snow et al., 1992; 1997	CBA -	USA n=1,360 6th grade	School	Adolescent Decision Making programme: Decision-making, group process skills, social network utilisation	NR	NR
Fearnow-Kenney et al., 2003	RCT -	USA n=653 13-18 years	School	All Stars Senior: Two activities per week,; general health education programme	NR	Teachers
Shope et al., 1996b; Shope et al., 1998	CBA -	USA n=422 5th-8th grade	School	Based on AMPS : Tobacco and drug use in addition to alcohol; 7 lessons in 5th-6th grade; 8 lessons in 7th-8th grade	NR	Teachers
Slater et al., 2006	RCT +	USA n=4,216 mean 12.2 years	School, community	Be Under Your Own Influence/All Stars: Two year programme, 13 sessions first year, 7 booster sessions second year; printed media material in school, community-based participation campaign with workshops	NR	Teachers, community leaders
Dedobbeleer & Desjardins, 2001	NRCT -	Canada n=791 6th and 8th grade	School, community, family	Coalition for Youth Quality of Life : Three year programme; youth educational programmes (first year only), parent education programmes, alternative activities, youth mobilisation and support systems for youth in trouble.	Refusal skills	Teachers
Dukes et al., 1996; 1997	CBA -	USA n=849 5th-6th grade	School	DARE (Colorado Springs): DARE curriculum	NR	NR
Ennett et al., 1994; Rosenbaum et al., 1994; Rosenbaum & Hanson, 1998	NRCT +	USA n=1,803 10-11 years	School	DARE (Illinois): standard 16 week DARE curriculum	Social influence theory	Police
Clayton et al., 1991	RCT -	USA n=2,091 11-12 years	School	DARE (Kentucky): 16 weeks; DARE curriculum	Social influence theory	External, police
Clayton et al., 1996	See Clayton et al., 1991	USA n=2,071 6th grade	School	DARE (Kentucky): 16 weeks; DARE curriculum	NR	External, police

Table 5.9. Summary of programme content: classroom-based (substance use including alcohol)

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Bennett, 1995	See Clayton et al., 1991	USA n=1,801 6th grade	School	DARE (Kentucky): DARE curriculum		External, police
Lynam et al., 1999	See Clayton et al., 1991	USA n=1,002 11-12 years	School	DARE (Kentucky): DARE curriculum	NR	Police
Ringwalt et al., 1991	RCT -	USA n=1,402 mean 10.4 years	School	DARE (North Carolina): 17 sessions; core DARE curriculum	Social influence theory	Police
Harmon, 1993	CBA +	USA n=708 mean 10.3 vears	School	DARE (South Carolina): DARE curriculum	NR	Police
Perry et al., 2003	RCT +	USA n=6,237 7th grade	School, family	DARE vs. DARE Plus: 10 session, core DARE curriculum; or DARE Plus, 10 session DARE curriculum + 4 session, peer- led, parental involvement programme, extracurricular activities over two school years, neighbourhood action teams	Social influence theory	Police, teachers, peers
Simons-Morton et al., 2005	RCT +	USA n=2,651 6th grade	School	Going Places: Three year programme, 18 sessions 6th grade, 12 sessions 7th grade, 6 sessions 8th grade; social skills curriculum, parent education, school environment change	Social skills training, social development theory, social cognitive theory	Teachers
Cuijpers et al., 2001; Smit et al., 2003	NRCT +	The Netherlands n=1,930 mean 12.4 years	School	Healthy School and Drugs Project: Three year programme, three lessons per year; educational lessons, activities, videos and brochures, refusal skills, increasing self- esteem	Behaviour change theory	Teachers
Brewer, 1991	RCT +	USA n=54 10th grade	School	Here's Looking at You 2000: 9 sessions (40-minute duration); social skills training curriculum, videos.	Problem behaviour theory, social learning theory	Certified school psychologist
Hecht et al., 2003; Gosin et al., 2003; Kulis et al., 2005 ^a ; Kulis et al., 2007 ^b	RCT -	USA n=6,035 mean 12.5 years	School, community	Keepin it REAL: 10 lessons (7th grade), booster sessions (8th grade); resistance and avoidance skills, classroom videos and televised PSAs	Social learning theory, communication competence theory	Teachers

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Warren et al., 2006	RCT +	USA n=4,734 7th grade	School	Keepin it REAL: see other Keepin it REAL	Social learning theory, communication competence theory	Teachers
Botvin et al., 1990a; Botvin et al., 1995a	RCT +	USA n=4,466 7th grade	School	LST: 15 sessions; booster sessions in grade 8 and 9; demonstration, behavioural rehearsal, feedback and reinforcement, homework assignments	Cognitive-behavioural theory, social influence theory	Teachers
Botvin et al., 1990b	RCT +	USA n=1,311 7th grade	School	LST: 20 sessions in 7th grade, with and without booster sessions; LST curriculum	NR	Teachers, Peers
Botvin et al., 1995b	RCT +	USA n=757 7th grade	School	LST: Two sessions per week (40-minutes duration); demonstration, behavioural rehearsal, feedback and reinforcement, homework assignments; peer-led story-telling, videos and demonstrations	NR	Teachers, Peers
Botvin et al., 1997	NRCT -	USA n=833 11-15 years	School	LST: 15 sessions; group discussion, demonstration, group modelling, behavioural rehearsal, feedback and reinforcement, and homework	NR	Teachers
Botvin et al., 2001a; Botvin et al., 2001b; Griffin et al., 2003	RCT +	USA n=3,621 mean 12.9 years	School	LST: 15 sessions, 10 booster sessions in 8th grade; group discussion, demonstration, modelling, behavioural rehearsal, feedback and reinforcement, homework	NR	Teachers
Fraguela et al., 2003	NRCT -	Spain n=1,029 14.3 years	School	LST: 16 sessions; LST curriculum + component focusing on leisure activities	NR	Teacher or research staff
Smith et al., 2004;Vicary et al., 2004	RCT +	USA n=732 7th grade	School	LST: vs. Infused-LST: LST: 15 sessions 7th grade, booster sessions 8th grade and 9th grade; standard LST curriculum I-LST: LST core components taught across different subject areas	NR	Teachers

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Spoth et al., 2002; 2005; 2008	RCT +	USA n=1,673 7th grade	School, family	LST vs. LST + SFP 10-14 LST: 15 sessions 7th grade, booster sessions 8th grade; standard LST curriculum. LST + SFP 10-14: as above + 7 sessions SFP; discussions, skill-building activities, videotapes, games	SPF: biopsychosocial model LST: social learning theory, problem behaviour theory	SPF: facilitators LST: Teachers
Eisen et al., 2002	RCT +	USA n=7,426 11-14 years	School	Lions Quest Skills for Adolescence: One year programme, 40 sessions (35-45 minutes each); social competency and refusal skills training	NR	External
Lennox & Cecchini, 2008	NRCT +	USA n=995 12-20 years	School	NARCONON [™] drug education curriculum: Eight modules; health motivation, social skills, social influence recognition, knowledge development.	NR	Other
Caplan et al., 1992	NRCT -	USA n=282 11-14 years	School	Positive Youth Development Programme: Six classes over 15 weeks (50-minutes each); curriculum covered stress management, self-esteem, problem solving, substance use and health information, promotion of social and personal competence.	Social skills training, social competence	Health educators, Teachers
Bell et al., 1993; Ellickson & Bell, 1990	RCT +	USA n=6,527 7th-8th grade	School, family	Project ALERT: 11 lessons (8 in 7th grade; 3 in 8th grade); reinforcement of group norms, role-play, skills rehearsal	Social influence theory, health belief model, social learning theory self-efficacy theory	Teachers, Peers
Ellickson et al., 2003	RCT +	USA n=4,689 7th-8th grade	School, family	Revised Project ALERT: Two year programme, 14 lessons (11 in 7th grade, 3 in 8th grade); normative education, social influence programme, resistance skills training	Social influence model, health belief model, social learning model, self-efficacy theory	Teachers
Graham et al., 1990	RCT +	USA n=5,070 12 years	School	Project SMART : 12 sessions; social skills programme, or affect management programme	NR	Health educators
Sussman et al., 1998; Sun et al., 2006	RCT +	USA n=1,578 mean 16.8 years	School (continuation high schools)	Project TND: Nine sessions; listening skills, alternative coping skills, making non-drug choices	NR	Health staff educators

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)			
Dent et al., 2001	RCT -	USA n=1,208 14-17 years	School (general high school)	Project TND: Three week programme (9 sessions); drug abuse prevention, skills to change, effective listening skills, effective communication skills, self control, myths about drug use, nature and consequences of drug use, effects of drug use on others, decision making skills.	Decision skills	Teachers			
Sussman et al., 2003	RCT +	USA n=1,037 mean 16.7 years	School	Project TND: 12 sessions (45 minutes each); cognitive misperception correction activities, social skills, listening skills, decision making skills	NR	Health educators (assisted self instruction)			
Faggiano et al., 2008	RCT +	7 European countries n=7,079 12-14 years	School, family	Unplugged : 12 one-hour units. Delivered in three formats: basic curriculum, or with the addition of (1) peer activities or (2) parent activities (three interactive workshops).	Social influence theory	Teachers, peers			
^a Reported data on a reported current sub-	^a Reported data on a subsample of students who reported their race or ethnicity as Mexican American, Mexican or Chicano; ^b Reported data on a subsample of students who reported current substance use at baseline								

Ten programmes, All Stars Senior, Be Under Your Own Influence/All Stars, a curriculum based on the original AMPS, Coalition for Youth Quality of Life, Going Places, Healthy School and Drugs Project, Keepin it REAL, Lions Quest Skills for Adolescence, Project Alert and Project Towards No Drug Abuse (Project TND), were delivered solely by teachers. Project Alert and the Unplugged curriculum were taught primarily by teachers, with or without assistance from peer leaders, and versions of the Life Skills Training programme were designed to be delivered by teachers or peer leaders. The provider of the Adolescent Decision Making programme (Snow et al., 1992, 1997) was not reported and external facilitators including health educators (Project SAVE, Positive Youth Development programme), the police (DARE), and certified school psychologists (Here's Looking at You, 2000) were utilised as providers in the remaining programmes.

The theoretical framework was not reported for nine programmes (Adolescent Decision Making programme; All Stars Senior; a curriculum based on the original AMPS; Be Under Your Own Influence/All Stars; Coalition for Youth Quality of Life; Lions Quest Skills for Adolescence; NARCONON drug education curriculum; Positive Youth Development Programme; and Project SAVE. Six programmes (Going Places, Healthy School and Drugs Project, Here's Looking at You 2000, Keepin it REAL, LST, and Project ALERT) were based on a combination of theories. The most commonly applied theory was social influence theory (n=5 programmes).

The overall number of students recruited to participate in the included studies ranged from 54 to over 6,000 students. Evaluation of ten programmes (DARE, Project TND, Healthy School and Drugs Project, Going Places, Be Under Your Own Influence/All Stars, Project ALERT, Keepin it REAL, Project SAVE, Unplugged and Lion's Quest Skills for Adolescence) were based on sample sizes of more than 1,500 students.

The majority of programmes targeted students aged 14 or younger. Three programmes, All Star Seniors, Healthy School and Drugs Project and the NARCONON drug education curriculum targeted students over a broad range of ages, and two programmes, Project TND and Here's Looking at You 2000 targeted older adolescents.

Evaluations of two programmes were based on immediate post-test follow-up only: All Stars Senior and the Healthy School and Drugs Project. The Healthy School and Drugs Project was a 3-year programme but follow-up data were not reported beyond the final year of the programme. Long-term follow-up data were reported for nine programmes: DARE, LST, Keepin' it REAL, Project ALERT, Project TND, Adolescent Decision Making programme, a substance use prevention curriculum based on AMPS, Be Under Your Own Influence/All Stars, and the Coalition for Youth Quality of Life.

5.3.2 Quality assessment

Of the 34 studies, 23 were based on RCT designs, seven were non-randomised controlled trials and four were CBA studies. Twenty-two of the 23 RCTs were based on cluster randomisation. All RCTs were reported to have been based on random assignment but in the majority of studies the methods of randomisation and whether an adequate concealment method was used were not reported. Seven studies (Botvin et al., 1990b, 1995b; Brewer, 1991; Hecht et al., 2003; Warren et al., 2006; Caplan et

al., 1992; Ennett et al., 1994) did not report how many participants were lost to follow-up. Losses to follow-up greater than 20% were reported in seven studies (Bell et al., 1993; Botvin et al., 1990a; Clayton et al., 1991; Simons-Morton et al., 2005; Slater et al., 2006; Cuijpers et al., 2001; Dedobbeleer & Desjardins, 2001). Outcomes measures were reported as reliable across the majority of RCTs and NRCTs, but six studies (Clayton et al., 1991; Eisen et al., 2002; Ellickson et al., 2003; Hecht et al., 2003; Ennett et al., 1994; Fraguela et al., 2003) either did not report on the validity or reliability of the outcome measures used or this aspect of study design was poorly reported. Baseline comparability was poorly reported or not reported on in seven studies (Clayton et al., 1991; Fearnow-Kenney et al., 2003; Graham et al., 1990; Hecht et al., 2003; Simons-Morton et al., 2005; Sussman et al., 2003; Cuijpers et al., 2001) and in some studies it was not clear how many participants were assigned to intervention and control groups. Five studies (Bell et al., 1993; Brewer, 1991; Ellickson et al., 2003; Faggiano et al., 2008; Slater et al., 2006) reported that all participants had been analysed in intention to treat analyses. The quality of the four CBA studies was generally poor, but one study (Harmon, 1993) was rated moderate quality. None of the studies clearly examined baseline comparability between intervention and control groups and only one study (Harmon, 1993) reported on the follow-up of study participants.

5.3.3 Findings

5.3.3.1 Short-term results (<6 months)

Short-term follow-up data was presented for ten programmes: DARE (Bennett, 1995; Clayton et al., 1991; Ennett et al., 1994; Rosenbaum et al., 1994; Harmon, 1993; Ringwalt et al., 1991); LST (Botvin et al., 1990a; Botvin et al., 1995a; Botvin et al., 1997; Botvin et al., 2001a; Botvin et al., 2001b; Griffin et al., 2003; Fraguela et al., 2003); infused LST (Smith et al., 2004; Vicary et al., 2004); Project ALERT (Ellickson & Bell, 1990; Bell et al., 1993); All Stars Senior (Fearnow-Kenney et al., 2003); a substance use curriculum based on AMPS (Shope et al., 1996b; Shope et al., 1998); Lions Quest Skills for Adolescence (Eisen et al., 2002), the Healthy School and Drugs Project (Cuijpers et al., 2001; Smit et al., 2003), the Positive Youth Development Programme (Caplan et al., 1992), and the Unplugged curriculum (Faggiano et al., 2008).

Knowledge and understanding

Short-term effects on knowledge were examined for three programmes, LST (Botvin et al., 1990a; 2001a), a programme based on the original AMPS curriculum (Shope et al., 1996b), and the Healthy School and Drugs Project (Cuijpers et al., 2001). Botvin et al (1990a) reported that drinking knowledge was significantly higher in students who had received at least 60% of the LST programme relative to the control group (p<0.001). Botvin et al (2001a) found that delivery of the LST programme to inner-city minority students was associated with greater drinking knowledge among intervention students compared to controls (p<0.05). Shope et al (1996b) found that delivery of a substance use prevention programme based on the AMPS curriculum had a significant effect on substance-related knowledge but not knowledge of pressures to use substances or skills knowledge. Cuijpers et al

(2001) reported that significant intervention effects of the Healthy School and Drugs programme were found on alcohol knowledge at both the end of the programme and at the mid-programme follow-up.

Attitudes and values

Short-term effects on attitudes and values were examined across four programmes, DARE (Bennett, 1995; Clayton et al., 1991; Ennett et al., 1994; Rosenbaum et al., 1994; Harmon, 1993; Ringwalt et al., 1991), LST (Botvin et al., 1990a; 1997; 2001a), the Healthy School and Drugs Project (Cuijpers et al., 2001) and the Positive Youth Development Programme (Caplan et al., 1992). Two studies of the delivery of DARE in Kentucky (Clayton et al., 1991; Bennett, 1995) found that students in the DARE group reported a significant increase in negative attitudes towards alcohol compared to students in the control group (p<0.01). Clayton et al (1991) also found that there were no effects of DARE on selfesteem or peer-pressure resistance. Ennett et al (1994) found that DARE had a significant, positive effect on participant's self-esteem, but no effects on any of the other variables measuring students' attitudes towards drugs. Rosenbaum et al (1994) reported that the only significant effect of DARE on attitudes and beliefs was a significant effect on the perceived media influence on beer drinking. Harmon (1993) found that DARE students reported higher levels of belief in prosocial norms (p<0.01), reported less association with drug using peers (p<0.01), had more negative attitudes towards substances (p<0.01). They found no difference between DARE and control students on commitment and attachment to school or self-esteem. Ringwalt et al (1991) found that compared with control students, DARE students perceived alcohol costs to be higher, and the media's portrayal of beer drinking to be more favourable. Botvin et al (1990a) found that among students who had received at least 60% of the LST programme there was no difference in drinking attitudes compared to control students. Based on a study among inner-city minority students, Botvin et al (1997) reported that future intentions to drink beer or wine within the next year were lower in the intervention group compared to the control group (p<0.01). There were no intervention effects on intentions to drink liquor. Intervention students reported significantly lower normative expectations for adult and peer drinking (p<0.01 and p<0.001, respectively). There was no difference between intervention and control students in terms of anti-drinking attitudes. Botvin et al (2001a) found LST students had lower peer and adult normative expectations for drinking (both p<0.05). Significant intervention effects were found for the Healthy School and Drugs Project (Cuijpers et al., 2001) on measures of attitudes towards alcohol at end of the second year of the three-year programme. However, findings were not significant at the end of the first year of the programme or at the end of the programme (three years from baseline). Measures of self-efficacy towards alcohol use showed significant intervention effects at the end of the first programme year, but not at the second or third years. Caplan et al (1992) found that relative to intervention students, the intentions of control students increased significantly with respect to beer and hard liquor (both p<0.05).

Personal and social skills

Short-term effects on personal and social skills were examined for DARE (Ennett et al., 1994; Harmon, 1993; Ringwalt et al., 1991) and LST (Botvin et al., 1997). Ennett et al (1994) found that there were no effects of DARE on social skills. Two studies (Ringwalt et al., 1991; Harmon, 1993) found that

compared to control students, students who participated in DARE had higher levels of assertiveness. Harmon (1993) also reported that DARE students had more positive or prosocial peer associations than control, but there were no differences between DARE and control students' use of coping strategies or social integration. Botvin et al (1997) found that inner-city minority students who participated in the LST programme were more likely than controls to report use of refusal skills (p<0.05). However, there was no difference between groups in terms of skills in relation to decision-making, anxiety reduction, communication and social assertiveness.

Health and social outcomes relating to alcohol use and sexual health

Short-term effects on health outcomes related to alcohol use were examined for ten programmes: DARE (Bennett, 1995; Clayton et al., 1991; Ennett et al., 1994; Rosenbaum et al., 1994; Harmon, 1993; Ringwalt et al., 1991), LST (Botvin et al., 1990a, 1995a, 1997, 2001a, 2001b; Fraguela et al., 2003); infused LST (Smith et al., 2004; Vicary et al., 2004), Project ALERT (Ellickson & Bell, 1990; Bell et al., 1993; Ellickson et al., 2003), Healthy School and Drugs Project (Cuijpers et al., 2001); All Stars Seniors (Fearnow-Kenney et al., 2003), a substance use prevention programme based on AMPS (Shope et al., 1996b, 1998), Lions Quest Skills for Adolescence (Eisen et al., 2002), the Positive Youth Development Programme (Caplan et al., 1992), and the Unplugged curriculum (Faggiano et al., 2008).

There were no short-term effects of All Stars Seniors (Fearnow-Kenney et al., 2003) or the Lions Quest Skills for Adolescence programme (Eisen et al., 2002; 2003) on alcohol use, and the short-term effects of the DARE programme on alcohol use were mixed. Five studies (Clayton et al., 1991; Bennett, 1995; Ennett et al., 1994; Rosenbaum et al., 1994; Ringwalt et al., 1991) found that there were no effects of the programme on alcohol use. Rosenbaum et al (1994) also found that DARE exposure had no effects on the initiation of alcohol use. Harmon (1993) found that DARE students reported less alcohol use in the last year compared to control students (p<0.05), but that students did not differ in terms of their alcohol use in the past month. Two studies (Botvin et al., 1990a; Fraguela et al., 2003) of LST found no effects of the intervention on alcohol use. Botvin et al (1990a) found that delivery of the LST programme over three years did not have significant effects on drinking frequency or amount in a high fidelity sample. The study by Fraguela et al (2003) replicated the LST programme with a sample of Spanish students, finding no impact of the intervention on alcohol use. Two studies reported positive intervention effects. Botvin et al (1997) found that compared to control students, inner-city minority students who received a 15-session LST programme reported drinking alcohol less often (p<0.01) and consumed significantly less alcohol (p<0.001). Among seventh grade inner-city students who participated in LST (Botvin et al., 2001a; Botvin et al., 2001b; Griffin et al., 2003), mean scores for drinking frequency and drinking quantity were significantly lower in the intervention group than the control group (both p<0.05). However, when the intracluster correlations were taken into account these differences in alcohol use became non-significant. Smith et al (2004) found that at the end of seventh grade (immediate post-test), neither LST nor infused LST had significant effects on alcohol use in male students. In addition, for females a significant reduction in the frequency of alcohol use was observed in the LST condition only. Short-term follow-up results of the original

Project ALERT (Ellickson & Bell, 1990; Bell et al., 1993) were reported at the end of the seventh grade, midway through the two-year programme, and at the beginning and end of eighth grade (before and immediately after delivery of the second year of the curriculum). At the end of seventh grade, compared to control students, baseline non-drinkers were less likely to have initiated drinking in the subsequent 3-months (p<0.05) and were less likely to report drinking monthly (p<0.05). However, at the end of eighth grade follow-up (immediate post-test), there was no difference between intervention and control students on any measure of alcohol use, with the exceptions of students in the teen leader condition who reported more alcohol use in the past month than controls at the beginning of the eighth grade, although the difference was not significant. Ellickson et al (2003) found that there were no significant effects on initial or current drinking behaviours, but students who received the revised Project ALERT programme reported lower overall alcohol misuse scores (p<0.05) and fewer negative consequences arising from their alcohol use (p<0.05). Further analysis of the sample revealed that effects were greatest among students who had tried alcohol three or more times in the past month. Compared to students who had not received a substance abuse prevention programme based on AMPS (Shope et al., 1996b; Shope et al., 1998), intervention students reported significantly less use of alcohol (p<0.001). Cuijpers et al (2001) reported significant effects of the three-year Healthy School and Drugs programme on measures of alcohol use at all follow ups (1, 2 and 3 years from baseline, respectively). At the 2-year follow-up, after the delivery of the alcohol specific intervention components, significant positive effects were found for the proportion of students who drank, however none of the other outcomes relating to use of alcohol showed significant effects of the intervention. At the 3-year follow-up, when all intervention components had been delivered, the authors reported significant positive effects of the intervention on the proportion of weekly users of alcohol and measures of the number of drinks per week consumed and drinks per occasion, as well as the proportion of participants who reported drinking. Relative to control students, Caplan et al (1992) reported that intervention students who participated in the Positive Youth Development Programme significantly reduced their frequency of having three or more drinks on a single occasion (p<0.05), and the amount of beer, wine, or liquor they usually consumed on one occasion (p<0.05).

Three studies (Botvin et al., 1990a; 1997; 2001a) that examined the LST programme found positive intervention effects on drunkenness. Among students who were exposed to 60% or more of the curriculum, compared to control students, those who taught the programme by teachers who received training by video reported significantly fewer occasions of drunkenness (p<0.05). In two studies (Botvin et al., 1997; 2001a) utilising samples of inner-city students, students who received the LST programme were less likely than controls to have got drunk (p<0.05 and p<0.01, respectively). Smith et al (2004) found that for females both LST and infused LST were associated with a significant reduction in the frequency of binge drinking. Students who participated in a substance abuse prevention programme based on the AMPS curriculum reported significantly lower levels of alcohol misuse than controls. Students who received the revised Project ALERT programme (Ellickson et al., 2003) were significantly less likely to engage in drinking that resulted in negative consequences (including getting sick, getting in a physical fight, getting in trouble at home or school, or doing something they later regretted). Faggiano et al (2008) examined the effectiveness of the Unplugged

curriculum as part of the EU-Dap study. Three versions of the curriculum were examined, incorporating peer and parental components. However, the main analyses were conducted on all four intervention arms pooled together. A statistically significant effect of the Unplugged curriculum (Faggiano et al., 2008) was observed at the short-term follow-up (3 months) for any episode and for frequent episodes of drunkenness in the past 30 days. There were no effects of the Lions Quest Skills for Adolescence programme (Eisen et al., 2002) on binge drinking in the past month.

5.3.3.2 Medium-term results (up to 12 months)

Medium-term follow-up data were reported for 12 programmes: DARE (Bennett, 1995; Ennett et al., 1994); LST (Botvin et al., 1990b; Botvin et al., 2001a; Fraguela et al., 2003); LST + SPF (Spoth et al., 2002; 2005; 2008); infused LST (Smith et al., 2004; Vicary et al., 2004); Project TND (Sussman et al., 1998; Dent et al., 2001); a programme based on the original AMPS curriculum (Shope et al., 1996b; Shope et al., 1998); Going Places (Simons-Morton et al., 2005); Here's Looking at You 2000 (Brewer, 1991); Lions Quest Skills for Adolescence (Eisen et al., 2002); the NARCONON drug education curriculum (Lennox & Cecchini, 2008); Project Alert (Bell et al., 1993) and Project SAVE (Graham et al., 1990).

Knowledge and understanding

Medium-term effects on knowledge were examined for three programmes: LST (Botvin et al., 1990b; 2001a), a substance use prevention curriculum based on the original AMPS (Shope et al., 1996b; 1998); and the NARCONON drug education curriculum (Lennox & Cecchini, 2008). Two studies (Botvin et al., 1990b; Botvin et al., 2001a) of the LST programme among predominantly White, middle class students and inner-city students, respectively, found positive medium-term effects of the intervention on knowledge. Botvin et al (1990b) found that a 20-session version of the programme had significant effects on drinking knowledge among students in the peer-led booster and non-booster groups (p<0.001 and p<0.01, respectively), but not among students in the teacher-led booster group. Short-term effects on knowledge of the effects of substance use were maintained at the 1-year follow-up in a study of a substance abuse prevention programme based on the AMPS curriculum (p<0.05; Shope et al., 1996b; Shope et al., 1998). However, as reported at PT, there were no intervention effects on knowledge of pressures to use substances or knowledge of skills. At 6-months follow-up, Lennox and Cecchini (2008) found that students who received the NARCONON drug education curriculum had a significantly better knowledge of the content of the programme than control students.

Attitudes and values

Medium-term intervention effects on attitudes and values were examined across four programmes: DARE (Bennett, 1995; Ennett et al., 1994), LST (Botvin et al., 1990b; 2001a), Lion's Quest Skills for Adolescence (Eisen et al., 2002) and the NARCONON drug education curriculum (Lennox & Cecchini, 2008). Bennett (1995) reported that participation in DARE was associated with an increase in negative attitudes towards alcohol; however this finding only reached significance among students who were average achievers. A second study of the DARE curriculum (Ennett et al., 1994) found no effects of the intervention on attitudes towards drugs (including alcohol) or self-esteem. Among a

predominantly white sample of students, Botvin et al (1990b) found that participation in LST was associated with an increase in drinking attitudes (i.e. more negative attitudes towards alcohol) among those who received peer-led booster and non-booster sessions. Students who were assigned to the teacher-led booster condition reported more positive attitudes towards alcohol at the medium-term follow-up. A study based on a sample of inner-city students who participated in the LST programme (Botvin et al., 2000a, 2001b), found that the intervention reduced intentions to drink alcohol (p<0.05), increased negative attitudes towards alcohol (p<0.01) and decreased peer and adult normative expectations (p<0.01 and p<0.05, respectively). The Lion's Quest Skills for Adolescence programme (Eisen et al., 2002; 2003) did not impact on students' attitudes and beliefs in terms of self-efficacy, intentions, or perceived peer use. Lennox and Cecchini (2008) reported that after controlling for baseline differences, control group students reported a greater tendency to plan to get drunk in the next year compared with the intervention group (p<0.01).

Personal and social skills

One study (Ennett et al., 1994) reported on the medium-term impact of the DARE curriculum on social skills, finding no impact of the intervention on this measure.

Health and social outcomes relating to alcohol use and sexual health

Medium-term intervention effects on alcohol use were reported for 11 programmes: DARE (Bennett, 1995; Ennett et al., 1994); LST (Botvin et al., 1990b; 2001a; Fraguela et al., 2003); LST vs. LST + SPF 10-14 (Spoth et al., 2002); LST vs. I-LST (Smith et al., 2004; Vicary et al., 2004); Project TND (Sussman et al., 1998; Dent et al., 2001); a substance abuse prevention curriculum based on AMPS (Shope et al., 1996b; Shope et al., 1998); Going Places (Simons-Morton et al., 2005); Here's Looking at You 2000 (Brewer, 1991); Lions Quest SFA (Eisen et al., 2002; 2003); NARCONON drug education curriculum (Lennox et al., 2008); and Project SAVE (Graham et al., 1990).

Seven programmes, infused-LST (Smith et al., 2004), a substance abuse prevention curriculum based on AMPS (Shope et al., 1996b; Shope et al., 1998), Going Places (Simons-Morton et al., 2005), Here's Looking at You 2000 (Brewer, 1991), Lions Quest SFA (Eisen et al., 2002; 2003), the NARCONON drug education curriculum (Lennox et al., 2008), and Project Alert (Bell et al., 1993) were found to have no medium-term impact on alcohol use. The effects of DARE were weak; Bennett (1995) reported that intervention students who received the DARE curriculum reported lower rates of past year alcohol use, however this finding only reached significance among low achieving students only, and a second study of the medium-term effects of DARE (Ennett et al., 1994) reported no intervention impacts. Among predominantly white, middle class students, Botvin et al (1990b) found that LST had significant positive effects on weekly and monthly drinking, and drinking frequency at the 1-year follow-up. In addition, students in the peer booster condition reported consuming less alcohol per occasion than students in the control group (p<0.05). However, students in the teacher booster condition reported higher levels of both weekly and monthly drinking than control students (both p<0.001), and drank more frequently (p<0.001). Inner-city students who received the LST programme in seventh grade (Botvin et al., 2001a) reported drinking less frequently and consuming less alcohol that control students (both p<0.01) at the 1-year follow-up. These findings remained significant when

adjusted to account for clustering. Fraguela et al (2003) found that students who were taught the LST programme by members of their research team reported significantly lower consumption of beer (but not spirits) than control students at the 1-year follow-up. However, the significance of these findings was not reported. Spoth et al (2002; 2005) examined the combination of LST with the Strengthening Families Programme (SFP). Significantly fewer students in the LST + SFP group were 'new users' at the 1-year follow-up relative to the control and LST only groups (both p<0.05), but there were no statistically significant effects on regular alcohol use in either intervention group. Sussman et al (1998) found no effects of Project TND on alcohol use for the whole sample of students, however, further analyses based on baseline alcohol use showed that at the 1-year follow-up, continuation school students in the classroom-only group who reported higher levels of alcohol use at baseline had significantly lower alcohol use than control students (p<0.01). There were no effects of the programme in baseline nonusers. Dent et al (2001) examined the effects of Project TND in a general high school sample. They also found no effect of the curriculum on alcohol use among baseline nonusers and students reporting lower levels of alcohol use in a general high school sample. Among students who reported higher levels of baseline alcohol use, intervention students reported significantly lower alcohol use at the 1-year follow-up compared to the control students. The Project SMART programme (Graham et al., 1990) had significant effects on the alcohol use index (p=0.03) at 1-year follow-up.

Inner-city students who participated in LST (Botvin et al., 2001a; 2001b) reported getting drunk less frequently and significantly fewer occasions of binge drinking than controls (both p<0.05). Smith et al (2004) found no effects of LST or an infused-LST curriculum on binge drinking or drunkenness at the 1-year follow-up in this study. There were also no intervention effects for three further programmes: a substance use prevention programme based on AMPS (Shope et al., 1996; 1998), Lions Quest Skills for Adolescence (Eisen et al., 2002; 2003) and the NARCONON drug prevention curriculum (Lennox et al., 2008).

5.3.3.3 Long-term results (>12 months)

Long-term follow-up results were reported for 11 programmes: DARE (Bennett, 1995; Clayton et al., 1996; Dukes et al., 1996; 1997; Ennett et al., 1994; Rosenbaum & Hanson, 1998; Lynam et al., 1999); DARE Plus (Perry et al., 2003); LST (Botvin et al., 1990a; Botvin et al., 1995a; Botvin et al., 1995b; Botvin et al., 2001a; Botvin et al., 2001b; Griffin et al., 2003; Fraguela et al., 2003); infused LST (Spoth et al., 2002; 2005; 2008); Keepin It REAL (Hecht et al., 2003; Gosin et al., 2003; Kulis et al., 2005, 2007; Warren et al., 2006); Project ALERT (Ellickson et al., 1993); Project TND (Sussman et al., 1998; Sun et al., 2006; Sussman et al., 2003); Adolescent Decision Making programme (Snow et al., 1992; 1997); a programme based on the original AMPS curriculum (Shope et al., 1996b; Shope et al., 1998); Be Under Your Own Influence/All Stars (Slater et al., 2006); and the Coalition for Youth Quality of Life (Dedobbeleer & Desjardins, 2001).

Knowledge and understanding

Two studies (Botvin et al., 2001a; Shope et al., 1996b) reported long-term intervention effects on knowledge for two programmes, LST, and a substance use curriculum based on AMPS. Neither programme had long-term impacts on knowledge, 2- and up to 6-years after intervention, respectively.

Attitudes and values

Long-term intervention effects on attitudes and values related to alcohol use were examined across five programmes: DARE (Bennett, 1995; Clayton et al., 1996; Dukes et al., 1996; 1997; Ennett et al., 1994; Lynam et al., 1999); DARE Plus (Perry et al., 2003); LST (Botvin et al., 1995b; Botvin et al., 2001a); Keepin it REAL (Hecht et al., 2003; Gosin et al., 2003; Kulis et al., 2005); and the Coalition for Youth Quality of Life (Dedobbeleer & Desjardins, 2001).

The long-term effects of DARE on attitudes were equivocal. Three studies (Bennett, 1995; Dukes et al., 1996; 1997; Ennett et al., 1994) found that DARE had no long-term effects on attitudes, but Clayton et al (1996) found that students who participated in DARE had more negative general and specific drug attitudes. Clayton et al (1996) also reported that the DARE curriculum had a long-term positive effect on capability to resist peer pressure and peer norms for drug use. However, two studies (Dukes et al., 1996, 1997; Lynam et al., 1999) reported no long-term effects on resistance to peer pressure, and there were no long-term effects of DARE on self-esteem (Ennett et al., 1994; Lynam et al., 1999). Perry et al (2003) examined the effectiveness of an enhanced version of the DARE programme, DARE Plus. At the 18-month follow-up, males who received the DARE Plus programme were less likely than those in the control group to show increases in alcohol intentions (p<0.05). There were no differences between intervention and control females. Botvin et al (1995b) found that minority students who received the LST programme reported significantly lower intentions to use beer or wine, or hard liquor in the future compared to the information only control group (p<0.01 and p<0.05, respectively). Students who received a culturally focused intervention reported significantly lower intentions to use beer or wine relative to the information only control (p<0.01), but only marginally fewer intentions to use hard liquor (p=0.06). At 2-years follow-up, Botvin et al (2001a) found that there were no effects of LST on drinking attitudes or peer drinking norms. Among students who participated in Keepin it REAL (Hecht et al., 2003; Gosin et al., 2003), significant intervention effects were on positive expectancies and descriptive norms. There were no intervention effects on alcohol resistance strategies, intentions, self-efficacy or norms. Among a subsample of students who reported their race or ethnicity as Mexican American, Mexican or Chicano, Kulis et al (2005) found that the intervention had no effects on attitudes and belief measures. Sixth grade students who participated in the Coalition for Youth Quality of Life project reported significantly greater change scores on the measure of self-esteem at the 10-month follow-up (p<0.05) (Dedobbeleer & Desjardins, 2001). However, there were no intervention effects on awareness of drug and alcohol problems, intentions to become involved in prevention activities, or enabling factors at either the 10- or 30-month follow-up.

Personal and social skills

Two studies (Ennett et al., 1994; Dedobbeleer & Desjardins, 2001) examined long-term programme effects on personal and social skills for two programmes, DARE and the Coalition for Youth Quality of Life, respectively. Ennett et al (1994) found that DARE had no long-term impact on students' social skills. Students who participated in the Coalition for Youth Quality of Life (Dedobbeleer & Desjardins, 2001) reported an increase in relationship quality with their fathers at the 18-month follow-up, but this effect was not sustained at the 30-month follow-up. There was no impact of the intervention on students' relationships with their mothers.

Health and social outcomes relating to alcohol use and sexual health

Long-term intervention effects on alcohol use were examined across 11 programmes and the following programmes were shown to have no long-term effects on alcohol use: DARE (Bennett, 1995; Clayton et al., 1996; Dukes et al., 1996; 1997; Ennett et al., 1994; Rosenbaum & Hanson, 1998; Lynam et al., 1999); Project ALERT (Ellickson et al., 1993); Project TND (Sussman et al., 1998; Sun et al., 2006; Sussman et al., 2003); Based on AMPS (Shope et al., 1996b; Shope et al., 1998); and the Coalition for Youth Quality of Life (Dedobbeleer & Desjardins, 2001). One programme, the Adolescent Decision Making Programme (Snow et al., 1992; 1997) was shown to have negative effects, with a higher proportion of intervention students than control students reporting alcohol use at the 2-year follow-up.

Five studies reported long-term follow-up data on Botvin's LST programme. Intervention with ethnic minority students produced reductions in drinking frequency and amount at 2 years (Botvin et al., 1995b). However, three studies, including two studies by other research groups (Botvin et al., 1990b; Fraguela et al., 2003; Smith et al., 2004) in Spain and the USA, respectively, found that LST did not have long-term positive effects indicating that there may be issues with the transferability of the programme to other settings. There were no long-term effects of LST or LST + SPF 10-14 on alcohol use (Spoth et al., 2005; 2008). Hecht et al (2003) reported that use of alcohol was found to have increased over time in both the intervention and control groups. However, the increase was significantly less in intervention students. Analysis of the Mexican American, Black/White, and Multicultural versions of 'Keepin' it REAL' revealed that students in each intervention condition reported increased alcohol use over the course of the study. However, increases were significantly smaller in each of the intervention conditions compared to control with regards to alcohol use at 14months. Perry et al (2003) examined the effectiveness of an enhanced version of the DARE programme, DARE Plus. At the 18-month follow-up, boys who received the DARE Plus programme were less likely than those in the control group to show increases in past year and past month alcohol use (p<0.05). There were no differences between intervention and control conditions among girls.

Seven studies (Perry et al., 2003; Botvin et al., 1995a, 2001a; Spoth et al., 2002, 2005; Smith et al., 2004; Shope et al., 1996b, 1998) examined long-term intervention effects on excessive drinking across three programmes, DARE Plus, LST and a curriculum based on AMPS. There were no long-term effects of the AMPS-based curriculum on alcohol misuse and positive effects of the LST + SPF 10-14 curriculum on drunkenness were only significant at 2-year follow-up. At 18-month follow-up,

females who received the DARE Plus curriculum were less likely to report increases in ever having been drunk, compared with girls in DARE only schools (p<0.05). However, there were no other differences between conditions among girls. Three studies of LST (Botvin 1990a, 1995a, 2001a) examined intervention effects on excessive drinking. All three studies found significant, positive intervention effects on drunkenness and one study (Botvin et al., 2003) reported reduced rates of binge drinking among intervention students compared to control students, two years after intervention.

5.3.4 Summary and evidence statements

A total of 52 articles were identified that examined 17 classroom-based programmes aiming to prevent substance use (including alcohol) across 37 studies.

5.3.4.1 Knowledge and understanding

Overall, four studies (Botvin et al., 1990a; Botvin et al., 2001a; Shope et al., 1996b; Cuijpers et al., 2001), including evaluations of LST, AMPS, DARE, and the Healthy School and Drugs Project, respectively, examined programme effects on alcohol-related knowledge. Overall both short and medium-term increases in alcohol knowledge were reported but these were not sustained long-term.

5.3.4.2 Attitudes and values

Twenty-one studies reported outcomes relating to alcohol or substance use attitudes and behavioural intentions across the following programmes: DARE (Bennett, 1995; Clayton et al., 1991; Dukes et al, 1996; Ennett et al., 1994; Lynam et al., 1999; Rosenbaum et al., 1994; Harmon, 1993; Ringwalt et al., 1991), DARE Plus (Perry et al., 2003); LST (Botvin et al., 1990a; 1990b; 1995b; 1997; 2001a), the Healthy School and Drugs Project (Cuijpers et al., 2001), Lions Quest Skills for Adolescence (Eisen et al., 2002); Positive Youth Development programme (Caplan et al., 1992), NARCONON drug education curriculum (Lennox et al., 2008), the Coalition for Youth Quality of Life (Dedobbeleer & Desjardins, 2001) and Keepin' It REAL (Hecht et al., 2003; Kulis et al., 2005).

The majority of studies of the DARE programme found increases in negative views towards alcohol in the short-term (Bennett, 1995; Clayton et al., 1991; Ennett et al., 1994; Harmon, 1993; Ringwalt et al., 1991) but these effects were not consistently sustained in the medium to long-term (Bennett, 1995; Clayton et al., 1996; Dukes et al., 1996, 1997; Ennett et al., 1994). Studies of the LST programme showed primarily no programme effects on attitudes towards alcohol use in the short-term (Botvin et al., 1990a; 1997), but did show medium-term effects (Botvin et al., 1990b; 2001a), although these were not sustained long-term (Botvin et al., 2001a). No other studies demonstrated long-term effects on attitudes toward alcohol use (Caplan et al., 1992; Cuijpers et al., 2001).

Eight studies (Botvin et al., 1995b; 1997; 2001a; Caplan et al., 1992; Eisen et al., 2002; Lennox & Cecchini, 2008; Perry et al., 2003; Hecht et al., 2003) also examined programme impacts on behavioural intentions to drink (or use substances). With all but one study, of the Keepin it REAL programme (Hecht et al., 2003), showing a decrease in intentions to drink or get drunk. Findings relating to perceived peer norms regarding alcohol or substances were reported for evaluations of DARE (Harmon, 1993; Ringwalt et al., 1991; Clayton et al., 1996), LST (Botvin et al., 1997; 2001a), Lion's Quest Skills for Adolescence (Eisen et al., 2002) and Keepin it REAL (Hecht et al., 2003; Kulis

et al., 2005). There was no effect of Lion's Quest Skills for Adolescence or Keepin it REAL on peer norms for alcohol or substance use. DARE participants showed decreases in beliefs of positive peer views relating to drug use in the short-term, but these were not sustained long-term. Similarly studies evaluating LST showed only short-term effects on attitudes towards peer norms.

5.3.4.3 Personal and social skills

Five studies (Ennett et al., 1994; Harmon, 1993; Ringwalt et al., 1991; Botvin et al., 1997; Dedobbeleer & Desjardins, 2001), including evaluations of DARE, LST and the Coalition for Youth Quality of Life, reported on personal and social skills related to alcohol. There were no effects of DARE on social skills over the short-, medium- or longer term (Ennett et a., 1994), but there were short-term positive impacts of the programme on assertiveness and prosocial peer associations (Ringwalt et al., 1991; Harmon, 1993). There were also short-term impacts of LST on refusal skills (Botvin et al., 1997), but not on other skills relating to decision-making, anxiety reduction, communication and social assertiveness. One study (Dedobbeleer & Desjardins, 2001) examined programme effects of the Coalition for Youth Quality of Life on student's relationship quality with their parents, finding a long-term programme effect on relationship quality with fathers but not mothers.

5.3.4.4 Health and social outcomes relating to alcohol use and sexual health

Four programmes, the Positive Youth Development Programme, the Unplugged programme, a revised version of Project Alert, and the Healthy School and Drug Project had positive short-term programme effects on alcohol use. However, the findings of Positive Youth Development Programme (Caplan et al., 1992) were limited by the poor quality of the study. The study of the Healthy School and Drugs Project (Cuijpers et al., 2002) was good quality and significant effects on weekly drinking, and quantity per week and per occasion at the end of the three-year programme were reported. Statistically significant effects of the Unplugged curriculum (Faggiano et al., 2008) were found in relation to drunkenness, but the impact of the programme on other measures of alcohol use was not explored in the study. Effects on alcohol misuse were also found for a revised version of Project Alert (Ellickson et al., 2003) but the programme had no effects on initiation of, or current, alcohol consumption. Positive longer term effects were demonstrated for two programmes, Keepin It REAL (Hecht et al., 2003 and Be Under Your Own Influence/All Stars (Slater et al., 2006), which combined school and media intervention components. The two-year programme had significant, positive effects on alcohol use across all three versions of the programme, but effects were largest for the multicultural version of the curriculum. The strongest evidence of effectiveness came from a series of studies which examined Botvin's LST (Botvin et al., 1990a; 1990b; 1995b; 1997; 2001a; Fraguela et al., 2003). Two studies (Botvin et al., 1997; 2001a) found positive short- and medium-term effects on drinking frequency and binge-drinking, and these were sustained long-term (Botvin et al., 1995b; 2001a). However, replication of this study in a Spanish cohort of students (Fraguela et al., 2003) did not demonstrate that the programme was effective.

Eight programmes had inconsistent or no effects on alcohol use: DARE, Going Places, Here's Looking at You 2000, Lion's Quest SFA, All Stars Senior, Project Alert, Project SMART, Project TND,

NARCONON drug education curriculum. In addition, one programme, the Adolescent Decision Making programme (Snow et al., 1992; 1997) had potentially harmful long-term effects.

Evidence statement 3

- 3 (g) There is moderate evidence from two RCTs, one NRCT and one CBA study¹ to suggest that classroom-based substance use programmes are effective at improving knowledge relating to substance use and its effects in the short- to medium-term, but that these effects are not sustained in the long-term. Findings may only be partially applicable to the UK as studies were conducted outside the UK and may not be generalisable beyond the populations studied.
- 3 (h) There is moderate evidence from 12 RCTs, seven NRCTs and two CBA studies² to suggest that classroom-based substance use programmes may have mixed effects on students' substance use-related attitudes and values. There is moderate evidence from seven RCTs, four NRCTs and two CBA studies³ to suggest that these programmes may impact on attitudes to substance use in the short- to medium-term, and further evidence from five RCTs and three NRCTs⁴ to suggest that they may have a positive impact on long-term behavioural intentions. There is weak evidence from six RCTs, one NRCT and one CBA study⁵ to suggest that classroom-based substance use programmes have no medium- to long-term effects on peer norms. Findings may only be partially applicable to the UK as studies were conducted in the USA and may not be generalisable beyond the populations studied.
- 3 (i) There is weak and inconsistent evidence from one RCT, three NRCTs and one CBA study⁶ to determine the effects of classroom-based substance use programmes on young peoples' personal and social skills.
- 3 (j) There is moderate evidence from 14 RCTs, two NRCTs and two CBA studies⁷ to suggest that the following classroom-based substance use programmes may have inconsistent or no effects on alcohol use: DARE, Going Places, Lion's Quest SFA, All Stars Senior, Project Alert, Project SMART, Project TND, NARCONON drug education curriculum. In addition, there is weak evidence from one CBA study⁸ to suggest that the Adolescent Decision Making Programme may have potentially harmful long-term effects on alcohol consumption.
- 3 (k) There is weak evidence from two RCTs and one NRCT⁹ to suggest that programmes which combine school-based curriculums with additional components may have positive effects on alcohol consumption. The Healthy School and Drugs Project, a three-year programme, which included a nine lesson teacher-led curriculum, formulation of school policy on substance use and involvement of parents, had short-term effects on alcohol use, but longer term effects of the programme have not been examined. Positive longer term effects were demonstrated for two programmes, Keepin It REAL and Be Under Your Own Influence/All Stars, which both combined school-based curriculums with media intervention components. Findings may only be partially applicable to the UK as studies were conducted in the Netherlands and the USA and may not be generalisable beyond the populations studied.

- 3 (I) There is moderate evidence from three RCTs¹⁰ to suggest that Life Skills Training has positive short-, medium- and long-term effects on drinking frequency and binge drinking. However, there is moderate evidence from three RCTs and one NRCT¹¹ to suggest that there may be issues with the transferability of LST to other settings. Findings may only be partially applicable to the UK as studies were conducted in Spain and the USA and may not be generalisable beyond the populations studied.
- Botvin et al., 1990a (RCT +); Botvin et al., 2001a (RCT +); Shope et al., 1996b (CBA -); Cuijpers et al., 2001 (NRCT +)
- ² Bennett, 1995 (RCT -); Clayton et al., 1991 (RCT -); Dukes et al, 1996, 1997 (CBA); Ennett et al., 1994 (NRCT +); Lynam et al., 1999 (RCT -); Rosenbaum et al., 1994 (NRCT +); Harmon, 1993 (CBA +); Ringwalt et al., 1991 (RCT -); Perry et al., 2003 (RCT +); Botvin et al., 1990a (RCT +); Botvin et al., 1990b (RCT +); Botvin et al., 1990b (RCT +); Botvin et al., 1995b (RCT +); Botvin et al., 1997 (NRCT -); Botvin et al., 2001a (RCT +); Cuijpers et al., 2001 (NRCT +); Eisen et al., 2002 (RCT +); Caplan et al., 1992 (NRCT -); Lennox & Cecchini, 2008 (NRCT +); Dedobbeleer & Desjardins, 2001 (NRCT -); Hecht et al., 2003 (RCT -); Kulis et al., 2005 (RCT -)
- Bennett, 1995 (RCT -); Clayton et al., 1991 (RCT -); Clayton et al., 1991 (RCT -); Dukes et al, 1996, 1997 (CBA); Ennett et al., 1994 (NRCT +); Harmon, 1993 (CBA +); Ringwalt et al., 1991 (RCT -); Botvin et al., 1990a (RCT +); Botvin et al., 1990b (RCT +); Botvin et al., 1997 (NRCT -); Botvin et al., 2001a (RCT +); Cuijpers et al., 2001 (NRCT +); Caplan et al., 1992 (NRCT -)
- ⁴ Botvin et al., 1995b (RCT +); Botvin et al., 1997 (NRCT -); Botvin et al., 2001a (RCT +); Caplan et al., 1992 (NRCT -); Eisen et al., 2002 (RCT +); Lennox & Cecchini, 2008 (NRCT +); Perry et al., 2003 (RCT +); Hecht et al., 2003 (RCT -)
- ⁵ Harmon, 1993 (CBA +); Ringwalt et al., 1991 (RCT -); Clayton et al., 1996 (RCT -); Botvin et al., 1997 (NRCT -); Botvin et al., 2001a (RCT +); Eisen et al., 2002 (RCT +); Hecht et al., 2003 (RCT -); Kulis et al., 2005 (RCT -)

⁵ Ennett et al., 1994 (NRCT +); Harmon, 1993 (CBA +); Ringwalt et al., 1991 (RCT -); Botvin et al., 1997 (NRCT -); Dedobbeleer & Desjardins, 2001 (NRCT -)

- Bennett et al., 1995 (RCT -); Lynam et al., 1999 (RCT -); Dukes et al., 1996; 1997 (CBA -); Ennett et al., 1994 (NRCT +); Clayton et al., 1991, 1996 (RCT -); Ringwalt et al., 1991 (RCT -); Harmon, 1993 (CBA +); Eisen et al., 2002 (RCT +); Fearnow-Kenney et al., 2003 (RCT -); Bell et al., 1993 (RCT +); Ellickson et al., 2003 (RCT +); Graham et al., 1990 (RCT +); Simons-Morton et al., 2005 (RCT +); Brewer, 1991 (RCT +); Sussman et al., 1998 (RCT +); Sussman et al., 2003 (RCT +) Dent et al., 2001 RCT -); Lennox & Cecchini, 2008 (NRCT +)
- Snow et al., 1992; 1997 (CBA -)
- ⁹ Cuijpers et al., 2002 (NRCT +); Hecht et al., 2003; Kulis et al., 2005, 2007 (all RCT -); Slater et al., 2006 (RCT +)
- ¹⁰Botvin et al., 1990a (RCT+); Botvin et al., 1995b (RCT +); Botvin et al., 2001a (RCT +)
- ¹¹Botvin et al., 1990b (RCT +); Fraguela et al., 2003 (NRCT -); Smith et al., 2004 (RCT +); Spoth et al., 2005; 2008 (RCT +)

Author Study		Intervention	Comparator	Follow-up	Outcomes			
Author	design	Intervention	Comparator	Follow-up	Knowledge	Attitudes	Skills	
Bennett, 1995	RCT -	DARE n=NR	Drugs unit only n=NR	PT n=NR	-	↑ negative attitudes towards alcohol*	-	
Botvin et al., 1990a; Botvin et al., 1995a	RCT +	LST 1-day teacher workshops or teacher training by video n=NR	NR n=NR	PT n=3,684 ^a (83%)	↑ drinking knowledge***	 NS drinking attitudes ↑ interpersonal skills knowledge^{NR} NS communication skills knowledge NS personality variables 	-	
Botvin et al., 1997	NRCT -	LST n=NR	Usual curriculum n=NR	PT n=721 (87%)	-	 ↓ future intentions to drink beer or wine** NS future intentions to drink liquor ↓ normative expectations, peers** ↓ normative expectations, adults*** NS anti-drinking attitudes NS advertising 	↑ refusal skills* NS anxiety reduction NS communication NS social assertiveness NS decision-making	
Botvin et al., 2001a; Botvin et al., 2001b; Griffin et al., 2003	RCT +	LST n=2,144	Usual curriculum n=1,477	end of 7 th grade n=NR	↑ drinking knowledge*	 ✓ normative expectations, peers* ✓ normative expectations, adults* 	-	
Caplan et al., 1992	NRCT -	Positive Youth Development Programme n=109	No intervention n=173	PT NR	-	NS attitudes ↓ intentions to use beer* ↓ intentions to use hard liquor*	-	
Clayton et al., 1991	RCT - (cluster)	DARE n=1,438	Drugs unit only n=487	PT (4 months) n=NR	-	 ↑ negative attitudes towards alcohol** NS self-esteem NS peer pressure resistance 	-	
Cuijpers et al., 2001; Smit et al., 2003	NRCT +	Healthy School and Drugs Project n=1,156	No intervention n=774	1 year ^b n=1,405 (74%)	NS knowledge	NS attitudes ↑ self-efficacy*	-	
Cuijpers et al., 2001; Smit et al., 2003	NRCT +	Healthy School and Drugs Project n=1,156	No intervention n=774	2 years ^b n=1,405 (74%)	↑ knowledge***	↓ pro attitudes* NS self-efficacy	-	

Table 5.10. Classroom-based substance use prevention programmes: short-term programme effects on knowledge, skills and attitudes

Author	Study	Intonyontion	Comparator	Fellow up	Outcomes			
Author	design	Intervention		1 Ollow-up	Knowledge	Attitudes	Skills	
Cuijpers et al., 2001;	NRCT +	Healthy School and Drugs Project n=1,156	No intervention n=774	3 years⁵ n=1,405 (74%)	↑ knowledge***	NS attitudes NS self-efficacy	-	
Ennett et al., 1994	NRCT +	DARE n=18 schools	No intervention n=18 schools	PT n=NR	-	NS attitudes towards drugs ↑ self-esteem*	NS social skills	
Harmon, 1993	CBA +	DARE n=341	No intervention n=367	20 weeks n=602 (85%)	-	 ↑ belief in prosocial norms** ↓ association with drug using peers** ↑ negative attitudes towards substances** NS commitment and attachment to school NS rebellious behaviour NS self-esteem 	 ↑ assertiveness* ↑ positive or prosocial peer associations* NS coping strategies NS social integration 	
Ringwalt et al., 1991	RCT -	DARE n=685	Delayed intervention n=585	PT n=1270 (91%)	-	 ↑ perceived alcohol costs to be higher** ↑ negative attitudes towards drugs** ↓ belief in positive peer attitudes toward drug use*** NS self-esteem 	↑ assertiveness	
Rosenbaum et al., 1994	NRCT +	DARE n=18 schools	No intervention n=18 schools	PT n=NR	-	 ↑ perceived media influence on beer drinking* NS other attitudes/beliefs 	-	
Shope et al., 1996b; Shope et al., 1998	CBA -	Based on AMPS n=308	No intervention n=134	PT n=442 (23%)	 ↑ effects of substance use* NS pressures to use substances NS skills 	-	-	
*p≤0.05; **p≤0.01; ** ^a High fidelity sample	*p≤0.001; ↑ in (students exp	crease relative to com osed to >60% or more	nparator;	relative to con 62% of the orig	nparator; NS not significa inal sample); ^b Follow-up	nt; - outcome not reported from baseline		

Author	Study	Intervention	Comparator	Follow-up	-110	Dutcomes	
Aution	design	intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other
Bennett, 1995	RCT -	DARE n=NR	Drugs unit only n=NR	PT	NS past year alcohol use	-	-
Clayton et al., 1991	RCT -	DARE n=1,438	Drugs unit only n=487	PT (4 months) n=NR	NS alcohol use	-	-
Ennett et al., 1994	NRCT +	DARE n=18 schools	No intervention n=18 schools	PT n=NR	NS alcohol use (rural students only)	-	-
Rosenbaum et al., 1994	NRCT +	DARE n=18 schools	No intervention n=18 schools	PT n=NR	NS initiation of alcohol use NS increased use of alcohol NS quitting behaviour	-	-
Harmon, 1993	CBA +	DARE n=341	No intervention n=367	20 weeks n=602 (85%)		-	-
Ringwalt et al., 1991	RCT -	DARE n=685	Delayed intervention n=585	PT n=1270 (91%)	NS current alcohol use NS lifetime alcohol use	-	-
Botvin et al., 1990a; Botvin et al., 1995a	RCT +	LST 1-day teacher workshops or teacher training by video n=NR	NR n=NR	PT n=3,684 ^a (83%)	NS drinking frequency NS drinking amount		-
Botvin et al., 1997	NRCT -	LST n=NR	Usual curriculum n=NR	PT n=721 (87%)	 ↓ drinking index** ↓ drinking amount*** 	↓ drunkenness*	-
Botvin et al., 2001a; Botvin et al., 2001b; Griffin et al., 2003	RCT +	LST n=2,144	Usual curriculum n=1,477	end of 7 th grade NR	 		-
Fraguela et al., 2003	NRCT -	LST; researcher- (n=235) or teacher-led (n=309)	No intervention n=485	PT 80-90%	NS alcohol use	-	-
Smith et al., 2004;Vicary et al., 2004	RCT +	LST (n=234); I- LST (n=297)	No intervention n=201	end of 7 th grade n=704 (96%)	Ψ frequency (LST, females only)*	↓ binge drinking* (females only) NS drunkenness	-

Table 5.11. Classroom-based substance use prevention programmes: short-term programme effects on health and social outcomes

Author	Study	Intervention	Comparator	Follow-up	Outcomes			
Author	design	intervention			Alcohol use	Heavy alcohol use	Other	
Ellickson & Bell, 1990			Usual curriculum n=10 schools	3 months ^b n=3,852 (60%)	Non-users: ↓ initiation (peer leader only*) ↓ current (peer leader only*) <i>Experimenters</i> + users: ↓ current (users, adult health educators only**) NS monthly ^d NS weekly ^e	-	-	
	RCT +	Adult health educator, n=10 schools Adult health educator + peer leaders, n=10 schools		12 months ^b n=3,852 (60%)	Non-users: NS initiation NS current Experimenters + users: ↓ past month (experimenters, peer leader only*) NS monthly ^d NS weekly ^e			
				15 months ^b n=3,852 (60%)	Non-users: NS initiation NS current Experimenters + users: NS current NS monthly ^d NS weekly ^e			
Ellickson et al., 2003	RCT +	Project ALERT (revised) n=2,553	Usual curriculum n=1,723	18 months ^b n=4,276 (79%)	-	✓ overall alcohol misuse score* NS high risk drinking		
Fearnow-Kenney et al., 2003	RCT -	All Stars Senior n=406	NR n=247	PT (~80%)	NS alcohol use	-	-	
Shope et al., 1996b; Shope et al., 1998	CBA -	Based on AMPS n=308	No intervention n=134	PT n=442 (23%)	↓ alcohol use***	-	-	
Cuijpers et al., 2001; Smit et al., 2003	NRCT +	Healthy School and Drugs Project n=1,156	No intervention n=774	1 year ^b n=1,405 (74%)	 ✓ proportion who drink* NS proportion weekly drinkers NS drinks/week ✓ drinks/occasion*** ✓ lifetime alcohol prevalence* 	-	-	

Author	Study	Intervention	Compositor	Follow	Outcomes			
Author	design	intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other	
Cuijpers et al., 2001; Smit et al., 2003	NRCT +	Healthy School and Drugs Project n=1,156	No intervention n=774	2 years ^b n=1,405 (74%)	 ✓ proportion who drink** NS proportion weekly drinkers NS drinks/week NS drinks/occasion ✓ lifetime alcohol prevalence** 	-	-	
Cuijpers et al., 2001; Smit et al., 2003	NRCT +	Healthy School and Drugs Project n=1,156	No intervention n=774	3 years ^b n=1,405 (74%)	 ✓ proportion who drink** ✓ proportion weekly drinkers* ✓ drinks/week*** ✓ drinks/occasion*** ✓ lifetime alcohol prevalence* 	-	-	
Eisen et al., 2002; 2003	RCT +	Lions Quest Skills for Adolescence n=NR	Usual curriculum n=NR	PT n=6,239 (84%)	NS alcohol use prevalence NS lifetime alcohol use NS 30-day alcohol use	NS 30-day binge drinking	-	
Caplan et al., 1992	NRCT -	Positive Youth Development Programme n=109	No intervention n=173	PT NR	$oldsymbol{\Psi}$ alcohol per occasion*		-	
Faggiano et al., 2008	RCT +	Unplugged Basic, n=1,190 Parent, n=1,164 Peer, n=1,193	No intervention n=3,532	3 months n=6,370 (90%)	-	 	-	
[*] p≤0.05; **p≤0.01; ** [*] ^a High fidelity sample taken into account; ^d	*p≤0.001; 个 in (students exp Eleven or more	crease relative to con osed to >60% or more e times in the past yea	nparator; Ψ decreas e of the programme; ar or three or more c	e relative to con 62% of the orig lays in the past	nparator; NS not significant; - outcom inal sample); ^b Follow-up from baselir month; ^e 6+ days in past month	e not reported e; [°] NS, when intraclass corr	elations were	

Author	Study	Intervention	Comparator	Follow up	Outcon		
Author	design	Intervention	Comparator	Follow-up	Knowledge	Attitudes	Skills
Bennett, 1995	RCT -	DARE n=NR	Drugs unit only n=NR	7 th grade	-	↑ negative attitudes towards alcohol (average achieving students only)*	-
Ennett et al., 1994	NRCT +	DARE n=18 schools	No intervention n=18 schools	1 year	-	NS attitudes towards drugs NS self-esteem	NS social skills
Botvin et al., 1990b	RCT +	LST Teacher; peer; teacher + booster; peer + booster n=NR	NR n=NR	1 year n=998 (76%)	↑ drinking knowledge (peer + booster***/non- booster**)	 ↑ drinking attitudes** ↓ locus of control (peer + booster only**) <i>Teacher</i> + booster vs. control: ↑ pro drinking attitudes* 	-
Botvin et al., 2001a; Botvin et al., 2001b; Griffin et al., 2003	RCT +	LST n=2,144	Usual curriculum n=1,477	1 year NR	↑ drinking knowledge*	 ↓ intentions to drink alcohol* ↑ negative attitudes towards alcohol** ↓ normative expectations, peers** ↓ normative expectations, adults* 	-
Shope et al., 1996b; Shope et al., 1998	CBA -	Based on AMPS n=507	No intervention n=530	1 year N=442 (23%)	 ↑ effects of substance use* NS pressures to use substances NS skills 	-	-
Eisen et al., 2002	RCT +	Lions Quest Skills for Adolescence n=NR	Usual curriculum n=NR	1 year n=5,691 (77%)	-	NS self-efficacy NS intentions to drink alcohol NS perceived harm NS refusal efficacy NS perceived peer use	-

Table 5.12. Classroom-based substance use prevention programmes: medium-term programme effects on knowledge, attitudes and skills

Author	Study	Intervention	Comparator	Follow-up	Outcomes			
Aution	design	Intervention			Knowledge	Attitudes	Skills	
Lennox & Cecchini, 2008	NRCT +	NARCONON™ drug education curriculum n=464	Delayed control n=531	6 months n=726 (73%)	↑ programme content***	 ↑ people risk harming themselves by having one or two drinks nearly every day* NS people risk harming themselves by having five or more drinks ↓ intentions to get drunk** ↑ wrong to drink regularly*** 	-	
*p≤0.05; **p≤0.01; ***	*p≤0.001; ↑ in	crease relative to corr	nparator; 🗸 decrease	relative to con	nparator; NS not significant;	- outcome not reported		

Table 5.13. Classroom-based substance use prevention programmes: medium-term programme effects on health and social outcomes

Author	Study	udy Intervention	Comparator	Follow-up	Outcomes			
Aution	design	intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other	
Bennett, 1995	RCT -	DARE n=NR	Drugs unit only n=NR	7 th grade n=NR	past year alcohol use (low achieving students only*)	-	-	
Ennett et al., 1994	NRCT +	DARE n=18 schools	No intervention n=18 schools	1 year n=NR	NS alcohol use	-	-	
Botvin et al., 1990b	RCT +	LST Teacher; peer; teacher + booster; peer + booster n=NR	NR n=NR	1 year n=998 (76%)	 ✓ weekly alcohol use*** ✓ monthly alcohol use*** ✓ drinking frequency** ✓ alcohol per occasion (peer + booster only*) <i>Teacher</i> + booster vs. control: ↑ weekly alcohol use*** ↑ monthly alcohol use*** ↑ drinking frequency** 	-	-	
Botvin et al., 2001a; Botvin et al., 2001b; Griffin et al., 2003	RCT +	LST n=2,144	Usual curriculum n=1,477	1 year NR	 ✓ drinking frequency** ✓ drinking quantity** 	 ✓ drunkenness frequency* ✓ binge drinking* 	-	

Author	Study	Intervention	Comparator	Follow-up	Outcomes			
Author	design	intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other	
Fraguela et al., 2003	NRCT -	LST; researcher- (n=235) or teacher- led (n=309)	No intervention n=485	1 year NR	 ↓ consumption of beer (research-led only**) ↓ consumption of spirits (teacher-led only*) 	-	-	
Smith et al., 2004;Vicary et al., 2004	RCT +	LST (n=234); I-LST (n=297)	No intervention n=201	end of 8 th grade n=659 (90%)	NS frequency	NS drunkenness NS binge drinking	-	
Spoth et al., 2002; 2005; 2008	RCT +	LST, n=621; or LST + SFP 10-14, n=549	Four leaflets mailed to families n=494	1 year N=1,361 (82%)	 	-	-	
Sussman et al., 1998	RCT +	Project TND Classroom, n=7 schools Classroom + school as community, n=7 schools	Usual curriculum n=7 schools	1 year n=1,074 (67%)		-	-	
Dent et al., 2001	RCT -	Project TND n=13 classes	No intervention n=13 classes	1 year n=679 (63%)		-	-	
Bell et al., 1993	RCT +	Project ALERT Adult health educator, n=10 schools Adult health educator + peer leaders, n=10 schools	Usual curriculum n=10 schools	12 months n=4,837 (74%)	NS initiation NS past month NS past year NS monthly NS weekly NS daily	-	-	
Shope et al., 1996b; Shope et al., 1998	CBA -	Based on AMPS n=507	No intervention n=530	1 year N=442 (23%)	NS alcohol use	NS alcohol misuse		
Simons-Morton et al., 2005	RCT +	Going Places n=692	No intervention n=628	up to 1 year n=1,320 (50%)	NS drinking stage	-	-	
Brewer, 1991	RCT +	Here's Looking at You 2000 n=18	Videotape or no intervention n=36	6 months	NS alcohol use	-	-	

Author	Study	Intervention	Comparator	Follow-up	Outcomes			
Aution	design	intervention			Alcohol use	Heavy alcohol use	Other	
Eisen et al., 2002; 2003	RCT +	Lions Quest Skills for Adolescence n=NR	Usual curriculum n=NR	1 year n=5,691 (77%)	NS lifetime alcohol use NS 30-day alcohol use	NS 30-day binge drinking	-	
Lennox & Cecchini, 2008	NRCT +	NARCONON™ drug education curriculum n=464	Delayed control n=531	6 months n=726 (73%)	NS alcohol use	NS drunkenness	-	
Graham et al., 1990	RCT +	Project SAVE Social skills programme, n=6 schools; Affect management programme, n=6 schools	Usual curriculum n=12 schools	1 year (70%)	↓ alcohol use*	-	-	
*p≤0.05; **p≤0.01; ***	′p≤0.001; ↑ in	crease relative to comp	arator; 🗸 decrease	relative to con	nparator; NS not significant; -	 outcome not reported 		

Table 5.14. Classroom-based substance use prevention programmes: long-term programme effects on knowledge, attitudes and skills

Author	Study	Intervention	Comparator	Follow up	Outcomes		
Aution	design	intervention		Follow-up	Knowledge	Attitudes	Skills
Bennett, 1995	DOT	DARE	Drug unit only	8 th grade	-	NS negative attitudes towards alcohol	-
	RCI -	n=NR	n=NR	9 th grade	-	NS negative attitudes towards alcohol	-
Clayton et al., 1996	RCT - (cluster)	DARE n=23 schools	Other drug education programmes n=8 schools	up to 5 years	-	 ↑ general and specific drug attitudes* ↑ capability to resist peer pressure* ↑ peer norms for drug use* 	-
Dukes et al., 1996; 1997	CBA -	DARE n=497	Delayed intervention n=352	Up to 6 years n=NR	-	NS pro drug use attitudes NS resistance to peer pressure	-
Ennett et al., 1994	NRCT +	DARE n=18 schools	No intervention n=18 schools	2 years n=NR		NS attitudes towards drugs NS self-esteem	NS social skills
Author	Study	Intervention	Comparator	Follow-up	Outcomes		
---------------------------------------------------------------------------	---------------------------------------	----------------------------------------------------------------	------------------------------------	-----------------------------	------------------------------	---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	---------------------------------------------------------------
Author	design	intervention	Comparator	Follow-up	Knowledge	Attitudes	Skills
Lynam et al., 1999	RCT -	DARE n=762	NR n=240	10 years n=NR	-	NS peer pressure resistance NS self-esteem	-
Perry et al., 2003	RCT +	DARE, n=2,226; DARE Plus, n=2,221	Delayed intervention n=1,790	Up to 18 months (84%)	-	↓ change in alcohol behaviour and intentions (DARE Plus, boys only*)	-
Botvin et al., 1995b	RCT +	LST, n=NR Culturally focused intervention (CFI), n=NR	Information only n=NR	2 years n=456 (60%)		 ✓ intentions to use beer or wine** ✓ intentions to use hard liquor (LST only*) 	-
Botvin et al., 2001a; Botvin et al., 2001b; Griffin et al., 2003	RCT +	LST n=2,144	Usual curriculum n=1,477	2 years NR	NS drinking knowledge	NS pro-drinking attitudes NS peer drinking norms	-
Hecht et al., 2003; Gosin et al., 2003	RCT -	Keepin it REAL n=25 schools	Usual curriculum n=10 schools	up to 14 months (84%)	-	 ✓ positive expectancies* ↑ descriptive norms* NS alcohol resistance strategies NS self-efficacy NS intentions NS norms 	-
Kulis et al., 2005	RCT -	Keepin it REAL n=2,397	Usual curriculum n=1,005	14 months NR	-	NS refusal confidence NS intent to accept NS positive expectancies NS norms	-
Shope et al., 1996b; Shope et al., 1998	CBA -	Based on AMPS n=308	No intervention n=134	up to 6 years	NS knowledge	-	-
		Coolition for Youth		18 months NR	-	NS intentions to become involved in prevention activities ↑ self-esteem*	↑ relationship with father* NS relationship with mother
Dedobbeleer & Desjardins, 2001	NRCT - Quality of Life n=4 schools	Quality of Life n=4 schools	No intervention n=6 schools	30 months n=320 (40%)	-	NS drug and alcohol problems NS intentions to become involved in prevention activities NS self-esteem	NS relationship with father/mother

Author	Study	Intervention	Comparator	Follow-up		Outcomes	
Aution	design	Intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other
Bonnott 1005	PCT	DARE	NR	8 th grade	NS past year alcohol use	-	-
Denneu, 1995	KUT-	n=NR	n=NR	9 th grade	NS past year alcohol use	-	-
Clayton et al., 1996	RCT - (cluster)	DARE n=23 schools	Other drug education programmes n=8 schools	up to 5 years	NS alcohol use	-	-
Dukes et al., 1996; 1997	CBA -	DARE n=399	Delayed intervention n=271	Up to 6 years	NS alcohol use	-	-
Ennett et al., 1994	NRCT +	DARE n=6 schools	No intervention n=6 schools	2 years	NS alcohol use		-
Rosenbaum & Hanson, 1998	NRCT +	DARE n=975	No intervention n=823	up to 6 years	NS lifetime alcohol use NS last month alcohol use	-	-
Lynam et al., 1999	RCT -	DARE n=762	NR n=240	10 years	NS alcohol use	-	-
Perry et al., 2003	RCT +	DARE, n=2,226; DARE Plus, n=2,221	Delayed intervention n=1,790	Up to 18 months (84%)	 NS alcohol use ↓ change in past year alcohol use (DARE Plus, boys only*) ↓ change in past month alcohol use (DARE Plus, boys only*) 		-
Botvin et al., 1990a; Botvin et al., 1995a	RCT +	LST 1-day teacher workshops or teacher training by video n=NR	NR n=NR	3 years n=3,597 (81%)	NS weekly alcohol use NS monthly alcohol use	NS 3+ drinks per occasion ↓ drunkenness (workshops*/training by video**)	-
Botvin et al., 1995b	RCT +	LST, n=NR Culturally focused intervention (CFI), n=NR	Information only n=NR	2 years n=456 (60%)	 ↓ drinking frequency*** ↓ drinking amount*** 	↓ drunkenness***	-

Table 5.15. Classroom-based substance use prevention programmes: long-term programme effects on health and social outcomes

Author	Study	Intervention	Comparator	Follow-up	In	Outcomes	
Aution	design	Intervention	comparator	1 Ollow-up	Alcohol use	Heavy alcohol use	Other
Botvin et al., 1995b	RCT +	Culturally focused intervention (CFI) n=NR	LST n=NR	2 years (60%)	 ↓ drinking frequency** ↓ drinking amount* 		-
Botvin et al., 2001a; Botvin et al., 2001b; Griffin et al., 2003	RCT +	LST n=2,144	Usual curriculum n=1,477	2 years NR	-	$oldsymbol{\Psi}$ binge drinking**	-
Fraguela et al., 2003	NRCT -	LST; researcher-led n=235 teacher-led n=309	No intervention n=485	2-3 years 40-36%	NS alcohol use	-	-
Spoth et al., 2002; 2005; 2008	L	LST, n=621; LST + SFP 10-14, n=549	Four leaflets mailed to families n=494	2 years n=1,198 (73%)	NS regular alcohol use		-
	K01+			5 years n=1,237 (74%)	NS 'new users'	NS drunkenness	-
Hecht et al., 2003; Gosin et al., 2003	RCT -	Keepin it REAL n=25 schools	Usual curriculum n=10 schools	up to 14 months (84%)		-	-
Kulis et al., 2005	RCT -	Keepin it REAL n=2,397	Usual curriculum n=1,005	14 months NR		-	-
Kulis et al., 2007	RCT -	Keepin it REAL n=1050	NR n=314	up to 14 months	 ↑ reduced or recently discontinued alcohol use*** ↑ transition to reduced alcohol use* 	-	-
Warren et al., 2006	RCT +	Keepin it REAL n=3314	Usual curriculum n=1420	14 months NR	NS alcohol use	-	-
Sussman et al., 1998; Sun et al., 2006	RCT +	Project TND Classroom, n=7 schools Classroom + school as community, n=7 schools	Usual curriculum n=7 schools	up to 5 years	NS alcohol use	-	-
Sussman et al., 2003	RCT +	Project TND n=NR	Usual curriculum n=NR	2 years n=NR	NS alcohol use	-	-

Author	Study	Intervention	Comparator	Follow-up		Outcomes	
Aution	design	Intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other
Ellickson et al., 1993	RCT +	Project ALERT n=3,247	Usual curriculum n=2,165	up to 12 th grade	NS alcohol use	-	-
Snow et al., 1992; 1997	CBA -	Adolescent Decision Making programme n=680	NR n=680	2 years (79%)	↑ alcohol use*	-	-
Shope et al., 1996b; Shope et al., 1998	CBA -	Based on AMPS n=507	No intervention n=530	up to 6 years (25%)	NS alcohol use	NS alcohol misuse	-
Slater et al., 2006	RCT +	Be Under Your Own Influence/All Stars In-school media + All Stars; In- school media only; All Stars only n=NR	No intervention n=NR	2 years (69%)		-	-
Dedobbeleer & Desjardins, 2001	NRCT -	Coalition for Youth - Quality of Life n=4 schools	No intervention n=6 schools	18 months n=NR	NS alcohol drinking frequency NS alcohol consumed per typical occasion	NS drug and alcohol problems	-
				30 months n=320 (40%)	NS alcohol drinking frequency NS alcohol consumed per typical occasion	-	-
*p≤0.05; **p≤0.01; ***	*p≤0.001; ↑ ir	crease relative to con	nparator; 🗸 decrease	relative to con	nparator; NS not significant; -	outcome not reported	
*p≤0.05; **p≤0.01; *** ^a Based on school as	NRCT - ʿp≤0.001; 个 ir level of analy:	Quality of Life n=4 schools ncrease relative to con sis	n=6 schools	30 months n=320 (40%) relative to con	NS alcohol drinking frequency NS alcohol consumed per typical occasion aparator; NS not significant; -	- outcome not reported	-

5.4 Brief behavioural or single session interventions

5.4.1 Overview of evidence identified

A total of 15 articles (Argentos, 1991; Dempster et al., 2006; Werch et al., 1996a, 1996b, 1998, 2000a, 2000b, 2001, 2003a, 2003b, 2005a, 2005b, 2005c, 2008a, 2008b) were identified that reported on evaluations of seven brief behavioural or single session intervention approaches across 13 studies. For the purposes of this review, brief behavioural and single session intervention approaches were defined as short (i.e. lasting no longer than one hour) or single session interventions, which were delivered outside of the lesson format or wider curriculum. Eight articles examined the STARS (Start Taking Alcohol Risks Seriously) for Families programme developed by Werch et al. Three articles (Werch et al., 2000a; 2001; 2003a) reported on a two-year version of the STARS programme and were grouped together as a single study.

Twelve of the studies were conducted in the USA and one study was conducted in the UK (Dempster et al., 2006). All thirteen studies were primarily school-based but six studies (Werch et al., 1998, 2000a, 2000b, 2003b, 2005a, 2008b) examined interventions which incorporated materials targeting parents. A range of providers were utilised across the included studies. Eight studies (Werch et al., 1996a, 1996b, 1998, 2000a, 2000b, 2003b, 2003b, 2005a, 2005b) examined interventions delivered by school nurses, of which three studies (Werch et al., 1996a, 2003b, 2003b, 2003b, 2005b) combined delivery with a physician, teacher or fitness professional. Two studies (Werch et al., 2008a, 2008b) were based on mailed intervention materials and therefore did not involve a provider in the delivery. The remaining three studies (Dempster et al., 2006; Werch et al., 2005c; Argentos, 1991) were delivered by a consultant, trained research staff and a motivational speaker, respectively.

The theoretical framework for intervention was not reported for one study (Werch et al., 2005c). For studies that did report the theoretical framework, all but one study (Argentos, 1991) was based on multiple theories.

The sample sizes in the included studies ranged from 104 to 704 students. The programmes examined targeted students across a range of ages. Five studies (Werch et al., 1996a, 1996b, 1998, 2000a, 2005a) of the STAR for Families programmes and one study (Werch et al., 2003b) of the Project SPORT intervention targeted students aged 14 years or younger. Werch et al (2000b) examined a version of the STAR for Families programme which targeted students in junior high, aged 12-15 years. The six remaining studies (Argentos, 1991; Dempster et al., 2006; Werch et al., 2005b, 2005c, 2008a, 2008b) examined programmes that targeted studies aged over 14 years, up to the age of 18. The length of follow-up across the included studies ranged from PT to 12 months. No studies reported on the long-term follow-up of brief behavioural or single session programmes.

Table 5.16. Alcohol education: brief behavioural or single session interventions

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Dempster et al., 2006	NRCT -	UK n=182 15-16 years	School	Single session (20 minutes) to communicate to young people the dangers of binge drinking	Theory of planned behaviour	Consultant
Werch et al., 1996a	RCT +	USA n=104 mean 13.8 years	School	STARS for Families : Self-instructional module and audiotape, brief health consultation with nurse or doctor, follow-up consultation with trained peers between PT and follow-up	Transtheoretical model of change, health belief model, social learning theory, behavioural self-control	Physician, School nurse
Werch et al., 1996b	RCT ++	USA N= 138 Mean age 12.2 years	School	STARS for Families : 6 focused weekly follow-up consultations; brief health consultation with nurse	Multi-component motivational stages prevention model, health belief model, social learning theory, behavioural self- control	School nurse
Werch et al., 1998	RCT +	USA n=211 mean 12.1 years	School, family	STARS for Families : Brief health consultation with nurse; letter to parents; up to 9 family-based prevention lessons	Multi-component motivational stages prevention model, health belief model, social learning theory, behavioural self- control	School nurse
Werch et al., 2000a; 2001; 2003a	RCT +	USA N=650 mean 12.1 years	School, family	STARS for Families : Brief health consultation with nurse; up to 10 prevention postcards sent to parents; follow-up consultation in second year; four family take-home lessons	Transtheoretical model of change, multi-component motivational stages prevention model,	School nurse
Werch et al., 2000b	RCT +	USA n=178 7th-9th grade	School, family	STARS for Families : Telephone-based nurse consultation; 10 prevention postcards sent to parents	Multi-component motivational stages prevention model, social cognitive theory	School nurse
Werch et al., 2003b	RCT ++	USA n=454 mean 13.2 years	School, family	Project SPORT Sport Plus: Health fitness screen and health consultation with a nurse, plus alcohol preventive consultation with a nurse Sport Plus Parent: As above, plus five parental cards mailed once a week	Social cognitive theory, multi-component motivational stages prevention model	School nurse, Teacher
Werch et al., 2005a	RCT +	USA N= 448 Mean 13.4 years	School, family	STARS for Families: Single vs. multiple drug prevention, brief health consultation with nurse, eight prevention postcards sent to parents	Health belief model, social cognitive theory, behavioural self control	School nurse

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Werch et al., 2005b	RCT ++	USA n=604 mean 15.2 years	School	Project SPORT: Health behaviour screen; brief health consultation; take-home fitness prescription and prevention flyer	Integrative Behaviour- Image Model	Nurse, fitness professionals
Werch et al., 2005c	RCT +	USA N= 232 mean 17 years	School	Alcohol beverage tailored programme: Brief screening instrument; brief alcohol risk reduction consultation; tip sheet reinforcing key messages	NR	Trained research staff
Argentos, 1991	CBA -	USA n=350 9th-10th grade	School	Programme Kickoff: 36 hours over one week; motivational speaker, prevention curriculum, group discussion and role play, t-shirts promoting drug free lifestyles	Social learning theory	Motivational speaker
Werch et al., 2008a	RCT +	USA n=375 mean 17 years	School	Plan for Success: printed text and scripted messages; health promotion and avoidance of health risks. Three interventions: (1) Plan for Success goal clarification survey; (2) goal clarification + goal planning; (3) goal clarification + career consultation.	Behaviour-image model, prospect theory and message framing.	NA
Werch et al., 2008b	RCT +	USA n=704 mean 15.2 years	School, family	Brief image-based messages : Three parent/ caregiver postcards, read and talk about each of four health and fitness facts found on the card with their teen	Prospect theory	NA

5.4.2 Quality assessment

Eleven studies (Werch et al., 1996a, 1996b, 1998, 2000a, 2000b, 2003b, 2005a, 2005b, 2005c, 2008a, 2008b) were based on RCT designs, and involved the randomisation of individual students to intervention and control groups. However, two studies by Werch et al (2003b, 2008a) involved randomisation to three intervention arms and did not utilise a control group for comparison. The study by Dempster et al (2006) was based on an NRCT design and Argentos et al (1991) reported on a CBA study. All 11 RCTs were rated good or moderate quality. Although detail on the methods of allocation were lacking in some studies, outcomes were reported to be valid and reliable across all of these studies and attrition rates were generally low. The study by Dempster et al (2006) was rated as poor quality because information was lacking on allocation and the baseline comparability between intervention and control students. In addition, although outcomes were generally well reported the scope of the analyses was lacking. The CBA study by Argentos (1991) was also rated poorly. Details of the study methodology were poorly reported, and it was not clear if the intervention and control groups were balanced at baseline as little information on the demographics of participants were reported.

5.4.3 Findings

5.4.3.1 Short-term results (<6 months)

Twelve studies (Argentos, 1991; Dempster et al., 2006; Werch et al., 1996a, 1996b, 1998, 2000a, 2003b, 2005a, 2005b, 2005c, 2008a, 2008b) reported on the short-term effects of brief behavioural and single session interventions.

Knowledge and understanding

None of studies examined short-term intervention effects on knowledge and understanding.

Attitudes and values

Ten studies (Argentos, 1991; Dempster et al., 2006; Werch et al., 1996a; 1996b; 2000a; 2003b; 2005a; 2005b; 2005c; 2008a) examined intervention effects on a range of measures related to attitudes and values in relation to alcohol use. Compared to students in the control group, students who participated in the one week Programme "Kickoff" (Argentos, 1991) reported an increased belief that their alcohol (and other drug) use might result in serious consequences (p<0.01). They also demonstrated significantly higher levels of understanding about popular myths regarding alcohol use and other drugs. Dempster et al (2006) reported that one month after receiving a single session on the dangers of binge drinking³, intervention students appraised binge drinking behaviour compared to the control group (p<0.001 and p<0.01, respectively). Intervention students also tended to show a stronger intention to stop binge drinking than the control group, but this finding did not reach significance. Four studies (Werch et al., 1996a; 1996b; 2000a; 2005a) examined the short-term

³ Defined by the authors as drinking half the recommended maximum weekly consumption of alcohol in a single session.

effects of the STARS for Families programme. Werch et al (1996a) reported that 10 weeks after participating the programme, intervention students reported significantly fewer peer expectations to drink alcohol, less intention to use alcohol in the future, and less intention to try alcohol compared to students who received an alcohol education booklet (all p<0.05). Werch et al (1996b) found that compared to a no intervention control, STARS for Families had no effects on students' perceptions of drinking consequences or their intentions for future alcohol use. Werch et al (2005a) examined STARS for Families (alcohol only) versus a multiple substance intervention (STARS Plus), compared a post-card only control. There were no short-term effects of either intervention on perceived peer alcohol use, but students who received the alcohol only programme reported less susceptibility to peer alcohol use and greater levels of alcohol incompatibility than controls (both p<0.05). Evaluation of two-year version of the STARS for Families programmes (Werch et al., 2000a; 2001) revealed inconsistent effects of the intervention of risk factors for alcohol use. Following the first year of the programme there were no differences in alcohol risk factors between intervention and control students. However, at the end of the programme, students in commuter schools reported fewer expectancy beliefs, greater motivations to avoid alcohol use and an overall lower number of alcohol risk factors (all p<0.05). There was no difference on any of these measures among students in (inner-city) community schools. Two studies (Werch et al., 2003b; 2005b) examined effects of the Project SPORT intervention. Werch et al (2003b) reported that students who received the Sport Plus Parent programme had greater increases in negative expectancy beliefs and self-control over time than students in the Sport or Sport Plus groups. At 3 months, compared to a no intervention control (Werch et al., 2005b), Project SPORT youth reported significantly greater negative alcohol expectancy beliefs (indicating greater protection against alcohol use) (p<0.05), behavioural capability (p<0.01), perceived susceptibility (p<0.05), parental monitoring (p<0.05), and parent/child communication (p<0.05). There were no intervention effects on resistance self-efficacy, self-control, value incompatibility or positive parent/child relationship. SPORT participants showed less risk for alcohol use compared to control participants, on measures of intentions to drink in the future (p<0.01), alcohol attitudes (p<0.05), and 'influenceability' (p<0.01). No effects on positive expectancy beliefs, subjective norms or perceived peer prevalence of alcohol were found. An alcohol beverage tailored intervention (Werch et al., 2005c) had significant, positive effects (all p<0.05) on the following alcohol risk factors: 'influenceability' for beer, wine, distilled spirits, and malt liquor consumption; perceived peer prevalence for wine, flavoured coolers, and fortified wine consumption; perceived susceptibility for beer and wine consumption; and perceived severity for beer, wine and distilled spirit consumption. A brief behavioural intervention strategy addressing positive images, delivered alone or in combination with either goal setting or career consultation, was associated with reductions in intentions to use alcohol and increases in the belief that alcohol interferes with other behaviours (Werch et al., 2008a).

Personal and social skills

Two studies (Werch et al., 2003b; 2005b) examined the short-term effects of the Project SPORT intervention on personal and social skills. Students who received the SPORT intervention plus parental cards reported improvements in self-control following intervention, compared to no change

across the other versions of the programme examined. Compared to no intervention, there were no effects of Project SPORT on self-control or parent-child relationships.

Health and social outcomes relating to alcohol use and sexual health

Twelve studies examined short-term intervention effects on alcohol use. Six studies (Argentos, 1991; Dempster et al., 2006; Werch et al., 1996b; Werch et al., 1998; Werch et al., 2003b; Werch et al., 2005c) found that the brief behavioural or single session intervention examined were not effective in reducing alcohol use.

Two studies (Werch et al., 1996a; 2001) demonstrated positive short-term effects of the STARS for Families intervention. Werch et al (1996a) reported that the STARS intervention had significant effects on 30-day quantity and frequency of alcohol use at 10 weeks follow-up (after delivery of the peer follow-up consultation), and STARS students reported using less alcohol than control students. However, there was no difference between intervention and control students who received an alcohol education booklet on the measure of recent alcohol use. Comparing community and specialist schools at the end of the first year of a two year intervention, Werch et al (2001) found intervention associated reductions in last week and last month drinking in STARS students in community schools (both p<0.05) but not in their specialist school counterparts. At the end of the two-year programme (Werch et al., 2000a) however, there was no difference between intervention and control students in specialist schools on any of the measures of alcohol use. Conversely, intervention students in specialist schools were more likely than control students to report that they did not drink and were less likely to reported heavy alcohol use (both p<0.05). There was no difference between intervention and control students in commuter schools on the measures of 7- or 30-day alcohol use.

Werch et al (2003b) found that there were reductions in alcohol use across all participants who received a sport consultation. Significant reductions were reported on 3 of 6 alcohol use measures: 30-day heavy drinking, alcohol problems and alcohol use initiation. However, there was no difference between groups (Sport, Sport Plus, or Sport Plus Parent). At 3-months follow up, students who received the Project SPORT intervention (Werch et al., 2005b) reported significantly less alcohol frequency, quantity and heavy use in the last 30 days compared to no intervention control students (all p<0.001). More students were in an earlier stage of alcohol initiation⁴ (p<0.001) and length (p<0.01) of alcohol initiation was also less. Intervention based on brief behavioural intervention messages to parents which prompted them to discuss health and fitness with their children was associated with reduced alcohol use frequency compared to students who received fitness flyers (p<0.05).

Werch et al (2008a) examined the Plan for Success intervention, with and without a goal setting or career consultation component. There was no control arm included in the study. Across all three intervention arms there was a reduction in the length of time students had been drinking alcohol but no impact on quantity or frequency. For students whose parents received brief image-based

⁴ This measure reflected student's stage of alcohol initiation ranging from a strong precontemplation stage (will never try alcohol) to a maintenance stage (drinking for longer than 6 months).

messages (Werch et al., 2008b) there was a reduction in alcohol use frequency and problems, fourmonths after intervention, compared to control students who received fitness flyers.

5.4.3.2 Medium-term results (up to 12 months)

Four studies (Werch et al., 1998; 2000b; 2003a; 2005b) examined medium-term intervention effects of three versions of the STARS for Families intervention and the Project SPORT intervention, respectively.

Knowledge and understanding

None of the included studies examined medium-term intervention effects on knowledge.

Attitudes and values

Three studies (Werch et al., 2000b; 2003a; 2005b) examined medium-term impacts on attitudes and values. Werch et al (2000b) examined the effects of the STARS programme among students attending sports physical examinations. At the 6-months post-test fewer intervention than control students from suburban and rural schools, but not urban schools, reported that they intended to use alcohol in next 6 months. At 12 months follow-up, after a two-year STARS intervention, Werch et al (2003a) reported that compared to control students, intervention students in specialist schools (but not community schools) were less likely to report intentions to use alcohol, had significantly greater motivations to avoid alcohol and an overall lower alcohol risk score (all p<0.01). There were no differences between intervention and control students in community or commuter schools on the measures of influenceability, peer prevalence or expectancy beliefs. At 12 months follow-up, Project SPORT participants reported significantly more protective factors and fewer risk factors for alcohol use. Further analyses showed that Project SPORT participants had significantly better parent/child communication (p<0.01), and marginally more positive parent/child relationships (p=0.05), compared to controls who received generic alcohol prevention and health promotion materials, but less protection on perceived susceptibility (p<0.05). There was no significant difference between intervention and control students in intentions to drink in the next 6 months.

Personal and social skills

None of the included studies examined medium-term intervention effects on personal and social skills.

Health and social outcomes relating to alcohol use and sexual health

One study (Werch et al., 2000b) examined the effect of the STARS programme at 6-months in a sample of students recruited from sports physical examinations. Compared to control students who received no intervention, fewer intervention students reported drinking during the previous month (p<0.05) and drinking heavily (p<0.05). There were no significant effects on drinking in the last 7 days. Two studies (Werch et al., 1998; Werch et al., 2003a) examined the effectiveness of the STARS for Families programme at 1 year. Neither version of the programme examined had significant effects on alcohol use behaviours in the medium-term, compared to an alcohol education booklet control. However, the analyses of 30-day heavy use showed an effect in favour of the 2-year version of the STARS for Families programme (Werch et al., 2003). The short-term effects of the Project SPORT programme had declined by the 1-year follow-up (Werch et al., 2005b), although the effects still

favoured the intervention, and intervention students reported using alcohol for a briefer period of time than control students (p<0.05).

5.4.3.3 Long-term results (>12 months)

None of the included studies reported long-term follow-up data.

5.4.4 Summary and evidence statements

Thirteen studies (Argentos, 1991; Dempster et al., 2006; Werch et al., 1996a; 1996b; 1998; 2000a; 2000b; 2003b; 2005a; 2005b; 2005c; 2008a; 2008b) were identified that examined seven brief behavioural or single session intervention approaches relevant to alcohol education. Eight studies (Werch et al., 1996a, 1996b, 1998, 2000a, 2000b, 2003b, 2005a, 2005b) examined interventions delivered by school nurses, of which three studies (Werch et al., 1996a, 2003b, 2005b) combined delivery with a physician, teacher or fitness professional. Three further studies (Dempster et al., 2006; Werch et al., 2005c; Argentos, 1991) were delivered by a consultant, trained research staff and a motivational speaker, respectively, and two studies (Werch et al., 2008a; 2008b) were based on mailed intervention materials.

5.4.4.1 Knowledge and understanding

None of the studies examined intervention effects on knowledge or understanding.

5.4.4.2 Attitudes and values

Eleven studies (Argentos, 1991; Dempster et al., 2006; Werch et al., 1996a; 1996b; 2000a; 2003b; 2005a; 2005b; 2005c; 2008a) examined intervention effects on students' attitudes and values. Short-term increases in negative views of alcohol and/or its consequences were found in six studies, which included evaluations of the one week Programme 'Kickoff' (Argentos, 1991), a single session on the dangers of binge drinking (Dempster et al., 2006), Project SPORT (Werch et al., 2003b; 2005b), an alcohol beverage tailored programme (Werch et al., 2005c), and a brief behavioural intervention strategy addressing positive images (Werch et al., 2008a). Furthermore, a decrease in students' alcohol expectancies (their own or their views of their peers') was found for two studies of Project SPORT (Werch et al., 2003b; 2005b) and among students who attended specialist schools in one study of STARS for families (Werch et al., 2000a). Brief behavioural or single session intervention approaches appeared to have inconsistent short- and medium-term effects on student's intentions to drink. Across five studies (Werch et al., 1996a; 2000b; 2003a; 2005b; 2008a), which examined a range of brief behavioural intervention approaches, there were decreases in students' intentions to drink alcohol However, two studies (Dempster et al., 2006; Werch et al., 1996b) reported no effects on intentions to stop binge drinking or drinking alcohol respectively.

5.4.4.3 Personal and social skills

Short-term intervention effects of the Project SPORT intervention on personal and social skills were examined in two studies (Werch et al., 2003b; 2005b). There was no impact of the intervention on self-control or parent-child relationships compared to no intervention, but relative to other versions of the programme, students who received additional parental components reported greater self-control.

5.4.4.4 Health and social outcomes relating to alcohol use and sexual health

Six studies (Werch et al., 1996a; 1996b; 1998; 2000a; 2000b; 2005a) reported on the short- to medium-term effectiveness of various versions of the STARS for Families programme. The programme had inconsistent effects on the quantity and frequency of alcohol use, and alcohol use in the last 7- or 30-days in both the short- and medium-term. However, STARS for Families did have a positive effect on heavy drinking in the past month, and analyses of 30-day heavy use showed an effect in favour of the 2-year version of the STARS for Families programme (Werch et al., 2000a; 2001; 2003a). Two studies (Werch et al., 2003b; 2005b) reported on the short-term effects of two sport-based brief behavioural intervention programmes. There were no differential intervention effects for a sports consultation when students received an additional preventive consultation on alcohol (Werch et al., 2003b), but compared to no intervention (Werch et al., 2005b), Project SPORT was shown to have short-term effects on frequency and quantity of alcohol use, and heavy alcohol use in the past month. By the 1-year follow-up, programme effects had declined although the direction of effect still favoured the intervention. An intervention based on brief image-based messages for adolescents and their parents had positive, short-term effects on alcohol use frequency and problems (Werch et al., 2008b). An alcohol tailored beverage programme (Werch et al., 2005c) and a brief behavioural intervention founded on the Behaviour-Image Model (Werch et al., 2008a) had inconsistent effects on alcohol use, and two further programmes, a single session on the dangers of binge drinking (Dempster et al., 2006), and the one week Programme "Kickoff" (Argentos, 1991), had no effects on alcohol use.

Evidence statement 4

- 4 (d) There is moderate evidence from nine RCTs, one NRCT and one CBA study¹ to suggest that brief behavioural or single session intervention approaches relating to alcohol use may have mixed effects on attitudes and values relating to alcohol use. There is moderate evidence from four RCTs, one NRCT and one CBA study² to suggest that these programmes may have positive short-term effects on how student's view alcohol use and its consequences and further evidence from five RCTs and one NRCT³ to suggest that brief behavioural intervention approaches have mixed or inconsistent effects on intentions to drink. Findings may only be partially applicable to the UK as the majority of studies were implemented in the USA and may not be generalisable beyond the populations studied. In addition, the emphasis of the STARS for Families and Project SPORT interventions on abstinence may be of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- 4 (e) There is inconsistent evidence from two studies⁴ to determine the effects of brief behavioural and single session intervention approaches on personal and social skills.
- 4 (f) There is moderate evidence from five RCTs⁵ to suggest that brief behavioural intervention approaches based on nurse-led consultations, such as the STARS for Families and Project SPORT programmes, can produce short-term reductions in alcohol use, but further moderate evidence from three RCTs⁶ to suggest that these effects may not be sustained in the mediumterm. There is weak evidence from two RCTs, one NRCT and one CBA study⁷ to suggest that

other brief behavioural and single session intervention approaches may have a limited impact on alcohol consumption. These findings may only be partially applicable to the UK as the majority of studies were implemented in the USA and may not be generalisable beyond the populations studied. In addition, the emphasis of the STARS programme on abstinence may be of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.

¹ Argentos, 1991 (CBA -); Dempster et al., 2006 (NRCT -); Werch et al., 1996a (RCT +); Werch et al., 1996b (RCT ++); Werch et al., 2000a (RCT +); Werch et al., 2000b (RCT +); Werch et al., 2003b (RCT ++); Werch et al., 2005a (RCT +); Werch et al., 2005b (RCT ++); Werch et al., 2005c (RCT +); Werch et al., 2008a (RCT +)
 ² Argentos, 1991 (CBA -); Dempster et al., 2006 (NRCT -); Werch et al., 2003b (RCT ++); Werch et al., 2005b (RCT ++); Werch et al., 200

³ Dempster et al., 2006 (NRCT -); Werch et al., 1996a (RCT +); Werch et al., 1996b (RCT ++); Werch et al., 2000b (RCT +); Werch et al., 2003a (RCT +); Werch et al., 2005b (RCT ++); Werch et al., 2008a (RCT +)

⁴ Werch et al., 2003b (RCT ++); Werch et al., 2005b (RCT ++)

⁵ Werch et al., 1996a (RCT +); Werch et al., 1996b (RCT ++); Werch et al., 2000a; 2001; (RCT +); Werch et al., 2005a (RCT +); Werch et al., 2005b (RCT ++)

³ Werch et al., 1998 (RCT +); Werch et al., 2003a (RCT +); Werch et al., 2005b (RCT ++)

['] Argentos, 1991 (CBA -); Dempster et al., 2006 (NRCT -); Werch et al., 2005c (RCT +); Werch et al., 2008a (RCT +);

Study	Deting	Intervention	Compositor		Outcomes		
Study	Rating	Intervention	Comparator	Follow-up	Knowledge	Attitudes	Skills
Argentos, 1991	CBA -	Programme Kickoff n=280	No intervention n=70	PT NR	-	 ↑ belief that their alcohol (and other drug) use might result in serious consequences** ↑ understanding of popular myths regarding alcohol use and other drugs 	-
Dempster et al., 2006	NRCT -	Alcohol brief intervention n=133	No intervention n=49	1 month n=NR	-	 ↑ negative appraisal binge drinking** ↑ belief could control binge drinking behaviours* NS intentions to stop binge drinking 	-
Werch et al., 1996a	RCT +	STARS for Families n=52	Alcohol education booklet n=52	PT NR 10 weeks n=101 (97%)	-	 ↑ intentions to stop or reduce drinking* ↑ perceived susceptibility to alcohol-related health problems (likely to get sick if drunk)* NS resistance self-efficacy NS perceived prevalence of drinking among peers or adults ↓ perceived prevalence of drinking among adults** NS perceived prevalence of drinking among peers ↑ perceived susceptibility to alcohol-related health problems (likely to become addicted by drinking offen)* 	-
Werch et al., 1996b	RCT ++	STARS for Families n=68	No intervention n=70	3 months n=124 (90%)	-	NS resistance self-efficacy NS drinking consequences NS intentions for alcohol use	-
Werch et al., 2000a; 2001	RCT +	STARS for Families Community n=107 Specialist n=174	Alcohol education booklet Community n=109 Specialist n=175	end of 6 th grade n=569 (88%)	-	NS alcohol risk factor measures (influenceability, peer prevalence, expectancy beliefs, motivations to avoid)	-

Study	Dating	Intervention	Comparator	Follow up	Outcomes		
Sludy	Rating	intervention	Comparator	Follow-up	Knowledge	Attitudes	Skills
Werch et al., 2000a; 2001	RCT +	STARS for Families Community, n=97 Specialist, n=157	Alcohol education booklet Community, n=100 Specialist, n=161	end of 7 th grade n=515 (79%)	-	 NS influenceability NS peer prevalence ✓ expectancy beliefs (specialist schools only*) ↑ motivations to avoid alcohol use (specialist schools only**) ✓ total alcohol risk (specialist schools only*) 	-
Werch et al., 2005a	RCT +	STARS for families, n=150 STARS Plus, n=149	Postcards only n=149	3 months n=433 (97%)	-	 NS peer alcohol use ↓ peer alcohol susceptibility (STARS only*) ↓ alcohol incompatibility (STARS only*) 	-
Werch et al., 2003b	RCT ++	Project SPORT Sport, n=152 Sport Plus, n=150 Sport Plus Parent, n=152	NA	3 months n=444 (98%)	-	 ↑ negative expectancy beliefs (Sport Plus Parent)* ↓ peer prevalence (control only*) 	↑ self-control (Sport Plus Parent*)
Werch et al., 2005b	RCT ++	Project SPORT n=302	No intervention n=302	3 months n=584 (97%)	-	 ↑ negative expectancy beliefs* ↑ behavioural capability** ↑ perceived susceptibility* ↑ parental monitoring* ↑ parent/child communication* NS resistance self-efficacy NS value incompatibility ↓ intentions to drink in the future** ↓ pro alcohol attitudes* ↓ influenceability** NS expectancy beliefs NS subjective norms NS perceived peer prevalence 	NS positive parent child relationship NS self-control

Study	Poting	Intervention	Comparator	Follow-up	Outcomes				
Study	Raing	intervention			Knowledge	Attitudes	Skills		
Werch et al., 2005c	RCT +	Alcohol beverage tailored programme n=115	Alcohol education booklet n=117	PT n=201 (87%)	-	 ✓ influenceability for beer, wine, distilled spirits and malt liquor* ✓ perceived peer prevalence for wine, flavoured coolers, and fortified wine* ↑ perceived susceptibility for beer and wine consumption* ↑ perceived severity for beer, wine and distilled spirits* 	-		
Werch et al., 2008a	RCT +	Plan for Success (1) Planning, n=113; (2) planning + goal setting, n=113; (3) planning + career consultation, n=109	NA	PT n=335 (93%)	-	 ↓ intention to use alcohol** ↑ belief alcohol interferes with other behaviours** 	-		
*p≤0.05; **p≤0.0 ^a Adjusted for ba baseline alcohol	*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported ^a Adjusted for baseline substance use; ^b 30-day heavy drinking, alcohol problems, alcohol use initiation declined over time in intervention and control groups; ^c Adjusted for baseline alcohol use								

Table 5.18. Brief behavioural and single session interventions: short-term programme effects on health and social outcomes

Study	Dating	Intervention	Comparator	Follow up			
Sludy	Rating	intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other
Argentos, 1991	CBA -	Programme Kickoff n=280	No intervention n=70	PT n=NR	NS alcohol use	-	-
Dempster et al., 2006	NRCT -	Alcohol brief intervention n=133	No intervention n=49	1 month n=NR	-	NS binge drinking	NS number of units of alcohol consumed in a single session
Werch et al., 1996a	RCT +	STARS for Families n=52	Alcohol education booklet n=52	PT n=NR	NS alcohol quantity last month NS 30-day alcohol frequency NS recent alcohol use	NS heavy drinking	-

Study	Dating	Intervention	Comparator	Follow up			
Study	Rating	Intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other
Werch et al., 1996a	RCT +	STARS for Families n=52	Alcohol education booklet n=52	10 weeks n=101 (97%)	 ✓ alcohol quantity last month* ✓ 30-day alcohol frequency* NS recent alcohol use 	NS heavy drinking	-
Werch et al., 1996b	RCT ++	STARS for Families n=68	No intervention n=70	3 months n=124 (90%)	NS 7-day use NS 30-day use	↓ 30-day heavy use*	-
Werch et al., 1998	RCT +	STARS for Families n=106	Alcohol education booklet n=105	PT n=187 (89%)	NS 30-day use NS 7-day use NS alcohol frequency NS alcohol quantity	NS 30-day heavy use NS heavy alcohol use	-
Werch et al., 2001; 2000a RCT +	RCT +	STARS for Families Community, n=107 Specialist, n=174	Alcohol education booklet Community, n=109 Specialist, n=175	end of 6 th grade n=569 (88%)	 ✓ initiation (community schools only*) ✓ 7-day use (community schools only*) ✓ 30-day use (community schools only*) 	Ψ 30-day heavy use (community schools only**)	-
		STARS for Families Community, n=97 Specialist, n=157	Alcohol education booklet Community, n=100 Specialist, n=161	end of 7 th grade n=515 (79%)	 ✓ initiation (commuter schools only*) NS 7-day use NS 30-day use 		-
Werch et al., 2005a	RCT +	STARS for families, n=150 STARS Plus, n=149	Postcards only n=149	3 months n=433 (97%)	NS 30-day quantity ↓ 30-day frequency ^c (STARS only*)	NS heavy alcohol use	✓ alcohol-related use problems*
Werch et al., 2003b	RCT ++	Project SPORT Sport, n=152 Sport Plus, n=150 Sport Plus Parent, n=152	NA	3 months n=444 (98%)	NS alcohol use measures ^b	-	-
Werch et al., 2005b	RCT ++	Project SPORT n=302	No intervention n=302	3 months n=584 (97%)	 ↓ 30-day frequency*** ↓ 30-day quantity*** ↓ length of alcohol use** ↓ stage of alcohol initiation*** 		NS alcohol problems
Werch et al., 2005c	RCT +	Alcohol beverage tailored programme n=115	Alcohol education booklet n=117	PT n=201 (87%)	NS 30-day frequency NS 30-day quantity	NS 30-day heavy use	-

Study Ratin		Intervention	Comparator	Follow-up						
Sludy	Raing	intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other			
Werch et al., 2008a	RCT +	Plan for Success (1) Planning, n=113; (2) planning + goal setting, n=113; (3) planning + career consultation, n=109	NA	PT n=335 (93%)	 ✓ length of alcohol use* NS 30-day alcohol frequency NS 30-alcohol use quantity 	-	-			
Werch et al., 2008b	RCT +	Brief image based messages n=182	Fitness flyers n=202	4 months n=NR (94%)	$oldsymbol{\Psi}$ alcohol use frequency*	-				
*p≤0.05; **p≤0.0 ^a Adjusted for ba baseline alcoho										

Table 5.19. Brief behavioural and single session interventions: medium-term programme effects on knowledge, attitudes and skills

Study	Poting	Intervention	Comparator	Follow-up	Outcomes			
Sludy	Rating	intervention			Knowledge	Attitudes	Skills	
Werch et al., 2000b	RCT +	STARS for Families n=NR	No intervention n=NR	6 months n=163 (92%)	-	↓ intentions to use alcohol in next 6 months (suburban and rural only*)	-	
Werch et al., 2003a	RCT +	STARS for Families n=325	Alcohol education booklet n=325	1 year n=507 (78%)	-	NS influenceability NS peer prevalence NS expectancy beliefs ↓ intentions to use alcohol (commuter schools only**) ↑ motivations to avoid alcohol use (commuter schools only**) ↓ total alcohol risk (commuter schools only**)	-	
Werch et al., 2005b	RCT ++	Project SPORT n=302	No intervention n=302	12 months n=514 (85%)	-	 ↑ alcohol protective factors** ↓ alcohol risk factors* 	-	
*p≤0.05; **p≤0.0)1; ***p≤0.00	1; ↑ increase relative to	o comparator; 🗸 decrea	ase relative to comp	arator; NS not significar	nt; - outcome not reported		

Churchy	Deting	Intervention	Compositor	Fallow or		Outcomes	
Study	Rating	Intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other
Werch et al., 1998	RCT +	STARS for Families n=106	Alcohol education booklet n=105	1 year n=147 (70%)	NS 7-day use NS 30-day use NS alcohol frequency	NS 30-day heavy use	NS alcohol quantity NS heavy alcohol use
Werch et al., 2000b	RCT +	STARS for Families n=NR	No intervention n=NR	6 months n=163 (92%)	NS 7-day use ✔ 30-day use*	ullet 30-day heavy use *	-
Werch et al., 2003a	RCT +	STARS for Families n=325	Alcohol education booklet n=325	1 year n=507 (78%)	NS lifetime alcohol use NS 30-day alcohol use NS length of drinking	NS 7-day alcohol use	-
Werch et al., 2005b	RCT ++	Project SPORT n=302	No intervention n=302	1 year n=514 (85%)	NS 30-day frequency NS 30-day quantity ↓ length of alcohol use* NS stage of alcohol initiation	NS 30-day heavy use	NS alcohol problems
*p≤0.05; **p≤0.0	01; ***p≤0.00	1; ↑ increase relative to	o comparator; 🗸 decrea	ase relative to com	parator; NS not significant; -	outcome not reported	

Table 5.20. Brief behavioural and single session interventions: medium-term programme effects on health and social outcomes

5.5 Multicomponent school- and community-based programmes

5.5.1 Overview of evidence identified

Nine studies (Perry et al., 1996; Komro et al., 1999; 2001; 2008; Perry et al., 2002; Williams et al., 1995; Toomey et al., 1996; Johnson et al., 1990; Chou et al., 1998) were identified that examined two multicomponent, school- and community-based programmes: Project Northland and the Midwest Prevention Project. Both programmes were based in communities in the USA and combined comprehensive school-based curriculums, with community-based activities and parental involvement components. A variety of stakeholders were involved in delivery of both programmes.

The theoretical framework for the intervention was not reported for the Midwest Prevention project, but Project Northland was reported to be based on the theory of triadic influence, and Perry's planning model for adolescent health promotion. The sample size for both programmes exceeded 1,500 students and both programme primarily targeted students in the sixth to the eighth grade. A second phase of the Project Northland programme targeted students in 11th and 12th grade. Both studies were delivered over a three year period but follow-up results were only reported at the end of the programme and medium and long-term follow-up results were not available for either study.

5.5.2 Quality assessment

All nine studies identified for inclusion were based on an RCT design. On the whole the quality of the studies was generally judged to be of moderate quality. Only one study (Chou et al., 1998) was rated poor quality, and this was because there were differences between intervention and control participants at baseline and the study suffered from a high level of attrition with almost 60% of participants failing to provide complete data at all follow-ups. Of the studies rated moderate quality, details were lacking regarding the method of randomisation and the authors did not always sufficiently report on baseline differences between intervention and control students.

Table 5.21. Alconol education. multicomponent school and community programm	Table 5.21. Alcoh	ol education: multicor	nponent school and	community pro	grammes
-----------------------------------------------------------------------------	-------------------	------------------------	--------------------	---------------	---------

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Perry et al., 1996; Komro et al., 2001	RCT +	USA n=2,351 6th-8th grade	School, community, family	Project Northland (Phase I): Three year programme; parental involvement/educational curriculum, behavioural curricula, peer participation, community task force activities	NR	Peers, teachers, adult volunteers
Perry et al., 2002	RCT +	USA n=2953 11-12 th grade	School, community, family	Project Northland (Phase II): interim curriculum (9 th grade); classroom curriculum (11 th grade), 11 postcards for parents, print media campaign, peer action teams, community action teams	NR	Teachers, Peers
Williams et al., 1995	RCT +	USA n=2,351 6 th grade	School, community, family	Project Northland (Slick Tracey Home Team): 4 weekly sessions; activity story books, small group discussion, family fun night.	NR	Teachers, Peers
Toomey et al., 1996	RCT +	USA n=1,028 7 th grade	School, community, family	Project Northland (Amazing Alternatives!): kickoff session for parent; 8-week peer led classroom curriculum; peer participation programme to create alcohol free alternative activities; home programme booklets mailed to parents; information for parents	NR	Teachers, Peers
Komro et al., 1999	RCT +	USA n=1,236 6 th grade	School, community, family	Project Northland: peer leadership activities during Phase I	NR	Teachers, Peers
Komro et al., 2008	RCT +	USA N=4,259 6 th grade	School, community, family	Project Northland (Chicago): Three year programme. See Perry et al., 1996 for details	Theory of triadic influence, Perry's planning model for adolescent health promotion	Teachers, peers
Johnson et al., 1990	RCT +	USA n=1,607 age NR	School, community	Midwest Prevention Project (Kansas City): 10 sessions; drug resistance skills, parent organisation programme, training for community leaders, mass media	NR	NR
Chou et al., 1998	RCT -	USA n=3,412 7th grade	School, community, family	Midwest Prevention Project (Indianapolis): see Johnson et al., 1990 for details	NR	Teachers

5.5.3 Findings

5.5.3.1 Short-term results (<6 months)

All nine studies (Perry et al., 1996; Komro et al., 1999, 2001, 2008; Perry et al., 2002; Williams et al., 1995; Toomey et al., 1996; Johnson et al., 1990; Chou et al., 1998) reported on the short-term effects of two long-term comprehensive multicomponent programmes.

Knowledge and understanding

None of the studies examined short-term intervention effects on knowledge.

Attitudes and values

Perry et al (1996) reported that students in the intervention district during Phase I of Project Northland had significantly lower scores on the peer influence scale at the end of the 8th grade compared to students in comparison districts. However, there were no significant differences between intervention and control communities on the self-efficacy or perceived access to alcohol scales. Among baseline non-users, students in the intervention districts had significantly lower scores at the end of 8th grade on the peer influence scale, and greater self-efficacy to refuse alcohol, relative to students in control districts. No difference was found between intervention and control baseline users. Perry et al (2002) reported that there were no differences in the trajectories of students' perceptions of peer influence to use alcohol or their perceived access to alcohol during phase two of Project Northland. During the interim phase when students were in ninth and tenth grade, students in intervention schools were significantly more likely to experience increased perceptions of peer influence to use alcohol and to decrease their self-efficacy to refuse alcohol. Komro et al (2008) examined a replication of Project Northland study among an urban, low-income population. Compared to the usual school curriculum, there were no effects of Project Northland on any of the measures of attitudes and values, including intentions, peer norms or self-efficacy.

Personal and social skills

None of the studies examined short-term intervention effects on personal and social skills

Health and social outcomes relating to alcohol use and sexual health

Perry et al (1996) reported that significant positive effects of Project Northland were found at the end of 8th grade (end of Phase one). Students in the intervention districts had significantly lower scores (p<0.05) on the tendency to use alcohol scale than students in control districts. In addition, nonusers of alcohol at baseline in the intervention communities reported lower scores than nonusers in control communities at the end of 8th grade (p<0.01), but not at the end of 6th or 7th grade. In addition for all students, the percentages who reported alcohol use in the past month and past week were significantly lower in the intervention group at the end of 8th grade. For baseline nonusers, intervention students had significantly lower monthly and weekly alcohol use at the end of the 8th grade. The percentage of students who reported past year alcohol use were also significantly lower among baseline nonusers in the intervention districts at the end of the 7th grade (p<0.05) and 8th grades (p<0.01). Perry et al (2002) reported that students in the intervention schools were also

significantly less likely to increase their tendency to use alcohol and binge drinking during phase two of Project Northland. However, no differences were found on other measures, although intervention students were marginally less likely than control students to have increased their past month alcohol use during the 11th and 12th grades (p>0.05). During the interim phase (9th and 10th grades), students in the intervention schools were significantly more likely than control students to increase their alcohol use on all measures. Komro et al (1999) examined the effectiveness of two peer leadership components of the 7th grade Project Northland intervention, 'Amazing Alternatives!' programme. At the end of the 7th grade, students who were elected peer leaders had higher scores on the Alcohol Use Tendency Scale⁵ than did those who had participated as volunteer peer leaders (p<0.01), students who did not participate as peer leaders (p<0.05) and (although not significant) students who participated as both elected and volunteer peer leaders (p>0.05). By the end of eighth grade no significant differences remained on the Alcohol Use Tendency Scale. Two studies examined the effects of Project Northland behavioural curricula. Williams and colleagues (1995; RCT +) found that at the end of the sixth grade the 'Slick Tracy Home Team Programme', had no significant effects on alcohol use after controlling for baseline differences between intervention and control students. Toomey et al (1996) found that the 'Amazing Alternatives! Home Programme' had no significant effects on any measure of alcohol use.

Johnson et al (1990) reported that the MPP had no significant effects on alcohol use at the 3-year follow-up when students were in ninth and tenth grade. Chou et al (1998) reported the programme showed a secondary prevention effect on decreasing alcohol use at 6 months after the intervention. The effect was also marginally significant for alcohol use at the 1.5-year follow-up with results of the logistic regression analysis indicating that the secondary prevention effect diminished over time.

5.5.3.2 Medium-term results (up to 12 months)

None of the included studies reported medium-term follow-up data.

5.5.3.3 Long-term results (>12 months)

None of the included studies reported long-term follow-up data.

5.5.4 Summary and evidence statements

Eight studies (Perry et al., 1996; Komro et al., 2001; Perry et al., 2002; Williams et al., 1995; Toomey et al., 1996; Komro et al., 2008; Johnson et al., 1990; Chou et al., 1998) were identified that examined two multicomponent, school- and community-based programmes: Project Northland and the Midwest Prevention Project (MPP). Project Northland was a two phase community trial designed to prevent alcohol use and alcohol related problems among young adolescents. Risk factors for alcohol use were targeted through school-, home- and community-based intervention, including parental involvement/education programmes, behavioural curricula, peer leadership programmes, and community-wide task forces activities. The MPP consisted of four intervention components: (1) a 10

⁵ A summary of alcohol use and intentions to use alcohol based on occasions of alcohol use (lifetime, last year, and last month), intentions to use alcohol (when 21 years or older, next year, month, and week).

session school-based programme focused on drug resistance skills training, additional homework sessions encouraging involvement with parents; (2) a parent organisation programme for reviewing school prevention policies and parent training in parent-child communication skills; (3) training of community leaders in organising a drug abuse prevention taskforce; and (4) mass media coverage.

5.5.4.1 Knowledge and understanding

None of the studies examined intervention effects on knowledge or understanding.

5.5.4.2 Attitudes and values

During Phase I of Project Northland (Perry et al., 1996; Komro et al., 2001), although there was evidence of positive intervention effects on peer influence, the intervention had no impact on selfefficacy or perceived access to alcohol. In the interim phase of the project, students who participated in Project Northland were more likely than controls to experience increased perceptions of peer influence to use alcohol and to decrease their self-efficacy to refuse alcohol (Perry et al., 2002).

5.5.4.3 Personal and social skills

None of the studies examined intervention effects on personal and social skills.

5.5.4.4 Health and social outcomes relating to alcohol use and sexual health

Project Northland significantly reduced growth in binge drinking and tendency to use alcohol during Phase I and II of the programme (Perry et al., 1996; 2002; Komro et al., 2001), however, during the interim phase of the programme the growth in alcohol use was greater among intervention students than control students. The three-year MPP (Johnson et al., 1990; Chou et al., 1998) did not have significant effects on alcohol use in one cohort of ninth/tenth grade students, but a short-term secondary prevention effect was reported in a second cohort.

Evidence statement 5

- 5 (d) There is no evidence from eight RCTs¹ to determine the impact of multicomponent, school- and community-based programmes on knowledge, or personal and social skills.
- 5 (e) There is moderate evidence from three RCTs² to suggest that Project Northland, a long-term multicomponent, school- and community-based programme, has no effects on attitudes and values related to alcohol consumption.
- 5 (f) There is moderate evidence from two RCTs³ to suggest that the Midwest Prevention Project has no effects on alcohol consumption and inconsistent evidence from five RCTs⁴ to suggest that Project Northland may have mixed effects on alcohol consumption. Two RCTs⁵ showed reductions in alcohol consumption in a rural population, particularly among younger adolescents, but replication of the programme among an urban sample⁶ showed that the programme was not effective. Findings may only be partially applicable to the UK as studies were implemented in the USA and may not be generalisable beyond the populations studied.

Perry et al., 1996; Komro et al., 2001 (both RCT +); Perry et al., 2002 (RCT +); Williams et al., 1995 (RCT +); Toomey et al., 1996 (RCT +); Komro et al., 2008 (RCT +); Johnson et al., 1990 (RCT +); Chou et al., 1998 (RCT -) ² Perry et al., 1996; Komro et al., 2001 (both RCT +); Perry et al., 2002 (RCT +); Komro et al., 2008 (RCT ++)

³ Johnson et al., 1990 (RCT +); Chou et al., 1998 (RCT -)

⁴ Perry et al., 1996; Komro et al., 2001 (both RCT +); Perry et al., 2002 (RCT +); Williams et al., 1995 (RCT +); Toomey et al., 1996 (RCT +); Komro et al., 2008 (RCT +)

⁵ Perry et al., 1996; Komro et al., 2001 (both RCT +); Perry et al., 2002 (RCT +)

⁶ Komro et al., 2008 (RCT ++)

Study	Poting	Intervention	Comparator	Follow-up	Outcomes		
Sludy	Rating	intervention			Knowledge	Attitudes	Skills
Perry et al., 1996; Komro et al., 2001	RCT +	Project Northland (Phase 1) n=NR	Usual curriculum n=NR	up to end of 8 th grade (81%)	-	 ✓ peer influence* NS self-efficacy NS perceived access to alcohol 	-
Perry et al., 2002	RCT +	Project Northland (Phase 2) n=NR	Usual curriculum n=NR	up to end of 12 th grade (85%)	-	NS peer influence NS self-efficacy NS perceived access to alcohol	-
Komro et al., 2008	RCT ++	Project Northland (Chicago) n=29 schools	Usual curriculum n=32 schools	up to end of 8 th grade n=3,802 (95%)	-	NS alcohol intentions NS norms supportive of use NS perceived outcomes of supportive use NS resistance self- efficacy NS access to alcohol	-
*p≤0.05; **p≤0.01; ***p≤	0.001; 个 incre	ase relative to comp	arator; 🗸 decreas	e relative to con	nparator; NS not significant	; - outcome not reported	

Table 5.22. Multicomponent programmes: short-term programme effects on knowledge, attitudes and skills

Table 5.23. Multicomponent programmes: short-term programme effects on health and social outcomes

Study	Poting	Intervention	Comparator	Follow-up	Outcomes		
Study	Rating	Intervention			Alcohol use	Heavy alcohol use	Other
Perry et al., 1996; Komro et al., 2001	RCT +	Project Northland (Phase 1) n=NR	Usual curriculum n=NR	up to end of 8 th grade (81%)	ullet weekly alcohol use* ullet monthly weekly use*	-	
Perry et al., 2002	RCT +	Project Northland (Phase 2) n=NR	Usual curriculum n=NR	up to end of 12 th grade (85%)	NS growth rate: weekly use NS growth rate: monthly use		
Williams et al., 1995	RCT +	Project Northland (Slick Tracey Home Team) n=NR	Usual curriculum n=NR	end of 6 th grade n=2,201 (94%)	NS alcohol use	-	-

Study	Poting	Intervention	Comparator	Follow up		Outcomes		
Sludy	Katiliy	intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other	
Toomey et al., 1996	RCT +	Project Northland (Amazing Alternatives) n=521	No intervention n=507	up to end of 8 th grade (83%)	NS alcohol use	-	-	
Komro et al. 1999	RCT +	Project Northland Peer leadership	Project Northland No peer	end of 7 th grade n=NR	-	-	↑ tendency to use alcohol (elected peer leaders*)	
Konno et al., 1999		(elected or voluntary) n=NR	leadership n=NR	end of 8 ^m grade (78%)	-	-	NS tendency to use alcohol	
Komro et al., 2008	RCT ++	Project Northland (Chicago) n=29 schools	Usual curriculum n=32 schools	up to end of 8 th grade n=3,802 (95%)	NS alcohol use	-	-	
Johnson et al., 1990	RCT +	MMP (Kansas City) n=NR	Community components only n=NR	PT (3 years) (84%)	NS alcohol use	-	-	
				PT (6 months) NR		-	-	
Chou et al., 1998	RCT -	MMP (Indianapolis)	Usual curriculum	PT (1.5 years) NR	↓ alcohol use (baseline users**)	-	-	
		n=1,904	1=1,508	2.5 years NR	NS alcohol use	-	-	
				3.5 years NR	NS alcohol use	-	-	
*p≤0.05; **p≤0.01; ***p≤	≦0.001; ↑ incre	ase relative to comp	arator; 🗸 decreas	se relative to con	nparator; NS not significant;	 outcome not reported 		

5.6 Peer support and counselling programmes

5.6.1 Overview of evidence identified

Five studies (Colnes, 2001; Padget et al., 2005; Webster et al., 2002; Bremberg & Arborelius, 1994; Valentine et al., 1998) examined five peer support and counselling programmes for students. Three studies (Colnes, 2001; Padget et al., 2005; Webster et al., 2002) examined peer support programmes. Colnes (2001) examined the effects of a four-day residential training retreat for peer leaders and Padget et al (2005) examined the impact of peer leadership training on high school students who taught the Protecting You/Protecting Me (PY/PM) curriculum to elementary students⁶. Webster et al (2002) examined the effects of participation in a peer support programme for year 7 students, who were supported by peers in year 11. Two studies (Bremberg & Arborelius, 1994; Valentine et al., 1998) examined counselling programmes, both of which included individual and group counselling sessions.

Three studies were conducted in the USA, and one each in Australia and Sweden. All five programmes were school-based only and the provider for three peer support programmes (Colnes, 2001; Padget et al., 2005; Webster et al., 2002) was peers alone. The two counselling programmes (Bremberg & Arborelius, 1994; Valentine et al., 1998) were delivered by health counsellors or educational psychology students, respectively. Only one study (Bremberg & Arborelius, 1994) reported the theoretical framework for intervention and was based on multiple theories (coping behaviour theory, self-efficacy theory, and social modelling). The sample size recruited across the included studies ranged from 76 to 428 and four of the five studies targeted older adolescents. Follow-up ranged from immediate post-test to a maximum of 6 months.

5.6.2 Quality assessment

Of the five studies identified for inclusion, one study (Colnes, 2001) was based on an RCT design; two studies (Padget et al., 2005; Valentine et al., 1998) were NRCTs; and two studies (Bremberg & Arborelius, 1994; Webster et al., 2002) were based on CBA designs. The RCT by Colnes (2000) appeared to have been adequately conducted, but the sample size for the study was small and the reported method of randomisation was inadequate. The two NRCTs differed in quality. Adequate details of the study methodology were reported for the study by Padget et al (2005) and overall the study was rated moderate quality, but there were differences in ethnicity and baseline alcohol use between intervention and comparison groups in the study by Valentine et al (1998). In addition, no data was reported on attrition of subjects and few methodological details were reported and the study was rated poorly. Neither CBA study (Bremberg & Arborelius, 1994; Webster et al., 2002) was of high quality nor was sufficient detail reported to adequately assess the methodological quality. Both studies were consequently rated poorly.

⁶ The impact of this programme on its elementary school participants is discussed in Jones et al (2009).

	_					
Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Colnes, 2001	RCT +	USA n=76 mean 15.4 years	School	Super Leaders: Four day residential training; after school peer leadership programme and activities	Social skills training	Peers
Padget et al., 2005	NRCT +	USA n=401 High school	School	PYPM : One lesson per week for 8 weeks over 5 years; peer leader training	NR	Peers
Webster et al., 2002	CBA -	Australia n=428 mean 12 years	School	10-16 sessions (45 minutes each); peer support programme, confidence and individuality development	NR	Peers
Bremberg & Arborelius, 1994	CBA -	Sweden n=124 15-16 years	School	It's Your Decision: Six sessions; group discussions and individual counselling	Coping behaviour, self- efficacy theory, social modelling	Health counsellor (teacher, school social worker, or school nurse)
Valentine et al., 1998	NRCT -	USA n=336 mean 13; 15 years	School	Urban Youth Connection: Individual, paired or group counselling	NR	Educational psychology students

Table 5.24. Alcohol education: peer support and counselling programmes

5.6.3 Findings

5.6.3.1 Short-term results (<6 months)

Knowledge and understanding

One study (Padget et al., 2005) examined the effects of a peer support prevention programme, which focused on reducing the dangers associated with drink driving. Students who taught the PY/PM programme to elementary school students demonstrated an increase in knowledge relating to the effects of alcohol (p<0.001) and the risks of high levels of alcohol use (p<0.05).

Attitudes and values

Two studies reported short-term intervention effects of two peer leadership programmes on attitudes and values. Colnes (2000) found that both intervention and control students had negative attitudes towards substance use at baseline and follow-up. There was no difference between intervention students who taught the PY/PM programme and controls (Padget et al., 2005) on any of the measures of attitudes and values including intentions or self-efficacy. At 4 months post baseline, intervention students who participated in the Its Your Decision counselling programme (Bremberg & Arborelius, 1994) reported a greater reduction in self attributed psychological problems related to alcohol (p<0.05). There was no difference in accidents related to alcohol, peer problems related to alcohol, parental problems related to alcohol or perceived lack of control of drinking.

Personal and social skills

Padget et al (2005) examined the effects of a peer support programme on the dangers of drinking and

driving. There was no evidence that the programme affected changes in riding with impaired drivers or driving after drinking.

Health and social outcomes relating to alcohol use and sexual health

Two studies reported short-term intervention effects of two peer support programmes on alcohol use. Colnes (2000) reported that there was no significant change in frequency of alcohol use in either peer leaders who participated in the Super Leaders programme or the control students between baseline and 4-months follow-up. Means for both the intervention and control groups at pre and post test indicated that students in both groups were largely abstinent from alcohol at baseline and follow-up. Padget et al (2005) found that students who taught the PY/PM programme reported lower levels of binge drinking at post-test relative to control students (p<0.05). However, there was no significant difference in the number of students reporting recent alcohol use at post-test.

There were no short-term effects of the It's Your Decision counselling programme (Bremberg & Arborelius, 1994) on any measure of alcohol consumption (quantity, frequency, frequency of getting drunk) and Valentine et al (1998) found that there was no difference in use of different alcohol drinks in middle school students between intervention and comparison students. Among high school students, there was a greater proportion of users of liquor, beer, wine, and 'alcopops' in the intervention group than the control group.

5.6.3.2 Medium-term results (up to 12 months)

Knowledge and understanding

None of the included studies examined medium-term intervention effects on knowledge and understanding.

Attitudes and values

None of the included studies examined medium-term intervention effects on attitudes and values.

Personal and social skills

None of the included studies examined medium-term intervention effects on personal and social skills.

Health and social outcomes relating to alcohol use and sexual health

One study (Webster et al., 2002) reported on the medium-term effects of a peer support programme on alcohol use. Webster et al (2002) found that the programme had no effects on the alcohol use. Over 6 months of follow-up, participants in both intervention and comparison schools showed an increase in the enjoyment and use of alcohol and there was no difference in the pattern of change between groups.

5.6.3.3 Long-term results (>12 months)

None of the included studies examined the long-term intervention effects of peer and counselling support programmes.

5.6.4 Summary and evidence statements

Five studies (Colnes, 2001; Padget et al., 2005; Webster et al., 2002; Bremberg & Arborelius, 1994; Valentine et al., 1998) examined peer support and counselling programmes for students. Three studies (Colnes, 2001; Padget et al., 2005; Webster et al., 2002) examined peer leadership or support programmes and two studies (Bremberg & Arborelius, 1994; Valentine et al., 1998) examined counselling programmes, both of which included individual and group counselling sessions.

5.6.4.1 Knowledge and understanding

None of the studies examined intervention effects on knowledge or understanding.

5.6.4.2 Attitudes and values

Three studies (Colnes, 2001; Padget et al., 2005; Bremberg & Arborelius, 1994) examined short-term intervention effects on attitudes and values. For two peer leadership programmes there appeared to be modest impacts on attitudes to alcohol. One study of peer leadership training (Colnes, 2001) found that both intervention and control students maintained negative attitudes towards substance use, but a second study (Padget et al., 2005) found positive intervention effects among students who taught the PY/PM programme on attitudes about the effects of alcohol use and the risks of high levels of alcohol use. One study of a counselling programme found that the programme had a positive impact on the number of psychological problems that students' attributed to their alcohol use (Bremberg & Arborelius, 1994).

5.6.4.3 Personal and social skills

None of the studies examined intervention effects on personal and social skills.

5.6.4.4 Health and social outcomes relating to alcohol use and sexual health

Two studies (Bremberg & Arborelius, 1994; Valentine et al., 1998) reported on the short-term effects of in-school counselling programmes. Neither programme was shown to be consistently effective. In addition, one programme had potentially harmful effects on high school students' alcohol consumption. Three studies reported short-term data on the effectiveness of peer support programmes. Two of the programmes examined were shown to have no effects on the alcohol use of participants (Colnes, 2000; Webster et al., 2002). However, peer leaders who taught the PY/PM programme reported lower levels of binge drinking (Padget et al., 2005).

Evidence statement 6

There is inconsistent evidence from one RCT, two NRCTs and two CBA studies¹ to determine the effectiveness of counselling and peer support on attitudinal and behavioural outcomes related to alcohol use. Findings may only be partially applicable to the UK as studies were implemented in the USA and may not be generalisable beyond the populations studied.

¹ Colnes, 2001 (RCT +); Padget et al., 2005 (NRCT +); Webster et al., 2002 (CBA -); Bremberg & Arborelius, 1994 (CBA -); Valentine et al., 1998 (NRCT -)

Author	Study	Intervention	Comparator	Follow-up	Outcomes		
Author	design	Intervention	Comparator		Knowledge	Attitudes	Skills
Colnes, 2001	RCT +	Super Leaders n=38	No intervention n=38	PT n=66 (87%)		NS attitudes	
Padget et al., 2005	NRCT +	PY/PM n=218	PAL only n=183	PT n=329 (82%)	 ↑ effects of alcohol*** ↑ risks of high levels of alcohol use* NS risks of low levels of alcohol use 	NS future intentions NS self-efficacy NS sources of alcohol	NS riding with an impaired driver NS driving after drinking
Bremberg & Arborelius, 1994	CBA -	It's Your Decision n=65	No intervention n=59	up to 2 months 87%	-	NS problems perceived to be related to alcohol use	-
*p≤0.05; **p≤0.01; ***p	≤0.001; ↑ increa	ase relative to comp	arator; 🗸 decreas	se relative to con	nparator; NS not significant;	- outcome not reported	

Table 5.25. Peer support and counselling programmes: short-term programme effects on knowledge, attitudes and skills

Table 5.26. Peer support and counselling programmes: short-term programme effects on health and social outcomes

Author	Study	Intervention	Comparator	Follow-up	Outcomes		
Addio	design	Intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other
Colnes, 2001	RCT +	Super Leaders n=38	No intervention n=38	PT n=66 (87%)	NS frequency of alcohol use	-	-
Padget et al., 2005	NRCT +	PY/PM n=218	PAL only n=183	PT n=329 (82%)	NS recent alcohol use	$oldsymbol{ u}$ binge drinking*	-
Bremberg & Arborelius, 1994	CBA -	It's Your Decision n=65	No intervention n=59	up to 2 months 87%	NS alcohol consumption	-	-
Valentine et al., 1998	NRCT -	Urban Youth Connection n=187	No intervention n=149	PT NR	 ✓ 30-day beer use (high programme participation only) ↑ 30-day wine use (all students) 	-	-
*p≤0.05; **p≤0.01; ***p	≤0.001; ↑ incre	ase relative to comp	arator; 🗸 decreas	e relative to con	nparator; NS not significant; -	outcome not reported	

Table 5.27. Peer support and counselling programmes	s: medium-term programme effects on health and social o	outcomes
-----------------------------------------------------	---------------------------------------------------------	----------

Author	Study design	Intervention	Comparator	Follow-up	Outcomes					
					Alcohol use	Heavy alcohol use	Other			
Webster et al., 2002	CBA -	Peer support n=235	Not reported n=193	up to 6 months n=326	NS alcohol use	-	-			
*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported										

5.7 Review of published economic evaluations

Two studies (Swisher et al., 2004; Pentz, 1998) were identified that met the criteria for inclusion in the review of published economic evaluations. Swisher et al (2004) assessed the cost-effectiveness of the standard Life Skills Training programme and infused Life Skills Training and Pentz (1998) assessed the costs, benefits and cost-effectiveness of the Midwestern Prevention Project.

5.7.1 Review of Swisher et al (2004)

5.7.1.1 Overview

Swisher et al (2004) assessed the cost-effectiveness of the standard Life Skills Training programme and infused Life Skills Training. The study was conducted alongside the cluster RCT reported on by Smith and colleagues (2004) (see Section 5.3). The aim of the economic evaluation was to compare the costs and effects of Infused Life Skill Training (I-LST) with those of Life Skill Training (LST).

5.7.1.2 Summary of effectiveness data

Details of the effectiveness of LST and I-LST (Smith et al., 2004) are discussed in Section 5.3. Briefly, the study examined the standard LST curriculum compared to an infused approach to the curriculum, which had no set number of lessons and was incorporated across subject areas. Nine middle schools were randomly assigned to three conditions: LST; I-LST; or control. The setting for the study was schools of lower socio economic status in rural communities in Pennsylvania, USA. Effectiveness data was not clearly presented in the economic study.

5.7.1.3 Summary of resource utilisation and cost data

The authors only included additional costs required to implement either I-LST or LST. Therefore the cost elements included in the analysis broadly covered the costs of teacher training, lesson development and programme time and, student and teacher materials. The authors did not report how the cost data were obtained or the year in which costs were expressed. Total cost elements and associated cost data are presented for the programme in Table 5.28.

Cost elements		Infused LST		Standard LST	Difference
7 th grade					
Total 7 th grade		51384.32	\$	32041.25	19343.07
Total 7 th grade by Student	\$	129.11	\$	95.65	33. <i>4</i> 6
8 th grade					
Total 8 th grade	\$	46442.11	\$	20822.01	25620.10
Total 8 th grade by Student	\$	116.69	\$	62.16	54.33
Total both grade		97826.43	\$	52863.26	44963.18
Total both grades by Student		245.80	\$	157.80	87.99

Table 5.28. Cost elements (reproduced from Swisher et al., 2004)
5.7.1.4 Summary of cost-effectiveness data

The authors describe a CEA with calculations of incremental costs of the LST and I-LST programmes per year per student. After one year, the authors reported that the standard LST programme was more cost-effective than I-LST by \$33.46 per student. In the second year, the authors reported that LST had no effects and cost \$62.16 per student, but that whereas I-LST was more costly at \$116.69 per student, it reduced smoking among females and was therefore more cost-effective. The authors undertook an analysis of the project costs of the programme over three years. In the third year, they projected that the cost per student for delivery of the programme over two school grades would be \$91.65 for I-LST compared to \$92.63 for LST. The 3-year total costs of the two programmes were estimated at \$109,429.04 and \$93,088.17, respectively.

5.7.1.5 Comments

The description of the interventions and their effects were not clearly reported. The description of resource use and unit costs of each alternative was adequately reported but not clearly tabulated. The authors refer indirectly to the use of marginal costs, considering the costs of both interventions as incremental compared to normal practice. No discount rate appears to have been applied although the authors refer to a time horizon. The authors report that the study design used is a cost-effectiveness analysis (CEA), however no incremental cost-effective ratios (ICERs) are expressed or reported and it is unclear how costs were related to outcomes (effects). What the authors call a sensitivity analysis is in fact a continuation model projecting the results of year 2 to year 3. Given the poor methodological and reporting quality of the study, the lack of clarity as to the effects described, the setting and the considerable attrition of participants in the original trial, the results of the study are unlikely to be generalisble to a UK context.

5.7.2 Review of Pentz (1998)

5.7.2.1 Overview

Pentz (1998) assessed the costs, benefits and cost-effectiveness of the Midwestern Prevention Project (MPP). Approximate costs, benefits, and cost-effectiveness were calculated from 5-year follow-up (6-year) outcome data and operational costs.

5.7.2.2 Summary of effectiveness data

As reported in the effectiveness review, the overall research design of the MPP included a quasiexperimental trial in Kansas City, Kansas and Kansas City, Missouri, followed by an experimental trial in Indianapolis. Pentz (1998) reported on the whole sample of students who received the MPP, which included students based in 107 junior or middle schools and 62 senior schools (approximately 26,000 adolescents entered the intervention each year). Follow-up was initially at 5 years but the text reports on the follow-up of a sub sample of 1,002 participants into adulthood. The programme described by Pentz (1998) consisted of five components: (1) mass media programming (31 programmes a year for the first 3 years); (2) a school programme (13 sessions in the sixth/seventh grade and 5 booster sessions the following year); (3) a parenting programme (education and coordination of parents with school policy in years 2 and 3 and towards the end of middle school); (4) community organisation (community leader training, organisation, planning, and implementation of community prevention campaigns); and (5) local policy change in years 4 and 5⁷.

The effects of the MPP were reported based on a random sample of 5,055 students from most of the Kansas City area. The author reported an accumulated 3-year net reduction of 9% in the incidence of monthly drunkenness, decreasing to approximately 2.5% by year 5. The effects on alcohol consumption appeared to be maintained into adulthood.

5.7.2.3 Summary of resource utilisation and cost data

Approximate costs for each prevention component of the MPP for each year of delivery are shown in Table 5.29.

Prevention component	1984-85	1985-86	1986-87	1987-88	1988-89	1989-90
Programme development	150	250	45	180	96	90
Training	44	88	118	162	176	176
Implementation	179	316	609	667	834	865
Institutionalisation	-	22	44	44	66	88
Programme subtotal	373	676	816	1053	1166	1229
Research/evaluation	246	470	565	776	879	925
Total (research + programme)	768	1146	1381	1829	2045	2144
Dollars per family (total)	102	51	37	36	34	30

 Table 5.29. Approximate direct costs of the MPP by component (in thousands of dollars)

However, the author estimated that costs of delivering the MPP as a "packaged product" would be less than those presented. Therefore, the yearly cost of the MPP was estimated at \$31 per family (year of costs was not reported) in a large city of over 1 million, although start-up costs were estimated to be more than treble this figure. However, the author cautions that this may be a conservative estimate of cost as it does not take into account constant development costs and the "ownership" of the programme by the community, which may impose further change on the design and content of the intervention.

5.7.2.4 Summary of cost-effectiveness data

The results of the cost-benefit analysis (CBA) presented by the author show a \$700 net saving per family per year resulting from a reduction in monthly drunkenness. Cost benefits ratios are also favourable (ratio to \$1 spent on prevention to saving is \$1:1.69). Costs and benefits were based on 26,000 new families being added per year to the prevention programme.

⁷ This was an additional component not described in the study by Johnson and colleagues (1990) that was included in the review of effectiveness.

The authors also undertook a CEA. The MPP was compared to drug education "as usual", which was estimated to cost \$6 per student. Usual drug education programmes were assumed to have no effects on alcohol or other substance use behaviours. Results of the CEA are shown in Table 5.30.

Variables		MPP	Traditional drug education	Extra cost (ΔC)	Extra effect (ΔE)				
Average cost	С	\$31	\$6	\$25	-				
Reduction in monthly drunkenness	E	2.5%	0	-	2.5%				
Incremental analysis									
ICER	ΔC/ ΔΕ	\$25/2.5	5 = \$10	-	-				

Compared to "usual" drug education the ICER of the MPP was reported to be equal to the ratio of its incremental cost per incremental effects, equivalent to \$10 per net reduction in the incidence of monthly drunkenness. The authors do not report whether discount rates were applied in the analyses.

5.7.2.5 Comments

The economic evaluation is based on effectiveness data presented in a conference abstract and was therefore not included in the review of effectiveness (Section 5). In addition, the authors made the assumption that "usual" drug education had little or no effect on alcohol use behaviours. However, given from the results presented that the intervention appears to be effective into adulthood, it is possible that the intervention may be exportable into a UK setting and the costs may represent value for money when compared to the benefits.

5.7.3 Summary and evidence statements

Two studies (Swisher et al., 2004; Pentz, 1998) were identified that met the criteria for inclusion in the review of published economic evaluations. Swisher et al (2004) assessed the cost-effectiveness of standard and infused LST, and Pentz (1998) assessed the costs, benefits and cost-effectiveness of the MPP. Evaluations of the effectiveness of these programmes were identified and included in Section 5.3.

The standard LST programme was found to be more cost effective than I-LST by \$33.46 per student after 1 year of intervention delivery. In the second year, however, standard LST had no effects and the authors conclude that I-LST is more cost-effective. The 3-year total costs of the two programmes were estimated at \$109,429.04 and \$93,088.17, respectively.

The results of the cost-benefit analysis (CBA) of the MPP demonstrated a \$700 net saving per family per year resulting from a reduction in the incidence of monthly drunkenness. Cost benefits ratios were also shown to be favourable (ratio to \$1 spent on prevention to saving is \$1:1.69). Compared to "usual" drug education the ICER of the MPP was reported to be equal to the ratio of its incremental cost per incremental effects, equivalent to \$10 per net reduction in the incidence of monthly drunkenness.

Evidence statement 7

There is inconsistent evidence from two economic evaluation studies¹ to determine the costeffectiveness of school-based interventions that aim to prevent or reduce alcohol use in young people under 18 years old. This evidence may be of limited applicability to a UK context because cost and benefit estimates were based on data from studies conducted in the USA.

¹ Swisher et al., 2004 (CEA -); Pentz, 1998 (CBA/CEA +)

6 Sex and relationships education

A total of 75 articles met the criteria for inclusion in the review of sex and relationships education programme. Nine articles were systematic reviews and meta-analyses, 65 articles reported on evaluations of sex and relationships education interventions, and one article was an economic evaluation study.

6.1 Systematic reviews and meta-analyses

Nine systematic reviews and meta-analyses were identified that examined the effectiveness of sex and relationships education interventions aimed at children aged 11-19 years, including abstinence and abstinence plus safer-sex programmes, safer-sex promotion and sexuality interventions.

6.1.1 Quality assessment

The reviews undertaken by Oakley et al (1995), Sales et al (2006) and Underhill et al (2007; 2008) were all rated 'SR ++'. These reviews were well conducted and included clearly reported methodology, which detailed a sufficiently rigorous search of the literature, and demonstrated how study quality was considered during analysis of the results. The quality of the meta-analysis by Franklin et al (1997) was judged to be adequate, but details around whether studies were similar enough to combine were poorly addressed by the authors and the article was rated 'SR +'. The review undertaken by Bennett and Assefi (2005) did not include a quality assessment of included studies, but the study was judged to be generally well-conducted and rated 'SR +'. Similarly, study quality was not reported by Robin et al (2004) or Pedlow and Carey (2003), but other aspects of the reviews were adequately covered and these were rated 'SR +'. The review undertaken by Kirby and colleagues (1994) was rated 'SR +'. The review addressed an appropriate and clearly focused question but did not assess quality of the included studies and the description of the literature search was not reported in sufficient detail to ascertain whether it was sufficiently rigorous.

6.1.2 Findings

Nine systematic reviews evaluated abstinence only and abstinence plus safer-sex promotion programmes (Underhill et al., 2007, 2008; Bennett & Assefi, 2005), safer-sex promotion (Oakley et al., 1995; Franklin et al., 2007; Kirby et al., 1994; Pedlow and Carey, 2003; Robin et al., 2004) and sexual behaviour-focused interventions (Sales et al., 2006). Findings from three reviews that examined abstinence-only and abstinence-plus programmes (Underhill et al., 2007, 2008; Bennett and Assefi, 2005) indicated that there is no evidence that these interventions cause harmful or adverse effects on relevant behavioural outcomes such as sexual activity. Underhill et al (2007) reported that abstinence-only programmes were ineffective at reducing sexual-risk behaviour, pregnancy and STI incidence and Bennett and Assefi (2005) suggested that any positive abstinence programme effects were only modest and short-term. Regarding abstinence-plus programmes, Bennett and Assefi (2005) reported that studies that included education about contraception significantly influenced knowledge and use of contraception. Underhill et al (2008) indicated that evidence for abstinence-plus programmes reducing HIV risk behaviours was limited, but that results consistently favoured abstinence-plus groups over controls for a range of behavioural outcomes including sexual activity, number of partners, sexual

initiation and condom use and significantly improved knowledge about AIDS/HIV. Additionally, Kirby et al (1994) concluded that there was insufficient evidence to draw conclusions about the effectiveness of abstinence-only programmes on sexual or contraceptive behaviours. The suggestion therefore is that abstinence-only programmes have limited effects or are ineffective for preventing or reducing sexual risk behaviours. Oakley et al (1995) also found evidence that abstinence only education may have an adverse effect and actually increase sexual experimentation among students. For abstinence-plus programmes, which incorporate information on safer sex, there is evidence of positive effects on sexual-risk behaviours, and knowledge and use of contraceptives (Bennett & Assefi, 2005).

Franklin et al (1997) reported that pregnancy prevention interventions were more effective at impacting upon contraception use and pregnancy rates than on sexual activity and that contraception knowledge-building or distribution programmes were more effective than interventions such as abstinence programmes. Positive results of HIV prevention interventions were noted by Pedlow and Carey (2003), but effective programmes produced small effect sizes. Oakley et al (1995) reported that there is no evidence that providing young people with information or contraception results in increases in risky sexual behaviours and similarly Kirby et al (1994) reported finding no evidence that sexuality and AIDS education increases sexual activity. The implications are that evidence points towards safer-sex interventions having some limited effects for preventing unwanted behaviours, but that they do not appear to increase the risk of these behaviours occurring.

Four reviews (Kirby et al., 1994; Pedlow and Carey, 2003; Robin et al., 2004; Sales et al., 2006) identified characteristics of interventions that were effective. Evidence suggests that more effective studies were theoretically based with social cognitive theory the most successfully applied theory, (Kirby et al., 1994; Pedlow and Carey, 2003; Sales et al., 2006) and used trained adult educators (Robin et al., 2004; Sales et al., 2006), although it was recognised that trained peer providers could also be effective (Robin et al., 2004). Two reviews reported that successful interventions applied interactive strategies (Robin et al., 2004; Sales et al., 2006) and two highlighted the importance of including highly specific content focusing on reducing sexual risk behaviour such as skills about condom use or refusing sex (Kirby et al., 1994; Robin et al., 2004).

6.1.3 Summary and evidence statements

Nine systematic reviews evaluated abstinence only and abstinence plus safer-sex promotion programmes (Underhill et al., 2007, 2008; Bennett & Assefi, 2005), safer-sex promotion (Oakley et al., 1995; Franklin et al., 2007; Kirby et al., 1994; Pedlow and Carey, 2003; Robin et al., 2004) and sexual behaviour-focused interventions (Sales et al., 2006). Findings from three reviews that examined abstinence-only and abstinence-plus programmes (Underhill et al., 2007, 2008; Bennett and Assefi, 2005) indicated that abstinence-only programmes have limited effects or are ineffective for preventing or reducing sexual risk behaviours. In addition, Oakley et al (1995) found evidence to suggest that abstinence only education may have an adverse effect and actually increase sexual experimentation among students. For programmes that incorporated information on safe sex and use of contraception, there was evidence from five reviews (Underhill et al., 2008; Pedlow & Carey, 2003; Franklin et al.,

1997; Kirby et al., 1994; Oakley et al., 1995) to suggest that interventions may have effects on preventing sexual risk behaviours, but that these effects tend to be modest. There was no evidence that sexuality and AIDS education increased sexual activity.

Evidence statement 8

- 8 (d) There is strong evidence from three systematic reviews¹ to suggest that abstinence-only programmes have limited effects or are ineffective for preventing or reducing sexual risk behaviours.
- 8 (e) There is moderate evidence from five systematic reviews² to suggest that interventions incorporating information on safer sex and contraceptive use may have positive, but limited effects on preventing sexual risk behaviours. There is no evidence that such programmes increase the occurrence of sexual activity among young people.
- 8 (f) There is moderate evidence from four systematic reviews³ to suggest that effective characteristics of sexual risk reduction interventions include: (1) a theoretical basis; (2) use of trained adult health educators as providers; and (3) provision of highly specific content focusing on sexual risk reduction.
- ¹ Underhill et al., 2007, 2008 (both SR ++); Bennett and Assefi, 2005 (SR +); Kirby et al., 1994 (SR +); Oakley et al., 1995 (SR ++)
- ² Pedlow & Carey, 2003 (SR +); Franklin et al., 1997 (SR +); Kirby et al., 1994 (SR +); Oakley et al., 1995 (SR ++); Underhill et al., 2008
- ³ Kirby et al., 1994 (SR +); Pedlow and Carey, 2003 (SR +); Robin et al., 2004 (SR +); Sales et al., 2006 (SR ++)

Author (Year)	Design	Inclusion/exclusion	Number of studies	Findings
Franklin et al., 1997	SR +	Studies with a primary focus on the primary prevention of adolescent	32 studies	Programmes had a greater effect on contraceptive use and pregnancy rates, than on sexual activity. Contraceptive knowledge building programmes and contraceptive distribution
		pregnancy		only programmes).
Bennett & Assefi,	SR +	School-based programmes reporting intervention impact on	16 studies	Some abstinence-only and abstinence-plus programmes can change sexual behaviours, however effects relatively modest and only lasted short-term. No evidence that abstinence-plus programmes increased sexual activity.
2005		sexual and contraceptive behaviour. USA only.		Programmes that offered contraceptive education significantly influenced students' knowledge and use of contraception.
				Insufficient evidence to determine if school-based abstinence-only programmes delay the onset of intercourse or affect other sexual or contraceptive behaviours.
Kirby et al., 1994	SR +	School-based interventions, impact on sexual or contraceptive behaviours measured	16 studies	I wo curriculums that specified delaying the onset of sexual intercourse as a clear goal - Postponing Sexual Involvement and Reducing the Risk - delayed the initiation of sex. Postponing Sexual Involvement significantly reduced the frequency of intercourse. Only two of the eight programmes (AIDS Prevention for Adolescents in School and an untitled curriculum by Schinke et al) included significantly increased contraceptive use among all experienced youths, but two additional programmes (Postponing Sexual Involvement and Reducing the Risk) increased contraceptive use among specific groups of students.
Oakley et al., 1995	SR ++	Sexual health interventions for young people aged 0-19 years	65 studies; 48 school- based	No evidence that providing practical information and contraception leads to sexual risk taking behaviour, but there is evidence that chastity education may encourage sexual experimentation.
Underhill et al., 2007	SR ++	Abstinence-only interventions with HIV prevention as stated goal	8 studies	No evidence that abstinence-only programs can effectively encourage abstinent behaviour; although programs did not appear to cause harm. Overall, programmes were ineffective for preventing or decreasing sexual activity.
Underhill et al., 2008	SR ++	Abstinence-plus programs designed to influence behaviour change on at least one outcome measure related to HIV transmission	37 studies	No evidence that abstinence-plus programmes increase HIV risk among youth participants in high-income countries and multiple evaluations found that the programmes can decrease HIV risk. No conclusive evidence that abstinence-plus programmes can reduce STI incidence and limited evidence that abstinence-plus programmes can reduce pregnancy incidence; however, direction of effects consistently favoured abstinence-plus programmes. Programmes had mixed effects on sexual behaviour

 Table 6.1. Sex and relationships education: systematic reviews and meta-analyses

Author (Year)	Design	Inclusion/exclusion	Number of studies	Findings
Sales et al., 2006	SR ++	School, community or clinic based interventions or interventions developed for special populations	39 studies; 13 school- based	Successful school-based interventions appeared to be theoretically based, implemented by trained teachers or health educators, and include a variety of skills and knowledge building didactic and interactive activities.
Pedlow & Carey, 2003	SR +	HIV risk reduction interventions targeting teenagers	22 studies	HIV risk reduction interventions have been shown to be effective but are associated with small effect sizes. Most effective studies emphasised a theoretical framework, most often Social Cognitive Theory. Interventions with multiple sessions or long doses have been no more successful than those with shorter doses.
Robin et al., 2004	SR +	Behavioural interventions targeting HIV, STD, and pregnancy for young people aged 13 years or younger	24 studies; 9 school-based	Programmes that produced positive effects: (1) used trained adult facilitators, and two other programmes with positive effects also used trained peer facilitators; (2) included content that was specific to reducing sexual risk behaviour such as refusal of unwanted sex and condom-use skills; and (3) commonly employed interactive and participatory educational strategies.

6.2 UK-based studies

6.2.1 Overview of evidence identified

Twelve UK studies, evaluating seven programmes, were identified that could be defined as predominantly sex and relationships education. Typically these programmes focused on areas of sexual health including contraception, sexually transmitted infections, sexual health services and relationships. Seven studies (Henderson et al., 2007; Mellanby et al., 1995; 2001; Stephenson et al., 2004; 2008; Tucker et al., 2007; Wight et al., 2002) reported on evaluations of three comprehensive school-based programmes, A PAUSE, RIPPLE and SHARE (Sexual Health and Relationships: Safe, Happy and Responsible). Two studies (Mellanby et al., 1995; 2001) examined the A PAUSE sex education programme. Mellanby et al (2001) conducted a follow-on study of the original controlled, repeat cross-sectional study by Mellanby et al (1995). The follow-on study (Mellanby et al., 2001) compared the effects of the peer-led elements of the programme to adult delivery of the same material. The RIPPLE study (Stephenson et al., 2004; 2008) examined the effectiveness of a peer-led sex education programme compared to teacher-led SRE. Three studies (Wight et al., 2002; Henderson et al., 2007; Tucker et al., 2007) examined the effectiveness of the SHARE programme. Two studies (Wight et al., 2002; Henderson et al., 2007) examined the medium- to long-term effects of the programme in the same sample of studies and Tucker et al (2007) undertook a replication study of the programme. Two studies (Gillies et al., 1990; Bellingham et al., 1993) reported on evaluations of the Streetwize UK AIDS education comic and Denman et al (1995) examined a theatre in education programme. Two studies (Graham et al., 2002; Magnusson et al., 2004) examined single lessons focusing on emergency contraception and contraceptive services, respectively. The majority of studies took place within normal school hours while one evaluation of the Streetwize programme (Bellingham and Gillies, 1993) was set within youth training centres (YTCs) that catered for young people who had just left school.

For 11 studies, no detail of the theory that the programme was based on was provided. The theoretical basis for the A PAUSE programme (Mellanby et al., 1995) was applied social learning theory. Teachers were the programme provider for three programmes (an emergency contraception education programme [Graham et al., 2002]; SHARE [Henderson et al., 2007; Tucker et al., 2007; Wight et al., 2002]) and one evaluation of the Streetwize programme (Gillies et al., 1990), with centre trainers providing the intervention in the second evaluation (Bellingham and Gillies, 1993).

Author	Study design and rating	Setting	Baseline population	Programme components	Theory	Provider
Gillies et al., 1990	CBA -	School	n=284 students, 14 years	Streetwize UK : HIV comic, containing 'real life' stories about HIV transmission, quizzes, information and useful contact numbers; accompanied by teachers guide and suggestions for role play and peer group approaches	Not reported	Teachers
Bellingham & Gillies, 1993	RCT +	Youth Training Centres	n=337, 16-19 years	Streetwize comic: one session in youth training centres; information about HIV, AIDS, body fluids and transmission, sex and risk, sexual relationships, and social and sexual behaviour and attitudes.	Not reported	Other: trainers
Denman et al., 1995	CBA -	School	n=807 students; 13-14 years	Theatre in HIV and AIDS education: 30 minute play and 1.5 hour workshop; HIV/AIDS prevention	Not reported	External: theatre group
Graham et al., 2002	RCT ++	School	n=3,234 year 10 students	One two-hour lesson for pupils; in-service teacher training about emergency contraception	Not reported	Teachers
Magnusson et al., 2004	NRCT -	School	n=589 students, 13-14 years	One lesson promotion of contraceptive services; location, opening hours, service provided and confidentiality policy.	Not reported	Health professional
Mellanby et al., 1995	NRCT +	School	n=6,573 students; 15-16 years	A PAUSE Program: 25-30 one-hour lessons over three years; puberty, contraception, reproductive health, assertiveness training, and negotiation in relationships.	Social learning theory	Doctor, Teacher, Peer led
Mellanby et al., 2001	CBA +	School	n=1,675 year 9 students	A PAUSE Program: 10 sessions; puberty, contraception, reproductive health, assertiveness training, and negotiation in relationships.	Social learning theory	Other: teacher and nurse; peers
Stephenson et al., 2004; 2008	RCT ++	School	n=8,766 year 9 students, mean 13.7 years	RIPPLE: Three one-hour sessions; relationships, STIs, condoms, contraception, sexual health services and sexual communication	Not reported	Peer led
Wight et al., 2002	RCT ++	School	n=7,616 students, 13-15 years	SHARE (Sexual Health and Relationships: Safe, Happy and Responsible): 20 sessions over two years (10 in year 9, 10 in year 10) aimed at reducing unsafe sexual behaviour and unwanted pregnancy and improving the quality of relationships	Not reported	Teachers

Author	Study design and rating	Setting	Baseline population	Programme components	Theory	Provider
Henderson et al., 2007	RCT ++	School	n=4,215 female students, 13-15 years	SHARE (Sexual Health and Relationships: Safe, Happy and Responsible): 20 sessions over 2 years (10 in year 9, 10 in year 10) aimed at reducing unsafe sexual behaviour and unwanted pregnancy and improving the quality of relationships	Not reported	Teachers
Tucker et al., 2007	CBA -	School	n=4,324 students, median 14.5 years	SHARE (Sexual Health and Relationships: Safe, Happy and Responsible): Revised teaching materials, multidisciplinary staff training, planned multidisciplinary classroom delivery by teachers and nurses, and access to sexual health services at drop- in centres for pupils.	Not reported	Teachers

The RIPPLE programme (Stephenson et al., 2004; 2008) was led by peers and an intervention based on a single lesson about contraceptive services (Magnusson et al., 2004) was provided by external health professionals, while the A PAUSE programme (Mellanby et al., 1995; 2001) involved doctor-, nurse-, teacher- and peer-led sessions.

The studies varied greatly in the number of participants recruited, from 284 to 7,616, with seven of the evaluations (Graham et al., 2002; Henderson et al., 2007; Mellanby et al., 2004; Mellanby et al., 2001; Stephenson et al., 2004; 2008; Tucker et al., 2007) involving more than 1,500 participants. Six studies (Graham et al., 2002; Henderson et al., 2007; Stephenson et al., 2004; 2008; Tucker et al., 2007 and Wight et al., 2002) discussed the use of power calculations to ensure that an appropriate sample size required to detect a significant effect was selected.

Eleven studies (Denman et al., 1995; Graham et al., 2002; Magnusson et al., 2004; Henderson et al., 2007; Tucker et al., 2007; Wight et al., 2002; Gillies et al., 1990 Stephenson 2004; 2008; Mellanby et al., 1995; 2001) evaluated programmes that targeted adolescents aged between 13 and 15 years. A second evaluation of the Streetwize comic programme was set in youth training centres (YTCs) (Bellingham and Gillies, 1993) and involved older adolescents aged 16-19 years.

Length of follow-up varied greatly across studies, from immediate post-test to over four years. Evaluation of the Theatre in HIV and AIDS education programme (Denman et al., 1995), the Streetwize UK comic (Gillies et al., 1990; Bellingham & Gillies, 1993) and an intervention based on a single lesson on emergency contraception (Magnusson et al., 2004) were based on immediate post-test or short-term follow-up only. Effects of two interventions, the A PAUSE programme (Mellanby et al., 1995; 2001) and an intervention based on a single lesson on emergency contraception (Graham et al., 2002) were examined over the medium-term. Evaluation of the A PAUSE programme was based on follow-up at both immediate post-test (Mellanby et al., 2001) and 1-year (Mellanby et al., 1995). Evaluation of the RIPPLE and SHARE programmes included both medium- and long-term follow-up, at 6-, 12- and 18-months (Stephenson et al., 2004; Tucker et al., 2007), and over four years after intervention (Henderson et al., 2007; Stephenson et al., 2008). Two medium-term evaluations of SHARE (Tucker et al., 2007) and the A PAUSE programme (Mellanby et al., 1995) were based on cross-sectional surveys.

6.2.2 Quality assessment

Of the 12 studies, six (Bellingham and Gillies, 1993; Graham et al., 2002; Henderson et al., 2007; Stephenson et al., 2004; 2008; Wight et al., 2002) were RCTs; two studies were based on an NRCT design (Magnusson et al., 2004; Mellanby et al., 1995) and five studies were CBA studies (Denman et al., 1995; Gillies et al., 1990; Mellanby et al., 2001; Tucker et al., 2007). All RCT and NRCTs were based on cluster randomisation conducted at the organisational level (school: Graham et al., 2002; Henderson et al., 2007; Magnusson et al., 2004; Mellanby et al., 1995; Stephenson et al., 2004; 2008; Wight et al., 2002; training centre: Bellingham & Gillies, 1993). For one study (Magnusson et al., 2004) it was not possible to determine the unit of analysis, but for six RCTs the unit of analysis was not matched to unit of randomisation. Five studies including the emergency contraception education programme (Graham et al., 2002), RIPPLE (Stephenson et al., 2004; 2008) and two evaluations of

the SHARE programme (Henderson et al., 2007; Wight et al., 2002) were rated good quality (++ rating). Graham et al (2002) and Stephenson et al (2004; 2008) provided details of randomisation, using computer generated methods so were rated good quality (++) on this criteria, whereas all other RCTs did not report details about their methods of randomisation, making it impossible to judge whether selection bias had been adequately minimised. Outcomes were generally well reported, but did not always include behavioural outcomes (Denman et al., 1995; Gillies et al., 1990; Mellanby et al., 2001). The analysis undertaken was also well reported in general although effect sizes or significance values were not able to be determined in one study (Denman et al., 1995). One RCT (Stephenson et al., 2004) reported intention to treat analysis. One study, an NRCT (Magnusson et al., 2004), was rated poor quality (- rating) because of poor analyses, short-term follow-up and a lack of details regarding the method of allocation to intervention and control groups.

6.2.3 Findings

6.2.3.1 Short-term results (<6 months)

Five studies (Bellingham and Gillies, 1993; Denman et al., 1995; Gillies et al., 1990; Magnusson et al., 2004; Mellanby et al., 2001) evaluated short-term outcomes for four interventions: the Streetwize UK comic; a Theatre in HIV and AIDS education programme; an intervention based on a single lesson about contraceptive services; and the A PAUSE programme.

Knowledge and understanding

Four studies (Bellingham and Gillies, 1993; Denman et al., 1995; Gillies et al., 1990; Mellanby et al., 2001) examined short-term knowledge outcomes with generally positive results. Three studies (Bellingham and Gillies, 1993; Denman et al., 1995; Gillies et al., 1990) examined the impact of two programmes (Streetwize comic; Theatre in HIV and AIDS) on knowledge relating to HIV and AIDS. A study of a theatre in HIV and AIDS production (Denman et al., 1995), and two studies of the Streetwize comic intervention, one with young people attending YTCs (Bellingham and Gillies, 1993) and a second with school pupils (Gillies et al., 1990), reported positive intervention effects on HIV and AIDS knowledge (both p<0.001). The peer-led version of the A PAUSE Programme (Mellanby et al., 2001) was found to have had a lesser impact on knowledge related to STIs in comparison to the adult-led version (p<0.01).

Attitudes and values

All five studies evaluated outcomes relating to attitudes and results indicated that programmes were less effective at influencing participants' attitudes than knowledge. One evaluation of the Streetwize comic with young people attending YTCs (Bellingham and Gillies, 1993) measured behavioural intentions relating to contraception and reported no significant differences between the intervention and comparison groups. This evaluation of the Streetwize comic programme also found that the intervention did not influence attitudes to HIV and AIDS, although a second study of the Streetwize comic with school pupils (Gillies et al., 1990) reported that the programme positively affected attitudes to HIV prevention through only having one partner (p<0.01). Denman et al (1995) reported on ten measures of attitude, but only found significant positive programme effects for three outcomes relating

to risk perceptions of AIDS (p<0.001); children with HIV (p<0.05) and condom use showing people care for each other (p<0.001) and not for attitudes about condoms reducing the risk of HIV, condom responsibility or confidence in using condoms. An intervention based on a single lesson about contraceptive services (Magnusson et al., 2004) positively affected student's perceptions of contraception and health clinic availability (p<0.001). Short-term evaluation of the A PAUSE Programme (Mellanby et al., 2001), a peer- and adult-led sex education programme for Year 9 pupils, indicated positive effects of the programme on participants' beliefs about males and females who have sex, and perceptions of sexual prevalence amongst peers in the peer-led arm of the programme (p<0.01) in comparison to the adult-led group. However, the programme had a less of an influence on attitudes towards the benefits of having sex, with no difference between the peer- and adult-led groups in their perceptions in the beneficial effects of sex.

Personal and social skills

Two studies (Bellingham and Gillies, 1993; Mellanby et al., 2001) reported on short-term outcomes relating to skills including communication and assertiveness for the Streetwize UK comic and A PAUSE programme, respectively. Bellingham and Gillies (1993) reported that the Streetwize UK comic had no significant effects on HIV communication among young people attending YTCs. Evaluation of the A PAUSE programme (Mellanby et al., 2001) revealed programme effects on assertiveness among females (p<0.05), but not males. However, there were no differences in assertiveness skills between students in the peer- and adult-led versions of the programme.

Health and social outcomes relating to alcohol use and sexual health

Two studies (Magnusson et al., 2004, Bellingham and Gillies, 1993) reported short-term health outcomes for the intervention based on a single lesson about contraceptive services and an evaluation of the Streetwize UK comic with young people attending YTCs, respectively. Both studies reported that the interventions examined had no significant effects on condom or contraception use at last sex (Bellingham and Gillies, 1993) or in general (Magnusson et al., 2004). Bellingham and Gillies (1993) also reported that Streetwize UK comic had no effect on the participants' number of sexual partners.

6.2.3.2 Medium-term results (up to 12 months)

Five studies (Graham et al., 2002; Mellanby et al., 1995; Stephenson et al., 2004; Tucker et al., 2007; Wight et al., 2002) reported medium-term follow-up data for four interventions including a teacher-led intervention about emergency contraception (Graham et al., 2002) and three comprehensive sex education programmes, RIPPLE (Stephenson et al., 2004), SHARE (Wight et al., 2002; Tucker et al., 2007) and A PAUSE (Mellanby et al., 1995). Evaluation of the A PAUSE programme (Mellanby et al., 1995) and one study of the SHARE programme (Tucker et al., 2007) utilised a repeated cross-sectional design and followed up different cohorts of students at each survey point.

Knowledge and understanding

All five studies reported on knowledge outcomes. A teacher-led intervention about emergency contraception (Graham et al., 2002) was effective at increasing knowledge about emergency

contraception, six months after delivery (p<0.01). The effects of the peer-led RIPPLE programme (Stephenson et al., 2004) were examined with regard to the programme's effects on students' knowledge of methods to prevent STIs, local sexual health services and of the emergency contraceptive pill. There were no significant effects of the programme on student's knowledge of emergency contraception or local health services, but compared to students who received teacher-led sex education, there was an increase in knowledge of methods to prevent STIs among intervention females, but not males. An evaluation of the effects of the RIPPLE programme on the peer-leaders themselves (Strange et al., 2002) found that they gained knowledge on emergency contraception (p<0.05), as well as improving their awareness of STI prevention (p<0.001). Two evaluations of the SHARE sex education programme found that the programme had significant and positive effects on knowledge about sexual health (p<0.01; Wight et al., 2002) and STIs (p<0.001; Tucker et al., 2007). The A PAUSE Programme (Mellanby et al., 1995) had a strong positive effect on the accuracy of participants' perceptions about sexual activity amongst their peers (p<0.001).

Attitudes and values

Four studies (Graham et al., 2002; Mellanby et al., 1995; Stephenson et al., 2004; Tucker et al., 2007) examined medium-term intervention effects on attitudes and values for four interventions: a teacherled intervention about emergency contraception (Graham et al., 2002), and three sex education programmes RIPPLE (Stephenson et al., 2004), SHARE (Tucker et al., 2007) and A PAUSE (Mellanby et al., 1995). Graham et al (2002) reported that there were no effects of a teacher-led intervention about emergency contraception on behavioural intentions to use emergency contraception and Stephenson et al (2004) found no significant differences between students who participated in the peer-led RIPPLE programme and control students in terms of their intentions about sex or condom use. In addition, there were no effects of the RIPPLE programme on perceptions about condom or contraceptive availability, or on peer leaders' attitudes about condoms or sex (Strange et al., 2002), although providers did gain more positive attitudes about homosexual relationships. The evaluation of the SHARE programme (Tucker et al., 2007) found positive effects on many attitudes to condom use including those concerning price (p<0.05); sexual enjoyment (p<0.001); embarrassment (p<0.001) and their ability to prevent AIDS (p<0.001); and STIs (p<0.01). Tucker et al (2007) also reported that the SHARE programme had positive effects on behavioural intentions to discuss condoms with a partner (p<0.05), intentions to own condoms (p<0.001), and on confidence in condom acquisition and use (p<0.001). There were no significant effects on of the A PAUSE programme (Mellanby et al., 1995) on students' beliefs towards sex.

Personal and social skills

One study (Stephenson et al., 2004), of the RIPPLE sex education programme, examined participants' confidence in discussing contraception or sex with a partner, but found no significant intervention effects. The RIPPLE programme was reported to have a large and positive effect upon peer-leaders' confidence to deliver sex education lessons to peers (Strange et al., 2002).

Health and social outcomes relating to alcohol use and sexual health

Three studies (Graham et al., 2002; Mellanby et al., 1995; Wight et al., 2002) of a teacher-led intervention about emergency contraception and the A PAUSE and SHARE sex education programmes, respectively, examined medium-term programme effects on outcomes related to sexual health. Graham et al (2002) found that a teacher-led intervention on emergency contraception had no effects on emergency contraception use or sexual experience. Wight et al (2002) reported that the SHARE sex education programme had no effects on condom use overall or at first sex, or on contraception or condom use at last sex. Participation in SHARE (Wight et al., 2002) did not significantly influence the number of unwanted pregnancies, or feelings of regret or pressure at first sex, although a small positive programme effect on regret at first sex with most recent partner (p<0.05) was reported for males. Tucker et al (2007) found that, based on a cross-sectional survey of students who participated in the SHARE programme in 2003 compared to control students, there were no effects of the programme on the number of students who reported having sex. One study of the A PAUSE programme, based on cross-sectional follow-up (Mellanby et al., 1995), indicated a positive effect of the programme on whether participants had ever had sex (OR 1.45; 95% CI 1.13, 1.87).

6.2.3.3 Long-term results (>12 months)

Three studies (Henderson et al., 2007; Stephenson et al., 2004; 2008), which evaluated two comprehensive sex education programmes, SHARE and RIPPLE, respectively, reported long-term follow-up data. Follow-up ranged from 18 months to over four years.

Knowledge and understanding

Only the RIPPLE programme was examined in terms of its impact on outcomes related to knowledge. At 18-months follow-up (Stephenson et al., 2004), the programme had positive effects on participants' knowledge relating to STIs (p<0.001), although this effect was only apparent for males. This was in contrast to the 6-month follow-up, where a positive effect had been seen for females (p<0.01) but not males. As at 6-month follow-up, there were no long-term effects of the peer-led RIPPLE programme on knowledge about emergency contraception or local sexual health services, at 18 months (Stephenson et al., 2004) or four years (Stephenson et al., 2008).

Attitudes and values

Only the RIPPLE programme was examined in terms of its impact on outcomes related to attitudes and values. At 18-months follow-up (Stephenson et al., 2004), there were no significant effects of the RIPPLE programme on students' attitudes or beliefs towards sex, or on perceptions concerning the availability of contraception (Stephenson et al., 2004).

Personal and social skills

One study (Stephenson et al., 2004) reported on long-term skills outcomes for the RIPPLE sex education programme. At 18-month follow-up, (Stephenson et al (2004) reported that the programme had positively affected confidence in condom use amongst females (p<0.001) but not males. However, at 18 month follow-up, RIPPLE (Stephenson et al., 2004) also had a negative effect on girls'

confidence in refusing sex (p<0.05). There was no effect of the RIPPLE programme on quality of relationships or confidence in discussing sex or contraception.

Health and social outcomes relating to alcohol use and sexual health

Both the RIPPLE and SHARE sex education programmes were evaluated in terms of their long-term effects on outcomes related to sexual health (Stephenson et al., 2004; 2008; Henderson et al., 2007). The results generally indicated that these programmes were not consistently effective at improving health outcomes relating to sexual activity, conceptions and contraception in the longer term.

At the 18-month follow-up (Stephenson et al., 2004), females who received the peer-led RIPPLE programme were significantly less likely to report having sex by age 16 years compared to females in the control group (p<0.001). There was no difference between intervention and control males. Based on a 4-year evaluation of the RIPPLE programme, at age 18 there was no difference in the proportion of males or females who had had sex (Stephenson et al., 2008).

Based on long-term follow-up at 18-months and four years (Stephenson et al., 2004; 2008), respectively, there appeared to be no effects of the RIPPLE on contraception use or STI diagnosis. The long-term effect of the programme on unintended pregnancies was less clear. At 18 months follow-up (Stephenson et al., 2004), the programme was found to have had no effect on unintended pregnancies, but at the four-year follow-up (Stephenson et al., 2008), fewer pregnancies were reported amongst intervention students compared to control students (OR 0.62; 95% CI: 0.42, 0.91). There were no effects of the RIPPLE programme on abortion or birth rates (Stephenson et al., 2008). Long-term evaluation of the SHARE programme (Henderson et al., 2007) found that at four and a half year follow-up, there were no programme effects on termination and conception rates.

6.2.4 Summary and evidence statements

The 12 UK studies that were identified included three brief HIV prevention interventions (Denman et al., 1995; Bellingham and Gillies, 1993; Gillies et al., 1990), two interventions based on a single lesson about emergency contraception (Graham et al., 2002) and contraceptive services (Magnusson et al., 2004), and seven studies evaluating three comprehensive SRE programmes: A PAUSE (Mellanby et al., 1995; 2001); RIPPLE (Stephenson et al., 2004; 2008); and SHARE (Henderson et al., 2007; Tucker et al., 2007; Wight et al., 2002).

6.2.4.1 Knowledge and understanding

Two studies (Tucker et al., 2007, Wight et al., 2002) that evaluated SHARE reported positive effects of the programme on knowledge about STIs (Tucker et al., 2007) and sexual health (Wight et al., 2002). The peer-led RIPPLE programme had non-significant effects on knowledge relating to emergency contraception and sexual health services, but had positive effects on students' medium-and long-term knowledge of STIs for females and males, respectively (Stephenson et al., 2004). The adult-led version of the A PAUSE programme was found to have had a greater impact on students' knowledge of STIs than the peer-led version of the programme (Mellanby et al., 2001). Three studies (Denman et al., 1995; Bellingham and Gillies, 1993; Gillies et al., 1990), including one of a theatre in AIDS/HIV education programme (Denman et al., 1995) and two evaluations of the Streetwize UK

comic (Bellingham and Gillies, 1993; Gillies et al., 1990) reported beneficial short-term intervention effects on knowledge of HIV and AIDS. Graham et al (2002) reported positive effects of a teacher-led intervention about emergency contraception on knowledge of contraception and intrauterine contraceptive devices.

6.2.4.2 Attitudes and values

The effects of three comprehensive SRE programmes on attitudes were mixed. Compared to teacherled SRE, the peer-led RIPPLE programme (Stephenson et al., 2004; 2008) was found to have had no effects on attitudes about using condoms or sex at medium- or long-term follow-up. However, positive programme effects of the SHARE programme (Tucker et al., 2007) were found on attitudes concerning condom use and preventing STIs. Both evaluations of the A PAUSE programme (Mellanby et al., 1995; 2001) indicated positive short- and medium-term effects of the programme on perceptions around sexual activity amongst peers. Mellanby et al (1995) found no effects of the programme on beliefs about sexual intercourse, but Mellanby et al (2001) reported that relative to the adult-led version of the programme, the peer-led version had positive on students' short-term attitudes towards the beneficial effects of sex. Results relating to confidence using condoms were mixed. Positive medium-term effects of the peer-led RIPPLE programme (Stephenson et al., 2004) were reported for females but not males, and there were no significant effects of the programme on whether students had discussed contraception with a partner at short- or long-term follow-up. Tucker et al (2007) reported positive programme effects of the SHARE programme on self-efficacy to use condoms, but not on condom acquisition.

Two brief HIV interventions, the Streetwize UK comic and a Theatre in HIV and AIDS education programme, had limited effects on short-term attitudes towards HIV and contraception. Gillies et al (1990) reported a positive effect on attitudes towards having one partner to prevent HIV among school pupils who received the Streetwize UK comic, but among young people in YTCs, Bellingham and Gilles (1993) found no effect of the comic on attitudes about HIV or intentions towards condom use. Denman et al. (1995) also examined attitudes concerning HIV and condoms and reported mainly non-significant findings. There were no significant effects of a teacher-led intervention (Graham et al., 2002) on emergency contraception on intentions to use emergency contraception, but an intervention based on a single lesson about contraceptive services (Magnusson et al., 2004) found that the intervention had a positive effect on perceptions of contraception and health clinic availability.

6.2.4.3 Personal and social skills

Results for skills outcomes were limited with few studies reporting on these outcomes. Two studies (Stephenson et al., 2004; Mellanby et al., 2001) of the RIPPLE and A PAUSE programmes, respectively, evaluated effects on personal and social skills. Stephenson et al (2004) found no medium-term intervention effects of RIPPLE on relationship quality and although Mellanby et al (2001) found no effect of the A PAUSE programme on assertiveness in boys, there was a positive short-term effect of the programme on assertiveness in girls. Bellingham and Gillies (1990) reported no effects of the Streetwize UK comic on short-term communication about HIV.

6.2.4.4 Health and social outcomes relating to alcohol use and sexual health

Few of the studies found significant programme effects on health outcomes related to sexual health. All three of the SRE programmes, A PAUSE (Mellanby et al., 1995), RIPPLE (Stephenson et al., 2008) and SHARE (Tucker et al., 2007), were evaluated in terms of their effects on sexual initiation. There were positive medium- and long-term effects of the A PAUSE (Mellanby et al., 1995) and RIPPLE (Stephenson et al., 2004) programmes on the number of students who reported ever having had sex, although for RIPPLE this effects was only apparent among females at the 18-month follow-up. Tucker et al (2007) found no effect of SHARE on the number of students who reported having had sex. There were no effects of either the RIPPLE (Stephenson et al., 2004; 2008) or SHARE (Wight et al., 2002) programmes on use of condoms or other forms of contraception. There were no medium-term effects of the SHARE (Wight et al., 2002) or RIPPLE (Stephenson et al., 2004) programmes on rates of unintended pregnancies, however, Stephenson et al (2008) found that at age 20, students who participated in the peer-led RIPPLE programme were less likely to have been pregnant. However, only 49% of the original sample was followed up at age 20. There were no long-term effects of the RIPPLE (Stephenson et al., 2007) programmes on termination rates.

Bellingham and Gillies (1993) found no effects of the Streetwize UK comic on students' number of sexual partners or condom use at last sex, among young people attending YTCs. In addition, there were no effects of two interventions based on a single lesson about emergency contraception (Graham et al., 2002) and contraceptive services (Magnusson et al., 2004), respectively, on contraception use at follow up. The teacher-led intervention about emergency contraception (Graham et al., 2002) also had no effects on sexual experience.

Evidence statement 9

- 9 (f) There is moderate evidence from two RCTs and two CBA studies¹ to suggest that comprehensive sex education programmes may be effective at increasing students' knowledge about STIs in the short- to long-term. In addition, there is weak evidence from one RCT and two CBA studies² to suggest that brief interventions focusing on HIV prevention, such as Theatre in Education or a comic-based intervention, may have short-term positive effects on knowledge about HIV/AIDs. This evidence is directly applicable as these studies were conducted in the UK.
- 9 (g) Overall, there is inconsistent evidence from two RCTs and two CBA studies³ to determine the effects of comprehensive sex education programmes on attitudes and values relating to sexual health. However, there is weak evidence from one CBA study⁴ to suggest that a two-year sex education programme aimed at reducing unsafe sexual behaviour and unwanted pregnancy (SHARE) may have positive effects on long-term attitudes and intentions regarding condoms. There is moderate evidence from two RCTs, one NRCT and two CBA studies⁵ to suggest that brief or single session interventions focusing on HIV prevention or contraceptives and contraceptive services may have a limited impact on students' attitudes and values. This evidence is directly applicable as the study was conducted in the UK.

- 9 (h) There is inconsistent evidence from two RCTs and one CBA study⁶ to determine the effects of UK-based SRE approaches on personal and social skills.
- 9 (i) There is moderate evidence from one RCT and one NRCT⁷ to suggest that comprehensive SRE programmes, that include peer-led sessions such as RIPPLE and A PAUSE, may delay sexual initiation, but strong evidence from three RCTs and one NRCT⁸ to suggest that SRE programmes and single session interventions focusing on contraceptives and contraceptive services may have no impacts on condom or contraceptive use. This evidence is directly applicable as the studies were conducted in the UK.
- 9 (j) There is mixed evidence from three RCTs⁹ on the effects of comprehensive SRE programmes on outcomes relating to pregnancy. There is moderate evidence from one RCT¹⁰ of the peerled RIPPLE programme to suggest that this programme may reduce teenage pregnancy, but not abortion, rates in the long-term and strong evidence from one RCT¹¹ of the teacher-led SHARE programme to suggest that this programme has no long-term effects on conceptions or terminations. This evidence is directly applicable as the studies were conducted in the UK.

- ³ Mellanby et al., 1995 (NRCT +); Mellanby et al., 2001 (CBA +); Stephenson et al., 2004; 2008 (RCT ++); Tucker et al., 2007 (CBA -)
- ⁴ Tucker et al., 2007 (CBA -)
- ⁵ Bellingham and Gillies, 1993 (RCT +); Denman et al., 1995 (CBA -); Gillies et al., 1990 (CBA -); Graham et al., 2002 (RCT ++); Magnusson et al., 2004 (NRCT -)
- ⁶ Bellingham and Gillies, 1993 (RCT +); Mellanby et al., 2001 (CBA +); Stephenson et al., 2004 (RCT ++)
- ⁷ Mellanby et al., 1995 (NRCT +); Stephenson et al., 2004 (RCT ++)
- ⁸ Stephenson et al., 2004; 2008 (both RCT ++); Wight et al., 2002 (RCT ++); Graham et al., 2002 (RCT ++); Magnusson et al., 2004 (NRCT -)
- ⁹ Henderson et al., 2007 (RCT ++); Stephenson et al., 2004; 2008 (both RCT ++); Wight et al., 2002 (RCT ++)
- ¹⁰ Stephenson et al., 2008 (RCT ++)
- ¹¹Henderson et al., 2007 (RCT ++)

¹ Mellanby et al., 2001 (CBA +); Stephenson et al., 2004 (RCT ++); Tucker et al., 2007 (CBA -); Wight et al., 2002 (RCT ++)

² Denman et al., 1995 (CBA -); Bellingham and Gillies, 1993 (RCT +); Gillies et al., 1990 (CBA -)

Study	Pating	Intervention	Comparator	Follow-up		Outcomes			
Study	Kaung	Intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills		
Bellingham & Gillies, 1993	RCT +	Streetwize comic n=173	No intervention n=164	2 weeks n=241 (72%)	↑ HIV/AIDS***	NS HIV/AIDS NS condom intentions	NS HIV communication with partner		
Gillies et al., 1990	CBA -	Streetwize UK n=122	No intervention n=162	2 weeks n=227 (80%)	↑ HIV/AIDS***	↑ HIV prevention through having one partner**	-		
Denman et al., 1995	CBA -	Theatre in HIV and AIDS education n=276	Did not take part in the performance or workshop n=531	PT (1 day) n=680 (84%)	↑ HIV/AIDS***	 ↑ everybody is at risk from AIDS*** ↑ other perceptions of the risk of AIDS NS attitudes towards condom use 	-		
Magnusson et al., 2004	NRCT -	Contraceptive services n=NR	Normal sex education n=NR	up to 6 months n=512 (87%)	-	↑ contraception availability and clinic location***	-		
Mellanby et al., 2001	CBA +	A PAUSE Peer-led n=1,064	A PAUSE Adult-led n=611	PT (1 week) n=1,320 (79%)	↓ STDs***	 ↑ perceived normative sex prevalence*** ↓ beneficial effects of sex* ↑ stereotypes about females having sex** ↓ stereotypes about males having sex** 	 ↑ assertiveness (females)* NS assertiveness (males) 		
*p≤0.05; **p≤0.0)1; * ^{**} p≤0.001; •	increase relative t	to comparator; 🗸 de	crease relative to co	mparator; NS not signif	ficant; - outcome not reported			

Table 6.3. UK SRE studies: short-term programme effects on knowledge, attitudes and skills

Table 6.4. UK SRE studies: short-term programme effects on health and social outcomes

					Health outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Bellingham & Gillies, 1993	RCT +	Streetwize UK comic n=173	No intervention n=164	2 weeks n=241 (72%)	-	NS number of partners	NS condom use at last sex	-	-
Magnusson et al., 2004	NRCT -	Contraceptive services n=NR	Normal sex education n=NR	~ 6 months n=512 (87%)	-	-	NS contraception use	-	-
*p≤0.05; **p≤0.	.01; ***p≤0.001;	↑ increase relative	to comparator; 🗸	decrease relative	to comparator; NS	S not significant; -	outcome not reported	d	

Study	Poting	Intervention	Comparator	Follow up		Outcomes	
Study	Katiliy	Intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills
Graham et al., 2002	RCT ++	Emergency contraception n=1,552	Teachers did not receive training N=1,682	6 months n=2,632 (81%)	 ↑ emergency contraception** ↑ intrauterine device* 	NS intention to use emergency contraception	-
Mellanby et al., 1995	NRCT +	A PAUSE n=1,175	Normal education n=5,398	1 year ^a NA	-	NS beliefs about sexual intercourse ↑ perceived normative sex prevalence***	-
Stephenson et al., 2004	RCT ++	RIPPLE n=4,516	Teacher-led SRE 4,050	6 months n=7,770 (88%)	 ↑ methods to prevent STIs (girls only**) NS emergency contraception NS local sexual health services 	NS availability of condoms or contraceptives NS condom use NS sex	NS confidence in discussing contraception or sex with partner
Tucker et al., 2007	CBA -	SHARE n=2,760	Non-SHARE programs n=1,564	1 year ^a NA	∱ STIs***	 ↑ condoms: embarrassment***; reduce sexual enjoyment***; to discuss condoms with partner*; prevent AIDS***; condom use prevents STIs**; price*; intention to own*** ↑ belief in planning to protect from STIs* ↑self-efficacy condom use*** NS self-efficacy condom 	-
Wight et al., 2002	RCT ++	SHARE n=3,616	Normal sex education n=6,000	6 months n=5,854 (87%)	↑ sexual health**	-	-
*p≤0.05; **p≤0.0 ^a cross-sectional)1; ***p≤0.001; ∕ follow-up surve	↑ increase relative t	o comparator; 🗸 de	crease relative to co	omparator; NS not significa	ant; - outcome not reported	

Table 6.5. UK SRE studies: medium-term programme effects on knowledge, attitudes and skills

						Health outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions	
Graham et al., 2002	RCT ++	Emergency contraception n=1,552	Teachers did not receive training N=1,682	6 months n=2,632 (81%)	NS sexual experience	-	NS use of emergency contraception	-	-	
Mellanby et al., 1995	NRCT +	A PAUSE n=1,175	Normal education n=5,398	1 year ^a NA		-	-	-	-	
Tucker et al., 2007	CBA -	SHARE n=2,760	Non-SHARE programs n=1,564	1 year ^a NA	NS reported sex	-	-	-	-	
Wight et al., 2002	RCT ++	SHARE n=3,616	Normal sex education n=6,000	6 months n=5,854 (87%)	 NS regret at first sex NS pressure at first sex ✓ regret at first sex with most recent partner (males only*) 	-	NS condom use: mean score; first sex; last sex NS contraception at last sex	-	NS unwanted pregnancies	
*p≤0.05; **p≤0. ^a Based on cros	01; ***p≤0.001; ss-sectional follo	↑ increase relative ow-up survey	to comparator; Ψ of	decrease relative	to comparator; NS	not significant; - o	outcome not reported			

Study	Pating	Intorvention	Comparator	Follow-up	Outcomes			
	Kaung	intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills	
Stephenson et al., 2004	RCT ++	RIPPLE n=4,516	Teacher-led SRE n=4,050	18 months n=6,656 (76%)	↑ STIs (boys only***) NS emergency contraception NS local sexual health services	NS availability of condoms or contraceptives NS condom use NS sex	 ↑ confidence condom use (girls only**) ↓ sex refusal-efficacy (girls only*) NS confidence in discussing contraception or sex with partner NS quality of relationship 	
Stephenson et al., 2008	RCT ++	RIPPLE n=4,516	Teacher-led SRE n=4,050	4 years n=4,310 (49%)	NS emergency contraception NS sexual health services	NS	-	
*p≤0.05; **p≤0.0)1; ***p≤0.001; ^	increase relative t	o comparator; Ѵ de	crease relative to co	omparator; NS not significa	ant; - outcome not reported		

Table 6.7. UK SRE studies	: long-term programme	effects on knowledge, attitudes and	skills
---------------------------	-----------------------	-------------------------------------	--------

Table 6.8. UK SRE studies: long-term programme effects on health and social outcomes

					Health outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Stephenson et al., 2004	RCT ++	RIPPLE n=4,516	Teacher-led SRE n=4,050	18 months n=6,656 (76%)	 ✓ ever had sex (girls only***) NS same sex experience 	-	NS contraception use at last sex	-	NS unintended pregnancies
Stephenson et al., 2008	RCT ++	RIPPLE n=4,516	Teacher-led SRE n=4,050	4 years n=4,310 (49%)	NS ever had sex NS regret or pressure at first or last sex	-	NS use of contraception NS unprotected first sex	NS STI diagnosis	 ✓ pregnancies NS abortion rates NS birth rates
Henderson et al., 2007	RCT ++	SHARE n=2,080	Normal sex education n=2,135	4.5 years N=4196 (99.5%)	-	-	-	-	NS termination and conception rates
*p≤0.05; **p≤0.	01; ***p≤0.001;	↑ increase relative	to comparator; 🗸	decrease relative	to comparator; NS n	ot significant; - o	outcome not reported	d	

6.3 Abstinence-only programmes

6.3.1 Overview of evidence identified

Ten articles (Blake et al., 2001; Borawski et al., 2005; Christopher & Roosa, 1990; Denny et al., 1999; 2002; 2006; Donnelly et al., 2001; Jorgensen et al., 1993; Roosa & Christopher, 1990; Trenholm et al., 2008) were identified that evaluated eight programmes defined as abstinence only programmes across nine studies. These programmes encouraged and promoted abstinence as the best and only way to prevent pregnancy, HIV and other STIs. All nine studies were conducted in the USA and were delivered within schools during normal lesson hours. Additionally, one study reported on a replication of the Success Express Programme (Roosa and Christopher, 1990), which was offered in schools and community centres. In addition, Project Taking Charge (Jorgensen et al., 1993) included three evening sessions for parents.

Four programmes were teacher-led (For Keeps [Borawski et al., 2005]; Sex Can Wait [Denny et al., 1999; Denny and Young, 2006]; Project Taking Charge [Jorgensen et al., 1993] and Project C.A.R.E. [Donnelly et al., 2001]) and one programme, Managing the Pressures before Marriage (MPM; Blake et al., 2001), was peer-led. Three studies did not report details of the providers for three programmes (Success Express Programme [Christopher and Roosa, 1990; Roosa and Christopher, 1990] and My Choice, My Future!/Recapturing the Vision (MCMF/RTV) [Trenholm et al., 2008]). For five programmes no theoretical base was reported (For Keeps; Success Express Programme; Sex Can Wait; Project Taking Charge; MCMF, RTV). Project C.A.R.E (Donnelly et al., 2001) applied social learning theory and one programme, MPM (Blake et al., 2001) applied social learning and social cognitive theories.

Where reported, sample size varied greatly amongst the included studies from 91 to 1,715 students. One study evaluating Sex Can Wait (Denny et al., 1999) did not describe the number of students, instead reporting that the sample consisted of students from 15 schools. No authors reported whether their study was sufficiently powered to detect an intervention effect. However, two studies discussed that the small sample size of sexually inexperienced students (Borawski et al., 2005) and high non-participation in the programme (Trenholm et al., 2008) might have reduced power.

Five programmes targeted adolescents between the ages of 11 to 14 years (MPM; For Keeps; Success Express Programme; Project C.A.R.E.; MCMF/ RTV). Additionally, the Sex Can Wait programme (Denny et al., 1999; 2002; Denny & Young, 2006) targeted students in middle (aged 11-14 years) and high school (aged 14-18 years). Project Taking Charge (Jorgensen et al., 1993), also targeted older adolescents aged over 14 years. The length of follow-up varied between studies. Evaluation of three programmes (Success Express Programme; Project C.A.R.E; MPM) and one evaluation of the Sex Can Wait programme (Denny et al., 1999) was based on immediate post-test only. Evaluation of one programme, Project Taking Charge (Jorgensen et al., 1993) was based on follow-up at six months and evaluation of the three-year MCMF and RTV programmes took place between 42 and 78 months after baseline data was collected. A second evaluation of Sex Can Wait (Denny and Young, 2006) followed up participants at 18 months.

Author	Study design and rating	Setting	Baseline population	Programme components	Theory	Provider
Blake et al., 2001	RCT +	School	USA n=351 8 th grade students	Managing the Pressures before Marriage (MPM): five one-hour sessions plus homework aiming to increase parent-child communication about sex	Skills based	Peer led
Borawski et al., 2005	NRCT +	School	USA n=1,096 students, 12-13 years	For Keeps: five 40-minute sessions emphasising the benefits of abstinence and consequences of early sexual activity; how condoms and contraception does not protect against the emotional consequences of sex; the need for development of resistance skills	Not reported	Teachers
Christopher & Roosa, 1990	NRCT -	School	USA n=320 students, mean age 12.8 years	Success Express: 6 sessions designed to teach behaviours, attitudes and skills consistent with abstinence; graduation ceremony	Not reported	NR
Roosa & Christopher, 1990	NRCT -	School + community	USA n=339 students, mean age 13 years	Success Express : six sessions over six weeks focusing on self-esteem and family values, growth and development that occur during puberty, media and peer pressures, assertiveness training, and goal-setting skills.	No reported	NR
Denny et al., 1999, 2002	NRCT -	School	USA Students from 15 middle and high schools	Sex Can Wait: 5 weeks curriculum on self-esteem, reproductive anatomy, physiology, changes associated with puberty, values and decision making skills; development and enhancement of communication skills; and goal setting and life planning	Not reported	Teachers
Denny & Young, 2006	NRCT -	School	USA Middle school n=698 students High school n=337 students	Sex Can Wait: Five weeks, 24 lessons on self-esteem, reproductive anatomy, physiology, changes associated with puberty, values and decision making skills; development and enhancement of communication skills; and goal setting and life planning	Not reported	Teachers
Donnelly et al., 2001	NRCT -	School	USA n=839 students, 6 th -8 th grade	Project C.A.R.E./Sex Can Wait: 23 sessions over 1 year to promote abstinence until marriage; Goal setting, decision making, communication skills, self esteem enhancement	Social learning theory	Teachers
Jorgensen et al, 1993	NRCT +	School	USA n=91 students, mean age 14.4 years	Project Taking Charge : 6 weeks abstinence promotion with classroom and parental components	Not reported	Teachers
Trenholm et al., 2008	RCT -	School	USA n=448 Middle School students	My Choice, My Future! (MCMF) 52 lessons over three years; abstinence education	Not reported	Not reported

Table 6.9. Sex and relationships education: abstinence-only programmes

Author	Study design and rating	Setting	Baseline population	Programme components	Theory	Provider
Trenholm et al., 2008	RCT -	School	USA n=480 middle school students	ReCapturing the Vision (RCV): One year daily class; abstinence education		NR

6.3.2 Quality assessment

Of the nine studies, two were RCTs (Trenholm et al., 2008; Blake et al., 2001) and seven were NRCTs (Borawski et al., 2005; Christopher and Roosa, 1990; Denny et al., 1999; Denny and Young, 2006; Donnelly et al., 2001; Jorgensen et al., 1993; Roosa and Christopher, 1990). The RCT by Trenholm et al (2008) was based on individual randomisation of students and Blake et al (2001) randomised at the level of the classroom taking into consideration the probability of type I errors. The study by Blake et al (2001) did not include a 'no intervention' control; rather they compared the effects of the MPM curriculum with and without five homework assignments. Of the two RCT, one study was rated moderate for quality (Blake et al., 2001; RCT +) and one study was rated poor (Trenholm et al., 2008; RCT -). Neither RCT provided details of the method of randomisation so it was difficult to judge whether selection bias was minimised, although both studies stated that random methods were applied. Blake et al (2001) reported outcomes to be reliable, and outcomes and analysis were reported to a high standard. Trenholm et al (2008) did not report on reliability of outcomes and it was not always possible to determine which programme (MCMF or RCV) outcomes were attributable to. Follow up was inadequate in Blake et al (2001) at one week and was unclear and variable in Trenholm et al (2008) at between 42 and 78 months from baseline. Neither study provided the initial sample size so it was impossible to determine attrition rates.

Five of the NRCTs were rated poor (NRCT -) for quality (Christopher and Roosa, 1990; Denny et al., 1990; Denny and Young, 2006; Donnelly et al., 2001; Roosa and Christopher, 1990) and two were rated moderate (NRCT +) for quality (Borawski et al., 2005; Jorgensen et al., 1993). Follow up times were generally inadequate (Borawski et al., 2005; Christopher and Roosa, 1990; Denny et al., 1999; Donnelly et al., 2001; Roosa and Christopher, 1990) and in one study with long-term outcomes (Denny and Young, 2006) attrition was very high at 50%. Attrition was a concern in two other studies, being either high (Christopher and Roosa, 1990) or not calculable due to post-test numbers only being presented (Denny et al., 1999). Analysis and outcomes were reasonably reported across studies except in three studies (Christopher and Roosa, 1990; Donnelly et al., 2001; Roosa and Christopher, 1990) with inadequate reporting of significance levels or effect sizes. Methodology was generally well reported, but lacked detail in three studies (Donnelly et al., 2001; Jorgensen et al., 1993; Roosa and Christopher, 1990).

6.3.3 Findings

6.3.3.1 Short-term results (<6 months)

Seven studies (Blake et al., 2001; Borawski et al., 2005; Christopher and Roosa, 1990; Denny et al., 1999; Denny and Young, 2006; Donnelly et al., 2001; Roosa and Christopher, 1990) reported short-term data for five programmes: MPM; For Keeps; Sex Can Wait, Project C.A.R.E., MPM and the Success Express Programme.

Knowledge and understanding

Four studies (Blake et al., 2001; Borawski et al., 2005; Denny et al., 1999; Denny and Young, 2006) examined short-term impacts on knowledge. One study (Blake et al., 2001) that examined MPM and

two studies of Sex Can Wait (Denny et al., 1999; Denny and Young, 2006) reported that there were no intervention effects on knowledge about abstinence or the curriculum. One study (Borawski et al., 2005) of the For Keeps programme reported significant positive effects on knowledge about HIV and STDs (p<0.001).

Attitudes and values

All seven studies that evaluated short-term outcomes reported on attitudes and values. Denny et al (1999) reported significant, positive effects of the Sex Can Wait curriculum on attitudes towards abstention (p<0.01) among high school students but not among middle school students. However, high school students reported a corresponding increase in hopelessness. Five studies (Blake et al., 2001; Borawski et al., 2005; Denny et al., 1999; Denny & Young, 2006; Roosa & Christopher, 1990) examined behavioural intentions and beliefs regarding sex and abstinence. There was no difference in beliefs supporting delayed sex between MPM only students and students who received the enhanced MPM (Blake et al., 2001), however, students who received the enhanced curriculum were less likely to report intentions to have sex before finishing high school (p<0.01). Students who participated in For Keeps (Borawski et al., 2005) were more likely than control students to report a belief in abstinence until marriage (p<0.001) and until older (p<0.01), and were less likely to report intentions to have sex in the next three months (p<0.05) or the next year (p<0.01). However, For Keeps participants also reported fewer intentions than control students to use condoms in the future (p<0.001). Two studies of the Sex Can Wait programme (Denny et al., 1999; Denny and Young, 2006) found significant, positive effects of the Sex Can Wait curriculum on intentions to remain abstinent among high school students (both p<0.001), but not middle school students. There was no difference in the pre-marital sexual beliefs or sexual behaviours of intervention and control participants in the replication study of the Success Express programme (Roosa & Christopher, 1990). Three studies (Donnelly et al., 2001; Christopher & Roosa, 1990; Roosa & Christopher, 1990) examined programme effects on self-esteem, for the Project CARE and Success Express programmes, respectively, finding no effects on this measure.

Blake et al (2001) also examined the effects of the enhanced MPM on refusal efficacy, finding significant positive effects of the enhanced programme compared to the MPM only curriculum on sexual (p<0.01) and substance use (p<0.001) refusal efficacy. There were no significant effects of the Sex Can Wait programme (Denny et al., 1999; Denny and Young, 2006) on decision making. Additionally one study (Borawski et al., 2005) examined the short-term impact of the For Keeps programme on condom use self-efficacy and sexual impulse control but found no significant differences between intervention and control groups.

Personal and social skills

Four studies (Blake et al., 2001; Christopher and Roosa, 1990; Roosa and Christopher, 1990; Donnelly et al., 2001) reported mainly non-significant short-term outcomes relating to student-family communications for three programmes (Success Express; MPM; Project C.A.R.E). Parent-child communication about sex (p<0.05) and school (p<0.001) was significantly higher among students who received the enhanced MPM relative to the curriculum only version of the programme (Blake et

al., 2001). No significant programme effects on family communication were reported in either study evaluating the Success Express programme (Christopher and Roosa, 1990; Roosa & Christopher, 1990) and there were no differences in family support between intervention and control students who participated in Project C.A.R.E (Donnelly et al., 2001).

Health and social outcomes relating to alcohol use and sexual health

Short-term outcomes related to sexual health were reported by six of the seven studies. The purpose of one study (Donnelly et al., 2001) was to examine the effects of an abstinence-only programme, Project CARE, on substance use outcomes and therefore only reported health outcomes relating to substance use.

Four studies (Christopher and Roosa, 1990; Denny et al., 1999; Denny and Young, 2006; Jorgensen et al., 1993) of three programmes, Success Express Programme; Sex Can Wait; Project Taking Charge, reported outcomes relating to initiation of sex and presented mixed findings. One study (Christopher and Roosa, 1990) that evaluated the Success Express programme found that intervention students reported more lifetime sexual experience than control students (p<0.05), an outcome found to be non-significant in a second evaluation of this programme (Roosa and Christopher, 1990). Both evaluations of Sex Can Wait (Denny et al., 1999; Denny & Young, 2006) found no effect of the programme on sexual initiation among middle school students. However, among high school students a positive effect of the intervention on whether students had ever been sexually activity was found (both p<0.001). The same pattern of results was apparent for the impact of Sex Can Wait on past month sexual activity, with non-significant results for middle school students but significant and positive outcomes for high school students (both p<0.01). Two studies (Blake et al., 2001; Borawski et al., 2005) of the MPM and For Keeps programmes, respectively, examined intervention effects on recent sexual activity finding no significant effects of either intervention. Borawski et al (2005), however, found significant positive effects of the For Keeps programme on frequency of sex (p<0.05), having two or more partners (p<0.01) and multiple episodes of sex (p<0.05).

Donnelly et al (2001) reported that Project C.A.R.E, an abstinence only intervention based on Sex Can Wait, had no effects on alcohol use or substance use, except for use of crack or methamphetamines for which negative effects were demonstrated (p<0.05). One further study (Blake et al., 2001) of MPM examined alcohol related outcomes and reported a small, but significant positive effect of the enhanced curriculum compared to MPM only, on lifetime (p≤0.05) and recent (p≤0.05) alcohol use.

6.3.3.2 Medium-term results (up to 12 months)

One study (Jorgensen et al., 1993), which examined the effects of Project Taking Charge reported medium-term follow-up results.

Knowledge and understanding

Jorgensen et al (1993) reported that Project Taking Charge had positive effects on knowledge of sexual development (p<0.05) and sexually anatomy (p<0.001), along with knowledge about STIs

(p<0.05). No medium-term programme effects were found for knowledge about pregnancy complications.

Attitudes and values

There were no effects of Project Taking Charge (Jorgensen et al., 1993) on the following measures related to attitudes and values: sexual values, attitudes or intentions and educational aspirations.

Personal and social skills

There were no effects of Project Taking Charge (Jorgensen et al., 1993) on participants' self-esteem or communication with parents.

Health and social outcomes relating to alcohol use and sexual health

Jorgensen et al (1993) examined whether participation in Project Taking Charge was associated with delays in the initiation of sex intercourse. Among participants who were sexually inexperienced at baseline, intervention students were less likely than control students to report the initiation of sexual activity at the 6-month follow-up (p=0.05). However, the authors caution that the sample size for analysis was small (n=50) and that the results should be interpreted with caution.

6.3.3.3 Long-term results (>12 months)

Two studies (Denny and Young, 2006; Trenholm et al., 2008), reporting on evaluations of three programmes (Sex Can Wait, MCMF and RTV), reported long-term follow-up data. Trenholm et al (2008) described follow-up at 42-78 months past baseline, equivalent to six months to three and a half years post-test for participants who took part in the three year programme.

Knowledge and understanding

There were positive long-term effects on knowledge reported in both studies (Denny and Young, 2006; Trenholm et al., 2008). Middle and high school students who participated in Sex Can Wait (Denny and Young, 2006) demonstrated significantly better knowledge of topics covered in the curriculum than controls (p<0.001), an outcome that had been non-significant at previous post-test evaluations (Denny et al., 1999; Denny and Young, 2006). Outcomes for knowledge about sexually transmitted infections (p<0.001) and the risks of having unprotected sex (p<0.01) were significantly greater amongst students in MCMF (Trenholm et al., 2008) than controls. However, there were no significant differences in knowledge for RTV students and controls.

Attitudes and values

Denny and Young (2006) reported on the long-term effects of Sex Can Wait on students' attitudes and found no significant programme effects on middle school students' intentions to remain abstinent. For high school students however, intervention students remained significantly more likely to demonstrate favourable intentions (p<0.05) as they had done at post-test.

Personal and social skills

None of the studies examined long-term programme effects on personal and social skills.

Health and social outcomes relating to alcohol use and sexual health

Both studies reported long-term intervention effects on sexual behaviours. At the 18-month follow-up, middle school students who participated in Sex Can Wait (Denny and Young, 2006) were significantly less likely have initiated sexual activity (p<0.05) and were less likely to report sexual intercourse in the last 30 days (p<0.001), compared to control students. For high school students however there were no significant differences between intervention and control students on either of these measures (Denny & Young, 2006). This is in contrast to the immediate post-test results reported in both Sex Can Wait evaluations which showed positive behavioural outcomes for high school, but not middle school, students. No significant effects on student abstinence at six months to three and a half years were reported following MCMF or RTV (Trenholm et al., 2008) and neither of these interventions demonstrated any effects on pregnancy rates or condom use.

6.3.4 Summary and evidence statements

The nine studies evaluating abstinence only programmes were similar in population, including mainly younger adolescents aged 12-14 years. Three studies (Denny et al., 1999; Denny and Young, 2006; Jorgensen et al., 1993) included older populations of adolescents (aged >14 years)

6.3.4.1 Knowledge and understanding

Knowledge outcomes were reported in six studies (Blake et al., 2001; Borawski et al., 2005; Denny et al., 1999; Denny and Young, 2006; Jorgensen et al., 1993; Trenholm et al., 2008). The most commonly reported knowledge outcome was concerning knowledge of STIs, for which two studies (Borawski et al., 2005; Jorgensen et al., 1993) reported positive short- and medium-term programme effects, respectively. A third study, Trenholm et al (2008), evaluated two programmes and found that only MCMF and not RTV affected participants' long-term STI knowledge. Neither study evaluating Sex Can Wait (Denny et al., 1999; Denny and Young, 2006) found short-term differences between programme and control high and middle school students on knowledge of topics covered in the curriculum, but Denny and Young (2006) reported positive long-term effects amongst high school students only. There was no difference in knowledge relating to abstinence among students who received an enhanced version of the MPM compared to those who received the curriculum only version of the programme (Blake et al., 2001).

6.3.4.2 Attitudes and values

Eight studies examined attitudes and values, and abstinence-only programmes were generally found to have had a positive effect on participants' beliefs and attitudes towards abstinence. Seven studies (Blake et al., 2001; Borawski et al., 2005; Christopher and Roosa, 1990; Denny et al., 1999; Denny and Young, 2006; Jorgensen et al., 1993; Roosa and Christopher, 1990) reported on attitudes towards abstinence or around age of first sex and found mostly positive programme effects. The Success Express programme (Christopher and Roosa, 1990; Roosa and Christopher, 1990) reported mainly non-significant programme effects on attitudes. Borawski et al (2005) reported finding positive effects of For Keeps on intentions to remain abstinent until marriage, until older and in the next three months and both evaluations of Sex Can Wait found positive programme effects on abstinence

intentions for high school, but not middle school students. Jorgensen et al (1993) however reported no effect of Project Taking Charge on intentions to have sex. No programmes reported an adverse programme effect on attitudes towards abstinence or intentions to have sex.

6.3.4.3 Personal and social skills

Eight studies (Blake et al., 2001; Borawski et al., 2005; Christopher and Roosa, 1990; Denny et al., 1999; Denny and Young, 2006; Donnelly et al., 2001; Jorgensen et al., 1993; Roosa and Christopher, 1990) examined outcomes relating to personal and social skills and generally reported no significant differences between intervention and control participants at follow-up. Just one study (Blake et al., 2001) found positive programme effects for skills outcomes: reporting students who had received the MPM programme had more favourable results for parent-child communication outcomes. Results from the other studies indicated that abstinence-only programmes did not affect communication.

6.3.4.4 Health and social outcomes relating to alcohol use and sexual health

Eight studies (Christopher & Roosa, 1990; Blake et al., 2001; Borawski et al., 2005; Denny et al., 1999: Denny & Young, 2006: Jorgensen et al., 1993; Roosa & Christopher, 1990; Trenholm et al., 2008) examined effects on health outcomes related to sexual health and two studies (Blake et al., 2001; Donnelly et al., 2001) examined effects on health outcomes related to substance use. These studies demonstrated that abstinence-only programmes generally had no significant effects on the initiation of sexual activity, with the exception of the Sex Can Wait programme (Denny et al., 1999; Denny & Young, 2006). Two studies of Sex Can Wait (Denny et al., 1999; Denny and Young, 2006) reported positive short-term programme effects on whether high school, but not middle school students, reported sexual intercourse, ever or in the last 30 days. However at long-term follow-up, 18months after intervention, Denny and Young (2006) found positive programme effects for middle school, but not high school students, on these measures. Jorgensen et al (1993) reported a positive impact of Project Taking Charge on the initiation of sexual activity, but as the sample size for the analyses was small the results are unlikely to be generalisable beyond the study. Based on short-term follow-up, Blake et al (2001) found no differences in sexual behaviour among students who received an enhanced version of the MPM programme compared to the curriculum only version. Borawski et al (2005) also reported a lack of programme effects of the For Keeps on recent sexual intercourse at short-term follow-up, but noted reductions in the frequency of sexual intercourse, number of sexual partners and multiple episodes of sex. Only one abstinence-only programme, Success Express (Christopher and Roosa, 1990), had a negative effect on health outcomes. Intervention students reported greater lifetime sexual experience than controls. However, in a replication study of this programme (Roosa & Christopher, 1990) there was no difference between intervention and control students on this measure. No significant long-differences differences between programme and control participants were reported for number of pregnancies following the MCMF and RTV programmes (Trenholm et al., 2008). Two studies (Borawski et al., 2005; Trenholm et al., 2008) reported outcomes relating to contraception and reported that abstinence programme participants were no more or less likely to use contraception than controls.

Two studies (Donnelly et al., 2001; Blake et al., 2001) examined the effects of abstinence only programmes and alcohol and other substance use. There was no impact of the Project CARE intervention (Donnelly et al., 2001), but students who received the enhanced version of the MPM programme reported less lifetime and recent alcohol use than students who received the curriculum only version of the programme.

Evidence statement 10

- 10 (e) There is moderate evidence from one RCT and two NRCTs¹ to suggest that abstinence only programmes may have positive short- to long-term effects on knowledge relating to STIs. This evidence may only be partially applicable because the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- 10 (f) There is weak evidence from four NRCTs² to suggest that abstinence only programmes may have short-term positive effects on attitudes and intentions towards sexual activity. There is weak evidence from one NRCT³ to judge the impact of abstinence programs on long-term intentions. This evidence may only be partially applicable because the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- 10 (g) There is moderate evidence from four NRCTs and two RCTs⁴ to suggest that abstinence only programmes may have no impact on the initiation of sexual behaviours or the maintenance of sexual abstinence. In addition, there is moderate evidence from one RCT and three NRCTs⁵ to suggest that abstinence only programmes may have no impact on or increase sexual activity. This evidence may only be partially applicable because the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- 10 (h) There is weak evidence from one RCT⁶ to suggest that abstinence only education programmes may have no impact on long-term pregnancy rates and contraception use. This evidence may only be partially applicable because the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- ¹ Borawski et al., 2005 (NRCT +); Jorgensen et al, 1993 (NRCT +); Trenholm et al., 2008 (RCT -)
- ² Borawski et al., 2005 (NRCT +); Denny et al., 1999 (NRCT -); Denny and Young, 2006 (NRCT -); Roosa and Christopher, 1990 (NRCT -)
- ³ Denny et al., 2006 (NRCT -)
- ⁴ Blake et al., 2001 (RCT +); Borawski et al., 2005 (NRCT +); Denny et al., 1999 (NRCT -); Jorgensen et al, 1993 (NRCT +); Roosa and Christopher, 1990 (NRCT -); Trenholm et al., 2008 (RCT -)
- ⁵ Blake et al., 2001 (RCT +); Borawski et al., 2005 (NRCT +); Denny et al., 1999 (NRCT -); Denny and Young, 2006 (NRCT -)
- ⁶ Trenholm et al., 2008 (RCT -)

Table 6 10	Abstingncg-only	programmes: s	hort-term programn	na offects on knowledge	attitudes and skills
	Abstillence-only	programmes. s	non-term programm	ie eliects on knowledge	, attitudes and skins

Study	Poting	Intervention	Comparator	Follow-up	Outcomes					
Study	Nating				Knowledge	Attitudes and values	Skills			
Blake et al., 2001	RCT +	MPM curriculum + homework n=190	MPM curriculum only n=161	1 week	NS abstinence	 ✓ belief that should expect sex if had sex before* ✓ belief would have sex before finishing high school** ↑ sexual refusal efficacy** ↑ substance use refusal efficacy*** 	 ↑ frequency of parent-child communication ↑ parent-child communication about prevention strategies** ↑ parent-child communication about consequences of sex* ↑ parent-child communication about school*** NS parent-child communication about puberty and sexual expectations 			
Borawski et al., 2005	NRCT +	For Keeps n=1,096	NR n=973	16-25 weeks (66%)	↑ HIV/STDs***	 ↑ belief in abstinence until older ** ↑ belief in abstinence until marriage*** ↓ intentions to have sex, next 3 months* ↓ intentions to have sex, next year** ↓ intentions to use condoms*** NS condom-use efficacy NS sexual impulse control 	-			
Christopher and Roosa, 1990	NRCT -	Success Express n=191	No intervention n=129	PT	-	NS perceived: best age for first sex; age expected for first sex; best age for marriage; lifetime or friends lifetime sexual involvement NS self-esteem	NS family communication			
Denny et al., 1999	NPCT	Sex Can Wait Middle school, n=287	Regular curriculum n=320	РТ	NS curriculum	NS attitudes NS hopelessness NS abstinence intentions NS self-efficacy NS decision making	-			
	NKCI -	Sex Can Wait High school, n=195	Regular curriculum n=92	PT	NS curriculum	 ↑ attitudes** ↑ hopelessness** ↑ abstinence intentions*** NS self-efficacy NS decision making 	-			
Study	Poting	Intervention	Comparator	Follow up	Outcomes					
-----------------------------------	----------------	-----------------------------------------	-------------------------------------	-------------------	---------------------	----------------------------------------------------------------------------------------------------------------------------------------------------------	---------------------------------	--	--	--
Sludy	Rating	intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills			
Denny and Young, 2006	NRCT -	Sex Can Wait Middle school, n=326	n=372	PT (87%)	NS curriculum	NS abstinence intentions NS hopelessness NS self efficacy NS decision making	-			
Denny and Young, 2006	NRCT -	Sex Can Wait High school, n=226	n=111	PT (85%),	NS curriculum	 ↑ beliefs about abstinence** ↑ intention to remain abstinent*** NS self efficacy NS decision making 	-			
Donnelly et al., 2001	NRCT -	Project C.A.R.E. n=413	Regular education n=426	PT	-	NS self-esteem	NS social support from families			
Roosa and Christopher, 1990	NRCT -	Success Express n=339	No intervention n=129	PT	-	 ↑ expected age to have first sex* NS pre-marital sex beliefs NS attitude to abstinence NS self-esteem 	NS family communication			
*p≤0.05; **p≤0.0	01; ***p≤0.001	; 🛧 increase relative	to comparator; $oldsymbol{\psi}$ to	decrease relative	e to comparator; NS	not significant; - outcome not report	ted			

Table 6.11. Abstinence-only programmes: short-term programme effects on health and social outcomes

							Health outcome	S	
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	Conceptions	Substance Use
Blake et al., 2001	RCT +	MPM curriculum + homework n=190	MPM curriculum only n=161	1 week	NS sexual intercourse, ever	NS recent sexual intercourse	-	-	 ↓ lifetime alcohol use* ↓ recent alcohol use
Borawski et al., 2005	NRCT +	For Keeps n=1,096	NR n=973	16-25 weeks (66%)	-	NS recent sexual intercourse ↓ frequency of sexual intercourse* ↓ 2+ sexual partners** ↓ multiple episodes of sex*	NS consistent condom use	-	-

							Health outcome	S	
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	Conceptions	Substance Use
Christopher & Roosa, 1990	NRCT -	Success Express n=191	No intervention n=129	PT	↑ lifetime sexual behaviour*	-	-	-	-
Roosa & Christopher, 1990	NRCT -	Success Express n=339	No intervention n=129	PT	NS lifetime sexual behaviour	-	-	-	-
Denny et al.,	NPCT	Sex Can Wait Middle school, n=287	Regular curriculum n=320	PT	NS sexual intercourse, ever	NS sexual intercourse, last 30 days	-	-	-
1999	NRC1 -	Sex Can Wait High school, n=195	Regular curriculum n=92	PT			-	-	-
Denny &	NRCT -	Sex Can Wait n=326 (Middle School)	n=372	PT (87%)	NS sexual intercourse, ever	NS sexual intercourse, last 30 days	-	-	-
Young, 2006		Sex Can Wait n=226 (High School)	n=111	PT (85%)			-	-	-
Donnelly et al., 2001	NRCT -	Project C.A.R.E. n=413	Regular education n=426	PT	-	-	-	-	 ↓ crack; methamphetamines use* NS alcohol; tobacco; cannabis; inhalants; crack; heroin use
*p≤0.05; **p≤0.	01; ***p≤0.00 [·]	1; 🛧 increase rela	tive to comparat	or; 🗸 decrease re	elative to compara	ator; NS not significar	nt; - outcome not re	eported	

Table 6.12. Abstinence-only programmes: medium-term programme effects on knowledge, attitudes and skills

Study	Pating	Intervention	Comparator	Follow-up		Outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills			
Jorgensen et al, 1993	Jorgensen et al, 1993 NRCT + Project Taking n=52 No intervention n=39 No intervention n=39 No intervention n=39 No intervention NR NR NR NR NS sexual anatomy*** NR NR NS pregnancy complications NS self-esteem NS sexual values NS sexual attitudes and intentions NS educational aspirations NS self-esteem									
*p≤0.05; **p≤0.0	*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported									

Table 6.13. Abstinence-only programmes: medium-term programme effects on health and social outcomes

							Health outcomes	5	
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	Conceptions	Substance Use
Jorgensen et al, 1993	NRCT +	Project Taking Charge n=52	No intervention n=39	6 months NR		-	-	-	
*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported ^a Based on small sample (n=50)									

Table 6.14. Abstinence-only programmes: long-term programme effects on knowledge, attitudes and skills

Study	Poting	Intervention	Comparator	Follow up		Outcomes	
Sludy	Rating	intervention	Comparator	Follow-up Knowledge		Attitudes and values	Skills
Denny and	NPCT	Sex Can Wait Middle school, n=326	n=372	18 months (34%)	NS curriculum	NS abstinence intentions NS hopelessness NS decision making NS self efficacy	-
Young, 2006	NRC1 -Sex Can Wait High school, n=22618 months (72%)↑ curriculum***	↑ curriculum***	NS beliefs about abstinence ↑ intention to remain abstinent* NS decision making NS self efficacy	-			
Trenholm et al., 2008	RCT -	My Choice, My Future! n=286 (FU)	No sex education n=162	42-78 months	↑ STDs*** ↑ unprotected sex risks**	-	-

Study	Pating	Intervention	Comparator	Follow-up		Outcomes			
Study	Kating	intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills		
Trenholm et al., 2008	RCT -	ReCapturing the Vision n=275	No sex education n=205	42-78 months	NS STDs NS unprotected sex risks	-	-		
*p≤0.05; **p≤0.0	*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported								

Table 6.15. Abstinence-only programmes: long-term programme effects on health and social outcomes

							Health outcomes	6	
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	Conceptions	Substance Use
Denny and Young, 2006	NRCT -	Sex Can Wait n=326 (Middle School)	n=372	18 months (34%)	↓ sexual intercourse, ever*	↓ sexual intercourse, last 30 days***	-	-	-
Denny and Young, 2006	NRCT -	Sex Can Wait n=226 (High School)	n=111	18 months (72%)	NS sexual intercourse, ever	NS sexual activity, last 30 days	-	-	-
Trenholm et	PCT	My Choice, My Future! n=286	No sex education n=162	42-78 months (82%)	NS abstinence	-	NS condom use	NS pregnancies	-
al., 2008	KUT-	ReCapturing the Vision n=275	No sex education n=205	42-78 months (82%)	NS abstinence	-	NS condom use	NS pregnancies	-
*p≤0.05; **p≤0.0	01; ***p <mark>≤0.001</mark>	; 1 increase relat	ve to comparato	or; 🗸 decrease rela	ative to compara	tor; NS not significant	; - outcome not re	ported	

6.4 Abstinence-plus programmes

6.4.1 Overview of evidence identified

A total of 24 articles were identified that reported on the evaluation of 15 abstinence plus programmes across 18 studies. Abstinence plus programmes were defined as those that reported an emphasis on abstinence as the safest way to avoid HIV/STI infection and pregnancy, but also promoted safer sex through the use of contraceptives. Four articles (Coyle et al., 1999; 2001; Basen-Enquist et al., 2001; Kirby et al., 2004) reported on evaluations of the Safer Choices programme across the same sample of students and were grouped as a single study (Coyle et al., 1999). Three articles (Aten et al., 2002; Siegel et al., 1998; 2001) reported on evaluations of the Rochester AIDS Prevention Project for Youth, although both evaluations appeared to report on largely the same sample of students an additional school was included in the study reported on by Siegel et al (2001) and Aten et al (2002).

All 24 studies were conducted in North America, in the USA (n=22) and Canada (n=2). All 15 programmes were primarily delivered in school during scheduled lesson time. Two programmes, Safer Choices (Coyle et al., 1999) and the Youth AIDS Prevention Project (YAPP; Levy et al., 1995), combined school-based curriculums with activities for parents, and two programmes, a sex education programme based on Health Belief Model and Social learning Theory (HBM-SLT; Eisen et al., 1990) and Focus on Kids (Stanton et al., 2006), were designed to be delivered in schools or in the community. Five programmes were solely taught by teachers (an AIDS prevention programme [Walter & Vaughan, 1993]; HBM-SLT curriculum [Eisen et al., 1990], Positive Prevention [LaChausse, 2006], Reducing the Risk [Kirby et al., 1991] and Safer Choices [Coyle et al., 1999; 2001]) or peers (Protection Express [Caron et al., 2004]). Four programmes, Be Proud! Be Responsible! (BPBR; Borawski et al., 2009), a modified version of Reducing the Risk (Zimmerman et al., 2008), the Rochester AIDS Prevention Project for Youth (RAPP; Siegel et al., 1998; 2001), and Skills for Healthy Relationships (Wright, 1998), were taught by teachers in combination with other providers including adult educators, peers and school nurses. Four programmes, Draw the Line/Respect the Line (Coyle et al., 2004), Focus on Kids (Stanton et al., 2005), Postponing Sexual Involvement (Aarons et al., 2000), and YAPP (Levy et al., 1995), were taught by external health educators and for two programmes, Teen Incentives (Smith, 1994) and an STI prevention curriculum (Boyer & Shafer, 1997), it was not clear who the provider was.

The theoretical base for the curriculum or programme was not reported for three programmes: an STI prevention programme (Boyer & Shafer, 1997), Teen Incentives (Smith, 1994), and an AIDS prevention programme (Walter & Vaughan, 1993). Eight programmes (Be Proud! Be Responsible!, Safer Choices, Draw the Line/Respect the Line, Health Belief Model and Social learning Theory curriculum, Reducing the Risk, Positive Prevention, Focus on Kids, and Skills for Healthy Relationships) were based on a combination of theories. The theories applied most often were social cognitive theory (n=7 programmes), social learning theory and the theory of reasoned action (both n=3 programmes).

Table 6.16. Sex and relationships education: abstinence-blus programmes	Table 6.16. Sex and	relationships	education:	abstinence-	plus	programmes
-------------------------------------------------------------------------	---------------------	---------------	------------	-------------	------	------------

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider
Aarons et al., 2000	RCT +	USA n=522 students; mean 12.8 years	School	Postponing Sexual Involvement: Eight lessons over two month in 7 th grade and booster sessions in 8 th grade; reproductive health, postponing sexual involvement peer-led curriculum, voluntary booster sessions covering a range of health issues.	Social cognitive theory	Peer led and health professional
Borawski et al., 2009	RCT +	USA n=1,357 students; mean 15 years	School	Be Proud! Be Responsible : Six, 50-minute modules, booster session; abstinence promotion, information and skill building about safer sex practices.	Social cognitive theory, theory of reasoned action, theory of planned behaviour	Predominantly, health education teachers, but in three pairs of schools, approximately a quarter of the classes were taught by school nurses
Boyer & Shafer, 1997	NRCT -	USA n=695 students; mean 14.4 years (range 13-17 years)	School	Three sessions over three days on STI prevention; didactic knowledge and skills building sessions	Not reported	Not reported
Caron et al., 2004	NRCT +	Canada n=945 junior high school students; n=477 senior high school students	School	Protection Express : HIV/STI prevention; Peer training programme for senior high school students, developed educational presentation on one of five topics: postponing sexual intercourse, communication and assertiveness in relationships, equality in relationships, conditions to a healthy relationship, and condom use. Peer presentations subsequently presented to junior students	Social cognitive theory	Peer led
Coyle et al., 1999; 2001	RCT +	USA n=4,310 9 th grade students	School, family	Safer Choices : 20 sessions over 2 school years (10 sessions per year); HIV/STI and pregnancy prevention curriculum, role playing, role model stories, parent newsletters, homework assignments, school-community linkages	Social cognitive theory, social influence theory	Teachers
Coyle et al., 2004	RCT +	USA n=2,829 students; mean 11.5 years	School	Draw the Line/Respect the Line: 20 sessions; HIV/AIDS and pregnancy prevention, limit setting and refusal skills	Social cognitive theory, social inoculation theory	Teachers
Eisen et al., 1990; 1992	RCT +	USA n=1,444 students; 13-19 years	School or community	8-12 hour sex education programme based on Health Belief Model and Social learning Theory; information, discussion of emotions, decision-making, and personal responsibility.	Health belief model, social learning theory	Teachers

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider
Hubbard et al., 1998	NRCT +	USA n=532 students; grades 9-12	School	Reducing the Risk: Sixteen lesson sexuality curriculum; skill development, setting behavioural goals, activities to personalise information on the risks of unprotected sex, support for personal values and groups norms against unprotected intercourse.	Social learning theory, social influence theory	Teachers
Kirby et al., 1991	NRCT +	USA n=1,033 students; mean 15.3 years	School	Reducing the Risk: 15 classes over one school year; abstinence and safer sex promotion; promotion of parent/child discussions.	Social learning theory, social inoculation theory, cognitive behaviour theory	Teachers
LaChausse, 2006	RCT -	USA n=353 9 th grade students	School	Positive Prevention : Six lessons, 45 minutes each; interactive activities regarding HIV/STIs, risks or early sexual involvement, resisting social pressures for sexual involvement.	Social learning theory, cognitive behavioural theory	Teachers
Levy et al., 1995; Weeks et al 1995	RCT +	USA n=2,392 7 th grade students	School, family	Youth AIDS Prevention Project: 10 sessions in 7 th grade (1 day/2 weeks) and 5 additional sessions (over 1 week) in 8 th grade; HIV/AIDS, pregnancy and STI prevention, decision-making skills and resistance/negotiation skills, homework assignment, orientation meeting for parents.	Social cognitive theory	Master's level health educators
Siegel et al., 1998	NRCT +	USA n=3,696 students; mean 13 years (middle school students); mean 17 years (high school students)	School	Rochester AIDS Prevention Project for Youth: 12 sessions for middle school students (in 7 th grade), 12 sessions for high school students (10 th , 11 th or 12 th grade); Curriculum emphasised self-esteem and decision-making, discussion and skills-based activities on sexuality, STIs, pregnancy and HIV/AIDS.	Theory of reasoned action	Teacher, Adult Health Educator or Peer led
Siegel et al., 2001; Aten et al., 2002	NRCT +	USA n=4,001 students; mean 13 years (middle school students); mean 17 years (high school students)	School	Rochester AIDS Prevention Project for Youth : See Siegel et al., 1998 for intervention details	Theory of reasoned action	Teacher, Adult Health Educator or Peer led
Smith, 1994	RCT -	USA n=120 students; mean 15.1 years	School	Teen Incentives Program : Once weekly sessions for eight weeks; interpersonal skills training, career mentorship, role playing, writing and acting out skits	Not reported	Not reported

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider
Stanton et al., 2006	RCT +	USA n=1,131 students; 12-16 years	School or community	Focus on Kids : Eight sessions, 1½ hours each, 1-2 day community; decision-making, goal setting, communication, negotiating, consensual relationships, and information regarding abstinence, safer sex, drugs, alcohol and drug selling.	Social cognitive theory, protection motivation theory	Sex education specialists
Walter & Vaughan, 1993	NRCT +	USA n=1,201 students; mean 15.7 years	School	Six session AIDS prevention programme; facts about AIDS transmission and prevention, correcting misrepresentations regarding AIDS risk behaviours, negotiation skills associated with delaying sex and condom use, knowledge and skills around obtaining and using condoms.	Not reported	Teachers
Wright 1998	NRCT +	Canada n=4,512 students; 13-16 years	School	Skills for Healthy Relationships : Twenty hours of skills building activities (31 activities); knowledge acquisition, skills development, motivational supports, attitudes development.	Theory of reasoned action, theory of planned behaviour, self- efficacy theory	Teachers, Peer leaders
Zimmerman et al., 2008	RCT +	USA n=1,944 students; 13-16 years	School	Reducing the Risk: 16-17 sessions; modified RTR curriculum to include videos with music, peer facilitation, role playing, games and prizes	Not reported	Teacher, Peer led

The overall number of students recruited to participate in the included studies ranged from 120 students to over 4,500 students. Evaluation of six programmes (Reducing the Risk, Youth AID Prevention Project, Draw the Line/Respect the Line, RAPP, Safer Choices and Skills for Healthy Relationships) was based on a sample size of more than 1,500 students. Power calculations or mention of sample size needed to detect significant effect was only discussed in four studies (Caron et al., 2004; Coyle et al 2004; LaChausse, 2006; Walter & Vaughan, 1993).

The programmes examined in the included studies targeted a range of age groups and school year levels. Three programmes, Draw the Line/Respect the Line, Postponing Sexual Involvement and YAPP, targeted students aged between 11 and 13 years. Six programmes (RAPP, Focus on Kids, Protection Express, Skills for Healthy Relationships, an unnamed STI prevention programme and the HBM-SLT curriculum) targeted students across a range of ages, from approximately 12-13 years up to age 16-17, and six programmes (Safer Choices, Positive Prevention, Teen Incentives Programme, Be Proud! Be Responsible!, an unnamed AIDS prevention programme, Reducing the Risk) targeted older adolescents (>14 years).

Programmes were evaluated over a range of follow-up periods. Three studies (Boyer & Shafer, 1997; Smith, 1994; Levy et al., 1995) reported immediate post-test results only, and the majority of studies reported follow-up periods of less than 12 months. Evaluations of three programmes, Reducing the Risk (Hubbard et al., 1998; Kirby et al., 1991), Skills for Healthy Relationships (Wright, 1998) and Safer Choices (Coyle et al., 2001), were based on long-term follow-up greater than 12 months, at 18-, 24- and 31-months respectively.

6.4.2 Quality assessment

Of the 18 studies identified, ten were RCTs and eight were NRCTs. Nine RCTs (Aarons et al., 2000; Borawski et al., 2009; Coyle et al., 1999; 2004; Eisen et al., 1990; LaChausse, 2006; Levy et al., 1995; Stanton et al., 2006; Zimmerman et al., 2008) were based on cluster randomisation and one study involved individual randomisation of students (Smith, 1994). Cluster randomisation was conducted at school district (Levy et al., 1995), school (Aarons et al., 2000; Borawski et al., 2009; Coyle et al., 1999; 2004; Zimmerman et al., 2008) and classroom level (Eisen et al., 1990; LaChausse, 2006; Stanton et al., 2006). For five of the cluster RCTs (Aarons et al., 2000; Eisen et al., 1990; LaChausse, 2006; Levy et al., 1995; Stanton et al., 2006) the unit of analysis was not matched to the unit of randomisation or authors did not report methods for adjusting for intraclass correlations. Study quality was rated moderate (+ rating) across the majority of the studies and only one study, an RCT, was rated poor (- rating). Although all of the RCTs were reported as randomised, no further details were reported about the method of randomisation, with the exception of Borawski et al (2009), so it was not possible to judge whether selection bias had been adequately minimised. Borawski et al (2009) reported using a two-stage, double-blinded randomisation procedure, and was rated good quality (++ rating) on this particular design aspect. Details on outcomes were generally well reported across the included RCTs and NRCTs. Outcomes measures were reported to be reliable in seven RCTs (Borawski et al., 2009; Coyle et al., 1999; 2004; Eisen et al., 1990; LaChausse, 2006; Stanton et al., 2006; Zimmerman et al., 2008) and three NRCTs (Boyer & Schaffer, 1997; Caron et al., 2004; Wright,

1998), and relevant outcomes were judged to have been reported across all studies. The quality of the analyses conducted was also generally good, but only one RCT (Stanton et al., 2006) reported that an intention-to-treat analysis had been undertaken. The majority of studies based their analysis on students who had completed both baseline and follow-up surveys. One study, RCT (Smith, 1994), was rated poor quality (- rating) because the reporting quality of the study was poor and it was difficult to interpret the statistical results presented.

6.4.3 Findings

6.4.3.1 Short-term results (<6 months)

Eleven studies (Aarons et al., 2000; Borawski et al., 2009; Boyer & Shafer, 1997; Coyle et al., 1999; Eisen et al., 1990; Levy et al., 1995; Siegel et al., 1998; Smith, 1994; Stanton et al., 2006; Walter & Vaughan, 1993; Wright, 1998) reported short-term data on the effects of 11 programmes: a postponing sexual involvement curriculum, BPBR, an STI/HIV prevention curriculum, Safer Choices, HBM-SLT curriculum, YAPP, RAPP, Teen Incentives, Focus on Kids, an AIDS-preventive curriculum, and Skills for Healthy Relationships.

Knowledge and understanding

Nine studies (Aarons et al., 2000; Borawski et al., 2009; Boyer & Shafer, 1997; Coyle et al., 1999; Eisen et al., 1990; LaChausse, 2006; Siegel et al., 1998; Walter & Vaughan, 1993; Wright, 1998) examined short-term intervention effects on sexual health knowledge, five of which reported positive effects. Aarons et al (2006) found that the Postponing Sexual Involvement curriculum was only effective at increasing contraceptive knowledge among male intervention students, and had no short-term effects on knowledge related to reproductive sexual services among males or females.

Attitudes and values

Ten studies (Aarons et al., 2000; Borawski et al., 2009; Coyle et al., 1999; Eisen et al., 1990; Levy et al., 1995; LaChausse, 2006; Siegel et al., 1998; Stanton et al., 2006; Walter & Vaughan, 1993; Wright, 1998) examined short-term intervention effects on attitudes and values.

Six studies (Aarons et al., 2006; Borawski et al., 2009; Levy et al., 1995; Siegel et al., 1998; Stanton et al., 2006; Wright, 1998) reported on changes in behavioural intentions. Four studies (Aarons et al., 2006; Borawski et al., 2009; Levy et al., 1995; Stanton et al., 2006) found no intervention effects on behavioural intentions, including intentions to have sex and intentions for condom use, for four programmes (a postponing sexual involvement curriculum, BPBR, YAPP and Focus on Kids). Female students who received the Postponing Sexual Involvement curriculum (Aarons et al., 2000) reported greater intentions to not have sex in the next six months, compared to control students at the end of seventh grade. However, less than six months later at the beginning of eighth grade, there was no difference in intentions between male or female intervention students and controls. Although Levy et al (1995) found that students in the intervention and control groups did not differ in their intentions to have sex or use condoms; intervention students were more likely to consider using condoms with foam if they planned on being sexually active in the next 12 months. For students who participated in

the RAPP (Siegel et al., 1998), only high school intervention students demonstrated significantly greater behavioural intentions than control students. In the study by Wright (1998), there were significantly higher levels of intentions to engage in preventive behaviours in the intervention group compared to the control group (p<0.0001).

Seven studies (Aarons et al., 2006; Borawski et al., 2009; Coyle et al., 1999; Eisen et al., 1990; LaChausse, 2006; Walter & Vaughan, 1993; Wright, 1998) examined impact on attitudes and beliefs related to sexual behaviour. There were no significant differences between female intervention students who received the Postponing Sexual Involvement curriculum (Aarons et al., 2006) and control students on attitudes toward delayed initiation of sex or delayed childbearing, but among males, intervention students reported more positive attitudes towards delayed childbearing, but not delayed sex, at the end of seventh grade and the beginning of eighth grade. Males and females who participated in Be Proud! Be Responsible! (Borawski et al., 2009) reported significantly higher condom use prevention beliefs, but there were no intervention effects on other beliefs related to condom use, whether condom use was hedonistic, and towards abstinence. Coyle et al (1999) reported short-term intervention effects of the Safer Choices programme on students' attitudes toward condom use, but there were no intervention effects on attitudes about sex. Intervention students expressed more positive attitudes toward condom use and decreased barriers toward condom use, higher self-efficacy for condom use, and higher levels of risk perception than control students. Eisen et al (1990) found that there was no difference in health beliefs between intervention and control students following delivery of a curriculum based on the Health Belief Model and Social Learning Theory. At the 1-month post-test, there were no effects of the Positive Prevention curriculum (LaChausse, 2006) on attitudes towards abstention, or self-efficacy for abstinence or condom use. Walter and Vaughan (1993) found significant improvements on four of five belief outcomes for students who received a special AIDSpreventive curriculum compared to controls: perceived susceptibility of acquiring AIDS (p<0.01); perceived benefits and barriers for engaging in AIDS preventive behaviour (p<0.01 and p<0.05, respectively); and perceptions about the commonness of involvement in AIDS prevention (p<0.01). There was no difference in participants' perceptions about the acceptability of involvement of AIDS preventive behaviour but significant improvements were found among intervention students on the measure of self-efficacy, which assessed participants' degree of certainty regarding their ability to success fully perform AIDS-preventive actions (p<0.01). Intervention group students who received the Skills for Healthy Relationships curriculum (Wright, 1998) reported significantly more positive attitudes towards homosexuals and people living with AIDS/HIV than control group students at the end of the programme (p<0.0001).

Personal and social skills

Three studies (Borawski et al., 2009; Boyer & Shafer, 1997; Wright, 1998) examined short-term intervention effects on personal and social skills related to sexual risk behaviours. Borawski et al (2009) reported a significant difference between male intervention who participated in BPBR, and control students on the following efficacy outcomes: impulse control, condom negotiation skills, and condom technical skills. A knowledge-and cognitive-behavioural skills-building intervention to prevent

STIs (Boyer & Shafer, 1997) had a significant short-term impact on sexual risk prevention skills (p <0.05) and substance use prevention skills (p<0.001). Intervention group students who participated in Skills for Healthy Relationships (Wright, 1998) reported higher levels of prevention skills⁸ than the control group (p<0.0001).

Health and social outcomes relating to alcohol use and sexual health

Six studies reported short-term programme effects on the initiation of sexual involvement. Of these, five studies (Borawski et al., 2009; Coyle et al., 1999; Eisen et al., 1990; Walter & Vaughan, 1993; Wright, 1998) found no intervention effects for the following five programmes: BPBR; Safer Choices; HBM-SLT curriculum; an AIDS-preventive curriculum, Skills for Healthy Relationships. Aarons et al (2000) evaluated an intervention that included reproductive health classes, health risk screening and the Postponing Sexual Involvement curriculum. At the end of the seventh grade, female (but not male) students had significantly higher virginity rates than control students. However, less than six months later at the beginning of eighth grade, there was no difference in virginity rates between intervention and control students of either gender.

Intervention effects on frequency of sexual intercourse and/or number of partners were examined in seven studies (Borawski et al., 2009; Boyer & Shafer, 1997; Coyle et al., 1999; Levy et al., 1995; Smith, 1994; Stanton et al., 2006; Walter & Vaughan, 1993). Four studies (Borawski et al., 2009; Coyle et al., 1999; Levy et al., 1995; Stanton et al., 2006) reported no intervention effects on frequency of sexual intercourse. Smith (1994) reported that students who participated in the Teen Incentives programme decreased their sexual activity (absolute and relative frequency) compared to control students who received written materials on contraception and decision-making. Boyer and Shafer (1997) found no effects of an STI/HIV prevention curriculum on number of sexual partners but Walter and Vaughan (1993) found that an AIDS-preventive curriculum had short-term effects (up to 3 months) on sexual risk behaviours. Compared to control students, students who participated in the curriculum in 9th or 11th grade reported less involvement in sexual intercourse with high-risk sexual partners (p<0.05) and were more likely to be sexually monogamous (p<0.05).

Eight studies (Aarons et al., 2000; Borawski et al., 2009; Boyer & Shafer, 1997; Coyle et al., 1999; Levy et al., 1995; Smith, 1994; Stanton et al., 2006; Walter & Vaughan, 1993) examined intervention effects on contraceptive use; four studies (Borawski et al., 2009; Boyer & Shafer, 1997; Levy et al., 1995; Stanton et al., 2006) found no intervention effects for the following programmes: BPBR, an STI/HIV prevention curriculum, and Focus on Kids. Aarons et al (2000) found that female intervention students who participated in a two-year curriculum reported greater use of contraception at the end of the programme compared to control students, both at the post-test at the end of seventh grade and at the beginning of eighth grade. Among sexually experienced students, those who participated in Safer Choices (Coyle et al., 1999) reported fewer acts of sexual intercourse without a condom in the past 3 months compared to the control group (mean difference [SE] 0.50 [0.31]; p=0.03) and were more likely to have used condoms or protection against pregnancy at last intercourse (condoms: OR 1.91;

⁸ Communication and behavioural skills relevant to prevention

95% CI 1.13, 3.21; other contraception: OR 1.62; 95% CI 1.05, 2.50). Two other studies reported intervention effects on contraceptive use. Smith (1994) reported that students who participated in the Teen Incentives programme increased their use of contraceptives relative to control students (p<0.05) and students who participated in an AIDS-preventive curriculum (Walter and Vaughan, 1993) were more likely to use condoms (p<0.05) than control students.

Coyle et al (1999) also found that there was no difference between students who received Safer Choices and control students in their use of alcohol and other drugs before sex in the past 3 months, and whether they had been tested for HIV or other STIs. Walter and Vaughan (1993) reported that participation in an AIDS-preventive curriculum was associated with a favourable trend for STI incidence, but the difference between intervention and control students did not reach significance.

6.4.3.2 Medium-term results (up to 12 months)

Eleven studies (Aarons et al., 2000; Borawski et al., 2009; Caron et al., 2004; Coyle et al., 2004; Eisen et al., 1990; Kirby et al., 1991; LaChausse, 2006; Siegel et al., 2001; Stanton et al., 2006; Wright, 1998; Zimmerman et al., 2008) reported medium-term follow-up data for 11 programmes: Postponing Sexual Involvement, Be Proud! Be Responsible!, Protection Express, Draw the Line/Respect the Line, HBM-SLT curriculum, Reducing the Risk (standard and modified), Positive Prevention, RAPP, Focus on Kids, and Skills for Healthy Relationship.

Knowledge and understanding

Eight studies (Aarons et al., 2000; Borawski et al., 2009; Coyle et al., 2004; Kirby et al., 1991; LaChausse, 2006; Siegel et al., 2001; Wright, 1998; Zimmerman et al., 2008) examined medium-term programme effects on knowledge. Five studies (Borawski et al., 2009; Coyle et al., 2004; Kirby et al., 1991; Siegel et al., 2001; Wright, 1998) reported that significant increases in knowledge were sustained among intervention students at medium-term follow-up for five programmes: BPBR, Draw the Line/Respect the Line, Reducing the Risk, RAPP, and Skills for Healthy Relationships. Aarons et al (2000) found programme effects on knowledge related to contraceptives among males intervention students only. Furthermore, there were no programme effects on knowledge related to reproductive health services among either gender. Two studies (LaChausse, 2006; Zimmerman et al., 2008) reported no medium-term intervention effects on knowledge.

Attitudes and values

Nine studies (Aarons et al., 2000; Caron et al., 2004; Coyle et al., 2004; Kirby et al., 1991; LaChausse, 2006; Siegel et al., 2001; Stanton et al., 2006; Wright, 1998; Zimmerman et al., 2008) examined medium-term programme effects on attitudes.

Six studies (Aarons et al., 2000; Caron et al., 2004; Siegel et al., 2001; Stanton et al., 2006; Wright, 1998) examined programme impact on behavioural intentions. There were no medium-term effects of the Postponing Sexual Involvement curriculum (Aarons et al., 2000) on intentions to not have sex in the next six months among females and among males, intentions to not have sex were significantly lower among intervention students (OR 0.32; 95% CI 0.13, 0.80). Junior high students who participated in Protection Express were more likely than control students to report intentions to remain

abstinent and to use condoms (both p<0.001). However, among senior high students who participated in the programme as peer leaders there were no intervention effects on behavioural intentions. Among middle school who participated in RAPP (Siegel et al., 2001), intervention students reported safer behavioural intentions regarding sexual behaviour and substance use than controls (females, p<0.05; males, p<0.01), but there was no difference among high school students who received the programme. Stanton et al (2006) found no differences in intentions to engage in sex among students who participated in Focus on Kids and control students who received an environmental health intervention. Compared to the control group, students who participated in the Skills for Healthy Relationships programme (Wright, 1998) reported significantly higher levels of intentions to engage in preventive behaviours in the intervention group.

Nine studies (Aarons et al., 2000; Caron et al., 2004; Coyle et al., 2004; Kirby et al., 1991; LaChausse, 2006; Siegel et al., 2001; Stanton et al., 2006; Zimmerman et al., 2008) examined medium-term intervention effects on attitudes, beliefs and/or self-efficacy. Following delivery of booster sessions in the eighth grade, there were no medium-term effects of the Postponing Sexual Involvement curriculum on attitudes to delaying sex or childbearing, or self-efficacy to refuse sex. Caron et al (2004) reported that the Protection Express curriculum had positive effects on junior high intervention students' attitudes to abstinence, and role beliefs and perceived self-efficacy for abstention and condom use. Senior high students, who acted as peer leaders, also reported changes in attitudes and beliefs in relation to abstinence and condom use after participation in the programme relative to the control group. Coyle et al (2004) found that boys who received the Draw the Line/Respect the Line programme had more positive attitudes toward not having sex than control students (p<0.01), however two studies (LaChausse, 2006; Zimmerman et al., 2008) reported no programme effects on attitudes towards delaying sexual involvement for Positive Prevention and Reducing the Risk, respectively. Students who participated in Skills for Healthy Relationships (Wright, 1998) reported significantly more positive attitudes towards homosexuals and people living with AIDS/HIV than control group students. Of three studies (Coyle et al., 2004; Kirby et al., 1991; Zimmerman et al., 2008) that examined students' perceptions or beliefs toward peer norms for sexual involvement, significant positive programme effects were reported in two studies (Coyle et al., 2004; Kirby et al., 1991) for Draw the Line/Respect the Line and the original evaluation of Reducing the Risk, respectively, but not for a modified version of Reducing the Risk (Zimmerman et al., 2008). Coyle et al (2004) also found that male students who participated in Draw the Line/Respect the Line had stronger sexual limits (p<0.01), and were less likely to place themselves in situations that could lead to sexual behaviours (p<0.001), than control students. Four studies (LaChausse, 2006; Siegel et al., 2001; Stanton et al., 2006; Zimmerman et al., 2008) examined intervention effects on self-efficacy. LaChausse (2006) reported that there were no significant differences between intervention and control students in selfefficacy to abstain from sexual intercourse, but that intervention students reported a higher selfefficacy to use condoms (mean [SD]: 7.41 [0.90] vs. 6.74 [1.26]; p=0.001). In the study by Siegel et al (2001), compared to control students, self-efficacy was significantly higher among middle school and high school female students who participated in the RAPP (p<0.05 and p<0.01, respectively). Stanton et al (2006) reported higher perceptions of self efficacy and response efficacy regarding abstinence

among participants in Focus on Kids compared to control students. Intervention students also demonstrated significantly higher perceptions of self efficacy to use condoms. Zimmerman et al (2008) found no significant differences between intervention and control students on three measures of self-efficacy (refusal, condom and situational).

Personal and social skills

Two studies (Kirby et al., 1991; Wright, 1998) examined medium-term programme effects on personal and social skills related to sexual risk reduction. Kirby et al (1991) found that the Reducing the Risk curriculum impacted on student's communication with their parents at 6-months; intervention participants and their parents were more likely than control students to have ever discussed abstinence (p<0.01) and contraception (p<0.01). However, there was no difference between group in their level of communication with their parents about pregnancy and STIs. Students who participated in the Skills for Healthy Relationships curriculum (Wright, 1998) reported higher levels of prevention skills compared to the control group.

Health and social outcomes relating to alcohol use and sexual health

Nine studies examined medium-term programme effects on initiation of sexual intercourse. Four studies (Borawski et al., 2009; Caron et al., 2004; Kirby et al., 1991; Wright, 1998) found that there were no medium-term effects of the following curriculums on initiation of sexual intercourse: BPBR, Protection Express, Reducing the Risk and Skills for Healthy Relationships. Aarons et al (2000) reported that, at the end of 8th grade, female students who had participated in the Postponing Sexual Involvement curriculum had significantly higher virginity rates than control students (OR 1.88; 95% CI 1.02, 3.47). Coyle et al (2004) found that young men, but not young women, who received the Draw the Line/Respect the Line curriculum were significantly less likely than control students to report ever having sex, at the end of 9th grade (age 14-15 years) one year after receiving the intervention. At 6months follow-up, students who received the Positive Prevention curriculum (LaChausse, 2006) were also significantly less likely to have had sex compared to control students (OR 0.19, 95% CI 0.07, 0.51) Based on a composite measure of initiation or onset of sexual experience, Siegel et al (2001) found that rates of initiation were lower among middle school females who received the RAPP curriculum in the medium-term than among control groups (p<0.01). Zimmerman et al (2008) found that when combined, participants who received the traditional or modified version of Reducing the Risk were significantly less like to have initiated sexual activity at the end of 10th grade (age 15-16 years; one year after intervention) than control participants (p<0.05).

Programme effects on frequency of sexual intercourse and/or number of partners were examined in six studies (Borawski et al., 2009; Coyle et al., 2004; Kirby et al., 1991; LaChausse, 2006; Siegel et al., 2001; Stanton et al., 2006). These studies found that none of the programmes examined (BPBR, Draw the Line/Respect the Line, Reducing the Risk, Positive Prevention, RAPP and Focus on Kids) had any effect on these outcomes at the medium-term follow-up.

Nine studies (Aarons et al., 2000; Borawski et al., 2009; Caron et al., 2004; Coyle et al., 2004; Eisen et al., 1990; Kirby et al., 1991; LaChausse, 2006; Stanton et al., 2006; Zimmerman et al., 2008)

examined medium-term programme effects on contraceptive use, of which seven studies (Borawski et al., 2009; Coyle et al., 2004; Eisen et al., 1990; Kirby et al., 1991; LaChausse, 2006; Stanton et al., 2006; Zimmerman et al., 2008) found no effects for the following programmes: BPBR, Draw the Line/Respect the Line, HBM-SLT curriculum, Reducing the Risk (standard and modified), Positive Prevention, and Focus on Kids. Aarons et al (2000) found that female, but not male, intervention students who received the Postponing Sexual Involvement curriculum, in addition to health risk screening and reproductive health classes reported greater use of contraception at the end of eighth grade compared to control students. Caron et al (2004) reported that at 9-months follow-up, senior high intervention students who acted as peers were more likely than control students to report using a condom consistently with their regular or occasional sexual partner (p<0.01). However, junior high intervention and control students did not differ with respect to condom use.

Two studies (Eisen et al., 1990; Kirby et al., 1991) examined the medium-term effects of a HLM-SLT curriculum and Reducing the Risk, respectively, on pregnancy. Neither study identified significant programme effects on this outcome.

6.4.3.3 Long-term results (>12 months)

Four studies (Coyle et al., 2001; Kirby et al., 1991; Hubbard et al., 1998; Wright, 1998) reported longterm follow-up data for three programmes: Safer Choices, Reducing the Risk and Skills for Healthy Relationships.

Knowledge and understanding

Three studies (Coyle et al., 2001; Kirby et al., 1991; Wright, 1998) examined long-term programme effects on sexual health knowledge. After 31-months follow-up, Coyle et al (2001) reported that effects on knowledge of HIV and STDs remained significantly higher for intervention students who participated in Safer Choices compared to control students (mean difference [SE] HIV 0.11 [0.02]; STIs 0.09 [0.02]; both p=0.00). Kirby et al (1991) and Wright (1998) also reported sustained long-term increases in knowledge among students exposed to abstinence plus programmes compared to controls.

Attitudes and values

At 31-months follow-up, Coyle et al (2001) found that Safer Choices students expressed significantly more positive attitudes than comparison students about condoms (mean difference [SE] 0.07 [0.02]; p=.01) and reported significantly fewer barriers to condom use (mean difference [SE] -0.11 [0.04]; p=0.01), greater self-efficacy for condom use (mean difference [SE] 0.11 [0.03]; p=0.00) and perceived HIV and STD risks to be higher (mean difference [SE] HIV 0.11 [0.05]; p<0.05; STIs 0.09 [0.04]; p<0.05). However, there was no difference between intervention and control students on the following measures: attitudes about sex; normative beliefs about condoms; self efficacy for refusing sex; and self efficacy for communication. Despite medium-term effects on beliefs about the sexual involvement of peers, at 18-months follow-up (Kirby et al., 1999) there was no difference between intervention students who participated in Reducing the Risk and control students on this outcome. Wright (1998) reported that both short- and medium-term effects on attitudes and intentions were

maintained at the end of 11th grade. Intervention group students reported significantly more positive attitudes towards homosexuals and people living with AIDS/HIV than control group students, and significantly higher levels of intentions to engage in preventive behaviours in the intervention group compared to the control group.

Personal and social skills

Despite a short-term impact on communication with parents (Coyle et al., 1999), there were no longterm effects of the Safer Choices curriculum (Coyle et al., 2001) on students' level of communication with their parents. Kirby et al (1991) found that medium-term effects on parental communication were sustained long-term for abstinence (p<0.05) but not for contraception. As at previous follow-ups (Kirby et al., 1991), there was no difference between Reducing the Risk and control students in communication with parents about pregnancy and STDs, or in intentions to use skills for avoiding unprotected intercourse. Wright (1998) found that short- and medium-term intervention effects on prevention skills were sustained at the long-term follow-up among students who participated in the Skills for Healthy Relationship programme, compared to controls. However, there was no difference in self-esteem between intervention and control students.

Health and social outcomes relating to alcohol use and sexual health

Two studies (Kirby et al., 1991; Hubbard et al., 1998) demonstrated that the Reducing the Risk curriculum was partially effective at reducing the age of initiation over the long-term. At 18-months, Kirby et al (1991) found that fewer students in the intervention group had initiated intercourse compared to the control group (29% vs. 38%; p<0.05). However, the results of the logistic regression analysis of the proportions initiating intercourse did not reach significance (p=0.13). In a replication of the Reducing the Risk curriculum, Hubbard et al (1998) also found that fewer intervention students than control students were sexually active at 18-months follow-up (28% vs. 43%; p<0.05). However, this study experienced large losses to follow-up (only 36% of students were followed up) and consequently students followed up at 18 months were unlikely to be representative of the baseline sample. For the other programmes with long-term follow-up, Safer Choices and Skills for Healthy Relationships, there were no intervention effects on initiation of sexual intercourse.

Kirby et al (1991) examined the long-term effects of the Reducing the Risk curriculum on frequency of sexual intercourse, finding that there were no statistically significant differences between the intervention and control group. Kirby et al (1991) also found that there was no long-term effects of the Reducing the Risk curriculum on contraceptive use or pregnancy. Hubbard et al (1998) reported that the curriculum had a long-term impact on use of contraception by new initiates, but as previously stated the results of this study were confounded by large losses to follow-up. There appeared to be long-term effects of the Safer Choices curriculum at long-term follow-up. At 31 months, compared to control students, intervention students reported fewer acts of sexual intercourse without a condom in the last 3 months (mean difference [SE] 0.63 [0.23]; p=0.05) and fewer sexual partners without a condom in the last 3 months (mean difference [SE] 0.73 [0.14]; p=0.02). Intervention students were also significantly more likely to have used condoms (OR 1.68; 95% CI 1.02, 2.76) or other pregnancy prevention methods at last intercourse (OR 1.76; 95% CI 1.01, 30.7).

6.4.4 Summary and evidence statements

A total of 18 studies were identified that evaluated 15 sex and relationships education programmes based on an abstinence-plus approach. Abstinence plus programmes were defined as those that reported an emphasis on abstinence as the safest way to avoid HIV/STI infection and pregnancy, but also promoted safer sex through the use of contraceptives. All 15 programmes were developed and evaluated in North America; 13 programmes were developed in the USA and two programmes were developed in Canada. Two programmes, Safer Choices and YAPP, incorporated activities for parents and one programme, Protection Express, was exclusively peer-led. The programmes identified tended to target older adolescents (>14 years) or students across a range of ages, from the age of 13 upwards.

6.4.4.1 Knowledge and understanding

Thirteen studies (Aarons et al 2000; Borawski et al., 2009; Boyer & Shafer, 1997; Coyle et al., 1999; 2001; Coyle et al., 2004; Eisen et al., 1990; Kirby et al., 1991; LaChausse, 2006; Siegel et al., 1998; 2001; Walter & Vaughan, 1993; Wright, 1998; Zimmerman et al., 2008) examined intervention effects on sexual health knowledge. These studies found that abstinence-plus programmes were generally effective in improving sexual health knowledge in the short- and medium-term. In addition, three studies (Coyle et al., 2001; Kirby et al., 1991; Wright, 1998) examined long-term programme effects on knowledge for three programmes, Safer Choices, Reducing the Risk and Skills for Healthy Relationships. All three studies reported sustained long-term increases in knowledge among students exposed to abstinence plus programmes compared to controls.

6.4.4.2 Attitudes and values

Fourteen studies (Aarons et al., 2000; Borawski et al., 2009; Coyle et al., 1999; Coyle et al., 2004; Eisen et al., 1990; Kirby et al., 1991; LaChausse, 2006; Levy et al., 1995; Siegel et al., 1998; 2001; Stanton et al., 2006; Walter & Vaughan, 1993; Wright, 1998; Zimmerman et al., 2008) examined short- and medium-term intervention effects on attitudes and values. The results of these studies demonstrated that intervention effects on behavioural intentions, attitudes to sexual behaviour, and self-efficacy were inconsistent and there was no clear indication of the direction of effects. In addition, some programmes, such as Draw the Line/Respect the Line, had differing effects on male and female students. Two studies (Coyle et al., 2001; Wright, 1998) examined long-term effects on attitudes and values for Safer Choices, Reducing the Risk and Skills for Healthy Relationships, respectively. Safer Choices had long-term positive effects on students' attitudes to and self-efficacy for condom use, but did not impact attitudes or self-efficacy related to sex. Wright (1998) found that positive effects of the Skills for Healthy Relationships programme on attitudes and intentions related to HIV prevention were sustained long-term.

6.4.4.3 Personal and social skills

Six studies (Borawski et al., 2009; Boyer & Shafer, 1997; Coyle et al., 1999; 2001; Kirby et al., 1991; Wright, 1998) examined intervention effects on personal and social skills related to sexual risk behaviours for five programmes that incorporated skills building activities (BPBR, an STI/HIV

prevention curriculum, Safer Choices, Reducing the Risk and Skills for Healthy Relationships). All five programmes had some positive short- or medium-term effects on skills relating to sexual risk prevention. Two studies (Kirby et al., 1991; Coyle et al., 1999; 2001) examined effects on communication with parents for Reducing the Risk and Safer Choices, respectively. Despite a short-term impact on communication with parents (Coyle et al., 1999), there were no long-term effects of the Safer Choices curriculum (Coyle et al., 2001) on students' level of communication with their parents. There were mixed effects of Reducing the Risk on communication, there were medium- to long-term positive effects on students' communication with their parents about abstinence, but positive medium-term effects on communication about contraception were not sustained. There was no difference between Reducing the Risk and control students in communication with parents about pregnancy and STDs, or in intentions to use skills for avoiding unprotected intercourse in the long-term.

6.4.4.4 Health and social outcomes relating to alcohol use and sexual health

All 18 studies examined intervention effects on a range of health outcome measures relating to sexual health. Across these studies there was no indication of short-term programme effects on the initiation of sexual involvement and at the medium-term follow-up, programme effects were largely inconsistent with four studies (Borawski et al., 2009; Caron et al., 2004; Kirby et al., 1991; Wright, 1998) reporting no programme effects on initiation of sexual intercourse. There was some evidence of a positive impact for five studies (Aarons et al., 2000; Coyle et al., 2004; LaChausse, 2006; Siegel et al., 2001; Zimmerman et al., 2008), although for three studies (Aaron et al., 2000; Coyle et al., 2004; Siegel et al., 2001) there were contrasting outcomes among intervention males and females. Two studies (Kirby et al., 1991; Hubbard et al., 1998) demonstrated that the Reducing the Risk curriculum was partially effective at reducing the age of initiation over the long-term, but methodological limitations of the study by Hubbard et al (1998) and lack of statistical significance in logistic regression analyses in the study by Kirby et al (1991) limited the interpretation of this finding. For two further programmes with long-term follow-up, Safer Choices (Coyle et al., 2001) and Skills for Healthy Relationships (Wright, 1998), there were no intervention effects on initiation of sexual intercourse. Programme effects on frequency of sexual intercourse and number of sexual partners were found to be inconsistent or non-existent in the short-, medium- and long-term across 11 studies (Borawski et al., 2009; Boyer & Shafer, 1997; Coyle et al., 1999; 2004; Levy et al., 1995; Kirby et al., 1991; LaChausse, 2006; Siegel et al., 2001; Smith, 1994; Stanton et al., 2006; Walter & Vaughan, 1993).

Short- and medium-term intervention effects on contraceptive use were found to be inconsistent across 14 studies (Aarons et al., 2000; Borawski et al., 2009; Boyer & Shafer, 1997; Caron et al., 2004; Coyle et al., 1999; 2004; Eisen et al., 1990; Kirby et al., 1991; LaChausse, 2006; Levy et al., 1995; Stanton et al., 2006; Smith, 1994; Walter & Vaughan, 1993; Zimmerman et al., 2008). Of these studies, six studies (Aarons et al., 2000; Caron et al., 2004; Coyle et al., 1999; Smith, 1994; Walter & Vaughan, 1993) reported short- to medium-term effects on condom use, with increased use of condoms and contraceptives among intervention students relative to controls. Kirby et al (1991) reported that there was no long-term effects of the Reducing the Risk curriculum on contraceptive use,

but in a replication study of Reducing the Risk, Hubbard et al (1998) reported that the curriculum had a long-term impact on use of contraception by new initiates. However, the results of this study were confounded by large losses to follow-up. There appeared to be long-term effects of the Safer Choices curriculum on contraceptive use. At long-term follow-up, compared to control students, intervention students reported fewer acts of sexual intercourse without a condom and fewer sexual partners without a condom in the last 3 months. Intervention students were also significantly more likely to have used condoms or other pregnancy prevention methods at last intercourse, than controls

Two studies (Eisen et al., 1990; Kirby et al., 1991) examined the medium- to long-term effects of a HLM-SLT curriculum and Reducing the Risk, respectively, on pregnancy. Neither study identified significant programme effects on this outcome.

Evidence statement 11

- 11 (f) There is moderate evidence from three RCTs and three NRCTs¹ to suggest that programmes that emphasise abstinence but that also promote safer sex, may produce short- to long-term term improvements in sexual health-related knowledge. This evidence may only be partially applicable because as these studies were conducted in the USA and may not be generalisable beyond the populations studied. In addition, the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- 11 (g) There is inconsistent evidence from four RCTs and four NRCTs² to determine the effects of abstinence-plus programmes on attitudes, behavioural intentions and self-efficacy relating to sexual behaviour.
- 11 (h) There is inconsistent evidence from one RCT and two NRCTs³ to determine the long-term effects of abstinence-plus programmes on communication with parents. There is moderate evidence from one RCT and one NRCT⁴ to suggest abstinence-plus programmes that incorporate skills building activities may have positive, short and medium- to long-term effects on skills relevant to prevention. This evidence may only be partially applicable because as these studies were conducted in the USA and may not be generalisable beyond the populations studied. In addition, the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- 11 (i) There is moderate evidence from five RCTs and five NRCTs⁵ to suggest that abstinence-plus programmes may not have a consistent short-, medium- or long-term impact on the initiation of sexual activity or the maintenance of abstinence. In addition, there is moderate evidence from four RCTs and two NRCTs⁶ to suggest that abstinence plus programmes may not have an impact on frequency of sexual activity and risky sexual behaviours. This evidence may only be partially applicable because as these studies were conducted in the USA and may not be generalisable beyond the populations studied. In addition, the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.

- 11 (j) There is moderate evidence from six RCTs and three NRCTs⁷ to suggest that abstinence-plus programmes may not have a consistent impact on condom and other contraceptive use, and moderate evidence from one RCT and one NRCT⁸ to suggest that abstinence-plus programmes have no medium- to long-term impact on pregnancy rates. This evidence may only be partially applicable because as these studies were conducted in the USA and may not be generalisable beyond the populations studied. In addition, the programme's emphasis on abstinence is of limited relevance to PSHE delivery in secondary schools focusing on SRE and alcohol education.
- ¹ Borawski et al., 2009 (RCT +); Coyle et al., 2001 (RCT +); Coyle et al., 2004 (RCT +); Kirby et al., 1991 (NRCT +); Siegel et al., 2001 (NRCT +); Wright 1998 (NRCT +)
- ² Borawski et al., 2009 (RCT +); Caron et al., 2004 (NRCT +); Coyle et al., 2004 (RCT +); Kirby et al., 1991 (NRCT +); LaChausse, 2006 (RCT +); Siegel et al., 2001 (NRCT +); Wright, 1998 (NRCT +); Zimmerman et al., 2008 (RCT +)
- ³ Coyle et al., 2001 (RCT +); Hubbard et al., 1998 (NRCT +); Kirby et al., 1991 (NRCT +)
- ⁴ Borawski et al., 2009 (RCT +); Wright, 1998 (NRCT +)
- ⁵ Borawski et al., 2009 (RCT +); Caron et al., 2004 (NRCT +); Coyle et al., 2001 (RCT +); Coyle et al., 2004 (RCT +); Hubbard et al., 1998 (NRCT +); Kirby et al., 1991 (NRCT +); LaChausse, 2006 (RCT +); Siegel et al., 2001 (NRCT +); Wright 1998 (NRCT +); Zimmerman et al., 2008 (RCT +)
- ⁶ Borawski et al., 2009 (RCT +); Coyle et al., 2004 (RCT +); Kirby et al., 1991 (NRCT +); LaChausse, 2006 (RCT +); Siegel et al., 2001 (NRCT +); Stanton et al., 2006 (RCT +)
- ⁷ Caron et al., 2004 (NRCT +); Coyle et al., 2001 (RCT +); Coyle et al., 2004 (RCT +); Eisen et al., 1990 (RCT +); Kirby et al., 1991 (NRCT +); Hubbard et al., 1998 (NRCT +); LaChausse, 2006 (RCT +); Stanton et al., 2006 (RCT +); Zimmerman et al., 2008 (RCT +)
- ³ Eisen et al., 1990 (RCT +); Kirby et al., 1991 (NRCT +)

STI/HIV

prevention

n=210

NRCT +

Boyer &

Shafer, 1997

Didactic

education

n=303

NS beliefs: abstinence

 $\mathbf{\Psi}$ intentions to have sex*

NS intentions to use a

condom NS perceived peer beliefs: sex/condom use

-

Study	Poting	Intervention	Comparator	Follow up		Outcomes	
Sludy	Rating	Intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills
Aarons et al.,	PCT -	Postponing Sexual	No intervention	PT	 ↑ contraception (males only*) NS reproductive health services 	 ✓ intentions to not have sex (females only*) ↑ self-efficacy to refuse sex (females only*) NS attitudes: delayed sex ↑ attitudes: delayed childbearing (boys only*) 	NS parent or boy/girlfriend communication
2000		Involvement n=262	n=260	<6 months	 ↑ contraception (males only*) NS reproductive health services 	NS intentions to not have sex NS self-efficacy to refuse sex NS attitudes: delayed sex ↑ attitudes: delayed childbearing (boys only*)	NS parent or boy/girlfriend communication
				PT	↑ STIs*** ↑ condoms***	 ↑ beliefs: condom use prevention** NS beliefs: condom use NS beliefs condom use hedonistic NS beliefs: abstinence NS intentions to have sex ↑ intentions to use a condom* 	 ↑ condom technical skills*** ↑ impulse control** ↑ condom negotiation skills**
Borawski et al., 2009	RCT +	BPBR n=631	Usual curriculum n=726	4 months		 NS beliefs: condom use prevention ↑ beliefs: condom use NS beliefs condom use hedonistic 	↑ condom negotiation skills**

4-months

(97%)

ΡT

n=513 (74%)

↑ STIs**

↑ condoms***

↑ STIs*

Table 6.17. Abstinence-plus programmes: short-term programme effects on knowledge, attitudes and skills

↑ condom technical

skills**

NS impulse control

↑ sexual risk prevention

skills*

↑ substance use

prevention skills***

Study	Pating	Intervention	Comparator	Follow-up		Outcomes			
Study	Katiliy	Intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills		
Coyle et al., 1999; Coyle et al., 2001	RCT +	Safer Choices n=NR	Knowledge- based HIV prevention curriculum n=NR	End of first year n=3,677 (85%)	↑ HIV*** ↑ other STIs***	 ↑ condom positive attitudes* ↓ condom use barriers*** ↑ risk perception HIV***/other STIs* ↑ self-efficacy: condom use*** NS attitudes about sex NS self-efficacy: refusing sex NS self-efficacy: communication 	↑ parent communication*		
Eisen et al., 1990	RCT +	HBM-SLT curriculum N=722	Community- based programmes N=722	PT N=1,328 (92%)	$m{\Lambda}$ sexual knowledge*	NS health beliefs	-		
LaChausse, 2006	RCT +	Positive Prevention N=216	Usual curriculum N=137	1 month	NS HIV infection and AIDS	NS attitudes towards abstention NS self-efficacy: abstinence NS self-efficacy: condoms	-		
Levy et al., 1995; Weeks et al., 1995	RCT +	YAPP N=1,459	Basic AIDS education N=933	PT N=1,669 (70%)	-	 ↑ consider using condoms with foam*** NS intentions to have sex NS intentions to use condoms 	-		
Siegel et al., 1998	NRCT +	RAPP n=2,437 (1,402 health educator; 1,035 peer educator)	Usual curriculum n=1,259	PT N=2758 (75%)	↑ knowledge ^a **	↑ behavioural intentions (high school only; females*/males**)			

Study	Poting	Intervention	Comparator	Follow-up			
Sludy	Rating	Intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills
Stanton et al., 2006	RCT +	Focus on Kids N=870	Environmental health intervention N=261	3 months N=898 (79%)	-	 NS abstinence: self-efficacy, response efficacy, response costs; extrinsic/intrinsic rewards; severity; vulnerability NS intentions to be abstinent NS condoms: self-efficacy, response efficacy, extrinsic/intrinsic rewards; severity; vulnerability ✓ response costs* NS intentions to be use condoms 	-
Walter & Vaughan, 1993	RCT +	AIDS-preventive curriculum n= 667	No intervention n= 534	3 months	↑ AIDS transmission and prevention***	 ↑perceived susceptibility of acquiring AIDS** ↑perceived benefits for engaging in AIDS preventive behaviour** ↓ perceived barriers for engaging in AIDS preventive behaviour* ↑ perceptions about the commonness of involvement in AIDS prevention** ↑ self-efficacy to perform AIDS-preventive actions** 	-
Wright 1998	NRCT +	Skills for Health Relationships N=2606	Usual curriculum N=1906	PT	↑ HIV/AIDS***	 ↑ positive attitudes towards homosexuals and people living with AIDS/HIV*** ↑ intentions to engage in preventive behaviours*** 	↑ prevention skills*** ↓ self-esteem**
*p≤0.05; **p≤0.0	01; ***p≤0.001; •	↑ increase relative t	to comparator; 🗸 de	crease relative t	to comparator; NS not significa	ant; - outcome not reported	
Significance no	or reported						

Table 6.18. Abstinence-plus programmes: short-term programme effects on health and social outcomes

					Health outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Aarons et al.,	DCT .	Postponing Sexual	No	PT n=503 ^a	↑ virginity (females only)*	-	↑ used contraceptives/ condoms last time had sex (female only)*	-	-
2000	KUT +	Involvement n=262	n=260	<6 months n=564 ^a	NS virginity	-	↑ used contraceptives/ condoms last time had sex (female only)*	-	-
Borawski et al., 2009	RCT +	BPBR N=631	Usual curriculum N=726	PT, 4 months	NS ever had intercourse	NS frequency	NS frequency of unprotected intercourse	-	-
Boyer & Shafer, 1997	NRCT +	STI/HIV prevention N=210	Didactic education N=303	PT n=513 (74%)	-	NS number of sexual partners	NS condom use	-	-
Coyle et al., 1999; Coyle et al., 2001	RCT +	Safer Choices n=NR	Normal education n=NR	End of first year n=3,677 (85%)	NS	 ✓ without a condom (past 3 months)* NS number of times had intercourse (past 3 months) 	 ↑ condom use at first intercourse* ↑ condom use at last intercourse* 	NS tested for HIV/STI	-
Eisen et al., 1990; Eisen et al., 1992	RCT +	HBM-SLT curriculum N=722	Community- based programmes N=722	PT N=1,328 (92%)	NS abstinence maintenance	-	-	-	-
Levy et al., 1995; Weeks et al., 1995	RCT +	YAPP N=1,459	Basic AIDS education N=933	PT N=1,669 (70%)	-	NS sex in last 30 days	 ↑ condoms with foam** NS engaged in protective behaviours 	-	-
Smith, 1994	RCT -	Teen Incentives N=60	Written materials N=60	PT N=95 (79%)	-	↓ frequency*	↑ contraception*	-	-

					Health outcomes						
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions		
Stanton et al., 2006	RCT +	Focus on Kids N=870	Environmental health intervention N=261	3 months	-	NS rates of sexual intercourse (past 6 months)	NS condom use	-	-		
Walter & Vaughan, 1993	RCT +	AIDS-preventive curriculum n= 667	No intervention n= 534	3 months	NS abstinence	 ✓ involvement in sexual intercourse with high-risk partners* ↑ sexual monogamy* 	↑ condom use*	NS STI incidence	-		
Wright 1998	NRCT +	Skills for Health Relationships N=2606	Usual curriculum N=1906	PT	NS sexual experience	-	-	-	-		
*p≤0.05; **p≤0. ^a follow-up base	.01; ***p≤0.00 ed on cross-s)1; ↑ increase relati ectional samples	ve to comparator;	↓ decrease relation	ative to comparator	; NS not significant;	- outcome not report	ed			

Table 6.19. Abstinence-plus programmes: medium-term programme effects on knowledge, attitudes and skills

Study	Dating	Intervention	Comparator	Follow-up	Outcomes			
Sludy	кашу	intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills	
Aarons et al., 2000	RCT +	Postponing Sexual Involvement n=262	No intervention n=260	12-months n=510	 ↑ contraceptives (males only*) NS reproductive health services 	NS intentions to not have sex NS self-efficacy to refuse sex NS attitudes: delayed sex NS attitudes: delayed childbearing	NS parent or boy/girlfriend communication	
Borawski et al., 2009	RCT +	BPBR n=631	Usual curriculum n=726	12-months (92%)	↑ STIs*** ↑ condoms*	NS beliefs: condom use prevention; condom use; condom use hedonistic; abstinence NS intentions to have sex or use a condom NS perceived peer beliefs: sex/condom use	 ↑ condom negotiation skills (male only**) NS impulse control NS condom negotiation skills 	

Study	Poting	Intervention	Comparator	Eollow up	Outcomes				
Sludy	Rating Intervention Comparator Follow		Follow-up	Knowledge	Attitudes and values	Skills			
Caron et al., 2004	NRCT +	Protection Express n=369 junior high	Usual curriculum n=329 junior high	9-months (70%)	-	Postponing sexual intercourse: ↑ intentions*** ↑ attitudes*** ↑ role beliefs*** ↑ perceived self-efficacy*** ↑ anticipated regret NS perceived behavioural control NS personal normative beliefs Condom use: ↑ intentions*** ↑ role beliefs*** NS attitudes NS behavioural control NS personal normative beliefs NS personal normative beliefs NS perceived self-efficacy NS anticipated regret	-		
Caron et al., 2004	NRCT +	Protection Express n=147 senior high	Usual curriculum n=159 senior high	9-months (70%)	-	Postponing sexual intercourse: NS intentions ↑ attitudes*** ↑ behaviour control (direct only)*** ↑ personal normative beliefs** ↑ role beliefs*** ↑ perceived self-efficacy*** Condom use: NS intentions NS attitudes ↑ behaviour control (indirect only)*** NS personal normative beliefs NS anticipated regret NS role beliefs ↑ perceived self-efficacy***	-		
Coyle et al., 2004	RCT +	Draw the Line/Respect the Line N=NR	Usual curriculum N=NR	Until end of ninth grade	↑ HIV and condoms (boys only***)	 ↑ attitudes favouring reasons for not having sex (boys only**) ↓ peer normative beliefs favouring sex (boys only**) ↑ sexual limits (boys only**) ↓ situations that could lead to sexual behaviour (males only***) 	-		

Study	Pating	Intervention	Comparator	Follow-up		Outcomes	Outcomes		
Sludy	кашу	Intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills		
Kirby et al., 1991	NRCT +	Reducing the Risk n= 586	Usual curriculum n= 447	6 months NR	↑ contraceptive knowledge***	↓ belief that peers are having sex**	 ↑ communication with parents: abstinence** ↑ communication with parents: contraception** NS communication with parents: pregnancy NS communication with parents: STIs 		
LaChausse, 2006	RCT +	Positive Prevention N=216	Usual curriculum N=137	6 months	NS HIV infection and AIDS	NS Attitudes toward sexual abstention NS self-efficacy for abstention ↑ self-efficacy to use condoms***	-		
Siegel et al., 2001; Aten et al., 2002	NRCT +	RAPP n=1,404 health educator; 1,020 peer educator; 313 regular health educator	Usual curriculum n= 1264	Mean duration 41.2 weeks	↑ knowledge ^a (middle school only: females***/males**)	 ↑ sex self-efficacy (females only; middle school*/high school**) ↑ 'preferable' behavioural intentions (middle school only: females*/males**) 	-		
Stanton et al., 2006	RCT +	Focus on Kids N=870	Environmental health intervention N=261	6 months N=938 (83%);	-	 ↑ self efficacy to abstain** ↑ response efficacy for abstinence* NS abstinence: response costs; extrinsic/intrinsic rewards; severity; vulnerability NS intentions to be abstinent ↑ self efficacy to use condoms* NS condoms: response efficacy, extrinsic/intrinsic rewards; severity; vulnerability; response costs NS intentions to be use condoms 	-		

Ctudy	Deting	Intervention	Compositor		Outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills		
Stanton et al., 2006	RCT +	Focus on Kids N=870	Environmental health intervention N=261	9 months N=904 (80%)	-	NS abstinence: self-efficacy, response efficacy, response costs; extrinsic/intrinsic rewards; severity; vulnerability NS intentions to be abstinent NS condoms: self-efficacy, response efficacy, extrinsic/intrinsic rewards; severity; vulnerability; response costs NS intentions to be use condoms	-		
Wright 1998	NRCT +	Skills for Health Relationships N=2606	Usual curriculum N=1906	Grade 10	↑ HIV/AIDS***	 ↑ positive attitudes towards homosexuals and people living with AIDS/HIV*** ↑ intentions to engage in preventive behaviours*** 	$m{\Lambda}$ prevention skills**		
Zimmerman et al., 2008	RCT +	Reducing the Risk Standard, n=NR Modified, n=NR	Standard curriculum N=NR	End of 10 th grade (52%)	NS	NS peer sexual activity, attitudes about waiting to have sex, self- efficacy, response to sexual pressure	-		
*p≤0.05; **p≤0.0 ^a human reprodu)1; ***p≤0.001; ∕ uction, decision	increase relative t making, communica	o comparator; $oldsymbol{\psi}$ de tion with others con	crease relative to cerning sexual n	o comparator; NS not sign natters, HIV/AIDS and oth	nificant; - outcome not reported er STIs, high risk behaviours and oth	er sexuality items		

					Health outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Aarons et al., 2000	RCT +	Postponing Sexual Involvement n=262	No intervention n=260	1-year n=510 ^ª	↑ virginity (females only)*	-	↑ used contraceptives /condom last time had sex (female only)*	-	-
Borawski et al., 2009	RCT +	BPBR N=631	Usual curriculum N=726	12 months	NS ever had intercourse	NS frequency	NS frequency of unprotected intercourse	-	-
Caron et al., 2004	NRCT +	Protection Express n=369 junior high, n=147 senior high	Usual curriculum n=329 junior high, n=159 senior high	9-months (70%)	NS postponing sex	-	↑ condom use (senior high only**)	-	-
Coyle et al., 2004	RCT +	Draw the Line/Respect the Line N=NR	Usual curriculum N=NR	Until end of ninth grade	↓ Ever had sex (boys only**)	NS Had sex in the last 12 months NS occasions of sexual intercourse NS number of sexual partners	NS condom use	-	-
Eisen et al., 1990; Eisen et al., 1992	RCT +	HBM-SLT curriculum N=722	Community- based programmes N=722	12 months N=888 (62%)	-	-	NS contraceptive use at first intercourse NS use of effective method at most recent intercourse ^b NS contraceptive efficiency ^b	-	NS pregnancy
Kirby et al., 1991	NRCT +	Reducing the Risk n= 586	Usual curriculum n= 447	6 months NR	NS initiated intercourse	NS frequency	NS contraceptive practice NS unprotected intercourse	-	NS pregnancy

					Health outcomes						
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions		
LaChausse, 2006	RCT +	Positive Prevention N=216	Usual curriculum N=137	6 months N=287 (81%)		NS frequency	NS frequency of condom use	-	-		
Siegel et al., 2001; Aten et al., 2002	NRCT +	RAPP n=1,404 health educator; 1,020 peer educator; 313 regular health educator	Usual curriculum n= 1264	Mean duration 41.2 weeks	 ✓ initiation/ onset sexual experience (middle school females only**) 	NS risky sexual behaviour	-	-	-		
Stanton et al., 2006	RCT +	Focus on Kids N=870	Environmental health intervention N=261	6 and 9 months	-	NS rates of sexual intercourse (past 6 months)	NS condom use	-	-		
Wright 1998	NRCT +	Skills for Health Relationships N=2606	Usual curriculum N=1906	All follow- ups	NS sexual experience	-	-	-	-		
Zimmerman et al., 2008	RCT +	Reducing the Risk Standard, n=NR Modified, n=NR	Standard curriculum N=NR	End of 10 th grade N=1,424	Ψ initiation of sexual activity*	-	NS frequency of condom use NS condom use at last sex	-	-		
*p≤0.05; **p≤0. ^a follow-up bas	.01; ***p≤0.00 ed on cross-s	01; ↑ increase relati sectional samples; ^b	ve to comparator; sexually inexperie	↓ decrease rela nced at baseline	ative to comparator	; NS not significant;	- outcome not report	ed			

Table 6.21. Abstinence-plus programmes: long-term programme effects on knowledge, attitudes and skills

Study	Poting	Intervention	Comparator	Follow up	Outcomes			
Study	Rating	intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills	
Coyle et al., 1999; Coyle et al., 2001	RCT +	Safer Choices n=NR	Knowledge- based HIV prevention curriculum n=NR	31 months n=3,058 (71%)	↑ HIV*** ↑ other STIs***	 ↑ condom positive attitudes** ↓ condom use barriers** ↑ risk perception HIV*/other STIs* ↑ self-efficacy for condom use*** NS attitudes about sex; NS normative beliefs about condoms NS self efficacy: refusing sex NS self efficacy: communication 	NS communication with parents	
Hubbard et al., 1998	NRCT +	Reducing the Risk N=267	Usual curriculum N=265	18 months n=212 (36%)	-	-	 ↑ communication with parents: contraception* ↑ communication with parents: protection from HIV/STIs* NS communication with parents: pregnancy NS communication with parents: abstinence 	
Kirby et al., 1991	NRCT +	Reducing the Risk n= 586	Usual curriculum n= 447	18 months N=758 (73%)	↑ contraceptive knowledge***	NS belief that peers are having sex	 ↑ communication with parents: abstinence* NS communication with parents: contraception NS communication with parents: pregnancy NS communication with parents: STIs 	
Wright 1998	NRCT +	Skills for Health Relationships N=2606 ↑ increase relative t	Usual curriculum N=1906 o comparator: V de	Grade 11	↑ HIV/AIDS*** o comparator: NS not signific	 ↑ positive attitudes towards homosexuals and people living with AIDS/HIV*** ↑ intentions to engage in preventive behaviours** ant: - outcome not reported 	↑ prevention skills*	

Table 6.22. Abstinence-plus programmes: Long-term programme effects on health and social outcomes

	Rating	Intervention	Comparator	Follow-up	Health outcomes				
Study					Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Coyle et al., 2001	RCT +	Safer Choices n=NR	Normal education n=NR	31 months n=3,058 (71%)	NS	↓ without a condom (past 3 months)*	 ↑ condom use at last intercourse* ↑ use of protection at last intercourse* 	NS tested for HIV/STI	-
Kirby et al., 1991	NRCT +	Reducing the Risk n= 586	Usual curriculum n= 447	18 months N=758 (73%)	 ↓ initiated intercourse* (NB: NS on log regressions) 	NS frequency	NS contraceptive practice NS unprotected intercourse	-	NS pregnancy
Hubbard et al., 1998	NRCT +	Reducing the Risk N=267	Usual curriculum N=265	18 months n=212 (36%)		-	↑ use of contraception by new initiates*	-	-
Wright, 1998	NRCT +	Skills for Health Relationships N=2606	Usual curriculum N=1906	Grade 11 (73%; 79%)	NS sexual experience	-	-	-	-
*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported									

6.5 HIV and sexual risk-reduction programmes

6.5.1 Overview of evidence identified

A total of 11 studies reported on school-based HIV prevention and sexual risk-reduction programmes. Studies were defined by their specific focus on HIV prevention and HIV risk-behaviour, sexual risk-behaviour or a combination of both.

Of the 11 studies, six were conducted in North America (Coyle et al., 2006, Fisher et al., 2002, Lemieux et al., 2008, Mitchell-DiCenso et al., 1997, Roberto et al., 2007, Workman et al., 1996) and the remaining five studies were conducted in European countries, including Italy (Borgia et al., 2005), Spain (Traeen, 2003), Norway (Kvalem et al., 1996), Sweden (Larsson et al., 2006), and the Netherlands (Schaalma et al., 1996). All programmes were delivered in school time however one programme (Roberto et al., 2007) also included six computer-based activities to be completed outside of school time. Two programmes were delivered solely by teachers (Safe sex and pregnancy prevention [Traeen, 2003]; AIDS/STI prevention curriculum [Schaalma et al., 1996]). A further two were delivered in combination with teachers; with one also using peer educators (HIV prevention curriculum [Fisher et al., 2002]) and another also using public health nurses and community professionals (McMasters Teen Programme [Mitchell-DiCenso et al., 1997]). Three programmes solely used peer educators (an AIDS and sexuality education programme [Kvalem et al., 1996]; Students Working Against AIDS Together [SWAAT; Lemieux et al., 2008]; and a programme based on social learning theory [Borgia et al., 2005]). One programme used health educators to deliver the intervention (All4You [Coyle et al., 2006]), one used undergraduate psychology students (CBT HIV prevention [Workman et al., 1996]), one used nurses/midwives and medical students (contraception programme [Larsson et al., 2006]) and one study was delivered using the internet (computer-based education [Roberto et al., 2007]).

The theoretical base was not reported for the contraception programme (Larsson et al., 2006). Five programmes were underpinned by a single theoretical base (SLT programme, HIV prevention education, McMasters Teen Programme, Computer-based education, CBT HIV prevention). However, five others used a combination of theories (All4You, Safe sex and pregnancy prevention, AIDS and sexuality education, SWAAT, AIDS/STI prevention curriculum). Overall social learning theory was the most commonly used theory in HIV/sexual risk-reduction programmes (n=5 programmes). Theory of planned behaviour (n=2 programmes), the information-motivation-behavioural skills (IMB) model (n=2 programmes) and the cognitive behavioural model (n=2 programmes) were also applied more often than others.

Table 6.23. Sex and relationship	os education: HIV and sexual	risk-reduction programmes
----------------------------------	------------------------------	---------------------------

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider
Borgia et al., 2005	RCT +	Italy n=1,697 students, median 18 years (high school)	School	5 sessions over 10 hours the aim was to increase knowledge of HIV, address social influences and group norms, improve decision making, communication and negotiation skills and related self-efficacy, place risks related to specific contexts and behaviour in proper dimension and abolish prejudice and stigma towards persons with AIDS.	Social learning theory	Peer educators
Coyle et al., 2006	RCT ++	USA n=988 students, 14- 18 years	School	All4you! : 14 sessions over 26 hours with 9 delivered in classrooms and 5 delivered in visits to volunteer sites. Programme included skills based HIV/STD and pregnancy prevention curriculum; service-learning activities involving visits to volunteer sites.	Social development theory, social cognitive theory, theory of planned behaviour, theory of reasoned action	Health educators
Fisher et al., 2002	NRCT ++	USA n=1,577 students, mean age 14.8 years	School	Combined peer plus curriculum HIV prevention intervention delivered in 5 lessons.	Information- motivation- behavioural skills (IMB) model	Teachers and peer educators
Kvalem et al., 1996	RCT -	Norway n=2,088 high school students, range 16- 20 years	School	AIDS and sexuality education programme focussing on STI and pregnancy prevention delivered in 10-14 hours over a two day period.	Cognitive social learning theory, social influence theory	Peer educators
Larsson et al., 2006	NRCT +	Sweden n=461 students, mean age 17.25 years	School	Educational programme aimed to improve knowledge of, attitudes towards and practices relating to condom and emergency hormonal contraceptive use. The multi- component programme included one 20-minute lesson about the emergency contraceptive pill (nurse/midwife); one session of three 40-minute lessons by educators from the Love Emergency (medical students) within one month after the first lesson focused on attitudes and values towards different contraceptive methods, including rehearsal of condom skills; VIP card for free condoms; telephone number to access individual counselling from nurse/midwife.	Not reported	Nurse/midwife and medical students

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider
Lemieux et al., 2008	CBA -	USA n=422 students, mean age 16 years	School	Students Working Against Aids Together : Music-based HIV prevention intervention. Main components were the creation, recording and distribution of HIV prevention themed music and promotional materials; in-class presentations. Programme took place over five months, four months preparation to create a CD and a month of CD distribution.	Information- motivation- behavioural skills model, natural opinion leader model	Peer educators
Mitchell- DiCenso et al., 1997	RCT +	USA n=2,309 students, mean age 12.6 years (intervention group)	School	McMasters Teen Programme : curriculum-based programme aimed at providing skills training. Ten one hour sessions included accurate information about reproductive system and adolescent development; relationship strategies; emotional communication skills; sexual problem solving skills.	Cognitive behavioural model	Teachers, public health nurses, community professionals
Roberto et al., 2007	RCT +	USA n=402 students, 10 th grade, mean age 15.5 years	School	Computer-based HIV prevention/sexual health programme based on an extended parallel process model. Delivered over 7 weeks and included six computer-based activities completed outside class time.	Extended parallel process model	Computer-based (via internet)
Schaalma et al., 1996	RCT +	Netherlands n=3,142 students, 9 th and 10 th grade	School	AIDS/STI prevention curriculum. Key components of the study focused on Knowledge about AIDS, STDs, transmission, prevention, and risk perception; attitudes toward safe sex in general and condom use; values, social influences, and communication skills regarding the prevention of AIDS and STDs; self-efficacy beliefs regarding negotiating skills and practicing condom use.	Social cognitive theory, communication theories, fear arousing communication, risk perception, social comparison, social/psychological inoculation theories	Teachers
Traeen, 2003	RCT +	Norway n=1,183 students, age 16-17 years	School	Safe sex and unwanted pregnancy reduction programme focusing on social skills and life skills plus knowledge based.	Social constructionism, social learning theory and sexual script theory.	Teachers
Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider
-------------------------	-------------------------------	--------------------------------------------	---------	------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	---------------------------------	-----------------------------------------
Workman et al., 1996	RCT -	USA n=60 students, mean age 15 years	School	Cognitive-behavioural HIV/AIDS prevention intervention focusing on sexual risk-reduction delivered over 12 weeks with 30 minutes per week. Key components focused on sexual-social values clarification, sexual-social decision making, reproductive-sexual anatomy and physiology, contraceptives, STDs, AIDS myths and facts, and sexual assertiveness and communication skills.	Cognitive-behavioural theory	Undergraduate psychology students

The overall number of students recruited to participate in the included studies ranged from 60 students to 3,142 students. Overall, five programmes (SLT programme, HIV prevention curriculum, AIDS and sexuality education, McMasters Teen Programme, AIDS/STI prevention curriculum) were based on samples greater than 1,500 students. The use of power calculations or an appropriate sample size to detect a significant effect was discussed in only 3 studies (Borgia et al., 2005, Traeen, 2003, Larsson et al., 2006) with two studies recognising the limitations of their samples (Borgia et al., 2005; Traeen, 2003). A further two studies provided enough information to determine that sample size was appropriate (Fisher et al., 2002, Mitchell-DiCenso et al., 1997). One study had a very small sample which was insufficiently powered (Workman et al., 1996). However, five studies did not provide sufficient information to determine whether the sample size was appropriate. (Coyle et al., 2006; Kvalem et al., 1996; Lemieux et al., 2008; Roberto et al., 2007; Schaalma et al., 1996).

The included studies focused on different ages and school years. One programme (McMasters Teen Programme) targeted young people 12-14 years with a mean age of 12.6 years. Two studies included a wide age range of students from age 13-14 years to 18-19 years (HIV prevention curriculum, All4you). A further two studies focused on older adolescents ranging between 16-20 years (AIDS and sexuality education, contraception programme) and two other studies focused on one age group/grade (aged 18 years; SLT programme, 15-16 years; Computer-based education).

Follow up times varied amongst programme evaluations. Two studies (Roberto et al., 2007; Workman et al., 1996) reported immediate post-test results only and three (Borgia et al., 2005; Lemieux et al., 2008; Schaalma et al., 1996) reported a short follow-up time of 5 months or less. Three studies (Fisher et al., 2002; Kvalem et al., 1996; Larsson et al., 2006) reported maximum follow-up times of 12 months and three studies (Coyle et al., 2006; Mitchell-DiCenso et al., 1997; Traeen, 2003) reported follow-up times greater than 12 months at 18 months (Coyle et al., 2006; Traeen, 2003) and 4 years (Mitchell-DiCenso et al., 1997).

6.5.2 Quality Assessment

Of the 11 included studies, eight were RCTs, two were NRCT and one was based on a CBA study design. Of the eight RCTs, seven (Borgia et al., 2005; Coyle et al., 2006; Kvalem et al., 1996; Mitchell-DiCenso et al., 1997; Roberto et al., 2007, Schaalma et al., 1996, Traeen, 2003) were based on cluster randomisation and one study used individual student randomisation (Workman et al., 1996). Cluster randomisation was conducted at school level in six studies (Borgia et al., 2005; Coyle et al., 2006; Mitchell-DiCenso et al., 1997; Roberto et al., 2007; Schaalma et al., 1996; Traeen, 2003) and at classroom level in one study (Kvalem et al., 1996). The unit of analysis in three of the RCTs (Borgia et al., 2005; Kvalem et al., 1996; Traeen, 2003) did not match the unit of allocation nor was any adjustment reported in the studies. The majority of studies were rated as moderate quality (+ rating). One RCT (Coyle et al., 2006) was rated good quality (++ rating) presenting a high quality matched study design and providing thorough details of the methodology and results. Only one study reported intention to treat analysis (Fisher et al., 2002) and this NRCT was rated good quality. However, three studies were rated as poor quality (- rating). Methodological data were limited in some studies and results were not always reported fully. Outcome measures were reported to be reliable in seven RCTs

(Borgia et al., 2005; Coyle et al., 2006; Mitchell-DiCenso et al., 1997; Roberto et al., 2007; Schaalma et al., 1996; Traeen, 2003; Workman et al., 1996), in one NRCT (Fisher et al., 2002), and in one CBA study (Lemieux et al., 2008). Relevant outcomes were reported across all included studies.

6.5.3 Findings

6.5.3.1 Short-term results (<6 months)

Six studies (Borgia et al., 2005; Fisher et al., 2002; Lemieux et al., 2008; Roberto et al., 2007; Schaalma et al., 1996; Workman et al., 1996) reported short-term outcomes on six programmes: a social learning theory-based (SLT) programme, an HIV prevention curriculum, SWAAT, a computer-based education programme, AIDS/STI prevention curriculum, a cognitive behavioural therapy-based (CBT) HIV prevention intervention.

Knowledge and understanding

Five studies (Borgia et al., 2005; Fisher et al., 2002; Roberto et al., 2007; Schaalma et al., 1996; Workman et al., 1996) reported short-term results related to sexual health and HIV knowledge and understanding. All five programmes reported significant improvements in knowledge relating to, for example, pregnancy, AIDS and prevention behaviours. One exception was in the peer educator delivery intervention (Fisher et al., 2002) which showed non-significant results compared to the teacher and combined methods of delivery.

Attitudes and values

Five studies (Borgia et al., 2005; Fisher et al., 2002; Lemieux et al., 2008; Roberto et al., 2007; Schaalma et al., 1996) reported short-term results of their programme on participants' attitudes and values.

Behavioural intention data (e.g. condom use) were reported by four studies (Fisher et al., 2002; Lemieux et al., 2008; Schaalma et al., 1996). Only one study (Schaalma et al., 1996) showed an increase in condom use intentions and students receiving the combined programme within the HIV prevention curriculum (Fisher et al., 2002) reported intentions to engage in HIV prevention behaviour that were significantly higher than the control group (p<0.05).

Five studies (Borgia et al., 2005; Fisher et al., 2002; Lemieux et al., 2008; Roberto et al., 2007; Schaalma et al., 1996) reported on attitudes and values relating to sexual behaviour. No significant short-term effects for attitudes towards condom use or social normative views of condom use were seen in those receiving SWAAT intervention (Lemieux et al., 2008). Intervention students receiving the SLT HIV prevention programme (Borgia et al., 2005) showed increases in attitudes and risk perceptions. Schaalma et al (1996) reported that students receiving an AIDS/STI prevention curriculum demonstrated significant increases in positive attitude and self-efficacy beliefs in addition to increased perceptions of subjective norms, peer behaviour, and risk appraisal. Roberto et al (2007) reported non-significant results relating to condom self-efficacy and refusal self-efficacy. Intervention students also expressed increased attitudes towards waiting to have sex and expressed significantly increased levels of situational self-efficacy. However, the intervention group also demonstrated

reduced perception of their own susceptibility to HIV/STIs. Fisher et al (2002) reported short-term results from a computer-based education programme on HIV prevention attitudes and norms stratified by intervention group (classroom, combined, peer) and sexual experience at baseline. Findings showed mixed effects on different groups.

Personal and social skills

Five studies (Borgia et al., 2005; Fisher et al., 2002; Roberto et al., 2007; Workman et al., 1996) explored short-term intervention effects on personal and social skills related to sexual behaviour and relationships. Two HIV focused studies (Borgia et al., 2005; Lemieux et al., 2008) reported significant increases in prevention skills⁹. A HIV prevention curriculum (Fisher et al., 2002) found significant increased in behavioural skills ¹⁰ in both their sexually inexperienced (p<0.01) and sexually experienced (p<0.05) intervention students exposed to a combined classroom and peer-based programme. However the intervention groups using either classroom only or peer only interventions showed no significant differences to the control group. Intervention students in one computer-based HIV and sexual health education programme (Roberto et al., 2007) reported significantly higher levels of condom negotiation skills compared to the control group (p<0.05). Workman and colleagues (1996) found no impact of their cognitive behavioural therapy-based HIV prevention intervention on their intervention students when examining: sexual decision-making, sexual assertiveness, level of comfort discussing AIDS preventative behaviour skills.

Health and social outcomes relating to alcohol use and sexual health

Five studies (Borgia et al., 2005; Fisher et al., 2002; Lemieux et al., 2008; Roberto et al., 2007; Schaalma et al., 1996) reported short-term health and social outcomes related to sexual health and alcohol use behaviours. Of these, two studies (Borgia et al., 2005; Schaalma et al., 1996) reported no significant effects on health outcomes in the following programmes: social learning theory-based HIV programme; AIDS/STI prevention curriculum. Roberto et al (2007) reported that students in the experimental group were significantly less likely to initiate sexual activity between pre-test and posttest compared to the control group (p<0.01). No studies examined short-term programme effects on conceptions.

Three studies examined the programme effects on frequency and/or number of sexual partners (Borgia et al., 2005; Roberto et al., 2007; Schaalma et al., 1996). All three studies reported no programme effects. Short-term effects on contraceptive use were reported by five studies (Borgia et al., 2005; Fisher et al., 2002; Lemieux et al., 2008; Roberto et al., 2007; Schaalma et al., 1996). Two studies (Borgia et al., 2005; Schaalma et al., 1996) found no intervention effects on frequency or consistent use of condoms in the past two or three months in students receiving either a social learning theory-based HIV programme or an AIDS/STI prevention programme. One study (Roberto et al., 2007) found no effects on condom use at last sexual intercourse in those students receiving a

⁹ Communication and negotiation skills; Borgia et al., 2007. Self-efficacy to perform prevention behaviours; Lemieux et al., 2008.

¹⁰ Abstinence and condom acquisition and use.

computer-based HIV education programme. Intervention students receiving the SWAAT curriculum (Lemieux et al., 2008) reported significant improvements in condom use behaviours in the preceding three months (p<0.05). Compared to the control students receiving their usual curriculum, sexually active students receiving the combined classroom/peer HIV intervention and those receiving the peer only HIV intervention reported significant increases in condom use in the past three months (p<0.05) (Fisher et al., 2002). Lemieux and colleagues (2008) also found short-term programme effects relating to uptake of testing for sexually transmitted infections in the experimental group receiving the SWAAT intervention.

6.5.3.2 Medium-term results (up to 12 months)

Seven studies (Coyle et al., 2006; Fisher et al., 2002; Kvalem et al., 1996; Larsson et al., 2006; Mitchell-DiCenso et al., 1997; Traeen, 2003; Workman et al., 1996) examined medium-term effects from seven programmes: All4you!, an HIV prevention curriculum, an AIDS/STI prevention programme, a contraception programme, McMaster's Teen Programme, a safe sex and pregnancy prevention programme, and a CBT HIV prevention programme.

Knowledge and understanding

Two studies (Coyle et al., 2006; Larsson et al., 2006) reported the medium-term impact of their programme on knowledge and understanding. Coyle and colleagues (2006) reported two medium-term follow up periods, one at 6 months and one at 12 months. At the six month follow-up a significant increase in HIV and condom knowledge (p<0.05) was found in the intervention group compared to the control group. However, this was not significant at 12 month follow-up. Non-significant results were also reported for general condom knowledge at both 6 and 12 month follow-up. Knowledge of emergency hormonal contraception (EHC) was examined at a 12 month follow-up of students receiving a contraception education programme. Students in the intervention group showed significant increases (p<0.01) in knowledge of EHC use at 72 hours after unprotected sex. However, no significant improvements were found in knowledge of EHC use on the first day after intercourse or of the side effects relating to EHC.

Attitudes and values

Two studies (Coyle et al., 2006; Larsson et al., 2006) examined medium-term programme effects on attitudes and values. Coyle and colleagues (2006) reported that students participating in the All4You! programme showed no significant difference in behavioural intentions relating to condom use at both 6 and 12 months compared to the control group. Furthermore, attitudes and beliefs regarding condom use for pregnancy prevention and for sexual intercourse generally were significantly higher in the control group receiving their usual curriculum than in the intervention group. Although, there was no significantly higher (p<0.05) increase in condom self-efficacy to get and use condoms compared to the intervention group at 12 month follow-up. A number of non-significant effects on attitudes and beliefs were reported at both 6 and 12 month follow-up as follows: attitudes towards condoms generally; condoms for STI prevention; perceptions of number of sexually active peers and peers

using condoms; perceptions of peer beliefs regarding condom use; optimism; fatalism; community orientation; connectedness to caring adult outside family circle; and self-efficacy to abstain. Larsson and colleagues (2006) found that attitudes and values of participants in a contraception programme remained relatively unchanged when observed at 12 month follow-up with the exception of a significant decrease in the view that EHC is a kind of abortion (p<0.05) and an increase in condom self-efficacy to buy condoms (p<0.05). Non-significant changes in contraceptive attitudes and values were observed in the following cases: attitude towards using EHC; feeling embarrassed to buy condoms; view that condoms are a man's responsibility; that contraceptives will be influenced by EHC; that an EHC increase would increase the risk of unprotected sex; could imagine discussing condoms; could imagine using condoms if they had a new partner; would recommend EHC to a friend.

Personal and social skills

No studies reported medium-term programme effects on personal and social skills.

Health and social outcomes relating to alcohol use and sexual health

Seven studies (Coyle et al., 2006; Fisher et al., 2002; Kvalem et al., 1996; Larsson et al., 2006; Mitchell-DiCenso et al., 1997; Traeen, 2003; Workman et al., 1996) examined medium-term effects from seven programmes. Three studies (Coyle et al., 2006; Kvalem et al., 1996; Mitchell-DiCenso et al., 1997) found no significant medium-term programme effects on initiation of sexual intercourse at either 6 (Coyle et al., 2006) or 12 month follow-up for the following programmes: All4you!; AIDS/STI education; McMaster's Teen Programme. Participants in the All4you! (Coyle et al., 2006) programme showed a significantly lower frequency of sexual intercourse in the intervention group compared to the control group receiving their usual curriculum. However, this was no longer significant at 12 month follow-up. There was no significant difference seen for number of unprotected partners (steady and non-steady) in the past three month or number of previous partners. Workman and colleagues (1996) also reported no significant effects on HIV prevention behaviours in students receiving the CBT HIV prevention programme.

Medium-term programme effects on contraception were reported in five studies (Coyle et al., 2006; Fisher et al., 2002; Kvalem et al., 1996; Larsson et al., 2006; Traeen, 2003). One study (Kvalem et al., 1996) reported no significant effects on condom use. Students participating in the All4you! programme (Coyle et al., 2006) showed significant decreases in unprotected sex (overall (p<0.01) and with non-steady partners (p<0.05)) in the past 3 months and in condom use at last sexual intercourse (p<0.01). However, no effects were sees at 12 month follow-up for these aspects or for frequency of unprotected sex with steady partners or effective pregnancy prevention behaviour at last sexual intercourse. At six months follow-up Kvalem et al (1996) found significantly increased levels of condom use in the intervention group compared to the control (p<0.001). However, these were not sustained at 12 month follow-up. Students participating in the contraception programme showed increased condom use at 12 months follow-up (p<0.05) and showed no significant differences in lifetime use of EHC compared to the control group. Fisher and colleagues (2002) reported mixed results from their intervention groups with significant increases in condom use seen in the classroom intervention group (p<0.01) and no significant differences seen in the peer and combined

peer/classroom intervention groups compared to the control. Students exposed to the safe sex and pregnancy prevention intervention (Traeen, 2003) who were sexually inexperienced at baseline showed significant increases in use of condoms (p<0.05) and oral/hormonal contraceptives (p<0.05) at first sex. However, no significant differences were seen in other methods of contraception at first sex or in the use of condoms or oral/hormonal contraception at most recent sex. Increases were sees in interrupted intercourse at both first sex and most recent sex (p<0.05) compared to controls.

One study (Coyle et al., 2006) examined the medium-term programme effects on health outcomes relating to sexually transmitted infection and alcohol and drug use. Findings in both the six month follow-up and the 12 month follow-up showed no significant programme effects on the number of HIV or STI tests taken or the use of drugs/alcohol prior to sexual intercourse in the past three months. Two studies also (Coyle et al., 2006; Mitchell-DiCenso et al 1997) reported no significant programme effects on pregnancy rates compared to control groups.

6.5.3.3 Long-term results (>12 months)

Three studies (Coyle et al., 2006; Mitchell-DiCenso et al 1997; Traeen, 2003) examined long-term programme effects for three interventions: All4you!, McMaster's Teen Programme, and a safe sex and pregnancy prevention programme.

Knowledge and understanding

Programme effects on knowledge and understanding were examined by one study (Coyle et al., 2006). Students participating in the intervention showed significant increases in HIV and condom knowledge (p<0.05) as well as general condom knowledge (p<0.05) compared to the control group receiving the usual curriculum.

Personal and social skills

One study (Coyle et al., 2006) reported on behavioural intention and attitudes and values relating to condom use and peers and self-efficacy at 18 months follow-up. However, no significant effects were shown in the intervention group at this time.

Health and social outcomes relating to alcohol use and sexual health

Three studies (Coyle et al., 2006; Mitchell-DiCenso et al 1997; Traeen, 2003) examined long-term health outcomes relating to sexual health and alcohol. Two studies (Coyle et al., 2006; Mitchell-DiCenso et al 1997) reported programme effects on initiation at first intercourse and similar to the medium-term results no significant effects were found. One study (Coyle et al., 2006) reported on frequency and number of sexual partners at 18 months follow-up and similar to medium-term effects there were no significant positive effects on number of unprotected partners or past number of sexual partners. In addition, there was a significant negative effect on number of unprotected non-steady sexual partners (p<0.05) in the intervention group compared to the control. Three studies (Coyle et al., 2006; Mitchell-DiCenso et al 1997; Traeen, 2003) reported long-term effects on contraception use

6.5.4 Summary and evidence statements

Overall, 11 studies (Borgia et al., 2005; Coyle et al., 2006; Fisher et al., 2002; Kvalem et al., 1996; Larsson et al., 2006; Lemieux et al., 2008; Mitchell-DiCenso et al., 1997; Roberto et al., 2007; Schaalma et al., 1996; Traeen, 2003; Workman et al., 1996) were identified that examined HIV and sexual risk-reduction programmes.

6.5.4.1 Knowledge and understanding

Knowledge outcomes were reported by seven studies (Borgia et al., 2005; Coyle et al., 2006; Fisher et al., 2002; Larsson et al., 2006; Roberto et al., 2007; Schaalma et al., 1996; Workman et al., 1996). General HIV or sexual health knowledge was reported most frequently. However, only two studies (Coyle et al., 2006; Larsson et al., 2006) explored medium- or long-term knowledge outcomes and they focused on knowledge of HIV and condom use or the emergency hormonal contraceptive (EHC) pill. Medium-term effects were inconsistent with evidence of increased knowledge of EHC use on the third day after unprotected sex. No effect on HIV and condom knowledge was seen at 12 months follow-up, yet there was a positive effect at 18 months. Both studies included older adolescents in their sample (Coyle et al., 2006: 14-18 years; Larsson et al., 2006: mean 17.25 years).

6.5.4.2 Attitudes and values

Outcomes relating to attitudes were reported by seven studies (Borgia et al., 2005; Coyle et al., 2006; Fisher et al., 2002; Lemieux et al., 2008; Larsson et al., 2006; Roberto et al., 2007; Schaalma et al., 1996). A wide variety of outcomes were reported across studies and often specific outcomes were only reported within one study, thus preventing comparison. Condom self-efficacy, perception of social norms and condom use/prevention intentions were the outcomes most commonly reported across studies. Lemieux et al (2008) and Roberto et al (2007) reported no programme effects on condom self-efficacy. Whereas Coyle et al (2006) reported no programme effects for condom self-efficacy at six and 18 months follow-up and a reduction at 12 months follow-up. Perceptions of social norms increased in both the classroom (sexually experienced at baseline) and combined groups (sexually inexperienced at baseline) receiving a HIV prevention curriculum but not in the other groups. Lemieux et al (2008) reported no programme effects on support. Schaalma et al (1996) found an increase in positive perceptions of subjective norms at 4-8 weeks follow-up. A significant improvement in condom use/prevention intentions was seen in Schaalma et al (1996). However, four studies (Fisher et al., 2002; Lemieux et al., 2008; Coyle et al., 2006; Larsson et al., 2006) reported primarily no effect on condom use or prevention intentions.

6.5.4.3 Personal and social skills

Short-term programme effects on personal and social skills were reported by five studies (Borgia et al., 2005; Fisher et al., 2002; Lemieux et al., 2008; Roberto et al., 2007; Workman et al., 1996). Findings were predominantly positive and indicated programme effects overall on prevention or behavioural skills in two studies (Borgia et al., 2005; Lemieux et al., 2008). Participants receiving the combined programme in the HIV prevention curriculum showed an increase in behavioural skills. Participants in

the computer-based educational programme reported an increase in condom negotiation skills. Only one study (Workman et al., 1996) reported no effects on any personal or social skills.

6.5.4.4 Health and social outcomes relating to alcohol use and sexual health

Eleven studies reported health outcomes from HIV and sexual risk-reduction programmes. Impact on sexual initiation was explored in four studies (Roberto et al., 2007; Coyle et al., 2006; Kvalem et al., 1996; Mitchell-DiCenso et al., 1997) and only short or medium-term outcomes were reported. One study (Roberto et al., 2007) reported positive results on age at sexual initiation. All other studies showed no programme effects. Furthermore, three studies (Borgia et al., 2005; Roberto et al., 2007; Schaalma et al., 1996) reported no impact on the rate of sexual activity and four studies (Borgia et al., 2005; Coyle et al., 2006; Roberto et al., 2007; Schaalma et al., 1996) showed no effect on the number of sexual partners participants reported. One study (Borgia et al., 2005) split analysis by whether participants were sexually active at baseline or not, whereas it was not clear if this was the case in others. Five studies (Fisher et al., 2002; Lemieux et al., 2008; Coyle et al., 2006; Kvalem et al., 1996; Larsson et al., 2006) showed positive short-term programme effects on condom use and one study (Traeen, 2003) showed programme effects on protected intercourse. The long-term programme effects were limited and one study (Borgia et al., 2005) showed no short-term programme effects on condom use. However, the mean age of the study population was 18 years which may have had an influence on the results if relationship status was not accounted for in an older adolescent population. One study (Mitchell-DiCenso et al., 1997) targeting a younger cohort reported only long-term effects on condom use, with positive increase in condom use among males at 2 years follow-up. Limited outcomes were presented on HIV/STI testing, alcohol or drug use and pregnancy. One study (Lemieux et al., 2008) reported a significant increase in HIV testing, however all other outcomes showed no difference compared to the control groups.

Evidence statement 12

- 12 (g) There is moderate evidence from five RCTs, two NRCTs and one CBA study¹ to suggest that HIV and sexual risk-reduction programmes can improve sexual health and HIV knowledge in the short-, medium-and long-term. This evidence may be only partially applicable to the UK as five of the studies were conducted in the USA, one in Italy and one in the Netherlands and may not be generalisable beyond the populations studied.
- 12 (h) There is mixed evidence from four RCTs and one NRCT² that examined the effects of HIV and sexual risk-reduction programmes on young people's≥(14 years) attitudes and values towards sexual health and alcohol. This evidence may be only partially applicable to the UK as studies were carried out in the USA, Italy and the Netherlands and may not be generalisable beyond the populations studied.
- 12 (i) There is moderate evidence from two RCTs, one NRCT and one CBA study³ to suggest that HIV and sexual risk-reduction programmes may improve personal and social skills including behavioural prevention skills and condom negotiation skills in the short-term. There was no evidence to determine the effects of HIV and sexual risk-reduction on personal and social skills in the medium- to long-term. This evidence may be only partially applicable to the UK as

studies were carried out in the USA and Italy and may not be generalisable beyond the populations studied.

- 12 (j) There is moderate evidence from seven RCTs⁴ to suggest that HIV and sexual risk-reduction programmes may have no effects on sexual initiation, frequency of sexual activity or number of sexual partners. This evidence may be only partially applicable to the UK as studies were carried out in the USA, Norway, Italy and Netherlands and may not be generalisable beyond the populations studied.
- 12 (k) There is strong evidence from three RCTs, two NRCTs and one CBA study⁶ to suggest that HIV and sexual risk-reduction programmes can increase condom use or protected intercourse in the short- to medium-term. However, there was moderate evidence from two RCTs⁷ to suggest that the long-term effects of HIV and sexual risk-reduction programmes on contraceptive use may be limited. This evidence may be only partially applicable to the UK as studies were carried out in the USA, Norway, and Sweden and may not be generalisable beyond the populations studied.
- 12 (I) There is moderate evidence from two RCTs⁸ to suggest that HIV and sexual risk-reduction programmes have no medium- to long-term effect on sexually transmitted infections, alcohol and drug use or on conceptions. As both studies were implemented in the USA findings may be only partially applicable to the UK and may not be generalisable beyond the populations studied.
- ¹ Borgia et al., 2005 (RCT +); Coyle et al., 2006 (RCT ++); Fisher et al., 2002 (NRCT ++); Larsson et al., 2006 (NRCT +); Lemieux et al., 2008 (CBA -); Roberto et al., 2007 (RCT +); Schaalma et al., 1996 (RCT +); Workman et al., 1996 (RCT -)
- ² Borgia et al., 2005 (RCT +); Coyle et al., 2006 (RCT ++); Fisher et al., 2002 (NRCT ++); Roberto et al., 2007 (RCT +); Schaalma et al., 1996 (RCT +)
- ³ Borgia et al., 2005 (RCT +); Fisher et al., 2002 (NRCT ++); Lemieux et al., 2008 (CBA -); Roberto et al., 2007 (RCT +)
- ⁴ Borgia et al., 2005 (RCT +); Coyle et al., 2006 (RCT ++); Kvalem et al., 1996 (RCT -); Mitchell-DiCenso et al., 1997 (RCT +); Roberto et al., 2007 (RCT +); Schaalma et al., 1996 (RCT +); Workman et al., 1996 (RCT -)
- ⁵ Coyle et al., 2006 (RCT ++); Kvalem et al., 1996 (RCT -); Mitchell-DiCenso et al., 1997 (RCT +)
- ⁶ Borgia et al., 2005 (RCT +); Coyle et al., 2006 (RCT ++); Roberto et al., 2007 (RCT +); Schaalma et al., 1996 (RCT +)
- ⁷ Coyle et al., 2006 (RCT ++); Fisher et al., 2002 (NRCT ++); Lemieux et al., 2008 (CBA -); Kvalem et al., 1996 (RCT -); Larsson et al., 2006 (NRCT +); Traeen, 2003 (RCT +)
- ⁸ Coyle et al., 2006 (RCT ++); Traeen, 2003 (RCT +)
- ⁹ Coyle et al., 2006 (RCT ++); Mitchell-DiCenso et al., 1997 (RCT +)

Study	Pating	Intonyoption	Comparator	Follow-up		Outcomes	
Sludy	каші	Intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Personal/social skills
Borgia et al., 2005 ³	RCT +	Peer-led n=613	Teacher-led n=682	5 months	↑ Knowledge	↑ Attitudes, risk-perception	↑Prevention skills
Fisher et al., 2002	NRCT ++	HIV prevention curriculum Classroom, n=310 Peer, n=381 Combined, n=296	Usual curriculum n=589	3 months	Sexually inexperienced: ↑ HIV prevention information (classroom***, combined*** only) Sexually experienced: ↑ HIV prevention information (classroom***, combined***, peer**)	Sexually inexperienced: ↑ attitudes (classroom**, combined** only) ↑ norms (combined only*) ↑intentions (combined only*) Sexually experienced: ↑ attitudes (peer**, combined** only ↑ norms (classroom only*) NS intentions	Sexually inexperienced: ↑ behavioural skills (combined only**) Sexually experienced: ↑ behavioural skills (combined only*)
Lemieux et al., 2008	CBA -	SWAAT n=NR	No intervention n=NR	3 months	-	NS condom use ^a NS social normative support for condom use ^a NS intention to use condoms	↑HIV prevention behavioural skills*
Roberto et al., 2007	RCT +	Computer-based education n=181	No intervention n=221	PT	↑ knowledge***	 ↑attitude towards waiting* NS condom self-efficacy, refusal self-efficacy ↑ situational self-efficacy* ↓susceptibility** 	↑condom negotiation*
Schaalma et al., 1996	RCT +	AIDS/STI prevention curriculum n=NR	Usual curriculum n=NR	4-8 weeks	↑ knowledge***	 ↑ risk appraisal** ↑ positive attitudes*** ↑ positive perceptions of subjective norms** ↑ positive perceptions of peer behaviour* ↑ self-efficacy beliefs* ↑ condom use intentions** 	-
Workman et al., 1996	RCT -	CBT HIV prevention n=30	Womanhood development n=30	PT (1 week)	↑ HIV knowledge**	-	NS sexual decision-making NS sexual assertiveness NS level of comfort discussing AIDS preventative behaviour
*p≤0.05; **p≤0.0 ^a Authors claime	01; ***p≤0.001; ∕ ed the programm	Increase relative to ne provided protective	o comparator;	crease relative t iificant decrease	o comparator; NS not signi es were seen in controls	ticant; - outcome not reported	

Table 6.24. HIV and sexual risk-reduction programmes: short-term programme effects on knowledge, attitudes and skills

						Неа	alth outcomes		
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Borgia et al., 2005 ³	RCT +	Peer-led n=613	Teacher-led n=682	5 months	-	NS number of partners, past 3 months ^a	NS frequency of condom use, past 3 months	-	-
Fisher et al., 2002	NRCT ++	HIV prevention curriculum, classroom n=310 Peer n=381 Combined n=296	Usual curriculum n=589	3 months	-	-	↑ condom use, past 3 months ^a (combined*, peer* only)	-	-
Lemieux et al., 2008	CBA -	SWAAT n=NR	No intervention n=NR	3 months	-	-	↑Condom use, past 3 months*	↑ HIV testing**	-
Roberto et al., 2007	RCT +	Computer-based education n=181	No intervention n=221	PT	↓ initiation of sexual activity**	NS number of partners	NS condoms use at last intercourse	-	-
Schaalma et al., 1996	RCT +	AIDS/STI prevention curriculum n=NR	Usual curriculum n=NR	4-8 weeks	-	NS sexual risk index ^b	-	-	-
*p≤0.05; **p≤0. ^a sexually expe	01;	↑ increase relative t ine; ^b score ranging f	to comparator; $oldsymbol{\psi}$ c from 0 (no sexual i	decrease relative t ntercourse) to 3 (i	to comparator; NS inconsistently using	not significant; - outc g condoms with two c	ome not reported or more sex partners	;)	

Table 6.25. HIV and sexual risk-reduction programmes: short-term programme effects on health and social outcomes

Chudu	Deting	Intoniontion	Commenter			Outcomes	
Study	Rating	Intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Personal/social Skills
Coyle et al., 2006	RCT ++	All4you! n=597	Usual curriculum n=391	6 months (73%)	↑ HIV & condoms* NS condoms	NS condoms generally NS condoms for STI protection ↓ condoms for pregnancy protection* ↓ condoms for intercourse* NS normative beliefs NS condom use intentions NS condom self-efficacy NS self-efficacy to abstain	-
				12 months (62%)	NS HIV & condoms NS condoms	NS attitudes and beliefs regarding condoms NS normative beliefs NS condom use intentions ↓ condom self-efficacy* NS self-efficacy to abstain	-
Larsson et al., 2006	NRCT +	Contraception programme n=282	No intervention n=179	12 months n=367 (94%)	↑EHC on 3 rd day** NS EHC on 1 st day NS EHC side effects	 ↓ EHC is a kind of abortion* ↑could imagine buying condoms* NS other attitudes towards EHC and condoms 	-
*p≤0.05; **p≤0.0 EHC emergency	01; ***p≤0.001; ∕ y hormonal cont	increase relative t raception	o comparator; 🗸 de	crease relative t	to comparator; NS not sig	nificant; - outcome not reported	

Table 6.26. HIV and sexual risk-reduction programmes: medium-term programme effects on knowledge, attitudes and skills

Table 6.27. HIV and sexual risk-reduction programmes: medium-term programme effects on health and social outcomes

							Health outcomes		
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Coyle et al., 2006	RCT ++	All4you! n=597	Usual curriculum n=391	6 months (73%)	NS sexual initiation	NS Number of unprotected partners ^a ↓ frequency of sexual intercourse ^a * NS number of unprotected steady/non-steady partners ^a NS number of previous sexual partners	 ✓ frequency of unprotected intercourse^a ✓ frequency of unprotected intercourse^a (steady partner only*) ↑ condom use^b NS effective pregnancy prevention^b 	NS number of HIV tests NS number of tests for STIs	NS pregnancy since baseline

					Health outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Coyle et al., 2006	RCT ++	All4you! n=597	Usual curriculum n=391	12 months (62%)	NS sexual initiation	NS Number of unprotected partners ^a NS Frequency of intercourse ^a NS number of unprotected steady/non-steady partners ^a NS number of sexual partners previous	NS Frequency of unprotected intercourse ^a NS Frequency of unprotected intercourse ^a (steady partner, non-steady partner) NS condom use ^b NS effective pregnancy prevention ^b	NS number of HIV tests NS number of tests for STIs	NS pregnancy since baseline
Fisher et al., 2002	NRCT ++	HIV prevention Classroom, n=310 Peer, n=381 Combined, n=296	Usual curriculum n=589	12 months	-	-	↑ condom use, past year (classroom only ^c **)	-	-
Kyalom ot		AIDS and	No	6 months (76%)	-	-	↑ condom use***	-	-
al., 1996	RCT -	education n=284	intervention n=801	12 months (69%)	NS initiation of first intercourse	-	NS condom use	-	-
Larsson et al., 2006	NRCT +	Contraceptio n programme n=282	No intervention n=179	12 months	-	-	↑ condom use* NS used EHC, ever	-	-
Mitchell- DiCenso et al., 1997	RCT +	McMasters Teen Programme n=2,309	Usual curriculum n=1,666	12 months	NS initiation of first intercourse	-	-	-	NS first pregnancy

					Health outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Traeen, 2003	RCT +	Safe sex and pregnancy prevention n=416	No intervention n=197	6-7 months (69%)	-	-	At first intercourse ^c : NS some contraception at first sex, no contraception, condoms before orgasm, emergency contraception ^b ↑ interrupted at first sex, condoms, hormonal/oral contraception At most recent sex ^c : NS some contraception, no contraception, no contraception, no contraception, no contraception, condoms before orgasm, condoms, hormonal/oral contraception, emergency contraception. ↑ interrupted	-	-
Workman et al., 1996	RCT -	CBT HIV prevention n=30	Womanhood development n=30	PT (1 week)	-	NS AIDS preventative behaviours	-	-	-
*p≤0.05; **p≤0 ^a in past 3 mo).01; ***p≤0.0 nths; ^b at last	001; ↑ increase intercourse; ^c se	relative to comp exually experience	arator; V decr	ease relative to	comparator; NS not sign	ificant; - outcome not repo	orted	

Table 6.28. HIV and sexual risk-reduction programmes: long-term programme effects on knowledge, attitudes and skills

Study	Pating	Intervention	Comparator	Follow-up		Outcomes	
Study	Rating	intervention	Comparator	i onow-up	Knowledge	Attitudes and values	Skills
Coyle et al., 2006	RCT ++	All4you! n=597	Usual curriculum n=391	18 months	↑ HIV & condoms* ↑ condoms*	NS attitudes and beliefs regarding condoms NS normative beliefs NS condom use intentions NS condom self-efficacy NS self-efficacy to abstain	-
*p≤0.05; **p≤0.0	01; ***p≤0.001; /	increase relative t	o comparator; 🗸 de	crease relative t	o comparator; NS not sign	ificant; - outcome not reported	

							Health outcomes		
Study	Rating	Intervention	Comparator	Follow- up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Coyle et al., 2006	RCT ++	All4you! n=597	Usual curriculum n=391	18 months	NS sexual initiation among those inexperience at baseline	NS Number of unprotected partners ^a NS Frequency of intercourse ^a NS Number unprotected steady partners ^a ↑ Number unprotected non- steady partners ^a * NS number of previous sexual partners	NS Frequency of unprotected intercourse ^a NS Frequency of unprotected intercourse ^a (steady) partner/non-steady) NS condom use ^b NS effective pregnancy prevention ^b	NS number of HIV tests NS number of tests for STIs	NS pregnancy since baseline
Traeen, 2003	RCT +	Safe sex and pregnancy prevention n=416	No intervention n=197	1-2 years (56%)	-	-	NS contraceptive use at first sex ^c	-	-
Mitchell- DiCenso et al., 1997	RCT +	McMasters Teen Programme n=2,309	Usual curriculum n=1,666	2-4 years	NS initiation of first intercourse	-	↑ always condom use in males, 2 years	-	NS first pregnancy
*p≤0.05; **p≤0. ^a past 3 months	01; ***p≤0.00 ; ^b last interco	01; ↑ increase re ourse; ^c sexually	elative to compar inexperienced a	rator; ↓ de t baseline,	crease relative to ^c sexually experie	comparator; NS not sign nced at baseline.	ificant; - outcome not re	ported	

Table 6.29. HIV and sexual risk-reduction programmes: long-term programme effects on health and social outcomes

6.6 Other school-based approaches

6.6.1 Overview of evidence identified

A total of included seven studies were identified that reported on six different programme approaches relating to sexual health. Two studies examined school-based clinic programmes (Stout et al., 1996; Teitler, 1997), two studies (Somers et al., 2001; 2006) reported findings from one programme using baby simulators (Baby Think it Over), and three studies (Lewis et al., 1999; Paine-Andrews et al., 1999; Vincent et al., 2004) examined the effects of combined community and school-based programmes. School-based clinic studies were included where the evaluation examined a health clinic located directly within the school grounds. Stout et al (1996) examined school-based clinic programmes at three pairs of schools (sites A, B and C), which had a centre that had been operating for five school months at the time of the baseline survey. All three centres had varying operating budgets, staffing levels and qualifications, and constraints on their practice, but all three faced the constraint of not being able to prescribe or dispense contraception. Teitler (1997) evaluated the impact of nine school-based health resource centres where students could receive reproductive health information, condoms and general health referrals. Vincent et al (2004) examined the long-term effects of the School/Community Programme for Sexual Risk Reduction Among Teens, which was originally conducted in a rural county in South Carolina. Two studies (Lewis et al., 1999; Paine-Andrews et al., 1999) reported on the replication of this programme in three Kansas communities (Franklin County, Geary County and one neighbourhood in Wichita). Paine-Andrews et al (1999) examined the effects of all three community partnerships, and Lewis et al (1999) examined the effects of the initiative in Geary County only. All three studies (Lewis et al., 1999; Paine-Andrews et al., 1999; Vincent et al., 2004) examined the effects of the programme at the county level by comparing outcomes in intervention and non-intervention communities.

All studies were conducted in North America. Two studies (Stout et al., 1996; Teitler, 1997) were carried out in schools and consisted of two school based clinics (Stout et al., 1996; Teitler, 1997) neither of which reported a theoretical base. A further three studies evaluated complex interventions conducted out in schools and the community (Lewis et al., 1999; Paine-Andrews et al., 1999; Vincent et al., 2004), two studies (Lewis et al., 1999; Vincent et al., 2004) did not report a theoretical base for the intervention, but the intervention in the third study (Paine-Andrews et al., 1999) was underpinned by theory of change theory. The Baby Think it Over programme was carried out in both school and the home (Somers et al., 2001; 2006) with no theoretical base in the initial study but with cognitive development theory underpinning the second study.

Table 6.30. Sex and relationshi	p education: other	school-based	approaches

Author	Study design and rating	Setting	Baseline population	Programme components	Theory	Provider
Stout et al., 1996	CBA -	School	USA n=1870 – site A n=778 – site B n=1,651, site C	School health promotion clinics; open 16-40 hours per week	Not reported	Clinic staff in school
Teitler 1997	CBA -	School	USA n=945 (at FU) students, mean 16 years	Drop in centres at school providing information, condoms and health referrals	Not reported	Health professionals and graduate interns
Lewis et al., 1999	ITS -	School + community	USA n=NR, 9 th -12 th grade students	Reducing the Risk : K-12 th Grade Curriculum; increased access to health services and contraceptives; media efforts to increase awareness; peer education and support; supervised activities; programs in the faith community; community lineages	Not reported	Teachers
Paine- Andrews et al., 1999	CBA -	School + community	USA Participant details NR	Pregnancy prevention programme over five years; Enhanced sexuality education for teachers and parents; comprehensive, age-appropriate sexuality education from K-12; increased access to health services; collaboration with school administrators; use of mass media; increased awareness and involvement of the entire community in teenage pregnancy prevention, peer support and education; alternative activities for young people; and involvement of the faith community.	Theory of change	Various not described
Somers et al., 2001	CBA -	School + home	USA 100 students, mean age years: intervention17.1; control 16.4	Baby Think It Over : 48 hours at home and school; computerised infant simulator	Not reported	NA
Somers et al., 2006	NRCT -	School + home	USA 230 students, mean age years: intervention 15.8; control 16.6	Baby Think It Over : Two nights and three days; computerised infant simulator	Cognitive development	NA
Vincent et al., 2004	ITS -	School + community	USA n=NR	Abstinence and contraception promotion interventions from K-12 grades in one county	Not reported	Health educators

One study examined three school health promotion clinics (Stout et al., 1996) where services were delivered by clinic staff. A second school-based clinic (Teitler, 1997) provided information, referrals and condoms to students was delivered by health professionals and graduate interns. The Baby Think it Over Programme was primarily delivered by the baby simulators used by young people over a period of 2-3 days. External professionals (i.e. health educators) were used to deliver the abstinence and safe sex programmes (Vincent et al., 2004). Reducing the Risk programme was the only intervention delivered by teachers. Further, the community-based pregnancy prevention programme (Paine-Andrews et al., 1999) used a variety of people to deliver the intervention however details for this were unclear.

The number of students recruited into the programmes ranged from 100 students to 1,870 students. However, the number of participants was not reported in three studies (Lewis et al., 1999; Paine-Andrews et al., 1999; Vincent et al., 2004). Only one study (Stout et al., 1996) was based on a sample of more than 1,500 participants and as this study employed three different clinic sites as intervention sites only two of the three sites had intervention groups over 1,500. Appropriate sample size to power the analysis conducted was only discussed in one case (Teitler, 1997) and authors concluded that the sample was too small to capture small or medium size effects.

Programmes targeted a range of age groups. Two studies did not clearly record the age group of their participating students (Paine-Andrews et al., 1999; Vincent et al., 2004). Reducing the Risk and both clinic-based interventions targeted students aged 14-18 years in grades 9-12 (Stout et al., 1996) and Teitler (1997) reported a mean age of 16 years for participants. The participants in the first Baby think it Over programme (Somers et al., 2001) had a mean age of 17.1 years whereas the second cohort (Somers et al., 2006) were younger with a mean age of 15.8 years. Interventions duration varied by programme with Baby Think It Over (Somers et al., 2001; 2006) carried out over 48-36 hours, curriculum-based programmes took place regularly over several years (Lewis et al., 1999; Paine-Andrews et al., 1999; Vincent et al., 2004) and school-based clinics open to students throughout the school week and available over a two year period (Stout et al., 1996; Teitler, 1997).

Programmes reported a range of follow-up times. One study (Somers et al., 2006) reported immediate post-test results only and one other (Somers et al., 2001) reported follow-up results at less than six months. Two studies (Lewis et al., 1999; Stout et al., 1996) reported long-term follow-up between 2-3 years. One study (Vincent et al., 2004) explored data relating to girls aged 14-17 years in one county receiving abstinence and safe sex education over a 20 year period. A further two other studies (Paine-Andrews et al., 1999; Teitler, 1997) did not report a clear follow-up time.

6.6.2 Quality assessment

Of the seven studies included, one (Somers et al., 2006) was a NRCT. Four studies (Paine-Andrews et al., 1999; Somers et al., 2001; Stout et al., 1996; Teitler, 1997) were CBA studies and two studies (Vincent et al., 2004; Lewis et al., 1999) were based on an interrupted time series. Study quality was rated poor across all seven studies (- rating). The Baby Think it Over programme (Somers et al., 2006) was based on varied follow-up times for participants, it was not clear how bias was minimised and contamination may have been significant in this study. Limited details on the intervention and the

reliability of the measures used were provided. The four CBA studies (Paine-Andrews et al., 1999; Somers et al., 2001; Stout et al., 1996; Teitler, 1997) were all rated poor quality. Studies were judged to have provided limited details of their methodology, intervention, and participants in both the intervention and control groups. Follow-up times were also generally too short to be classed as meaningful in some studies, although outcome measures were deemed appropriate overall. The two interrupted time series studies (Lewis et al., 1999; Vincent et al., 2004) were not well reported in terms of the methodology used for the time series. Neither study reported a clearly defined point in time over which the intervention occurred, other than the year of intervention, nor was it clear if the intervention occurred independently of other changes over time.

6.6.3 Findings

6.6.3.1 Short-term results (<6 months)

Two studies (Somers et al., 2001; 2006) reported short-term programme effects for the Baby Think It Over, infant simulator programme.

Knowledge and understanding

None of the studies identified examined short-term programme effects on knowledge and understanding.

Attitudes and values

Two studies (Somers et al., 2001; 2006) using the Baby think it over programme reported short-term effects on participants' attitudes and values. Non-significant effects on behavioural intentions regarding safe sex and child bearing were reported in both studies. The programme also had no effect on attitudes towards pre-marital sex, child rearing, pregnancy, or future orientation. However, Somers et al (2006) did report a decrease in the intervention students' views regarding other people's acceptance of teenage pregnancy ($p \le 0.05$).

Personal and social skills

None of the studies identified examined short-term programme effects on personal and social skills.

Health and social outcomes relating to alcohol use and sexual health

Two studies (Somers et al., 2001; 2006) of the Baby Think It Over programme examined short-term health outcomes on participants. There no effects of the intervention on age of first sex, frequency of sex or contraceptive use.

6.6.3.2 Medium-term results (up to 12 months)

None of the studies identified examined medium-term programme effects.

6.6.3.3 Long-term results (>12 months)

Five studies (Stout et al., 1996; Teitler, 1997; Lewis et al., 1999; Paine-Andrews et al., 1999; Vincent et al., 2004) examined long-term effects intervention effects for two school-based health centres and three school- and community-based programmes, respectively.

Knowledge and understanding

None of the studies identified examined medium-term programme effects on knowledge and understanding.

Attitudes and values

Stout et al (1996) examined the effects of school-based health centres across three pairs of school sites defined as sites A, B and C. At one site, site B, there was a decrease in positive views towards abstention before marriage relative to the comparison site (p<0.05), but at a second site, site C, an increase in the number of people favouring abstention before marriage was reported relative to the comparison site (p<0.05). Interviews with staff at site C indicated that they placed a great emphasis on encouraging abstinence, a factor which is likely to have influenced outcomes at this site. At the third site, site A, there was no difference between the intervention and comparison sites in reproductive health attitudes.

Personal and social skills

Stout et al (1996) reported that there were no significant programme effects of health clinics on participants' sexuality dialogue with parents, at any of the three intervention sites, compared to comparison sites.

Health and social outcomes relating to alcohol use and sexual health

Three studies (Lewis et al., 1999; Paine-Andrews et al., 1999; Vincent et al., 2004) reported long-term programme effects for two multicomponent community- and school-based intervention programme approaches. Vincent et al (2004) examined the long-term effects of the School/Community programme in a rural county in South Carolina, by comparing 20 years of pregnancy rates among girls aged 14-17 in the intervention county with three comparison counties. Based on three approaches to the analyses, the authors found that: (1) overall, there was a general downturn in teen pregnancy rates across the entire state between 1981 and 2000; (2) in the intervention county rates fell from 54.8 to 32.1 pregnancies per 1,000 females; and (3) the intervention county started higher and ended lower than the comparison communities. From this the authors concluded that the intervention had had a positive effect on teenage pregnancies in the intervention county. Paine-Andrews et al (1999) examined the effects of the replication of the School/Community programme on estimated pregnancy rates and birth rates across three communities in Kansas, USA. Overall, although estimated pregnancy rates decreased among females aged 14-17 in two intervention communities (Franklin County and Geary County), there was no significant difference in pregnancy rates between the intervention and comparison communities. Changes in birth rates were mixed in two intervention communities in Wichita, and overall there was no significant difference between intervention and control communities on this measure. Lewis et al (1999) examined the effects of the school/community partnership approach in Geary County in more detail. In the intervention community the estimated pregnancy rate for 14-17 year olds decreased between 1991 and 1996 (the partnership approach was introduced in 1993), in comparison to an increase in comparison communities. However, based on an average of the estimated pregnancy rate for the three years

before and during the intervention, the relative decrease seen in the intervention community was not found to be significant.

Two studies examined the long-term effects of two school-based health centres. Teitler (1997) reported on the evaluation of the health resource centres programme which was implemented in nine schools in Philadelphia, USA. There were no significant effects of the health resource centres on initiation of sex, sex in the past month or condom use. However, the authors reported that the direction of the effect was in favour of the health centres on all of these measures. Stout et al (1996) examined the effects of school-based health centres across three study sites. At study site A, relative to the control school, students who used the health centre reported a reduction in contraceptive use (p<0.05) and there was no difference in reproductive health outcomes. At site B, relative to the control site, clinic participants reported an increase in contraception use at first sex and an increase in contraception initiation in the first six months of initiating sex. However, no other programme effects were observed. Clinic participants at site C reported a decrease in sexual activity relative to participants at the control site (p<0.05), but among sexually active students there were no changes in contraceptive behaviour. The authors note that the emphasis on abstinence at this site might have contributed to the reduction in sexual activity.

6.6.4 Summary and evidence statements

A total of included seven studies were identified that reported on six programmes relating to other school-based approaches to sex and relationships education

6.6.4.1 Knowledge and understanding

No studies in this category reported on knowledge outcomes.

6.6.4.2 Attitudes and values

Three studies (Stout et al., 1996; Somers et al., 2001; 2006) that reported on school-based health clinics and an infant simulation intervention, respectively, reported few effects on attitudes. There were inconsistent effects on reproductive attitudes across three school health clinic sites (Stout et al., 1996), which appeared to be largely due to the different operational goals of the sites studied. Findings from the Baby Think It Over programme (Somers et al., 2001; 2006) indicated no effects of the intervention on attitudes and values towards child bearing or safe sex.

6.6.4.3 Personal and social skills

Only one study (Stout et al., 1996), which examined three school-based health clinics, reported programme effects on personal and social skills. Stout et al (1996) examined the impact of participation in school-based health clinics on sexuality communication between students and parents, but they proved to have no effect on this outcome.

6.6.4.4 Health and social outcomes relating to alcohol use and sexual health

There were no effects of the Baby Think it Over infant simulation programme on any of the sexual behaviour measures examined (Somers et al., 2001; 2006). Three studies (Lewis et al., 1999; Paine-Andrews et al., 1999; Vincent et al., 2004) that examined a school and community partnership

approach to tackling teenage pregnancy, generally found that although there were reductions in pregnancy rates among 14-17 year olds at the intervention sites, these reductions were not found to be significant compared to non-intervention sites. However, one study (Vincent et al., 2004) that examined pregnancy rates over 20 years concluded that the intervention had had a positive effect on teenage pregnancies. Two studies (Stout et al., 1996; Teitler, 1997), which examined the long-term effects of school-based health centres, found that these programmes did not have consistent effects on participant's sexual behaviour.

Evidence statement 13

- 13 (e) There is inconsistent evidence from one NRCT and two CBAs¹ to determine the effects of school-based clinics or an infant simulation intervention on knowledge, attitudes, values and personal and social skills.
- 13 (f) There is weak evidence from one NRCT and one CBA study² to suggest that infant simulation programmes have no effect on health outcomes related to sexual health. This evidence may be only partially applicable to the UK as studies were carried out in the USA and may not be generalisable beyond the populations studied.
- 13 (g) There is weak evidence from two CBA studies and one ITS³ to suggest that a comprehensive school- and community-based approach to teenage pregnancy may produce modest reductions in teenage pregnancy rates. This evidence may be only partially applicable to the UK as studies were carried out in the USA and may not be generalisable beyond the populations studied.
- 13 (h) There is inconsistent evidence from two CBAs⁴ to determine the effects of school-based clinics on health outcomes related to sexual health.
- ¹ Stout et al., 1996 (CBA -); Somers et al., 2001 (CBA -); Somers et al., 2006 (NRCT -)
- ² Somers et al., 2001 (CBA -); Somers et al., 2006 (NRCT -);
- ³ Lewis et al., 1999 (Interrupted time series -); Paine-Andrews et al., 1999 (CBA -); Vincent et al., 2004 (CBA -)

⁴ Stout et al., 1996 (CBA -); Teitler, 1997 (CBA -)

Study	Doting	Intervention	Comparator			Outcomes	
Sludy	Rating	intervention	Comparator	Follow-up	Knowledge	Attitudes and values	Skills
Somers et al., 2001	CBA -	Baby Think It Over n=84	No intervention n=16	6-10 weeks n=NR	-	NS contraceptive/sexual attitudes NS pre-marital sex NS pregnancy NS child bearing NS child rearing	-
Somers et al., 2006	NRCT -	Baby Think It Over n=113	No intervention n=117	PT n=NR	-	 ✓ Perception of others' acceptance about teen pregnancy* NS realism about child rearing NS future orientation NS sex and chid bearing intentions NS pre-marital sex 	-
*p≤0.05; **p≤0.0	01; ***p≤0.001; ^	Increase relative t	o comparator; 🗸 de	crease relative to co	mparator; NS not signific	cant; - outcome not reported	

Table 6.31. Other school-based SRE approaches: short-term programme effects on knowledge, attitudes and skills

Table 6.32. Other school-based SRE approaches: short-term programme effects on health and social outcomes

					Health outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Somers et al., 2001	CBA -	Baby Think It Over n=84	No intervention n=16	6-10 weeks n=NR	-	NS sexual behaviours	-	-	-
Somers et al., 2006	NRCT -	Baby Think It Over n=113	No intervention n=117	PT n=NR	NS age first had sex	NS frequency of sex; oral sex	NS contraception use	-	-
*p≤0.05; **p≤0.0	*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported								

Study	Doting	Intervention	Comparator	Follow-up	Outcomes				
Sludy	Rating				Knowledge	Attitudes and values	Skills		
Stout et al., 1996	CBA -	Site A n=739	Site A n=1131	2 years n=NR	2 years n=NR	-	NS reproductive health attitudes	NS sexuality dialog with parents	
		Site B n=354	Site B n=424			-	abstention before marriage*	NS sexuality dialog with parents	
		Site C n=1126	Site C n=525		-	A abstention before marriage*	NS sexuality dialog with parents		
*p≤0.05; **p≤0.	*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported								

Table 6.33. Other school-based SRE approaches: long-term programme effects on knowledge, attitudes and skills

Table 6.34. Other school-based SRE approaches: long-term programme effects on health and social outcomes

		Intervention		Follow-up	Health outcomes				
Study	Rating		Comparator		Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Lewis et al., 1999	ITS -	Multicomponent school and community intervention n=1 site	Non- intervention matched counties n=NR	3 years n=NR	-	-	-	-	NS pregnancy rates
Paine- Andrews et al., 1999	CBA -	Multicomponent school and community intervention n=4 sites	Non- intervention matched counties n=4 sites	NR	-	-	-	-	NS pregnancy rates NS birth rates
Vincent et al., 2004	ITS -	Multicomponent school and community intervention n=1 site	Non- intervention counties n=3 sites	20 years NA	-	-	-	-	
Stout et al., 1996	CBA -	Site A n=739	Site A n=1131	2 years n=NR	-	NS sexual activity	↓ contraception use*	↓ binge drinking*	NS emotional health NS health service utilisation

			Comparator	Follow-up	Health outcomes				
Study	Rating	Intervention			Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs	Conceptions
Stout et al., 1996	CBA -	Site B n=354	Site B n=424	2 years n=NR	-	NS sexual activity	 ↑ contraception use at first sex* ↑ initiation of contraception within first 6 months of sex* 	-	-
Stout et al., 1996	CBA -	Site C n=1126	Site C n=525	2 years n=NR	-		NS contraception use	-	-
Teitler 1997	CBA -	Students in schools with Health Resource Centres n=348 (at FU)	Students in schools without Health Resource Centres n=597 (at FU)	NR n=945	NS initiation of sex	NS past month sex	NS condom use at last sex; in last month	-	-
*p≤0.05; **p≤0.0)1; ***p <mark>≤0.001</mark>	; 🛧 increase relative	to comparator; 🗸	decrease relative t	o comparator; NS	onot significant; -	outcome not reporte	d	

6.7 Review of published economic evaluations

One study (Wang et al., 2000) was identified that met the criteria for inclusion in the review of published economic evaluations. Wang et al (2000) evaluated the cost-effectiveness and cost benefit of Safer Choices, an abstinence-plus programmes.

6.7.1 Review of Wang et al (2000)

6.7.1.1 Overview

The aim of the study by Wang et al (2000) was to evaluate the cost-effectiveness and cost benefits of Safer Choices, an abstinence-plus sex and relationships education programme, compared to a standard, information-based curriculum.

6.7.1.2 Summary of effectiveness data

Effectiveness estimates were drawn from an evaluation of the Safer Choices programme by Coyle et al (1999). The economic evaluation was based on 7-month follow-up data and a sample of 3677 ninth-grade students who had completed both the baseline and follow-up surveys. At the 7-month follow-up, the authors reported that a significant increase in condom and contraceptive use at last intercourse was found (both p<0.05); 67% of intervention students compared to 52% of the control group reported using condoms at last intercourse, and 68% of intervention students reported using contraception at last intercourse compared to 57% of the control group.

6.7.1.3 Summary of resource utilisation and cost data

The authors adapted the Bernoulli model to translate increases in condom use into cases of HIV and other STDs (Chlamydia, gonorrhoea, and pelvic inflammatory disease [PID]) averted. The model estimated the probability of HIV infection based on four variables: number of sexual partners, number of sexual contacts with each partner, HIV prevalence, and probability of transmission. In addition, a pregnancy model was developed to translate increases in contraceptive use into cases of pregnancy averted. Full details of the variables used in the model were reported in the article. To examine the cost-benefit of Safer Choices, cases of HIV, other STIs and pregnancies averted were translated into medical and social costs averted. Cost data for the base-case analysis were based on the private sector perspective, but the use of the both public and private sector costs were explored in sensitivity analyses. Medical costs for HIV, Chlamydia, gonorrhoea, PID and pregnancy were drawn from previously published studies or estimated by the authors. For societal costs related to HIV infection and pregnancy the authors used estimates drawn from previously published studies and assumptions made by the authors. Social costs including lost productivity for HIV infection, and costs relating to earning-related outcomes, public assistance, and other consequences for childbearing.

A retrospective cost analysis was conducted to estimate intervention costs which consisted of programme costs and the costs of condoms and oral contraceptives. Direct programme costs included teacher training, teaching, lesson materials and site coordination as shown in Table 36. The costs of developing and evaluating the programme were excluded. The costs of condoms and oral contraceptives were estimated as the product of the number of condoms or oral contraceptives used

within 1-year among intervention students and their sexual partners and the wholesale price of condoms or oral contraceptives. The total costs to implement the programme were estimated at \$102,852 (1994 US\$) and the costs of condoms and oral contraceptives were estimated at \$2,391, resulting in total intervention costs of \$105,243.

Component	Total cost, \$ (1994)
Teacher training	29,183.10
Teaching	16,400.00
Peer facilitators	25,436.80
Site coordinator training	1,640.00
Site coordination	23,616.00
Curriculum packages	3,904.00
Implementation materials	490.00
Activity kits	590.000
Photocopies for students	430.35
Photocopies for teachers	11.59
Videos	1150.00
Total	102,851.84

6.7.1.4 Summary of cost-effectiveness data

Overall, 0.12 cases of HIV, 24.30 cases of Chlamydia, 2.77 cases of gonorrhoea, 5.86 cases of PID and 18.50 cases of pregnancy were estimated to have been averted over a 1-year time horizon. The total costs averted were calculated at \$279,519 (\$139,806 medical costs and \$139,713 social costs). Overall, the net benefit of the programme was \$174,276 and the benefit-cost ratio was 2.65, indicating that for every \$1 spent on the programme, \$2.65 were saved in medical and societal costs.

The authors reported that the results were not cost-saving in two scenarios: (1) low probability of HIV or STD transmission, low percentage of students using contraceptives, high contraceptive failure rate, low medical costs, and low prevalence or incidence rate; and (2) low probability of HIV or STD transmission, low percentage of students using contraceptives, high contraceptive failure rate, low medical costs, and low condom use per act.

6.7.1.5 Comments

The economic evaluation answered a well-defined question, evaluating the cost-effectiveness and cost benefits of a school-based abstinence-plus programme relative to a standard, information-based HIV curriculum. Data on the impact of the programme on condom use were drawn from an RCT and translated into cases of health outcomes averted using an analytic model. The authors developed a static model with a 1-year time horizon, and so did not incorporate population-level transmission dynamics. The authors made several assumptions regarding the data used in the model, but the robustness of the results were examined in a multivariate sensitivity analysis. Costs were considered from a medical and societal perspective and the authors appeared to have considered all the important costs and consequences of the two perspectives adopted. However, the conclusions of the

analysis were not based on an overall index or ratio of costs to effects. In addition, the use of a static model and short time horizon means that the true benefits of the intervention may have been considerably underestimated. The generalisbility of the study to a UK context is unclear as the data used in the evaluation is based on studies conducted in the USA, and other US population estimates. However, the authors state that the methods and data used were conservative and it is possible that the intervention may be cost saving in a UK context.

6.7.2 Summary and evidence statements

One study (Wang et al., 2000) was identified that met the criteria for inclusion in the review of published economic evaluations. Wang et al (2000) evaluated the cost-effectiveness and cost benefits of a school-based sex and relationships education programme, Safer Choices. Evaluations of the effectiveness of this programme were identified and are included in Section 6.2.

Overall the net benefit of the Safer Choices programmes was \$174,276 and the benefit-cost ratio was 2.65, indicating that for every \$1 spent on the programme, \$2.65 were saved in medical and societal costs.

Evidence statement 14

There is moderate evidence from one economic evaluation study¹ to suggest that a sex and relationships education programme, Safer Choices, may be cost-effective and cost saving. This evidence may be of limited applicability to a UK context because cost and benefit estimates were based on data from studies conducted in the USA.

¹ Wang et al., 2000 (CEA +)

6.36. \$	Sex and	relationships	education:	summary of	of economic	evaluation s	studies
----------	---------	---------------	------------	------------	-------------	--------------	---------

Study details	Research question	Methods of estimation for	Results	Confounders, potential
		costs and benefits		sources of bias and other
				comments
Wang et al (2000)	Research question: To evaluate	The authors adapted the	Total intervention costs were estimated at	Intervention effectiveness
	the cost-effectiveness and cost	Bernoulli model to translate	\$105,243 and 0.12 cases of HIV, 24.30 cases	based on 7-month outcome
Country/currency:	benefit of Safer Choices	increases in condom use into	of Chlamydia, 2.77 cases of gonorrhoea, 5.86	data.
US\$ (1994)		cases of HIV and other STDs	cases of PID and 18.50 cases of pregnancy	
	Population: High school	(Chlamydia, gonorrhoea, and	were estimated to have been averted over a 1-	Model did not incorporate
CEA +	students and their sexual	pelvic inflammatory disease)	year time horizon. The total costs averted were	population-level transmission
	partners	averted. Pregnancy model was	calculated at \$279,519 (\$139,806 medical	dynamics
		developed to translate increases	costs and \$139,713 social costs).	
	Intervention: Safer Choices, a	in contraceptive use into cases		Short time horizon
	school based education	of pregnancy averted, and cases	In the cost-benefit analysis, the net benefit was	
	programme designed to prevent	of adverse health outcomes	\$174,276 and the benefit-cost ratio was 2.65.	No ratio of cost to benefits
	HIV, other STDs, and	averted were translated into		presented.
	pregnancy.	medical and social costs	Results were not cost-saving in two scenarios,	
		averted.	when the following variable estimates were	Study limited by some of the
	Perspective: Medical and		used: low probability of HIV or STD	assumptions made, e.g. all
	societal	A retrospective cost analysis	transmission, low percentage of students using	relationships were of a
		was conducted to estimate	contraceptives, high contraceptive failure rate,	heterosexual nature and that
		intervention costs which	low medical costs, and low prevalence or	the sexual partners of
		consisted of programme costs	incidence rate/low condom use per act.	intervention students were
		and the costs of condoms and		from a pool outside of the
		oral contraceptives.		intervention group.
			1	

7 General health education programmes

No systematic reviews or meta-analyses were identified for inclusion in the review of general health education programmes, and in addition no economic evaluation studies were identified. Nine articles that reported on evaluations of general health education programmes were identified. Studies were defined as general health education if they focused on a broad range of health behaviours, but incorporated as a minimum, a focus on both alcohol use and sexual health.

7.1 General health education programmes

7.1.1 Overview of evidence identified

Overall, nine studies (Bond et al., 2004; Flay et al., 2004; Harrington et al., 2001; O'Donnell et al., 1999; 2002; Moberg & Piper, 1990; McNeal et al., 2004; Patton et al., 2006; Piper et al., 2000) reported on the evaluation of six general health education programmes that reported relevant alcohol and sexual education outcomes. Two studies (O'Donnell et al., 1999; 2002) examined the effectiveness of the Reach for Health (RFH) curriculum, and the effects of the inclusion of an additional community youth service (CYS) component. Although both studies appeared to be based on the same sample of students from two middle schools, O'Donnell et al (2002) reported on a subsample of students who met eligibility criteria and dropped the 'no intervention' control group from the analyses. Two studies (Harrington et al., 2001; McNeal et al., 2004) examined the All Stars intervention programme, which was designed to reduce substance use, sexual behaviour and violence. McNeal et al (2004) extended the analyses presented in the study by Harrington et al (2001) by breaking down the composite substance use measure into separate types of drugs (cigarettes, cannabis, alcohol and inhalants). Bond et al (2004) and Patton et al (2006) examined the Gatehouse Project, a 2-year programme which was designed to improve emotional wellbeing and reduce health risk behaviours. The study by Bond et al (2004) was based on longitudinal follow-up of students who received the programme in year 8 and annually thereafter until the end of Year 10. Patton et al (2006) surveyed subsequent cross-sectional samples of year 8 students over five years. The two remaining studies examined the Healthy for Life health promotion programme (Piper et al., 2000), which targeted poor nutrition and eating patterns, tobacco, alcohol and cannabis use, and Flay et al (2004) examined the effects of a social development curriculum, with and without additional community components, delivered as part of the Aban Aya project.

Table 7.1. General health education programmes

Author	Study design and rating	Setting	Baseline population	Programme components	Theory	Provider
Bond et al., 2004	RCT ++	School	Australia n=2,678 13-14 years	Gatehouse Project: Multi-component study focusing on emotional well-being and risk behaviours carried out in 40 hours over 10 weeks. Addressing risk and protective factors and delivered in year 8	Not reported	Teachers and other staff members
Flay et al., 2004	RCT ++	School + community	USA n=644 10-12 years	Aban Aya: Three years, 16- 21 lessons per year; building cognitive behavioural skill, school + community intervention (SCI) and social development curriculum (SCD) groups. SCI = SDC + parental support, community development school climate		University-based health educators
Harrington et al., 2001	RCT +	School	USA n=2,289 students, mode 12 years	All Stars: 1 year; 14 entire class, four small group and four one on one sessions; debates, games and general discussion	Not reported	Teachers; External (specialists, All Star facilitators)
McNeal et al., 2004	RCT +	School	USA n=2,289 students, 11-13 years	All Stars: 22 sessions; substance use, sex and violence prevention	Social learning theory	Teachers; External (specialists, All Star facilitators)
Moberg & Piper, 1990	CBA +	School	USA n=197 students 12-14 years	Project Model Health: 32 hour programme delivered throughout the semester focusing on a range of health outcomes including reduction in drink driving and risk-taking attitudes.	Social learning theories	Teams of college- age instructors
O'Donnell et al., 1999; 2002	RCT +	School, community	USA n=1,157 students in 12-14 years	Reach for Health curriculum + Community Youth Service programme: 40 lessons and three hours per week community placements; curriculum focused on health risks related to substance use, sexual health, violence, STI, and unintended pregnancy.	Health belief model; social learning theory	Teachers
Patton et al., 2006	RCT +	School	Australia n=2,546 students, 13-14 years	Gatehouse Project : Three year programme including a 10 week problem solving curriculum in 8 th grade	Not reported	Teachers and other staff members
Piper et al., 2000	RCT -	School, community	USA n=2,483 students, 11-14 years	The Healthy for Life Programme: containing a peer, family and community component and focusing general health behaviours including alcohol use.	Social influences model	Teachers and peer leaders

Five of the six programmes were conducted in North America and the remaining study (the Gateway Project; Bond et al., 2004; Patton et al., 2006) was carried out in Australia. All six programmes were primarily delivered in schools with four solely school-based and two using combined school and community elements. Aban Aya (Flay et al., 2004) focused on building cognitive behavioural skills and consisted of two intervention groups; one using social development curriculum and one using the social development curriculum in conjunction with parental support and community development. Founded on social learning theory, the All Stars programme (Harrington et al., 2001; McNeal et al., 2004) employed a curriculum-based approach to prevent substance use, sex and violence. Project Model Health (Moberg & Piper, 1990) was curriculum-based and reported a range of health outcomes including drink driving and risk-taking attitudes. Founded on health belief models and social learning theory, the Reach for Health (RFH) curriculum (O'Donnell et al., 1999; 2002) with a community youth service (CYS) programme employed primarily school-based educational sessions along with community placements. Gatehouse Project (Bond et al., 2004; Patton et al., 2006) was a multicomponent programme carried out solely in schools focusing on emotional well-being and risk behaviours. The social learning theories-based programme Health for Life (Piper et al., 2000) focused on general health but had an alcohol prevention and sexual behaviour aspect and contained a peer. family and community element.

The RFH curriculum (O'Donnell et al., 1999; 2002) used only teachers to deliver the intervention. A further three studies (All Stars [Harrington et al., 2001; McNeal et al., 2004]; Gatehouse Project [Bond et al., 2004; Patton et al., 2006]; and the Healthy for Life programme [Piper et al., 2000]) supported teacher delivery with specialist facilitators, other staff members and peer leaders respectively. Two other studies, of the Aban Aya project (Flay et al., 2004) and Project Model Health (Moberg & Piper, 1990) were not provided by teachers but used university-based educators and teams of college-age educators, respectively, to deliver the programmes. Age and grade of participating students varied slightly in students receiving general health interventions. However, all studies reported participating students between ages 11-14 years.

Sample sizes varied across programmes from 197 students to 2,676 students. Evaluation of three programmes (All Stars [Harrington et al., 2001; McNeal et al., 2004]; Gatehouse Project [Patton et al., 2006; Bond et al., 2004]; and the Healthy for Life programme [Piper et al., 2000]) was based on a sample size greater than 1,500 students. Power calculations were reported in five of the nine studies (Harrington et al., 2001; McNeal et al., 2004; Patton et al., 2006; Bond et al., 2004; Piper et al., 2000). Studies were sufficiently powered in all five studies, however, one study recognised that some subsamples were smaller than desired and could have reduced the power and thus the precision of the estimates (Piper et al., 2000). Evaluations took place over differing follow-up periods. Evaluation of Project Model Health was based on follow-up at 20 months (Moberg & Piper, 1990). O'Donnell et al (1999; 2002) reported post-test results along with findings at two years for the Reach for Health Curriculum. The Gatehouse Project (Bond et al., 2004) was evaluated at one and two years follow-up. The study of the Healthy for Life Programme (Piper et al., 2000) followed-up students until 10th grade and follow up of participants in All Stars was conducted at one year post intervention. The second

study of the Gateway Project (Patton et al., 2006) consisted of three cross-sectional studies, conducted at 2-year intervals.

7.1.2 Quality assessment

Of the nine studies, eight were RCTs and one was a CBA study. Six RCTs (Flay et al., 2004; O'Donnell et al., 1999; 2002; Patton et al., 2006; Bond et al., 2004; Piper et al., 2000) were based on cluster randomisation and two studies (All Stars [Harrington et al., 2001; McNeal et al., 2004]) were based on individual randomisation. Cluster randomisation was carried out at school district (Gatehouse Project [Bond et al., 2004]), school (Aban Aya [Flay et al., 2004]; Gatehouse Project [Patton et al., 2006]; The Healthy for Life Programme [Piper et al., 2000]) and classroom (RFH + CYS [O'Donnell et al., 1999; 2002]) level. In three of the cluster RCTs the unit of randomisation did not match the unit of analysis. However, in all three cases (Flay et al., 2004; Bond et al., 2004; Piper et al., 2000) the analyses were adjusted for the intraclass correlations, with one study reporting an intention to treat analysis (ITT) (Bond et al., 2004). All but three of the RCTs were rated moderate (+ rating) for quality. Two RCT studies (Bond et al., 2004; Flay et al., 2004) were rated good quality (++ rating), as methods of randomisation were clear and appropriate, and design and outcome measures were clearly reported and reliable. One RCT (Piper et al., 2000) was rated poor quality (- rating) as the method of randomisation used was subject to bias. Although assignment to the intervention and control groups was random, schools allocated to the intervention arm chose which version of the curriculum to deliver. One CBA study (Moberg & Piper, 1990) was also rated as moderate quality. Attrition rates were unclear in this study. The reliability of outcome measures was not reported for one programme (RFH +CYS [O'Donnell et al., 1999; 2002]) and another did not report full details of outcomes (Harrington et al., 2001). However, outcome measures were judged to be relevant and reliable in all other studies. Follow-up times were deemed to be meaningful across all included studies.

7.1.3 Findings

7.1.3.1 Short-term results (<6 months)

Three studies (O'Donnell et al., 1999; Bond et al., 2004; Patton et al., 2006) reporting on two programmes, the RHS curriculum + CYS programme and the Gateway Project, respectively, reported immediate post-test programme effects.

Knowledge and understanding

None of the identified studies examined short-term intervention effects on knowledge and understanding.

Attitudes and values

Bond et al (2004) examined the effects of the Gateway Project on self-reported anxiety/depressive symptoms, finding that the intervention had no impact on this outcome.

Personal and social skills

Bond et al (2004) also examined the effects of the Gateway Project on student's social relationships in terms of their availability of attachments and conflictual relationships. There were no effects of the intervention on either of these measures.

Health and social outcomes relating to alcohol use and sexual health

Short-term programme effects of the RFH + CYS programme (O'Donnell et al., 1999) showed no effect of the RFH curriculum only, compared to a no intervention control, on sexual activity or an index of sexual behaviour¹¹. However, students who received the RFH curriculum and participated in CYS activities were less likely to report recent sexual activity (p<0.05) and reported lower scores on the sexual behaviour index (p<0.05), than control students. Two studies (Bond et al., 2004; Patton et al., 2006) examined the short-term impact of the Gateway Project. At the end of the 2-year intervention programme, both studies found that the intervention had no effects on alcohol or other substance use, and Patton et al (2006) also found no intervention effects on early initiation of sexual activity. At the end of the one-year All Stars programme there were no differences between intervention and control students in terms of their sexual activity. However, compared to both the teacher condition and control, specialist intervention students showed a higher mean level of substance use (p<0.05) (Harrington et al., 2001).

7.1.3.2 Medium-term results (up to 12 months)

Five studies (Harrington et al., 2001; McNeal et al., 2004; Bond et al., 2004; Piper et al., 2000; Moberg & Piper, 1990) reported medium-term programme effects for four programmes: All Stars; The Healthy for Life Programme; Gatehouse Project; Project Model Health.

Knowledge and understanding

One study (Moberg & Piper, 1990) examined programme effects relating to participants' knowledge of health. However, no effects were shown at one year follow-up.

Attitudes and values

Three studies (Bond et al., 2004; Moberg & Piper, 1990; Piper et al., 2000) examined short-term intervention effects on attitudes and values. Bond et al (2004) reported that there were no effects of the Gatehouse Project on self-reported anxiety/depressive symptoms. Moberg and Piper (1990) found significant effects of the Project Model Health programme; intervention students reported more positive attitudes towards sex¹² and reported fewer intentions to use substances, compared to control students. There were no intervention effects on self-esteem. Piper et al (2000) reported that participants receiving the intensive, 1-year version of the Healthy for Life Programme were less likely

¹¹ Four category index, scored as follows: (1) no lifetime experience with intercourse; (2) past but no recent intercourse; (3) recent, always protected intercourse (both condoms and contraceptive use); and (4) recent, unprotected intercourse.

¹² Belief that students should postpone the onset of sexual intercourse; students who are sexually active should use contraception.

to perceive high rates of substance use among their peers (p<0.001). Students who received the 3year, age appropriate version of the programme did no differ from control students on this measure.

Personal and social skills

None of the included studies examined medium-term intervention effects on personal and social skills.

Health and social outcomes relating to alcohol use and sexual health

Four studies (Harrington et al., 2001; McNeal et al., 2004; Bond et al., 2004; Moberg & Piper, 1990; Piper et al., 2000) examined medium-term effects on health outcomes relating to sexual health and alcohol for three general health programmes (All Stars; The Healthy for Life Programme; Gatehouse Project). There were no effects of the All Stars programme (McNeal et al., 2004) on sexual activity in the past month, but students who received the teacher-led version of the intervention showed a decrease in their use of alcohol in the past month, compared to controls (p<0.05). There was no impact of the intensive version of Healthy for Life programme on past month sexual intercourse, but students receiving the age appropriate intervention were more likely than control students to report having had sexual intercourse in the past month (p<0.05). At one year follow-up, students who received the age-appropriate version of the Healthy for Life Programme (Piper et al., 2000) were also more likely than control students to have used alcohol in the past month (p<0.05). There was no difference between intervention students who received the intensive version of the programme and control students on this measure. There were no medium-term effects of the Gatehouse project (Bond et al., 2004) on alcohol use measures, including any drinking and binge drinking. Findings from the evaluation of Project Model Health (Moberg & Piper, 1990) revealed no effects of the programme on past month alcohol use or alcohol use frequency. However, at the 1-year follow-up, students who participated in Project Model Health were less likely to report they had been sexually active in the past month, compared to controls (Moberg and Piper, 1990).

7.1.3.3 Long-term results (>12 months)

Five studies (Piper et al., 2000; Flay et al., 2004; O'Donnell et al., 1999; 2002; Patton et al., 2006) reported long-term findings for five programmes: The Healthy for Life Programme; Gatehouse Project; Aban Aya; Project Model Health; RFH + CYS programme.

Knowledge and understanding

None of the included studies examined long-term programme effects on knowledge and understanding.

Attitudes and values

At follow-up in 10th grade, there was no difference between intervention students who participated in the Healthy for Life programme and control students in their perceptions of peer substance use.

Personal and social skills

None of the included studies examined long-term programme effects on personal and social skills.
Health and social outcomes relating to alcohol use and sexual health

Five studies (Piper et al., 2000; Flay et al., 2004; O'Donnell et al., 2002; Patton et al., 2006) reported long-term finding on health outcomes for five programmes: The Healthy for Life Programme; Gatehouse Project; Aban Aya; RFH + CYS programme.

Two studies (Patton et al., 2006; O'Donnell et al., 2002) reporting on two programmes (Gatehouse Project; RFH + CYS programme) examined programme effects on sexual initiation. There were effects of the Gatehouse Project (Patton et al., 2006) on early sexual initiation. Results from a crosssectional survey of year 8 students taken two years after intervention showed that students in intervention schools were less likely to report early initiation of sexual intercourse (OR 0.55 95% CI: 0.37,0.83). At the 2-year follow-up, students who received the RFH + CYS programme (O'Donnell et al., 2002) were less likely to report sexual initiation after receiving either one year (OR 0.49; 95% CI: 0.25, 0.99) or two years (OR 0.32; 95% CI: 0.25, 0.99) of the combined programme, compared to students who received the RFH curriculum only. Students in the combined programme arm were also less likely to report recent sexual activity after one (OR 0.48; 95% CI: 0.24, 0.96) or two (OR 0.39; 95% CI: 0.20, 0.76) years of the programme. Piper et al (2000) found no change in past month sexual activity in 10th grade students who had received either the intensive or age appropriate version of Healthy for Life, despite finding a negative treatment effect on this measure at the ninth grade followup. Findings from the Aban Aya programme (Flay et al., 2004) at three years follow-up showed a decrease in sexual activity among males in the Social/Community Intervention (SCI) group (p<0.05) and an increase in condom use (p<0.05), but not in the Social Development Curriculum (SDC). A reduction in substance use was also seen among males in both intervention groups (p=0.05) and students in the SCI group also showed decreases in provoking behaviours and school delinguency (p<0.05). No significant programme effects were found for females on any of these measures. No programme effects on substance use were present at any cross-sectional observation periods in the Gatehouse Project (Patton et al., 2006). Students who participated in the Health for Life programme (Piper et al., 2000) reported increases in past month alcohol use in both the age appropriate and intensive intervention groups (p<0.05), in addition to an increase in the overall substance use scale for those students receiving the age appropriate intervention (p<0.05).

7.1.4 Summary and evidence statements

A total of nine studies (Bond et al., 2004; Flay et al., 2004; Harrington et al., 2001; O'Donnell et al., 1999; 2002; Moberg & Piper, 1990; McNeal et al., 2004; Patton et al., 2006; Piper et al., 2000) were identified that examined sex and relationship or alcohol education within a general health education programme.

7.1.4.1 Knowledge and understanding

Only one study (Moberg & Piper, 1990) examined programme effects on knowledge however results showed no difference between the intervention and control groups.

7.1.4.2 Attitudes and values

Two studies (Moberg & Piper, 1990; Piper et al., 2000) detailed the impact of their programmes on participants' attitudes and values. Findings from Project Model Health (Moberg & Piper, 1990) showed that at one year follow-up intervention students displayed improved health attitudes towards postponing sex and using contraceptives in addition to a positive reduction in their attitudes and intentions towards alcohol and other drugs. Piper et al (2000) found a medium-term impact on students' perceptions of peer attitudes towards alcohol and other drugs in those receiving the intensive condition of the Healthy for Life programme. However, by two years follow-up there was no effect.

7.1.4.3 Personal and social skills

No studies reported on programme effects on personal and social skills.

7.1.4.4 Health and social outcomes relating to alcohol use and sexual health

Nine studies (Bond et al., 2004; Flay et al., 2004; Harrington et al., 2001; O'Donnell et al., 1999; 2002; Moberg & Piper, 1990; McNeal et al., 2004; Patton et al., 2006; Piper et al., 2000) examined health outcomes relating to sex and relationship or alcohol education from general health education programmes. Only two programmes, the RFH curriculum (O'Donnell et al., 1999; 2002) and the Gatehouse Project (Patton et al., 2006), explored the intervention impact on age of sexual initiation. One programme (O'Donnell et al., 1999; 2002) incorporating a community element showed positive results at medium and long-term follow-up. Two programmes, the RFH curriculum (O'Donnell et al., 1999; 2002) and the Aban Aya project (Flay et al., 2004), incorporated a community element into their programme design and both reported a positive impact on sexual behaviour in their intervention groups exposed to the community element. The All Stars programme showed inconsistent results across two studies (Harrington et al., 2001; McNeal et al., 2004) and also produced an increase in substance use in the specialist condition. The Healthy for Life Programme (Piper et al., 2000) also produced long-term increases in alcohol use at two years follow-up. The Gatehouse Project (Bond et al., 2004; Patton et al., 2006) showed no programme effects for any health outcomes.

Evidence statement 15

- 15 (d) There is inconsistent evidence from one RCT and one CBA study¹ to determine the effect of general health education programmes on knowledge, attitudes and values relating to sexual health and alcohol use.
- 15 (e) There is moderate evidence from three RCTs² to suggest that general health education programmes incorporating an intensive community intervention element in conjunction with a curriculum base may have a positive effect on sexual behaviour and substance use. The evidence may only be partially applicable to the UK as programmes were implemented in the USA and focused primarily on black and minority ethnic groups. As such their generalisbility may be limited to the populations studied.
- 15 (f) There is moderate evidence from five RCTs³ to suggest that curriculum-based, general health education programmes have no impact on, and in some cases may have a negative impact

on, sexual behaviours and alcohol use. Studies were based on two programmes conducted in the USA and Australia and therefore the evidence may not be generalisable beyond the populations studied.

- ¹ Moberg & Piper, 1990 (CBA +); Piper et al., 2000 (RCT -)
- ² Flay et al., 2004 (RCT ++); O'Donnell et al., 1999, 2002 (both RCT +)
- ³ Harrington et al., 2001 (RCT +); McNeal et al., 2004 (RCT +); Piper et al., 2000 (RCT -); Bond et al., 2004 (RCT ++); Patton et al., 2006 (RCT +)

7.1.5 Summary tables

Table 7.2. General health education: short-term programme effects on knowledge, attitudes and skills

Study	Poting	Intervention	Comparator	Follow-up	Outcomes					
Study	Katiliy				Knowledge	Attitudes and values	Personal/social skills			
Bond et al., 2004	RCT ++	Gateway Project n=1,335	No intervention n=1,342	PT (>90%)	-	NS depressive symptoms	NS social relations			
*p≤0.05; **p≤0.01; ***	*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported									

Table 7.3. General health education: short-term programme effects on health and social outcomes

		Intervention	Comparator	Follow-up				
Study	Rating				Age of initiation	Frequency/ Number of partners	Contraceptive use	Alcohol use
Bond et al., 2004	RCT ++	Gateway Project n=1,335	No intervention n=1,342	PT (>90%)	-	-	-	NS any drinking NS regular use NS binge drinking
Harrington et al., 2001	RCT +	All Stars Specialist condition, n=269 Teacher condition, n=287	n=739	PT (1 year)	-	NS sexual activity	-	↑ past month substance use (specialist condition only*)
O'Donnell et al., 1999	RCT +	Reach for Health, n=222 ^a Reach for Health + CYS, n=255 ^a	No intervention n=584 ^a	PT n=1,061 (92%)	-	 ✓ recent sex (RHS + CYS only*) ✓ sexual risk behaviour index (RHS + CYS only*) 	-	-
Patton et al., 2006	RCT +	Gateway Project n=1,158	No intervention n=1,428	PT n=NA ^c	NS early initiation	-	-	NS substance use
*p≤0.05; **p≤0.01; ** ^a only students who c	*p≤0.001; ↑ in completed both	crease relative to con baseline and follow-u	nparator;	relative to com eligible samp	nparator; NS not sign le at 10 th grade; ^c follo	ificant; - outcome not	reported s-sectional surveys	

Study	Poting	Intervention	Comparator	Follow-up	Outcomes			
Study	Kating	Intervention		Follow-up	Knowledge	Attitudes and values	Personal/social skills	
Bond et al., 2004	RCT ++	Gateway Project n=1,335	No intervention n=1,342	1 year (>90%)	-	NS depressive symptoms	NS social relations	
Moberg & Piper, 1990	CBA +	Project Model Health n=115	No intervention n=82	1 year	NS health knowledge	 ↑ sexual attitudes to postpone sex and use contraception* ↓ substance use and intent index* NS self-esteem 	-	
Piper et al., 2000	RCT -	Healthy for Life Age appropriate, n=827 Intensive, n=758	No intervention n=898	9 th grade	-	↓ perceptions of peer substance use (intensive condition only***)	-	
*p≤0.05; **p≤0.01; **	*p≤0.001; ↑ in	crease relative to con	nparator; 🗸 decrease	relative to con	nparator; NS not significant;	- outcome not reported		

Table 7.5. General health education: medium-term programme effects on health and social outcomes

	Rating	Intervention	Comparator	Follow-up	Sex			
Study					Age of initiation	Frequency/ Number of partners	Contraceptive use	Alcohol use
Bond et al., 2004	RCT ++	Gatehouse Project n=1,335	No intervention control n=1,343	1 year (>90%)	-	-	-	NS Any drinking NS Regular drinker NS Binge drinking
McNeal et al., 2004	RCT +	All Stars Specialist condition, n=NR Teacher condition, n=NR	Normal health education n=NR	1 year n=1822 (72%)	-	NS past month sexual activity	-	↓ past month alcohol use (Teacher condition only*)
Moberg & Piper, 1990	CBA +	Project Model Health n=115	No intervention n=82	1 year (74%)	-	✓ sexual activity, past month**	-	NS alcohol use, past month NS alcohol use, quantity/frequency

	Rating	Intervention	Comparator	Follow-up	Sexual health outcomes			
Study					Age of initiation	Frequency/ Number of partners	Contraceptive use	Alcohol use
Piper et al., 2000	RCT -	Healthy for Life Age appropriate, n=827 Intensive, n=758	No intervention n=898	9 th grade (80%)	-	↑ sexual intercourse, past month (age appropriate only***)	-	↑ alcohol use, past month (intensive condition only*)
*p≤0.05; **p≤0.01; ***	'p≤0.001; ↑ ind	crease relative to com	parator; 🗸 decrease i	relative to com	parator; NS not signifi	cant; - outcome not re	ported	

Table 7.6. General health education: long-term programme effects on knowledge, attitudes and skills

Study	Rating	Intervention	Comparator	Follow-up	Outcomes				
Sludy					Knowledge	Attitudes and values	Personal/social skills		
Piper et al., 2000	RCT -	Healthy for Life Age appropriate, n=827 Intensive, n=758	No intervention n=898	10 th grade (68%)	-	NS perceptions of peer substance use	-		
*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported									

Table 7.7. General health education: long-term programme effects on health and social outcomes

		Intervention	Comparator	Follow-up	Health outco			
Study	Rating				Age of initiation	Frequency/ Number of partners	Contraceptive use	Alcohol use
Patton et al., 2006	RCT +	Gateway Project n=966 ^a	No intervention n=1,497 ^a	2 years NA ^a	$oldsymbol{\Psi}$ early initiation	-	-	NS substance use
Flay et al., 2004	RCT ++	Aban Aya Social development curriculum, n=204	Health Enhancement	3 years	-	NS recent sex	NS condom use	
		Aban Aya School/ community, n=185	n=184		-		↑ condom use (boys only*)	✓ substance use (boys only*)

					Health outco					
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	Alcohol use		
Piper et al., 2000	RCT -	Healthy for Life Age appropriate, n=827 Intensive, n=758	No intervention n=898	10 th grade (68%)	-	NS sexual intercourse, past month	-	↑ alcohol use, past month*		
O'Donnell et al., 2002	RCT +	RFH + CYS (1 or 2 programme years) n=NR	RFH curriculum only (1 or 2 programme years) n=NR	2 years n=195 (77% ^b)			-	-		
*p≤0.05; **p≤0.01; ** ^a follow-up based on	*p≤0.05; **p≤0.01; ***p≤0.001; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported ^a follow-up based on cross-sectional surveys									

8 Discussion

8.1 Alcohol and drug education programmes

A total of 119 articles met the criteria for inclusion in the review of alcohol and drug education programmes. Fourteen articles were systematic reviews and meta-analyses, 103 articles reported on the evaluation of an alcohol or substance use education programme, and two articles were economic evaluation studies. Of the 103 articles, 74 reported on evaluations of classroom-based programmes; 20 of which were alcohol specific, and 54 of which focused on substance use including alcohol. Also identified were 15 articles that reported on evaluations of brief or single session interventions, nine articles reporting on two multicomponent school- and community-based programmes and five articles reporting on evaluations of peer support and/or counselling programmes.

8.1.1 Systematic reviews and meta-analyses

A total of 14 systematic reviews and meta-analyses were identified for inclusion. The majority of the reviews identified examined the effectiveness of programmes targeting substance use including alcohol, and only three reviews (Foxcroft et al., 2002; Loveland-Cherry, 2003; Spoth et al., 2008) focused specifically on the prevention of alcohol use. Foxcroft et al (2002) found that there was no consistent evidence to determine which programmes were effective over the short to medium-term, but highlighted three programmes which were effective over the longer term. These included the family-based, Strengthening Families programme, and two school-based programmes, Botvin's LST and a culturally-focused curriculum for Native American students. Spoth et al (2008) highlighted promising evidence from six additional programmes, Keepin it REAL, the Midwest Prevention Project, Project Northland, Healthy School and Drugs, Project ALERT, and SHAHRP. Two reviews (Cuijpers, 2002; Gottfredson and Wilson, 2003) identified evidence to suggest that peer leaders strengthened the effects of school-based interventions, although Gottfredson and Wilson (2003) found that any beneficial effects of peer involvement were lost when they were combined with teacher-led activities. White et al (2004) did not find any evidence to suggest that any particular agency or external contributor was more effective than another.

8.1.2 Classroom-based programmes

A total of 74 articles were identified for inclusion that reported on evaluations of classroom-based programmes. Overall, 20 articles were identified for inclusion that reported on the evaluation of 12 alcohol education programmes across 15 studies and 54 articles were identified that examined 22 classroom-based substance use (including alcohol) prevention programmes across 34 studies.

Of the 15 studies identified for inclusion that examined alcohol education programmes, nine studies were RCTs, three were NRCTs and three were CBA studies. The 12 alcohol education programmes were primarily classroom-based curriculums, but two programmes incorporated additional materials and activities for parents. The programmes identified targeted students across a range of age groups; eight programmes targeted students aged 14 or younger and four programmes were targeted at older

adolescents. Across eight studies (Bagnall, 1990; McBride et al., 2004; Morgenstern et al., 2009; Newton et al., 2009; Schnepf, 2002; Vogl et al., 2009; Newman et al., 1992; Shope et al., 1996a) that examined intervention effects on knowledge related to alcohol use there were indications that alcoholspecific education programmes generally increased alcohol or curriculum knowledge over the shortterm. However, effects on medium- and long-term knowledge acquisition were weaker. Eight studies (Bagnall, 1990; Baumann, 2006; McBride et al., 2004; Morgenstern et al., 2009; Newton et al., 2009; Schnepf, 2002; Vogl et al., 2009; Wilhelmsen et al., 1994) examined young people's alcohol-related attitudes and values, finding non-significant programme effects across the majority of programmes. However, the SHAHRP programme (McBride et al., 2004), which was based on a harm reduction approach, had positive short- and long-term effects on students' alcohol-related attitudes. Short-term increases in safer alcohol-related attitudes were also reported by Wilhelmsen et al (1994) who examined a highly-role specific programme compared to a less-role specific alcohol programme. Few studies examined intervention effects on personal and social skills. Intervention impacts on a range of alcohol-related measures were examined across the included studies. The SHAHRP programme (McBride et al., 2004) appeared to have the most consistent effects on short-term alcohol use, and additionally had effects on hazardous/harmful drinking. Medium- to long-term effects on alcohol consumption were found to be limited. Studies were either methodologically poor, as in the case of the AAPT programme, or reported conflicting or diminished effects. For example, 17-months after delivery of the SHAHRP programme (McBride et al., 2004) the positive short-term effects appeared to be declining. Although intervention effects favoured SHAHRP, differences between intervention and control students in terms of their alcohol consumption and other measures of alcohol use including harmful/hazardous drinking were no longer significant. There were no long-term effects of a longer term version of the AMPS programme (Shope et al., 1996a) on alcohol consumption, but there did appear to be intervention effects on alcohol misuse.

Of the 34 studies, which examined 22 classroom-based substance use (including alcohol) prevention programmes, 23 were based on RCT designs, seven were NRCTs and four were CBA studies. Although all of the programmes were primarily classroom-based, five programmes combined school components with family- and/or community-based components. Two studies (Perry et al., 2003; Spoth et al., 2002, 2005, 2008) combined two originally school-based only programmes (DARE and LST, respectively) with components targeting parents. The majority of programmes targeted students aged 14 or younger. Four studies (Botvin et al., 1990a; Botvin et al., 2001a; Shope et al., 1996b; Cuijpers et al., 2001), including evaluations of LST, AMPS, DARE, and the Healthy School and Drugs Project, respectively, examined programme effects on alcohol-related knowledge. Overall both short and medium-term increases in alcohol knowledge were reported but these were not sustained long-term. Twenty-one studies reported outcomes relating to alcohol or substance use attitudes and behavioural intentions. There were inconsistent effects on attitudes towards alcohol use and peer norms, but eight studies (Botvin et al., 1995b; 1997; 2001a; Caplan et al., 1992; Eisen et al., 2002; Lennox & Cecchini, 2008; Perry et al., 2003; Hecht et al., 2003), which examined behavioural intentions indicated generally positive programme effects on intentions to drink or get drunk. Intervention effects on personal and social skills were examined across a small number of programmes but found to be

inconsistent. Four programmes, the Positive Youth Development Programme, the Unplugged programme a revised version of Project Alert, and the Healthy School and Drug Project had positive short-term programme effects on alcohol use. However, the findings of the Positive Youth Development Programme (Caplan et al., 1992) were limited by the poor quality of the study. Positive longer term effects were demonstrated for two programmes, Keepin It REAL (Hecht et al., 2003) and Be Under Your Own Influence/All Stars (Slater et al., 2006), which combined school and media intervention components. The strongest evidence of effectiveness came from a series of studies which examined Botvin's LST (Botvin et al., 1990a; 1990b; 1995b; 1997; 2001a; Fraguela et al., 2003). Two studies (Botvin et al., 1997; 2001a) found positive short- and medium-term effects on drinking frequency and binge-drinking, and these were sustained long-term (Botvin et al., 1995b; 2001a). However, replication of the programme by other research groups suggests that there may be issues with the transferability of LST to other settings (Fraguela et al., 2003; Smith et al., 2004; Spoth et al., 2005; 2008).

8.1.3 Brief behavioural or single session interventions

A total of 15 articles were identified that reported on evaluations of seven brief or single session intervention approaches across 13 studies. Of the 13 studies, 11 studies were RCTs, one was an NRCT and one was a CBA study. All 13 studies were primarily school-based but six studies examined interventions which incorporated materials targeting parents. A range of providers were utilised including school nurses, physicians, teachers, fitness professionals, consultants, trained research staff and motivational speakers. Two studies were based on mailed intervention materials and therefore did not involve a provider in the delivery.

None of the studies examined intervention effects on knowledge or understanding but 11 studies (Argentos, 1991; Dempster et al., 2006; Werch et al., 1996a; 1996b; 2000a; 2003b; 2005a; 2005b; 2005c; 2008a) examined intervention effects on students' attitudes and values. Across these studies, there was an indication of positive intervention effects in the short-term, resulting in increases in negative views of alcohol and/or its consequences and a decrease in alcohol-related expectancies. Brief behavioural or single session intervention approaches appeared to have inconsistent short- and medium-term effects on student's intentions to drink. Few studies examined intervention effects on personal and social skills, and for two studies (Werch et al., 2003b; 2005b) that impacts on selfcontrol and parent-child relationships intervention effects were mixed. Eight studies (Werch et al., 1996a; 1996b; 1998; 2000a; 2000b; 2003b; 2005a; 2005b) that examined STARS for Families and Project SPORT, two brief intervention approaches based on nurse consultations, indicated mixed, but generally positive effects of this programme approach on alcohol consumption and heavy drinking in the short- to medium-term. An alcohol tailored beverage programme (Werch et al., 2005c) and a brief intervention founded on the Behaviour-Image Model (Werch et al., 2008a) had inconsistent effects on alcohol use, and two further programmes, a single session on the dangers of binge drinking (Dempster et al., 2006), and the one week Programme "Kickoff" (Argentos, 1991), had no effects on alcohol use.

8.1.4 Multicomponent school- and community-based programmes

Nine studies (Perry et al., 1996; Komro et al., 1999; 2001; 2008; Perry et al., 2002; Williams et al., 1995; Toomey et al., 1996; Johnson et al., 1990; Chou et al., 1998) were identified that examined two multicomponent, school- and community-based programmes: Project Northland and the Midwest Prevention Project. Both programmes were based in communities in the USA and combined comprehensive school-based curriculums, with community-based activities and parental involvement components. All nine studies identified for inclusion were based on an RCT design.

None of the studies examined intervention effects on knowledge or understanding, or personal and social skills. There no were effects of Project Northland on attitudes and values and neither of the studies of the MPP examined intervention effects on these outcomes. Project Northland significantly reduced growth in binge drinking and tendency to use alcohol during Phase I and II of the programme (Perry et al., 1996; 2002; Komro et al., 2001), however, during the interim phase of the programme the growth in alcohol use was greater among intervention students than control students. The three-year MPP (Johnson et al., 1990; Chou et al., 1998) did not have significant effects on alcohol use in one cohort of ninth/tenth grade students, but a short-term secondary prevention effect was reported in a second cohort.

8.1.5 Peer support and counselling programmes

Five studies examined five peer support and counselling programmes. Of the five studies identified for inclusion, one study was based on an RCT design; two studies were NRCTs; and two studies were based on CBA designs. All five programmes were school-based only and the provider for three peer support programmes (Colnes, 2001; Padget et al., 2005; Webster et al., 2002) was peers alone. The two counselling programmes (Bremberg & Arborelius, 1994; Valentine et al., 1998) were delivered by health counsellors or educational psychology students, respectively.

None of the studies examined intervention effects on knowledge or understanding, or personal and social skills. Three studies (Colnes, 2001; Padget et al., 2005; Bremberg & Arborelius, 1994) examined short-term intervention effects on attitudes and values. For two peer leadership programmes there appeared to be modest impacts on attitudes to alcohol and one study of a counselling programme found that the programme had a positive impact on the number of psychological problems that students' attributed to their alcohol use (Bremberg & Arborelius, 1994). Neither of the counselling programmes (Bremberg & Arborelius, 1994; Valentine et al., 1998) was shown to be effective in reducing alcohol consumption, and one programme (Valentine et al., 1998) had potentially harmful effects on high school students' alcohol consumption. There were inconsistent effects of peer support programmes on alcohol use.

8.1.6 Review of published economic evaluations

Two studies (Swisher et al., 2004; Pentz, 1998) were identified that met the criteria for inclusion in the review of published economic evaluations. Swisher et al (2004) assessed the cost-effectiveness of standard and infused LST, and Pentz (1998) assessed the costs, benefits and cost-effectiveness of the MPP. The standard LST programme was found to be more cost effective than I-LST by \$33.46

per student after 1 year of intervention delivery. In the second year, however, standard LST had no effects and the authors concluded that I-LST was more cost-effective. The 3-year total costs of the two programmes were estimated at \$109,429.04 and \$93,088.17, respectively. The results of the cost-benefit analysis (CBA) of the MPP demonstrated a \$700 net saving per family per year resulting from a reduction in the incidence of monthly drunkenness. Cost benefits ratios were also shown to be favourable (ratio to \$1 spent on prevention to saving is \$1:1.69). Compared to "usual" drug education the ICER of the MPP was reported to be equal to the ratio of its incremental cost per incremental effects, equivalent to \$10 per net reduction in the incidence of monthly drunkenness.

8.2 Sex and relationships education

A total of 75 articles met the criteria for inclusion in the review of sex and relationships education programme. Nine articles were systematic reviews and meta-analyses, 65 articles reported on evaluations of sex and relationships education interventions, and one article was an economic evaluation study.

8.2.1 Systematic reviews and meta-analyses

Nine systematic reviews evaluated abstinence only and abstinence plus safer-sex promotion programmes (Underhill et al., 2007, 2008; Bennett & Assefi, 2005), safer-sex promotion (Oakley et al., 1995; Franklin et al., 2007; Kirby et al., 1994; Pedlow and Carey, 2003; Robin et al., 2004) and sexuality-focused interventions (Sales et al., 2006). Findings from three reviews that examined abstinence-only and abstinence-plus programmes (Underhill et al., 2007, 2008; Bennett and Assefi, 2005) indicated that abstinence-only programmes have limited effects or are ineffective for preventing or reducing sexual risk behaviours. In addition, Oakley et al (1995) found evidence to suggest that abstinence only education may have an adverse effect and actually increase sexual experimentation among students. For programmes that incorporated information on safe sex and use of contraception, there was evidence from five reviews (Underhill et al., 2008; Pedlow & Carey, 2003; Franklin et al., 1997; Kirby et al., 1994; Oakley et al., 1995) to suggest that interventions may have effects on preventing sexual risk behaviours, but that these effects tend to be modest. There was no evidence that sexuality and AIDS education increased sexual activity.

8.2.2 UK-based studies

Twelve UK studies, evaluating seven programmes, were identified that could be defined as predominantly sex and relationships education. Of the 12 studies, six were RCTs, two studies were based on an NRCT design, and five were CBA studies. Seven studies (Henderson et al., 2007; Mellanby et al., 1995; 2001; Stephenson et al., 2004; 2008; Tucker et al., 2007; Wight et al., 2002) reported on evaluations of three comprehensive school-based programmes, A PAUSE, RIPPLE and SHARE, respectively. Two studies (Gillies et al., 1990; Bellingham et al., 1993) reported on evaluations of the Streetwize UK, AIDS education comic, Denman et al (1995) examined a theatre in education programme and two studies (Graham et al., 2002; Magnusson et al., 2004) examined single lessons on emergency contraception and contraceptive services, respectively.

Across four studies (Tucker et al., 2007, Wight et al., 2002; Stephenson et al., 2004; Mellanby et al., 2001) that examined three comprehensive school-based programmes, there were indications that these programmes had significant effects on knowledge about STIs. Three studies (Denman et al., 1995; Bellingham and Gillies, 1993; Gillies et al., 1990) that examined a theatre in AIDS/HIV education programme and the Streetwize UK comic, respectively, had positive impacts on knowledge about HIV, and a teacher-led intervention about emergency contraception (Graham et al., 2002) increased knowledge relating to contraception. The effects of three comprehensive SRE programmes on attitudes were mixed. There were inconsistent or no effects of the peer-led RIPPLE programme and A PAUSE, but there were positive programme effects of the SHARE programme (Tucker et al., 2007) on attitudes concerning condom use and STI prevention, on self-efficacy to use condoms. Other intervention approaches focusing on HIV prevention (Denman et al., 1995; Bellingham and Gillies, 1993; Gillies et al., 1990) and emergency contraception (Graham et al., 2002) had limited impacts on attitudes and values. Results for skills outcomes were limited with few studies reporting on these outcomes. In addition, few of the studies found significant programme effects on health outcomes related to sexual health. There were positive medium- and long-term effects of the A PAUSE (Mellanby et al., 1995) and RIPPLE (Stephenson et al., 2004) programmes on the number of students who reported ever having had sex, although for RIPPLE this effect was only apparent among females at the 18-month follow-up. There were no effects of either the RIPPLE (Stephenson et al., 2004; 2008) or SHARE (Wight et al., 2002) programmes on use of condoms or other forms of contraception. There was mixed evidence on the effects of these programmes on pregnancy. There were no medium-term effects of the SHARE (Wight et al., 2002) or RIPPLE (Stephenson et al., 2004) programmes on rates of unintended pregnancies, however, Stephenson et al (2008) found that at age 20, students who participated in the peer-led RIPPLE programme were less likely to have been pregnant. There were no long-term effects of the RIPPLE (Stephenson et al., 2008) or SHARE (Henderson et al., 2006) programmes on abortion rates. Other intervention approaches focusing on HIV prevention (Denman et al., 1995; Bellingham and Gillies, 1993), and single lessons about emergency contraception (Graham et al., 2002) and contraceptive services (Magnusson et al., 2004), respectively, had no impact on sexual behaviours.

8.2.3 Abstinence-only programmes

Ten articles were identified that evaluated eight programmes defined as abstinence-only programmes across nine studies. These programmes encouraged and promoted abstinence as the best and only way to prevent pregnancy, HIV and other STIs. Of the nine studies, two were RCTs and seven were NRCTs. Four programmes were teacher-led and one was peer-led. The provider for three programmes was not reported. Abstinence-only programmes were generally targeted at students younger than 14 years, with the exception of the Sex Can Wait programme, which consists of upper elementary, middle and high school components.

Knowledge outcomes were reported in six studies (Blake et al., 2001; Borawski et al., 2005; Denny et al., 1999; Denny and Young, 2006; Jorgensen et al., 1993; Trenholm et al., 2008) and overall abstinence-only programmes appeared to be effective at increasing knowledge related to STIs in the

short- to medium-term. Evidence was lacking on the longer term effects on knowledge. Eight studies examined effects on attitudes and values, and abstinence-only programmes were generally found to have had a positive effect on participants' beliefs and attitudes towards abstinence, and no programmes reported an adverse programme effect on attitudes towards abstinence or intentions to have sex. Results from the studies that examined intervention effects on parent-child communication indicated that abstinence-only programmes did not affect communication. Eight studies (Christopher & Roosa, 1990; Blake et al., 2001; Borawski et al., 2005; Denny et al., 1999; Denny & Young, 2006; Jorgensen et al., 1993; Roosa & Christopher, 1990; Trenholm et al., 2008) demonstrated that abstinence-only programmes had non-significant or inconsistent effects on the initiation of sexual activity. Two studies (Borawski et al., 2005; Trenholm et al., 2008) reported outcomes relating to contraception and no impact on contraception use. One abstinence-only programme, Success Express (Christopher and Roosa, 1990), had a negative effect; at follow-up intervention students reported greater lifetime sexual experience than controls. However, in a replication study of this programme (Roosa & Christopher, 1990) there was no difference between intervention and control students on this measure.

8.2.4 Abstinence-plus programmes

A total of 24 articles were identified that reported on the evaluation of 15 abstinence plus programmes across 18 studies. Abstinence plus programmes were defined as those that reported an emphasis on abstinence as the safest way to avoid HIV/STI infection and pregnancy, but also promoted safer sex through the use of contraceptives. Of the 18 studies identified, ten were RCTs and eight were NRCTs. Two programmes, Safer Choices and YAPP, incorporated activities for parents and one programme, Protection Express, was exclusively peer-led. The programmes identified tended to target older adolescents (>14 years) or students across a range of ages, from the age of 13 years and upwards.

Fourteen studies (Aarons et al 2000; Borawski et al., 2009; Boyer & Shafer, 1997; Coyle et al., 1999; 2001; 2004; Eisen et al., 1990; Kirby et al., 1991; LaChausse, 2006; Siegel et al., 1998; 2001; Walter & Vaughan, 1993; Wright, 1998; Zimmerman et al., 2008) examined intervention effects on sexual health knowledge, finding that abstinence-plus programmes were generally effective in improving sexual health knowledge in the short- and medium-term. In addition, three studies (Coyle et al., 2001; Kirby et al., 1991; Wright, 1998) reported sustained long-term increases in knowledge among students exposed to abstinence plus programmes compared to controls. The results of 14 studies (Aarons et al., 2000; Borawski et al., 2009; Coyle et al., 1999; Coyle et al., 2004; Eisen et al., 1990; Kirby et al., 1991; LaChausse, 2006; Levy et al., 1995; Siegel et al., 1998; 2001; Stanton et al., 2006; Walter & Vaughan, 1993; Wright, 1998; Zimmerman et al., 2008) demonstrated that intervention effects on behavioural intentions, attitudes to sexual behaviour, and self-efficacy were inconsistent and there was no clear indication of the direction of effects. In addition, some programmes, such as Draw the Line/Respect the Line, had differing effects on male and female students. Across five programmes that incorporated skills building activities (Borawski et al., 2009; Boyer & Shafer, 1997; Coyle et al., 1999; 2001; Kirby et al., 1991; Wright, 1998) there were positive short- to medium-term effects on skills relating to sexual risk prevention. There was no indication of short-term intervention

effects on the initiation of sexual involvement and at the medium-term follow-up, programme effects were largely inconsistent with four studies (Borawski et al., 2009; Caron et al., 2004; Kirby et al., 1991; Wright, 1998) reporting no effects on initiation of sexual intercourse. In addition, intervention effects on frequency of sexual intercourse, number of sexual partners and contraceptive use were found to be inconsistent or non-existent in the short-, medium- and long-term. Two studies (Eisen et al., 1990; Kirby et al., 1991) examined the medium- to long-term effects of a HLM-SLT curriculum and Reducing the Risk, respectively, on pregnancy. Neither study identified significant programme effects on this outcome.

8.2.5 HIV and sexual risk-reduction programmes

Overall, 11 studies were identified that examined HIV and sexual risk-reduction programmes. Studies were defined by their specific focus on HIV prevention and HIV risk-behaviour, sexual risk-behaviour or a combination of both. Of the 11 included studies, eight were RCTs, two were NRCT and one was based on a CBA study design. All programmes were delivered in school time however one programme included six computer-based activities to be completed outside of school time. The included studies focused on different ages and school years. One programme targeted students aged 12-14 years, but in general, programmes were targeted at older adolescents.

Seven studies (Borgia et al., 2005; Coyle et al., 2006; Fisher et al., 2002; Larsson et al., 2006; Roberto et al., 2007; Schaalma et al., 1996; Workman et al., 1996) examined intervention effects on general HIV and sexual health knowledge, finding significant effects on knowledge over the short-term. Two studies (Coyle et al., 2006; Larsson et al., 2006), which explored medium- to long-term effects on knowledge of HIV and contraception reported significant effects. Outcomes relating to attitudes were reported by seven studies (Borgia et al., 2005; Coyle et al., 2006; Fisher et al., 2002; Lemieux et al., 2008; Larsson et al., 2006; Roberto et al., 2007; Schaalma et al., 1996). Condom self-efficacy, perception of social norms and condom use/prevention intentions were the outcomes most commonly reported across these studies but intervention effects were found to be inconsistent with no clear direction of effect. Short-term programme effects on personal and social skills were predominantly positive and included positive programme effects on behavioural prevention skills and condom negotiation skills (Borgia et al., 2005; Fisher et al., 2002; Lemieux et al., 2008; Roberto et al., 2007; Workman et al., 1996). However, none of the included studies examined medium- to long-term impacts on skills. Intervention impacts on sexual initiation were explored in four studies (Roberto et al., 2007; Coyle et al., 2006; Kvalem et al., 1996; Mitchell-DiCenso et al., 1997), which overall indicated inconsistent effects on this outcome. Further studies indicated no impact on sexual activity or the numbers of sexual partners. Six studies (Fisher et al., 2002; Lemieux et al., 2008; Coyle et al., 2006; Kvalem et al., 1996; Larsson et al., 2006; Traeen, 2003) showed positive short-term programme effects on condom use and protected intercourse, but longer term programme effects on contraception use appeared to be limited. Limited outcomes were presented on HIV/STI testing, alcohol or drug use and pregnancy.

8.2.6 Other school-based approaches

Seven studies were identified that reported on six different programme approaches relating to sexual health; one was a NRCT, five were CBA studies and one study was based on an interrupted time series design. Two studies examined school-based clinic programmes (Stout et al., 1996; Teitler, 1997), two studies (Somers et al., 2001; 2006) reported findings from one programme using baby simulators (Baby Think it Over), and three studies (Lewis et al., 1999; Paine-Andrews et al., 1999; Vincent et al., 2004) examined the effects of combined community and school-based programmes.

None of the studies examined intervention effects on knowledge or understanding. Three studies (Stout et al., 1996; Somers et al., 2001; 2006), which examined school-based health clinics and an infant simulation intervention, respectively, reported inconsistent effects on attitudes towards sexual health and only one study examined programme effects on personal and social skills. There were no effects of the Baby Think it Over infant simulation programme on any of the sexual behaviour measures examined (Somers et al., 2001; 2006). Three studies (Lewis et al., 1999; Paine-Andrews et al., 1999; Vincent et al., 2004) that examined a school and community partnership approach to tackling teenage pregnancy, generally found that although there were not found to be significant compared to non-intervention sites. However, one study (Vincent et al., 2004) that examined pregnancy rates over 20 years concluded that the intervention had had a positive effect on teenage pregnancies. Two studies (Stout et al., 1996; Teitler, 1997), which examined the long-term effects of school-based health centres, found that these programmes did not have consistent effects on participant's sexual behaviour.

8.2.7 Review of published economic evaluations

One study (Wang et al., 2000) was identified that met the criteria for inclusion in the review of published economic evaluations. Wang et al (2000) evaluated the cost-effectiveness and cost benefits of a school-based sex and relationships education programme, Safer Choices. Overall the net benefit of the Safer Choices programmes was \$174,276 and the benefit-cost ratio was 2.65, indicating that for every \$1 spent on the programme, \$2.65 were saved in medical and societal costs. The generalisbility of the study to a UK context was unclear as the data used in the evaluation was based on studies conducted in the USA, and utilised other US population estimates. However, the authors state that the methods and data used were conservative and it is possible that the intervention may be cost saving in a UK context.

8.3 General health education programmes

8.3.1 Systematic reviews and meta-analyses

No systematic reviews or meta-analyses were identified for inclusion in the review of general health education programmes.

8.3.2 General health education programmes

Overall, nine studies reported on the evaluation of six general health education programmes that reported relevant alcohol and sexual education outcomes. Of the nine studies, eight were RCTs and one was a CBA study. All six programmes were primarily delivered in schools. Four programmes were solely school-based and two programmes incorporated both school and community elements. All six programmes targeted young adolescents aged less than 14 years.

One study (Moberg & Piper, 1990) examined programme effects on knowledge, finding no significant effects of Project Model Health on curriculum-specific knowledge. However, there were positive medium-term effects of this programme on attitudes towards postponing sex and using contraceptives, and attitudes and intentions towards alcohol and other drugs. Piper et al (2000) found a medium-term impact on students' perceptions of peer attitudes towards alcohol and other drugs in those receiving the intensive condition of the Healthy for Life Programme. However, there was no longer term effect of this programme. None of the studies examined intervention effects on personal and social skills. Two programmes, the RFH curriculum (O'Donnell et al., 1999; 2002) and the Aban Aya project (Flay et al., 2004), which incorporated community components, had positive effects on sexual behaviour in the medium- to long-term. Four school-based programmes (Bond et al., 2004; Harrington et al., 2006; Piper et al., 2000) had either no effect or harmful effects on sexual behaviours and alcohol use.

8.3.3 Review of published economic evaluations

No published economic evaluation studies were identified for inclusion in the review of general health education programmes.

8.4 Strengths and limitations

The review of the effectiveness and cost-effectiveness of PSHE in secondary schools focusing on SRE and alcohol education was based on a comprehensive and systematic literature review. Over 11,000 titles and abstracts were screened for inclusion for the review, and over 900 full text articles were reviewed. In addition, the review has been conducted using a standardised and transparent approach, adhering to protocols for the development of NICE public health guidance.

8.4.1 Quality of the included studies

The studies identified for inclusion in the review were based on a range of study designs, with over half of the studies based on an RCT design. The study quality was variable across the included studies and overall one third of the included studies were rated poor quality. Approximately one tenth of studies achieved a good rating (++), with the remainder rated moderate quality (~60%). The vast majority of studies did not describe the source population or source area from which study participants were drawn, and therefore it was not possible to determine the eligibility of the selected populations or area included. Few studies reported how selection bias was minimised. Across the RCTs identified for inclusion the authors reported little more than that randomisation had been undertaken. The actual methods of randomisation and whether allocation had been adequately concealed was rarely reported. For studies that were based on non-random assignment it was also

rarely reported how confounding and bias were minimised or how individuals or clusters were allocated to intervention or control arms. In addition, only two studies, both RCTs, reported whether investigators were blind to intervention and comparison groups. It was difficult to judge whether contamination was acceptably low across the included studies as few studies discussed whether contamination had or was likely to have occurred. For studies that randomised at the classroom or individual level there was the possibility that contamination occurred within schools, but for studies that randomised at the school level the possibility of contamination was minimised. Attrition rates varied across studies and some studies failed to account for all participants at study conclusion. Outcome measures were reported to be reliable in around two thirds of studies, which a proportion of studies reporting inter- or intra-rater reliability scores. Follow-up time varied greatly across the included studies from immediate post-test to over four years of follow-up. Around two-thirds of studies reported what was judged to be a follow-up time of adequate length. A variety of approaches were taken to the analysis of results. Few studies reported that an intention to treat analysis had been undertaken and few authors discussed whether the studies were adequately powered. Analytical methods appeared to be appropriate in the majority of studies, but estimates of effect size were not reported or calculable in some studies. Overall, the reporting guality of the studies was lacking meaning that the criteria on the study quality checklist were often noted as not reported.

8.4.2 Applicability

As highlighted in previous reviews conducted by the lead author and colleagues (Jones et al., 2007; 2009), there is a lack of prevention initiatives originating from the UK, particularly in the field of alcohol education. UK-based evaluations of a range of intervention approaches to SRE were identified for inclusion in the review and this provides applicable evidence from which to generate recommendations for practice and policy in England. However, the research literature on both SRE and alcohol education continues to be dominated by programmes from the USA, which may have a limited generalisbility to the UK context, because of their emphasis on abstinence.

8.4.3 How and why programmes worked

Due to the short timescales over which the review was conducted and the volume of evidence identified, it was beyond the scope of the review to undertake a full examination of how and why programmes that demonstrated effectiveness worked. However, the systematic reviews identified for inclusion the review highlighted characteristics of interventions that were effective.

For SRE programmes, more effective studies were theoretically based with social cognitive theory the most successfully applied theory (Kirby et al., 1994; Pedlow and Carey, 2003; Sales et al., 2006) and used trained adult educators (Robin et al., 2004; Sales et al., 2006), although it was recognised that trained peer providers could also be effective (Robin et al., 2004). Two reviews reported that successful interventions applied interactive strategies (Robin et al., 2004; Sales et al., 2006) and two highlighted the importance of including highly specific content focusing on reducing sexual risk behaviour such as skills about condom use or refusing sex (Kirby et al., 1994; Robin et al., 2004).

For alcohol education programmes, interactive programmes targeting alcohol have been found to be more effective than non-interactive programmes (Tobler et al., 2000; Cuijpers, 2002). Cuijpers (2002) proposed further quality criteria suggesting that programmes should be based on the social influence model and focus on normative education, commitment not to use substances, and intentions not to use. In addition, Cuijpers (2002) reported that life-skills training and/or the use of peer leaders strengthened the effects of school-based interventions.

8.5 Research recommendations

The review has identified a number of gaps in the evidence and future research should aim to address the following key research recommendations:

- There needs to be further evaluation of the effectiveness and cost-effectiveness of PSHE approaches in secondary school focusing on alcohol education, which are currently being delivered or planned in the UK;
- Full economic evaluation studies are required of PSHE approaches focusing on both SRE and alcohol education that consider both the costs and consequences of implementing these types of interventions and programmes.
- Future research should consider the relationship between alcohol use and sexual health.

Improvements in study design and the quality of reporting are required with respect to all types of studies and the following are recommendations to improve the methodology of future studies:

- Improved reporting of methods is required, particularly with regard to methods for the allocation of participants and clusters (e.g. methods of randomisation), allocation concealment, procedures for blinding, and follow-up of participants. Reporting standards could be improved by following guidelines on reporting, such as the CONSORT statement for RCTs and TREND statement for non-randomised studies.
- Standardisation of outcomes is required. Across the included studies a range of attitudinal and behavioural measures were reported and consequently it was not possible to synthesise outcomes across studies. Also when considering which outcomes to incorporate, there needs to be a consideration of the age and level maturity of the sample targeted (e.g. with regard to studies of SRE programmes consideration should be given to the relationship status of participants).
- Some studies were conducted with inadequate sample sizes, and future research studies should be sufficiently powered to detect intervention effects.
- Future research studies should incorporate an adequate length of follow-up.

9 Conclusions

9.1 Alcohol and drug education programmes

The evidence suggests that classroom-based programmes, regardless of whether the focus is on alcohol alone or as one of a number of substances, may have beneficial effects on alcohol-related knowledge, particularly in the short- to medium-term. Programme effects on attitudes and values were mixed and inconsistent across a range of intervention approaches and the evidence was insufficient to draw conclusions about the impact of these programmes on personal and social skills. Overall, the findings of the review of alcohol and drug education programmes highlight that there is a lack of clear, medium- to long-term evidence for the effectiveness of school-based alcohol education programmes on health and social outcomes relating to alcohol use. In addition, the applicability to a UK context of those programmes that have demonstrated effectiveness, such as LST, STARS for Families and Project SPORT, is limited. There is lack of evidence on which to draw conclusions about the cost-effectiveness of alcohol and drug education programmes. Further good quality UK-based research of promising or novel intervention approaches, including assessment of cost-effectiveness, is required in order to improve the evidence base on which to make UK-based policy and practice recommendations for PSHE focusing on alcohol education.

9.2 Sex and relationships education programmes

There were consistently positive programme effects on acquisition of sexual health knowledge, across the SRE education programmes that examined this outcome, regardless of whether the programme emphasised abstinence or not. A range of outcomes were reported with regards to attitudes and values and programme effects were mixed or inconsistent across these measures. It was therefore not possible to draw unequivocal conclusions about the impact of SRE programmes on attitudes and values relating to sexual health. The evidence suggests that while abstinence-only programmes have no effects on health and social outcomes related to sexual health, programmes that incorporate information on safer sex and contraceptive use may have positive, but limited effects on the prevention of sexual risk behaviours, in particular limited effects on contraceptive use. Although the applicability of some of these programmes to a UK context is limited, these conclusions are supported by the evidence drawn from studies conducted in the UK. There is lack of evidence on which to draw conclusions about the cost-effectiveness of SRE programmes. Further good quality UK-based research of promising or novel intervention approaches, including assessment of cost-effectiveness, is required in order to add to the evidence base on which to make UK-based policy and practice recommendations for PSHE focusing on SRE.

9.3 General health education programmes

There was a lack of evidence on which to draw clear conclusions about the effects of the general health education programmes on knowledge, and attitudes and values relating to alcohol use and sexual health. The evidence suggests that general health education programmes, which incorporate an intensive community intervention element in conjunction with a curriculum base may have a positive effect on sexual behaviour and substance use. There were no effects, or in some cases

harmful effects on sexual health and alcohol use outcomes for curriculum-based, general health education programmes. In addition, the applicability of these programmes to a UK context was limited.

10 References

Bellis M, Hughes K, Calafat A, et al. (2008). Sexual uses of alcohol and drugs and the associated health risks: a cross sectional study of young people in nine European cities. BMC Public Health, 8(1):155.

Choquet M, Manfredi R (1992). Sexual intercourse, contraception, and risk taking behaviour among unselected French adolescents aged 11-20 years. Journal of Adolescent Health 13: 623-630.

Department for Children Schools and Families (2008) Review of sex and relationship Education (SRE) in Schools. London, Department for Children Schools and Families.

Department for Education and Employment. (2000) Sex and relationship education guidance. London, Department for Education and Employment.

Department for Education and Skills (2005) National Healthy Schools Status, a guide for schools. London, Department for Education and Skills.

Department for Education and Skills (2006) Teenage Pregnancy: Accelerating the Strategy to 2010. London, Department for Education and Skills.

Department for Education and Skills (2007) Improving Access to Sexual Health Services for Young People in Further Education Setting. London, Department for Education and Skills.

Department of Health (2007) Safe. Sensible. Social. The next steps in the National Alcohol Strategy. London: Department of Health.

Department of Health. (2009) Health profiles for England 2008. London, Department of Health.

Dye C, Upchurch DM (2006). Moderating effects of gender on alcohol use. Implications for condom use at first intercourse. Journal of School Health, 76 (3):111-116.

Emmerson (2008) National Mapping of On-site Sexual Health Services in Education Settings. Sex Education Forum.

Emmerson (2007) SRE and young people in further education: A review of provision and practice in England. Pastoral Care in Education, 25 (3):39-48.

Health Protection Agency (2009). Selected STI diagnoses and diagnosis rates from GUM clinics in the UK: 2004-2008. London, Health Protection Agency.

Hibell B, Andersson B, Bjarnasson T et al. (2004). The 2003 ESPAD report: alcohol and other drug use among students in 35 countries. Stockholm: Swedish Council for Information on Alcohol and Other Drugs.

Jones L, Bates G, Downing J, Sumnall H, Bellis MA (2009) A review of the effectiveness and costeffectiveness of personal, social and health education in primary schools focusing on sex and relationship education for young people aged 5 to 11 years. Liverpool John Moores University Kim-Godwin YS, Clements C, Bullers S et al. (2007). Sexual behaviors and drinking patterns among middle school and high school students in southeastern North Carolina. Journal of School Nursing 23 (4):214-221.

Lowry R, Holtzman D, Trueman BI, et al. (1994). Substance use and HIV related sexual behaviours among US high school students: are they related? Am J Public Health, 84: 1116-20.

MacDonald A (2009) Independent review of the proposal to make Personal, Social, Health and Economic (PSHE) education statutory. London, Department for Children, Schools and Families.

Miller JW, Naimi TS, Brewer R, Everett Jones S. (2007). Binge drinking and associated health risk behaviours among high school students. Pediatrics, 119: 76-85.

Prime Minister's Strategy Unit (2007) Alcohol Harm Reduction Strategy. London, Strategy Unit.

Qualifications and Curriculum Authority (2007) The National Curriculum 2007. London, Qualifications and Curriculum Authority.

Ramisetty-Mikler S, Caetano R, Goebert D, Nishimura S. (2004). Ethic variation in drinking, drug use, and sexual behaviour among adolescents in Hawaii. J School Health, 74: 16-22.

Robertson JA, Plant MA (1988). Alcohol, sex and risks of HIV. Drug Alcohol Dependence, 22: 75-78.

Appendix 1. References to included studies

Aarons SJ, Jenkins RR, Raine TR, et al (2000) Postponing sexual intercourse among urban junior high school students - A randomized controlled evaluation. Journal of Adolescent Health 27(4):236-247.

Argentos, M. S. (1991). The evaluation of a drug and alcohol prevention program in a secondary school setting. Dissertation Abstracts International 52 (4-B).

Assefi NP, Bennett SE (2005) School-based teenage pregnancy prevention programs: a systematic review of randomized controlled trials. Journal of Adolescent Health 36(1):72-81.

Aten MJ, Siegel DM, Enaharo M, Auinger P (2002) Keeping middle school students abstinent: Outcomes of a primary prevention intervention. Journal of Adolescent Health 31(1):70-78.

Bagnall, G. (1990). Alcohol education for 13 year olds--does it work? Results from a controlled evaluation. British Journal of Addiction 85 (1): 89-96.

Basen-Engquist K, Coyle KK, Parcel GS, et al (2001) Schoolwide effects of a multicomponent HIV, STD, and pregnancy prevention program for high school students. Health Education & Behavior 28(2):166-85.

Battistich V, Dean BJ, Gosselink CA, Zavela KJ (2004) Say yes first: follow up of a five-year rural drug prevention program. Journal of Drug Education 34(1):73-88.

Baumann BD (2006) Alcohol use and dating violence: A high-school prevention model, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 66.

Bell RM, Ellickson PL, Harrison ER (1993) Do drug prevention effects persist into high school? How project ALERT did with ninth graders.[see comment]. Preventive Medicine 22(4):463-83.

Bellingham K, Gillies P (1993) Evaluation of an AIDS education programme for young adults, in Journal of Epidemiology and Community Health, pp 134-8.

Bennett, G. T. (1995). An exploration of patterns of drug use and of the effectiveness of a substance abuse prevention program according to adolescents' level of academic achievement. Dissertation Abstracts International Section A: Humanities and Social Sciences 55 (10-A).

Blake SM, Simkin L, Ledsky R, et al (2001) Effects of a parent-child communications intervention on young adolescents' risk for early onset of sexual intercourse. Family Planning Perspectives 33(2):52-61.

Bond, L., Patton, G., Glover, S. et al. (2004). The Gatehouse Project: can a multilevel school intervention affect emotional wellbeing and health risk behaviours? Journal of Epidemiology and Community Health 58 (12): 997-1003.

Borawski EA, Trapl ES, Adams-Tufts K, et al (2009) Taking be proud! Be responsible! To the suburbs: A replication study. Perspectives on Sexual and Reproductive Health 41:12-22.

Borawski EA, Trapl ES, Lovegreen LD, et al (2005) Effectiveness of abstinence-only intervention in middle school teens. American Journal of Health Behavior 29(5):423-34.

Borgia P, Marinacci C, Perucci CA, Schifano P (2005) Is peer education the best approach for HIV prevention in schools? Findings from a randomized controlled trial. Journal of Adolescent Health 36(6):508-516.

Botvin GJ, Baker E, Dusenbury L, et al (1990) Preventing Adolescent Drug Abuse through a Multimodal Cognitive- Behavioral Approach: Results of a 3-Year Study. Journal of Consulting and Clinical Psychology 58(4):437-46.

Botvin GJ, Baker E, Dusenbury L, et al (1995a) Long-term follow-up results of a randomized drug abuse prevention trial in a white middle-class population. Journal of the American Medical Association 273(14):1106-1112.

Botvin GJ, Baker E, Filazzola AD, Botvin EM (1990) A cognitive-behavioral approach to substance abuse prevention: One-year follow-up. Addictive Behaviors 15(1):47-63.

Botvin GJ, Griffin KW, Diaz T, Ifill-Williams M (2001) Preventing binge drinking during early adolescence: one- and two-year follow-up of a school-based preventive intervention. Psychology of Addictive Behaviors 15(4):360-5.

Botvin GJ, Griffin KW, Diaz T, et al. (2001a). Drug abuse prevention among minority adolescents: posttest and one-year follow-up of a school-based preventive intervention. Prevention Science 2 (1): 1-13.

Botvin GJ, Griffin KW, Diaz T, et al. (2001b). Preventing binge drinking during early adolescence: one- and two-year follow-up of a school-based preventive intervention. Psychology of Addictive Behaviors 15 (4): 360-365.

Botvin GJ, Schinke SP, Epstein JA, et al. (1995b). Effectiveness of culturally focused and generic skills training approaches to alcohol and drug abuse prevention among minority adolescents: Two-year follow-up results. Psychology of Addictive Behaviors 9 (3): 183-194.

Boyer CB, Shafer MA, Tschann J (1997) Evaluation of a Knowledge- and Cognitive-Behavioral Skills-Building Intervention to Prevent STDs and HIV Infection in High School Students. Adolescence 32(125):25-42.

Bremberg S, Arborelius E. (1994). Effects on adolescent alcohol consumption of a school based student-centred health counselling programme. Scandinavian Journal of Social Medicine 22 (2): 113-119.

Brewer LC (1991). Social skills training as a deterrent to entry level drug experimentation among 15year-old adolescents. Unpublished PhD dissertation, University of Pennsylvania.

Caplan M, Weissberg RP, Grober JS, et al. (1992). Social competence promotion with inner-city and suburban young adolescents: Effects on social adjustment and alcohol use. Journal of Consulting and Clinical Psychology 60 (1): 56-63.

Caron F, Godin G, Otis J, Lambert LD (2004) Evaluation of a theoretically based AIDS/STD peer education program on postponing sexual intercourse and on condom use among adolescents attending high school. Health Education Research 19(2):185-197.

Chou CP, Montgomery S, Pentz MA, et al. (1998). Effects of a community-based prevention program on decreasing drug use in high-risk adolescents. American Journal of Public Health 88 (6): 944-948.

Christopher FS, Roosa MW (1990) An evaluation of an adolescent pregnancy prevention program: Is "Just Say No" enough? Family Relations 39(1):68-72.

Clayton RR, Cattarello AM, Johnstone BM (1996) The effectiveness of drug abuse resistance education (project DARE): 5- year follow-up results. Preventive Medicine 25(3):307-318.

Clayton RR, Cattarello A, Day L, Walden KP (1991). Persuasive communication and drug prevention: An evaluation of the DARE program. In: Donohew, L., Sypher, H. E., and Bukoski, W. J. (ed) Persuasive communication and drug abuse prevention, pp 295-313. Hillsdale, NJ, England: Lawrence Erlbaum Associates, Inc.

Coggans N, Cheyne B, McKellar S (2003). The Life Skills Training Drug Education Programme: a review of research.

Colnes RM (2001). The effectiveness of a school-based substance use prevention program. Dissertation Abstracts International: Section B: The Sciences and Engineering 61 (8-B): pp-

Coyle K, Basen EK, Kirby D, et al (1999) Short-Term Impact of Safer Choices: A Multicomponent, School-Based HIV, Other STD, and Pregnancy Prevention Program. Journal of School Health 69(5):181-88.

Coyle K, Basen-Engquist K, Kirby D, et al (2001) Safer choices: Reducing teen pregnancy, HIV, and STDs. Public Health Reports 116(SUPPL. 1):82-93.

Coyle KK, Kirby DB, Maran BV, Gomez CA (2004) Draw the line/respect the line: a randomized trial of a middle school intervention to reduce sexual risk behaviors. American Journal of Public Health 94(5):843-852.

Coyle KK, Kirby DB, Robin LE, et al (2006) All4You! A randomized trial of an HIV, other STDs, and pregnancy prevention intervention for alternative school students. AIDS Education & Prevention 18(3):187-203.

Robin L, Dittus P, Whitaker D, et al (2004) Behavioral interventions to reduce incidence of HIV, STD, and pregnancy among adolescents: a decade in review. Journal of Adolescent Health 34(1):3-26.

Cuijpers P, Jonkers R, de Weerdt I, de Jong A (2002) The effects of drug abuse prevention at school: the 'Healthy School and Drugs' project. Addiction 97(1):67-73.

Dedobbeleer N, Desjardins S (2001) Outcomes of an ecological and participatory approach to prevent alcohol and other drug "abuse" among multiethnic adolescents. Substance Use & Misuse 36(13):1959-91.

Dempster M, Newell G, Cowan G, Marley J (2006) Facing up to binge drinking: reducing binge drinking in adolescent males. British Dental Journal 201(9):587-90; discussion 578.

Denman S, Pearson J, Moody D, The effects of drug abuse prevention at school: the 'Healthy School and Drugs' project (1995) Theatre in education on HIV and AIDS: a controlled study of school children's knowledge and attitudes. Health Education Journal 54(1):3-17.

Denny G, Young M (2006) An evaluation of an abstinence-only sex education curriculum: an 18month follow-up. Journal of School Health 76(8):414-22.

Denny G, Young M, Rausch S, Spear C (2002) An evaluation of an abstinence education curriculum series: sex can wait. American Journal of Health Behavior 26(5):366-77.

Denny G, Young M, Spear CE (1999) An evaluation of the "sex can wait" abstinence education curriculum series. American Journal of Health Behavior 23(2):134-144.

Dent CW, Sussman S, Stacy AW (2001) Project Towards No Drug Abuse: generalizability to a general high school sample.[erratum appears in Prev Med. 2005 Jun;36(6):539-40]. Preventive Medicine 32(6):514-20.

Donnelly J, Ferraro H, Eadie C (2001) Effects of a health and relationship education program on drug behaviors. North American Journal of Psychology 3(3):453-462.

Dukes RL, Stein JA, Ullman JB (1997). Long-term impact of Drug Abuse Resistance Education (D. A. R. E.): Results of a 6-year follow-up. Evaluation Review 21 (4): 483-500.

Dukes RL, Ullman JB, Stein JA (1996). Three-year follow-up of Drug Abuse Resistance Education (D.A.R.E.). Evaluation Review 20 (1): 49-66.

Dusenbury L, Falco M, Lake A (1997). A review of the evaluation of 47 drug abuse prevention curricula available nationally. Journal of School Health 67 (4): 127-132.

Eisen M, Zellman GL, Massett HA, Murray DM (2002) Evaluating the Lions-Quest "Skills for Adolescence" drug education program: First-year behavior outcomes. Addictive Behaviors 27(4):619-632.

Eisen M, Zellman GL, McAlister AL (1990) Evaluating the impact of a theory-based sexuality and contraceptive education program. Family Planning Perspectives 22(6):261-271.

Eisen M, Zellman GL, McAlister AL (1992) A Health Belief Model-Social Learning Theory approach to adolescents' fertility control: findings from a controlled field trial. Health Education Quarterly 19(2):249-62.

Ellickson PL, Bell RM (1990) Drug prevention in junior high: A multi-site longitudinal test. Science 247(4948):1299-1305.

Ellickson PL, McCaffrey DF, Ghosh-Dastidar B, Longshore DL (2003) New Inroads in Preventing Adolescent Drug Use: Results from a Large-Scale Trial of Project ALERT in Middle Schools. American Journal of Public Health 93(11):1830-1836.

Ennett ST, Rosenbaum DP, Flewelling RL, et al. (1994). Long-term evaluation of drug abuse resistance education. Addictive Behaviors 19 (2): 113-125.

Fearnow-Kenney MD, Wyrick DL., Jackson-Newsom J, et al. (2003). Initial indicators of effectiveness for a high school drug prevention program. American Journal of Health Education 34 (2): 66-71.

Fisher JD, Bryan AD, Fisher WA, Misovich SJ (2002) Information-motivation-behavioral skills modelbased HIV risk behavior change intervention for inner-city high school youth. Health Psychology 21(2):177-86.

Flay BR, Graumlich S, Segawa E, et al (2004) Effects of 2 Prevention Programs on High-Risk Behaviors among African American Youth: A Randomized Trial. Archives of Pediatrics and Adolescent Medicine 158(4):377-384.

Fraguela JAG, Martin AL, Trinanes ER (2003). Drug abuse prevention in the school: Four-year followup of a programme. Psychology in Spain

Franklin C, Grant D, Corcoran J, et al (1997) Effectiveness of Prevention Programs for Adolescent Pregnancy: A Meta- Analysis. Journal of Marriage and the Family 59(3):551-67.

Gillies PA, Stork A, Bretman M (1990) Streetwize UK: a controlled trial of an AIDS education comic. Health Education Research 5(Mar 90):27-33.

Gosin M, Marsiglia, FF, Hecht ML. (2003). keepin' it R.E.A.L.: A drug resistance curriculum tailored to the strengths and needs of pre-adolescents of the southwest. (84 refs.). Journal of Drug Education 33 (2): 119-142.

Gottfredson DC, Wilson DB (2003) Characteristics of effective school-based substance abuse prevention. Prevention Science 4:27-38.

Graham A, Moore L, Sharp D, Diamond I (2002) Improving teenagers' knowledge of emergency contraception: cluster randomised controlled trial of a teacher led intervention. BMJ 324(7347):1179.

Graham JW, Johnson CA, Hansen WB, et al. (1990). Drug use prevention programs, gender, and ethnicity: evaluation of three seventh-grade Project SMART cohorts. Preventive Medicine 19 (3): 305-313.

Griffin KW, Botvin GJ, Nichols TR, Doyle MM (2003) Effectiveness of a universal drug abuse prevention approach for youth at high risk for substance use initiation. Preventive Medicine 36(1):1-7.

Harmon MA (1993). Reducing the risk of drug involvement among early adolescents. Evaluation Review 17 (2): 221-239.

Harrington NG, Giles SM, Hoyle RH, et al (2001) Evaluation of the All Stars character education and problem behavior prevention program: effects on mediator and outcome variables for middle school students. Health Education & Behavior 28(5):533-548.

Hecht ML, Marsiglia FF, Elek E, et al. (2003). Culturally grounded substance use prevention: an evaluation of the keepin' it R.E.A.L. curriculum. Prevention Science 4 (4): 233-248.

Henderson M, Wight D, Raab GM, et al (2007) Impact of a Theoretically Based Sex Education Programme (SHARE) Delivered By Teachers On NHS Registered Conceptions And Terminations: Final Results Of Cluster Randomised Trial. BMJ 334(7585):133-136.

Hubbard BM, Giese ML, Rainey J (1998) A Replication Study of Reducing the Risk, a Theory-Based Sexuality Curriculum for Adolescents. Journal of School Health 68(6):243-47.

Johnson CA., Pentz MA., Weber MD, et al. (1990). Relative effectiveness of comprehensive community programming for drug abuse prevention with high-risk and low-risk adolescents. Journal of Consulting & Clinical Psychology 58 (4): 447-456.

Jorgensen SR, Potts V, Camp B (1993) Project Taking Charge: Six-month follow-up of a pregnancy prevention program for early adolescents. Family Relations 42(4):401-06.

Kirby D, Barth RP, Leland N, Fetro JV (1991) Reducing the risk: Impact of a new curriculum on sexual risk-taking. Family Planning Perspectives 23(6):253-263.

Kirby D, Short L, Collins J, et al (1994) School-based programs to reduce sexual risk behaviors: a review of effectiveness. Public Health Reports 109(3):339-60.

Kirby DB, Baumler E, Coyle KK, et al (2004) The "Safer Choices" intervention: Its impact on the sexual behaviors of different subgroups of high school students. Journal of Adolescent Health 35(6):442-452.

Klitzner M, Gruenewald PJ, Bamberger E, et al. (1994). A quasi-experimental evaluation of Students Against Driving Drunk. American Journal of Drug & Alcohol Abuse 20 (1): 57-74.

Komro KA, Perry CL, Veblen-Mortenson S, et al (2008) Outcomes from a randomized controlled trial of a multi-component alcohol use preventive intervention for urban youth: project northland Chicago. Addiction 103(4):606-18.

Komro KA, Perry CL., Williams CL. et al. (2001). How did Project Northland reduce alcohol use among young adolescents? Analysis of mediating variables. Health Education Research 16 (1): 59-70.

Kreft IG (1998). An illustration of item homogeneity scaling and multilevel analysis techniques in the evaluation of drug prevention programs. Evaluation Review 22 (1): 46-77.

Kulis S, Nieri T, Yabiku S, et al (2007) Promoting reduced and discontinued substance use among adolescent substance users: Effectiveness of a universal prevention program. Prevention science 8:35-49.

Kulis S, Marsiglia FF, Elek E, et al. (2005). Mexican/Mexican American adolescents and keepin' it REAL: an evidence-based substance use prevention program. Children & Schools 27 (3): 133-145.

Kvalem IL, Sundet JM, Rivo KI, et al (1996) The effect of sex education on adolescents' use of condoms: applying the Solomon four-group design. Health Education Quarterly 23(1):34-47.

LaChausse RG (2006) Evaluation of the Positive Prevention HIV/STD curriculum. American Journal of Health Education 37(4):203-210.

Larsson M, Eurenius K, Westerling R, Tyden T (2006) Evaluation of a sexual education intervention among Swedish high school students. Scandinavian Journal of Public Health 34(2):124-131.

Lemieux AF, Fisher JD, Pratto F (2008) A Music-Based HIV Prevention Intervention for Urban Adolescents. Health Psychology 27(3):349-357.

Lennox RD, Cecchini MA (2008) The NARCONON drug education curriculum for high school students: a non-randomized, controlled prevention trial. Substance Abuse Treatment, Prevention, & Policy 3:8.

Levy SR, Perhats C, Weeks K, et al (1995) Impact of a School-Based AIDS Prevention Program on Risk and Protective Behavior for Newly Sexually Active Students. Journal of School Health 65(4):145-51.

Lewis RK, Paine-Andrews A, Fisher J, et al (1999) Reducing the risk for adolescent pregnancy: evaluation of a school/community partnership in a Midwestern military community. Family & Community Health 22(2):16-31.

Loveland-Cherry CJ (2005) Alcohol, children, and adolescents. Annual Review of Nursing Research 23:135-77.

Lynam DR., Milich R, Zimmerman R, et al. (1999). Project Dare: No Effects at 10-Year Follow-up. Journal of Consulting & Clinical Psychology 67 (4): 590-593.

Magnusson J, Kendall S, Oakley L, Townsend J (2004) Promoting contraceptive services to teenagers. Community Practitioner 77(10):381-384.

McBride N, Farringdon F, Midford R, et al (2003) Early unsupervised drinking--reducing the risks. The School Health and Alcohol Harm Reduction Project. Drug & Alcohol Review 22(3):263-76.

McBride N, Farringdon F, Midford R, et al (2004) Harm minimization in school drug education: final results of the School Health and Alcohol Harm Reduction Project (SHAHRP). Addiction 99(3):278-91.

McBride N, Midford R, Farringdon F, Phillips M (2000) Early result from a school alcohol harm minimization study: The school health and alcohol harm reduction project. Addiction 95(7):1021-1042.

McNeal RB, Jr., Hansen WB, Harrington NG, Giles SM (2004) How All Stars works: an examination of program effects of mediating variables. Health Education & Behavior 31(2):165-180.

Mellanby AR, Newcombe RG, Rees J, Tripp JH (2001) A comparative study of peer-led and adult-led school sex education. Health Education Research 16(4):481-92.

Mellanby AR, Phelps FA, Crichton NJ, Tripp JH (1995) School sex education: an experimental programme with educational and medical benefit. BMJ 311(7002):414-7.

Mitchell-DiCenso A, Thomas BH, Devlin MC, Goldsmith CH, et al (1997) Evaluation of an educational program to prevent adolescent pregnancy. Health Education and Behavior 24(3):300-12.

Moberg DP, Piper DL (1990). An outcome evaluation of Project Model Health: a middle school health promotion program. Health Education Quarterly 17 (1): 37-51.

Newton NC, Vogl LE, Teesson M, Andrews G (2009) CLIMATE Schools: alcohol module: crossvalidation of a school-based prevention programme for alcohol misuse. Australian & New Zealand Journal of Psychiatry 43(3):201-7.

Oakley A, Fullerton D, Holland J, et al (1995) Sexual health education interventions for young people: a methodological review. BMJ 310:158-162.

O'Donnell L, Stueve A, Doval AS, et al (1999) The effectiveness of the Reach for Health Community Youth Service learning program in reducing early and unprotected sex among urban middle school students. American Journal of Public Health 89(2):176-182.

O'Donnell L, Stueve A, O'Donnell C, et al (2002) Long-term reductions in sexual initiation and sexual activity among urban middle schoolers in the reach for health service learning program. Journal of Adolescent Health 31(1):93-100.

Padget A, Bell ML, Shamblen SR, Ringwalt C (2005) Effects on high school students of teaching a cross-age alcohol prevention program. Journal of Drug Education 35(3):201-216.

Paine-Andrews A, Harris KJ, Fisher JL, et al (1999) Effects of a replication of a multicomponent model for preventing adolescent pregnancy in three Kansas communities. Family Planning Perspectives 31(4):182-189.

Palmer RF, Graham JW, White EL, et al. (1998). Applying multilevel analytic strategies in adolescent substance use prevention research. Preventive Medicine 27 (3): 328-336.

Patton GC, Bond L, Carlin JB, et al (2006) Promoting social inclusion in schools: A group-randomized trial of effects on student health risk behavior and well-being. American Journal of Public Health 96(9):1582-1587.

Pedlow CT, Carey MP (2003) HIV sexual risk-reduction interventions for youth: A review and methodological critique of randomized controlled trials. Behavior Modification 27(2):135-190.

Perry CL, Komro KA, Veblen-Mortenson S, et al (2003) A randomized controlled trial of the middle and junior high school D.A.R.E. and D.A.R.E. Plus programs. Archives of Pediatrics and Adolescent Medicine 157(2):178-184.

Perry CL, Williams CL, Veblen-Mortenson S, et al. (1996). Project Northland: outcomes of a communitywide alcohol use prevention program during early adolescence. American Journal of Public Health 86 (7): 956-965.

Piper DL, Moberg DP, King MJ (2000) The Healthy for Life project: behavioral outcomes. Journal of Primary Prevention 21(1):47-73.

Ringwalt C, Ennett ST, Holt KD (1991). An outcome evaluation of project DARE (Drug Abuse Resistance Education). Health Education Research 6 (3): 327-337.

Roberto AJ, Zimmerman RS, Carlyle KE, Abner EL (2007) A computer-based approach to preventing pregnancy, STD, and HIV in rural adolescents. Journal of Health Communication 12(1):53-76.

Roosa MW, Christopher FS (1990) Evaluation of an Abstinence-Only Adolescent Pregnancy Prevention Program: A Replication. Family Relations 39(4):363-67.

Rosenbaum DP, Hanson GS (1998). Assessing the effects of school-based drug education: A sixyear multilevel analysis of project DARE. (60 refs.). Journal of Research in Crime and Delinquency 35 (4): 381-412.

Rosenbaum DP, Flewelling RL, Bailey SL, et al. (1994). Cops in the classroom: A longitudinal evaluation of drug abuse resistance education (DARE). Journal of Research in Crime and Delinquency 31 (1): 3-31.

Sales JM, Milhausen RR, DiClemente RJ (2006) A decade in review: Building on the experiences of past adolescent STI/HIV interventions to optimise future prevention efforts. Sexually Transmitted Infections 82(6):431-436.

Schaalma HP, Kok G, Bosker RJ, et al (1996) Planned development and evaluation of AIDS/STD education for secondary school students in The Netherlands: short-term effects. Health Education Quarterly 23(4):469-87.

Schnepf CM (2002) Comparing the effects of a peer- and a teacher-implemented alcohol education program for urban adolescents, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 63.

Shope JT, Copeland LA, Marcoux BC, Kamp ME (1996) Effectiveness of a school-based substance abuse prevention program. Journal of Drug Education 26(4):323-37.

Shope JT, Copeland LA, Maharg R, Dielman TE (1996) Effectiveness of a high school alcohol misuse prevention program. Alcoholism: Clinical and Experimental Research 20(5):791-798.

Shope JT, Copeland LA, Kamp ME, et al. (1998). Twelfth grade follow-up of the effectiveness of a middle school-based substance abuse prevention program. Journal of Drug Education 28 (3): 185-197.

Shope JT, Kloska DD, Dielman TE, et al. (1994). Longitudinal evaluation of an enhanced alcohol misuse prevention study (AMPS) curriculum for grades six-eight. Journal of School Health 64 (4): 160-166.

Shortt AL, Hutchinson DM, Chapman R, Toumbourou JW (2007) Family, school, peer and individual influences on early adolescent alcohol use: first-year impact of the Resilient Families programme. Drug & Alcohol Review 26(6):625-34.

Siegel DM, Aten MJ, Enaharo M (2001) Long-term effects of a middle school- and high school-based human immunodeficiency virus sexual risk prevention intervention. Archives of Pediatrics & Adolescent Medicine 155(10):1117-26.

Siegel DM, Aten MJ, Roghmann KJ, Enaharo M (1998) Early effects of a school-based human immunodeficiency virus infection and sexual risk prevention intervention. Archives of Pediatrics and Adolescent Medicine 152(10):961-970.

Simons-Morton B, Haynie D, Saylor K, et al. (2005). The effects of the Going Places program on early adolescent substance use and antisocial behavior. Prevention Science 6 (3): 187-197.

Slater MD, Kelly KJ, Edwards RW, et al (2006) Combining in-school and community-based media efforts: Reducing marijuana and alcohol uptake among younger adolescents. Health Education Research 21(1):157-167.

Smit F, Cuijpers P, Lemmers L, et al (2003) Same prevention, different effects? Effect modification in an alcohol misuse prevention project among high-school juniors. Drugs: Education, Prevention and Policy 10:185-193.

Smith EA, Swisher JD, Vicary JR, et al (2004) Evaluation of Life Skills Training and infused-Life Skills Training in a rural setting: outcomes at two years. Journal of Alcohol and Drug Education 48(1):20.

Smith MA (1994) Teen Incentives Program: evaluation of a health promotion model for adolescent pregnancy prevention. Journal of Health Education 25(1):24-9.

Snow DL, Tebes JK, Arthur MW, Tapasak RC (1992) Two-year follow-up of a social-cognitive intervention to prevent substance use. Journal of Drug Education 22(2):101-14.

Snow DL, Tebes JK, Ayers TS (1997). Impact of two social-cognitive interventions to prevent adolescent substance use: Test of an amenability to treatment model. Journal of Drug Education 27 (1): 1-17.

Somers CL (2006) Teenage Pregnancy Prevention and Adolescents' Sexual Outcomes: An Experiential Approach. American Secondary Education 34(2):21-24.

Somers CL, Gleason JH, Johnson SA, Fahlman MM (2001) Adolescents' and Teachers' Perceptions of a Teen Pregnancy Prevention Program. American Secondary Education 29(3):51-66.

Spoth RL, Randall GK, Trudeau L, et al (2008) Substance use outcomes 51/2 years past baseline for partnership-based, family-school preventive intervention. Drug and Alcohol Dependence 96(1-2):57-68.

Spoth RL, Redmond C, Trudeau L, Shin C (2002) Longitudinal substance initiation outcomes for a universal preventive intervention combining family and school programs. Psychology of Addictive Behaviors 16(2):129-34.

Stanton B, Harris C, Cottrell L, Li X, et al (2006) Trial of an urban adolescent sexual risk-reduction intervention for rural youth: a promising but imperfect fit. The Journal of Adolescent Health 38(1):e25-55.e36.

Stephenson J, Strange V, Allen E, et al (2008) The long-term effects of a peer-led sex education programme (RIPPLE): A cluster randomised trial in schools in England. PLoS Medicine 5(11):1579-1590.

Stephenson JM, Strange V, Forrest S, et al (2004) Pupil-led sex education in England (RIPPLE study): Cluster-randomised intervention trial. Lancet 364(9431):338-346.

Stout JW, White LC, Alexander T (1996) Oregon School-Based Health Centers: A Follow-Up Report. Portland: Oregon State Department of Human Resources.

Sun W, Skara S, Sun P, et al. (2006). Project Towards No Drug Abuse: Long-term substance use outcomes evaluation. Preventive Medicine 42 (3): 188-192.

Sussman S, Dent CW, Stacy AW, et al. (1998). One-year outcomes of Project Towards No Drug Abuse. Preventive Medicine 27 (4): 632-642.

Sussman S, Sun P, McCuller WJ, et al. (2003). Project Towards No Drug Abuse: Two-year outcomes of a trial that compares health educator delivery to self-instruction. (27 refs.). Preventive Medicine 37 (2): 155-162.

Teitler JO (1997) Does condom availability make a difference? An evaluation of Philadelphia's health resource centers. Family Planning Perspectives 29(3):123-127.

Tobler NS, Stratton HH (1997) Effectiveness of school-based drug prevention programs: A metaanalysis of the research. Journal of Primary Prevention 18(1):71-128.

Tobler NS. (1993). Meta-analysis of adolescent drug prevention programs. Dissertation Abstracts International 55 (11).

Tobler NS, Roona MR, Ochshorn P, et al. (2000). School-based adolescent drug prevention programs: 1998 Meta-analysis. Journal of Primary Prevention 20 (4): 275-336.

Toomey TL (1996). An Alcohol Primary Prevention Program for Parents of 7th Graders: The Amazing Alternatives! Home Program. Journal of Child and Adolescent Substance Abuse 5 (4): 35-53.

Traeen B (2003) Effect of an intervention to prevent unwanted pregnancy in adolescents: a randomized, prospective study from Nordland County, Norway, 1999-2001. Journal of Community and Applied Social Psychology 13(3):207-23.

Trenholm C, Devaney B, Fortson K, et al (2008) Impacts of Abstinence Education on Teen Sexual Activity, Risk of Pregnancy, and Risk of Sexually Transmitted Diseases. Journal of Policy Analysis and Management 27:22-276.

Tucker JS, Fitzmaurice AE, Imamura M, et al (2007) The effect of the national demonstration project Healthy Respect on teenage sexual health behaviour. European Journal of Public Health 17(1):33-41.

Underhill K, Montgomery P, Operario D (2008) Abstinence-plus programs for HIV infection prevention in high-income countries. Cochrane Database of Systematic Reviews (1).

Underhill K, Operario D, Montgomery P (2007) Abstinence-only programs for HIV infection prevention in high-income countries. The Cochrane database of systematic reviews (4):CD005421-NaN.

Valentine J, Griffith J, Ruthazer R, et al. (1998). Strengthening causal inference in adolescent drug prevention studies: Methods and findings from a controlled study of the urban youth connection program. Drugs & Society 12 (1-2): 127-143.

Vicary JR, Henry KL, Bechtel LJ, et al. (2004). Life skills training effects for high and low risk Rural Junior High School Females. Journal of Primary Prevention 25 (4): 399-416.

Vincent M, Drane JW, Joshi P, et al (2004) Sustained reduction in adolescent pregnancy rates through school and community-based education, 1982-2000. American Journal of Health Education 35(2):76-86.

Vogl L, Teesson M, Andrews G, et al (2009) A computerized harm minimization prevention program for alcohol misuse and related harms: randomized controlled trial. Addiction 104(4):564-576.

Walter HJ, Vaughan RD (1993) AIDS risk reduction among a multiethnic sample of urban high school students. Journal of the American Medical Association 270(6):725-730.

Wang LY, Davis M, Robin L, et al (2000) Economic evaluation of safer choices: A school-based human immunodeficiency virus, other sexually transmitted diseases, and pregnancy prevention program. Archives of Pediatrics and Adolescent Medicine 154(10):1017-1024.

Warren JR, Hecht ML, Wagstaff DA, et al. (2006). Communicating Prevention: The Effects of the keepin' it REAL Classroom Videotapes and Televised PSAs on Middle-School Students' Substance Use. Journal of Applied Communication Research 34 (2): 209-227.

Webster RA, Hunter M, Keats JA (2002) Evaluating the effects of a peer support programme on adolescents' knowledge, attitudes and use of alcohol and tobacco. Drug and Alcohol Review 21(1):7-16.

Weeks K, Levy SR, Zhu C, et al (1995) Impact of a school-based AIDS prevention program on young adolescents' self-efficacy skills. Health Education Research 10(3):329-344.

Werch C, Jobli E, Moore MJ, et al (2005a) A brief experimental alcohol beverage-tailored program for adolescents. Journal of Studies on Alcohol 66(2):284-290.

Werch CC, Moore MJ, DiClemente CC, et al (2005b) A multihealth behavior intervention integrating physical activity and substance use prevention for adolescents. Prevention Science 6(3):213-26.

Werch CE, Anzalone DM, Brokiewicz LM, et al (1996) An intervention for preventing alcohol use among inner-city middle school students. Archives of Family Medicine 5(3):146-52.

Werch CE, Bian H, Moore MJ, et al (2008a) Brief multiple behavior health interventions for older adolescents. American journal of health promotion : AJHP 23(2):92-6.

Werch CE, Carlson JM, Owen DM, et al (2001) Effects of a stage-based alcohol preventive intervention for inner-city youth. Journal of Drug Education 31(2):123-38.

Werch CE, Carlson JM, Pappas DM, et al (2000a) Effects of a brief alcohol preventive intervention for youth attending school sports physical examinations. Substance Use and Misuse 35(3):421-432.

Werch CE, Moore MJ, DiClemente CC (2008b) Brief image-based health behavior messages for adolescents and their parents. Journal of Child & Adolescent Substance Abuse 17(4):19-41.

Werch CE, Owen DM, Carlson JM, DiClemente CC, Edgemon P, Moore M (2003a) One-year followup results of the STARS for Families alcohol prevention program. Health Education Research 18(1):74-87.

Werch CE, Pappas DM, Carlson JM, et al (2000b) Evaluation of a brief alcohol prevention program for urban school youth. American Journal of Health Behavior 24(2):120-31.

Werch CE, Owen DM (2002). latrogenic effects of alcohol and drug prevention programs. (43 refs.). Journal of Studies on Alcohol 63 (5): 581-590.

Werch CE, Pappas DM, Carlson JM, et al. (1998). Short- and long-term effects of a pilot prevention program to reduce alcohol consumption. Substance Use & Misuse 33 (11): 2303-2321.

Werch C, Moore M, DiClemente CC, et al. (2003b). A sport-based intervention for preventing alcohol use and promoting physical activity among adolescents. Journal of School Health 73 (10): 380-388.

White D, Buckley E, Hassan J (2004). Literature review on the role of external contributors in school drug, alcohol and tobacco education. London: Department for Education and Skills.

Wight D, Raab GM, Henderson M, et al (2002) Limits of teacher delivered sex education: Interim behavioural outcomes from randomised trial. British Medical Journal 324(7351):1430-1433.

Wilhelmsen BU, Laberg JC, Klepp K-I (1994) Evaluation of Two Student and Teacher Involved Alcohol Prevention Programmes. Addiction 89(9):1157-1165.

Williams CL, Perry CL, Dudovitz B, et al. (1995). A Home-Based Prevention Program for Sixth-Grade Alcohol Use: Results from Project Northland. Journal-of-Primary-Prevention 16 (2): 125-147.

Workman GM, LaVome Robinson W, Cotler S, Harper GW (1996) A school-based approach to HIV prevention for inner-city African- American and Hispanic adolescent females. Journal of Prevention and Intervention in the Community 14(1-2):41-60.

Wright NP (1998) Effects of a high school sexuality education program, in Dissertation Abstracts International Section A: Humanities and Social Sciences, vol 58.

Zimmerman RS, Cupp PK, Donohew L, et al (2008) Effects of a school-based, theory-driven HIV and pregnancy prevention curriculum. Perspectives on Sexual and Reproductive Health 40:42-51.
Appendix 2. References to excluded studies

• Study did not meet design criteria for inclusion (n=320)

Anon (1993) California Programs To Prevent and Reduce Drug, Alcohol, and Tobacco Use among In-School Youth: An Interim Report about Tobacco Use. Report No. 2, pp 18. Southwest Regional Lab., Los Alamitos, CA.

Anon (1997) School-Based HIV-Prevention Education--United States, 1994. Journal of School Health 67(3):103-105.

Anon (2001) Halfway There: A Prescription for Continued Progress in Preventing Teen Pregnancy, pp 42. National Campaign To Prevent Teen Pregnancy, Washington, DC.

Anon (2007) Review of Comprehensive Sex Education Curricula, pp 40. Administration for Children and Families (DHHS), Washington, DC.

Anon (2008) Preventing Teenage Pregnancy in Looked-after Children. Community Care:24-25.

Abel EM, Greco M (2008) A Preliminary Evaluation of an Abstinence-Oriented Empowerment Program for Public School Youth. Research on Social Work Practice 18:223-230.

Aggleton P, Brownjohn C, Chalmers H, Tyrer P (2005) SRE: a programme to support community nurses. Community Practitioner 78(6):199-200.

Allen JP, Philliber S (1991) Evaluating Why and How the Teen Outreach Program Works: Years 3-5 of the Teen Outreach National Replication (1986/87-1988/89), pp 12. Association of Junior Leagues, Inc., New York, NY.

Allensworth DD (1994) The research base for innovative practices in school health education at the secondary level. Journal of School Health 64(5):180-188.

Alstead M, Campsmith M, Halley CS, Hartfield K, Goldblum G, Wood RW (1999) Developing, Implementing, and Evaluating a Condom Promotion Program Targeting Sexually Active Adolescents. AIDS Education and Prevention 11(6):497-512.

Archer E, Cahill M (1991) Building Life Options: School-Community Collaborations for Pregnancy Prevention in the Middle Grades, pp 147. Academy for Educational Development, Washington, DC.

Arnold EM, Smith TE, Harrison DF, Springer DW (1999) The effects of an abstinence-based sex education program on middle school students' knowledge and beliefs. Research on Social Work Practice 9:10-24.

Arnold EM, Smith TE, Harrison DF, Springer DW (2000) Adolescents' knowledge and beliefs about pregnancy: the impact of "ENABL". Adolescence 35(139):485-98.

Ashworth CS, DuRant RH, Newman C, Gaillard G (1992) An evaluation of a school-based AIDS/HIV education program for high school students. Journal of Adolescent Health 13(7):582-588.

Ateka GK, Selwyn BJ (2007) Adolescent human immunodeficiency virus and sexually transmitted disease prevention programmes: are gender-blind approaches justified? Public Health 121(9):682-689.

Bagnall G, Lockerbie L (1996) HIV and AIDS education for senior school pupils in Scotland: a quantitative evaluation. Health Education Journal 55(2):141-55.

Barnes ND, Harrod SE (1993) Teen pregnancy prevention: a rural model using school and community collaboration. School Counselor 41(2):137-40.

Barnett JE, Hurst CS (2004) Do adolescents take 'baby think it over' seriously? Adolescence 39(153):65-75.

Basen EK, Parcel GS, Harrist R, Kirby D, Coyle K, Banspach S, Rugg D (1997) The Safer Choices Project: Methodological Issues in School-Based Health Promotion Intervention Research. Journal of School Health 67(9):365-71.

Bearss N, Santelli JS, Papa P (1995) A pilot program of contraceptive continuation in six schoolbased clinics. Journal of Adolescent Health 17(3):178-83.

Beath L, MacRae-Stevens D, Walters J (1998a) SIECCAN Newsletter: vol. 33, no. 3. Student health services: comprehensive sexual health & STD programming in local secondary schools. Canadian Journal of Human Sexuality 7(4):371-374.

Beath L, MacRae-Stevens D, Walters J (1998b) Student Health Services: Comprehensive sexual health and STD programming in local secondary schools. Canadian Journal of Human Sexuality 7(4):371-373.

Beavet T, Thompson J (1996) Parents talking: sex education in secondary schools. Pastoral Care in Education 14:11-18.

Bechtel LJ, Vicary J, Swisher J, Smith E, Hopkins A, Henry K, Minner D (2006) An interdisciplinary approach for the integration and diffusion of substance abuse prevention programs. American Journal of Health Education 37(4):219-226.

Becker MG, Barth RP (2000) Power through choices: the development of a sexuality education curriculum for youths in out-of-home care. Child Welfare 79(3):269-82.

Bell ML, Bliss K (1997) Project SAVE: Evaluation of Pilot Test Results, pp 70. Texas Alcoholic Beverage Commission, Austin.

Bishop JB (2000) An environmental approach to combat binge drinking on college campuses. Journal of College Student Psychotherapy 15(1):15-30.

Blake SM, Ledsky R, Goodenow C, Sawyer R, Lohrmann D, Windsor R (2003) Condom Availability Programs in Massachusetts High Schools: Relationships with Condom Use and Sexual Behavior. American Journal of Public Health 93(6):955-962. Boothroyd RI (2004) Analyzing the contribution of environmental change to prevent adolescent pregnancy: A study of three multi-component community initiatives in Kansas, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 65.

Borgia P, Spadea T, Perucci CA, De Pascali V, Fano V, Schifano P, Abeni DDC (1997) Limited effectiveness of a school-based HIV prevention campaign in Italy: a randomised controlled trial. European Journal of Public Health 7(4):411-418.

Brooks RG, Schwendinger A, Smith TE, Spaulding-Givens J, Steen JA (2005) Gender differences in adolescent attitudes and receptivity to sexual abstinence education. Children & Schools 27(1):45-50.

Brown JH, And O (1995) In Their Own Voices: Students and Educators Evaluate California School-Based Drug, Alcohol, and Tobacco, Education (DATE) Programs, pp 63. Pacific Inst. for Research and Evaluation, Bethesda, MD.

Brown HN, Saunders RB (2002) College-bound sisters. Exploring one pregnancy prevention program. AWHONN Lifelines 6(2):146-51.

Brown L, Hutchinson P, Karim AM, Macintyre K, Magnani R (2005a) The impact of life skills education on adolescent sexual risk behaviors in KwaZulu-Natal, South Africa. Journal of Adolescent Health 36(4):289-304.

Brown SA (2001) Facilitating change for adolescent alcohol problems: A multiple options approach, in Innovations in adolescent substance abuse interventions., pp 169-187.

Brown SA, Anderson KG, Schulte MT, Sintov ND, Frissell KC (2005b) Facilitating youth self-change through school-based intervention. Addictive Behaviors 30(9):1797-1810.

Brown SM (1997) Design, implementation and evaluation of a pilot sexuality education curriculum for middle school special education students, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 58.

Buckhalt JA, And O (1990) Rural Drug Abuse Prevention: Establishing Needs and Implementing Programs, pp 15.

Buysse A, Van Oost P (1997) Impact of a school-based prevention programme on traditional and egalitarian adolescents' safer sex intentions. Journal of Adolescence 20(2):177-188.

Cagampang HH (1997) Education now and babies later (ENABL): Life history of a campaign to postpone sexual involvement. Family Planning Perspectives 29(3):109-114.

Campbell C, Foulis C-A (2002) Creating Contexts That Support Youth-Led HIV Prevention in Schools. Society in Transition 33(3):312-329.

Carlson CE (1990) HIPP: A comprehensive school-based substance abuse program with cooperative community involvement. Journal of Primary Prevention 10(4):289-302.

Carter DSG (1999) A Whole-School Approach to Adolescent Peer-Leader Development for Affective Learning in Health-Related Curricula. Research Papers in Education Policy and Practice 14(3):295-319.

Carter DSG, Carter SM (1993) The Peer Leader AIDS Education Project, pp 23.

Carter SM, Carter DS (1993) Gender differentiated receptivity to sexuality education curricula by adolescents. Health Education Research 8(2):233-43.

Castro-Vazquez G, Kishi I (2002) "If you say to them that they have to use condoms, some of them might use them. It is like drinking alcohol or smoking": An educational intervention with Japanese senior high school students. Sex Education 2:105-117.

Centers for Disease C, Prevention (1996) School-based HIV-prevention education--United States, 1994. MMWR - Morbidity & Mortality Weekly Report 45(35):760-5.

Chan W, Lederman RP, Roberts-Gray C (2004) Sexual risk attitudes and intentions of youth aged 12-14 years: survey comparisons of parent-teen prevention and control groups. Behavioral Medicine 29(4):155-163.

Chang VN (1993) Prevent and empower: a student-to-student strategy with alcohol abuse. Social Work in Education 15(4):207-214.

Chapman RJ (1990) The La Salle University FIPSE Grant: A Review and Evaluation of the Alcohol and Other Drug Program, 1 September 88 through 30 September 90, pp 43. La Salle Univ., Philadelphia, PA.

Chatterji P (2001) Applying cost analysis methods to school-based prevention programs. Prevention Science 2(1):45-55.

Christopher FS (1995) Adolescent Pregnancy Prevention. Family Relations 44(4):384-91.

Cohen DA (2004) Comparing the cost-effectiveness of HIV prevention interventions. Journal of Acquired Immunodeficiency Syndrome 37(3):1404-1414.

Cole BP, Nelson TD, Steele RG (2008) An evaluation of a peer-based HIV/AIDS education program as implemented in a suburban high school setting. Journal of HIV/AIDS Prevention in Children & Youth 9(1):84-97.

Coll KM (1998) Effectiveness of an Alcohol Education Program Among Community College Students. Community College Journal of Research and Practice 22(2):123-31.

Collins J, Robin L, Wooley S, Fenley D, Hunt P, Taylor J, Haber D, Kolbe L (2002) Programs-That-Work: CDC's Guide to Effective Programs that Reduce Health-Risk Behavior of Youth. Journal of School Health 72(3):93-99.

Craft A (1991) The Living Your Life Programme. British Journal of Special Education 18(4):157-60.

Craft A (1994) Sex education for students with severe learning difficulties. International Journal of Adolescent Medicine and Health 7(2):133-140.

D'Amico EJ, Fromme K (2002) Brief prevention for adolescent risk-taking behavior. Addiction 97(5):563-74.

Damond ME, And O (1993) The Evaluation of Setting and a Culturally Specific HIV/AIDS Curriculum: HIV/AIDS Knowledge and Behavioral Intent of African American Adolescents. Journal of Black Psychology 19(2):169-89.

Dark LS (1996) Peer approaches for increasing HIV awareness on a college campus. ABNF Journal 7(2):54-6.

Davis CN, Sr. (1995) Family Life Education Programs and Student Attitudes, pp 12.

De GJF, And O (1994) Teacher Philosophy and Program Implementation and the Impact on Sex Education Outcomes. Journal of Research and Development in Education 27(4):265-70.

Deck DD (2004) Addressing Adolescent Substance Abuse: An Evaluation of Washington's Prevention and Intervention Services Program. 2001-03 Final Report, pp 72. Washington Office of the State Superintendent of Public Instruction, Olympia.

DeMairo P, Dischell J, Jouthe SA, Horner A (2008) The Teen Outreach Reproductive Challenge: Improving Adolescent Health Care Delivery through Peer Education Projects, in American Journal of Sexuality Education, vol 3, pp 1-17.

DeRidder LM (1993) Teenage Pregnancy: Etiology and Educational Interventions. Educational Psychology Review 5(1):87-107.

Devaney B, Johnson A, Maynard R, Trenholm C (2002) The Evaluation of Abstinence Education Programs Funded Under Title V Section 510: Interim Report, pp 57. Mathematica Policy Research, Princeton, NJ.

DiClemente RJ, Crosby RA (2006) Preventing sexually transmitted infections among adolescents: 'the glass is half full'. Current Opinion in Infectious Diseases 19(1):39-43.

Dielman TE (1994) School-based research on the prevention of adolescent alcohol use and misuse: methodological issues and advances. Journal of Research on Adolescence 4(2):271-93.

Dietz E (1996) Best practices in preventing drug abuse and violence. Intervention in School and Clinic 31(5):313-18.

Donati S, Medda E, Spinelli A, Grandolfo ME (2000) Sex education in secondary schools: An Italian experience. Journal of Adolescent Health 26(4):303-308.

Dotterer S, Axiotis IR (1994) Sex-Tac-Toe. Journal of American College Health 43(3):137-9.

Drolet J, Clark K (1994) The Sexuality Education Challenge: Promoting Healthy Sexuality in Young People. ETR Associates, , Santa Cruz, CA.

Dryfoos JG (1991) Adolescents at risk: A summation of work in the field: Programs and policies. Journal of Adolescent Health 12(8):630-637.

Dunn L, Ross B, Caines T, Howorth P (1998) A school-based HIV/AIDS prevention education program: Outcomes of peer-led versus community health nurse-led interventions. Canadian Journal of Human Sexuality 7(4):339-345.

Dycus S, Costner GM (1990) Healthy Early-Adolescent Development (11-13 Year Olds): Implementing a Human Sexuality Curriculum for Seventh Graders. Elementary School Guidance and Counseling 25(1):46-53.

Dykeman C, Nelson JR (1996) Students' Evaluations of the Effectiveness of Substance Abuse Education: The Impact of Different Delivery Modes. Journal of Child & Adolescent Substance Abuse 5(2):43-61.

Ebert M (2003) Peer programs in prevention of sexually transmitted infections. Acta Dermatovenerologica Alpina, Pannonica et Adriatica 12(2):77-79.

Ebreo A, Feist-Price S, Siewe Y, Zimmerman RS, Main DS (2002) Effects of peer education on the peer educators in a school-based HIV prevention program: where should peer education research go from here? Health Education and Behavior 29(4):411-26.

Elliott KJ, Lambourn AJ (1999) Sex, drugs and alcohol: Two peer-led approaches in Tamaki Makaurau/Auckland, Aotearoa/New Zealand. Journal of Adolescence 22(4):503-513.

Emmerson L (2008) On-site sexual health services in further education are the norm. Education & Health 26(1):10-12.

Ertle V (1993) Sharing Your Success III. Summaries of Successful Programs and Strategies Supporting Drug-Free Schools and Communities. Volume III, pp 175. Western Regional Center for Drug-Free Schools and Communities, Portland, OR.

Faggiano F, Richardson C, Bohrn K, Galanti MR (2007) A cluster randomized controlled trial of school-based prevention of tobacco, alcohol and drug use: The EU-Dap design and study population. Preventive Medicine 44(2):170-173.

Ferguson JL, And O (1991) Experimental Evaluation of "AIDS and You": A Pamphlet for High School Special Education Students, pp 9.

Fisher G, And O (1992) Evaluation of HIV Prevention and Comprehensive Health Education Activities, pp 25.

Fisher LA, Scott DM, Chase JM, Smith MA (2000) Peer-Led Alcohol Education program: a pharmacy student-led program for seventh-graders. Journal of the American Pharmaceutical Association 40(1):82-6.

Fletcher A, Bonell C, Hargreaves J (2008) School effects on young people's drug use: a systematic review of intervention and observational studies (Structured abstract), in Journal of Adolescent Health, vol 42, pp 209-220.

Fletcher A, Harden A, Brunton G, Oakley A, Bonell C (2008) Interventions addressing the social determinants of teenage pregnancy. Health Education 108:29-39.

Fongkaew W, Fongkaew K, Muecke M (2006) HIV/sexual and reproductive health program for HIV prevention: the youth-adult partnership with schools approach. Journal of the Medical Association of Thailand 89(10):1721-32.

Forrest S, Strange V, Oakley A, Johnson A, Stephenson J, Black S, Flux A, Charleston S (2004) What do young people want from sex education: The results of a needs assessment from a peer-led sex education programme. Culture, Health and Sexuality 6(4):337-354.

Franklin C, Corcoran J (2000) Preventing adolescent pregnancy: a review of programs and practices. Social Work 45(1):40-52.

Frisiras S, Lagiou A, Sourtzi P, Vidalaki M (1991) Sex education and family planning messages in Greek school books. Planned Parenthood in Europe 20(1):16-7.

Fromme K, Brown SA (2000) Empirically based prevention and treatment approaches for adolescent and young adult substance use. Cognitive and Behavioral Practice 7(1):61-64.

Frost JJ, Forrest JD (1995) Understanding the impact of effective teenage pregnancy prevention programs. Family Planning Perspectives 27(5):188-195.

Gardner SP, Boellaard R (2007) Does youth relationship education continue to work after a high school class? A longitudinal study. Family Relations 56:490-500.

Garman JF, Crider DA (2002) HIV/AIDS Education: A Six-Year Comparison of Secondary Level Effectiveness, pp 33.

Gay KEM (1997) Collaborative school-based research: Evaluation of an academically integrated HIV/AIDS prevention curriculum for middle school students, in Dissertation Abstracts International Section A: Humanities and Social Sciences, vol 57.

Gaydos CA, Hsieh YH, Galbraith JS, Barnes M, Waterfield G, Stanton B (2008) Focus-on-Teens, sexual risk-reduction intervention for high-school adolescents: Impact on knowledge, change of risk-behaviours, and prevalence of sexually transmitted diseases. International Journal of STD and AIDS 19(10):704-710.

Ghosh-Dastidar B, Longshore DL, Ellickson PL, McCaffrey DF (2004) Modifying pro-drug risk factors in adolescents: results from Project ALERT. Health Education & Behavior 31(3):318-335.

Gilliland L, Scully J (2005) STI-HIV prevention: a model program in a school-based health center. Nursing Clinics of North America 40(4):681-689.

Glanz K, Maddock JE, Shigaki D, Sorensen CA (2003) Preventing underage drinking: a "roll of the dice". Addictive Behaviors 28(1):29-38.

Glover BL (1995) DINOS (drinking is not our solution): Using computer programs in middle school drug education. Elementary School Guidance & Counseling 30(1):55-62.

Goldsworthy R, Schwartz N (2008) Development and evaluation of a multimedia-enhanced STD/HIV curriculum for middle schools. Journal of Educational Multimedia and Hypermedia 17:413-444.

Goodman J (1991) Redirecting sexuality education for young adolescents. Curriculum and Teaching 6:12-22.

Gorman DM (2002) The "science" of drug and alcohol prevention: The case of the randomized trial of the Life Skills Training program. International Journal of Drug Policy 13(1):21-26.

Graham JW, And O (1991) Modeling Transitions in Latent Stage-Sequential Processes: A Substance Use Prevention Example. Journal of Consulting and Clinical Psychology 59(1):48-57.

Griffin P, Perrotti J, Priest L, Muska M (2002) It Takes a Team! Making Sports Safe for Lesbian, Gay, Bisexual, and Transgender Athletes and Coaches. An Education Kit for Athletes, Coaches, and Athletic Directors, pp 93. Women's Sports Foundation.

Gruen SM, Hayes E, Fritsch-deBruyn R (1991) Setting up a school-based sexual education program to help prevent AIDS and other sexually transmitted diseases. Nurse Practitioner 16(8):47-51.

Guttmacher S, Lieberman L, Ward D, Freudenberg N, Radosh A, Jarlais DD (1997) Condom availability in New York City public high schools: Relationships to condom use and sexual behavior. American Journal of Public Health 87(9):1427-1433.

Guzman BL, Casad BJ, Schlehofer SMM, Villanueva CM, Feria A (2001) C.A.M.P.: A Community-Based Approach to Promoting Safe Sex Behavior in Adolescence, vol 13, pp 40.

Haggerty KP, MacKenzie EP, Skinner ML, Harachi TW, Catalano RF (2006) Participation in "parents who care": predicting program initiation and exposure in two different program formats, in The Journal of Primary Prevention, vol 27, pp 47-65.

Haines MP, Barker GP, Rice R (2003) Using social norms to reduce alcohol and tobacco use in two midwestern high schools, in The social norms approach to preventing school and college age substance abuse: A handbook for educators, counselors, and clinicians., pp 235-244. Jossey-Bass.

Hanna KM, Hanrahan S, Hershey J, Greer D (1997) Evaluation of the effect of an AIDS prevention program on high school students' sexual risk-taking behavior. Issues in Comprehensive Pediatric Nursing 20(1):25-34.

Harden A, Weston R, Oakley A (1999) A review of the effectiveness and appropriateness of peerdelivered health promotion interventions for young people. Evaluation of Health Promotion and Social Interventions:180.

Hardy JB, Zabin LS (1991) Adolescent Pregnancy in an Urban Environment: Issues, Programs, and Evaluation. University Press of America, Lanham, MD.

Harvey J, McLachlin J (1998) STD teaching in high schools. Canadian Journal of Human Sexuality 7(4):383-384.

Haynes NM (1993) Reducing the Risk of Teen Parenthood: An Approach That Works. NASSP Bulletin 77(557):36-40.

Henderson M, Wight D, Raab GM, Abraham C, Parkes A, Scott S, Hart G, Reading R (2007) Impact of a theoretically based sex education programme (SHARE) delivered by teachers on NHS registered conceptions and terminations: final results of cluster randomised trial. Child: Care, Health and Development 33(3):349-349.

Higgins D, King R, Witthaus D (2001) Pride and prejudice: facilitating change in the attitudes of students toward gay men and lesbians. Health Promotion Journal of Australia 12(3):238-242.

Hoyt HH, Broom BL (2002) School-based teen pregnancy prevention programs: A review of the literature, vol 18, pp 11-17.

Hughes C, Julian R, Richman M, Mason R, Long G (2008) Harnessing the power of perception : reducing alcohol-related harm among rural teenagers. Youth Studies Australia 27:26-35.

Hulton LJ (2007) An evaluation of a school-based teenage pregnancy prevention program using a logic model framework. The Journal of School Nursing 23:104-110.

Jacobs CD, Wolf EM (1995) School Sexuality Education and Adolescent Risk-Taking Behavior. Journal of School Health 65(3):91-95.

James WH (1992) A Report on Restructuring Schools: Reducing Alcohol and Other Drug Abuse among Adolescents through a School-Community-University Partnership Program in the Federal Way School District, pp 59. Washington Univ., Seattle. Center for the Study and Teaching of At- Risk Students.

Jarmon BBJ (1998) The Real Life Cost Model: An Early Intervention Strategy for Preventing Teen Fatherhood and Motherhood. Reclaiming Children and Youth Journal of Emotional and Behavioral Problems 7(1):34-38.

Jason LA, Curie CJ, Townsend SM, Pokorny SB, Katz RB, Sherk JL (2002) Health promotion interventions. Child & Family Behavior Therapy 24(1/2):67-83.

Johnson G, Jefferson ACR (2001) HIV/AIDS Prevention: Effective Instructional Strategies for Adolescents with Mild Mental Retardation. TEACHING Exceptional Children 33(6):28-32.

Jorgensen SR (1991) Project Taking Charge: An evaluation of an adolescent pregnancy prevention program. Family Relations 40(4):373-380.

Jurich JA, Adams RA, Schulenberg JE (1992) Factors related to behavior change in response to AIDS. Family Relations 41(1):97-103.

Kerpelman JL, Pittman JF, Adler-Baeder F (2008) Identity as a moderator of intervention-related change: Identity style and adolescents' responses to relationships education. Identity 8:151-171.

Kidger J (2004) 'You realise it could happen to you': The benefits to pupils of young mothers delivering school sex education, vol 4, pp 185-197.

Kims A (1999) Reinforcing Resistance to Drug and Alcohol Use through Teen Role Models. NERCRD Project Report, pp 7. Northeast Regional Center for Rural Development, University Park, PA.

Kinsman J, Harrison S, Kengeya-Kayondo J, Kanyesigye E, Musoke S, Whitworth J (1999) Implementation of a comprehensive AIDS education programme for schools in Masaka District, Uganda. AIDS Care - Psychological and Socio-Medical Aspects of AIDS/HIV 11(5):591-601.

Kirby D (1992) School-based programs to reduce sexual risk-taking behaviors. Journal of School Health 62(7):280-7.

Kirby D (1999) Reflections on Two Decades of Research on Teen Sexual Behavior and Pregnancy. Journal of School Health 69(3):89-94.

Kirby D (2001) Emerging Answers: Research Findings on Programs To Reduce Teen Pregnancy, pp 242. National Campaign To Prevent Teen Pregnancy, Washington, DC.

Kirby D (2002) The impact of schools and school programs upon adolescent sexual behavior. Journal of Sex Research 39(1):27-33.

Kirby D, DiClemente RJ (1994) School-based interventions to prevent unprotected sex and HIV among adolescents, in Preventing AIDS: Theories and methods of behavioral interventions., pp 117-139.

Kirby D, Korpi M, Barth RP, Cagampang HH (1997) The impact of the postponing sexual involvement curriculum among youths in California. Family Planning Perspectives 29(3):100-108.

Klaus H (1996) The Teen STAR Program, pp 11.

Klaus H (1998) The International Teen STAR Program, pp 10. Teen STAR Program, Bethesda, MD.

Komro KA, And O (1994) Peer Participation in Project Northland: A Community-Wide Alcohol Use Prevention Project. Journal of School Health 64(8):318-22.

Komro KA, And O (1996) Peer-Planned Social Activities for Preventing Alcohol Use among Young Adolescents. Journal of School Health 66(9):328-34.

Komro KA, Toomey TL (2002) Strategies to prevent underage drinking. Alcohol Research & Health 26(1):5-14.

Koo HP, Dunteman GH, George C, Green Y, Vincent M (1994) Reducing adolescent pregnancy through a school- and community-based intervention: Denmark, South Carolina, revisited. Family Planning Perspectives 26(5):206-211+217.

Kourtis AP, Kraft JM, Gavin L, Kissin D, McMichen-Wright P, Jamieson DJ (2006) Prevention of sexually transmitted human immunodeficiency virus (HIV) infection in adolescents. Current HIV Research 4(2):209-219.

Koutakis N, Stattin H, Kerr M (2008) Reducing youth alcohol drinking through a parent-targeted intervention: The Orebro prevention program. Addiction 103(10):1629-1637.

Kralewski J, Stevens-Simon C (2000) Does mothering a doll change teens' thoughts about pregnancy?[see comment]. Pediatrics 105(3):E30.

Kreft IGG (1997) The Evaluation of Drug Prevention Programs and Its Effect on Alcohol Consumption of Junior High School Students, pp 35.

Lagana L, Hayes DM (1993) Contraceptive health programs for adolescents: a critical review. Adolescence 28(110):347-59.

Lamelas CAC, Henderson DL (1996) Adolescent Sex Education Knowledge in American Schools of Mexico, pp 26.

Larkin J, Andrews A, Mitchell C (2006) Guy talk: Contesting masculinities in HIV prevention education with Canadian youth. Sex Education 6:207-221.

Laurs MR (1990) Alcohol Prevention in New Zealand: Working towards a National Alcohol Policy. The International Journal of the Addictions 25:513-532.

Lawrence P, Swain J (1993) Sex education programmes for students with severe learning difficulties in further education and the problem of evaluation. Disability, Handicap & Society 8(4):405-421.

Lazebnik R, Grey SF, Ferguson C (2001) Integrating substance abuse content into an HIV riskreduction intervention: A pilot study with middle school-aged hispanic students. Substance Abuse 22(2):105-117.

Levy SR, Weeks K, Handler A, Perhats C, Franck JA, Hedeker D, Zhu C, Flay BR (1995) A longitudinal comparison of the AIDS-related attitudes and knowledge of parents and their children. Family Planning Perspectives 27(1):4-10.

Lewis J, Knijn T (2001) A comparison of English and Dutch sex education in the classroom. Education & Health 19(4):59-65.

Lilja J, Wilhelmsen BU, Larsson S, Hamilton D (2003) Evaluation of drug use prevention programs directed at adolescents. Substance Use and Misuse 38(11-13):1831-1863+1923-1926.

Ling DC, Tremblay CH (2005) AIDS education, condom demand, and the sexual activity of American youth. Health Economics 14(8):851-867.

Lloyd K, Lyth N (2003) Evaluation of the use of drama in sex and relationship education. Nursing Times 99(47):32-4.

LoSciuto L, Steinman RB (2004) A re-evaluation of project pride, a redesigned school-based drug abuse prevention program. Journal of Drug Education 34(2):155-66.

Lustig SL (1994) The AIDS prevention magic show: avoiding the tragic with magic. Public Health Reports 109(2):162-7.

Lutzer VD, Day VN (2001) The Healthy Relationships Program: Preventing Sexual Assault of Youth. Reclaiming Children and Youth 10(3):158-61.

Lynagh M, Knight J, Schofield MJ, Paras L (1999) Lessons learned from the Hunter Region Health Promoting Schools Project in New South Wales, Australia. Journal of School Health 69(6):227-32. Lynagh M, Schofield MJ, Sanson-Fisher RW (1997) School health promotion programs over the past decade: A review of the smoking, alcohol and solar protection literature. Health Promotion International 12(1):43-60.

MacKillop J, Ryabchenko KA, Lisman SA (2006) Life Skills Training outcomes and potential mechanisms in a community implementation: A preliminary investigation. Substance Use and Misuse 41(14):1921-1935.

MacKinnon DP, Johnson CA, Pentz MA, Dwyer JH, Hansen WB, Flay BR, Wang EY (1991) Mediating mechanisms in a school-based drug prevention program: first-year effects of the Midwestern Prevention Project. Health Psychology 10(3):164-72.

Mahat G, Scoloveno MA, Ruales N, Scoloveno R (2006) Preparing peer educators for teen HIV/AIDS prevention. Journal of Pediatric Nursing 21(5):378-84.

Makin J, Butler S (2001) 'Tell me about it' : a community-based project to reduce the rate of teen pregnancy in Wagga Wagga. Youth Studies Australia 20:49-52.

Mann RE, Smart RG (1997) Can school alcohol education programmes contribute to declines in drinking among students? Drugs: Education, Prevention and Policy 4(2):131-7.

Mantell JE, Harrison A, Hoffman S, Smit JA, Stein ZA, Exner TM (2006) The Mpondombili Project: preventing HIV/AIDS and unintended pregnancy among rural South African school-going adolescents. Reproductive Health Matters 14(28):113-22.

Marechal C, Choquet M (1990) Alcohol prevention among adolescents: a French experiment. Drug & Alcohol Dependence 26(3):227-33.

Marick J (2002) HIV/AIDS peer education: A rural health project. The Journal of School Nursing 18:41-47.

Matjasko JL (2001) Are School-Based Prevention and Intervention Programs Effective? Their Impact on At-Risk Adolescent Development, in Dissertation Abstracts International, A: The Humanities and Social Sciences, vol 62, pp 791-a.

Mattox JR, II (1997) A Review of Interventions To Increase Driving Safety among Teenage Drivers, pp 26.

May C (1991) Research on alcohol education for young people: a critical review of the literature. Health Education Journal 50(4).

McCowan RJ, Roberts SW (2004) The effectiveness of infant simulators. Adolescence 39(155):475-487.

McKay A (2000) Common questions about sexual health education. The Canadian Journal of Human Sexuality 9(2):129-137.

McKay A (2000) Prevention of Sexually Transmitted Infections in Different Populations: A Review of Behaviourally Effective and Cost-Effective Interventions. Canadian Journal of Human Sexuality 9(2):95-120.

Merakou K, Kourea-Kremastinou J (2006) Peer education in HIV prevention: An evaluation in schools. European Journal of Public Health 16(2):128-132.

Milburn K (1995) A critical review of peer education with young people with special reference to sexual health. Health Education Research.

Mitchell JM (2003) AIDS Education and Citizenship Development in a University/K-12 Partnership, pp 10.

Montoya ID, Atkinson J, McFaden WC (2003) Best characteristics of adolescent gateway drug prevention programs. Journal of Addictions Nursing 14(2):75-84.

Moore KA, And O (1995) Adolescent Pregnancy Prevention Programs: Interventions and Evaluations, pp 508. Child Trends, Inc., Washington, DC.

Moore L, Graham A, Diamond I (2003) On the Feasibility of Conducting Randomised Trials in Education: Case study of a sex education intervention. British Educational Research Journal 29:673-689.

Morton M, Nelson L, Walsh C, Zimmerman S, Coe RM (1996) Evaluation of a HIV/AIDS education program for adolescents. Journal of Community Health 21(1):23-35.

Murphy JG, Correia CJ, Barnett NP (2007) Behavioral economic approaches to reduce college student drinking. Addictive Behaviors 32:2573-2585.

Murray DM, Clark MH, Wagenaar AC (2000) Intraclass correlations from a community-based alcohol prevention study: the effect of repeat observations on the same communities. Journal of Studies on Alcohol 61(6):881-90.

Murray-Johnson L, Witte K, Liu WY, Hubbell AP, Sampson J, Morrison K (2001) Addressing cultural orientations in fear appeals: promoting AIDS-protective behaviors among Mexican immigrant and African American adolescents and American and Taiwanese college students. Journal of Health Communication 6(4):335-58.

Nader PR, Rissel C, Rowling L, Marshall BJ, Sheehan MM, Northfield JR, Maher S, Carlisle R, St LLH, Stewart DE, Parker E, Gillespie A, Stokes H, Mukherjee D, Nutbeam D, Mitchell A, Ollis D, Watson J (2000) School Health Programs in Australia - A Special Insert. Journal of School Health 70(6):247-264.

Nielsen SK (1995) An ecological assessment of the school-based substance abuse psychological services for Native American youth, in Dissertation Abstracts International Section A: Humanities and Social Sciences, vol 55.

No authorship i (2006) Section III: Promoting Health and Well-Being: Effective Interventions and Resources.

Noell J, Ary D, Duncan T (1997) Development and evaluation of a sexual decision-making and social skills program: "the choice is yours--preventing HIV/STDs". Health Education & Behavior 24(1):87-101.

O CWR, Jr. (1999) Report of a Study of Successful Programs of HIV Prevention Education in Colleges and Universities, pp 46. National Association of Student Personnel Administrators, Inc.

O'Hara P, Messick BJ, Fichtner RR, Parris D (1996) A peer-led AIDS prevention program for students in an alternative school. Journal of School Health 66(5):176-82.

Olsen J, And O (1992) Student Evaluation of Sex Education Programs Advocating Abstinence. Adolescence 27(106):369-80.

Olsen JA, And O (1991) The Effects of Three Abstinence Sex Education Programs on Student Attitudes toward Sexual Activity. Adolescence 26(103):631-41.

Orange C (2002) Project Northland and the Power of Preventing Underage Drinking. Reclaiming Children and Youth 11(3):164-67.

Ormson S (2008) Blackpool Sixth Form College uses peer-peer support to promote health. Education & Health:12-13.

O'Sullivan LF, Cooper-Serber E, Kubeka M, Harrison A (2007) Body concepts: Beliefs about the body and efforts to prevent HIV and pregnancy among a sample of young adults in South Africa. International Journal of Sexual Health 19:69-80.

Otto N, Middleton J, Freker J (2002) Making Schools Safe: An Anti-Harassment Training Program from the Lesbian & Gay Rights Project of the American Civil Liberties Union, pp 99.

Out JW, Lafreniere KD (2001) Baby Think It Over: using role-play to prevent teen pregnancy. Adolescence 36(143):571-82.

Ozcebe H, Akin L (2003) Effects of peer education on reproductive health knowledge for adolescents living in rural areas of Turkey [2]. Journal of Adolescent Health 33(4):217-218.

Ozer EJ, Weinstein RS, Maslach C (1997) Adolescent AIDS prevention in context: the impact of peer educator qualities and classroom environments on intervention efficacy. American Journal of Community Psychology 25(3):289-323.

Paine-Andrews A, Vincent ML, Fawcett SB, Campuzano MK, et al. (1996) Replicating a community initiative for preventing adolescent pregnancy: From South Carolina to Kansas. Family & Community Health 19(1):14-30.

Panzer RA (2008) The effects of fear versus norm appeals and directive versus cognitively flexible designs in abstinence-centered multimedia education on teen sexual attitudes, intentions and behaviors, in Dissertation Abstracts International Section A: Humanities and Social Sciences, vol 68.

Pedrazzini A, McGowan H, Lucking L, Johanson R (2000) 'The trouble with sex - It always gets in the way': An evaluation of a peer-produced teenage pregnancy video. British Journal of Family Planning 26(3):131,133-135.

Peerbhoy D, Bourke C (2007) Icebreaker: The Evaluation. Health Education Journal 66:262-276.

Petersen J, Atkinson P, Petrie S, Gibin M, Ashby D, Longley P (2009) Teenage pregnancy-new tools to support local health campaigns. Health & Place 15(1):300-7.

Petronyte G, Zaborskis A, Veryga A (2007) Risk factors for alcohol use among youth and main aspects of prevention programs. Medicina (Kaunas, Lithuania) 43(2):103-9.

Phelps FA, Mellanby AR, Crichton NJ, Tripp JH (1994) Sex education: the effect of a peer programme on pupils (aged 13-14 years) and their peer leaders. Health Education Journal 53(2):127-39.

Philip MR (1994) SPARK Peer Helpers Program, 1992-93 Evaluation. OER Report, pp 64. New York City Board of Education, Brooklyn, NY. Office of Educational Research.

Pick De Weiss S, Givaudan M, Givaudan S (1993) Planeando Tu Vida: sex and family life education: fundamentals of development, implementation, and evaluation. International Journal of Adolescent Medicine & Health 6(3-4):211-24.

Pinkleton BE, Austin EW, Cohen M, Chen Y, Fitzgerald E (2008) Effects of a peer-led media literacy curriculum on adolescents' knowledge and attitudes toward sexual behavior and media portrayals of sex. Health Communication 23(5):462-473.

Piotrow PT, Rimon JG, 2nd (1995) "Enter-educate." Reaching youth with messages of sexual responsibility. Planned Parenthood Challenges (1):41-5.

Pittman V, Gahungu A (2006) Comprehensive Sexuality Education or Abstinence-Only Education: Which Is More Effective? Online Submission 3:32-91.

Puentes WJ, Wassel M (2003) Using peer health education to enhance family life education. Journal of School Nursing 19(6):313-8.

Raab M (1998) Condom availability in high school does not increase teenage sexual activity but does increase condom use. Family Planning Perspectives 30(1):48-50.

Rafferty Y, Radosh A (2000) AIDS prevention and condom availability in an urban school system: facilitating factors and programme challenges. Health Education Journal 59(1):50-69.

Raffoul PR, Lyles E (1993) If You Drink, an alcohol education program. Computers in Human Services 10(1):71-74.

Raj A, Decker MR, Murray JE, Silverman JG (2007) Gender differences in associations between exposure to school HIV education and protective sexual behaviors and sexually transmitted disease/HIV diagnosis among high school students. Sex Education 7(2):191-199.

Rector R (2002) The Effectiveness of Abstinence Education Programs in Reducing Sexual Activity among Youth. The Heritage Foundation Backgrounder, pp 11. Heritage Foundation, Washington, DC.

Reed NA, et al. (1992) An AIDS Prevention Program for Adolescents with Special Learning Needs. Journal of School Health 62(5):195-97.

Remez L (1994) Course on AIDS changed behavior and attitudes of high school students. Family Planning Perspectives 26(1):46-48.

Ringwalt C, Bliss K (2006) The Cultural Tailoring of a Substance Use Prevention Curriculum for American Indian Youth. Journal of Drug Education 36:19-177.

Roberto AJ, Zimmerman RS, Carlyle KE, Abner EL, Cupp PK, Hansen GL (2007) The effects of a computer-based pregnancy, STD, and HIV prevention intervention: a nine-school trial. Health Communication 21(2):115-24.

Rodrick AL, Bhavnagri NP (1997) Strategies for Overcoming Obstacles in AIDS Education for Preteens. Childhood Education 73(2):70-76.

Rohrbach LA, And O (1994) Parental Participation in Drug Abuse Prevention: Results from the Midwestern Prevention Project. Journal of Research on Adolescence 4(2):295-317.

Rolston B, Schubotz D, Simpson A (2005) Sex education in Northern Ireland schools: A critical evaluation, vol 5, pp 217-234.

Ross C, Richard L, Potvin L (1998) One year outcome evaluation of an alcohol and drug abuse prevention program in a Quebec high school. Canadian Journal of Public Health Revue Canadienne de Sante Publique. 89(3):166-70.

Sabatino M (1996) Sounding the Alarm on Safe and Drug-Free Schools. Safe and Drug-Free Schools and Communities. Final Report, 1995-96, pp 69. Austin Independent School District, TX. Dept. of Accountability, Student Services, and Research.

Safer LA, Harding CG (1993) Under pressure program: using live theatre to investigate adolescents' attitudes and behavior related to drug and alcohol abuse education and prevention. Adolescence 28(109):135-48.

Sager RW, Jr. (1993) The Effects of a Kansas Education Class on Students' Knowledge and Attitudes of Human Immunodeficiency Virus and Acquired Immunodeficiency Syndrome, pp 98.

Saltz E, Perry A, Cabral R (1994) Attacking the Personal Fable: Role-Play and Its Effect on Teen Attitudes toward Sexual Abstinence. Youth and Society 26(2):223-242.

Samuels S, Smith M (1993) Condoms in the Schools. Henry J. Kaiser Family Foundation, Menlo Park, CA.

Sanderson CA (2000) The effectiveness of a sexuality education newsletter in influencing teenagers' knowledge and attitudes about sexual involvement and drug use. Journal of Adolescent Research 15(6):674-681.

Scales P (1990) Developing capable young people: An alternative strategy for prevention programs. The Journal of Early Adolescence 10(4):420-438.

Schacht RM (1991) Alcohol and Substance Abuse Prevention Curriculum Project: An Assessment of the Effectiveness of the Implementation of the Beauty Way Curriculum and of the Curriculum Itself:

Technical Report on Evaluation Data, pp 89. Northern Arizona Univ., Flagstaff. Inst. for Human Development.

Schinke S, Schwinn T (2005) Gender-specific computer-based intervention for preventing drug abuse among girls. American Journal of Drug and Alcohol Abuse 31(4):609-616.

Schuster MA, Bell RM, Berry SH, Kanouse DE (1998) Impact of a high school condom availability program on sexual attitudes and behaviors. Family Planning Perspectives 30(2):67-72.

Scott DM, Surface JL, Friedli D, Barlow TW (1999) Effectiveness of Student Assistance Programs in Nebraska Schools. Journal of Drug Education 29(2):165-74.

Scott F, Chambers LW, Underwood J, Walter S, Pickard L (1990) AIDS seminars for senior grades in secondary school. Canadian Journal of Public Health 81(4):290-295.

Scott JL, Giedt A, Dimitrievich E (1998) A community medicine project involving an educational session in sexually transmitted diseases for high school students. South Dakota Journal of Medicine 51(9):345-6.

Scott PRD, Milsom GA, Milsom GL (1995) Teachers and parents too - An assessment of Dutch sexual health education. British Journal of Family Planning 21(1):20-21.

Selwyn N, Powell E (2007) Sex and Relationships Education in Schools: The Views and Experiences of Young People, in Health Education, vol 107, pp 219-231.

Sheehan M, Schonfeld C, Hindson E, Ballard R (1995) Alcohol education in an indigenous community school in Queensland, Australia. Drugs: Education, Prevention and Policy 2(3):259-73.

Shegog R, Markham C, Peskin M, Dancel M, Coton C, Tortolero S (2007) "It's your game": an innovative multimedia virtual world to prevent HIV/STI and pregnancy in middle school youth. Studies in Health Technology & Informatics 129(Pt 2):983-7.

Short R (1998) Teaching safe sex in school. International Journal of Gynaecology & Obstetrics 63 Suppl 1:S147-50.

Short RV (2004) Teaching safe sex in English schools. Lancet 364(9431):307-308.

Sidebottom A, Birnbaum AS, Nafstad SS (2003) Decreasing Barriers for Teens: Evaluation of a New Teenage Pregnancy Prevention Strategy in School-Based Clinics. American Journal of Public Health 93(11):1890-1892.

Sigelman C, And O (1996) Mexican-American and Anglo-American Children's Responsiveness to a Theory-Centered AIDS Education Program. Child Development 67(2):253-66.

Sigfusdottir ID, Thorlindsson T, Kristjansson AL, Roe KM, Allegrante JP (2009) Substance use prevention for adolescents: the Icelandic Model. Health Promotion International 24(1):16-25.

Silvia ES, Thorne J (1997) School-Based Drug Prevention Programs: A Longitudinal Study in Selected School Districts. Final Report. Executive Summary, pp 33. Research Triangle Inst., Research Triangle Park, N.C.

Silvia ES, Thorne J, Tashjian CA (1997) School-Based Drug Prevention Programs: A Longitudinal Study in Selected School Districts. Final Report, pp 171. Research Triangle Inst., Research Triangle Park, NC.

Simkin L, Radosh A, Nelsesteun K, Silverstein S (2003) Building Emergency Contraception Awareness among Adolescents. A Toolkit for Schools and Community-Based Organizations, pp 144. Academy for Educational Development, Inc., New York, NY.

Skager R (2007) Replacing ineffective early alcohol/drug education in the United States with ageappropriate adolescent programmes and assistance to problematic users. Drug & Alcohol Review 26(6):577-84.

Slater MD, And O (1996) Adolescent Counterarguing of TV Beer Advertisements: Evidence for Effectiveness of Alcohol Education and Critical Viewing Discussions. Journal of Drug Education 26(2):143-58.

Smith MU, Katner HP (1992) A Controlled Experimental Evaluation of Three AIDS Prevention Activities for Improving Knowledge, Changing Attitudes, and Decreasing Risk Behaviors of High School Seniors, pp 5.

Smith MU, Katner HP (1995) Quasi-experimental, evaluation of three AIDS prevention activities for maintaining knowledge, improving attitudes, and changing risk behaviors of high school seniors. AIDS Education and Prevention 7(5):391-402.

Smylie L, Maticka-Tyndale E, Boyd D, The Adolescent Sexual Health Planning Committee C (2008) Evaluation of a school-based sex education programme delivered to Grade Nine students in Canada. Sex Education 8:25-46.

Snegroff S (1995) Communicating about sexuality: a school/community program for parents and children. Journal of Health Education 26(1):49-52.

Somers CL, Fahlman MM (2001) Effectiveness of the "Baby Think It Over" teen pregnancy prevention program. Journal of School Health 71(5):188-196.

Song EY, Pruitt BE, McNamara J, Colewell B (2000) A Meta-Analysis Examining Effects of School Sexuality Education Programs on Adolescents' Sexual Knowledge, 1960-1997. Journal of School Health 70(10):413-16.

Soon T, Chan RK, Goh CL (1995) Project youth inform--a school-based sexually transmitted disease/acquired immune deficiency syndrome education programme. Annals of the Academy of Medicine, Singapore 24(4):541-6.

Sousa AP, Soares I, Vilar D (2007) Lessons learnt from a secondary school sex education program in Portugal. Sex Education 7:35-45.

Spadea T, Schifano P, Borgia P, Perucci CA (1999) The balance of positive and negative effects of specific messages in the evaluation of interventions for preventing HIV infection. European Journal of Epidemiology 15(2):109-117.

Stammers T (2007) Sexual Health in Adolescents. BMJ 334(7585):103-104.

Stephenson JM, Oakley A, Charleston S, Brodala A, Fenton K, Petruckevitch A, Johnson AM (1998) Behavioural intervention trials for HIV/STD prevention in schools: Are they feasible? Sexually Transmitted Infections 74(6):405-408.

Stephenson JM, Oakley A, Johnson AM, Forrest S, Strange V, Charleston S, Black S, Copas A, Petruckevitch A, Babiker A (2003) A school-based randomized controlled trial of peer-led sex education in England. Controlled Clinical Trials 24(5):643-657.

Stevens MM, Mott LA, Youells F (1996) Rural adolescent drinking behavior: three year follow-up in the New Hampshire substance abuse prevention study. Adolescence 31(121):159-66.

Strachan W, Gorey KM (1997) Infant simulator lifespace intervention: Pilot investigation of an adolescent pregnancy prevention program. Child & Adolescent Social Work Journal 14(3):171-180.

Strange V, Forrest S, Oakley A (2002) What influences peer-led sex education in the classroom? A view from the peer educators. Health Education Research 17(3):339-349.

Sulak PJ, Herbelin SJ, Fix DD, Kuehl TJ (2006) Impact of an adolescent sex education program that was implemented by an academic medical center. American Journal of Obstetrics & Gynecology 195(1):78-85.

Sundar M, Riley SC, Glasier A (2000) A feasibility study of adolescent sex education: Medical students as peer educators in Edinburgh schools. British Journal of Obstetrics and Gynaecology 107(3).

Sunwood J, Brenman A, Escobedo J, Philpott T, Allman K, Mueller J, Jaeger J, Brown LK, Cole FS (1995) School-based AIDS education for adolescents. Journal of Adolescent Health 16(4):309-15.

Sussman S (2004) Project ALERT reduces initiation of cigarette and marijuana use in 12-14 year olds. Evidence-Based Mental Health 7(2):53-54.

Swain R, McNamara M (1997) The effects of a participative programme on Irish pupils' attitudes to HIV/AIDS. Health Education Research 12(2):267-273.

Thakor HG, Kumar P (2000) Impact assessment of school-based sex education program amongst adolescents. Indian Journal of Pediatrics 67(8):551-8.

Thomas CL, Dimitrov DM (2007) Effects of a Teen Pregnancy Prevention Program on Teens' Attitudes toward Sexuality: A Latent Trait Modeling Approach. Developmental Psychology 43:13-185.

Thomas RE, Lorenzetti D, Spragins W (2008) Mentoring of children and adolescents for preventing drug and alcohol use. Cochrane Database of Systematic Reviews (4).

Van BJW, Render GF, Moon CE (1995) A Report of Outcomes of Project D.A.R.E. with Eighth Grade Students, pp 17.

Vanderschmidt HF, Lang JM, Knight-Williams V, Vanderschmidt GF (1993) Risks among inner-city young teens: The prevalence of sexual activity, violence, drugs, and smoking. Journal of Adolescent Health 14(4):282-288.

Vicary JR, Karshin CM (2002) College alcohol abuse: A review of the problems, issues, and prevention approaches. Journal of Primary Prevention 22:299-331.

Vincent ML, And O (1991) Projected Public Sector Savings in a Teen Pregnancy Prevention Program. Journal of Health Education 22(4):208-13.

Wagner EF, Tubman JG, Gil AG (2004) Implementing school-based substance abuse interventions: Methodological dilemmas and recommended solutions. Addiction 99(SUPPL. 2):106-119.

Walker-Shaw M (1993) Applying community organization to developing health promotion programs in the school community. Journal of School Health 63(2):109-11.

Walters ST, Bennett ME (2000) Addressing drinking among college students: A review of the empirical literature. Alcoholism Treatment Quarterly 18(1):61-77.

Walters ST, Bennett ME, Noto JV (2000) Drinking on campus. What Do we know about reducing alcohol use among college students? Journal of Substance Abuse Treatment 19(3):223-8.

Warwick I, Aggleton P (2004) Building on Experience: A Formative Evaluation of a Peer Education Sexual Health Project in South Africa. London Review of Education 2:17-153.

Watson DW, McCuller WJ, Bisesi L, Tanamly S, Sim T, Kavich JA (2004) Preparing adolescents for success in school: A school-based indicated prevention model for adolescents in the alternative high school setting. Journal of HIV/AIDS Prevention in Children and Youth 6(2):7-31.

Watson N, Robertson A (1996) A qualitative evaluation of HIV and AIDS education to senior school children in Lothian Region, Scotland. Health Promotion International 11(4):291-8.

Watt S, Horn E, Higgins C (1997) Teaching education about sexuality: a Scottish regional case study. British Journal of Special Education 24(3):115-23.

Weed S, Jensen L (1993) A second year evaluation of three abstinence sex education programs. Journal of Research & Development in Education 26(2):92-96.

Weis L (1998) Learning To Speak Out in an Abstinence Based Sex Education Group: Gender and Race Work in an Urban Magnet School, pp 38.

Weis L, Fine M (2003) Extraordinary Conversations in Public Schools.

Wight D, Abraham C (2000) From psycho-social theory to sustainable classroom practice: Developing a research-based teacher-delivered sex education programme. Health Education Research 15(1):25-38.

Williams CL, Perry CL (1998) Lessons from Project Northland: preventing alcohol problems during adolescence. Alcohol Health & Research World 22(2):107-16.

Williams KL, Doyle MS, Taylor BA, Ferguson G (1992) Addressing sexual orientation in a public high school. Journal of School Health 62(4):154-157.

Wilson KL, Goodson P, Pruitt BE, Buhi E, Davis-Gunnels E (2005) A review of 21 curricula for abstinence-only-until-marriage programs. Journal of School Health 75(3):90-8.

Wodarski JS, Bordnick PS (1994) Teaching adolescents about alcohol and driving: A 2-year follow-up study. Research on Social Work Practice 4(1):28-39.

Wolk LJ, Rosenbaum R (1995) The benefits of school-based condom availability: Cross-sectional analysis of a comprehensive high school-based program. Journal of Adolescent Health 17(3):184-188.

Wong WCW, Lee A, Tsang KKA, Lynn H (2006) The impact of AIDS/sex education by schools or family doctors on Hong Kong Chinese adolescents. Psychology, Health and Medicine 11(1):108-116.

Wood E, Senn CY, Desmarais S, Verberg N (2000) Offering Sexual Health Fairs To Supplement Existing Sex Education Programs: An Evaluation of Adolescent Students' Knowledge Needs. Alberta Journal of Educational Research 46(4):356-71.

Wood E, Shakeshaft A, Gilmour S, Sanson-Fisher R (2006) A systematic review of school-based studies involving alcohol and the community. Australian & New Zealand Journal of Public Health 30(6):541-9.

Woodside M, Bishop RM, Miller LT, Swisher JD (1997) Experimental evaluation of 'the images within': an alcohol education and prevention program. Journal of Drug Education 27(1):53-65.

Wren PA, Janz NK, Carovano K, Zimmerman MA, Washienko KM (1997) Preventing the spread of AIDS in youth: Principles of practice from 11 diverse projects. Journal of Adolescent Health 21(5):309-317.

Wynn SR, Schulenberg J, Kloska DD, Laetz VB (1997) The Mediating Influence of Refusal Skills in Preventing Adolescent Alcohol Misuse. Journal of School Health 67(9):390-95.

Yampolskaya S, Brown EC, Vargo AC (2004) Assessment of Teen Pregnancy Prevention Interventions Among Middle School Youth. Child & Adolescent Social Work Journal 21:69-83.

Yanqin L, Quanqing Z, Ping L, Zhifang W, Yi W, Xuecheng L (2001) Evaluation on the effect of a short-term intervention of sexual/reproductive health education among rural middle school students. Journal of Xi'an Medical University, English Edition 13(2):89-93.

Yawn BP, Yawn RA (1993) Adolescent pregnancies in rural America: A review of the literature and strategies for primary prevention. Family & Community Health 16(1):36-45.

Young C (1991) Alcohol, Drugs, Driving and You: A comprehensive program to prevent adolescent drinking, drug use, and driving. Journal of Alcohol and Drug Education 36(2):20-25.

Younoszai TM, Lohrmann DK, Seefeldt CA, Greene R (1999) Trends from 1987 to 1991 in alcohol, tobacco, and other drug (ATOD) use among adolescents exposed to a school district-wide prevention intervention. Journal of Drug Education 29(1):77-95.

Zabin LS (1992) School-linked reproductive health services: The Johns Hopkins program, in Preventing adolescent pregnancy: Model programs and evaluations., pp 156-184.

Zabin LS (1994) Addressing adolescent sexual behavior and childbearing: self-esteem or social change? Womens Health Issues 4(2):92-7.

Zagumny MJ, Thompson MK (1997) Does D.A.R.E. Work? An Evaluation in Rural Tennessee. Journal of Alcohol and Drug Education 42(2):32-41.

Zanetta R, Moacyr R, Lancarotte I (2008) Bringing up students in the Healthy Lifestyle Multiplier Students program, Sao Paulo, Brazil. Preventing Chronic Disease 5(3):A98.

• Population targeted by the intervention(s) did not meet review criteria (n=202)

Abatemarco DJ, West B, Zec V, Russo A, Sosiak P, Mardesic V (2004) Project Northland in Croatia: a community-based adolescent alcohol prevention intervention. Journal of Drug Education 34(2):167-179.

Abbaspour Z, Saidian M, Abedi P (2007) Peer education vs health provider education in knowledge and attitude about prevention and transmission of AIDS in high school students. Pakistan Journal of Medical Sciences 23(1):108-110.

Abolfotouh MA (1995) The impact of a lecture on AIDS on knowledge, attitudes and beliefs of male school-age adolescents in the Asir Region of southwestern Saudi Arabia. Journal of Community Health 20(3):271-81.

Ackermann ME (2008) The importance of gender and readiness to change in the prediction of drinking and negative consequences of first-year student drinkers, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 68.

Adams TB, Evans DR, Shreffler RM, Beam KJ (2006) Development and Evaluation of Theory-Based Alcohol Education Programs. Journal of Alcohol and Drug Education 50(3):21-30.

Agha S (2002) An evaluation of the effectiveness of a peer sexual health intervention among secondary-school students in Zambia. AIDS Education and Prevention 14(4):269-281.

Agha S, Van Rossem R (2004) Impact of a school-based peer sexual health intervention on normative beliefs, risk perceptions, and sexual behavior of Zambian adolescents. Journal of Adolescent Health 34(5):441-452.

Alexander SL (1999) The effects of a self-efficacy enhancement program on self-efficacy and condom use among college students, in Dissertation Abstracts International Section A: Humanities and Social Sciences, vol 59.

Anavberokhai MO (2006) Adolescents Sexual Behavioural Response to HIV/AIDS Education. Journal of Human Ecology 20(2):129-133.

Anderko L, Uscian M (2002) Academic-community partnerships as a strategy for positive change in the sexual behavior of rural college-aged students. Nursing Clinics of North America 37(2):341-9.

Ang A, Kinsler J, Morisky DE, Sneed CD (2004) Evaluation of a school-based intervention for HIV/AIDS prevention among Belizean adolescents. Health Education Research 19(6):730-738.

Antunes MC, Stall RD, Paiva V, Peres CA, Paul J, Hudes M, Hearst N (1997) Evaluating an AIDS sexual risk reduction program for young adults in public night schools in São Paulo, Brazil. AIDS (London, England) 11(Suppl 1):S121-S127.

Aplasca MR, Siegel D, Mandel JS, Santana-Arciaga RT, Paul J, Hudes ES, Monzon OT, Hearst N (1995) Results of a model AIDS prevention program for high school students in the Philippines. AIDS 9 Suppl 1:S7-13.

Asamoah A (1999) Effects of adult-led didactic presentation and peer-led discussion on AIDSknowledge of junior secondary school students. IFE Psychologia: An International Journal 7(2):213-228.

Asher KM (2009) Educating college students through judicial response: Examining the effectiveness of judicial sanctions for alcohol-related violations, in Dissertation Abstracts International Section A: Humanities and Social Sciences, vol 69.

Baer JS, Kivlahan DR, Fromme K, Marlatt GA (1994) Secondary prevention of alcohol abuse with college student populations: A skills-training approach, in Alcohol use and misuse by young adults.,(Howard GS, Nathan PE eds), pp 83-108.

Barnett LA, Far JM, Mauss AL, Miller JA (1996) Changing perceptions of peer norms as a drinking reduction program for college students. Journal of Alcohol and Drug Education 41(2):39-62.

Battistich V, Dean BJ, Gosselink CA, Zavela KJ (2004) Say yes first: follow-up of a five year rural drug prevention program. Journal of Drug Education 34:73-88.

Bauman DW, 3rd (1993) Peer education in the residential context. Journal of American College Health 41(6):271-2.

Bendtsen P, Johansson K, Akerlind I (2006) Feasibility of an email-based electronic screening and brief intervention (e-SBI) to college students in Sweden. Addictive Behaviors 31(5):777-87.

Benson TA, Deskins MM, Eakin D, Flood AM, McDevitt-Murphy ME, Murphy JG, Torrealday O, Vuchinich RE (2004) A comparison of personalized feedback for college student drinkers delivered with and without a motivational interview. Journal of Studies on Alcohol 65(2):200-203.

Bersamin M, Paschall MJ, Fearnow-Kenney M, Wyrick D (2007) Effectiveness of a Web-based alcohol-misuse and harm-prevention course among high- and low-risk students. Journal of American College Health 55(4):247-54.

Blair KE (2000) Evaluation of the effectiveness of a physician intervention program to reduce alcohol consumption among college students, in Dissertation Abstracts International Section A: Humanities and Social Sciences, vol 61.

Blume AW, Marlatt GA (2004) Motivational Enhancement as a Brief Intervention for College Student Drinkers, in Handbook of motivational counseling: Concepts, approaches, and assessment.,(Miles Cox W, Klinger E eds), pp 409-420. John Wiley & Sons td.

Brewer KK (1997) Behavioral preventions of sexually transmitted diseases and unintended pregnancies in college students, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 58.

Brondino MJ (1998) Message framing effects on risky decision-making in the context of AIDS prevention programming, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 58.

Bryan AD, Aiken LS, West SG (1996) Increasing condom use: evaluation of a theory-based intervention to prevent sexually transmitted diseases in young women. Health Psychology 15(5):371-82.

Cabezon C, Vigil P, Rojas I, Leiva ME, Riquelme R, Aranda W, Garcia C (2005) Adolescent pregnancy prevention: An abstinence-centered randomized controlled intervention in a Chilean public high school. Journal of Adolescent Health 36(1):64-69.

Caceres CF, Rosasco AM, Mandel JS, Hearst N (1994) Evaluating a school-based intervention for STD/AIDS prevention in Peru. Journal of Adolescent Health 15(7):582-591.

Cai Y, Hong H, Shi R, Ye X, Xu G, Li S, Shen L (2008) Long-term follow-up study on peer-led schoolbased HIV/AIDS prevention among youths in Shanghai. International Journal of STD and AIDS 19(12):848-850.

Carey KB, Scott-Sheldon LAJ, Carey MP, DeMartini KS (2007) Individual-level interventions to reduce college student drinking: a meta-analytic review. Addictive Behaviors 32(11):2469-94.

Caron SL, Carter DB, Davis CM, Macklin E (1997) Evaluating the effectiveness of workshop interventions on contraceptive use among first-year college students. Journal of Psychology & Human Sexuality 9(3-4):99-120.

Cartagena RG, Veugelers PJ, Kipp W, Magigav K, Laing LM (2006) Effectiveness of an HIV Prevention Program for Secondary School Students in Mongolia. Journal of Adolescent Health 39(6):925.e9-925.e16.

Caudill BD, Luckey B, Crosse SB, Blane HT, Ginexi EM, Campbell B (2007) Alcohol risk-reduction skills training in a national fraternity: a randomized intervention trial with longitudinal intent-to-treat analysis, in Journal of Studies on Alcohol and Drugs, pp 399-409.

Cheng Y, Lou CH, Mueller LM, Zhao SL, Yang JH, Tu XW, Gao ES (2008) Effectiveness of a School-Based AIDS Education Program among Rural Students in HIV High Epidemic Area of China. Journal of Adolescent Health 42(2):184-191. Chernoff RA (2001) An evaluation of a brief HIV/AIDS prevention intervention using normative feedback to promote risk reduction among sexually active college students, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 62.

Chernoff RA, Davison GC (2005) An evaluation of a brief HIV/AIDS prevention intervention for college students using normative feedback and goal setting. AIDS Education & Prevention 17(2):91-104.

Chiauzzi E, Green TC, Lord S, Thum C, Goldstein M (2005) My student body: a high-risk drinking prevention web site for college students. Journal of American College Health 53(6):263-74.

Cin SD, MacDonald TK, Fong GT, Zanna MP, Elton-Marshall TE (2006) Remembering the Message: The Use of a Reminder Cue to Increase Condom Use following a Safer Sex Intervention. Health Psychology 25(3):438-443.

Clapp JD, Stanger L (2003) Changing the College AOD Environment for Primary Prevention. Journal of Primary Prevention 23(4):515-23.

Cohen D (1996) The health belief model: Utility in predicting AIDS risk behavior in college students, in Dissertation Abstracts International Section A: Humanities and Social Sciences, vol 56.

Collins PJ (2006) Message framing and sexual abstinence messages, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 66.

Croom K, Lewis D, Marchell T, Lesser ML, Reyna VF, Kubicki-Bedford L, Feffer M, Staiano-Coico L (2009) Impact of an online alcohol education course on behavior and harm for incoming first-year college students: short-term evaluation of a randomized trial. Journal of American College Health 57(4):445-54.

Curtin L, Stephens RS, Bonenberger JL (2001) Goal setting and feedback in the reduction of heavy drinking in female college students. Journal of College Student Psychotherapy 15(3):17-37.

Czuchry M, Sia TL, Dansereau DF (1999) Preventing alcohol abuse: an examination of the "Downward Spiral" game and educational videos. Journal of Drug Education 29(4):323-35.

Daboer JC, Ogbonna C, Jamda MA (2008) Impact of health education on sexual risk behaviour of secondary school students in Jos, Nigeria. Nigerian Journal of Medicine: Journal of the National Association of Resident Doctors of Nigeria 17(3):324-9.

Dalrymple L, du Toit MK (1993) The evaluation of a drama approach to AIDS education. Educational Psychology 13(2):147-154.

DeJong W, Schneider SK, Towvim LG, Murphy MJ, Doerr EE, Simonsen NR, Mason KE, Scribner RA (2006) A multisite randomized trial of social norms marketing campaigns to reduce college student drinking. Journal of Studies on Alcohol 67(6):868-879.

DiPasquale JA (1990) HIV infection. An educational program on prevention for college freshmen. Cancer Nursing 13(3):152-7. Donohue B, Allen DN, Maurer A, Ozols J, DeStefano G (2004) A Controlled Evaluation of Two Prevention Programs in Reducing Alcohol Use Among College Students at Low and High Risk for Alcohol Related Problems. Journal of Alcohol and Drug Education 48:13-33.

Dotterer S, Olds RS, Raub CG, Sharp KE, Thombs DL (2004) A close look at why one social norms campaign did not reduce student drinking. Journal of American College Health 53(2):61-68.

Eddy MS (1992) College Alcohol and Drug Abuse Prevention Programs: An Update. ERIC Digest, pp 4. ERIC Clearinghouse on Higher Education, Washington, DC.

Eggleston AM (2008) Components analysis of a brief intervention for college drinkers, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 68.

Ehrhardt BL, Krumboltz JD, Koopman C (2007) Training Peer Sexual Health Educators: Changes in Knowledge, Counseling Self-Efficacy, and Sexual Risk Behavior. American Journal of Sexuality Education 2:39-55.

Elliott JC, Carey KB, Bolles JR (2008) Computer-based interventions for college drinking: a qualitative review. Addictive Behaviors 33(8):994-1005.

Esere MO (2008) Effect of Sex Education Programme on at-risk sexual behaviour of school-going adolescents in Ilorin, Nigeria. African Health Sciences 8(2):120-5.

Evans AE, Edmundson-Drane EW, Harris KK (2000) Computer-assisted instruction: an effective instructional method for HIV prevention education?, in The Journal of adolescent health : official publication of the Society for Adolescent Medicine, pp 244-51.

Fisher JD, Fisher WA, Misovich SJ, Kimble DL, Malloy TE (1996) Changing AIDS risk behavior: effects of an intervention emphasizing AIDS risk reduction information, motivation, and behavioral skills in a college student population. Health Psychology 15(2):114-23

Fitzgerald AM, Stanton BF, Terreri N, Shipena H, Li X, Kahihuata J, Ricardo IB, Galbraith JS, De Jaeger AM (1999) Use of western-based HIV risk-reduction interventions targeting adolescents in an african setting. Journal of Adolescent Health 25(1):52-61.

Fongkaew W, Fongkaew K, Suchaxaya P (2007) Early adolescent peer leader development in HIV prevention using youth-adult partnership with schools approach. JANAC: Journal of the Association of Nurses in AIDS Care 18:60-71.

Fromme K, Corbin W (2004) Prevention of heavy drinking and associated negative consequences among mandated and voluntary college students. Journal of Consulting and Clinical Psychology 72(6):1038-1049.

Fromme K, Orrick D (2004) The Lifestyle Management Class: A harm reduction approach to college drinking. Addiction Research & Theory 12:335-351.

Gallant M, Maticka-Tyndale E (2004) School-Based HIV Prevention Programmes for African Youth. Social Science & Medicine 58(7):1337-1351.

George Washington Univ., Washington, DC. School of Education and Human Development.

Geshi M, Hirokawa K, Taniguchi T, Fujii Y, Kawakami N (2007) Effects of alcohol-related health education on alcohol and drinking behavior awareness among Japanese junior college students: a randomized controlled trial. Acta Medica Okayama 61(6):345-54.

Givaudan M, Leenen I, Van De Vijver FJR, Poortinga YH, Pick S (2008) Longitudinal study of a School based HIV/AIDS early prevention program for Mexican Adolescents. Psychology, Health and Medicine 13(1):98-110.

Givaudan M, Van de Vijver FJR, Poortinga YH, Leenen I, Pick S (2007) Effects of a school-based life skills and HIV-prevention program for adolescents in Mexican high schools. Journal of Applied Social Psychology 37:1141-1162.

Glautier S, Rigney U, Willner P (2001) Motivation for alcohol assessed by multiple variable interval schedule behavior: Effects of reward size and alcohol cues. Behavioural Pharmacology 12(2):81-89.

Goertzel TG, Bluebond-Langner M (1991) What is the impact of a campus AIDS education course? Journal of American College Health 40(2):87-92.

Gonzalez G, Clement V (1994) Research and Intervention. Preventing Substance Abuse in Higher Education, pp 129. Network of Colleges and Universities Committed to the Elimination of Drug and Alcohol Abuse, Washington, DC.

Gregory BM (2001) College alcohol and life skills study with student-athletes, in Dissertation Abstracts International Section A: Humanities and Social Sciences, vol 62.

Griffin SY (2005) Factors that affect college students' readiness to change their drinking behavior, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 65.

Guarna J (2001) Altering binge drinking behavior: An intervention based on the attention-allocation model, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 61.

Guth LJ, Hewitt GC, Smith S, Fisher MS (2000) Student Attitudes toward AIDS and Homosexuality: The Effects of a Speaker with HIV. Journal of College Student Development 41(5):503-12.

Guth LJ, Lopez DF, Fisher MS (2002) In-Person versus Internet Training: A Comparison of Student Attitudes toward Homosexuality. NASPA Journal 40(1).

Gyarmathy VA, McNutt LA, Molnar A, Morse DL, DeHovitz J, Ujhelyi E, Szamado S (2002) Evaluation of a comprehensive AIDS education curriculum in Hungary - the role of good educators. Journal of Adolescence 25(5):495-508.

Hahn EJ, Hall LA, Rayens MK, Myers AV, Bonnel G (2007) School- and home-based drug prevention: Environmental, parent, and child risk reduction. Drugs: Education, Prevention & Policy 14:319-331.

Haines M, Spear SF (1996) Changing the perception of the norm: a strategy to decrease binge drinking among college students. Journal of American College Health 45(3):134-40.

Harris JL, Ludwig M (1996) The Trading Cards Program: An Evaluation of Use of High School Role Models for Drug Abuse Prevention. Journal of Health Education 27(3):183-86.

Helion AM, Reddy DM, Kies AL, Morris DR, Wilson CM (2008) Influence of communicator's race on efficacy of an HIV/STD prevention intervention among African American and Caucasian college students. Public Health Nursing 25(5):440-50.

Hernandez DV, Skewes MC, Resor MR, Villanueva MR, Hanson BS, Blume AW (2006) A pilot test of an alcohol skills training programme for Mexican-American college students. International Journal of Drug Policy 17:320-328.

Horn PA, Brigham TA (1996) A self-management approach to reducing AIDS risk in sexually active heterosexual college students. Behavior and Social Issues 6(1):3-21.

Huang H, Ye X, Cai Y, Shen L, Xu G, Shi R, Jin X (2008) Study on peer-led school-based HIV/AIDS prevention among youths in a medium-sized city in China. International Journal of STD and AIDS 19(5):342-346.

Hunter Fager J, Mazurek Melnyk B (2004) The effectiveness of intervention studies to decrease alcohol use in college undergraduate students: an integrative analysis. Worldviews on Evidence-Based Nursing 1(2):102-19.

James S, Reddy P, Ruiter RAC, McCauley A, Van Den Borne B (2006) The impact of an HIV and AIDS life skills program on secondary school students in Kwazulu-Natal, South Africa. AIDS Education and Prevention 18(4):281-294.

James S, Reddy PS, Ruiter RAC, Taylor M, Jinabhai CC, Van Empelen P, Van Den Borne B (2005) The effects of a systematically developed photo-novella on knowledge, attitudes, communication and behavioural intentions with respect to sexually transmitted infections among secondary school learners in South Africa. Health Promotion International 20(2):157-165.

Johnsson KO, Berglund M (2006) Comparison between a cognitive behavioural alcohol programme and post-mailed minimal intervention in high-risk drinking university freshmen: results from a randomized controlled trial, in Alcohol and alcoholism (Oxford, Oxfordshire), vol 41, pp 174-80.

Jones SG, Patsdaughter CA, Jorda ML, Hamilton M, Malow R (2008) SENORITAS: an HIV/sexually transmitted infection prevention project for Latina college students at a Hispanic-serving university. Journal of the Association of Nurses in AIDS Care 19(4):311-9.

Kamal AS (1999) Adult awareness strategies in AIDS health education, in Dissertation Abstracts International Section A: Humanities and Social Sciences, vol 60.

Karnell AP, Cupp PK, Zimmerman RS, Feist PS, Bennie T (2006) Efficacy of an American Alcohol and HIV Prevention Curriculum Adapted for Use in South Africa: Results of a Pilot Study in Five Township Schools. AIDS Education and Prevention 18:16-310.

Kashibuchi M, Sakamoto A (2001) The educational effectiveness of a simulation/game in sex education. Simulation & Gaming 32(3):331-343.

Kiene SM, Barta WD (2006) A brief individualized computer-delivered sexual risk reduction intervention increases HIV/AIDS preventive behavior. Journal of Adolescent Health 39(3):404-10.

Kinsler J, Sneed CD, Morisky DE, Ang A (2004) Evaluation of a school-based intervention for HIV/AIDS prevention among Belizean adolescents. Health Education Research 19(6):730-8.

Kinsman J, Nakiyingi J, Kamali A, Carpenter L, Quigley M, Pool R, Whitworth J (2001) Evaluation of a comprehensive school-based aids education programme in rural Masaka, Uganda. Health Education Research 16(1):85-100.

Kirkpatrick WR (1993) The influence of an AIDS awareness program on knowledge, attitudes, opinions, behaviors, and behavioral intentions of college students, in Dissertation Abstracts International, vol 53.

Knaus CS, Pinkleton BE, Austin EW (2000) The ability of the AIDS Quilt to motivate information seeking, personal discussion, and preventative behavior as a health communication intervention. Health Communication 12(3):301-316.

Knight CL (1995) HIV prevention and heterosexual college students: The impact of video instruction on the "safer" sexual behaviors of sexually active men, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 56.

Kritch KM, Bostow DE, Dedrick RF (1995) Level of interactivity of videodisc instruction on college students' recall of AIDS information. Journal of Applied Behavior Analysis 28(1):85-6.

Kuhn L, Steinberg M, Mathews C (1994) Participation of the school community in AIDS education: an evaluation of a high school programme in South Africa. AIDS Care 6(2):161-71.

Kyrychenko P, Kohler C, Sathiakumar N (2006) Evaluation of a School-Based HIV/AIDS Educational Intervention in Ukraine. Journal of Adolescent Health 39(6):900-907.

LaBrie JW, Huchting K, Tawalbeh S, Larimer M, Neighbors C, Pedersen ER, Thompson AD, Shelesky K (2008) A Randomized Motivational Enhancement Prevention Group Reduces Drinking and Alcohol Consequences in First-Year College Women. Psychology of Addictive Behaviors 22(1):149-155.

LaBrie JW, Hummer JF, Neighbors C, Pedersen ER (2008) Live interactive group-specific normative feedback reduces misperceptions and drinking in college students: a randomized cluster trial. Psychology of Addictive Behaviors 22(1):141-8.

Lambert EC (2001) College students' knowledge of human papillomavirus and effectiveness of a brief educational intervention.[see comment]. Journal of the American Board of Family Practice 14(3):178-83.

Larimer ME, Cronce JM (2002) Identification, prevention and treatment: a review of individual-focused strategies to reduce problematic alcohol consumption by college students. Journal of Studies on Alcohol - Supplement (14):148-63.

Larimer ME, Cronce JM (2007) Identification, prevention, and treatment revisited: Individual-focused college drinking prevention strategies 1999-2006. Addictive Behaviors 32(11):2439-2468.

Larimer ME, Kilmer JR, Lee CM (2005) College Student Drug Prevention: A Review of Individually-Oriented Prevention Strategies. Journal of Drug Issues 35:431-456. Larsen JD, Kozar B (2005) Evaluation of a computer administered alcohol education program for college students. Journal of Alcohol and Drug Education 49:69-83.

Lau HC (2007) Development and evaluation of a single-session expectancy challenge intervention to reduce alcohol use among heavy drinking college students, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 67.

Lederman LC, Stewart LP, Russ TL (2007) Addressing college drinking through curriculum infusion: A study of the use of experience-based learning in the communication classroom. Communication Education 56:476-494.

Lee MJ, Bichard SL (2006) Effective message design targeting college students for the prevention of binge-drinking: basing design on rebellious risk-taking tendency. Health Communication 20(3):299-308.

Li X, Stanton B, Wang B, Mao R, Zhang H, Qu M, Sun Z, Wang J (2008) Cultural adaptation of the Focus on Kids program for college students in China. AIDS Education & Prevention 20(1):1-14.

Licciardone JC (2003) Outcomes of a Federally Funded Program for Alcohol and Other Drug Prevention in Higher Education. American Journal of Drug and Alcohol Abuse 29(4):803-827.

Lindemann DF (2004) Comparing two condom use components of a HIV/AIDS intervention for college students, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 64.

Lindemann DF, Brigham TA, Harbke CR, Alexander T (2005) Toward errorless condom use: a comparison of two courses to improve condom use skills. AIDS & Behavior 9(4):451-7.

Lou C-h, Zhao Q, Gao E-S, Shah IH (2006) Can the Internet be used effectively to provide sex education to young people in China? Journal of Adolescent Health 39(5):720-8.

Magnani R, MacIntyre K, Karim AM, Brown L, Hutchinson P (2005) The impact of life skills education on adolescent sexual risk behaviors in KwaZulu-Natal, South Africa. Journal of Adolescent Health 36(4):289-305.

Martinez-Donate AP, Hovell MF, Zellner J, Sipan CL, Blumberg EJ, Carrizosa C (2004) Evaluation of two school-based HIV prevention interventions in the border city of Tijuana, Mexico. Journal of Sex Research 41(3):267-278.

Martiniuk ALC, O'Connor KS, King WD (2003) A cluster randomized trial of a sex education programme in Belize, Central America.[see comment]. International Journal of Epidemiology 32(1):131-6.

Mayhew MJ, Caldwell RJ, Hourigan AJ (2008) The Influence of Curricular-Based Interventions within First-Year "Success" Courses on Student Alcohol Expectancies and Engagement in High-Risk Drinking Behaviors. NASPA Journal 45(1):49-72.

Mba CI, Obi SN, Ozumba BC (2007) The impact of health education on reproductive health knowledge among adolescents in a rural Nigerian community. Journal of Obstetrics & Gynaecology 27(5):513-7.

Meyer WA, Steyn M (1992) AIDS Preventive Education and Life Skills Training Programme for Secondary Schools: Development and Evaluation, pp 156. Human Sciences Research Council, Pretoria (South Africa).

Michael KD, Curtin L, Kirkley DE, Jones DL, Harris R, Jr. (2006) Group-based motivational interviewing for alcohol use among college students: An exploratory study. Professional Psychology: Research and Practice 37:629-634.

Montgomery AJ (1996) AIDS education: knowledge, sexual attitudes and sexual behavioral responses of selected college students. ABNF Journal 7(2):57-60.

Moore MJ, Soderquist J, Werch C (2005) Feasibility and efficacy of a binge drinking prevention intervention for college students delivered via the Internet versus postal mail. Journal of American College Health 54(1):38-44.

Murphy JG (2003) A comparison of personal feedback for college student drinkers delivered with and without a counseling session, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 64.

Murphy JG, Duchnick JJ, Vuchinich RE, Davison JW, Karg RS, Olson AM, Smith AF, Coffey TT (2001) Relative efficacy of a brief motivational intervention for college student drinkers, in Psychology of addictive behaviors : journal of the Society of Psychologists in Addictive Behaviors, pp 373-9.

Nelson TF, Weitzman ER, Wechsler H (2005) The effect of a campus-community environmental alcohol prevention initiative on student drinking and driving: Results from the "A Matter of Degree" program evaluation. Traffic Injury Prevention 6(4):323-330.

Nokes KM (1996) Intervention to promote safer sexual behaviors and educate peers about HIV/AIDS. Journal of Nursing Education 35(5):227-9.

Oswalt SB, Shutt MD, English E, Little SD (2007) Did it work? Examining the impact of an alcohol intervention on sanctioned college students. Journal of College Student Development 48:543-557.

Ozcebe H, Akin L, Aslan D (2004) A peer education example on HIV/AIDS at a high school in Ankara. Turkish Journal of Pediatrics 46(1):54-59.

Parwej S, Kumar R, Walia I, Aggarwal AK (2005) Reproductive health education intervention trial. Indian Journal of Pediatrics 72(4):287-91.

Paschall MJ, Bersamin M, Fearnow-Kenney M, Wyrick D, Currey D (2006) Short-Term Evaluation of a Web-based College Alcohol Misuse and Harm Prevention Course (College Alc). Journal of Alcohol and Drug Education 50:49-65.

Peeler CM (2001) An analysis of the effects of a course designed to reduce the frequency of high-risk sexual behavior and heavy drinking (immune deficiency), in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 61.

Peeler CM, Far J, Miller J, Brigham TA (2000) An analysis of the effects of a program to reduce heavy drinking among college students. Journal of Alcohol and Drug Education 45(2):39-54.

Peleg A, Neumann L, Friger M, Peleg R, Sperber AD (2001) Outcomes of a brief alcohol abuse prevention program for Israeli high school students. Journal of Adolescent Health 28(4):263-9.

Perkins HW, Craig DW (2002) A Multifaceted Social Norms Approach To Reduce High-Risk Drinking: Lessons from Hobart and Williams Smith Colleges, pp 44. Higher Education Center for Alcohol and Other Drug Prevention, Newton, MA.

Perold JJ, Schoeman JB, Visser MJ (2004) Evaluation of HIV/AIDS prevention in South African schools. Journal of Health Psychology 9(2):263-280.

Peterson KE (1992) AIDS education and prevention at colleges and universities. AIDS prevention and treatment: Hope, humor, and healing.:121-141.

Petrie J, Bunn F, Byrne G (2007) Parenting programmes for preventing tobacco, alcohol or drugs misuse in children <18: a systematic review. Health Education Research 22(2):177-191.

Ploem C, Byers ES (1997) The effects of two AIDS risk-reduction interventions on heterosexual college women's AIDS-related knowledge, attitudes and condom use. Journal of Psychology & Human Sexuality 9(1):1-24.

Polonec LD, Major AM, Atwood LE (2006) Evaluating the believability and effectiveness of the social norms message "most students drink 0 to 4 drinks when they party". Health Communication 20(1):23-34.

Rainer JP (1994) The effect of intimate exposure to alcohol abuse on the acquisition of knowledge about drinking. Journal of Addictive Diseases 13(2):21-33.

Rajbhandari MMS (2008) Impact of Sexuality Education in Preventing STD-HIV/AIDS among Teenagers of School Going Students, pp 56.

Rasler ML (1993) Development of a Sex Education Syllabus for Health Science at American River College. Emergence of Higher Education in America, pp 103.

Rehnberg T, Barabasz M (1994) The Effect of a Health Belief Intervention on Safer Sex Practices, pp 10.

Reis J, Riley W, Baer J (2000) Interactive multimedia preventive alcohol education: An evaluation of effectiveness with college students. Journal of Educational Computing Research 23(1):41-65.

Reis J, Trockel M, Wall A (2003) Promoting Student Support for Alcohol Misuse Prevention on Campus: The Role of Secondhand Consequence Expectancies. NASPA Journal 40(2).

Reis J, Tymchyshyn P (1992) A longitudinal evaluation of computer-assisted instruction on contraception for college students. Adolescence 27(108):803-11.

Roberts LJ, Neal DJ, Kivlahan DR, Baer JS, Marlatt GA (2000) Individual drinking changes following a brief intervention among college students: clinical significance in an indicated preventive context, in Journal of Consulting and Clinical Psychology, pp 500-5.

Sain R (2006) Analysis of a program to increase condom use by sexually active college students, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 66.

Saleh MA, al-Ghamdi YS, al-Yahia OA, Shaqran TM, Mosa AR (1999) Impact of health education program on knowledge about AIDS and HIV transmission in students of secondary schools in Buraidah city, Saudi Arabia: an exploratory study. Eastern Mediterranean Health Journal 5(5):1068-75.

Sanderson CA (1999) Role of relationship context in influencing college students' responsiveness to HIV prevention videos. Health Psychology 18(3):295-300.

Sanderson CA, Jemmott JB, III (1996) Moderation and mediation of HIV-prevention interventions: Relationship status, intentions, and condom use among college students. Journal of Applied Social Psychology 26(23):2076-2099.

Sankaranarayan S, Naik E, Reddy PS, Gurunani G, Ganesh K, Gandewar K, Singh KP, Vermund SH (1996) Impact of school-based HIV and AIDS education for adolescents in Bombay, India. Southeast Asian Journal of Tropical Medicine & Public Health 27(4):692-5.

Sawyer RG, Pinciaro P, Bedwell D (1997) How peer education changed peer sexuality educators' self-esteem, personal development, and sexual behavior. Journal of American College Health 45(5):211-7.

Schenker II, Greenblatt CL (1993) Israeli youth and AIDS: Knowledge and attitude changes among high school students following an AIDS education program. Israel Journal of Medical Sciences 29(10 SUPPL.):41-47.

Schroeder CM, Prentice DA (1998) Exposing pluralistic ignorance to reduce alcohol use among college students. Journal of Applied Social Psychology 28(23):2150-2180.

Scott KJ, Miller G, Gridley B (1997) Voluntary alcohol education: Whom do we reach? Journal of College Student Development 38(3):305-307.

Shapiro Cohen EL (2001) High-risk sexual behavior in the context of alcohol use: An intervention for college students, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 61.

Sharmer LA (2000) Evaluation of alcohol education on attitude, knowledge and self-reported behavior of college students, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 61.

Shen LX, Hong H, Cai Y, Jin XM, Shi R (2008) Effectiveness of peer education in HIV/STD prevention at different types of senior high schools in Shanghai, People's Republic of China. International Journal of STD and AIDS 19(11):761-767.

Shulman HA, Sweeney B, Gerler ER (1995) A computer-assisted approach to preventing alcohol abuse: Implications for the middle school. Elementary School Guidance & Counseling 30(1):63-77.

Smith BH (2004) A randomized study of a peer-led, small group social norming intervention designed to reduce drinking among college students. Journal of Alcohol and Drug Education 47(3):67-75.

Smith EA, Dickson LL (1993) The impact of a condom desensitization program on female college students. Health Values: The Journal of Health Behavior, Education & Promotion 17(3):21-31.

Spradlin K (2007) Integration of the Transtheoretical Model of Change and the Elaboration Likelihood Model in approaching HIV/AIDS prevention, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 68.

Stamper GA (2007) The evaluation of program fidelity in two college alcohol interventions, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 67.

Stamper GA, Smith BH, Gant R, Bogle KE (2004) Replicated findings of an evaluation of a brief intervention designed to prevent high-risk drinking among first-year college students: Implications for social norming theory, vol 48, pp 53-72.

Stanton BF, Li X, Kahihuata J, Fitzgerald AM, Neumbo S, Kanduuombe G, Ricardo IB, Galbraith JS, Terreri N, Guevara I, Shipena H, Strijdom J, Clemens R, Zimba RF (1998) Increased protected sex and abstinence among Namibian youth following a HIV risk-reduction intervention: A randomized, longitudinal study. AIDS 12(18):2473-2480.

Steffian G (1999) Correction of normative misperceptions: an alcohol abuse prevention program. Journal of Drug Education 29(2):115-38.

Stiff J, And O (1990) Learning about AIDS and HIV Transmission in College-Age Students. Communication Research 17(6):743-58.

Stock ML (2008) It only takes once: Influence of sexual risk status, social comparison, and a public service announcement on absent-exempt cognitions, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 68.

Straub DM, Pomputius P, Boyer CB, Someillan NS, Perrin K (2007) HIV prevention education and testing among youth: is there a correlation? Journal of Adolescent Health 41(1):105-108.

Taylor EL (1992) Impact of an AIDS Education Course on University Student Attitudes. Health Education 23(7):418-22.

Thato R, Jenkins RA, Dusitsin N (2008) Effects of the culturally-sensitive comprehensive sex education programme among Thai secondary school students. Journal of Advanced Nursing 62(4):457-69.

Thomas N (1998) The effectiveness of a peer-led, self-management program on reducing AIDS risk in college students, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 59.

Thompson SC, Kyle D, Swan J, Thomas C, Vrungos S (2002) Increasing condom use by undermining perceived invulnerability to HIV. AIDS Education & Prevention 14(6):505-14.

Toomey TL, Lenk KM, Wagenaar AC (2007) Environmental policies to reduce college drinking: An update of research findings. Journal of Studies on Alcohol and Drugs 68(2):208-219.

Turner JC, Garrison CZ, Korpita E, Waller J, Addy C, Hill WR, Mohn LA (1994) Promoting responsible sexual behavior through a college freshman seminar. AIDS Education & Prevention 6(3):266-77.

Turner JC, Korpita E, Mohn LA, Hill WB (1993) Reduction in sexual risk behaviors among college students following a comprehensive health education intervention. Journal of American College Health 41(5):187-93.

Vachhrajani B (1996) Impact of self instructional material on sex education and attitude towards sex. Indian Journal of Psychometry & Education 27(1):27-31.

Valus KL (1999) Do AIDS peer educators have increased safer sex behaviors? (immune deficiency), in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 60.

Visser MJ (2005) Life skills training as HIV/AIDS preventive strategy in secondary schools: evaluation of a large-scale implementation process. SAHARA J: Journal of Social Aspects of HIV/AIDS Research Alliance 2(1):203-16.

Visser MJ (2007) HIV/AIDS prevention through peer education and support in secondary schools in South Africa. SAHARA J: Journal of Social Aspects of HIV/AIDS Research Alliance 4(3):678-94.

Visser MJ, Schoeman JB, Perold JJ (2004) Evaluation of HIV/AIDS Prevention in South African Schools. Journal of Health Psychology 9(2):263-280.

Vrungos SM (2003) Increasing HIV intervention effectiveness using an enhanced motivational approach: An evaluation of a multi-component motivation-skills program on HIV risk reduction behavior and condom use in heterosexual college students, in Dissertation Abstracts International: Section B: The Sciences and Engineering, vol 63.

Walker D, Gutierrez JP, Torres P, Bertozzi SM (2006) HIV prevention in Mexican schools: Prospective randomised evaluation of intervention. British Medical Journal 332(7551):1189-1192.

Walters ST (2000) In praise of feedback: An effective intervention for college students who are heavy drinkers. Journal of American College Health 48(5):235-238.

Walters ST, Vader AM, Harris TR (2007) A controlled trial of web-based feedback for heavy drinking college students, in Prevention science : the official journal of the Society for Prevention Research, pp 83-8.

Walters ST, Vader AM, Harris TR, Field CA, Jouriles EN (2009) Dismantling motivational interviewing and feedback for college drinkers: A randomized clinical trial. Journal of Consulting and Clinical Psychology 77:64-73.

Wechsler H, Nelson TE, Lee JE, Seibring M, Lewis C, Keeling RP (2003) Perception and reality: a national evaluation of social norms marketing interventions to reduce college students' heavy alcohol use. Journal of Studies on Alcohol 64(4):484-94.

Weinhardt LS, Carey KB, Carey MP (2000) HIV risk sensitization following a detailed sexual behavior interview: a preliminary investigation. Journal of Behavioral Medicine 23(4):393-8.

Weitzman ER, Nelson TF, Lee H, Wechsler H (2004) Reducing drinking and related harms in college: Evaluation of the "a matter of degree" program. American Journal of Preventive Medicine 27(3):187-196.

Wellings K, Wadsworth J, Johnson AM, Field J, Whitaker L, Field B (1995) Provision of sex education and early sexual experience: The relation examined. British Medical Journal 311(7002):417-420.

Wiefferink CH, Poelman J, Linthorst M, Vanwesenbeeck I, Van Wijngaarden JCM, Paulussen TGW (2005) Outcomes of a systematically designed strategy for the implementation of sex education in Dutch secondary schools. Health Education Research 20(3):323-333.

Wilson CE (2004) Raising college students' alcohol awareness: a successful program at the University of Virginia. Journal of Emergency Nursing 30(1):64-7.

Wright SS, Kyes KB (1996) The effects of safer-sex stories on college students' attitudes toward condoms. Journal of Psychology & Human Sexuality 8(4):1-18.

Wyrick DL, Fearnow-Kenney MD, Wyrick CH, Bibeau DL (2005) College alcohol education and prevention: A case for distance education. Journal of Alcohol and Drug Education 49(4):5-14.

Yosefy C, Reuven Viskoper J, Friedman M, Biton Y, Glick S (1998) Health education in Israeli public schools by undergraduate medical students. Education for Health 11(1):73-81.

Yzer MC, Fisher JD, Bakker AB, Siero FW, Misovich SJ (1998) The effects of information about AIDS risk and self-efficacy on women's intentions to engage in AIDS preventive behavior. Journal of Applied Social Psychology 28(20):1837-1852.

Zellner J, Martinez-Donate A, Hovell M, Sipan C, Blumberg E, Carrizosa C, Rovniak L (2006) Feasibility and use of school-based condom availability programs in Tijuana, Mexico. AIDS & Behavior 10(6):649-658.

Ziemelis A, Bucknam RB, Elfessi AM (2002) Prevention efforts underlying decreases in binge drinking at institutions of higher education. Journal of American College Health 50(5):238-253.

Zisserson RN, Palfai T, Saitz R (2007) 'No-contact' interventions for unhealthy college drinking:efficacy of alternatives to person-delivered intervention approaches. Substance Abuse 28(4):119-31.

Intervention(s) examined was not based in a school setting (n=38)

Allen JP, Philliber S, Herrling S, Kuperminc GP (1997) Preventing teen pregnancy and academic failure: Experimental evaluation of a developmentally based approach. Child Development 64:729-742.

Babor TE, Higgins-Biddle J, Dauser D, Higgins P, Burleson JA (2005) Alcohol screening and brief intervention in primary care settings: implementation models and predictors, in Journal of Studies on Alcohol, vol 66, pp 361-8.
Baldwin JA, Daley E, Brown EJ, August EM, Webb C, Stern R, Malow R, Devieux JG (2008) Knowledge and Perception of STI/HIV Risk Among Rural African-American Youth: Lessons Learned in a Faith-Based Pilot. Journal of HIV/AIDS Prevention in Children and Youth 9(1):97-114.

Benshoff JM, Alexander SJ (1993) The Family Communication Project: Fostering Parent-Child Communication about Sexuality. Elementary School Guidance and Counseling 27(4):288-300.

Brody GH, Kogan SM, Chen Yf, Murry VM (2008) Long-Term Effects of the Strong African American Families Program on Youths' Conduct Problems. Journal of Adolescent Health 43(5):474-481.

Cohen DA, Rice JC (1995) A Parent-targeted Intervention for Adolescent Substance Use Prevention: Lessons Learned. Evaluation Review 19(2):159-80.

D'Amico EJ, Orlando Edelen M (2007) Pilot Test of Project CHOICE: A Voluntary Afterschool Intervention for Middle School Youth. Psychology of Addictive Behaviors 21(4):592-598.

Dilorio C, McCarty F, Resnicow K, Lehr S, Denzmore P (2007) REAL men: a group-randomized trial of an HIV prevention intervention for adolescent boys, in American Journal of Public Health, vol 97, pp 1084-9.

Downs JS, Murray PJ, Bruine de Bruin W, Penrose J, Palmgren C, Fischhoff B (2004) Interactive video behavioral intervention to reduce adolescent females' STD risk: a randomized controlled trial, in Social science & medicine (1982), pp 1561-72.

Fager JH, Melnyk BM (2004) The effectiveness of intervention studies to decrease alcohol use in college undergraduate students: an integrative analysis (Structured abstract), in World Views on Evidence-Based Nursing, pp 102-119.

French RS, Mercer CH, Kane R, Kingori P, Stephenson JM, Wilkinson P, Grundy C, Lachowycz K, Jacklin P, Stevens M, Brooker S, Wellings K (2007) What Impact Has England's Teenage Pregnancy Strategy Had on Young People's Knowledge of and Access to Contraceptive Services? Journal of Adolescent Health 41(6):594-601.

Gao E-S, Lou C-H, Shen Y, Wang B (2004) Effects of a community-based sex education and reproductive health service program on contraceptive use of unmarried youths in Shanghai. Journal of Adolescent Health 34(5):433-440.

Harrison C, Parker P, Honey S (2005) Stepping Stones: parenting skills in the community. Community Practitioner 78(2):58-62.

Jemmott IJB, Jemmott LS, Fong GT (1998) Abstinence and safer sex HIV risk-reduction interventions for African American adolescents a randomized controlled trial. Journal of the American Medical Association 279(19):1529-1536.

Jemmott JB, III, And O (1992) Reductions in HIV Risk-Associated Sexual Behaviors among Black Male Adolescents: Effects of an AIDS Prevention Intervention. American Journal of Public Health 82(3):372-77.

Kipke MD, Boyer C, Hein K (1993) An evaluation of an AIDS Risk Reduction Education and Skills Training (ARREST) program. Journal of Adolescent Health 14(7):533-539.

Lederman RP, Mian TS (2003) The parent-adolescent relationship education (PARE) program: a curriculum for prevention of STDs and pregnancy in middle school youth. Behavioral Medicine 29(1):33-41.

Lee J, Donlan W, Paz J (2009) Culturally competent HIV/AIDS prevention: understanding program effects on adolescent beliefs, attitudes, and behaviors. Journal of HIV/AIDS & Social Services 8(1):57-80.

Meltzer IJ, Fitzgibbon JJ, Leahy PJ, Petsko KE (2006) A youth development program: Lasting impact. Clinical Pediatrics 45(7):655-660.

Midford R, McBride N, Farringdon F, Woolmer J (2000) The impact of a Youth Alcohol Forum: What changes for the participants? International Journal of Health Promotion and Education 38(2):65-70.

Miller BC, And O (1993) Impact Evaluation of FACTS & Feelings: A Home-Based Video Sex Education Curriculum. Family Relations 42(4):392-400.

Pearlman DN, Camberg L, Wallace LJ, Symons P, Finison L (2002) Tapping youth as agents for change: evaluation of a peer leadership HIV/AIDS intervention, in The Journal of adolescent health : official publication of the Society for Adolescent Medicine, pp 31-9.

Philliber S, Kaye JW, Herrling S, West E (2002) Preventing pregnancy and improving health care access among teenagers: an evaluation of the Children's Aid Society-Carrera Program. Perspectives on Sexual & Reproductive Health 34(5):244-252.

Pinkerton SD (2000) Economic evaluation of HIV risk reduction intervention in African-American male adolescents. Journal of Acquired Immune Deficiency Syndromes and Human Retrovirology 25:164-172.

Rohrbach LA, Johnson CA, Mansergh G, Fishkin SA, Neumann FB (1997) Alcohol-related outcomes of the Day One Community Partnership. Evaluation and Program Planning 20(3):315-322.

Schinke SP, Cole KCA, Fang L (2009) Gender-Specific Intervention to Reduce Underage Drinking Among Early Adolescent Girls: A Test of a Computer-Mediated, Mother-Daughter Program. Journal of Studies on Alcohol and Drugs 70(1):70-77.

Smith LW, Brown HN, Lewallen LP, Penny JM (2006) Religious activities and health outcomes: the College Bound Sisters Program. Issues in Mental Health Nursing 27(2):199-212.

Spoth R, Redmond C, Shin C, Azevedo K (2004) Brief Family Intervention Effects on Adolescent Substance Initiation: School-Level Growth Curve Analyses 6 Years Following Baseline. Journal of Consulting and Clinical Psychology 72(3):535-542.

Spoth R, Shin C, Guyll M, Redmond C, Azevedo K (2006) Universality of effects: an examination of the comparability of long-term family intervention effects on substance use across risk-related

subgroups. Prevention science : the official journal of the Society for Prevention Research 7(2):209-24.

Spoth RL, Guyll M, Day SX (2002) Universal family-focused interventions in alcohol-use disorder prevention: cost-effectiveness and cost-benefit analyses of two interventions. Journal of Studies on Alcohol 63(2):219-28.

St Pierre TL, And O (1995) A 27-Month Evaluation of a Sexual Activity Prevention Program in Boys & Girls Clubs across the Nation. Family Relations 44(1):69-77.

Summerville G, Canova K (2006) The Power of Plain Talk: Exploring One Program' Influence on the Adolescent Reproductive Health Field, pp 34. Public/Private Ventures, Philadelphia, PA.

Toomey TL, And O (1996) An Alcohol Primary Prevention Program for Parents of 7th Graders: The Amazing Alternatives! Home Program. Journal of Child and Adolescent Substance Abuse 5(4):35-53.

Turner-Musa JO, Rhodes WA, Harper PTH, Quinton SL (2008) Hip-hop to prevent substance use and HIV among African-American youth: A preliminary investigation. Journal of Drug Education 38(4):351-365.

Turrisi R, Jaccard J, Taki R, Dunnam H, Grimes J (2001) Examination of the short-term efficacy of a parent intervention to reduce college student drinking tendencies. Psychology of Addictive Behaviors 15(4):366-72.

Villalbi JR (2000) Effects on attitudes, knowledge, intentions and behaviour of an AIDS prevention programme targeting secondary school adolescents. Promotion et Education 7(3):17-22.

Villarruel AM, Jemmott IJB, Jemmott LS (2006) A randomized controlled trial testing an HIV prevention intervention for Latino youth. Archives of Pediatrics and Adolescent Medicine 160(8):772-777.

Walker KE, Kotloff LJ (1999) Plain Talk: Addressing Adolescent Sexuality through a Community Initiative. A Final Evaluation Report, pp 143. Public/Private Ventures, Philadelphia, PA.

• Intervention targeted 'at risk' or high risk population (n=4)

Carter-Jessop L, Franklin LN, Heath Jr JW, Jimenez-Irizarry G, Peace MD (2000) Abstinence education for urban youth. Journal of Community Health 25(4):293-304.

Conrod PJ, Stewart SH, Comeau N, Maclean AM (2006) Efficacy of cognitive-behavioral interventions targeting personality risk factors for youth alcohol misuse. Journal of Clinical Child & Adolescent Psychology 35(4):550-63.

Longshore D, Ellickson PL, McCaffrey, DF, St Clair PA (2007) School-Based Drug Prevention among At-Risk Adolescents: Effects of ALERT Plus. Health Education and Behavior 34:18-66 Watson DW, Bisesi L, Tanamly S (2004) Substance Use HIV Intervention Program (SHIP): A School-Based Indicated Prevention Model for Juvenile Offenders in the Alternative School Setting. Journal of Correctional Education 55:24-259.

• Intervention examined was not alcohol education and/SRE related (n=17)

Casey MK, Timmermann L, Allen M, Krahn S, Turkiewicz KL (2009) Response and self-efficacy of condom use: A meta-analysis of this important element of AIDS education and prevention, vol 74, pp 57-78.

Cuijpers P (2002) Peer-led and adult-led school drug prevention: a meta-analytic comparison. Journal of Drug Education 32(2):107-119.

Dixon DJ, McLearen AM (2002) The Effectiveness of an Alcohol Abuse Prevention Program among High School Students in Rural Missouri. Rural Educator 24(1):18-21.

Feltey KM, Ainslie JJ, Geib A (1991) Sexual Coercion Attitudes among High School Students: The Influence of Gender and Rape Education. Youth and Society 23(2):229-250.

Foshee VA, Linder GF, Bauman KE, Langwick SA, Arriaga XB, Heath JL, McMahon PM, Bangdiwala S (1996) The Safe Dates Project: theoretical basis, evaluation design, and selected baseline findings 5:S39-47

Elder RW, Nichols JL, Shults RA, Sleet DA, Barrios LC, Compton R (2005) Effectiveness of schoolbased programs for reducing drinking and driving and riding with drinking drivers: a systematic review. American Journal of Preventive Medicine 28(5 Supplement):288-304.

Griffin KW, Botvin GJ, Nichols TR (2004) Long-term follow-up effects of a school-based drug abuse prevention program on adolescent risky driving, in Prevention science : the official journal of the Society for Prevention Research, pp 207-12.

McCambridge J, Strang J (2004) The efficacy of single-session motivational interviewing in reducing drug consumption and perceptions of drug-related risk and harm among young people: Results from a multi-site cluster randomized trial. Addiction 99(1):39-52.

McCambridge J, Strang J (2005) Deterioration over time in effect of Motivational Interviewing in reducing drug consumption and related risk among young people. Addiction 100(4):470-478.

Moyers PA, Stoffel VC (2004) An evidence-based and occupational perspective of interventions for persons with substance-use disorders. American Journal of Occupational Therapy 58(5):570-586.

Sheehan M, Schonfeld C, Ballard R, Schofield F, Najman J, Siskind V (1996) A three year outcome evaluation of a theory based drink driving education program. Journal of Drug Education 26(3):295-312.

Shope JT, Elliott MR, Raghunathan TE, Waller PF (2001) Long-term follow-up of a high school alcohol misuse prevention program's effect on students' subsequent driving. Alcoholism: Clinical & Experimental Research 25(3):403-10.

Simons-Morton B, Haynie D, Saylor K, Crump AD, Chen R (2005) Impact analysis and mediation of outcomes: the Going Places program. Health Education & Behavior 32(2):227-41.

Weisz AN, Black BM (2001) Evaluating a Sexual Assault and Dating Violence Prevention Program for Urban Youth. Social Work Research 25(2):89-100.

Wilkins TT (2000) The "Stay Alive From Education" (SAFE) program: Description and preliminary pilot testing. Journal of Alcohol and Drug Education 45(2):1-11.

Wright V, Akers SW, Rita S (2000) The Community Awareness Rape Education (CARE) program for high school students. Journal of Emergency Nursing 26(2):182-5.

Yates BA, Dowrick PW (1991) Stop the Drinking Driver: A Behavioral School-Based Prevention Program. Journal of Alcohol and Drug Education 36(2):12-19.

Appendix 3. Summary of quality assessment

Table 10.1Quality assessment: Randomised controlled trials

	Aarons, et al., 2000	Bellingham, et al., 1993	Blake, et al., 2001	Borawski, et al., 2009	Borgia, et al., 2005	Coyle, 1999	Coyle, 2004	Coyle, 2006	Eisen, 1990
Section 1: Population								•	•
1.1 Is the source population or source area well described?	NR	-	+	NR	NR	NR	++	NR	++
1.2 Is the eligible population or area representative of the source population or area?	NR	+	++	NR	+	NR	++	-	++
1.3 Do the selected participants or areas represent the eligible population or area?	NR	++	++	NR	+	NR	++	-	NR
Section 2: Method of allocation	1	1					1		
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	+	+	NR	++	++	+	NR	+	NR
2.2 Were interventions (and comparisons) well described and appropriate?	-	+	++	++	+	++	+	++	++
2.3 Was the allocation concealed?	NR	NR	NR	+	NR	NR	NR	NA	NR
2.4 Were participants and/or investigators blind to exposure and comparison?	NR	-	-	NR	+	NR	NR	NA	NR
2.5 Was the exposure to the intervention and comparison adequate?	-	+	++	++	+	++	++	+	-
2.6 Was contamination acceptably low?	NR	NR	NR	++	++	++	++	NR	NR
2.7 Were other interventions similar in both groups?	NR	NA	++	NR	++	++	++	NR	+
2.8 Were all participants accounted for at study conclusion?	++	++	-	+	+	+	+	+	+
2.9 Did the setting reflect usual UK practice?	+	++	++	NA	+	NR	+	+	NA
2.10 Did the intervention or control comparison reflect usual UK practice?	+	++	+	NA	+	NR	+	+	+
Section 3: Outcomes									
3.1 Were outcome measures reliable?	NR	++	++	++	++	++	++	+	+

	Aarons, et al., 2000	Bellingham, et al., 1993	Blake, et al., 2001	Borawski, et al., 2009	Borgia, et al., 2005	Coyle, 1999	Coyle, 2004	Coyle, 2006	Eisen, 1990		
3.2 Were all outcome measurements complete?	++	++	++	NA	++	NA	+	++	++		
3.3 Were all important outcomes assessed?	++	++	++	++	++	++	++	++	++		
3.4 Were outcomes relevant?	++	++	++	++	++	++	++	++	++		
3.5 Were there similar follow-up times in exposure and comparison groups?	++	++	++	++	++	++	++	++	NR		
3.6 Was follow-up time meaningful?	+	-	-	++	+	++	++	++	++		
Section 4: Analyses	1	Π		I			T	1	1		
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	++	NR	+	++	-	+	++	+	++		
4.2 Was Intention to treat (ITT) analysis conducted?	NR	NR	NA	NR	NR	NR	NR	NR	NR		
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	NR	+	NR	NR	-	NR	+	NR	NR		
4.4 Were the estimates of effect size given or calculable?	++	++	++	-	+	++	NR	++	NR		
4.5 Were the analytical methods appropriate?	+	+	++	+	+	++	+	++	++		
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	+	++	NR	-	+	++	++	++	NR		
Section 5: Summary			-								
5.1 Are the study results internally valid (i.e. unbiased)?	+	+	+	+	+	+	+	++	+		
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	+	++	+	+	+	-	+	-	+		
a RCT based on randomisation at the individua	al level										
NR – not reported; NA – not applicable	NR – not reported; NA – not applicable										

Table 10.2Quality assessment: Randomised controlled trials continued

	Flay, et al., 2004	Graham, et al., 2002	Graham, et al., 1990	Harrington, et al., 2001	Komro, et al., 2008	Kulis, et al., 2007	Kvalem, et al., 1996	Lachausse, 2006	Levy, et al., 1995
Section 1: Population				•					
1.1 Is the source population or source area well described?	++	++	-	++	+	+	NR	+	NR
1.2 Is the eligible population or area representative of the source population or area?	++	++	-	++	++	++	NR	+	NR
1.3 Do the selected participants or areas represent the eligible population or area?	++	++	-	++	++	++	NR	+	NR
Section 2: Method of allocation							-		
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	++	++	NR	NR	++	NR	+	+	+
2.2 Were interventions (and comparisons) well described and appropriate?	++	++	+	++	++	+	++	++	++
2.3 Was the allocation concealed?	NR	++	NR	NR	NR	NR	NR	NR	NR
2.4 Were participants and/or investigators blind to exposure and comparison?	NR	-	NR	NR	NR	NR	NR	NR	NR
2.5 Was the exposure to the intervention and comparison adequate?	++	++	NR	NR	++	++	+	NR	++
2.6 Was contamination acceptably low?	++	++	NR	NR	++	NR	NR	NR	NA
2.7 Were other interventions similar in both groups?	++	NR	NA	++	++	+	NR	NA	NR
2.8 Were all participants accounted for at study conclusion?	++	NR	NR	NR	++	+	++	+	+
2.9 Did the setting reflect usual UK practice?	-	++	++	++	++	-	NA	++	+
2.10 Did the intervention or control comparison reflect usual UK practice?	++	++	++	++	++	-	NA	++	-
Section 3: Outcomes	-	-	-						
3.1 Were outcome measures reliable?	++	+	++	++	++	++	NR	++	NR

	Flay, et al., 2004	Graham, et al., 2002	Graham, et al., 1990	Harrington, et al., 2001	Komro, et al., 2008	Kulis, et al., 2007	Kvalem, et al., 1996	Lachausse, 2006	Levy, et al., 1995	
3.2 Were all outcome measurements complete?	++	NA	++	++	++	++	NA	++	NA	
3.3 Were all important outcomes assessed?	++	+	++	++	++	+	+	++	+	
3.4 Were outcomes relevant?	++	++	++	++	++	+	+	++	+	
3.5 Were there similar follow-up times in exposure and comparison groups?	++	++	++	++	++	++	++	++	+	
3.6 Was follow-up time meaningful?	++	+	++	++	++	++	++	+	-	
Section 4: Analyses	1	1	1	1	1		1			
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	++	++	++	++	++	++	+	+	+	
4.2 Was Intention to treat (ITT) analysis conducted?	NR	++	NR	NA	++	++	NR	NR	NR	
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	NR	++	++	++	++	++	NR	+	NR	
4.4 Were the estimates of effect size given or calculable?	++	++	++	++	++	NR	+	++	++	
4.5 Were the analytical methods appropriate?	++	++	++	++	++	-	+	+	+	
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	++	++	++	++	NR	+	-	++	++	
Section 5: Summary				1						
5.1 Are the study results internally valid (i.e. unbiased)?	++	++	-	+	++	+	-	+	+	
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	++	++	-	++	++	+	-	+	-	
a RCT based on randomisation at the individua	a RCT based on randomisation at the individual level									
NR – not reported; NA – not applicable										

Table 10.3Quality assessment: Randomised controlled trials continued

	McNeal, et al., 2004	Mitchell- Dicenso, et al., 1997	Morgenstern, et al., 2009	Newton, et al., 2009	O'Donnell, et al., 1999	Patton, et al., 2006	Roberto, et al., 2007	Schaalma, et al., 1996	Shortt, et al., 2007	
Section 1: Population										
1.1 Is the source population or source area well described?	-	+	NR	NR	NR	NR	-	NR	NR	
1.2 Is the eligible population or area representative of the source population or area?	++	++	+	NR	-	NR	NR	NR	NR	
1.3 Do the selected participants or areas represent the eligible population or area?	++	++	NR	NR	+	NR	NR	NR	NR	
Section 2: Method of allocation										
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	NR	+	+	+	NR	+	+	+	+	
2.2 Were interventions (and comparisons) well described and appropriate?	++	++	+	++	+	+	+	++	+	
2.3 Was the allocation concealed?	NR	NR	+	NR	NR	NR	NR	NR	++	
2.4 Were participants and/or investigators blind to exposure and comparison?	NR	NR	NR	NR	NR	NR	NR	NR	NR	
2.5 Was the exposure to the intervention and comparison adequate?	NR	++	+	+	NR	++	+	++	NR	
2.6 Was contamination acceptably low?	NR	NR	++	+	NR	++	+	++	++	
2.7 Were other interventions similar in both groups?	NA	NA	NR	++	NR	+	NR	NR	NR	
2.8 Were all participants accounted for at study conclusion?	NR	+	++	++	-	NA	-	+	++	
2.9 Did the setting reflect usual UK practice?	++	++	+	+	+	NA	NA	NA	NR	
2.10 Did the intervention or control comparison reflect usual UK practice?	++	++	+	+	+	NA	NA	NA	NR	
Section 3: Outcomes										
3.1 Were outcome measures reliable?	++	++	+	++	NR	+	+	+	++	

	McNeal, et al., 2004	Mitchell- Dicenso, et al., 1997	Morgenstern, et al., 2009	Newton, et al., 2009	O'Donnell, et al., 1999	Patton, et al., 2006	Roberto, et al., 2007	Schaalma, et al., 1996	Shortt, et al., 2007
3.2 Were all outcome measurements complete?	++	++	NA	NA	+	NA	NA	NA	NA
3.3 Were all important outcomes assessed?	++	++	++	++	+	+	+	+	++
3.4 Were outcomes relevant?	++	++	++	++	++	+	+	+	++
3.5 Were there similar follow-up times in exposure and comparison groups?	++	++	++	++	NR	+	+	+	+
3.6 Was follow-up time meaningful?	++	++	++	+	-	+	-	-	+
Section 4: Analyses	1	1	1	1	1		T		
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	++	NR	+	+	NR	+	+	NR	NR
4.2 Was Intention to treat (ITT) analysis conducted?	NR	++	++	NR	+	++	NR	-	NR
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	++	++	NR	NR	NR	++	NR	NR	NR
4.4 Were the estimates of effect size given or calculable?	++	++	++	+	+	+	-	+	++
4.5 Were the analytical methods appropriate?	++	++	++	+	+	+	+	++	++
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	++	++	++	+	++	+	-	-	++
Section 5: Summary									
5.1 Are the study results internally valid (i.e. unbiased)?	+	+	++	+	+	+	+	+	+
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	+	+	+	-	+	+	-	-	++
a RCT based on randomisation at the individu	ual level								
NR – not reported; NA – not applicable									

Table 10.4Quality assessment: Randomised controlled trials continued

	Smith, 1994	Spoth, et al., 2008	Stanton, et al., 2006	Stephenson, et al., 2004	Traeen, 2003	Vogl, et al., 2009	Weeks, et al., 1995	Werch, et al., 2008	Werch, et al., 2008
Section 1: Population									
1.1 Is the source population or source area well described?	NR	NR	NR	-	NR	NR	NR	NR	+
1.2 Is the eligible population or area representative of the source population or area?	NR	++	NR	+	NR	NR	NR	NR	NR
1.3 Do the selected participants or areas represent the eligible population or area?	NR	NR	NR	+	NR	NR	NR	NR	NR
Section 2: Method of allocation									
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	+	+	+	++	+	+	+	+	NR
2.2 Were interventions (and comparisons) well described and appropriate?	+	+	+	++	+	++	++	++	+
2.3 Was the allocation concealed?	NR	NR	NR	NR	NR	NR	NR	NR	NR
2.4 Were participants and/or investigators blind to exposure and comparison?	NR	NR	NR	NR	NR	++	NR	NA	NR
2.5 Was the exposure to the intervention and comparison adequate?	+	++	+	+	NR	+	+	+	++
2.6 Was contamination acceptably low?	-	NR	NR	+	++	NA	NR	NR	-
2.7 Were other interventions similar in both groups?	NR	NA	NR	NR	NR	NR	-	+	+
2.8 Were all participants accounted for at study conclusion?	++	++	+	++	++	++	++	+	+
2.9 Did the setting reflect usual UK practice?	NR	++	NR	++	++	+	+	NA	NR
2.10 Did the intervention or control comparison reflect usual UK practice?	NR	++	NR	+	++	+	+	NA	NR
Section 3: Outcomes									
3.1 Were outcome measures reliable?	++	++	++	NR	++	++	NR	+	++

	Smith, 1994	Spoth, et al., 2008	Stanton, et al., 2006	Stephenson, et al., 2004	Traeen, 2003	Vogl, et al., 2009	Weeks, et al., 1995	Werch, et al., 2008	Werch, et al., 2008
3.2 Were all outcome measurements complete?	NA	++	NA	+	+	++	+	NA	++
3.3 Were all important outcomes assessed?	+	++	++	++	++	++	+	++	+
3.4 Were outcomes relevant?	+	++	++	++	++	++	++	++	+
3.5 Were there similar follow-up times in exposure and comparison groups?	+	++	++	++	++	++	++	++	++
3.6 Was follow-up time meaningful?	-	++	+	++	++	++	+	+	-
Section 4: Analyses	1	1	1			1	1	1	
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	+	++	+	+	NR	+	++	+	++
4.2 Was Intention to treat (ITT) analysis conducted?	NR	++	++	-	NR	NR	NR	NR	NR
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	NR	++	NR	NR	+	+	NR	NR	NR
4.4 Were the estimates of effect size given or calculable?	-	++	+	++	+	+	+	+	+
4.5 Were the analytical methods appropriate?	-	++	+	+	++	+	+	+	+
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	-	++	+	++	++	+	++	+	NR
Section 5: Summary			_					-	
5.1 Are the study results internally valid (i.e. unbiased)?	-	++	+	++	+	+	+	+	+
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	-	++	+	+	+	+	+	-	+
a RCT based on randomisation at the individual level									
NR – not reported; NA – not applicable									

Table 10.5Quality assessment: Randomised controlled trials continued

	Wight, et al., 2002	Workman, et al., 1996	Zimmerman, et al., 2008
Section 1: Population			
1.1 Is the source population or source area well described?	NR	NR	NR
1.2 Is the eligible population or area representative of the source population or area?	NR	NR	NR
1.3 Do the selected participants or areas represent the eligible population or area?	NR	NR	NR
Section 2: Method of allocation			
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	+	+	+
2.2 Were interventions (and comparisons) well described and appropriate?	+	++	++
2.3 Was the allocation concealed?	NR	NR	NR
2.4 Were participants and/or investigators blind to exposure and comparison?	NR	NR	NR
2.5 Was the exposure to the intervention and comparison adequate?	+	++	++
2.6 Was contamination acceptably low?	NR	NR	NR
2.7 Were other interventions similar in both groups?	NR	NR	NR
2.8 Were all participants accounted for at study conclusion?	++	-	+
2.9 Did the setting reflect usual UK practice?	++	-	+
2.10 Did the intervention or control comparison reflect usual UK practice?	++	-	+
Section 3: Outcomes			
3.1 Were outcome measures reliable?	NR	+	++

	Wight, et al., 2002	Workman, et al., 1996	Zimmerman, et al., 2008
3.2 Were all outcome measurements complete?	+	NA	++
3.3 Were all important outcomes assessed?	++	+	++
3.4 Were outcomes relevant?	++	+	++
3.5 Were there similar follow-up times in exposure and comparison groups?	NR	++	-
3.6 Was follow-up time meaningful?	++	-	++
Section 4: Analyses	1	1	
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	++	NR	+
4.2 Was Intention to treat (ITT) analysis conducted?	NR	NR	-
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	++	NR	NR
4.4 Were the estimates of effect size given or calculable?	++	-	-
4.5 Were the analytical methods appropriate?	+	+	+
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	++	-	++
Section 5: Summary	Γ	Γ	
5.1 Are the study results internally valid (i.e. unbiased)?	++	-	+
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	+	-	+
a RCT based on randomisation at the individual level			
NR – not reported; NA – not applicable			

Table 10.6. Quality assessment: Non-randomised controlled trials

	Borawski, et al., 2005	Boyer, et al., 1997	Caron, et al., 2004	Dempster, et al., 2006	Denman, et al., 1995	Denny, et al., 1999	Denny, et al., 2006	Donnelly, et al., 2001	Fisher, et al., 2002
Section 1: Population									
1.1 Is the source population or source area well described?	-	++	-	NR	NR	NR	NR	-	NR
1.2 Is the eligible population or area representative of the source population or area?	+	NR	+	NR	NR	NR	NR	NR	NR
1.3 Do the selected participants or areas represent the eligible population or area?	+	NR	+	NR	NR	NR	NR	NR	NR
Section 2: Method of allocation			1			1			
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	NR	-	NR	-	-	-	-	NR	-
2.2 Were interventions (and comparisons) well described and appropriate?	+	+	+	+	+	+	+	+	++
2.3 Was the allocation concealed?	NR	NA	NR	NR	NA	NR	-	NR	NA
2.4 Were participants and/or investigators blind to exposure and comparison?	-	NA	-	NR	NA	NR	NA	NR	NA
2.5 Was the exposure to the intervention and comparison adequate?	-	++	+	+	+	NR	NR	NR	++
2.6 Was contamination acceptably low?	-	++	-	NR	-	NR	+	NR	++
2.7 Were other interventions similar in both groups?	+	NA	++	NR	NR	NR	NR	NA	++
2.8 Were all participants accounted for at study conclusion?	+	+	+	NR	+	-	-	NR	-
2.9 Did the setting reflect usual UK practice?	-	+	+	++	NR	NA	NA	-	++
2.10 Did the intervention or control comparison reflect usual UK practice?	-	+	+	++	NR	NA	NA	++	+
Section 3: Outcomes	1		1	1	1	1	1		1
3.1 Were outcome measures reliable?	NR	+	+	+	NR	+	+	+	++
3.2 Were all outcome measurements complete?	+	++	+	NR	NA	NA	NR	NR	++

	Borawski, et al., 2005	Boyer, et al., 1997	Caron, et al., 2004	Dempster, et al., 2006	Denman, et al., 1995	Denny, et al., 1999	Denny, et al., 2006	Donnelly, et al., 2001	Fisher, et al., 2002		
3.3 Were all important outcomes assessed?	+	++	+	++	-	+	+	+	++		
3.4 Were outcomes relevant?	+	++	++	++	+	+	+	++	++		
3.5 Were there similar follow-up times in exposure and comparison groups?	+	++	++	++	+	+	+	++	++		
3.6 Was follow-up time meaningful?	+	+	++	-	-	-	+	-	++		
Section 4: Analyses	1	1	1				1				
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	+	++	++	NR	+	NR	NR	+	++		
4.2 Was Intention to treat (ITT) analysis conducted?	NR	NR	NR	NR	NR	-	NA	NA	++		
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	NR	NR	+	NR	NR	NR	NR	NR	++		
4.4 Were the estimates of effect size given or calculable?	+	++	NR	-	-	+	NR	-	++		
4.5 Were the analytical methods appropriate?	+	++	+	+	+	+	+	-	++		
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	+	++	-	-	-	+	NR	-	NR		
Section 5: Summary				•		-					
5.1 Are the study results internally valid (i.e. unbiased)?	+	+	+	-	-	-	-	-	++		
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	+	+	+	-	-	-	-	+	++		
a RCT based on randomisation at the individual	a RCT based on randomisation at the individual level										
NR – not reported; NA – not applicable	R – not reported; NA – not applicable										

Table 10.7. Quality assessment: Non-randomised controlled trials continued

	Hubbard, et al., 1998	Jorgensen, et al., 1993	Kirby, 1991	Larsson, et al., 2006	Lennox, et al., 2008	Magnusson, et al., 2004	Mellanby, et al., 1995	Roosa, et al., 1990	Siegel, 1998
Section 1: Population									
1.1 Is the source population or source area well described?	-	++	NR	-	NR	NR	++	NR	NR
1.2 Is the eligible population or area representative of the source population or area?	-	++	NR	NR	+	NR	++	NR	NR
1.3 Do the selected participants or areas represent the eligible population or area?	-	++	NR	NR	+	NR	++	NR	NR
Section 2: Method of allocation				•	•				
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	+	+	+	NR	+	-	NR	-	-
2.2 Were interventions (and comparisons) well described and appropriate?	+	++	++	+	+	+	NR	-	+
2.3 Was the allocation concealed?	NR	NR	NR	NA	NR	NR	NR	-	NR
2.4 Were participants and/or investigators blind to exposure and comparison?	NR	NR	NR	NR	NR	NR	-	NR	NR
2.5 Was the exposure to the intervention and comparison adequate?	-	NR	NR	+	NR	+	++	-	+
2.6 Was contamination acceptably low?	NR	NR	+	NA	NR	++	NR	-	++
2.7 Were other interventions similar in both groups?	NR	NA	NR	NR	+	NR	NR	NR	NR
2.8 Were all participants accounted for at study conclusion?	-	NR	-	+	+	+	NA	-	-
2.9 Did the setting reflect usual UK practice?	+	++	+	NA	-	++	++	-	+
2.10 Did the intervention or control comparison reflect usual UK practice?	+	++	+	NA	-	++	+	-	+
Section 3: Outcomes									
3.1 Were outcome measures reliable?	NR	++	NR	NR	++	NR	+	+	NR

	Hubbard, et al., 1998	Jorgensen, et al., 1993	Kirby, 1991	Larsson, et al., 2006	Lennox, et al., 2008	Magnusson, et al., 2004	Mellanby, et al., 1995	Roosa, et al., 1990	Siegel, 1998
3.2 Were all outcome measurements complete?	+	++	++	NA	++	NA	++	NA	+
3.3 Were all important outcomes assessed?	+	++	+	+	++	+	++	++	+
3.4 Were outcomes relevant?	++	++	++	+	++	+	++	++	++
3.5 Were there similar follow-up times in exposure and comparison groups?	NR	++	+	+	++	+	NR	++	++
3.6 Was follow-up time meaningful?	++	+	++	-	+	-	++	-	+
Section 4: Analyses	1					ſ			
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	++	++	++	+	+	NR	NR	NR	+
4.2 Was Intention to treat (ITT) analysis conducted?	-	NR	NR	NR	NR	NR	NA	NR	-
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	NR	++	NR	+	++	NA	+	NR	NR
4.4 Were the estimates of effect size given or calculable?	++	++	+	+	+	-	++	-	NR
4.5 Were the analytical methods appropriate?	+	++	+	+	+	-	++	+	+
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	+	++	+	+	NR	-	++	-	NR
Section 5: Summary				1		1	1		
5.1 Are the study results internally valid (i.e. unbiased)?	+	+	+	+	+	-	+	-	+
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	+	+	+	-	+	-	++	-	+
a RCT based on randomisation at the individu	al level								
NR – not reported; NA – not applicable									

Table 10.8. Quality assessment: Non-randomised controlled trials continued

	Somers, 2006	Tucker, et al., 2007	Walter, et al., 1993	Wright, 1998
Section 1: Population				
1.1 Is the source population or source area well described?	NR	+	NR	NR
1.2 Is the eligible population or area representative of the source population or area?	NR	NR	NR	NR
1.3 Do the selected participants or areas represent the eligible population or area?	NR	NR	NR	NR
Section 2: Method of allocation				
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	-	NA	+	+
2.2 Were interventions (and comparisons) well described and appropriate?	-	NA	++	++
2.3 Was the allocation concealed?	+	NR	NR	NR
2.4 Were participants and/or investigators blind to exposure and comparison?	NR	NR	NR	NR
2.5 Was the exposure to the intervention and comparison adequate?	+	NR	+	++
2.6 Was contamination acceptably low?	++	NR	NR	++
2.7 Were other interventions similar in both groups?	NR	NR	NR	NR
2.8 Were all participants accounted for at study conclusion?	-	NA	-	++
2.9 Did the setting reflect usual UK practice?	+	++	++	NR
2.10 Did the intervention or control comparison reflect usual UK practice?	-	++	+	NR
Section 3: Outcomes				
3.1 Were outcome measures reliable?	-	NR	NR	++
3.2 Were all outcome measurements complete?	++	++	++	NA

	Somers, 2006	Tucker, et al., 2007	Walter, et al., 1993	Wright, 1998
3.3 Were all important outcomes assessed?	+	++	++	++
3.4 Were outcomes relevant?	++	++	++	++
3.5 Were there similar follow-up times in exposure and comparison groups?	-	+	+	++
3.6 Was follow-up time meaningful?	-	+	-	++
Section 4: Analyses				
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	-	NR	-	++
4.2 Was Intention to treat (ITT) analysis conducted?	NR	NA	NR	NR
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	NR	++	++	NR
4.4 Were the estimates of effect size given or calculable?	+	++	+	-
4.5 Were the analytical methods appropriate?	+	++	++	+
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	NR	++	++	-
Section 5: Summary				
5.1 Are the study results internally valid (i.e. unbiased)?	-	-	+	+
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	+	+	+	+
a RCT based on randomisation at the individual level				
NR – not reported; NA – not applicable				

Table 10.9. Quality assessment: Controlled before and after studies

	Christopher, et al., 1990	Gillies, et al., 1990	Lemieux, et al., 2008	Paine- Andrews, et al., 1999	Mellanby et al., 2001	Somers, et al., 2001	Stout, et al., 1996	Teitler, 1997	Trenholm, et al., 2008
Section 1: Population			I	I	L		L	L	
1.1 Is the source population or source area well described?	+	NR	-	+	NR	++	+	NR	+
1.2 Is the eligible population or area representative of the source population or area?	+	NR	NR	NA	NR	+	++	NR	+
1.3 Do the selected participants or areas represent the eligible population or area?	+	NR	NR	NA	NR	+	++	NR	+
Section 2: Method of allocation	•		•	•	•				
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	-	-	+	-	-	NR	+	NA	+
2.2 Were interventions (and comparisons) well described and appropriate?	++	-	+	+	+	++	-	+	+
2.3 Was the allocation concealed?	NR	NR	-	NA	-	NR	-	NR	NR
2.4 Were participants and/or investigators blind to exposure and comparison?	NR	NR	-	NA	-	NR	NR	NR	NR
2.5 Was the exposure to the intervention and comparison adequate?	+	NR	++	+	+	NR	NR	-	+
2.6 Was contamination acceptably low?	-	++	NR	NA	+	NR	NR	NR	-
2.7 Were other interventions similar in both groups?	NR	NR	NR	NA	+	NA	NA	NR	+
2.8 Were all participants accounted for at study conclusion?	-	+	NR	NA	+	NR	NR	NA	+
2.9 Did the setting reflect usual UK practice?	+	NR	NR	NA	+	++	+	+	-
2.10 Did the intervention or control comparison reflect usual UK practice?	-	NR	NR	NA	+	++	+	+	-
Section 3: Outcomes									
3.1 Were outcome measures reliable?	+	NR	+	+	+	+	++	NR	+

	Christopher, et al., 1990	Gillies, et al., 1990	Lemieux, et al., 2008	Paine- Andrews, et al., 1999	Mellanby et al., 2001	Somers, et al., 2001	Stout, et al., 1996	Teitler, 1997	Trenholm, et al., 2008
3.2 Were all outcome measurements complete?	++	NA	+	NA	-	++	++	+	+
3.3 Were all important outcomes assessed?	-	-	++	+	-	++	+	-	+
3.4 Were outcomes relevant?	++	+	++	+	+	++	++	++	+
3.5 Were there similar follow-up times in exposure and comparison groups?	++	+	NR	NR	+	++	NR	++	+
3.6 Was follow-up time meaningful?	-	-	NR	NR	-	+	NR	++	++
Section 4: Analyses		1		1	1				
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	++	+	NR	NR	-	+	+	-	+
4.2 Was Intention to treat (ITT) analysis conducted?	NR	NR	NA	NA	-	NR	NA	NA	NR
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	NR	NR	NR	NA	NR	NR	NR	-	NR
4.4 Were the estimates of effect size given or calculable?	+	-	NR	-	+	++	NR	+	NR
4.5 Were the analytical methods appropriate?	+	+	++	-	+	++	+	+	+
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	-	-	+	-	+	++	NR	+	NR
Section 5: Summary	1	[1	1	1				
5.1 Are the study results internally valid (i.e. unbiased)?	-	-	-	-	+	-	-	-	+
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	+	-	+	-	+	-	+	+	+
a RCT based on randomisation at the individu NR – not reported; NA – not applicable	al level								

Table 10.10. Quality assessment: Other study types

	Lewis, et al., 1999	Strange, et al., 2002	Vincent, et al., 2004
Section 1: Population			
1.1 Is the source population or source area well described?	+	NR	++
	+	NR	+
1.2 Is the eligible population or area representative of the source population or area?			
1.3 Do the selected participants or areas represent the eligible population or area?	NR	++	+
Section 2: Method of allocation	Г	I	Г
2.1 Allocation to intervention (or comparison). How was selection bias minimised?	NA	NA	NR
2.2 Were interventions (and comparisons) well described and appropriate?	+	++	-
2.3 Was the allocation concealed?	NA	NA	NR
2.4 Were participants and/or investigators blind to exposure and comparison?	NA	NA	NR
2.5 Was the exposure to the intervention and comparison adequate?	+	NA	NR
2.6 Was contamination acceptably low?	NR	NA	NR
2.7 Were other interventions similar in both groups?	NR	NA	NR
2.8 Were all participants accounted for at study conclusion?	NR	++	NA
2.9 Did the setting reflect usual UK practice?	+	++	-
2.10 Did the intervention or control comparison reflect usual UK practice?	+	++	NR
Section 3: Outcomes			
3.1 Were outcome measures reliable?	NR	NR	NA
3.2 Were all outcome measurements complete?	+	++	++

	Lewis, et al., 1999	Strange, et al., 2002	Vincent, et al., 2004
3.3 Were all important outcomes assessed?	+	-	-
3.4 Were outcomes relevant?	++	+	++
3.5 Were there similar follow-up times in exposure and comparison groups?	NA	NA	++
3.6 Was follow-up time meaningful?	++	-	++
Section 4: Analyses	1	T	
4.1 Were exposure and comparison groups similar at baseline? If not, were these adjusted?	NA	NA	NR
4.2 Was Intention to treat (ITT) analysis conducted?	NA	NR	NA
4.3 Was the study sufficiently powered to detect an intervention effect (if one exists)?	NR	+	NR
4.4 Were the estimates of effect size given or calculable?	NA	+	NR
4.5 Were the analytical methods appropriate?	+	+	+
4.6 Was the precision of intervention effects given or calculable? Were they meaningful?	+	-	+
Section 5: Summary	I	Γ	
5.1 Are the study results internally valid (i.e. unbiased)?	-	-	-
5.2 Are the findings generalisable to the source population (i.e. externally valid)?	+	-	+
a RCT based on randomisation at the individual level			
NR – not reported; NA – not applicable			

Table 10.11. Quality assessment for published economic evaluation studies

Study id Include a	entification uthor, title, reference, year of publication	Pentz, 1998	Swisher et al., 2004	Wang et al., 2000
Evaluati	on criterion		•	•
1.	Was a well-defined question posed in answerable form?	Yes	Partly, the use of a do-nothing comparator is specified in the cluster randomised trial but not really mentioned in the economic evaluation	Yes
1.1	Did the study examine both costs and effects of the service(s) or programme(s)?	Yes	Partly, costs are listed (but not resource use items) but effects are not specified nor quantified.	Yes
1.2	Did the study involve a comparison of alternatives?	Yes	Yes: Infused Life Skill Training (ILST) with Life Skill Training (LST) compared to control (standard practice).	Yes, vs. standard, information only HIV programme
1.3	Was a viewpoint for the analysis stated and was the study placed in any particular decision-making context?	Yes	Yes, a societal viewpoint and set in a school	Private sector in base case, public sector considered in sensitivity analyses
2.	Was a comprehensive description of the competing alternatives given (that is, can you tell who? did what? to whom? where? and how often?)?	Partly	No, especially the ILST description which is vague	Partially
2.1	Were any important alternatives omitted?	The text is a book chapter so alternatives are not mentioned in detail. There is a control intervention in the relevant C-RCT but the intervention may not be the same as that mentioned as an alternative in the ICER.	No clear description of incremental costs	No

Study id	entification	Pentz, 1998	Swisher et al., 2004	Wang et al., 2000	
Include a	author, title, reference, year of publication				
2.2	Was (Should) a do-nothing alternative (be) considered?	No	Yes, in the C-RCT, see answer to 1 above	No, not really practical as would likely be some intervention in schools related to HIV	
3.	Was the effectiveness of the programmes or services established?	Difficult to say because the results of the C-RCT are not mentioned but a figure of 9% reduction over 3 years of drunkenness is mentioned in the paper.	Yes	Yes	
3.1	Was this done through a randomised, controlled clinical trial? If so, did the trial protocol reflect what would happen in regular practice?	There is a discrepancy between the C-RCT results and description of intervention and its effects in the economic evaluation.	Yes a C-RCT. Probably	Yes	
3.2	Was effectiveness established through an overview of clinical studies?	No, C-RCT	No primary research	NA	
3.3	Were observational data or assumptions used to established effectiveness? If so, what are the potential biases in results?	Not clear	No	NA	
4.	Were all the important and relevant costs and consequences for each alternative identified?	Probably	No	Probably	
4.1	Was the range wide enough for the research question at hand?	Probably	Yes	Yes, the authors reported all costs incorporated into the analyses.	
4.2	Did it cover all relevant viewpoints? (Possible viewpoints include the community or social viewpoint, and those of patients and third-party payers.)	Yes	Only societal viewpoint	Assumptions made re: societal costs. i.e. for societal costs for PID were 50% of private sector costs	

Study id Include a	entification author, title, reference, year of publication	Pentz, 1998	Swisher et al., 2004	Wang et al., 2000	
4.3	Were capital costs, as well as operating costs, included?	N/A	No, for good reasons only staff and material costs were considered	Not reported	
5.	Were costs and consequences measured accurately in appropriate physical units (for example, hours of nursing time, number of physician visits, lost work-days, gained life-years)?	Yes	Costs were expressed, partly without resource consumption/unit cost breakdown. Effects were not defined (i.e. cost per smoking habit year gained at end of year 2).	Not reported	
5.1	Were any of the identified items omitted from measurement? If so, does this mean that they carried no weight in the subsequent analysis?	Νο	No	No	
5.2	Were there any special circumstances (for example, joint use of resources) that made measurement difficult? Were these circumstances handled appropriately?	Not clear	Appeared to be	No	
6.	Were costs and consequences valued credibly?	Not clear		Yes	
6.1	Were the sources of all values clearly identified? (Possible sources include market values, patient or client preferences and views, policy-makers' views and health professionals' judgements.)	Not clear	No	Yes	
6.2	Were market values employed for changes involving resources gained or depleted?	Not clear	Mainly staff salaries and student materials. Impossible to say if valuation was realistic	NR	
6.3	Where market values were absent (for example, volunteer labour), or did not reflect actual values (for example, clinic space donated at reduced rate), were adjustments made to approximate market values?	Not clear	Not applicable	NR	

Study id	entification	Pentz, 1998	Swisher et al., 2004	Wang et al., 2000
include a	author, title, reference, year of publication			
6.4	Was the valuation of consequences appropriate for the question posed (that is, has the appropriate type or types of analysis – cost-effectiveness, cost-benefit, cost-utility – been selected)?	Yes	No, not mentioned	NR
7.	Were costs and consequences adjusted for differential timing?	No		Yes
7.1	Were costs and consequences which occur in the future 'discounted' to their present values?	N/A	No	Medical costs discounted at 5%
7.2	Was any justification given for the discount rate used?	N/A	N/A	No
8.	Was an incremental analysis of costs and consequences of alternatives performed?	Yes		No
8.1	Were the additional (incremental) costs generated by one alternative over another compared to the additional effects, benefits or utilities generated?	Yes	Incremental costs were expressed (vaguely compared to standard practice), but no ICERs were calculated	No
9.	Was allowance made for uncertainty in the estimates of costs and consequences?	No		Partially
9.1	If data on costs or consequences were stochastic, were appropriate statistical analyses performed?	No	No	NR
9.2	Were study results sensitive to changes in the values (within the assumed range for sensitivity analysis, or within the confidence interval around the ratio of costs to consequences)?	N/A	Sensitivity analysis presented is a projection to year 3 of costs per student or teacher of each programme.	Yes, appropriate multivariate sensitivity analyses were conducted on key variables
10.	Did the presentation and discussion of study results include all issues of concern to users?	Partly	Partly, the issue of generalisability of results and meaning of the evaluation are not discussed	No

Study id Include a	entification author, title, reference, year of publication	Pentz, 1998	Swisher et al., 2004	Wang et al., 2000		
10.1	Were the conclusions of the analysis based on some overall index or ratio of costs to consequences (for example, cost-effectiveness ratio)? If so, was the index interpreted intelligently or in a mechanistic fashion?	Yes	No	No summary benefit measure used		
10.2	Were the results compared with those of others who have investigated the same question? If so, were allowances made for potential differences in study methodology?	Νο	Not directly			
10.3	Did the study discuss the generalisability of the results to other settings and patient/client groups?	No No		Yes, briefly considered, acknowledge that the cost-effectiveness of the programme may vary by geographic region.		
10.4	Did the study allude to, or take account of, other important factors in the choice or decision under consideration (for example, distribution of costs and consequences, or relevant ethical issues)?	No	No	No		
10.5	Did the study discuss issues of implementation, such as the feasibility of adopting the 'preferred' programme given existing financial or other constraints, and whether any freed resources could be redeployed to other worthwhile programmes?	Yes	No	No		
OVERAL	L ASSESSMENT OF THE STUDY					
How well	was the study conducted? Code ++, + or -	+	-	+		
Are the r this guide	results of this study directly applicable to the patient group targeted by eline?	Possibly. Difficulties in implementation are not discussed.	No, vague text, rural US situation, predominantly white males and low- socio economic status. Absence of description of effects.	Possibly, based on conservative estimates. But programmes based on abstinence-plus approach		

Appendix 4. Summary of quality assessment for studies included in Jones et al (2007)

Table 10.12. Quality assessment for systematic reviews and meta-analyses

- 1.1 The study addresses an appropriate and clearly focused question
- 1.2 A description of the methodology used is included
- 1.3 The literature search was sufficiently rigorous to identify all relevant studies
- 1.4 Study quality is assessed and taken into account
- 1.5 There are enough similarities between the studies selected to make combining them reasonable
- Key: ++ Well covered + Adequately covered Poorly covered NR Not reported N/A Not applicable

Reference(s)	Questions	Coding				
Reference(o)	1.1	1.2	1.3	1.4	1.5	ocumg
Bruvold 1990	+	-	NR	NR	+	-
Coggan et al., 2003	+	-	++	-	N/A	+
Cuijpers 2002	++	-	-	NR	+	-
Dusenbury et al., 1997	+	-	+	+	N/A	+
Foxcroft et al., 2002	++	++	++	++	N/A	++
Loveland-Cherry 2003	++	+	++	++	+	+
Skara and Sussman 2003	++	++	+	-	+	+
Tobler, 1993	+	+	-	+	+	+
Tobler et al., 1997	+	++	-	++	++	+
Tobler et al., 2000	++	+	-	++	+	+
Werch and Owen 2002	++	+	+	-	×	+
White et al., 2004	++	++	++	+	N/A	++

Table 10.13. Quality assessment for RCTs

- 1.1 The study addressed an appropriate and clearly focused question
- 1.2 The assignment of participants to intervention groups is randomised
- 1.3 An adequate concealment method is used
- 1.4 Participants and investigators are kept 'blind' about intervention allocation
- 1.5 The intervention and control groups are similar at the start of the trial
- 1.6 The only difference between groups is the intervention under investigation
- 1.7 All relevant outcomes are measured in a standard, valid and reliable way
- 1.8 What percentage of the participants or clusters recruited into each intervention arm of the study dropped out before the study was completed?
- 1.9 All participants are analysed in the groups to which they were allocated? (ITT)
- 1.10 Where the study is carried out at more than one site, results are comparable for all sites

Reference(s)	Question											
	1.1	1.2	1.3	1.4	1.5	1.6	1.7	1.8	1.9	1.10		
Baumann, 2006	~~	×	×	N/A	~	~	~~	42% from intervention and 25% from control	~	N/A	-	
Bennett, 1995	See Clayton et al., 1991, 1996											
Bond et al., 2004	~~~	~~~	×	N/A	~~~	×	~~	10%	<i>\ \ \</i>	×	++	
Botvin et al., 1990a, 1995a	~~	~	×	N/A	~~	×	~~	24% lost at 1 year follow-up	×	×	+	
Botvin et al., 1990b	~~	~~	×	N/A	~~	~~	~~	NR.	×	×	+	

Key: $\checkmark \checkmark \checkmark$ Well covered $\checkmark \checkmark$ Adequately covered \checkmark Poorly covered \star Not addressed NR Not reported N/A Not applicable

Reference(s)	Question										Rating
	1.1	1.2	1.3	1.4	1.5	1.6	1.7	1.8	1.9	1.10	
Botvin et al., 1995b	~~~	~~	NR	N/A	~~	~~	√√	NR	NR	×	+
Botvin et al., 2001a; 2001b; Griffin et al., 2003	~~	×	×	N/A	$\checkmark\checkmark$	×	<i>√ √ √</i>	17% not followed up.	×	×	+
Brewer, 1991	$\checkmark \checkmark \checkmark$	$\checkmark \checkmark \checkmark$	NR	NR	$\checkmark \checkmark \checkmark$	$\checkmark \checkmark \checkmark$	<i>√√√</i>	NR	$\checkmark \checkmark \checkmark$	N/A	+
Chou et al., 1998	~~	~~~	~	×	√√√	√√√	<i>√√√</i>	Complete cases Intervention (42.72%) control (41.03%)	$\sqrt{\sqrt{2}}$	~~	-
Clayton et al., 1991, 1996	~~	~	×	×	×	×	~	Overall, 45% rate of attrition.	×	×	-
Colnes 2000	VV	~~	×	N/A	~ ~ ~	×	~~~	47%	×	×	+
Dent et al., 2001	~~~	~	NR	N/A	~~	~~	$\checkmark\checkmark$	37%	~	NR	-
Donaldson et al., 1995, 2000	**	~	NR	N/A	NR	NR	$\checkmark\checkmark\checkmark$	NR	NR	N/A	-
Eisen et al., 2002	~~	~	×	N/A	$\checkmark\checkmark$	NR	✓	16% didn't complete PT survey	×	×	+
Ellickson et al.,1990, 1993a	vv	~~	NR	NR	~	~	~~	40%	NR	N/A	+
Ellickson et al.,1993b	**	<i>√√√</i>	NR	NR	$\checkmark\checkmark$	~~	~~	125% failed to take 10th or 12th grade survey.	$\checkmark\checkmark$	~~~	+
Ellickson et al., 2003	~~	~~	NR	NR	~~	~	~	9%	$\checkmark\checkmark$	×	+

Reference(s)	Question											
	1.1	1.2	1.3	1.4	1.5	1.6	1.7	1.8	1.9	1.10		
Fearnow- Kenney et al., 2003	~~	~	×	N/A	×	×	~~	~20%	×	×	-	
Graham et al., 1990	~~~	~	~	NR	~	~~	~~~	30%	NR	NR	+	
Hansen & Graham 1991	~~	~~	NR	NR	~~	~	~	20%	NR	N/A	-	
Hecht et al., 2003; Gosin et al., 2003	~~	~~	NR	NR	NR	NR	NR	NR	NR	N/A	-	
Johnson et al., 1990	~~	~~	N/A	N/A	~~	~~	~~	32%	NR	NR	+	
Komro et al., 1999	~ ~ ~	~~	×	N/A	N/A	N/A	~~~	NR	√ √	~~	+	
Kreft 1998	~~	~~	NR	N/A	NR	NR	~		NR	N/A	-	
Kulis et al., 2005	~ ~ ~	~	×	N/A	~	~~	~~~	NR	×	NR	-	
Lynam et al., 1999					See Cla	ayton et al., 1991	, 1996					
Newman et al., 1992	~~~	~	NR	N/A	NR	~	~~~	NR	NR		-	
Palmer et al., 1998	~~	~~	NR	NR	~~	~~	NR		NR	N/A	-	
Perry et al., 1996; Komro et al., 2001	~~~~~~~~~~~~~	~	×	N/A	~~	×	~~	19% lost at end of 8th grade	~	×	+	
Perry et al., 2003	~~	×	×	N/A	~~	~	~~	0.16	×	~	+	

Reference(s) Question										Rating	
	1.1	1.2	1.3	1.4	1.5	1.6	1.7	1.8	1.9	1.10	
Perry et al., 2002	~~~	~	×	~~	~~	×	~~	15% in 98	√	×	+
Piper et al., 2000	~~~	~	×	N/A	~~	×	~~~	32%	×	×	-
Ringwalt et al., 1991	~	×	×	N/A	~~	×	~~	9% drop out	×	×	-
Shope et al., 1992	~~	×	×	N/A	~	×	~	28% at 2.5 yr	×	×	-
Shope et al., 1994	~~	×	×	NR	~	×	~~	31%	NR	NR	-
Simons-Morton et al., 2005	~~~	~	×	N/A	~	~~	~~~	~50% lost to follow-up	×	NR	+
Slater et al., 2006	~~	~	×	N/A	~~~	~~~	~~~	31.4% over 2 years	√ √	~~	+
Smith et al., 2004; Vicary et al., 2004	~~	~	×	N/A	~~	×	~~	~10%	×	×	+
Spoth et al., 2002; 2005	~~~	~	×	N/A	~~	~~	~~	14%	√	×	+
Sussman et al., 1998; Sun et al., 2006	~~	~~	×	NR	$\checkmark\checkmark$	~	$\checkmark\checkmark$	~45%	NR	N/A	+
Sussman et al., 2003	~~	~~	×	N/A	NR	NR	~~	~45%	NR	N/A	+
Toomey et al., 1996					Se	e Perry et al., 199	96				
Warren et al., 2006	~~	~~	×	NR	~~	~~	~~	NR	NR	N/A	+

Reference(s)	Question											
	1.1	1.2	1.3	1.4	1.5	1.6	1.7	1.8	1.9	1.10		
Werch et al., 1996a	~~~	44	×	NR	~ ~ ~	<i>~~</i>	~~~	3.8% dropped out of the intervention group and 1.9% dropped out of the control group.	444	N/A	+	
Werch et al 1996b	~~~	~~	×	N/A	$\checkmark\checkmark\checkmark$	~~~~~~~~~~~~~	~~~	12% intervention group, and 9% control group.	$\checkmark\checkmark\checkmark$	N/A	++	
Werch et al., 1998	~~	~~	×	N/A	$\checkmark\checkmark$	~	~~	11%	×	N/A	+	
Werch et al., 2000a, 2001, 2003a	~~~	~	×	N/A	$\checkmark\checkmark$	~~	~~	23% from intervention group and 21% from the control group.	×	N/A	+	
Werch et al., 2000b	~ ~ ~	~~	NR	N/A	$\checkmark\checkmark$	~~	~ ~ ~	NR	×	N/A	+	
Werch et al., 2003b	$\checkmark \checkmark \checkmark$	~~~	$\checkmark\checkmark$	N/A	$\checkmark\checkmark$	~	~ ~ ~	2%	$\checkmark\checkmark$	N/A	++	
Werch et al., 2005a	~~~	~~	×	N/A	~ ~ ~	~~	~~~	3.3%	×	N/A	+	
Werch et al., 2005b	<i>~~</i>	~~~	NR	N/A	~ ~ ~	~~~	~~~	42 and 48 students dropped out at 12 months from the two arms respectively.	NR	N/A	++	
Werch et al., 2005c	~~~	~	NR	N/A	$\checkmark\checkmark$	~~~	~~~	13%	NR	N/A	+	
Williams et al., 1995					Se	e Perry et al., 19	96	·		•	•	
Table 10.14. Quality assessment for NRCT

- 1.11 The study addressed an appropriate and clearly focused question
- 1.12 The assignment of participants to intervention groups is randomised
- 1.13 An adequate concealment method is used
- 1.14 Participants and investigators are kept 'blind' about intervention allocation
- 1.15 The intervention and control groups are similar at the start of the trial
- 1.16 The only difference between groups is the intervention under investigation
- 1.17 All relevant outcomes are measured in a standard, valid and reliable way
- 1.18 What percentage of the participants or clusters recruited into each intervention arm of the study dropped out before the study was completed?
- 1.19 All participants are analysed in the groups to which they were allocated? (ITT)
- 1.20 Where the study is carried out at more than one site, results are comparable for all sites
- Key: VV Well covered V Adequately covered V Poorly covered Not addressed NR Not reported N/A Not applicable

Reference(s)	Question											
	1.1	1.2	1.3	1.4	1.5	1.6	1.7	1.8	1.9	1.10	·······································	
Botvin et al., 1997	~~	N/A	N/A	N/A	~~	×	~~	14%	×	×	-	
Caplan et al., 1992	~~	N/A	N/A	N/A	~~	~	~~~	NR	×	×	-	
Cuijpers et al., 2001; Smit et al., 2003	√ √	N/A	N/A	N/A	NR	NR	~~~	32% of intervention group; 20% dropped out from the control group	×	×	+	
Dedobbeleer & Desjardins, 2001	√ √	N/A	N/A	N/A	~~	~~	~ ~ ~	60% lost to follow at 30 months	NR	NR	-	

Reference(s)	Question											
11010101100(0)	1.1	1.2	1.3	1.4	1.5	1.6	1.7	1.8	1.9	1.10		
Ennett et al., 1994;	~~~	N/A	N/A	N/A	~~	~~	~	12% not followed up at 2 years (7th/8th grade); attrition at 6 years not clear/	NR	NR	+	
Fraguela et al., 2003	~~~~~~~~~~~~~	N/A	N/A	N/A	~~	~	~	60%	×	×	-	
McBride et al., 2000, 2003, 2004	~ ~ ~	N/A	N/A	N/A	~	×	√ √	~25%	×	×	+	
Padget et al., 2005	~~	N/A	N/A	N/A	~~	~~	$\checkmark\checkmark\checkmark$	18%	×	×	+	
Rosenbaum al., 1994	See Ennett et al., 1994											
Rosenbaum & Hanson, 1998	See Ennett et al., 1994											
Schnepf, 2002	~~	N/A	N/A	N/A	~~	×	~~	NR	×	N/A	-	
Valentine et al., 1998	~~~	N/A	N/A	N/A	√ √	<i>√ √</i>	√ √	71% middle school, 48% high school completed	~ ~ ~	N/A	-	
Webster et al., 2002	~~~	N/A	N/A	N/A	NR	<i>√ √ √</i>	$\checkmark\checkmark$	28% intervention and 19% control students	$\checkmark\checkmark\checkmark$	×	+	
Wilhelmsen et al., 1994	~~~	N/A	N/A	N/A	×	NR	✓ ✓	Not clear	NR	✓	-	

Table 10.15. Quality assessment for controlled before and after studies

- 1.1 Contemporaneous data collection
- Score DONE pre and post intervention periods for study and control sites are the same.
- Score NOT CLEAR if it is not clear in the paper, e.g. dates of collection are not mentioned in the text.
- Score NOT DONE if data collection was not conducted contemporaneously during pre and post intervention periods for study and control sites.
- 1.2 Appropriate choice of control site
- Studies using second site as controls:
- Score DONE if study and control sites are comparable with respect to dominant reimbursement system, level of care, setting of care and academic status.
- Score NOT CLEAR if not clear from paper whether study and control sites are comparable.
- Score NOT DONE if study and control sites are not comparable.
- 1.3 Baseline measurement

• Score DONE if performance or patient outcomes were measured prior to the intervention, and no substantial differences were present across study groups (e.g. where multiple pre intervention measures describe similar trends in intervention and control groups);

• Score NOT CLEAR if baseline measures are not reported, or if it is unclear whether baseline measures are substantially different across study groups;

- Score NOT DONE if there are differences at baseline in main outcome measures likely to undermine the post intervention differences (e.g. are differences between the groups before the intervention similar to those found post intervention).
- 1.4 Characteristics for studies using second site as control

• Score DONE if the authors state explicitly that the primary outcome variables were assessed blindly OR the outcome variables are objective e.g. length of hospital stay, drug levels as assessed by a standardised test;

- Score NOT CLEAR if not specified in the paper;
- Score NOT DONE if the outcomes were not assessed blindly.
- 1.5 Blinded assessment of primary outcome(s)

PSHE 9.4 Secondary school and FE review

• Score DONE if the authors state explicitly that the primary outcome variables were assessed blindly OR the outcome variables are objective e.g. length of hospital stay, drug levels as assessed by a standardised test;

• Score NOT CLEAR if not specified in the paper;

• Score NOT DONE if the outcomes were not assessed blindly.

1.6 Protection against contamination

Studies using second site as control

• Score DONE if allocation was by community, institution, or practice and is unlikely that the control group received the intervention;

• Score NOT CLEAR if providers were allocated within a clinic or practice and communication between experimental and group providers was likely to occur;

• Score NOT DONE if it is likely that the control group received the intervention (e.g. cross-over studies or if individuals rather than providers were randomised).

1.7 Reliable primary outcome measure(s)

• Score DONE if two or more raters with at least 90% agreement or kappa greater than or equal to 0.8 OR the outcome is obtained from some automated system e.g. length of hospital stay, drug levels as assessed by a standardised test;

• Score NOT CLEAR if reliability is not reported for outcome measures that are obtained by chart extraction or collected by an individual;

• Score NOT DONE if agreement is less than 90% or kappa is less than 0.8.

1.8 Follow up of professionals (protection against exclusion bias)

• Score DONE if outcome measures obtained 80-100% subjects allocated to groups. (Do not assume 100% follow-up unless stated explicitly.);

• Score NOT CLEAR if not specified in the paper;

• Score NOT DONE if outcome measures obtained for less than 80% of individuals allocated to groups.

1.9 Follow up of individuals

• Score DONE if outcome measures obtained 80-100% of individuals allocated to groups or for individuals who entered the study. (Do not assume 100% follow-up unless stated explicitly.);

• Score NOT CLEAR if not specified in the paper;

• Score NOT DONE if outcome measures obtained for less than 80% of individuals allocated to groups or for less than 80% of individuals who entered the study.

Reference(s)	Question									Coding
	1.1	1.2	1.3	1.4	1.5	1.6	1.7	1.8	1.9	
Argentos et al., 1991	Done	Not clear	Not clear	Not clear	Done	Not clear	Done	Not clear	Not clear	-
Bagnall, 1990	Not clear	Not clear	Not clear	Not clear	Done	Not clear	Done	Not clear	Not clear	-
Bremberg & Arborelius, 1994	Done	Not clear	Done	Done	Done	Not clear	Not clear	Not clear	Not clear	-
Dukes et al., 1996, 1997	Done	Not clear	Not clear	Not clear	Not done	Not clear	Not clear	Not done	Not done	-
Harmon, 1993	Done	Done	Not done	Done	Done	Done	Done	Done	Done	+
Klitzner et al., 1994	Done	Done	Done	Done	Done	Done	Not clear	Not done	Not done	+
Moberg & Piper, 1990	Done	Not clear	Done	Done	Not done	Done	Done	Not clear	Not clear	+
Shope et al., 1996a	Done	Not clear	Done	Not clear	Not done	Done	Not clear	Done	Done	+
Shope et al., 1996b; Shope et al., 1998	Not clear	Not clear	Not clear	Not clear	Done	Done	Done	Not clear	Not clear	-
Snow et al., 1992	Done	Not clear	Not clear	Not done	Not clear	-				
Snow et al., 1997	Not clear	Not clear	Not clear	Not done	Not done	Not clear	Not clear	Not clear	Not clear	-

Appendix 5. Conversion table for English key stages and US grade equivalents

Age	England		USA		
		Year		Grade	
0-4		-		-	
4-5	Pre-School	-	Pre Kindergarten	-	
5-6	Primary School	1	Kindergarten	-	
6-7	(Key Stage 1)	2		1	
7-8	Junior School (Key stage 2)	3	Elementary School	2	
8-9		4		3	
9-10		5		4	
10-11		6		5	
11-12		7		6	
12-13	Lower Secondary (Key stage 3)	8	Middle School	7	
13-14		9		8	
14-15	Upper Secondary	10		9	
15-16	(Key stage 4)	11	High School	10	
16-17	6th Form Collogo	-		11	
17-18	our ronn College	-		12	

Supplement C

A review of the effectiveness and cost effectiveness of alcohol and sex and relationship education for all children and young people aged 5-19 years in community settings

Final report

Lisa Jones, Geoff Bates, Jennifer Downing, Harry Sumnall, Mark A Bellis

Centre for Public Health, Liverpool John Moores University



.

January 2010

Contribution of authors (alphabetical)

Geoff Bates	Study selection, data extraction and quality assessment. Contributed to the writing of all sections of the report.
Mark Bellis	Overall management responsibility for the project. Commented on various drafts of the report.
Jennifer Downing	Study selection, data extraction and quality assessment. Contributed to the writing of all sections of the report.
Lisa Jones	Management responsibility for the project and responsible for overall content of the final report. Study selection, data extraction and quality assessment. Contributed to the writing of all sections of the report.
Harry Sumnall	Management responsibility for the project. Commented on various drafts of the report.

Acknowledgements

In addition to the authors of the report we would like to acknowledge the contribution of Michelle Wareing (MW), Olivia Wooding (OW), Katrina Stredder (KW) and Angelina Kurtev (AK) from the Centre for Public Health, LJMU for their assistance in full text screening and data extraction.

Contents

G	lossary	
A	bbrevia	tionsix
E	xecutive	e summaryxi
1	Intro	duction1
	1.1	Aims and objectives1
	1.2	Research question1
2	Back	ground2
3	Meth	odology5
	3.1	Search strategy
	3.2	Inclusion and exclusion criteria
	3.3	Data extraction strategy7
	3.4	Quality assessment strategy7
	3.5	Methods of analysis/synthesis
4	Sum	mary of study identification9
	4.1	Review of effectiveness and cost-effectiveness9
5	Prog	rammes targeting alcohol use11
	5.1	Systematic reviews and meta-analyses11
	5.2	Programmes delivered within social or community settings14
	5.3	Programmes delivered to families
	5.4	Programmes delivered to parents
	5.5	Programmes involving the wider community or mass media
	5.6	Review of published economic evaluations45
6	Prog	rammes targeting sexual health
	6.2	Programmes delivered within social and community settings

6.3	Programmes delivered within healthcare settings	72
6.4	Programmes delivered to families	85
6.5	Programmes delivered to parents	97
6.6	Programmes involving the wider community or mass media	106
6.7	Programmes targeting vulnerable young people	109
7 Pro	ogrammes targeting multiple health behaviours	114
7.1	Programmes delivered within social, healthcare and community settings	114
7.2	Programmes delivered to families or parents	119
8 Dis	scussion	128
8.1	Programmes targeting alcohol use	128
8.2	Programmes targeting sexual health	130
8.3	Programmes targeting multiple behaviours	134
8.4	Strengths and limitations	136
8.5	Research recommendations	137
9 Co	nclusions	139
9.1	Programmes targeting alcohol use	139
9.2	Programmes targeting sexual health	139
9.3	Programmes targeting multiple behaviours	140
9.4	Summary	140
10	References	141
Append	dices	
Append	lix 1. References to included studies	143
Append	lix 2. References to excluded studies	150
Append	lix 3. Quality assessment tables	176

Tables of figures

Figure 4.1. Process of study selection9
Table of tables
Table 2.1. Risks and protective factors associated with drug misuse. 4
Table 4.1. Summary of study designs identified from inclusion 10
Table 5.1. Summary of findings from systematic reviews and meta-analyses: Programmes targeting alcohol use 13
Table 5.2. Summary of programme content: programmes delivered within social, healthcare or community settings
Table 5.3. Programmes delivered in social, healthcare and community settings: effects on knowledge, skills and attitudes 18
Table 5.4. Programmes delivered in social, healthcare and community settings: effects on health and social outcomes related to alcohol use
Table 5.5. Summary of programme content: programmes delivered in families 26
Table 5.6. Programmes delivered in families: effects on knowledge, attitudes and skills
Table 5.7. Programme delivered to families: effects on health and social outcomes related to alcohol use 30
Table 5.8. Summary of programme content: programmes delivered to parents
Table 5.9. Programmes delivered to parents: effects on knowledge, attitudes and skills
Table 5.10. Programmes delivered to parents: effects on health and social outcomes related to alcohol use 40
Table 5.11. Summary of programme content: programmes involving the wider community or mass media 43
Table 5.12. Programmes involving the wider community or mass media: effects on knowledge, attitudes and skills 44
Table 5.13. Programmes involving the wider community or mass media: effects on health and social outcomes related to alcohol use
Table 5.14. Summary of intervention costs (Spoth et al., 2002) 46
Table 6.1. Summary of findings from systematic reviews and meta-analyses: Sexual health

Table 6.2. Summary of programme content: programmes delivered in social and community settings
Table 6.3. Programme delivered in social and community settings: effects on knowledge, attitudes and skills 65
Table 6.4. Programme delivered in social and community settings: effects on health and social outcomes related to sexual health
Table 6.5. Summary of programme content: programmes delivered within healthcare settings 80
Table 6.6. Programme delivered in healthcare settings: effects on knowledge, attitudes and skills 81
Table 6.7. Programme delivered in healthcare settings: effects on health and social outcomes related to sexual health 82
Table 6.8. Summary of programme content: programmes delivered to families 91
Table 6.9. Programmes delivered to families: effects on knowledge, attitudes and skills
Table 6.10. Programmes delivered to families: intervention effects on health and social outcomes related to sexual health
Table 6.11. Summary of programme content: programme delivered to parents 102
Table 6.12. Programme delivered to parents: effects on knowledge, attitudes and skills 103
Table 6.13. Programme delivered to parents: effects on health and social outcomes related to sexual health 105
Table 6.14. Summary of programme content: programme involving the wider community or mass media 108
Table 6.15. Programme involving the wider community or mass media: effects on health and social outcomes related to sexual health
Table 6.16. Summary of programme content: programmes targeting vulnerable young people 112
Table 6.17. Programmes targeting vulnerable young people: effects on knowledge, attitudes and skills
Table 6.18. Programmes targeting vulnerable young people: effects on health and social outcomes related to sexual health
Table 7.1. Summary of programme content: programmes delivered in social, healthcare or community settings

Table 7.2. Programmes delivered in social, healthcare or community settings: effects on knowledge
attitudes and skills
Table 7.3. Programmes delivered in social, healthcare or community settings: effects on health and
social outcomes related to sexual health and alcohol use
Table 7.4. Summary of programme components: programmes delivered to families and parents 123
Table 7.5. Programmes delivered to families and parents: effects on knowledge, attitudes and skills
Table 7.6. Programmes delivered to families and parents: effects on health and social outcomes
related to sexual health and alcohol use 126
Table 10.1. Quality assessment: randomised controlled trials (individual)
Table 10.2. Quality assessment: randomised controlled trials (cluster) 177
Table 10.3. Quality assessment: other study designs 178
Table 10.4. Quality assessment: systematic reviews and meta-analyses
Table 10.5. Quality assessment for published economic evaluation studies 180

Glossary

Abstinence-only programmes	Programmes that encourage and promote abstinence as the best and only way to prevent pregnancy, HIV and other STIs.
Abstinence-plus programmes	Programmes that emphasise abstinence as the safest way to prevent pregnancy, HIV and other STIs, but also promote safer sex through the use of contraceptives.
American school grades	Education is divided into 3 levels: elementary school, junior high (or middle) school and high school.
Benefit-cost ratio	The benefits of a programme divided by its costs. One way of presenting the results of a cost-benefit analysis.
Bias	Deviation of results or inferences from the truth, or processes leading to such deviation. Any trend in the collection, analysis, interpretation, publication or review of data that can lead to conclusions that are systematically different from the truth.
Binge drinking	Consuming large quantities of alcohol over a short period of time. Often associated with drinking to become intoxicated.
Cost-effectiveness analysis	An economic evaluation technique in which outcomes are measured in natural units.
Cluster randomisation	A trial where the unit of randomisation is a cluster of participants (e.g. a school).
Controlled before and after study (CBA)	Intervention groups are tested and data collected before and after the intervention has been administered. Differ from controlled non-randomised trials in that participants are not allocated to intervention or control groups, but rather a 'convenience' control sample is used.
Effect size	Effect size is a term used for a family of indices that measure the magnitude of the relationship between variables or treatment effect. Effect sizes are commonly used in meta-analyses as unlike significance tests these indices are independent of sample size.
Generalisability	The extent to which the results and conclusions from a study may be validity transposed to other situations.
Intention to treat analysis	A method of data analysis in which all participants are analysed in the group they were assigned to at randomisation regardless of treatment adherence.

Internal validity	How well the study has minimised sources of bias and how likely it is that the intervention caused the observed outcomes.
Long-term outcome	Study outcomes evaluated at greater than one year post- intervention.
Medium-term outcome	Study outcomes evaluated at six months to one year post- intervention.
Mass media	Means of communication that reach large numbers of people in a short time, such as television, newspapers, magazines, and radio.
Mean difference	The difference between two means divided by an estimate of the within group standard deviation.
Meta-analysis	The combination of quantitative evidence from a number of studies.
Net present value (NPV)	The benefits of an intervention minus its costs, taking into account the discount rate.
Net (social) benefit	An NPV, which considers social benefits
Non-Randomised Controlled Trial	These are trials where participants or clusters are allocated between intervention and control groups but the allocation is not randomised or quasi-randomised (e.g. alternate allocation).
Odds ratio	The odds of the event occurring in one group (e.g. intervention) divided by the odds of the event occurring in the other group (e.g. control).
Randomised Controlled Trial	Individuals or, defined groups of individuals (clusters) are randomised to either an intervention or a control group. If well implemented, randomisation should ensure that intervention and control groups only differ in their exposure to treatment.
Short-term outcomes	Study outcomes evaluated at less than six months post- intervention.
Systematic review	A method of locating, appraising and synthesising evidence from primary studies, which adheres to a scientific methodology.
Uncontrolled before and after study	Intervention groups are tested and data collected before and after the intervention has been administered. No control group is used for comparison purposes.

Abbreviations

AESOP	AIDS Evaluation of Street Outreach Project
BPBR	Be Proud! Be Responsible!
CAS	Children's AID Society
СВА	Controlled before and after study
CI	Confidence interval
CTS	Cross-sectional time series
DCSF	Department for Children, Schools and Families
DfES	Department for Education And Skills
DH	Department of Health
ESOL	English Speakers for Other Languages
FOK	Focus on Kids
HEART	Heart Power! For Hispanics
HIV	Human immunodeficiency virus
ImPACT	Informed Parents And Children Together
ISFP	Iowa Strengthening Families Programme
ITT	Intention to treat
LST	Life Skills Training
MDHP	Mother/Daughter Health Promotion curriculum
MDRR	Mother/Daughter HIV risk reduction
NA	Not applicable
NICE	National Institute for Health and Clinical Excellence
NNT	Number needed to treat
NR	Not reported
NRCT	Non-Randomised Controlled Trial
NS	Non-significant
OR	Odds Ratio
PARE	Parent-Adolescent Relationship Education
PATH	Parent Preadolescent Training for HIV Prevention
PDFY	Preparing for the Drug Free Years
PSHE	Personal Social and Health Education
PT	Post-test
PWC	Parents Who Care
RAP	Reaching Adolescents and Parents
RCT	Randomised Controlled Trial
SAAF	Strong African-American Families
SE	Standard error
SHAPE	Sharing Healthy Adolescent and Parent Experiences
SR	Systematic Review
SRE	Sex and relationships education

- STANDStudents Together Against Negative DecisionsSTISexually transmitted infection
- STI Sexually transmitted infection
- UBA Uncontrolled before and after study
- YPYD Young People's Youth Development
- YUTHE Youth United Through Health Education

Executive summary

BACKGROUND

This review sought to identify effective and cost-effective interventions and programmes that addressed health literacy and personal skills in relation to alcohol use and sexual health in community-based settings, including parent-targeted and family-based approaches.

METHODS

The methods for the review followed NICE protocols for the development of NICE public health guidance. Eighteen databases were searched for effectiveness and cost-effectiveness studies published since 1990. One reviewer screened all titles and abstracts and full text screening was undertaken independently by two reviewers. Data extraction and quality assessment were undertaken by one reviewer and checked for accuracy by a second reviewer. Each study was also graded (++, + or -) based on the extent to which the design and execution of the study minimised the potential sources of bias. Results of the data extraction and quality assessment for each study of effectiveness and cost-effectiveness were presented in structured tables and as a narrative summary.

PROGRAMMES TARGETING ALCOHOL USE

Thirty-one articles met the criteria for inclusion in the review of community-based programmes targeting alcohol use by young people. Four articles were systematic reviews and/or meta-analyses, three articles reported on studies that examined intervention or programmes delivered within social, healthcare and community settings, 20 articles reported on studies that examined programmes or interventions delivered to families or parents, and three studies examined interventions or programmes that involved the wider community or mass media. One economic evaluation study was also identified.

Systematic reviews and meta-analyses

Four articles covering three systematic reviews and meta-analyses were identified for inclusion. One review examined interventions and programmes aimed at the primary prevention of alcohol use across a range of populations and settings, and two reviews interventions and programmes delivered to parents and families, respectively. One good quality review found that although there was no consistent evidence to determine which programmes were effective over the short to medium-term, one family-based programme, the Iowa Strengthening Families programme (ISFP), was effective over the longer term. The systematic reviews of interventions and programmes delivered to parents and families also highlighted the long-term effectiveness of this programme.

Evidence statement 1

There is strong evidence from three systematic reviews to suggest that a family-based programme, lowa Strengthening Families (ISFP), can produce long-term reductions (greater than 3 years) in alcohol use and heavy alcohol use.

Programmes delivered in social, healthcare and community settings

Three studies were identified that examined interventions and programmes targeting alcohol use, which were delivered in social, healthcare and community settings. All three studies were conducted within youth and after school agencies and were based in the USA. Of the three studies identified, two were cluster RCTs and one was based on CBA design. One study examined an interactive CD-ROM intervention designed to reduce early alcohol use, and two studies examined substance use prevention programmes which targeted migrant families, and adolescents enrolled in after school programmes, respectively. None of the studies examined intervention effects on knowledge and understanding. Short-term increases in perception of harm were reported in two studies of one CD-ROM intervention and one substance use prevention programme, respectively, but this effect was not sustained over the longer term. One study also found no impact of an after-school, youth development programme on participants' drug beliefs and there was no impact of a culturally tailored, substance use prevention programme on participants' susceptibility to alcohol. Intervention effects on personal and social skills were examined in one study of a CD-ROM intervention, which found a short-term intervention impact on assertion skills. Two studies, of a CD-ROM intervention and a culturally tailored, substance use prevention programme, respectively, found no intervention effects on health and social outcomes related to alcohol use. However, substance use remained low among both intervention and control participants throughout these studies. One study, which targeted older children (mean age 15 years) in after school programmes, however, reported a positive short- to medium-term effect on alcohol use.

Evidence statement 2

- 2 (a) There is inconsistent evidence from two RCTs and one CBA study to determine the effects of interventions and programmes delivered in social, healthcare and community settings on attitudes and values related to alcohol use.
- 2 (b) There is inconsistent evidence from two RCTs and one CBA study to determine the effects of interventions and programmes delivered in social and community settings on alcohol use. However, there is weak evidence from one CBA study to suggest that programmes that target older children may impact on alcohol consumption. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.

Programmes delivered to parents and families

Twenty studies were identified that examined programmes and interventions delivered to parents and families, which targeted adolescent alcohol or substance use. Evaluations of nine programmes delivered to families were reported on across fifteen studies and five studies examined parent-targeted interventions. Eighteen studies were RCTs, one was an NRCT and one was a CBA. A range of intervention approaches were examined across these studies, including home- and community-based interventions. Effects on knowledge and understanding were only examined in one study of a family-based programme and none of the parent-targeted interventions examined this outcome. Short-term intervention effects on attitudes and values related to alcohol use were found for two

family-based programmes but for parent-targeted interventions there was no clear effect on parental attitudes to adolescent drinking. Both family-based and parent-targeted interventions appeared to produce short-term improvements in parent-child communication. Two studies of CD-ROM based interventions showed positive programme effects on family communication skills and involvement skills and a culturally tailored programme had a short-term positive effect on parental communication. Short-term intervention effects on parent-child communication were found for three studies of parenttargeted interventions; two studies reported more frequent or recent parent-child communication about alcohol and one study showed positive long-term effects on parent-child communication regarding family rules about alcohol and alcohol related situations. Eleven studies examined the effects of family-based programmes on health and social outcomes related to alcohol use across eight programmes. Three programmes demonstrated non-significant effects on alcohol use, but across four programmes, short- and long-term positive effects on alcohol use were reported. In addition, six studies of four family-based programmes reported positive intervention effects on initiation of alcohol use in the medium- to long-term. The Iowa Strengthening Families Programme (ISFP) also had long-term positive effects on drunkenness and drinking without parental permission and long-term follow up of the Preparing for the Drug Free Years (PDFY) revealed a positive effect of this programme on women's alcohol abuse in early adulthood. Two studies examined the effects of parent-targeted interventions. One study found no intervention effects but a second study, of a programme that promoted zero tolerance to adolescent alcohol use, reported positive intervention effects on youth drinking and drunkenness.

Evidence statement 3

- 3 (a) There is no evidence from one RCT to determine the effect of programmes aimed at families on knowledge and understanding relating to alcohol use
- 3 (b) There is moderate evidence from two RCTs to suggest that programmes delivered to families may have short-term positive effects on attitudes and values related to alcohol. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.
- 3 (c) There is moderate evidence from two RCTs to suggest that programmes delivered to families which target family interaction may have positive effects on family communication, parental monitoring and parental rules about alcohol. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.
- 3 (d) There is moderate evidence from 11 RCTs to suggest that programmes delivered to families may have mixed effects on health outcomes related to alcohol use. Three RCTs showed no intervention effects on alcohol use. One RCT of a brief, family focused intervention (Iowa Strengthening Families Program) showed long-term reductions in alcohol use, initiation of alcohol use, and drunkenness and one RCT of a culturally-tailored family-based programme showed a long-term effect on initiation of alcohol use. In addition, one RCT of a CD-ROM intervention with parental involvement showed long-term reductions in monthly alcohol use.

Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.

3 (e) There is weak evidence from one RCT to suggest that physician-led interventions may have a long-term negative impact on alcohol use. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.

Evidence statement 4

- 4 (a) There is inconsistent evidence from one NRCT and two RCTs to determine the effects of interventions delivered to parents on attitudes and values relating to alcohol. However, there is weak evidence from one RCT to suggest that programmes aimed at parents can have positive short-term effects on young people's attitudes towards family rules and their influence as a deterrent for drinking. These findings may be only partially applicable to the UK as this study was implemented in the USA and may not be generalisable beyond this population.
- 4 (b) There is moderate evidence from two RCTs and one CBA study to suggest that interventions delivered to parents may have a positive short- to potentially long-term effect on parent-child communication about alcohol. These findings may be only partially applicable to the UK as they were not implemented in a UK setting and may not be generalisable beyond the populations studied.
- **4 (c)** There is insufficient and inconsistent evidence from one NRCT and one RCT to determine the effect of interventions delivered to parents on health and social outcomes relating to alcohol use among young people.

Programmes involving the wider community or mass media

Three studies were identified that examined programmes involving the wider community or mass media. All three studies were based on a CBA design. Two studies examined mass media programmes delivered in communities in the USA and New Zealand, respectively, and one study examined a 5-year community-based health promotion programme for adolescents on an American Indian Reservation. None of the included studies examined intervention effects on knowledge and understanding, or on personal and social skills. Only one study examined impacts on attitudes and values towards alcohol use, findings showed no effects of a long-term mass media programme or the community-wide campaign targeting American Indian adolescents on alcohol use.

Evidence statement 5

5 (a) There is weak evidence from one CBA study to suggest interventions and programmes involving mass media have no effect on attitudes and values related to alcohol use. These findings may be only partially applicable to the UK as the study was not implemented in a UK

setting and may not be generalisable beyond the populations studied.

5 (b) There is moderate evidence from three CBA studies to suggest that interventions and programmes involving the wider community or mass media have no effects on alcohol use by young people. Findings may only be partially applicable to the UK as the studies were conducted in the USA and New Zealand and may not be generalisable beyond the populations studied.

Review of published economic evaluations

One study was identified that met the criteria for inclusion in the review of published economic evaluations. The study evaluated the cost-effectiveness and net benefits of two brief, family-focused interventions, the ISFP and PDFY, compared to a minimal intervention approach. Overall the net benefit for the ISFP was \$5,923 per family and \$2,697 per family for PDFY. The benefit-cost ratios were 9.60 and 5.85, indicating that for every \$1 spent on the ISFP and PDFY, \$9.60 and \$5.85, respectively, were saved in medical costs. The generalisbility of the study to a UK context was unclear as the data used was based on studies conducted in the USA. In addition, projected alcohol use disorder rates were calculated based on US population data.

Evidence statement 6

There is moderate evidence from one economic evaluation study to suggest that programmes delivered to families may be cost-effective and cost saving. This evidence may be of limited applicability to a UK context because cost and benefit estimates were based on data from studies conducted in the USA.

PROGRAMMES TARGETING SEXUAL HEALTH

A total of 49 articles met the criteria for inclusion in the review of community-based programmes targeting young people's sexual health. Nine articles were systematic reviews and/or meta-analyses, 20 articles reported on studies that examined interventions or programmes delivered within social, healthcare and community settings, 15 articles reported on studies that examined programmes or interventions delivered to families or parents, two articles reported on studies that examined interventions or programmes that involved the wider community or mass media, and three articles reported on studies which examined interventions for vulnerable young people. No economic evaluation studies were identified for inclusion.

Systematic reviews and meta-analyses

Nine systematic reviews and meta-analyses were identified that examined the effectiveness of interventions and programmes across a range of settings and populations that targeted young people's sexual health behaviours. One systematic review focused on interventions and programmes that targeted sexual risk taking among young homeless people. Findings from six systematic reviews indicated that community-based programmes can affect sexual risk behaviours of young people. In particular, HIV prevention and sexual risk reduction programmes were effective in increasing condom use and reducing pregnancy. However, these programmes were found to have a limited impact on adolescent sexual activity. According to one systematic review, successful community-based

interventions were theoretically based, tailored to the target population, implemented by trained facilitators, and the content was diverse and delivered using a wide variety of methods.

Evidence statement 7

- 7 (a) There is strong evidence from five systematic reviews and meta-analyses to suggest that intervention and programmes delivered in a range of community settings can have a positive impact on young people's sexual risk behaviours, in particular, condom use and pregnancy.
- 7 (b) There is strong evidence from one systematic review to suggest that effective communitybased interventions and programmes are: (1) theoretically based; (2) tailored to the target population, (3) implemented by trained facilitators; (4) based on diverse content; and (5) delivered using a wide variety of methods.
- 7 (a) There is strong evidence from one systematic review to suggest that effective clinic-based programmes include: (1) a focus on a single gender or ethnic group; (2) HIV/STI education with skills building activities (e.g. condom application); (3) condom negotiation and sexual communication components; and (4) personalised risk assessments.

Programmes delivered in social, healthcare and community settings

Twenty studies were identified that examined interventions or programmes delivered within social, healthcare or community settings. Nine studies examined group education sessions or skills-based training interventions delivered in social and community settings, and five further studies in social and community settings, respectively, examined peer-led interventions, the Children's Aid Society (CAS) Carrera programme and a theatre production designed to inform young people about HIV. Six studies were conducted in healthcare settings including family planning clinics and primary care practices; four of which examined group-based education and/or skills-based interventions for sexually active young women, and two that examined interventions based around a health practitioner-led sexual health consultation. Of the included studies, 15 were RCTs, three were NRCTs, and two were CBA studies.

Across four studies that examined group education sessions or skills-based training interventions in community settings there were positive intervention effects on knowledge and understanding over the short- to medium-term. In addition, the three-year, CAS-Carrera programme had a positive impact on knowledge. There was no effect of a peer counselling intervention on knowledge, but two peer leadership interventions had positive effects on levels of knowledge among the peer leaders themselves. Four studies of interventions that specifically targeted sexually active young females in healthcare settings, reported consistent short- to medium-term improvements in sexual health-related knowledge among intervention participants. In addition, two studies of health practitioner-led sexual health consultations reported significant short-term increases in knowledge among intervention participants relative to controls.

Short-term decreases in intentions to engage in risky sexual intercourse were reported among black male adolescents who participated in an AIDS risk reduction intervention and an abstinence-based HIV risk reduction intervention resulted in short-term reductions in intentions to engage in any sexual intercourse. Across three studies that examined group education sessions and skills-based training interventions in community settings there were short-term increases in intervention participants' perception of their vulnerability to HIV infection. However, this effect was not sustained in the medium-term. Two studies found no effects of a theatre production intervention or peer leadership intervention, respectively, on HIV attitudes at follow-up. There were indications of positive intervention effects of group education sessions and skills-based training interventions delivered in community settings on attitudes and values related to condom use. However, these effects were not consistent, and were not maintained over the medium-term. There were fairly consistent positive intervention effects on condom use attitudes across three studies, which examined group-based education and skills-based interventions for sexually active young women in healthcare settings, and one study that examined a primary care-based sexual risk assessment and education intervention. Two studies found short-term positive intervention effects of a CD-ROM mediated intervention and an abstinencebased HIV risk reduction intervention, respectively, on attitudes towards abstinence. A CD-ROM intervention and education and skills training programme had positive effects on behavioural skills but results from five studies presented mixed findings in relation to effects of programmes and intervention delivered in social, healthcare and community settings on communication.

Across five studies that examined group-based sessions and/or skills training in community settings, short- to medium-term effects on sexual intercourse were reported in four studies; one study reported no programme effects and one poorly conducted study reported a potentially harmful effect. The CAS-Carrera programme had a positive effect on sexual activity among females, but there were no effects of health practitioner-led sexual health consultations or peer interventions. Intervention effects on frequency of sexual intercourse and number of sexual partners were limited. Across four studies conducted in community settings, only one study reported a positive intervention effect and across four studies conducted in healthcare settings, there were inconsistent intervention effects on these outcomes. Intervention effects on condom use and unprotected intercourse were more consistent. Across six studies that examined group-based sessions and skills training in community and healthcare settings, there were positive short- to medium-term intervention effects on measures of condom use, and some evidence from three studies of a positive intervention effect on frequency of unprotected intercourse. There were no effects of an HIV theatre production or peer counselling intervention on contraceptive use or frequency of unprotected sex, but the CAS-Carrera programme positively influenced both condom and hormonal contraceptive use among females. This programmes also had a positive effect on pregnancy, with a reduction in pregnancies among intervention females. There was no effect of a peer counselling intervention or peer leadership programme on pregnancy rates. Three studies examined intervention effects on STI infection and/or diagnosis, finding mixed intervention effects. However, medium-term positive effects on STI diagnosis were reported in one study of a skills-based HIV/STI intervention for sexually active females.

Evidence statement 8

- 8 (a) There is moderate evidence from five RCTs, one NRCT and one CBA study to suggest that group-based education and/or skills-based interventions, youth development programmes and peer leadership interventions delivered in social and community settings may have a positive short-to medium-term impact on knowledge and understanding related to sexual health. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 8 (b) There is inconsistent evidence from five RCTs, one NRCT and one CBA study on which to determine the effects of interventions and programmes delivered in social and community settings on attitudes and values related to sexual health. There was moderate evidence from three RCTs to suggest that group-based education and/or skills-based interventions may have positive short-, but not long-term, effects on attitudes and values related to condom use. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.

Evidence statement 8 continued

- 8 (c) There is weak evidence from two RCTs to suggest that group-based education and/or skillsbased interventions delivered in social and community settings may have a positive short-term impact on behavioural skills related to sexual health. There was no evidence on which to determine intervention effects on communication skills. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 8 (d) There is moderate evidence from four RCTs and one CBA study to suggest that group-based education and/or skills-based interventions may have limited effects on sexual activity. Although reductions in the likelihood of sexual intercourse were reported across four RCTs⁶ there was only evidence from one RCT of intervention effects on frequency of sexual intercourse or number of sexual partners. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 8 (e) There is weak evidence from four RCTs to suggest that group-based education and/or skillsbased interventions delivered in social and community settings may have positive short-term impacts on condom use and frequency of unprotected intercourse. There is weak evidence from one RCT to suggest that these effects may diminish over the medium term. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 8 (f) There is moderate evidence from one RCT to suggest that youth development programmes that target disadvantaged young people may have a positive impact on sexual behaviours among females, including sexual activity, condom use and pregnancy. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.

Evidence statement 9

- 9 (a) There is strong evidence from six RCTs to suggest that interventions and programmes delivered in healthcare settings may produce short- to medium-term improvements in sexual health-related knowledge. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 9 (b) There is strong evidence from three RCTs to suggest that group-based education and/or skillsbased interventions specifically targeting sexually active young women in healthcare settings may have short- to medium-term positive effects on condom use attitudes. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.

Evidence statement 9 continued

- 9 (c) There is inconsistent evidence from three RCTs on which to determine the effects of interventions and programmes delivered in healthcare settings on sexual health-related communication. However, there is strong evidence from one RCT to suggest that a genderand culturally-tailored intervention for African American females may have a positive impact on communication with sexual partners and condom use skills. This evidence may only be partially applicable because these studies were conducted in the USA and focused on an ethnic population specific to the USA.
- 9 (d) There is moderate evidence from two RCTs to suggest that interventions and programmes based on health practitioner-led sexual health consultations may have a limited impact on sexual behaviours, including sexual activity and condom and other contraceptive use. This evidence may only be partially applicable because these studies were conducted in the USA.
- 9 (e) There is strong evidence from four RCTs to suggest that group-based education and/or skillsbased interventions specifically targeting sexually active young women in healthcare settings may not have a consistent impact on sexual activity or numbers of sexual partners. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 9 (f) There is strong evidence from four RCTs to suggest that group-based education and/or skillsbased interventions specifically targeting sexually active young women in healthcare settings may have a short- to medium-term positive impact on condom and other contraceptive use, and unprotected intercourse. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 9 (g) There is inconsistent evidence from three RCTs on which to determine the effects of interventions and programmes delivered in healthcare settings on STIs. However, there is strong evidence from one RCT to suggest that a skill-based HIV/STI intervention may have a positive medium-term impact on STI diagnosis. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations specific to the USA.

Programmes delivered to parents and families

Fifteen studies were identified that examined interventions and programmes delivered to parents and families, which targeted adolescent sexual health. Ten studies evaluated seven programmes delivered to adolescents and their families and five studies examined parent-targeted interventions. Programmes and interventions were delivered in a variety of settings, including at home and in community-based settings. Eleven studies were RCTs, two studies were NRCTs, and two were CBA studies. Both family-based and parent-targeted interventions demonstrated positive influences on knowledge related to sexual health in the short-, medium- and long-term, with improvements seen in both parent and adolescent knowledge related to sexual health. Programmes and interventions delivered to families did not appear to be effective at influencing adolescent's attitudes and intentions towards resisting or delaying sex and across three studies that examined parent-targeted

interventions, there were inconsistent effects on intentions. There were mixed effects on parent-child communication across both family-based and parent-targeted interventions. Nine studies that examined family-programmes found no clear intervention effects on communication, but in general positive effects were found across four studies that examined parent-targeted interventions. Across five studies that examined the effects of family-based programmes on health and social outcomes related to sexual health the results suggested that programmes and interventions delivered to families may not affect sexual behaviour. Two studies found no intervention aimed at mothers and their adolescent children found no long-term effects on abstinence or involvement in intimate sexual behaviours. There were, however, limited but positive effects of this programme on condom use. There were positive short-term effects of two parent education programmes on initiation of sexual activity and behavioural risks related to early sexual initiation, respectively. However, lack of clear intervention effects were reported in two further studies of parent-targeted interventions.

Evidence statement 10

- 10 (a) There is moderate evidence from five RCTs and one NRCT to suggest that interventions and programmes delivered to families may improve knowledge in the short- to long-term. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.
- 10 (b) There is moderate evidence from five RCT and one NRCT to suggest that interventions and programmes delivered to families may not influence adolescent's attitudes or intentions regarding abstinence or delaying sex. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.
- 10 (c) There is moderate evidence from seven studies to suggest that programmes and interventions delivered to families may not influence parent-child communication. There is weak evidence from two CBA studies to suggest that intensive, family-focused interventions may have positive short-term effects on family communication. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.
- 10 (d) There is weak evidence from three RCT and two CBA studies to suggest that programmes delivered to families may not have effects on adolescent sexual behaviour. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.

Evidence statement 11

- 11 (a) There is moderate evidence from one RCT to suggest that training for mothers to be their daughters' primary HIV educator may produce short-term improvements in sexual health-related knowledge and understanding. The evidence may only be partially applicable to the UK as this study was conducted in the USA and focused on ethnic populations specific to the USA.
- 11 (b) There is inconsistent evidence from three RCTs and one NRCT on which to determine the effects of intervention and programmes delivered to parents on sexual health-related attitudes and values.
- 11 (c) There is weak evidence from three RCTs and one NRCT to suggest that interventions delivered to parents may improve parent-child communication about sexual health topics. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.
- 11 (d) There is inconsistent evidence from four RCTs on which to determine the effects of programme delivered to parents on their children's sexual behaviour.
- 11 (e) There is moderate evidence from one RCT to suggest that delivery of HIV prevention content by mothers may be as equally effective as that of health experts. The evidence may only be partially applicable to the UK as this study was conducted in the USA and focused on ethnic populations specific to the USA.

Programmes involving the wider community or mass media

Two studies were identified that examined interventions which involved the wider community or mass media. One study examined a mass media intervention and the second study examined a community outreach programme. Both studies were based on cross-sectional time series. Neither of the included studies examined intervention effects on knowledge, attitudes and skills and both analysed population-level changes in pregnancy and STI rates, respectively, as measures of effectiveness. Both studies reported positive intervention effects at a population level, however the study of the mass media programme did not adequately control for natural fluctuations in the data and therefore it is not clear whether these or intervention effects were responsible for the differences seen in the intervention and control communities.

Evidence statement 12

- 12 (a) There is no evidence from two CTS on which to determine the effects of interventions and programmes involving the wider community or mass media on knowledge, attitudes and skills related to sexual health.
- 12 (b) There is weak evidence from one CTS to suggest that a programme of community outreach may have a positive impact on STI rates among young people. Findings may only be partially applicable to the UK as the study was not conducted in a UK setting and may not be generalisable beyond the population studied.

Programmes targeting vulnerable groups

Three studies examined the effectiveness of community-based programmes that targeted vulnerable populations. All three studies examined interventions which specifically targeted young homeless people. Intervention approaches examined were street outreach, a brief group-based sexual health intervention and a community reinforcement approach combined with HIV prevention. Intervention effects on knowledge and skills were examined in only one study and none of the included examined intervention of ffects on attitudes and values. There were limited effects of a brief sexual health intervention on knowledge related to AIDS and other STIs, and on communication and self-efficacy. Health and social outcomes related to sexual health were examined in all three studies, two of which reported no intervention effects. One study found a positive effect on the frequency of condom use among younger participants in a programme which combined a community reinforcement approach with HIV prevention content.

Evidence statement 13

- 13 (a) There is insufficient evidence from one NRCT to determine effects of interventions and programmes targeting vulnerable populations on sexual health-related knowledge and understanding, and personal and social skills.
- 13 (b) There is weak and inconsistent evidence from two NRCT and one RCT on which to determine effects of interventions and programmes targeting vulnerable populations on health and social outcomes relating to sexual health.

PROGRAMMES TARGETING MULTIPLE BEHAVIOURS

No systematic reviews or meta-analyses were identified for inclusion in the review of programmes targeting multiple health behaviours. Five articles were identified that reported on evaluations of programmes and interventions that addressed both alcohol use and sexual health. Two articles reported on studies that examined interventions or programmes delivered in social, healthcare or community settings and three articles reported on studies that examined intervention studies that examined intervented on studies that examined interventions or programmes delivered in social, healthcare or community settings and three articles reported on studies that examined interventions or programmes delivered to families or parents.

Programmes delivered in social, healthcare and community settings

Two studies examined interventions and programmes which targeted both sexual health and alcohol use. One study examined the effects of a sexual activity prevention programme for young people enrolled in Boys and Girls Clubs, which was part of a wider programme designed to prevent substance use, and a second study evaluated an intensive, multicomponent youth development programme. One study was an NRCT and a second study was based on a CBA design. Neither of the included studies examined intervention effects on knowledge and understanding, or on personal and social skills. However, both studies examined intervention effects on attitudes and values. One study of a sexual activity prevention programme found a favourable reduction in sexual attitudes but only among sexually experienced participants who received the intervention without additional booster sessions. The youth development programme had potentially harmful effects on attitudes, with female intervention participants more likely than control participants to report that they expected to be a

parent by age 20. Both studies examined intervention effects on health and social outcomes related to sexual health, and one study also examined intervention effects on alcohol use. The effects of the sexual activity prevention programme were inconsistent across the two intervention conditions examined. The youth development programme had a negative impact on participant's sexual behaviour, particularly among intervention females who were significantly more likely than controls to engage in heterosexual sexual intercourse and more likely to become pregnant. There was no effect of the programme on male participants or on participant's alcohol use.

Evidence statement 14

- 14 (a) There is weak and inconsistent evidence from one NRCT and one CBA study on which to determine the effects of programmes delivered in social and community settings on attitudes and values related to sexual health and alcohol use.
- 14 (b) There is weak and inconsistent evidence from one NRCT on which to determine the effects of programmes delivered in social and community settings that seek to address both sexual health and alcohol use.
- 14 (c) There is weak evidence from one CBA study to suggest that youth development programmes, which target young females at behavioural risk, may have a negative effect on sexual behaviours. This evidence is applicable as the study was conducted in the UK.

Programmes delivered to parents and families

Five studies examined three programmes delivered to families or parents, which targeted both alcohol use and sexual health, in addition to other risk behaviours. All three were universal prevention programmes that combined parenting, youth and family components. All five studies were conducted in the USA and were based on an RCT design. Two studies examined programmes that specifically targeted African American and Hispanic populations, respectively. None of the included studies examined intervention effects on knowledge and understanding. Across four studies that examined intervention effects on attitudes and values towards risky behaviours there were indications of mixed intervention effects. One study found positive long-term effects of both self-directed and group-based versions of a universal substance use and problem behaviour prevention programme on attitudes towards substance use and there were also long-term positive programme effects of a parental monitoring intervention for African American families on attitudes and values related to a range of risky behaviours. Three studies examined intervention effects on personal and social skills, finding mixed programme effects on parent/family-child communication. One study found positive effects of a culturally-tailored programme on communication, family functioning and positive parenting, and a second study found a positive effect of a parental monitoring intervention for African American families on parent-child communication about HIV/AIDS. Four studies examined intervention effects on health outcomes related to alcohol use and sexual health. Short- to medium-term reductions in alcohol drinking were found for African American families, who received a parental monitoring intervention, but this reduction was not sustained and no other significant programme effects were found for health outcomes related to alcohol use. One study of a culturally-tailored programme reported a long-term decrease in incidence rates for STIs and unsafe sex at last sexual intercourse

among Hispanic adolescents who received an additional parent-targeted component. Although, shortterm benefits of a parental monitoring intervention were also reported, these differences were not sustained and over the longer term there were no additional positive effects on sexual behaviour of the parent-targeted intervention among African American adolescents who had received a community-based risk reduction intervention.

Evidence statement 15

- 15 (a) There is mixed evidence from four RCTs regarding the effects of interventions and programmes delivered to families and parents on attitudes and values related to risky behaviours.
- 15 (b) There is moderate evidence from two RCTs to suggest that interventions and programmes delivered to families and parents, and which target alcohol use and sexual health, may improve parent-child communication and family functioning. This evidence may only be partially applicable to the UK as these studies were conducted in the USA and focused on ethnic populations specific to the USA.
- 15 (c) There is moderate evidence from two RCTs to suggest that interventions and programmes delivered to parents and which target alcohol use and sexual health may not provide long-term additional benefits in terms of health and social outcomes related to sexual health and alcohol use beyond those conferred through interventions and programmes which directly target young people. This evidence may only be partially applicable to the UK as these studies were conducted in the USA and focused on ethnic populations specific to the USA.

CONCLUSIONS

The results of this systematic review suggest that programmes and interventions delivered to families may be effective in reducing adolescent alcohol consumption and that group-based sessions and/or skills training programmes in community and healthcare settings may be effective in increasing condom use and reducing the frequency of unprotected intercourse among adolescents. In addition, programmes and interventions delivered to families and parents appeared to be effective in increasing parent-child communication about alcohol use and sexual health. However, the applicability of the evidence identified may not be generalisable to the UK and good quality UK-based research of promising or novel intervention approaches, including assessment of cost-effectiveness, is required in order to build the evidence base on which to make UK-based policy and practice recommendations.

Programmes targeting alcohol use

There was a lack of evidence on which to draw conclusions about the effects of programmes and interventions that targeted adolescent alcohol use on knowledge and understanding. There were positive effects of programmes and interventions delivered to families on attitudes and values related to alcohol use, but programmes and interventions delivered to parents or within social, healthcare and community settings appeared to have no impact on these outcomes. Programmes and interventions delivered to families and parents produced short- and long-term improvements in parent-child communication, and programmes and interventions delivered to families had positive effects on both

alcohol use and initiation of alcohol use. Programme effects on health and social outcomes related to alcohol use were mixed and inconsistent across programmes and interventions delivered to parents, in social, healthcare or community settings, or to the wider community. The family-focused ISFP was highlighted across three systematic reviews as showing particular promise; this programme, which was designed to enhance family protective and resiliency processes and to reduce family-based risk factors associated with child behaviour problems, had positive, long-term effects on a range of outcomes related to alcohol use and has been shown to be cost-effective and potentially cost saving.

Programmes targeting sexual health

The evidence suggests that programmes and interventions delivered in social, healthcare and community settings and to families and parents may have beneficial effects on sexual health-related knowledge in the short- to long-term. A range of outcomes were reported with regards to attitudes and values and programmes effects were mixed across these measures. The evidence suggests that while programme and interventions targeting adolescent sexual health may not impact on attitudes towards sexual intercourse, programme and interventions delivered in healthcare settings may positively impact on condom use attitudes. Programmes and interventions delivered to families and in social, healthcare and community settings had mixed and inconsistent effects on communication, but programmes and intervention delivered to parents appeared to have positive effects on parent-child communication. There appeared to be no effects of programmes and interventions delivered to families and parents on adolescent sexual behaviour, and programmes and interventions delivered in social, healthcare and community settings had limited and inconsistent effects on sexual activity including frequency of intercourse and number of sexual partners. However, the evidence suggests that group-based sessions and/or skills training programmes in community and healthcare settings may increase condom use and reduce the frequency of unprotected sex. In addition, a youth development approach showed promise, with effects on a range of sexual health outcomes for females. There was a lack of evidence on which to draw conclusions about the effects of programmes involving the wider community or mass media or those targeting vulnerable populations.

Programmes targeting multiple behaviours

There was a lack of evidence on which to draw conclusions about the effects of programmes and interventions that targeted multiple behaviours on knowledge and understanding, and there was evidence of mixed and inconsistent effects of these programmes on attitudes and values. Programmes and interventions delivered to parents and families had long-term positive effects on communication, but intervention effects on health and social outcomes related to sexual health were less clear. There was no evidence supporting the effectiveness of programmes and interventions delivered in social, healthcare and community settings and interventions and programmes delivered to parents did not appear to provide additional long-term benefits beyond those conferred through intervention and programmes which directly targeted young people.

1 Introduction

1.1 Aims and objectives

The aim of the review was to identify effective and cost-effective interventions and programmes that address health literacy and personal skills in relation to alcohol use and sexual health. This was defined as alcohol education and/or sex and relationships education (SRE) delivered in isolation or as part of a wider programme in family, social, healthcare and community settings.

1.2 Research question

The review aimed to address the following key research questions:

- 1) What services, interventions, programmes, policies or strategies for children and young people aged 5 years and above are effective and cost-effective in contributing to the achievement of the "Every Child Matters" outcomes related to sexual health and alcohol?
- 2) What elements/components of those services, interventions, programmes, policies or strategies for children and young people aged 5 years and above are effective and costeffective in contributing to the achievement of the "Every Child Matters" outcomes related to sexual health and alcohol?

2 Background

Interventions aiming to prevent, delay or reduce risk-taking behaviours are delivered at several different levels: individual, community and population. However, regardless of the level at which interventions are delivered their effects are rarely limited to just one level (NICE, 2007). This systematic review examined the effectiveness of both alcohol and sexual health interventions delivered in a community setting. Separate from school-based interventions, community-based interventions may include, for example, interventions targeted at families, parents, and young people outside school in after school clubs or youth clubs. Community-based interventions are used to reach young people who may not be in education or training and are also used to target vulnerable groups in the wider community, for example, such young homeless people. The need for interventions involving community-based and outreach initiatives for the prevention of sexual health and the prevention of alcohol use has been previously recognised (MedFASH, 2008)

Young people aged 16-19 years report that lessons at school are their primary source of sexual health information. However, females report parents to be their preferred source of information and for males, parents are reported a close second to school lessons (33.3% compared to 34.4% respectively). The difference between actual sources of information and preferred sources of information is most notable in males; with 1 in 12 reporting parents as their main source of information, compared to 1 in 3 who would prefer their main source of information to be their parents (Wellings et al., 2006). Often difficulty in addressing sensitive issues and a lack of parental communication skills can contribute to an inability for parents and children to openly discuss topics such as alcohol and sex. Poor family relationships and poor parental support have been highlighted as contributing risk-factors for teenage pregnancy (DfES, 2006). Strong family bonds, parental monitoring and family rules have also been cited as important contributing factors to prevention interventions (DfES, 2004) (see Table 2.1).

As young people develop, their primary source of education and information regarding, for example, sexual behaviour and alcohol use can move from parents, to school, to peers, particularly during adolescence and the need to adhere to peer norms can conflict with or override previous influential norms. At this developmentally vulnerable time females with parents who drink show higher rates of initial alcohol use (Duncan et al., 2006). Young people's drinking is predicted more by a mother's drinking behaviour than a father's drinking behaviour, indicating that young peoples' drinking is influenced through modelling (White et al., 2000). Further research has shown that being white, being from a single-parent family, having deviant peers and friends who encourage alcohol use predicts increased rates of alcohol use from age 9-16 years (Duncan et al., 2006). Strict parental rules about drinking alcohol have been shown to have a protective influence on frequent or heavy consumption of alcohol in young people and to delay the initiation of alcohol use (van der Vorst et al., 2005; van der Vorst et al., 2006). However, if parents are permissive regarding, for example, early alcohol use or if they engage in heavy alcohol use themselves then their children are more likely to tend towards early or heavy drinking (van der Vorst et al., 2009).
Similarly, young white teenagers who report sexually permissive peer norms, perceived peer approval of teenage sexual behaviour, perceived peer sexual behaviour and greater levels of sexual communication with peers have a higher susceptibility and higher odds of initiating sexual intercourse (L'Engle & Jackson, 2008). Other studies also support the findings that normative peer sexual behaviour predicts sexual initiation in young teenagers and show that perceived peer values are more strongly related to sexual initiation than actual peer reports of attitudes and values (Sieving et al., 2006). Findings also show that the media can act as a sexual socialising agent for young teenagers (L'Engle & Jackson, 2008) and be a super peer in terms of sexual influences, giving the impression to young girls that early sexual behaviour is acceptable (Brown et al., 2005). Moderating factors for sexual initiation, similar to those from alcohol studies, are stronger parent-child relationships, particularly with the mother. In addition, parental monitoring and strong links to school (e.g. feeling good about being in school, expecting to finish school and go on to further education) can have protective effects.

Although education on alcohol and sexual health features in school curricula, the contribution and support of parents and carers is vital to ensure that consistent and accurate public health messages are conveyed to young people to prevent risk-taking behaviour and facilitate behaviour-change (DfES, 2004; DCSF, 2008). This also includes the provision of parenting support to develop improved communication (DfES, 2006). An example of a well received programme for parent training reported in the UK is 'Speakeasy' which is a community-based education programme solely targeting parents. The aim of the programme is to help parents communicate well with their children about sex, sexual health and relationships by increasing parental knowledge and self-efficacy through group work (Ramm & Coleman, 2008). However, this programme is yet to be evaluated using a control group or with the inclusion of feedback from young people.

Early intervention is necessary, particularly in cases where young people are experiencing behavioural, emotional or social difficulties, which are risk factors for alcohol use and sexual risk-taking. As such parenting programmes are being implemented to help support parents and promote child well-being which could subsequently affect alcohol and sexual behaviour. Parent early intervention pathfinder programme, a DCSF funded programme primarily aimed at addressing antisocial behaviour, recommends a national roll-out of parenting programmes for parents with children aged 8-13 years (Lindsay et al., 2008). Programmes such as those recommended in the pathfinder programme (Strengthening Families, Strengthening Communities and Incredible Years) may precede and complement parenting programmes such as Speakeasy.

A whole school approach to education goes further and emphasises that consistent messages to those taught in schools ought to be conveyed not only through parents and family members but within the wider community also (DfES, 2004). Educational interventions delivered in the community also include media campaigns such as 'want respect, use a condom' and 'know your limits'. In addition community interventions can be supported by population level interventions such as that included in the Alcohol Harm Reduction Strategy to changing labels on alcohol, provide information at point of

sale, review advertising regulations and ensure that alcohol is not used positively in advertisements (The Strategy Unit, 2004).

Vulnerable Groups	Risk Factors	Protective Factors
Homeless	Chaotic home environment	Strong family bonds
Looked after	Parents who misuse drugs or	Experiences of strong
School truants	suffer from mental illness	parental monitoring with clear
Pupils excluded from schools	Behavioural disorders	family rules
Sexually abused	Lack of parental nurturing	Family involvement in the
Prostitutes	Inappropriate or aggressive	lives of children
In contact with mental health or	classroom behaviour	Successful school
criminal justice system	School failure	experiences
Children of parents with drug	Poor coping skills	Strong bonds with local
problems	Low commitment to school	community activities
	Friendship with deviant peers	A caring relationship with at
	Low socio-economic status	least one adult
	Early age of first drug use	
	Being labelled as a drug misuser	

Table 2.1. Risks and protective factors associated with drug¹ misuse.

(Source: The Right Responses – Managing and making policy for drug-related incidents in schools [Drugscope, 1999], taken from DfES, 2004)

¹ 'Drugs' refers to all drugs including medicines, volatile substances, alcohol, tobacco and illegal drugs.

3 Methodology

3.1 Search strategy

Systematic searches of electronic databases and websites were undertaken to identify studies that examined the effectiveness and/or cost-effectiveness of alcohol education and/or SRE delivered in community settings in isolation or as part of a wider programme of study such as PSHE or its equivalents. Searches were conducted across a range of health, education and social care databases as shown in Box 3.1.

Box 3.1. Health, education and social care databases

- ASSIA (Applied Social Science Index and Abstracts)
- CINAHL (Cumulative Index of Nursing and Allied Health Literature)
- Database of Abstracts of Reviews of Effectiveness (DARE)
- The Cochrane Library
- EMBASE
- ERIC
- British Education Index
- Australian Education Index
- HMIC
- MEDLINE
- PsycINFO
- Sociological Abstracts
- Social Science Citation Index
- EPPI Centre databases
- The Campbell Collaboration
- C2-SPECTR & C2-PROT Campbell Collaboration

Economic evaluation studies were identified by searching the following major health economics databases:

- NHS Economic Evaluations Database (NHS EED)
- EconLit

3.2 Inclusion and exclusion criteria

3.2.1 Population

Studies were eligible for inclusion if they included children aged 5 to 19 years old in community and outreach settings.

Studies were eligible for inclusion if they were undertaken in the UK, Western Europe, Australia, New Zealand, Canada and the USA.

3.2.2 Interventions

Studies were eligible for inclusion if they examined interventions that focused on SRE and/or alcohol education. Relevant intervention approaches included:

- Interventions and programmes delivered within social, healthcare or community settings
- Interventions and programmes delivered to families or parents
- Intervention and programmes involving the wider community or mass media

3.2.3 Comparator(s)

Studies were eligible for inclusion if they compared the intervention of interest against a no intervention control or against another intervention approach.

3.2.4 Outcomes

Studies were eligible for inclusion only if they examined the primary outcomes of interest:

- Health and social outcomes relating to alcohol use and sexual health
- Personal and social skills

The following secondary outcomes were assessed but only where a study reported a primary outcome of interest:

- Knowledge and understanding
- Attitudes and values

3.2.5 Study design

Systematic reviews, meta-analyses, randomised controlled trials, controlled non-randomised studies and controlled before and after studies that compared a community-based intervention against no intervention or another type of intervention were eligible for inclusion in the assessment of effectiveness.

Studies were eligible for inclusion in the assessment of cost-effectiveness if they were economic evaluations conducted alongside trials, modelling studies and analyses of administrative databases. Only full economic evaluations that compared two or more options and considered both costs and consequences (including cost-effectiveness, cost utility and cost-benefit analyses) were included.

3.3 Data extraction strategy

All titles and abstracts retrieved were screened by one reviewer (LJ, GB and JD) according to the inclusion/exclusion criteria described above. Relevant articles were retrieved in full, and full text screening was undertaken independently by two reviewers (LJ, GB, JD, MW, OW, KS and AK). Disagreements were resolved through consensus and where necessary a third reviewer was consulted.

One reviewer (LJ, GB and JD) independently extracted and assessed the quality of the individual studies into an Access database. All data extraction and quality assessment were independently checked for accuracy by a second reviewer. The results of the data extraction are presented in an addendum to this report.

3.4 Quality assessment strategy

The quality of the studies was assessed according to criteria set out in the NICE Centre for Public Health Excellence Methods Manual (2009). Each of the effectiveness and cost-effectiveness studies was graded using a code, ++, + or – based on the extent to which the potential sources of bias had been minimised:

- ++ All or most of the criteria have been fulfilled. Where they have not been fulfilled the conclusions are thought very unlikely to alter.
- + Some of the criteria have been fulfilled. These criteria that have not been fulfilled or not adequately described are thought unlikely to alter the conclusions.
- Few or no criteria have been fulfilled. The conclusions of the study are thought likely or very likely to alter.

Results of the quality assessment are presented in Appendix 4 and 5.

3.5 Methods of analysis/synthesis

3.5.1 Effectiveness studies

The results of the data extraction and quality assessment for each study of effectiveness are presented in structured tables and as a narrative summary. The possible effects of study quality on the effectiveness data and review findings are also discussed within the text of the review.

Studies are grouped according to (1) focus (alcohol or sex and relationships education) and (2) setting (social, health or community; family; parent; or community-wide or mass-media). Where reported in the original publications, effect sizes (e.g. odds ratios, Cohen's d) are presented. Where effect sizes were not reported significant (p<0.05; p<0.01; p<0.001) and non-significant changes in outcomes of the intervention(s) relative to the comparison group are presented.

Where sufficient data are available, intervention effect sizes will be calculated and presented as odds ratios (OR) for dichotomous data and as mean differences for continuous data in an addendum to this report to be prepared for the PDG meeting in February. Forest plots will be generated for single studies using RevMan (version 5). Heterogeneity between the included studies was assessed by considering differences in (a) the study population, (b) intervention approach, (c) outcome measures, and (d) study quality. However, given the anticipated heterogeneity between the included studies it was judged to be unlikely that pooling would be appropriate or feasible.

3.5.2 Published economic evaluations

Details of each identified published economic evaluation, together with a critical appraisal of its quality were presented in structured tables and as a narrative summary. For economic studies conducted alongside trials, the validity of the included studies was assessed by considering the source of the resource use and effectiveness data, the methods used to measure and calculate costs, the methods of analysis used and the generalisability of the results to the UK population.

4 Summary of study identification

4.1 Review of effectiveness and cost-effectiveness

A total of 12,108 references were identified from the literature searches. Following screening of titles and abstracts, 531 articles were identified as potentially relevant and attempts were made to source the full text articles. Of these articles, 91 were not available. These studies were therefore not subject to further screening and a total of 440 full-text articles were screened against the inclusion criteria for the study. The process of study selection is summarised in Figure 1.



Figure 4.1. Process of study selection

4.1.1 Included studies

A total of 87 studies met the criteria for inclusion in the review of effectiveness and cost-effectiveness. Of these, 31 studies examined interventions targeting alcohol use, 51 studies examined interventions targeting sexual health and the remaining five studies examined interventions which targeted both alcohol use and sexual health. Thirteen articles were systematic reviews and/or meta-analyses, 62 studies were based on experimental designs of which 53 used random assignment to allocate participants to intervention and comparison conditions. Eleven observational studies were identified for inclusion including nine controlled before and after (CBA) studies, and two cross-sectional time series (CTS). One economic evaluation study was also identified. A summary of the study designs identified for inclusion in the review is summarised on Table 4.1.

	-	-							
Section	Total	SR/MA	RCT	NRCT	СВА	CTS	Economic evaluation		
Alcohol use	31	4	21	1	4	-	1		
Sexual health	49	9	27	7	4	2	-		
Multiple behaviours	7	-	5	1	1	-	-		
Total	87	13	53	9	9	2	1		
SR/MA - systematic review or meta-analysis; RCT - randomised controlled trial; NRCT -									
nonrandomised contro	lled trial; C	BA – contro	olled before	and after; (CTS – cross	s sectional t	time series		

Table 4.1. Summary of study designs identified from inclusion

4.1.2 Excluded studies

A total of 353 articles did not meet the criteria for inclusion in the review and were excluded for the following reasons:

- Study design did not meet design criteria for inclusion in the review, n=213
- Population targeted by the intervention(s) did not meet the review criteria, n=75
- Intervention examined was not based in a relevant setting, n=53
- Intervention or intervention was not alcohol education or SRE related, n=9
- Duplicates or foreign language, n=3

References for the excluded studies are presented in Appendix 2.

5 Programmes targeting alcohol use

A total of 31 articles met the criteria for inclusion in the review of community-based programmes targeting alcohol use by young people. Four articles were systematic reviews and/or meta-analyses, four articles reported on studies that examined intervention or programmes delivered within social, healthcare and community settings, 19 articles reported on studies that examined programmes or interventions delivered to families or parents, and three studies examined interventions or programmes that involved the wider community or mass media. One economic evaluation study was also identified that examined the cost-effectiveness and cost-benefits of the Iowa Strengthening Families Programme (ISFP) and Preparing for the Drug Free Years (PDFY).

5.1 Systematic reviews and meta-analyses

5.1.1 Overview of evidence identified

Four systematic reviews were identified for inclusion that examined community-based interventions and programmes targeting alcohol use among young people. Two publications of the Cochrane review by Foxcroft and colleagues were identified (Foxcroft et al., 2002; 2003). This review sought to identify and summarise rigorous evaluations of psychosocial and educational interventions aimed at the primary prevention of alcohol misuse by young people. One review (Petrie et al., 2007) examined the effectiveness of parenting programmes to prevent tobacco, alcohol or drug abuse in children, and the fourth article identified (Smit et al., 2008) aimed to quantify the effectiveness of family interventions in reducing adolescent drinking through a meta-analysis of RCTs. All three reviews focused primarily on the inclusion of RCTs.

5.1.2 Quality assessment

All three reviews were rated good quality. They addressed appropriate and clearly focused questions and a good description of the methodology used to conduct the reviews was reported. All three reviews were based on rigorous searches of the literature and assessed study quality. The synthesis of study data was undertaken appropriately across all three reviews.

5.1.3 Findings

Of the community-based studies reviewed by Foxcroft and colleagues (2002; 2003), the ISFP was highlighted as showing particular promise over the long-term. The authors conducted an intention-to-treat reanalysis, reporting a number needed to treat (NNT) of 9 for this programme at the four-year follow-up. This indicates that for every 9 individuals who receive the intervention, there will be one fewer person reporting that they have ever used alcohol (NNT 9; 95% CI 5, infinity), used alcohol without permission (NNT 9; 95% CI 5, 160), or ever been drunk (NNT 9; 95% CI 5, 327).

Petrie and colleagues (2007) reported that the strongest evidence supported interventions targeting preteen and early adolescent children. Parenting programmes highlighted as effective included the Iowa Strengthening Families Program (ISFP) and Preparing for the Drug Free Years (PDFY). The authors reported that effective interventions focused on developing strategies to involve adolescents

in family activities, in order to maintain familial bonds and manage conflict. Effective intervention also placed an emphasis on parental engagement in an activity-based programme.

Smit and colleagues (2008) included nine RCTs in their meta-analysis, the main findings of which suggested a favourable effect of family-based interventions on alcohol initiation (OR 0.71; 95% CI 0.54, 0.94) and frequency of alcohol use (Cohen's d -0.25; 95% CI -0.37, -0.12) in adolescents. However there was evidence of heterogeneity across the pooled studies on the measure of alcohol initiation. Longitudinal analyses conducted by the authors pointed to the success of the ISFP and PDFY.

5.1.4 Summary and evidence statements

Three systematic reviews and meta-analyses were identified for inclusion. One review (Foxcroft et al., 2002; 2003) examined interventions and programmes aimed at the primary prevention of alcohol use across a range of populations and settings. Two further reviews (Petrie et al., 2007; Smit et al., 2008) examined interventions and programmes delivered to parents and families, respectively.

Foxcroft et al (2002; 2003) found that although there was no consistent evidence to determine which programmes were effective over the short to medium-term, one family-based programme, the Strengthening Families programme, was effective over the longer term. The reviews by Petrie and colleagues (2007) and Smit and colleagues (2008) also highlighted the long-term effectiveness of the Strengthening Families programmes.

Evidence statement 1

The is strong evidence from three systematic reviews¹ to suggest that a family-based programme, Strengthening Families, can produce long-term reductions (greater than 3 years) in alcohol use and heavy alcohol use.

¹ Foxcroft et al., 2002; 2003 (SR ++); Petrie et al., 2007 (SR ++); Smit et al., 2008 (SR ++)

Author (year)	Design	Inclusion/exclusion	Number of studies	Findings
Foxcroft et al (2002; 2003)	SR ++	Psychosocial and educational interventions aimed at the primary prevention of alcohol misuse by young people aged up to 25 years	56 studies	Twenty studies demonstrated evidence of ineffectiveness. No firm conclusions about the effectiveness of prevention in the short and medium-term were possible. But over the longer term (>3 years), the Strengthening Families Programme showed more promise as an effective prevention intervention.
Petrie et al (2007)	SR ++	Parenting programmes to prevent tobacco, alcohol or drug abuse in children	20 studies	Strongest evidence related to interventions and programmes that had been undertaken with preteen and early adolescent children. Effective interventions focussed on developing strategies to involve adolescents in family activities to maintain familial bonds and manage conflict. Also, an emphasis on parental engagement in an activity-based programme.
Smit et al (2008)	SR ++	Family interventions that focused on reducing adolescent drinking.	18 studies	Main findings pointed to a favourable effect of family interventions on alcohol initiation and frequency of alcohol consumption among young people. The effects were maintained over time. Studies that examined group-based interventions and programmes tended to report a stronger intervention effect that interventions targeting individual families.

Table 5.1. Summary of findings from systematic reviews and meta-analyses: Programmes targeting alcohol use

5.2 **Programmes delivered within social or community settings**

5.2.1 Overview of evidence identified

Three studies (Elder et al., 2002; Schinke et al., 2005; Tebes et al., 2007) were identified that examined programmes delivered within social, healthcare or community settings, which targeted alcohol use among young people. These studies were conducted within youth and after school agencies and all three were conducted in the USA. Schinke et al (2005) examined an interactive CD-ROM intervention designed to reduce early alcohol use, and Elder et al (2002) and Tebes et al (2007) examined substance use prevention programmes which targeted migrant families, and adolescents enrolled in after school programmes, respectively.

The theoretical basis for intervention was not reported in two of the three studies. The interactive CD-ROM intervention (Schinke et al., 2005) was based on a combination of theories, including social cognitive theory, problem-behaviour theory, peer-cluster theory and family networks theory.

The number of participants recruited across the included studies ranged from 304 (Tebes et al., 2007) to 660 (Elder et al., 2002) students. One study (Schinke et al., 2005) examined interventions that targeted children aged 10-12 years, and another (Tebes et al., 2007) examined an intervention that targeted students with a mean age of 15 years. The study by Elder et al. (2002) did not specify the age range of the adolescents targeted in their study. One study (Elder et al., 2002) reported long-term follow-up data (>12 months); the study by Tebes and colleagues (2007) reported 12-months of follow-up and Schinke and colleagues (2005) reported immediate post-test results only.

5.2.2 Quality assessment

Of the three studies identified, two were cluster RCTs (Elder et al., 2002; Schinke et al., 2005) and one was based on CBA design (Tebes et al., 2007). The RCT by Elder and colleagues (2002) was rated moderate. The authors did not report the number of participants assigned to the intervention and control groups, although other aspects of the study were adequately reported. The RCT by Schinke and colleagues (2005) was rated poor quality as insufficient information was reported to determine whether the analyses were conducted appropriately and whether the outcomes measures were reliable. The quality of the CBA study by Tebes and colleagues (2007) was rated moderate. The intervention and comparison conditions were well described and appropriate, contamination was acceptably low, and all important and relevant outcomes were examined.

5.2.3 Findings

5.2.3.1 Knowledge

None of the included studies examined intervention effects on knowledge.

5.2.3.2 Attitudes and values

Three studies (Elder et al., 2002; Schinke et al., 2005; Tebes et al., 2007) examined intervention effects on alcohol and substance-related attitudes. Elder and colleagues (2002) found that there were

no effects of the 'Sembrano Salud' programme, which targeted migrant families, on participant's susceptibility to drinking. The CD-ROM intervention, Thinking not Drinking, examined by Schinke and colleagues (2005) had a positive impact on participant's perception of the harms of alcohol at post-test in comparison to a 'no intervention' control group (p<0.05). A short-term positive intervention effect on participants' perception of the harms of alcohol was also reported in the study by Tebes and colleagues (2007) that examined an after-school, youth development programme, Adolescent Decision-Making for the Positive Youth Development Collaborative (ADM-PYDC), designed to prevent substance use. At post-test, intervention participants reported an increased perception of risk of harm compared with the control group (p<0.01), but there was no difference on this measure at the 12-month follow-up. In addition, there was no significant difference between the intervention and control group in their attitudes towards drugs over follow-up.

5.2.3.3 Personal and social skills

Only one study (Schinke et al., 2005) examined intervention effects on personal and social skills. At immediate post-test, participants who received an interactive CD-ROM intervention scored more positively on the measure of assertion skills compared to their control counterparts (p<0.001).

5.2.3.4 Health and social outcomes related to alcohol use

All three studies examined intervention effects on health and social outcomes related to alcohol use. There were no effects of the interactive CD-ROM intervention (Schinke et al., 2005) on participants' alcohol use at post-test, but the authors reported that the frequency of substance use was low among the study participants, who were a median age of 11 at baseline. Frequency of alcohol prevalence was also reported to be low among those who participated in the evaluation of Sembrano Salud (Elder et al., 2002) and there was no difference between intervention and control groups in terms of 30-day drinking at any follow-up (OR 1.21; 95% CI 0.74, 1.97). The after-school, youth development programme examined by Tebes and colleagues (2007) was found to have had a positive effect on alcohol use. Between baseline and the 1-year follow-up, reductions in alcohol use were found to be significantly greater among the intervention group relative to the control group (OR 0.37; 95% CI 0.15-0.90).

5.2.4 Summary and evidence statements

Three studies were identified for inclusion that examined interventions and programme targeting alcohol use, which were delivered in social, healthcare and community settings. All three studies (Elder et al., 2002; Schinke et al., 2005; Tebes et al., 2007) were conducted within youth and after school agencies and were based in the USA.

None of the studies examined intervention effects on knowledge and understanding. Short-term increases in perception of harm were reported in two studies (Schinke et al., 2005; Tebes et al., 2007), but this effect was not sustained at the 12-month follow-up in the study by Tebes and colleagues (2007). Tebes and colleagues (2007) also found no impact of an after-school, youth development programme on participants' drug beliefs and there was no impact of Sembrano Salud (Elder et al., 2002) on participants' susceptibility to alcohol. One study (Schinke et al., 2007) examined intervention

effects on personal and social skills, finding a short-term intervention impact of an interactive CD-ROM intervention on assertion skills.

Two studies (Elder et al., 2002; Schinke et al., 2005), conducted within youth agencies, reported that there were no intervention effects on health and social outcomes related to alcohol use, and that substance use remained low among participants. However, one study (Tebes et al., 2007), which targeted older children (mean age 15 years) in after school programmes reported a positive short- to medium-term effect on alcohol use.

Evidence statement 2

- 2 (a) There is inconsistent evidence from two RCTs and one CBA study¹ to determine the effects of interventions and programmes delivered in social, healthcare and community settings on attitudes and values related to alcohol use.
- 2 (b) There is inconsistent evidence from two RCTs and one CBA study¹ to determine the effects of interventions and programmes delivered in social and community settings on alcohol use. However, there is weak evidence from one CBA study² to suggest that programmes that target older children may impact on alcohol consumption. Findings may only be partially applicable to the UK as the study was conducted in the USA and may not be generalisable beyond the populations studied.

¹ Elder et al., 2002 (RCT +); Schinke et al., 2005 (RCT -); Tebes et al., 2007 (CBA +)

² Tebes et al., 2007 (CBA +)

Table 5.2. Summary of programme content: programmes	delivered within social, healthcare or community settings
-----------------------------------------------------	-----------------------------------------------------------

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Elder et al., 2002	RCT (cluster) +	USA n=660 Majority Mexican age NR	School (evenings)	Sembrano Salud : Eight weekly, 2-hour sessions: presentation of information, modelling and behavioural rehearsal; developing parental support through enhanced parent-child communication. Additional components were telephone booster calls and three newsletters.	NR	Mexican American group leaders
Schinke et al., 2005	RCT (cluster) -	USA n=489 54% African American; 30% Hispanic, 11% White; 5% other 10-12 years	Youth agencies	Thinking Not Drinking: Ten weekly, 45 minute sessions. Interactive CD-ROM; goal setting, coping, media literacy, peer pressure, assertiveness training and preventive strategies	Social cognitive theory, problem- behaviour theory, peer-cluster theory and family networks theory	CD-ROM
Tebes et al., 2007	CBA +	USA n=304 76% African American; 20% Hispanic; 4% White; <1% other mean 15 years	Youth agencies	Adolescent Decision-Making for the Positive Youth Development Collaborative (ADM-PYDC): 18 sessions; understanding and coping with stress, decision-making, information about tobacco, alcohol and drugs, and applying decision-making.	NR	Community group leaders

			Comparator		Outcomes			
Study	Rating	Intervention		Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills	
			First aid/home safety n=NR	PT (97%)	-	NS susceptibility to drinking	-	
Elder et al., RCT 2002 (cluster	RCT (cluster) +	Sembrano Salud n=NR		1 yr (89%)	-	NS susceptibility to drinking	-	
				2 yr (81%)	-	NS susceptibility to drinking	-	
Schinke et al., 2005	RCT (cluster) -	Thinking Not Drinking n=329	No intervention n=160	PT (100%)	-	† perceived harm of alcohol*	1 assertion skills***	
Tebes et al., 2007	CBA +	ADM-PYDC n=149	Other after- school activities n=155	PT (NR)	-	f perception of risk of harm** NS drug beliefs	-	
				1 yr (Int 62%; Con 58%)	-	NS perception of risk of harm NS drug beliefs	-	
*p<0.05; **p<0.0)1; ***p<0.001;	p value not reported	d; ↑ increase relati	ve to comparat	tor; ↓ decrease relative to co	mparator; NS not significant; - ou	itcome not reported	

Table 5.3. Programmes	delivered in social, healthcare and c	community settings: effects on ki	nowledge, skills and attitudes
-----------------------	---------------------------------------	-----------------------------------	--------------------------------

Cturdy	Dating	la temperati em	Commenter	Follow up	Health and social outcomes			
Study	Rating	Intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other	
Elder et al., 2002		Sembrano Salud n=NR	First aid/home safety educational programme n=NR	PT (97%)	NS 30-day drinking	-	-	
	RCT (cluster) +			1 yr (89%)	NS 30-day drinking	-	-	
				2 yr (81%)	NS 30-day drinking	-	-	
Schinke et al., 2005	RCT (cluster) -	Thinking Not Drinking n=329	No intervention n=160	PT (100%)	NS alcohol use	-	-	
Tebes et al., 2007	CBA +	ADM-PYDC n=149	Other after- school activities n=155	1 yr (Int 62%; Con 58%)	↓ alcohol use [¶]	-	-	
*p<0.05; **p<0.0	01; ***p<0.001;	p value not reported	d; 1 increase relati	ve to comparat	tor; ↓ decrease relative to com	nparator; NS not significant; - ou	itcome not reported	

Table 5.4. Programmes delivered in social, healthcare and community settings: effects on health and social outcomes related to alcohol use

5.3 Programmes delivered to families

5.3.1 Overview of evidence identified

Fifteen articles were identified that evaluated nine programmes delivered to families, which focused on alcohol or substance use, all of which were implemented in the USA. Three programmes, evaluated in three studies (Loveland-Cherry et al., 1999; Schinke et al 2004; Schinke et al., 2009), were alcohol use prevention interventions while the remaining seven programmes, examined in twelve studies (Bauman et al., 2000; Brody et al., 2004; 2006; Gerrard et al., 2006; Johnson et al., 1996; Jones et al., 2005; Mason et al., 2009; Murry et al., 2007; Spoth et al., 1999; Spoth et al., 2001; 2004; Stevens et al., 2002), focused on substance use prevention, including alcohol. Four studies (Spoth et al., 1999; 2001; 2004; Mason et al., 2009) reported on the large-scale evaluation of two brief, family-focused programmes, the ISFP and PFDY. Two papers evaluated participants from both programmes (Spoth et al., 2001; 2004), one paper evaluated ISFP participants only (Spoth et al., 1999) and one paper provided a long-term evaluation of the PDFY participants only (Mason et al., 2009). Four studies (Brody et al., 2004; 2006; Gerrard et al., 2006; Murry et al., 2007) reported on an evaluation of the Strong African American Families programme, which targeted parenting practices. Two studies (Jones et al., 2005; Stevens et al., 2002) reported on the Dartmouth Prevention Project that examined the effects of two physician-led interventions delivered in a primary care setting, one of which sought to prevent early drinking and smoking.

Interventions took place in settings including at home, primary care settings and community centres; although for the majority of studies the setting was not detailed. Similarly, the person or persons providing the interventions was poorly reported. Three studies (Spoth et al., 1999; 2001; 2004) examined interventions based on video presentations and two interventions (Schinke et al., 2004; Schinke et al., 2009) were delivered via CD-ROM.

Eight studies (Bauman et al., 2000; Loveland-Cherry et al., 1999; Mason et al., 2009; Schinke et al., 2004; 2009; Spoth et al., 1999; 2001; 2004) reported the theoretical basis for intervention. The ISFP and PDFY programmes (Spoth et al., 1999; Spoth et al., 2001; Mason et al., 2009) were based upon the biopsychosocial model and social development model, respectively. For two CD-ROM based interventions (Schinke et al., 2004; Schinke et al., 2009), a CD-ROM intervention with and without parental involvement component drew on social learning theory, problem behaviour theory and family interaction theory, and a gender-specific, computer mediated intervention was informed by family interaction theory. Bauman and colleagues (2000) reported a combination of social and health theories including the health belief model, social learning theory and social inoculation theory in the development of materials for the Family Matters programme and the alcohol use prevention intervention examined by Loveland-Cherry and colleagues (1999) was based on theories of social cognition and problem behaviours.

The number of participants taking part in studies ranged from 120 adolescents and their parents (Johnson et al., 1996) to 3,111 families (Stevens et al., 2002). Young people taking part in programmes were generally in early adolescence, aged 11-15, with the exception of one alcohol

misuse prevention intervention (Loveland Cherry, 1999) that was implemented in the 4th grade (aged 9-10 years). Programme evaluations varied in length. One programme (Schinke et al., 2009) was based on short-term follow-up only, but all other studies included a medium- or- long term evaluation. Long-term evaluations (>12 months follow-up) were provided for the following programmes: SAAF (Brody et al., 2004; 2006), PDFY (Spoth et al., 2001; 2004; Mason et al., 2009), ISFP (Spoth et al., 2001; 2004), physician-led primary care-based intervention (Jones et al., 2005; Stevens et al., 2002), an alcohol misuse prevention intervention (Loveland-Cherry, 1999) and a CD-ROM intervention (Schinke et al., 2004).

5.3.2 Quality assessment

All 15 studies identified were RCTs and included three studies that were based on individual randomisation (Bauman et al., 2000; Loveland-Cherry 1999; Schinke et al., 2009). The remaining twelve studies were based on cluster randomisation at the level of county (SAAF: Brody et al., 2004; 2006; Gerrard et al., 2006; Murry et al., 2007); family (Johnson et al., 1996); clinic (Jones et al., 2005; Stevens et al., 2002); school (ISFP/PDFY: Mason et al., 2009; Spoth et al., 1999; 2001; 2004) or community site (Schinke et al., 2004). None of the RCTs were rated good quality and three were rated poor quality (Bauman et al., 2000; Johnson et al., 1996; Loveland-Cherry, 1999), with the remainder rated moderate quality. All 12 studies described allocation to intervention and control groups as randomised but failed to detail further methods of randomisation and concealment of allocation was only detailed in the evaluations of the ISFP and PDFY (Mason et al., 2009; Spoth et al., 1999; 2001; 2004). Studies were generally rated as good or moderate for quality relating to outcome measures but two studies did not report on reliability of outcomes (Bauman et al., 2000; Loveland-Cherry et al., 1999) and three studies were rated as not assessing all important outcomes (Bauman et al., 2000; Johnson et al., 1996; Jones et al., 2005). The study by Stevens et al (2002) was generally a good quality study but the authors did not discuss attrition although it was clear that not all participants were followed up. The study was therefore rated moderate. Outcomes across all studies were deemed relevant and only one study (Schinke et al., 2009) was rated moderate for length of follow-up time. However, the length of follow-up was generally good, with medium to long-term followup results across all other studies. One study (Loveland-Cherry et al., 1999) did not report on baseline comparability between groups but this was undertaken in the remaining studies. Analytical methods were poorly reported in two studies (Bauman et al., 2000; Johnson et al., 1996) where effect sizes for outcomes were not presented or calculable and levels of significance not presented for all outcomes. In seven studies analytical methods were generally rated as good (Brody et al., 2001; 2004; Mason et al., 2009; Schinke et al., 2004; Spoth et al., 1999; 2001; 2004).

5.3.3 Findings

5.3.3.1 Knowledge and understanding

One study (Johnson et al., 1996) examined intervention effects on knowledge and understanding. Johnson and colleagues (1996) reported a positive intervention effect (p<0.001) on knowledge of

alcohol and other drugs at post-test and the 12-month follow-up, among those who participated in the community-based, Creating Lasting Connections programme.

5.3.3.2 Attitudes and values

Measures relating to attitudes and values were reported in evaluations of three programmes: SAAF (Brody et al., 2001; 2004); a substance misuse prevention intervention (Jones et al., 2005) and a CD-ROM based intervention (Schinke et al., 2009). Brody and colleagues (2004; 2009) reported finding positive post-test effects of the SAAF on youth protective factors (p<0.05) that included resistance efficacy, goal-directed future orientation, images of drinkers and negative attitudes about sex and alcohol, but these effects were non-significant at the long-term follow-up. Short-term evaluation of a CD-ROM intervention for girls and their mothers (Schinke et al., 2009) suggested positive effects of the intervention on girls' beliefs about underage drinking, intentions not to drink and self-efficacy to avoid alcohol (all p<0.05) compared to control participants. Jones and colleagues (2005) evaluated a physician-led primary care-based intervention. At the 3-year follow-up, the authors found a negative association between intervention group boys and externalizing problem behaviour (p<0.01). Effects for girls were non-significant, as were effects on internalising of problems.

5.3.3.3 Personal and social skills

For four programmes, SAAF (Brody et al., 2001; 2004), Creating Lasting Connections (Johnson et al., 1996) and two CD-ROM based interventions (Schinke et al., 2004; 2009), measures relating to personal and social skills were reported. Brody and colleagues (2004) reported a positive effect of the SAAF on communicative parenting (p<0.05) at post-test, but the effect was no longer significant at long-term follow-up. Two CD-ROM based interventions (Schinke et al., 2004; 2009) produced positive effects on skills outcomes. Schinke and colleagues (2009) reported positive short-term intervention effects on a range of measures including mother and daughter reported parental rules, daughter's reported parental monitoring, family conflict management skills and communication with mother (all p<0.05). Mother-reported communication with daughter and parental monitoring did not significantly differ between intervention and control groups. Long-term evaluation of a CD-ROM intervention with and without a parent involvement component (Schinke et al., 2004) revealed positive intervention effects on levels of family involvement (p<0.05) among those who received the additional parental component, and a positive effect on peer influence outcomes (p<0.001). Johnson and colleagues (2006) reported non-significant effects of the Creating Lasting Connections programme on family communication, and youth- and parent-reported levels of bonding at post-test and long-term follow-up. However, there were positive programme effects on youth involvement in the setting up of rules about alcohol and other drugs (p<0.001) at post-test. This effect was no longer significant by long-term follow-up, and no significant intervention effects were found on the existence of rules about alcohol and other drugs or for non-substance use behaviours.

5.3.3.4 Health and social outcomes related to sexual health

Measures related to alcohol behaviour were evaluated for eight programmes: Family Matters (Bauman et al., 2000); SAAF (Brody et al., 2004; 2006); a physician-led primary care-based intervention (Stevens et al., 2002; Jones et al., 2005); a home-based family intervention (Loveland-

Cherry et al., 1999); ISFP and PDFY (Mason et al., 2009; Spoth et al., 1999; 2001; 2004); a CD-ROM intervention with and without a parent involvement component (Schinke et al., 2004); and a CD-ROM intervention for girls and their mothers (Schinke et al., 2009). Studies examined alcohol use over a period of time and/or initiation of alcohol use.

Eight studies evaluated the impact of seven programmes on alcohol use (Bauman et al., 2000; Stevens et al., 2002; Jones et al., 2005; Loveland-Cherry, 1999; Schinke et al., 2004; 2009; Spoth et al., 2001; Mason et al., 2009). Three of these programmes had no effects on alcohol use among young people who had already initiated alcohol use. There were non-significant effects of the Family Matters programme (Bauman et al., 2000) on past 30-day drinking at 3- and 12-month follow-up, and no long-term effects of a home-based family intervention (Loveland-Cherry et al., 1999). There were positive effects of two CD-ROM interventions (Schinke et al., 2004; 2009). Young people who received a CD-ROM intervention with and without a parental involvement component reported less monthly alcohol use than controls over medium- to long-term follow-up (all p<0.001), and at the 3-year follow-up, participants in the CD-ROM plus parental involvement condition reported less monthly use that those who received the CD-ROM intervention only (p<0.05). In addition, short-term positive effects of a CD-ROM intervention for girls and their mothers (Schinke et al., 2009) were found for alcohol use in the past week (p<0.01), month (p<0.05) and year (p<0.05). Long-term evaluation of the ISFP and PDFY (Spoth et al., 2001) revealed positive long-term effects of both programmes on alcohol use in the past 30 days and on a composite index of alcohol use², compared to controls (p<0.05 and p<0.01, respectively). Mason and colleagues (2009) examined the long-term impact on PDFY on rates of alcohol abuse. Based on ten years of follow-up, women who received the intervention were significantly less likely than controls to report alcohol abuse at age 22 (p<0.05), while the intervention had no significant effect on alcohol abuse among men. There were potentially harmful effects of the intervention examined in the Dartmouth Prevention Project (Stevens et al., 2002; Jones et al., 2005). Participants who received a physician-led health consultation designed to promote parental communication about alcohol and smoking reported significantly higher levels of drinking than participants who received a consultation about bicycle helmet use, seatbelt use or gun storage at both the 24- and 36-month follow-up (24 months: OR 1.27; 95% CI 1.03, 1.55 and 36 months: OR 1.30; 95% CI 1.07, 1.57).

Six studies (Brody et al., 2004; 2006; Loveland-Cherry et al., 1999; Spoth et al., 1999; 2001; 2004) evaluated the impact of four programmes on initiation of alcohol use. At post-test and 24-months follow-up, Brody and colleagues (2004; 2006) found positive effects of the SAAF on initiation of alcohol use (p<0.05) and Loveland-Cherry and colleagues (1999) reported significant long-term effects on alcohol initiation (p<0.05) of a home-based family intervention. Similarly, medium- and long-term evaluations of the ISFP programme (Spoth et al., 1999; 2001) found positive intervention effects on initiation of alcohol use (1-year follow-up: p<0.05; 2-years follow-up: p<0.01; 3-years follow-up:

² Composite measure of three items concerning lifetime behaviours and one concerning recent use: lifetime use, lifetime use without parental permission, lifetime drunkenness, and past month use.

p<0.01); although no significant effect was seen among PDFY participants. Spoth and colleagues (2004) also examined the impact of the ISFP and PDFY on growth in initiation of alcohol use over 6 years from the 6th to the 12th grade (age 11-12 to 17-18 years) finding that the ISFP delayed alcohol initiation among intervention participants (p<0.05). Spoth and colleagues (2001) also found that in 10th grade (age 15-16 years), ISFP participants who had initiated alcohol use during the study were less likely to have been drunk and to have drank without parental permission (both p<0.01), but again no significant programme effects on these behaviours were demonstrated among PDFY participants.

5.3.4 Summary and evidence statements

Outcomes relating to knowledge were examined in one study (Johnson et al., 1996), which reported a positive intervention effect at post-test and 1-year follow-up (both p<0.001) on knowledge of alcohol and other drugs among participants in the Creating Lasting Connections programme. Four studies evaluated intervention effects on attitudes and values (Brody et al., 2004; 2009; Jones et al., 2005; Schinke et al., 2009). Positive effects on attitudes about drinking were reported at short-term follow-up for SAAF (Brody et al., 2004) and a CD-ROM intervention for mothers and their daughters (Schinke et al., 2009). However, at long-term follow-up of SAAF (Brody et al., 2009) no significant effects were reported on attitudes. Evaluations of two CD-ROM based interventions (Schinke et al., 2004; 2009) showed positive programme effects on family communication skills and involvement skills. The SAAF (Brody et al., 2004) had a short-term positive effect on parental communication, but this finding was no longer significant at long-term follow-up (Brody et al., 2006). The Creating Lasting Connections programme (Johnson et al., 2006) had non-significant effects on family communication, but this finding and rule setting at both short- and long-term follow-up times for all but one measure of youth involvement in rule setting.

Eleven studies examined intervention effects on health and social outcomes related to alcohol use across eight programmes. Three programmes (Bauman et al., 2000; Jones et al., 2005; Loveland-Cherry, 1999) demonstrated non-significant effects on alcohol use, but across four programmes (Schinke et al., 2004; 2009; Spoth et al., 2001; Mason et al., 2009), short- and long-term positive effects on alcohol use were reported. In addition, six studies (Brody et al., 2004; 2006; Loveland-Cherry et al., 1999; Spoth et al., 1999; 2001; 2004) of four programmes reported positive intervention effects on initiation of alcohol use in the medium- to long-term. In addition, to its effects on alcohol use and initiation of alcohol use, the ISFP had long-term positive effects on drunkenness and drinking without parental permission (Spoth et al., 2004). Long-term follow-up of the PDFY (Mason et al., 2009) revealed a positive programme effect on women's alcohol abuse in early adulthood.

Evidence statement 3

- 3 (a) There is no evidence from one RCT¹ to determine the effect of programmes aimed at families on knowledge and understanding relating to alcohol use
- 3 (b) There is moderate evidence from two RCTs² to suggest that programmes delivered to families may have short-term positive effects on attitudes and values related to alcohol. Findings may

only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.

- 3 (c) There is moderate evidence from two RCTs³ to suggest that programmes delivered to families which target family interaction may have positive effects on family communication, parental monitoring and parental rules about alcohol. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.
- 3 (d) There is moderate evidence from 11 RCTs⁴ to suggest that programmes delivered to families may have mixed effects on health outcomes related to alcohol use. Three RCTs⁵ showed no intervention effects on alcohol use. One RCT⁶ of a brief, family focused intervention (Iowa Strengthening Families Program) showed long-term reductions in alcohol use, initiation of alcohol use, and drunkenness and one RCT⁷ of a culturally-tailored family-based programme (Strong African American Families) showed a long-term effect on initiation of alcohol use. In addition, one RCT⁸ of a CD-ROM intervention with parental involvement showed long-term reductions in monthly alcohol use. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.
- 3 (e) There is weak evidence from one RCT⁹ to suggest that physician-led interventions may have a long-term negative impact on alcohol use. Findings may only be partially applicable to the UK as the study was conducted in the USA and may not be generalisable beyond the populations studied.
- ¹ Johnson et al., 1996 (RCT -)
- ² Brody et al., 2004 (RCT +); Schinke et al., 2009 (RCT +)
- ³ Schinke et al., 2004 (RCT +); Schinke et al., 2009 (RCT +)
- ⁴ Bauman et al., 2000 (RCT -); Brody et al., 2004; 2006 (RCT +); Jones et al., 2005 (RCT +); Loveland-Cherry, 1999 (RCT -); Schinke et al., 2004 (RCT +); Schinke et al., 2009 (RCT +); Spoth et al., 1999 (RCT +); Spoth et al., 2001; 2004 (RCT +); Mason et al., 2009 (RCT +)
- ⁵ Bauman et al., 2000 (RCT -); Jones et al., 2005 (RCT +); Loveland-Cherry, 1999 (RCT -)
- ⁶ Spoth et al., 1999 (RCT +); Spoth et al., 2001; 2004 (RCT +)
- ⁷ Brody et al., 2004; 2006 (RCT +)
- ⁸ Schinke et al., 2004 (RCT +)
- ⁹ Stevens et al., 2002 (RCT +)

Table 5.5. Summary of programme c	content: programmes	delivered in families
-----------------------------------	---------------------	-----------------------

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Bauman et al., 2000	RCT (individual) -	USA n=203 families 73% White; 13% Black; 9% Hispanic; 5% other 12-14 years	Family	Family Matters : families received four booklets aimed at preventing the onset of substance use	Theories of socialisation; value expectancy theory; health belief model; social learning theory; social inoculation theory	NR
Brody et al., 2004; 2006; Gerrard et al., 2006; Murry et al., 2007	RCT (cluster) +	USA n=332 100% African American mean 11.2 years	Family	Strong African American Families Program : seven two-hour meetings to prevent drinking including separate and joint parent and child components.	NR	NR
Johnson et al., 1996	RCT (cluster) -	USA n=120 families 16% African American 12-14 years	Family; church communities	Creating Lasting Connections: three training modules for parents and adolescents and case-management services focusing on delaying the onset and reducing the frequency of alcohol and other drug use	NR	NR
Jones et al., 2005	RCT (cluster) +	USA n=2153 families 97% White mean 11 years	Family	Substance use prevention education and family communication intervention; family discussion and mailed brochures and quarterly newsletters	NR	Physician and nurse
Stevens et al., 2002	RCT (cluster) +	USA n=3,111 Ethnicity=NR mean 11 years	Primary care (paediatric)	Dartmouth Prevention Project : Single session health consultation and 12 newsletters over 36 months; child and parent discussions about alcohol and tobacco use; family signed contract agreeing to talk about the risks at home and develop a family policy; mailed print materials; biannual telephone call and incentives.	NR	Physician
Loveland- Cherry et al.,1999	RCT (individual) -	USA n=892 families 86% European American 10-11 years	Family; in the home	Alcohol misuse prevention intervention including three- hour sessions at home; family meetings and phone calls; booster sessions; semi-annual newsletters	Social cognitive theory; problem behaviour theory	NR

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Schinke et al., 2004	RCT (cluster) +	USA n=514 youths 54% black, 30% Hispanic, 11% white, 5% Asian or other mean 12 years	Family	CD-ROM based alcohol prevention intervention for adolescents including 10 45-minute sessions plus booster sessions. The parent intervention consisted of a video and two newsletters plus a workshop booster session. A second intervention group were not exposed to the parent intervention.	Family interaction theory; social learning theory; problem behaviour theory	CD-ROM
Schinke et al., 2009	RCT (individual) +	USA n=202 mother and daughter (mean age 12.2 years) dyads 67.8% White, 14.1% Latina, 9.5% Black, .5% Asian, 8% other	Family; in the home	Computer modules aiming to reduce underage alcohol consumption through improving mother-daughter relationships and teaching skills to avoid drinking. Dyads undertook 14 modules over 3 weeks focusing on communication; relationships; conflict management; alcohol refusal skills; peer norms and the media	Family interaction theory; social learning theory	CD-ROM
Spoth et al., 1999; Spoth et al., 2001; 2004	RCT (cluster) +	USA n=446 families 10-12 years 98% White	Family	Iowa Strengthening Families Program (ISFP): Seven weekly two-hour sessions aiming to prevent substance use. Parents and children attended separate and joint sessions where they were exposed to a skills-based curriculum and engaged in activities to increase family cohesiveness	Biopsychosocial model	Video
Spoth et al., 2001; 2004 Mason et al., 2009	RCT (cluster) +	USA n=429 families 10-12 years 98% White	Family	Preparing for the Drug Free Years (PDFY): Parents attended four education and skill-based sessions and were accompanied by their children for one additional session	Social development model	Video

					Outcomes		
Study	Rating	Intervention	Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills
Brody of al		SAAF	Received 3	PT n=NR	-	1 youth protective factors*	t communicative parenting**
2004; 2006	RCT +	n=182	leaflets n=150	24 months n=305 (92%)	-	NS youth protective factors	NS
				PT n=120	↑ alcohol and other drug knowledge***	-	 † youth involvement in setting up AOD rules*** NS family communication or bonding with mother, father or siblings (youth and parent reports), family meeting practices
Johnson et al., 1996	RCT -	Creating Lasting Connections n=59 (Post-test n given only)	Received no intervention n=61 (Post- test n given only)	12 months n=120	1 alcohol and other drug knowledge***	-	NS family meeting practices, family rules about ATOD, family rules about non-AOD youth behaviour, youth involvement in setting non-AOD rules, family communication (parent or youth report), bonding with mother (parent report), bonding with father (youth report), bonding with siblings (parent report)
Jones et al., 2005	RCT +	Physician-led intervention n=1,235 families ^a	Discussed safety issue n=918 families ^a	36 months n=2,153 (70%)	-	 ↓ externalizing of problems (boys) NS externalizing of problems (girls) NS internalizing of problems 	-
Schinke et al., 2004	RCT +	CD ROM + parent intervention n=NR	No intervention n=NR	3 years n=469 (91%)	-	-	family involvement*

Table 5.6. Programmes delivered in families: effects on knowledge, attitudes and skills

Study	Rating	Intervention	Comparator	Follow-up	Outcomes			
					Knowledge and understanding	Attitudes and values	Personal and social skills	
Schinke et al., 2009	RCT +	NR n=NR	Waiting list control group n=NR	2 months n=199 (99%)	-	 ↑ girls' beliefs about underage drinking* ↑ girls' self-efficacy for avoiding alcohol* ↑ intention not to drink alcohol until adulthood* 	 f girls' communication with mother* f girls reported parental rules about drinking* f girls reported parental monitoring* f girls reported family conflict management skills* f girls' alcohol refusal skills* f mother reported rules about drinking* NS mother reported communication; parental monitoring 	
*p<0.05; **p<0.01; ***p<0.001; [¶] p value not reported; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported ^a Follow-up sample								

Ctudy	Rating	Intervention	Comparator	Follow-up	Health and social outcomes			
Study					Alcohol use	Heavy alcohol use	Other	
Bauman et al., 2000	RCT -	Family Matters n=99 families	No intervention n=104 families	3 mo n=NR	NS past 30 day drinking	-	-	
				12 mo n=74 (71%)	NS past 30 day drinking	-	-	
Brody et al	RCT +	Strong African American Families Program n=182	Received 3 leaflets n=150	PT n=NR	↓ alcohol initiation*	-	-	
2004; 2006				24 mo n=305 (92%)	↓ alcohol initiation*	-	-	
	RCT (cluster) +	Dartmouth Prevention Project n=1,780	Consultation on bicycle helmet, seatbelt use or gun storage n=1,331	12 mo (NR)	NS drinking	-	-	
Stevens et al., 2002				24 mo (NR)	↑ drinking [¶]	-	-	
				36 mo (NR)	↑ drinking [¶]	-	-	
Jones et al., 2005	RCT +	NR At FU n=1,235 families	Discussed with doctor an unrelated safety issue At FU n=918 families	36 mo n=2,153 (70%)	NS alcohol use	-	-	
Loveland- Cherry et al., 1999	RCT -	NR n=90 at FU	No intervention n=338 at FU	4 yrs n=428 (48%)	↓ alcohol initiation** NS alcohol use (among prior drinkers)	-	-	
Schinke et al., 2004	RCT +	CD ROM + parent intervention n=NR	No intervention n=NR	3 yrs n=469 (91%)	↓ past 30-day alcohol use***	-	-	
Schinke et al., 2009	RCT +	NR n=NR	Waiting list control group n=NR	2 mo n=199 (99%)	 past week alcohol consumption** past month alcohol consumption* past year alcohol consumption** 			

Table 5.7. Programme delivered to families: effects on health and social outcomes related to alcohol use

Study	Poting	Intervention	Comparator	Follow-up	Health and social outcomes			
Study	Rating				Alcohol use	Heavy alcohol use	Other	
Spoth et al., 1999	RCT +	ISFP n=238	Minimal contact n=208	1 yr n=317 (71%)	↓ alcohol initiation index scores*	-	-	
				2 yrs n=294 (66%)	↓ alcohol initiation index scores**	-	-	
Spoth et al., 2001	RCT +	ISFP n=238	Minimal contact n=208	4 yrs ^a n=447 (67%)	↓ alcohol initiation** ↓ past 30-day alcohol use* ↓ alcohol composite use index score**	↓ new user ever been drunk**	↓ new user drank without parental permission**	
Spoth et al., 2004	RCT +	ISFP n=238	Minimal contact n=208	6 yrs ^a n=304 (46%)	↓ growth in lifetime alcohol use*	-	-	
Spoth et al., 2001	RCT +	PDFY n=221	Minimal contact n=208	4 yrs ^a n=447 (67%)	NS alcohol initiation ↓ past 30-day alcohol use* ↓ alcohol composite use index score*	NS new user ever been drunk	NS new user drank without parental permission;	
Spoth et al., 2004	RCT +	PDFY n=221	Minimal contact n=208	6 yrs ^a n=304 (46%)	NS growth in lifetime alcohol use	-	-	
Mason et al., 2009	RCT +	PDFY n=221	Minimal contact n=208	10 yrs ^a n=313 (73%)	-	-	↓ rate of alcohol abuse (women only*)	
*p<0.05; **p<0.01; ***p<0.001; ¹ p value not reported; 1 increase relative to comparator; 4 decrease relative to comparator; NS not significant; - outcome not reported ^a Follow-up from baseline								

5.4 Programmes delivered to parents

5.4.1 Overview of evidence identified

A total of five studies (Beatty et al., 2008; Carlson et al., 2000; Cohen and Rice, 1995; Koutakis et al., 2008; Toomey et al., 1996) reported on alcohol prevention programmes which targeted parents. Three studies were conducted in North America (Carlson et al., 2000; Cohen and Rice, 1995; Toomey et al., 1996), one was conducted in Australia (Beatty et al., 2008) and one was conducted in Sweden (Koutakis et al., 2008).

Two programmes were delivered solely in the home; Carlson and colleagues (2000) examined information postcards that addressed alcohol risk and protective factors (Carlson et al., 2000) and Beatty and colleagues (2008) examined a parent-directed intervention designed to encourage parent-child communication. Two programmes used home-delivered materials in conjunction with community-based group sessions (Koutakis et al., 2008; Toomey et al., 1996) and one programme was delivered in schools to parents (Cohen and Rice, 1995).

No theoretical base for intervention was reported for two studies (Koutakis et al., 2008; Toomey et al., 1996). Diffusion of Social Innovations theory and social cognitive theory underpinned the intervention examined by Beatty and colleagues (2008), whilst cognitive behavioural theory was used in an unnamed alcohol prevention programme (Cohen and Rice, 1995) and social cognitive theory was used in conjunction with the health belief model in the STARS for Families programme (Carlson et al., 2000).

The overall number of participants recruited in the included studies ranged from 478 (Carlson et al., 2000) to 2,278 (Cohen and Rice, 1995). Overall, two studies (Beatty et al., 2008; Cohen and Rice, 1995) were based on samples of over 1,000 parents, however, one other study (Toomey et al., 1996) included a sample student population over 1,000. The use of power calculations or an appropriate sample size to detect a significant effect was discussed in only one study (Beatty et al., 2008), a further two studies provided enough information to determine that sample sizes were appropriate (Koutakis et al., 2008; Toomey et al., 1996). However, two additional studies (Cohen and Rice, 1995; Carlson et al., 2000) did not provide sufficient information to determine if the sample was sufficiently powered.

The included studies focused primarily on parents of young people aged between 10-13 years old. However one study (Koutakis et al., 2008) was aimed at parents with children aged 13-16 years. Follow-up times varied with two studies (Cohen and Rice, 1995; Koutakis et al., 2008) reporting posttest data using mean longitudinal data; one (Carlson et al., 2000) reported a short follow-up time of two months; another study (Beatty et al., 2008) did not clearly report the follow-up time used and one study reported findings at post-test and after one year (Toomey et al., 1996).

5.4.2 Quality assessment

Of the five included studies three were RCTs, one was based on an NRCT design and one was a CBA study. All three RCTs (Beatty et al., 2008; Cohen and Rice, 1995; Toomey et al., 1996) were based on cluster randomisation with randomisation conducted at school level in all three studies. The unit of analysis did not match the unit of allocation in two studies (Beatty et al., 2008; Cohen & Rice, 1995); Beatty and colleagues (2008) examined the effect of clustering but within-school correlations were not found to be significant and no adjustments were made, while Cohen and Rice (1995) did not describe how any potential clustering effect was accounted for. One study (Koutakis et al., 2008) was rated good quality presenting a high quality matched study design which controlled for study bias and clearly presented study details. One RCT (Beatty et al., 2008) was rated moderate quality providing details of power calculations and accounting for attrition levels. Three studies (Carlson et al., 2000; Cohen and Rice, 1995; Toomey et al., 1996) were rated poor quality (- rating). Methodological data were limited and in some instances the internal validity of the study was not clear. Outcome measures were reported to be reliable in four studies (Beatty et al., 2008; Carlson et al., 2000; Cohen and Rice, 1995; Toomey et al., 1996) and relevant in all studies.

5.4.3 Findings

5.4.3.1 Knowledge

None of the included studies examined intervention effects on knowledge and understanding.

5.4.3.2 Attitudes and values

Three studies (Cohen and Rice, 1995; Koutakis et al., 2008; Toomey et al., 1996) examined intervention effects on alcohol-related attitudes and values. Cohen and Rice (1995) examined effects on attitudes and values among two cohorts. In relation to the impact of the intervention on parental attitudes and values, there was no difference in parenting behaviours between groups across time. When children's perceptions of parenting behaviours were stratified by the onset of children's alcohol use (drinker vs. non-drinker), children who became drinkers showed larger declines in parental respect, parental rapport and parental monitoring compared with children who remained non-drinkers. However, parents whose children reported having substance-using peers consistently perceived their parents-child relationship as having higher indexes of rapport (p<0.001), respect (p<0.001), and monitoring (p<0.001). Comparing drinkers to non-drinkers across the two cohorts revealed an increase in students' perceptions of parents' respect for their child and students' perceptions of parental monitoring (Cohort 1: p<0.001, p<0.01; Cohort 2: both p<0.01 respectively). However, within cohort 1 there were no programme effects on students' perceptions of parent-child rapport, whereas cohort 2 showed a significant increase in perception (p<0.01). There were no programme effects on students' perceptions of their parents' knowledge of their friends in either cohort. Koutakis and colleagues (2008) found significant programme effects of a zero tolerance alcohol prevention programme (the Örebro Prevention Programme) on the maintenance of strict parental attitudes towards underage drinking in the intervention group compared to the control group (p<0.001). The Amazing Alternatives! Home Programme (Toomey et al., 1996) showed no effect on parental attitudes regarding young people aged 18-20 drinking; allowing teens to drink when they are seniors in school; family rules about drinking; perception of how many parents of their children's friends they knew; their contact with other parents regarding alcohol-related issues; rules or systems for unsupervised periods; the frequency at which they check parties have adult supervision; and whether they always monitored their teen's whereabouts. At post-test no programme effects were seen on students' alcohol use intentions for the next week, month, year or when they were 21 years old. Furthermore, no intervention effects were seen on students' views that their parents would allow them to drink alcohol when they were high school seniors. At post-test students reported significant increases in family rules against drinking (p<0.01); talking about the consequences if caught drinking (p<0.05) and parents' rules as a reason not to drink alcohol (p<0.05). However, these effects were not sustained at the long-term follow-up (>1 year).

5.4.3.3 Personal and social skills

Three studies (Beatty et al., 2008; Carlson et al., 2000; Toomey et al., 1996) reported intervention effects on personal and social skills. All outcome measures for personal and social skills focused on communication. Beatty and colleagues (2008) reported greater parent-child discussion about drinking alcohol; recent discussion about alcohol; more alcohol-related topics discussed in the intervention group compared to the control. Furthermore, the intervention group reported a greater perception of engagement with their child. Findings from the STARS programme (Carlson et al., 2000) showed short-term effects on the frequency at which parents spoke to their child about avoiding alcohol (p<0.05) and on how recently parents had spoken to their child (last 30 days; p<0.05). At long-term follow-up (>1 year) Toomey and colleagues (1996) found that more parents reported a higher proportion of parent-child discussion about the consequences if caught drinking (p<0.05); alcoholrelated situations (p<0.01); alcohol messages in the mass media (p<0.01); and encouraged their child and friends to gather at home (p<0.05). Students also reported a higher proportion of parent-child discussion related to family rules about alcohol (p<0.001); consequences for breaking the rules (p<0.01); and problems they could have with alcohol use (p<0.05). However, students reported nonsignificant changes in discussions about having friends over to the house; alcohol messages in the mass media; good eating habits and sex education.

5.4.3.4 Health outcomes related to alcohol use

Two studies (Koutakis et al., 2008; Toomey et al., 1996) reported alcohol-related health outcomes. The Örebro Prevention Programme (Koutakis et al., 2008) was found to result in a decrease in reported youth drinking, drunkenness and past 30 day drunkenness (p<0.001) in the intervention group. Furthermore, subgroup analysis revealed that of those categorised as early starters in alcohol use, those in the intervention group reported lower rates of drunkenness (p<0.01). The Amazing Alternatives! Home Program (Toomey et al., 1996) found no programme effects on lifetime alcohol use; past year alcohol use; or past month alcohol use at either post-test or at long-term follow-up.

5.4.4 Summary and evidence statements

Five studies (Beatty et al., 2008; Carlson et al., 2000; Cohen and Rice, 1995; Koutakis et al., 2008; Toomey et al., 1996) were identified that examined alcohol prevention interventions aimed at parents of young people. Two studies (Carlson et al., 2000; Toomey et al., 1996) were primarily delivered using postcards or booklets and one additional study (Beatty et al., 2008) used information tools in conjunction with workshops. The remaining two studies (Cohen and Rice, 1995; Koutakis et al., 2008) used trained facilitators or project workers to deliver the programme.

None of the included studies presented outcomes relating to knowledge or understanding. Three studies (Cohen and Rice, 1995; Koutakis et al., 2008; Toomey et al., 1996) examined programme effects on attitudes and values. Intervention effects on parental attitudes towards underage drinking were seen in one study (Koutakis et al., 2008). However, Toomey and colleagues reported no programme effects on parental attitudes to young people's drinking. Improvements in parents' perceptions of their parent-child monitoring and parental respect for their child were reported in one study (Cohen and Rice, 1995). Short-term intervention effects on parent-child communication about alcohol and family rules were reported in one study (Toomey et al., 1996). However, effects were not maintained long-term. Three studies (Beatty et al., 2008; Carlson et al., 2000; Toomey et al., 1996) reported intervention effects on personal and social skills relating to alcohol use. All three studies reported increases in parent-child communication about alcohol. Two studies (Beatty et al., 2008; Carlson et al., 2000) either increased frequency or reported more recent parent-child communication about alcohol. One study (Toomey et al., 1996) showed positive long-term effects on parent-child communication about alcohol. Two studies (reported by both parents and their children).

Two studies (Koutakis et al., 2008, Toomey et al., 1996) reported health outcomes relating to alcohol use. No programme effects were reported by Toomey and colleagues. However, Koutakis and colleagues reported positive intervention effects on youth drinking, past month drunkenness and drunkenness. Positive intervention effects on drunkenness also extended to a subgroup categorised as early starters in alcohol use.

Evidence statement 4

- 4 (a) There is inconsistent evidence from one NRCT and two RCTs¹ to determine the effects of interventions delivered to parents on attitudes and values relating to alcohol. However, there is weak evidence from one RCT² to suggest that programmes aimed at parents can have positive short-term effects on young people's attitudes towards family rules and their influence as a deterrent for drinking. These findings may be only partially applicable to the UK as this study was implemented in the USA and may not be generalisable beyond this population.
- 4 (b) There is moderate evidence from two RCTs and one CBA study³ to suggest that interventions delivered to parents may have a positive short- to potentially long-term effect on parent-child communication about alcohol. These findings may be only partially applicable to the UK as they

were not implemented in a UK setting and may not be generalisable beyond the populations studied.

- 4 (c) There is insufficient and inconsistent evidence from one NRCT and one RCT⁴ to determine the effect of interventions delivered to parents on health and social outcomes relating to alcohol use among young people.
- ¹ Koutakis et al., 2008 (NRCT ++); Toomey et al., 1996 (RCT -); Cohen and Rice, 1995 (RCT -)
- ² Toomey et al., 1996 (RCT -)
- ³ Beatty et al., 2008 (RCT -); Toomey et al., 1996 (RCT -); Carlson et al., 2000 (CBA -)
- ⁴ Koutakis et al., 2008 (NRCT ++); Toomey et al., 1996 (RCT -)

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Beatty et al., 2008	RCT (cluster) +	Australia n=1,201 Ethnicity=NR 10-11 years	Home, community	Self-directed intervention; parents received five communication sheets containing self-help information and activities.	Diffusion of social innovation theory	Communication sheets/Peer leaders
Carlson et al., 2000	CBA -	USA n=478 parents 65% Black, 30% White, 5% Other 6 th grade	Home	STARS for Families: 10 postcards were mailed, two per week. With parents using an average of 6.49 cards to facilitate discussion.	Health belief model, Social cognitive theory	Postcards
Cohen and Rice, 1995	RCT (cluster) -	USA Cohort 1= 1,034; 15% Asian, 32% Hispanic, 38% White, 4 % black, 11% other 5^{th} Grade Cohort 2 = 1,244, 15% Asian, 27% Hispanic, 40% White, 4% Black, 15% Other NR	Schools	Substance use prevention focusing on parental skills training around substances; Drug refusal skills, family rules about drugs. It incorporated 4 sessions for cohort 1 and 3 sessions for cohort 3.	Cognitive-behavioural model	Facilitators
Koutakis et al., 2008	NRCT ++	Sweden n= 811 children; n= 651 parents, 13-16 yrs old Ethnicity=NR	Schools	The Örebro Prevention Programme: An alcohol prevention programme promoting zero-tolerance to alcohol use, promoting leisure activities, parental influence on adolescents, parent-child contracts. Conducted over five semesters, with one 30 minute long meeting per semester.	Not reported	External project workers
Toomey et al., 1996	RCT (cluster) -	USA n=1,028 children n=521 parents, grade 7 Primarily White	Home, community	Amazing Alternatives! Home Programme: Aim was to improve communication between parents and their 7th graders concerning alcohol-related issues, to improve parenting skills like monitoring and to reduce underage drinking. The intervention was delivered in four booklets to parents and two focus groups with parents.	Not reported	Booklets/other

Table 5.8. Summary of programme content: programmes delivered to parents

	Rating	Intervention	Comparator	Follow-up	Outcomes			
Study					Knowledge and understanding	Attitudes and values	Personal and social skills	
Beatty et al., 2008	RCT (cluster) +	Self-directed intervention n=353 (29%)	No intervention n=848 (71%)	1 mo n=830 (69%)	-	-	 † discussions with child about drinking alcohol*** † spoken with child about alcohol recently*** † perceived engagement with child** † discussion alcohol-related topics*** 	
Carlson et al., 2000	CBA-	STARS n=237 (parents)	No intervention n=237 (parents)	2 mo (NR)	-	-	 ↑ talked to their child about avoiding alcohol 10+ times in the past year* ↑ talked to their child in past 30 days* 	
Cohen & Rice, 1995	RCT (cluster) -	Cohort 1 n=NR (drinkers vs. non- drinkers)	No intervention n=NR	PT, annually over 4 years (NR)	-	 † perception of parent respect for child*** NS perception of parent-child rapport † perception of parental monitoring*** NS perception of parent's knowledge of children's friends 	-	
		Cohort 2 n=NR (drinkers vs. non- drinkers)			-	the perception of parent respect for child** 1 perception of parent-child rapport** 1 perception of parental monitoring** NS perception of parent's knowledge of children's friends	-	
Koutakis et al., 2008	NRCT ++	Örebro Prevention Programme n= 393 children; n= 339 parents	No intervention n=418 students; n=312 parents	PT (Mean over time) (students=85%; parents=71%)	-	1 strict parental attitudes towards underage drinking***	-	

Table 5.9. Programmes delivered to parents: effects on knowledge, attitudes and skills
						Outcomes	
Study	Rating	Intervention	Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills
				PT	-	 NS parental attitudes^a † young person's attitudes to: family rules against youth drinking**; talked about consequences if caught drinking*; parents' rules as a reason not to use alcohol* NS My parents will allow me to drink when I am a high school senior; NS alcohol use intentions 	-
Toomey et al., 1996	RCT (cluster) -	Amazing Alternatives! Home Program n=257 parents (49%)	No intervention n= 264 parents (51%)	>1 year (students=83%; parents=>90%)	-	NS young person's attitudes NS alcohol use intentions	 † parental discussion with child: consequences if caught drinking*; alcohol-related situations**; alcohol messages in the mass media**; encourage child and friends to gather at home* † young person's discussion with parent: family rules about alcohol***; consequences for breaking rules**; problems could have with alcohol use*; (NS having friends over to house; NS alcohol messages in the mass media; NS good eating habits; NS sex education)
*p<0.05; **p<0 ^a Included: not a about alcohol-r	.01; ***p<0.001 allowing teens t related situation	; ¹ p value not reporte o drink when senior; s?; special rules or s	ed; 1 increase rel family rules agai systems for unsur	ative to comparato nst drinking; how r pervised periods; c	r; ↓ decrease relativ nany parents of your heck parties for adu	re to comparator; NS not significant; r child's friends do you know?; how It supervision; always monitor teen's	; - outcome not reported often do you contact other parents s whereabouts

Study	Deting	Intervention	Compositor	Fellow	Health and social outcomes			
	Rating	intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other	
Koutakis et al., 2008	NRCT ++	Örebro Prevention Programme students= 1,183; parents= 1,022	No intervention students=122; parents= 999	PT (Mean over time) students=85%; Parents=71%)	↓ youth drinking***	↓ drunkenness*** ↓ past month drunkenness*** Early starters - ↓ drunkenness**	-	
Toomey et al., 1996	RCT (cluster) -	Amazing Alternatives! Home Program youth= NR; parents= 257	No intervention	PT	NS lifetime alcohol use NS past year alcohol use NS past month alcohol use	-	-	
			youth= NR; parents= 264	>1 year (students=83%; parents=>90%)	NS lifetime alcohol use NS past year alcohol use NS past month alcohol use	-	-	
*p<0.05; **p<0.	.01; ***p<0.001	; [¶] p value not reporte	ed; 1 increase rel	ative to comparato	r; ↓ decrease relative to comp	arator; NS not significant; - ou	tcome not reported	

Table 5.10. Programmes delivered to parents: effects on health and social outcomes related to alcohol use

5.5 Programmes involving the wider community or mass media

5.5.1 Overview of the evidence identified

Three studies (Cheadle et al., 1995; Flynn et al., 2006; Kypri et al., 2005) were identified that examined programmes involving the wider community or mass media. Cheadle and colleagues (1995) examined a 5-year community-based health promotion programme for adolescents on an American Indian Reservation. Programme components included classes, skills development programmes, alcohol- and drug-free events and a public information campaign. Flynn and colleagues (2006) and Kypri and colleagues (2005) examined mass media intervention programmes delivered in communities in the USA and New Zealand, respectively. The campaign examined by Flynn and colleagues (2006) was delivered over four years and included television and radio messages directed at young people, in addition to messages directed at parents and training videos for alcohol retailers. The focus of the campaign examined by Kypri and colleagues (2005) was on highlighting the risks of supplying alcohol to young people. The campaign lasted one and half months and included advertisements and publicity via local media outlets in addition to media events.

Only one study reported the theoretical base for intervention. The mass media programme examined by Flynn et al., (2006) was based on social cognitive theory. Two studies examined programmes that were targeted at the wider community (Cheadle et al., 1995; Kypri et al., 2005) and Flynn and colleagues (2006) examined a mass media intervention directed towards young people as they matured from Grades 4-5 into Grades 7-8 (approximately from age 10 to age 12-14 years).

5.5.2 Quality assessment

All three studies (Cheadle et al., 1995; Flynn et al., 2006; Kypri et al., 2005) were based on a CBA design and the quality of all three studies was rated moderate. The interventions and comparisons examined were well described and appropriate, contamination was acceptably low, and all three studies examined important and relevant outcomes.

5.5.3 Findings

5.5.3.1 Knowledge and understanding

None of the included studies examined intervention effects on knowledge.

5.5.3.2 Attitudes and values

One study (Flynn et al., 2006) examined intervention effects on alcohol-related attitudes and values. There were no effects of the media campaign on mediators of alcohol use, which included expectations related to alcohol use, perceived parental and peer norms, perceived peer prevalence, perceived access to alcohol and whether or not participants felt confident refusing offers of alcohol.

5.5.3.3 Personal and social skills

None of the included studies examined intervention effects on personal and social skills

5.5.3.4 Health and social outcomes related to alcohol use

There were no effects of either mass media programme (Flynn et al., 2006; Kypri et al., 2005) on alcohol use and the community-wide campaign targeting American Indian adolescents also failed to show any effects on alcohol consumption.

5.5.4 Summary and evidence statements

Three studies (Cheadle et al., 1995; Flynn et al., 2006; Kypri et al., 2005) were identified that examined programmes involving the wider community or mass media. Two studies (Flynn et al., 2006; Kypri et al., 2005) examined mass media intervention programmes delivered in communities in the USA and New Zealand, respectively, and one study (Cheadle et al., 1995) a 5-year community-based health promotion programme for adolescents on an American Indian Reservation.

None of the included studies examined intervention effects on knowledge and understanding, or on personal and social skills. There were no effects of a long-term mass media programme (Flynn et al., 2006) on mediators of alcohol use. There were no effects of either mass media programme (Flynn et al., 2006; Kypri et al., 2005) or a community-wide campaign targeting American Indian adolescents on alcohol use.

Evidence statement 5

- 5 (a) There is weak evidence from one CBA study¹ to suggest interventions and programmes involving mass media have no effect on attitudes and values related to alcohol use. Findings may only be partially applicable to the UK as the study was conducted in the USA and may not be generalisable beyond the populations studied.
- 5 (b) There is moderate evidence from three CBA studies² to suggest that interventions and programmes involving the wider community or mass media have no effects on young people's alcohol use. Findings may only be partially applicable to the UK as all the studies were set outside the UK and may not be generalisable beyond the populations studied.

² Flynn et al., 2006 (CBA +); Kypri et al., 2005 (CBA +); Cheadle et al., 1995 (CBA +)

¹ Flynn et al., 2006 (CBA +)

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Flynn et al., 2006	CBA +	USA n=16 school districts Ethnicity=NR 12-14 years	Mass media	32 television and 23 radio messages over 4 years; directed to youth as they matured from Grades 4-5 into Grades 7-8; radio messages directed toward their parents; training video for retail clerks	Social cognitive theory	NA
Kypri et al., 2005	CBA +	New Zealand n=872 New Zealand European (82%), Maori (8%), Samoan (1%), Chinese (1%) and Other (9%) 15-19 years	Mass media	'Think before you buy under-18s drink' campaign : Local newspaper and radio advertisements; radio and print media interviews with community workers; media events, billboard advertisements, distribution of printed material and presentation of campaign information at point of sale.	NR	NA
Cheadle et al., 1995	CBA +	USA n=6 communities Ethnicity not fully reported but included American Indian, White Hispanic and Asian 9th and 12th grade	American Indian Reservation	Classes, skills development programs, alcohol- and drug- free events, and public campaigns.	NR	Various

Table 5.11. Summary of programme content: programmes involving the wider community or mass media

Table 5.12. Programmes involving	a the wider communit	v or mass media: effects on	knowledge, attitudes and skills
·	g		

	_					Outcomes		
Study	Rating	Intervention	Comparator	Irator Follow-up Knowledge and understanding Attitudes and values		Personal and social skills		
Flynn et al., 2006	CBA +	Media campaign n=8 school districts	No intervention n=8 school districts	Annual surveys (NA)	-	NS mediators of alcohol use	-	
*p<0.05; **p<0.01; ***p<0.001;								

Table 5.13. Programmes involving the wider community or mass media: effects on health and social outcomes related to alcohol use

Study	Deting	Intervention	Comparator	Follow-up	Health and social outcomes			
Study	Kating	intervention	Comparator	Follow-up	Alcohol use	Heavy alcohol use	Other	
Cheadle et al., 1995	CBA +	Unnamed n=1 community	No intervention n=5 nonurban, communities	Annual surveys over 3 years (NA)	NS alcohol use measures	-	-	
Flynn et al., 2006	CBA +	Media campaign n=8 school districts	No intervention n=8 school districts	Annual surveys (NA)	NS beer drinking	-	-	
Kypri et al., 2005	CBA +	Media campaign n=2 communities	No intervention n=1 community	PT (NA)	-	NS binge drinking	NS supplied alcohol for unsupervised drinking	
*p<0.05; **p<0.0	01; ***p<0.001;	p value not reported	d; 1 increase relati	ve to compara	tor; ↓ decrease relative to com	parator; NS not significant; - ou	itcome not reported	

5.6 Review of published economic evaluations

One study (Spoth et al., 2002) was identified that met the criteria for inclusion in the review of published economic evaluations. Spoth and colleagues (2002) evaluated the cost-effectiveness and net benefits of the ISFP and PDFY.

5.6.1 Review of Spoth and colleagues (2002)

5.6.1.1 Overview

The purpose of the study was to evaluate two interventions designed for general populations by estimating: (1) the cost per case of alcohol-use disorder prevented; (2) benefit-cost ratios; and (3) net benefits per participating family.

5.6.1.2 Summary of effectiveness data

Effectiveness estimates were drawn from a cluster randomised trial of two universal, family focused interventions, the seven-session ISFP and the five-session PDFY (Spoth et al., 1999). At the end of the 4-year study (Spoth et al., 2004), data were available from 478 (72%) of the original participants (study details and effectiveness data are presented in section 5.3)

Estimates of the number of alcohol-use disorder cases prevented per 100 families treated were calculated. For each age, the total number of children who had initiated alcohol use was multiplied by the proportion of future alcohol-use disorders expected among persons who initiated alcohol use at that age (taken from Grant & Dawson, 1997). The rate of future alcohol-use disorders expected for each condition was calculated by dividing the number of cases expected by the total number of persons assigned to each condition and multiplying by 100 to give the number of alcohol use disorder cases expected per 100 families treated. To estimate the number of future alcohol-use disorder cases prevented per 100 families, the rate for each intervention group was subtracted from the rate for the control condition. The projected disorder rate for participants who had not reported alcohol-use onset by the 4-year follow-up was calculated as the average of all of the rates associated with ages 18 and older (also taken from Grant & Dawson, 1997).

The projected lifetime alcohol use disorder rate calculated for the two intervention and control groups were: ISFP 37.65% (SE 3.59); PDFY 40.46% (SE 3.74); and control 43.17% (SE 3.68). Therefore the number of alcohol use disorder cases prevented per 100 families treated was estimated to be 5.53 for the ISFP and 2.72 for the PDFY.

5.6.1.3 Summary of resource utilisation and cost data

Both direct and indirect costs were assessed and expressed in dollars spent per participant (adjusted for inflation into 1992 dollar equivalents). Facility costs were not included, and the authors ignored the costs of providing informational materials to the minimal contact control group (\$1.25 per family) as they considered there to be no appreciable opportunity costs associated with participation in the control arm of the study. Intervention costs are summarised in Table 5.11. Intervention costs per 100 families for ISFP and PDFY were \$68,856 and \$55,567, respectively.

	Intervent	tion cost ^a
Cost source	ISFP ^b	PDFY ^c
Facilitator ads	\$550	\$550
Facilitator training ^d	\$25,758	\$32,648
Family training materials ^e	\$2,776	\$10,602
Family participation incentives ^f	\$13,620	\$10,994
Site management ^g	\$5,385	\$3,715
Programme facilitation ^h	\$31,972	\$10,875
Child care	\$4,620	\$2,948
Parent travel	\$455	\$483
Total costs (1994 \$)	\$85,136	\$72,815
Inflation adjusted costs (1992 \$) ⁱ	\$80,562	\$68,903

Table 5.14. Summary of intervention costs (Spoth et al., 2002)

^aIntervention costs include those incurred by additional participating families who did not complete study assessments. ^bAmong pretested families, 117 participated in the ISFP. ^cAmong pretested families, 124 participated in the PDFY. ^dIncludes trainer travel, time and lodging; facilitator travel, time and meals; training materials (manuals and videotapes). ^eFor ISFP, duplication of materials in facilitator manuals and supplies; for PDFY, cost of family activity books. ^fGrocery and video rental coupons, snacks. ^gLocal staff providing access to rooms where training held, setting up rooms, making arrangements for audiovisual equipment. ^hPreparation, facilitator time, travel. ⁱAdjustment accounts for increases in the Consumer Price Index of 3.0% from 1992 to 1993 and 2.6% from 1993 to 1994 (Bureau of Labor Statistics, 2000).

Reproduced from Spoth and colleagues (2002)

5.6.1.4 Summary of cost-effectiveness data

The cost per case prevented was calculated by dividing the number of alcohol-use disorder cases prevented per 100 families into the cost per 100 families treated. For the ISFP, the estimated cost for each case of alcohol-use disorder prevented was \$12,459 (\$68,856/5.53). Corresponding costs for the PDFY condition were \$20,439 (\$55,567/2.72).

The present value of the total lifetime benefit realised by the prevention of a single alcohol-use disorder was calculated to be \$119,633. The benefit-cost ratio equalled the benefit per case prevented, divided by the cost per case prevented. For the ISFP the benefit-cost ratio was 9.60 (i.e. \$9.60 was saved for every dollar invested). For PDFY, the benefit-cost ratio equalled 5.85.

The net benefit in the ISFP condition was \$5,923 per family (0.0553 cases prevented per family treated x \$119,633 benefit per case prevented - \$689 in intervention costs per family treated). The corresponding net benefit in the PDFY condition was \$2,697 per family (0.0272 × \$119,633 - \$556).

5.6.1.5 Comments

The economic evaluation answered a well-defined question, evaluating the cost-effectiveness and net benefits of two brief, family-focused interventions, the ISFP and PDFY, compared to a minimal intervention approach. Effectiveness estimates were drawn from a good-quality cluster RCT and used to estimate the impact of the programme on adult alcohol use disorder. The authors reported that the methods used in their analyses were likely to have resulted in a conservative estimate of the number of future alcohol use disorder cases. Costs were considered from a societal perspective, but although

the costs included were reported there was little description of how costs were valued. The generalisability of the study to a UK context is unclear as the data used in the evaluation is based on studies conducted in the USA. In addition, projected alcohol use disorder rates were calculated based on US population data.

5.6.2 Summary and evidence statements

One study (Spoth et al., 2002) was identified that met the criteria for inclusion in the review of published economic evaluations. Spoth and colleagues (2002) examined the cost-effectiveness and cost benefits of two family-focused interventions, the ISFP and the PDFY. Evaluations of the effectiveness of these programmes were identified and are included in Section 5.3.

Overall the net benefit was \$5,923 per family for the ISFP and \$2,697 per family for PDFY. The benefit-cost ratios were 9.60 and 5.85, indicating that for every \$1 spent on the ISFP and PDFY, \$9.60 and \$5.85, respectively, were saved in medical costs.

Evidence statement 6

There is moderate evidence from one economic evaluation study¹ to suggest that programmes delivered to families may be cost-effective and cost saving. This evidence may be of limited applicability to a UK context because cost and benefit estimates were based on data from studies conducted in the USA.

¹ Spoth et al., 2002 (CEA/CBA +)

6 Programmes targeting sexual health

A total of 49 articles met the criteria for inclusion in the review of community-based programmes targeting young people's sexual health. Nine articles were systematic reviews and/or meta-analyses, 20 articles reported on studies that examined intervention or programmes delivered within social, healthcare and community settings, 15 articles reported on studies that examined programmes or interventions delivered to families or parents, two articles reported on studies that examined interventions or programmes that involved the wider community or mass media, and three articles reported on studies which examined interventions for vulnerable young people.

6.1.1 Systematic reviews and meta-analyses

6.1.2 Overview of evidence identified

Nine systematic reviews and meta-analyses were identified that examined the effectiveness of interventions and programmes that targeted young people's sexual health behaviours. One review (Arnold and Rotheram-Borus, 2007). All nine reviews examined interventions and programmes delivered across a range of settings.

6.1.3 Quality assessment

Four reviews (DiCenso et al., 2002; Franklin et al., 1997; Underhill et al., 2007; 2008) were rated good quality (++ rating) and covered all of the criteria on the quality assessment tool indicating a high standard of methodology. Four reviews (Guyatt et al., 2000; Pedlow & Carey, 2003; Robin et al., 2004; Sales et al., 2006) were rated moderate quality (+ rating), which although generally good quality, did not assess and/or take into account the quality of the included studies. One review (Arnold and Rotheram-Borus, 2009) was rated poor quality as the authors did not provide a detailed description of the methodology used, and it was not clear whether the literature searches undertaken were sufficiently rigorous. In addition, this review neither assessed nor took into account the quality of the included studies.

6.1.4 Findings

DiCenso and colleagues (2002) concluded from the findings of their review that primary prevention strategies do not delay the initiation of sexual intercourse or improve the use of birth control among young men and women. However, across three reviews (Franklin et al., 1997; Robin et al., 2004; Sales et al., 2006) there was consensus that HIV prevention and sexual risk reduction programmes were effective in increasing condom use and reducing pregnancy, but that there was a limited impact of these programmes on adolescent sexual activity. There were different findings across these three reviews in relation to the impact of setting on effectiveness. Franklin and colleagues (1997) reported that community-based programmes were more effective than school-based programmes in terms of increasing contraceptive use and reducing pregnancy, and that clinic-based programmes more effectively influenced contraceptive behaviour outcomes than non-clinic programmes. However, Robin and colleagues (2004) concluded from the findings of their review that no key study features or programme characteristics clearly distinguished studies with positive, null, and negative effects from

each other. Sales and colleagues (2006) didn't examine the relative comparability of setting but did highlight features of successful community- and clinic-based interventions and programmes. Successful community-based interventions were theoretically based, tailored to the target population, implemented by trained facilitators, and the content was diverse and delivered using a wide variety of methods. The characteristics of successful clinic-based programmes included a focus on a single gender or ethnic group, HIV/STI education with skills building activities (e.g. condom application), condom negotiation and sexual communication components, and personalised risk assessments. The reviews conducted by Underhill and colleagues (2007; 2008) and Pedlow and Carey (2003) leant further support to the evidence that community-based programmes can affect risk behaviours.

Guyatt and colleagues (2000) compared the results of RCTs and observational studies of interventions to prevent adolescent pregnancy in order to explore study design as a possible determinant of outcome. They found that observational studies yielded systematically greater estimates of treatment effects than RCTs. The authors cautioned that where feasible, recommendations for practice should be derived from randomised trials.

Arnold & Rotheram-Borus (2009) included evaluations of six programmes in their review of HIV prevention programmes for young homeless people. The authors reported that all six programmes identified were relatively intensive, including a range of 5 to 35 sessions (minimum of 5) and all programmes supported the notion that homeless youth are in need of intensive and prolonged help. All six programmes were designed as adjuncts to existing services, and intervention approaches included a runaway shelter, services delivered by mental health providers or social services and community outreach. The authors found that programmes which demonstrated success in recruiting and retaining youth were present/future-oriented, skill-based interventions aimed at increasing the youth's ability to reduce behaviours that lead to HIV.

6.1.5 Summary and evidence statements

Nine systematic reviews and meta-analyses were identified that examined the effectiveness of interventions and programmes across a range of settings and populations that targeted young people's sexual health behaviours. One review (Arnold & Rotheram-Borus, 2007) focused on interventions and programmes that targeted sexual risk taking among young homeless people.

Findings from six reviews (Franklin et al., 1997; Pedlow & Carey, 2003; Robin et al., 2004; Sales et al., 2006; Underhill et al., 2007; 2008) indicated that community-based programmes can affect sexual risk behaviours of young people. In particular, HIV prevention and sexual risk reduction programmes were effective in increasing condom use and reducing pregnancy (Franklin et al., 1997; Robin et al., 2004; Sales et al., 2006). However, they may have a limited impact on adolescent sexual activity. According to Sales and colleagues (2006) successful community-based interventions were theoretically based, tailored to the target population, implemented by trained facilitators, and the content was diverse and delivered using a wide variety of methods.

Evidence statement 7

- 7 (b) There is strong evidence from five systematic reviews and meta-analyses¹ to suggest that interventions and programmes delivered in a range of community settings can have a positive impact on young people's sexual risk behaviours, in particular, condom use and pregnancy.
- 7 (c) There is strong evidence from one systematic review² to suggest that effective communitybased interventions and programmes are: (1) theoretically based; (2) tailored to the target population, (3) implemented by trained facilitators; (4) based on diverse content; and (5) delivered using a wide variety of methods.
- 7 (d) There is strong evidence from one systematic review² to suggest that effective clinic-based programmes include: (1) a focus on a single gender or ethnic group; (2) HIV/STI education with skills building activities (e.g. condom application); (3) condom negotiation and sexual communication components; and (4) personalised risk assessments.
- ¹ Franklin et al., 1997 (SR +); Pedlow & Carey, 2003 (SR +); Robin et al., 2004 (SR +); Sales et al., 2006 (SR ++); Underhill et al., 2007; 2008 (both SR ++)

² Sales et al., 2006 (SR ++)

Author (Year)	Design	Inclusion/exclusion	Number of studies	Findings
Arnold & Rotheram- Borus, 2009	SR -	Programmes that had an HIV prevention focus and targeted homeless youth.	6 studies	Programmes with demonstrated success in recruiting and retaining youth are present/future-oriented, skill-based interventions aimed at increasing the youth's ability to reduce behaviours that lead to HIV.
DiCenso et al., 2002	SR ++	Primary prevention strategies aimed at delaying sexual intercourse, improving use of birth control and reducing incidence of unintended pregnancy in adolescents	26 studies	Primary prevention strategies do not delay the initiation of sexual intercourse or improve the use of birth control among young men and women.
Franklin et al., 1997	SR +	Studies with a primary focus on the primary prevention of adolescent pregnancy	32 studies	Programmes had a greater effect on contraceptive use and pregnancy rates, than on sexual activity. Contraceptive knowledge building programmes and contraceptive distribution programmes are more effective than other sex education programmes (e.g. abstinence-only programmes).
Guyatt et al., 2000	SR +	Randomised trials and observational studies of interventions to prevent adolescent pregnancy	30 studies	Summary odds ratios for the observational studies showed a significant intervention benefit (p<0.05) while the randomised trials did not show a benefit for any outcome in either females or males. Observational studies yield systematically greater estimates of treatment effects than randomized trials of adolescent pregnancy prevention interventions.
Pedlow & Carey, 2003	SR +	HIV risk reduction interventions targeting teenagers	22 studies	HIV risk reduction interventions have been shown to be effective but are associated with small effect sizes. Most effective studies emphasised a theoretical framework, most often Social Cognitive Theory. Interventions with multiple sessions or long doses have been no more successful than those with shorter doses.
Robin et al., 2004	SR +	Behavioural interventions targeting HIV, STI, and pregnancy for young people aged 13 years or younger	39 studies; 2 community-based; 2 clinic-based; 1 home based; 4 mixed settings	Programmes that produced positive effects: (1) used trained adult facilitators, and two other programmes with positive effects also used trained peer facilitators; (2) included content that was specific to reducing sexual risk behaviour such as refusal of unwanted sex and condom-use skills; and (3) commonly employed interactive and participatory educational strategies.
Sales et al., 2006	SR ++	School, community or clinic based interventions or interventions developed for special populations	24 studies; 12 clinic-based; 9 special populations; 5 community-based	Most successful community based interventions were theoretically based, tailored to the target population, implemented by trained facilitators, and the content was diverse and delivered using a wide variety of methods.

Table 6.1. Summary of findings from systematic reviews and meta-analyses: Sexual health

Author (Year)	Design	Inclusion/exclusion	Number of studies	Findings
Underhill et al., 2007	SR ++	Abstinence-only interventions with HIV prevention as stated goal	2 family-based studies; 1 community-based study	No evidence that abstinence-only programs can effectively encourage abstinent behaviour; although programs did not appear to cause harm. Overall, programmes were ineffective for preventing or decreasing sexual activity.
Underhill et al., 2008	SR ++	Abstinence-plus programs designed to influence behaviour change on at least one outcome measure related to HIV transmission	21 community- based studies; 5 family-based studies	No conclusive evidence that abstinence-plus programmes can reduce STI incidence and limited evidence that abstinence-plus programmes can reduce pregnancy incidence; however, direction of effects consistently favoured abstinence-plus programmes. Programmes had mixed effects on sexual behaviour. Authors reported that the results from the studies of community-based programmes suggest that a number of community-based abstinence-plus programs can affect risk behaviour

6.2 Programmes delivered within social and community settings

6.2.1 Overview of evidence identified

Fourteen studies examined interventions or programmes delivered within social and community setting that focused on preventing risky sexual behaviours among young people. These studies were conducted within various community-based agencies including youth agencies (Di Noia & Schinke, 2007; Elliott et al., 1996; Ferguson, 2000; Kipke et al., 1993; Philliber, 2002; Postrado & Nicholson, 1992), schools (Jemmott et al., 1992; 1998; Pearlman et al., 2002; Smith et al., 2000; Villarruel et al., 2006), recreation centres (Stanton et al., 1996; 1997) and a housing development (Sikkema et al., 2005).

Of the 14 studies, seven (Di Noia & Schinke, 2007; Jemmott et al., 1992; Jemmott et al., 1998; Postrado & Nicholson, 1992; Stanton et al., 1996; 1997; Villarruel et al., 2006) examined interventions based on group education sessions targeting risky sexual behaviours and two (Kipke et al., 1993; Sikkema et al., 2005) examined skills-based training interventions. Three studies (Ferguson, 2000; Pearlman et al., 2002; Smith et al., 2000) examined peer-led interventions, including a peer counselling programme (Ferguson, 2000) and peer-led leadership programmes (Pearlman et al., 2002; Smith et al., 2000), respectively. Philliber and colleagues (2002) examined the Children's AID Society (CAS) Carrera programme that focused on youth development for young people enrolled in after school programmes and Elliott and colleagues (1996) examined a theatre production designed to inform young people about HIV.

The theoretical basis for intervention was not reported in seven studies (Elliott et al., 1996; Ferguson, 2000; Jemmott et al., 1992; Pearlman et al., 2002; Philliber et al, 2002; Postrado & Nicholson, 1992; Sikkema et al., 2005). Interventions examined in five studies (Di Noia & Schinke, 2007; Jemmott et al., 1998; Kipke et al., 1993; Smith et al., 2000; Villarruel et al., 2006) were based on multiple theories. The most commonly applied theories were the theory of reasoned action and the theory of planned behaviour. The intervention examined by Stanton and colleagues (1996; 1997) was based on a single theory, protection motivation theory.

The number of participants recruited ranged from 74 (Smith et al., 2000) to 1,172 (Sikkema et al., 2005) adolescents. A range of different age groups were targeted across the included studies. The youngest age targeted was 9 years (Stanton et al., 1996; 1997) and the oldest age targeted was 18 (Villarruel et al., 2006). Eight studies (Di Noia & Schinke, 2007; Elliott et al., 1996; Ferguson, 2000; Jemmott et al., 1992; Kipke et al., 1993; Pearlman et al., 2002; Postrado & Nicholson, 1992; Sikkema et al., 2005) were based on short-term follow-up only (<6 months), three studies (Jemmott et al., 1998; Smith et al., 2000; Villarruel et al., 2006) were based on medium-term follow-up (up to 12 months) and two studies (Philliber et al., 2002; Stanton et al., 1996; 1997) reported long term results (>12 months).

6.2.2 Quality assessment

Of the 14 studies identified, nine were RCTs (Di Noia and Schinke, 2007; Jemmott et al., 1992; 1998; Kipke et al., 1993; Philliber et al., 2002; Sikkema et al., 2005; Stanton et al., 1996; 1997; Villarruel et al., 2006), three were NRCTs (Elliott et al., 1996; Ferguson, 2000; Pearlman et al., 2002), and two were CBA studies (Postrado & Nicholson, 1992; Smith et al., 2000). Of the nine RCTs, four were based on cluster randomisation (Di Noia and Schinke, 2007; Sikkema et al., 2005; Stanton et al., 1996; 1997); two studies (Di Noia and Schinke, 2007; Sikkema et al., 2005) randomised at the community site level and two studies (Stanton et al., 1996; 1997) of the Focus on Kids programme randomised groups of friends. Two well-reported and conducted RCTs (Jemmott et al., 1998; Villarruel et al., 2006) based on individual randomisation, were rated strong for quality. These studies appropriately allocated participants to intervention and control conditions, reported a range of relevant and reliable outcomes and conducted appropriate analyses. Seven RCTs (Jemmott et al., 1992; Kipke et al., 1993; Philliber et al., 2002; Di Noia and Schinke, 2007; Sikkema et al., 2005; Stanton et al., 1996, 1997), including four cluster RCTs were rated moderate quality. Although these studies were generally well-reported and reported appropriate methods for allocation, and relevant and reliable outcomes, none of these studies reported whether they were sufficiently powered to detect an intervention effect and only one (Stanton et al., 1996) reported that an ITT analyses had been undertaken. The quality of the three NRCTs was moderate and they were generally well reported. However, the study by Elliott et al (1996) was subject to large losses to follow-up and only reported short-term follow-up and was consequently rated poorly. The two CBA studies (Postrado and Nicholson, 1992; Smith et al., 2000) were rated poorly. The study conducted by Postrado and Nicholson (1992) was at risk of bias as participants self-selected into the intervention and control conditions, and the method of allocation was not clear in the study by Smith et al (2000).

6.2.3 Findings

6.2.3.1 Knowledge and understanding

Ten studies (Di Noia and Schinke, 2007; Elliott et al., 1996; Ferguson, 2000; Jemmott et al., 1992; 1998; Kipke et al., 1993; Pearlman et al., 2002; Philliber et al., 2002; Smith et al., 2000; Stanton et al., 1996) examined intervention effects on knowledge across a range of follow-up periods. For four of the five studies that examined group education sessions or skills-based interventions there were positive intervention impacts on knowledge. The Keepin' it Safe CD-ROM intervention (Di Noia and Schinke, 2007), which specifically targeted females, was shown to have improved knowledge at post-test among intervention participants compared to wait-list controls (p<0.001). At post-test and the 3-month follow-up, black male adolescents who participated in the BPBR programme (Jemmott et al., 1992) had greater knowledge about AIDS than controls who received a career opportunities intervention (p<0.001 and p<0.01, respectively). In a second evaluation of the BPBR programme (Jemmott et al., 1998), African American male and female adolescents who received a safer sex education version of the programme had significantly higher condom use knowledge compared to those who received the control or an abstinence only version of the programme (both p<0.001). Participants in both the abstinence and safer sex intervention groups reported significantly greater knowledge about HIV risk

reduction than controls (both p<0.001). Kipke and colleagues (1993) reported that adolescents who participated in the ARREST programme had significantly higher levels of knowledge at post-test than controls (p<0.001). Only one programme failed to have an impact on knowledge. Stanton et al (1996) found that there was no difference in knowledge at either the 6- or 12-month follow-up among African American adolescents who participated in the Focus on Kids programme and controls who received weekly HIV prevention sessions.

Three studies examined peer-led interventions. There was no effect of a peer counselling programme for African American females (Ferguson et al., 2000) on knowledge at post-test, and at the 3-month follow-up, control participants reported a significant increase in knowledge compared to the intervention group (p<0.01). Pearlman et al (2002) found an effect of a peer leadership intervention on knowledge among the peer leaders themselves. At 9-months, new peer leaders reported significantly higher levels of knowledge about HIV/AIDS than comparison youth (p<0.01). In addition, Smith et al (2000) found that a peer leadership intervention, Students Together Against Negative Decisions (STAND), had a significant effect on participants knowledge of risk behaviours relative to comparison students (p<0.05). There were no effects of a theatre production on participants' knowledge about HIV (Elliott et al., 1996). Philliber and colleagues (2002) found that gains in knowledge were significantly greater among Carrera programme participants than controls (p<0.001).

6.2.3.2 Attitudes and values

Seven studies (Di Noia and Schinke, 2007; Elliott et al., 1996; Jemmott et al., 1992; 1998; Kipke et al., 1993; Stanton et al., 1996; Smith et al., 2000) examined programme effects on attitudes and values. This included four studies (Di Noia and Schinke, 2007; Jemmott et al., 1992; 1998; Stanton et al., 1996) that examined interventions based on group education sessions targeting risky sexual behaviours and one study (Kipke et al., 1993) that examined a skills-based intervention.

Di Noia and Schinke (2007) examined a CD-ROM mediated HIV prevention intervention for adolescent females. At post-test, intervention participants reported higher scores on the following attitudinal measures: perceived vulnerability to HIV (p<0.01), perceived efficacy and enjoyment of condoms (p<0.01 and p<0.05, respectively), and perceived efficacy and enjoyment of abstinence (p<0.01 and p<0.001, respectively). There was no difference between groups on the measure of partner norms or participants' self-efficacy for low-risk activities. At post-test, compared to a career opportunities intervention, Black male adolescents who participated in the BPBR programme (Jemmott et al., 1992) expressed less favourable attitudes toward risky sexual behaviours (p<0.01) and reported weaker intentions to engage in such behaviours (p<0.001). At 3-months, BPBR participants reported weaker intentions to engage in risky sexual behaviour in the next 3 months compared to control participants (p<0.01), but there was no difference in attitudes regarding risky sexual behaviours. In a further study of the BPBR programme, Jemmott and colleagues (1998) examined two versions of the programme. At post-test, compared to participants in the control group and those who received the safer-sex intervention, adolescents in the abstinence group believed more strongly that practicing abstinence would prevent pregnancy and AIDS (p<0.001 and p<0.05, respectively), expressed less favourable attitudes toward sexual intercourse (p<0.001 and p<0.01,

respectively) and reported weaker intentions of having sexual intercourse (both p<0.05) in the next 3 months. There was no difference between abstinence and control participants on attitudinal measures related to condom use. However, participants who received the safer sex intervention were more likely than participants in the control and abstinence groups to report condom prevention beliefs (both p<0.001), condom hedonistic beliefs (both p<0.001), condom availability control beliefs (p<0.05 and p<0.001, respectively). They also reported significantly higher impulse beliefs than controls and higher self-efficacy to use condoms (both p<0.05, respectively). There was no effect of either the abstinence or safer sex intervention on participant's intentions to use condoms compared to controls. At post-test, adolescents who received AIDS education and skills training in the ARREST programme (Kipke et al., 1993) reported a significant decrease in negative attitudes towards AIDS compared to the control group, (p<0.05) and an increase in the perception that adolescents are at risk of becoming HIV infected (p<0.01). At the 6-month follow-up, compared to control participants, African American adolescents who participated in Focus on Kids (FOK; Stanton et al., 1996) were significantly more likely to report an intention to use condoms (p<0.05). However this difference was not apparent at the 12 month follow-up. The authors also examined perceptions, finding that intervention participants perceived greater peer use of condoms (p<0.05) and perceived themselves as more vulnerable to HIV infection at the 6-month, but not the 12-month follow-up. There were no differences between intervention and control students at either follow-up on other attitudinal measures, which included perceptions of extrinsic and intrinsic rewards relevant to condom use, vulnerability and self-efficacy.

Elliott and colleagues (1996) examined the effects of a theatre production designed to teach young people about HIV. There was no impact of the intervention on attitudes at post-test or 2-month follow-up. Smith et al (2000) examined the effects of a peer leadership intervention, but at follow-up there were no differences on the HIV Prevention Attitude Scale or on the Condom Attitude Scale between intervention and control participants. However, STAND participants had significantly greater gains in condom use self-efficacy compared to the comparison group (p<0.01), but there was no difference in refusal skills self-efficacy.

6.2.3.3 Personal and social skills

Three studies (Di Noia and Schinke, 2007; Kipke et al., 1993; Smith et al., 2000) examined programme effects on skills. Di Noia and Schinke (2007) examined the effects of a CD-ROM intervention on sexual assertiveness and sexual communication. Females who received the intervention reported higher scores at post-test on sexual assertiveness than wait-list controls (p<0.001), but there was no difference between the groups in levels of communication. Kipke and colleagues (1993) found that compared to controls, ARREST participants had increased behavioural skills at post-test for giving a reason for refusing to engage in risk-related activities (p<0.001) and for proposing alternative lower risk activities (p<0.001). Young people who participated in the peer leadership programme, STAND (Smith et al., 2000), were significantly more likely than comparison students to report speaking with their friends about birth control/condoms and STIs (both p<0.01), but there was no difference in the frequency of conversations with parents or other adults, or on the Dyadic Sexual Communication Scale or Health Protection Communication Scale.

6.2.3.4 Health and social outcomes related to sexual health

Eight studies (Jemmott et al., 1992; Jemmott et al., 1998; Kipke et al., 1993; Postrado and Nicholson, 1992; Stanton et al., 1996; 1997; Sikkema et al., 2005; Villarruel et al., 2006) that tested the effects of interventions based on group education sessions or skills training examined their effects on health outcomes related to sexual health. Controlling for pre-intervention behaviour, at 3-months follow-up, black adolescents who participated in the BPBR programme (Jemmott et al., 1992) were less likely than controls to report having engaged in risky sexual behaviours (p<0.01). Across different sexual behaviours the effects of the intervention were fairly consistent. There was no difference between groups in whether participants had sex, but intervention participants reported having sex on fewer days, with fewer women, and with fewer women who were involved in sexual relationships with other men. Intervention participants also reported fewer occasions of sex without a condom and were less likely to report having had anal intercourse with women. In a further evaluation of the effects of the BPBR programme, Jemmott and colleagues (1998) examined an abstinence and safer sex version of the programme. At 3-months follow-up, abstinence intervention participants were significantly less likely to report having sexual intercourse compared to controls (OR 0.45; 95% CI 0.23, 0.86; p<0.05) but not safer sex intervention participants (OR 0.54; 95% CI 0.28, 1.07). At the 3-month follow-up, those who received the safer sex intervention were more likely to report consistent condom use relative to the control group (OR 3.38; 95% CI 1.25, 9.16; p<0.05) and the abstinence group (OR 3.10; 95% CI 0.99, 9.73; p<0.05). In addition, safer sex intervention participants reported a higher frequency of condom use than controls (p<0.05), were less likely to report having unprotected sexual intercourse (OR 0.35; 95% CI 0.13, 0.95; p<0.05) and reported fewer days of unprotected intercourse (p<0.05). At the 6-month follow-up there were no significant differences between adolescents in the abstinence group and adolescents in the control or safer sex group on any of the sexual behaviour measures (in past 3 months), except that adolescents in the safer sex group reported a higher frequency of condom use relative to controls (p<0.05). At 12-months follow-up, compared to adolescents in the control group, there was a higher frequency of condom use among adolescents in both the abstinence group (p<0.05) and the safer sex group (p<0.01). Postrado and Nicholson (1992) examined interventions targeting young females enrolled in Girls Incorporated member organisations. Girls who did not participate in the Growing Together programme were more likely to have initiated sexual intercourse at post-test than those who did participate (OR 2.6; p<0.05). Participation in the Will Power Won't Power programme was not associated with initiation of sexual intercourse. There was no difference in risk-related sexual behaviours between participants who received the ARREST education and skills training programme (Kipke et al., 1993) and the control group on the following measures: number of sexual encounters, number of sexual partners, and use of condoms. Sikkema and colleagues (2005) examined the effects of an HIV prevention intervention, with and without the addition of a community-level intervention. At long-term follow-up, adolescents who received the additional community intervention were more likely to have remained abstinent compared to the comparison group (adjusted OR 1.97; 95% CI 1.06, 3.67; p<0.05), although the difference between community and workshop participants was not significant (OR 1.72; 95% CI 0.94, 3.16). In addition, compared to the comparison group, condom use rates were higher in both the community (OR 2.50;

95% CI 1.01, 6.22) and workshop (OR 2.23; 95% CI 0.99, 5.03) condition groups. The difference between the community and workshop groups was not significant. Stanton and colleagues (1996) examined the effects of the Focus on Kids programme on self-reported condom use. At the 6-month follow-up, condom use was significantly greater overall among intervention participants than control participants (p<0.05), but at 12-months this difference was no longer significant. Further analysis of the data up to the 18 months follow-up found that at 6- and 18-months follow-up, control youth were more likely than intervention youths to have engaged in unprotected sex, and cumulatively in the post-intervention period, intervention youth were less likely to have engaged in unprotected sex than control youths. Villarruel and colleagues (2006) examined the ¡Cuidate! programme, an adaptation of the BPBR programme for Latino adolescents. At 12-months follow-up, adolescents who participated in the programme were less likely to report sexual intercourse (OR 0.66; 95% CI 0.46, 0.96), multiple partners (OR 0.53; 95% CI 0.31, 0.90), fewer days of unprotected intercourse (RR 0.47; 95% CI 0.26, 0.84) and more likely to report using condoms consistently (OR 1.91; 95% CI 1.24, 2.93). There were no significant effects of the intervention on condom use at last sex (OR 1.45; 95% CI 0.97, 2.15) or the proportion of days of protected sex.

Three studies (Ferguson, 2000; Pearlman et al., 2002; Smith et al., 2000) examined the effects of peer-led programmes on health and social outcomes related to sexual health. Ferguson (2000) examined the impact of peer counselling in a pregnancy prevention programme. At the 3-month follow-up, none of the participants in either the intervention or control group had become pregnant. There was no significant delay in sexual intercourse for participants in either group and there was no difference between groups in the use of effective methods of contraception. There were also limited effects of a peer leadership intervention examined by Pearlman and colleagues (2002). At 9-months follow-up, there were no significant differences between new peer leaders and a comparison group on a measure of sexual risk taking behaviour. Smith and colleagues (2000) examined STAND, a peer leadership programme targeting high school students. At the 8-month follow-up, there was no significant difference between the intervention and comparison groups in the number of participants who were 'non-virgins' at follow-up. However, STAND participants were more likely to be 'consistent' condom users (p<0.05). There was no difference in the number of participants who reported condom use at last intercourse or in the number of condom-protected or unprotected instances of intercourse. In addition, there was no difference in the number of participants who reported being involved in a pregnancy, but STAND participants were less likely to have been diagnosed with an STI (p<0.01). Alcohol and other drug use in conjunction with intercourse were not frequently reported in either the intervention or comparison group.

Philliber and colleagues (2002) examined the effects of the CAS-Carrera youth development programme which targeted disadvantaged adolescents. At the end of the three year programme, the odds of becoming pregnant (or causing pregnancy) were significantly reduced among female, but not male, programme participants compared with controls (OR 0.31; p<0.01). Female programme participants were also less likely to be sexually active (OR 0.52; p<0.05) and were more likely to have

used a condom and a hormonal method at last intercourse (OR 2.37; p<0.05). There were no significant programme effects on males.

There were no significant effects of a theatre production on the number of participants who reported having unprotected sex in the previous 2-months (Elliott et al., 1996). However, at the 2-month follow-up, significantly more theatre participants than seminar participants reported changing their behaviour in response to the intervention (p<0.01). Those reporting a behavioural change in both groups said that they had become more cautious about sex or at least bought and carried condoms more than before.

6.2.4 Summary and evidence statements

Fourteen studies examined interventions or programmes delivered within social and community setting that focused on preventing risky sexual behaviours among young people. Nine studies (Di Noia and Schinke, 2007; Jemmott et al., 1992; Jemmott et al., 1998; Kipke et al., 1993; Postrado and Nicholson, 1992; Sikkema et al., 2005; Stanton et al., 1996; 1997; Villarruel et al., 2006) examined group education sessions or skills-based training interventions. Three studies (Ferguson, 2000; Pearlman et al., 2002; Smith et al., 2000) examined peer-led interventions, including a peer counselling programme (Ferguson, 2000) and peer leader leadership programmes (Pearlman et al., 2002; Smith et al., 2000), respectively. Philliber and colleagues (2002) examined the CAS-Carrera programme that focused on youth development for disadvantaged young people enrolled in after school programmes and Elliott et al (1996) examined a theatre production designed to inform young people about HIV.

Across four studies (Di Noia and Schinke, 2007; Jemmott et al., 1992; 1998; Kipke et al., 1993), that examined group education sessions or skills-based training interventions there were positive intervention effects on knowledge and understanding over the short- to medium-term. In addition, the three-year, Carrera programme (Philliber et al., 2002) had a positive impact on knowledge. There was no effect of a peer counselling intervention (Ferguson, 2000) on knowledge, but two peer leadership interventions (Pearlman et al., 2002; Smith et al., 2000) had positive effects on levels of knowledge among the peer leaders themselves.

Seven studies (Di Noia and Schinke, 2007; Elliott et al., 1996; Jemmott et al., 1992; 1998; Kipke et al., 1993; Stanton et al., 1996; Smith et al., 2000) examined intervention effects on a range of attitudes and values related to sexual health. Short-term decreases in intentions to engage in risky sexual intercourse were reported in the study of the BPBR programme which targeted black male adolescents (Jemmott et al., 1992) and an abstinence-based version of the programme resulted in short-term reductions in intentions to engage in any sexual intercourse. Across three studies (Di Noia and Schinke, 2007; Kipke et al., 1993; Stanton et al., 1996) there were short-term increases in intervention participants' perception of their vulnerability to HIV infection. However, this effect was not sustained in the medium-term (Stanton et al., 1996). Two studies (Elliott et al., 1996; Smith et al., 2000) found no effects of a theatre production intervention or peer leadership intervention, respectively, on HIV attitudes at follow-up. Across four studies (Di Noia and Schinke, 2007; Jemmott

et al., 1998; Smith et al., 2000; Stanton et al., 1996) there were indications of positive intervention effects on attitudes and values related to condom use. However, these effects did not appear to be consistent and were not maintained over the medium-term (Stanton et al., 1996). Two studies (Di Noia and Schinke, 2007; Jemmott et al., 1998) found short-term positive intervention effects of a CD-ROM mediated intervention and an abstinence-based version of the BPBR curriculum, respectively, on attitudes towards abstinence.

Intervention effects on personal and social skills were examined across three studies (Di Noia and Schinke, 2007; Kipke et al., 1993; Smith et al., 2000). A CD-ROM intervention (Di Noia and Schinke, 2007) and education and skills training programme (Kipke et al., 1993) had positive effects on behavioural skills but results from two studies (Di Noia and Schinke, 2007; Smith et al., 2000) presented mixed findings in relation to effects on communication.

Eight studies (Ferguson, 2000; Jemmott et al., 1992; 1998; Philliber et al., 2002; Postrado and Nicholson, 1992; Sikkema et al., 2005; Smith et al., 2000; Villarruel et al., 2006) examined intervention effects on sexual activity. Across five studies (Jemmott et al., 1992; 1998; Postrado and Nicholson, 1992; Sikkema et al., 2005; Villarruel et al., 2006) that examined group-based sessions and/or skills training, short- to medium-term effects on sexual intercourse were reported in four studies (Jemmott et al., 1992; Postrado and Nicholson, 1992; Sikkema et al., 2005; Villarruel et al., 2006), and one study (Jemmott et al., 1998) reported no programme effects. The Carrera programme (Philliber et al., 2002) had a positive effect on sexual activity among females but there were no effects of two peer interventions (Ferguson, 2000; Smith et al., 2000). Intervention effects on frequency of sexual intercourse and number of sexual partners was limited. Across four studies (Jemmott et al., 1998; Kipke et al., 1993; Pearlman et al., 2002; Villarruel et al., 2006) that examined these outcomes only one study (Villarruel et al., 2006) reported a positive intervention effect. Nine studies (Elliott et al., 1996; Ferguson, 2000; Jemmott et al., 1998; Kipke et al., 1993; Philliber et al., 2002; Sikkema et al., 2005; Smith et al., 2000; Stanton et al., 1996; Villarruel et al., 2006) examined intervention effects on contraceptive use. Across four of the five studies that examined group-based sessions and/or skills training, there were positive short-term intervention effects on measures of condom use (Jemmott et al., 1998; Sikkema et al., 2005; Stanton et al., 1996; Villarruel et al., 2006), and some evidence from two studies (Jemmott et al., 1998; Villarruel et al., 2006) of a positive intervention effect on frequency of unprotected intercourse. However, over the longer term, intervention effects appeared to diminish. There were no effects on a HIV theatre production (Elliott et al., 1996) or peer counselling intervention (Ferguson, 2000) on contraceptive use or frequency of unprotected sex, but the Carrera programme (Philliber et al., 2002) positively influenced both condom and hormonal contraceptive use among females. There was no effect of a peer counselling intervention (Ferguson, 2000) or peer leadership programme (Smith et al., 2000) on pregnancy rates, but again the Carrrea programme (Philliber et al., 2002) had a positive effect, with a reduction in pregnancies among intervention females.

Evidence statement 8

- 8 (a) There is moderate evidence from five RCTs, one NRCT and one CBA study¹ to suggest that group-based education and/or skills-based interventions, youth development programmes and peer leadership interventions delivered in social and community settings may have a positive short- to medium-term impact on knowledge and understanding related to sexual health. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 8 (b) There is inconsistent evidence from five RCTs, one NRCT and one CBA study² on which to determine the effects of interventions and programmes delivered in social and community settings on attitudes and values related to sexual health. There was moderate evidence from three RCTs³ to suggest that group-based education and/or skills-based interventions may have positive short-, but not long-term, effects on attitudes and values related to condom use. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 8 (c) There is weak evidence from two RCTs⁴ to suggest that group-based education and/or skillsbased interventions delivered in social and community settings may have a positive short-term impact on behavioural skills related to sexual health. There was no evidence on which to determine intervention effects on communication skills. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 8 (d) There is moderate evidence from four RCTs and one CBA study⁵ to suggest that group-based education and/or skills-based interventions may have limited effects on sexual activity. Although reductions in the likelihood of sexual intercourse were reported across four RCTs⁶ there was only evidence from one RCT⁷ of intervention effects on frequency of sexual intercourse or number of sexual partners. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 8 (e) There is weak evidence from four RCTs⁸ to suggest that group-based education and/or skillsbased interventions delivered in social and community settings may have positive short-term impacts on condom use and frequency of unprotected intercourse. There is weak evidence from one RCT⁹ to suggest that these effects may diminish over the medium term. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 8 (f) There is moderate evidence from one RCT¹⁰ to suggest that youth development programmes that target disadvantaged young people may have a positive impact on sexual behaviours among females, including sexual activity, condom use and pregnancy. This evidence may only be partially applicable because these studies were conducted in the USA and focused on

ethnic populations, specific to the USA.

- ¹ Di Noia and Schinke, 2007 (RCT +); Jemmott et al., 1992 (RCT +); Jemmott et al., 1998 (RCT ++); Kipke et al., 1993 (RCT +); Pearlman et al., 2002 (NRCT +); Philliber et al., 2002 (RCT +); Smith et al., 2000 (CBA -)
- ² Di Noia and Schinke, 2007 (RCT +); Elliott et al., 1996 (NRCT -); Jemmott et al., 1992 (RCT +); Jemmott et al., 1998 (RCT ++); Kipke et al., 1993 (RCT +); Stanton et al., 1996 (RCT +); Smith et al., 2000 (CBA -)
- ³ Di Noia and Schinke, 2007 (RCT +); Jemmott et al., 1998 (RCT ++); Stanton et al., 1996 (RCT +)
- ⁴ Di Noia and Schinke, 2007 (RCT +); Kipke et al., 1993 (RCT +)
- ⁵ Jemmott et al., 1992 (RCT +); Jemmott et al., 1998 (RCT ++); Postrado and Nicholson, 1992 (CBA -); Sikkema et al., 2005 (RCT +); Villarruel et al., 2006 (RCT ++)
- ⁶ Jemmott et al., 1992 (RCT +); Jemmott et al., 1998 (RCT ++); Sikkema et al., 2005 (RCT +); Villarruel et al., 2006 (RCT ++)
- ⁷ Villarruel et al., 2006 (RCT ++)
- ⁸ Jemmott et al., 1998 (RCT ++); Sikkema et al., 2005 (RCT +);Stanton et al., 1996 (RCT +); Villarruel et al., 2006 (RCT ++)
- ⁹ Stanton et al., 1996 (RCT +)
- ¹⁰ Philliber et al., 2002 (RCT +)

Table 6.2. Summary of programme content: programmes delivered in social and community settings	
Table 0.2. Summary of programme content. programmes derivered in Social and community settings	

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Di Noia and Schinke, 2007	RCT (cluster) +	USA n=204 females 29% Hispanic; 54% Black; 4% White or other; 15% NR 11-14 years	Youth agencies	Keepin' it Safe : Six weekly sessions; HIV/AIDS knowledge and perceived vulnerability to HIV infection, sexual decision making, self-efficacy, sexual communication and assertiveness, and risk reduction skills building.	Health Belief Model, theory of reasoned action, theory of planned behaviour, self-efficacy theory	CD-ROM
Elliott et al., 1996	NRCT -	UK n=10 projects Ethnicity NR mean 15-16 years	Youth agency	Theatre production designed to inform young people about HIV, especially the modes of transmission; explore attitudes and emotional issues associated with HIV and inform participants about safer sex practices, in particular proper condom use.	NR	Theatre company
Ferguson, 2000	NRCT -	USA n=63 females 100% African American 12-16 years	Youth agency	Eight weekly, 2-hour peer counselling sessions including group discussions and role play. Focus on sexual behaviour, reproduction, STIs, contraceptives and hygiene.	NR	Trained peer counsellors
Jemmott et al., 1992	RCT (individual) +	USA n=157 males 100% Black mean 15 years	School (weekends)	Be Proud! Be Responsible! (BPBR) : Five hour AIDS risk reduction course covering information about risks associated with intravenous drug use and specific sexual activities	NR	Trained facilitators
Jemmott et al., 1998	RCT (individual) ++	USA n=659 100% African American mean 12 years	School (weekends)	BPBR: Eight, 1-hour modules delivered over consecutive Saturdays; (1) abstinence intervention designed to increase HIV/STI knowledge, strengthen behavioural beliefs and skills supporting abstinence; (2) safer sex intervention designed to increase HIV/STI knowledge and condom use.	Social cognitive theory, reasoned action, theory of planned behaviour.	Adult facilitator or peer co-facilitator
Kipke et al., 1993	RCT (individual) +	USA n=87 59% Latino; 41% African American 12-16 years	Youth agencies	ARREST (AIDS Risk Reduction Education and Skills Training): Three training sessions; AIDS education; instruction on how to use condoms; and decision-making, communication and assertiveness skills training. Take home exercises.	Health Belief Model, Social Learning Theory	NR
Pearlman et al., 2002	NRCT +	USA n=168 36% Hispanic; 28% White; 17% Black; 18% other mean 15-16 years	School	Project Teen Health : peer leadership course; ongoing group work with an adult advisor to learn about HIV and related skills	NR	NR

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Philliber et al., 2002	RCT (individual) +	USA n=484 56% Black; 42% Hispanic; 2% other 13-15 years	Youth agencies	Children's AID Society-Carrera programme : Five days/week over the school year; Work-related intervention; academic component; comprehensive family life and sexuality education; arts component; individual sports component. Supplemented by mental health care and medical care.	NR	Community organisers
Postrado and Nicholson, 1992	CBA -	USA n=412 females 75% Black; 15% White; 11% Hispanic or other 12-17 years	Youth agency	Will Power/Won't Power (WPWP): 6 sessions; Group-building exercises, introduction to relationships and basic assertiveness skills Growing Together (GT): 5 sessions; one parents only session; other sessions focused on physical and emotional changes during puberty, anatomy of reproduction, myths and facts about sexuality and getting pregnant, and other related topics.	NR	NR
Sikkema et al., 2005	RCT (cluster) +	USA n=1,172 51% African American; 20% Asian; 10% East African; 5% White; 3% Hispanic; 11% other mean 15 years	Housing developments	Two, three-hour training sessions; (1) Workshop intervention; HIV/STI education, skills training, sexual negotiation skills, condom use skills, and risk behaviour self-management; (2) Community-level intervention; as workshop condition, followed by a multi-component community intervention	NR	NR
Smith et al., 2000	CBA -	USA n=74 58% African American; 39% White; 3% other mean 16 years	School (evenings)	STAND: Peer leadership programme; 36 hours over 4 months; team-building exercises, contraceptive demonstrations, visit to local health department, skills practice, visits from an AIDS specialist physician and nurse, and optional parent/teen activities	Diffusion of innovations theory, transtheoretical model	AIDS Education Specialist, middle school counsellor, college interns
Stanton et al., 1996; 1997	RCT (cluster) +	USA n=383 100% African American 9-15 years	Recreation centres	Focus on Kids : Eight weekly meetings; sessions focused on one or more Protection Motivation Theory constructs from difference perspectives. Facts regarding AIDS, STIs, contraception and human development were also provided, as were condoms.	Protection Motivation Theory	Interventionists
Villarruel et al., 2006	RCT (individual) ++	USA n=656 100% Latino 13-18 years	School (weekends)	¡Cuidate! : Eight hours over two consecutive Saturdays; abstinence and condom use were presented as culturally accepted and effective ways to prevent STIs.	Social cognitive theory, theories of reasoned action and planned behaviour.	Trained facilitators

					Outcomes			
Study	Rating	Intervention	Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills	
Di Noia and Schinke, 2007	RCT (cluster) +	Keepin' it Safe n=111	Wait list control n=93	PT (75%)	† ***	 t vulnerability to HIV** condom use efficacy** enjoyment of condoms* abstinence efficacy** enjoyment of abstinence*** 	↑ sexual assertiveness*** NS communication	
Elliott et al		HIV theatre	Health	PT (NR)	NS	NS attitudes	-	
1996	NRCT -	production n=132	seminars n=85	2 mo (int=34%; con=43%)	NS	NS attitudes	-	
Ferguson, 2000		NRCT - Peer counselling n=33	No	PT (NR)	NS	-	-	
	NRCT -		intervention n=30	3 mo (int=91%; con=73%)	↑ **	-	-	
Jemmott et al., 1992	RCT (individual) +	al) + Be Proud! Be Responsible! n=85	Career opportunities n=72	PT (NR)	1 AIDS***	 ↓ favourable attitudes toward risky sexual behaviours** ↓ intentions to engage in risky sexual behaviours*** 	-	
				3 mo (NR)	† AIDS**	NS favourable attitudes toward risky sexual behaviours** ↓ intentions to engage in risky sexual behaviours**	-	
Jemmott et al., 1998	RCT (individual) ++	Be Proud! Be Responsible! Safer sex, n=218	Health promotion control n=214	PT (NR)	↑ condom use*** ↑ HIV risk reduction***	 t condom prevention beliefs*** t condom hedonistic beliefs*** t condom availability control beliefs* t impulse control beliefs* NS negotiation skills beliefs NS technical skills beliefs t condom use self-efficacy* NS condom use intentions NS abstinence-related outcomes 	-	

Table 6.3. Programme delivered in social and community settings: effects on knowledge, attitudes and skills

Study	Rating	Intervention	Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills
Jemmott et al., 1998	RCT (individual) ++	Be Proud! Be Responsible! Abstinence, n=215	Health promotion control n=214	PT (NR)	NS condom use ↑ HIV risk reduction***	 ↑ abstinence prevention beliefs*** ↑ abstinence goal-attainment beliefs* ↓ attitudes toward sexual intercourse*** ↓ intentions to have sexual intercourse, next 3 mo* NS condom-related outcomes 	-
Jemmott et al., 1998	RCT (individual) ++	Be Proud! Be Responsible! Safer sex, n=218	Be Proud! Be Responsible! Abstinence, n=215	PT (NR)	↑ condom use*** ↑ HIV risk reduction***	 ↓ abstinence prevention beliefs* NS abstinence goal-attainment beliefs ↑ attitudes toward sexual intercourse** ↑ intentions to have sexual intercourse* ↑ condom prevention beliefs*** ↑ condom hedonistic beliefs*** ↑ condom availability control beliefs*** NS impulse control beliefs NS negotiation skills beliefs NS condom use self-efficacy NS condom use intentions 	-
Kipke et al., 1993	RCT (individual) +	ARREST n=41	Wait list control n=46	PT (99%)	† ***	 ↓ negative attitudes towards AIDS* ↑ perception that adolescents are at risk of becoming HIV infected 	1 behavioural skills***
Pearlman et al., 2002	NRCT +	Project Teen Health n=97	Usual sex education n=71	PT (NR)	† **	1 perception of oneself as a 'change agent'**	-
Philliber et al., 2002	RCT (individual) +	CAS-Carrera programme n=242	Regular youth programme n=242	3 yr (81%)	t¶	-	-

						Outcomes	
Study	Rating	Intervention	Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills
Smith et al., 2000	CBA -	STAND n=21	NR n=53	PT, 8 mo (NR)	↑ risk behaviour*	NS HIV prevention attitudes NS condom attitudes ↑ condom use self-efficacy** NS refusal skills self-efficacy	NS communication with parents, other adults or partners
Stanton et al., 1996; 1997	RCT (cluster) +	Focus on Kids) + n=206	Weekly HIV prevention sessions n=177	6 mo (79%)	NS	↑ condom use intentions* ↑ perception of peer condom use [¶] ↑ perceived vulnerability to HIV infection [¶]	-
				12 mo (73%)	NS	NS condom use intentions NS perception of peer condom use NS perceived vulnerability to HIV infection	-
*p<0.05; **p<0.0	01; ***p<0.001; ¹	^I p value not reporte	d; 1 increase relat	tive to compara	ator; ↓ decrease relative to	comparator; NS not significant; - outo	come not reported

					Sexual health outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs/ Conceptions	Other
Elliott et al., 1996	NRCT -	HIV theatre production n=132	Health education seminars n=85	2 mo (int=34%; con=43%)	-	-	NS unprotected sex	-	1 changed behaviour in response to intervention**
Ferguson, 2000	NPCT	Peer counselling	No	PT (NR)	NS delay in sexual intercourse	-	NS use of effective methods of contraception	NS pregnancy	-
	NKC1 -	n=33	n=30	3 mo (int=91%; con=73%)	NS delay in sexual intercourse	-	NS use of effective methods of contraception	NS pregnancy	
Jemmott et al., 1992	RCT (individual) +	BPBR n=85	Career opportunities n=72	3 mo (NR)	↓ risky sexual behaviour**		-	-	-
Jemmott et al., 1998	RCT	T BPBR	Health promotion	3 mo (int=99%; con=97%)	NS sexual intercourse	NS frequency of intercourse	<pre>1 consistent condom use* 1 frequency condom use* ↓ unprotected sex* ↓ frequency of unprotected sex*</pre>	-	-
	(Individual) ++	Sater sex, n=218	control n=214	6 mo (int=95%; con=99%)	NS sexual intercourse	NS frequency of intercourse	NS consistent condom use f frequency condom use* NS unprotected sex NS frequency of unprotected sex	-	-

Table 6.4. Programme delivered in social and community settings: effects on health and social outcomes related to sexual health

					Sexual health outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs/ Conceptions	Other
Jemmott et al., 1998	RCT (individual) ++	BPBR Safer sex, n=218	Health promotion control n=214	12 mo (int=94%; con=95%)	NS sexual intercourse	NS frequency of intercourse	NS consistent condom use f frequency condom use** NS unprotected sex NS frequency of unprotected sex	-	-
Jemmott et al., 1998	RCT (individual) ++	BPBR Abstinence, n=215	Health promotion control n=214	3, 6, 12 mo (int=93%; con=95%)	↓ sexual intercourse (3 mo only*)	NS frequency of intercourse	NS consistent condom use f frequency condom use (12 mo only*) NS frequency of unprotected sex	-	-
Jemmott et al., 1998	RCT (individual) ++	BPBR Safer sex, n=218	Be Proud! Be Responsible! Abstinence, n=215	3, 6, 12 mo	NS sexual intercourse	NS frequency of intercourse	 t consistent condom use (3 mo only*) NS frequency condom use NS frequency unprotected sex 	-	-
Kipke et al., 1993	RCT (individual) +	ARREST n=41	Wait list control n=46	PT (99%)	-	NS number of sexual partners NS number of sexual encounters	NS use of condoms	-	-
Pearlman et al., 2002	NRCT +	Project Teen Health n=97	Usual sex education n=71	PT (NR)	-	NS sexual risk taking behaviour	-	-	-

						Sex	ual health outcor	es			
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs/ Conceptions	Other		
Philliber et al., 2002	RCT (individual) +	Carrera programme n=242	Regular youth programme n=242	3 yr (81%)	↓ sexually active (females only [¶])	-	↑ condom use at last intercourse (females only [¶]) ↑ hormonal contraceptive at last intercourse (females only [¶])	↓ pregnancy (females only [¶])	-		
Postrado and	CBA -	WPWP n=257	No intervention n=155	PT (NR)	NS initiation of sexual intercourse	-	-	-	-		
1992		GT n=84	No intervention n=328	PT (NR)	↓ initiation of sexual intercourse*	-	-	-	-		
Sikkema et al., 2005	RCT (cluster) +	HIV prevention intervention Workshop, n=428 Community, n=392	Standard AIDS education session n=352	2 mo (65%)	1 abstinence (Community vs. control*)	-	1 condom use (community vs. control ¹ ; workshop vs. control ¹)	-	-		
Smith et al., 2000	CBA -	STAND n=21	NR n=53	PT, 8 mo (NR)	NS 'non- virgins'	-	1 consistent condom use*	↓ STI diagnosis** NS pregnancy	-		
Stanton et al.,	RCT	Focus on Kids	HIV prevention n=177	6 mo (79%)	-	-	1 condom use*	-			
1996; 1997	(cluster) +	n=206		12 mo (73%)	-	-	NS condom use	-	-		

Study						Se	xual health outcor	nes	
	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs/ Conceptions	Other
Villarruel et al., 2006	RCT (individual) ++	;Cuidate! n=312	Health promotion control n=344	12 mo (84%)	↓ sexual intercourse	↓ multiple partners	↑ consistent condom use NS condom use at last intercourse ↓ days of unprotected intercourse NS proportion of days of unprotected sex	-	-
*p<0.05; **p<0.0	01; ***p<0.001; ¹	ⁱ p value not reporte	d; 1 increase rela	tive to compara	ator; ↓ decrease r	elative to compara	tor; NS not significa	ant; - outcome not i	reported

6.3 Programmes delivered within healthcare settings

6.3.1 Overview of the evidence identified

Six studies (Boekeloo et al., 1999; Danielson et al., 1990; DiClemente et al., 2004; Downs et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al., 2005) were identified that examined interventions and programme delivered in healthcare settings including family planning clinics and primary care practices.

DiClemente and colleagues (2004) and Morrison-Beedy and colleagues (2005) both examined group education sessions for sexually active young females attending family planning clinics. Two other studies based in healthcare settings (Downs et al., 2004; Jemmott et al., 2005) also specifically targeted sexually experienced young females. Downs and colleagues (2004) examined an interactive video intervention and Jemmott and colleagues (2005) examined a skills-based HIV prevention intervention. Two studies (Boekeloo et al., 1999; Danielson et al., 1990) examined interventions based on health consultations. Boekeloo and colleagues (1999) examined a sexual risk assessment and education intervention and Danielson and colleagues (1990) examined a reproductive health consultation and slide tape programme, which specifically targeted young males.

The number of participants included in the studies ranged from 62 (Morrison-Beedy et al., 2005) to 682 (Jemmott et al., 2005), and a range of different age groups were targeted. The youngest age targeted was 12 years (Boekeloo et al., 1999; Jemmott et al., 2005) and the oldest was 19 (Morrison-Beedy et al., 2005; Jemmott et al., 2005). Five of the six studies examined gender specific interventions, with four studies (DiClemente et al., 2004; Downs et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al., 2005) focusing on interventions targeting females, and one study (Danielson et al., 1990) focusing on an intervention for males. Two studies (DiClemente et al., 2004; Jemmott et al., 2005) targeted African American, and African American and Latino adolescents, respectively. One study (Morrison-Beedy et al., 2005) was based on short-term follow-up only (<6 months) and four studies (Boekeloo et al., 1999; DiClemente et al., 2004; Downs et al., 2004; Jemmott et al., 2005) were based on medium-term follow-up (up to 12 months). The length of follow-up for the study by Danielson and colleagues (1990) was not reported.

6.3.2 Quality assessment

All six studies were RCTs and based on individual randomisation. Two well-reported and conducted studies (DiClemente et al., 200; Jemmott et al., 2005) were rated strong for quality. These studies appropriately allocated participants to intervention and control conditions, reported a range of relevant and reliable outcomes and conducted appropriate analyses. Three studies (Boekeloo et al., 1999; Downs et al., 2004; Morrison-Beedy et al., 2005) were rated moderate quality. Although these studies were generally well-reported and reported appropriate methods for allocation, and relevant and reliable outcomes, none of these studies reported whether they were sufficiently powered to detect an intervention effect or whether an ITT analyses had been undertaken. One study (Danielson et al., 1990) was rated poor quality. The authors did not report the overall sample size for the study, the length of follow-up was not clear and the authors did not discuss attrition.

6.3.3 Findings

6.3.3.1 Knowledge and understanding

All six studies (Boekeloo et al., 1999; Danielson et al., 1990; DiClemente et al., 2004; Morrison-Beedy et al., 2005; Jemmott et al., 2005; Downs et al., 2004) examined intervention effects on knowledge and understanding.

Four studies (DiClemente et al., 2004; Morrison-Beedy et al., 2005; Jemmott et al., 2005; Downs et al., 2004) examined interventions that specifically targeted sexually active young females. Two studies, delivered in family planning clinics had significant effects on HIV knowledge. DiClemente et al (2004) reported that intervention participants had higher HIV prevention knowledge scores than control participants at both the 6- and 12-month follow-ups (both p<0.001). Morrison-Beedy and colleagues (2005) found that at the 3-month follow-up, the intervention group scored significantly higher on HIVrelated knowledge (p<0.001). A skills-based HIV prevention intervention (Jemmott et al., 2005) was also found to have impacted on knowledge. African American and Latino females who received a skills-based intervention scored higher on HIV/STI knowledge and condom use knowledge postintervention, compared to those who received a health promotion control (p<0.01), and reported greater gains in condom use knowledge compared to those who received an information-based intervention (p<0.01). However, there were no effects of an interactive video intervention at any follow-up (1-, 3- and 6-months) on general or specific knowledge related to STIs (Downs et al., 2004). The authors noted that knowledge improved in the both the intervention and control conditions over time. In this particular study the two control conditions were content-matched and topic-matched to the intervention, respectively.

Two studies (Boekeloo et al., 1999; Danielson et al., 1990) that examined personal health consultations both reported intervention impacts on knowledge. At immediate post-test, Boekeloo and colleagues (1999) reported that adolescents in the intervention group were more likely than control adolescents to know that HIV is transmitted through oral and anal intercourse. However, the significance of this finding was not reported. Danielson and colleagues (1990) did not report the period over which participants were followed up but found that knowledge about ways to protect against STIs was significantly associated with the intervention (p<0.001).

6.3.3.2 Attitudes and values

Five studies (Boekeloo et al., 1999; Danielson et al., 1990; DiClemente et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al., 2005) examined intervention effects on attitudes and values.

Two studies (Boekeloo et al., 1996; Danielson et al., 1990) examined the impact of health consultations. Boekeloo and colleagues (1999) reported that at immediate post-test, adolescents in the intervention group were more likely than control adolescents to believe that their doctor thought they should use condoms if they had sexual intercourse (p<0.05), believe that they should use condoms if they had sexual intercourse (p<0.05), and less likely to believe it would be hard to refuse sex with a partner who refused condom use (p<0.05). Perceived susceptibility to HIV and other STIs,

condom use self-efficacy, and beliefs about abstinence did not differ between groups. Danielson and colleagues (1990) examined intervention effects on sexual attitudes. Coercive sexual attitudes³ were inversely associated with the intervention (OR 0.74; p<0.05). The association was weak and not statistically significant among those who had already been sexually active at baseline, but was stronger and statistically significant among those who had not yet become sexually active at baseline (OR 0.67; p<0.01).

Three studies (DiClemente et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al., 2005) examined interventions specifically targeting sexually active young women. DiClemente and colleagues (2004) found that over a 12-month follow-up period, African American females who received group education sessions focusing on HIV prevention reported fewer barriers to condom use (p<0.01), more favourable attitudes toward using condoms (p<0.001), and higher condom use self-efficacy scores (p<0.001). Jemmott and colleagues (2005) also reported positive effects for a skills-based intervention that targeted African American and Latino adolescent females. At post-test, compared to a health promotion control group, skills-based intervention group participants reported greater intentions to use condoms (p<0.01), greater condom use hedonistic beliefs⁴ (p<0.01), greater sexual partner approval of condoms (p<0.01), higher technical skills beliefs (p<0.05), and higher impulse control beliefs (p<0.05). Participants who received the information-based intervention also reported higher scores compared to the health promotion control group on the following measures: condom use intentions (p<0.01); condom use hedonistic beliefs (p<0.01); technical skills beliefs (p<0.05), and impulse control beliefs (p<0.05). Based on 3-months of follow-up, Morrison-Beedy and colleagues (2005) found that females who received an HIV prevention intervention scored significantly higher than the control group on confidence in condom use (p<0.05), and lower on the cons of condom use (p<0.05). However, there was no difference between groups in risk perception, readiness, behavioural intentions, or pros of condom use.

6.3.3.3 Personal and social skills

Three studies (Boekeloo et al., 1999; DiClemente et al., 2004; Morrison-Beedy et al., 2005) examined intervention effects on skills. Boekeloo and colleagues (1999) found that at post-test, intervention participants who received a health consultation reported significantly more discussion with their physician about 11 of 13 topics regarding sexuality than control adolescents (all p<0.05). However, there was no difference between intervention and control adolescents in their discussions with their parents on these topics. Two studies (DiClemente et al., 2004; Morrison-Beedy et al., 2005) examined interventions which specifically targeted sexually experienced young females. Over 12-months of follow-up, African American females who received an HIV prevention intervention (DiClemente et al., 2004) reported more frequent discussions with male sex partners about HIV prevention compared to controls (p<0.01) and scored significantly higher on a measure of the condom use skills (p<0.001). Morrison-Beedy and colleagues (2004) found a potentially negative effect of a group-based HIV

³ Based on two items: "A girl who leads you on should go all the way", and "I might stop seeing some if she refused me".

⁴ A measure concerning the belief that condoms do not interfere with sexual enjoyment
intervention. At the 3-month follow-up, control participants reported talking with their partners about safer sex more often than did intervention participants (p<0.05).

6.3.3.4 Health and social outcomes related to sexual health

All six studies (Boekeloo et al., 1999; Danielson et al., 1990; DiClemente et al., 2004; Downs et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al 2005) examined intervention effects on health and social outcomes related to sexual health.

Boekeloo and colleagues (1999) and Danielson and colleagues (1990) examined the effects of health practitioner-led sexual health consultations. Boekeloo and colleagues (1999) examined whether an STI risk assessment and education intervention had an impact on sexual activity, condom use and STI rates. Bivariate analyses, revealed that there were no statistically significant differences between groups regarding vaginal, anal, and/or oral sexual intercourse in the last 3 months or lifetime, or the number of vaginal intercourse partners (last 3 months or lifetime), at either the 3- or 9-month follow-up. Based on a mixed model regression, which controlled for baseline sexual experience and doctor, intervention participants were found to be more likely to have had vaginal intercourse at the 3-month follow-up than controls (OR 2.46; 95% CI 1.04, 5.84), but not at the 9-month follow-up (OR 1.64; 95% CI 0.81, 3.34). Among adolescents sexually active in the last 3 months, there were no significant differences between intervention and control participants regarding condom use at last intercourse at 9-months follow-up (OR 1.00; 95% CI: 0.31, 3.24), but the rate was greater among intervention participants at the 3-month follow-up (OR 18.1; 95% CI 1.3, 256.0). There was no difference between intervention and control participants in their reported STI diagnoses, STI treatment or pregnancies during the last 3 months at either follow-up. At the 9-month follow-up, more control participants reported genital signs of possible STIs than intervention participants (p<0.05). Danielson and colleagues (1990) reported that there was no statistically significant effect of the intervention, which specifically targeted male adolescents, on sexual activity status at follow-up. When confounding variables were controlled for, the association between the intervention and contraceptive effectiveness was statistically significant among the larger population of all males who were sexually active at follow-up (OR 1.51; p<0.05), and particularly among those who were not sexually active at baseline (OR 2.53; p<0.01). A partner's use of the pill at last intercourse was significantly associated with the intervention among all participants who were sexually active at follow-up (OR 1.66; p<0.05).

Four studies (DiClemente et al., 2004; Downs et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al., 2005) examined interventions specifically targeting sexually active young females. DiClemente and colleagues (2004) found that relative to participants who received a general health promotion intervention, African American females who received a group-based HIV intervention were more likely to report using condoms consistently in the past 30 days and 6 months, and to have used a condom at last vaginal sexual intercourse, at both the 6- and 12-month follow-up and in analyses conducted over the entire 12-month follow-up. Intervention participants were less likely to report having a new vaginal sex partner at the 6-month follow-up, and over the 12-month follow-up period. In addition, participants who received the intervention were significantly less likely to self-report a pregnancy at the 6-month follow-up, but there was no difference between groups on this outcome at 12-months, or

PSHE x.x Community review

over the entire follow-up period. Compared to controls, HIV intervention participants were more likely to report condom protected sex acts, both in the 30 days and 6-months preceding the 6- and 12month follow-up, and over the entire 12-month follow-up. They also reported significantly fewer unprotected vaginal intercourse episodes and a higher frequency of putting condoms on their partners. The results presented by the authors suggested that the intervention had an effect on Chlamydia infections (OR 0.17; 95% CI 0.03, 0.92; p<0.05), but no difference between groups were observed for Trichomonas vaginalis (OR 0.37; 95% CI 0.09, 1.46) or Gonorrhoea (OR 0.14; 95% CI 0.01, 3.02). Downs and colleagues (2004) examined an interactive video intervention, finding that intervention participants were more likely to have been completely abstinent between baseline and the 3-month follow-up than controls (OR 2.50; p<0.05). However, at the 6-month follow-up there was no difference between groups on this measure. There was no significant difference between intervention and control participants in how often they reported using condoms between baseline and the 3-month follow-up, or at the 6-month follow-up. There was no difference in the number of condom failures between conditions at the 3-month follow-up, but at the 6-month follow-up, participants in the video condition reported fewer condom failures in the past 3 months compared to controls (p<0.05). Participants in the video condition were significantly less likely to report having been diagnosed with an STI compared to controls (OR 2.79; p=0.05). However, the only disease with sufficient power to detect a difference was Chlamydia (OR 7.75; p=0.05). Jemmott and colleagues (2005) reported that participants who received a skill-based HIV intervention reported less frequent unprotected sexual intercourse at the 12-month follow-up compared to those who received a health-promotion control (p<0.01) or an information-based intervention (p<0.05). There was no difference between the groups on these measures at the 3- or 6-month follow-up. At the 12-month follow-up, skill-based intervention participants reported fewer sexual partners than controls (p<0.05), and they were less likely to report having multiple partners (p<0.01). No differences in the reported number of sexual partners were observed at the 3- or 6-month follow-up. There was no difference in STI rates between groups at the 6-month follow-up. However, at 12-months, participants who received the skill-based intervention were significantly less likely to have an STI than were those in the health promotion control group (p<0.05). Morrison-Beedy and colleagues (2005) found that a group-based HIV intervention for sexually active females had limited effects on sexual risk behaviours. Although overall risk behaviour scores were lower within the HIV intervention group relative to the control group, none of the individual risk outcomes, (including vaginal sex with/without condom, received/gave oral sex, alcohol or drug use before sex, and number of partners) were significantly different between the two groups.

6.3.4 Summary and evidence statements

Six studies (Boekeloo et al., 1999; Danielson et al., 1990; DiClemente et al., 2004; Downs et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al., 2005) were conducted in healthcare settings including family planning clinics and primary care practices. Four studies (DiClemente et al., 2004; Downs et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al., 2005) examined group-based education and/or skills-based interventions that specifically targeted sexually active young women. Two studies (Boekeloo et al., 1999; Danielson et al., 1990) examined interventions based around a health practitioner-led sexual health consultation.

All six studies (Boekeloo et al., 1999; Danielson et al., 1990; DiClemente et al., 2004; Morrison-Beedy et al., 2005; Jemmott et al., 2005; Downs et al., 2004) examined intervention effects on knowledge and understanding. Across four studies (DiClemente et al., 2004; Morrison-Beedy et al., 2005; Jemmott et al., 2005; Downs et al., 2004) of interventions that specifically targeted sexually active young females, there were consistent short- to medium-term improvements in sexual health-related knowledge among intervention participants. In addition, two studies (Boekeloo et al., 1999; Danielson et al., 1990) of health consultations reported significant short-term increases in knowledge among intervention participants relative to controls. Five studies (Boekeloo et al., 1999; Danielson et al., 1990; DiClemente et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al., 2005) examined intervention effects on attitudes and values. There were fairly consistent positive intervention effects on condom use attitudes across three studies (DiClemente et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al., 2005), which examined group-based education and/or skills-based interventions specifically targeting sexually active young women, and one study (Boekeloo et al., 1996) that examined a sexual risk assessment and education intervention. Three studies (Boekeloo et al., 1999; DiClemente et al., 2004; Morrison-Beedy et al., 2005) examined intervention effects on personal and social skills related to communication. There were inconsistent effects on communication; one study (DiClemente et al., 2004) reported medium-term positive effects on intervention participants' communication with their partners and a further study (Boekeloo et al., 1999) found positive short-term effects on adolescents' communication with their doctor, but not their parents. One study (Morrison-Beedy et al., 2005) found a potentially negative effect of a group-based HIV intervention.

All six studies (Boekeloo et al., 1999; Danielson et al., 1990; DiClemente et al., 2004; Downs et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al 2005) examined intervention effects on health and social outcomes related to sexual health. Two studies (Boekeloo et al., 1999; Danielson et al., 1990) of health practitioner-led sexual health consultations found no intervention effects on sexual activity, but there were weak short-term effects on condom and other contraceptive use. Across four studies (DiClemente et al., 2004; Downs et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al 2005) that examined interventions specifically targeting sexually active young females, there were inconsistent intervention effects on sexual activity, frequency of intercourse and number of partners. However, intervention effects on condom use and unprotected intercourse were more consistent, with two studies (DiClemente et al., 2004; Jemmott et al., 2005) reporting medium-term positive effects on these outcomes. Three studies (Boekeloo et al., 1999; Downs et al., 2004; Jemmott et al., 2005) examined intervention effects on STI infection and/or diagnosis, finding mixed intervention effects. However, medium-term positive effects on STI diagnosis were reported in one study (Jemmott et al., 2005) of a skills-based HIV/STI intervention.

Evidence statement 9

- 9 (a) There is strong evidence from six RCTs¹ to suggest that interventions and programmes delivered in healthcare settings may produce short- to medium-term improvements in sexual health-related knowledge. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 9 (b) There is strong evidence from three RCTs² to suggest that group-based education and/or skillsbased interventions specifically targeting sexually active young women in healthcare settings may have short- to medium-term positive effects on condom use attitudes. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 9 (c) There is inconsistent evidence from three RCTs³ on which to determine the effects of interventions and programmes delivered in healthcare settings on sexual health-related communication. However, there is strong evidence from one RCT⁴ to suggest that a genderand culturally-tailored intervention for African American females may have a positive impact on communication with sexual partners and condom use skills. This evidence may only be partially applicable because this study were conducted in the USA and focused on an ethnic population specific to the USA.
- 9 (d) There is moderate evidence from two RCTs⁵ to suggest that interventions and programmes based on health practitioner-led sexual health consultations may have a limited impact on sexual behaviours, including sexual activity and condom and other contraceptive use. This evidence may only be partially applicable because these studies were conducted in the USA.
- 9 (e) There is strong evidence from four RCTs⁶ to suggest that group-based education and/or skillsbased interventions specifically targeting sexually active young women in healthcare settings may not have a consistent impact on sexual activity or numbers of sexual partners. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 9 (f) There is strong evidence from four RCTs⁶ to suggest that group-based education and/or skillsbased interventions specifically targeting sexually active young women in healthcare settings may have a short- to medium-term positive impact on condom and other contraceptive use, and unprotected intercourse. This evidence may only be partially applicable because these studies were conducted in the USA and focused on ethnic populations, specific to the USA.
- 9 (g) There is inconsistent evidence from three RCTs³ on which to determine the effects of interventions and programmes delivered in healthcare settings on STIs. However, there is strong evidence from one RCT⁸ to suggest that a skill-based HIV/STI intervention may have a positive medium-term impact on STI diagnosis. This evidence may only be partially applicable because this study were conducted in the USA and focused on ethnic populations specific to the USA.

- ¹ Boekeloo et al., 1999 (RCT +); Danielson et al., 1990 (RCT -); DiClemente et al., 2004 (RCT ++); Downs et al., 2004 (RCT +); Jemmott et al., 2005 (RCT ++); Morrison-Beedy et al., 2005 (RCT +)
- ² DiClemente et al., 2004 (RCT ++); Jemmott et al., 2005 (RCT ++); Morrison-Beedy et al., 2005 (RCT +)
- ³ Boekeloo et al., 1999 (RCT +); DiClemente et al., 2004 (RCT ++); Morrison-Beedy et al., 2005 (RCT +)
- ⁴ DiClemente et al., 2004 (RCT ++);
- ⁵ Boekeloo et al., 1999 (RCT +); Danielson et al., 1990 (RCT -)
- ⁶ DiClemente et al., 2004 (RCT ++); Downs et al., 2004 (RCT +); Jemmott et al., 2005 (RCT ++); Morrison-Beedy et al., 2005 (RCT +)
- ⁷ Boekeloo et al., 1999 (RCT +); Downs et al., 2004 (RCT +); Jemmott et al., 2005 (RCT ++)
- ⁸ Jemmott et al., 2005 (RCT ++)

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Boekeloo et al., 1999	RCT (individual) +	USA n=215 ~70% African American 12-15 years	Primary care	ASSESS: Sexual risk assessment and education intervention. Single session with doctor two brochures that addressed skills and self-efficacy for sexual health, community resources brochures, and two brochures for parents about how to discuss sex and drug risks with teens.	Social cognitive theory, Theory of Reasoned Action	Primary care doctors
Danielson et al., 1990	RCT (individual) -	USA n=522 males Ethnicity NR 15-18 years	Primary care	Reproductive health consultation and slide tape programme. Included information on reproductive anatomy, fertility, hernia, testicular self-examination, STIs, contraception, couple communication and access to health services.	NR	Health care practitioner
DiClemente et al., 2004	RCT (individual) ++	USA n=522 females 100% African American 14-18 years	Family medicine clinic	Four, 4-hour sessions; ethnic and gender pride, HIV risk reduction strategies and the importance of healthy relationships.	Social cognitive theory, theory of gender and power	Trained African American female health educator and two African American peer educators
Downs et al., 2004	RCT (individual) +	USA n=300 females 75% African American; 15% White; 10% other 14-18 years	Healthcare settings	Interactive video intervention, which covered negotiation behaviours with sexual partners, condom efficacy, and information about reproductive health and viral and bacterial STIs	NR	Video
Jemmott et al., 2005	RCT (individual) ++	USA n=682 females 68% African American; 32% Latino 12-19 years	Hospital- based family planning clinic	Single session interventions. (1) Educational videotapes illustrated correct condom use with a demonstration model and depicted effective negotiation of condom use; (2) Participants practiced the skills needed to use condoms	Cognitive behavioural theory	Female, African American facilitators
Morrison- Beedy et al., 2005	RCT (individual) +	USA n=62 females 59% White; 29% Black; 10% Hispanic; 2% Asian 15-19 years	Family planning centre	Four, 2-hour sessions; HIV-related information and behavioural skills components (assertiveness, self- efficacy, and negotiation) combined with motivational enhancement strategies.	Information- Motivation- Behavioural Skills Model	Trained female interventionists

Table 6.5. Summary of programme content: programmes delivered within healthcare settings

				Outcomes				
Study	Rating	Intervention	Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills	
Boekeloo et al., 1999	RCT (individual) +	ASSESS n=101	No intervention n=114	PT (NR)	↑ HIV transmission [¶]	↑ beliefs about condom use [¶] NS condom use self-efficacy NS perceived HIV susceptibility NS abstinence beliefs	1 discussion with physician about sexuality topics [¶]	
Danielson et al., 1990	RCT (individual) -	Reproductive health consultation n=262	Wait list control n=260	NR	1 ways to protect against STIs***	↓ coercive sexual attitudes*	-	
DiClemente et	RCT	HIV prevention	General health	6 mo	1 HIV prevention**	↑ condom attitudes** ↓ condom barriers** ↑ condom use self-efficacy**	<pre>1 condom use skills***</pre>	
al., 2004	(Individual) ++	n=251	n=271	12 mo	1 HIV prevention**	<pre>f condom attitudes** NS condom barriers f condom use self-efficacy**</pre>	↑ condom use skills*** ↑ communication frequency*	
Downs et al.,	RCT	Video	Alternative delivery	3 mo	NS general STI NS specific STI	-	-	
2004	(individual) +	n=NR	formats n=NR	6 mo	NS general STI NS specific STI	-	-	
lemmett et RCT		Skills-based n=235	Health promotion control n=219	3, 6, 12 mo	↑ HIV/STI risk reduction*** ↑ condom use***	 f condom use intention** f condom use hedonistic beliefs*** f sexual partner approval** f technical skills beliefs* f impulse control beliefs* NS negotiation skill beliefs 	-	
al., 2005	(Individual) ++	Information n=228	Health promotion control n=219	3, 6, 12 mo	↑ HIV/STI risk reduction*** ↑ condom use***	 ↑ condom use intention*** ↑ condom use hedonistic beliefs*** NS sexual partner approval ↑ technical skills beliefs* ↑ impulse control beliefs* NS negotiation skill beliefs 	-	

Table 6.6. Programme delivered in healthcare settings: effects on knowledge, attitudes and skills

				Follow-up		Outcomes	
Study	Rating	Intervention	Comparator		Knowledge and understanding	Attitudes and values	Personal and social skills
Jemmott et al., 2005	RCT (individual) ++	Skills-based n=235	Information n=228	3, 6, 12 mo	NS HIV/STI risk reduction*** 1 condom use***	NS condom use intention NS condom use hedonistic beliefs NS sexual partner approval NS technical skills beliefs NS impulse control beliefs NS negotiation skill beliefs	-
Morrison- Beedy et al., 2005 RCT (individual) + HIV risk reduction intervention n=33 Health promotion control n=29 3 mo 1*** Confidence in condom use* 1 cons of condom use* NS risk perception NS readiness J communication with partners about safe sex*							
*p<0.05; **p<0.0	01; ***p<0.001; ¹	ⁱ p value not reporte	d; 1 increase rela	tive to compara	ator; ↓ decrease relative to	comparator; NS not significant; - out	come not reported

Table 6.7. Programme delivered in healthcare settings: effects on health and social outcomes related to sexual health

			Comparator			Healt	h and social outco	mes	
Study	Rating	Intervention		Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs/ Conceptions	Other
			No	3 mo (int=92%; con=94%)	NS vaginal, anal or oral sexual intercourse	NS number of vaginal intercourse partners	↑ condom use [¶]	NS STI diagnoses/ treatment NS pregnancy	-
Boekeloo et al., 1999	RCT (individual) +	ASSESS n=101	intervention n=114	9 mo (int=93%; con=90%)	NS vaginal, anal or oral sexual intercourse	NS number of vaginal intercourse partners	NS condom use	↓ genital signs of possible STIs [¶] NS STI treatment NS pregnancy	-
Danielson et al., 1990	RCT (individual) -	Unnamed n=262	Wait list control n=260	NR	NS sexual activity status	-	 f effective contraception* f partner's use of pill at last intercourse* 	-	-

			Comparator			Healt	h and social outco	mes	
Study	Rating	Intervention		Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs/ Conceptions	Other
DiClemente et al., 2004	RCT	HIV prevention intervention n=251	General health promotion n=271	6 mo (int=90%; con=90%)	-	↓ new vaginal sex partner, past 30 d*	NS consistent condom use, past 30 d 1 consistent condom use, past 6 mo** 1 condom use at last intercourse***	-	-
	(incividual) ++			12 mo (int=87%; con=89%)	-	NS new vaginal sex partner in past 30 days	 Consistent condom use, past 30 d* 1 consistent condom use, past 6 mo* 1 condom use at last intercourse*** 	-	-
		Interactive video	Alternative	3 mo (NR)	1 abstinence*		NS condom use frequency NS condom failures	-	-
Downs et al., 2004	RCT (individual) +	intervention n=NR	delivery formats n=NR	6 mo (NR)	NS abstinence		NS condom use frequency ↓ condom failures, past 3 mo*	↓ diagnosed with Chlamydia*	-

					Health and social outcomes						
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs/ Conceptions	Other		
Jemmott et	RCT (individual)	HIV/STI reduction intervention	Health promotion	3, 6 mo (NR)	-	NS number of partners, past 3 mo NS multiple partners, past 3 mo	NS frequency sex without condom use, past 3 mo	-	↓ frequency sex drugs/alcohol (both*) NS frequency unprotected sex drugs/alcohol		
al., 2003	++	Skills-based, n=235	n=219	12 mo (NR)	-	↓ multiple partners, past 3 mo** ↓ number of partners, past 3 mo*	↓ frequency sex without condom use, past 3 mo**	↓ tested positive STI*	NS frequency sex drugs/alcohol ↓ frequency unprotected sex drugs/alcohol*		
Jemmott et al., 2005	RCT (individual) ++	HIV/STI reduction intervention Information, n=228	Health promotion control n=219	3, 6, 12 mo (NR)	-	NS multiple partners, past 3 mo NS number of partners, past 3 mo	NS frequency sex without condom use, past 3 mo	NS tested positive STI	NS frequency sex drugs/alcohol NS frequency unprotected sex drugs/alcohol		
Jemmott et al., 2005	RCT (individual) ++	HIV/STI reduction intervention Skills-based, n=235	HIV/STI reduction intervention Information, n=228	3, 6, 12 mo (NR)	-	NS multiple partners, past 3 mo NS number of partners, past 3 mo	↓ frequency sex without condom use, past 3 mo (12 mo FU only*)	NS tested positive STI	↓ frequency sex drugs/alcohol (3 mo only*) NS frequency unprotected sex drugs/alcohol		
Morrison- Beedy et al., 2005	RCT (individual) +	HIV risk reduction intervention n=33	Health promotion control n=29	3 mo (48%)	NS received/ gave oral sex NS vaginal sex with/without condom	NS number of partners	-	-	NS alcohol or drug use before sex		
*p<0.05; **p<0.0)1; ***p<0.001; [¶]	p value not reporte	d; 1 increase rela	tive to compara	ator; ↓ decrease re	elative to comparat	or; NS not significan	t; - outcome not re	eported		

6.4 Programmes delivered to families

6.4.1 Overview of evidence identified

Ten studies (Anderson et al., 1999; Dilorio et al., 2006; Lederman et al., 2004; 2008; McBride et al., 2007; McKay et al., 2004; Miller et al., 1993; Scheinberg et al., 1997; Winett et al., 1992; 1993) evaluated seven programmes delivered to families which targeted adolescent sexual health, all of which were conducted in the USA. Two studies (Winett et al., 1992; 1993) evaluated interventions that aimed to prevent HIV and AIDS while eight studies (Anderson et al., 1999; Dilorio et al., 2006; Lederman et al., 2004; 2008, McKay et al., 2004; McBride et al., 2007; Miller et al., 1993; Scheinberg et al., 1997) evaluated programmes that focused on delaying sexual initiation and reducing sexual risk behaviours. Two studies (Anderson et al., 1999; Miller et al., 1993) included a specific focus on improving family communication about sexual behaviours. The setting where the intervention or programme was delivered varied, and included sessions after school, in the home and in community centres. The person or persons providing the intervention was poorly reported across the included studies, and were reported for only one programme, the CHAMP family programme (McKay et al., 2004; McBride et al., 2007), which was provided by a variety of providers including mental health interns, parents and community consultants respectively.

Six evaluations detailed the theory or theories that were the basis for intervention. Three programmes including RAP (Anderson et al., 1999), PARE (Lederman et al., 2008) and SHAPE (Scheinberg et al., 1997) were based on social learning theory. The development of the SHAPE programme (Scheinberg et al., 1997) also informed by social cognitive theory. Keepin' It R.E.A.L (Dilorio et al., 2006) was based on social cognitive theory and problem behavioural theory.

Sample size varied between studies, ranging from 49 families (Winett et al., 1992) to 804 adolescentparent dyads (Lederman et al., 2004). All programmes involved children in early adolescence aged between 10 to 15 years and their parents, and all programmes had at least one session for both parent and child together. One programme (Keepin' It R.E.A.L, Dilorio et al., 2006) only included mothers while all other programmes recruited both mothers and fathers, or an appropriate other adult. Post-intervention follow-up time varied across the included studies. The period of follow-up for three programmes (CHAMP Family Program, McKay et al., 2004; McBride et al., 2007; an HIV prevention video, Winett et al., 1992; Winett et al., 1993; SHAPE, Scheinberg et al., 1997) was six months or less, with the CHAMP family programme and SHAPE evaluated at post-test only. Two studies, including evaluations of Facts and Feelings (Miller et al., 1993) and one evaluation of PARE (Lederman et al., 2004) reported follow-up periods of 12 months or less. Long-term evaluation (>12 months) was reported by three studies including evaluations of RAP (Anderson et al., 1999), Keepin' It R.E.A.L (Dilorio et al., 2006), and the second evaluation of PARE (Lederman et al., 2008).

6.4.2 Quality assessment

Of the 10 studies identified, seven were RCTs, one was an NRCT and two were CBA studies. Five RCTs (Lederman et al., 2004; 2008; Miller et al., 1993; Winett et al., 1992; 1993) were based on individual randomisation while the remaining two RCTs (Anderson et al., 1999; Dilorio et al., 2006)

were based on cluster randomisation, by group (Anderson et al., 1999) and community site (Dilorio et al., 2006). Quality of the RCTs varied with one study rated good for guality (Dilorio et al., 2006), two studies rated as moderate (Miller et al., 1993; Winett et al., 1993) and four rated as poor (Anderson et al., 1999; Lederman et al., 2004; 2008; Winett et al., 1992). The quality of the NRCT was rated as poor (Scheinberg et al., 1997). Two CBA studies that evaluated the CHAMP family programme (McKay et al., 2004; McBride et al., 2007) were rated poor quality due to being based on post-test follow-up only and poor reporting of outcomes. All seven RCTs were reported as randomised, but did not detail methods of randomisation or describe whether allocation was appropriately concealed. Outcome measures were reliable in all studies and rated good quality in three studies (Lederman et al., 2008; Miller et al., 1993; Winett et al., 1993). In all RCT studies, outcomes were rated as being relevant and generally all important outcomes were assessed. RCTs were rated as moderate or good for length of follow-up time, with the exception of one study (Winett et al., 1993) where follow-up time was less than six months. Baseline comparability of groups was poorly reported or not reported in five RCTs (Anderson et al., 1999; Dilorio et al., 2006; Lederman et al., 2004; Lederman et al., 2008; Miller et al., 1993) and an ITT analysis was only reported to have been undertaken in one study (Dilorio et al., 2006).

6.4.3 Findings

6.4.3.1 Knowledge and understanding

For six programmes, eight studies (Dilorio et al., 2006; Lederman et al., 2008; McKay et al., 2004; McBride et al., 2007; Miller et al., 1993; Scheinberg et al., 1997; Winett et al., 1992; 1993) reported outcomes relating to knowledge and understanding. For four programmes including Keepin' It R.E.A.L (Dilorio et al., 2006), CHAMP family programme (McKay et al., 2004; McBride et al., 2007), SHAPE (Scheinberg et al., 1997) and an HIV prevention video intervention (Winett 1992; 1993) only short-term follow-up results, at six months or less were presented. Two studies (Miller et al., 1993; Lederman et al., 2009) reported medium- to long-term follow-up results, at 1- and 2-years, respectively.

Across the eight studies that examined intervention effects on knowledge and understanding, there were generally positive intervention effects. Evaluations of four programmes (Keepin' It R.E.A.L; CHAMP family programme; an unnamed HIV prevention video and PARE) examined knowledge about HIV and AIDS. Positive intervention effects were reported for both adolescent (p<0.05) and mother's (p<0.01) HIV knowledge following the Keepin' It R.E.A.L programme (Dilorio et al., 2006), for those who received the social cognitive theory-based intervention (SCT) in comparison to controls and those who received the life skills intervention (LSK). In two studies (Winett et al., 1992; 1993) of a home-based HIV prevention video, positive intervention effects were found on knowledge about HIV for both adolescents and their parents in the short-term (all comparisons, p<0.001). Lederman and colleagues (2008) also found that PARE had long-term positive effects on knowledge about preventing HIV and pregnancy (2-years follow-up; p<0.05). The CHAMP family programme (McKay et al., 2004; McBride et al., 2007) had no impact on knowledge about HIV at post-test, and although

SHAPE (Scheinberg et al., 1997) had a positive effect on children's knowledge at post-test (p<0.05) there was no significant impact upon parents. Miller and colleagues (1993) reported that the Facts and Feelings intervention had positive effects on child (p<0.001) and father (p<0.01) sexual knowledge, but had no significant impact on mother's knowledge.

6.4.3.2 Attitudes and values

Eight studies (Anderson et al., 1999; Dilorio et al., 2006; Lederman et al., 2004; 2008; McKay et al., 2004; McBride et al., 2007; Miller et al., 1993; Scheinberg et al., 1997) examined intervention effects on attitudes and values for six programmes. Adolescent attitudes and intentions regarding abstinence was the most common outcome evaluated and on the whole non-significant programme effects were reported. There was no significant effect of RAP (Anderson et al., 1999) on reasons why not to have sex in the short- or medium-term and there was no effect of Facts and Feelings (Miller et al., 1993) on intentions to have sex, or attitudes towards abstinence or sex. However, this programme positively influenced father's (p<0.05) and mother's (p<0.01) abstinence values in comparison to control participants. Dilorio and colleagues (2006) found a non-significant association between participation in Keepin' It R.E.A.L and adolescent intentions to have sex or use condoms at the 2-year follow-up. Based on short-term follow-up, Scheinberg and colleagues (1997) found that participation in SHAPE had no significant effects on adolescent attitudes towards abstinence or parent attitudes towards sex. Based on short-term follow-up, adolescent intentions to postpone sex were positively associated with participation in PARE (Lederman et al., 2004; p<0.01), however the programme did not impact upon attitudes towards risk behaviours. Long-term evaluation of PARE (Lederman et al., 2008) revealed that there was no significant effect of the programme on self-efficacy to resist sex.

Two studies (McKay et al., 2004; McBride et al., 2007) examined the impact of the CHAMP family programme on attitudes towards HIV and AIDS, finding no significant programme effects. One study (Dilorio et al., 2006) examined attitudes related to parent-child communication. Based on 2-year follow-up of Keepin' It R.E.A.L participants, Dilorio and colleagues (2006) found positive intervention effects on mothers' intentions to communicate about sex with their child (p<0.01) and on their comfort in doing so (p<0.001).

6.4.3.3 Personal and social skills

Ten studies (Anderson et al., 1999; Dilorio et al., 2006; Lederman et al., 2004; 2008; McKay et al., 2004; McBride et al., 2007; Miller et al., 1993; Scheinberg et al., 1997; Winett et al., 1992; 1993) examined the impact of eight programmes on personal and social skills. Nine studies (Anderson et al., 1999; Dilorio et al., 2006; Lederman et al., 2004; 2008; McKay et al., 2004; McBride et al., 2007; Miller et al., 1993; Scheinberg et al., 1997) examined intervention effects on parent-child communication. Two studies (Lederman et al., 2004; Scheinberg et al., 1997) of the PARE and SHAPE programmes, respectively, found no significant effects on communication about sexual behaviours at short-term follow-up in intervention families compared to controls. The CHAMP family programme (McKay et al., 2004; McBride et al., 2007) was found to have had significant short-term effects on family decision making (p<0.05), parental monitoring, family decision making and comfort in communication (all p<0.01). However, CHAMP intervention families also reported significantly higher

levels of family conflict compared to controls (p<0.01). At post-test, there was a positive effect of the RAP programme (Anderson et al., 1999) on parent-child communication (p<0.05), but at the 1-year follow-up this difference was no longer significant between intervention and control families. Two studies (Lederman et al., 2008; Dilorio et al., 2006) examined intervention effects on communication over the long-term. Long-term evaluation of the PARE programme (Lederman et al., 2008) revealed a non-significant effect of the programme on parent-child communication, in addition, although the Keepin' It R.E.A.L programme (Dilorio et al., 2006) had a significant effect on mother-daughter discussion as reported by mothers, there was no significant effect on communication as reported by daughters.

Winett and colleagues (1992; 1993) evaluated two home-based HIV video interventions. Both evaluations found a short-term positive effect of the interventions on family problem-solving skills (both p<0.05), but no effect on teen assertiveness or teen problem solving skills. Scheinberg and colleagues (1997) found that the SHAPE programme had no impacts in the short-term on adolescent's comfort accessing contraception, sexual decision-making or assertiveness.

6.4.3.4 Health and social outcomes related to sexual health

Five studies (Anderson et al., 1999; Dilorio et al., 2006; McKay et al., 2004; McBride et al., 2007; Miller et al., 1993) examined the effects of six programmes (RAP, Facts and Feelings, Keepin' It R.E.A.L, and the CHAMP family programme) on health and social outcomes related to sexual health.

Long-term evaluation of Keepin' It R.E.A.L (Dilorio et al., 2006) indicated no significant effects of either intervention condition on abstinence or involvement in intimate sexual behaviours, however, among sexually active participants, those in the SCT and LSK groups were more likely than controls to report that they had used a condom the last time they had sex, although the significance of this finding was not clearly reported. Two studies (Miller et al., 1993; Anderson et al., 1999) examined intervention effects on outcomes related to sexual health in the medium-term finding no effect of either intervention at follow-up, on sexual behaviours and pregnancy, respectively. Short-term evaluation of the CHAMP family programme (McKay et al., 2004; McBride et al., 2007) revealed a significant reduction in the time adolescents spent in situations of sexual possibility (p<0.01).

6.4.4 Summary and evidence statements

Ten studies (Anderson et al., 1999; Dilorio et al., 2006; Lederman et al., 2004; 2008; McBride et al., 2007; McKay et al., 2004; Miller et al., 1993; Scheinberg et al., 1997; Winett et al., 1992; 1993) evaluated seven programmes delivered to families which targeted adolescent sexual health. All ten studies were conducted in the USA and the young people involved were in early adolescence. Across the included studies there was a focus on delaying sexual initiation, reducing sexual risk behaviours and improving parent-child communication about sexual behaviours.

Seven studies (Dilorio et al., 2006; McKay et al., 2004; McBride et al., 2007; Miller et al., 1993; Scheinberg et al., 1997; Winett et al., 1992; Winett et al., 1993) examined intervention effects on knowledge and understanding. The results of these studies demonstrated that interventions and

PSHE x.x Community review

programmes delivered to families can have positive influences on knowledge related to sexual health in the short- (Scheinberg et al., 1997; Winett 1992; Winett et al., 1993), medium- (Miller et al., 1993) and long-term (Dilorio et al., 2006; Lederman et al., 2008). Improvements were reported in adolescent knowledge and knowledge among mothers (Dilorio et al., 2006) and fathers (Miller et al., 1993). Only one programme, CHAMP (McKay et al., 2004; McBride et al., 2007), was found to have had no impact on knowledge at follow-up. Eight studies (Anderson et al., 1999; Dilorio et al., 2006; Lederman et al., 2004; 2008; McKay et al., 2004; McBride et al., 2007; Miller et al., 1993; Scheinberg et al., 1997) examined intervention effects on attitudes and values. The most common outcomes examined were attitudes towards abstinence and adolescent's intentions to remain abstinent or have sex. Generally, across six studies (Anderson et al., 1999; Dilorio et al., 2006; Lederman et al., 2004; 2008; Miller et al., 1993; Scheinberg et al., 1997), the results suggested that programmes and interventions delivered to families were not effective at influencing adolescent's attitudes and intentions regarding resisting or delaying sex. Six studies reported finding no significant differences between intervention and control groups at follow-up regarding adolescent's attitude towards abstinence (Anderson et al., 1999; Miller et al., 1993; Scheinberg et al., 1997), or intentions to have sex (Dilorio et al., 2006; Miller et al., 1993), self-efficacy to resist sex (Lederman et al., 2008), and one study (Lederman et al., 2004) reported inconsistent effects, with effects on intentions to delay sex at medium-term follow-up but not on attitudes towards risk behaviours. Ten studies (Anderson et al., 1999; Dilorio et al., 2006; Lederman et al., 2004; 2008; McKay et al., 2004; McBride et al., 2007; Miller et al., 1993; Scheinberg et al., 1997; Winett et al., 1992; 1993) examined the impact of eight programmes on personal and social skills. Across nine studies, which examined effects on parent-child communication, no clear intervention effects were found.

Five studies (Anderson et al., 1999; Miller et al., 1993; Dilorio et al., 2006; McKay et al., 2004; McBride et al., 2007) examined effects on health and social outcomes related to sexual health for four programmes. The results suggested that programmes and interventions delivered to families do not affect sexual behaviour. Two studies (Anderson et al., 1999; Miller et al., 1993) found no intervention effects on pregnancy rates or sexual behaviour, respectively, and long-term evaluation of Keepin' It R.E.A.L (Dilorio et al., 2006) found no intervention effects on abstinence or involvement in intimate sexual behaviours. There were, however, limited but positive effects of this programme on condom use.

Evidence statement 10

- 10 (a) There is moderate evidence from five RCTs and one NRCT¹ to suggest that interventions and programmes delivered to families may improve knowledge in the short- to long-term. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.
- 10 (b) There is moderate evidence from five RCT and one NRCT² to suggest that interventions and programmes delivered to families may not influence adolescent's attitudes or intentions regarding abstinence or delaying sex. Findings may only be partially applicable to the UK as

all the studies were conducted in the USA and may not be generalisable beyond the populations studied.

- 10 (c) There is moderate evidence from seven studies² to suggest that programmes and interventions delivered to families may not influence parent-child communication. There is weak evidence from two CBA studies³ to suggest that intensive, family-focused interventions may have positive short-term effects on family communication. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.
- 10 (d) There is weak evidence from three RCT and two CBA studies⁴ to suggest that programmes delivered to families may not have effects on adolescent sexual behaviour. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.
- ¹ Dilorio et al., 2006 (RCT ++); Scheinberg et al., 1997 (NRCT -); Winett et al., 1992 (RCT -); Winett et al., 1993 (RCT +); Miller et al., 1993 (RCT +); Lederman et al., 2008 (RCT -)
- Anderson et al., 1999 (RCT -); Dilorio et al., 2006 (RCT ++); Lederman et al., 2004 (RCT -); Lederman et al., 2008 (RCT -); Miller et al., 1993 (RCT +); Scheinberg et al., 1997 (NRCT -)
- ³ McKay et al., 2007, McBride et al., 2007 (CBA -)
- ⁴ Anderson et al., 1999 (RCT -); Miller et al., 1993 (RCT +); Dilorio et al., 2006 (RCT ++); McKay et al., 2004 (CBA -); McBride et al., 2007 (CBA -)

Table 6.8. Summary of programme content: programmes delivered to families

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Anderson et al., 1999	RCT (cluster) -	USA n=251 adolescents (mean age 10.6) and their parents 46% Hispanic; 21% African American; 13% European- American; 6% Asian American; 2% Native American; 5% other; 8% unknown	Family; summer; after-school and in- school classes	RAP Reaching Adolescents and Parents: six adolescent-only; one adult-only and one joint sessions to improve parent-child communication and delay sexual debut	Social learning theory	NR
Dilorio et al., 2006	RCT (cluster) ++	USA n=582 adolescents (mean age 12 years) and their mothers Ethnicity=NR	Family; boys and girls club members	Keepin' It R.E.A.L: seven two-hour sessions over 14 weeks. Participants received either a life skills or social cognitive theory based intervention that aimed to delay sexual initiation and increase condom use	Social cognitive theory; problem behaviour theory	NR
Lederman et al., 2004	RCT (individual) -	USA n=804 parent and child dyads 38% Hispanic; 26% African American; 25% White; 10% Other (children aged 11- 15 years)	Family; after- school	PARE Parent-Adolescent Relationship Education: participants received four two and a half-hour sessions over a four week period plus three booster sessions to reduce sexual risk behaviours	NR	NR
Lederman et al., 2008	RCT (individual) -	USA n=192 families 36% Hispanic; 29% African American; 24% White; 11% Asian or Other (adolescents aged 11-15 years)	Family	PARE Parent-Adolescent Relationship Education: participants received four two and a half-hour sessions over a four week period plus three booster sessions to reduce sexual risk behaviours	Social learning theory; cognitive behavioural theory	NR

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
McKay et al., 2004; McBride et al., 2007	CBA -	USA n=564 4 th and 5 th grade children and their families ~99% African American	Family	CHAMP Family Programme: 12 90-minute weekly meetings aimed at delaying initiation of sexual intercourse and reducing time spent in situations of sexual possibility	NR	Mental health interns; community consultants; parents
Miller et al., 1993	RCT (individual) +	USA n=548 families Mothers and fathers were 93% and 97% White respectively (adolescent mean age 13.9 years)	Family; in the home	Facts and Feelings: intervention included six 15-20 minute videos to increase parent-child communication about sexual issues and to delay the likelihood of sexual initiation with or without accompanying newsletters that provided supplementary information	NR	NR
Scheinberg et al., 1997	NRCT -	USA n=122 participants from n=61 families Majority White, 5% Latino, 4% Asian	Family; classroom curriculum	SHAPE (Sharing Healthy Adolescent and Parent Experiences): Parents and children attended six two- hour sessions together where they were exposed to a curriculum aiming to delay sexual intercourse and to prevent risky sexual behaviours	Social learning theory; social cognitive theory, relational ethics	NR
Winett et al., 1992	RCT (individual) -	USA n=49 families (adolescents aged 12-14 years) Ethnicity=NR	Family; in the home	NR: families viewed four HIV prevention videos at home that focused on educating about HIV; problem-solving, assertiveness, coping and communication skills	NR	Video
Winett et al., 1993	RCT (individual) +	USA n=69 families (adolescents aged 12-14 years) Ethnicity=NR	Family; in the home	NR: families viewed 135 minutes of HIV prevention video that included education about HIV and teen risk behaviour and health issues; family and problem-solving skills and teen assertiveness	NR	Video

					Outcomes			
Study	Rating	Intervention	Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills	
Anderson et	RCT	RAP	Delayed	PT (NR)	-	NS reasons why participants "would not have sex now"	↑ parent child communication*	
al., 1999	(cluster) -	n=185	n=66	12 mo n=251 (54%)	-	NS reasons why participants "would not have sex now"	NS parent child communication	
				4 mo n=547 (94%)	↑ adolescent HIV knowledge ^a * ↑ mother's HIV knowledge ^a *	-	-	
Dilorio et al., 2006	RCT (cluster) ++	Keepin' It R.E.A.L n=381 (LSK n=187; SCT n=194)	1 hour HIV session n=201	24 months n=524 (90%)	-	 f mothers' intentions to discuss sex with child** f mothers' comfort in discussing sex with child*** NS child's comfort talking to mother about sex NS would end sexual activity until older NS would use a condom every time they have sex NS outcome expectations and self-efficacy for abstinence 	1 mothers reporting discussion with child about sex in past 3 months** NS child's reported communication with mother	
Lederman et al., 2004	RCT (individual) -	PARE n=90	Traditional intervention delivery n=714	3-6 mo n=NR	-	 t intentions to postpone sexual involvement** NS expectancies about consequences of sexual behaviour; NS attitudes towards risk behaviours NS perceptions of parents' disapproval to involvement in risk behaviours 	NS discourse with parents about sexual and other risk behaviours	
Lederman et al., 2008	RCT (individual) -	PARE n=90	Attention Control n=102	2 yrs n=NR	1 protection against pregnancy and HIV transmission*	NS self-efficacy for resisting sex	† parents reported having definite rules about child's behaviour* NS parent child communication	

Table 6.9. Programmes delivered to families: effects on knowledge, attitudes and skills

					Outcomes			
Study	Rating	Intervention	Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills	
McKay et al., 2004; McBride et al., 2007	CBA -	CHAMP n=274	Did not receive intervention n=290	PT n=465 (82%)	NS knowledge about HIV/AIDS	NS attitudes towards HIV/AIDS	 f family decision making* f parental monitoring and supervision** f family communication regarding sensitive issues** f communication comfort** f children reported higher family conflict** 	
Miller et al., 1993	RCT (individual) +	Facts & Feelings n=258 (video only n=132; video + newsletter n=126)	Did not receive intervention n=290	1 yr n=504 (92%)	↑ child's sexual knowledge*** ↑ father's sexual knowledge** NS mother's sexual knowledge	 father's abstinence values* mother's abstinence values*** NS child's abstinence values; intentions to have sex before marriage or in the next year NS child's acceptability for pressuring for sex; child's peer's sexual values; child's family's sexual values; family or peer influence on child's sexual values 	↑ child, father and mother reported parent child communication about sex*** ↑ frequency in communication NS frequency in communication at delayed post-test	
Scheinberg et al., 1997	NRCT -	SHAPE II n=NR	Reduced intervention n=NR	PT n=118 (97%)	1 children's' knowledge test scores* NS parent's knowledge test scores	 1 attitudes towards homosexuality* 1 child's satisfaction with social relationships* NS child's attitudes about birth control; sexuality; gender roles; abstinence NS child's self-esteem; satisfaction with sexuality NS parent attitudes towards sexual behaviour 	 I children's social decision making scores* NS child engaging in social activities NS child's comfort engaging in social activities; talking with parents or friends about sex or birth control; comfort accessing or using birth control NS child's sexual decision making; communication; assertiveness; birth control assertiveness skills 	

						Outcomes						
Study	Rating	Intervention	Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills					
Winett et al.,	RCT	NR	Waiting list	PT n=NR	↑ teen knowledge about HIV*** ↑ parent knowledge about HIV***	-	f family problem solving skills*** NS teen assertiveness skills NS teen problem-solving skills					
1992	(individual) -) - n=NR	n=NR	6 mo n=NR (94%)	↑ teen knowledge about HIV* ↑ parent knowledge about HIV***	-	↑ family problem-solving skills* NS teen assertiveness skills NS teen problem-solving skills					
Winett et al.,	RCT	NR	Reduced	PT n=69 families (100%)	↑ knowledge about HIV***	-	f family problem-solving skills*** NS teen assertiveness skills NS teen problem-solving skills					
1993	(individual) +	n=NR	n=NR	4 mo n=46 families (67%)	↑ knowledge about HIV***	-	↑ family problem-solving skills*** NS teen assertiveness skills NS teen problem-solving skills					
*p<0.05; **p<0.0 aSCT and contro	*p<0.05; **p<0.01; ***p<0.001; ¹ p value not reported; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported ^a SCT and control groups in comparison to LSK group ^b SCT group in comparison to LSK and control groups ^c video plus group in comparison to video only group											

						Health and so	cial outcomes	
Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs/ Conceptions
Anderson et al., 1999	RCT (cluster) -	RAP n=185	Delayed intervention n=66	12 months n=251 (54%)	-	-	-	NS pregnancy rates
Dilorio et al., 2006	RCT (cluster) ++	Keepin' It R.E.A.L n=381 LSK, n=187; SCT, n=194)	1 hour HIV prevention session n=201	24 months n=524 (90%)	NS abstinence; intimate behaviours; sexual possibility situations	-	↑ condom use at last sex*	-
McKay et al., 2004; McBride et al., 2007	CBA -	CHAMP n=274	Did not receive intervention n=290	Post-test n=465 (82%)	↓ time in situations of sexual possibility**	-	-	-
Miller et al., 1993	RCT (individual) +	Facts & Feelings n=258 (video only n=132; video + newsletter n=126)	Did not receive intervention n=290	12 mo n=504 (92%)	NS sexual behaviour	-	-	-
*p<0.05; **p<0.0 ^a LSK group com	01; ***p<0.001; ¹ npared to SCT a	p value not reporte nd control groups	d; 1 increase relat	ive to compara	ator; ↓ decrease relative	e to comparator; NS no	ot significant; - outcome	not reported

Table 6.10. Programmes delivered to families: intervention effects on health and social outcomes related to sexual health

6.5 Programmes delivered to parents

6.5.1 Overview of evidence identified

Five studies (Dancy et al., 2006; Dilorio et al., 2007; Forehand et al., 2007; Gustafson, 1998; O'Donnell et al., 2005) examined interventions and programmes delivered to parents that targeted adolescent sexual health. All five studies were conducted in the USA. Programmes were delivered in a variety of settings, with parents as the focus. One intervention was delivered at home to parents via CD-ROM (Saving Sex for Later; O'Donnell et al., 2005), others were delivered in community-based settings using facilitators to provide training sessions to parents and their children (Parents Matter!; Forehand et al., 2007), health experts (The Mother/Daughter HIV risk-reduction [MDRR]; Dancy et al., 2006), parish nurses (Let's Talk: Sex is for Love; Gustafson, 1998) or community-based workers (REAL Men programme; Dilorio et al., 2007).

The theoretical base underpinning the intervention was not reported in two studies (Forehand et al., 2007; Gustafson, 1998), cognitive behavioural skills theory was used in one study (Dancy et al., 2006), social cognitive theory was reported in another study (Dilorio et al., 2007), and the Saving Sex for Later programme (O'Donnell et al., 2005) was reportedly based on both diffusion of innovations theory and the theory of planned behaviour.

The numbers of participating parents and children were not clear for all studies. The total number of parents and youth involved was reported for three studies (Dilorio et al., 2007; O'Donnell et al., 2005; Gustafon, 1998), and ranged from 58 to 674 parents and 63 to 846 children. For the MDRR programme (Dancy et al., 2006), the authors reported only the numbers of daughters included (n=262). A total number of 1,115 participants was reported for the Parents Matter! Programme (Forehand et al., 2007). Power calculations or sample size was mentioned in only one study (REAL Men programme; Dilorio et al., 2007) and was reported to be poor due to small numbers and clusters. A further two studies provided enough information to determine that sample sizes were appropriate (O'Donnell et al., 2005; Forehand et al., 2007). However, insufficient information was provided to determine the appropriate size of the samples in two studies (Dancy et al., 2006; Gustafson, 1998).

The majority of studies focused on young people aged between 9-14 years. However, one study (Gustafson, 1998) focused on a slightly other population aged 12-16 years with a mean age of 14 years. One study (Gustafson, 1998) reported immediate post-test results only and Dancy and colleagues (2006) reported follow-up within 1-2 weeks after participating children had completed their training. A short follow-up time of three months was reported by O'Donnell and colleagues. Thus only two studies (Forehand et al., 2007; Dilorio et al., 2007) reported both short and medium term follow-up.

6.5.2 Quality assessment

Of the five included studies four were based on an RCT design and one used an NRCT design. Of the four RCTs, two were based on cluster randomisation (Dancy et al., 2006; Dilorio et al., 2007) and two were based on individual randomisation (Forehand et al., 2007; O'Donnell et al., 2005). Cluster

randomisation was conducted at community-based intervention site level. The unit of analysis did not match the unit of allocation in one study (Forehand et al., 2007) and no adjustment was reported. All studies were reported as moderate quality (+ rating) and two studies (Forehand et al., 2007; O'Donnell et al., 2005) reported intent to treat analysis. Outcome measures were reported to be reliable in all studies. Relevant outcomes were reported across all included studies.

6.5.3 Findings

6.5.3.1 Knowledge and understanding

Knowledge outcomes were reported in one study (Dancy et al., 2006), which found no significant increase in HIV transmission knowledge in the MDRR group, in comparison to a group who received HIV education delivered by a health expert. However, a significant increase in HIV transmission knowledge was found in the MDRR group in comparison to a group who received a nutrition and exercise intervention.

6.5.3.2 Attitudes and values

Four studies (Dancy et al., 2006; Dilorio et al., 2007; Gustafson, 1998; O'Donnell et al., 2005) examined intervention effects on attitudes and values related to sexual health.

Dancy and colleagues (2006) found no effects of the MDRR intervention on intentions to refuse sex in comparison to a group receiving HIV education taught by health experts (Health Expert HIV Risk Reduction curriculum; HERR). However, in comparison to a group receiving a nutrition and exercise intervention (Mother/Daughter Health Promotion curriculum; MDHP), MDRR participants reported greater intentions to refuse sex (p<0.05). The REAL Men Programme (Dilorio et al., 2007) had nonsignificant short-term effects on fathers' intentions to discuss sex-related topics and sons' intentions to delay sexual intercourse. However, medium-term findings showed significant increases in both reported intentions among intervention participants compared to controls who participated in a nutrition and exercise programme (both p<0.05). Gustafson (1998), reported post-tests results indicating positive programme effects of the Let's Talk programme on Satisfaction with Personal Sexuality scale (p<0.05) and Clarity of Personal Sexual Values scale (p<0.05). Hoever, there was no change in young people's attitudes towards sexuality, values of fidelity, attitude towards the use of force in sexual activity or in their intentions regarding sexual intercourse. O'Donnell and colleagues (2005) reported positive programme effects of the Saving Sex for Later programme on parents' views of their influence over their child's risk-taking behaviour. In logistic regression analyses, parents in the intervention group scored higher than controls on parental influence (adjusted OR 2.15; 95% CI 1.36, 3.41; p<0.001)..

6.5.3.3 Personal and social skills

All five studies reported outcomes relating to personal and social skills. Most studies reported intervention effects on communication. The REAL Men programme (Dilorio et al., 2007) had inconsistent short- and medium-term effects on father and son reports of sexual health-related communication. In comparison to controls who participated in a nutrition and exercise programme, at

both the 3- and 12-month follow-up, intervention fathers reported a positive increase in discussion (p<0.05), with no difference reported at 6-month follow-up. However, sons in the intervention group reported no significant increases, in sex-related communication compared to controls at any follow-up period. Forehand and colleagues (2007) found higher mean changes (p<0.05) in parental reports of sexual communication and parental responsiveness to sexual communication in the short- to mediumterm among parents who received an enhanced communication intervention compared those who received a single session communication intervention and controls who received a general health intervention. The effects of the enhanced intervention on children's reports of sexual communication and parental responsiveness to sexual communication were less consistent; higher mean changes were observed at post-test for both measures compared to the single session and control groups, but not at subsequent follow-ups. Compared to a no intervention control, parents who participated in the Let's Talk programme (Gustafson, 1998) had a significantly greater improvement in their scores on a scale measuring the quality of communication with their child (p<0.05). However, there was no difference on other measures of parental norms (sexual values of fidelity, frequency of communication, and monitoring) or on measures of social support behaviours (family cohesion and shared family activities). O'Donnell and colleagues (2005) found positive intervention effects of the Saving Sex for Later programme on parents' reports of communication (adjusted OR 2.45; 95% CI 1.53, 3.92; p<0.001) and self-efficacy (adjusted OR 1.94; 95% CI 1.21, 3.11; p<0.01), but not on parental monitoring (adjusted OR 1.84; 95% CI 0.91, 3.72). Children whose parents participated in the programme reported positive programme effects on family rules and family support (both p<0.05).

6.5.3.4 Health and social outcomes related to sexual health

Four studies (Dancy et al., 2006; Dilorio et al., 2007; Forehand et al., 2007; O'Donnell et al., 2005) examined intervention effects on health and social outcomes related to sexual behaviours. Dancy and colleagues (2006) reported that there was no difference in sexual activity among MDRR participants and in those who received HIV education taught by health experts (HERR). However, in comparison to a group receiving a nutrition and exercise intervention (MDHP), MDRR participants were less likely to be sexually active at post-test (p<0.05). The REAL Men Programme (Dilorio et al., 2007) had no short or medium-term programme effects on intimate behaviours, or on sexual abstinence rates. However, positive medium-term programme effects were reported for unprotected intercourse. Intervention participants were less likely than controls who participated in a nutrition and exercise programme to report ever having sexual intercourse without a condom (p<0.05). Forehand and colleagues (2007) found that children whose parents attended an enhanced communication intervention group were no more or less likely to be at sexual risk than those in the control group (RR 1.04; 95% CI 0.73, 1.46) or the single session groups (RR 0.98; 95% CI, 0.69, 1.39). Youth in the Saving Sex for Later programme (O'Donnell et al., 2005) showed significant decreases in behavioural risks compared to controls (p<0.05).

6.5.4 Summary and evidence statements

Five studies (Dancy et al., 2006; Dilorio et al., 2007; Forehand et al., 2007; Gustafson, 1998; O'Donnell et al., 2005) were identified that examined sexual health interventions aimed at parents.

One study (O'Donnell et al., 2005) examined intervention effects on a programme delivered via CD-ROM. Two studies (Dancy et al., 2006; Gustafson, 1998) used medically trained people to deliver the programme and two others (Dilorio et al., 2007; Forehand et al., 2007) used community-based workers and facilitators respectively.

One study (Dancy et al., 2006) found that, compared to a nutrition and exercise programme, an intervention which trained mother's to be their daughters' primary HIV educators had short-term significant effects on knowledge of HIV transmission. Four studies (Dancy et al., 2006; Dilorio et al., 2007; Gustafson, 1998; O'Donnell et al., 2005) examined intervention effects on sexual behaviour attitudes and values. Across three studies (Dancy et al., 2006; Dilorio et al., 2007; Gustafson, 1998) that examined effects on intentions there were inconsistent results. Dancy and colleagues (2006) found short-term positive effects of an HIV risk reduction intervention and positive medium-term programme effects were seen in one study (Dilorio et al., 2007). However, Gustafson (1998) found no programme effects on intentions towards sexual intercourse. O'Donnell and colleagues (2005) reported positive programme effects on parents attitudes, including an increase in parents' reported of parental influence on their children's risk-taking behaviour. All five studies (Dancy et al., 2006; Dilorio et al., 2007; Forehand et al., 2007; Gustafson, 1998; O'Donnell et al., 2005) examined outcomes relating to personal and social skills. Dancy and colleagues (2006) found short-term positive effects of an HIV risk reduction intervention on self-efficacy to refuse sex. In addition, generally positive programme effects were reported across the remaining studies with regards to communication, with the exception of the study by Dilorio and colleagues (2007). They found that reports were inconsistent between fathers and sons regarding the communication of sex-related topics with fathers reporting more positively than their sons. Gustafson (1998) reported a positive programme effect on quality of communication and Forehand and colleagues (2007) reported an increase in sexual communication based on reports from both parents and their children who received an enhanced communication intervention; although parent reports were found to be more positive over the medium term. O'Donnell and colleagues (2005) reported positive programme effects on communication across a range of measures including communication, self-efficacy and monitoring.

Four studies (Dancy et al., 2006; Dilorio et al., 2007; Forehand et al., 2007; O'Donnell et al., 2005) examined health and social outcomes related to sexual behaviour. There were positive short-term effects of two parent education programmes (Dancy et al., 2006; O'Donnell et al., 2005) on initiation of sexual activity and behavioural risks related to early sexual initiation, respectively. However, lack of clear intervention effects were reported in two further studies (Dilorio et al., 2007; Forehand et al., 2007).

Evidence statement 11

- 11 (a) There is moderate evidence from one RCT¹ to suggest that training for mothers to be their daughters' primary HIV educator may produce short-term improvements in sexual health-related knowledge and understanding. The evidence may only be partially applicable to the UK as this study was conducted in the USA and focused on ethnic populations specific to the USA.
- 11 (b) There is inconsistent evidence from three RCTs and one NRCT² on which to determine the effects of intervention and programmes delivered to parents on sexual health-related attitudes and values.
- 11 (c) There is weak evidence from three RCTs and one NRCT³ to suggest that interventions delivered to parents may improve parent-child communication about sexual health topics. Findings may only be partially applicable to the UK as all the studies were conducted in the USA and may not be generalisable beyond the populations studied.
- 11 (d) There is inconsistent evidence from four RCTs⁴ on which to determine the effects of programme delivered to parents on their children's sexual behaviour.
- 11 (e) There is moderate evidence from one RCT¹ to suggest that delivery of HIV prevention content by mothers may be as equally effective as that of health experts. The evidence may only be partially applicable to the UK as this study was conducted in the USA and focused on ethnic populations specific to the USA.
- ¹ Dancy et al., 2006 (RCT +)
- ² Dancy et al., 2006 (RCT +); Dilorio et al., 2007 (RCT +); Gustafson, 1998 (NRCT +); O'Donnell et al., 2005 (RCT +)
- ³ Dilorio et al., 2007 (RCT +); Forehand et al., 2007 (RCT +); O'Donnell et al., 2005 (RCT +); Gustafson, 1998 (NRCT +)
- ⁴ Dancy et al., 2006 (RCT +); Dilorio et al., 2007 (RCT +); Forehand et al., 2007 (RCT +); O'Donnell et al., 2005 (RCT +)

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Dancy et al., 2006	RCT (cluster) +	USA n=262 daughters, 100% African American mean 12.4 years	Community	The Mother/Daughter HIV risk-reduction (MDRR) : Aimed to reduce sexual activity, increase HIV transmission knowledge, self-efficacy and intention to refuse sex. Mother's were actively involved with the programme and had 12 weeks training.	Cognitive behavioural skills, Fishbein and Ajzen's behavioural intentions; Collins' community-other- mothers	Health experts
Dilorio et al., 2007	RCT (cluster) +	USA n=554 (fathers and sons) primarily African American 13-14 years old	Boys and Girls Clubs	REAL Men programme: Programme consisted of lectures, discussions, role-plays, games, videotapes and homework as well as weekly goals. Fathers received seven two hour sessions and their sons received one (final) session.	Social cognitive theory	NR
Forehand et al., 2007	RCT (individual) +	USA n=1,115 100% African American 9-12 years old	Community	Parents Matter! : Sexual risk-reduction programme including group sessions focussing on increasing parents' communication about sexual topics. The programme was delivered over five, 2.5 hour sessions using enhanced communication.	Not Reported	Facilitators
Gustafson, 1998	NRCT +	USA n=58 families Majority White 12-16 yrs	Community	Let's Talk: Sex is for Love: Parenting workshop and in- home exercises to complete as a family. Three hour workshop session; four weekly in-home exercises.	Not Reported	Parish nurse
O'Donnell et al., 2005	RCT (individual) +	USA n=846 children n=674 parents 62% Black; 29% Hispanic; 8% other 5 th or 6 th grade	Home	Saving Sex for Later : A CD-based intervention to improve parental communication relating to sexual behaviour. The programme disseminated one CD every 10 weeks for six months.	Diffusion of innovation model, theory of planned behaviour	CD-ROM

Table 6.11. Summary of programme content: programme delivered to parents

		Intervention			Outcomes			
Study	Rating		Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills	
Dancy et al.,	RCT	MDRR	MDHP n=62	PT (91%)	1 knowledge of HIV transmission**	1 intention to refuse sex*	† self-efficacy to refuse sex**	
2006	(cluster) +	n=103	HERR n=97	PT (91%)	NS knowledge of HIV transmission	NS intention to refuse sex	NS self-efficacy to refuse sex	
Dilorio et al., 2007	RCT (cluster) +	REAL Men programme n=141 fathers (52%)	7 session nutrition and exercise programme n= 132 fathers (48%)	3 mo (NR)	-	NS intent to discuss sex- related topics (fathers) NS intentions about delaying sexual intercourse (youth)	 topics (fathers)* NS discussion of sex- related topics (youth) 	
				6 mo (NR)	-	NS Intent to discuss sex- related topics (fathers) NS intentions about delaying sexual intercourse (youth)#	NS discussion of sex- related topics (fathers) NS discussion of sex- related topics (youth)	
				12 mo (80%)	-	 1 intent to discuss sex- related topics (fathers)* 1 Intentions about delaying sexual intercourse (youth)* 	 ↑ discussion of sex-related topics (fathers)* NS discussion of sex- related topics (youth) 	
	RCT (individual) +	Parents Matter! Enhanced n=378	Communication /General health n= 366	PT (NR)	-	-	† sexual communication (parent and child report)* † responsiveness	
				6 mo (NR)	-	-	f sexual communication (parent report)	
Forehand et al., 2007				12 mo (int=84%; con=70%)	-	-	† sexual communication (parent report)	
		ividual) + Parents Matter! Comm Single session /Gene n=371 n		PT (NR)	-	-	↑ sexual communication (parent and child report)* ↑ responsiveness	
			/General health	6 mo (NR)	-	-	-	
			1= 300	12 mo (int=74%; con=70%)	-	-	-	

					Outcomes							
Study	Rating	Intervention	Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills					
				PT (NR)	-	-	 f sexual communication (parent report) f sexual communication (child report)* 					
Forehand et	RCT (individual) +	Parents Matter! Enhanced	Parents Matter! Single session	6 mo (NR)	-	-	f sexual communication (parent report)					
al., 2007		n=378	n=371	12 mo (Enhanced =84%; Single=74 %)	-	-	† sexual communication (parent report)					
Gustafson, 1998	NRCT +	Let's Talk: Sex is for Love n= 34 families	No intervention n= 24 families	PT (int=78%; con=97%)	-	 † 'Satisfaction with Personal Sexuality' scale* † 'Clarity of Personal Sexual Values'* NS attitude toward sexuality NS sexual values of fidelity NS attitude toward the use of pressure and force in sexual activity NS intentions of sexual intercourse 	 [†] Quality of Communication with Teen scale* NS sexual values of fidelity NS frequency of communication NS frequency of monitoring NS family cohesion NS shared familiy activities NS quality of communication with mother or father NS frequency of communication, understanding personal sexual response NS skills to avoid sexual pressure 					
O'Donnell et al., 2005	RCT (individual) +	Saving Sex for Later n=423 children n=337 parents	No intervention n=423 children n= 337 parents	3 mo (68%)	-	↑ parental influence (parent report)***	Parents reports - t communication*** NS monitoring t self-efficacy** Youth reports – t family rules* t family support*					
*p<0.05; **p<0.0 to those not sex	01; ***p<0.001; ¹ zually active	p value not report	*p<0.05; **p<0.01; ***p<0.001; ¹ p value not reported; ↑ increase relative to comparator; ↓ decrease relative to comparator; NS not significant; - outcome not reported; #Limited to those not sexually active									

Study			Comparator		Health and social outcomes			
	Rating	Intervention		Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs/ Conceptions
Dancv et al	RCT	MDRR	MDHP n=62	PT (91%)	↓ sexual activity*	-	-	-
2006	(cluster) +	n=103	HERR n=97	PT (91%)	NS sexual activity			
Dilorio et al., 2007	RCT (cluster) +	REAL Men programme n=141 fathers (52%)	7 session nutrition and exercise programme n= 132 fathers (48%)	3 mo (NR)	NS intimate behaviours NS sexual abstinence	-	NS ever sexual intercourse without condom	-
				6 mo (NR)	NS intimate behaviours NS sexual abstinence	-	NS ever sexual intercourse without condom	-
				12 mo (80%)	NS intimate behaviours NS sexual abstinence	-	↓ ever sexual intercourse without condom	-
Forehand et al., 2007	RCT (individual) +	Parents Matter! Enhanced n=378	Communication/ general health n= 366	12 mo (int=84%; con=70%)	NS at sexual risk	-	-	-
		Parents Matter! single session n=371	Communication/ general health n= 366	12 mo (int=74%; con=70%)	NS at sexual risk	-	-	-
		Parents Matter! Enhanced single session (En n=378 n=371 S	12 mo (Enhanced=84%; Single=74%)	NS at sexual risk	-	-	-	
O'Donnell et al., 2005	RCT (individual)+	Saving Sex for Later n= 423 children n= 337 parents	No intervention n= 423 children n= 337 parents	3 mo (68%)	↓ behavioural risks*	-	-	-
*p<0.05; **p<0	.01; ***p<0.001	; ¹ p value not repo	rted; 1 increase rela	ative to comparator;	↓ decrease relative to com	nparator; NS not	significant; - outcome no	ot reported

Table 6.13. Programme delivered to parents: effects on health and social outcomes related to sexual health

6.6 Programmes involving the wider community or mass media

6.6.1 Overview of evidence identified

Two studies were identified (Doniger et al., 2001; Sieverding et al., 2005) that examined interventions that involved the wider community or mass media. Both studies were conducted in the USA. Doniger et al (2001) examined a mass media intervention, Not Me, Not Now, which involved paid television and radio advertising, billboards, posters distributed in schools in addition to educational materials for parents and education programming in schools. The duration of the programme was five years. Sieverding et al (2005) examined the Youth United through Health Education (YUTHE) outreach programme. Peer educators targeted young people between the ages of 12 and 22 years to undertake sexual risk assessment and provide information on STIs and STI screening. Young people who participated in the intervention were also provided with condoms.

Neither study reported the theoretical model underpinning intervention. As the interventions examined targeted the wider community it was not clear how many young people received the intervention in either study.

6.6.2 Quality assessment

Both studies were based on cross-sectional time series (Doniger et al., 2001; Sieverding et al., 2005). Sieverding and colleagues (2005) examined Chlamydia rates over a 5-year period (1998-2002) and Doniger and colleagues (2001) examined pregnancy rates for 15-17 year olds over a 3-year period (1993-1996). The study by Doniger and colleagues (2001) was rated poorly as the study was likely to be subject to bias, as the analyses did not appear to adequately account for natural variations in the data over time. The study by Sieverding and colleagues (2005) appeared to have been appropriately conducted and was rated moderate quality.

6.6.3 Findings

6.6.3.1 Knowledge and understanding

None of the included studies examined intervention effects on knowledge and understanding.

6.6.3.2 Attitudes and values

None of the included studies examined intervention effects on attitudes and values.

6.6.3.3 Personal and social skills

None of the included studies examined intervention effects on personal and social skills.

6.6.3.4 Health and social outcomes related to sexual health

Doniger et al (2001) examined the effects of Not Now, Not Me, an abstinence-oriented communications programme. Based on the analysis of pregnancy rates for 15-17 year olds across five geographic areas, the authors noted a statistically significant downward trend between 1993 and 1996 in four areas, including the intervention area. Based on further analyses of the slope of a regression line the authors reported that the rate of decline was fastest in the intervention area.

Sieverding and colleagues (2005) examined the impact of a community outreach programme on STI rates over a 5-year period. For both males and females in the intervention neighbourhood, Chlamydia rates remained relatively stable over the 5-year period (1998-2002) and both males and females in the comparison neighbourhood were significantly more likely to have Chlamydia than those in the intervention neighbourhoods (females: OR 3.0; 95% CI 2.3, 3.9; p<0.001 / males: OR 2.9; 95% CI 2.0, 4.4; p<0.001). As there were much fewer cases of Chlamydia in the youngest adolescents, the authors further examined rates in the older youth aged 18–22. Among females and males aged 18–22, those in the comparison neighbourhood were significantly more likely to have Chlamydia than their counterparts in the intervention neighbourhood (females: OR 2.3; 95% CI 1.7, 3.2; p<0.001 / males: OR 2.3; 95% CI 1.5, 3.5; p<0.001).

6.6.4 Summary and evidence statements

Two studies were identified (Doniger et al., 2001; Sieverding et al., 2005) that examined interventions that involved the wider community or mass media. Doniger and colleagues (2001) examined a mass media intervention, Not Me, Not Now, and Sieverding and colleagues (2005) examined the Youth United through Health Education (YUTHE) outreach programme.

Neither of the included studies examined intervention effects on knowledge, attitudes and skills. Both studies analyses population-level changes, in pregnancy (Doniger et al., 2001) and STI rates (Sieverding et al., 2005), respectively. Both studies reported positive intervention effects at a population level, however the study by Doniger and colleagues (2001) did not adequately control for natural fluctuations in the data and therefore it is not clear whether these or intervention effects were responsible for the differences seen in the intervention and control communities.

Evidence statement 12

- 12 (a) There is no evidence from two CTS¹ on which to determine the effects of interventions and programmes involving the wider community or mass media on knowledge, attitudes and skills related to sexual health.
- 12 (b) There is weak evidence from one CTS² to suggest that a programme of community outreach may have a positive impact on STI rates among young people. Findings may only be partially applicable to the UK as the study was conducted in the USA and may not be generalisable beyond the population studied.
- ¹ Doniger et al., 2001 (CTS –); Sieverding et al., 2005 (CTS +)
- ² Sieverding et al., 2005 (CTS +)

Table 6.14. Summary	v of programme content: program	nme involving the wider co	ommunity or mass media

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Sieverding et al., 2005	CTS +	USA n=NR 87% African American 12-22 years	Community outreach	Youth United Through Health Education: Outreach programme; sexual risk assessment, information on STIs and STI screening sites, role model stories and condoms	NR	Peer educators
Doniger et al., 2001	CTS -	USA n=NR Ethnicity=NR 15-17 years	Mass media	Not Me, Not Now: Paid television and radio advertising, billboards, posters distributed in schools, educational materials for parents and an educational series presented in schools (Postponing Sexual Involvement), sponsorship of community events, website	Social learning theory, consumer information processing theory	NA

Table 6.15. Programme involving the wider community or mass media: effects on health and social outcomes related to sexual health

Study	Rating	Intervention	Comparator	Follow-up	Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs/ Conceptions
Doniger et al., 2001	CTS -	Not Me, Not Now n=NR	No intervention n=NR	1993-1996	-	-	-	1 rate of decline in pregnancy rates ¹
Sieverding et al., 2005	CTS +	YUTHE n=1 neighbourhood	No intervention n=1 neighbourhood	1998-2002	-	-	-	↓ rate of Chlamydia infection***
*p<0.05; **p<0.01; ***p<0.001; 1 p value not reported; 1 increase relative to comparator; \downarrow decrease relative to comparator; NS not significant; - outcome not reported								

6.7 Programmes targeting vulnerable young people

6.7.1 Overview of evidence identified

Three studies (Gleghorn et al., 1997; Rew et al., 2007; Slesnick and Kang, 2008) examined community interventions aimed at vulnerable groups. All three studies examined interventions which specifically targeted young homeless people. Gleghorn and colleagues (1997) examined an HIV prevention intervention that combined street outreach, storefront prevention services and tailored prevention materials including posters, T-shirts, condom packets and magazines. Rew and colleagues (2007) examined a brief sexual health intervention, which consisted of group sessions based around role-play and discussion. The study by Slesnick and Kang (2008) examined an integrated cognitive-behavioural and HIV prevention intervention, which combined a community reinforcement approach with HIV prevention content drawn from the Becoming a Responsible Teen programme.

All three studies were conducted in North America and were delivered in the community using outreach workers (Gleghorn et al., 1997), trained health educators (Rew et al., 2007) or therapists (Slesnick and Kang, 2008). One study (Gleghorn et al., 1997) did not report a theoretical basis. Rew and colleagues (2007) reported using theory of reasoned action and social cognitive theory and Slesnick and Kang (2008) reported using cognitive behavioural theory. The sample size was small in one study (Slesnick and Kang, 2008; n=180; n=96 intervention, n=84 control) and reasonably large in two studies (Gleghorn et al., 1997; T1=429, T2=717: Rew et al., 2007; n=572). However, no studies reported power calculations or limitations based on sample size.

All three studies examined homeless youth aged over 19 years. However, samples also included youth aged 12 years thus justifying the studies' inclusion. Follow-up times were unclear in one study (Gleghorn et al., 1997), Rew and colleagues (2007) reported follow-up assessments at immediate post-test and 3-6 weeks after the end of the programme. Slesnick and Kang (2008) reported follow-up assessments at 3 and 6 months after baseline assessment, however data were presented as changes over time.

6.7.2 Quality assessment

Of the three studies, one (Slesnick and Kang, 2008) was an RCT design based on individual randomisation. The two remaining studies were based on NRCT study designs. The RCT study was rated moderate quality (+ rating) and used an intent-to-treat design. However, the possibility of contamination was mentioned. One NRCT study (Rew et al., 2007) was also rated moderate quality and considered the possibility of contamination in the choice of research design. The NRCT study by Gleghorn and colleagues (1997) was rated poor quality as details of the study were not fully reported and scales used were not validated. Relevant outcomes were reported across all three studies.

6.7.3 Findings

6.7.3.1 Knowledge and understanding

Only one study (Rew et al., 2007) reported outcomes relating to knowledge and understanding. The authors reported an increase in HIV and STI knowledge in both the intervention and control groups at follow-up (both p<0.001). However, findings showed that for the intervention group, knowledge decreased from baseline to post-test and remained stable to the 3-6 week follow-up. For the control group, knowledge remained stable from baseline to post-test and then decreased at the 3-6 week follow-up.

6.7.3.2 Attitudes and values

None of the included studies examined intervention effects on attitudes and values.

6.7.3.3 Personal and social skills

Only one study (Rew et al., 2007) reported outcomes relating to personal and social skills. Selfefficacy for breast self-examination in women and testicular self-examination in men increased from baseline to post-test then remained stable. However, this was also the case for the control group reporting self-efficacy for testicular self-exam. No programme effects were seen for condom selfefficacy or assertive communication.

6.7.3.4 Health and social outcomes related to sexual health

All three studies examined health and social outcomes related to sexual health. Gleghorn and colleagues (1997) found no significant effects of a street outreach programme on young homeless people's use of condoms with either a main or casual partner and there was no effect of the programme on HIV-related referrals. In addition, Rew and colleagues (2007) found no effects of a group-based brief sexual health intervention on safe sex behaviours or on sexual risk-taking behaviours. Slesnick and Kang (2008) found that although there was no overall effect of combined community reinforcement therapy and HIV prevention sessions, frequency of condom use increased among subpopulations in both the intervention and control groups. Younger intervention participants (14-18 years) and older control and intervention participants (18-22 years) increased their condom use frequency at 6 months post-baseline. Intervention youths aged 14-18 years were more likely than control youths to report more frequent condom use (p<0.01).

6.7.4 Summary and evidence statements

Three studies (Gleghorn et al., 1997, Rew et al., 2007, Slesnick and Kang, 2008) were identified that examined the effectiveness of community-based programmes on vulnerable populations. One programme (AESOP) was delivered by outreach workers, another (unnamed sexual health intervention) used healthcare educators and a third (community reinforcement approach) used therapists to deliver the intervention.

Intervention effects on knowledge and skills were examined by one study (Rew et al., 2007) and none of the included examined intervention effects on attitudes and values. There were limited effects of a brief sexual health intervention (Rew et al., 2007) on knowledge relating to AIDS and other STIs, and
on communication and self-efficacy. Health and social outcomes related to sexual health were examined in all three studies, two of which reported no intervention effects (Gleghorn et al., 1997, Rew et al., 2007). Slesnick and Kang (2008) found a positive effect on the frequency of condom use among younger participants in a programme which combined a community reinforcement approach with HIV prevention content.

Evidence statement 13

- 13 (a) There is insufficient evidence from one NRCT¹ to determine effects of interventions and programmes targeting vulnerable populations on sexual health-related knowledge and understanding, and personal and social skills.
- 13 (b) There is inconsistent evidence from two NRCT and one RCT² on which to determine effects of interventions and programmes targeting vulnerable populations on health and social outcomes relating to sexual health.
- ¹ Rew et al., 2007 (NRCT +)
- ² Gleghorn et al., 1997 (NRCT -); Rew et al., 2007 (NRCT +); Slesnick and Kang, 2008 (RCT +)

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
Gleghorn et al., 1997	NRCT -	USA T1, n=429, T2, n=717, White: T1=155, T2=169 12-23 years	Outreach with homeless/ contact at youth centre	AIDS Evaluation of Street outreach Project (AESOP): This programme aimed to reduce youth HIV risk behaviours in homeless populations with the use of outreach workers. The intervention was compared to sites with limited HIV services and no regular outreach with no subculture-specific interventions and no youth- oriented HIV prevention centres	NR	Outreach workers
Rew et al., 2007	NRCT +	USA n=572, majority White ~58%, also African American, Asian American, Hispanic, American Indian, Multi-ethnic, Other 16-23 years	Community	A sexual health promotion intervention with homeless youth using eight one hour taught sessions to increase sexual health knowledge and self-efficacy.	Theory of reasoned action, Social cognitive theory	Healthcare educators
Slesnick and Kang, 2008	RCT (individual) +	USA n=180, 13% native American, 1% Asian, 3% African American, 30% Hispanic, 41% White, 12% Other 14-22 years	Community	Community reinforcement approach : An HIV risk- reduction programme using trained therapists to provide skills building and education to homeless youth. Compared to youth only accessing drop-in centres with links to case management, support, youth and community services and HIV testing and counselling.	Cognitive behavioural theory	Therapists

Table 6.16. Summary of programme content: programmes targeting vulnerable young people

		Intervention	Comparator	Follow-up	Outcomes			
Study	Rating				Knowledge and understanding	Attitudes and values	Personal and social skills	
Rew et al., 2007	NRCT +	Unnamed - sexual health intervention n=196	No intervention n= control only: 287, both int & cont:89	PT, 3-6 weeks (NR)	NS HIV/STI knowledge	-	NS Condom self-efficacy NS Assertive communication 1 self-efficacy for breast self-exam in women** 1 self-efficacy for testicular self exam in men***	
*p<0.05; **p<0.	01; ***p<0.001; ¹	^I p value not reported	d; ↑ increase relati	ve to compara	tor; ↓ decrease relative to com	parator; NS not significant; - o	utcome not reported	

Table 6.17. Programmes targeting vulnerable young people: effects on knowledge, attitudes and skills

Table 6.18. Programmes targeting vulnerable young people: effects on health and social outcomes related to sexual health

		Intervention	Comparator	Follow-up	Sexual health outcomes			
Study	Rating				Age of initiation	Frequency/ Number of partners	Contraceptive use	STIs/ Conceptions Other
Gleghorn et al., 1997	NRCT -	AESOP n= T1=246, T2=392	Sites without regular outreach n= T1=183, T2=325	NR	-	-	NS condom use at last sex with main partner NS condom use at last sex with casual partner	NS HIV referrals
Rew et al., 2007	NRCT +	Brief sexual health intervention n=196	No intervention n= control only: 287, both int & cont:89	PT, 3-6 weeks (NR)	-	NS safe sex behaviours NS sexual risk- taking behaviours	-	-
Slesnick and Kang, 2008	RCT (individual) +	Community reinforcement approach + HIV prevention n= 96	Treatment as usual n= 84	3 mo (73%), 6 mo (86%)	-	-	frequency of condom use (younger participants only**)	-
*p<0.05; **p<0.0)1; ***p<0.001; [¶]	^I p value not reported	d; ↑ increase relati	ve to compara	tor; ↓ decrease relative	e to comparator; NS not	significant; - outcome r	not reported

7 Programmes targeting multiple health behaviours

No systematic reviews or meta-analyses were identified for inclusion in the review of programmes targeting multiple health behaviours, and in addition no economic evaluation studies were identified. Five articles, which reported on evaluations of programmes and interventions that addressed both alcohol use and sexual health, were identified. Two articles reported on studies that examined interventions or programmes delivered in social, healthcare or community settings and three articles reported on studies that examined interventions or programmes delivered in social, healthcare or community settings and three articles reported on studies that examined interventions or programmes delivered in social, healthcare or community settings and three articles reported on studies that examined interventions or programmes delivered to families or parents.

7.1 Programmes delivered within social, healthcare and community settings

7.1.1 Overview of evidence identified

Two studies (St Pierre et al., 1995; Wiggins et al., 2009) examined programmes which targeted both sexual health and alcohol use. St Pierre et al (1995) examined the effects of Stay SMART, which targeted young people enrolled in Boys and Girls Clubs, with and without the addition of a peer leadership component. Wiggins et al (2009) evaluated the effectiveness of the Young People's Youth Development (YPYD) programme in reducing teenage pregnancy, substance use and other outcomes. The programme targeted young people considered to be at risk of teenage conception, substance misuse or exclusion from school.

The numbers of participants included in the two studies (St Pierre et al., 1995; Wiggins et al., 2009) were 359 and 2,724, respectively, and both studies targeted young people between the ages of 13 and 15 years. Both studies reported long-term follow-up, of 27- and 18-months, respectively.

7.1.2 Quality assessment

The study by St Pierre and colleagues (1995) was an NRCT. Participants were allocated to the intervention and control groups within club groupings by the researchers. Overall, the study design used was rated poorly. The authors excluded participants who did not complete a set number of sessions, which limited the generalisibility of the study, and resulted in large losses to follow-up. The study by Wiggins and colleagues (2009) was based on a CBA design. Allocation was not controlled by the research team and participants for a comparison group were recruited from agencies that had not received funds to run the YPYD programme. Overall, given the limitations of the study design it was well reported and appeared to have been well conducted. The study was rated moderate quality.

7.1.3 Findings

7.1.3.1 Knowledge and understanding

None of the included studies examined intervention effects on knowledge and understanding.

7.1.3.2 Attitudes and values

Both studies examined intervention effects on attitudes and values. St Pierre et al (1995) reported that Stay SMART only participants who were sexually active at baseline perceived significantly fewer social benefits from engaging in sexual activity across all three follow-ups compared to the Stay SMART + boosters participants (p<0.01) and control participants (p<0.01). Among participants who were virgins at baseline there were no significant effects of either intervention condition. Wiggins et al (2009) found that female participants were more likely than control participants to report that they expected to be a parent by age 20 (weighted adjusted OR 1.61; 95% CI: 1.07, 2.41; p<0.05).

7.1.3.3 Personal and social skills

None of the included studies examined intervention effects on personal and social skills.

7.1.3.4 Health and social outcomes related to sexual health

Both studies examined intervention effects on health and social outcomes relating to sexual health and alcohol use. St Pierre et al (1995) reported that there was no impact on sexual behaviour at the 15-months follow-up of either intervention condition among all participants. Stay SMART participants who were sexually active at baseline reported significantly less sexual behaviour compared to the Stay SMART + booster group and the control group at 27-months (both p<0.05). No statistically significant effects were observed among participants who were virgins at baseline. Wiggins et al (2009) found that the YPYD programme had a negative impact on participant's sexual behaviour. At follow-up, significantly more pregnancies were reported among females in the YPDP groups than in the comparison group (weighted adjusted OR 5.48; 95% CI 2.18, 13.75; p<0.01) and significantly more females in the YPDP group than in the comparison group reported heterosexual sex at follow-up 2 (weighted adjusted OR 3.48; 95% CI 1.49, 8.12). There was no difference in rates of monthly drunkenness among programme and comparison participants.

7.1.4 Summary and evidence statements

Two studies (St Pierre et al., 1995; Wiggins et al., 2009) examined programmes which targeted both sexual health and alcohol use. St Pierre et al (1995) examined the effects of Stay SMART, which targeted young people enrolled in Boys and Girls Clubs, with and without the addition of a peer leadership component. Wiggins et al (2009) evaluated the effectiveness of the Young People's Youth Development (YPYD) programme in reducing teenage pregnancy, substance use and other outcomes.

Neither of the included studies examined intervention effects on knowledge and understanding, or on personal and social skills. However, both studies examined intervention effects on attitudes and values. St Pierre and colleagues (1995) found a favourable reduction in sexual attitudes but only among sexually experienced participants who received the intervention without the additional booster sessions. The YPYD programme (Wiggins et al., 2009) had potentially harmful effects on attitudes, with female intervention participants more likely than control participants to report that they expected to be a parent by age 20.

Both studies examined intervention effects on health and social outcomes related to sexual health, and Wiggins and colleagues (2009) examined effects on alcohol use. The effects of the Stay SMART intervention were inconsistent across the two intervention conditions examined. The YPYD programme (Wiggins et al., 2009) had a negative impact on participant's sexual behaviour, particularly among intervention females who were significantly more likely than controls to engage in

heterosexual sexual intercourse and more likely to become pregnant. There was no effect of the programme on male participants or on participant's alcohol use.

Evidence statement 14

- 14 (a) There is weak and inconsistent evidence from one NRCT and one CBA study¹ on which to determine the effects of programmes delivered in social and community settings on attitudes and values related to sexual health and alcohol use.
- 14 (b) There is weak and inconsistent evidence from one NRCT² on which to determine the effects of programmes delivered in social and community settings that seek to address both sexual health and alcohol use.
- 14 (c) There is weak evidence from one CBA study³ to suggest that youth development programmes, which target young females at behavioural risk, may have a negative effect on sexual behaviours. This evidence is applicable as the study was conducted in the UK.
- ¹ St Pierre et al., 1995 (NRCT -); Wiggins et al., 2009 (CBA +)
- ² St Pierre et al., 1995 (NRCT -)
- ³ Wiggins et al., 2009 (CBA +)

Table 7.1. Summar	of programme content: programmes delivered in s	ocial, healthcare or community settings

Author	Study design and rating	Baseline population	Setting	Programme components	Theoretical base	Provider(s)
St Pierre et al., 1995	NRCT -	USA n=359 45% White, 14% Hispanic, 42% Black 13-14 years	Boys & Girls Clubs	Stay SMART: 12 sessions + 8 booster sessions; including components of LST. SMART leaders peer leader programme designed to build upon skills and knowledge.	NR	NR
Wiggins, 2009	CBA +	UK n=2,724 Int/Con: Black or minority ethnic 23%/20% 13-15 years	Youth agencies	Young People's Youth Development programme: 6-10 hours, one week a year; Education, training/employment opportunities, life skills, mentoring, volunteering, health education, arts, sports and advice on accessing services.	NR	Youth agency staff

Table 7.2. Programmes delivered in social, healthcare or community settings: effects on knowledge, attitudes and skills

	Rating	Intervention	Comparator	Follow-up	Outcomes			
Study					Knowledge and understanding	Attitudes and values	Personal and social skills	
St Pierre et al., 1995	NRCT -	Stay SMART only n=119 Stay SMART + boosters n=117	No intervention n=123	3, 15, 27 mo (76%, 55% 42%)	-	↓ perceived social benefits from engaging in sexual activity (Stay SMART only vs. control; non-virgins only**)	-	
		Stay SMART + boosters n=117	Stay SMART only n=119	3, 15, 27 mo (NR)	-	↑ perceived social benefits from engaging in sexual activity (non-virgins only**)	-	
Wiggins, 2009	CBA +	YPYDP n=1,637	No intervention n=1,087	9, 18 mo (int=43%; con=31%)	-	↑ expected to be parent by age 20 (females only*)		
*p<0.05; **p<0.0	01; ***p<0.001; ¹	p value not reporte	d; 1 increase rela	ive to compara	ator; ↓ decrease relative to cor	nparator; NS not significant; - o	utcome not reported	

Ctudy	Dating	Intervention	Comparator		Health and social outcomes		
Study	Rating	Intervention	Comparator	Follow-up	Sexual health	Alcohol use	
St Pierre et	NRCT -	Stay SMART only n=119 Stay SMART + boosters n=117	No intervention n=123	3, 15 mo (NR)	NS sexual intercourse	-	
				27 mo (76%, 55% 42%)	↓ sexual intercourse (Stay SMART only vs. control; non-virgins only*)	-	
. ,		Stay SMART + Stay SMA boosters only n=117 n=119	Stay SMART	3, 15 mo (NR)	NS sexual intercourse	-	
			n=119	27 mo (NR)	↑ sexual intercourse (non-virgins only*)	-	
Wiggins, 2009	CBA +	YPYDP n=1,637	No intervention n=1,087	9, 18 mo (int=43%; con=31%)	↑ pregnancy (females only**) ↑ heterosexual sex (females only [¶])	NS drunkenness	
*p<0.05; **p<0.0	01; ***p<0.001; ¹	^I p value not reporte	d; 1 increase rela	tive to compara	ator; \downarrow decrease relative to comparator; NS no	t significant; - outcome not reported	

Table 7.3. Programmes delivered in social, healthcare or community settings: effects on health and social outcomes related to sexual health and alcohol use

7.2 Programmes delivered to families or parents

7.2.1 Overview of evidence identified

Five studies (Haggerty et al., 2007; Prado et al., 2007; Stanton et al., 2000; 2004; Wu et al., 2003) examined interventions and programmes delivered to families or parents, which targeted both alcohol use and sexual health. Haggerty and colleagues (2007) examined the Parents Who Care programme, which was designed to prevent substance use and other problem behaviours. Stanton and colleagues (2000; 2004; Wu et al., 2003) examined a parental monitoring intervention, Informed Parents and Children Together (ImPACT) that was designed to reduce substance use, sexual risk behaviours and truancy. Prado and colleagues (2007) examined Familias Unidas + Parent Preadolescent Training for HIV Prevention (PATH), a programme that specifically targeted Hispanic parents and aimed to prevent adolescent substance use and unsafe sexual behaviours.

All five studies were conducted in North America and were delivered in a community setting by trained facilitators, project staff or family consultants. One study, of the Familias Unidas + PATH programme (Prado et al., 2007), reported using an ecodevelopmental theory to underpin the programme, whereas Stanton and colleagues (2000; 2004; Wu et al., 2003) reported using a social cognitive model in conjunction with protection motivation theory, and the basis for intervention in the study by Haggerty and colleagues (2007) was the social development model.

Sample sizes ranged from 237 parent-child dyads (Stanton et al., 2000) to 817 families (Stanton et al., 2004; Wu et al., 2003). Prado and colleagues (2007) reported acceptable power calculations of 80% and, although power was not specified in the study write up, enough information was provided in the study by Stanton and colleagues (2004) to determine that the sample size was appropriate. Haggerty and colleagues (2007) did not specify whether their study was sufficiently powered.

The mean age of youth participants recruited to participate in the included studies was a mean 13-14 years. Evaluations of all three programmes (Haggerty et al., 2007; Prado et al., 2007; Stanton et al., 2004) were based on short and medium-term follow-up (6 and 12 months) and two studies (Haggerty et al., 2007; Stanton et al., 2004; Wu et al., 2003) also reported data from long-term follow-up assessments up to 24 months.

7.2.2 Quality assessment

All five studies were based on an RCT design. Two articles of the FOK + ImPACT evaluation (Stanton et al., 2004; Wu et al., 2003) utilised cluster randomisation and three were based on individual randomisation (Haggerty et al., 2007; Prado et al., 2007; Stanton et al., 2000). The unit of analysis matched the unit of allocation in the cluster RCT and intraclass correlations were used to adjust the findings (Stanton et al., 2004; Wu et al., 2003). Of the RCTs based on individual randomisation, two (Prado et al., 2007; Haggerty et al., 2007) reported that their analyses were based on an intention to treat design. The study by Stanton and colleagues (2004; Wu et al., 2003) was rated good quality (++ rating) as it presented a good research design and the methodology and results were well reported. The remaining three studies (Prado et al., 2007; Haggerty et al., 2007; Stanton et al., 2000) were

rated moderate quality (+ rating) and were also clearly presented. However, validity of scales were not reported in the study by Prado and colleagues (2007), two studies (Prado et al., 2007; Haggerty et al., 2007) lacked information to judge whether allocation to intervention and control groups had been adequately concealed, and the study by Stanton and colleagues (2000) did not carry out their analyses on an intent to treat basis and power calculations were lacking. Outcomes were deemed relevant in all five studies.

7.2.3 Findings

7.2.3.1 Knowledge and understanding

None of the included examined intervention effects on knowledge and understanding.

7.2.3.2 Attitudes and values

Four studies (Haggerty et al., 2007; Stanton et al., 2000; Wu et al., 2003; Stanton et al., 2004) examined intervention effects on attitudes and values towards risk behaviour. Haggerty and colleagues (2007) evaluation of Parents Who Care found no significant programme effects at posttest and medium-term follow-up on attitudes towards and perceived harm of substance abuse. However, by long-term follow-up significantly less favourable attitudes to substance abuse were reported (p<0.05), but differences between groups' perceived harms of substance abuse remained non-significant. There were limited effects of ImPACT (Stanton et al., 2000) on parent and adolescent agreement regarding their involvement in risky behaviours and Wu and colleagues (2003) found no difference in risk taking intentions between FOK + ImPACT and FOK only participants at 6- and 12- month follow-up. Based on longer term follow-up of FOK + ImPACT participants (Stanton et al., 2004), compared to the FOK only group, there were positive programme effects on self-efficacy for stopping having sex until older (p<0.01), getting condoms (p<0.05), refusing sex without a condom (p<0.01), refusing sex if asked by a partner (p<0.05), not feeling the need to have sex with a long-time partner (p<0.05), not needing to have sex even if all friends are having sex (p<0.05), and overall response efficacy (p<0.05).

7.2.3.3 Personal and social skills

Three studies (Prado et al., 2007; Stanton et al., 2000; 2004) reported outcomes relating to personal and social skills. Growth curve analysis from the Familias Unidas + PATH programme (Prado et al., 2007) showed that compared to the ESOL (English for Speakers of Other Languages) + PATH the intervention group showed increased family functioning (p<0.05), positive parenting (p<0.05), and parent-adolescent communication (p<0.05). In comparison, relative to a second control group ESOL + HEART (HeartPower! For Hispanics), the intervention group compared showed greater increases in family functioning (p<0.001) and positive parenting (p<0.05). However, there was no difference in parent-adolescent communication between these groups. Two studies (Stanton et al., 2000; Wu et al., 2003) examined intervention effects on adolescent and parent perceptions of parental monitoring and communication, and there were mixed effects of FOK + ImPACT on adolescent's perceptions of parental communication and monitoring. At 6-month follow-up,

FOK + ImPACT participants reported significantly higher perceptions of parental monitoring, but there were no differences in perceptions of monitoring or open communication at the 12-month follow-up. Stanton and colleagues (2004) found a positive programme effect for communication with the family about HIV/AIDS, but only in the FOK + ImPACT + boosters intervention group in comparison to the FOK only group. No effects of FOK + ImPACT were seen for condom-related skills, which included asking for condoms in a clinic or store, putting condoms on correctly, feeling able to convince a sexual partner to use a condom, asking partner about past relationships, or wanting to wait until they were older to have sex again.

7.2.3.4 Health and social outcomes related to sexual health

Four studies (Haggerty et al., 2007; Prado et al., 2007; Wu et al., 2003; Stanton et al., 2004) presented findings related to health outcomes for both alcohol and sexual behaviours. No programme effects were seen for Familias Unidas + PATH (Prado et al., 2007) for past 90-day alcohol use, or unprotected sexual behaviours when compared to either control group (ESOL + PATH: HEART + PATH). Compared to the ESOL + PATH control group, the intervention group showed decreased rates of STIs and unsafe sex at last intercourse (p<0.05). However, compared to the ESOL + HEART control group, the intervention group showed better outcomes only for decreased incidence of STIs (p<0.05). Haggerty and colleagues (2007) reported no significant effects of either version of Parents Who Care on substance use initiation at long-term follow-up and a reduction in sexual initiation was only seen among African American participants who received the group administered version of the programme. There were also limited effects of ImPACT + FOK (Wu et al., 2003; Stanton et al., 2004). At the 6-month follow-up, compared to FOK only participants. FOK + ImPACT participants were less likely to report having had sexual intercourse (p<0.05), unprotected intercourse (p<0.01) and drinking alcohol (p<0.05). However, at 12-month follow-up, the only significant difference that remained between FOK + ImPACT and FOK only participants was on the measure of alcohol use (p<0.01). At the 24-month follow-up (Stanton et al., 2004) there were no significant differences between FOK + ImPACT and FOK only participants on any of the sexual health or alcohol use measures, but participants who received FOK + ImPACT + boosters were less likely to report being pregnant or getting a girl pregnant (p<0.05) compared to FOK only participants. Participants who received FOK + ImPACT + boosters also showed a positive effect only in the proportion of people asking a recent partner if they always used condoms (p<0.01).

7.2.4 Summary and evidence statements

Five studies (Haggerty et al., 2007; Prado et al., 2007; Stanton et al., 2000; 2004; Wu et al., 2003) examined interventions and programmes delivered to families or parents, which targeted both alcohol use and sexual health. All five studies were conducted in the USA and were delivered by either project staff or trained facilitators.

None of the included studies examined intervention effects on knowledge and understanding. Across four studies (Haggerty et al., 2007; Stanton et al., 2000; Wu et al., 2003; Stanton et al., 2004) that examined intervention effects on attitudes and values towards risky behaviours there were indications

of mixed intervention effects. Haggerty and colleagues (2007) found positive long-term effects of the Parents Who Care programme on attitudes towards substance use and there were also long-term positive programme effects of FOK + ImPACT (Stanton et al., 2004) on attitudes and values related to a range of risky behaviours. Three studies (Prado et al., 2007; Stanton et al., 2000; 2004) examined intervention effects on personal and social skills, finding mixed programme effects on parent/family-child communication. Prado and colleagues (2007) found positive intervention effects on communication, family functioning and positive parenting, and Stanton and colleagues (2004) found a positive effect of the FOK + ImPACT + boosters condition on parent-child communication about HIV/AIDS.

Four studies (Haggerty et al., 2007; Prado et al., 2007; Wu et al., 2003; Stanton et al., 2004) examined intervention effects on health outcomes related to alcohol use and sexual health. Short-term to medium-term reductions in alcohol drinking were found for participants who received FOK + ImPACT, but this reduction was not sustained and no other significant programme effects were found for health outcomes related to alcohol use. One study (Prado et al., 2007) reported a decrease in incidence rates for STIs and unsafe sex at last sexual intercourse in the intervention group compared to controls (ESOL + PATH). Although, short-term benefits of FOK + ImPACT were also reported, these differences were not sustained and over the longer term there were no additional positive effects on sexual behaviour of the ImPACT programme among young people who had received a risk reduction programme (FOK; Stanton et al., 2004).

Evidence statement 15

- 15 (a) There is mixed evidence from four RCTs¹ regarding the effects of intervention and programmes delivered to families and parents on attitudes and values related to risky behaviours.
- 15 (b) There is moderate evidence from two RCTs² to suggest that interventions and programme delivered to families and parents, and which target alcohol use and sexual health, may improve parent-child communication and family functioning. This evidence may only be partially applicable to the UK as these studies were conducted in the USA and focused on ethnic populations specific to the USA.
- 15 (c) There is moderate evidence from two RCTs² to suggest that interventions and programme delivered to parents and which target alcohol use and sexual health, may not provide long-term additional benefits in terms of health and social outcomes related to sexual health and alcohol use beyond those conferred through interventions and programmes which directly target young people. This evidence may only be partially applicable to the UK as these studies were conducted in the USA and focused on ethnic populations specific to the USA.
- ¹ Haggerty et al., 2007 (RCT +); Stanton et al., 2000 (RCT +); Wu et al., 2003 (RCT +); Stanton et al., 2004 (RCT ++)

² Prado et al., 2007 (RCT +); Stanton et al., 2004 (RCT +)

Table 7.4. Summary of programme components: programmes delivered to families and parents

Author	Study design and rating	Baseline population	Setting	Programme components	Theory	Provider
Haggerty et al., 2007	RCT (individual) +	USA n=331 families; 51% European American; 49% African American mean 13.8 years	Family	Parents Who Care: Participants received either a self- administered or parent and adolescent administered substance abuse and problem behaviours prevention intervention	Social development theory	Family consultants
Prado et al., 2007	RCT (individual) +	USA n=266 youth n=266 parents 100% Hispanic mean 13.4 years	Schools	Familias Unidas + PATH : HIV and sexual risk-reduction programme. Included two parent centred modules which also included adolescent participation in family visits and discussion circles with facilitators. The programme was conducted over 49 hours, in a 36 month period.	Eco- developmental theory	Trained facilitators
Stanton et al., 2000	RCT (individual) +	USA n=237 dyads 100% African American median 13.6 years	Family	Informed Parents and Children Together (ImPACT): Parents and children viewed a 22-minute educational video about AIDS, condoms and risky behaviours.	NR	Video
Wu et al., 2003; Stanton et al., 2004	RCT (cluster) ++	USA n=817 adolescents 100% African American 13-16 years	Community	Focus on Kids (FOK) + ImPACT : Risk reduction programme for alcohol, smoking, drugs and sexual behaviour. Included parent and school components delivered over eight 1.5 hour sessions and conducted four 90 minute booster sessions, role play and 20 minute video sessions.	Social cognitive model; Protection motivation theory	Other

						Outcomes	Outcomes		
Study	Rating	Intervention	Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills		
		Parents Who Care	No	PT n=313 (95%)	-	NS attitudes towards substance abuse; perceived harm of substance abuse	-		
Haggerty et RCT al., 2007 (individua	RCT	(SA n=107; PA n=118)	n=106	12 months n=306 (92%)	-	NS attitudes towards substance abuse; perceived harm of substance abuse	-		
	(individual) +	PWC: SA n=107	No intervention n=106	24 months n=304 (92%)	-	↓ favourable attitudes towards substance abuse* NS perceived harm of substance abuse	-		
		PWC: PA n=118	No intervention n=106	24 months n=304 (92%)	-	↓ favourable attitudes towards substance abuse* NS perceived harm of substance abuse	-		
Prado et al.,	RCT	RCT dividual) + Familias Unidas + PATH n=91	ESOL + PATH n=84	36 mo (80%)	-	-	 family functioning* positive parenting* parent-adolescent communication* 		
2007	(individual) +		ESOL + HEART n=91	36 mo (80%)	-	-	 f family functioning*** f positive parenting* NS parent-adolescent communication 		
		ImPACT II) + n=NR	Goal for IT! n=NR	2 months n=209 dyads (88%)	-	NS parent and youth agreement regarding having a boyfriend/girlfriend/drank alcohol/had sex	NS youth and parental reports of communication and monitoring		
Stanton et al., 2000	RCT (individual) +			6 months n=204 dyads (86%)	-	 f parent and youth agreement regarding having a boyfriend/girlfriend* NS parent and youth agreement regarding having drank alcohol/had sex 	 ↑ youth performing condom skills correctly*** ↑ parents performing condom skills correctly** NS youth and parental reports of communication and monitoring 		

Table 7.5. Programmes delivered to families and parents: effects on knowledge, attitudes and skills

					Outcomes		
Study	Rating	Intervention	Comparator	Follow-up	Knowledge and understanding	Attitudes and values	Personal and social skills
Wu et al., RCT 2003 (cluster) +		FOK + ImPACT (with or without boosters) n=496	FOK only n=321	6 mo n=608 (74%)	-	-	 NS risk taking intention † perceptions of parental monitoring** NS perceptions of parental communication
	RCT (cluster) +			12 mo n=580 (71%)	-	-	NS risk taking intention NS perceptions of parental monitoring NS perceptions of open communication † perceptions of problem communication
Stanton et RCT al., 2004 (cluster) +	FOK + ImPACT n=258	FOK only n=321	24 mo (60%)	-	-	NS talked with family member/other adult about AIDS/HIV NS asked recent sexual partner if condom always used	
	(cluster) +	FOK + ImPACT + boosters n=238	FOK only n=321	24 mo (60%)	-	-	 talked with family member/other adult about AIDS/HIV* asked recent sexual partner if condom always used**
Stanton et RCT al., 2004 (cluster	RCT	FOK + ImPACT + boosters n=238	FOK + ImPACT n=258	24 mo (60%)	-	-	↑ talked with family member/other adult about AIDS/HIV* NS asked recent sexual partner if condom always used
	(cluster) +	Both intervention groups n=496	FOK only n=321	24 mo (60%)	-	 t overall self-efficacy** t overall response efficacy* NS overall response cost 	NS talked with family member/other adult about AIDS/HIV 1 asked recent sexual partner if condom always used*
*p<0.05; **p<0	0.01; ***p<0.001	; ¹ p value not repo	rted; 1 increase r	elative to comparato	r; ↓ decrease relative to com	nparator; NS not significant; - ou	tcome not reported

Study Rati	Pating	Intervention	Comparator	Follow-up	Health and social outcomes			
Study	Rating	Intervention	Comparator	i onow-up	Sexual health	Alcohol use		
Haggerty et al., 2007	RCT (individual) +	Parents Who Care: SA n=107	No intervention n=106	24 mo n=304 (92%)	NS sexual initiation	NS initiation of alcohol use		
		Parents Who Care: PA n=118	No intervention n=106	24 mo n=304 (92%)	↓ sexual initiation (African American youth only*)	NS initiation of alcohol use		
Prado et al., R(2007 (individ	RCT	Familias Unidas	ESOL + PATH n=84	36 mo (80%)	NS unprotected sexual behaviour ↓ rates of incidence of STIs* ↓ unsafe sex at last sexual intercourse*	NS past 90-day alcohol use		
	(individual) +	n=91	ESOL + HEART n=91	36 mo (80%)	NS unprotected sexual behaviour ↓ rates of incidence of STIs* NS unsafe sex at last sexual intercourse	NS past 90-day alcohol use		
Wu et al.,	RCT	RCT Focus on Kids (cluster) + n=496	Received Focus on Kids	6 mo n=608 (74%)	↓ sexual intercourse* ↓ unprotected sex** NS sexual risk behaviour	↓ drank alcohol*		
2003	(cluster) +		only n=321	12 mo n=580 (71%)	NS sexual intercourse NS unprotected sex NS sexual risk behaviour	↓ drank alcohol**		
Stanton et al., 2004	RCT (cluster) +	FOK + ImPACT n=258	FOK only n=321	24 mo (60%)	NS sexual intercourse ↓ anal sex* ↓ been, or gotten a girl, pregnant* NS used birth control at last intercourse NS used condom at last intercourse	NS drank alcohol		
		FOK + ImPACT + boosters n=238	FOK only n=321	24 mo (60%)	NS sexual intercourse NS anal sex NS been, or gotten a girl, pregnant NS used birth control at last intercourse NS used condom at last intercourse	NS drank alcohol		

Table 7.6. Programmes delivered to families and parents: effects on health and social outcomes related to sexual health and alcohol use

Study	Rating	Intervention	Comparator	Follow-up	Health and social outcomes		
Sludy	Raing	intervention		Follow-up	Sexual health	Alcohol use	
Stanton et al.,	, RCT (cluster) +	FOK + ImPACT + boosters n=238	FOK + ImPACT n=258	24 mo (60%)	NS sexual intercourse NS anal sex ↑ been, or gotten a girl, pregnant* NS used birth control at last intercourse NS used condom at last intercourse	NS drank alcohol	
2004		Both intervention groups n=496	FOK only n=321	24 mo (60%)	NS sexual intercourse NS anal sex NS been, or gotten a girl, pregnant NS used birth control at last intercourse NS used condom at last intercourse	NS drank alcohol	
*p<0.05; **p<0.0	01; ***p<0.001; [•]	p value not reporte	d; 1 increase rela	tive to compara	ator; ↓ decrease relative to comparator; NS not	significant; - outcome not reported	

8 Discussion

8.1 Programmes targeting alcohol use

A total of 31 articles met the criteria for inclusion in the review of community-based programmes targeting alcohol use by young people. Four articles were systematic reviews and/or meta-analyses, three articles reported on studies that examined intervention or programmes delivered within social, healthcare and community settings, 20 articles reported on studies that examined programmes or interventions delivered to families or parents, and three studies examined interventions or programmes that involved the wider community or mass media. One economic evaluation study was also identified that examined the cost-effectiveness and cost-benefits of the Iowa Strengthening Families Programme (ISFP) and Preparing for the Drug Free Years (PDFY).

8.1.1 Systematic reviews and meta-analyses

Three systematic reviews and meta-analyses examined community-based interventions and programmes that targeted alcohol use among young people. One review (Foxcroft et al., 2002; 2003) examined interventions and programmes aimed at the primary prevention of alcohol use across a range of populations and settings. Two further reviews (Petrie et al., 2007; Smit et al., 2008) examined interventions and programmes delivered to parents and families, respectively. Foxcroft et al (2002; 2003) found that although there was no consistent evidence to determine which programmes were effective over the short to medium-term, one family-based programme, the ISFP, was effective over the longer term. The reviews by Petrie and colleagues (2007) and Smit and colleagues (2008) also highlighted the long-term effectiveness of the ISFP.

8.1.2 Programmes delivered in social, healthcare and community settings

Three studies were identified that examined interventions and programmes targeting alcohol use, which were delivered in social, healthcare and community settings. All three studies (Elder et al., 2002; Schinke et al., 2005; Tebes et al., 2007) were conducted within youth and after school agencies and were based in the USA.

None of the studies examined intervention effects on knowledge and understanding. Short-term increases in perception of harm were reported in two studies (Schinke et al., 2005; Tebes et al., 2007), but this effect was not sustained over the longer term. One study (Tebes et al., 2007) also found no impact of an after-school, youth development programme on participants' drug beliefs and there was no impact of a culturally tailored programme (Elder et al., 2002) on participants' susceptibility to alcohol. One study (Schinke et al., 2005) examined intervention effects on personal and social skills, finding a short-term intervention impact of an interactive CD-ROM intervention on assertion skills. Two studies (Elder et al., 2002; Schinke et al., 2005), conducted within youth agencies, found no intervention effects on health and social outcomes related to alcohol use. However, substance use remained low among both intervention and control participants throughout the study. One study (Tebes et al., 2007), which targeted older children (mean age 15 years) in after school programmes reported a positive short- to medium-term effect on alcohol use.

8.1.3 Programmes delivered to parents and families

A total of 20 studies were identified that examined programmes and interventions delivered to parents and families, which targeted adolescent alcohol or substance use. Evaluations of nine programmes delivered to families were reported on across fifteen studies (Bauman et al., 2000; Brody et al., 2004; 2006; Gerrard et al., 2006; Johnson et al., 1996; Jones et al., 2005; Loveland-Cherry et al., 1999; Mason et al., 2009; Murry et al., 2007; Schinke et al., 2004; 2009; Spoth et al., 1999; 2001; 2004; Stevens et al., 2002) and five studies (Beatty et al., 2008; Carlson et al., 2000; Cohen and Rice, 1995; Koutakis et al., 2008; Toomey et al., 1996) examined parent-targeted interventions.

Effects on knowledge and understanding were only examined in one study of a family-based programme and none of the parent-targeted interventions examined this outcome. Short-term intervention effects on attitudes and values related to alcohol use were found for two family-based programmes (Brody et al., 2004; 2009; Schinke et al., 2009) but for parent-targeted interventions there was no clear effect on parental attitudes to adolescent drinking (Cohen and Rice, 1995; Koutakis et al., 2008; Toomey et al., 1996). Both family-based and parent-targeted interventions appeared to produce short-term improvements in parent-child communication. Two CD-ROM based interventions (Schinke et al., 2004; 2009) showed positive programme effects on family communication skills and involvement skills and the culturally tailored SAAF (Brody et al., 2004) had a short-term positive effect on parental communication. Short-term intervention effects on parent-child communication were found for three studies (Beatty et al., 2008; Carlson et al., 2000; Toomey et al., 1996) of parent-targeted interventions; two studies (Beatty et al., 2008; Carlson et al., 2000) reported more frequent or recent parent-child communication about alcohol and one study (Toomey et al., 1996) showed positive long-term effects on parent-child communication regarding family rules about alcohol and alcohol related situations. Eleven studies examined intervention effects of family-based programmes on health and social outcomes related to alcohol use across eight programmes. Three programmes (Bauman et al., 2000; Jones et al., 2005; Loveland-Cherry, 1999) demonstrated nonsignificant effects on alcohol use, but across four programmes (Schinke et al., 2004; 2009; Spoth et al., 2001; Mason et al., 2009) short- and long-term positive effects on alcohol use were reported. In addition, six studies (Brody et al., 2004; 2006; Loveland-Cherry et al., 1999; Spoth et al., 1999; 2001; 2004) of four family-based programmes reported positive intervention effects on initiation of alcohol use in the medium- to long-term. The ISFP also had long-term positive effects on drunkenness and drinking without parental permission (Spoth et al., 2004), and long-term follow-up of the PDFY (Mason et al., 2009) revealed a positive effect of this programme on women's alcohol abuse in early adulthood. Two studies (Koutakis et al., 2008, Toomey et al., 1996) examined the effects of parenttargeted interventions. One study (Toomey et al., 1996) found no intervention effects but a second study (Koutakis et al., 2008) reported positive intervention effects on youth drinking, and past month drunkenness.

8.1.4 Programmes involving the wider community or mass media

Three studies (Cheadle et al., 1995; Flynn et al., 2006; Kypri et al., 2005) were identified that examined programmes involving the wider community or mass media. Two studies (Flynn et al., 2006;

Kypri et al., 2005) examined mass media intervention programmes delivered in communities in the USA and New Zealand, respectively, and one study (Cheadle et al., 1995) examined a 5-year community-based health promotion programme for adolescents on an American Indian Reservation. None of the included studies examined intervention effects on knowledge and understanding, or on personal and social skills. Only one study (Flynn et al., 2006) examined impacts on attitudes and values towards alcohol use, findings no effects of a long-term mass media programme on mediators of alcohol use. In addition, there were no effects of either mass media programme (Flynn et al., 2006; Kypri et al., 2005) or a community-wide campaign targeting American Indian adolescents on alcohol use.

8.1.5 Review of published economic evaluations

One study (Spoth et al., 2002) was identified that met the criteria for inclusion in the review of published economic evaluations. The study evaluated the cost-effectiveness and net benefits of two brief, family-focused interventions, the ISFP and PDFY, compared to a minimal intervention approach. Overall the net benefit was \$5,923 per family for the ISFP and \$2,697 per family for PDFY. The benefit-cost ratios were 9.60 and 5.85, indicating that for every \$1 spent on the ISFP and PDFY, \$9.60 and \$5.85, respectively, were saved in medical costs. The generalisability of the study to a UK context was unclear as the data used in the evaluation is based on studies conducted in the USA. In addition, projected alcohol use disorder rates were calculated based on US population data.

8.2 **Programmes targeting sexual health**

A total of 49 articles met the criteria for inclusion in the review of community-based programmes targeting young people's sexual health. Nine articles were systematic reviews and/or meta-analyses, 20 articles reported on studies that examined intervention or programmes delivered within social, healthcare and community settings, 15 articles reported on studies that examined programmes or interventions delivered to families or parents, two articles reported on studies that examined interventions or programmes that involved the wider community or mass media, and three articles reported on studies which examined interventions for vulnerable young people.

8.2.1 Systematic reviews and meta-analyses

Nine systematic reviews and meta-analyses were identified that examined the effectiveness of interventions and programmes across a range of settings and populations that targeted young people's sexual health behaviours. One review (Arnold and Rotheram-Borus, 2007) focused on interventions and programmes that targeted sexual risk taking among young homeless people. Findings from six reviews (Franklin et al., 1997; Pedlow and Carey, 2003; Robin et al., 2004; Sales et al., 2006; Underhill et al., 2007; 2008) indicated that community-based programmes can affect sexual risk behaviours of young people. In particular, HIV prevention and sexual risk reduction programmes were effective in increasing condom use and reducing pregnancy (Franklin et al., 1997; Robin et al., 2004; Sales et al., 2006). However, they may have a limited impact on adolescent sexual activity. According to Sales and colleagues (2006) successful community-based interventions were

theoretically based, tailored to the target population, implemented by trained facilitators, and the content was diverse and delivered using a wide variety of methods.

8.2.2 Programmes delivered in social, healthcare and community settings

A total of 20 studies were identified that examined interventions or programmes delivered within social, healthcare or community settings. Nine studies (Di Noia and Schinke, 2007; Jemmott et al., 1992; Jemmott et al., 1998; Kipke et al., 1993; Postrado and Nicholson, 1992; Sikkema et al., 2005; Stanton et al., 1996; 1997; Villarruel et al., 2006) examined group education sessions or skills-based training interventions delivered in community settings. Three studies (Ferguson, 2000; Pearlman et al., 2002; Smith et al., 2000) examined peer-led interventions, including a peer counselling programme (Ferguson, 2000) and peer leader leadership programmes (Pearlman et al., 2002; Smith et al., 2000), respectively. Philliber and colleagues (2002) examined the CAS-Carrera programme that focused on youth development for disadvantaged young people enrolled in after school programmes and Elliott et al (1996) examined a theatre production designed to inform young people about HIV. Six studies (Boekeloo et al., 1999; Danielson et al., 1990; DiClemente et al., 2004; Downs et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al., 2005) were conducted in healthcare settings including family planning clinics and primary care practices. Four studies (DiClemente et al., 2004; Downs et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al., 2005) examined group-based education and/or skillsbased interventions that specifically targeted sexually active young women. Two studies (Boekeloo et al., 1999; Danielson et al., 1990) examined interventions based around a health practitioner-led sexual health consultation.

Across four studies (Di Noia and Schinke, 2007; Jemmott et al., 1992; 1998; Kipke et al., 1993), that examined group education sessions or skills-based training interventions in community settings there were positive intervention effects on knowledge and understanding over the short- to medium-term. In addition, the three-year, CAS-Carrera programme (Philliber et al., 2002) had a positive impact on knowledge. There was no effect of a peer counselling intervention (Ferguson, 2000) on knowledge, but two peer leadership interventions (Pearlman et al., 2002; Smith et al., 2000) had positive effects on levels of knowledge among the peer leaders themselves. Four studies (DiClemente et al., 2004; Morrison-Beedy et al., 2005; Jemmott et al., 2005; Downs et al., 2004) of interventions that specifically targeted sexually active young females in healthcare setting, reported consistent short- to medium-term improvements in sexual health-related knowledge among intervention participants. In addition, two studies (Boekeloo et al., 1999; Danielson et al., 1990) of health practitioner-led sexual health consultations reported significant short-term increases in knowledge among intervention participants relative to controls. Short-term decreases in intentions to engage in risky sexual intercourse were reported in the study of the community-based, BPBR programme which targeted Black male adolescents (Jemmott et al., 1992) and an abstinence-based version of the programme resulted in short-term reductions in intentions to engage in any sexual intercourse. Across three studies group that examined group education sessions and skills-based training interventions in community settings (Di Noia and Schinke, 2007; Kipke et al., 1993; Stanton et al., 1996) there were short-term increases in intervention participants' perception of their vulnerability to HIV infection.

However, this effect was not sustained in the medium-term (Stanton et al., 1996). Two studies (Elliott et al., 1996; Smith et al., 2000) found no effects of a theatre production intervention or peer leadership intervention, respectively, on HIV attitudes at follow-up. There were indications of positive intervention effects of group education sessions and skills-based training interventions (Di Noia and Schinke, 2007; Jemmott et al., 1998; Smith et al., 2000; Stanton et al., 1996) in community settings on attitudes and values related to condom use. However, these effects did not appear to be consistent and were not maintained over the medium-term (Stanton et al., 1996). There were fairly consistent positive intervention effects on condom use attitudes across three studies (DiClemente et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al., 2005), which examined group-based education and skills-based interventions specifically targeting sexually active young women in healthcare settings, and one study (Boekeloo et al., 1996) that examined a primary care-based sexual risk assessment and education intervention. Two studies (Di Noia and Schinke, 2007; Jemmott et al., 1998) found short-term positive intervention effects of a CD-ROM mediated intervention and an abstinence-based version of the BPBR curriculum, respectively, on attitudes towards abstinence. A CD-ROM intervention (Di Noia and Schinke, 2007) and education and skills training programme (Kipke et al., 1993) had positive effects on behavioural skills but results from five studies (Di Noia and Schinke, 2007; Smith et al., 2000; Boekeloo et al., 1999; DiClemente et al., 2004; Morrison-Beedy et al., 2005) presented mixed findings in relation to effects on communication.

Across five studies (Jemmott et al., 1992; 1998; Postrado and Nicholson, 1992; Sikkema et al., 2005; Villarruel et al., 2006) that examined group-based sessions and/or skills training in community settings, short- to medium-term effects on sexual intercourse were reported in four studies (Jemmott et al., 1992; Postrado and Nicholson, 1992; Sikkema et al., 2005; Villarruel et al., 2006), and one study (Jemmott et al., 1998) reported no programme effects. The CAS-Carrera programme (Philliber et al., 2002) had a positive effect on sexual activity among females, but there were no effects of health practitioner-led sexual health consultations (Boekeloo et al., 1999; Danielson et al., 1990) or peer interventions (Ferguson, 2000; Smith et al., 2000). Intervention effects on frequency of sexual intercourse and number of sexual partners were limited. Across four studies (Jemmott et al., 1998; Kipke et al., 1993; Pearlman et al., 2002; Villarruel et al., 2006) conducted in community settings, only one study (Villarruel et al., 2006) reported a positive intervention effect and across four studies (DiClemente et al., 2004; Downs et al., 2004; Jemmott et al., 2005; Morrison-Beedy et al 2005) conducted in healthcare settings, there were inconsistent intervention effects on these outcomes. Intervention effects on condom use and unprotected intercourse were more consistent. Across six studies that examined group-based sessions and skills training in community and healthcare settings, there were positive short- to medium-term intervention effects on measures of condom use (DiClemente et al., 2004; Jemmott et al., 1998; 2005; Sikkema et al., 2005; Stanton et al., 1996; Villarruel et al., 2006), and some evidence from three studies (Jemmott et al., 1998; 2005; Villarruel et al., 2006) of a positive intervention effect on frequency of unprotected intercourse. There were no effects of an HIV theatre production (Elliott et al., 1996) or peer counselling intervention (Ferguson, 2000) on contraceptive use or frequency of unprotected sex, but the CAS-Carrera programme (Philliber et al., 2002) positively influenced both condom and hormonal contraceptive use among

females. There was no effect of a peer counselling intervention (Ferguson, 2000) or peer leadership programme (Smith et al., 2000) on pregnancy rates, but the CAS-Carrrea programme (Philliber et al., 2002) had a positive effect, with a reduction in pregnancies among intervention females. Three studies (Boekeloo et al., 1999; Downs et al., 2004; Jemmott et al., 2005) examined intervention effects on STI infection and/or diagnosis, finding mixed intervention effects. However, medium-term positive effects on STI diagnosis were reported in one study (Jemmott et al., 2005) of a skills-based HIV/STI intervention delivered in a healthcare setting.

8.2.3 Programmes delivered to parents and families

Fifteen studies were identified that examined intervention and programmes delivered to parents and families, which targeted adolescent sexual health. Ten studies (Anderson et al., 1999; Dilorio et al., 2006; Lederman et al., 2004; 2008; McBride et al., 2007; McKay et al., 2004; Miller et al., 1993; Scheinberg et al., 1997; Winett et al., 1992; 1993) evaluated seven programmes delivered to adolescents and their families and five studies (Dancy et al., 2006; Dilorio et al., 2007; Forehand et al., 2007; Gustafson, 1998; O'Donnell et al., 2005) examined parent-targeted interventions.

Both family-based and parent-targeted interventions demonstrated positive influences on knowledge related to sexual health in the short- (Dancy et al., 2006; Scheinberg et al., 1997; Winett 1992; Winett et al., 1993), medium- (Miller et al., 1993) and long-term (Dilorio et al., 2006; Lederman et al., 2008), with improvements seen in both parent and adolescent knowledge (Dilorio et al., 2006; Miller et al., 1993) related to sexual health. Programmes and interventions delivered to families (Anderson et al., 1999; Dilorio et al., 2006; Lederman et al., 2004; 2008; Miller et al., 1993; Scheinberg et al., 1997) did not appear to be effective at influencing adolescent's attitudes and intentions towards resisting or delaying sex and across three studies (Dancy et al., 2006; Dilorio et al., 2007; Gustafson, 1998) that examined effects of parent-targeted interventions on intentions there were inconsistent results. There were mixed effects on parent-child communication across both family-based and parent-targeted interventions. Eight studies (Anderson et al., 1999; Dilorio et al., 2004; McBride et al., 2007; Miller et al., 1993; Scheinberg et al., 1997), which examined family-programmes found no clear intervention effects on communication, but in general positive effects were found across four studies (Dilorio et al., 2007; Forehand et al., 2007; O'Donnell et al., 2005; Gustafson, 1998) that examined parent-targeted interventions.

Across five studies (Anderson et al., 1999; Miller et al., 1993; Dilorio et al., 2006; McKay et al., 2004; McBride et al., 2007) that examined the effects of family-based programmes on health and social outcomes related to sexual health the results suggested that programmes and interventions delivered to families may not affect sexual behaviour. Two studies (Anderson et al., 1999; Miller et al., 1993) found no intervention effects on pregnancy rates or sexual behaviour, respectively, and one study (Dilorio et al., 2006) found no long-term effects of an intervention aimed at mothers and their adolescent children on abstinence or involvement in intimate sexual behaviours. There were, however, limited but positive effects of this programme on condom use. There were positive short-term effects of two parent education programmes (Dancy et al., 2006; O'Donnell et al., 2005) on initiation of sexual

activity and behavioural risks related to early sexual initiation, respectively. However, lack of clear intervention effects were reported in two further studies (Dilorio et al., 2007; Forehand et al., 2007).

8.2.4 Programmes involving the wider community or mass media

Two studies were identified (Doniger et al., 2001; Sieverding et al., 2005) that examined interventions that involved the wider community or mass media. Doniger and colleagues (2001) examined a mass media intervention, Not Me, Not Now, and Sieverding and colleagues (2005) examined the Youth United through Health Education (YUTHE) outreach programme. Neither of the included studies examined intervention effects on knowledge, attitudes and skills. Both studies analysed population-level changes, in pregnancy (Doniger et al., 2001) and STI rates (Sieverding et al., 2005), respectively, as measures of effectiveness. Both studies reported positive intervention effects at a population level, however the study by Doniger and colleagues (2001) did not adequately control for natural fluctuations in the data and therefore it is not clear whether these or intervention effects were responsible for the differences seen in the intervention and control communities.

8.2.5 Programmes targeting vulnerable populations

Three studies (Gleghorn et al., 1997, Rew et al., 2007, Slesnick and Kang, 2008) were identified that examined the effectiveness of community-based programmes on vulnerable populations. All three studies examined interventions which specifically targeted young homeless people. Intervention approaches examined were street outreach (Gleghorn et al., 1997), a brief group-based sexual health intervention (Rew et al., 2007), and a community reinforcement approach combined with HIV prevention (Slesnick and Kang, 2008). Intervention effects on knowledge and skills were examined in one study (Rew et al., 2007) and none of the included studies examined intervention effects on attitudes and values. There were limited effects of a brief sexual health intervention (Rew et al., 2007) on knowledge relating to AIDS and other STIs, and on communication and self-efficacy. Health and social outcomes related to sexual health were examined in all three studies, two of which reported no intervention effects (Gleghorn et al., 1997, Rew et al., 2007). Slesnick and Kang (2008) found a positive effect on the frequency of condom use among younger participants in a programme which combined a community reinforcement approach with HIV prevention content.

8.3 **Programmes targeting multiple behaviours**

No systematic reviews or meta-analyses were identified for inclusion in the review of programmes targeting multiple health behaviours. Five articles were identified that reported on evaluations of programmes and interventions that addressed both alcohol use and sexual health. Two articles reported on studies that examined interventions or programmes delivered in social, healthcare or community settings and three articles reported on studies that examined intervention studies that examined intervented on studies that examined interventions or programmes delivered in social, healthcare or community settings and three articles reported on studies that examined interventions or programmes delivered to families or parents.

8.3.1 Programmes delivered within social, healthcare and community settings

Two studies (St Pierre et al., 1995; Wiggins et al., 2009) examined programmes which targeted both sexual health and alcohol use. St Pierre and colleagues (1995) examined the effects of Stay SMART,

which targeted young people enrolled in Boys and Girls Clubs, with and without the addition of a peer leadership component, and Wiggins and colleagues (2009) evaluated the effectiveness of the Young People's Youth Development (YPYD) programme in reducing teenage pregnancy, substance use and other outcomes.

Neither of the included studies examined intervention effects on knowledge and understanding, or on personal and social skills. However, both studies examined intervention effects on attitudes and values. St Pierre and colleagues (1995) found a favourable reduction in sexual attitudes but only among sexually experienced participants who received the intervention without the additional booster sessions. The YPYD programme (Wiggins et al., 2009) had potentially harmful effects on attitudes, with female intervention participants more likely than control participants to report that they expected to be a parent by age 20. Both studies examined intervention effects on health and social outcomes related to sexual health, and Wiggins and colleagues (2009) also examined intervention effects on alcohol use. The effects of the Stay SMART intervention were inconsistent across the two intervention conditions examined. The YPYD programme (Wiggins et al., 2009) had a negative impact on participant's sexual behaviour, particularly among intervention females who were significantly more likely than controls to engage in heterosexual sexual intercourse and more likely to become pregnant. There was no effect of the programme on male participants or on participant's alcohol use.

8.3.2 Programmes delivered to families or parents

Five studies (Haggerty et al., 2007; Prado et al., 2007; Stanton et al., 2000; 2004; Wu et al., 2003) examined three programmes delivered to families or parents, which targeted both alcohol use and sexual health, in addition to other risk behaviours. None of the included studies examined intervention effects on knowledge and understanding. Across four studies (Haggerty et al., 2007; Stanton et al., 2000; Wu et al., 2003; Stanton et al., 2004) that examined intervention effects on attitudes and values towards risky behaviours there were indications of mixed intervention effects. Haggerty and colleagues (2007) found positive long-term effects of both self-directed and group-based versions of a universal substance use and problem behaviour prevention programme on attitudes towards substance use and there were also long-term positive programme effects of a parental monitoring intervention (Stanton et al., 2004) on attitudes and values related to a range of risky behaviours. Three studies (Prado et al., 2007; Stanton et al., 2000; 2004) examined intervention effects on personal and social skills, finding mixed programme effects on parent/family-child communication. Prado and colleagues (2007) found positive effects of a culturally-tailored programme on communication, family functioning and positive parenting, and Stanton and colleagues (2004) found a positive effect of a parental monitoring programme on parent-child communication about HIV/AIDS. Four studies (Haggerty et al., 2007; Prado et al., 2007; Wu et al., 2003; Stanton et al., 2004) examined intervention effects on health outcomes related to alcohol use and sexual health. Shortterm to medium-term reductions in alcohol drinking were found for participants who received a parental monitoring intervention (Wu et al., 2003; Stanton et al., 2004), but this reduction was not sustained and no other significant programme effects were found for health outcomes related to alcohol use. One study (Prado et al., 2007) of a culturally-tailored programme reported a long-term

decrease in incidence rates for STIs and unsafe sex at last sexual intercourse among those who received an additional parent-targeted component. Although, short-term benefits of a parental monitoring intervention (Wu et al., 2003) were also reported, these differences were not sustained and over the longer term there were no additional positive effects on sexual behaviour of the intervention among young people who had received a risk reduction intervention (Stanton et al., 2004).

8.4 Strengths and limitations

This review of the effectiveness and cost-effectiveness of community-based interventions and programmes that address health literacy and personal skills in relation to alcohol use and sexual health was based on a comprehensive and systematic literature review. Over 12,000 titles and abstracts were screened for inclusion in the review, and over 400 full text articles were reviewed. In addition, the review has been conducted using a standardised and transparent approach, adhering to NICE protocols for the development of public health programme guidance.

8.4.1 Quality of the included studies

The studies identified for inclusion in the review were based on a range of study designs. However, the vast majority were based on an RCT design, of which half were cluster RCTs. The quality of the included studies was generally moderate or good, with approximately 30% of studies receiving a poor rating for quality. In general studies did not describe the source population or source area from which study participants were drawn, and it was therefore, frequently not possible to determine the eligibility of the selected populations or areas included. Methods of randomisation were well described in approximately one fifth of RCTs, which generally also reported that allocation was adequately concealed and that participants and/or investigators were blinded. Across the remaining RCTs, the authors reported that randomisation had been undertaken but did not describe the actual method of randomisation or how allocation was concealed. For studies based on non-random assignment, authors rarely reported how confounding and bias were minimised or how individuals or clusters were allocated to intervention or comparison groups. Few authors examined or commented on contamination and it was therefore difficult to judge whether contamination was acceptably low across the included studies. Attrition rates varied across the included studies, but over half of studies accounted for all participants at follow-up. Outcome measures were reported to be reliable across the majority of the included studies, and were deemed to be relevant. Follow-up times varied across the included studies, from immediate post-test to ten years, but two thirds of studies reported what was judged to be a meaningful follow-up time. Intervention and comparison groups were similar at baseline, or adjustments were made for differences, across the vast majority of studies. Few studies reported undertaking an intention to treat analysis (~25%) or reported whether studies were adequately powered (~20%). Analytical methods appeared to be appropriate in the majority of studies, but estimates of effect sizes were not reported or calculable in approximately one third of the included studies.

8.4.2 Applicability and transferability

As highlighted in previous reviews conducted by the lead author and colleagues, there is a lack of prevention initiatives originating from the UK which have been subject to evaluation and peerreviewed publication. The research literature identified for this review was, as in previous reviews, dominated by programmes conducted in the USA, which focused on minority ethnic populations specific to the USA. Both African American and Hispanic adolescents have been identified as populations at high risk of HIV infection (Jemmott et al., 2005), and Hispanic adolescents report higher levels of substance use and unprotected sexual intercourse than non-Hispanic White and African American adolescents (Prado et al., 2007). However in the UK, although black and minority ethnic (BME) groups may be at higher risk of STI infection (DH, 2001) survey data indicates that drug use is generally lower among BME populations (Edmonds et al., 2005). In this review we identified the YPYD programme (Wiggins et al., 2009) which was implemented in England and whose development was informed by the US-based, Carrera programme. Despite promising effects of the Carrera programme, Wiggins and colleagues (2009) found an adverse effect of the intervention among an English population.

8.5 Research recommendations

The review has identified a number of gaps in the evidence and future research should aim to address the following key research recommendations:

- There needs to be further evaluation of the effectiveness and cost-effectiveness of alcohol education and SRE approaches delivered in community settings, which are currently being delivered or planned in the UK;
- Full economic evaluation studies are required of community-based approaches focusing on both SRE and alcohol education that consider both the costs and consequences of implementing these types of interventions and programmes.
- Future research should consider the relationship between alcohol use and sexual health.

Improvements in study design and the quality of reporting are required with respect to all types of studies and the following are recommendations to improve the methodology of future studies:

- Improved reporting of methods is required, particularly with regard to methods for the allocation of participants and clusters (e.g. methods of randomisation), allocation concealment, procedures for blinding, and follow-up of participants. Reporting standards could be improved by following guidelines on reporting, such as the CONSORT statement for RCTs and TREND statement for non-randomised studies.
- Standardisation of outcomes is required. Across the included studies a range of attitudinal and behavioural measures were reported and consequently it was not possible to synthesise outcomes across studies. Also when considering which outcomes to incorporate, there needs to be a consideration of the age and level maturity of the sample targeted (e.g. with regard to

studies of SRE programmes consideration should be made of the relationship status of participants).

- Some studies were conducted with inadequate sample sizes, and future research studies should be sufficiently powered to detect intervention effects.
- Future research studies should incorporate an adequate length of follow-up.

9 Conclusions

9.1 Programmes targeting alcohol use

There was a lack of evidence on which to draw conclusions about the effects of programmes and interventions that targeted adolescent alcohol use on knowledge and understanding. There were positive effects of programmes and interventions delivered to families on attitudes and values related to alcohol use, but programmes and interventions delivered to parents or within social, healthcare and community settings appeared to have no impact on these outcomes. Programmes and interventions delivered to families had positive effects on both alcohol use and interventions delivered to families had positive effects on both alcohol use and interventions. Programmes and interventions delivered to families had positive effects on both alcohol use were mixed and inconsistent across programmes and interventions delivered to parents, in social, healthcare or community settings, or to the wider community. The family-focused ISFP was highlighted across three systematic reviews as showing particular promise; this programme, which was designed to enhance family protective and resiliency processes and to reduce family-based risk factors associated with child behaviour problems, had positive, long-term effects on a range of outcomes related to alcohol use and has been shown to be cost-effective and potentially cost saving.

9.2 **Programmes targeting sexual health**

The evidence suggests that programmes and interventions delivered in social, healthcare and community settings and to families and parents may have beneficial effects on sexual health-related knowledge in the short- to long-term. A range of outcomes were reported with regards to attitudes and values and programmes effects were mixed across these measures. The evidence suggests that while programmes and interventions targeting adolescent sexual health may not impact on attitudes towards sexual intercourse, programmes and interventions delivered in healthcare settings may positively impact on condom use attitudes. Programmes and interventions delivered to families and in social, healthcare and community settings had mixed and inconsistent effects on communication, but programmes and interventions delivered to parents appeared to have positive effects on parent-child communication. There appeared to be no effects of programmes and interventions delivered to families and parents on adolescent sexual behaviour, and programmes and interventions delivered in social, healthcare and community settings had limited and inconsistent effects on sexual activity including frequency of intercourse and number of sexual partners. However, the evidence suggests that group-based sessions and/or skills training programmes in community and healthcare settings may increase condom use and reduce the frequency of unprotected sex. In addition, a youth development approach showed promise, with effects on a range of sexual health outcomes for females. There was a lack of evidence on which to draw conclusions about the effects on programmes involving the wider community or mass media or those targeting vulnerable populations.

9.3 Programmes targeting multiple behaviours

There was a lack of evidence on which to draw conclusions about the effects of programmes and interventions that targeted multiple behaviours on knowledge and understanding, and there was evidence of mixed and inconsistent effects of these programmes on attitudes and values. Programmes and interventions delivered to parents and families had long-term positive effects on communication, but intervention effects on health and social outcomes related to sexual health were less clear. There was no evidence supporting the effectiveness of programmes and interventions delivered in social, healthcare and community settings and interventions and programmes delivered to parents did not appear to provide additional long-term benefits beyond those conferred through interventions and programmes which directly target young people.

9.4 Summary

The results of this systematic review suggest that programmes and interventions delivered to families may be effective in reducing adolescent alcohol consumption and that group-based sessions and/or skills training programmes in community and healthcare settings may be effective in increasing condom use and reducing the frequency of unprotected intercourse among adolescents. In addition, programmes and interventions delivered to families and parents appeared to be effective in increasing parent-child communication about alcohol use and sexual health. However, the applicability of the evidence identified may not be generalisable to the UK and good quality UK-based research of promising or novel intervention approaches, including assessment of cost-effectiveness, is required in order to build the evidence base on which to make UK-based policy and practice recommendations.

10 References

Brown JD, Halpern CT, L'Engle KL (2005) Mass media as a sexual super peer for early maturing girls. Journal of Adolescent Health 36:420-427.

DCSF (2008) Review of sex and relationship education (SRE) in schools. London, DCSF.

DfES (2004) Drugs, guidance for schools. London, DfES.

DfES (2004) Healthy living blueprint for schools. London, DfES.

DfES (2006) Extended schools and health services – working together for better outcomes for children and families. London, DfES.

DfES (2006) Teenage pregnancy: Accelerating the strategy to 2010. London, DfES.

DH (2001) Better prevention, better services, better sexual health. The national strategy for sexual health and HIV. London, DH.

Duncan SC, Duncan TE, Strycker LA (2006) Alcohol use from ages 9 to 16: a cohort-sequential latent growth model. Drugs and Alcohol Dependence.81:71-81.

L'Engle KL, Jackson C (2008) Socialization influences on early adolescents' cognitive susceptibility and transition to sexual intercourse. Journal of Research on Adolescence 18(2):353-378.

Lindsay G, Davies H, Band S, et al (2008) Parenting Early Intervention Pathfinder Evaluation. London, DCSF.

MedFASH (2008) Progress and priorities – working together for high quality sexual health. Review of the National Strategy for Sexual Health and HIV. London, MedFASH.

NICE (2007) Behaviour change at population, community and individual levels. London, NICE.

Ramm J, Coleman L (2008) Evaluation of the effects of the Birmingham Speakeasy course. East Sussex, Trust for the Study of Adolescence.

Sieving RE, Eisenberg ME, Pettingell S, Skay C (2006) Friends' influence on adolescents' first sexual intercourse. Perspectives on Sexual and Reproductive Health 38(1):13-19.

The Strategy Unit (2004) Alcohol Harm Reduction Strategy for England. London, the Strategy Unit.

van der Vorst H, Engles RCME, Meeus W, Dekovic M (2006) The impact of alcohol-specific rules, parental norms about early drinking and parental alcohol use on adolescents' drinking behavior. Journal of Child Psychology and Psychiatry 47(12):1299-1306.

van der Vorst H, Engles RCME, Meeus W, et al (2005) The role of alcohol-specific socialization in adolescents' drinking behaviour. Addiction 100:1464-1476.

van der Vorst H, Vermulst AA, Meeus W, et al (2009) Identification and predication of drinking trajectories in early and mid-adolescence. Journal of Clinical Child & Adolescent Psychology, 38(3):329-341.

White HR, Johnson V, Buyske S (2000) Parental modelling and parenting behaviour effects on offspring alcohol and cigarette use, a growth curve analysis. Journal of Substance Abuse 12:287-310.

Appendix 1. References to included studies

Anderson NLR, Koniak-Griffin D, Keenan CK, et al (1999) Evaluating the outcomes of parent-child family life education.... Scholarly Inquiry for Nursing Practice 13(3):211-239.

Arnold EM, Rotheram-Borus MJ (2009) Comparisons of prevention programs for homeless youth. Prevention Science 10(1):76-86.

Bauman KE, Ennett ST, Foshee VA, et al (2000) Influence of a family-directed program on adolescent cigarette and alcohol cessation. Prevention Science 1:227-237.

Beatty SE, Cross DS, Shaw TM (2008) The impact of a parent-directed intervention on parent-child communication about tobacco and alcohol. Drug and Alcohol Review 27(6):591-601.

Boekeloo BO, Schamus LA, Simmens SJ, et al (1999) A STD/HIV prevention trial among adolescents in managed care. Pediatrics 103:107-115.

Brody GH, Murry VM, Gerrard M, et al (2004) The Strong African American Families Program: Translating Research into Prevention Programming. Child Development 75(3):917.

Brody GH, Murry VM, Kogan SM, et al (2006) The Strong African American Families Program: A Cluster-Randomized Prevention Trial of Long-Term Effects and a Mediational Model. Journal of Consulting and Clinical Psychology 74(2):356-366.

Carlson JM, Moore MJ, Pappas DM, et al (2000) A pilot intervention to increase parent-child communication about alcohol avoidance. Journal of Alcohol & Drug Education 45(2):59-71.

Cheadle A, Pearson D, Wagner E, et al (1995) A community-based approach to preventing alcohol use among adolescents on an American Indian reservation. Public Health Reports 110(4):439-447.

Cohen DA, Rice JC (1995) A Parent-targeted Intervention for Adolescent Substance Use Prevention: Lessons Learned. Evaluation Review 19(2):159-180.

Dancy BL, Crittenden KS, Talashek M (2006) Mother's effectiveness as HIV risk reduction educators for adolescent daughters. Journal of Health Care for the Poor & Underserved 17(1):218-240.

Danielson R, Marcy S, Plunkett A, et al (1990) Reproductive health counseling for young men: what does it do? Family Planning Perspectives 22(3):115-121.

Di Noia J, Schinke SP (2007) Gender-specific HIV prevention with urban early-adolescent girls: outcomes of the Keepin' It Safe Program. AIDS Education & Prevention 19(6):479-488.

DiCenso A, Guyatt G, Willan A, Griffith L (2002) Interventions to reduce unintended pregnancies among adolescents: systematic review of randomised controlled trials. BMJ 324(7351):1426.

DiClemente RJ, Wingood GM, Harrington KF, et al (2004) Efficacy of an HIV prevention intervention for African American adolescent girls: a randomised controlled trial. JAMA 292(2):171-179.

Dilorio C, Resnicow K, McCarty F, et al (2006) Keepin' It R.E.A.L.! Results of a Mother-Adolescent HIV Prevention Program. Nursing Research 55(1):43-51.

Dilorio C, McCarty F, Resnicow K, et al (2007) REAL Men: a group randomised trial of an HIV prevention intervention for adolescent boys. American Journal of Public Health 97:1084-1089.

Doniger AS, Adams E, Utter CA, Riley JS (2001) Impact evaluation of the "Not me, Not now" abstinence-oriented, adolescent pregnancy prevention communications program, Monroe County, New York. Journal of Health Communication 6(1):45-60.

Downs JS, Murray PJ, Bruine de Bruin W, et al (2004) Interactive video behavioral intervention to reduce adolescent females' STD risk: a randomized controlled trial. Social Science & Medicine 59:1561-1572.

Elder JP (2002) Tobacco and alcohol use-prevention program for Hispanic migrant adolescents. American Journal of Preventive Medicine 23(4):269-275.

Elliott L, Gruer L, Farrow K, et al (1996) Theatre in AIDS education -- a controlled study. AIDS Care 8(3):321-341.

Ferguson SL (2000) Evaluation of the effects of peer counseling in a culturally-specific adolescent pregnancy prevention program for African American females. Unpublished dissertation. University of Virginia, Charlottesville, VA.

Flynn BS, Worden JK, Bunn JY, et al (2006) Mass media and community interventions to reduce alcohol use by early adolescents. Journal of Studies on Alcohol 67(1):66-74.

Forehand R, Armistead L, Long N, et al (2007) Efficacy of a parent-based sexual-risk prevention program for African American preadolescents: a randomized controlled trial. Archives of Pediatrics & Adolescent Medicine 161(12):1123-1130.

Foxcroft D, Ireland D, Lowe G, Breen R (2002) Primary prevention for alcohol misuse in young people. Cochrane Database of Systematic Reviews, Issue 3.

Foxcroft DR, Ireland D, Lister-Sharp DJ, Lowe G, Breen R (2003) Longer-term primary prevention for alcohol misuse in young people: a systematic review. Addiction 98(4):397-412.

Franklin C, Grant D, Corcoran J, et al (1997) Effectiveness of prevention programs for adolescent pregnancy: a meta-analysis. Journal of Marriage and the Family 59:551-567.

Gerrard M, Gibbons FX, Brody GH, et al (2006) A theory-based dual-focus alcohol intervention for preadolescents: the Strong African American Families Program. Psychology of Addictive Behaviors 20(2):185-195.

Gleghorn AA, Clements KD, Marx R, et al (1997) The impact of intensive outreach on HIV prevention activities of homeless, runaway, and street youth in San Francisco: the AIDS Evaluation of Street Outreach Project (AESOP). AIDS & Behavior 1(4):261-272.

Gustafson CZ (1998) The effects of a health promotion program on parental attitudes and behaviors and adolescent sexual well-being. Unpublished dissertation. University of Michigan, Dearborn, MI.

Guyatt GH, DiCenso A, Farewell V, et al (2000) Randomized trials versus observational studies in adolescent pregnancy prevention. Journal of Clinical Epidemiology 53(2):167-174.

Haggerty KP, Skinner ML, MacKenzie EP, Catalano RF (2007) A randomized trial of Parents Who Care: effects on key outcomes at 24-month follow-up. Prevention Science 8(4):249-260.

Jemmott JB, Jemmott LS, Fong GT (1992) Reductions in HIV Risk-Associated Sexual Behaviors among Black Male Adolescents: Effects of an AIDS Prevention Intervention. American Journal of Public Health 82:372-377.

Jemmott J, Jemmott L (1998) Abstinence and safer sex: HIV risk-reduction interventions for African American adolescents. Journal of the American Medical Association 279:1529-1536.

Jemmott JB, Jemmott LS, Braverman PK, Fong GT (2005) HIV/STD risk reduction interventions for African American and Latino adolescent girls at an adolescent medicine clinic. A randomized controlled trial. Archives of Pediatrics & Adolescent Medicine 159:440-449.

Johnson K, Bryant D, Strader T, Bucholtz G, et al. (1996) Reducing alcohol and other drug use by strengthening community, family, and youth resiliency: An evaluation of the Creating Lasting Connections Program. Journal of Adolescent Research 11(1):36-67.

Jones DJ, Olson AL, Forehand R, et al (2005) Original research: A family-focused randomized controlled trial to prevent adolescent alcohol and tobacco use: The moderating roles of positive parenting and adolescent gender. Behavior Therapy 36(4):347-355.

Kipke MD, Boyer C, Hein K (1993) An evaluation of an AIDS risk reduction education and skills training (ARREST) program. Journal of Adolescent Health 14(7):533-539.

Koutakis N, Stattin H, Kerr M (2008) Reducing youth alcohol drinking through a parent-targeted intervention: The Orebro Prevention Program. Addiction 103(10):1629-1637.

Kypri K, Dean J, Kirby S, et al (2005) 'Think before you buy under-18s drink': evaluation of a community alcohol intervention. Drug & Alcohol Review 24(1):13-20.

Lederman RP, Chan W, Roberts-Gray C (2004) Sexual risk attitudes and intentions of youth aged 12-14 years: survey comparisons of parent-teen prevention and control groups. Behavioral Medicine 29(4):155-163. Lederman RP, Chan W, Roberts-Gray C (2008) Parent-adolescent relationship education (PARE): program delivery to reduce risks for adolescent pregnancy and STDs. Behavioral Medicine 33(4):137-145.

Loveland-Cherry CJ, Ross LT, Kaufman SR (1999) Effects of a home-based family intervention on adolescent alcohol use and misuse. Journal of Studies on Alcohol - Supplement (S13):94-102.

Mason WA, Kosterman R, Haggerty KP, et al (2009) Gender moderation and social developmental mediation of the effect of a family-focused substance use preventive intervention on young adult alcohol abuse. Addictive Behaviors 34(6-7):599-605.

McBride CK, Baptiste D, Traube D, et al (2007) Family-based HIV preventive intervention: child level results from the CHAMP family program. Social Work in Mental Health 5(1-2):203-221.

McKay MM, Chasse KT, Paikoff R, et al (2004) Family-level impact of the CHAMP Family Program: a community collaborative effort to support urban families and reduce youth HIV risk exposure. Family Process 43(1):79-94.

Miller BC, Norton MC, Jenson GO, et al (1993) Impact evaluation of Facts & Feelings: a home-based video sex-education curriculum. Family Relations 42:392-400.

Morrison-Beedy D, Carey MP, Kowalski J, Tu X (2005) Group-based HIV risk reduction intervention for adolescent girls: evidence of feasibility and efficacy. Research in Nursing & Health 28(1):3-15.

Murry VM, Berkel C, Brody GH, et al (2007) The Strong African American Families Program: longitudinal pathways to sexual risk reduction. Journal of Adolescent Health 41(4):333-342.

O'Donnell L, Stueve A, Agronick G, et al (2005) Saving sex for later: an evaluation of a parent education intervention. Perspectives on Sexual & Reproductive Health 37(4):166-174.

Pearlman DN, Camberg L, Wallace LJ, et al (2002) Tapping youth as agents for change: evaluation of a peer leadership HIV/AIDS intervention. Journal of Adolescent Health 31(1):31-39.

Pedlow CT, Carey MP (2003) HIV sexual risk-reduction interventions for youth: A review and methodological critique of randomized controlled trials. Behavior Modification 27(2):135-190.

Petrie J, Bunn F, Byrne G (2007) Parenting programmes for preventing tobacco, alcohol or drugs misuse in children <18: a systematic review. Health Education Research 22(2):177-191.

Philliber S, Kaye JW, Herrling S, West E (2002) Preventing pregnancy and improving health care access among teenagers: an evaluation of the Children's Aid Society-Carrera Program. Perspectives on Sexual & Reproductive Health 34(5):244-252.

Postrado LT, Nicholson HJ (1992) Effectiveness in delaying the initiation of sexual intercourse of girls aged 12-14. Two components of the Girls Incorporated Preventing Adolescent Pregnancy Program. Youth & Society 23(3):356-379.
Prado G, Pantin H, Briones E, et al (2007) A randomized controlled trial of a parent-centered intervention in preventing substance use and HIV risk behaviors in Hispanic adolescents. Journal of Consulting and Clinical Psychology 75(6):914-926.

Rew L, Fouladi RT, Land L, Wong YJ (2007) Outcomes of a Brief Sexual Health Intervention for Homeless Youth. Journal of Health Psychology 12:818-832.

Robin L, Dittus P, Whitaker D, et al (2004) Behavioral interventions to reduce incidence of HIV, STD, and pregnancy among adolescents: a decade in review. Journal of Adolescent Health 34(1):3-27.

Sales JM, Milhausen RR, DiClemente RJ (2006) A decade in review: building on the experiences of past adolescent STI/HIV interventions to optimise future prevention efforts. Sexually Transmitted Infections 82(6):431-436.

Scheinberg CA (1997) The impact of parent participation in sex education on adolescent sexual knowledge, attitudes, locus of control, and behavior. Unpublished doctoral dissertation. Pacifica Graduate Institute, Carpinteria CA.

Schinke SP (2005) Alcohol abuse prevention among high-risk youth: computer-based intervention. Journal of Prevention & Intervention in the Community 29(1-2):117-130.

Schinke SP, Cole KC, Fang L (2009) Gender-specific intervention to reduce underage drinking among early adolescent girls: a test of a computer-mediated, mother-daughter program. Journal of Studies on Alcohol and Drugs 70:70-77.

Schinke SP, Schwinn TM, Di Noia J, Cole KC (2004) Reducing the risks of alcohol use among urban youth: three-year effects of a computer-based intervention with and without parent involvement. Journal of Studies on Alcohol 65(4):443-449.

Sieverding J, Boyer CB, Siller J, et al (2005) Youth United Through Health Education: building capacity through a community collaborative intervention to prevent HIV/STD in adolescents residing in a high STD prevalent neighborhood. AIDS Education & Prevention 17(4):375-386.

Sikkema KJ, Anderson ES, Kelly JA, et al (2005) Outcomes of a randomized, controlled communitylevel HIV prevention intervention for adolescents in low-income housing developments. AIDS 19(14):1509-1516.

Slesnick N, Kang MJ (2008) The impact of an integrated treatment on HIV risk behavior among homeless youth: a randomized controlled trial. Journal of Behavioral Medicine 31(1):45-59.

Smit E, Verdurmen J, Monshouwer K, Smit F (2008) Review: Family interventions and their effect on adolescent alcohol use in general populations; a meta-analysis of randomized controlled trials. Drug and Alcohol Dependence 97(3):195-206.

Smith MU, Dane FC, Archer ME, et al (2000) Students Together Against Negative Decisions (STAND): evaluation of a school-based sexual risk reduction intervention in the rural South. AIDS Education & Prevention 12(1):49-71.

Spoth R, Redmond C, Lepper H (1999) Alcohol initiation outcomes of universal family-focused preventive interventions: One- and two-year follow-ups of a controlled study. Journal of Studies on Alcohol S13:103-111.

Spoth RL, Redmond C, Shin C (2001) Randomized trial of brief family interventions for general populations: Adolescent substance use outcomes 4 years following baseline. Journal of Consulting and Clinical Psychology 69(4).

Spoth RL, Guyll M, Day SX (2002) Universal family-focused interventions in alcohol-use disorder prevention: Cost-effectiveness and cost-benefit analyses of two interventions. Journal of Studies on Alcohol 63(2):219-228.

Spoth R, Redmond C, Shin C, Azevedo K (2004) Brief family intervention effects on adolescent substance initiation: school-level growth curve analyses 6 years following baseline. Journal of Consulting and Clinical Psychology 72(3):535-542.

St Pierre TL, Mark MM, Kaltreider L, Aikin KJ (1995) A 27-Month Evaluation of a Sexual Activity Prevention Program in Boys & Girls Clubs across the Nation Family Relations 44(1):69-77.

Stanton BF, Li X, Ricardo I, et al (1996) A randomized, controlled effectiveness trial of an AIDS prevention program for low-income African-American youths. Archives of Pediatrics & Adolescent Medicine 150(4):363-372.

Stanton B, Fang X, Li X, et al (1997) Evolution of risk behaviors over 2 years among a cohort of urban African American adolescents. Archives of Pediatrics & Adolescent Medicine 151(4):398-406.

Stanton BF, Li X, Galbraith J, et al (2000) Parental underestimates of adolescent risk behavior: a randomized, controlled trial of a parental monitoring intervention. The Journal of adolescent health : official publication of the Society for Adolescent Medicine 26(1):18-26.

Stanton BC, Cole M, Galbraith J, et al (2004) Randomized trial of a parent intervention: Parents can make a difference in long-term adolescent risk behaviors, perceptions, and knowledge. Archives of Pediatrics and Adolescent Medicine 158(10):947-955.

Stevens MM, Olson AL, Gaffney CA, et al (2002) A pediatric, practice-based, randomized trial of drinking and smoking prevention and bicycle helmet, gun, and seatbelt safety promotion. Pediatrics 109(3):490-498.

Tebes JK, Feinn R, Vanderploeg JJ, et al (2007) Impact of a positive youth development program in urban after-school settings on the prevention of adolescent substance use. Journal of Adolescent Health 41(3):239-248.

Toomey TL, Williams CL, Perry CL, et al (1996) An alcohol primary prevention program for parents of 7th graders: the Amazing Alternatives! Home Program. Journal of Child & Adolescent Substance Abuse 5(4):35-54.

Underhill K, Montgomery P, Operario D (2008) Abstinence-plus programs for HIV infection prevention in high-income countries. Cochrane Database of Systematic Reviews, Issue 1.

Underhill K, Operario D, Montgomery P (2007) Abstinence-only programs for HIV infection prevention in high-income countries. Cochrane Database of Systematic Reviews, Issue 4.

Villarruel AM, Jemmott JB, Jemmott LS (2006) A randomized controlled trial testing an HIV prevention intervention for Latino youth. Archives of Pediatrics & Adolescent Medicine 160(8):772-777.

Wiggins M, Bonell C, Sawtell M, et al (2009) Health outcomes of youth development programme in England: prospective matched comparison study. BMJ 339:b2534.

Winett RA, Anderson ES, Moore JF, et al (1992) Family/media approach to HIV prevention: results with a home-based, parent-teen video program. Health Psychology 11(3):203-206.

Winett RA, Anderson ES, Moore JF, et al (1993) Efficacy of a home-based human immunodeficiency virus prevention video program for teens and parents. Health Education Quarterly 20(4):555-567.

Wu Y, Stanton BF, Galbraith J, et al (2003) Sustaining and broadening intervention impact: a longitudinal randomized trial of 3 adolescent risk reduction approaches. Pediatrics 111(1):e32-e38.

Appendix 2. References to excluded studies

1. Study did not meet design criteria for inclusion (n=213)

Aguilera S, Plasencia AV, Aguilera S, Plasencia AV (2005) Culturally appropriate HIV/AIDS and substance abuse prevention programs for urban Native youth. Journal of Psychoactive Drugs 37(3):299-304.

Allamani A, Sani IB, Centurioni A, Ammannati P (2007) Preliminary evaluation of the educational strategy of a community alcohol use action research project in Scandicci (Italy). Substance Use & Misuse 42(12-13):2029-2040.

Allen D, Coombes L, Foxcroft D, Foxcroft DR (2007) Cultural accommodation of the Strengthening Families Programme 10-14: UK Phase I study. Health Education Research 22(4):547-560.

Allen JP, Philliber S, Hoggson N (1990) School-based prevention of teen-age pregnancy and school dropout: process evaluation of the national replication of the Teen Outreach Program. American Journal of Community Psychology 18(4):505-524.

Andreasson S, Sjostrom E, Branstrom R, Andreasson S, Sjostrom E, Branstrom R (2007) A sixcommunity prevention trial to reduce alcohol and drug use-related problems in Sweden: planning and early findings. Substance Use & Misuse 42(12-13):2017-2027.

Anon (1991) Drug-Free Schools & Communities: Program Planning Guidelines & Community Inventory.

Anon (1993) Reaching African-American Youth Who Live in High-Risk Environments. Technical Assistance Bulletin.

Anon (1998) Reaching young men: 2 successful approaches. Contracept Technol Update 19(8):101-102.

Anon (1999a) Compendium of HIV Prevention Interventions with Evidence of Effectiveness. From CDC's HIV/AIDS Prevention Research Synthesis Project.

Anon (1999b) Tennessee KIDS COUNT: The State of the Child in Tennessee, 1999. Tennessee Commission on Children and Youth, Tennessee KIDS COUNT, Andrew Johnson Tower, 9th Floor, 710 James Robertson Parkway, Nashville, TN 37243-0800. Tel: 800-264-0904 (Toll Free); Tel: 615-741-2633; Fax: 615-741-5956; e-mail: tccy@mail.state.tn.us; Web site: http://www.state.tn.us/tccy.

Anon (2001a) A Community Case Study on Underage Drinking Prevention.

Anon (2001b) Halfway There: A Prescription for Continued Progress in Preventing Teen Pregnancy. National Campaign To Prevent Teen Pregnancy, 1776 Massachusetts Avenue, NW, #200, Washington, DC 20036 (\$5). Tel: 202-478-8500; Fax: 202-478-8588; e-mail: campaign@teenpregnancy.org;. Web site: http://www.teenpregnancy.org.

Anon (2004) Interventions for alcohol use and alcohol use disorders in youth. Alcohol Research & Health: the Journal of the National Institute on Alcohol Abuse & Alcoholism 28(3):163-174.

Anon (2008) Study assesses impact of parent intervention about tobacco and alcohol. DATA: The Brown University Digest of Addiction Theory & Application 27(8):NaN.

Ashcraft C (2008) So Much More than "Sex Ed": Teen Sexuality as Vehicle for Improving Academic Success and Democratic Education for Diverse Youth. American Educational Research Journal 45(3):631-667.

As-Sanie S, Gantt A, Rosenthal MS, As-Sanie S, Gantt A, Rosenthal MS (2004) Pregnancy prevention in adolescents.[see comment]. American Family Physician 70(8):1517-1524.

August GJ (2001) Prevention of adolescent drug abuse: Targeting high-risk children with a multifaceted intervention model -- The early risers "Skills for success" program. Applied & Preventive Psychology 10(2).

Bailey KA (2004) Pilot randomized controlled trial of a brief alcohol intervention group for adolescents. Drug and Alcohol Review.23(2).

Baldwin JA, Daley E, Brown EJ, August EM, Webb C, Stern R, Malow R, DÇ,vieux JG (2008) Knowledge and perception of STI/HIV risk among rural African-American youth: lessons learned in a faith-based pilot program. Journal of HIV/AIDS Prevention in Children & Youth 9(1):97-115.

Baptiste DR, Paikoff RL, McKay MM, Madison-Boyd S, Coleman D, Bell C, Baptiste DR, Paikoff RL, McKay MM, Madison-Boyd S, Coleman D, Bell C (2005) Collaborating with an urban community to develop an HIV and AIDS prevention program for black youth and families. Behavior Modification 29(2):370-416.

Barnes ND, Harrod SE (1993) Teen pregnancy prevention: a rural model using school and community collaboration. School Counselor 41(2):137-140.

Barth RP (1991) Preventing adolescent abuse: Effective intervention strategies and techniques. Journal of Primary Prevention 11(3):193-205.

Bauman KE, Ennett ST, Foshee VA, Pemberton M, Hicks K (2001) Correlates of participation in a family-directed tobacco and alcohol prevention program for adolescents. Health Education & Behavior 28(4):440-461.

Bearinger LH, Sieving RE, Ferguson J, Sharma V, Bearinger LH, Sieving RE, Ferguson J, Sharma V (2007) Global perspectives on the sexual and reproductive health of adolescents: patterns, prevention, and potential. Lancet 369(9568):1220-1231.

Beatty SE (2003) A randomised comparison trial to evaluate an in-home parent-directed drug education intervention. Publisher: Curtin University of Technology, Bentley WA, 2003., Bentley-WA.

Benshoff JM, Alexander SJ (1993) The Family Communication Project: Fostering Parent-Child Communication about Sexuality. Elementary School Guidance and Counseling 27(4):288-300.

Biglan A, Hinds E, Biglan A, Hinds E (2009) Evolving prosocial and sustainable neighborhoods and communities. Annual Review of Clinical Psychology 5:169-196.

Boothroyd RI Analyzing the contribution of environmental change to prevent adolescent pregnancy: A study of three multi-component community initiatives in Kansas. Dissertation Abstracts International: Section B: The Sciences and Engineering.65(4-B).

Bosworth K (1997) Drug Abuse Prevention: School-based Strategies That Work. ERIC Digest.

Boyer CB, Sieverding J, Siller J, Gallaread A, Chang YJ (2007) Youth United Through Health Education: community-level, peer-led outreach to increase awareness and improve noninvasive sexually transmitted infection screening in urban African American youth. Journal of Adolescent Health 40(6):499-506.

Brock GC, Beazley RP (1995) Using the Health Belief Model to Explain Parents' Participation in Adolescents' At-Home Sexuality Education Activities. Journal of School Health 65(4):124-128.

Brown-Peterside P, Laraque D (1997) A community research model: a challenge to public health. American Journal of Public Health 87(9):1563-1564.

Brown EJ (2002) Recruitment feasibility and HIV prevention intervention acceptability among rural North Florida blacks. Journal of Community Health Nursing 19(3):147-161.

Brown EJ, Wells S (2005) A faith-based integrated substance abuse and HIV prevention program for rural African American adolescents. Journal of the American Psychiatric Nurses Association 11(6):344-351.

Brown HN, Saunders RB, Brown HN, Saunders RB (2002) College-bound sisters. Exploring one pregnancy prevention program. AWHONN Lifelines 6(2):146-151.

Brown SA (2001) Facilitating change for adolescent alcohol problems: A multiple options approach. [References], in Innovations in Adult Substance Abuse Interventions,(Wagner EFW, Holly B (Ed) ed. Elsevier.

Burrows MO (1998) A holistic peer education program to reduce STD infection among transient young adults in a resort community. Canadian Journal of Human Sexuality 7(4):365-370.

Busen NH, Engebretson JC, Busen NH, Engebretson JC (2008) Facilitating risk reduction among homeless and street-involved youth. Journal of the American Academy of Nurse Practitioners 20(11):567-575.

Butts JB, Hartman S (2002) Project BART: effectiveness of a behavioral intervention to reduce HIV risk in adolescents... Becoming a Responsible Teen. MCN: The American Journal of Maternal Child Nursing 27(3):163-171.

Cagampang HH, Barth RP, Korpi M, Kirby D (1997) Education Now and Babies Later (ENABL): life history of a campaign to Postpone Sexual Involvement. Family Planning Perspectives 29(3):109-114.

Caldera DB (2007) Impact of a statewide home visiting program on parenting and on child health and development. Child Abuse and Neglect 31(8):801-827.

Callaway M, Suedfeld P (1995) Residents' assessment of a community-based alcohol initiative in the Canadian Arctic. Arctic Medical Research 54(4):184-191.

Card JJ (1999) Teen pregnancy prevention: do any programs work? Annual Review of Public Health 20:257-285.

Carlson CE (1990) HIPP: A comprehensive school-based substance abuse program with cooperative community involvement. Journal of Primary Prevention 10(4):289-302.

Carter R (1998) Evaluation Research in Context: A Community Application for Youth and Family Programs. Family & Consumer Sciences Research Journal 26(3):346-363.

Chandler RK, Baker JG, Griffin M (1996) The impact of home-based sex education on rural adolescents sexual behavior and attitudes. Journal of Adolescent Health 18(2):136-136.

Cheadle AW (2001) The effect of neighborhood-based community organizing: Results from the Seattle Minority Youth Health Project. Health Services Research 36(4).

Christopher FS (1995) Adolescent Pregnancy Prevention. Family Relations 44(4):384-391.

Coggan C, Disley B, Patterson P (1998) Reports from the field. Community based intervention on adolescent risk taking: using research for community action. Injury Prevention 4(1):58-62.

Coombes L, Allen D, Marsh M, Foxcroft D (2009) The Strengthening Families Programme (SFP) 10-14 and substance misuse in Barnsley: The perspectives of facilitators and families. Child Abuse Review 18(1).

Cornelius JB, Lawrence JS (2009) Receptivity of African American adolescents to an HIV-prevention curriculum enhanced by text messaging. Journal for Specialists in Pediatric Nursing 14(2):123-132.

Cremeens JL, Usdan SL, Brock-Martin A, Martin RJ, Watkins K (2008) Parent-Child Communication to Reduce Heavy Alcohol Use among First-Year College Students. College Student Journal 42(1):152-163.

D'Onofrio CN (1997) The Prevention of Alcohol Use by Rural Youth. For full text: http://www.nida.nih.gov/PDF/Monographs/Monograph168/Download168.html.

D'Souza CM, Shrier LA (1999) Prevention and intervention of sexually transmitted diseases in adolescents. Current Opinion in Pediatrics 11(4):287-291.

de Anda D (2002) The GIG: An Innovative Intervention to Prevent Adolescent Pregnancy and Sexually Transmitted Infection in a Latino Community. Journal of Ethnic & Cultural Diversity in Social Work 11(3-4).

Deas DEMA, Deas Ddme (2008) Evidence-based treatments for alcohol use disorders in adolescents. Pediatrics 121(S4).

Delgado HM, Austin SB (2007) Can media promote responsible sexual behaviors among adolescents and young adults? Current Opinion in Pediatrics 19(4):405-411.

DeLisle S, Wasserheit JN (1999) Accelerated Campaign to Enhance STD Services (ACCESS) for youth: successes, challenges, and lessons learned. Sexually Transmitted Diseases 26(4 Suppl):S28-S41.

DeMairo P, Dischell J, Jouthe SA, Horner A (2008) The Teen Outreach Reproductive Challenge: Improving Adolescent Health Care Delivery through Peer Education Projects. American Journal of Sexuality Education 3(1):1-17.

Diamond S, Diamond Ssdio (2009) Building Xperience: A multilevel alcohol and drug prevention intervention. American Journal of Community Psychology 43(3-4):292-312.

DiCenso AG (2003) Review: Primary prevention strategies do not improve contraceptive use or reduce pregnancies in adolescents. Evidence-Based Medicine 8(1).

DiClemente RJ, Crosby RA (2006) Preventing sexually transmitted infections among adolescents: 'the glass is half full'. Current Opinion in Infectious Diseases 19(1):39-44.

Dolcini MM, Harper GW, Boyer CB, Watson SE, Anderson M, Pollack LM, Chang JY (2008) Preliminary findings on a brief friendship-based HIV/STI intervention for urban African American youth: Project ORE. Journal of Adolescent Health 42(6):629-633.

Doolittle M, Smith R (1997) Planning a Comprehensive Approach to Safe and Drug-Free Schools and Communities. Title IV Safe and Drug-Free Schools and Communities Evaluation, 1996-97. Publication Number 96.15.

Doswell WM (1999) Sexually risky behaviour in adolescents was reduced by a safer sex, condom based intervention [commentary on Jemmott JB 3rd, Jemmott LS, Fong GT. Abstinence and safer sex HIV risk-reduction interventions for African American adolescents: a randomized controlled trial. JAMA 1998 May 20;279(19):1529-36]. Evidence-Based Nursing 2(1):14-15.

Dryfoos JG (1990) A review of interventions to prevent pregnancy. Advances in Adolescent Mental Health 4:121-135.

Edwards ED, Seaman J (1995) A community approach for Native American drug and alcohol prevention programs: A logic model framework. Alcoholism Treatment Quarterly 13(2):43-62.

Ertle V (1993) Sharing Your Success III. Summaries of Successful Programs and Strategies Supporting Drug-Free Schools and Communities. Volume III.

Feudo R, Vining-Bethea S, Shulman LC, Shedlin MG, Burleson JA (1998) Bridgeport's Teen Outreach and Primary Services (TOPS) Project: a model for raising community awareness about adolescent HIV risk. Journal of Adolescent Health 23(2):49-59.

Fischer RL (2003) Getting the Word Out: Evaluating the Effectiveness of a Pregnancy PreventionCampaign for Pre-Teens. Journal of Community Practice 10(3):1-22.

Fisher DA (2000) Environmental Strategies To Prevent Alcohol Problems on College Campuses. For full text: http://www.udetc.org/documents/EnviroStrat.pdf.

Fluhr JD, Oman RF, Allen JR, Lanphier MG, McLeroy KR (2004) A collaborative approach to program evaluation of community-based teen pregnancy prevention projects. Health Promotion Practice 5(2):127-138.

Foster DG, Klaisle CM, Blum M, Bradsberry ME, Brindis CD, Stewart FH (2004) Expanded statefunded family planning services: estimating pregnancies averted by the Family PACT Program in California, 1997-1998. American Journal of Public Health 94(8):1341-1347.

Foster HW, And O (1990) A Model for Increasing Access: Teenage Pregnancy Prevention. Journal of Health Care for the Poor and Underserved:46.

Franklin C, Corcoran J (2000) Preventing adolescent pregnancy: a review of programs and practices. Social Work 45(1):40-53.

Fulkerson JA, Pasch KE, Perry CL, Komro K (2008) Relationships Between Alcohol-related Informal Social Control, Parental Monitoring and Adolescent Problem Behaviors Among Racially Diverse Urban Youth. Journal of Community Health 33:425-433.

Fullilove RE, Green L, Fullilove MT (2000) The Family to Family program: a structural intervention with implications for the prevention of HIV/AIDS and other community epidemics. AIDS 14(S1):S63-S67.

Galbraith J, Ricardo I, Stanton B, Black M, et al. (1996) Challenges and rewards of involving community in research: An overview of the "Focus on Kids" HIV risk reduction program. Health Education Quarterly 23(3):393-394.

Gardner SE, And O (1994) Signs of Effectiveness II: Preventing Alcohol, Tobacco, and Other Drug Use: A Risk Factor/Resiliency-Based Approach.

Gebhardt TL, Kaphingst K, DeJong W (2000) A campus-community coalition to control alcohol-related problems off campus: an environmental management case study.[see comment]. Journal of American College Health 48(5):211-215.

Giesbrecht N (2003) Preventing Alcohol-Related Problems in the US Through Policy: Media Campaigns, Regulatory Approaches and Environmental Interventions. Journal of Primary Prevention.24(1).

Gil AG, Tubman JG, Wagner EF (2004) Culturally sensitive substance abuse intervention for Hispanic and African American adolescents: empirical examples from the Alcohol Treatment Targeting Adolescents in Need (ATTAIN) project. Addiction 99(S2):140-150.

Gillis HL (1991) Project Choices: Update on the Effectiveness of Adventure Activities Used for Treating Addictions.

Glik D, Halpert-Schilt E, Zhang W (2001) Narrowcasting risks of drinking during pregnancy among African American and Latina adolescent girls. Health Promotion Practice 2(3):222-233.

Glik DPMMAEK (2008) Fetal alcohol syndrome prevention using community-based narrowcasting campaigns. Health Promotion Practice 9(1):93-103.

Goplerud EN (1991) Preventing Adolescent Drug Use: From Theory to Practice. OSAP Prevention Monograph-8.

Goulart M, Madover S (1991) An Aids Prevention Program for Homeless Youth. Journal of Adolescent Health 12(7):573-575.

Gould JR A prevention program for adolescent substance abuse: Self-agency, community participation and religious perception. Dissertation Abstracts International Section A: Humanities and Social Sciences.64(10-A).

Green HH, DocumÇ,t PI (2005) Parent peer education: lessons learned from a community-based initiative for teen pregnancy prevention. Journal of Adolescent Health 37(3S):S100-SNaN.

Greening L, Stoppelbein L, Jackson M (2001) Health education programs to prevent teen pregnancy. Journal of Adolescent Health 28(4):257-258.

Greenwood P (2008) Prevention and Intervention Programs for Juvenile Offenders. Future of Children 18(2):185-210.

Gregor MAS (2003) Feasibility of using an interactive laptop program in the emergency department to prevent alcohol misuse among adolescents. Annals of Emergency Medicine 42(2):276-284.

Gronbaek M, Stroger U, Strunge H, Moller L, Graff V, Iversen L (2001) Impact of a 10-year nationwide alcohol campaign on knowledge of sensible drinking limits in Denmark. European Journal of Epidemiology 17(5):423-427.

Guyll M, Guyll Mgie (2004) Family-Focused Preventive Interventions: Evaluating Parental Risk Moderation of Substance Use Trajectories. Journal of Family Psychology 18(2):293-301.

Harrington M (2001) Evaluation of Free To Grow, Phase II: Detailed Profile of the Free To Grow Project in California. Final Report. Publications Department, Mathematica Policy Research, Inc., P.O. Box 2393, Princeton, NJ 08543-2393 (Document no. PR01-14, \$3.40). Tel: 609-799-3535; Fax: 609-799-0005. For full text: http://www.mathematica-mpr.com/PDFs/ftgcaprofile.pdf.

Hawkins EH Preventing substance abuse in American Indian and Alaska native youth: Promising strategies for healthier communities. Marlatt, G Alan(Ed):New-621.

Hernandez LP (1996) DAYS La Familia Community Drug and Alcohol Prevention Program: Familycentered model for working with inner-city Hispanic families. Journal of Primary Prevention 16(3):255-272.

Hillman E, Hovell MF, Williams L, Hofstetter R, Burdyshaw C, Rugg D, Atkins C, Elder J, Blumberg E (1991) Pregnancy, STDS, and AIDS prevention: evaluation of New Image Teen Theatre. AIDS Education & Prevention 3(4):328-340.

Holder HD (2000) Community prevention of alcohol problems. Addictive Behaviors 25(6):843-859.

Holder HD (2004) Community prevention of young adult drinking and associated problems. Alcohol Research & Health 28(4):245-250.

Holder HD, Gruenewald PJ, Ponicki WR, Treno AJ, Grube JW, Saltz RF, Voas RB, Reynolds R, Davis J, Sanchez L, Gaumont G, Roeper P (2000) Effect of community-based interventions on high-risk drinking and alcohol-related injuries. JAMA 284(18):2341-2347.

Holder HD, Reynolds RI (1997) Application of local policy to prevent alcohol problems: experiences from a community trial. Addiction 92(S2):S285-S292.

Holder HD, Saltz RF, Grube JW, Voas RB, Gruenewald PJ, Treno AJ (1997) A community prevention trial to reduce alcohol-involved accidental injury and death: overview. Addiction 92(S2):S155-S171.

Holleran LK, Taylor-Seehafer MA, Pomeroy EC, Neff JA (2005) Substance Abuse Prevention for High-Risk Youth: Exploring Culture and Alcohol and Drug Use. Alcoholism Treatment Quarterly 23(2-3):165-184.

Holmila M (1995) Community action on alcohol: Experiences of the Lahti Project in Finland. Health Promotion International 10(4):283-291.

Huckle T, Conway K, Casswell S, Pledger M (2005) Evaluation of a regional community action intervention in New Zealand to improve age checks for young people purchasing alcohol. Health Promotion International 20(2):147-156.

Hutchinson MK, Hutchinson MKk (2007) Reconceptualizing adolescent sexual risk in a parent-based expansion of the theory of planned behavior. Journal of Nursing Scholarship 39(2):141-146.

Ichiyama MA, Fairlie AM, Wood MD, Turrisi R, Stanger L, Francis D, Ray AE (2008) A randomized trial of a parent-based intervention with incoming college students. Alcoholism-Clinical and Experimental Research 32(6):244A-244A.

James WH (1992) A Report on Restructuring Schools: Reducing Alcohol and Other Drug Abuse among Adolescents through a School-Community-University Partnership Program in the Federal Way School District.

Janz NK, Zimmerman MA, Wren PA, Israel BA, Freudenberg N, Carter RJ (1996) Evaluation of 37 AIDS prevention projects: Successful approaches and barriers to program effectiveness. Health Education Quarterly 23(1):80-97.

Jason LA, Curie CJ, Townsend SM, Pokorny SB, Katz RB, Sherk JL (2002) Health promotion interventions. Child & Family Behavior Therapy 24(1/2):67-83.

Johns MJ, Moncloa F, Gong EJ (2000) Teen Pregnancy Prevention Programs: Linking Research and Practice. Journal of Extension 38(4).

Johnson K, Noe T, Collins D, Strader T, Bucholtz G (2000) Mobilizing church communities to prevent alcohol and other drug abuse: a model strategy and its evaluation. Journal of Community Practice 7(2):1-28.

Johnson RL, Stanford PD, Douglas W, Jr., Botwinick G, Marino E (2001) High-risk sexual behaviors among adolescents engaged through a street-based peer outreach program--(the Adolescent HIV Project). Journal of the National Medical Association 93(5):170-177.

Kaplan C (2009) Review: behavioural counselling reduces sexually transmitted infections in adults and adolescents. Evidence-Based Nursing 12(2):46-47.

Kemp M (2006) Promoting the Health and Wellbeing of Young Black Men Using Community-Based Drama. Health Education 106(3):186-200.

Kerber CS, Schlenker E, Kerber CS, Schlenker E (2006) Use of a breathalyzer in a local bar: A community education project. Journal of Nursing Education 45(11):455-457.

King W, Nu'Man J, Fuller TR, Brown M, Smith S, Howell AV, Little S, Patrick P, Glover L, King W, Nu'Man J, Fuller TR, Brown M, Smith S, Howell AV, Little S, Patrick P, Glover L (2008) The diffusion of a community-level HIV intervention for women: lessons learned and best practices. Journal of Women's Health 17(7):1055-1066.

Kirby D, Kirby D (2002) Effective approaches to reducing adolescent unprotected sex, pregnancy, and childbearing. Journal of Sex Research 39(1):51-57.

Kirby D, Miller BC (2002) Interventions Designed To Promote Parent-Teen Communication about Sexuality. New Directions for Child and Adolescent Development 97:93-110.

Klein JD, Matos AM, Matos Auerbach M (2002) Improving adolescent health outcomes. Minerva Pediatrica 54(1):25-39.

Klitzner M, And O (1990) The Assessment of Parent-Led Prevention Programs: A National Descriptive Study. Journal of Drug Education:25.

Klitzner M, Gruenewald PJ, Bamberger E (1990) The assessment of parent-led prevention programs: a preliminary assessment of impact. Journal of Drug Education 20(1):77-94.

Komro KA (2002) Strategies to prevent underage drinking. Alcohol Research & Health 26(1):5-12.

Koo HP, Dunteman GH, George C, Green Y, Vincent M (1994) Reducing adolescent pregnancy through a school- and community-based intervention: Denmark, South Carolina, revisited. Family Planning Perspectives 26(5):206-213.

Kourtis APK (2006) Prevention of sexually transmitted human immunodeficiency virus (HIV) infection in adolescents. Current HIV Research 4(2):Apr.

Kumpfer KL, Alvarado R, Whiteside HO, Kumpfer KL, Alvarado R, Whiteside HO (2003) Family-based interventions for substance use and misuse prevention. Substance Use & Misuse 38(11-13):1759-1787.

LaBrie JW, Pedersen ER, Lamb TF, Bove L (2006) HEADS UP! A nested intervention with freshmen male college students and the broader campus community to promote responsible drinking. Journal of American College Health 54(5):301-305.

Lederman RP, Mian TS (2003) The Parent-Adolescent Relationship Education (PARE) Program: a curriculum for prevention of STDs and pregnancy in middle school youth. Behavioral Medicine 29(1):33-42.

Lesesne CA, Lewis KM, White CP, Green DC, Duffy JL, Wandersman A, Lesesne CA, Lewis KM, White CP, Green DC, Duffy JL, Wandersman A (2008) Promoting science-based approaches to teen pregnancy prevention: proactively engaging the three systems of the interactive systems framework. American Journal of Community Psychology 41(3-4):379-392.

Letters P, Stathis S, Letters P, Stathis S (2004) A mental health and substance abuse service for a youth detention centre. Australasian Psychiatry 12(2):126-129.

Lewis RK, Paine-Andrews A, Fawcett SB, Francisco VT, Richter KP, Copple B, Copple JE (1996) Evaluating the effects of a community coalition's efforts to reduce illegal sales of alcohol and tobacco products to minors. Journal of Community Health 21(6):429-437.

Little PMD, Harris E (2003) A Review of Out-of-School Time Program Quasi-Experimental and Experimental Evaluation Results. Out-of-School Time Evaluation Snapshot. Harvard Family Research Project, Harvard Graduate School of Education, 3 Garden Street, Cambridge, MA 02138. Tel: 617-495-9108; Fax: 617-495-8594; e-mail: hfrp@gse.harvard.edu; Web site: http://www.hfrp.org. For full text: http://www.gse.harvard.edu/hfrp/content/projects/afterschool/resources/snapshot1.pdf.

Logan BN (1991) Adolescent Substance Abuse Prevention: An Overview of the Literature. Family & Community Health 13(4):25-36.

MacKinnon DP, Nohre L, Pentz MA, Stacy AW (2000) The alcohol warning and adolescents: 5-year effects. American Journal of Public Health 90(10):1589-1595.

Maguin E, Safyer A, Nochajski T, Dewit D, Macdonald S (2003) The impact of a family-based alcohol prevention program on children's externalizing behavior problems. Alcoholism Clinical and Experimental Research 27(5 Supplement):72A.

Manlove J, Terr-Humen E, Papillo AR, Franzetta K, Williams S, Ryan S (2002) Preventing Teenage Pregnancy, Childbearing, and Sexually Transmitted Diseases: What the Research Shows. American Teens. Child Trends Research Brief. Child Trends, 4301 Connecticut Ave., NW, Suite 100, Washington, DC 20008 (Stock no. 2002-31, \$5). Tel: 202-362-5580; Fax: 202-362-5533; Web site: http://www.childtrends.org. For full text: http://www.childtrends.org/PDF/KnightReports/K1Brief.pdf.

Marcus MT, Walker T, Swint JM, Smith BP, Brown C, Busen N, Edwards T, Liehr P, Taylor WC, Williams D, Von SK (2004) Community-based participatory research to prevent substance abuse and HIV/AIDS in African-American adolescents. Journal of Interprofessional Care 18(4):347-360.

Mattox JR, II (1997) A Review of Interventions To Increase Driving Safety among Teenage Drivers.

McKay MBMMCTDEBCE (2007) Adapting a family-based HIV prevention program for HIV-infected preadolescents and their families: Youth, families and health care providers coming together to address complex needs. Social Work in Mental Health 5(3-4):146.

Melton AP, Chino M, May PA, Gossage JP (2000) Promising Practices and Strategies To Reduce Alcohol and Substance Abuse among American Indians and Alaska Natives. An OJP Issues & Practices Report. Full text at Web site: http://www.ojp.usdoj.gov/americannative/promise.pdf.

Meltzer IJ, Fitzgibbon JJ, Leahy PJ, Petsko KE (2006) A youth development program: lasting impact. Clinical Pediatrics 45(7):655-661.

Mikhailovich K, Arabena K (2005) Evaluating an indigenous sexual health peer education project. Health Promotion Journal of Australia 16(3):189-194.

Miller T, Blewden M, Zhang J-fEMA, Miller Tmpo (2004) Cost savings from a sustained compulsory breath testing and media campaign in New Zealand. Accident Analysis & Prevention 36(5).

Mizuno Y, Kennedy M, Weeks-Norton K, Myllyluoma J, Mizuno Y, Kennedy M, Weeks-Norton K, Myllyluoma J (2002) An examination of adolescents who were and were not exposed to "Teens Stopping AIDS": reaching the hard-to-reach. Journal of Health Communication 7(3):197-203.

Montoya ID, Atkinson J, McFaden WC (2003) Best characteristics of adolescent gateway drug prevention programs. Journal of Addictions Nursing 14(2):75-84.

Moore KA, And O (1995) Adolescent Pregnancy Prevention Programs: Interventions and Evaluations.

Mosena PW (2004) Peer Advocates for Health: A Community-Based Program to Improve Reproductive Health Knowledge and Lifestyle Choices among Adolescent Males. International Journal of Men's Health 3(3):221-240.

Nissen LB, Nissen LB (2007) Reclaiming futures: communities helping teens overcome drugs, alcohol and crime--a new practice framework for juvenile justice. Journal of Psychoactive Drugs 39(1):51-58.

Nitz K (1999) Adolescent pregnancy prevention: a review of interventions and programs. Clinical Psychology Review 19(4):457-471.

Norman J (1999) Peer sexuality education. Social Policy 30(1):30-31.

O'Donnell L, Wilson-Simmons R, Dash K, Jeanbaptiste V, Myint U, Moss J, Stueve A (2007) "Saving Sex for Later": Developing a Parent-Child Communication Intervention to Delay Sexual Initiation among Young Adolescents. Sex Education: Sexuality 7(2):107-125.

Ottenritter N, Barnett L (2002) HIV Prevention Strategies for Community Colleges: Lessons Learned from Bridges to Healthy Communities. AACC Project Brief. Community College Press, P.O. Box 311, Annapolis Junction, MD 20701 (Order # 1546: pack of 20, \$20). Tel: 800-250-6557 (Toll Free); Fax: 301-604-0158; e-mail: aaccpub@pmds.com. For full text: http://www.aacc.nche.edu.

Paikoff RL (1996) Adapting developmental research to intervention design: Applying developmental psychology to an AIDS prevention model for urban African American youth. Journal of Negro Education 65(1):44-59.

Paine-Andrews A, Fisher J, Paine-Andrews Aake (2002) Analyzing the contribution of community change to population health outcomes in an adolescent pregnancy prevention initiative. Health Education & Behavior 29(2):183-193.

Paine-Andrews A, Vincent M (1996) Replicating a community initiative for preventing adolescent pregnancy: From South Carolina to Kansas. Family & Community Health 19(1):14-30.

Parker L (1990) The Missing Component in Substance Abuse Prevention Efforts: A Native American Example. Contemporary Drug Problems no. 2:251-270

Pearson JF (1991) Preventing unwanted pregnancies. BMJ 303(6803):598.

Peckham S Community development interventions for the prevention of teenage pregnancy and supporting young women and lone mothers, in Promoting the health of teenage and lone mothers: setting a research agenda/ A report of a Health Education Authority Expert Working Group chaired by Kaye Wellings. Health Education Authority, London.

Peckham S (1993) Preventing unintended teenage pregnancies. Public Health 107(2):125-133.

Perez SM, Duany LA (1992) Reducing Hispanic Teenage Pregnancy and Family Poverty: A Replication Guide. Final Version. National Council of La Raza, Publications Department, 810 First Street, N.E., Suite 300, Washington, DC 20002 (\$15).

Perry CL, Williams CL, Komro KA, Veblen-Mortenson S, Forster JL, Bernstein-Lachter R, Pratt LK, Dudovitz B, Munson KA, Farbakhsh K, Finnegan J, McGovern P (2000) Project Northland high school interventions: community action to reduce adolescent alcohol use. Health Education & Behavior 27(1):29-51.

Pierre N, Cox J (1997) Teenage pregnancy prevention programs. Current Opinion in Pediatrics 9(4):310-316.

Pilgrim C, Abbey A, Hendrickson P, Lorenz S (1998) Implementation and impact of a family-based substance abuse prevention program in rural co/mmunities. Journal of Primary Prevention 18(3):341-361.

Pinkerton SD, Cecil H, Holtgrave DR (1998) HIV/STD prevention interventions for adolescents: costeffectiveness considerations. Journal of HIV/AIDS Prevention & Education for Adolescents & Children 2(2):5-32.

Plotnick RD (1993) The effect of social policies on teenage pregnancy and childbearing. Families in Society 74(6):324-328.

Podschun GD (1993) Teen Peer Outreach-Street Work Project: HIV prevention education for runaway and homeless youth. Public Health Reports 108(2):150-156.

Puentes WJ, Wassel M (2003) Using peer health education to enhance family life education. Journal of School Nursing (Allen Press) 19(6):313-319.

Rebchook GM, Kegeles SM, Huebner D, Team TR, Rebchook GM, Kegeles SM, Huebner D (2006) Translating research into practice: the dissemination and initial implementation of an evidence-based HIV prevention program. AIDS Education & Prevention 18(4 Suppl A):119-136. Rector R (2002) The Effectiveness of Abstinence Education Programs in Reducing Sexual Activity among Youth. The Heritage Foundation Backgrounder. Heritage Foundation, 214 Massachusetts Avenue, N.W., Washington, DC 20002-4999. Tel: 202-546-4400. For full text: http://www.heritage.org/library/backgrounder.

Rehnman C, Larsson J, Andreasson S, Rehnman C, Larsson J, Andreasson S (2005) The beer campaign in Stockholm--Attempting to restrict the availability of alcohol to young people. Alcohol 37(2):65-71.

Resnick MD (2007) Challenges and prospects for community-partnered research. Journal of Adolescent Health.40(6):487-488.

Rohrbach LA (1997) Alcohol-related outcomes of the Day One Community Partnership. Evaluation and Program Planning 20(3):315-322.

Rotheram-Borus MJ (1991) HIV and adolescents. Journal of Primary Prevention 12(1):65-82.

Rotheram-Borus MJ (2000) Expanding the Range of Interventions to Reduce HIV among Adolescents. AIDS 14(S1):S33-S40.

Rotheram-Borus MJ, O'Keefe Z, Kracker R, Foo HH (2000) Prevention of HIV among adolescents. Prevention Science 1(1):15-30.

Roye CF, Hudson M (2003) Developing a culturally appropriate video to promote dual-method use by urban teens: rationale and methodology. AIDS Education & Prevention 15(2):148-149.

Safer LA, Harding CG (1993) Under pressure program: using live theatre to investigate adolescents' attitudes and behavior related to drug and alcohol abuse education and prevention. Adolescence 28(109):135-148.

Sallett AJ (1990) A Proactive Alcohol and Drug Abuse Prevention Program.

Schilling RF, McAlister AL (1990) Preventing drug use in adolescents through media interventions. Journal of Consulting & Clinical Psychology 58(4):416-424.

Schinke SP (1992) Multicomponent, school-based strategies to prevent HIV infection and sexually transmitted diseases among adolescents: Theory and research into practice. Research on Social Work Practice 2(3):364-379.

Segest E, And O (1990) Free Condoms in Youth Clubs in Copenhagen. Journal of Adolescence 13(1):17-24.

Shaffer HJD (2005) The epidemiology of college alcohol and gambling policies. Harm Reduction Journal 2(1):1-20.

Share RA, Stacks JS (2006) From the field. Youth-adult partnership in community organizing: a case study of the My Voice Counts! campaign. Journal of Community Practice 14(4):113-128.

Silvern J (1991) Strengthening Youth and Family Resistance to Alcohol and Other Drug Abuse. Special Focus. Family Resource Coalition Report 10(3). Smith MU (2000) STAND: A peer educator training curriculum for sexual risk reduction in the rural south. Preventive Medicine: An International Journal Devoted to Practice and Theory 30(6):441-449.

StafstrÇôm M, Ostergren P, Larsson S, Lindgren B, Lundborg P (2006) A community action programme for reducing harmful drinking behaviour among adolescents: the Trelleborg Project. Addiction 101(6):813-824.

Stafstrom M, Ostergren PO, Stafstrom M, Ostergren PO (2008) A community-based intervention to reduce alcohol-related accidents and violence in 9th grade students in southern Sweden: the example of the Trelleborg project. Accident Analysis & Prevention 40(3):920-925.

Stephens TT, Braithwaite R, Robillard A, Finnie R, Colbert SJ (2002) A community-based approach to eliminating racial and health disparities among incarcerated populations: the HIV example for inmates returning to the community. Health Promotion Practice 3(2):255-264.

Stevens MM, Mott LA, Youells F (1996) Rural adolescent drinking behavior: three year follow-up in the New Hampshire substance abuse prevention study. Adolescence 31(121):159-166.

Summerville G, Canova K (2006) The Power of Plain Talk: Exploring One Program' Influence on the Adolescent Reproductive Health Field, in Public/Private Ventures, pp Dec.

Svenson GR, Ç-stergren P, Merlo J, RǾstam L (2002) Action control and situational risks in the prevention of HIV and STIs: individual, dyadic, and social influences on consistent condom use in a university population. AIDS Education & Prevention 14(6):515-532.

Thomas MH (2000) Abstinence-based programs for prevention of adolescent pregnancies: a review. Journal of Adolescent Health 26(1):5-17.

Torrence WA, Guidry JJ (2007) An initiative for the development of a rural church-based adolescent sexuality education intervention. American Journal of Health Education 38(3):172-175.

Townsend J, Richards R (2000) Promoting the positive: the Northern Territory's "Choose Yourself" youth alcohol and smoking campaign. Health Promotion Journal of Australia 10(2):135-140.

Treno AJ, Remer LG (2007) The Sacramento Neighborhood Alcohol Prevention Project: Outcomes From a Community Prevention Trial. Journal of Studies on Alcohol and Drugs 68(2):197-207.

Trimble JE, Beauvais F (2001) Prevention of Alcoholism, Drug Abuse, and Health Problems among American Indians and Alaska Natives: An Introduction and Overview.

Van Stelle KR (1998) Alcohol and drug prevention among American Indian families: The Family Circles Program. Drugs & Society 12(1-2):53-60.

Vincent ML, Paine-Andrews A, Fisher J, Devereaux RS, Dolan HG, Harris KJ, Reininger B (2000) Replication of a community-based multicomponent teen pregnancy prevention model: Realities and challenges. Family & Community Health 23(3):28-45.

Visser AP, van BP, van Bilsen P (1994) Effectiveness of sex education provided to adolescents. Patient Education & Counseling 23(3):147-160.

Wagenaar AC (1999) Communities Mobilizing for Change on Alcohol: Lessons and results from a 15community randomized trial. Journal of Community Psychology 27(3):315-326.

Wagenaar AC, Murray DM, Gehan JP, Wolfson M, Forster JL, Toomey TL, Perry CL, Jones WR (2000) Communities mobilizing for change on alcohol: outcomes from a randomized community trial. Journal of Studies on Alcohol 61(1):85-94.

Wagenaar AC, Perry CL (1994) Community Strategies for the Reduction of Youth Drinking: Theory and Application. Journal of Research on Adolescence 4(2):319-345.

Walker-Shaw M (1993) Applying community organization to developing health promotion programs in the school community. Journal of School Health 63(2):109-111.

Wang B, Meier A, Shah I, Li X (2006) The impact of a community-based comprehensive sex education program on Chinese adolescents' sex-related knowledge and attitudes. Journal of HIV/AIDS Prevention in Children & Youth 7(2):43-65.

Ward K, Waters J (1999) The impact of targeted prevention programs for adolescents at high risk for HIV transmission. Journal of HIV/AIDS Prevention & Education for Adolescents & Children 3(1/2):51-78.

Weiss R (2001) Realistic program for teens discourages drunk driving. Health Progress 82(6):10-11.

Williams CLP (1998) Design and implementation of parent programs for a community-wide adolescent alcohol use prevention program. Journal of Prevention and Intervention in the Community 17(2):1998.

Wilson N, Minkler M, Dasho S, Wallerstein N, Martin AC (2008) Getting to social action: the Youth Empowerment Strategies (YES!) project. Health Promotion Practice 9(4):395-404.

Wren PA, Janz NK, Carovano K, Zimmerman MA, Washienko KM (1997) Preventing the spread of AIDS in youth: principles of practice from 11 diverse projects. Journal of Adolescent Health 21(5):309-317.

Zane NA (1998) Dosage-related changes in a culturally-responsive prevention program for Asian American youth. Drugs and Society 12(1-2).

Zibalese-Crawford M (1997) A creative approach to HIV/AIDS programs for adolescents. Social Work in Health Care 25(1/2):73-89.

Zweig JM, Van Ness A (2001) The National Study of Girl Neighborhood Power: An Out-of-School Program for Girls Ages 9 to 14. Urban Institute, 2100 M Street, N.W., Washington, DC 20037. Tel: 202-833-7200; Fax: 202-429-0687; e-mail: pubs@ui.urban.org. For full text: http://www.urban.org/pdfs/GNP-Study.pdf.

2. Population targeted by the intervention(s) did not meet review criteria (n=64)

(1995) A Hispanic/Latino Family Approach to Substance Abuse Prevention. CSAP Cultural Competence Series 2.

(2002) How To Reduce High-Risk College Drinking: Use Proven Strategies, Fill Research Gaps. Final Report of the Panel on Prevention and Treatment. For full text: http://www.collegedrinkingprevention.gov/Reports/Panel02/Panel02_TOC.aspx.

Anderko L, Uscian M (2000) The effectiveness of a community-level HIV/STD prevention program in a three-county rural area. Family & Community Health 23(3):46-58.

Boyer CB, Barrett DC, Peterman TA, Bolan G (1997) Sexually transmitted disease (STD) and HIV risk in heterosexual adults attending a public STD clinic evaluation of a randomized controlled behavioral risk-reduction intervention trial. AIDS 11(3):359-367.

Brindis CD, Geierstanger SP, Wilcox N, McCarter V, Hubbard A (2005) Evaluation of a peer provider reproductive health service model for adolescents. Perspectives on Sexual & Reproductive Health 37(2):85-92.

Buka SL, Birdthistle IJ (1999) Long-term effects of a community-wide alcohol server training intervention. Journal of Studies on Alcohol 60:27-36.

Burgos M, Reininger B, Richter DL, Coker AL, AlegrÇða M, Vera M, Saunders R (2000) Correlates of sexually transmitted infections among street-based female adolescent sex workers in Puerto Rico: implications for community health. International Quarterly of Community Health Education 20(3):253-265.

Caron SL, Carter DB, Davis CM, Macklin E (1997) Evaluating the effectiveness of workshop interventions on contraceptive use among first-year college students. Journal of Psychology & Human Sexuality 9(3-4):99-120.

Chung PJ, Borneo H, Kilpatrick SD, Lopez DM, Travis R, Jr., Lui C, Khandwala S, Schuster MA (2005) Parent-adolescent communication about sex in Filipino American families: a demonstration of community-based participatory research. Ambulatory Pediatrics 5(1):50-56.

Clapp JD, Johnson M, Lange JE, Russell C, Shillington A, Voas RB (2005) Reducing DUI among US college students: results of an environmental prevention trial. Addiction 100(3):327-334.

Croom K, Lewis D, Marchell T, Lesser ML, Reyna VF, Kubicki-Bedford L, Feffer M, Staiano-Coico L (2009) Impact of an online alcohol education course on behavior and harm for incoming first-year college students: short-term evaluation of a randomized trial. Journal of American College Health 57(4):445-454.

Darbes L, Crepaz N, Lyles C, Kennedy G, Rutherford G (2008) The efficacy of behavioral interventions in reducing HIV risk behaviors and incident sexually transmitted diseases in heterosexual African Americans. AIDS 22(10):1177-1194.

Dearing B, Caston RJ, Babin J (1991) The Impact of a Hospital Based Educational Program on Adolescent Attitudes toward Drinking and Driving. Journal of Drug Education no. 4(pp. 349-359).

East P, Kiernan E, ChÇvez G (2003) An evaluation of California's Adolescent Sibling Pregnancy Prevention Program. Perspectives on Sexual & Reproductive Health 35(2):62-71.

Ehrhardt AA, Exner TM, Hoffman S, Silberman I, Yingling S, ms-Skinner J, Smart-Smith L (2002) HIV/STD risk and sexual strategies among women family planning clients in New York: Project FIO. AIDS & Behavior 6(1):1-14.

Fitter M, Urquhart R (2008) Awareness of emergency contraception: a follow-up report. Journal of Family Planning & Reproductive Health Care 34(2):111-114.

Flowers P, Hart GJ, Williamson LM, Frankis JS, Der GJ (2002) Does bar-based, peer-led sexual health promotion have a community-level effect amongst gay men in Scotland? International Journal of STD & AIDS 13(2):102-108.

Friedman AS, Terras A, Glassman K (2002) Multimodel substance use intervention program for male delinquents. Journal of Child & Adolescent Substance Abuse 11(4):43-66.

Fromme K (1994) The Alcohol Skills Training Program: A group intervention for young adult drinkers. Journal of Substance Abuse Treatment 11(2):143-154.

Godin G, Naccache H, Cote F, Leclerc R, Frechette M, Alary M (2008) Promotion of safe sex: evaluation of a community-level intervention programme in gay bars, saunas and sex shops. Health Education Research 23(2):287-298.

Gollub EL, Brown EL, Savouillan M, Waterlot J, Coruble G (2002) A community-based safer-sex intervention for women: results of a pilot study in south-eastern France. Culture, Health & Sexuality 4(1):21-42.

Hays RB, Kegeles SM, Rebchook GM (2003) The Mpowerment Project: community-building with young gay and bisexual men to prevent HIV. American Journal of Community Psychology 31(3-4):301-312.

Hingson RW, Howland J, Hingson RW, Howland J (2002) Comprehensive community interventions to promote health: implications for college-age drinking problems. Journal of Studies on Alcohol (S14):226-240.

Hupp SDA (2003) The development and validation of the Parent Instruction-Giving Game With Youngsters (PIGGY) in a head start population. Dissertation Abstracts International: Section B: The Sciences and Engineering 63(11-B).

Ichiyama MA, Fairlie AM, Wood MD, Turrisi R, Francis DP, Ray AE, Stanger LA, Ichiyama MA, Fairlie AM, Wood MD, Turrisi R, Francis DP, Ray AE, Stanger LA (2009) A randomized trial of a parentbased intervention on drinking behavior among incoming college freshmen. Journal of Studies on Alcohol & Drugs Supplement.(16):67-76. Kahn JG, Kegeles SM, Hays R, Beltzer N (2001) Cost-effectiveness of the Mpowerment Project, a community-level intervention for young gay men. Journal of Acquired Immune Deficiency Syndromes 27(5):482-491.

Kamb ML, Fishbein M, Douglas JM, Rhodes F, Rogers J, Bolan G, Zenilman J, Hoxworth T, Malotte CK, latesta M, Kent C, Lentz A, Graziano S, Byers RH, Peterman TA (1998) Efficacy of risk-reduction counseling to prevent human immunodeficiency virus and sexually transmitted diseases: a randomized controlled trial. Project RESPECT Study Group. JAMA : the journal of the American Medical Association 280(13):1161-1167.

Kegeles SM, Hays RB, Coates TJ (1996) The Mpowerment Project: a community-level HIV prevention intervention for young gay men.

Kiene SM (2006) A Brief Individualized Computer-Delivered Sexual Risk Reduction Intervention Increases HIV/AIDS Preventive Behavior. Journal of Adolescent Health 39(3):404-410.

Kneidek T (1995) Project Imani: Changing the Way the World Works for People: A Guide to Family-Based Alcohol and Other Drug Education for Young Children.

Larimer ME, Cronce J (2002) Identification, prevention, and treatment: A review of individual-focused strategies to reduce problematic alcohol consumption by college students. Journal of Studies on Alcohol S14:148-163.

Lauby JL, Smith PJ, Stark M, Person B, Adams J (2000) A community-level HIV prevention intervention for inner-city women: results of the women and infants demonstration projects. American Journal of Public Health 90(2):216-223.

Leslie M (1991) An analysis of community intervention strategies designed to improve AIDS-related knowledge, attitudes and beliefs, and health practices among high-risk reproductive-age women. Dissertation Abstracts International 52:2828.

Maguin E (1994) The path to alcohol problems through conduct problems: A family-based approach to very early intervention with risk, (Boyd GMH, Jan (Ed) ed, pp Current-124.

Marion LN, Finnegan L, Campbell RT, Szalacha LA (2009) The Well Woman Program: A Community-Based Randomized Trial to Prevent Sexually Transmitted Infections in Low-Income African American Women. Research in Nursing & Health 32(3):274-285.

Marsh JC (1991) Evaluation of Hull House teen pregnancy and parenting program. Evaluation and Program Planning.14(1-2).

Maxwell AE, Bastani R, Warda U (2002) Pilot test of a single-session AIDS workshop for young Hispanic U.S. immigrants. Journal of Immigrant Health 4(2):73-79.

McCambridge J, Strang J (2004) The efficacy of single-session motivational interviewing in reducing drug consumption and perceptions of drug-related risk and harm among young people: results from a multi-site cluster randomized trial. Addiction 99(1):39-52.

McCartt AT, Hellinga LA, Wells JK, McCartt AT, Hellinga LA, Wells JK (2009) Effects of a college community campaign on drinking and driving with a strong enforcement component. Traffic Injury Prevention 10(2):141-147.

Meyer L, Job SN, Bouyer J, Bouvet E, Spira A (1991) Prevention of sexually transmitted diseases: a randomised community trial. Journal of Epidemiology and Community Health 45(2):152-158.

Nelson TF, Weitzman ER, Wechsler H (2005) The effect of a campus-community environmental alcohol prevention initiative on student drinking and driving: results from the "a matter of degree" program evaluation. Traffic Injury Prevention 6(4):323-330.

Nyamathi AM, Flaskerud J, Bennett C, Leake B, Lewis C (1994) Evaluation of two AIDS education programs for impoverished Latina women. AIDS Education & Prevention 6(4):296-310.

Opuni KA, And O (1994) The Northeast Adolescent Project: A Collaborative Effort to Address Teen-Age Pregnancy in Houston, Texas. Journal of School Health 64(5):212-214.

Peterson PL, Baer JS, Wells EA, Ginzler JA, Garrett SB (2006) Short-Term Effects of a Brief Motivational Intervention to Reduce Alcohol and Drug Risk among Homeless Adolescents. Psychology of Addictive Behaviors 20(3):254-264.

Pick De WS, Givaudan M, Givaudan S, Pick De Weiss S (1993) Planeando Tu Vida: sex and family life education: fundamentals of development, implementation, and evaluation. International Journal of Adolescent Medicine & Health 6(3-4):211-224.

Pinkerton SD, Holtgrave DR, DiFranceisco WJ, Stevenson LY, Kelly JA (1998) Cost-effectiveness of a community-level HIV risk reduction intervention. American Journal of Public Health 88(8):1239-1242.

Ploem C, Byers ES (1997) The effects of two AIDS risk-reduction interventions on heterosexual college women's AIDS-related knowledge, attitudes and condom use. Journal of Psychology & Human Sexuality 9(1):1-24.

Proude EM, D'Este C, Ward JE (2004) Randomized trial in family practice of a brief intervention to reduce S risk in young adults. risk by FPs. As primary prevention efforts for young adults remain important, randomized controlled trials of larger scope are needed. Family Practice 21(5):537-544.

Roeper PJ (2000) A long-term community-wide intervention to reduce alcohol-related traffic injuries: Salinas, California. Drugs: Education, Prevention & Policy 7(1).

Saltz RF, Welker LR, Paschall MJ, Feeney MA, Fabiano PM (2009) Evaluating a Comprehensive Campus-Community Prevention Intervention to Reduce Alcohol-Related Problems in a College Population. Journal of Studies on Alcohol and Drugs 70(S16):21-27.

Schlapman N, Cass PS (2000) Project: HIV prevention for incarcerated youth in Indiana. Journal of Community Health Nursing 17(3):151-158.

Sigelman C, Derenowski E, Woods T, Mukai T, feld-Liro C, Durazo O, Maddock A (1996) Mexican-American and Anglo-American children's responsiveness to a theory-centered AIDS education program. Child Development 67(2):253-266. Sulak PJ, Herbelin S, Kuehl AL, Kuehl TJ (2005) Analysis of knowledge and attitudes of adult groups before and after attending an educational presentation regarding adolescent sexual activity. American Journal of Obstetrics & Gynecology 193(6):1945-1955.

Turrisi R, Jaccard J, Taki R, Dunnam H, Grimes JEMA, Turrisi Rrbe (2001) Examination of the shortterm efficacy of a parent intervention to reduce college student drinking tendencies. Psychology of Addictive Behaviors 15(4).

Turrisi R, Larimer ME, Mallett K, Turrisi Rrpe (2009) A randomized clinical trial evaluating a combined alcohol intervention for high-risk college students. Journal of Studies on Alcohol and Drugs 70(4):555-567.

Vrungos SM (2003) Increasing HIV intervention effectiveness using an enhanced motivational approach: An evaluation of a multi-component motivation-skills program on HIV risk reduction behavior and condom use in heterosexual college students. Dissertation Abstracts International: Section B: The Sciences and Engineering.63(7-B).

Walls CT, Lauby J, Lavelle K, Derby T, Bond L (1998) Exposure to a community-level HIV prevention intervention: who gets the message. Journal of Community Health 23(4):281-300.

Wolitski RJ (2006) Relative efficacy of a multisession sexual risk-reduction intervention for young men released from prisons in 4 states. American Journal of Public Health 96(10):1854-1862.

Wood MD, DeJong W, Fairlie AM, Lawson D, Lavigne AM, Cohen F (2009) Common Ground: An Investigation of Environmental Management Alcohol Prevention Initiatives in a College Community. Journal of Studies on Alcohol and Drugs 70(S16):96-105.

Yonas M, Baker D, Cornwell EE, III, Chang D, Phillips J, Paradise J, Paradise M, Sutton E, Elihu A (2005) Readiness to change and the role of inpatient counseling for alcohol/substance abusing youth with major trauma. Journal of Trauma 59(2):466-470.

Ziersch A, Gaffney J, Tomlinson DR (2000) STI prevention and the male sex industry in London: evaluating a pilot peer education programme. Sexually Transmitted Infections 76(6):447-453.

Zimmerman RS, Palmgreen PM, Noar SM, Lustria ML, Lu HY, Horosewski ML (2007) Effects of a Televised Two-City Safer Sex Mass Media Campaign Targeting High-Sensation-Seeking and Impulsive-Decision-Making Young Adults. Health Education & Behavior:826.

3. Intervention(s) examined was not based in a community setting (n=53)

Allen JP, Philliber S (1991) Evaluating Why and How the Teen Outreach Program Works: Years 3-5 of the Teen Outreach National Replication (1986/87-1988/89). Association of Junior Leagues International, 660 First Avenue, New York, NY 10016-3241.

Allen JP, Philliber S, Herrling S, Kuperminc GP (1997) Preventing teen pregnancy and academic failure: experimental evaluation of a developmentally based approach.

Arnold EM, Smith TE, Harrison DF, Springer DW (2000) Adolescents' knowledge and beliefs about pregnancy: the impact of 'enabl.'. Adolescence 35(139):485-486.

Botvin G, Baker E, Dusenbury L, Tortu S, Botvin E (1990) Preventing adolescent drug abuse through a multimodal cognitive behavioural approach: results of a 3 year study. Journal of Consulting and Clinical Psychology 58:437-446.

Botvin GJ, Baker E, Dusenbury L, Botvin EM, Diaz T (1995) Long-term follow-up results of a randomized drug abuse prevention trial in a white middle-class population. JAMA : the journal of the American Medical Association 273:1106-1112.

Boulter LT (2007) The effectiveness of peer-led FAS/FAE prevention presentations in middle and high schools... Fetal alcohol syndrome... fetal alcohol effects. Journal of Alcohol & Drug Education 51(3):7-27.

Brown EJ (2002) Recruitment feasibility and HIV prevention intervention acceptability among rural North Florida blacks. Journal of Community Health Nursing 19(3):147-161.

Christopher FS, Roosa MW (1990) An evaluation of an adolescent pregnancy prevention program: Is 'Just say no' enough ? Family Relations 39:68-72.

Collins J, Robin L, Wooley S, Fenley D, Hunt P, Taylor J, Haber D, Kolbe L, Collins J, Robin L, Wooley S, Fenley D, Hunt P, Taylor J, Haber D, Kolbe L (2002) Programs-that-work: CDC's guide to effective programs that reduce health-risk behavior of youth. Journal of School Health 72(3):93-99.

Coyle K, Basen-Engquist K, Kirby D, Parcel G, Banspach S, Collins J, Baumler E, Carvajal S, Harrist R (2001) Safer choices: reducing teen pregnancy, HIV, and STDs. Public Health Reports 116 Suppl 1:82-93.

Coyle K, Kirby D, Parcel G, Basen-Engquist K, Banspach S, Rugg D, Weil M (1996) Safer choices: a multicomponent school-based HIV/STD and pregnancy prevention program for adolescents. Journal of School Health 66(3):89-95.

D'Amico EJ, Edelen MO, D'Amico EJ, Edelen MO (2007) Pilot test of Project CHOICE: a voluntary afterschool intervention for middle school youth. Psychology of Addictive Behaviors 21(4):592-598.

Danoff AL (1996) The effects of multiple preventive intervention strategies on alcohol and other drug knowledge and behavior, social competence, and self-esteem of at-risk youth. Dissertation Abstracts International: Section B: The Sciences and Engineering.58(1-B).

Dedobbeleer N, Desjardins S (2001) Outcomes of an ecological and participatory approach to prevent alcohol and other drug "abuse" among multiethnic adolescents. Substance Use & Misuse 36(13):1959-1991.

Eischens A, Komro KA, Perry CL, Bosma LM, Farbakhsh K (2004) The Association of Extracurricular Activity Participation with Substance Use Among Youth in the DARE Plus Project. American Journal of Health Education 35(2):68-75.

Eisen M, Zellman GL (1990) Evaluating the impact of a theory-based sexuality and contraceptive education program. Family Planning Perspectives 22(6):261-262.

Ellickson PL, McCaffrey DF, Ghosh-Dastidar B, Longshore DL, Ellickson PL, McCaffrey DF, Ghosh-Dastidar B, Longshore DL (2003) New inroads in preventing adolescent drug use: results from a large-scale trial of project ALERT in middle schools. American Journal of Public Health 93(11):1830-1836.

Frost JJ, Forrest JD (1995) Understanding the Impact of Effective Teenage Pregnancy Prevention Programs. Family Planning Perspectives 27(5):188-195.

Gottsegen E, Philliber WW (2001) Impact of a Sexual Responsibility Program on Young Males. Adolescence 36(143):427-433.

Guzman BL, Casad BJ, Feria A, Schlehofer-Sutton MM, Villanueva CM (2003) A community-based approach to promoting safe sex behaviour in adolescence. Journal of Community & Applied Social Psychology 13(4):269-283.

Guzman BL, Casad BJ, Schlehofer-Sutton MM, Villanueva CM, Feria A (2001) C.A.M.P.: A Community-Based Approach to Promoting Safe Sex Behavior in Adolescence.

Hahn EJ, Hall LA, Rayens MK, Myers AV, Bonnel G (2007) School- And Home-Based Drug Prevention: Environmental, Parent, and Child Risk Reduction. Drugs: Education 14(4):319-331.

Johnson CA (1990) Relative effectiveness of comprehensive community programming for drug abuse prevention with high-risk and low-risk adolescents. Journal of Consulting and Clinical Psychology 58(4):447-456.

Jones DJ (2005) A Family-Focused Randomized Controlled Trial to Prevent Adolescent Alcohol and Tobacco Use: The Moderating Roles of Positive Parenting and Adolescent Gender. Behavior Therapy 36(4):347-355.

Jorgensen SR (1991) Project Taking Charge: An Evaluation of an Adolescent Pregnancy Prevention Program. Family Relations 40(4):373-380.

Kirby D, Korpi M, Barth RP, Cagampang HH (1997) The impact of the Postponing Sexual Involvement curriculum among youths in California. Family Planning Perspectives 29(3):100-108.

Komro KA (2006) Cross-Cultural Adaptation and Evaluation of a Home-Based Program for Alcohol Use Prevention among Urban Youth: The "Slick Tracy Home Team Program." The Journal of Primary Prevention 27(2):135-154.

Komro KA, Perry CL, Murray DM, Veblen-Mortenson S, Williams CL, Anstine PS (1996) Peer-planned social activities for preventing alcohol use among young adolescents. Journal of School Health 66(9):328-334.

Komro KA, Perry CL, Veblen-Mortenson S, Farbakhsh K, Toomey TL, Stigler MH, Jones-Webb R, Kugler KC, Pasch KE, Williams CL (2008) Outcomes from a randomized controlled trial of a multi-

component alcohol use preventive intervention for urban youth: Project Northland Chicago. Addiction 103(4):606-619.

Kulis S, Marsiglia FF, Elek E, Dustman P, Wagstaff DA, Hecht ML (2005) Mexican/Mexican American adolescents and keepin' it REAL: an evidence-based substance use prevention program. Children & Schools 27(3):133-146.

Kulis S, Nieri T, Yabiku S, Stromwall L, Kulis Skae (2007) Promoting reduced and discontinued substance use among adolescent substance users: Effectiveness of a universal prevention program. Prevention Science 8(1):35-49.

MacIntosh JM (2008) Theatre-based peer education for youth: A powerful medium for HIV prevention, sexuality education and social change. Dissertation Abstracts International 68(8).

MacKillop J, Ryabchenko KA, Lisman SA, MacKillop J, Ryabchenko KA, Lisman SA (2006) Life skills training outcomes and potential mechanisms in a community implementation: a preliminary investigation. Substance Use & Misuse 41(14):1921-1935.

Magnusson J, Kendall S, Oakley L, Townsend J (2004) Promoting contraceptive services to teenagers. Community Practioner 77(10):381-384.

McBride D, Gienapp A (2000) Using randomized designs to evaluate client-centered programs to prevent adolescent pregnancy. Family Planning Perspectives 32(5):227-236.

Olsen J, Weed S (1992) Student evaluation of sex education programs advocating abstinence. Adolescence 27(106):369-370.

Perry CL, Williams CL, Komro KA, Veblen-Mortenson S, Stigler MH, Munson KA, Farbakhsh K, Jones RM, Forster JL, Perry CL, Williams CL, Komro KA, Veblen-Mortenson S, Stigler MH, Munson KA, Farbakhsh K, Jones RM, Forster JL (2002) Project Northland: long-term outcomes of community action to reduce adolescent alcohol use. Health Education Research 17(1):117-132.

Perry CL, Williams CL, Veblen-Mortenson S, Toomey TL, Komro KA, Anstine PS, McGovern PG, Finnegan JR, Forster JL, Wagenaar AC, Wolfson M (1996) Project Northland: outcomes of a communitywide alcohol use prevention program during early adolescence. American Journal of Public Health 86(7):956-966.

Richards-Colocino N, McKenzie P, Newton R (1996) Project Success: Comprehensive intervention services for middle school high-risk youth. Journal of Adolescent Research 11(1):130-163.

Roberts SW, McCowan RJ (2004) The Effectiveness of Infant Simulators. Adolescence 39(155):475-487.

Rohrbach LA, And O (1994) Parental Participation in Drug Abuse Prevention: Results from the Midwestern Prevention Project. Journal of Research on Adolescence 4(2):295-317.

Simkin L, And O (1992) Evaluation of the HIV/AIDS Education Program/Including Condom Availability, 1990-1992. OREA Report.

Slater MD, Kelly KJ, Edwards RW, Thurman PJ, Plested BA, Keefe TJ, Lawrence FR, Henry KL (2006) Combining in-school and community-based media efforts: reducing marijuana and alcohol uptake among younger adolescents. Health Education Research 21(1):157-167.

Smyer R (1991) Key Issues in Education: Drug-Free Schools. 1990-91 Program Evaluation of Federally Funded Drug-Free Schools Programs.

Spoth RL, Redmond C, Trudeau L, Shin C, Spoth RL, Redmond C, Trudeau L, Shin C (2002) Longitudinal substance initiation outcomes for a universal preventive intervention combining family and school programs. Psychology of Addictive Behaviors 16(2):129-134.

Swanson DJ, Zegers KM, Zwaska AA (2004) Implementing a Social Norms Approach to Reduce Alcohol Abuse on Campus: Lessons Learned in the Shadow of 'The World's Largest Six-Pack'. The Social Science Journal 41(4):621-635.

Tait RJ, Hulse GK, Tait RJ, Hulse GK (2003) A systematic review of the effectiveness of brief interventions with substance using adolescents by type of drug. Drug & Alcohol Review 22(3):337-346.

Tanner JF, Jr., Anne RM, Ladd SD, Tanner JFJ, nne Raymond M, Ladd SD (2009) Evaluating a community saturation model of abstinence education: an application of social marketing strategies. Health Marketing Quarterly 26(1):27-41.

Tucker JSFA (2007) The effect of the national demonstration project Healthy Respect on teenage sexual health behaviour. The European Journal of Public Health 17(1):33-41.

Williams CL (1995) A home-based prevention program for sixth-grade alcohol use: Results from project Northland. Journal of Primary Prevention 16(2):125-147.

Williams CL, Grechanaia T, Romanova O, Komro KA, Perry CL, Farbakhsh K (2001) Russian-American partners for prevention. Adaptation of a school-based parent-child programme for alcohol use prevention. European Journal of Public Health 11(3):314-321.

Williams CL, Perry CL, Farbakhsh K, Veblen-Mortenson S (1999) Project Northland: comprehensive alcohol use prevention for young adolescents, their parents, schools, peers and communities. Journal of Studies on Alcohol 13(S13):112-124.

Wood E, Shakeshaft A, Gilmour S, Sanson-Fisher R (2006) A systematic review of school-based studies involving alcohol and the community. Australian & New Zealand Journal of Public Health 30(6):541-550.

4. Intervention targeted 'at risk' or high risk population (n=11)

(1993) From the Centers for Disease Control and Prevention. Assessment of street outreach for HIV prevention--selected sites, 1991-1993. JAMA 270(22):2675.

Barrett KA Multiple-family intervention with adolescent substance abusers. Dissertation Abstracts International.52(2-B).

Goldberg P (1994) SPARK Peer Helper Program, 1993-94. OER Report. Office of Educational Research, New York City Board of Education, 110 Livingston Street, Room 740, Brooklyn, NY 11201.

Hanlon TE, Bateman R (2002) An early community-based intervention for the prevention of substance abuse and other delinquent behavior. Journal of Youth and Adolescence 31(6):459-471.

Harrell A, Cavanagh S, Sridharan S (1999) Evaluation of the Children at Risk Program: Results 1 Year after the End of the Program. Research in Brief. National Criminal Justice Reference Service, Box 6000, Rockville, MD 20849-6000. Tel: 800-851-3420 (Toll Free); e-mail: askncjrs@ncjrs.org. For full text: http://www.ojp.usdoj.gov/nij.

Hawke J, Jainchill N, Messina M (2005) Post-treatment outcomes among adjudicated adolescent males and females in modified therapeutic community treatment. Substance Use & Misuse 40(7):975-996.

Marvel F, Rowe CL, Colon-Perez L, DiClemente RJ, Liddle HA, Marvel F, Rowe CL, Colon-Perez L, DiClemente RJ, Liddle HA (2009) Multidimensional family therapy HIV/STD risk-reduction intervention: an integrative family-based model for drug-involved juvenile offenders. Family Process 48(1):69-84.

Oliva G, Rienks J, Udoh I, Smith CD (2005) A university and community-based organization collaboration to build capacity to develop, implement, and evaluate an innovative HIV prevention intervention for an urban African American population. AIDS Education & Prevention 17(4):300-317.

Orte C, Touza C, Ballester L, March M (2008) Children of Drug-Dependent Parents: Prevention Programme Outcomes. Educational Research 50(3):249-260.

Powers S, And O (1992) Quantitative and Qualitative Methods in the Evaluation of an Alcohol and Other Drug Abuse Prevention Program for High Risk Families with Preschool Children.

Walton FR (1991) Education versus schooling: Project LEAD: High expectation. Journal of Negro Education.60(3):441-453.

5. Intervention examined was not alcohol education and/SRE related (n=9)

Brindis, C. D., Llewelyn, L., Marie, K., Blum, M., Biggs, A. and Maternowska, C. (2003) Meeting the reproductive health care needs of adolescents: California's Family Planning Access, Care, and Treatment Program. Journal of Adolescent Health 32, 79-91.

Hanks DE, III (1996) An Implementation and Evaluation of the Curriculum, "Taking Care", Designed to Address the Needs of Young Adult and Adolescent Unwed Fathers. Dissertation Abstracts International 56(12).

Hingson R, McGovern T, Howland J, Heeren T, Winter M, Zakocs R (1996) Reducing alcoholimpaired driving in Massachusetts: the Saving Lives Program. American Journal of Public Health 86(6):791-798. Holder HD, Saltz RF, Grube JW, Treno AJ, Reynolds RI, Voas RB, Gruenewald PJ (1997) Summing up: lessons from a comprehensive community prevention trial. Addiction 92(S2):S293-S301.

Huckle T, Greenaway S, Broughton D, Conway K, Huckle T, Greenaway S, Broughton D, Conway K (2007) The use of an evidence-based community action intervention to improve age verification practices for alcohol purchase. Substance Use & Misuse 42(12-13):1899-1914.

Klein JD (2003) Improving adolescent preventive services through state, managed care, and community patnerships. Journal of Adolescent Health 32(S6):91-97.

Merritt BL (2006) An intervention for at-risk Chicano adolescents using their dating relationships. Dissertation Abstracts International: Section B: The Sciences and Engineering.67(8-B).

Stewart TCP (2009) Evaluation of an adolescent hospital-based injury prevention program. Journal of Trauma - Injury, Infection and Critical Care 66(5):1451-1460.

Vappie-Aydin RC (2008) Reducing the rate of recidivism for first-time juvenile offenders with the Parent Monitoring Program, PMP: A family counseling intervention program. Dissertation Abstracts International 69(5):2008.

Appendix 3. Quality assessment tables

Table 10.1. Quality assessment: randomised controlled trials (individual)

Deference	Po	pulati	on	r	Nethod	d of all	ocatio	n to in	terven	tion (c	or com	pariso	n)			Outc	omes					Anal	yses			Sum	mary
Reference	1.1	1.2	1.3	2.1	2.2	2.3	2.4	2.5	2.6	2.7	2.8	2.9	2.10	3.1	3.2	3.3	3.4	3.5	3.6	4.1	4.2	4.3	4.4	4.5	4.6	5.1	5.2
Bauman et al ., 2000	NR	NR	NR	+	+	NR	NR	+	+	NR	-	+	+	NR	+	-	++	NR	++	+	NR	NR	-	+	NR	-	+
Boekeloo et al., 1999	NR	NR	NR	+	++	NR	NA	+	NR	NR	++	NR	NR	+	NR	++	++	+	+	+	NR	NR	++	+	++	+	-
Danielson et al., 1990	NR	NR	NR	+	++	NR	+	+	NR	NR	NR	NR	NR	NR	NR	+	+	+	NR	NR	NR	NR	+	+	+	-	-
DiClemente et al., 2004	NR	NR	NR	++	++	+	NR	++	NR	++	++	NR	NR	++	NR	++	++	++	++	++	NR	++	++	++	++	++	-
Downs et al., 2004	NR	NR	NR	++	++	NR	NR	++	NR	NR	NR	NR	NR	++	NR	++	++	++	++	+	NR	-	+	+	+	+	-
Forehand et al., 2007	+	+	++	+	++	NR	NA	-	NR	+	++			++	++	++	++	++	++	++	++	++	+	++	-	+	+
Haggerty et al., 2007	NR	NR	NR	+	++	NR	NR	-	NR	NR	++	+	+	++	++	+	++	++	++	NR	++	NR	+	+	+	+	+
Jemmott et al., 1992	NR	NR	NR	+	+	NR	NR	+	NR	+	NR	NR	NR	++	NR	++	++	+	+	+	NR	NR	+	+	+	+	-
Jemmott et al., 1998	NR	NR	NR	++	++	NR	NR	++	NR	++	++	NR	NR	+	NR	++	++	++	++	++	++	++	++	++	++	++	-
Jemmott et al., 2005	NR	NR	+	++	++	++	+	++	NR	++	++	NR	NR	++	NR	++	++	++	++	++	NR	NR	++	++	++	++	+
Kipke et al., 1993	NR	NR	NR	+	++	NR	NR	++	NR	++	+	NR	NR	+	NR	+	+	++	-	++	NR	NR	+	+	+	+	-
Lederman et al., 2008	+	NR	+	+	-	NR	NR	NR	NR	NR	-	++	+	++	+	-	+	++	++	NR	NR	NR	-	+	-	-	+
Lederman et al., 2004	NR	NR	NR	+	+	NR	NR	+	NR	NR	NR	NR	NR	+	NR	-	+	+	+	NR	NR	NR	-	+	-	-	-
Loveland- Cherry et al., 1999	NR	NR	NR	+	-	NR	NR	NR	NR	NR	-	++	+	NR	++	+	++	++	++	NR	NR	NR	+	+	+	-	+
Miller et al., 1993	+	+	+	+	+	NR	NR	NR	NR	NR	++	-	+	++	++	+	++	++	++	NR	NR	NR	-	+	++	+	-
Morrison-Beedy et al., 2005	NR	NR	NR	+	++	NR	NR	++	NR	++	-	NR	NR	++	NR	++	++	++	+	+	NR	NR	+	+	+	+	-
O'Donnell et al., 2005	+	++	++	++	++	NA	NA	+	++	NR	++	NR	NR	++	++	++	++	++	+	++	++	++	+	+	+	+	+
Philliber et al., 2002	NR	NR	NR	+	++	NR	NR	++	NR	++	+	NR	NR	+	NR	+	+	+	++	+	+	NR	+	+	-	+	-
Prado et al., 2007	+	++	++	++	++	NR	NA	++	NR	+	NR	NR	-	-	++	++	++	++	++	+	++	++	-	+	+	+	-

Deference	Population			I	Nethod	d of all	ocatio	n to in	terven	tion (c	or com	pariso	n)			Outc	omes					Anal	yses			Sum	mary
Reference	1.1	1.2	1.3	2.1	2.2	2.3	2.4	2.5	2.6	2.7	2.8	2.9	2.10	3.1	3.2	3.3	3.4	3.5	3.6	4.1	4.2	4.3	4.4	4.5	4.6	5.1	5.2
Schinke et al., 2009	NR	NR	NR	+	+	NR	NR	+	NR	NR	++	NR	NR	++	NR	++	++	++	+	++	+	NR	-	++	+	+	-
Slesnick et al., 2008	++	++	++	++	++	+	NR	++	-	+	++	+	+	++	++	+	++	NR	+	++	++	NR	-	+	++	+	++
Stanton et al., 2000	NR	NR	NR	++	++	++	NR	-	NR	NR	+	-	+	++	++	+	++	+	-	++	NR	NR	+	++	++	+	+
Villarruel et al., 2006	NR	NR	NR	++	++	++	NR	++	+	++	+	NR	NR	+	+	++	++	++	++	++	++	NR	++	++	++	++	-
Winett et al., 1992	NR	NR	NR	+	++	NA	NA	+	NA	NR	-	++	NA	+	+	++	++	++	+	+	-	NR	-	+	+	-	+
Winett et al., 1993	+	NR	NR	+	++	NR	++	NR	NA	NR	++	+	+	++	-	+	++	++	-	++	NR	NR	-	+	+	+	+

Table 10.2. Quality assessment: randomised controlled trials (cluster)

Defenence	Po	opulati	on		Metho	d of all	ocatio	n to in	terven	tion (d	or com	pariso	n)			Outc	omes					Anal	yses			Sum	mary
Reference	1.1	1.2	1.3	2.1	2.2	2.3	2.4	2.5	2.6	2.7	2.8	2.9	2.10	3.1	3.2	3.3	3.4	3.5	3.6	4.1	4.2	4.3	4.4	4.5	4.6	5.1	5.2
Anderson et al., 1999	NR	NR	NR	+	+	NR	NR	++	NR	NR	+	+	+	+	+	-	+	++	++	NR	-	-	+	+	-	-	+
Beatty et al., 2008	+	+	+	+	+	NA	NA	+	NR	NR	++	NR	NR	++	++	++	++	NR	NR	-	NA	+	+	+	++	+	+
Brody et al., 2004; 2006	+	NR	NR	+	++	NR	NR	++	NR	NR	++	NR	NR	+	NR	++	++	++	++	++	NR	NR	++	++	++	+	+
Cohen & Rice, 1995	-	NR	++	NR	+	NR	NR	+	NR	NR	NR	NR	NR	+	++	++	++	NR	++	++	NR	NR	++	++	++	-	-
Dancy et al., 2006	++	++	++	+	++	NR	NR	++	NR	+	++	NR	NR	++	++	+	+	++	-	++	NR	NR	+	+	++	+	+
Di Noia & Schinke, 2007	NR	NR	NR	+	++	NR	+	++	++	++	+	NR	NR	++	NR	+	++	++	-	++	-	NR	++	++	++	+	
DiLorio et al., 2006	+	NR	NR	+	++	NR	NR	NR	+	NR	NR	+	+	+	++	++	++	++	++	-	++	NR	-	+	++	++	+
Dilorio et al., 2007	+	+	+	+	+	NA	NA	+	NR	+	++	NR	NR	++	++	++	++	++	++	++	NR	-	+	++	+	+	+
Elder et al., 2002	NR	NR	NR	+	++	NR	+	++	++	++	-	NR	NR	+	NR	+	+	++	++	-	NR	NR	+	+	+	+	-
Johnson et al., 1996	NR	NR	NR	+	+	NR	NR	+	NR	NR	NR	NR	NR	+	NR	-	+	+	++	+	NR	NR	-	+	-	-	-
Jones et al., 2005	NR	NR	NR	+	-	NA	NA	NR	NA	NR	++	++	+	++	++	-	++	++	++	+	NR	NR	+	+	++	+	+
Schinke et al., 2004	NR	NR	NR	+	++	NR	NR	++	NR	NR	++	+	+	++	++	+	++	++	++	++	+	NR	+	++	++	+	+

Deference	Po	opulati	on		Metho	d of all	ocatio	n to in	terven	tion (c	or com	pariso	n)			Outc	omes					Anal	yses			Sum	mary
Reference	1.1	1.2	1.3	2.1	2.2	2.3	2.4	2.5	2.6	2.7	2.8	2.9	2.10	3.1	3.2	3.3	3.4	3.5	3.6	4.1	4.2	4.3	4.4	4.5	4.6	5.1	5.2
Schinke et al., 2005	NR	NR	NR	+	+	NR	NR	+	++	NR	++	NR	NR	NR	NR	-	+	++	-	+	NR	NR	-	+	-	-	-
Sikkema et al., 2005	NR	NR	NR	+	++	NR	NR	++	++	++	++	NR	NR	++	NR	++	++	++	+	++	NR	NR	++	++	++	+	-
Spoth et al., 1999; 2001; 2004	NR	NR	NR	+	+	+	NR	NR	NR	NR	+	+	+	+	++	+	++	++	++	++	NR	NR	+	++	++	+	+
Stanton et al., 2004	+	-	-	+	++	NR	NA	+	NR	+	++	NR	NR	++	++	++	++	++	++	++	NR	+	NR	++	++	+	++
Stanton et al., 1996	NR	NR	NR	+	++	NR	NR	++	NR	+	+	NR	NR	+	NR	+	+	++	++	+	++	NR	+	+	+	+	-
Stevens et al., 2002	NR	NR	NR	++	++	NR	NR	++	++	++	-	NR	NR	++	NR	++	++	++	++	++	NR	NR	++	++	++	+	-
Toomey et al., 1996	+	-	+	-	+	NA	NA	+	-	-	++	NR	NR	++	++	++	++	++	++	NR	NA	+	-	+	++	-	+
Wu et al., 2003	NR	NR	NR	+	++	NR	NR	++	++	NR	-	NR	NR	+	NR	++	++	++	++	++	NR	NR	+	++	+	+	-

Table 10.3. Quality assessment: other study designs

Poforonoo	Study	Po	opulat	ion	м	ethod	of all	ocatior	n to in	terven	tion (or con	nparise	on)			Outc	omes					Anal	yses			Sum	mary
Reference	design	1.1	1.2	1.3	2.1	2.2	2.3	2.4	2.5	2.6	2.7	2.8	2.9	2.10	3.1	3.2	3.3	3.4	3.5	3.6	4.1	4.2	4.3	4.4	4.5	4.6	5.1	5.2
Elliott et al., 1996	NRCT	NR	NR	NR	-	+	+	+	+	+	NR	+	NR	NR	NR	NR	+	+	+	-	NR	NR	NR	-	+	-	-	-
Ferguson, 2000	NRCT	+	NR	NR	-	+	-	NR	+	-	-	+	NR	NR	+	NR	+	+	+	+	NR	-	+	-	+	-	-	-
Gustafson et al., 1998	NRCT	NR	NR	NR	-	+	-	NR	+	++	NR	+	NR	NR	+	NR	+	+	+	-	-	NR	NR	+	+	+	+	-
Koutakis et al., (2008)	NRCT	+	++	++	NR	+	NR	NR	++	++	+	++	NR	NR	NR	++	++	++	++	++	++	++	++	++	+	++	++	++
Pearlman et al., 2002	NRCT	NR	NR	NR	-	++	-	NR	++	NR	++	NR	NR	NR	+	NR	+	+	+	+	+	+	NR	-	+	-	+	-
Rew et al., 2007	NRCT	+	+	++	NR	-	NR	NR	NA	++	NR	NR	NR	NR	++	++	++	++	NR	-	++	NA	NR	+	++	++	+	+
Scheinberg et al., 1997	NRCT	NR	NR	NR	-	+	-	-	NR	NR	+	++	+	+	I	++	+	++	+	-	++	NR	-	-	+	+	•	-
St Pierre et al., 1995	NRCT	NR	NR	NR	-	+	-	NR	+	++	NR	-	NR	NR	I	NR	+	+	++	++	-	-	NR	+	+	+	•	-
Cheadle et al., 1995	CBA	+	NR	NR	-	+	-	-	+	+	NR	NA	NA	NA	NR	NR	+	+	+	NA	+	NA	NR	+	+	+	+	-
Flynn et al., 2006	CBA	NR	NR	NR	NR	++	NR	NR	++	++	+	NA	NR	NR	++	NR	+	+	++	++	NA	NA	NR	+	++	+	+	-

Poforonoo	Study	Po	pulati	on	Μ	lethod	of all	ocatior	n to in	terver	tion (or com	paris	on)			Outc	omes					Anal	yses			Sum	mary
Reference	design	1.1	1.2	1.3	2.1	2.2	2.3	2.4	2.5	2.6	2.7	2.8	2.9	2.10	3.1	3.2	3.3	3.4	3.5	3.6	4.1	4.2	4.3	4.4	4.5	4.6	5.1	5.2
Gleghorn et al., 1997	СВА	++	++	++	NR	+	NR	NA	-	++	+	NA	NR	NR	-	++	++	++	NR	NR	++	NA	NR	+	++	-	-	+
Kypri et al., 2005	CBA	NR	NR	NR	-	+	-	NR	+	NR	NR	+	NR	NR	+	NR	+	+	NR	-	NR	NR	+	+	+	+	+	-
McKay et al., 2004	CBA	+	NR	NR	-	++	NR	NA	NR	NR	NR	++	+	+	NR	+	+	++	-	-	-	NR	NR	I	+	++	-	+
Postrado & Nicholson, 1992	CBA	NR	NR	NR	-	+	-	-	+	NR	NR	NR	NR	NR	+	NR	-	+	NR	-	+	NR	NR	+	+	+	-	-
Smith et al., 2000	CBA	NR	NR	NR	-	+	NR	NR	++	++	NR	NR	NR	NR	+	NR	++	++	++	+	+	NR	NR	-	+	-	-	-
Tebes et al., 2007	CBA	NR	NR	NR	NR	+	NR	NR	+	++	NR	+	NR	NR	+	NR	+	+	+	++	+	+	NR	+	+	+	+	-
Wiggins et al., 2009	CBA	NR	NR	NR	-	+	-	NR	+	NR	+	+	+	++	++	NR	++	++	+	++	+	NR	+	++	++	++	+	-
Doniger et al., 2001	CTS	NR	NR	NR	-	++	-	-	++	NA	NR	NA	NR	NR	+	NR	+	+	NA	NA	NA	NA	NA	-	+	-	-	
Sieverding et al., 2005	CTS	NR	NR	NR	-	+	NA	NA	+	++	NR	NA	NR	NR	NR	NR	+	+	NR	+	+	NA	NA	NA	+	NA	+	-
NRCT – Non-ra	andomised	contro	olled tr	ial; CB	SA – co	ontrolle	d befc	ore and	after s	study;	CTS –	cross-	section	nal time	series	;												

Table 10.4. Quality assessment: systematic reviews and meta-analyses

Poforonoo(c)	Questions					Coding
Reference(S)	1.1	1.2	1.3	1.4	1.5	Coung
Arnold & Rotherham- Borus (2009)	-	-	-	NR	NA	-
DiCenso et al., 2002	++	++	++	++	++	++
Foxcroft et al., 2002	++	++	++	++	NA	++
Foxcroft et al., 2003	++	++	++	+	NA	+
Franklin et al., 1997	++	++	+	++	++	++
Guyatt et al., 2000	+	++	++	-	NA	+
Pedlow & Carey, 2003	++	+	+	-	NA	+
Petrie et al., 2007	++	++	+	++	NA	++
Robin et al., 2004	+	+	+	-	NA	+

Poforonoo(c)	Questions					Coding					
Reference(S)	1.1	1.2	1.3	1.4	1.5	coung					
Sales et al., 2006	++	++	++	-	NA	+					
Smit et al., 2008	++	++	++	++	++	++					
Underhill et al., 2007	++	++	++	++	NA	++					
Underhill et al., 2008 ++ ++ ++ ++ NA ++											
NA – not applicable; NR	R – not reported; ++ well co	overed; + adequately add	ressed; - poorly addresse	d							

Table 10.5. Quality assessment for published economic evaluation studies

Study i Include	dentification author, title, reference, year of publication	Spoth et al., 2002
Evaluat	tion criterion	
1.	Was a well-defined question posed in answerable form?	Yes
1.1	Did the study examine both costs and effects of the service(s) or programme(s)?	Yes
1.2	Did the study involve a comparison of alternatives?	Yes
1.3	Was a viewpoint for the analysis stated and was the study placed in any particular decision-making context?	Yes, societal
2.	Was a comprehensive description of the competing alternatives given (that is, can you tell who? did what? to whom? where? and how often?)?	Yes, both interventions were described.
2.1	Were any important alternatives omitted?	No
2.2	Was (Should) a do-nothing alternative (be) considered?	Yes, the intervention was compared to a minimal contact intervention.
3.	Was the effectiveness of the programmes or services established?	Yes, in a cluster RCT
3.1	Was this done through a randomised, controlled clinical trial? If so, did the trial protocol reflect what would happen in regular practice?	Yes, as above
3.2	Was effectiveness established through an overview of clinical studies?	NA
3.3	Were observational data or assumptions used to established effectiveness? If so, what are the potential biases in results?	NA

Study in Include	dentification author, title, reference, year of publication	Spoth et al., 2002
4.	Were all the important and relevant costs and consequences for each alternative identified?	Yes
4.1	Was the range wide enough for the research question at hand?	They appeared to be.
4.2	Did it cover all relevant viewpoints? (Possible viewpoints include the community or social viewpoint, and those of patients and third-party payers.)	Again they appeared to be, the authors discussed using the human capital approach to value the societal cost of alcohol disorders.
4.3	Were capital costs, as well as operating costs, included?	Not clear.
5.	Were costs and consequences measured accurately in appropriate physical units (for example, hours of nursing time, number of physician visits, lost work-days, gained life-years)?	Not reported
5.1	Were any of the identified items omitted from measurement? If so, does this mean that they carried no weight in the subsequent analysis?	Not reported
5.2	Were there any special circumstances (for example, joint use of resources) that made measurement difficult? Were these circumstances handled appropriately?	Not reported
6.	Were costs and consequences valued credibly?	Partially
6.1	Were the sources of all values clearly identified? (Possible sources include market values, patient or client preferences and views, policy-makers' views and health professionals' judgements.)	Not reported
6.2	Were market values employed for changes involving resources gained or depleted?	Not reported
6.3	Where market values were absent (for example, volunteer labour), or did not reflect actual values (for example, clinic space donated at reduced rate), were adjustments made to approximate market values?	ΝΑ
6.4	Was the valuation of consequences appropriate for the question posed (that is, has the appropriate type or types of analysis – cost-effectiveness, cost-benefit, cost-utility – been selected)?	Yes
7.	Were costs and consequences adjusted for differential timing?	Yes
7.1	Were costs and consequences which occur in the future 'discounted' to their present values?	Yes, discount rate of 3% applied
7.2	Was any justification given for the discount rate used?	Yes, recommended by the Panel on Cost- effectiveness in Health and Medicine
8.	Was an incremental analysis of costs and consequences of alternatives performed?	No
8.1	Were the additional (incremental) costs generated by one alternative over another compared to the additional effects, benefits or utilities generated?	No
9.	Was allowance made for uncertainty in the estimates of costs and consequences?	Yes
9.1	If data on costs or consequences were stochastic, were appropriate statistical analyses performed?	Not reported

Study in Include	dentification author, title, reference, year of publication	Spoth et al., 2002
9.2	Were study results sensitive to changes in the values (within the assumed range for sensitivity analysis, or within the confidence interval around the ratio of costs to consequences)?	Sensitivity analyses conducted on key variables, favourable still reported.
10.	Did the presentation and discussion of study results include all issues of concern to users?	Partially.
10.1	Were the conclusions of the analysis based on some overall index or ratio of costs to consequences (for example, cost-effectiveness ratio)? If so, was the index interpreted intelligently or in a mechanistic fashion?	No.
10.2	Were the results compared with those of others who have investigated the same question? If so, were allowances made for potential differences in study methodology?	Partially.
10.3	Did the study discuss the generalisability of the results to other settings and patient/client groups?	Partially.
10.4	Did the study allude to, or take account of, other important factors in the choice or decision under consideration (for example, distribution of costs and consequences, or relevant ethical issues)?	No.
10.5	Did the study discuss issues of implementation, such as the feasibility of adopting the 'preferred' programme given existing financial or other constraints, and whether any freed resources could be redeployed to other worthwhile programmes?	No
OVERA	LL ASSESSMENT OF THE STUDY	
How we	II was the study conducted? Code ++, + or -	+
Are the	results of this study directly applicable to the patient group targeted by this guideline?	Unknown, estimates presented were conservative but are based on USA population estimates.
Supplement D



Prevention of sexually transmitted infections (STIs): a review of reviews into the effectiveness of non-clinical interventions

Evidence Briefing Update

Jennifer Downing, Lisa Jones, Penny A. Cook, and Mark A. Bellis

Centre for Public Health Castle House North Street Liverpool John Moores University L3 2AY Tel: 0151 231 4447 Fax: 0151 231 4515 Email: j.downing1@ljmu.ac.uk

-1-. Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

Acknowledgements

We would like to thank all the staff at the Centre for Public Health. Special thanks are also due to Hannah Madden, Dr Harry Sumnall and Suzy Hargreaves.

Content

ACKNOW	LEDGEMENTS	2
CONTEN	٢	3
APPENDI	CES	5
EXECUTI	VE SUMMARY	6
1 INTROE	DUCTION	15
1.1 Backg	jround	15
1.2 Conte	xt: sexual health in England	16
1.2.1	Policy context	16
1.2.2	Epidemiology – prevalence of STIs	17
1.2.3	Sexual behaviour	18
1.2.4	Inequalities	20
1.2.5	Concentral framework	20
1.2.6	Conceptual framework	20
2 METH	HODOLOGY	21
2.1 Ide	ntification of the literature	21
2.1.1	Electronic searches	21
2.1.2	Selection and filtering	22
2.2 C+:	tical appreciaal	22
2.2 Uri	Accessing the quality of the reviewe	८८
2.2.1	Categorisation of reviews	23 23
2.2.2	Calegonsation of reviews	20
2.3 An	alysis and synthesis	24
2.3.1	Évidence statements	24
24 Pre	esentation of data	24
2		
3 THE	RESULTS	26
3.1 Su	mmary of identification of papers	26
011 04		
3.2 Su	mmary of reviews in categories 1-3	27
4 THE	EVIDENCE	29
7 11161		23
4.1 Ind	ividual level interventions	29
4.1.1	Clinic-based sexual health education	29
4.1.2	Partner notification	31
4.1.3	Improved communication between parents and adolescents (parental/family intervention)	31
4.2 Gro	oup-level interventions	33
4.2.1	Clinic-based interventions	33

4.2.2	Linking schools to clinic services and/or providing condoms in schools	36
4.2.3	School curriculum-based sex education	3/
4.2.4	Siliali-group work	39
4.2.3		40
4.3 C	community level interventions	41
4.3.1	Community outreach	41
4.4 S	ocio-political interventions	42
4.4.1	Legislation, policy, equality work, regulation	42
4.4.2	Resource allocation	43
4.5 F	eatures of effective interventions	44
4.5.1	Use of theoretical models in developing interventions	44
4.5.2	Emphasis on risk reduction (e.g. promoting condom use or reduction in number of partners), rather than promotion of abstinence only	45
4.5.3	Use of behavioural skills and social skills training, including self-efficacy	46
4.5.4	Provision of basic, accurate information through clear, unambiguous messages	47
4.5.5	Facilitator training	48
4.5.6	Of appropriate duration, length and intensity	49
4.5.7	Use of peers and community opinion leaders	51
4.5.8	Targeted and tailored (in terms of age, gender, culture, development etc), making use of nee assessment or formative research	eds 52
4.6 N	Iulti-component and multi-level interventions	53
4.7 C	ost-effectiveness of sexual health interventions	54
4.8 lı	nterventions to address inequalities in sexual health	54
5 DIS	CUSSION, CONCLUSIONS AND RESEARCH RECOMMENDATIONS	56
5.1 N	lethodological issues	56
5.1.1	Outcome measures of effectiveness	56
5.1.2	Methodological issues: behavioural indicators	57
5.1.3	Methodological issues: limitations identified by the reviews	58
5.1.4	Methodological issues: limitations and gaps in the evidence	58
5.1.5	Limitations of this review of reviews	59
5.2 C	conclusion and recommendations for research	59
5.2.1	What works in STI prevention interventions	60
5.2.2	What interventions are cost effective: conclusions and research recommendations	69
5.2.3	What works to reduce inequalities in sexual health: conclusions and research	
_	recommendations	69
5.2.4	Implications for policy and practice	70
REED	ENCES	11

Appendices

APPENDIX A. EXAMPLE OF A SEARCH STRATEGY	71
APPENDIX B. CAT SCREENING TOOL	73
APPENDIX C. LIST OF REVIEWS BY CATEGORY 1-5	75
APPENDIX D. IN DEPTH SUMMARIES OF CATEGORY ONE, TWO AND THREE REVIEWS	78
APPENDIX E. DETAILS OF THE PRIMARY STUDIES INCLUDED IN EACH REVIEW	9 8

Executive summary

Introduction

In 2004, the Health Development Agency (HDA; now the National Institute for Health and Clinical Excellence) published *Prevention of sexually transmitted infections (STIs): a review of reviews into the effectiveness of non-clinical interventions* (Ellis and Grey, 2004). This evidence briefing updates the original report, assessing the effectiveness of interventions to impact on the factors which influence the sexual risk behaviours for STI transmission. As in the original Evidence Briefing, the aims of this update were to:

- Identify and synthesise review-level material to highlight 'what works' to prevent or reduce sexual risks and promote sexual health
- Highlight conflicting evidence, gaps in the evidence and provide recommendations for research.

This updated evidence briefing focuses upon particular types and features of interventions in order to draw out findings on the effectiveness of interventions for specific populations to reduce inequalities in sexual health.

Policy

In 2001, the Department of Health published the first *National Strategy for Sexual Health and HIV* (DH, 2001). This set out five main aims, including reducing the transmission of STIs (with a national goal of a 25% reduction in newly gonorrhoea infections by 2007), reducing prevalence of undiagnosed STIs and reducing stigma associated with STIs.

Following on from this, the white paper *Choosing Health: making healthy choices easier* (DH, 2004) included sexual health as one of its six key priorities. This has encouraged a renewed determination to tackle sexual ill-health, and additional resources have been provided in order to do so. In the action plan the Government outlined its plans to improve sexual health services in a new £300m programme over the next three years.

Epidemiology

National data on STIs reveal that chlamydia remained the most common STI diagnosed at GUM clinics in 2005, with the number diagnosed rising by 4.9% (104,733 to 109,958) with 53% (57,577) of those diagnoses among women. Gonorrhoea diagnoses have shown a decrease in the last year, with 19,392 diagnoses in 2005, 13% fewer than 2004. In 2005, 2,814 cases of syphilis were reported, an increase of 123% since 2004 (2,282 cases) and an increase of 273% since 2001 (753). Genital warts are caused by infection with two distinct subtypes of the human papilloma virus (HPV). Many other subtypes are linked to cervical cancer, but because they do not cause symptoms they remain undiagnosed. Diagnoses of genital warts increased by 1.3% between 2004 and 2005 (80,055 to 81,137). In 2005, 50% of diagnoses were in heterosexual men, 47% in women and 3% in men who have sex with men (MSM). The number of cases of herpes simplex virus (HSV) identified in GUM clinics represents only a small proportion of the total number

of cases. In 2005, 19,837 cases were diagnosed in GUM clinics, up 4% on the previous year, with the highest numbers in both men and women in the 25 to 34 year age group (2,838 and 3,628 respectively) (HPA, 2006).

Sexual behaviour

The sexual behaviour of the population is an important determinant of the rates of STIs. The second National Survey of Sexual Attitudes and Lifestyles (NATSAL) in 2000, provides the most recent data on sexual behaviour among the general population in Britain (Johnson et al., 2001; Wellings et al., 2001). Since the first NATSAL in 1990, there have been notable changes in sexual behaviour during the intervening decade, these include:

- A greater number of lifetime partners from 8.6 to 12.7 for men, and from 3.7 to 6.5 for women
- Lower median age at first intercourse from 17 years among 16 to 19 year olds in NATSAL 1990 to 16 years among 16 to 19 years old in NATSAL 2000
- A greater proportion with concurrent partnerships in the past year (two or more partners at the same time) from 11.4% to 14.6% for men (20% for 15 to 24 year olds); and from 5.4% to 9% for women (15% for 15 to 24 year olds)
- A greater proportion with two or more partners in the past year and who did not use condoms consistently from 13.6% to 15.4% in men; and from 7.1% to 10.1% in women, and
- A greater proportion of men reporting ever having had a homosexual partner from 3.6% to 5.4%.

Data on sexual behaviour needs to be interpreted with caution, acknowledging that unprotected sex is not the same as unsafe sex. Similarly, data on the number of partners can provide an indication of potential STI risk, even where intercourse is protected.

Inequalities

The statistics on incidence clearly show how STIs disproportionately affect communities already suffering from considerable inequalities relating to their sexual orientation, ethnicity and gender. Statistics show that Black Caribbean populations continue to be disproportionately affected by gonorrhoea (UK Collaborative Group for HIV and STI Surveillance, 2005), for example, one study (Low et al., 1997) showed that gonorrhoea rates among some inner city black and minority ethnic groups were ten times higher than in white communities.

Cost-effectiveness

Aside from the obvious ethical reasons for averting infections, the economic arguments are substantial. First, there are the costs associated with diagnosing and treating STIs. Second, there are also the health and societal costs of dealing with the consequences of poor sexual health, including pelvic inflammatory disease (which can cause ectopic pregnancies and infertility), HIV, cervical and other genital cancers, hepatitis, chronic liver disease and liver cancer, recurrent genital herpes, bacterial vaginosis and premature delivery, unintended pregnancies and abortions, psychological consequences of sexual coercion and abuse, and poor educational, social and economic opportunities for teenage mothers (DH,

2001). In a recent guide for planning and commissioning services, Payne and O'Brien (2005) concluded that investment in sexual health interventions is good value for money and can also be cost-saving. Examples of cost-saving interventions were listed as: wide-spread condom provision; outreach safe sex training for high risk groups; school education programmes; needle exchange services; screening programmes; high quality rapid access to STI services; wide choice of contraceptive services; and abortion services provided with minimal delay.

Conceptual framework

This review draws on the same conceptual framework as the original review of reviews. For full details of the framework see Ellis and Grey (2004).

Methodology

A standardised methodology was developed by the HDA (now the National Institute for Health and Clinical Excellence) for the analysis and synthesis of review-level evidence (Swann et al., 2003b, Kelly et al., 2002). The strategies used in this update were based on an updated 2003 HDA search strategy.

The following process was applied:

All databases were searched from January 2001 to January 2006 for references published in the English language. Results were downloaded into a Reference Manager database. Two reviewers independently appraised all of the titles and abstracts of the identified references to determine whether to retrieve the full paper on the basis of the following criteria:

- English language only
- Published since 2001
- Systematic review, synthesis, meta-analysis or literature review
- Relevance to STI prevention or sexual health promotion
- Presents (and synthesises) data from primary evaluation studies of intervention effectiveness.

This evidence briefing update excludes:

- Screening and treatment of STIs
- Interventions that focus mainly on the prevention of teenage pregnancy or HIV
- Reviews covered in the original evidence briefing.

A joint decision was made as to whether the full paper would be retrieved for critical appraisal; if the reviewers disagreed, or no clear decision could be made on the basis of the title or abstract, the full paper was obtained for appraisal. Two reviewers appraised the identified papers (including journal articles, book chapters and reports) independently. A joint decision was made regarding whether the paper was to be classified as Category 1, 2, 3, 4 or 5 based upon how well the paper satisfied the criteria of the appraisal tool.

In total 10 papers were judged to be Category 1, 2 or 3 and went on to analysis and synthesis. This compares to 14 Category 1, 2 or 3 papers from Ellis and Gray, 2004. Category 1 and 2 papers were used to derive evidence statements.

The categorisation process is only briefly described here, for further information see Ellis and Grey (2004). A joint decision was made regarding whether the paper was to be classified as Category 1, 2, 3, 4 or 5. The 'data pool' for the review of reviews consists of those reviews categorised as 1, 2 or 3. The categories are:

Category 1

The review satisfied the criteria in both stages one and two of the CAT. Category 1 reviews are included within the data pool and analysed to derive evidence statements.

Category 2

The review passed stage one of the CAT, but failed to meet all the criteria within stage two – usually because it was not clear how the review had assessed the quality of the included studies. Category 2 reviews are also included within the data pool and analysed to derive evidence statements.

Category 3

The research question(s) was judged to be highly pertinent, and the paper is a review of interventions; however it failed to pass stage one of the CAT – it is not clear how the included studies were identified and the review had additional shortcomings. Category 3 reviews are included within the data pool and so may provide some evidence about effectiveness, but these reviews alone are insufficient to inform conclusions about effectiveness.

Category 4

This paper is either not a review of effectiveness of interventions, or it is not a review in its own right (e.g. it may extract findings from another review). However it is relevant and contains useful policy, background, epidemiological or interpretive information.

Category 5

The paper is not directly relevant and is therefore not used.

Category 1, 2 and 3 reviews are used to derive 'evidence statements' about types and features of interventions. Where relevant review evidence has been considered in conjunction with that of the original Evidence Briefing (Ellis and Grey, 2004) and evidence statements were derived based upon the combined evidence. The evidence statements reflect the strength of the conclusions made by the review(s), the outcomes used to judge effectiveness, the category of the review, and any inconsistencies within and between reviews. The reviews are used to derive the following evidence statements:

- **Sufficient review-level evidence** clear evidence/conclusions from at least one Category 1 review, with no conflicting evidence/conclusions between Category 1 reviews
- Tentative review-level evidence tentative evidence/conclusions from Category 1 review; or conflicting conclusions from Category 1 reviews; or clear conclusions from at least one Category 2 review
- Insufficient review-level evidence no evidence/conclusions from Category 1 reviews and only tentative evidence/conclusions from Category 2 reviews; or clear evidence/conclusions from Category 3 reviews
- No review-level evidence no evidence/conclusions from Category 1, 2 or 3 reviews.

Findings and discussion and conclusions

The **Evidence** section (4) considers the evidence by individual, group, community and socio-political level interventions. In addition, we consider the 'features of effectiveness'; cost-effectiveness; and, evidence of interventions to reduce inequalities in sexual health.

The **Findings**, discussion and conclusions section (5) address the key questions posed by the review of reviews, namely:

What works to prevent STI transmission? What works to reduce the risk behaviours for STI transmission? What works to address the determinants of STI risk?

We found that most reviews analysed only intermediate health outcomes (e.g. behaviour) as opposed to health promotion outcomes (e.g. knowledge or skills), as such there was insufficient evidence available with which to make conclusions. However, we present the features of effectiveness elicited from the evidence. Some evidence statements have not changed from the original review (stated as 'No change from Ellis and Grey, 2004') and all evidence statements ought to be considered in conjunction with those of the previous review (Ellis and Grey, 2004).

Features of effective interventions

There is **sufficient review-level evidence** to conclude that interventions are more likely to be effective if they include the following features:

- The use of theoretical models (No change from Ellis and Grey, 2004)
- Use of behavioural skills training, including self-efficacy (No change from Ellis and Grey, 2004)
- Provision of basic, accurate information through clear, unambiguous messages (No change from Ellis and Grey, 2004)
- Use of targeted and tailored interventions (in terms of age, gender, culture, etc.), making use of needs assessment or formative research (No change from Ellis and Grey, 2004).

There is **tentative review-level evidence** to conclude that interventions are more likely to be effective if they include the following features:

- Have an emphasis on risk reduction (No change from Ellis and Grey, 2004)
- Use a trained facilitator
- Extend the delivery of their intervention (e.g. using 'booster' sessions).
- Use peers and community opinion leaders (No change from Ellis and Grey, 2004).

Clinic-based sexual health promotion

There is **tentative review-level evidence** to conclude that individual risk counselling can be effective.

Partner notification

There is **sufficient review-level evidence** to conclude that partner notification is an effective means of detecting new infections (No change from Ellis and Grey, 2004).

Improved communication between parents and adolescents

We conclude that there is **insufficient review-level evidence** to support or discount the effectiveness of parent/adolescent communication (No change from Ellis and Grey, 2004).

Clinic-based interventions

There is insufficient review-level evidence to conclude that:

• Clinic-based interventions aimed at adolescents are effective at reducing the sexual risk behaviour of adolescents.

There is tentative review-level evidence to conclude that:

• Clinic-based interventions using behavioural skills are an effective way to reduce the sexual risk behaviour of clinic attendees (No change from Ellis and Grey, 2004).

There is **insufficient review-level evidence** to support or discount the effectiveness of parental inclusion in clinic-based interventions.

School-based programmes/sex education

There is sufficient review-level evidence to conclude that:

• School-based sex education can be effective in reducing adolescents' sexual risk behaviour (No change from Ellis and Grey, 2004).

There is tentative review-level evidence to conclude that:

 Sex education is more effective if begun before the onset of sexual activity (No change from Ellis and Grey, 2004).

There is insufficient review-level evidence to conclude that:

- School-based interventions linked to clinical services and/or to the provision of condoms in schools are more effective (No change from Ellis and Grey, 2004)
- Condom availability in schools is not associated with increased sexual activity
- Parent involvement in school-based interventions is effective at improving parent/adolescent communication thereby reducing adolescent risk-taking behaviour.

Small group work

There is **sufficient review-level evidence** to conclude that small-group work interventions can be effective in reducing sexual risk behaviour (No change from Ellis and Grey, 2004).

Detached education and outreach by professionals

There is **tentative review-level evidence** to conclude that detached education and outreach work by professionals is effective in reducing sexual risk behaviour.

Community level interventions

There is **insufficient review-level evidence** to conclude that community outreach is effective at reducing sexual risk-taking behaviour.

Socio-political interventions

There is **insufficient review-level evidence** to support or discount the effectiveness of legislation or policy.

There is **no review-level evidence** to reach a conclusion about the effectiveness of equality work, regulation, facilitation, resource allocation, or organisation and delivery of services.

Are multi-component and multi-level interventions more likely to be effective in influencing sexual risk behaviours?

There is **sufficient review-level evidence** to conclude that the most successful interventions are multicomponent interventions. There is **insufficient review-level evidence** to support or discount the effectiveness of multi-level interventions (No change from Ellis and Grey, 2004).

What works to reduce inequalities in sexual health?

There is **no review-level evidence** on interventions that aim to reduce inequalities in sexual health. In particular, there was **no review-level evidence** relative to socio-economic status (No change from Ellis and Grey, 2004).

What interventions are cost-effective?

This update found **no new review level evidence** to add to the findings in the previous Evidence Briefing that reported **tentative review-level evidence** (see Ellis and Grey, 2004 for further information).

Implications for policy and practice

Limitations of the evidence (section 5.1) leave us to conclude that we are unable to make recommendations for policy and practice. It is very important to note that '**no evidence' of 'insufficient evidence' does not equal evidence of ineffectiveness**. It simply means that we are unable to support or discount the effectiveness of interventions from the review-level evidence provided. We have, however, made recommendations for research. We also emphasise that practitioners and policy makers should not consider this review of review alone, but in conjunction with other the original Evidence Briefing (Ellis and Grey, 2004), non-review evidence and other relevant source information.

Research recommendations

The **Conclusions and research recommendations** (section 5.2) highlights the research needed specific to target groups and types of interventions. The following is a summary of general research recommendations. Some recommendations have not changed from the original Evidence Briefing (stated as 'No change from Ellis and Grey (2004)' followed by their recommendation) and all recommendations ought to be considered in conjunction with those of the previous review (Ellis and Grey, 2004).

- No change from Ellis and Grey (2004). A key research recommendation is that intervention evaluations should measure outcomes relating to the personal and structural determinants of risk (e.g. knowledge, attitudes, skills, behavioural intentions, access to condoms, peer norms). This should be done in addition to the measurement of intermediate health outcomes (e.g. changes in behaviour) and, where appropriate (e.g. for large-scale multi-component interventions or programmes), health outcomes (e.g. changes in STI incidence). In turn, reviews should cease to exclude studies that only include data on health promotion outcomes alongside their effects on intermediate health outcomes and health outcomes. Reviews should fully report all of these outcomes where possible.
- No change from Ellis and Grey (2004). It is imperative that data on intermediate health outcomes (i.e. sexual behaviour surveys) are more specific about the context in which certain behaviours take place. Primary studies and future reviews should therefore focus on more meaningful risk indicators when making judgements about the relative effectiveness of interventions in influencing so-called 'risky' behaviours. Where feasible and appropriate these should be correlated with biological behavioural indicators and other socio-demographic variables. In conclusion, we need a consensus on the appropriate indicators for assessing the effectiveness of STI prevention programmes with different target populations.
- There is a need for more rigorous evaluations of UK-based STI prevention interventions. It is
 necessary for researchers to clearly report the methodology that they use, including their methods
 for randomisation. Future research should include appropriate comparison groups and should
 consider longer follow-up periods. In these cases research should report: levels of attrition, intent-

to-treat analysis, and effect sizes. There is also a need for primary studies to be standardised in their reporting.

In addition to the recommendations from Ellis and Grey (2004), we recommend that further
research is needed on the cost-effectiveness of STI prevention interventions in different settings.
There is also a need for further research aimed at vulnerable 'high-risk' groups in the UK such as
commercial sex workers (CSWs) and looked-after children. Finally, there is a need for further
research to explore the effectiveness of interventions that aim to address inequalities and sociopolitical issues specific to target populations.

1 Introduction

1.1 Background

In 2004, the Health Development Agency (HDA) published Prevention of sexually transmitted infections (STIs): a review of reviews into the effectiveness of non-clinical interventions (Ellis and Grey. 2004). Since then, sexual health has continued to be at the forefront of policies, guidelines and initiatives, and these have identified the HDA's Evidence Briefing as the resource for ascertaining what works in preventing STIs. In 2004, the Department of Health (DH) produced its plan to improve the health of the population Choosing health: making healthy choices easier. In this key national policy document, sexual health is named as one of six key priorities, and the HDA evidence briefings identified as the source of information on effective interventions. In 2005, DH published the action plan Delivering choosing health which sets key targets around STIs, rolling out chlamydia screening and access to genitourinary medicine (GUM) clinics within 48 hours (DH, 2005). In 2005, the HDA joined with the National Institute for Clinical Excellence (NICE) to form the new National Institute of Health and Clinical Excellence (and retained the acronym NICE). The new organisation continues the commitment to build on the evidence base in public health. In the light of the Government's continued commitment to sexual health and HIV, and the necessity to ground STI prevention in good evidence, it was deemed necessary to update the 2004 Evidence Briefing to ensure that the most recent review-level evidence is available for planning services and improving the sexual health of the population.

NICE evidence briefings report on **reviews of reviews**, sometimes referred to as tertiary level research. They consist of detailed expositions of the strengths and weaknesses of the evidence from reviews, identification of gaps in the evidence, an analysis of future primary and secondary research needs, and a discussion of the implications of the evidence for policy and practice. The full rationale for carrying out reviews of reviews is given in the 2004 Evidence Briefing on STI prevention, as is a detailed consideration of the limitations of such a methodology. One such limitation is that of the inevitable time-lag between publication of primary research material and review level material. To mitigate this somewhat, Ellis and Grey (2004) stated an intention to update the Evidence Briefing regularly to keep the evidence base as up to date as reasonably practicable. This document is the first such update, and should be used in conjunction with the Ellis and Grey (2004)'s original Evidence Briefing. Each document has a freestanding summary that is published separately. The documents are also published on and supported by the NICE website at www.publichealth.nice.org.uk. This website also contains electronic copies of, or means of access to, the original reviews upon which the evidence briefings draw (if they are in the public domain).

Reviews that focus primarily on teenage pregnancy or HIV prevention, while clearly relevant, are excluded on the basis that their findings are considered in detail by the HDA/NICE evidence briefings on HIV prevention (Ellis et al., 2003, and updated in 2006 by Downing et al.) and teenage pregnancy (Swann et al., 2003a). Professionals are encouraged to refer to these documents (see www.publichealth.nice.org.uk), since the outcomes for many of the included reviews are similar to those for STI prevention (e.g. age at first intercourse, number of partners, condom use). Many of the topic areas covered have a reference group (e.g. for HIV, teenage pregnancy) comprising key academics, practitioners and officials with relevant expertise. All of the NICE evidence briefings are externally peer reviewed.

1.2 Context: sexual health in England

1.2.1 Policy context

In 2001, the Department of Health published the first *National Strategy for Sexual Health and HIV* (DH, 2001b). This set out five main aims, including reducing transmission of STIs (with a national goal of a 25% reduction in newly gonorrhoea infections by 2007), reducing prevalence of undiagnosed STIs and reducing stigma associated with STIs.

As part of its commitments in the *Implementation Action Plan* (DH, 2002), the DH published two sexual health and HIV toolkits, one on commissioning (DH, 2003a) and one on health promotion (DH, 2003b). The sexual health promotion toolkit gives practical advice on making inter-agency work effective, running health promotion projects, managing group work, developing resources and managing outreach work. The document also emphasises the need for a lot more research, and the need for health professionals to evaluate and document their health promotion interventions, in terms of intermediate outcomes (e.g. knowledge, attitude) as well as behavioural outcomes (DH, 2003b).

The white paper Choosing Health: making healthy choices easier (DH, 2004) identified six key priorities:

- Tackling health inequalities
- Reducing the numbers of people who smoke
- Tackling obesity
- Improving sexual health
- Improving mental health and wellbeing, and
- Reducing harm and encouraging sensible drinking.

The spotlight on sexual health in this key health policy document has encouraged a renewed determination to tackle sexual ill-health, and provides additional resources in order to do so. In the action plan the Government outlined its plans to improve sexual health services in a new £300m programme over the next three years. The main aims are to:

- Use £50m to implement a national and regional sexual health campaign aimed at young men and women to promote condom use and explain the risks of unprotected sex
- Make a commitment to sexual health in England through additional funding to deliver multidisciplinary sexual health services in a range of settings
- Expand the national Chlamydia screening programme to cover the whole of England by March 2007 (with an additional £80m to help achieve this goal)
- Carry out an audit of contraceptive services in 2005 in order to improve service provision

- Focus on modernizing Genito-Urinary Medicine (GUM) clinics, with an investment of £130m over three years including upgraded prevention services with an additional £40m provided for this purpose
- Ensure that every individual referred to a GUM clinic has an appointment within 48 hours by 2008.

1.2.2 Epidemiology – prevalence of STIs

The surveillance of sexually transmitted infections relies on GUM clinicians reporting aggregated numbers of selected conditions by sex and age, and for some conditions, whether cases in men are heterosexual or from sex between men. These are submitted quarterly on KC60 forms. While there are limitations of the system (principally that only diagnoses made in GUM clinics are included, there are limited demographic data and no geographical marker), it is a good source of data on trends. More detailed information is sought from additional surveys, such as The Gonococcal Resistance to Antimicrobials Surveillance Programme (GRASP) survey to investigate resistance to gonorrhoea treatment and enhanced surveillance of outbreaks of previously rare infections such as syphilis and lymphogranuloma venereum (LGV).

Chlamydia remained the most common STI diagnosed at GUM clinics in 2005, with the number diagnosed rising by 4.9% (104,733 to 109,958) with 53% (57,577) of those diagnoses among women (HPA, 2006). In 2005, the highest rates of chlamydia in women were among young women aged 16 to 19 (1,359/100,000) while the men were slightly older, with the highest rate in men aged 20 to 24 years (1,070/100,000) (UK Collaborative Group for HIV and STI Surveillance, 2006). Factors that may have contributed to increases more recently include increased awareness of chlamydia and the roll out of chlamydia screening programmes. However, the rates calculated from GUM diagnoses represent only the tip of the iceberg when considering the prevalence of chlamydia in the general population. A study based on the first phase of the screening programme indicates that 10% of women and 13% of men aged under 25 years have chlamydia (LaMontagne et al 2004).

Gonorrhoea diagnoses have shown a decrease in the last year, with 19,392 diagnoses in 2005, 13% fewer than 2004. However, there was a 9.8% increase among MSM (3,994 to 4,388) (HPA, 2006). As was the case with chlamydia, the highest rates in 2005 were in young women between the ages of 16 and 19 years and men between the ages of 20 to 24 years. The GRASP 2005 survey shows that rates of gonorrhoea are higher in urban populations, among MSM, young women and some black ethnic minorities (UK Collaborative Group for HIV and STI Surveillance, 2006).

In 2005, 2,814 cases of syphilis were reported, an increase of 123% since 2004 (2,282 cases) and an increase of 273% since 2001 (753). In 2005, only 14.8% (418) of cases were in women, while among men infection among MSM account for 60% of all cases (HPA, 2006). Unlike chlamydia and gonorrhoea, it is not the younger age groups that experience the higher levels of syphilis. The highest prevalence is recorded in 2005 was among men aged 25 to 34 years (19/100,000) (UK Collaborative Group for HIV and STI Surveillance, 2006). Enhanced surveillance has revealed much detail about these localised outbreaks of a previously rare infection. In MSM, syphilis has been associated with oral sex and large numbers of

anonymous sex partners (Bellis et al., 2002), while heterosexual outbreaks have been associated with commercial sex work and exposure abroad (Simms et al., 2005).

Genital warts are caused by infection with two distinct subtypes of the human papilloma virus (HPV). Many other subtypes are linked to cervical cancer, but because they do not cause symptoms they remain undiagnosed. There are currently two new vaccines under consideration, both of which aim to protect against two of the common subtypes of HPV that cause cervical cancer and one that also includes protection against the two subtypes that cause warts (Steinbrook, 2006). Diagnoses of genital warts increased by 3.1% between 2005 and 2006 (81,201 to 83,745). In 2005, 50% of diagnoses were in heterosexual men, 47% in women and 3% in men who have sex with men (MSM) (HPA, 2006). In common with chlamydia and gonorrhoea, diagnosed men tend to be older than women with the highest male prevalence in those 20 to 24 years compare to females aged 16 to 19 years and 20 to 24 years (UK Collaborative Group for HIV and STI Surveillance, 2006). When considering all types of HPV infection, both those that cause warts and those associated with cancer, the prevalence is thought to be very high. An Edinburgh study showed that the prevalence of all types of HPV was 42% in women aged under 25 years and 20% of all women (Cuschieri et al., 2004). HPV infection is also linked to anal cancer, and its prevalence may be very high in men who have sex with men (57% of MSM in a USA study: Chin Hong et al., 2004).

Genital herpes is caused by the herpes simplex virus (HSV). Aside from HIV, genital herpes is the only incurable STI, often causing recurrent debilitating disease in those infected. Genital herpes is thought to be caused by one of two subtypes of the virus, HSV-1 and HSV-2. HSV-1 used to be more frequently associated with oral infection. The antibodies to the oral infection offer protection from the genital infection. However, the proportion of genital HSV infections attributed to HSV-1 is increasing, possibly because childhood exposure to HSV-1 is decreasing. Another possible link is to the reported increasing levels of oral sex among young people (Johnson et al., 2001). The number of cases of herpes simplex virus (HSV) identified in GUM clinics represents only a small proportion of the total number of cases. In 2005, 19,837 cases were diagnosed in GUM clinics, up 4% on the previous year, with the highest numbers in both men and women in the 25 to 34 year age group (2,838 and 3,628 respectively) (HPA, 2006).

1.2.3 Sexual behaviour

The sexual behaviour of the population is an important determinant of the rates of STIs. The second National Survey of Sexual Attitudes and Lifestyles (NATSAL) in 2000, provides the most recent data on sexual behaviour among the general population in Britain (Johnson et al., 2001; Wellings et al., 2001). Since the first NATSAL in 1990, there have been notable changes in sexual behaviour during the intervening decade:

- A greater number of lifetime partners from 8.6 to 12.7 for men; and from 3.7 to 6.5 for women
- A lower median age at first intercourse from 17 years among 16 to 19 year olds in NATSAL 1990 to 16 years among 16 to 19 years old in NATSAL 2000

- A greater proportion with concurrent partnerships in the past year (two or more partners at the same time) from 11.4% to 14.6% for men (20% for 15 to 24 year olds); and from 5.4% to 9% for women (15% for 15 to 24 year olds)
- A greater proportion with two or more partners in the past year and who did not use condoms consistently from 13.6% to 15.4% in men; and from 7.1% to10.1% in women
- A greater proportion of men reporting ever having had a homosexual partner from 3.6% to 5.4%.

However, comparison of successive age groups in the NATSAL 2000 survey showed that condom use has continued to increase in the last decade. The proportion of men and women reporting condom use at first intercourse increased from 60% of 25 to 29 year olds (who would have become sexually active in 1990 or thereabouts) to 80% of 16 to 19 year olds (who would have become sexually active around the year 2000).

The main UK wide source of information about gay men's behaviour is the annual Sigma Research surveys. Periodically, the survey asks whether respondents have had unprotected anal intercourse: the last three surveys to include this were 1995 and 2000 (Hickson et al., 2001) and 2004 (Weatherburn et al., 2005). MSM in 2000 reported increases in unprotected anal intercourse (UAI) with both regular and casual partners compared with 1995:

- A greater proportion have UAI with a regular partner from 42.3% in 1995 to 54.9% in 2000
- Fewer men having anal intercourse with a regular partner in 2000 always used a condom compared with men in 1995
- A greater proportion of men have casual UAI from 9.8% of all men in 1995 to 14.4% in 2000.

The 2004 survey (Weatherburn et al., 2005) did not repeat these questions. The 2004 findings on selfreported behaviour showed that the more sexual partners a man reported the more likely he was to have had an HIV or STI test within the last year (22% of men who had only one partner had an HIV test, and 24% an STI test, compared to 43% of men who had 15-29 partners having an HIV test and 54% were tested for STIs). Men who reported more sexual partners expressed a wish for more sexual education than men who had fewer sexual partners. However, a worryingly high number of men who had many different sexual partners were not tested; 55% of men who had sex with 30 or more partners in the last 12 months had not had an HIV test and 39% not had an STI test in the last year. The survey also revealed that 10% had paid or been paid for sex in the last year, with over 40% of men involved with commercial sex work citing gay websites as their most likely method of finding partners.

The 2005 survey (Hickson et al., 2006) also revealed that in the past year three quarters of all men questioned had engaged in sex with a man of unknown serostatus. If these findings are representative of the whole MSM population they indicate a widespread lack of communication around HIV status among MSM. The survey also revealed that 11% of men who had anal intercourse in the last year never used a condom and inconsistent condom use was as common as consistent condom use. Furthermore, 19% of

men whose last HIV test was negative or who had never been tested for HIV, had participated in receptive unprotected anal intercourse with a partner of unknown status in the last year. This figure rose to 32% of untested men with thirty or more partners.

Data on sexual behaviour needs to be interpreted with caution, acknowledging that unprotected sex is not the same as unsafe sex. Similarly, data on the number of partners can provide an indication of potential STI risk, even where intercourse is protected.

1.2.4 Inequalities

The statistics on incidence clearly show how STIs disproportionately affect communities already suffering from considerable inequalities relating to their sexual orientation, ethnicity and gender. Statistics show that Black Caribbean populations continue to be disproportionately affected by gonorrhoea (UK Collaborative Group for HIV and STI Surveillance, 2005), for example, one study (Low et al., 1997) showed that gonorrhoea rates among some inner city black and minority ethnic groups were ten times higher than in white communities.

1.2.5 Costs

Aside from the obvious ethical reasons for averting infections, the economic arguments are substantial. First, there are the costs associated with diagnosing and treating STIs. Second, there are also the health and societal costs of dealing with the consequences of poor sexual health, including pelvic inflammatory disease (which can cause ectopic pregnancies and infertility), HIV, cervical and other genital cancers, hepatitis, chronic liver disease and liver cancer, recurrent genital herpes, bacterial vaginosis and premature delivery, unintended pregnancies and abortions, psychological consequences of sexual coercion and abuse, and poor educational, social and economic opportunities for teenage mothers (DH, 2001). In a recent guide for planning and commissioning services, Payne and O'Brien (2005) conclude that investment in sexual health interventions is good value for money and can also be cost-saving. Examples of cost-saving interventions were listed as: wide-spread condom provision; outreach safe sex training for high risk groups; school education programmes; needle exchange services; screening programmes; high quality rapid access to STI services; wide choice of contraceptive services; and abortion services provided with minimal delay.

1.2.6 Conceptual framework

This review draws on the same conceptual framework as the original Evidence Briefing. For full details of the framework see Ellis and Grey (2004).

2 Methodology

A standardised methodology was developed by the HDA (now the National Institute for Health and Clinical Excellence) for the analysis and synthesis of review-level evidence (Swann et al., 2003b, Kelly et al., 2002). The specific methodology used in the review of reviews is outlined below.

2.1 Identification of the literature

2.1.1 Electronic searches

A wide-ranging set of search strategies was developed for the original Evidence Briefing by Ellis and Grey, (2004) in an attempt to identify as many types of 'review' relevant to the subject as possible.

Briefly, the topic strategy was developed to identify papers with a focus on any intervention with the goal of reducing the sexual risk of transmitting or acquiring an STI. The topic search was developed using general terms for STIs (e.g. STD, chlamydia, gonorrhoea). For each database, strategies were developed with a combination of Medical Subject Headings ('MESH') (using various 'operators' such as 'and', 'or', 'adjacent' and 'near') and text words (appropriately truncated, e.g. 'STI\$') in order to achieve as many relevant reviews as possible.

The strategies used in this update were based on an updated 2003 HDA search strategy, which included papers to December 2002. The Medline search strategy used is provided as an example in Appendix A.

The following databases and websites were searched:

- The Cochrane Library (including HTA database, DARE [Database of Abstracts of Reviews of Effects] and NHS EED [Economic Evaluation Database])
- The 'Wider Public Health' report
- MEDLINE
- TRIP (database with access to largest collection of 'evidence-based' material on the web)
- SIGN (Scottish Intercollegiate Guidelines Network)
- Health Evidence Bulletins Wales
- National Guidelines Clearing House
- NCCHTA (National Co-ordinating Centre for Health Technology Assessment) website
- NICE (National Institute for Clinical Excellence) website
- REFER (Research Findings Electronic Register)
- National Research Register
- Clinical Evidence
- EMBASE
- Sociological Abstracts
- PsycINFO
- Cinahl (Cumulative Index to Nursing and Allied Health Literature)

- Sociofile
- EPPI-Centre's *** Register of Reviews of Effectiveness (RoRE).***

All databases were searched from January 2001 to January 2006 for references published in the English language. Results were downloaded into a Reference Manager database.

2.1.2 Selection and filtering

Two reviewers independently appraised all of the titles and abstracts of the identified references to determine whether to retrieve the full paper on the basis of the following criteria:

- English language only
- Published since 2001
- Systematic review, synthesis, meta-analysis or literature review
- Relevance to STI prevention or sexual health promotion in the UK
- Presents (and synthesises) data from primary evaluation studies of intervention effectiveness.

This evidence briefing update excludes:

- Screening and treatment of STIs
- Interventions that focus mainly on the prevention of teenage pregnancy or HIV
- Reviews covered in the original evidence briefing.

A joint decision was made as to whether the full paper would be retrieved for critical appraisal; if the reviewers disagreed, or no clear decision could be made on the basis of the title or abstract, the full paper was obtained for appraisal.

2.2 Critical appraisal

Two reviewers appraised the identified papers (including journal articles, book chapters and reports) independently. The appraisal process sought to identify the extent to which papers were:

- *Systematic* do the review authors apply a consistent and comprehensive approach? If repeated, would this give the same results?
- *Transparent* are the review authors clear about the processes involved, the individual study results, the findings and conclusions drawn from these? Is there sufficient detail to repeat?
- *Analytically sound* do the authors undertake the appropriate methods of analysis and are they undertaken correctly?
- Relevant is the content relevant to the UK and the priority populations under consideration?

2.2.1 Assessing the quality of the reviews

The HDA's standard protocol (Swann et al., 2003; Kelly et al., 2002) and critical appraisal tool (CAT) was used to make judgements about the quality of the identified papers. See Appendix B for further details. For a further discussion about the assessment of quality please see the previous Evidence Briefing (Ellis and Grey, 2004).

2.2.2 Categorisation of reviews

The categorisation process is only briefly described here, for further information see Ellis and Grey (2004).

A joint decision was made regarding whether the paper was to be classified as Category 1, 2, 3, 4 or 5. The 'data pool' for the review of reviews consists of those reviews categorised as 1, 2 or 3. The categories are:

Category 1

The review satisfied the criteria in both stages one and two of the CAT. Category 1 reviews are included within the data pool and analysed to derive evidence statements.

Category 2

The review passed stage one of the CAT, but failed to meet all the criteria within stage two – usually because it was not clear how the review had assessed the quality of the included studies. Category 2 reviews are also included within the data pool and analysed to derive evidence statements.

Category 3

The research question(s) was judged to be highly pertinent, and the paper is a review of interventions; however it failed to pass stage one of the CAT – it is not clear how the included studies were identified and the review had additional shortcomings. Category 3 reviews are included within the data pool and so may provide some evidence about effectiveness, but these reviews alone are insufficient to inform conclusions about effectiveness.

Category 4

This paper is either not a review of effectiveness of interventions, or it is not a review in its own right (e.g. it may extract findings from another review). However it is relevant and contains useful policy, background, epidemiological or interpretive information.

Category 5

The paper is not directly relevant and is therefore not used.

2.3 Analysis and synthesis

Reviews categorised as 1, 2 or 3 were subject to analysis and synthesis. Detailed structured summaries were completed for each Category 1, 2 and 3 reviews (see Appendix D).

2.3.1 Evidence statements

Category 1, 2 and 3 reviews are used to derive 'evidence statements' about types and features of interventions. Where relevant review evidence has been considered in conjunction with that of the original Evidence Briefing (Ellis and Grey, 2004) and evidence statements were derived based upon the combined evidence. The evidence statements reflect the strength of the conclusions made by the review(s), the outcomes used to judge effectiveness, the category of the review, and any inconsistencies within and between reviews. The reviews are used to derive the following evidence statements:

- **Sufficient review-level evidence** clear evidence/conclusions from at least one Category 1 review, with no conflicting evidence/conclusions between Category 1 reviews
- Tentative review-level evidence tentative evidence/conclusions from Category 1 review; or conflicting conclusions from Category 1 reviews; or clear conclusions from at least one Category 2 review
- Insufficient review-level evidence no evidence/conclusions from Category 1 reviews and only tentative evidence/conclusions from Category 2 reviews; or clear evidence/conclusions from Category 3 reviews
- No review-level evidence no evidence/conclusions from Category 1, 2 or 3 reviews.

Where there was no change from the evidence statement in the original Evidence Briefing this was stated in the text and followed by the original evidence statement from Ellis and Grey (2004). Evidence statements remained the same where there was evidence from the same category or lower than in the original Evidence Briefing and/or where all current evidence agreed with the previous evidence.

Evidence statements were amended from the original Evidence Briefing (Ellis and Grey, 2004) where current evidence originated from a higher category. Evidence statements also changed where there was evidence from the same category or higher where the authors' conclusions were more definitive than those in the original Evidence Briefing¹.

2.4 Presentation of data

Complete bibliographic lists of the Category 1-5 papers are presented in the **References** (section 6).

The **Results** (section 3) contains a flow chart of the paper identification process and includes a summary table of the assessment of each Category 1-3 review according to the critical appraisal process.

¹ On one occasion (Cost-effectiveness section 4.7) the evidence statement was amended because there was no current cost-effectiveness evidence found to continue to maintain the original evidence statement in relation to current sexual health interventions.

^{-24-.} Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

Appendix C contains lists of all of the reviews by category 1-5.

Appendix D contains detailed summaries of Category 1-3 reviews. **Appendix E** contains the primary studies from Category 1 and 2 reviews.

The findings of Category 1, 2 and 3 reviews are analysed and synthesised in **section 4** (**The Evidence**). There are sections on different 'types' of individual, group, community and socio-political interventions, as well as a section on 'features of effective interventions'; other sections consider the evidence for cost effectiveness and interventions to reduce inequalities in sexual health.

Within each section, having weighed up the evidence a number of 'evidence statements' are made, based on the findings from the Category 1, 2 and 3 review papers (see section 3.2).

The **Discussion, conclusions and research recommendations (section 5)** considers methodological issues, gaps, inconsistencies and limitations of the review of reviews drawn out from the evidence statements in section 4, particularly in relation to the UK and in the context of the those reported in the previous Evidence Briefing.

3 The Results

3.1 Summary of identification of papers

A total of 2,319 titles and abstracts were screened for inclusion. Compared to 2,430 in Ellis and Grey, 2004. Of these, 99 papers were ordered as full papers and appraised using the CAT. In total, 10 papers were judged to be Category 1, 2 or 3 and went on to analysis and synthesis. A summary of the process of review identification is shown in Figure 1. Table 1 summarises the Category 1-3 review-level evidence.



Figure 1. Flowchart showing the process of identification and categorisation of papers

3.2 Summary of reviews in categories 1-3

Table 1: Category 1-3 review-level evidence

	Stage one						Stage two							
	1		2		2d	3			3d	4		5	-	
		Comprehensive search strategy			Assessment of quality of included studies				Quality of analysis					
Author and date*	Clear aim/ question inc. Setting population, intervention	2a Identify approp. range of source databases	2b Additional search strategies	2c ISpecifies search terms	specified	3a Definition of quality	3b Tool used to assess quality	3c How assessments generated	Assessment of the studies' quality	4a Study findings clear and consistent	4b Study findings analysed appropr- iately	Conclusions related to findings	CATEGORY	
DiClemente R.J. et al., (2005)	√	1	x	\checkmark	\checkmark	1	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	1	1	
Ward D.J. et al., (2005)***	√	1	(in part)	\checkmark	\checkmark	1	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	1	1	
Ward D.J. et al., (2004)	√	1	(in part)	\checkmark	\checkmark	1	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	1	1	
Manhart L.E. et al., (2005)	\checkmark	~	(in part)	\checkmark	\checkmark	1	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	1	1	
Robin L. et al. (2004)	~	1	(in part)	x	\checkmark	~	\checkmark	x	\checkmark	\checkmark	\checkmark	\checkmark	2	
Pedlow C.T and Carey M.P., (2004)	√	1	(in part)	x	\checkmark	1	\checkmark	x	\checkmark	\checkmark	\checkmark	1	2	
Shrier L. A., (2004)	1	x	x	x	x	-	-	-	-	-	-	-	3	
Schaalma H.P., et al., (2004)	√	x	x	x	x	-	-	-	-	-	-	-	3	
Schmiedl R. (2004)	~	x	x	×	x	-	-	-	-	-	-	-	3	
DiClemente R.J. et al., (2002)	1	x	x	x	x	-	-	-	-	-	-	-	3	

* Complete bibliographic details of Category 1, 2 and 3 review papers are provided in Appendix C. Detailed summaries of Category 1-3 reviews can be found in Appendix D.

** Additional search strategies involve follow-up of references/journals, consultation with experts in the field and grey literature searches.

*** Ward D. J. et al., (2005) is a journal article containing many of the same primary studies and findings from Ward D. J. et al., (2004).

4 The evidence

We follow each section with evidence statements regarding the effectiveness of interventions. Where there is no change from the evidence statement in the original Evidence Briefing this is stated in the text and followed by the original evidence statement from Ellis and Grey (2004). This Evidence Briefing also contains evidence statements on topics not previously included in the original Evidence Briefing (Ellis and Grey, 2004). Further, statements were amended from the original evidence briefing where current evidence was deemed to be of a higher quality. The quality of evidence was judged by the category from which it originated.

Detailed summaries of Category 1-3 reviews can be found in Appendix D. Details of the primary studies for each Category 1 and 2 reviews can be found in Tables 2-7 in Appendix E.

4.1 Individual level interventions

We found a total of eight Category 1, 2 and 3 reviews that included individual-level interventions. This compares with the eight Category 1, 2 and 3 reviews in the original evidence briefing (Ellis and Grey, 2004). The most frequently reported individual level intervention was clinic-based risk reduction interventions.

Category 1: DiClemente et al., 2005; Manhardt and Holmes, 2005; Ward et al., 2004; Ward et al., 2005.
Category 2: Robin et al., 2004; Pedlow and Carey, 2004.
Category 3: DiClemente et al., 2002; Shrier, 2004.

4.1.1 Clinic-based sexual health education

We found a total of six reviews that referred to primary studies covering sexual health education in a clinic setting. These findings add to the five Category 1 and 3 reviews identified in the original Evidence Briefing (Ellis and Grey, 2004) concerning individual counselling.

Outcomes

The most frequently reported outcomes were condom use, sexual activity and incidence of STIs.

Category 1: DiClemente et al., 2005; Manhardt and Holmes, 2005; Ward et al., 2004; Ward et al., 2005.Category 2: Robin et al., 2004; Pedlow and Carey, 2004.Category 3: No evidence.

Seven Category 1 and 2 reviews examined the evidence of effectiveness of clinic-based sexual health education (DiClemente et al., 2005; Manhardt and Holmes, 2005; Ward et al., 2004; Ward et al., 2005; Robin et al., 2004; Pedlow. and Carey, 2004). The main outcomes considered were condom use, sexual

activity and STI incidence. Sexual health knowledge and self-efficacy of condom use were also considered.

DiClemente and colleagues (2005) reviewed five individual-level clinic-based STI prevention interventions (Boekeloo et al., 1999; DeLameter et al., 2000; Mansfield et al., 1993; Scholes et al., 2003 and Schrier et al., 2001) (some overlap with Ward et al., 2004, 2005 and Pedlow and Carey, 2004). Interventions primarily included risk-assessment, condom attitudes and skills, STI education, culturally tailored information, and information on drug and needle use. One intervention (Scholes et al., 2003) involved sending out postal information. None of the interventions showed significant effects for all their outcomes however, the individual interventions that showed at least one significant impact in measurable outcomes (e.g. consistent condom use) in the intervention group compared to the control group incorporated risk-assessments and used tailored educational materials. Overall (including the results for group clinic interventions), there were significant results for condom-protected sex at six and 12 months follow up.

Manhardt and Holmes (2005) (some overlap with Ward et al., 2004, 2005 and Robin et al., 2004) included five additional individual-level interventions targeted at people attending STI clinics (Boyer et al., 1997; Kamb et al., 1998; Peterman et al., 2000; VCT Efficacy study group, 2000; EXPLORE Study Team, 2004). Intervention methods included cognitive-behavioural intervention, individual risk-reduction counselling and voluntary counselling and testing (VCT). All interventions examined STI incidence as the primary outcome. Findings of the VCT interventions showed that although a reduction in the incidence of STIs and HIV acquisition was seen, the effects were non-significant. However, results were more positive for counselling intervention based on the theory of reasoned action and social cognitive theory at reducing the incidence of STIs over a 12-month period, when compared to the efficacy of didactic messages" (S15). The findings in these studies add to those of the previous Evidence Briefing (Ellis and Grey, 2004), which found that there was tentative review-level evidence to conclude that individual risk counselling can be effective.

Ward and colleagues (2005) (some overlap with Ward et al., 2005, DiClemente et al., 2005, Manhardt and Holmes, 2005 and Robin et al., 2004) examined nine clinic-based interventions (see Table 5, Appendix E). Interventions primarily included counselling and skills-building techniques. Four out of the nine trials considering the outcome of laboratory confirmed STIs found greater reductions in their intervention groups relative to their controls, which was statistically significant (p<0.05) for two studies. However, pooled results did not indicate an overall effect (RR 1.00 (95% CI, 0.81 to 1.23)). Of the four studies considering clinically diagnosed STIs one found no reduction (Branson et al., 1998) whilst the remaining studies showed that participants in the intervention group had higher STI rates than the control group following the intervention. Two trials considering self-reported STIs also found no significant effects (Shrier et al., 2001;Metzler et al., 2000). Six out of seven trials reporting consistent condom use observed an increase in the intervention groups compared to the controls (Kamb et al., 1998; NIMH, 1998; Imrie et al., 2001; Kalichman et al., 1999; Shrier et al., 2001; Boyer et al., 1997) and pooled results suggested an overall

effect (RR 1.17 (1.10 to 1.25)). Also, five out of seven trials considering number of sexual partners found fewer sexual partners among their intervention groups compared to their control groups.

In addition to one study already mentioned (Boyer, 1997), Robin and colleagues (2004) examined one other clinic-based intervention that included risk-assessment, counselling on condom use and prevention, educational material and the offer of free condoms (Mansfield, 1993). However, analysis showed no effects on any behavioural outcome and the authors make no further comment about it.

Evidence statement: No change from Ellis and Grey (2004). There is tentative reviewlevel evidence to conclude that individual risk counselling can be effective in reducing sexual risk behaviour among clinic attendees.

4.1.2 Partner notification

We found one Category 1 review to add to the six Category 1, 2 and 3 reviews relevant to partner notification in the originals Evidence Briefing (Ellis and Grey, 2004).

Category 1: Manhardt and Holmes, 2005. Category 2: No evidence Category 3: No evidence

The previous Evidence Briefing (Ellis and Grey, 2004) found evidence to support the effectiveness of partner notification. In this update, Manhart and Holmes (2005), reported on one study relevant to partner notification (Peterman et al., 1997) that randomly assigned people diagnosed with syphilis to different intervention approaches. The methods used were: partner notification by patients within two days; immediate notification by disease-intervention specialist; or immediate notification by disease-intervention specialist; or immediate notification by disease-intervention specialist with optional blood draw in the field. Findings showed similar success for all three approaches in terms of locating and treating partners.

Evidence statement: No change from Ellis and Grey (2004). There is sufficient reviewlevel evidence to conclude that notification can be an effective means of newly detecting infections

4.1.3 Improved communication between parents and adolescents (parental/family intervention)

A total of four reviews were identified that discussed studies relevant to parental monitoring. These findings add to those previously reported from three Category 3 reviews in the original evidence briefing (Ellis and Grey, 2004).

Outcomes

The main outcomes examined were unprotected sex, pregnancy and sexual debut.

Category 1: No evidenceCategory 2: Robin et al., 2004; Pedlow and Carey, 2004.Category 3: DiClemente et al., 2002; Shrier, 2004.

Robin et al. (2004) reported on one individual-level intervention by Miller et al. (1993). This intervention involved adolescents and their families viewing six videos and supplementary printed material over a three-month period. However, this intervention showed null results, Robin et al. (2004) made no additional comments about this intervention.

Pedlow and Carey (2004) included three school-based studies that aimed to promote parent/adolescent communication (further details provided in section 4.2.5 (ii)). Pedlow and Carey (2004) noted that "Parents continue to be important, and parent-adolescent discussions about sex can protect teens from other influences that might encourage risky sex. Thus, encouraging parent-teen communication and assisting parents in monitoring teens' behaviour may be especially important for the prevention of HIV/STDs as well as unintended pregnancy" (p174).

Shrier (2004) and Diclemente et al. (2002) also highlighted the importance of parental monitoring. DiClemente et al. (2002) reported findings from several studies that involved parental monitoring and family support as a prevention intervention. Li et al. (2000a, 2000b; cited in DiClemente et al., 2002) examined parental monitoring to prevent unprotected sex. DiClemente at al. (2002) stated that "parental monitoring had a direct and stronger protective effect against adolescents' engagement in unprotected vaginal sex at the 2-, 3-, and 4-year follow-up" (p.173). DiClemente et al. (2001; cited in DiClemente et al., 2002) found that adolescents with perceived low parental monitoring were significantly more likely to participate in unprotected sex; have multiple sexual partners; and have partners who were not monogamous compared to others with perceived high parental monitoring. Romer et al. (1994) found that high levels of parental monitoring were related to lower odds of initiating sex before 11 years of age and greater likelihood of delayed sexual initiation in female adolescents have been reported by Metzler et al, (1994); Luster et al. (1997); Benda and DiBlasio, (1994); and Crosby et al. (2002). One study showed that behavioural interventions resulted in significant improvements in parental monitoring (Stanton et al., 2000).

Ellis and Grey (2004) found Category 3 evidence for greater communication between parents and adolescents. Our findings add to those from the original Evidence Briefing, however, we conclude that there is insufficient evidence to change the previous evidence statement.

Evidence statement: No change from Ellis and Grey (2004). There is insufficient reviewlevel evidence to support or discount the effectiveness of improved communication between parents and adolescents (parental/family involvement) at reducing the sexual risk behaviour of adolescents.

4.2 Group-level interventions

We found a total of nine Category 1, 2, and 3 reviews that included group level interventions. These add to those previously reported from 17 Category 1, 2, and 3 reviews in the original Evidence Briefing (Ellis and Grey, 2004). The most frequently reported group-level interventions were clinic-based.

Category 1: DiClemente et al., 2005; Manhart and Holmes, 2005; Ward et al., 2004; and Ward et al., 2005.

Category 2: Robin et al., 2004; Pedlow and Carey, 2004 Category 3: DiClemente et al., 2002; Schmiedl, 2004; Schaalma et al., 2004

4.2.1 Clinic-based interventions

We found a total of seven reviews that reported findings on clinic-based interventions.

Outcomes

The most frequently reported outcome was condom use.

(i) Clinic-based interventions aimed at adolescents

A total of five reviews reported findings on clinic-based interventions aimed at adolescents. The original Evidence Briefing did not contain any reviews that explored clinic-based interventions aimed at adolescents.

Category 1: DiClemente et al., 2005; Ward et al., 2004; Ward et al., 2005 Category 2: Robin et al., 2004; Pedlow and Carey, 2004 Category 3: No evidence

DiClemente et al. (2004) reported on five clinic or health service-based interventions (DiClemente et al., 2004; Gillmore et al., 1997; Metzler et al., 2000; Orr et al., 1996; Lawrence et al., 1995) all of which included adolescent populations. The interventions used group sessions that focused primarily upon safe sex, attitudes towards condoms, various skills training and assertion. Across the five interventions, outcomes were mixed. In three studies, participants receiving the intervention reported higher levels of consistent condom use at six- (DiClemente et al., 2004; Orr et al., 1996; St. Lawrence et al., 1995) and 12-month follow-up (DiClemente et al., 2004; St. Lawrence et al., 1995). One study (Metzler et al., 2000) found that the intervention group demonstrated a significant reduction in sexual partners at six-month follow-up. Gilmore et al. (1997) reported no significant effect of the intervention on condom use, number of sexual partners or refusing sex without a condom.

Ward et al. (2004) (the same studies as DiClemente et al., 2004) reported on two group-level clinic-based STI interventions targeted at adolescents (Metzler et al., 2000; Orr et al., 1996; also reported in Ward et al., 2005). Metzler et al. (2000) incorporated role-play and skills training into their interventions. Findings

showed that there was a significant decrease in the number of sexual partners who were strangers, in addition to significant decreases in the mean number of sexual partners and the mean number of non-monogamous sexual partners. Orr et al. (1996) also examined an intervention that included role-play, condom negotiation and skills training. Findings from this study showed that there was a significant increase in consistent condom use in both the intervention and control group. Ward et al. (2004) made no specific comment on clinic-based interventions specific to adolescents.

Pedlow and Carey (2004) (overlap with Ward et al., 2004; 2005; DiClemente et al., 2004) reported on two clinic-based interventions targeting adolescents that have not previously been mentioned² (DeLamater et al., 2000; Mansfield et al., 1993). These interventions included decision-making skills, booster sessions, condom skills, negotiation skills, and one also measured risk perception. However, the authors made no specific comment on the effectiveness of these interventions relative to their setting.

Robin et al. (2004) described one group-level health centre-based intervention aimed at adolescent African Americans (St. Lawrence et al., 1995; also reported in DiClemente et al., 2004). This randomised controlled trial involved small-group discussion modelling skills, skills practice, external speakers and roleplay. The intervention showed positive results on rate of sexual activity, condom use, and sexual initiation. However Robin et al. (2004) categorised this study as 'mixed' due to the fact that it showed null results when carried out on males in a state reformatory (St. Lawrence, 1999).

New evidence statement: There is insuffient review-level evidence to conclude that clinic-based interventions aimed at adolescents are effective at reducing the sexual risk behaviour of adolescents.

(ii) Clinic-based interventions aimed at the general clinic attendeesA total of three reviews reported findings on interventions aimed at general adult clinic attendees.

Category 1: Manhart and Holmes, 2005; Ward et al., 2004; Ward et al., 2005 Category 2: No evidence Category 3: No evidence

Manhart and Holmes (2005) reported details of six clinic-based interventions (seven studies) (Branson et al., 1998; NIMH, 1998; Shain et al., 1999; Shain et al., 2002; Hobfall et al., 2002; Baker et al., 2003; Cohen et al., 1991). Two interventions targeted both men and women and the remaining four interventions targeted women only. All six interventions involved small-group work including counselling, attitudes, skills and behaviour change. All but two of the studies (Branson et al., 1998; Hobfall et al., 2002) reported reductions in STI incidence. Cohen et al. (1991) compared group discussion on condoms in waiting rooms

² Pedlow and Carey, 2004 also included Orr et al., 1996 and Metzler et al., 2000.

^{-34-.} Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

(intervention group) to no group discussion (control), finding a non-significant 50% reduction in repeat clinic attendances with a subsequent STI for the intervention group.

Ward et al. (2004) reported on ten group-level clinic-based STI interventions, five of which have not previously been discussed and were aimed at general STI clinic attendees (Balmer et al., 1998; Imrie et al., 2001; Kalichman et al., 1999; O'Leary et al., 1996; Solomon et al., 1989). Ward et al. (2004) stated that, "This review suggests that behavioural interventions may be effective in increasing the proportion of GUM clinic patients reporting consistent condom use, though there were inconsistent effects on STI rates and other measures of sexual behaviour (e.g. number of sexual partners)" (p.53). Ward et al. (2005) also found "evidence that behavioural interventions increase consistent condom use, though effects on other aspects of behaviour were inconsistent" (p. 390).

New evidence statement: There is tentative review-level evidence to conclude that clinic-based interventions using behavioural skills are an effective way to reduce the sexual risk behaviour of clinic attendees.

(iii) Parental involvement in clinic-based interventions

A total of two reviews referred to parental involvement in clinic-based interventions. The original Evidence Briefing did not contain any reviews that explored parental involvement in clinic-based interventions.

Category 1: DiClemente et al., 2004 Category 2: No evidence Category 3: DiClemente et al., 2002

Although they provided no evidence of effectiveness DiClemente et al. (2004) highlighted the importance of involving parents as behaviour-change agents through improved parent-teen communication. They also maintained that clinicians who see young people are in a good position to provide parents with the information they need to facilitate parent-adolescent discussions.

DiClemente et al. (2002) also suggested that clinic-level interventions involving parental counselling may enable physicians to emphasise the importance of parental monitoring (p.175).

New evidence statement: There is insufficient review-level evidence to either support or discount the effectiveness of parental inclusion in clinic-based interventions at reducing the sexual risk behaviour of students.
4.2.2 Linking schools to clinic services and/or providing condoms in schools

One review we identified reported on five primary studies relevant to school-based STI prevention interventions. These reviews add to the five relevant Category 1, 2 and 3 reviews previously reported in the original Evidence Briefing (Ellis and Grey, 2004).

Outcome

The most frequently reported outcome was condom use.

Category 1: No evidence Category 2: No evidence Category 3: Schmiedl, 2004

Schmeidl (2004) reported on five school-based condom distribution interventions (Schuster et al., 1998; Wolk and Rosenbaum, 1995; Guttmacher et al., 1997; Kirby et al., 1999; Blake et al., 2003). Findings from Schuster et al. (1998) showed that there was no increase in the level of sexual activity at one-year followup. However, condom use increased among males, and there was an increase in the percentage of male students who had used condoms at recent first sex. Wolk and Rosenbaum (1995) found that condom distribution in schools was three times more beneficial to a sexually active student than the risk of encouraging a non-sexually active young person to become sexually active. Guttmacher et al. (1997) reported increased condom use in students with three or more partners in the last six months in a school with condom provision compared to a school without. Kirby et al. (1999) found that condom provision did not increase sexual activity in students. Blake et al. (2003) found that students from schools with condom provision were less likely to report sexual intercourse, more likely to receive condom use education, and where they did report sexual activity, were more likely to use condoms but were less likely to use any other method of contraception. Overall, findings showed that some young people received condoms but did not use them. In addition, there was no difference in pregnancy rates between intervention schools and schools without condom distribution schemes. None of the studies examined the effects of condom distribution on rates of STIs.

These findings add to those in the first Evidence Briefing and although they do not measure changes in STI rates they do show evidence of increased condom use. However, they original only from Category 3 papers therefore, there is insufficient evidence.

Evidence statement: No change from Ellis and Grey (2004). There is insufficient reviewlevel evidence to conclude that condom distribution in schools can be effective at reducing the sexual risk behaviour of students.

(i) Condom availability in schools and association with increased sexual activity

One review identified studies relevant to condom availability in schools. The original Evidence Briefing did not include evidence relating to an association between condom availability in schools and sexual activity.

Category 1: No evidence Category 2: No evidence Category 3: Schmiedl, 2004

Schmiedl (2004) stated that, "school-based condom availability programs were not associated with an increase in teen sexual activity, whereas data from most programs indicated increased use of condoms by teens who are sexually active, these programs should be seriously considered when developing school-based programs focused in preventing pregnancy, STDs, and HIV" (p.19).

New evidence statement: There is insufficient review-level evidence to support or discount the view that condom distribution in schools is not associated with increased sexual activity.

4.2.3 School curriculum-based sex education

We found a total of three reviews that reported on 52 relevant primary studies. These findings add to those previously reported by the relevant 14 Category 1, 2 and 3 reviews in the original Evidence Briefing (Ellis and Grey, 2004).

Outcomes

The most frequently reported outcomes were delay of sexual initiation, condom use, contraceptive use and frequency of sexual intercourse.

Category 1: No evidence Category 2: Robin et al., 2004; Pedlow and Carey, 2004 Category 3: Schaalma et al., 2004

Robin et al. (2004) reported on 12 primary studies that addressed school-based sex education (see Table 6, Appendix E). Eight studies reported positive intervention effects, two showed mixed effects and two, aimed at older adolescents, showed negative effects.

Pedlow and Carey (2004) reported on 13 school-based interventions (see Table 7, Appendix E). Findings showed that programmes targeting pre-adolescent youth (aged 9-12 years) were effective at delaying the onset of sexual intercourse and improving condom use among sexually active youth. With young adolescents (aged 13-15 years) interventions were effective at increasing condom use, reducing unprotected sex, and reducing the frequency of sexual intercourse.

Schaalma et al. (2004) reported on five school-based interventions (Abraham et al., 2002; Schaalma et al., 1996; Schaalma and Kok, 2001; Wight and Abraham, 2000; Paulussen et al., 1994). Schaalma et al.

(2004) noted that these interventions include aspects such as, discussion groups, role-play, rehearsal and modelling, which are methods that have proven to be effective at preventing risk behaviours.

The findings from the previous Evidence Briefing concluded that after-school based interventions, and the inclusion of skills building techniques such as role-play were effective at changing sexual behaviour.

Evidence statement: No change from Ellis and Grey (2004). There is sufficient reviewlevel evidence to conclude that school-based sex education can be effective in reducing the sexual risk behaviours of young people.

(i) Start sex education early, before the onset of sexual activity

We found one review to add to the tow Category 1 and 2 reviews in the original Evidence Briefing (Ellis and Grey, 2004).

Category 1: No evidence Category 2: No evidence Category 3: Shrier, 2004

Shrier (2004) concluded that the earlier young people receive sex education, the less likely they are to participate in risky sexual behaviour.

These comments agree with those provided by two reviews in the original evidence briefing (Ellis and Grey, 2004).

Evidence statement: No change from Ellis and Grey (2004). There is tentative reviewlevel evidence to conclude that sex education is more effective if begun before the onset of sexual activity.

(ii) Parents and school-based interventions

Two Category 2 and 3 reviews (Pedlow and Carey, 2004; DiClemente et al., 2002) were identified that reported on five primary studies relevant to parental inclusion in school-based interventions. The original Evidence Briefing did not include evidence of parent and school-based interventions.

Outcomes

The most frequently reported outcomes were improved parent-adolescent communications, multiple partners, and pregnancy.

Category 1: No evidence Category 2: Pedlow and Carey, 2004 Category 3: DiClemente et al., 2002 Pedlow and Carey (2004) reported details of three studies that included parental involvement (Coyle et al., 2001; Levy et al., 1995; Aarons et al., 2000). The interventions examined incorporated aspects such as completing homework assignments with parents, providing material to parents to encourage communication, meetings and newsletters for parents. Two studies found no change in parent and adolescent communication after the intervention, and one study (Coyle et al., 2001) found improved adolescent-parent communication at almost three years follow-up.

DiClemente et al. (2002) reported on one intervention aimed at reducing the sexual risk behaviour of young people in schools through parental inclusion in school-based intervention. Hawkins et al. (1991) examined a youth development programme that provided training to both young people and their parents in a school setting. This intervention decreased the odds of young people participating in risky sexual behaviour, such as multiple partners and pregnancy, during adolescence.

New evidence statement: There is insufficient review-level evidence to conclude that parental involvement in school-based interventions is effective at reducing the sexual risk behaviour of young people.

4.2.4 Small-group work

Six reviews were identified that reported on small group work. These findings add to the seven Category 1 and 3 reviews previously reported in the original evidence briefing (Ellis and Grey, 2004).

Outcomes

The most frequently reported outcome was condom use.

Category 1: DiClemente et al., 2004; Manhart and Holmes, 2004; Ward et al., 2004; Ward et al., 2005.Category 2: Robin et al., 2004; Pedlow and Carey, 2004Category 3: No evidence

DiClemente et al. (2004) reported on successful primary studies (DiClement et al., 2004; Metzler et al., 2000; Orr et al., 1996; St Lawrence et al., 1995) showing significant effects on outcomes including, for example, consistent condoms use in the intervention group compared to the control group. These studies incorporated clinic-based small-group work, and highlighted the importance of role-play and skills building.

Manhart and Holmes (2005) reported details of five small-group RCTs with passive follow-up (e.g. review of clinic records as opposed to actively testing individuals) (O'Donnell et al., 1998; Cohen et al., 1992a; Cohen et al., 1992b; Gollub et al., 2000; Imrie et al., 2001). Findings showed mixed results with one study (Imrie et al., 2001) demonstrating an increase in STI incidence in the intervention group.

Pedlow and Carey also included studies focused at pre- and young adolescents (9-12 years and 13-15 years) that included small group work. They found evidence of successful group-level interventions addressing cognitive factors, risk perception and skills development. Pedlow and Carey (2004) classified the design of the intervention by Kirby et al. (1997) as being a developmentally appropriate study for this target group as it included decision-making skills and preparation for risky situations.

Robin et al. (2004) reported details of 21 primary studies that included small group work with young people from communities, clinics, schools, detention facilities and a state reformatory (see Table 6, Appendix E). Robin et al. (2004) reported details of three primary studies that were undertaken with participants from schools and clinics as well as family planning clinics and community based organisations (Jemmott et al., 1992³; Kirby et al., 1997; Eisen et al., 1990). These interventions included, among other aspects, role-play, discussion, and videos. Findings from Jemmott et al. (1992) favoured the intervention group (who received a career planning intervention) for risky sexual behaviour index, frequency of sex, number of partners, occasions of sex without a condom and having had anal sex. Robin et al. (2004) deemed the studies by Kirby et al. (1997) and Eisen et al. (1990) to have negative effects due inconsistencies in the randomisation process. Robin and colleagues noted "the most common strategies for content delivery were small group discussion, role-playing and interactive and experiential exercises, use of media and interactive media, use of structured games, and lecture" (p.18).

Manhart and Holmes (2005) stated that, "Interventions delivered in small-group settings were as effective as those delivered to individuals" (S20) and other reviews have also found that group interventions although effective, are no more effective than individual-level interventions. Ward et al. (2004) carried out a pooled analysis and found no difference in effects between group interventions and those interventions using either an individual format or those with one or two sessions. Ward et al. (2005) reported that, "intervention format and length was not associated with trial results [group based relative risk (RR) 0.94 (0.70 to 1.25) versus individual based RR 1.16 (0.76 to 1.75)]" (p.390).

These findings add to the evidence of effectiveness reported in the previous evidence briefing (Ellis and Grey, 2004).

Evidence statement: No change from Ellis and Grey (2004). There is sufficient reviewlevel evidence to conclude that small group work involving skills-building activities can be effective at reducing the sexual risk behaviour of all target groups.

4.2.5 Detached education and outreach by health professionals

Two reviews were identified that reported details of education and outreach-based interventions employing group-level education and condom-distribution. The original Evidence Briefing did not include evidence of group level detached education and outreach by health professionals.

³ Also reported by Pedlow and Carey (2004).

^{-40-.} Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

Outcomes The most frequently reported outcome was STI incidence.

Category 1: Manhart and Holmes, 2005 Category 2: Robin et al., 2004 Category 3: No evidence

Condom distribution was a characteristic of three studies included by Manhart and Holmes (2005) (Feldblum et al., 2001; Fontanet et al., 1998; Celentano et al., 1998; also reported in Speizer et al., 2003). Feldblum and colleagues examined an intervention that included the provision of information, education and communication (IEC) as well as female condoms to female plantation workers in Kenya. The intervention had no impact on STI incidence. Fontanet et al. (1998) provided male and female condoms to brothel workers in Thailand. The intervention resulted in a non-significant decrease in STI incidence compared to brothels that only received male condoms.

Robin and colleagues (2004) reported on three community-based education programmes, all with positive outcomes on risky sexual behaviour (e.g. condom use) (Jemmott et al., 1998; Stanton et al, 1996; Stanton et al., 1997). Interventions included, among other aspects, small-group discussions, role-play, videos, and skills-building activities. However, these interventions were aimed at African American young people and their generalisability to the UK population of young people may be limited.

One additional group-level intervention reported by Manhart and Holmes (2005) addressed STI prevention for male Thai army conscripts using condom distribution at brothels targeted at men with an STI history. Findings showed significant, desirable effects on rates of STIs with greater than 80% reduction in STI recurrence compared to that in the control groups. Findings also showed a reduction in HIV incidence, risk behaviour, brothel visits and an improvement in consistent condom use.

New evidence statement: There is tentative review-level evidence to conclude that detached education and outreach by health professionals, including targeted condom distribution, is effective at reducing the sexual risk behaviour of high risk groups.

4.3 Community level interventions

Only one review identified reported on relevant community level interventions. This adds to the seven Category 1, 2 and 3 reviews included in the original Evidence Briefing (Ellis and Grey, 2004).

4.3.1 Community outreach

One review reported details of two community-level mixed modality interventions. The original Evidence Briefing did not include evidence relating to community outreach.

Outcomes The reported outcome was STI incidence.

Category 1: Manhart and Holmes, 2005. Category 2: No evidence Category 3: No evidence

Manhart and Holmes (2005) reported on two community-level interventions (Hayes et al., 2003; Kamali et al., 2003). Both interventions were mixed modality and included behavioural and treatment strategies. The study by Kamali et al. (2003) showed that communities receiving only information, education and communication (IEC) experienced a decrease in HSV-2 seroconversion compared to the communities receiving IEC and management of STI symptoms, which experienced decreased incidence of syphilis and gonorrhoea. However, all groups, including the control group, reported an increase in condom use. Hayes et al. (2003) carried out a four-component intervention that included school health and reproductive education, reproductive youth services, condom distribution and community activities. However, no intervention effects were found for STI or HIV incidence even though improvements in knowledge and behaviour were reported.

Evidence statement: There is insufficient review-level evidence to support or discount the effectiveness of community level outreach interventions on sexual risk behaviour.

4.4 Socio-political interventions

Three reviews were identified that addressed socio-political interventions. They add to the four Category 3 reviews reported in the original evidence briefing (Ellis and Grey, 2004).

Outcomes

The most frequently reported outcome was condom use.

4.4.1 Legislation, policy, equality work, regulation

We found three reviews to add to one Category 3 review regarding Legislation, policy, equality work and regulation.

Category 1: Manhart and Holmes, 2005 Category 2: No evidence Category 3: DiClemente et al., 2002; Schaalma et al., 2004.

Manhart and Holmes (2005) included one study (Egger et al., 2000) that reported on an intervention in Nicaragua where a health ordinance required motels to provide condoms to guests when requested.

Motels frequented by CSWs were randomly assigned to either (1) offering condoms when requested at check-in, (2) always offering condoms at check-in, or (3) placing condoms on each bed before check-in. Findings showed that both offering condoms at check-in and placing condoms on the bed resulted in significant increases in condom use.

DiClemente et al. (2002) discussed the benefits of parental involvement in school-based programmes (see section 4.2.3 (ii)). DiClemente and colleagues concluded that, "community policies that involve parents in the sex education programs offered to children and adolescents in schools may also be beneficial" (p.176).

In relation to sex education in schools, Schaalma et al. (2004) commented on the policies in The Netherlands that state that teenagers in school should be taught about condom use, contraceptive use and social skills relative to sexual health. Findings have shown that the theory and evidence based interventions implemented in The Netherlands have had favourable effects. However, Schaalma and colleagues also noted the negative effects policies can have on sex education in schools. For example, abstinence based sex education in US schools has shown little or no evidence of effectiveness yet continues to be implemented, and national guidelines in Tanzania discourage the promotion of condom and contraceptive information. Schaalma et al. (2003) argue that when policy or legislation constrains evidence-based practice the PRECEDE/PROCEED or Intervention Mapping approaches could be applied to politicise health promotion and mobilise community members to support and aid the establishment of health promotion policies.

Changed evidence statement: There is insufficient review-level evidence to support or discount the effectiveness of legislation, policy, equality work or regulation.

The previous evidence statement reported no review-level evidence to support or discount the effectiveness of legislation, policy, equality work or regulation.

4.4.2 Resource allocation

Four reviews were identified that referred to the issue of resource allocation. They add to the two Category 3 reviews included in the original Evidence Briefing (Ellis and Grey, 2004).

Category 1: Ward et al., 2004.Category 2: Robin et al., 2004.Category 3: DiClemente et al., 2002; Schaalma et al., 2004.

Although the reviews considered in this evidence briefing presented no evidence of effective models of resource allocation they have considered resource allocation when making conclusions regarding the effectiveness of interventions and suggestions for future research.

Ward et al. (2004) stated, "one of the studies to show success in reducing STI rates found no advantage of an enhanced four-week intervention over and above two 20-minute individually based counselling sessions. However, even if such an approach was shown to be effective in a UK setting, currently it could not be delivered as part of routine clinic care without increased resources, clinic time, and trained staff. In addition, the potential impact of such an intervention on STI rates is not clear. This will depend on the uptake of such an intervention were it offered, and crucially on the proportion of STIs thought to occur in those who have previously attended GUM clinics" (p.55).

DiClemente et al. (2002) discussed the importance of parental monitoring and suggested that parental counselling by clinicians may be effective as an STI intervention aimed at young people. However, they highlight that further resources would be needed to develop and undertake this intervention.

With regard to resource allocation for school-based interventions Schaalma et al. (2004) stated that, "high quality teacher training is required for high quality sexual health promotion in classrooms and this needs to be acknowledged in budget planning" (p.264). In light of the evidence that suggested interventions carried out over longer periods of time are more effective Robin et al. (2004) noted, "the effect of duration and intensity on interventions results in a particular challenge for school-based educators because they may have limited time to implement programs within a school year" (p.23). They also noted, "community-based organisations also need sufficient resources to implement programs in multiple sessions" (p.23).

Evidence statement: No change from Ellis and Grey (2004). There is no review-level evidence to support or discount the effectiveness of difference models of resource allocation.

4.5 Features of effective interventions

4.5.1 Use of theoretical models in developing interventions

Findings from five Category 1, 2 and 3 reviews add to those reported in 13 reviews from the original evidence briefing (Ellis and Grey, 2004) that concluded that many successful interventions were underpinned by theories of behaviour change.

Category 1: Manhart and Holmes, 2005; Ward et al., 2004; Ward et al., 2005.Category 2: Pedlow and Carey, 2004Category 3: Schaalma et al., 2004

With regard to theory-based interventions, Ward et al. (2004; 2005) concluded that interventions using social cognitive or related theories were related to intervention effectiveness, although pooled results were

not statistically significant [social cognition models RR 0.91 (0.71 to 1.16) versus other trials RR 1.19 (0.84 to 1.68)].

In relation to theory-based interventions, Manhart and Holmes (2005) concluded that "among the behavioural interventions reviewed, those that showed no effect were theory based as often as those that demonstrated benefit, and no underlying behavioural theory or approach (e.g., skills building, counselling, and motivational interviewing) was more often successful than another" (S20).

Pedlow and Carey (2004) concluded that the majority of theories applied to interventions aimed at young people were developed primarily for adults and do not focus upon the developmental determinants of sexual risk (p.181).

Schaalma et al. (2004) reported details of several studies that found evidence of effectiveness in interventions underpinned by cognitive behavioural theories. Schaalma et al. (2004) referred specifically to one study by Jemmott and Jemmott (2000) that examined 36 theoretically based interventions that were effective at changing behaviour to see if they were more effective on condom use. Findings showed that those studies with effects on mediators of behaviour change such as knowledge and intentions, also had greater effects on condom use and abstinence.

Evidence statement: No change from Ellis and Grey (2004). There is sufficient reviewlevel evidence to conclude that theory-based interventions are more likely to be effective.

4.5.2 Emphasis on risk reduction (e.g. promoting condom use or reduction in number of partners), rather than promotion of abstinence only

Findings from three Category 1, 2 and 3 reviews contribute to the evidence from eight reviews reported in the original evidence briefing (Ellis and Grey, 2004).

Category 1: DiClemente et al., 2004. Category 2: Robin et al., 2004 Category 3: Schaalma et al., 2004

DiClemente et al. (2004) noted that adolescents should understand the value of abstinence in accordance with recommendations (US). They also stated that adolescents should be informed about the use of condoms and contraceptives.

Although Robin et al. (2004) made no specific conclusions regarding risk reduction compared to abstinence only programmes they did highlight that more effective programmes were usually of a longer

duration and focused on developing knowledge and life skills, compared to shorter programmes that focused specifically on abstinence and condom-use skills.

From the evidence reviewed on abstinence-based interventions, Schaalma et al. (2004) concluded, "programs with an exclusive abstinence message generally have not succeeded in the promotion of abstinent sexual behaviour, although some have succeeded in changing attitudes towards abstinence" (p.265).

Evidence statement: No change from Ellis and Grey (2004). There is tentative reviewlevel evidence to conclude that interventions that promote risk reduction, rather than abstinence alone, are more likely to be effective.

Evidence statement: No change from Ellis and Grey (2004). There is insufficient reviewlevel evidence to support or discount the effectiveness of school-based abstinence-only approaches.

4.5.3 Use of behavioural skills and social skills training, including self-efficacy

Nine Category 1, 2 and 3 reviews were found to add to the 15 Category 1, 2 and 3 reviews in the original Evidence Briefing.

Category 1: DiClemente et al., 2004; Manhart and Holmes et al., 2005; Ward et al., 2004; Ward et al., 2005

Category 2: Pedlow and Carey, 2004; Robin et al., 2004

Category 3: DiClemente et al., 2002; Schaalma et al., 2004; Shrier, 2004

Ward et al. (2004; 2005) commented that trials of behavioural interventions showed a reduction in STI rates and was associated with increased consistent condom use. Many of the interventions included training such as condom use, training in how to deal with high-risk situations, and decision-making skills.

DiClemente et al. (2004) concluded that characteristics of effective interventions include technical skills, such as condom-use skills, and active learning skills, through the use of role-play, as well as self-efficacy through communication and assertiveness skills (p.215).

Manhart and Holmes (2005) included studies that used behavioural skills and although skills-building was listed as a feature of successful interventions, Manhart and Holmes (2005) concluded that no theory or approach (e.g. behavioural skills) was more often successful than another (S18).

With regard to interventions aimed at adolescents, Pedlow and Carey (2004) concluded that skills-training is effective (p.174).

Robin et al. (2004) found that typically, programmes that were effective at reducing sexual risk behaviours included skills-building activities such as communication skills, problem-solving, changing peer-norms and decision-making. However, Robin et al. (2004) also found that programmes more generally targeted towards increasing adolescent resiliency and competencies have also shown promising results.

DiClemente et al. (2002) suggested that clinic and community-based interventions should be further utilised in order to promote behavioural skill-based interventions through physician counselling and social/friendship/peer networks.

Although Shrier (2004) provided no evidence of effectiveness he/she maintained that the incorporation of behavioural skills such as partner notification, sexual negotiation and risk perception into prevention interventions can increase self-efficacy.

Schaalma et al. (2004) highlighted studies that call for behavioural-skills and social-skills development, including what they term "life skills" such as decision-making, communication and negotiation (p.262). Through these skills young people will learn how they would react in social situations that could lead to risky behaviour. They report on two successful programmes (three studies) that contain relevant aspects (Abraham et al., 2002; Schaalma et al., 1996; Schaalma and Kok, 2001). Schaalma et al. (2004) highlighted the relationship between self-efficacy and condom use and also identified four cognitions that have medium to large correlations with condom use, they are: (1) attitudes toward condoms (r = .32); (2) descriptive norms in relation to condom use (i.e., perceptions that others approve of and use condoms; r = .37; (3) intentions to use condoms (r = .43); and (4) pregnancy motivation (e.g. condoms should be used for contraceptive purposes, as well as STI protection; r = .37) (p.261). Schaalma et al. (2004) referred to a study by Jemmott and Jemmott (2000) that examined 36 theoretically based interventions that were effective at changing behaviour to see if they were more effective on condom use. Interventions were divided into those that had a small effect on these cognitions. Findings showed that those interventions with greater effects on cognitive mediators also had greater effects upon behaviour such as condom use "(ds = 0.15 for interventions with a small effect on mediators vs. ds = 0.41 for those with larger effects on mediators)" (p.262).

Evidence statement: No change from Ellis and Grey (2004). There is sufficient reviewlevel evidence to conclude that interventions that incorporate behavioural skills training (specifically negotiation skills) are more likely to be effective.

4.5.4 Provision of basic, accurate information through clear, unambiguous messages

Two reviews were found to add to the 10 Category 1, 2 and 3 reviews from the original Evidence Briefing (Ellis and Grey, 2004).

Category 1: DiClemente et al., 2004

Category 2: Pedlow and Carey, 2004 Category 3: No evidence

DiClemente et al. (2004) concluded that one of the fundamental aspects of clinic-based interventions aimed at young people and achieving positive effects on STI-associated risk behaviours, included timely and accurate STI related information in a clear and understandable format (p.215).

Pedlow and Carey, (2004) argued for developmentally appropriate prevention interventions and maintained that prevention concepts aimed at young people need to be clearly illustrated using personalised examples appropriate to their cognitive maturity.

Evidence statement: No change from Ellis and Grey, 2004. There is sufficient reviewlevel evidence to conclude that interventions which include the provision of basic, accurate information through clear, unambiguous messages are more likely to be effective at reducing sexual risk-taking behaviour.

4.5.5 Facilitator training

The findings from two Category 2 and 3 reviews explored the influence of facilitator training. Facilitator training was not covered in the original Evidence Briefing (Ellis and Grey, 2004).

Category 1: No evidence Category 2: Robin et al., 2004 Category 3: Schaalma et al., 2004

Robin et al. (2004) stated that, "Programs that produced positive effects used trained adult facilitators" (p.18). The authors went on to conclude that, "program facilitators' training may be more important than whether facilitators' and participants' demographic characteristics match" (p.23).

Schaalma et al. (2004) examined the role of school-based intervention facilitators. They stated, "Health promotion in classrooms depends upon the establishment of a "safe" classroom atmosphere. Young people should feel free to discuss intimate issues because discussing sexual behaviour may break taboos concerning the public discussion of sexuality" (p.264). Schaalma et al. (2004) concluded that, "sexual health promotion in schools is a specialised aspect of health promotion requiring particular expertise from planning through delivery. High quality teacher training is required for high quality sexual health promotion in classrooms" (p.264).

New evidence statement: There is tentative review-level evidence to conclude that school-based interventions that use trained adult facilitators are more likely to be effective.

4.5.6 Of appropriate duration, length and intensity

We found five Category 1 and 2 reviews to add to the six Category 1, 2 and 3 reviews included in the original Evidence Briefing (Ellis and Grey, 2004).

Category 1: Manhart and Holmes, 2005; Ward et al., 2004; Ward et al., 2005.Category 2: Robin et al., 2004; Pedlow and Carey, 2004.Category 3: No evidence

In this Evidence Briefing we have examined reviews relating to the extended delivery of interventions and booster sessions separately. In the original Evidence Briefing both issues were considered together under a general category of appropriate duration, length and intensity.

(i) Extended delivery

Category 1: Manhart and Holmes, 2005; Ward et al., 2004; Ward et al., 2005.Category 2: Robin et al., 2004; Pedlow and Carey, 2004.Category 3: No evidence

With specific reference to interventions aimed at clinic patients, Ward and colleagues (2004) found no evidence to suggest that duration or format affected the impact of interventions. Ward et al. (2005) also stated that, "there was no evidence that an intervention's success in reducing infections or increasing consistent condom use was related to its format or length" (p.392). They concluded that, "Intervention format and length was not associated with trial results (group based RR 0.94 (0.76 to 1.25) versus individual based RR 1.16 (0.76 to 1.75))" (p.390).

Considering clinic-based interventions, Ward and colleagues (2005) also noted that there is no difference in effectiveness between studies that received a multiple session enhanced intervention and a brief intervention. However, they stated that, "the control groups in these trials received an intervention that goes beyond current UK practice, making it difficult to generalise results or transfer effect estimates to a UK population" (p. 392).

Manhart and Holmes (2005) highlighted evidence that the number of sessions may affect the impact of the intervention, they stated that, "All trials of successful counseling interventions involved at least 2

counseling sessions. However, 2 interventions of 4 and 6 sessions had no effect on the incidence of STI[s]" (S20).

With reference to adolescents, findings from Pedlow and Carey (2004) showed that "extending the delivery of interventions is a promising strategy for reaching youth during developmental periods". They concluded that preliminary findings "suggest that it may be necessary to design, and fund, interventions to be delivered over time as youth mature rather than providing multiple interventions in a single-shot intervention" (p.182).

With specific reference to interventions aimed at adolescents, Robin and colleagues (2004) stated that, "Programs that were more than 15 hours long addressed a greater variety of knowledge topics (including violence prevention and drug and alcohol use) than did those of shorter duration. In addition, more programs that were longer than 15 hours included more general life skills (such as community service learning, career planning, and general problem solving) than programs of shorter duration, and included a greater variety of general life skills. In contrast, programs between 7 and 15 hours were more likely than longer or shorter programs to teach abstinence and condom-use skills, and to teach a greater variety of each" (p.18). Robin and colleagues report one study (Rotheram-Borus et al., 1998) that tested two versions of a study of the same duration delivered in a different number of sessions. Findings showed that the intervention delivered in more sessions (seven compared to three) was more effective.

This evidence adds to that of Shepherd et al. (2001), included in the original evidence briefing, which concluded that, "interventions should be sustained over longer periods of time" (Ellis and Grey, 2004) (p.50). It also adds to the reviews by Kirkby and Coyle (1997) and Kirby (1999) which concluded that, "effective programmes lasted a sufficient length of time to adequately complete a variety of activities" (Ellis and Grey, 2004) (p.50).

Changed evidence statement: There is tentative review-level evidence to conclude that extended delivery of an intervention relates to its effectiveness.

(ii) Booster sessions

Category 1: No evidence Category 2: Pedlow and Carey, 2004 Category 3: No evidence

Pedlow and Carey, 2004 commented on four studies (Coyle et al., 2001; Levy et al., 1995; Aarons et al., 2000; Shrier et al., 2001) that incorporated booster sessions into their interventions. They stated that, "All studies that provided booster sessions were effective in reducing sexual risk behaviours. Suggesting that extended delivery of interventions is a promising strategy for reaching youth during development periods.

The use of extended interventions is different from interventions with a longer "dose". Prior reviews have shown that intervention dose is not necessarily related to effectiveness" (p.175).

This evidence is in accordance with the recommendation by Kirkby et al. (1994), included in the original Evidence Briefing (Ellis and Grey, 2004), for future research to assess whether re-inforcement measures, such as booster sessions, are effective in sustaining the desired intervention outcome (Ellis and Grey, 2004) (p50).

Changed evidence statement: There is tentative review-level evidence to conclude that extended delivery using 'booster' sessions enhances intervention effectiveness.

4.5.7 Use of peers and community opinion leaders

We found three Category 1 and 2 reviews to add to the eight Category 1, 2 and 3 reviews from the original Evidence Briefing (Ellis and Grey, 2004).

Category 1: DiClemente et al., 2004. Category 2: Pedlow and Carey, 2004; Robin et al., 2004 Category 3: No evidence

DiClemente et al. (2004) reported on two interventions involving peers (DiClemente et al., 2004b; Gillmore et al., 1997). Only one study, (DiClemente et al., 2004b) which used a multi-component approach, reported positive outcomes for consistent condom use.

In relation to peer-led interventions, Robin et al. (2004) found one study (Jemmott et al., 1998) that suggested there was no difference in effectiveness between adult versus peer-led interventions.

Pedlow and Carey (2004) reported on several studies that aimed to improve peer norms or used peer educators. Six out of the seven studies measuring peer norms for condom use or abstinence reported improvements in peer norms. In addition, five out of six studies that aimed to improve peer norms also achieved reductions in sexual risk behaviour. Pedlow and Carey concluded, "improvements in peer norms for safer sex and sexual communication skills were strongly associated with reductions in sexual risk behaviour" (p.181). One promising strategy reported by Pedlow and Carey (2004) for directly influencing peer norms involved working with young people in existing friendship groups.

Evidence statement: No change from Ellis and Grey (2004). There is tentative reviewlevel evidence to conclude that interventions that use peers and community opinion leaders are effective at reducing risk-taking behaviour.

4.5.8 Targeted and tailored (in terms of age, gender, culture, development etc), making use of needs assessment or formative research

We found four relevant Category 1 and 2 reviews to add to the 11 Category 1, 2 and 3 reviews in the original Evidence Briefing (Ellis and Grey, 2004).

Category 1: DiClemente et al., 2004; Ward et al., 2004; Ward et al., 2005 Category 2: Pedlow and Carey, 2004 Category 3: No evidence

Ward et al. (2004) stated that, "study interventions that were appropriately tailored to their population following extensive formative research were more likely to reduce subsequent bacterial STIs than those that were not" (p. 55). Ward et al. (2004) also stated that, "the most effective interventions were developed through extensive formative research" (p.386). Ward et al. (2005) noted, with reference to clinic-based interventions, that, "extensive formative research, including interviews, focus groups, input from community representatives, and pilot testing, was reported by the four most effective trials" (p.390).

DiClemente et al. (2004) stated that all effective elements of interventions uncovered should be tailored to gender and race.

With regard to pre- and young adolescents, Pedlow and Carey (2004) noted that interventions "must be tailored to meet the unique needs of younger versus older youth or sexually naïve versus experienced teens" (p.174). Characteristics of successful and developmentally appropriate interventions were, for example, prevention interventions aimed at adolescents who have not yet initiated sex, booster sessions, illustrated abstract concepts with examples, peer educators and skill-building activities.

Pedlow and Carey (2004) also argued that literature relating to pre- and young adolescents is limited because it fails to measure menarche, which is a developmental transition that has implications for sexual risk behaviour.

Evidence statement: No change from Ellis and Grey (2004). There is sufficient reviewlevel evidence to conclude that interventions are more likely to be effective if they are targeted and tailored for specific populations, making appropriate use of formative research or needs assessments.

4.6 Multi-component and multi-level interventions

See also section 4.5 for 'features of effective interventions'. These findings from seven Category 1, 2 and 3 reviews suggest that interventions with multiple components are more likely to be effective than those without. This adds to the five Category 1 and 3 reviews in the original Evidence Briefing (Ellis and Grey, 2004).

Category 1: DiClemente et al., 2004; Ward et al., 2004; Ward et al., 2005; Manhart and Holmes, 2005Category 2: Pedlow and Carey, 2004; Robin et al., 2004.Category 3: Schhalma et al., 2004

Ward et al. (2004; 2005) concluded that the several clinic-based trials using behavioural interventions observed greater reduction in STI rates. Interventions used personal goal setting, condom negotiation, risk-perceptions, role-play, group discussion, modelling behaviours and skills-building.

DiClemente et al. (2004) summarised that effective interventions included the following cornerstones: (1) providing timely and accurate STI-prevention information in clear, understandable language, (2) developing and mastering social (e.g., sexual negotiation/communication) and technical competency (e.g., condom application) skills through observational learning and active learning techniques (e.g. role playing, preferably a series of graded-intensity of high-risk situations), (3) enhancing adolescents' self-efficacy to communicate assertively and effectively with sex partners, (4) motivating adolescents to use newly acquired STI-prevention knowledge and risk-reduction skills; and (5) tailoring all the aforementioned messages and activities for gender and, if possible, race (p.215). DiClemente et al. (2004) also commented that the next generation of clinic-based STI risk-reduction programs should include a two-tier system, with the first tier consisting of clinic-based counselling and the second tier consisting of community-based prevention accessed through physician referral (p.216).

Manhart and Holmes (2005) reported on two studies (Celentano et al., 2000; Lonczak et al., 2002). Celentano et al. (2000) examined a multi-component behavioural intervention aimed at Thai army conscripts. Participants receiving the intervention showed a greater than 80% reduction in STI recurrence than controls. However, units were not randomly assigned to the intervention or the control. Lonczak et al. (2002) examined an intervention carried out in public schools in high-crime areas using a multi-component design that included teacher training, child social and emotional skills, and parent training. The African American subset of the intervention group reported a statistically significant decrease in self-reported STIs by aged 21 years (odds ratio (OR) 0.11; p<0.01).

Pedlow and Carey (2004) commented that the use of multiple intervention strategies should be used in prevention intervention design targeting adolescents and they called for "the continued development of multifaceted interventions that address biological, psychological, and social influences on sexual behaviour" (p.182).

Robin et al. (2004) concluded that effective interventions contained skills-building activities such as sexual communication, decision-making and problem solving and that "this may reflect a shift toward multi-component interventions that target a variety of youth competencies" (p.23).

Schaalma et al. (2004) made the point that behaviour change cannot be achieved through increasing knowledge alone, they maintained that interventions also need to influence the "proximal cognitive determinants of decision-making and goal enactment....[and should]..be informed by cognitive theories that have been successfully applied to a variety of other health related behaviours" (p.262).

Evidence statement: No change from Ellis and Grey (2004). There is sufficient reviewlevel evidence to conclude that multi-component interventions are more likely to be effective than single-component interventions.

Evidence statement: No change from Ellis and Grey (2004). There is insufficient reviewlevel evidence to support or discount the effectiveness of multi-level interventions.

4.7 Cost-effectiveness of sexual health interventions

Category 1: No evidence Category 2: No evidence Category 3: No evidence

We found no review level evidence regarding the cost-effectiveness of STI prevention interventions to add to the findings of the previous evidence briefing.

Changed evidence statement: There is no new review-level evidence to support or discount the cost-effectiveness of STI interventions.

The original evidence statement declared that there was tentative review-level evidence to conclude that STI prevention interventions are cost effective, and can be cost saving. However, there is no current evidence to demonstrate that recent interventions are either cost effective or cost saving.

4.8 Interventions to address inequalities in sexual health

Category 1: No evidence Category 2: No evidence Category 3: No evidence We found no review level evidence regarding interventions to address inequalities in sexual health to add to the findings of the previous evidence briefing. As such we recommend no change from the previous evidence statement.

Evidence statement: No change from Ellis and Grey (2004). There is no review-level evidence to support or discount the effectiveness of interventions to address inequalities in sexual health.

5 Discussion, conclusions and research recommendations

This evidence briefing update aims to address the following research questions:

- What works to prevent STI transmission? What works to reduce the risk behaviours for STI transmission? What works to address the determinants of STI risk?
- Are multi-component and multi-level interventions more likely to be effective in influencing sexual risk behaviours?
- What works to reduce inequalities in sexual health?
- What interventions are cost-effective?

In the Evidence section (4), we considered the evidence from Category 1, 2 and 3 reviews regarding the effectiveness of STI prevention interventions giving consideration to the level and feature of interventions to inform our evidence statements regarding effectiveness.

In this section we consider the methodological issues. We explore methodological issues and provide research recommendations relating to outcome measures of effectiveness (5.1.1), behavioural indicators (5.1.2), limitations identified by the reviews (5.1.3), limitations and gaps in the evidence (5.1.4) and the limitations of this review (5.1.5).

Where we have not found additional review-level evidence to add to the recommendations of the original Evidence Briefing (Ellis and Grey, 2004) we have stated that there is no change and reported the previous recommendation. All recommendations ought to be considered in conjunction with those of the previous review (Ellis and Grey, 2004).

5.1 Methodological issues

5.1.1 Outcome measures of effectiveness

The original Evidence Briefing found limited information regarding the effects of STI prevention intervention approaches on health promotion outcomes such as health literacy (e.g. sexual negotiation, condom skills). We also found that the majority of reviews selected primary studies for inclusion based upon health outcomes (e.g. STI incidence) or made conclusions of intervention effectiveness based upon intermediate health outcomes (e.g. behaviour). We did find reviews that included primary studies with health promotion outcomes (DiClemente et al., 2004; Manhart and Holmes, 2005; Pedlow and Carey, 2004). However, only one review (Pedlow and Carey, 2004) made conclusions regarding health promotion outcomes (knowledge and self-efficacy; condom use skills and attitudes to condoms, respectively) and this was in conjunction with intermediate health outcomes.

Similar to the original evidence briefing, we found that reviews called for more robust research using intermediate health outcomes to determine effectiveness (DiClemente et al., 2004; Ward et al., 2004). Also, we found that instead of authors calling for small-scale studies to measure outcomes at different

stages along the causal pathway (which would include measuring the personal and structural determinants of risk (e.g. knowledge or skills)) (as recommended by Ellis and Grey, 2004), there were calls for guidance on how to scale-up successful small scale interventions. However, DiClemente et al. (2004) highlight that health outcomes (e.g. STI incidence) should be measured in clinic-based interventions. DiClemente et al. (2004) also emphasise that many of the clinic-based interventions they examined did not measure health outcomes and they recommend that future clinic-based studies should measure STI incidence. In addition to these comments, Pedlow and Carey (2004) recommended that risk perception be linked to actual behaviour, however they also recommended that there is a need to measure process outcomes of developmental factors.

For further discussion of the conclusions of effectiveness and their implications, see the Evidence (4) and Conclusions and recommendations for policy and practice (5.2) sections. We conclude that there are no further research recommendations to add to those from the original Evidence Briefing.

Recommendations: No change from Ellis and Grey (2004). A key research recommendation is that intervention evaluations should measure outcomes relating to the personal and structural determinants of risk (e.g. knowledge, attitudes, skills, behavioural intentions, access to condoms, peer norms). This should be done in addition to the measurement of intermediate health outcomes (e.g. changes in behaviour) and, where appropriate (e.g. for large-scale multi-component interventions or programmes), health outcomes (changes in STI incidence). In turn, reviews should cease to exclude studies that only include data on health promotion outcomes (e.g. knowledge or skills) alongside their effects on intermediate health outcomes (e.g. behaviour) and health outcomes (e.g. STI incidence). Reviews should fully report all of these outcomes where possible.

5.1.2 Methodological issues: behavioural indicators

The previous Evidence Briefing found that indicators of 'risk' were unsophisticated and defined as, for example, number of partners, condom use and contraceptive use. We found no evidence to show that the terminology for risk indicators has become more specific. As such we agree with the research recommendations from the previous Evidence Briefing.

Recommendations: No change from Ellis and Grey (2004). It is imperative that data on intermediate health outcomes (i.e. sexual behaviour surveys) are more specific about the context in which certain behaviours take place. Primary studies and future reviews should therefore focus on more meaningful risk indicators when making judgements about the relative effectiveness of interventions in influencing so-called 'risky' behaviours. Where feasible and appropriate these should be correlated with biological behavioural indicators and other socio-demographic variables. In conclusion, we need a consensus on the appropriate indicators for assessing the effectiveness of STI prevention programmes with different target populations.

5.1.3 Methodological issues: limitations identified by the reviews

The reviews in this update reported similar limitations to those reported in the original Evidence Briefing (Ellis and Grey, 2004). Reviews reported that their inclusion criteria limited their findings and although most reviews only included RCTs, authors reported on the poor quality of the primary studies (DiClemente et al., 2004; Manhart and Holmes, 2005; Ward et al., 2005) and called for further rigorous trials. One review noted that the results of studies were limited in their generalisability (Ward et al., 2004). Reviews also highlighted the weaknesses of the effect sizes due to differences in follow-up periods (Robin et al., 2004; DiClemente et al., 2004) as well as sample size and appropriateness of the comparison group (DiClemente et al., 2004).

Other inconsistencies reported referred to the differences among groups that were reported but not controlled for in the analysis (Robin et al., 2004). It has also been highlighted that interventions aimed at adolescents have reported inconsistent findings from studies that have not been replicated (Pedlow and Carey, 2004).

Other limitations expressed by the reviews included the absence of reporting rates of attrition, the absence of intent-to-treat analyses, lack of power analyses, the lack of description in the primary studies relating to RCT methods (DiClemente et al., 2004). In addition to these limitations Manhart and Holmes (2005) and Ward et al. (2005) also commented on the potential for publication bias in their reviews.

Recommendations: There is a need for more rigorous evaluations of UK-based STI prevention interventions. It is necessary for researchers to clearly report the methodology that they use, including their methods for randomisation. Future research should include appropriate comparison groups and should consider longer follow-up periods. In these cases research should report: levels of attrition, intent-to-treat analysis, and effect sizes. There is also a need for primary studies to be standardised in their reporting.

5.1.4 Methodological issues: limitations and gaps in the evidence

In the reviews included in this evidence briefing update we found many of the same limitations and gaps as those reported in the original Evidence Briefing (Ellis and Grey, 2004). As such the limitations we highlight here should not be considered as an exhaustive list but should be considered in conjunction with those of the original review.

The main issues we found with the current review level evidence are stated below. For implications of the limitations, inconsistencies and gaps in the evidence see the Conclusions and recommendations for research section (5.2).

There was a dearth of UK-based evidence on intervention effectiveness. Thus, one of the main limitations we found was the lack generalisability to the UK population. We found that much of the literature included

in the reviews focused upon adolescents with only very little review-level evidence concentrated on interventions with adults. Specifically, there was a lack of evidence relating to commercial sex workers (CSWs), men who have sex with men (MSM) and other high-risk groups in the UK.

It is also clear from the evidence that there is a lack of evidence relevant to community-based interventions (see section 4.3). To this gap in the evidence, we add that there is a lack of review-level evidence relating to the cost-effectiveness of STI prevention interventions in all areas including, for example, clinics, schools and youth services.

There is also a dearth of information regarding the impact of interventions on inequalities or interventions addressing socio-political issues. In particular, there is a lack of evidence addressing the impact of interventions on socio-economic status and on vulnerable groups (e.g. looked after children).

Recommendations: In addition to the recommendations from Ellis and Grey (2004), for example, to be more transparent about the process used; to include more details of the intervention delivery and; to publish null findings, we recommend that further research is needed on the cost-effectiveness of STI prevention interventions in different settings. There is also a need for further research aimed at vulnerable 'high-risk' groups in the UK such as CSWs and looked-after children. Finally, there is a need for further research to explore the effectiveness of interventions that aim to address inequalities and socio-political issues specific to target populations.

5.1.5 Limitations of this review of reviews

This update followed the same methodology as the original Evidence Briefing and therefore has all the same limitations reported by Ellis and Grey (2004). In addition to these limitations we add that although all efforts were made to retrieve shortlisted documents we were unable to obtain 8% of the literature, which may have resulted in some relevant paper being missed.

We also note that the reviews in this update will include many of the same primary studies that were included in the original Evidence Briefing. Our most recent Category 1 papers were published in 2005 (Manhart and Holmes, 2005; Ward et al., 2005), and the most recent primary studies presented within this level were published in 2004. This updates the evidence base of reviews by two years.

5.2 Conclusion and recommendations for research

In this section we consider the findings based upon the evidence statements and examine their relevance to the UK population in order to inform our recommendations that follow each section. Where we have found no further evidence to inform additional recommendations we have stated that there is no change from the original Evidence Briefing (Ellis and Grey, 2004) and repeat the previous recommendation.

5.2.1 What works in STI prevention interventions

5.2.1.1 Clinic-based sexual health promotion: conclusions and research recommendations

As discussed in the Evidence section (4) there was **tentative review-level evidence** for the effectiveness of individual risk counselling.

Of the studies involving individual risk counselling (see section 4.1.1), only two studies (VCT Efficacy Study Group, 2000; EXPLORE Study Group, 2004) considered counselling in conjunction with HIV testing. In addition, all of the studies were either carried out in the US or Africa and not in the UK.

Recommendations: No change from Ellis and Grey, 2004. We urgently need more UK-based evaluations of risk counselling interventions for STIs other than HIV. We urgently need more UK-based evaluations of skill-based training.

5.2.1.2 Partner notification

We found **sufficient review-level evidence** regarding the effectiveness of partner notification for detecting new infections.

Recommendations: No change from Ellis and Grey, 2004. More research is needed to understand the consequences of partner notification for infected persons and their partners.

5.2.1.3 Improved communication between parents and adolescents (parent/family intervention): conclusions and research recommendations

Overall, we concluded that there was **insufficient review-level evidence** regarding the effectiveness of improved communication between parents and adolescents at reducing sexual risk behaviour. Although several authors commented upon the effectiveness of interventions such as parental monitoring, it must be noted that none of the studies examined were carried out in the UK.

DiClemente et al. (2002) recommended that as a method of intervention, families could be mobilised to communicate important values, model appropriate behaviours, monitor adolescents' behaviour, and encourage protective behaviours. The improvement of familial communication and skills may strengthen the family unit and encourage adolescents, as well as other family members, to adopt and maintain protective behaviours.

Recommendations: We urgently need more UK-based evaluations of interventions that incorporate parental/adolescent communication/monitoring as a means to reduce sexual risk-taking behaviour.

5.2.1.4 Clinic-based interventions aimed at adolescents and general clinic attendees: conclusions and research recommendations

The evidence presented in sections (4.1.1) and (4.2.1) led us to conclude that there is **insufficient review-level evidence** of clinic-based interventions aimed at adolescents. We also found **tentative review-level evidence** for the effectiveness of clinic-based interventions, especially those that include skills training, for instance, to increase the levels of condom use.

However, only one study was carried out in a UK-based GUM clinic (Imrie et al., 2001) and the findings from this study showed that clinic-based interventions were ineffective at changing sexual risk behaviour. Although only one study, it shows that one cannot infer the transferability of studies from other countries. These findings suggest that there is a need for further rigorously evaluated primary studies in the UK. Studies should examine new interventions based on the format of those that have demonstrated effectiveness in US settings. In addition, studies should examine the effectiveness of behavioural interventions in GUM clinics, where new approaches to reducing infection rates are urgently needed (recommendations from Ward et al., 2004 and 2005).

Recommendations: There is a need for more primary studies to evaluate the components of effective clinic-based interventions. When sufficient primary studies are available a methodologically sound review of UK clinic interventions should be carried out in order to inform the evidence base in the UK.

Parental involvement in clinic-based interventions

We conclude that there is **insufficient review-level evidence** to support or discount the effectiveness of parent involvement in clinic-based interventions.

Recommendations: More UK research is needed to discern the effectiveness of parent involvement in clinic-based interventions and the effects of counselling/education on parent/adolescent communication and the subsequent impact (if any) on sexual risk-taking behaviour.

5.2.1.5 Group work: conclusions and research recommendations

From the evidence presented in section (4.2.4) we conclude that there is **sufficient review-level evidence** to conclude that small-group work, especially that involving skills-building activities, can be effective. Interventions reporting increased condom use outcomes frequently used role-play, games and discussion.

However, only one study was carried out in the UK (Imrie et al., 2001) and thus the findings from these reviews are limited in their generalisability. Although group work can be resource intensive, DiClemente et

al. (2004) recommended that further research should be carried out involving group work with clinicians or healthcare workers and patients.

Recommendations: More UK research is needed to discern the effectiveness of small group work relevant to specific target groups. It is imperative that evaluations of these interventions consider the cost-effectiveness and resource implications of this type of intervention compared to individual level interventions.

5.2.1.6 School-based sex education programmes: conclusions and research recommendations*

Overall, there is **sufficient review-level evidence** to conclude that school-based sex education can be effective at delaying the onset of sex and encouraging condom use in those who are already sexually active (section 4.2.3).

School-based sex education programmes most frequently included elements such as role-play, discussion and skills-training. Some school-based interventions also involved the provision of condoms and clinic services.

Evidence of effectiveness is primarily based upon interventions carried out in the US and Africa. However, one study was carried out in The Netherlands and another was carried out in Scotland. This one UK-based intervention adds to the two others reported in the original evidence briefing (Ellis and Grey, 2004), all from Category 3 reviews. As such, there remains a dearth of review-level evidence regarding UK studies of school-based sex education programmes.

We examined the review-level evidence concerning the provision of condoms in schools (section 4.2.2). We conclude that there is **insufficient review-level evidence** that condom availability in schools can be effective at increasing condom use in sexually active young people. However, the reported studies took place only in the US and may have limited generalisability to the UK population. Schmiedl et al. (2004) cited Cohen et al. (1999) who recommended that barriers to condom use be eliminated and condoms provided free of charge. They also stated that with the appropriate resources and support "school nurses [could] develop successful school-based condom disbursement programs that effectively reduce teen's exposure to STDs and HIV " (p.20).

Schaalma et al. (2004) also highlight the fact that health behaviour not only refers to individual behaviour but to the actions of groups and organisations also. Schaalma at al. (2004) stated that, "condom accessibility may depend on individual knowledge, motivation, and skills, but is also determined by the actions of legislators, health authorities, schools, and other decision-making groups. Therefore, interventions may be required at each of these levels if condom availability is to be increased among sexually active young people" (p260).

^{*} See evidence briefing on teenage pregnancy and parenthood (Swann et al., 2003a).

^{-62-.} Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

Recommendations: There is a need to address the practicality of providing clinic and condom services in schools. Evaluations of clinic services and condom distribution schemes in schools should also ensure that as well as addressing the effectiveness of condom distribution in schools, they address related issues of parental consent, motivation for sexual activity pre- and post-interventions, level of pubertal development and changing social norms as well as behavioural outcomes. Also, school-based interventions should be developmentally appropriate, and should use multiple strategies compatible with adolescents' cognitive ability, including managing emotions and risk perception (recommendation from Pedlow and Carey, 2004).

Further conclusions regarding the effects of condom distribution and sex education on the sexual practices of young people and further views on elements of effective school interventions were also elicited from the reviews. We have addressed these points separately below.

Condom distribution in schools does not increase sexual activity

We identified **insufficient review-level evidence** to conclude that condom distribution in schools does not increase sexual activity. One Category 3 review (Schmeidl, 2004) referred to several studies (Schuster et al., 1998; Wolk and Rosenbaum, 1995; Guffmacher et al., 1997; Kirby et al., 1999; Blake et al., 2003) and concluded that condom distribution does not increase sexual activity.

Recommendations: Once there are sufficient primary studies a methodologically sound review will be needed to support or discount the tentative conclusion that school-based condom distribution does not increase sexual activity.

Sex education should start early, prior to sexual initiation

Overall, we conclude that there was **tentative review-level evidence** that sex education should start early and before sexual debut in an effort to delay the onset of sexual activity. However, further UK research is necessary to discern the most effective age and format of pre-adolescent sex education.

Recommendations: Primary research is needed to evaluate school-based interventions and their effectiveness. In addition to the length, format and content of interventions, evaluations should consider effectiveness of interventions at different ages, gender, sexual experience and stage of pubertal development. When sufficient primary evidence is available a methodologically sound review should be carried out to inform the UK evidence base.

Parental involvement in school interventions

The evidence presented in section 4.2.3 (ii) leads us to conclude that there is **insufficient review-level evidence** for the effectiveness of parental involvement in school interventions that can reduce the number of partners and rate of pregnancy during adolescence.

Once again there no UK evidence identified to support this conclusion and further research is needed. Schaalma et al. (2004) recommended that, "health promotion planners could focus on mobilising parental support for comprehensive sex education" (p267).

Recommendations: More research is needed to explore how parental involvement in sexual health interventions could complement school-based interventions and impact adolescents' sexual risk taking behaviour.

5.2.1.7 Detached education and outreach by health professionals: conclusions and research recommendations

Overall, we conclude that there is **tentative review-level evidence** for the effectiveness of detached education, including condom distribution. The studies we found primarily focused upon CSWs and their customers, however, none of the studies took place within the UK and as such the findings may be limited in their generalisability.

Recommendations: More research is needed to explore the effectiveness of detached education and outreach for different target populations in the UK. Studies should also consider the effectiveness of individual intervention components and cost-effectiveness. Once sufficient primary studies are available a methodologically sound review of interventions ought to be undertaken.

5.2.1.8 Community level interventions: conclusions and research recommendations

In section 4.3 we report on studies of community level interventions from two reviews. We conclude that there is **insufficient review-level evidence** for the effectiveness of community outreach.

Recommendations: Further research is needed to determine the effectiveness of communitybased interventions in the UK, such as condom provision and community empowerment schemes as well as the effects of mass media on risk taking behaviour. Also, research should explore the potential harms of mass media interventions with reference to specific target groups and the impact upon clinic services.

5.2.1.9 Socio-political interventions: conclusions and research recommendations.

We found **insufficient review-level evidence** to support or discount the effectiveness of legislation, policy, equality work or regulation. We found two reviews that addressed skills-based sex education programmes available nationally in schools and we also found evidence of effective condom distribution policies. However, they do not refer to UK policies and are limited in their generalisability. Therefore, we conclude that there has been no change since the original evidence briefing and as such the previous recommendations remain.

Recommendations: No change from Ellis and Grey (2004). We urgently need more UK evaluations – and in turn, reviews- of socio-political interventions or actions. As it is difficult to conduct controlled trials of most of these types of 'interventions', it is likely that observational research will be most useful in looking at the impact of policies, regulations, and strategies to tackle wider determinants of risk such as education, employment, poverty and discrimination. More specifically, research is needed on use, acceptability and sexual behaviour. As well as research on the interventions themselves, there is also a need for research to identify the structural determinants of sexual risk for difference populations, which are often (but not exclusively) addressed by socio-political interventions.

5.2.1.10 Features of effective interventions:* conclusions and research recommendations

We examined the review-level evidence for features of effective interventions in section 4.5. We would like to emphasize that this does not constitute an exhaustive list of features within the reviews. However, they are features of successful interventions that have been specifically highlighted by the authors.

Use of theoretical models

We conclude that there is **sufficient review-level evidence** that theory-based interventions are effective at influencing sexual risk-taking behaviour. A number of reviews reported on the successful use of theoretical models (Manhart and Holmes, 2005; Ward et al., 2004; Ward et al., 2005; Pedlow and Carey, 2004; Schaalma et al., 2004). Although it is the case that studies are clearly reporting the theory used they do not explain the reason for their choice and why it is appropriate for the population under investigation.

Only one of the reviews (Pedlow and Carey, 2004) addressed issues for consideration regarding theoretical models. They argued that existing theoretical models should be modified and specifically aimed at young people.

Recommendations: It is important that studies report the theoretical model they use and their reason for its use. Evaluations ought to consider the effectiveness of each theory and their components on different target groups.

Emphasis on risk reduction

We conclude that there is **tentative review-level evidence** for the effectiveness of interventions that emphasise risk-reduction. There is **insufficient review-level evidence** to support or discount the effectiveness of school-based abstinence only approaches. DiClemente et al. (2004) concluded that interventions are more effective if they have an emphasis on risk reduction as opposed to abstinence alone.

^{*} Also, see evidence briefing on teenage pregnancy and parenthood (Swann et al., 2003a).

^{-65-.} Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

It is important to note that the majority of these studies focused solely upon adolescents and as such this has impacted on the recommendations.

Recommendations: In addition to the recommendation in the previous Evidence Briefing (Ellis and Grey, 2004) that 'We need a methodologically sound review of the effectiveness of riskreduction interventions, compared with abstinence-only approaches, for young people. In particular, it may be timely for a good quality review into the effectiveness of school-based abstinence-only programmes compared with programmes which emphasise risk reduction', we add that that it is necessary for school-based evaluations to consider age, sexual knowledge and sexual experience. It is also imperative that further research is carried out to discern the effectiveness of risk reduction on other target groups.

Use of behavioural skills training

Overall, there is **sufficient review-level evidence** to conclude that behavioural skills training is an important and effective component of STI prevention interventions (see section 4.5.3). Pedlow and Carey (2004) recommended the inclusion of skills such as emotion management, risk perception, communication skills and negotiation skills.

Recommendations: In addition to the previous Evidence Briefing recommendation that 'Research is needed to identify the most effective methods of behavioural skills training with different populations' we recommend that further rigorous research is needed to distinguish between the effectiveness of behavioural skills training specific to sexual health, behavioural skills that are more general 'social/life skills', and those skills that aim to increase adolescent

Provide basic accurate information

Evidence included in the original evidence briefing (Ellis and Grey, 2004) and in this update (see section 4.5.4) suggests that there is **sufficient review-level evidence** that information provided in a clear and accurate manner is likely to be effective in combination with, for example, behavioural skills-training.

Recommendations: No change from Ellis and Grey (2004). Primary and secondary research needs to measure and report the impact upon health promotion outcomes such as knowledge and awareness. Research is needed to identify the effective elements of training and materials to support the providers of sexuality information (teachers, parents and peers). Research in this area should attempt to explore the interaction between, and relative importance of, the nature of the provider (including their comfort with the issues and their relationship with the student), and the nature of the training or materials.

Trained facilitator

Evidence in section 4.5.5 suggests that there is **tentative review-level evidence to conclude** that trained adult facilitators are effective at delivering interventions in a school environment.

Recommendations: Further research is needed to support or discount the tentative finding that trained facilitators are effective at delivering school-based STI prevention interventions. Research is also needed to support or discount the effectiveness of teacher training on the delivery of sex education in schools.

Duration, length, and intensity

We identified **tentative review-level evidence** to conclude that extended delivery of an intervention is related to its effectiveness (see section 4.5.6). There is also **tentative review-level evidence** to support the use of 'booster' sessions to aid the extended delivery of interventions. Pedlow and Carey (2004) highlighted that extended interventions aimed at young people can allow researchers to explore transitional phases such as entering puberty and initiating sexual activity. Extended delivery, especially that using booster sessions allows researchers to reinforce the intervention months or years after the initial intervention has finished, thereby refreshing information and skills that may become more meaningful as sexual awareness and experience develops.

Robin et al. (2004) raised the difficulty of determining programme length, as multiple sexual risk-reduction interventions took place at the same time, thereby affecting the dosage received. They also reported on the failure of studies to state the duration and dose of interventions.

Recommendations: We agree with the original recommendation from Ellis and Grey (2004) that, 'Research needs to report more specific variables about the intervention's duration and intensity, for example the amount of personal exposure to the facilitator and the number and length of 'sessions'. Longer-term follow-up (i.e. well beyond six months, ideally up to one year and beyond) should assess the durability of measured effects and determine whether re-inforcement, for example booster sessions, are necessary to sustain outcomes'. We also recommend that researchers use models such as Intervention Mapping or PRECEDE/PROCEED model to ensure, through educational diagnosis, that optimal circumstances prevail for implementing the interventions (e.g. that it does not overlap with other planned programmes).

Use of peers and community leaders

Overall, there was **tentative review-level evidence** to conclude that the use of peers and community leaders was effective. However, none of the studies that used peers or community leaders were carried out in the UK. Also, as evidence suggests that trained adult facilitators are effective at delivering STI prevention interventions in schools, more research is needed to support or discount these findings and clarify their effectiveness with different target groups and types of interventions.

Recommendations: No change from Ellis and Grey (2004). We need methodologically sound primary research into the effectiveness of peer-led interventions and into the reasons for their success (or failure). A literature review should determine the availability of UK research for conducting a review of effectiveness in this area.

Use of targeted and tailored interventions

The evidence in section 4.5.8 leads us to conclude that there is **sufficient review-level evidence** for the effectiveness of targeted and tailored interventions.

Ward et al. (2005) explained that, "formative research seeks to identify the behaviours, motivations, and beliefs within the target population that lead to increased risk, and link these to the key elements of an interventions" (p.392). Ward et al. (2004; 2005) included studies that were based upon extensive formative research including work on cultural norms or sexual beliefs of different groups. Formative research can also include the use of needs assessments that gather information on cultural diversity and gender issues appropriate to target groups.

Pedlow and Carey (2004) highlighted some of the limitations of formative research targeting young people and stated that researchers have failed to "measure onset of menarche, a developmental transition with strong implications for sexual risk behaviour" (p.175). DiClemente et al. (2004) stated that studies need to be expanded to include other groups in addition to adolescents and consider their family and social environment (p.215).

Recommendations: We agree with the recommendations from the previous Evidence Briefing that more needs assessments and formative research relative to specific populations are required and that authors should share their findings (Schaalma et al., 2004); that studies ought to report on how this research contributed to the interventions; the transferability of the formative research; and that reviews ought to report on needs assessments and formative research (Ellis and Grey, 2004). To this we add that further formative research targeting young people ought to focus upon pubertal development and ought to include measures of cognitive functioning, future-time perspective, and decision-making. We also suggest that future research should not consider young people alone, but ought to consider their wider social influences of peers and parents (recommendations from Pedlow and Carey, 2004).

Multi-component and multi-level interventions

From the evidence presented in section 4.6 we conclude that there is **sufficient review-level evidence** that multi-component interventions are effective. These interventions incorporate elements such as education, counseling and skills-building (e.g. communication and goal enactment). There is **insufficient review-level evidence** to support or discount the effectiveness of multi-level interventions.

Recommendations: No change from Ellis and Grey (2004). We need further evaluations, and in turn reviews, of multi-component interventions. Research needs to investigate the relative contribution of the different components of interventions effectiveness. We also urgently need evaluations and reviews of multi-level interventions or programmes. Evaluations of multi-level interventions need to ensure that they capture all relevant health promotion outcomes (e.g. knowledge), peer norms, anti-discrimination policies alongside any intermediate health outcomes (e.g. STI incidence).

5.2.2 What interventions are cost effective: conclusions and research recommendations

Section 4.7 shows that we identified **no review-level evidence** relating to cost-effective interventions. Manhart and Holmes (2005), Ward et al. (2004) and Ward et al. (2005) all called for further research on the cost-effectiveness of STI prevention interventions to ensure that limited resources are effectively utilized.

Recommendations: We agree with the recommendations of the previous review (Ellis and Grey, 2004) that more UK studies of cost-effectiveness should be carried out and followed by a review of the cost-effectiveness literature. To this we add that information is needed on the cost-effectiveness of interventions specific to different settings and estimates for the relative cost-effectiveness of altering approaches.

5.2.3 What works to reduce inequalities in sexual health: conclusions and research recommendations

As can be seen in section 4.8 there is **no review-level evidence** regarding interventions to reduce inequalities in sexual health. In particular, we found no review-level evidence relating to socio-economic status, sexuality, gender and ethnicity.

Recommendations: No change from Ellis and Grey (2004). There is an urgent need to conduct evaluations of interventions to determine their effectiveness in addressing inequalities in sexual health; in particular, whether there is any differential impact of interventions according to the socio-economic and demographic characteristics (including sexuality, ethnicity, refugee status) of individuals within the population being targeted. All intervention research should therefore routinely record such socio-economic and demographic variables.

5.2.4 Implications for policy and practice

Due to the limitations of the evidence identified (see section 5.1) we are unable to make recommendations for policy and practice. We have, however, made recommendations for research. We also emphasise that practitioners and policy makers should not consider this review of reviews alone, but in conjunction with other the original Evidence Briefing (Ellis and Grey, 2004), non-review evidence and other relevant source information.

Appendix A. Example of a search strategy

Medline search

- 1 meta-analysis/
- 2 review literature/
- 3 (meta-analy\$ or meta analy\$ or metaanaly\$).ti,ab.
- 4 (systematic\$ adj4 (review\$ or overview\$)).mp.
- 5 meta-analysis.pt.
- 6 review.pt.
- 7 review.ti.
- 8 review literature.pt.
- 9 1 or 2 or 3 or 4 or 5 or 6 or 7 or 8
- 10 Case Report/
- 11 letter.pt.
- 12 historical article.pt.
- 13 review of reported cases.pt.
- 14 review, multicase.pt.
- 15 10 or 11 or 13 or 14
- 16 9 not 15
- 17 animal/
- 18 human/
- 19 17 not (17 and 18)
- 20 16 not 19
- 21 sexually transmitted diseases/
- 22 Sexually Transmitted Diseases, Bacterial/
- 23 Sexually Transmitted Diseases, Viral/
- 24 chancroid/
- 25 chlamydia infections/
- 26 lymphogranuloma venereum/
- 27 gonorrhea/
- 28 granuloma inguinale/
- 29 syphilis/
- 30 condylomata acuminata/
- 31 herpes genitalis/
- 32 chlamydia.ti,ab.
- 33 (gonorrhea or gonorrhoea).ti,ab.
- 34 syphilis.ti,ab.
- 35 (genital herpes or herpes genitalis).ti,ab.
- 36 chancroid.ti,ab.
- 37 lymphogranuloma venereum.ti,ab.
- 38 granuloma inguinale.ti,ab.
- 39 condylomata acuminata.ti,ab.
| 41 | sexually transmitted disease\$.ti,ab. |
|----|----------------------------------------------------------------------------------------------------------------------------------------|
| 42 | sexually transmitted infection\$.ti,ab. |
| 43 | venereal disease\$.ti,ab. |
| 44 | 21 or 22 or 23 or 24 or 25 or 26 or 27 or 28 or 29 or 30 or 31 or 32 or 33 or 34 or 35 or 36 or 37 or 38 or 39 or 40 or 41 or 42 or 43 |
| 45 | (hiv or aids).mp. |
| 46 | 45 not (44 and 45) |
| 47 | 44 not 46 |
| 48 | 20 and 47 |
| 49 | limit 48 to english language |
| 50 | limit 49 to english language |
| 51 | limit 49 to yr=2001-2006 |
| | |

Appendix B. CAT screening tool

The CAT is divided into two stages. The first stage assesses the strengths of the methods used to identify and select all of the available literature, since this is regarded as one of the most important factors in ensuring a balanced view of the evidence. If a paper passes this first stage, then the quality of its methodological analysis and the appropriateness of its conclusions are assessed.

Authors:				
Title:				
Source:				
Does this paper address your topic area? Yes	No	Unsur	e	
Circle whether the paper is a:				
 Systematic review Synthesis Other review (please specify) 				
Does it address (circle as appropriate):				
Effectiveness (interventions and treatments) Causatio Monitoring and surveillance trends Other (please specify)	on			
Does the paper have a clearly focused aim or research question? (1)	Yes	No	Unsure	
Consider whether the following are discussed:				
The population studied	Yes	No	Unsure	
The interventions given	Yes	No	Unsure	
The outcomes considered	Yes	No	Unsure	
inequalities	Yes	NO	Unsure	
Do the reviewers try to identify all relevant English language studies?	Yes	No	Unsure	
Consider whether details are given for:				
Databases searched (2a)	Yes	No	Unsure	
References followed up (2b)	Yes	No	Unsure	
Experts consulted (2b)	Yes	No	Unsure	
Grey literature searched (2b)	Yes	No	Unsure	
Years searched (2c)	Yes	No	Unsure	
Search terms specified (2c)	Yes	No	Unsure	
Indusion criteria described (2d)	Yes	No	Unsure	
Is it worth continuing?	Yes	No		
Why / Why not?				

Do the authors address the quality (rigour) of the included studies? (3, 3d)	Yes	No	Unsure
Consider whether the following are used: A rating system (3b) More than one assessor (3c)	Yes Yes	No No	Unsure Unsure
If study results have been combined, was it reasonable to do so?	Yes	No	Unsure
Consider whether the following are true: Are the results of included studies clearly displayed? (4a) Are studies addressing similar research questions? (4b)	Yes Yes	No No	Unsure Unsure
Are the studies sufficiently similar in design? (4b)	Yes	No	Unsure
Are the results similar from study to study (test of heterogeneity)? (4b)	Yes	No	Unsure
Are the reasons for any variation in the results discussed?	Yes	No	Unsure

What is the overall finding of the review? Consider: How the results are expressed (numeric – relative risks, etc); whether the results could be due to chance (p-values and confidence intervals).

Are sufficient data from individual studies included to media between data and interpretation/conclusions? (5)	te	Yes	No	Unsure
Does this paper cover all appropriate interventions and approaches for this field (within the aims of the study)?		Yes	No	Unsure
If no, what?				
Relevance				
Can the results be applied/are generalisable				
to a UK population/population group?		Yes	No	Unsure
Are there cultural differences from the UK?		Yes	No	Unsure
Are there differences in healthcare provision with the UK? Is the paper focused on a particular target group (age, sex,		Yes	No	Unsure
population sub-group etc)?		Yes	No	Unsure
Accept for inclusion in the evidence briefing?	Yes	No	Refer	to
		1,2,3	4,5	third party
Use to inform the review of effectiveness?		Yes	No	
Use to inform the background discussion?		Yes	No	
Additional commenter				

Appendix C. List of reviews by Category 1-5

Category 1

- DiClemente, R. J., Milhausen, R., Sales, J. M. et al. (2005). A programmatic and methodologic review and synthesis of clinic-based risk-reduction interventions for sexually transmitted infections: research and practice implications. *Seminars in Pediatric Infectious Diseases* 16 (3): 199-218.
- Manhart, L. E. and Holmes, K. K. (2005). Randomized controlled trials of individual-level, populationlevel, and multilevel interventions for preventing sexually transmitted infections: what has worked? *Journal of Infectious Diseases* 191 (Suppl 1): S7-24.
- Ward, D. J., Rowe, B., Pattison, H. et al. (2004). Behavioural interventions to reduce the risk of sexually transmitted infections in genitourinary medicine clinic patients: a systematic review. Birmingham, UK: West Midlands Health Technology Assessment Collaboration, Department of Public Health and Epidemiology, University of Birmingham.
- Ward, D. J., Rowe, B., Pattison, H. et al. (2005). Reducing the risk of sexually transmitted infections in genitourinary medicine clinic patients: a systematic review and meta-analysis of behavioural interventions. *Sexually Transmitted Infections* 81 (5): 386-393.

Category 2

- Pedlow, C. T. and Carey, M. P. (2004). Developmentally appropriate sexual risk reduction interventions for adolescents: rationale, review of interventions, and recommendations for research and practice. *Annals of Behavioral Medicine* 27 (3): 172-184.
- Robin, L., Dittus, P., Whitaker, D. et al. (2004). Behavioral interventions to reduce incidence of HIV, STD, and pregnancy among adolescents: a decade in review. *Journal of Adolescent Health* 34 (1): 3-26.

Category 3

- DiClemente, R. J., Crosby, R. A., and Wingood, G. M. (2002). Enhancing STD/HIV prevention among adolescents: the importance of parenteral monitoring. *Minerva Pediatrica* 54 (3): 171-177.
- Schaalma, H. P., Abraham, C., Gillmore, M. R. et al. (2004). Sex education as health promotion: what does it take? *Archives of Sexual Behavior* 33 (3): 259-269.
- Schmiedl, R. (2004). School-based condom availability programs. [Review] [23 refs]. *Journal of School Nursing* 20 (1): 16-21.
- Shrier, L. A. (2004). Sexually transmitted diseases in adolescents: biologic, cognitive, psychologic, behavioral, and social issues. *Adolescent Medicine Clinics* 15 (2): 215-234.

Category 4 – see Reference for full details

Author (s)	Year	Reason for not including in Category 1-3
Abdullah A.S., et al.	2004	Not a review, but points to gaps in the research
Gil A.G. & Tubman J.G.	2005	Not a synthesis. Includes relevant evidence.
Gott M.	2004	Not a synthesis, but points to gaps in the evidence
Irwin R.	2003	Not a review of prevention, but contains some information on evidence
Jackson, D. et al.	2004	Not a review of behavioural interventions
Kirby B.D.	2002	Not a synthesis, but contains information in Emerging Answers review
Mayaud, P. & Mabey D.	2004	Not a review of behavioural interventions, but has some useful information
Miller, K. E., et al.	2003	Not a review but has guidelines on prevention
Oberg C., et al.	2002	Not a review, but points to gaps in the research
Patel R.	2004	Not a synthesis, but contains recommendations
Rekart M.L.	2005	No risk behaviour included, includes preventative information
Reyna V.F.	2005	Not a review of prevention, but contains some information on evidence
Speizer I.S., et al.	2003	Developing countries
Zenilman J.M., et al.	2005	Not a review - includes some prevention programmes and theories

Category 5 – see Reference for full details

Author (s)	Year	Reason for not including in Category 1-3
Anonymous	2002	Not a review
Bergk A., et al.	2005	Not a review of STI prevention interventions
Blair, M.	2004	Not STI prevention
Bonner K.	2001	Focused on HIV prevention (male circumcision).
Bonsu IK.	2005	Not a review
Casper C. & Wald A.	2002	Condom effectiveness, not an STI prevention review
Chorba T. et al.	2004	Not a review of STI prevention
Coetzee N., et al.	2003	Duplicates Mathews et al., 2002
Cothran M.M. & White J.P.	2002	Not a review of STI prevention
Dehne K.L. & Riedner G.	2001	STI services not prevention interventions
Dulmus C.N.E. & Rapp- Paglicci L.A.E.	2005	Not a review
Elwy A.R.et al.	2002	Included in previous Evidence Briefing.
Fenton K.A. & Hughes G.	2003	Not a review
Feroli K.L. &Burstein G.R.	2003	Not a review of STI prevention interventions
Fitch J.T.	2001	Not a review
Fortenberry J.D.	2002	Included in previous Evidence Briefing.
Genuis S.J. & Genuis S.K.	2005	Not a review
Gilson R.J. & Mindel A.	2001	Not STI prevention
Glik D. et al.	2002	HIV focus
Heinz M.	2004	Not a review. No prevention interventions.
Henderson Z. et al.	2005	Not a review of STI prevention interventions
Hingson R.W. & Howland J.	2002	HIV and pregnancy interventions - exclusion criteria
Holmes K.K., et al.	2004	Condom effectiveness not an STI prevention intervention review
Honey E., et al.	2002	Cost effectiveness - but screening
Jungmann E.	2005	Not a review. About treatment not prevention.
Kirby D., et al.	2002	Included in the original review (Cat 3)
Klein J.D. & Matos A.M.	2002	Not prevention
Kotchick B.A. et al.	2001	Not a synthesis of prevention interventions
Lee J.D. & Clarke J.	2004	Condoms and not a review
Leone P.	2005	Not a review. Treatment not prevention.

Low N.	2004	Not a review of STI prevention. Effects of treatment.
Marra C.M.	2004	Not a review
Mathews, C. et al.	2004	Not a synthesis
Mathews, C. et al.	2002	Included in the original review (Cat 4)
Mathews, Cet al.	2002	Duplicate
Mathews, C. et al.	2002	Duplicate
Mathews, C., et al.	2001	Publication of Cochrane review (included in the original Evidence Briefing (Cat 1))
McEvoy M. & Coupey S.M.	2002	Not a review
Meade & Ickovics	2005	Not a synthesis
Minnis A.M. & Padian N.S.	2005	Not a review of behavioural interventions
Moran J.	2005	Not a review. About treatment of STIs.
Myer L., et al.	2001	Protocol, not review
Oberdorfer A., et al.	2002	Protocol, not a review
Pozniak A.	2003	Not a review of prevention intervention
Roberts C.	2005	Not a review of prevention
Robin L., et al.	2004	Duplicate
Ross M. W. & Williams M.L.	2002	Included in the previous review
Sangani P. et al.	2001	HIV and treatment exclusion criteria
Sarkar N.N.	2001	Not a review
Scholes D. et al.	2003	Not a review of STI prevention
Serrant-Green L.	2005	Not STI prevention
Sitruk-Ware R.	2005	Exclusion criteria, not a review of prevention interventions
Smith C.	2003	Overview of service structures.
Steenbeck A.	2004	Not a relevant population to the UK.
Sternberg P. & Hubley J.	2004	Not STI prevention review, focuses on HIV prevention.
Suarez T. & Miller J.	2001	Not a review of STI prevention
Sulak PJ.	2004	Not a review
Teran S., et al.	2001	Not a review
Thachil A. & Bhugra D.	2006	Not about STI prevention intervention effectiveness
Tillett J.	2005	Not a review
Tobin J.M.	2002	Cost effectiveness, but screening.
Toroitich R, et al.	2001	Protocol for a Cochrane review (withdrawn)
Toups M.L. & Holmes W.R.	2002	Not a synthesis
Tripp J., & Viner R.	2005	Not a review of prevention interventions
Warner L., et al.	2006	Condom effectiveness not an STI prevention intervention review
Weller , S. & Davis Beaty , K.	2002	Protocol, not a review
Wilson M.M.	2003	Not a review

Appendix D. In depth summaries of category one, two and three reviews

Category one reviews

Data pool:	
Population	Adolescents
Setting(s)	Clinic venues
Interventions	Clinic-based sexual risk-reduction interventions
Searches	Searches were conducted in: PubMed, Academic Search Premier, AltHealthWatch,
	MEDLINE, Health Source: Nursing/Academic Edition, PsycARTICLES, Cambridge
	Scientific Abstracts (Biological Sciences, AIDS and Cancer Research Abstracts), Lexis
	Nexis Health and Medical Journals, PsycInfo, Web of Science, Science Direct, Ingenta,
	and Journals at Ovid.
Selection/inclusion	Studies were included if they were:
criteria	(1) clinic-based (ie, the majority of sample had to be recruited from a medical clinic),
	(2) randomised controlled trials (RCTs) of interventions to decrease sexual risk-taking
	behavior among adolescents, and
	(3) published in peer-reviewed journals.
Quality assessment	16 methodological criteria were identified to assess the studies:
	(1) clear description of study site and sample; (2) specification of theoretical framework;
	(3) description of program implementation; (4) description of intervention content and
	behavior change techniques sufficiently detailed to
	permit replication; (5) description of content for control group treatment; (6) specification
	of length of follow-up; (7) use of blinding procedures to prevent bias; (8) specification of
	retention rates reported for each study condition; (9) adherence
	to intention-to-treat principles in the data analysis; (10) assessment of pretest
	equivalence on socio-demographic and behavioral factors between study conditions; (11)
	clear description of data analytic techniques; (12) specification of a
	measure of variability for the designated effect size; and (13) sample size justification.
	In addition to these criteria authors also used the JAMA checklist.
Types of studies	Randomised controlled trials
Number of studies	9 studies
Outcomes reported and/or	Sexual intercourse; number of partners; condom use
used for assessing	
'effectiveness'	
Review findings:	Only a few of the programmes included STI incidence as on outcome.
	• STI risk-reduction programs may not be of sufficient intensity or dosage to sustain

DiClemente R.J. et al. (2004) A programmatic and methodologic review and synthesis of clinic-based riskreduction interventions for sexually transmitted infections: research and practical implications

-78-. Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

		health-promoting effects over the course of time.
	•	There were many methodological limitations of the studies.
	•	The authors state that 'Findings suggest that clinic-based STI risk-reduction
		programmes have not yet reached a level of success that should be considered
		satisfactory'.
Gaps and inconsistencies	•	The authors report a plethora of methodological limitations such as sample size; lack
identified by the review:		of specified power analysis; the appropriateness of the control/comparison group;
		the absence of descriptive information regarding the randomisation process; the
		absence of appropriate follow up period; the absence of reporting rates of attrition;
		the absence of intention-to-treat.
Research recommendations:	•	More rigorous studies of clinic-based risk reduction interventions to reduce sexually
		transmitted infections need to be undertaken.
	•	Consider interventions based on paediatrician/healthcare worker group work with
		patients.
	•	Supplemental programmes ought to be considered to promote longer-term
		effectiveness.
	•	There is a need to improve research related to STI prevention technology.

Manhart L.E. and Homes K.K. (2005) Randomised controlled trials of individual-level, population-level and multilevel interventions for preventing sexually transmitted infections: What has worked? Data pool:

Data pool.	
Population	Any
Setting(s)	Any
Interventions	Behavioural interventions to reduce STIs
Searches	Searches were carried out in: MEDLINE, the Cochrane Central Register of controlled
	trials and unpublished presentations.
Selection/inclusion	Studies were included if they:
criteria	Used a randomised controlled trial.
	Included biological markers of a prospectively measure objective STI outcome.
	Were published in the English language.
	• Were Peer-reviewed literature or presentation at international conferences but not
	yet published.
Quality assessment	See selection/inclusion criteria.
Types of studies	Randomised controlled trials
Number of studies	15 studies included behavioural interventions (41 studies overall)
Outcomes reported and/or	STI rates; risky sexual behaviour
used for assessing	
'effectiveness'	
Review findings:	The review found that:
	Individual level
	• Risk-reduction counselling was the most common intervention used. These
	interventions resulted in a reduction in the rate of STIs, but were not statistically
	significant.
	One study that used ARRM and counselling without HIV testing showed no reduction
	in STI acquisition. However, it had low participation and follow-up rates.
	One counselling intervention study showed a non-significant reduction in HIV
	acquisition and also showed a significant reduction in unprotected anal intercourse.
	Group level
	Results of one study showed no difference in STI acquisition or risk behaviour
	between studies with a differing number of sessions.
	The majority of studies using small group interventions showed reductions in (study
	1) STI symptoms; incidence of gonococcal infection in men; (study 2) gonococcal
	and chlamydia infection in women; (study 3) new cases of STIs; (study 4) STIs and
	unplanned pregnancy. Only one study using small group interventions showed no
	difference between the intervention and comparison groups.
	One using information as well as condom distribution showed no difference between
	intervention and comparison groups.
	One study using information as well as male/female condom use showed a non-
	significant reduction in the incidence of STIs compared to the comparison group that
	had access to only male condoms.

C	ommunity level
	• The mixed modality intervention results showed that those receiving IEC
	showed a reduction in the HSV seroconversion and those receiving IEC and
	treatment showed a reduction in the incidence of both syphilis and gonorrhoea.
	As well as an increase in condom use (all groups).
	• One study that used a four component random assignment to school-based
	sexual health and reproductive health education, enhanced reproductive health
	services for youth, condom distribution, and community activities. Results
	showed no difference between intervention and comparison groups
N	on-RCT mentioned in the text, summary of findings show:
In	dividual level
(F	RCT)
	• Chart reviews, reported symptoms, and self-reported STIs all found no
	significant difference.
G	roup level
(F	RCT)
	Decreased incidence of STIs were reported in group interventions using video
	and discussion, role-play and condom negotiation (successful with men only).
	A cognitive behavioural intervention for MSM resulted in a non-significant
	reduction in unprotected and cox, however it also showed a significant increase
	A multiple back and internet in the institute of a second se
	A multi-behavioural intervention in mail military men resulted in a greater than
	80% reduction in STT recurrence.
	Non-significant reductions were found in repeat attendances to clinics providing
	group discussion and condoms.
(n	on-RCT)
	• Findings showed that condom distribution among brothel based CSWs resulted
	in a reduction in HIV and STIs
	Multi-component school-based interventions showed a decrease in self-reported
	STIs at aged 21 for the African American cohort.
(F	(CT)
	• Partner notification strategies (Peru) by either patient of healthcare worker
	showed no difference.
	Condom distribution in a hotel used by CSWs resulted in an increase in condom
	use.
	Overall no one theoretical framework was more successful than another
	No specific approach was more successful than another
	No specific approach was more successful than abouter.
	• All counselling sessions used at least two sessions however the number of
	sessions did not affect the success of the intervention.

Gaps and inconsistencies	Study quality
identified by the review:	Potential for publication bias
Research recommendations:	Due to the costs of intervention trials, criteria for prioritising and selecting intervention
	warranting evaluation are imperative.
	It is important to incorporate outcome and impact evaluations and cost-benefit analyses
	into future studies on effective interventions to ensure that limited resources are
	effectively utilised.

Ward D. J. (2004) Behavioural interventions to reduce the risk of sexually transmitted infections in Genitourinary Medicine Clinic patients: a systematic review Data pool:

٦

Data pool.			
Population	Any		
Setting(s)	Genitourinary medicine clinics (almost all in North America)		
Interventions	Behavioural interventions		
Searches	Searches were carried out in: MEDLINE, Cochrane database of systematic reviews, Database of abstracts of reviews if effectiveness (DARE), Health technology assessment (HTA), CINAHL, Embase, PsychINFO, Applied Social Sciences Index and Abstracts (ASSIA), Cochrane library of controlled clinical trials register (CCTR), National Research Register.		
Selection/inclusion	Studies were included if they:		
criteria	Used a randomised controlled trial.		
	 Had a study population recruited from those attending a GUM clinic or equivalent (public clinic offering free services and self-referral for the screening, diagnosis and treatment of STIs). Had a population that was not primarily recruited from those with known HIV infection or AIDS. The intervention was behavioural and was aimed at reducing the future likelihood of 		
	 acquiring an STI, but excluded the provision of HIV testing with or without counselling as the principle intervention. Outcomes were related to clinically or laboratory diagnosed STI rates, self-reported STI rates, or to self-report of quantifiable behavioural changes (e.g. condom use, number of partners etc.), but not attitudinal or knowledge based measures alone, and not related to partner notification alone. 		
Quality assessment	See selection/inclusion criteria. The authors also used a Jadad scoring system and included the method of randomisation, concealment, blinding, completeness of follow-up and the use of intention-to-treat analyses		
Types of studies	Randomised controlled trials		
Number of studies	14 studies		
Outcomes reported and/or	STIs; Condom use; number of partners etc.		
used for assessing 'effectiveness'			
Review findings:	 Four studies reported lower STI rates at follow-up in their intervention groups relative to the control group. The result was statistically significant for two of these studies. No clear pattern of intervention format or number of sessions and reported effects were noted [enhanced vs. control RR 0.78 (95% CI 0.65, 0.95), brief vs. control RR 0.82 (95% CI 0.68, 0.99)]. Duration of intervention was also not significant. 		

Gaps and inconsistencies • Few studies of behavioural interventions were identified by the review: Gaps and inconsistencies • Few studies of behavioural interventions were identified. • The quality of study reported in their generalisability, due to the population studied and the setting. • The quality of identified studies was frequently poor. • The quality of identified studies was frequently poor. • There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. • There was also showed low recruitment rates. • Research recommendations:		s one study (renya) presented statistically significant results in the reduction of one
 Results from self-reported STIs showed a non-significant reduction. Mixed results were reported for the number of characteristics of sexual partners. All but one study reported results that favoured increased consistent condom use among the intervention group and in two studies this is statistically significant (p<0.05). Only one study indicated a clear relationship between the intervention and an increase in the proportion of protected sexual encounters. Results show that those interventions using a theoretical framework based on social cognitive theory may have been more successful in increasing the rate of consistent condom use. Key point: "This review did not find that behavioural interventions consistently reduce STI rates in patients attending GUM clinics". "This review did find evidence that behavioural interventions may alter certain aspects of behaviour in the population of interest". Gaps and inconsistencies identified by the review: Few studies of behavioural interventions were identified. The results are limited in their generalisability, due to the population studied and the setting. The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervence. Studies also showed low recruitment rates. 		however, did not describe how they were found.
 Mixed results were reported for the number of characteristics of sexual partners. All but one study reported results that favoured increased consistent condom use among the intervention group and in two studies this is statistically significant (p<0.05). Only one study indicated a clear relationship between the intervention and an increase in the proportion of protected sexual encounters. Results show that those interventions using a theoretical framework based on social cognitive theory may have been more successful in increasing the rate of consistent condom use. Key point: "This review did not find that behavioural interventions consistently reduce STI rates in patients attending GUM clinics". "This review did find evidence that behavioural interventions may alter certain aspects of behaviour in the population of interest". Gaps and inconsistencies identified by the review: Few studies of behavioural interventions were identified. The results are limited in their generalisability, due to the population studied and the setting. The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. 		Results from self-reported STIs showed a non-significant reduction.
 All but one study reported results that favoured increased consistent condom use among the intervention group and in two studies this is statistically significant (p<0.05). Only one study indicated a clear relationship between the intervention and an increase in the proportion of protected sexual encounters. Results show that those interventions using a theoretical framework based on social cognitive theory may have been more successful in increasing the rate of consistent condom use. Key point: "This review did not find that behavioural interventions consistently reduce STI rates in patients attending GUM clinics". "This review did find evidence that behavioural interventions may alter certain aspects of behaviour in the population of interest". Few studies of behavioural interventions were identified. The results are limited in their generalisability, due to the population studied and the setting. The applicability of study results to UK practice also relates to the feasibility of introducing the experimental intervention and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. 		• Mixed results were reported for the number of characteristics of sexual partners.
among the intervention group and in two studies this is statistically significant (p<0.05). • Only one study indicated a clear relationship between the intervention and an increase in the proportion of protected sexual encounters. • Results show that those interventions using a theoretical framework based on social cognitive theory may have been more successful in increasing the rate of consistent condom use. Key point: "This review did not find that behavioural interventions consistently reduce STI rates in patients attending GUM clinics". "This review did find evidence that behavioural interventions may alter certain aspects of behaviour in the population of interest". Gaps and inconsistencies identified by the review: • Few studies of behavioural interventions were identified. • The results are limited in their generalisability, due to the population studied and the setting. • The results are limited in their generalisability, due to the population studied and the setting. • The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. • The quality of identified studies was frequently poor. • There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. • Studies also showed low recruitment rates. • There is a need for further research in this topic.		• All but one study reported results that favoured increased consistent condom use
(p<0.05). • Only one study indicated a clear relationship between the intervention and an increase in the proportion of protected sexual encounters. • Results show that those interventions using a theoretical framework based on social cognitive theory may have been more successful in increasing the rate of consistent condom use. Key point: "This review did not find that behavioural interventions consistently reduce STI rates in patients attending GUM clinics". "This review did find evidence that behavioural interventions may alter certain aspects of behaviour in the population of interest". Gaps and inconsistencies identified by the review: Few studies of behavioural interventions were identified. The results are limited in their generalisability, due to the population studied and the setting. The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. Research recommendations: There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM Future studies should aim to include patients who are representative of a UK GUM 		among the intervention group and in two studies this is statistically significant
 Only one study indicated a clear relationship between the intervention and an increase in the proportion of protected sexual encounters. Results show that those interventions using a theoretical framework based on social cognitive theory may have been more successful in increasing the rate of consistent condom use. Key point: "This review did not find that behavioural interventions consistently reduce STI rates in patients attending GUM clinics". "This review did find evidence that behavioural interventions may alter certain aspects of behaviour in the population of interest". Gaps and inconsistencies identified by the review: Few studies of behavioural interventions were identified. The results are limited in their generalisability, due to the population studied and the setting. The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. Research recommendations: There is a need for further research in this topic. 		(p<0.05).
increase in the proportion of protected sexual encounters. Results show that those interventions using a theoretical framework based on social cognitive theory may have been more successful in increasing the rate of consistent condom use. Key point: "This review did not find that behavioural interventions consistently reduce STI rates in patients attending GUM clinics". "This review did find evidence that behavioural interventions may alter certain aspects of behaviour in the population of interest". Gaps and inconsistencies identified by the review: Few studies of behavioural interventions were identified. The results are limited in their generalisability, due to the population studied and the setting. The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. Research recommendations: There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM 		• Only one study indicated a clear relationship between the intervention and an
 Results show that those interventions using a theoretical framework based on social cognitive theory may have been more successful in increasing the rate of consistent condom use. Key point: "This review did not find that behavioural interventions consistently reduce STI rates in patients attending GUM clinics". "This review did find evidence that behavioural interventions may alter certain aspects of behaviour in the population of interest". Gaps and inconsistencies identified by the review: Few studies of behavioural interventions were identified. The results are limited in their generalisability, due to the population studied and the setting. The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. Research recommendations: There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM 		increase in the proportion of protected sexual encounters.
cognitive theory may have been more successful in increasing the rate of consistent condom use. Key point: "This review did not find that behavioural interventions consistently reduce STI rates in patients attending GUM clinics". "This review did find evidence that behavioural interventions may alter certain aspects of behaviour in the population of interest". Gaps and inconsistencies identified by the review: • Few studies of behavioural interventions were identified. • The results are limited in their generalisability, due to the population studied and the setting. • The results of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. • The quality of identified studies was frequently poor. • There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. • Studies also showed low recruitment rates. • There is a need for further research in this topic.		• Results show that those interventions using a theoretical framework based on social
condom use. Key point: "This review did not find that behavioural interventions consistently reduce STI rates in patients attending GUM clinics". "This review did find evidence that behavioural interventions may alter certain aspects of behaviour in the population of interest". Gaps and inconsistencies identified by the review: • Few studies of behavioural interventions were identified. • The results are limited in their generalisability, due to the population studied and the setting. • The results are limited in their generalisability, due to the population studied and the setting. • The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. • The quality of identified studies was frequently poor. • There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. • Studies also showed low recruitment rates. Research recommendations: • There is a need for further research in this topic.		cognitive theory may have been more successful in increasing the rate of consistent
Key point: "This review did not find that behavioural interventions consistently reduce STI rates in patients attending GUM clinics". "This review did find evidence that behavioural interventions may alter certain aspects of behaviour in the population of interest". Gaps and inconsistencies identified by the review: Few studies of behavioural interventions were identified. The results are limited in their generalisability, due to the population studied and the setting. The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. Research recommendations: There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM 		condom use.
Gaps and inconsistencies • Few studies of behavioural interventions were identified. Gaps and inconsistencies • Few studies of behavioural interventions were identified. • The results are limited in their generalisability, due to the population studied and the setting. • The results are limited in their generalisability, due to the population studied and the setting. • The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. • The quality of identified studies was frequently poor. • There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. • Studies also showed low recruitment rates. Research recommendations: • There is a need for further research in this topic.		
Gaps and inconsistencies • Few studies of behavioural interventions were identified. identified by the review: • Few studies of behavioural interventions were identified. • The results are limited in their generalisability, due to the population studied and the setting. • The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. • The quality of identified studies was frequently poor. • There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. • Studies also showed low recruitment rates. Research recommendations: • There is a need for further research in this topic. • Future studies should aim to include patients who are representative of a UK GUM		Key point: "This review did not find that behavioural interventions consistently reduce STT
"This review did find evidence that behavioural interventions may alter certain aspects of behaviour in the population of interest". Gaps and inconsistencies identified by the review: Few studies of behavioural interventions were identified. The results are limited in their generalisability, due to the population studied and the setting. The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. Research recommendations: There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM 		rates in patients attending GUM clinics".
Gaps and inconsistencies • Few studies of behavioural interventions were identified. identified by the review: • Few studies of behavioural interventions were identified. • The results are limited in their generalisability, due to the population studied and the setting. • The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. • The quality of identified studies was frequently poor. • There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. • Studies also showed low recruitment rates. Research recommendations: • There is a need for further research in this topic. • Future studies should aim to include patients who are representative of a UK GUM		"This review did find evidence that behavioural interventions may alter certain aspects of
Gaps and inconsistencies Few studies of behavioural interventions were identified. The results are limited in their generalisability, due to the population studied and the setting. The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. Research recommendations: There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM 		behaviour in the population of interest".
Gaps and inconsistencies • Few studies of behavioural interventions were identified. identified by the review: • The results are limited in their generalisability, due to the population studied and the setting. • The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. • The quality of identified studies was frequently poor. • There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. • Studies also showed low recruitment rates. Research recommendations: • There is a need for further research in this topic.		
identified by the review: The results are limited in their generalisability, due to the population studied and the setting. The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. Research recommendations: There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM 	Gaps and inconsistencies	Few studies of behavioural interventions were identified.
 The results are initial in their generalisability, due to the population studied and the setting. The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. Research recommendations: There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM 	identified by the review:	• The results are limited in their generalisability, due to the population studied and the
 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM 		setting.
introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. • The quality of identified studies was frequently poor. • There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. • Studies also showed low recruitment rates. Research recommendations: • There is a need for further research in this topic. • Future studies should aim to include patients who are representative of a UK GUM		
services are over stretched, overcrowded and under resourced. • The quality of identified studies was frequently poor. • There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. • Studies also showed low recruitment rates. Research recommendations: • There is a need for further research in this topic. • Future studies should aim to include patients who are representative of a UK GUM		• The applicability of study results to UK practice also relates to the feasibility of
 The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. Research recommendations: There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM 		• The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current
 There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. Research recommendations: There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM 		• The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced.
studies and many reported a low rate of intervention adherence. • Studies also showed low recruitment rates. Research recommendations: • There is a need for further research in this topic. • Future studies should aim to include patients who are representative of a UK GUM		 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor.
 Studies also showed low recruitment rates. Research recommendations: There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM 		 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five
Research recommendations: • There is a need for further research in this topic. • Future studies should aim to include patients who are representative of a UK GUM		 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence.
 Future studies should aim to include patients who are representative of a UK GUM 		 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates
• I dure studies should aim to include patients who are representative of a OK Obin	Research recommendations:	 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. There is a need for further research in this topic.
clinic and include arouns who are most affected by STIs and suffer inequality in their	Research recommendations:	 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. There is a need for further research in this topic. Euture studies should aim to include patients who are representative of a LIK GLIM.
sexual health.	Research recommendations:	 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM clinic and include groups who are most affected by STIs and suffer inequality in their
Any proposed clinical trails should use rigorous methodology, including using a	Research recommendations:	 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM clinic and include groups who are most affected by STIs and suffer inequality in their sexual health.
control arm that simulates typical current UK practice, minimising bias by improving	Research recommendations:	 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM clinic and include groups who are most affected by STIs and suffer inequality in their sexual health. Any proposed clinical trails should use rigorous methodology, including using a
methods of blinding, and including laboratory based or biological diagnoses.	Research recommendations:	 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM clinic and include groups who are most affected by STIs and suffer inequality in their sexual health. Any proposed clinical trails should use rigorous methodology, including using a control arm that simulates typical current UK practice, minimising bias by improving
Trial should also consider cluster-randomisation by clinic or another natural unit	Research recommendations:	 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM clinic and include groups who are most affected by STIs and suffer inequality in their sexual health. Any proposed clinical trails should use rigorous methodology, including using a control arm that simulates typical current UK practice, minimising bias by improving methods of blinding, and including laboratory based or biological diagnoses.
Future research should consider addressing both clinical and cost-effectiveness and	Research recommendations:	 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM clinic and include groups who are most affected by STIs and suffer inequality in their sexual health. Any proposed clinical trails should use rigorous methodology, including using a control arm that simulates typical current UK practice, minimising bias by improving methods of blinding, and including laboratory based or biological diagnoses. Trial should also consider cluster-randomisation by clinic or another natural unit.
consider the likely uptake of behavioural interventions in this setting.	Research recommendations:	 The applicability of study results to UK practice also relates to the feasibility of introducing the experimental interventions into a GUM clinic setting as current services are over stretched, overcrowded and under resourced. The quality of identified studies was frequently poor. There was also evidence of differential follow-up rates between study arms in five studies and many reported a low rate of intervention adherence. Studies also showed low recruitment rates. There is a need for further research in this topic. Future studies should aim to include patients who are representative of a UK GUM clinic and include groups who are most affected by STIs and suffer inequality in their sexual health. Any proposed clinical trails should use rigorous methodology, including using a control arm that simulates typical current UK practice, minimising bias by improving methods of blinding, and including laboratory based or biological diagnoses. Trial should also consider cluster-randomisation by clinic or another natural unit. Future research should consider addressing both clinical and cost-effectiveness and

Ward D.J. et al. (2005) Reducing the risk of sexually transmitted infections in genitourinary medicine clinic patients: a systematic review and meta-analysis of behavioural interventions

		-		
Е)at	аι	nor	٦ŀ

Data pool:			
Population	Any		
Setting(s)	Genitourinary medicine clinics (almost all in North America)		
Interventions	Behavioural interventions		
Searches	Searches were conducted in:		
	MEDLINE, CINAHL, Embase, PsychINFO, Applied Social Sciences Index and Abstracts		
	(ASSIA), Cochrane library of controlled clinical trials register (CCTR), National Research		
	Register (1966 to January 2004).		
Selection/inclusion	Studies were included if:		
criteria	• They were randomised controlled trials of behavioural interventions aimed to reduce		
	the risk of STIs in patients attending GUM, or equivalent, clinics.		
Quality assessment	See selection/inclusion criteria.		
	The authors also used a Jadad scoring system and included the method of		
	randomisation, concealment, blinding, completeness of follow-up and the use of		
	intention-to-treat analyses.		
Types of studies	Randomised controlled trials		
Number of studies	14 studies		
Outcomes reported and/or	STI rates; changes is risky sexual behaviour		
used for assessing			
Review findings:	This study is a journal article using many of the 2004 Ward et al. report.		
	Four studies reported lower STI rates at follow-up in their intervention groups relative		
	to the control group. The result was statistically significant for two of these studies.		
	Pooled results do not indicate an overall effect (RR 1.00 (95% CI 0.81 to 1.23)).		
	However, visual examination of the Forest plot and statistical testing suggest		
	heterogeneity between trials, as such one ought not to rely heavily on the summary		
	Trials reporting significant effects had the prostent adherence and follow up rates		
	Inals reporting significant effects had the greatest adherence and follow up rates. They also appended mean analysis off and had been been appended appendix and follow up rates.		
	different		
	unerent.		
	Pormat and length of intervention was not significant.		
	Duration of intervention was found no to be significant. Theoretical framework was related to effectiveness, although peoled results were not		
	 medical framework was related to enectiveness, although pooled results were not significant. 		
	Four studies considered clinically diagnosed STIs. One study found no reduction and		
	others observed increased rates of STIs in the intervention groups. Pooled results		
	show an overall effect (RR 1.23 (1.01 to 1.50)) with no evidence of heterogeneity.		
	Studies considering self-reported diagnoses showed no significant effects.		
	• Six out of seven trials reporting consistent condom use observed a greater increase		

		in their intervention groups than controls.
	•	Trials also reported increases (non-significant) in condom protected sexual
		encounters, inconsistent effects on refusing unsafe sex, and using condoms with
		recent partners.
	•	Five out of seven trials reported fewer sexual partners among the intervention
		arouns than controls. However, two also reported fewer partners among the control
		group and controls. However, two also reported rewer partners among the control
		group.
Gaps and inconsistencies	٠	Trial quality was poor
identified by the review:	•	Only RCTs were included
	•	Publication bias was found for the laboratory confirmed infections (primary outcome).
Research recommendations:	٠	There is a need for research on behavioural interventions in a UK setting, where new
		approaches to reducing infection rates are urgently required.
	•	Information is needed on the likely effect size, acceptability, and cost-effectiveness
		of introducing behavioural interventions into the UK GUM setting.
1		
	•	Future studies should develop new interventions based on those that have been
	•	Future studies should develop new interventions based on those that have been
	•	Future studies should develop new interventions based on those that have been successful in US settings.

Category two reviews

Pedlow C.T. and Carey M.P. (2004) Developmentally appropriate sexual risk reduction interventions for adolescents: rationale review of intervention and recommendations for research and practice.

Sexual risk reduction interventions to change behaviour		
Searches were conducted using Medline, PsychINFO and Cinahl		
Randomised controlled trials were included if they:		
e February		
trial design;		
or younger s with pre- ondom use effective in equency of promoting sexual risk o achieved owever, no difficulty in re effective for positive ree studies unprotected or example successful titation.		
or svo effe equ pr sex o a ow diff re or cee unp or su itia		

	٠	Improvements in condom application skills (observing the application) were not
		associated with increased condom use and non-significant findings for condom use
		skills were associated with risk reduction outcomes in another study.
	•	Many studies provided appropriate communication skills training (21/25). Six studies
		measured sexual communication behavioural skills, all six demonstrated significant
		improvements at follow up and five of the six were also effective at reducing sexual
		risk behaviour.
	•	Mixed results were reported for those studies that provided risk reduction information
		to peers as a means to reduce peer pressure for risky sex.
	•	Studies measuring peer norms for condom use or abstinence showed improved peer
		norms. Studies with improved peer norms for sexual risk behaviour were also
		effective at delaying onset of sex and improving condom use.
	•	A school-based intervention addressing social influences at home, in school and in
		the community still showed effects two and a half years later on improved teen-
		parent communication, attitudes to condom use, increased condom use and less
		unprotected sex.
	•	Mixed results were observed in those interventions that considered parental
		influence.
Gaps and inconsistencies	•	Literature failed to measure the onset of menarche and its implication on sexual
identified by the review:		activity.
	•	Measurement for risk perception varied from study to study, greater use of
		standardised measures in the future will verify the importance of risk perception.
	•	Only one third of interventions provided multiple intervention exercises that were
		cognitively appropriate for adolescents.
	•	Only two interventions included exercises to assist teens in considering their risk
		behaviour.
	•	Only one intervention illustrated abstract concepts with concrete real-life examples to
		be consistent with adolescents' limited ability to hypothesise.
	•	Only one study measured a marker of cognitive development.
	•	Although many risk reduction interventions for youth have been implemented, their
		outcomes have been inconsistent and few studies have been replicated.
	•	Most of the theories applied were developed for adult interventions.
Research recommendations:	•	Modifying existing theoretical models will advance intervention development,
		implementation and evaluation.
	•	Researchers ought to use multiple strategies compatible with adolescent cognitive
		ability, including managing emotions associated with sexual risk reduction.
	•	Researchers ought to include measures of cognitive functioning, future-time
		perspective, and decision-making in formative research.
	•	Risk perception ought to be linked to actual behaviour and reporting findings for
		sexually inexperienced versus experienced youth.
	•	I he inclusion of communication skills and negotiation skills are encouraged.
	•	I he inclusion of communication skills and negotiation skills are encouraged. It is important to provide multiple opportunities for skills training that are consistent

•	Booster sessions may reinforce the skills acquisition as youth's relationships change.
•	There is a need to develop new, more effective risk reduction interventions that
	address biological, psychological and social influences on sexual behaviour.
•	Studies that compare different intervention strategies, identify effective components
	and measure process outcomes of developmental factors are needed.

Robin L. et al. (2004) Behavioural interventions to reduce incidence of HIV, STD and pregnancy among adolescents: A decade in review.

Data pool:	
Population	Adolescents and ethnic minorities
Setting(s)	Any
Interventions	Behavioural interventions to reduce sexual risk-taking
Searches	Six electronic databases were searched: Medline, Psychlit, Popline, ERIC, Sociofile and CHID.
	Unspecified manual searches were also carried out.
Selection/inclusi	Studies (randomised control trials or quasi-experimental design with controls) were included if
on criteria	they:
	Specified a theoretical basis for the intervention programme.
	• Provided information about the intervention (e.g. duration, content, facilitators).
	Defined clear aims.
	• Random assignment or matched control groups using a quasi-experimental design that
	matched units through stratification of risk behaviours and demographic variables.
	If researchers controlled statistically for any differences at baseline.
	• Studies were included if they had more than 16 participants per condition, followed
	participants for at least four weeks post-intervention, or had immediate pre- or post-test for
	interventions lasting four months or longer.
	• Had attrition rates of less than 40% at follow-up four weeks after the end of the
	intervention.
	• Studies were also included if they measured sexual intention for those 13 years or
	younger.
Quality	None
assessment	
Types of studies	Randomised control trials or quasi-experimental design with controls.
Number of	24 studies
studies	
studies Outcomes reported	Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours;
studies Outcomes reported and/or used for	Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence.
studies Outcomes reported and/or used for assessing	Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence.
studies Outcomes reported and/or used for assessing 'effectiveness'	Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence.
studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	 Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence. Most interventions were based on multiple theories however no clear pattern of study
studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	 Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence. Most interventions were based on multiple theories however no clear pattern of study method presented itself, the most commonly used theories were: social cognitive theories;
studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	 Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence. Most interventions were based on multiple theories however no clear pattern of study method presented itself, the most commonly used theories were: social cognitive theories; the Health Belief Model; social learning theories; and social influence theories.
studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	 Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence. Most interventions were based on multiple theories however no clear pattern of study method presented itself, the most commonly used theories were: social cognitive theories; the Health Belief Model; social learning theories; and social influence theories. Among commonly measured behaviours (8 out of 12 studies), condom use was affected
studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	 Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence. Most interventions were based on multiple theories however no clear pattern of study method presented itself, the most commonly used theories were: social cognitive theories; the Health Belief Model; social learning theories; and social influence theories. Among commonly measured behaviours (8 out of 12 studies), condom use was affected most consistently, and delayed initiation of sexual intercourse was least affected (4 of 11
studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	 Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence. Most interventions were based on multiple theories however no clear pattern of study method presented itself, the most commonly used theories were: social cognitive theories; the Health Belief Model; social learning theories; and social influence theories. Among commonly measured behaviours (8 out of 12 studies), condom use was affected most consistently, and delayed initiation of sexual intercourse was least affected (4 of 11 studies).
studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	 Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence. Most interventions were based on multiple theories however no clear pattern of study method presented itself, the most commonly used theories were: social cognitive theories; the Health Belief Model; social learning theories; and social influence theories. Among commonly measured behaviours (8 out of 12 studies), condom use was affected most consistently, and delayed initiation of sexual intercourse was least affected (4 of 11 studies). Among measures less commonly used, pregnancy or partner impregnation was recorded.
studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	 Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence. Most interventions were based on multiple theories however no clear pattern of study method presented itself, the most commonly used theories were: social cognitive theories; the Health Belief Model; social learning theories; and social influence theories. Among commonly measured behaviours (8 out of 12 studies), condom use was affected most consistently, and delayed initiation of sexual intercourse was least affected (4 of 11 studies). Among measures less commonly used, pregnancy or partner impregnation was recorded. Three studies recorded negative results; increased likelihood of males engaging in sexual
studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	 Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence. Most interventions were based on multiple theories however no clear pattern of study method presented itself, the most commonly used theories were: social cognitive theories; the Health Belief Model; social learning theories; and social influence theories. Among commonly measured behaviours (8 out of 12 studies), condom use was affected most consistently, and delayed initiation of sexual intercourse was least affected (4 of 11 studies). Among measures less commonly used, pregnancy or partner impregnation was recorded. Three studies recorded negative results; increased likelihood of males engaging in sexual intercourse in the last month compared to the control group; increased reports of
studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	 Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence. Most interventions were based on multiple theories however no clear pattern of study method presented itself, the most commonly used theories were: social cognitive theories; the Health Belief Model; social learning theories; and social influence theories. Among commonly measured behaviours (8 out of 12 studies), condom use was affected most consistently, and delayed initiation of sexual intercourse was least affected (4 of 11 studies). Among measures less commonly used, pregnancy or partner impregnation was recorded. Three studies recorded negative results; increased likelihood of males engaging in sexual intercourse in the last month compared to the control group; increased reports of pregnancy and STDs; less contraceptive use at most recent sexual encounter for females
studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	 Abstinence; reduced sexual activity or number of sexual partners; less risky sexual behaviours; reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence. Most interventions were based on multiple theories however no clear pattern of study method presented itself, the most commonly used theories were: social cognitive theories; the Health Belief Model; social learning theories; and social influence theories. Among commonly measured behaviours (8 out of 12 studies), condom use was affected most consistently, and delayed initiation of sexual intercourse was least affected (4 of 11 studies). Among measures less commonly used, pregnancy or partner impregnation was recorded. Three studies recorded negative results; increased likelihood of males engaging in sexual intercourse in the last month compared to the control group; increased reports of pregnancy and STDs; less contraceptive use at most recent sexual encounter for females who were sexually inexperienced at baseline or; less contraception efficiency among

	Once studies were aggregated into the 17 programmes and variants, 10 programmes had			
	positive effects, one had mixed effects and four had null effects and two had negative			
	effects.			
	• Studies with positive effects were published after 1995, included both males and females,			
	targeted African American youth and took place in schools.			
	Programmes that produced positive effects used trained adult facilitators.			
	• Effective programmes included content that was specific to reducing sexual risk behaviour			
	(e.g. sexual refusal strategies and condom-use skills).			
	• Effective programmes employed interactive and participatory educational strategies.			
	• Programmes that emphasised skills to reduce specific behaviours, interventions more			
	generally targeted toward increasing youth resiliency and competencies show promising			
	approaches to reducing sexual risk behaviour.			
Gaps and	• Limitations included differences among groups that were reported but not controlled for in			
inconsistencies	the analysis, and the unit of randomisation not matching the unit of analysis.			
identified by the review:	• Limits on the strength of effect sizes due to the differences in follow-up periods and			
	possible limits on information available from the primary studies included.			
Research				
recommendations:	Future programmes ought to focus upon appropriate skills, adapting programmes for length,			
	being clear what constitutes a given programme and decide who ought to facilitate them			
	Resiliency-based programmes ought to be further explored with regard to reducing the sexual			
	risk behaviour of adolescents.			
	Researchers should design studies that will clearly reveal which programme characteristics			
	drive positive effects in sexual risk-reduction.			

Category three reviews

Data pool:			
Population	Adolescents (US)		
Setting(s)	Any		
Interventions	Parental interventions		
Searches	Not specified		
Selection/inclusion	Observational studies of parental monitoring that report the impact on risk behaviour and		
criteria	clinic and community based STD/HIV interventions. No other criteria specified.		
Quality assessment	Not specified		
Types of studies	Observational studies		
Number of studies	Unclear		
Outcomes reported and/or	Condom use; risky sexual behaviour		
used for assessing			
'effectiveness'			
Review findings:	 Health risk behaviours: At least two studies have shown that lack of parental monitoring has been associated with less contraceptive use. Sexual risk behaviours: empirical investigations have found associations between lack of parental monitoring and 1) unprotected sexual activity, 2) earlier initiation of sexual activity, 3) sex with non-monogamous male partners among adolescent females. Also, noted was an increase in STDs, multiple sexual partners and risky sexual partners. Increased parental monitoring has been shown to help delay sexual initiation, report fewer sexual partners and less frequent sexual intercourse Other family-related variables: One study of African American females showed lack of parental monitoring was associated with 1) less frequent communication between adolescents and parents, 2) lower levels of perceived family support, 3) a living arrangement that did not include both parents living in the same home. Clinic-level interventions: counselling sessions with adolescents' parents have been recommended by two studies. Community-level interventions – community based organisations (churches, agencies, schools) could provide programmes for both adolescents and parents together, as well as support groups and discussion. 		
Gaps and inconsistencies	• Reliance on the validity of adolescents self-reports for assessment of outcome		
identified by the review:	measures.		
	• Psychological interventions that target sexual risk behaviours of adolescents have		
	failed to achieve lasting effects.		
Research recommendations:	 Families can be mobilised to communicate important values, model appropriate behaviours, monitor adolescents' behaviours, and encourage protective behaviours. Family-level interventions could be designed to provide parents with guidance for their adolescents. 		

DiClemente R.J., et al., (2002) Enhancing STD/HIV prevention among adolescents: the importance of parental monitoring.

Research should encourage strengthening the family unit as an intervention strategy
designed to promote adolescents' adoption and maintenance of health protective
behaviours.

Data pool:	
Population	Young people and sex education in schools
Setting(s)	Any
Interventions	School-based sex education programmes
Searches	Not specified
Selection/inclusion	School-based sex education programmes, no other specified.
criteria	
Quality assessment	Not specified
Types of studies	Not specified
Number of studies	Unclear
Outcomes reported and/or	Condom use, risky sexual behaviour and abstinence
used for assessing	
'effectiveness'	
Review findings:	 Interventions carried out within the ARRM framework have been shown to correlate with condom use and perceived control over behaviour (heterosexual studies). These results have also been found in studies applying theories of reasoned action, planned behaviour, and social cognitive theory. As such, 'education design must target psychological change beyond increases in knowledge', also, 'theories of sexual behaviour and sex education interventions demonstrates that despite the particularities of sexual behaviour, sexual health promotion should be informed by cognitive theories that have been successfully applied to a variety of other health-related behaviours'. Social rehearsal on sexual negotiation ought to be learned through instruction and guided practice (SHARE and Long Live Love programmes). Long Live Love showed favourable effects on social-cognitive mediators of consistent condom use and risk-reduction behaviour Reviews of abstinence programmes show that they have not succeeded in changing attitudes towards abstinence. However, very few abstinence programmes have been rigorously evaluated therefore results need to be interpreted cautiously. Health promotion should be evidence-based, needs-driven, and ecological in its perspective. In relation to the cognitive determinants of health behaviour, sexual health promotion draws on the same research base as other areas of health promotion. Teaching social skills relevant to sexual behaviour in classroom settings requires specialist expertise both in programme design and delivery by teachers or facilitators.
Gaps and inconsistencies	See Review findings
identified by the review:	
Research recommendations:	Researchers should continue to conduct and share the results of needs assessment
	research.
	• Researchers should target those who could potentially change the policy context in

Schaalma H. P. (2004) Sex education as health promotion: what does it take?

	which sex education is designed.
•	Schaalma concludes that, "To facilitate widespread adoption of effective sexual
	health promotion programmes, health promotion planners could focus on mobilising
	parental support for comprehensive sex education. This would involve designing
	community empowerment programmes that would be deemed effective if they
	prompted parents to act politically to demand comprehensive sex education (e.g. by
	writing to politicians, or schools and media advocacy)".

Data pool:							
Population	Young people (US)						
Setting(s)	School-based in US						
Interventions	Condom distribution in schools; abstinence						
Searches	Not stated						
Selection/inclusion	Not stated						
criteria							
Quality assessment	Not stated						
Types of studies	Unclear						
Number of studies	8 studies						
Outcomes reported and/or	Condom use; sexual initiation.						
used for assessing							
'effectiveness'							
Review findings:	• Findings from studies examining condom distribution on schools showed that						
	condom availability was not associated with an increase in teen sexual activity.						
	• Data also showed that most programmes indicated increased use of condoms by						
	sexually active young people.						
	• Parental consent is usually required for students to participate in school-base						
	condom distribution schemes (passive or active).						
	Counselling teens prior to condom distribution has also been used in some schools.						
	• The method of distribution is important to the success of the programme in schools,						
	free distribution in baskets as opposed to vending machines resulted in more uptake.						
	• The cost of condoms could be a barrier to condom use as one study showed a						
	reduction in uptake when a charge was added to each condom (after is had been						
	freely available). It also resulted in a decrease in the number of sites participating,						
	and a decrease in reported condom use among individuals with two or more						
	partners.						
Gaps and inconsistencies	None specified						
identified by the review:							
Research recommendations:	Recommend that barriers to condom use be eliminated and that condoms should be free						
	of charge (Cohen et al, 1999).						
	Schmiedl concludes that, "With the appropriate resources, support, and consideration of						
	the factors reviewed, however, school nurses can develop successful school-based						
	condom disbursement programs that effectively reduce teen's exposure to STDs and HIV						
	and that decrease teen pregnancy".						

Schmiedl R. (2004) School-based condom availability programs

Data pool:							
Population	Adolescents						
Setting(s)	Any (US)						
Interventions	Parental monitoring						
Searches	Not specified						
Selection/inclusion	Not specified						
criteria							
Quality assessment	Not specified						
Types of studies	Uncertain						
Number of studies	Unclear						
Outcomes reported and/or	Condom use; number of sexual partners; risky sexual behaviour						
used for assessing							
'effectiveness'							
Review findings:	Parental monitoring, Parent-adolescent communication and perceived family suppor						
	have been associated with safer sex cognitions and behaviour.						
	• One study found that virginal adolescents who perceived maternal approval of birth						
	control are more likely to initiate sexual activity over the following year. Those who						
	perceive maternal disapproval are more likely to remain virgins, delay initiation, have						
	less intercourse, use contraception, and not become pregnant.						
	Peer norms affect the sexual behaviour of young people.						
	• Abstinence only programmes are not effective in delaying the onset of intercourse.						
	School-based condom availability may or may not increase condom use.						
Gaps and inconsistencies	None stated						
identified by the review:							
Research recommendations:	Programmes designed to reduce sexually transmitted infections in adolescents should						
	take a focused integrated approach that addresses the modifiable biological, cognitive,						
	psychologic, behavioural, and social issues affecting adolescent sexual risk.						

Shrier L.A. (2004) Sexually transmitted diseases in adolescents: biologic, cognitive, psychologic, behavioural, and social issues.

Appendix E. Details of the primary studies included in each review.

Table 2. Category 1 review. DiClemente et al, 2004.

Author		Title							
	DiClemente R.J. et al., 2004	A programr interventio	natic and	methodologic review an xually transmitted infecti	d synthesis ons: resear	of clinic-base ch and practio	ed risk-reduction ce implications		
	Type of intervention	Author	Study Design	Outcome	Theory	Duration	Target group		
Individual level	Observational learning: audiotape risk-assessment and education with Dr follow-up	Boekeloo etal., (1999)	RCT	Condom use at last intercourse	SCT, TRA	One-time	Adolescents/mostly African American. Physician's office.		
	Observational learning, direct feedback: culturally specific STD education, condom attitudes and skills	DeLameter et al., (2000)	RCT	Condom use knowledge/Self- efficacy and intent to use condoms with regular partner	SRMIB, SET	One-time	Adolescents/African American. STD clinic.		
	Observational learning, direct feedback: STD education, condom efficacy, information on drug and needle use	Mansfield et al., (1993)	RCT	condom use; number of sexual partners; sexually active	None stated	One-time	multi-ethnic adolescents. Physician's visit.		
	Tailored material	Scholes et al., (2003)	RCT	Condom use	SST	Twice	multi-ethnic adolescents. Postal.		
	Observational learning, direct feedback:	Schrier et al., (2001)	RCT	Condom use; new partner	SCT, ТТМ	One-time	multi-ethnic adolescents. Clinic.		
	Observational learning, role play, direct feedback: Interactive group sessions focused on gender and ethnic pride, HIV, safe sex, healthy relationships	DiClemente et al., (2004)	RCT	Consistent condom use, new vaginal partner in last 30 days, self-reported pregnancy	SCT, TGP	Four consecutive Saturdays	Adolescents/African American. Community health agency.		
evel	Observational learning, role play, direct feedback: STD/HIV education, favourable attitudes towards condoms and partner communication	Gillmore et al., (1997)	RCT	Safe sex behaviour; condom use; number of sexual partners	SCT, TRA	Two sessions	Adolescents/African American and Caucasian. STD clinic & juvenile detention centre.		
Group I	Observational learning, role play, direct feedback: decision- making skills, goal and social skills, acceptance of negative thoughts	Metzler et al., (2000)	RCT	Condom use; number of sexual partners	SCT, IMBST	Five sessions	Adolescents/ white. STD clinic.		
	Observational learning, role play, direct feedback: STD education, condom negotiation and skills training, positive attitude towards condoms	Orr et al., (1996)	RCT	Condom use	НВМ	One-time	Multi-ethnic adolescents. Family planning and STI clinic.		
	Observational learning, role play, direct feedback: education and behaviour skills training, including condom use, assertion, refusal etc.	St. Lawrence et al., (1995)	RCT	Condom protected intercourse	SCT, IMBST	8 weekly meetings	Adolescents/African American. Health service.		

	Author	Title							
Mar	hart L.E. and Holmes K.K., 2005	Randomised controlled trials of individual-level population-level, and multi-level interventions for preventing sexually transmitted infections: what has worked?							
	Type of intervention	Author	Study Design	Outcome	Theory	Duration	Target group		
	Experimental, cognitive- behavioural intervention	Boyer et al. (1997)	RCT	STI acquisition	ARRM	Four sessions	Heterosexual men and women attending STI clinic		
	Individual risk reduction counselling	Kamb et al. (1998)	RCT	Reduction in STI incident	TRA, SCT	Four sessions	Heterosexual HIV- negative men and women attending clinic		
	Individual risk reduction counselling	Peterman et al. (2000)	RCT	Reduction in STI incident	TRA, SCT	Four sessions	Heterosexual HIV- negative men and women attending clinic		
Individual level	Voluntary counselling and testing (VCT)	VCT Efficacy study group (2000)	RCT	Reduction in STI incident	Client- centred HIV counselling model	Not specified	Individuals and couples from Tanzania, Kenya and Trinidad		
	Voluntary counselling and testing (VCT)	EXPLORE Study Group (2004)	RCT	HIV acquisition	None stated	10 sessions	MSM in the US		
	Unclear	El-Bassel et al. (2003)	RCT	Behavioural	None stated	Not stated	Clinic based		
	Unclear	James et al. (1998)	RCT	Behavioural	None stated	Not stated	Clinic based		
	Observational learning, direct feedback:	Schrier et al. (2001)	RCT	Condom use; new partner	SCT, TTM	One-time	multi-ethnic adolescents. Clinic.		
	Partner notification	Peterman et al. (1997)	RCT	Locating partners within 2 days	None stated	Not stated	People with syphilis		
	Small-group counselling	Branson et al. (1998)	RCT	Incident of gonorrhoea, chlamydia, HIV, syphilis infection	IMBSM	Five sessions (1 booster)	Men and women attending STI clinics		
e	Small-group risk reduction sessions focusing on attitudes, skills and strategies for behaviour change	NIMH multisite intervention group trial (1998)	RCT	STI incidence, condom use	None stated	Seven sessions	Men and women attending STI clinics		
Group lev	Small-group risk reduction	Shain et al. (1999)	RCT	Infection rates, unsafe sex, number of partners	ARRM	Three session	Latina and AA women attending STI clinics		
	Small-group risk reduction	Shain et al. (2002)	RCT	Infection rates, unsafe sex, number of partners	ARRM	Three session	Latina and AA women attending STI clinics		
	Small-group sessions including communal mastery and negotiation skills as well as HIV and safe sex information	Hobfall et al. (2002)	RCT	Risky sexual behaviour	None stated	Six session	Women attending clinics		

Table 3. Category 1 review. Manhart and Holmes, 2005.

	1	r	i	r	r	·	
	Small-group skills training	Baker et al. (2003)	RCT	Reduction in STI incidence; risky behaviour	RP	16 sessions	Heterosexual women in clinics or organisations
	Group intervention	Boyer et al. (2003)	RCT	STIS and unplanned pregnancy	IMBSM	Not stated	Female military recruits
	Information education and communication and male and female condom distribution	Feldblum et al. (2001)	RCT	STI rates	None stated	Not stated	Women from plantations in Kenya
	Male or female condoms available in brothels	Fontanet et al. (1998)	RCT	Incidence of STIs	None stated	Not stated	Sex workers in Thailand brothels
	Video and discussion	O'Donnell et al. (1998)	RCT	Behavioural	None stated	Not stated	Minority ethnic men. In clinics.
	Instruction on condom use	Cohen et al. (1992a)	RCT	Behavioural	None stated	Not stated	Unclear. In clinics.
	Discussion and role play for condom negotiation	Cohen et al. (1992b)	RCT	Behavioural	None stated	Not stated	Unclear. In clinics
	Counselling in types of barrier method contraception	Gollub et al. (2000)	RCT	Behavioural	None stated	Not stated	Women in clinics
	Cognitive-behavioural intervention	Imrie et al. (2001)	RCT	Behavioural	None stated	Not stated	MSM in clinics
	Multi-component	Celentano et al., 2000	Quasi- experim ental	Behavioural	None stated	Not stated	Thai army conscripts
	Clinic based group discussions	Cohen et al. (1991)	Quasi- experim ental	Behavioural	None stated	Not stated	Clinics
	4 component intervention; in schools SRE; reproductive services for youths; condom distribution; community activities	Hayes et al. (2003)	RCT	Incidence of STIs, knowledge and behaviour	None stated	Not stated	Individuals from 20 communities of youth in Mwanza, Tanzania
/ level	Information education and communication (IEC); IEC plus management of STI	Kamali et al. (2003)	RCT	Incidence of STIs; condom use	None stated	Not stated	Individuals from 18 communities in Masaka, Uganda
Community	Condom distribution	Celentano et al. 1998	Non- RCT	Behavioural	None stated	Not stated	Thai army conscripts
	Multi-component intervention with young people	Lonczak et al. (2002)	Non- RCT	Behavioural	None stated	Not stated	Young people in schools
	Health ordinance condom distribution	Egger et al. (2000)	RCT	Behavioural	None stated	Not stated	Motels frequented by CSWs

Table 4. Category 1 review. Ward et al., 2004.

Author		Title								
	Ward D.J. et al., 2004	Behavioural interventions to reduce the risk of sexually transmitted infections in genitourinary medicine clinic patients: a systematic review								
	Type of intervention	Author	Study Design	Outcome	Theory	Duration	Target group			
F	Individual risk reduction counselling	Kamb et al. (1998)	RCT	Cumulative incidence of gonorrhoea, chlamydia or HIV	TRA, SCT	Four sessions	Heterosexual HIV- negative men and women attending clinic			
Individual lev	Experimental, cognitive- behavioural intervention	Boyer et al. (1997)	RCT	STI acquisition	ARRM	Four sessions	Heterosexual men and women attending STI clinic			
	Observational learning, direct feedback:	Schrier et al. (2001)	RCT	Condom use; new partner	SCT, TTM	One-time	multi-ethnic adolescents. Clinic			
	Small-group risk reduction sessions focusing on attitudes, skills and strategies for behaviour change	NIMH multisite intervention group trial (1998)	RCT	STI incidence, condom use	None stated	Seven sessions	Men and women attending STI clinics			
	Group discussions and motivation; impediments to condom use	Balmer et al. (1998)	RCT	Number of new sexual partners and condom use with high-risk sexual contact	Unified theory	26 wkly 1 hr sessions	Men from STI clinic in Nairobi, Kenya			
	Observational learning, role play, direct feedback: decision-making skills, goal and social skills, acceptance of negative thoughts	Metzler et al. (2000)	RCT	Condom use; number of sexual partners	SCT, IMBST	Five sessions	Male and female adolescents/ white. US STI clinics.			
o level	Observational learning, role play, direct feedback: STD education, condom negotiation and skills training, positive attitude towards condoms	Orr et al. (1996)	RCT	Condom use	НВМ	One-time	Female multi-ethnic adolescents. STI and FP clinics			
Grou	Small-group risk reduction	Shain et al. (2002)	RCT	Infection rates, unsafe sex, number of partners	ARRM	Three session	Latina and AA women attending STI clinics			
	Enhanced group prevention programme. Mixed gender groups including video session with discussion.	Branson et al. (1998)	RCT	More than one sexual partner, condom use	IMBST	Four sessions and booster session	Men and women attending STI clinics			
	Group workshop - including goal setting personal motivation, dealing with high-risk situations etc.	Imrie et al. (2001)	RCT	Proportion not engaging in unprotected anal intercourse; of those who are, number with partner of different or unknown serostatus	TTM (RP, SLT, MI)	Single one day session	MSM from a UK GUM clinic			
	Culturally specific observational learning through video featuring information on living with HIV, rap music and comedy, condom use.	Kalichman et al. (1999)	RCT	Proportion always using condoms; refusing unsafe sex; number of unprotected acts of intercourse; proportion of sex acts with condom; number of sexual partner	IMBST	Two sessions	African American men from STI clinic			

Behavioural training including assertion training, self- management, active listening skills, condom use. Social skills training including modelling behaviour or rehearsing behaviour.	O'Leary et al. (1998)	RCT	Condom use; number of sexual partners; number of risky acts	SCT	Seven sessions	Men and women from 7 STI clinics
Culturally specific observational learning through video featuring black Americans and coupons for condoms.	Solomon et al. (1989)	RCT	Mean number of condom coupons redeemed	Not theory based	Not stated	Men and women from a STI clinic

Table 5. Category 1 review. Ward et al., 2005.

	Author	Title								
,	Ward D.J. et al., 2005	Reducing the risk of sexually transmitted infections in genitourinary medicine clinic patients: a systematic review and meta-analysis of behavioural interventions.								
	Type of intervention	Author	Study Design	Outcome	Theory	Duration	Target group			
ndividual level	Experimental, cognitive- behavioural intervention	Boyer et al. (1997)	RCT	STI acquisition	ARRM	Four sessions	Heterosexual men and women attending STI clinic			
	Observational learning, direct feedback:	Schrier et al.(2001)	RCT	Condom use; new partner; number of unprotected episodes	SCT, TTM	One-time	Young women with cervicitis or PID from a children's hospital or inpatient unit . Clinic			
	Negotiation	Maher et al. (2003)	RCT	Unclear	None stated	Three sessions	Black men attending clinics with definite or probable STIs			
	Small-group risk reduction sessions focusing on attitudes, skills and strategies for behaviour change	NIMH multisite intervention group trial (1998)	RCT	STI incidence, condom use	None stated	Seven sessions	Men and women attending STI clinics			
	Group discussions and motivation; impediments to condom use	Balmer et al. (1998)	RCT	Number of new sexual partners and condom use with high-risk sexual contact	Unified theory	26 weekly sessions	Men from a STI clinic in Kenya			
	Enhanced group prevention programme. Mixed gender groups including video session with discussion.	Branson et al. (1998)	RCT	More than one sexual partner, condom use	IMBST	Four sessions and booster session	Men and women attending STI clinics			
Group level	Group workshop - including goal setting personal motivation, dealing with high-risk situations etc.	Imrie et al. (2001)	RCT	Proportion not engaging in unprotected anal intercourse; of those who are, number with partner of different or unknown serostatus	TTM (RP, SLT, MI)	Single one day session	MSM from a UK GUM clinic			
	Culturally specific observational learning through video featuring information on living with HIV, rap music and comedy, condom use.	Kalichman et al. (1999)	RCT	Proportion always using condoms; refusing unsafe sex; number of unprotected acts of intercourse; proportion of sex acts with condom; number of sexual partner	IMBST	Two sessions	African American men from STI clinic			
	Observational learning, role play, direct feedback: decision- making skills, goal and social skills, acceptance of negative thoughts	Metzler et al. (2000)	RCT	Condom use; number of sexual partners	SCT, IMBST	Five sessions	Male and female adolescents/ white. STI clinic.			

Behavioural training including assertion training, self- management, active listening skills, condom use. Social skills training including modelling behaviour or rehearsing behaviour.	O'Leary et al. (1998)	RCT	Condom use; number of sexual partners; number of risky acts	SCT	Seven sessions	Men and women from 7 STI clinics
Observational learning, role play, direct feedback: STD education, condom negotiation and skills training, positive attitude towards condoms	Orr et al.(1996)	RCT	Condom use	НВМ	One-time	Multi-ethnic adolescents
Small-group risk reduction	Shain et al. (2002)	RCT	Infection rates, unsafe sex, number of partners	ARRM	Three sessions	Latina and AA women attending STI clinics
Culturally specific observational learning through video featuring black Americans and coupons for condoms.	Solomon et al. (1989)	RCT	Mean number of condom coupons redeemed	Not theory based	Not stated	Men and women from a STI clinic

Table 6. Category 2 review. Robin et al., 2004.

Author		Title								
	Robin L. et al., 2004	Beha	Behavioral interventions to reduce incidence of HIV, STD, and pregnancy among adolescents: a decade in review							
	Type of intervention	Author	Study Design	Outcome	Theory	Duration	Target group			
	Experimental, cognitive- behavioural intervention	Boyer et al. (1997)	RCT	STI acquisition	ARRM	Four sessions	Heterosexual men and women attending STI clinic			
Individual level	Videos, printed materials providing supplementary information and additional suggestions for discussions and activities	Miller et al. (1993)	RCT	Teen sexual behaviour, intentions	None stated	Six video- tapes	Young people predominantly white. With their families.			
	Individualised risk assessment for HIV, counselling on condom use, and HIV pamphlet, an offer of free condoms, a 20 minute counselling session with physician to discuss HIV prevention	Mansfield et al. (1993)	RCT	Ever had sex; condom use; number of partners	None stated	One-time	Clinic based sample of young people			
	Video games, interactive exercises, and role-play	Jemmott et al. (1992)	RCT	Sexual intent of risky behaviour; ever had sex; multiple partners; condom use; anal intercourse	РМ	One-time	Adolescent convenience sample from school, YMCA and clinic			
	Small-group discussion, videos, games, brainstorming, experimental analysis, role-play, skills building activities	Jemmott et al. (1998)	RCT	Condom use; unprotected sex; always use a condom	SCT	8 sessions	Community based, young female African Americans. Mean age 11.8 yrs.			
Group level	Small-group discussion, lactures, videos, games, role-play, acting storytelling, arts, crafts, community projects, "family tree" depicting relationships and situations, "graduation" ceremony	Stanton et al. (1996)	RCT	Intentions, behaviour (did vs. did not have sex) condom use	РМ	7 sessions	Community based sample of young African Americans. Mean age 11.4 yrs.			
	Small-group discussion, lactures, videos, games, role-play, acting storytelling, arts, crafts, community projects, "family tree" depicting relationships and situations, "graduation" ceremony	Stanton et al. (1997)	RCT	Intentions, behaviour (did vs. did not have sex) condom use	PM	7 sessions	Community based sample of young African Americans. Mean age 11.4 yrs.			
	Small group discussions, role-plays, skills practice, brainstorming	Magura et al. (1994)	Quasi- experimental	Multiple partners; risk partners; anal sex; condom use for vaginal, oral, and anal sex.	PST	4 sessions	Sample from detention facility for young people mixed ethnicity and all male			

Tasks in preventing drug/alcohol use, violence, and risky sexual behaviours	O'Donnell et al. (1999)	Quasi- experimental	Lifetime sexual experience; frequency of sex in past three months; recent condom use; recent contraceptive use' composite index of sexual risk	НВМ	80 sessions	School based sample of young mixed ethnicity people.
Sexual decision-making, communication skills, emphasis on norms to avoid risky sex through abstinence or condom use, parent-adolescent communication	Kirby et al. (1991)	Quasi- experimental	Ever had sex; frequency of sex; birth control use at first and most recent intercourse; birth control use all or most of the time; unprotected sex; pregnant or caused pregnancy	SLT, SI, SBT	15 sessions	School based sample of young mixed ethnicity people. Mean age 15.3 yrs.
Knowledge and skills based curriculum plus peer resources, parent education school- community linkages to health services and other planned safer-sex promotion activities	Coyle et al. (1999)	RCT	Sexual initiation, frequency of unprotected intercourse, number of sexual partners, condom use at first and last intercourse	SCT, SI	20 sessions	School based sample of young mixed ethnicity people.
Family planning services, small group discussions, role-plays, writing, and videotaped skits, a healthcare and monitoring component	Bayne Smith et al. (1994)	RCT	Frequency of sex; frequency of contraceptive use	None stated	8 sessions	School based sample of young mixed ethnicity people.
Structured discussions, group exercises, role-play, guest speakers, informational presentations	Allen et al. (1997)	Quasi- experimental	Ever been pregnant or caused pregnancy (or since baseline)	None stated	1-3 times per week during academic year	School based sample of young mixed ethnicity people.
Lectures, small group discussions, role-play.	Walter et al. (1993)	RCT	AIDS risk behaviour index; sexual involvement; condom use; high risk partner; STD diagnosis	HBM, SCT, SI	6 sessions	School based sample of young mixed ethnicity people.
Skills-building, modeled skills, class discussion, small group discussion, role-plays, games and anonymous question box, homework including parents	Weeks et al. (1995)	RCT	Behavioural intentions to use condoms and to use condoms with foam, ever had sex, recent sex, age of debut, condom use at first and last sex, condom use with foam in last 12 mts, 30 days	SCT, TPB, SI	10 sessions	School based sample of young mixed ethnicity people.
Skills-building, modeled skills, class discussion, small group discussion, role-plays, games and anonymous question box, homework including parents	Levy et al. (1995)	RCT	Behavioural intentions to use condoms and to use condoms with foam, ever had sex, recent sex, age of debut, condom use at first and last sex, condom use with foam in last 12 mts, 30 days	SCT, TPB, SI	10 sessions	School based sample of young mixed ethnicity people.

Skills-building, modeled skills, class discussion, small group discussion, role-plays, games and anonymous question box, homework including parents	Weeks et al. (1997)	RCT	Behavioural intentions to use condoms and to use condoms with foam, ever had sex, recent sex, age of debut, condom use at first and last sex, condom use with foam in last 12 mts, 30 days	SCT, TPB, SI	10 sessions	School based sample of young mixed ethnicity people.
Small group discussion, videos, modelling skills, skills practice, role play, HIV+ speakers	St Lawrence et al. (1995)	RCT	Frequency of sex; frequency of contraceptive use; number of sexual partners	IMBS	est. 8 sessions	Health centre based sample of young African American people. Mean age 15.3 yrs.
Small group discussion, modeled skills, skills practice, videos, role plays	St Lawrence et al. (1999)	RCT	Frequency of sex; frequency of contraceptive use; number of sexual partners in last 3 months	IMBS	6 sessions	State Reformatory based sample young males mixed ethnic group.
Games, role play, large and small group activities, discussion, question and answer sessions, modeling of skills and skills practice	Kirby et al. (1997)	RCT	Delay of sexual onset, frequency of intercourse, condom use, birth control use, STD and pregnancy rate	SLT, HBM	8 sessions	School based young mixed ethnicity group.
Decision-making, human sexuality including family planning, videos, slide presentations, class discussion	Howard et al. (1992)	Quasi- experimental	Had sex; frequency of intercourse; contraceptive use; occurrence of pregnancy	SI	4 sessions	School/clinic young predominantly black people.
Decision-making, human sexuality including family planning, videos, slide presentations, class discussion	Howard et al. (1990)	Quasi- experimental	Had sex; frequency of intercourse; contraceptive use; occurrence of pregnancy	SI	4 sessions	School/clinic young predominantly black people.
Class discussions, group activities, videos, slide presentations, and role play	Kirby et al. (1997)	Quasi- experimental	Initiation of sex; frequency of sex; number of sexual partners; use of contraceptives; pregnancy rates; rates of reported sexually transmitted infections	None stated	4 sessions	Sample from schools, clinics and community based organisations.
Lectures, simulations, leader-guided discussions, role play, games and videos	Eisen et al. (1990)	Quasi- experimental	Ever had sex; contraceptive use at first and most recent sex; consistency of contraceptive use; index of contraceptive efficiency; occurrence of pregnancy	HBM, SLT	4 sessions	Sample from schools and family planning clinics. Young people mixed sex and ethnicity.

Shaded areas denote the same study design.
Table 7. Category	2 review.	Pedlow and	Carey, 2004.
-------------------	-----------	------------	--------------

	Author	Title					
Pe	dlow C.T. and Carey M.P., 2004	Developmentally appropriate sexual risk reduction interventions for adolescents: rationale, review of interventions, and recommendations for research and practice.					
	Type of intervention	Author	Study Design	Outcome	Theory	Duration	Target group
Individual level	Individualised intervention	Shrier et al. (2001)	Not stated	Psychological; social	Not stated	Not stated	Female adolescents with an STI, median age 17.2 years. Clinic.
	Video games, interactive exercises, and role-play	Jemmott et al. (1992)	RCT	Sexual intent of risky behaviour; ever had sex; multiple partners; condom use; anal intercourse	РМ	one-time	Convenience sample from school, YMCA and clinic
	Small-group discussion, videos, games, brainstorming, experimental analysis, role-play, skills building activities	Jemmott et al. (1998)	RCT	Condom use; unprotected sex; always use a condom	SCT	8 sessions	Community based, young female African Americans. In schools.
	HIV intervention	Jemmott et al. (1999)	Not stated	Psychological; social	Not stated	Not stated	African American teens, mean age 13.2 years. In schools.
	Postponing sexual intercourse	Aarons et al. (2000)	Not stated	Biological; social	Not stated	Not stated	Males and females in school, mean age 12.8 years. From schools.
	Assess HIV/STD prevention	Boekeloo et al. (1999)	Not stated	Psychological; social	Not stated	Not stated	Males and females in school, 12-15 years
svel	Safer choices curriculum	Coyle et al. (2001)	Not stated	Biological; psychological; social	Not stated	Not stated	High school youth, 14-18 years
Group le	Health educator intervention; videotape intervention	DeLamater et al. (2000)	Not stated	Psychological; social	Not stated	Not stated	African American teens, aged 15-19 years. In clinics.
	Intervention 1. Video and comic book. Intervention 2. Group skills training, video and comic book	Gilmore et al. (1997)	Not stated	Psychological; social	Not stated	Not stated	Sexually active teens, 1. 14-19 years, 2. 13-19 years. In detention centre.
	AIDS risk reduction and skills training intervention	Kipke et al. (1993)	Not stated	Psychological; social	ARRM	Not stated	Minority youth aged 12-16 years. In schools.
	Class discussions, group activities, videos, slide presentations, and role play	Kirby et al. (1997)	Quasi- experimental	Initiation of sex; frequency of sex; number of sexual partners; use of contraceptives; pregnancy rates; rates of reported sexually transmitted infections	Not stated	4 sessions	Sample from schools, clinics and community based organisations
	Youth AIDS prevention	Levy et al. (1995)	Not stated	Biological; psychological; social	Not stated	Not stated	African American youth. In schools.
	AIDS education and counselling	Mansfield et al. (1993)	Not stated	Psychological	Not stated	Not stated	Adolescents with STD. In clinics.

-108-. Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

l	Monogamy, abstinence, condom choice	Metzler et al. (2000)	Not stated	Psychological; social	Not stated	Not stated	Sexually active teens. From clinics.
-	Tasks in preventing drug/alcohol use, violence, and risky sexual behaviours	O'Donnell et al. (1999)	Quasi- experimental	Lifetime sexual experience; frequency of sex in past three months; recent condom use; recent contraceptive use' composite index of sexual risk	НВМ	80 sessions	School based sample of young mixed ethnicity people
	Behavioural intervention	Orr et al. (1996)	Not stated	Psychological; social	Not stated	Not stated	Female adolescents with Chlamydia in clinics.
i	7 and 3 session HIV ntervention	Rotheram- Borus et al. (1998)	Not stated	Psychological; social	Not stated	7/3 sessions	High risk minority youth. From social services.
;	Skills training intervention	Slonim- Nevo et al. (1996)	Not stated	Psychological; social	Not stated	Not stated	Delinquent and abused youth, mean age 14.7 years. From welfare centre.
l	Decision-making AIDS ntervention	Stanton et al. (1996)	Not stated	Psychological; social	Not stated	Not stated	African American early youth aged 9- 15 years. Housing development.
	_ectures, small group discussions, role play.	Walter et al. (1993)	RCT	AIDS risk behaviour index; sexual involvement; condom use; high risk partner; STD diagnosis	HBM, SCT, SI	6 sessions	School based sample of young mixed ethnicity people.
1	Becoming a responsible een	St. Lawrence et al. (1999)	Not stated	Psychological; social	Not stated	Not stated	Incarcerated male youth, mean age 15.8 years
	Observational learning, role olay, direct feedback: education and behaviour skills training, including condom use, assertion, refusal etc.	St. Lawrence et al. (1995a)	RCT	Condom protected intercourse	SCT, IMBST	8 weekly meetings	Adolescents/African American
I	Behavioural skills training	St. Lawrence et al. (1995b)	Not stated	Psychological; social	Not stated	Not stated	Youth in treatment for substance abuse. Age range 13-17 years
	AIDS prevention curriculum	Walter and Vaughan (1993)	Not stated	Psychological; social	Not stated	Not stated	Urban minority youth. Aged 12-20 years. In schools.
l	HIV prevention	Workman et al. (1996)	Not stated	Psychological; social	Not stated	Not stated	Inner city minority female adolescents. In schools.

Theory definitions

SLT – Social Learning Theory; **CLT** – Cognitive Learning Theory; **SRMIB** – Self-Regulation Model of Illness Behaviour; **CBT** - Cognitive Behavioural Therapy; **ARRM** - AIDS risk-reduction Method; **RP** - Relapse Prevention; **HBM** - Health Belief Model; **ME/I** - Motivational Enhancement/Interviewing; **IMBSM** – Information Motivation Behavioural Skills Model; **PST** - Problem Solving Therapy; **TRA** - Theory of Reasoned Action; **PM** - Protection Motivation; **SI** - Social Influence; **SET** – Self-efficacy Theory; **TTM** – Transtheoretical Model; **TPB** – Theory of Planned Behaviour. **SCT** –Social Cognitive Theory.

Study design definitions

-109-. Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

RCT – Randomised Controlled Trial; **RSC-NC** – Repeat cross-sectional study without a comparison group; **PS-NC** – Panel study without a comparison.

Table 8. Cross over of primary studies for Category 1 and 2 reviews.

	Category 1 and 2 reviews							
Primary study	DiClemente et al., 2004	Manhart L.E. & Holmes K. K., 2005	Ward D.J. et al., 2004	Ward D.J. et al., 2005	Robin L. et al., 2004	Pedlow C.T. & Carey M.P., 2004		
Individual level								
Boekeloo et al., (1999)	\checkmark							
Boyer et al. (1997)		\checkmark	~	~	\checkmark			
DeLameter et al., (2000)	~							
El-Bassel et al.		\checkmark						
EXPLORE Study Group (2004)		~						
James et al.		✓						
Kamb et al. (1998)		✓	✓					
Maher et al.				~				
Mansfield et al.	~				✓			
Miller et al.					✓			
Peterman et al.		✓						
Peterman et al.		~						
Scholes et al., (2003)	~							
Schrier et al. (2001)	~	✓	~	~		✓		
VCT Efficacy study group (2000)		~						
Group level			1					
Aarons et al.						✓		
Allen et al.					✓			
Baker et al.		✓						
Balmer et al.			~	~				
Bayne Smith et					✓			
Boekeloo et al. (1999)*						✓		
Boyer et al. (2003)		\checkmark						
Branson et al. (1998)		✓	~	~				
Celentano et al., 2000		\checkmark						
Cohen et al. (1991)		\checkmark						
Cohen et al. (1992a)		\checkmark						
Cohen et al. (1992b)		\checkmark						
Coyle et al.					\checkmark			

^{*} Boekeloo et al. (1999) was categorised as group level for Pedlow and Carey (2004) and individual level for DiClemente et al. (2004). The intervention used both group education and one-to-one doctor follow-up. Pedlow and Carey classed it as a school based intervention and DiClemente et al. (2004) focused upon the one-to-one intervention.

-111-. Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

(1999)						
Coyle et al. (2001)						\checkmark
DeLamater et						✓
DiClemente et	~					
Eisen et al.					✓	
Feldblum et al.		✓				
Fontanet et al.		✓				
Gillmore et al.,	~					✓
Gollub et al.		~				
Hobfall et al.		✓				
Howard et al.					✓	
Howard et al.					✓	
Imrie et al. (2001)		✓	✓	✓		
Jemmott et al.					✓	✓
Jemmott et al.					✓	✓
Jemmott et al.						✓
Kalichman et al.			✓	✓		
Kipke et al.						✓
Kirby et al.					✓	
Kirby et al.					✓	
Kirby et al.					✓	✓
Levy et al.					✓	✓
Magura et al.					✓	
Mansfield et al.						✓
Metzler et al.	✓		✓	✓		✓
NIMH multisite						
group trial		\checkmark	\checkmark	\checkmark		
O'Donnell et al.		✓				
O'Donnell et al.					✓	✓
O'Leary et al.			~	~		
Orr et al. (1996)	✓		✓	✓		✓
Rotheram-						,
Borus et al. (1998)						~
Shain et al. (1999)		\checkmark				
Shain et al. (2002)		✓	✓	✓		
Slonim-Nevo et al. (1996)						✓
Solomon et al. (1989)			~	~		
St Lawrence et al. (1999)					\checkmark	✓

-112-. Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

St. Lawrence et	✓			\checkmark	~
al. (1995a)					
St. Lawrence et					\checkmark
al. (1995b)					
Stanton et al.				\checkmark	\checkmark
(1996)					
Stanton et al.				\checkmark	
(1997)					
Walter et al.				\checkmark	\checkmark
(1993)					-
Walter &					,
Vaughan					\checkmark
(1993)					
Weeks et al.				\checkmark	
(1995)					
Weeks et al.				\checkmark	
(1997)				•	
Workman et al.					\checkmark
(1996)					•
Community le	evel				
Celentano et al.		1			
1998		•			
Egger et al.		1			
(2000)		•			
Hayes et al.		1			
(2003)		•			
Kamali et al.		1			
(2003)		· ·			
Lonczak et al.		1			
(2002)		v			

References

Abdullah, A. S., Ebrahim, S. H., Fielding, R. et al. (2004). Sexually transmitted infections in travelers: implications for prevention and control. *Clinical Infectious Diseases* 39 (4): 533-538.

Anonymous (2002). Prevention and management of reproductive tract infections/sexually transmitted infections. *Indian Journal of Public Health* 46 (3): 115-116.

Bellis, M. A., Cook, P., Clark, P., Syed, Q. and Hoskins, A. (2002) Re-emerging syphilis in gay men: a case-control study of behavioural risk factors and HIV status. *Journal of Epidemiology and Community Health* 56: 235-236.

Bergk, A., Berg, T., and Neumann, U. P. (2005). Risk factors for the acquisition of hepatitis C. *Minerva Gastroenterologica e Dietologica* 51 (1): 7-14.

Blair, M. (2004). Sexually transmitted diseases: an update. Urologic Nursing 24 (6): 467-473.

Bonner, K. (2001). Male Circumcision as an HIV Control Strategy: Not a 'Natural Condom'. *Reproductive Health Matters* 9 (18): 143-155.

Bonsu, I. K. (2005). How contraception nurses can improve teenage sexual health. *Nursing Times* 101 (7): 34-36.

Casper, C. and Wald, A. (2002). Condom use and the prevention of genital herpes acquisition. *Herpes* 9 (1): 10-14.

Chin-Hong, P. V., Vittinghoff, E., Cranston, R.D., Buchbinder, S., Cohen, D., Colfax G., Da Costa, M., Darragh, T., Hess, E., Judson, F., Koblin, B., Madison, M., Palefsky, J. M. (2004) Age-specific prevalence of anal human papilloma virus infection in HIV-negative sexually active men who have sex with men: the EXPLORE study. *Journal of Infectious Diseases* 190: 2070–6.

Chorba, T., Scholes, D., Bluespruce, J. et al. (2004). Sexually transmitted diseases and managed care: an inquiry and review of issues affecting service delivery. *American Journal of Medical Quality* 19 (4): 145-156.

Coetzee, N., Guttmacher, S., Mathews, C. et al., (2004) Partner notification. *Clinical Evidence* (10): 1863-1870.

Cothran, M. M. and White, J. P. (2002). Adolescent behavior and sexually transmitted diseases: the dilemma of human papillomavirus. *Health Care for Women International* 23 (3): 306-319.

Cuschieri, K. S., Cubie, H. A., Whitley, M. W., Seagar, A. L., Arends, M. J., Moore, C., Gilkisson, G., McGoogan, E. (2004) Multiple high risk HPV infections are common in cervical neoplasia and young women in a cervical screening population. *Journal of Clinical Pathology* 57: 68-72.

Dehne, K. L. and Riedner, G. (2001). Sexually transmitted infections among adolescents: the need for adequate health services. *Reproductive Health Matters* 9 (17): 170-183.

Department of Health (2001). *National strategy for sexual health and HIV*. London: Department of Health. (Available at <u>www.doh.gov.uk</u>).

Department of Health (2002). *National strategy for sexual health and HIV: implementation action plan*. London: Department of Health. (Available at <u>www.doh.gov.uk</u>).

Department of Health (2003a). Effective commissioning of sexual health and HIV services: a sexual health and HIV commissioning toolkit for primary care trusts and local authorities. London: Department of Health (Available at <u>www.doh.gov.uk</u>).

Department of Health (2003b). *Effective sexual health promotion: a toolkit for primary care trusts and others working in the field of promoting good sexual health and HIV prevention*. London: Department of Health. (Available at <u>www.doh.gov.uk</u>).

Department of Health (2004). *Choosing Health: making healthy choices easier*. London: Department of Health. (Available at <u>www.doh.gov.uk</u>).

Department of Health (2005). *Delivering Choosing Health: making health choices easier*. London: Department of Health. (Available at <u>www.doh.gov.uk</u>).

DiClemente, R. J., Crosby, R. A., and Wingood, G. M. (2002). Enhancing STD/HIV prevention among adolescents: the importance of parenteral monitoring. *Minerva Pediatrica* 54 (3): 171-177.

DiClemente, R. J., Milhausen, R., Sales, J. M. et al. (2005). A programmatic and methodologic review and synthesis of clinic-based risk-reduction interventions for sexually transmitted infections: research and practice implications. *Seminars in Pediatric Infectious Diseases* 16 (3): 199-218.

Downing, J., Jones, L., Cook, P. A., Bellis, M. A. (Awaiting publication). *HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission. Evidence briefing update.* London: NICE.

Dulmus, C. N. E. and Rapp-Paglicci, L. A. E. (eds) (2005). *Handbook of preventive interventions for adults*. Hoboken, NJ, US: John Wiley & Sons, Inc.

Ellis, S. and Grey, A. (2004). *Prevention of sexually transmitted infections (STIs): a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing.* London: Health Development Agency.

Ellis, S., Barnett-Page, E., Morgan, A. et al. (2003). HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission. Evidence briefing. London: Health Development Agency.

Elwy, A. R., Hart, G. J., Hawkes, S. et al. (2002). Effectiveness of interventions to prevent sexually transmitted infections and human immunodeficiency virus in heterosexual men: a systematic review. *Archives of Internal Medicine* 162 (16): 1818-1830.

Fenton, K. A. and Hughes, G. (2003). Sexual behaviour in Britain: why sexually transmitted infections are common. *Clinical Medicine* 3 (3): 199-202.

Feroli, K. L. and Burstein, G. R. (2003). Adolescent sexually transmitted diseases: new recommendations for diagnosis, treatment, and prevention. *MCN, American Journal of Maternal Child Nursing* 28 (2): 113-118.

Fitch, J. T. (2001). Are condoms effective in reducing the risk of sexually transmitted disease? *Annals of Pharmacotherapy* 35 (9): 1136-1138.

Fortenberry, J. D. (2002). Clinic-based service programs for increasing responsible sexual behavior. *Journal of Sex Research* 39 (1): 63-66.

Genuis, S. J. and Genuis, S. K. (2005). Primary prevention of sexually transmitted disease: applying the ABC strategy. *Postgraduate Medical Journal* 81 (955): 299-301.

Gil, A. G. and Tubman, J. G. (2005). HIV/AIDS and STDs. In: Dulmus, C. N. and Rapp-Paglicci, L. A. (ed) *Handbook of preventive interventions for adults,* pp 226-251. Hoboken, NJ, US: John Wiley & Sons, Inc.

Gilson, R. J. and Mindel, A. (2001). Recent advances: Sexually transmitted infections. *British Medical Journal* 322 (7295): 1160-1164.

Glik, D., Nowak, G., Valente, T. et al. (2002). Youth performing arts entertainment-education for HIV/AIDS prevention and health promotion: practice and research. *Journal of Health Communication* 7 (1): 39-57.

Gott, M. (2004). Are older people at risk of sexually transmitted infections? A new look at the evidence. *Reviews in Clinical Gerontology* 14 (1)13.

Heinz, M. (2004). Sexually transmitted diseases. Endocrine Development 7: 197-212.

Henderson, Z., Tao, G., and Irwin, K. (2005). Sexually transmitted disease care in managed care organizations. *Infectious Disease Clinics of North America* 19 (2): 491-511.

Hickson, F., Reid, D., Weatherburn, P., Stephens, M., Brown, D. (2001) *Time for more: Findings from the National Gay Men's Sex Survey 2000.* London: Sigma Research (Available at <u>www.sigmaresearch.co.uk</u>).

Hingson, R. W. and Howland, J. (2002). Comprehensive community interventions to promote health: implications for college-age drinking problems. *Journal of Studies on Alcohol - Supplement* (14): 226-240.

Holmes, K. K., Levine, R., and Weaver, M. (2004). Effectiveness of condoms in preventing sexually transmitted infections. *Bulletin of the World Health Organization* 82 (6): 454-461.

Honey, E., Augood, C., Templeton, A. et al. (2002). Cost effectiveness of screening for Chlamydia trachomatis: a review of published studies. *Sexually Transmitted Infections* 78 (6): 406-412.

HPA (2006). Diagnoses and rates of selected STIs seen at GUM clinics: 1997-2006. National and regional level summary tables. Health Protection Agency, Centre for Infections, London.

Irwin, R. (2003). Reducing the transmission of sexually transmitted infections. *Professional Nurse* 18 (9): 497-500.

Jackson, D., Dallabetta, G., and Steen, R. (2004). Sexually transmitted infections: prevention and management. *Clinics in Occupational & Environmental Medicine* 4 (1): 167-188.

Johnson, A. M., Mercer, C. H., Erens, B., Copas, A. J., McManus, S., Wellings, K., Fenton, K., Korovessis, C., Macdowall, W., Nanchahal, K., Purdon, S., Field, J. (2001). Natsal 2000: Sexual behaviour in Britain: partnerships, practices, and HIV risk behaviours. *Lancet* 358 (9296): 1835-42.

Jungmann, E. (2005). Genital herpes. Clinical Evidence (13): 1987-2001.

Kelly, M. P., Swann, C., Morgan, A., Killoran, A., Naidoo, B. and Barnett-Page, E. (2002). *Methodological Problems in Constructing the Evidence Base in Public Health*. London: Health Development Agency.

Kirby, B. D. (2001). Understanding what works and what doesn't in reducing adolescent sexual risk-taking. *Family Planning Perspectives* 33 (6): 276-281.

Kirby, D. (2002). Effective approaches to reducing adolescent unprotected sex, pregnancy, and childbearing. *Journal of Sex Research* 39 (1): 51-57.

Klein, J. D. and Matos, A. M. (2002). Improving adolescent health outcomes. *Minerva Pediatrica* 54 (1): 25-39.

-117-. Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

Kotchick, B. A., Shaffer, A., Forehand, R. et al. (2001). Adolescent sexual risk behavior: a multi-system perspective. *Clinical Psychology Review* 21 (4): 493-519.

LaMontagne, D. S., Fenton K. A., Randall, S., Anderson, S., Carter P. (2004) Establishing the National Chlamydia Screening Programme in England: results from the first full year of screening. *Sexually Transmitted Infections* 80: 335–341.

Lee, J. D. and Clarke, J. (2004). Cover up or cool it? Sexual intercourse during therapy for bacterial sexually transmitted infections--a discussion of evidence for efficacy of condom use preventing transmission during an acute bacterial STI. *International Journal of STD & AIDS* 15 (5): 285-288.

Leone, P. (2005). Reducing the risk of transmitting genital herpes: advances in understanding and therapy. *Current Medical Research & Opinion* 21 (10): 1577-1582.

Low, N., Daker-White, G., Barlow, D. and Pozniak, A. (1997). Gonorrhoea in inner London: results of a cross sectional study. *British Medical Journal* 314 (7096): 1719-23.

Low, N. (2004). Chlamydia (uncomplicated, genital). Clinical Evidence (12): 2203-2211.

Manhart, L. E. and Holmes, K. K. (2005). Randomized controlled trials of individual-level, population-level, and multilevel interventions for preventing sexually transmitted infections: what has worked? *Journal of Infectious Diseases* 191 (Suppl 1): S7-24.

Marra, C. M. (2004). Syphilis and human immunodeficiency virus: prevention and politics. *Archives of Neurology* 61 (10): 1505-1508.

Mathews, C., Coetzee, N., Zwarenstein, M. et al. (2001). Strategies for partner notification for sexually transmitted diseases. In: *The Cochrane Database of Systematic Reviews*, Issue 4. Chichester, UK: John Wiley & Sons, Ltd.

Mathews, C., Coetzee, N., Zwarenstein, M. et al. (2002). A systematic review of strategies for partner notification for sexually transmitted diseases, including HIV/AIDS. *International Journal of STD & AIDS* 13 (5): 285-300.

Mathews, C., Coetzee, N., Zwarenstein, M. et al. (2004). Partner notification. *Clinical Evidence* (11): 2113-2120.

Mayaud, P. and Mabey, D. (2004). Approaches to the control of sexually transmitted infections in developing countries: old problems and modern challenges. *Sexually Transmitted Infections* 80 (3): 174-182.

McEvoy, M. and Coupey, S. M. (2002). Sexually transmitted infection. A challenge for nurses working with adolescents. *Nursing Clinics of North America* 37 (3): 461-474. Meade, C. S. and Ickovics, J. R. (2005). Systematic Review of Sexual Risk among Pregnant and

Mothering Teens in the USA: Pregnancy as an Opportunity for Integrated Prevention of STD and Repeat Pregnancy. *Social Science & Medicine* 60 (4): 661-678.

Miller, K. E., Ruiz, D. E., and Graves, J. C. (2003). Update on the prevention and treatment of sexually transmitted diseases. *American Family Physician* 67 (9): 1915-1922.

Minnis, A. M. and Padian, N. S. (2005). Effectiveness of female controlled barrier methods in preventing sexually transmitted infections and HIV: current evidence and future research directions. *Sexually Transmitted Infections* 81 (3): 193-200.

Moran, J. (2005). Gonorrhoea. Clinical Evidence (13): 2016-2023.

Myer, L., Morroni, C., Mathews, C. et al. (2001). Structural and community-level interventions for increasing condom use to prevent HIV and other sexually transmitted infections (Protocol). In: *The Cochrane Database of Systematic Reviews*, Issue.4. Chichester, UK: John Wiley & Sons, Ltd.

Oberdorfer, A., Oberdorfer, A. L., and Tran, D. T. (2002). Behavioural interventions for preventing hepatitis B and/or C (Protocol). In: *The Cochrane Database of Systematic Reviews,* Issue.2. Chichester, UK: John Wiley & Sons, Ltd.

Oberg, C., Hogan, M., Bertrand, J. et al. (2002). Health care access, sexually transmitted diseases, and adolescents: identifying barriers and creating solutions. *Current Problems in Pediatric & Adolescent Health Care* 32 (9): 320-339.

Patel, R. (2004). Educational interventions and the prevention of herpes simplex virus transmission. *Herpes* 11 (Suppl 3): 155A-160A.

Payne, N. and O'Brien R. (2005) *Health economics of sexual health: A guide for commissioning and planning.* London: Department of Health.

Pedlow, C. T. and Carey, M. P. (2004). Developmentally appropriate sexual risk reduction interventions for adolescents: rationale, review of interventions, and recommendations for research and practice. *Annals of Behavioral Medicine* 27 (3): 172-184.

Pozniak, A. (2003). What happened to sexually transmitted disease prevention? *Current Opinion in Infectious Diseases* 16 (1): 33-35.

Rekart, M. L. (2005). Sex-work harm reduction. Lancet 366 (9503): 2123-2134.

Reyna V. F. Risky decision making in childhood and adolescence: A fuzzy-Trace theory approach. In Jacobs J. E. and Klascynski P. A. (eds) (2005) *The development of judgement and decision making in children and adolescents*. London: Lawrence Erlbaum Associates, publishers.

Roberts, C. (2005). Genital herpes in young adults: changing sexual behaviours, epidemiology and management. *Herpes* 12 (1): 10-14.

Robin, L., Dittus, P., Whitaker, D. et al. (2004). Behavioral interventions to reduce incidence of HIV, STD, and pregnancy among adolescents: a decade in review. *Journal of Adolescent Health* 34 (1): 3-26.

Ross, M. W. and Williams, M. L. (2002). Effective targeted and community HIV/STD prevention programs. *Journal of Sex Research* 39 (1): 58-62.

Sangani, P., Rutherford, G., and Wilkinson, D. (2004). Population-based interventions for reducing sexually transmitted infections, including HIV infection. In: *The Cochrane Database of Systematic Reviews*, Issue 2. Chichester, UK: John Wiley & Sons, Ltd.

Sarkar, N. N. (2001). Adolescents' and young adults' sexual behaviour, STD and AIDS. *International Medical Journal* 8 (3): 173.

Schaalma, H. P., Abraham, C., Gillmore, M. R. et al. (2004). Sex education as health promotion: what does it take? *Archives of Sexual Behavior* 33 (3): 259-269.

Schmiedl, R. (2004). School-based condom availability programs. Journal of School Nursing 20 (1): 16-21.

Scholes, D., Anderson, L. A., Operskalski, B. H. et al. (2003). STD prevention and treatment guidelines: a review from a managed care perspective. *American Journal of Managed Care* 9 (2): 181-189.

Serrant-Green, L. (2005). Breaking traditions: sexual health and ethnicity in nursing research: a literature review. *Journal of Advanced Nursing* 51 (5): 511-519.

Shrier, L. A. (2004). Sexually transmitted diseases in adolescents: biologic, cognitive, psychologic, behavioral, and social issues. *Adolescent Medicine Clinics* 15 (2): 215-234.

Simms, I., Fenton, K. A., Ashton, M., Turner, K. M. E., Crawley-Boevey, E., Gorton, R. et al. (2005) The re-emergence of syphilis in the UK: the new epidemic phases. *Sexually Transmitted Diseases* 32: 220-226.

Sitruk-Ware, R. (2005). Delivery options for contraceptives. Drug Discovery Today 10 (14): 977-985.

Smith, C. (2001). Commissioned review: the national strategy for sexual health and HIV: a perspective from family planning. *Journal of Family Planning & Reproductive Health Care* 27 (4): 230-233.

Speizer, I. S., Magnani, R. J., and Colvin, C. E. (2003). The Effectiveness of Adolescent Reproductive Health Interventions in Developing Countries: A Review of the Evidence. *Journal of Adolescent Health* 33 (5): 348.

Steenbeek, A. (2004). Empowering health promotion: a holistic approach in preventing sexually transmitted infections among First Nations and Inuit adolescents in Canada. *Journal of Holistic Nursing* 22 (3): 254-266.

Steinbrook, R. (2006) The Potential of Human Papillomavirus Vaccines. *New England Journal of Medicine* 354 (11): 1109.

Sternberg, P. and Hubley, J. (2004). Evaluating men's involvement as a strategy in sexual and reproductive health promotion. *Health Promotion International* 19 (3): 389-396.

Suarez, T. and Miller, J. (2001). Negotiating risks in context: a perspective on unprotected anal intercourse and barebacking among men who have sex with men--where do we go from here?. *Archives of Sexual Behavior* 30 (3): 287-300.

Sulak, P. J. (2004). Adolescent sexual health. Journal of Family Practice (Suppl): S3-S4.

Swann, C., Bowe, K., McCormick, G. and Kosmin, M. (2003a). *Teenage pregnancy and parenthood: a review of reviews. Evidence briefing.* London: Health Development Agency.

Swann, C., Falce, C., Morgan, A., Kelly, M. P. and Powell, G. (2003b). *HDA evidence base. Process and quality standards manual for evidence briefings.* 2nd edition. London: Health Development Agency.

Teran, S., Walsh, C., and Irwin, K. L. (2001). Chlamydia trachomatis infection in women: bad news, good news, and next steps in prevention. *Journal of the American Medical Womens Association* 56 (3): 100-104.

Thachil, A. and Bhugra, D. (2005). Literature update: A critical review. *Sexual and Relationship Therapy* 20 (4): 464.

Tillett, J. (2005). Adolescents and informed consent: ethical and legal issues. *Journal of Perinatal & Neonatal Nursing* 19 (2): 112-121.

Tobin, J. M. (2002). Chlamydia screening in primary care: is it useful, affordable and universal? *Current Opinion in Infectious Diseases* 15 (1): 31-36.

Toroitich, R., Kuyoh, M., Grimes, D. A. et al. (2001). Strategies for increasing concurrent barrier use among women using oral contraceptives (Protocol). *The Cochrane Database of Systematic Reviews*, Issue 3. Chichester, UK: John Wiley & Sons, Ltd.

Toups, M. L. and Holmes, W. R. (2002). Effectiveness of abstinence-based sex education curricula: A review. *Counseling and Values* 46 (3): 240.

Tripp, J. and Viner, R. (2005). Sexual health, contraception, and teenage pregnancy. *BMJ* 330 (7491): 590-593.

UK Collaborative Group for HIV and STI Surveillance (2005) *Mapping the issues: HIV and other sexually transmitted infections in the United Kingdom: 2005.* London: Health Protection Agency, Centre for Infections.

UK Collaborative Group for HIV and STI Surveillance (2006) A complex Picture: HIV and other sexually transmitted infections in the United Kingdom: 2006. London: Health Protection Agency, Centre for Infections.

Ward, D. J., Rowe, B., Pattison, H. et al. (2004). Behavioural interventions to reduce the risk of sexually transmitted infections in genitourinary medicine clinic patients: a systematic review. Birmingham, UK: West Midlands Health Technology Assessment Collaboration, Department of Public Health and Epidemiology, University of Birmingham.

Ward, D. J., Rowe, B., Pattison, H. et al. (2005). Reducing the risk of sexually transmitted infections in genitourinary medicine clinic patients: a systematic review and meta-analysis of behavioural interventions. *Sexually Transmitted Infections* 81 (5): 386-393.

Warner, L., Stone, K. M., Macaluso, M. et al. (2006). Condom use and risk of gonorrhea and chlamydia: A systematic review of design and measurement factors assessed in epidemiologic studies. *Sexually Transmitted Diseases* 33 (1): 36-51.

Weatherburn P, Reid D, Hickson F, Hammond G, Stephens M. (2005) *Risk and reflexion: findings from the UK gay men's sex survey 2004.* London: Sigma Research. (Available at <u>www.sigmaresearch.co.uk</u>)

Weller, S., Davis, B. et al. (2002). The effectiveness of male condoms in prevention of sexually transmitted diseases (Protocol). In: *The Cochrane Database of Systematic Reviews*, Issue 4. Chichester, UK: John Wiley & Sons, Ltd.

Wellings, K., Nanchahal, K., MacDowall, W., McManus, S., Erens, B., Mercer, C. H., Johnson, A. M., Copas, A. J., Korovessis, C., Fenton, K. A. and Field, J. (2001). Sexual behaviour in Britain: early heterosexual experience. *Lancet* 358 (9296): 1843-50.

Wilson, M. M. (2003). Sexually transmitted diseases. *Clinics in Geriatric Medicine* 19 (3): 637-655.Zenilman, J. M. (2005). Behavioral interventions - rationale, measurement, and effectiveness. *Infectious Disease Clinics of North America* 19 (2): 541-562.



Jennifer Downing Lisa Jones Penny A. Cook Mark A. Bellis

Centre for Public Health Castle House North Street Liverpool John Moores University L3 2AY Tel: 0151 231 4447 Fax: 0151 231 4515 Email: j.downing1@ljmu.ac.uk

-124-. Prevention of STIs: a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. 2006.

Supplement E



HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission

Evidence Briefing Update

Jennifer Downing, Lisa Jones, Penny A. Cook, and Mark A. Bellis

Centre for Public Health Castle House North Street Liverpool John Moores University L3 2AY Tel: 0151 231 4447 Fax: 0151 231 4515 Email: j.downing1@ljmu.ac.uk

- 1 - HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission. 2006.

Acknowledgements

We would like to thank staff at the Centre for Public Health for their support, especially Dr Harry Sumnall, Suzy Hargreaves, Dr Kevin Corbett and Hannah Madden.

Contents

Ackn	owledgements	2
Conte	ents	3
Appe	ndices	6
1 E	Executive summary	7
2 lr	ntroduction	15
2.1	Background	15
2.2	Scope of the Evidence Briefing on HIV prevention	16
2.3	Context	17
2.4	Conceptual Framework	24
3 N	lethodology	27
3.1	Identification of the literature	27
3.2	Critical appraisal	28
3.3	Analysis and synthesis	30
3.4	Presentation of data	31
4 T	The Results	33
4.1	Summary of identification of papers	33
4.2	Summary of reviews in Categories 1-3	34
5 T	The evidence	42
5.1	Men who have sex with men (MSM)	42
5.2	Commercial sex workers (CSWs)	51
5.3	People with HIV (PWHIV)	54
5.4	African communities in the UK	59
5.5	Voluntary counselling and testing (VCT)	66
6 C	Cost-Effectiveness	73
7 T	heory-based interventions	76
8 D	Discussion	

8.1	What works to reduce the sexual risk of HIV transmission among the priority	
popu	lations? What works to change the modifying factors that influence sexual risk	
beha	viour for HIV transmission?	_ 81
8.2	The role of theory in intervention effectiveness	_ 82
8.3	Multi-component interventions	_ 82
8.4	Inequalities	_ 82
8.5	Cost-effectiveness	_ 83
8.6	Methodological issues	_ 83
8.7	Limitations, inconsistencies and gaps in the evidence	_ 84
8.8	Men who have sex with men (MSM)	_ 85
8.9	Commercial sex workers (CSWs)	_ 86
8.10	People with HIV (PWHIV)	_ 86
8.11	African communities in the UK	_ 87
8.12	Voluntary counselling and testing (VCT)	_ 88
8.13	Cost-effectiveness	_ 88
8.14	Theory-based interventions	_ 88
8.15	Limitations of the Evidence Briefing	_ 88
9 Im	plications for policy and practice	_ 90
9.1	Implications for policy and practice with men who have sex with men (MSM)	_ 90
9.2	Implications for policy and practice with commercial sex workers (CSWs)	_ 90
9.3	Implications for policy and practice with people with HIV (PWHIV)	_ 91
9.4	Implications for policy and practice with African communities in the UK	_ 91
9.5	Implications for policy and practice with voluntary counselling and testing (VCT) $_$	_ 91
10 I	Recommendations for research	_ 93
10.1	General recommendations for secondary research/reviews	_ 93
10.2	General recommendations for primary research	_ 93
10.3	Recommendations for research with men who have sex with men (MSM)	_ 94
10.4	Recommendations for research with commercial sex workers (CSWs)	_ 94
10.5	Recommendations for research with African communities in the UK	_ 94
10.6	Recommendations for research with people with HIV	94

10.7 Recommendations for research in relation to voluntary testing and counselling

(VCT) 95

11 References______96

Appendices

Appendix A. Theories of behaviour change	110
Appendix B. Example of a search strategy	117
Appendix C: CAT screening tool	119
Appendix D. List of reviews by Category 1-5	121
Appendix E. In depth summary of Category 1 and 2 reviews	
Appendix F: Tables of primary studies within level one and two review papers	

1 Executive summary

Introduction

This updated Evidence Briefing reports on a review of reviews assessing the effectiveness of interventions to impact on the factors – both personal and structural - which influence the sexual risk behaviours for HIV transmission. As in the original review of reviews, the aims were to:

- Identify and appraise all relevant reviews, including systematic reviews, literature reviews, syntheses and meta-analyses
- Analyse and synthesise these papers to highlight 'what works' to prevent or reduce the sexual risk of HIV transmission for the UK's priority populations
- Highlight conflicting evidence, and gaps in the evidence, and provide recommendations for research (as well as some initial implications for policy and practice).

This updated Evidence Briefing focuses on the priority populations for HIV transmission in the UK, which are: men who have sex with men (MSM), commercial sex workers (CSWs), African communities living in the UK, and people with HIV (PWHIV). We also consider the role of HIV voluntary counselling and testing (VCT) and examine the evidence relating to inequalities in health, the cost-effectiveness of interventions and the role of theory in intervention effectiveness.

The findings from this update will be the most useful to those commissioning and undertaking research, including practitioners and agencies outside academia. Although it can also inform policy and practice, the evidence presented here from reviews has a number of limitations and should be considered alongside the original review of reviews (Ellis et al., 2003), best available non-review evidence and information from other relevant sources.

Epidemiology

At the end of 2005 there were an estimated 63,500 (range: 59,500-68,800) people living with HIV in the UK, one third of whom were unaware of their infection (UK Collaborative Group for HIV and STI Surveillance, 2006). In the same year there were 785 (HPA, 2007) diagnoses of AIDS, contributing to a cumulative total of 22,199 by the end of 2005. Where country of infection was reported, the majority (75%, 2,760) of heterosexual infections in the UK in 2005 were reported to have occurred in African countries. There are a small but growing number of infections acquired within the UK from sex between men and women, from 245 in 2000 to 553 in 2005 (HPA, 2007). However, the actual level of transmission in the UK may be higher because of the tendency for doctors to classify patients as having been exposed in Africa if they had sex in both the UK and Africa (Brown, 2000). Unlinked anonymous testing carried out as part of a survey of African communities in London, Luton and West Midlands (the MAYISHA II project), revealed an overall HIV prevalence of 14% of those tested. The prevalence in each subgroup was indicative of the prevalence in the country of birth, with 26% (12/46)

of those born in southern Africa and 24% (102/434) of those born in south eastern and eastern Africa being HIV positive to 3.5% (8/230) of those born in the Horn of Africa. Although CSWs are at increased risk of HIV transmission (among other STIs) compared to the general population, they are not included in surveillance systems as separate from those infected via sex between men, sex between men and women, or injecting drug users (IDUs). As such, their contribution to the HIV epidemic in the UK is unknown even though it has been reported that they do not play a significant role in the spread of HIV (Evans et al., 2001).

Behaviour

The 2000 National Survey of Sexual Attitudes and Lifestyles (NATSAL) (Johnson et al., 2001; Wellings et al., 2001), found that there have been notable increases in risky sexual behaviour since the 1990 survey and although a greater proportion of individuals used condoms at first intercourse, this increase in condom use was offset by increases in risk in other areas. The NATSAL survey samples of minority ethnic groups were not large enough to enable conclusions to be drawn on the sexual behaviour of any particular minority ethnic group (8.8%) (Johnson et al., 2001). However, the MAYISHA studies (Fenton et al., 2002, MAYISHA II Collaborative group, 2005) present data on Africans in the UK. The MAYISHA II study states that: 17% women and 13% of men had been diagnosed with an STI in the previous five years; more men than women reported two or more sexual partners in the previous year (33% vs 18%); and 57% of the men compared to 49% of women reported condom use during their last sexual episode. Eight percent of men and women reported ever having had same sex partners, compared to 5% of NATSAL 2000 (general population) respondents.

Data from the 2004 Sigma Research, (Weatherburn et al, 2005) which reports on MSM, showed that 39% of participating men had not had an STI test in the last year. The survey also revealed that 10% had paid or been paid for sex in the last year, with over 40% of men involved with commercial sex work citing gay websites as their most likely method of finding partners. The 2005 Sigma research (Hickson et al., 2006) also showed that 19% of men whose last HIV test was negative or who had never been tested for HIV had participated in receptive unprotected anal intercourse with a partner of unknown status in the last year. This figure rose to 32% of untested men with thirty or more partners.

Policy

Following the National strategy for sexual health and HIV (DH, 2001b) and its subsequent *Implementation action plan* (DH, 2002) the Department of Heath have published two HIV toolkits; one aimed at commissioning (DH, 2003a) and one aimed at health promotion (DH, 2003b). General HIV prevention guidance, which has been developed by MedFASH (2003) and endorsed by the *Delivering choosing health action plan* (DH, 2005a), states that NHS and partner agencies, through a multi-agency planning group, should develop, implement and evaluate a local HIV prevention programme. This programme, among other things, should be informed by an up-to-date assessment of local HIV prevention need; should be multi-agency and multi-disciplinary, linking clinic-based, community and education-based initiatives and should be audited for impact. For prevention with African communities

the Department of Health has published *HIV and AIDS in African communities: A framework for better prevention and care* (DH, 2005b), which urges the involvement of communities in prevention initiatives as well as the thorough documentation of the prevention work being carried out.

Conceptual framework

This review draws on the same conceptual framework as the original Evidence Briefing. For full details of the framework see Ellis et al., 2003.

Methodology

Searches were carried out across 18 databases and websites. Two reviewers independently appraised all the titles and abstracts of the identified references to determine whether to retrieve the full paper on the basis of the following criteria:

- English language only
- Published since 2001
- Systematic review, synthesis, meta-analysis or literature review
- Relevance to HIV prevention in the UK and the priority populations identified
- Presents (and synthesises) data from primary evaluation studies of intervention effectiveness.

Two reviewers appraised the identified papers (including journal articles, book chapters and reports) independently using a critical appraisal tool. The appraisal process sought to identify the extent to which papers were systematic, transparent, analytically sound and relevant. A joint decision was made regarding whether the paper was to be classified as Category 1, 2, 3, 4 or 5 based on how well the paper satisfied the criteria of the appraisal tool. See Appendix C for further details.

In total, 32 papers were judged to be Category 1, 2 or 3 and went on to analysis and synthesis (Tables 1-5). Category 1 and 2 papers provided information on which to base evidence statements of sufficient, tentative, insufficient or no evidence, regarding the effectiveness of HIV prevention interventions.

Findings and discussion

The Evidence section (4) considers the evidence for each of the priority populations and makes 'evidence statements' by level of intervention. The voluntary counselling and testing (VCT) section (5), cost-effectiveness section (6) and theory-based evidence section (7) consider the evidence by priority and other specific populations (e.g. women) and evidence statements of effectiveness are made.

The Discussion section (8) addresses the key questions posed by the review of reviews, namely:

What works to reduce the sexual risk of HIV transmission among the priority populations in the UK? What works to change the modifying factors that influence sexual risk behaviours for HIV transmission?

We found that most reviews analysed only behavioural outcomes rather than health promotion outcomes. Although a limited number of studies included the effects of interventions on changes in modifying factors (e.g. knowledge or skills), there was insufficient evidence available with which to make conclusions.

Are theory-based interventions more likely to be effective?

We found that interventions using behavioural theories were more effective than interventions without a theoretical framework at reducing risky sexual behaviour. It was often the case that studies used more than one theory to inform their intervention. However, occasionally reviews failed to clearly report the theory used in their study.

Are multi-component interventions more likely to be effective?

The evidence suggests that multi-component group-work with MSM results in greater effectiveness. Evidence also suggests that voluntary counselling and testing (VCT) in combination with another component is more effective than VCT alone. We found no other evidence of effective multi-component interventions for MSM, CSWs, Africans or people with HIV.

What works to reduce inequalities in sexual risk for HIV transmission?

We found no evidence of interventions that addressed the issue of inequality. In particular we found no evidence relating to the effectiveness of HIV prevention interventions that depended upon socioeconomic status or vulnerable groups.

What interventions are cost-effective?

We found review-level evidence to suggest HIV prevention interventions can be cost-effective and cost-saving when aimed at MSM. However, we found insufficient evidence to make any conclusions regarding the cost-effectiveness of interventions for CSWs, Africans, and people with HIV.

Implications for policy and practice

The policy and practice implications reported below are based solely on the evidence statements of sufficient or tentative review-level evidence of effectiveness. As such, they consider only the evidence provided and retain the gaps and limitations that have been previously discussed. They ought not to

be considered alone but in conjunction with the policy and practice recommendations from the original review of reviews (Ellis et al., 2003), other non-review evidence and other relevant source information.

Implications for policy and practice with men who have sex with men (MSM)

The evidence presented for MSM populations in this update support cognitive-behavioural individuallevel intervention; cognitive-behavioural group work focusing upon risk-reduction, sexual negotiation and communication skills; peer-related group level interventions; and community-level interventions involving peer and popular opinion leaders. Very few of the interventions reported were carried out within the UK therefore some findings may not be transferable to the UK's MSM population. We recommend that future research should:

- Place interventions in relevant theoretical frameworks
- Undertake individual-level cognitive-behavioural interventions, including self-justification, risk-reduction, education, and counselling
- Undertake group-level peer-led interventions, that address, for example, issues of intimacy, relationships, communication skills, coping skills, interpersonal skills, risk-reduction and relapse prevention.

Implications for policy and practice with commercial sex workers (CSWs)

Our findings show that although there was evidence supporting the effectiveness of peer-led community level interventions with female CSWs in the previous evidence briefing (Ellis et al., 2003), these same methods were not effective when used with male sex workers.

Two Category 3 studies demonstrated effectiveness for peer-led interventions and condom distribution schemes. However, these interventions were not carried out in the UK and may represent different CSW and patron populations to those in the UK. Also, several studies were mentioned (by Albarracin et al., 2005) but no details were provided, and therefore we are unable to make any practice recommendations.

Implications for policy and practice with people with HIV (PWHIV)

In the review we made tentative evidence statements regarding the effectiveness of partner notification and small-group level interventions for people with HIV. Very few of the interventions reported were carried out within the UK and therefore not all findings may be transferable to the UK's PWHIV population. We recommend that future research should:

- Place interventions in relevant theoretical frameworks
- Undertake partner notification, including the choice between patient or provider referral
- Undertake small-group work, including stress management, HIV coping, and counselling.

Implications for policy and practice with African communities in the UK

The reviews identified for this update have revealed tentative evidence to conclude that group-level interventions may have a modest effect on the sexual risk behaviours of black and minority ethnic (BME) heterosexual adults. Tentative review-level evidence was also available to conclude that community-level interventions may be effective for females, adolescents and heterosexual males. However, interventions reported in this update took place outside the UK, primarily in the US with African American, Hispanic and multi-ethnic populations. Given the cultural differences to the UK's African populations (see African section in the Discussion, 8.11) we suggest that these findings are not transferable to the UK's African population, therefore we are unable to make any specific practice recommendations.

Implications for policy and practice with voluntary counselling and testing (VCT)

Based on the evidence identified for this update no further conclusions can be made regarding the effectiveness of VCT interventions, other than those stated in the original review of reviews (Ellis et al., 2003). However, we did conclude that there is tentative review level evidence to support the effectiveness of mass media interventions in influencing the uptake of HIV VCT in MSM, women, heterosexual men, and the general public. As there is insufficient evidence regarding the harms of HIV VCT we suggest that VCT interventions and promotion are aimed only at high-risk populations.

• Undertake mass media promotion for VCT aimed at MSM and African populations.

Recommendations for research

The research recommendations below build on those previously stated in the original review of reviews and aim to address the gaps that exist in the current intervention evidence.

General recommendations for secondary research/reviews

Although we have found evidence of high quality, well-reported reviews, almost all of them originate from outside of the UK. As such, all original recommendations remain applicable. To the secondary research recommendations we would add that there is a need to:

- Conduct a review of UK-based primary studies with respect to cost-effectiveness of HIV prevention interventions
- Conduct a review of good quality UK-based qualitative research of HIV prevention interventions
- Ensure that future studies report on the details of the population included in the intervention, and state the proportion of HIV positive people included
- Ensure that future studies report the details of the intervention level (individual, group, community).

General recommendations for primary research

To the primary research recommendations we would add that there is a need to:

- Underpin primary research with relevant theory
- Include full details of the theory/theories used to be reported
- Standardise the reporting of primary research
- Design research studies that analyse findings in order to identify the component characteristics that result in sexual risk reduction
- Explore further research on migrant populations resident in the UK who come from countries with high HIV prevalence
- Include more rigorous qualitative research of HIV prevention at all levels and with all target groups (including migrant populations living in the UK)
- Explore more research at the socio-political level for HIV prevention with all priority populations
- Explore more research that considers the potential effects of criminal prosecutions for the sexual transmission of HIV on onward transmission of the virus.

Recommendations for research with men who have sex with men (MSM)

In addition to the recommendations above and those in the original review of reviews, those below apply to MSM. There is a need for research to:

- Investigate HIV treatment optimism and its impact upon sexual risk-taking behaviour
- Consider interventions of sexual risk-taking, including social norms regarding intentional unsafe sexual practices, and use of the internet to solicit sexual partners
- Explore interventions using the primary outcome of serodiscordant or unknown status unprotected anal intercourse
- Look at disabled MSM and equality
- Examine African MSM and explore issues such as cultural impediments to sexual health promotion interventions
- Look at the intervention effectiveness and cost-effectiveness of individual versus group level interventions underpinned by cognitive-behavioural techniques
- Examine the effectiveness of interventions that include, for example, condom distribution, skills training, sexual communication, sexual health knowledge and HIV disclosure.

Recommendations for research with commercial sex workers (CSWs)

There is a need for research with CSWs to:

- Be carried out within a relevant theoretical framework
- Include an understanding of coping styles, social support needs, problem-solving skills and health promotion practices
- Explore the effectiveness of condom distribution schemes for both male and female CSWs
- Consider workplace interventions focusing on men who potentially use CSWs such as truck drivers

 Investigate interventions with male CSWs in the UK, including community-based out-reach interventions.

Recommendations for research with African communities in the UK

- There is an urgent need for UK-based culturally sensitive research with African women, men and adolescents to be carried out at all intervention levels
- There is also a need for specific research to be conducted with African migrant populations living in the UK.

Recommendations for research with people with HIV

There is a need for research with people with HIV to:

- Investigate HIV treatment optimism and its impact upon sexual risk-taking behaviour
- Explore the potential harms of partner notification
- Include community-based peer-led condom and contraceptive promotion
- Be aimed at serodiscordant couples
- Address the issue of access to services for people with HIV
- Explore the effects of interventions including coping skills and disclosure on people with HIV
- Examine the effectiveness of community-based organisation interventions compared to clinicbased interventions.

Recommendations for research in relation to voluntary testing and counselling (VCT)

• There is a need for UK-based research on the effects of an HIV test resulting in a positive test on the sexual risk behaviour of serodiscordant couples.

2 Introduction

2.1 Background

In 2003, the Health Development Agency (HDA) published HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission (Ellis et al., 2003). Since then, sexual health and HIV has continued to be at the forefront of policies, guidelines and initiatives, and these have identified the HDA review of reviews as a primary resource for ascertaining what works in the field of HIV prevention. The Medical Foundation for Sexual Health and HIV has published Recommended Standards for NHS HIV Services (including a section on prevention services) (MedFASH, 2003) and the Department of Health (DH) published its plan to improve the health of the population Choosing Health: making healthy choices easier (DH, 2004). In this key national policy document, sexual health was named as one of six key priorities, and the HDA evidence briefings were identified as the source of information on effective interventions. In 2005 DH published the action plan Delivering Choosing Health, which illustrates the economic benefit of reducing HIV transmission and endorses the MedFASH standards for HIV services. In 2005 the HDA joined with the National Institute for Clinical Excellence (NICE) to form the new National Institute of Health and Clinical Excellence (and retained the acronym NICE). The new organisation is committed to continue to build on the evidence base in public health. In light of the Government's continued commitment to sexual health and HIV and the necessity to ground HIV prevention in good evidence, it was deemed necessary to update the 2003 review of reviews to ensure that the most recent reviewlevel evidence is available for planning services and improving the sexual health of the population.

NICE evidence briefings report on **reviews of reviews**, sometimes referred to as tertiary level research. They consist of detailed expositions of the strengths and weaknesses of the evidence from reviews, identification of gaps in the evidence, an analysis of future primary and secondary research needs, and a discussion of the implications of the evidence for policy and practice. The full rationale for carrying out reviews of reviews is given in the 2003 Evidence Briefing on HIV prevention (Ellis et al., 2003). Each document has a freestanding summary that is published separately. The documents are also published on and supported by the NICE website at www.publichealth.nice.org.uk. This website also contains electronic copies of, or means of access to, the original reviews upon which the evidence briefings draw (if they are in the public domain).

The evidence from the previous 2003 briefing (Ellis et al., 2003) suggested that interventions with men who have sex with men are more likely to be effective if they are:

• Placed within the broader context of men's lives, addressing the range of factors which influence risk at both the personal level (e.g. knowledge, skills) and the structural level (e.g. discrimination towards gay men, gay community norms towards condoms).

- Tailored and targeted to specific sub-populations of men who have sex with men, for instance black gay men and working-class gay men.
- Multi-component (using small group work), focusing on risk reduction, sexual negotiation and communication skills training and rehearsal (e.g. through role-play or identifying 'triggers').
- There is some evidence that interventions delivered at the community level (particularly peer-led) can be effective in influencing the sexual risk behaviours for commercial sex workers.
- There is also evidence to conclude that HIV counselling and testing can influence sexual risk behaviours among individuals other than CSWs and MSM. However, there is not enough evidence to make conclusions about the harms of VCT.

The previous review of reviews was unable to identify any review-level evidence from which to make policy and practice recommendations for HIV prevention with African communities and with people with HIV.

As with the 2003 review of reviews, the current briefing tells us more about the gaps in research than it does about effective interventions. Therefore, it will be of greatest interest to policy makers, planners and commissioners of research operating at strategic levels, including those within the Department of Health. A secondary audience are those undertaking research, including practitioners and agencies outside of academia. In addition, this evidence briefing can be used to inform some areas of policy and practice. This evidence briefing will be published on the NICE website at www.publichealth.nice.org.uk.

2.2 Scope of the Evidence Briefing on HIV prevention

This updated evidence briefing reports on a review of reviews assessing the effectiveness of interventions to impact on the factors which influence the sexual risk behaviours for HIV transmission. As in the original Evidence Briefing, the aims were to:

- Identify and appraise all relevant reviews, including systematic reviews, literature reviews, syntheses and meta-analyses
- Analyse and synthesise these papers to highlight 'what works' to prevent or reduce the sexual risk of HIV transmission for the UK's priority populations
- Highlight conflicting evidence, and gaps in the evidence, and provide recommendations for research (as well as some initial recommendations for policy and practice).

The Evidence Briefing focused on:

- Interventions whose ultimate goal is to influence the sexual behaviour that puts individuals at risk of HIV infection – i.e. the sexual transmission of HIV
- Interventions that aim to impact on the structural determinants of HIV risk, including national and local policy, planning, organisational and community issues
- HIV voluntary counselling and testing (HIV VCT)
- Interventions that aim to address inequalities
- The cost effectiveness of interventions.

In relation to the above, the Evidence Briefing concentrated on the priority populations for HIV transmission in the UK, namely:

- Men who have sex with men (MSM)
- African communities in the UK
- Commercial sex workers (CSWs)
- People with HIV.

The Evidence Briefing excluded:

- The role of condom effectiveness, post-exposure prophylaxis (PEP), microbicides, treatment of sexually transmitted infections (STIs), or circumcision in reducing HIV transmission
- Interventions to prevent occupational, injecting drug use and mother-to-child transmission.

Separate evidence briefings address interventions to prevent both STIs (Ellis and Grey 2004, updated in 2006 by Downing et al.) and teenage pregnancy (Swann et al., 2003) have also been published.

2.3 Context

2.3.1 Policy context

In 2001, the Department of Health published the first *National Strategy for Sexual Health and HIV* (DH, 2001). This set out five main aims, including; to reduce transmission of HIV (with a national goal of a 25% reduction in newly acquired HIV by 2007); to reduce the prevalence of undiagnosed HIV; improve health and social care for people with HIV; and reduce stigma associated with HIV. The subsequent *Implementation Action Plan* (DH, 2002) made it clear that the DH would continue to support targeted work with MSM through the Community HIV and AIDS Prevention Strategy (CHAPS), coordinated by the Terrence Higgins Trust; and the DH encouraged commissioners and providers to adopt the associated *Making It Count* strategy (Hickson et al., 2000) for local prevention activity.

As part of its commitments in the *Action Plan*, in 2003 the DH published two sexual health and HIV commissioning toolkits, one on commissioning (DH, 2003a) and one on health promotion (DH, 2003b). The sexual health promotion toolkit gives practical advice on making inter-agency work effective, running health promotion projects, managing group work, developing resources and managing outreach work. The document also emphasises the need for more research, and the need for health professionals to evaluate and document their health promotion interventions, in terms of intermediate outcomes (e.g. knowledge, attitude) as well as behavioural outcomes (DH, 2003b).

The white paper *Choosing Health: making healthy choices easier* (DH, 2004) identified six key priorities:

- (1) tackling health inequalities
- (2) reducing the numbers of people who smoke
- (3) tackling obesity
- (4) improving sexual health
- (5) improving mental health and wellbeing, and
- (6) reducing harm and encouraging sensible drinking.

The spotlight on sexual health has renewed determination to tackle sexual ill-health in the population, and provides additional resources in order to do so. In the action plan the Government outlined its plans to improve sexual health services in a new £300m programme over the next three years. The main aims are to:

- Use £50m to implement a national and regional sexual health campaign aimed at young men and women to promote condom use and explain the risks of unprotected sex
- Make a commitment to sexual health in England through additional funding to deliver multidisciplinary sexual health services in a range of settings
- Expand the national Chlamydia screening programme to cover the whole of England by March 2007 (with an additional £80m to help achieve this goal)
- Carry out an audit of contraceptive services in 2005 in order to improve service provision
- Focus on modernising Genito-Urinary Medicine clinics, with an investment of £130m over three years including upgraded prevention services with an additional £40m provided for this purpose
- Ensure that every individual referred to a GUM clinic has an appointment within 48 hours by 2008.

Delivering Choosing Health (DH, 2005a) points out the economic benefit of reducing HIV transmission. While none of the aims directly addresses HIV prevention among the key groups identified here (MSM, African communities, CSWs, people with HIV), the action plan does endorse the MedFASH HIV standards for services (2003), which, in turn, devotes a section to HIV prevention services.
The MedFASH standards state that NHS and partner agencies, through a multi-agency planning group, should develop, implement and evaluate a local HIV prevention programme which:

- Is part of a broader sexual health promotion programme
- Is informed by an up-to-date assessment of local HIV prevention need
- Is adequately resourced in a way which reflects priorities identified through needs assessment
- Is developed and delivered with the involvement of target populations
- Is coordinated across primary and secondary healthcare, and community settings
- Is multi-agency and multidisciplinary, linking clinically based, community and education-based initiatives
- Includes work with specific population groups as well as with the general population
- Complements nationally funded and delivered HIV prevention campaigns
- Has protocols governing work undertaken with each population group
- Includes interventions at individual, small group and population level
- Is provided by appropriately skilled practitioners with knowledge of the issues and confidence in raising them sensitively with their clients and service users
- Is audited for impact.

The NHS and partner agencies should facilitate ongoing professional training and development for healthcare and other workers with a role in HIV prevention. This includes the development of understanding and skills to provide culturally competent prevention activities with different population groups, including black and minority ethnic communities.

The recently published *HIV* and *AIDS* in African communities: A framework for better prevention and *care* (DH, 2005b) describes some of the differences between African communities compared to the population as a whole in terms of their population structure (they are younger on average), education (a higher than average proportion with qualifications), and employment (lower than average). It goes on to summarise the issues, some of which include: the diversity of cultures; barriers caused by language; lack of awareness of services available; the double discrimination faced by African men who have sex with men (from the predominantly white gay population and from heterosexual Africans); and uncertainty within migrant and asylum seeking communities. The document also summarises the many research needs, including: in epidemiology and monitoring; sexual behaviour; gender issues; and effectiveness of health promotion. It urges the involvement of communities if interventions are to be appropriate and effective, and argues for the need to document prevention work that is being done. It recommends that specialist advice on welfare entitlement, housing, and legal status and immigration should be provided.

2.3.2 Epidemiology – prevalence of HIV in the UK

At the end of 2005 there were an estimated 63,500 (range: 59,500-68,800) people living with HIV in the UK, almost one third of whom were unaware of their infection (UK Collaborative Group for HIV

and STI Surveillance, 2006). During 2005 there were 7,645 new diagnoses of HIV, contributing to a total of 79,442 up to the end of 2005 (HPA, 2007). In the same year there were 785 diagnoses of AIDS, contributing to a cumulative total of 22,199 at the end of 2005. Not only has the number of HIV diagnoses been rising year on year (from 2,552 in 1990), the shape of the epidemic in the UK has changed over the years: during 2005, 32% (2,477) of HIV reports related to men who have sex with men, whereas this was the source of 67% of all reports in 1990. Diagnoses of HIV infections passed on during sex between men and women increased from 21% of all reports in 1990 to 57% in 2005 4,324) (HPA, 2007).

Where country of infection was reported, the majority (75%, 3,000) of heterosexual infections in the UK in 2005 were reported to have occurred in African countries. There are a small but growing number of infections acquired within the UK from sex between men and women, from 245 in 2000 to 583 in 2005 (HPA, 2007). However, the actual level of transmission in the UK may be higher because of the tendency for doctors to classify patients as having been exposed in Africa if they had sex in both the UK and Africa (Brown, 2000). Unlinked anonymous testing carried out as part of a survey of African communities in London, Luton and West Midlands (the MAYISHA II project), revealed an overall HIV prevalence of 14% of those tested. The prevalence in each subgroup was indicative of the prevalence in the country of birth, with 26% (12/46) of those born in southern Africa and 24% (102/434) of those born in south eastern and eastern Africa being HIV positive to 3.5% (8/230) of those born in the Horn of Africa. Of those anonymously testing positive, 66% (93/141) were undiagnosed based on their recall of previous voluntary testing (51 had never tested, 28 reported testing negative and 14 did not know/could not remember).

HIV therefore continues to impact unequally on particular communities, namely MSM and certain African communities. Non-sexual transmission routes of HIV accounted for relatively few diagnoses in 2005, with injecting drug use at 2% (162), mother to child transmission at 2% (117) and blood transfusion (generally from countries where blood is not routinely screened) representing well under 1% (20) of cases (HPA, 2007). It is generally accepted that the low proportion of cases attributed to injecting drug use is due to the prompt introduction of harm reduction such as syringe exchange schemes (EMCDDA, 2002).

Epidemiological information can also be used to measure the uptake of voluntary counselling and testing (VCT), and the proportion of those diagnosed as a result of VCT is one way to measure the effectiveness of this prevention activity. Estimates from the unlinked anonymous survey suggest that in 2005, 80% of MSM and 82% of heterosexuals attending GUM clinics had VCT. The proportion of HIV positive individuals remaining undiagnosed whose infection could have been diagnosed during that clinic visit were they to have been offered and accepted VCT fell from 63% in 2000 to 35% in 2005. Of those who did not have a test, at least 36% of MSM and 45% of heterosexuals had been offered a test. Of those who refused testing, 8% of MSM and under 1% of heterosexuals were HIV-infected (UK Collaborative Group for HIV and STI Surveillance, 2006).

2.3.3 Sexual behaviour

Sexual behaviour is a major factor determining the incidence of HIV. The 2000 National Survey of Sexual Attitudes and Lifestyles (NATSAL, 2000) provided the most recent data on sexual behaviour among the general population in Britain (Johnson et al., 2001; Wellings et al., 2001). The survey found that a greater proportion of individuals used condoms at first intercourse: 80% of 16-19 year olds in NATSAL 2000, compared with 60% of 25-29 year olds in NATSAL 2000 (the latter group were 15-19 years old at the time of the first survey in 1990). However, this increase in condom use was offset by increases in risk in other areas. Since the first NATSAL in 1990, the 2000 survey found:

- A greater number of lifetime partners from 8.6 to 12.7 for men; and from 3.7 to 6.5 for women
- A lower age at first intercourse from 17 years to 16 years for men; and from 21 years to 16 years for women
- A greater proportion with concurrent partnerships (two or more partners at the same time) 14.6% for men (20% for 15-24 year olds); and 9% for women (15% for 15-24 year olds)
- A greater proportion with two or more partners in the past year and who did not use condoms consistently from 13.6% to 15.4% in men; and from 7.1% to10.1% in women
- A greater proportion of men who reported ever having a partner of the same sex from 3.6% to 5.4%.

Although the NATSAL survey covered the entire UK population, the sample size was not adequate to make inferences about any particular minority ethnic group (8.8% of total sample) (Johnson et al., 2001). Specific studies on sexual behaviour in UK African communities of a large enough scale were not available until the publication of the MAYISHA studies (the first of which was a large pilot in London in 1999 (n=748 individuals) (Fenton et al., 2002), and the second had three sites, London, Luton and West Midlands in 2004 (n=1359 individuals) (MAYISHA II Collaborative Group, 2005). In the MAYISHA II study, 17% of women and 13% of men had been diagnosed with an STI in the previous five years, similar to the previous (London only) study figure of 16% for both sexes. More men than women reported two or more sexual partners in the previous year (33% vs 18%). Fifty seven percent of men compared to 49% of women reported condom use during the last sexual episode (this was higher than the previous, London only, survey figures of 46% and 43% respectively). Eight percent of men and women reported ever having had same sex partners, compared to 5% of NATSAL 2000 (general population) respondents.

Data on sexual behaviour needs to be interpreted with caution, acknowledging that unprotected sex is not the same as unsafe sex. Similarly, data on the number of partners can provide an indication of potential STI risk, even where intercourse is protected.

The main UK wide source of information about gay men's behaviour is the annual Sigma Research surveys. Periodically, the survey asks whether respondents have had unprotected anal intercourse:

the last three surveys to include this were 1995 and 2000 (Hickson et al., 2001) and 2004 (Weatherburn et al., 2005). MSM in 2000 reported increases in unprotected anal intercourse (UAI) with both regular and casual partners compared with 1995:

- A greater proportion have UAI with a regular partner from 42.3% in 1995 to 54.9% in 2000
- Fewer men having anal intercourse with a regular partner in 2000 always used a condom compared with men in 1995
- A greater proportion of men have casual UAI from 9.8% of all men in 1995 to 14.4% in 2000.

The 2004 survey (Weatherburn et al., 2005) did not repeat these questions. The 2004 findings on self-reported behaviour showed that the more sexual partners a man reported the more likely he was to have had an HIV or STI test within the last year (22% of men who had only one partner had an HIV test, and 24% an STI test, compared to 43% of men who had 15-29 partners having an HIV test and 54% were tested for STIs). Men who reported more sexual partners expressed a wish for more sexual education than men who had fewer sexual partners. However, a worryingly high number of men who had many different sexual partners were not tested; 55% of men who had an STI test in the last year. The survey also revealed that 10% had paid or been paid for sex in the last year, with over 40% of men involved with commercial sex work citing gay websites as their most likely method of finding partners.

The 2005 survey (Hickson et al., 2006) also revealed that in the past year three quarters of all men questioned had engaged in sex with a man of unknown serostatus. If these findings are representative of the whole MSM population they indicate a widespread lack of communication around HIV status among MSM. The survey also revealed that 11% of men who had anal intercourse in the last year never used a condom and inconsistent condom use was as common as consistent condom use. Furthermore, 19% of men whose last HIV test was negative or who had never been tested for HIV, had participated in receptive unprotected anal intercourse with a partner of unknown status in the last year. This figure rose to 32% of untested men with thirty or more partners.

2.3.4 Inequalities

The statistics on incidence clearly show how HIV disproportionately affects communities already suffering from considerable inequalities relating to their sexual orientation, ethnicity and gender. This can be further mitigated by socio-economic factors, for example MSM with lower educational achievement are at greater risk of acquiring HIV (Weatherburn et al., 1999).

Inequalities may persist after diagnosis. For example, the death rate of those with HIV in the north of England is higher than that in the south (Cosgrove et al., 2001a). Also, in a North West study, those from poorer areas were more likely to require hospital treatment after controlling for sex, age, stage of

disease and route of infection (Cook et al., 2004). In the context of prevention, if these findings are applicable to other areas of the UK, the increased burden on HIV services in areas with already higher prevalence may decrease resources for preventing onwards transmission.

The group suffering from the highest levels of absolute material deprivation, marginalisation and stigmatisation are asylum seekers. The prevalence of HIV among this group is likely to reflect that of their country of origin. Based on this assumption, the recent findings of an expert medical panel reporting to the All Party Parliamentary Group on AIDS (Gazzard et al., 2005), estimated there were approximately 900 asylum seekers with HIV entering the country in 2003/2004. Currently asylum seekers have the right to HIV treatment whilst seeking asylum. However, due to the policy of dispersal without reference to medical needs, asylum seekers may find themselves in areas where the medical services are unaware and unprepared for their health status and may not have sufficient expertise. Should their asylum application fail, individuals lose the right to free treatment for HIV. The independent expert medical panel argued that on public health grounds treatment should not be withdrawn, since HIV positive failed asylum seekers may not necessarily be deported within the target time period and could transmit HIV to others. They identified inconsistencies in policies between HIV and other infectious diseases such as TB (where treatment is available to all regardless of immigration status) and other STIs (free treatment to all), and recommended that HIV be reclassified as an STI. UNAIDS and WHO (2005) argue that it is essential to the fight against HIV/AIDS that universal access to treatment and care is achieved. However, they state that this is part of the wider fight against barriers to access that include stigma, discrimination, gender inequalities and other human rights violations. As such it is imperative that treatment be freely available to all those who are aware of their positive status as part of the global effort to combat the infection and to help to create a more effective environment for HIV prevention (Joint United Nations Programme on HIV/AIDS (UNAIDS) and World Health Organisation (WHO), 2005, p7).

2.3.5 Economic costs

Aside from the public health and humanitarian reasons for averting HIV infections (see section 2.3.4), the economic arguments are substantial. The monetary value of preventing a single onward transmission is estimated to be somewhere between £0.5 and £1 million in terms of individual health benefits and treatment costs (DH, 2001, p11). This economic benefit is highlighted again in *Choosing Health*, which stated that a 25% reduction in HIV incidence would save the NHS £500 million a year (DH, 2004).

The average lifetime treatment costs for an HIV positive individual is between £135,000 and £181,000, with the average annual cost per person for service provision for those on Highly Active Anti-Retroviral Therapy (HAART) put at approximately £15,000 (MedFASH, 2003). However, historical funding inequalities in England mean that treatment services in the north have received less funding per HIV positive patient than those in the south (Cosgrove et al., 2001a). The Government also give £16.5 million a year to local authorities for social care services for HIV care (DH, 2005a). The

voluntary HIV sector has a long history of involvement in health and social care of HIV. Analysis of the economics of the voluntary HIV sector (for the North West of England) estimated that for every £1 spent by the statutory sector on voluntary sector care, the statutory sector received £2 worth of services (after accounting for the combined effects of volunteering and non-statutory income), and therefore represent good value for money (Cosgrove et al., 2001b).

The previous Evidence Briefing of HIV prevention (Ellis et al., 2003) found no review-level evidence on cost effectiveness in HIV prevention relevant to the key priority populations. The STI Evidence Briefing (Ellis and Grey 2004) did identify tentative review level evidence in the reviews by Mackay (2000) and Pinkerton et al. (1998). Despite the evidence base for cost effectiveness being small, Mackay (2000) cited two studies of behaviourally effective interventions that have been shown to be cost-effective. One focused on HIV prevention, the other on HIV/STI and pregnancy prevention. He indicated that the high cost of treating HIV and STIs meant that appropriate interventions were likely to be cost saving and cost-effective. Pinkerton et al. (1998) concluded that due to the high costs of treating HIV/STIs and their consequences, most risk reduction interventions would be cost saving. These authors also stressed the greater societal savings from initiatives targeted at adolescents rather than adults, due to the reductions in life expectancy (for HIV), the reproductive health consequences for women (e.g. pelvic inflammatory disease (PID), ectopic pregnancy) and more opportunity for ongoing transmission. This updated Evidence Briefing presents new review-level evidence on cost effectiveness in section 6.

Treat with Respect, the expert medical panel's report on asylum seekers, included reference to the economics of denying treatment to those who have been refused asylum (Gazzard et al., 2005). This report suggests that in 2003/04 there were approximately 900 members of this population with HIV, and £13.5m per annum would be needed to treat them. However, this was considerably less than the lifetime costs of treating all those who may become infected if onwards transmission was not prevented. The expert group concluded that it made economic and public health sense to treat any one in the UK for HIV. It recommended that PCTs should be reimbursed for treatment, since small PCTs with a low prevalence of HIV find it difficult to fund treatment for dispersed asylum seekers.

2.4 Conceptual Framework

This review draws on the same conceptual framework as the original Evidence Briefing. For full details of the framework see Ellis et al. (2003). Here we provide a brief outline of the modifying factors and intervention levels.

Factors that influence sexual behaviour are personal modifying factors and structural modifying factors.

Personal modifying factors include:

- Knowledge and awareness such as awareness of HIV risk and protective measures available (e.g. condoms) and knowledge of serostatus
- Attitudes, motivations and intentions e.g. towards condom use and serostatus
- Beliefs and perceptions e.g. self-efficacy and self-esteem
- Skills e.g. in safe sex negotiation, sexual communication and condom use.

Structural modifying factors include:

- Economic factors e.g. funding
- Policy factors e.g. sexual law and regulations
- Societal factors e.g. community or peer norms regarding sexual behaviour such as condom use or abstinence and social attitudes towards, for example, discrimination towards MSM and access to services
- Organisational factors e.g. service organisation and the sustainability of available services.

Levels of HIV prevention interventions

Individual level interventions are one-to-one interactive interventions, including:

- Counselling (voluntary counselling and testing (VCT see below), one-to-one counselling and couples counselling)
- Individual cognitive-behavioural therapy (CBT)
- Telephone helplines
- Face-to-face detached or outreach work
- Some internet-based work

Voluntary counselling and testing (VCT)

VCT includes a range of interventions and is defined by UNAIDS (2004) as client-initiated testing to learn serostatus with pre-testing, provided either individually or in groups, and can be followed up by post-test counselling.

Group level interventions include:

- School-based sex education
- Small group work, including cognitive-behavioural therapy (CBT)

Community level interventions include:

- Media (mass media/small media)
- Condoms and lubrication provision
- Peer education and social diffusion
- Community empowerment and development
- Some internet interventions (chat rooms)
- Some organisational/institutional interventions

Socio-political level interventions include:

- Legislation (including sexual legislation and anti-discrimination laws)
- Equality work that reduces discrimination and social exclusion
- Facilitating interventions (e.g. research and development and multi-agency working)
- Resource allocation
- Regulation (e.g. labelling of condoms).

3 Methodology

A standardised methodology was developed by the HDA (now the National Institute for Health and Clinical Excellence) for the analysis and synthesis of review-level evidence (Kelly et al., 2002). The specific methodology used in this Evidence Briefing is outlined below.

3.1 Identification of the literature

3.1.1 Electronic searches

A wide-ranging set of search strategies was developed for the original Evidence Briefing by Ellis et al. (2003) in an attempt to identify as many types of 'review' relevant to the subject.

Briefly, the topic strategy was developed to identify papers with a focus on any intervention with the goal of reducing the sexual risk of transmitting or acquiring HIV. The topic search was developed using a combination of: specific HIV prevention terms; terms for interventions specific to the topic area; and general health promotion/health education/public health terms combined with topic terms such as HIV or sexual behaviour. For each database, strategies were developed with a combination of Medical Subject Headings ('MESH') (using various 'operators' such as 'and', 'or', 'adjacent' and 'near') and text words (appropriately truncated, e.g. 'HIV prevent\$') in order to achieve as many relevant reviews as possible.

The following exclusions were built into the searches in an attempt to limit the results to the scope of the Evidence Briefing:

- Developing countries
- AIDS treatment
- Vertical transmission (mother-to-child)
- Transmission related to injecting drug use
- Post-exposure prophylaxis.

The strategies used in this evidence briefing were based on an updated 2003 HDA search strategy. An example of a search strategy is provided in Appendix B.

The following databases and websites were searched:

- The Cochrane Library (including HTA database, DARE [Database of Abstracts of Reviews of Effects] and NHS EED [Economic Evaluation Database])
- The 'Wider Public Health' report
- MEDLINE
- TRIP (database with access to largest collection of 'evidence-based' material on the web)
- SIGN (Scottish Intercollegiate Guidelines Network)

- Health Evidence Bulletins Wales
- National Guidelines Clearing House
- NCCHTA (National Co-ordinating Centre for Health Technology Assessment) website
- NICE (National Institute for Health and Clinical Excellence) website
- REFER (Research Findings Electronic Register)
- National Research Register
- Clinical Evidence
- EMBASE
- Sociological Abstracts
- PsycINFO
- Cinahl (Cumulative Index to Nursing and Allied Health Literature)
- Sociofile
- EPPI-Centre's** Register of Reviews of Effectiveness (RoRE).***

All databases were searched from January 2001 to January 2006 for references published in the English language. Results were downloaded into a Reference Manager database.

3.1.2 Selection and filtering

Two reviewers independently appraised all the titles and abstracts of the identified references to determine whether to retrieve the full paper on the basis of the following criteria:

- English language only.
- Published since 2001.
- Systematic review, synthesis, meta-analysis or literature review.
- Relevance to HIV prevention in the UK and the priority populations identified.
- Presents (and synthesises) data from primary evaluation studies of intervention effectiveness.

A joint decision was made as to whether the full paper would be retrieved for critical appraisal; if the reviewers disagreed, or no clear decision could be made on the basis of the title or abstract, the full paper was obtained for appraisal.

3.2 Critical appraisal

Two reviewers appraised the identified papers (including journal articles, book chapters and reports) independently. The appraisal process sought to identify the extent to which papers were:

• *Systematic* – do the review authors apply a consistent and comprehensive approach? If repeated, would this give the same results?

- *Transparent* are the review authors clear about the processes involved, the individual study results, the findings and conclusions drawn from these? Is there sufficient detail to repeat?
- *Analytically sound* do the authors undertake the appropriate methods of analysis and are they undertaken correctly?
- Relevant is the content relevant to the UK and the priority populations under consideration?

3.2.1 Assessing the quality of the reviews

The HDA's standard protocol (Swann et al., 2003; Kelly et al., 2002) and critical appraisal tool (CAT) were used to make judgements about the quality of the identified papers. See Appendix C for further details.

For a further discussion about the assessment of quality please see the previous Evidence Briefing (Ellis et al., 2003).

3.2.2 Categorisation of reviews

The original Evidence Briefing categorised papers as 'Core', 'Supplementary' and 'Discarded'. However, for this update we used the gradient system introduced by Ellis and Grey (2004) in their Evidence Briefing into prevention of sexually transmitted infections. Those reviews that would have previously been categorised as 'Core' would now be categorised as Category 1 or 2 depending upon whether they passed stages one and two or only stage one. Reviews that would have previously been categorised as 'Supplementary' would now belong to Category 3. The categorisation process is only briefly described here, for further information see Ellis and Grey (2004).

A joint decision was made regarding whether the paper was to be classified as category 1, 2, 3, 4 or 5. The 'data pool' for the Evidence Briefing consists of those reviews categorised as 1, 2 or 3. The categories are:

Category 1

The review satisfied the criteria in both stages one and two of the critical appraisal tool (CAT) (see Tables 4-5). Category 1 reviews are included within the data pool and analysed to derive evidence statements.

Category 2

The review passed stage one of the CAT, but failed to meet all the criteria within stage two – usually because it was not clear how the review had assessed the quality of the included studies. Category 2 reviews are also included within the data pool and analysed to derive evidence statements.

Category 3

The research question(s) was judged to be highly pertinent, and the paper is a review of interventions; however it failed to pass stage one of the CAT – it is not clear how the included studies were identified and the review had additional shortcomings. Category 3 reviews are included within the data pool and so may provide some evidence about effectiveness, but these reviews alone are insufficient to inform conclusions about effectiveness.

Category 4

This paper is either not a review of effectiveness of interventions, or it is not a review in its own right (e.g. it may extract findings from another review). However it is relevant and contains useful policy, background, and epidemiological or interpretive information.

Category 5

The paper is not directly relevant and is therefore not used.

3.3 Analysis and synthesis

Reviews categorised as 1, 2 or 3 were subject to analysis and synthesis. Detailed structured summaries were completed for each Category 1 and 2 review (see Appendix E), and brief summaries of all Category 1, 2 and 3 reviews (see section 3.2).

The previous document by Ellis et al. (2003) outlined a conceptual framework for how interventions affect the modifying factors influencing sexual behaviour, and thus the approach to evidence of effectiveness taken in the Evidence Briefing. 'Interventions' were defined and discussed with regards to the appropriateness of different outcome measures for determining 'success' in HIV prevention.

Ellis et al. (2003) grouped and analysed the evidence according to each of the UK's priority populations. They then looked at the available data for interventions delivered at the individual, group, community and socio-political level. This update of that review carries on their approach.

3.3.1 Evidence statements

Category 1, 2 and 3 reviews were used to derive 'evidence statements' about types and features of interventions. Where relevant review evidence has been considered in conjunction with that of the original Evidence Briefing (Ellis et al., 2003) evidence statements were derived based upon the combined evidence. The evidence statements reflect the strength of the conclusions made by the review(s), the outcomes used to judge effectiveness, the category of the review, and any inconsistencies within and between reviews. The reviews are used to derive the following evidence statements:

• Sufficient review-level evidence – clear evidence/conclusions from at least one Category 1 review, with no conflicting evidence/conclusions between Category 1 reviews

- Tentative review-level evidence tentative evidence/conclusions from Category 1 review; or conflicting conclusions from Category 1 reviews; or clear conclusions from at least one Category 2 review
- Insufficient review-level evidence no evidence/conclusions from Category 1 reviews and only tentative evidence/conclusions from Category 2 reviews; or clear evidence/conclusions from Category 3 reviews
- No review-level evidence no evidence/conclusions from Category 1, 2 or 3 reviews.

Where there was no change from the evidence statement in the original Evidence Briefing this was stated in the text and followed by the original evidence statement from Ellis et al. (2003). Evidence statements remained the same where there was evidence from the same category or lower than in the original Evidence Briefing and/or where all current evidence agreed with the previous evidence.

Evidence statements were amended from the original Evidence Briefing (Ellis et al., 2003) where current evidence originated from a higher category. Evidence statements also changed where there was evidence from the same category or higher where the authors' conclusions were more definitive than those in the original Evidence Briefing or referred to different target populations than in the original Evidence Briefing (Ellis et al., 2003).

3.4 Presentation of data

Complete bibliographic lists of the Category 1-5 papers are presented in the **References (section 11)**.

The **Results** (section 4) contains brief summaries of Category 1-3 reviews and includes a summary table of the assessment of each Category 1-3 review according to the critical appraisal process.

Appendix E contains detailed summaries of Category 1 and 2 reviews.

The findings of Category 1, 2 and 3 reviews are analysed and synthesised in **section 5** (**The evidence**). Each priority population section begins with an overview of the relevant review papers. Within each priority population section there are sections on individual-level, group-level, community-level and socio-political level interventions, in addition to sections on inequalities. Within each section, having weighted the evidence, a number of 'evidence statements' were made, based on the findings from the Category 1, 2 and 3 review papers (see section 3.2.1). The findings of the cost-effectiveness review are analysed in **section 6** (**Cost-effectiveness**). There are sections on each of the priority populations and VCT. The findings on theory-based effective interventions are analysed in **section 7** (**Theory-based interventions**). There are sections on each of the priority populations and VCT. The findings on theory-based and the priority populations and VCT. The findings on theory-based effective interventions are analysed in **section 7** (**Theory-based interventions**). There are sections on each of the priority populations and VCT. The **Discussion (section 8)** considers methodological issues and limitations of the Evidence Briefing. In the **Recommendations (sections 8 and 9)**, gaps and inconsistencies in the evidence are drawn out

from the evidence statements from section 5, 6 and 7, particularly in relation to the UK and in the context of the those reported in the previous Evidence Briefing.

4 The Results

4.1 Summary of identification of papers

A total of 3,387 titles and abstracts were screened for inclusion. This compares to 800 in the original Evidence Briefing. Of these, 189 papers were ordered as full papers and appraised using the CAT. In total, 32 papers were judged to be Category 1, 2 or 3 and went on to analysis and synthesis. A summary of the process of review identification is shown in Figure 1. Tables 1-5 show a summary of the review-level evidence.



Figure 1. Flowchart showing the process of identification and categorisation of papers

4.2 Summary of reviews in Categories 1-3

Table 1. Summary of Category 1 reviews

		.				Relevance to areas covered						
Author and date*	Title	Dates covered	Interventions/scope	Country/countries of research**	No. studies	MSM	BME	CSW	PWHIV	VCT		
Elwy A.R. et al. (2002)	Effectiveness of interventions to prevent sexually transmitted infections and human immunodeficiency virus in heterosexual men.	Not specified	Interventions for heterosexual men. Reported separately on IDUs African- American men, minority at-risk clinic attendees (among others e.g. truck drivers, military etc.), interventions described are small group.	United States; Australia; Brazil; Senegal; India; Namibia; Scotland; Mozambique; Thailand.	28 studies							
Herbst J.H. et al. (2005)	A meta-analytical review of HIV behavioural interventions for reducing sexual risk behaviour of men who have sex with men.	up to July 2003	An analysis of international HIV prevention interventions for MSM. Included studies were based upon theoretical models.	United States; New Zealand; Canada; Puerto Rico; Mexico; England; Scotland; Australia; Brazil; Russia; Bulgaria.	33 studies							
Matthews C. et al. (2002)	A systematic review of strategies for partner notification for sexually transmitted diseases including HIV/AIDS.	1991 to 2001	An examination of partner notification. Includes two studies on PWHIV	United States	10 studies (2 PWHIV)							
Pedlow C.T. and Carey M.P. (2003)	HIV sexual risk-reduction interventions for youth: A review and methodological critique of randomized controlled trials.	Up to September 2000	To provide a review and methodological critique of HIV risk reduction interventions for adolescents that employed an RCT design and measured sexual risk behaviour as an outcome measure.	United States	22 studies							
Rees R. et al. (2004)	HIV health promotion and men who have sex with men (MSM): A systematic review of research relevant to the development and implementation of effective and appropriate interventions.	1992 to March 2003	Focuses on vulnerable men.	Scotland; Australia; England; United States.	26 studies							
Vidanapathirana J. et al. (2005)	Mass media interventions for promoting HIV testing.	1980 to April 2004	Mass media interventions in relation to changes in HIV testing	London; Australia; Canada; Israel; Scotland; United States.	14 studies							

* Complete bibliographic details of level 1, 2 and 3 review papers are provided in the References (section 11). **Country/countries where available from review of reviews

Table 2. Summary of Category 2 reviews

		Dates		Country/countries of	No.	Relevance to areas covered					
Author and date*	Title	covered	Interventions/scope	research**	studies	MSM	BME	CSW	PWHIV	VCT	
Albarracin D. et al. (2005)	A test of major assumptions about behaviour change: A comprehensive look at the effects of passive and active HIV-prevention interventions since the beginning of the epidemic.	1985 to September 2003	An analysis of theoretical interventions for behaviour change.	United states; Africa; Ghana; India; Italy; Peru; Zimbabwe; United Kingdom; Thailand; Nigeria; Mexico; Honduras; Hungary; Kenya; South Africa; Uganda; Senegal; Greece; Australia; Brazil; Namibia; Puerto Rico; Mozambique; Malawi; Holland.	191 studies						
Farnham P.G. et al. (2002)	Cost-effectiveness of counselling and testing to prevent sexual transmission of HIV in the United States.	1987 to 1999	Review the cost-effectiveness of HV counselling and testing in various settings and populations in preventing the sexual transmission of HIV.	United states	12 studies						
Johnson B.T. et al. (2003)	Interventions to reduce sexual risk for the Human Immunodeficiency Virus in adolescents, 1985- 2000	Up to December 2000	To summarise studies that have tested the efficacy of HIV sexual risk-reduction interventions in adolescents	United States; South Africa; Sweden; Netherlands; Namibia.	44 studies						
Johnson W.D et al. (2005)	HIV intervention research for men who have sex with men: a 7 year update	Up to 2005	Behavioural HIV prevention interventions for MSM	United states; Canada; New Zealand; England; Scotland	40 studies						
Johnson W.D. et al. (2002)	HIV prevention research for men who have sex with men: A systematic review and meta- analysis.	Up to June 1998	HIV prevention interventions for MSM	United States	9 studies						
Johnson W.D. et al. (2003)	Interventions to modify sexual risk behaviours for preventing HIV infection in men who have sex with men.	1988 to 1997	Behavioural interventions to reduce risk for HIV or STI transmission among MSM.	United States	13 studies						
Logan T.K. et al. (2002)	Women, sex and HIV: Social and contextual factors, meta-analysis of published interventions, and implications for practice and research.	1992 to 1999	Examines social and contextual factors related to HIV risk-behaviour in women	United States	30 studies						
Merzel C. & D'Afflitti J. (2003)	Reconsidering community-based health promotion: promise, performance, and potential.	1980 to 2001	Review US-based community health promotion programmes. Covers cardiovascular disease, smoking prevention, cancer prevention, substance abuse, HIV prevention and other health issues.	United States	11 HIV- related projects						

		Dates		Country/countries of	No.	Relevance to areas covered					
Author and date*	Title	covered	Interventions/scope	research**	studies	MSM	BME	CSW	PWHIV	VCT	
Mullen P.D. et al. (2002)	Meta-analysis of the effects of behavioural HIV prevention interventions on the sexual risk behaviour of sexually experienced adolescents in controlled studies in the United States.	1988 to 1998	An analysis of the effect of behavioural and social interventions on sexual risk of HIV among sexually active adolescents	United States	16 studies						
Neumann M.S et al. (2002)	Review and meta-analysis of HIV prevention intervention research for heterosexual adult populations in the United States.	1998 to 2002	Examines behavioural and social interventions and their effects on the adoption of safer sex behaviour and incidence of STDs.	United States; Tanzania; Thailand; Greece; Nicaragua; Honduras	24 studies						
Pinkerton S.D. et al. (2002a)	A review of the cost-effectiveness of interventions to prevent sexual transmission of HIV in the United States.	1996 to 2002	Review and summarise the literature on the cost-effectiveness of interventions to prevent sexual transmission of HIV in the US.	United States	16 studies						
Robin L. et al. (2004)	Behavioural interventions to reduce incidence of HIV, STD and pregnancy among adolescents: A decade in review.	1990s	A review of young people's risk reduction interventions and behavioural changes in predominantly ethnic minority groups.	United States	22 studies						
Semaan S. et al (2002a)	A meta-analysis of the effect of HIV prevention interventions on the sex behaviours of drug users in the United States.	Up to June 1998	Examines HIV intervention studies for reducing the sexual risk behaviours of drug users by reducing unprotected sex or increasing the use of male condoms.	United States	33 studies						
Van Empelen et al. (2003)	Effective methods to change sex-risk among drug users: a review of psychosocial interventions	Not clear	Examine the current state of knowledge on the effectiveness of HIV prevention interventions with respect to sexual behaviour among the population of drug users and the relation between effective intervention methods and their underlying theory.	United States; Australia	17 studies						
Weinhardt L.S. (2005)	HIV diagnosis and risk behaviour. In: Positive Prevention: Reducing transmission among people living with AIDS/HIV	1985 to November 2003	An examination of the impact of HIV testing and counselling on sexual risk behaviour. Studies were conducted in North America, Africa and Europe. Studies included individuals and couples. Groups are women, MSM, drug users, sero-discordant couples and mixed.	United States; Africa; England; Holland	21 studies						
Wilson B.D.M. and Miller R.M. (2003)	Examining strategies for culturally grounded HIV prevention: A review.	1985 to 2001	Culturally grounded research. Predominantly USA based studies looking at small group interventions and ethnic minorities (including two MSM studies)	United States; Puerto Rico	17 studies						

* Complete bibliographic details of level 1, 2 and 3 review papers are provided in the References (section 11).

**Country/countries where available from review of reviews.

Table 3. Summary of Category 3 reviews

		Dates		Country/countries of	No.	Relevance to areas covered					
Author and date	litie	covered covered		research**	studies	MSM	BME	CSW	PWHIV	VCT	
Elkavich A. et al (2005)	Young people living with HIV. In: Positive Prevention: Reducing transmission among people living with AIDS/HIV	Not clear	An evaluation of the intervention activities and the related theoretical frameworks	United States	4 studies						
Fortenberry J.D. (2002)	Clinic-based service programs for increasing responsible sexual behavior.	Not clear	US clinic-based service programmes for increasing responsible sexual behaviour, including (1) clinic-based educational/counselling, (2) school clinic- based condom distribution programmes, and (3) clinic-based STD/HIV screening programmes.	United States	37 studies						
Johnson-Masotti A.P. et al. (2003)	Efficacy and cost-effectiveness of the first generation of HIV prevention interventions for people with severe and persistent mental illness.	Not clear	Review of the efficacy and cost-effectiveness of RCTs of HIV prevention interventions.	United States	5 studies						
Kelly J.A. and Kalichman S.C. (2002)	Behavioral research in HIV/AIDS primary and secondary prevention: Recent advances and future directions.	Not clear	Review research advances made over the past decade in both HIV primary and secondary prevention, and emerging issues in behavioural research.	United States; Thailand; Kenya; Zimbabwe; Rwanda; Tanzania; Trinidad. Others unclear.	Not clear						
Kirby, D. (2002)	Effective approaches to reducing adolescent unprotected sex, pregnancy and childbearing.	1980 or later	Reviews four groups of programmes which have reasonably strong evidence including (a) sex and HIV education curricula with specified characteristics, (b) one-on-one clinician- patient protocols in health care settings, (c) service learning programmes, and (d) intensive youth development programmes.	United States	73 studies						
Metsch L.R. (2005)	Interventions in Community Settings. In: Positive Prevention: Reducing transmission among people living with AIDS/HIV.	Not clear	Interventions for prevention aimed at people living with HIV	United States	Not clear						
Pinkerton S.D. et al. (2002b)	Cost-effectiveness of community-level approaches to HIV prevention: A review.	Up to 1999	Review the published literature on the cost- effectiveness of community-level HIV prevention interventions.	United States	12 studies						
Pinkerton S.D. & Holtgrave D.R. (2002c)	Assessing the cost-effectiveness of alternative approaches to HIV prevention. In: <i>Beyond</i> <i>condoms: Alternative approaches to HIV</i> <i>prevention.</i>	Not clear	Overview of cost-effectiveness of alternative approaches to HIV prevention.	United States; Tanzania; UK; France; Nairobi; Zaire.	Not clear						
Ross M.W. and Williams M.L. (2002)	Effective targeted and community HIV/STD prevention programs.	Not clear	Overview of community interventions and interventions targeting specific groups at risk of HIV/STD.	France; United States; Uganda; Thailand; Kenya; Zimbabwe; Tanzania.	Not clear						

Author and date	Title	Dates	Interventions/seens	Country/countries of	No.	Relevance to areas covered				
	The	covered	interventions/scope	research**	studies	MSM	BME	CSW	PWHIV	VCT
Wolitski R.J. et al. (2005)	An Overview of Prevention with People Living with HIV. In: Positive Prevention: Reducing transmission among people living with AIDS/HIV.	Not clear	Examines prevention interventions with people living with HIV, including risk reduction interventions	United States	Not clear					

* Complete bibliographic details of level 1, 2 and 3 review papers are provided in the References (section 11).

**Country/countries where available from review of reviews.

Table 4. Results of the CAT for Category 1 and 2 papers

			Stage one						Sta	ge two			
	1		2		2d		3		3d		4	5	
	Clear aim/ question Comprehensive search strategy				Inclusion criteria	Asse	Assessment of quality of included studies			Quality o	of analysis	Concl- usions	
Author and date*	inc. Setting population, intervention	2a Identify approp. range of source databases	2b Additional search strategies **	2c Specifies search terms	specified	3a Definition of quality	3b Tool used to assess quality	3c How assessment s generated	studies' quality	4a Study findings presented clearly & consistently	4b Study findings analysed appropriatel y	related to findings	CATEGORY
Elwy A.R. et al. (2002)		year not specified											1
Herbst J.H. et al. (2005)				(available from authors)									1
Matthews C. et al. (2002)													1
Pedlow C.T. and Carey M.P. (2003)													1
Rees R. et al. (2004)			(in part)										1
Vidanapathirana J. et al. (2005)													1
Johnson W.D. et al. (2002)			(in part)										2
Johnson W.D. et al. (2003)			(in part)										2
Johnson W.D. et al. (2005)			(in part)										2
Albarracin D. et al. (2005)													2
Farnham P.G. et al. (2002)													2
Johnson B.T. et al. (2003)			(in part)										2
Logan T.K. et al. (2002)			(in part)										2

* Complete bibliographic details of level 1, 2 and 3 review papers are provided in the References (section 11).

** Additional search strategies involve follow-up of references/journals, consultation with experts in the field and grey literature searches.

Table 4. Results of the CAT for Category 1 and 2 papers (continued)

			Stage one						Stag	je two			
	1		2		2d		3		3d	4		5	
	Clear aim/ question	Clear aim/ question			Inclusion criteria	Assessment of quality of included studies			Assessme nt of the	Quality	of analysis	Concl- usions	
Author and date*	inc. Setting population, intervention	2a Identify approp. range of source databases	2b Additional search strategies **	2c Specifies search terms	specified	3a Definition of quality	3b Tool used to assess quality	3c How assessments generated	studies' quality	4a Study findings presented clearly & consistently	4b Study findings analysed appropriately	related to findings	CATEGORY
Merzel C. and D'Afflitti J. (2003)		1					1		1				2
Mullen P.D. et al. (2002)			(in part)										2
Neumann M.S. et al. (2002)													2
Pinkerton S.D. et al. (2002a)													2
Robin L. et al. (2004)			(in part)										2
Semaan S. et al. (2002)													2
Van Empelen et al. (2003)			(in part)										2
Weinhardt L.S. (2005)			(in part)										2
Wilson B.D.M. and Miller R.M. (2003)			(in part)										2

* Complete bibliographic details of level 1, 2 and 3 review papers are provided in the References (section 11).

** Additional search strategies involve follow-up of references/journals, consultation with experts in the field and grey literature searches.

Table 5. Results of the CAT for Category 3 papers

	Stage one						Stage two										
	1		2		2d		3		3d		4	5					
	Clear aim/ question inc.	Compre	Inclusion criteria	Assessment of quality of included studies				Quality	of analysis	Conclusions related to							
Author and date*	population, intervention	2a Identify approp. range of source databases	2b Additional search strategies **	2c Specifies search terms	specified	3a Definition of quality	3b Tool used to assess quality	3c How assessments generated	Assessment of the studies' quality	4a Study findings presented clearly & consistently	4b Study findings analysed appropriately	findings	CATEGORY				
Elkavich A. et al (2005)						-	-	-	-	-	-		3				
Fortenberry J.D. (2002)						-	-	-	-	-	-		3				
Johnson-Masotti A.P. et al. (2003)													3				
Kelly J.A. and Kalichman S.C. (2002)						-	-	-	-	-	-		3				
Kirby, D. (2002)						-	-	-	-	-	-		3				
Metsch L.R. (2005)						-	-	-	-	-	-		3				
Pinkerton S.D. et al. (2002b)													3				
Pinkerton S.D. & Holtgrave D.R. (2002)													3				
Ross M.W. and Williams M.L. (2002)						-	-	-	-	-	-		3				
Wolitski R.J. et al. (2005)						-	-	-	-	-	-		3				

* Complete bibliographic details of level 1, 2 and 3 review papers are provided in the References (section 11).

** Additional search strategies involve follow-up of references/journals, consultation with experts in the field and grey literature searches.

5 The evidence

We follow each section with evidence statements regarding the effectiveness of interventions. Where there is no change from the evidence statement in the original Evidence Briefing this is stated in the text and followed by the original evidence statement by Ellis et al. (2003). This Evidence Briefing also contains evidence statements on topics not previously included in the original Evidence Briefing (Ellis et al., 2003). Further, evidence statements were amended from the original Evidence Briefing. For further details see section 3.3.1.

We present review-level evidence on each of the following priority populations:

- Men who have sex with men (MSM)
- Commercial sex workers
- People with HIV
- African communities in the UK
- Voluntary counselling and testing (VCT)

5.1 Men who have sex with men (MSM)

5.1.1 The reviews

Men who have sex with men (MSM) is a broad term and includes gay and bisexual men, and those who have sex with men but do not self-identify as either gay or bisexual.

Category 1: Herbst et. al., 2005; Rees, 2004.

Category 2: Albarracin et al., 2005; Wilson and Miller, 2003; Johnson et al., 2003; Johnson et al., 2002; Johnson et al., 2005.

Category 3: No evidence

The original Evidence Briefing (Ellis et al., 2003) reported on two reviews. Although the current review level evidence incorporates some of the same primary studies it also reports on studies that were not previously included. The primary studies can be found in Table 6 Appendix F. The majority of the studies were based on group interventions, however each level of intervention was well represented in studies involving MSM.

Outcomes

The most frequently reported outcomes were: condom use, unprotected sex, and number of partners.

5.1.2 The findings

The findings from the evidence have been examined at the following levels: individual, group, community and socio-political. We present evidence identified by intervention level with an overview of the studies and examples of the interventions.

5.1.2.1 Individual-level intervention

Five Category 1 and 2 reviews (Herbst, 2005; Rees, 2004; Johnson, 2003; Johnson et al., 2005; Albarracin, 2005) examined individual-level interventions carried out with MSM populations. Two reviews did not address individual-level interventions (Johnson et al., 2003; Wilson and Miller, 2003). The most frequently used individual-level interventions were counselling, self-justification and diary keeping.

- A prevention intervention study (Dilley et al., 2002) reported in four reviews (Herbst, 2005; Rees et al., 2004; Johnson et al., 2005; Albarracin, 2005) used self-justification methods, and adds weight to Gold and Rosenthan's study (1995) included in the original Evidence Briefing. This study (description from Rees et al., 2004) included a comparison of three intervention groups with a control group. All groups received standard counselling for HIV testing. Intervention groups differed in whether or not they received an additional cognitive-behavioural session examining the self-justification of high-risk sexual behaviour and/or a sex diary.
- A similar study (Imrie et al., 2001; description from Rees et al., 2004), reported in three reviews (Herbst, 2005; Rees et al., 2004; Johnson et al., 2005), compared a one-day cognitivebehavioural workshop to standard counselling. The control group included brief sexual risk-taking counselling; offers of contact tracing, and possibility of referral to clinic or community prevention education or counselling services. The intervention group received everything that the control group received as well as a cognitive-behavioural workshop.
- Rees et al. (2004) stated that both these studies were methodologically 'sound'. Rees et al. (2004) pooled the findings from these studies and showed that interventions based upon cognitive-behavioural techniques were significantly more effective than standard counselling methods in reducing serodiscordant unprotected anal intercourse (UAI) when measured at six months follow-up (odds ratio (OR) 0.49; 95% confidence interval (CI) 0.29 0.84).
- Several other individual-level interventions included counselling methods, however mixed results were reported. Picciano et al., (2001; description from Johnson et al., 2005) provided feedback by telephone regarding a baseline risk-assessment to MSM practicing risky sexual behaviour. This intervention effect, and similar intervention effects with minimal or no HIV prevention, showed favourable (PR = 0.93), but not statistically significant, results (CI = 0.78, 1.10). The EXPLORE study team (2004;description from Johnson et al. (2005) included ten one-to-one counselling sessions followed by maintenance sessions every three months. These sessions covered: risk-

assessment; sexual communication; knowledge of HIV serostatus; alcohol and drug use; triggers for unsafe sex; and motivational interviewing. Johnson et al., (2005) state that, "this individual intervention, although reporting a modest reduction in unprotected sex, was accompanied by a substantial reduction in new HIV infections" (p. 584).

- Individual-level interventions also covered a variety of risk-reduction and counselling techniques including: intensive case management (Sorensen et al., 2003 in Johnson et al., 2005); an intervention based on HIV risk education (Rotherham-Borus et al., 2004 in Johnson et al., 2005); one intervention based on HIV prevention counselling and safe sex education (Rosser, 1990 in Herbst, 2005; Johnson, 2003; Johnson et al., 2005); counselling methods (Richardson et al., 2004 in Johnson et al., 2004), ARRM) (Patterson et al., 2003 in Johnson et al., 2005); and/or relapse prevention interventions (Gold and Rosenthal, 1998 in Herbst, 2005; Rees, 2004; Johnson et al., 2005).
- Johnson et al. (2005) included all eight individual-level interventions mentioned above in a subgroup analysis, which revealed a 5% reduction in unprotected sex (RR = 0.91; CI = 0.62, 1.34). Johnson et al. (2005) concluded that, "the most favourable effects among individual-level interventions were those that addressed losses" (p.580). The limited information available on biological outcomes suggested that higher risk clients may be better served by individual-level interventions rather than other methods that aimed to introduce clients to potential partners who were also high-risk.

This evidence adds to the reviews from the original Evidence Briefing that referred to one-to-one 'selfjustification' counselling sessions.

New evidence statement: There is tentative review-level evidence to support the effectiveness of cognitive-behavioural individual-level interventions in influencing the sexual risk behaviours of MSM.

This is a change from the original Evidence Briefing where it was stated that there was insufficient evidence to either support or discount the effectiveness of any individual-level interventions in influencing the sexual risk behaviours of MSM.

5.1.2.2 Group-level intervention

Seven Category 1 and 2 reviews (Herbst, et. al., 2005; Johnson et al., 2003; Johnson et al., 2002; Johnson et al., 2005; Rees, 2004; Albarracin et al., 2005; Wilson and Miller, 2003) were identified and these included a total of 20 original group-level interventions to add to the seven reported in the previous Evidence Briefing. The most frequently used interventions were cognitive-behavioural therapy, relapse prevention, and skills training.

- Several studies have incorporated specific cognitive-behavioural and risk-reduction elements into their interventions (Kelly et al., 1993; Shoptaw et al., in press). This builds upon the findings from studies taken from Oakley et al. (1996) and Aggleton (1994) in the original Evidence Briefing.
- Kelly et al. (1993) (description from Herbst et al., 2005) reported on a cognitive-behavioural intervention targeted at depressed, HIV positive MSM. This also included a social support group where individuals could share their feelings about HIV. The intervention was provided over four months, with four sessions totalling six hours. Shoptaw et al. (in press; description from Johnson et al., 2005), reported on a 16 week-long intervention for methamphetamine-dependent MSM. This incorporated group education; cognitive skills development, recognition of triggers and drugabstinence reward schemes.
- Carballo-Dieguez (2004; description from Johnson et al., 2005) carried out an eight-session intervention that incorporated goal-setting and self-efficacy. Kalichman et al. (2001; described in Johnson et al., 2005) reported on a support group intervention that aimed to create sexual health and relationship plans, develop disclosure and communication skills, and to teach hazards of co-infection with other STIs. This intervention was delivered in five 120 minute long sessions. Harding et al., (2004; described in Johnson et al., 2005) provided an introduction to sadomasochistic sex, including risk-taking, emotional aspects, STI infection and HIV transmission, rights and responsibilities, legal issues, the role of fantasy, limits and boundaries. This intervention was delivered in four sessions of seven hours. Johnson et al. (2005) included the results of small group interventions (some of which were mentioned in the original Evidence Briefing) in a subgroup analysis, which revealed a significantly favourable effect on unprotected sex (PR = 0.80; CI = 0.72, 0.89).
- Risk-reduction methods such as relapse prevention interventions have also been reported (Miller, 1995; Roffman et al., 1997; Roffman et al., 1998). Miller (1995; described in Herbst et al., 2005) carried out a single session relapse prevention workshop (five hours) targeted at MSM that aimed to increase self-efficacy, self-worth and confidence. Roffman and colleagues (1997; description in Herbst et al., 2005) undertook an intervention involving telephone counselling to develop relaxation and coping skills, as well as motivational enhancement. This intervention lasted 14 sessions, including five events, over 14 weeks and lasted for a minimum of 21 hours. This intervention was delivered by peer counsellors, social workers, and facilitators. Roffman and co-authors (1998; described in Herbst et al., 2005) reported on an intervention underpinned by several theories, including relapse prevention. The intervention incorporated cognitive-behavioural group counselling on HIV education, motivational enhancement and goal setting. Seventeen sessions were carried out over 18 weeks.
- Peer-delivered prevention interventions have been included in recent review level evidence addressing behaviours in MSM (Amirkhanian et al., 2003; Remafedi, 1994; Toro-Alfonso et al.,

2002; Wolitski et al., 2005; Stall et al., 1999). Amirkhanian et al., (2003; described in Herbst et al., 2005) reported on an intervention that included training peer leaders to deliver HIV prevention messages. Remafedi (1994; description from Herbst et al., 2005) detailed an intervention aimed at young people that individualised HIV risk-assessment and counselling as well as peer-education and referrals. This intervention lasted for three sessions of a minimum of four hours over two days. Goldbaum et al., 1999 (unpublished) in Herbst et al (2005) carried out a risk-reduction intervention aimed at non-gay identifying MSM. Toro-Alfonso et al. (2002; described in Herbst et al., 2005) reported a peer-delivered intervention that included small group meetings and workshops focused on intimacy, relationships, drug abuse, sexual behaviour and STI/HIV. This intervention was provided over five sessions totalling 15 hours. Stall et al. (1999; described in Herbst et al., 2005) carried out an intervention based on theories of social learning, risk-reduction and relapse prevention, aimed at substance-using MSM and included standard drug treatment as well as coping skills, interpersonal skills and discussion of sexual issues.

 Herbst et al., (2005) commented that, "Although no behavioural theory in particular was associated with significantly better intervention efficacy than any other theory, interventions based on the diffusion of innovation theory and the model of relapse prevention showed greater point estimates for reduction in risk behaviour in the intervention group relative to the comparison group" (p237).

These findings support those in the previous Evidence Briefing. Current findings also add to the reviews from the original Evidence Briefing for peer-related group level interventions.

Evidence statement: No change from Ellis et al. (2003). There is sufficient review-level evidence to conclude that cognitive-behavioural group work, focusing on risk-reduction, sexual negotiation and communication skills training can be effective in influencing the sexual risk behaviours of MSM.

New evidence statement: There is tentative review-level evidence to conclude that peer-related group-level interventions can be effective in influencing the sexual risk behaviours of MSM.

- Additional studies included in the review level evidence (Perry, 1991; Rosser et al., 1990; Rosser et al., 2002; Sampaio et al., 2002; Kalichman et al., 2001; Harding et al., 2004) examined a variety of interventions that differed by session length and included components e.g. counselling, motivational approaches, skills and communication training.
- Perry (1991; description from Johnson et al., 2002), reported on a counselling intervention targeted at MSM, prior- and post- HIV testing in addition to one of three experimental conditions

to reduce emotional distress: individual stress prevention therapy; interactive video of HIV information and relaxation techniques; or both. This intervention was delivered in multiple sessions, over several weeks. Rosser et al., (1990; description from Johnson, 2002) reported on an intervention that included HIV prevention counselling, a video on safer sex, a video on eroticising safer sex and a 'stop AIDS' workshop. Rosser et al., (2002; described in Herbst et al., 2005) delivered a sex health seminar to address risk factors, HIV risk assessment and sex communication. Sampaio et al., (2002; described in Johnson et al., 2005) included safer sex workshops with games, role-playing, small-group discussion using verbal and non-verbal communication. Topics included; basic information, clarification of misconceptions, recognition of risk, non-genital practices, safer-sex practices, and sexual negotiation.

Three other interventions (Dockrell et al., 1999; Martin et al., 2001; Turner and Heywood, 2000) were reported by Rees et al. (2004), but were described as methodologically 'unsound'. They included small-group work, responses to sexual behaviour following the revelation of an undetectable viral load and workshop sessions respectively.

5.1.2.3 Multi-component small-group work

The previous Evidence Briefing concluded that multi-component small-group work involving, for example, skills training and the improvement of self-efficacy was effective in influencing risk behaviour. We found no evidence to disagree with this statement.

Evidence statement: No change from Ellis et al., 2003. There is sufficient review-level evidence to conclude that building a range of components into group-level interventions contributes to their effectiveness in influencing the sexual risk behaviours of MSM.

5.1.2.4 Multi-session small-group work

The previous Evidence Briefing concluded that there was not enough evidence to be able to assess the effectiveness of the number of sessions included in an intervention on the effect of the intervention. We found no evidence to disagree with this statement.

Evidence statement: No change from Ellis et al., 2003. There is insufficient review-level evidence to either support or reject the conclusions that multi-session interventions are, in themselves, more effective than single session interventions in influencing the sexual risk behaviours of MSM.

5.1.2.5 Community-level intervention

Five Category 1 and 2 reviews (Herbst, et. al., 2005; Johnson et al., 2003; Johnson et al., 2005; Rees, 2004; Wilson and Miller, 2003) included a total of 10 original community-level interventions to

add to the six reported in the previous Evidence Briefing. The most frequently used interventions were peer-related.

 Rosser (1991; described in Herbst et al., 2005) reported on a mass media intervention, 'The grim reaper', which was a fear-based advertisement. However, this study did not have a comparison group to be able to compute an effect. The voluntary counselling and testing section (5.5) provides further evidence of the effects of mass media interventions aimed at MSM that have informed the evidence statement below.

New evidence statement: There is tentative review-level evidence to support the effectiveness of mass media interventions in influencing the uptake of HIV VCT in groups of MSM.

- Six reviews contained studies regarding peer interventions (Elford et al., 2001; Flowers et al., 2002; Kegeles et al., 2002; Miller et al., 1998; Shepherd et al., 1997; Zimmerman et al., 1997). To summarise, examples of the peer-led interventions reported were: HIV conversations; empowerment training of AIDS prevention and health care and information (a 2002 update of the 1996 study by Kegeles reported in the original Evidence Briefing); and peer-education.
- Hoff et al. (1997) and CDC ACDP (1999) (both described in Johnson et al., 2005) both reported outreach interventions.
- Dahl et al. (1997) (described in Rees et al., 2004) reported a community-wide social marketing intervention that Rees et al. (2004) described as methodologically 'unsound'.
- All community-level interventions reported here (except Dahl et al., 1997 and Rosser, 1991) were included in a meta-analysis by Johnson et al. (2005). They reported that community level interventions were statistically significant and favourable effect (PR = 0.86; CI = 0.76, 0.96).

These studies add to those included in the original review and consolidate the evidence statement regarding peers and opinion-led interventions.

Evidence statement: No change from Ellis et al. (2003). There is sufficient review-level evidence to conclude that community-level interventions involving peers and popular opinion leaders can be effective in influencing the sexual risk behaviours of MSM.

5.1.2.6 General issues concerning the effectiveness of interventions with MSM.

The previous Evidence Briefing concluded that there was not enough evidence to support the conclusion that the duration of an intervention was associated with the likelihood of success. Ellis et al. (2003) stated that those interventions targeted to specific risk groups were more likely to impact on sexual risk behaviour. We have found no evidence to disagree with these statements.

Evidence statement: No change from Ellis et al. (2003). There is insufficient review-level evidence to support or discount the effectiveness of 'brief' interventions in influencing the sexual risk behaviour of MSM.

Evidence statement: No change from Ellis et al. (2003). There is sufficient review-level evidence to conclude that interventions for MSM are more likely to be effective if they are targeted and tailored to the specific community and, ideally, if implementation follows according to the findings of formative research.

5.1.2.7 Socio-political intervention

None of the reviews presented data on the effectiveness of socio-political level interventions in HIV prevention or impact upon health outcomes or risk behaviour with MSM.

Evidence statement: No change from Ellis et al. (2003). There is no review-level evidence to either support or discount the effectiveness of any socio-political interventions in influencing the sexual risk behaviours of MSM.

5.1.2.8 Inequalities

None of the reviews presented data on interventions targeting inequalities for MSM.

Evidence statement: No change from Ellis et al. (2003). There is no review-level evidence to either support or discount the effectiveness of interventions in addressing inequalities in sexual health for MSM.

5.1.2.9 Cost-effectiveness

Section 6 provides full details of the three cost-effectiveness reviews relevant to MSM (Pinkerton et al., 2002a; Pinkerton et al., 2002b; Pinkerton and Holtgrave, 2002). The original Evidence Briefing (Ellis et al., 2003) did not contain any relevant cost-effectiveness reviews.

New evidence statement: There is tentative review-level evidence to conclude that HIV prevention interventions aimed at MSM are cost-effective, and can be cost-saving.

5.2 Commercial sex workers (CSWs)

The term 'commercial sex workers' (CSWs) includes both men and women who sell sex, working on the street and in off street locations (e.g. brothels, massage parlours).

Category 1: Herbst et al., 2005; Elwy et al., 2002. Category 2: Albarracin et al., 2005. Category 3: Ross et al., 2002; Kelly et al., 2002.

Outcomes

The most frequently reported outcome was frequency of unprotected sexual intercourse.

5.2.1 Findings

The literature has been examined at the following levels: individual, group, community and sociopolitical. As there were a small number of primary studies, a brief summary of each is provided for each intervention level.

The three Category 1 and 2 reviews contained nine prevention interventions related to CSWs. There was some overlap with the studies included in the original Evidence Briefing (Ellis et al., 2003). Seven studies were found in addition to the 15 included previously.

5.2.1.1 Individual-level intervention

Elwy et al. (2002) reported one individual-level study by Jackson et al. (1997) that involved targeting the clients of CSWs. Outreach workers provided individual counselling, HIV testing and condom skills including negotiation, demonstration and promotion as well a free supply of condoms. Findings showed a significant reduction in the number of STIs and the number of sexual partners. Elwy and colleagues (2002) reported that this intervention was a moderate-quality study that used a prospective uncontrolled design, they add no other comments.

The previous Evidence Briefing reported on HIV testing and counselling with additional information and skills training but reported that there was no review-level evidence to either support or discount the effectiveness of any other individual-level intervention for influencing the sexual risk behaviour of CSWs.

Changed evidence statement: There is insufficient review-level evidence to either support or discount the effectiveness of any other individual-level interventions in influencing the HIV sexual risk behaviours of CSWs.

5.2.1.2 Group-level intervention

Elwy et al. (2002) reported on one group-level intervention by Herbst et al. (1999). This intervention involved targeting the patrons of CSWs in their workplace (port workers) via outreach workers and discussion groups. Findings showed that there was a significant increase in condom use and a reduction in the number of reported partners. However, there was no change in attitudes towards condoms or knowledge of AIDS. Elwy and colleagues made no conclusions regarding the relevance to the UK population.

Evidence statement: No change from Ellis et al. (2003). There is insufficient review-level evidence to either support or discount the effectiveness of any group-level interventions in influencing the HIV sexual risk behaviours of CSWs.

5.2.1.3 Community-level intervention

Four studies, not included in the previous Evidence Briefing, reported on the role of popular opinion leaders ('diffusion of innovations'), and other community wide, peer-based interventions, including those targeted at the clients of CSWs.

- Miller et al. (1998; description taken from Herbst et al., 2005) targeted CSWs and patrons of gay bars (hustler bars). The 'hustler bar project' involved popular opinion leaders delivering HIV risk information to three hustler bars in New York. Peers therefore, became outreach workers. However, results showed that the intervention favoured the comparison group (delayed treatment or wait list).
- Albarracin (2005) included three studies related to CSWs; Ford et al. (2002), Ford et al. (1999) and Leonard et al. (2000). No details of these studies were provided in Albarracin. However Elwy et al. (2002) provided details of the intervention carried out by Leonard et al. (2000). This intervention involved peer-education in the workplace (truck workers) including discussions about STIs/HIV, condom use skills and communication skills. Findings showed that although there was no change in the number of sexual partners, there was a significant increase in condom use, HIV knowledge, and self-efficacy for using condoms. However, the findings were viewed as unreliable by Leonard and colleagues (2000) as the reported number of sexual partners was not corroborated by female sex workers.
- Ross et al. (2002) and Kelly et al. (2002) described two studies aimed at CSWs. One study in Thailand (Hanenberg et al., 1994) involved a mass distribution and advertising of condoms in all CSW establishments which, within two years, resulted in an increase in condom use from 14% to 97% and, over five years, a 79% decrease in the five major STIs in men. Another study based in Kenya and Zimbabwe (Ngugi et al., 1996) used peer-educators to distribute condoms and safe-

sex information to CSWs. After one year, condom use had increased in Kenya from 5% to 37% and in Zimbabwe from 18% to 66%.

The previous Evidence Briefing reported on HIV testing and counselling with additional information and skills training.

Evidence statement: No change from Ellis et al. (2003). There is tentative review-level evidence to conclude that interventions delivered at the community level (particularly peerled interventions) can be effective in influencing the HIV sexual risk behaviours of patrons and CSWs.

5.2.1.4 Intervention level unclear

 Albarracin et al. (2005) reported on three additional studies, which included prevention education aimed at Greek CSWs (Papaevangelou et al., 1988), and behavioural change in CSWs and their clients (Van Griensven et al., 1998; Walden et al., 1999). No other details of the studies were provided.

5.2.1.5 Socio-political intervention

None of the reviews presented data on the effectiveness of socio-political level interventions for HIV prevention with CSWs.

Evidence statement: No change from Ellis et al. (2003). There is no review-level evidence to support or discount the effectiveness of socio-political interventions in addressing the sexual health of CSWs.

5.2.1.6 Inequalities

None of the reviews presented data on the effectiveness interventions in addressing inequalities for HIV prevention with CSWs.

Evidence statement: No change from Ellis et al. (2003). There is no review-level evidence to support or discount the effectiveness of interventions in addressing the inequalities in sexual health for CSWs.

5.3 People with HIV (PWHIV)

5.3.1 The reviews

Reviews related to this population include evaluations of targeted prevention interventions for people diagnosed with HIV/AIDS.

Category 1: Mathews et al., 2002; Herbst et al., 2005.

Category 2: Albarracin et al., 2005; Johnson et al., 2005.

Category 3: Elkavich et al., 2005; Kelly and Kalichman, 2002; Metsch et al., 2005; Wolitski et al., 2005.

Outcomes

The most frequently reported outcomes were on techniques for individuals with HIV/AIDS diagnosis to inform previous partners of their health status ('partner notification').

5.3.2 The findings

Findings have been examined at the following levels: individual, group, community and socio-political. As there were a limited number of original studies, a brief summary of each is provided at each intervention level. Category three reviews have been briefly summarised for each level.

There was no overlap with reported studies from the original Evidence Briefing as this publication found no review level evidence to report on the effectiveness of interventions for PWHIV^{*}. As such many of the evidence statements relating to PWHIV are new.

5.3.2.1 Individual-level intervention

Six reviews reported on 10 individual level interventions. The most commonly reported interventions were partner notification and counselling.

 Landis (1992; reported in Mathews et al., 2002) described an assessment of the choice between either the provider or patient notifying previous partners (patients could notify all or some of their partners themselves. The remaining partners as well as those who failed to present to services after two weeks were contacted by counsellors). The comparison group included patient referral, and interview with a counsellor to discuss the process of notification. The results of this study showed that offering a choice between patient or provider referral compared to patient referral alone resulted in more partners being notified. In the experimental group, those provided with the

^{*} Two studies (Landis, 1992 & Coates et al., 1989) were reported in the supplementary review tables in the original review.

^{- 54 -} HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission, 2006.
choice notified more previous partners, and more of these partners subsequently tested HIV positive.

- Levy (1998; reported in Mathews et al., 2002) (trial ongoing). This study involved the choice between provider and patient referral; with index patients (patient taking part in the study) receiving help with identifying and naming partners as well as counselling about notification. Indigenous community workers notified those partners the index patients did not want to notify themselves, without revealing the identity of the index patient. This study also included community-based testing for the partner/partners of the index patient. The comparison group received patient referral, whereas index patients received help with identifying and naming partners, patients were also counselled about notification. Community-based testing was also provided for the partners of index patients in this group. In the group with the choice between provider and patient referral, 82% of index patients chose provider referral for at least one partner (covering 71% of partners), suggesting that provider referral is preferred and facilitates notification. However, at the time of reporting this study was ongoing and comparison group data were not available.
- Mathews et al. (2002) stated that "partner notification affects either of two outcomes, i.e. prevention of morbidity in those notified, or prevention of transmission to others – both are benefits" (p.297).
- Johnson et al. (2005) included four individual level interventions, (two of which were also reported in Metsch et al., 2005), (Patterson et al., 2003; Rotheram-Borus et al., 2004; Richardson et al., 2004; Sorensen et al., 2003). These interventions consisted of different designs and approaches including a social cognitive theory based intervention, which included sexual negotiation; prevention counselling; case management and a youth based 'staying healthy' intervention. Participants in these studies were predominantly MSM and the meta-analysis results for MSM studies (including studies with small proportions of HIV positive MSM where status was not an entry requirement) showed that although individual-level interventions for this population resulted in reductions in unprotected sex (RR = 0.91; CI = 0.62, 1.34), they were not statistically significant like those for small-group and community interventions.
- Of the Category 3 studies, Elkavich et al. (2005) reported on another individual level intervention that showed significant increases in condom use, 'outercourse' (i.e. non-penetrative sex), selfefficacy and safe sex knowledge (Butler et al., 2003). Kelly and Kallichman (2002) reported on three individual-level interventions (including one-to-one counselling) that were more likely to increase protected sex, condom use (including within serodiscordant couples) and result in a reduction in the incidence of HIV (Kamb et al., 1998; Allen et al., 1992a; Padian et al., 1993).

New evidence statement: There is tentative review-level evidence to support the effectiveness of partner notification and its influence on the sexual risk behaviours of people living with HIV.

5.3.2.2 Group-level intervention

Six reviews included studies of group-level interventions. The most commonly reported interventions were support groups.

- Coates et al. (1989; description taken from Herbst et al., 2005) examined an intervention involving stress reduction training with systematic relaxation, health habit change and stress management skills. A significant effect on the number of sexual partners was observed in a combined analysis with other studies (Herbst et al., 2005).
- Johnson et al. (2005) included Coates et al. (1989) in addition to five other group level interventions (Wolitski et al., 2005; Cleary et al., 1995; Kelly et al., 1993; Rotheram-Borus et al., 2001a; Kalichman et al., 2001). These additional interventions included peer-led work to build a sense of community; two support groups; small-group work aimed at improving HIV coping among others; and counselling. The majority of these small-group interventions reported in Johnson et al. (2005) included MSM participants. The results of a meta-analysis (including one other study with a small proportion of HIV positive MSM where status was not an entry requirement) showed that the effects were more favourable to this population than individual-level interventions (RR = 0.71; Cl = 0.51, 0.99).
- Two Category 3 level reviews (Elkavich et al., 2005; Metsch et al., 2005) referred to one additional group-level intervention which reported increases in active coping, fewer sexual partners and less unprotected sex (Rotheram-Borus et al., 2001b). Woltiski et al. (2005) and Metsch et al. (2005) also reported on a group-level intervention involving HIV positive drug users, which consisted of a harm-reduction intervention including methadone maintenance, counselling and case management, as well as manual-guided psychotherapy sessions. Results showed that those involved in the intervention group obtained lower addiction scores, and were less likely to engage in high-risk sexual behaviour and drug-use behaviours (Margolin et al., 2003).
- The review of Elkavich et al. (2005) included one study (Butler et al., 2003) aimed at young HIV positive haemophiliacs which showed significant pre- to post-test increases in safer sexual practices following the intervention.
- Kelly et al. (2002) described a relevant study by Kalichman (2000), which used a cognitivebehavioural intervention, including risk-reduction, aimed at PWHIV to reduce HIV transmission. Findings showed that the intervention group displayed significantly lower rates of unprotected

intercourse and greater condom use at follow up. Kelly et al. (2002) also referred to one study (Padian et al., 1993) involving serodiscordant couples and interventions such as risk-reduction education and condom use modelling. The intervention couples reported increased condom use from 49% to 88% at 16 month follow up.

New evidence statement: There is tentative review-level evidence to support the effectiveness of small-group level interventions in influencing the sexual risk behaviours of people living with HIV.

5.3.2.3 Community-level intervention

- Fogarty et al. (2001; reported in Albarracin, D., et al., 2005) (categorised as individual/group level in Metsch, 2005). This study involved peer-based condom and contraception promotion aimed at HIV positive African-American women. Compared to the comparison group, the women in the intervention group reported improved consistency in condom use, perceived condoms as more advantageous, and increased levels of self-efficacy for condom use.
- Two Category 3 level reviews (Metsch, 2005; Woltiski, 2005) reported on a skills-based behaviour change intervention conducted by Kalichman and colleagues (2001). This intervention recruited HIV positive men and women from AIDS services and clinics. Lower rates of unprotected sexual intercourse and greater condom use were reported in the group that received the cognitivebehavioural intervention.

Although additional studies have been identified since the original Evidence Briefing there is not enough evidence to support the effectiveness of community-level interventions for people living with HIV.

New evidence statement: There is insufficient review-level evidence to support the effectiveness of community-level interventions in influencing the sexual risk behaviours of people living with HIV.

5.3.2.4 Socio-political intervention

None of the reviews presented data relevant to the effectiveness of socio-political interventions for people living with HIV.

Evidence statement: No change from Ellis et al. (2003). There is no review-level evidence to either to support or discount the effectiveness of socio-political interventions in influencing the sexual risk behaviours of people living with HIV.

5.3.2.5 Inequalities

None of the reviews presented data relevant to the effectiveness of interventions in dealing with inequalities for people living with HIV.

Evidence statement: No change from Ellis et al. (2003). There is no review-level evidence to either to support or discount the effectiveness of socio-political interventions in influencing the sexual risk behaviours of people living with HIV.

5.4 African communities in the UK

As discussed in Section 2.3 (Context) there are concerns about the risk of HIV transmission among certain African communities in the UK.

5.4.1 The reviews

A total of 14 reviews met the criteria for Category 1-3 and included studies that were of some relevance to African communities in the UK.

Category 1: Elwy et al., 2002; Pedlow and Carey, 2003

Category 2: Albarracin et al., 2005; Johnson B.T. et al., 2003; Logan, Cole and Leukefeld, 2003; Merzel and D'Afflitti, 2003; Mullen et al., 2002; Neumann et al., 2002; Robin et al., 2004; Semaan et al., 2002a; van Empelen et al., 2003; Wilson and Miller, 2003

Category 3: Johnson-Masotti et al., 2003; Kirby, 2002; Fortenberry, 2002.

Of the 14 reviews, one (Wilson and Miller, 2003) specifically examined "culturally grounded" group interventions in mixed populations. Four (Johnson B.T. et al., 2003; Mullen et al., 2002; Pedlow and Carey, 2003; Robin et al., 2004) focused on adolescents, three (Elwy et al., 2002; Neumann et al., 2002; Logan et al., 2003) were of heterosexual adults, one (Johnson-Masotti et al., 2003) examined interventions aimed at the mentally ill, one (Fortenberry, 2002) explored clinic-based interventions, and one (Merzel and D'Afflitti, 2003) examined community-based health promotion across a number of health areas including HIV.

Outcomes

The most frequently reported outcome was condom use.

5.4.2 The findings

5.4.2.1 Individual-level intervention

Nine studies were covered by six reviews (Elwy et al., 2002; Johnson B.T. et al., 2003; Kirby, 2002; Logan et al., 2003; Mullen et al., 2002; Pedlow and Carey, 2003).

Pedlow and Carey's (2003) review identified six studies, all conducted in health care settings, which delivered interventions to individual adolescents. Five of these studies (Boekeloo et al., 1999; De Lamater et al., 2000; Gillmore et al., 1997; Mansfield et al., 1993; Orr et al. 1996) included a high proportion of African American or Hispanic adolescents. Two of the five interventions (Boekeloo et al., 1999; Orr et al. 1996) were effective in increasing condom use. Johnson B.T. et al. (2003), Kirby (2002) and Mullen et al. (2002) also covered these studies in their reviews.

- Logan et al. (2003) included three studies (Belcher et al., 1998; Grinstead et al., 1999; Kamb et al., 1998) of individual-level interventions targeted at black and minority ethnic (BME) groups in their review of interventions aimed at women (although the interventions were aimed at both men and women in the included studies). However, very few details were available and the reviewers did not draw any conclusions about the effectiveness of individual-level interventions.
- Elwy et al. (2002) described one primary study of an individual-level intervention in their review of interventions for heterosexual men. Wagstaff et al. (1999) targeted black men with an individual education session, educational video, or standard care and education programme. STI rates were reported to have reduced at six-months follow-ups. The reviewers made no conclusions about interventions delivered at this level.
- Fortenberry (2002) included several clinic-based counselling and condom distribution interventions. These interventions were aimed at a variety of populations including African American and Hispanic people. However, the author made no specific conclusions regarding interventions aimed at BME groups.

Although this update has identified a number of studies that were not covered by the previous Evidence Briefing, there is still no clear evidence on the effectiveness of individual-level interventions for BME groups.

Evidence statement: No change from Ellis et al. (2003). There is insufficient review-level evidence either to support or discount the effectiveness of any individual-level interventions in influencing the sexual risk behaviours of BME groups.

5.4.2.2 Group-level intervention

Nine reviews (Albarracin et al., 2005; Elwy et al., 2002; Pedlow and Carey, 2003; Logan et al., 2003; Mullen et al., 2002; Neumann et al., 2002; Robin et al., 2004; Wilson and Miller, 2003; Kirby, 2002) covered 53 primary studies, which examined a variety of interventions across a number of setting and populations of BME groups. The majority of the studies reported on interventions delivered to adolescents.

Adolescents

• Pedlow and Carey (2003) reviewed 12 studies, which evaluated group-level interventions and included a large proportion of African American, or multi-ethnic youth. Overall, they concluded that many HIV risk reduction interventions for adolescents have been shown to be effective, most frequently on condom use. They did not comment on their findings in relation to BME groups.

- Robin et al. (2004) also reviewed interventions for adolescents, including eight studies of grouplevel interventions (there was some overlap with Pedlow and Carey, 2003). As part of their conclusions they commented, "the majority of studies showing positive effects targeted African American youth" (p.18). Three studies with positive effects included in their review only recruited African American participants.
- Mullen et al. (2002) examined interventions for sexually experienced adolescents, including eight studies with a large proportion of BME groups. Overall, they found that "more protective outcomes" were associated with the interventions that were exclusively targeted at a single ethnic group (African American or Hispanic in this case).
- Based on meta-analysis of 44 studies, Johnson B.T. et al. (2003) found that there was no significant relationship between condom use and the racial composition of the sample.
- Kirby (2002) reviewed seven studies, all of which where covered by the other reviews. No conclusions were made regarding ethnicity in this review.

Overall, a number of reviews were identified which focused on HIV prevention in adolescents, however none specifically examined or drew conclusions about programmes targeting BME groups.

New evidence statement: There is insufficient review-level evidence to support or discount the effectiveness of group-level interventions in influencing the sexual risk behaviour of BME adolescents.

Adults

The previous Evidence Briefing (Ellis et al., 2003) found evidence from six RCTs and one other study supporting the effectiveness of small-group interventions for BME women. This update identified a number of further studies across four reviews (Elwy et al., 2002; Logan et al., 2003; Neumann et al., 2002; Wilson and Miller, 2003), which focused on group-level interventions in heterosexual adult populations.

Elwy et al. (2002) reviewed studies of interventions in heterosexual men. They identified a
number of studies that recruited from BME groups including African American and Hispanic
populations. However, they did not draw any conclusions specifically about interventions in
minority populations or interventions at the group-level. Overall, they identified no single
consistently effective approach but noted that interventions in more general populations of men
had all shown a positive behavioural intervention effect.

- Logan et al. (2003) reviewed the effectiveness of interventions with high-risk heterosexual adults. Seventeen of the studies identified included samples of African Americans of 55% or more, and three studies included Hispanics as the majority of the sample. Overall, meta-analysis of 30 articles found a small but significant positive effect of the interventions on condom use.
- Neumann et al. (2002) also reviewed interventions for heterosexual adults but included US-based studies only. All but one of the studies they identified had also been include by Logan and colleagues (2003). They included 10 studies in a meta-analysis (all of which recruited mixed populations and ethnicities) resulting in an OR of 0.81 (95% CI: 0.69-0.95), indicating that HIV risk-reduction interventions were associated with "significant and modest" effects on unsafe sexual behaviours.
- Johnson-Masotti et al. (2003) (cost-effectiveness review) reviewed interventions for people with severe and persistent mental illness. They included two studies (Otto-Salaj et al., 2001; Susser et al., 1998) where more than 50% of the sample comprised African Americans. Both studies examined small-group cognitive-behavioural skills training. Otto-Salaj et al. (2001) included both men and women, with women showing greater changes across a range of sexual risk behaviours compared to the men who showed a significant change in HIV risk knowledge only. Susser et al. (1998) recruited homeless men and found a decrease in risky sexual behaviours in the men receiving a skills training intervention compared to those receiving an AIDS education programme.

Two reviews (Semaan et al., 2002; van Empelen et al., 2003) focused on HIV prevention interventions for drug users. However, all of the studies included were covered by the other reviews identified for this update and the reviews included in the previous Evidence Briefing.

Overall there were a number of studies that have examined interventions for heterosexual adult populations, including samples from BME groups. However, no review was identified which specifically focused on BME groups and, therefore, it is difficult to draw conclusions about the effectiveness of group-level interventions for BME heterosexual adults.

New evidence statement: There is tentative review-level evidence to conclude that group-level interventions may have a modest effect on influencing the sexual risk behaviour of BME heterosexual adults.

5.4.2.3 Community-level intervention

In addition, to the five studies of community-level interventions included in the previous Evidence Briefing, five studies included across four reviews (Elwy et al., 2002; Albarracin et al., 2005; Neumann et al., 2002; Mullen et al., 2002) were identified for this update.

- Elwy et al. (2002) included two RCTs (O'Donnell et al. 1995; Solomon and DeJong, 1989), which examined the effectiveness of video-based education delivered to BME groups attending an STI clinic (O'Donnell et al., 1995 specifically targeted African American and Hispanic populations). Intentions to use condoms increased in the intervention groups in both studies.
- Mullen et al. (2002) included a single community-level study (Sellers et al., 1994) that examined condom distribution to Latino adolescents. Participants receiving the intervention reported a non-significant reduction in the number of sexual partners.
- Merzel and D'Afflitti (2003) examined community-based prevention programmes including 11 HIV prevention programmes, three of which were targeted at BME groups (Galbraith et al., 1996; Lauby et al., 2000; Sikkema et al., 2000). Sikkema et al. (2000) examined workshops conducted by opinion leaders with women in a housing development project and found significant increases in condom use, but no effect on the frequency of unprotected intercourse. Lauby et al. (2000) examined education by peer outreach workers in the community, peer-led workshops and the use of role model stories as part of the Women and Infants Demonstration Projects conducted across four communities. They found significant intervention effects for condom use with main partners, but no impact on use with 'other' partners. Galbraith et al., (1996) examined the Focus on Kids project, which included weekly small-group educational sessions focusing on decision-making skills in an urban African American community. There were higher rates of condom use and intention to use condoms in the intervention group compared with adolescents in the study's control arm.
- Albarracin et al. (2005) and Neumann et al. (2002) also included studies of community-level interventions (Deren et al., 1995; Lauby et al., 2000). However, few details of these studies were available or they were covered by the other reviews.

The studies identified for this update include adolescents and male heterosexuals and so build on the conclusions drawn in the previous Evidence Briefing about community-level interventions for BME women. However, in the reviews that did assess the quality of the identified studies, the quality was judged to be low and so the conclusions draw are tentative.

Changed evidence statement: There is tentative review-level evidence to conclude that community-level interventions may have an effect on the sexual risk behaviour of BME groups including women, adolescents and heterosexual males.

5.4.2.4 Socio-political intervention

None of the reviews identified for the update included studies of socio-political interventions for BME groups.

Evidence statement: No change from Ellis et al. (2003). There is no review-level evidence either to support or discount the effectiveness of any socio-political level interventions in influencing the sexual risk behaviours of BME groups.

5.4.2.5 Overall conclusions of the reviews regarding BME groups

None of the identified reviews drew any specific conclusions about interventions delivered to BME groups. The majority of the reviews focused on other populations (for example, adolescents or heterosexual males) that included samples of BME groups.

5.4.2.6 Culturally grounded interventions

Wilson and Miller (2003) reviewed 17 studies of interventions that explicitly sought to address cultural concepts. They identified two primary strategies that were used to integrate the concept of culture into prevention research: (1) attending to intervention presentation which primarily focused on the visual and audible characteristics, and (2) attending to intervention content which involved embedding cultural concepts into the design of intervention activities and messages. Some intervention design used a combination of these two strategies.

Six of the studies reviewed aimed to assess whether culturally grounded strategies enhanced intervention effectiveness. Matching facilitators' race and ethnicity to those of the participants appeared to have little extra effect on reducing HIV risk behaviours, but one study found that a culturally specific intervention was more useful in increasing knowledge about AIDS for a subgroup of participants. Four studies examined the difference between the use of a combination of the two strategies and the use of culturally dissimilar materials. Two studies examined culturally specific videos, both found that participants responded better to the culturally specific video compared to culturally unspecific material. Wilson and Miller (2003) reported that more studies were needed that examined the impacts of culture on the effectiveness of HIV prevention programmes.

Wilson and Miller (2003) concluded "in sum, these studies remind us that it is perhaps most important to focus on understanding and integrating cultural concepts into the intervention content, rather than rely exclusively on attending to presentation strategies. These studies also highlight the importance of

examining assumptions regarding what are salient cultural referents to any one group before designing an intervention program" (p.191).

New evidence statement: There is tentative review-level evidence to conclude that culturally grounded interventions have an effect on the sexual risk behaviour of BME groups.

5.4.2.7 Inequalities

None of the reviews presented any data about the effectiveness of interventions addressing inequalities in sexual health.

Evidence statement: No change from Ellis et al. (2003). There is no review-level evidence either to support or discount the effectiveness of interventions in addressing inequalities in sexual health for BME groups.

5.4.2.8 Cost-effectiveness

Three reviews were identified which presented data on the cost-effectiveness of interventions for BME groups. These findings are presented in section 6.1.4. The original Evidence briefing found no review-level evidence regarding cost-effectiveness for BME groups.

Changed evidence statement: There is insufficient review-level evidence to support or discount that HIV prevention interventions aimed at BME groups are cost-effective.

5.5 Voluntary counselling and testing (VCT)

VCT includes a range of interventions, described in Section 2.4 (Conceptual Framework).

5.5.1 The reviews

Eight review papers were identified that met the appraisal criteria including two Category 1 reviews (Elwy et al., 2002; Vidanapathirana et al., 2005), which are of some relevance to HIV voluntary counselling and testing:

Category 1: Elwy et al., 2002; Vidanapathirana et al., 2005
Category 2: Albarracin et al., 2005; Neumann et al., 2002; Semaan et al., 2002; Van Empelen et al., 2003; Weinhardt, 2005.
Category 3: Kelly and Kalichman, 2002

Only one of the included reviews (Weinhardt, 2005) specifically examined the impact of HIV VCT on sexual health behaviour. This review was an update of Weinhardt et al. (1999), which was included in the previous Evidence Briefing. The other reviews included primary studies relevant to VCT across a number of different populations. Vidanapathirana et al. (2005) examined mass media interventions to promote the uptake of VCT.

Outcomes

The reviews reported here do not include any findings on health promotion outcomes/intervention impact measures (e.g. knowledge, skills), only intermediate health outcomes (behavioural change).

5.5.2 The findings

5.5.2.1 Men who have sex with men (MSM)

Two reviews (Vidanapathirana et al., 2005, Weinhardt et al., 2005) included studies of VCT with MSM.

- Vidanapathirana et al. (2005) included a study by McOwan (2002) that examined the effects of a multimedia intervention (including posters, tabloid newspapers and leaflets) targeted at gay and bisexual men. The intervention had a positive effect on the uptake of HIV testing.
- Weinhardt et al. (2005) covered one additional study (Colfax et al., 2002) to those included in the previous Evidence Briefing. The study examined sexual risk behaviour in HIV-positive individuals during early seroconversion. The sample showed reductions in unprotected insertive and receptive anal intercourse with HIV-negative and unknown individuals following receipt of a HIVpositive test.

The previous Evidence Briefing found that there was conflicting evidence on the effectiveness of HIV VCT in influencing the sexual risk behaviours of MSM. Although this update has identified additional positive evidence in this area, it is insufficient to affect the conclusions drawn by Ellis et al. (2003).

Evidence statement: No change from Ellis et al. (2003). There is insufficient review-level evidence to either support or discount the effectiveness of HIV VCT in influencing the sexual risk behaviours of MSM, regardless of whether they test sero-positive or sero-negative.

New evidence statement: There is tentative review-level evidence to support the effectiveness of mass media interventions in influencing the uptake of HIV VCT in groups of MSM.

5.5.2.2 Injecting drug users (IDUs)

Three reviews (Semaan et al., 2002; van Empelen et al., 2003; Weinhardt, 2005) included a total of five additional studies that were relevant to VCT with injecting drug users. None of the reviews examined VCT interventions as a group so details of the individual studies are reported here.

- Deren et al. (1998) compared sexual risk behaviours in HIV-positive and HIV-negative injectors and crack users. There was a reduction in unprotected sex acts across all groups but a greater reduction was seen in seropositive individuals.
- Gibson et al. (1999) examined two interventions: (1) heroin users were randomised to either HIV counselling and testing with 50 minutes post test counselling, problem solving and condom use demonstration, or to receive an educational brochure; and (2) injecting drug users were randomised to receive HIV counselling and testing with 50 minutes post test counselling or the HIV counselling and testing (CT) plus problem solving and condom use demonstration. No differences were seen between any of the groups at 6- and 12-month follow-up, and as noted by Van Empelen et al. "no significant reductions in condom use were reported for the overall sample" (p.1600).
- Three of the studies (Cottler et al., 1998; Kotranski et al. 1998; McCoy et al., 1998) included a standard HIV CT programme compared to an enhanced programme. Kotranski et al. (1998) randomly assigned participants to the standard programme or an enhanced intervention aimed specifically at STI reduction, providing STI information, personalising STI risk, promoting condom use, and discussing barriers and strategies; Cottler et al. (1998) randomly assigned drug users to a standard two-session HIV CT programme or the standard programme plus an additional four two-hour sessions of peer-delivered intervention focused on AIDS, stress management, drug awareness and reduction of high-risk sexual behaviour; and McCoy and colleagues (1998) randomly assigned female African American drug users to groups receiving a standard HIV CT

intervention or an enhanced intervention, which combined the HIV CT with additional small-group sessions. Overall, there was no difference found between the standard and enhanced groups in terms of sexual behaviour, with the exception of the study by McCoy and colleagues (1998), although van Empelen et al. (2003) described the difference as only "marginally significant".

The previous Evidence Briefing identified five reviews that included 15 studies involving HIV VCT with IDUs. These reviews presented some evidence that HIV VCT can be effective in influencing the sexual behaviours of some IDUs, in particular those who learn that they are seropositive. Overall, the studies identified for the update do not provide any evidence that enhanced VCT programmes are more effective than standard VCT programmes.

Evidence statement: No change from Ellis et al. (2003). There is tentative review-level evidence to conclude that HIV VCT can be effective in influencing the sexual behaviours of some IDUs, in particular those who learnt that they were seropositive.

5.5.2.3 Women and heterosexual couples

Two reviews (Albarracin et al., 2005; Weinhardt, 2005) included a total of two studies that were not covered by the reviews included in the previous HIV evidence briefing. No details were available regarding the study (Flaskerud et al., 1997) included in Albarracin et al. (2005). The study included in the review by Weinhardt (2005), (Roth et al., 2001) was based on a sample from studies in Rwandan women included previously (Allen et al., 1992a; Allen et al., 1992b; Allen, 1993). The study found increased rates of condom use in serodiscordant couples and the effect was most pronounced in couples where the man was positive and had not been tested before. This finding is consistent with the review-level evidence presented in the previous Evidence Briefing.

Evidence statement: No change from Ellis et al. (2003). There is sufficient review-level evidence to conclude that HIV VCT can be effective in influencing the sexual risk behaviours of couples who learn that they are sero-discordant for HIV.

Vidanapathirana et al. (2005) examined four studies of mass media interventions targeted at women to promote the uptake of testing. The following interventions were examined across the four studies: two types of educational programmes (participatory education and didactic education with skill training) (Amero et al., 2001); two "gain-framed" (i.e. adopting safer sexual behaviours would result in healthier outcomes) and two "loss-framed" (i.e. continuing to take risks would result in a reduction of sexual health) videos (Apanovitch et al., 2003); and leaflets including HIV information (Simpson et al., 1998; 1999). All of the studies reported that the interventions were effective in promoting the uptake of HIV testing.

New evidence statement: There is sufficient review-level evidence to conclude that mass media interventions can be effective in influencing the uptake of HIV VCT in women.

5.5.2.4 Heterosexual men

One review (Elwy et al., 2002), which specifically focused on HIV prevention in heterosexual men, included two studies that examined the effectiveness of VCT and were not included in the previous Evidence Briefing. One of the studies was concerned with STI clinic attendees and is discussed in section 5.5.2.6. The other study (Jackson et al., 1997) comprised an on-site clinic providing counselling and HIV testing, and individual skills sessions in condom negotiation, demonstration and promotion, and HIV/STI risk reduction for transport workers in Mombasa, Kenya. The authors reported a significant intervention effect on lowering STI incidence and decreasing the number of sex partners of these men. There was no change in condom use.

In the previous Evidence Briefing, Ellis et al. (2003) identified review-level evidence to support the conclusion that HIV VCT in addition to other components can be effective in influencing sexual risk behaviours in heterosexual men who learn that they are seropositive. There is insufficient review-level evidence available from this update to draw further conclusions on the effectiveness of VCT for heterosexual men.

Evidence statement: No change from Ellis et al. (2003). There is tentative review-level evidence to conclude that HIV VCT plus another component (for example, offering referrals to other services) can be effective in influencing sexual risk behaviours in heterosexual men who learn that they are seropositive.

5.5.2.5 Commercial sex workers

There were no additional studies identified in the included reviews that were relevant to HIV VCT in populations of commercial sex workers. Three studies on CSW were included in the previous Evidence Briefing, but there was insufficient review-level evidence to draw any conclusions.

 Vidanapathirana et al. (2005) included a study, which although it targeted the general public, assessed the effects of the intervention (HIV testing uptake) on CSWs. The intervention was a nationwide multimedia AIDS educational campaign. The study used an interrupted time series design and found a positive change in HIV testing initially but this declined over time to less than the initial value. Evidence statement: No change from Ellis et al. (2003). There is insufficient review-level evidence either to support or discount the effectiveness of HIV VCT in influencing the HIV sexual risk behaviours of CSWs.

New evidence statement: There is insufficient review-level evidence either to support or discount the effectiveness of mass media interventions on the uptake of HIV VCT among CSWs.

5.5.2.6 STI clinic attendees

Two additional studies on STI clinic attendees were covered by three reviews (Albarracin et al., 2005; Elwy et al., 2002; Weinhardt, 2005). One study (Bentley et al., 1998) was included in two reviews and examined counselling (individual sessions providing education and condom use skills) and HIV testing for heterosexual men attending STI clinics in India. Condom use increased, however, there was no change in the number of sex partners reported.

 Weinhardt (2005) included a study (George et al., 1998) that examined rates of STI infection following a positive or negative HIV test at an STI clinic in London. Rates of STI infections were found to have decreased in both HIV-positive and HIV-negative patients. However, this outcome was not consistent with the overall finding by Weinhardt (2005) that the incidence of STIs is increased among HIV-negative participants after VCT.

Ellis et al. (2003) found evidence from a small number of studies that HIV VCT has a beneficial effect for people diagnosed seropositive and that combined interventions may be more effective. The studies identified in this update are consistent with this finding.

Evidence statement: No change from Ellis et al. (2003). There is tentative review-level evidence to conclude that HIV VCT can be effective in influencing HIV sexual risk behaviours of STI clinic attendees who learn that they are seropositive.

5.5.2.7 Mixed populations

One review (Weinhardt, 2005) drew conclusions about the effectiveness of VCT across a variety of settings and populations. As previously stated, this review was an update of the review included in the previous Evidence Briefing (Weinhardt et al., 1999).

 Weinhardt (2005) found that an HIV diagnosis, whether provided to an individual patient or to a couple, significantly reduced the occurrence of unprotected intercourse and reduced the number of sexual partners of individual HIV-positive patients. Significant increases in condom use were found in couples with at least one partner testing positive for HIV, but not in HIV-positive individuals relative to untested individuals.

Based on the review-level evidence presented in the previous Evidence Briefing there is a growing body of evidence to support the conclusion that HIV VCT can be effective in influencing the sexual risk behaviours of people who learn that they are seropositive.

Evidence statement: No change from Ellis et al. (2003). There is tentative review-level evidence to conclude that HIV VCT can be effective in influencing the sexual risk behaviours of people who learn that they are seropositive.

Vidanapathirana et al. (2005) identified five studies (Beck et al., 1987; Beck et al., 1990; Dwyer et al., 1988; Joshi et al., 1988; Maayan et al., 1989; Tesoriero and Sorin, 1992; Turner and Mutton, 1987) of mass media interventions which aimed to increase the uptake of HIV testing in the general population. All of the studies had an initial positive impact on the uptake of HIV testing. However, in four studies (Beck et al., 1990; Dwyer et al., 1988; Maayan et al., 1989; Tesoriero and Sorin, 1992) the effects of the intervention declined over time and in two studies fell to a value less than recorded prior to the intervention.

New evidence statement: There is sufficient review-level evidence to conclude that mass media interventions can have an initial positive effect on influencing the uptake of HIV VCT in the general public.

5.5.2.8 Overall conclusions of the reviews regarding VCT

Weinhardt (2005) concluded that the results of his review provided "a positive outlook in terms of reducing transmission risk behaviour following an HIV diagnosis" (p.50). However he also commented, "the question of the effect of HIV diagnosis on sexual risk behaviour is not one that can be answered once and laid to rest... [The] effect of the diagnosis is a moving target that depends on the context in which the diagnosis takes place" (p.59).

In their review of the evidence, Kelly and Kalichman (2002) concluded, "voluntary HIV counseling and testing appears partially successful as an HIV prevention strategy in western countries. Infected individuals and those in serodiscordant relationships reduce their risk behavior following testing, although meta-analysis shows limited behavior-change effects produced by counseling and testing among those who learn they are uninfected" (p.628).

Vidanapathirana et al. (2005) concluded "mass media interventions can have immediate and overall effects in promotion of HIV testing" (p.1). However they also noted that no long-term effects were shown and that the intervention did not have a significant impact on detecting seropositive status.

For some of the priority groups, for example CSWs, it is still not possible to draw conclusions about the effectiveness of HIV VCT because of a lack of review-level evidence. On the whole, very few further studies of HIV VCT were identified for this update that had not been covered in the previous Evidence Briefing.

5.5.2.9 Inequalities

None of the reviews identified presented any data about the effectiveness of HIV VCT in addressing inequalities in sexual health.

Evidence statement: No change from Ellis et al. (2003). There is no review-level evidence to support or discount the effectiveness of HIV VCT in addressing inequalities in sexual health

5.5.2.10 Cost-effectiveness

One review (Farnham et al., 2002) presented data on the cost-effectiveness of HIV counselling and testing. The results of this review are present in section 6.1.7.

New Evidence statement: There is insufficient review-level evidence to support or discount that HIV counselling and testing interventions are cost-effective.

6 Cost-Effectiveness

6.1.1 The reviews

Five reviews of the cost-effectiveness of HIV prevention interventions were identified that met the appraisal criteria:

Category 1: No evidence

Category 2: Farnham et al., 2002; Pinkerton, et al., 2002a Category 3: Johnson-Masotti, et al., 2002; Pinkerton, et al., 2002b; Pinkerton and Holtgrave, 2002

The reviews reported on a number of different populations. Three reviews (Pinkerton et al., 2002a; Pinkerton et al., 2002b; Pinkerton and Holtgrave, 2002) included studies relevant to MSM and BME populations. Farnham et al. (2002) reported exclusively on HIV counselling and testing interventions.

6.1.2 The findings

In the following sections, the evidence from cost-effectiveness reviews is considered according to the priority groups identified: MSM, African, CSW, PWHIV and VCT.

6.1.3 Men who have sex with men (MSM)

Category 1: No evidence

Category 2: Pinkerton et al., 2002a.

Category 3: Pinkerton et al., 2002b; Pinkerton and Holtgrave, 2002.

Of the 16 studies included in the review by Pinkerton et al. (2002a), five studies examined the costeffectiveness of prevention interventions for MSM. The interventions reviewed included a small-group, cognitive behavioural intervention, a safer sex skills-building session added to an educational session, "peer leader" community-level norm change focus, and an intervention comprising risk reduction counselling, peer education and HIV testing. All of the studies used cost-utility analysis and found that the interventions were cost-saving to society.

The review by Pinkerton et al. (2002b) included one additional study, which examined the costeffectiveness of community mobilisation through the Mpowerment Project. The cost per HIV infection averted was estimated to be between \$9,500 and \$18,300, far less than the lifetime medical costs of treating HIV disease.

The review by Pinkerton and Holtgrave (2002) did not include any additional cost-effectiveness studies. See Table 11, Appendix F for details of the primary studies included.

New evidence statement: There is tentative review-level evidence to conclude that HIV prevention interventions aimed at MSM are cost-effective, and can be cost-saving.

6.1.4 African communities in the UK

Category 1: No evidence

Category 2: Pinkerton et al., 2002a.

Category 3: Pinkerton et al., 2002b; Pinkerton and Holtgrave, 2002.

The review by Pinkerton et al. (2002a) included two studies that examined the cost-effectiveness of HIV prevention interventions, both of which focused on African American populations. The interventions examined were an intensive one-day workshop focusing on knowledge, attitudes and behaviours for adolescent males and free condoms distributed at community locations. The workshop intervention was found to be cost-effective (compared to treatment of HIV disease) when only those youth who reported sexual activity in the previous three months were included in the analysis. Distribution of free condoms was found to be cost-saving compared to the lifetime cost of HIV-related medical care.

The reviews by Pinkerton et al. (2002b) and Pinkerton and Holtgrave (2002) did not include any additional cost-effectiveness studies for this population. See Table 11, Appendix F for details of the primary studies included.

New evidence statement: There is insufficient review-level evidence to support or discount that HIV prevention interventions aimed at BME groups are cost-effective.

6.1.5 Commercial sex workers (CSW)

Category 1: no evidence Category 2: no evidence Category 3: Pinkerton and Holtgrave (2002)

The review by Pinkerton and Holtgrave (2002) made reference to a study that assessed the costeffectiveness of an intervention focused on increasing condom use among CSWs in Nairobi, Kenya. The programme provided free condoms and offered health education counselling. The annual cost per infection averted was estimated to be between \$8 and \$12, which was judged to be less costly than HIV/AIDS medical care (estimated to be between \$100 and \$1600 in Zaire and Tanzania). No reviews were identified that covered cost-effectiveness studies that were relevant to UK populations of CSW. New evidence statement: There is no review-level evidence to support or discount that HIV prevention interventions aimed at CSW are cost-effective.

6.1.6 People living with HIV (PWHIV)

Category 1: No evidence Category 2: No evidence Category 3: No evidence

No reviews were identified that included cost-effectiveness studies that are relevant to UK populations of PWHIV.

New evidence statement: There is no review-level evidence to support or discount that HIV prevention interventions aimed at PWHIV are cost-effective.

6.1.7 HIV voluntary counselling and testing (VCT)

Category 1: No evidence Category 2: Farnham et al., 2002. Category 3: No evidence

Farnham et al. (2002) reviewed 12 studies that examined the cost-effectiveness of HIV counselling and testing (CT) programmes and their effect on sexual behaviour change. See Table 11, Appendix F for details of the primary studies included.

Based on their review, the authors concluded that HIV CT programmes were generally cost-effective, particularly if targeted to groups of infected or high-risk individuals. However, they also noted that there is continuing debate about the effectiveness of HIV CT in changing sexual behaviour and the cost-effectiveness results should be viewed with caution.

New evidence statement: There is insufficient review-level evidence to support or discount that HIV counselling and testing interventions are cost-effective.

7 Theory-based interventions

7.1.1 The reviews

Fifteen reviews (see below) referred to theory prevention interventions however, only 13 (Herbst et al., 2005; Johnson, 2002; Johnson, 2005; Elwy et al., 2002; Robin et al., 2004; Johnson, 2003; Van Empelen et al., 2003; Elkavich et al., 2005; Kelly et al., 2002; Ross et al., 2002; Woltiski et al., 2005; Pedlow & Carey, 2003) provided full information regarding the theoretical basis for each intervention included in the review. The original Evidence Briefing (Ellis et al., 2003) found that none of their Core reviews considered the role of theory in intervention effectiveness. As such all of the evidence statements in this section are new.

Category 1: Elwy et al., 2002; Herbst et al., 2005; Johnson, 2002; Johnson, 2005; Johnson, 2003. **Category 2:** Robin et al., 2004; Van Empelen et al., 2003

Category 3: Elkavich et al., 2005; Kelly et al., 2002; Ross and Williams, 2002; Woltiski et al., 2005; Pedlow and Carey, 2003.

Findings

Summaries of the theories utilised in the studies can be found in Appendix A. Table 12 contains details of the primary studies and theoretical approaches used.

7.1.1.1 Men who have sex with men (MSM)

Twenty-four studies, in four Category 1 and 2 reviews (Herbst et al., 2005; Johnson, 2002; Johnson, 2005; Johnson, 2003), examined theoretically based interventions for MSM. Diffusion of innovations and relapse-prevention were the most frequently used single theoretical frameworks for prevention interventions targeting MSM, however, studies were most likely to use more than one theory to underpin their intervention (see Table 12, in Appendix F).

Johnson et al. (2002) stated that "interventions that include interpersonal skills training produced the most clearly favourable effects on unprotected sex" (S112). The effects of interpersonal skills training were also raised in Johnson et al. (2003). Findings in Herbst et al. (2005) (22 studies), showed positive and significant reductions in UAI in those interventions reporting any theoretical base. They also reported greater significant reductions in UAI risk in the intervention groups of studies based on the diffusion of innovations theory (OR = 0.62; 95% CI: 0.47-0.83) and those based upon the relapse prevention model (OR = 0.60; 95% CI: 0.47-0.78). However, findings showed no significant difference in control and comparison groups for those interventions underpinned by social learning theory (OR = 0.60; 95% CI: 0.33-1.09) or other behavioural theories (OR = 0.73; 95% CI: 0.45-1.20).

New evidence statement: There is tentative review-level evidence to conclude that interventions based upon theoretical models, and in particular diffusion of innovations and the model of relapse prevention, are effective in influencing the sexual risk behaviours of MSM.

7.1.1.2 People with HIV (PWHIV)

Herbst et al. (2005) Johnson et al. (2005) and Woltiski et al., (2005) all refer to one study (Kelly et al., 1993) with a theoretical framework that included men living with HIV. The authors did not make any specific conclusions regarding theoretically based intervention and PWHIV.

Elkavich et al. (2005), also refer to four additional studies relevant to PWHIV. Three studies by Rotheram-Borus et al. (2001a; 2001b in press)*, involved the use of social action model intervention aimed at young people. All three studies yielded significant and positive results. Elkavich et al. (2005) also reported the results of a study based upon the use of multiple theories (Butler et al., 2003), aimed at young HIV positive haemophiliacs which showed significant pre- to post-test increases in safer sexual practices.

Kelly et al. (2002) describes one relevant study by Kalichman (2000), which used a cognitivebehavioural intervention, including risk-reduction, aimed at PWHIV to reduce HIV transmission. Findings showed that the five session health maintenance intervention support displayed significantly lower rates of unprotected intercourse and greater condom use at follow up.

Woltiski et al. (2005) describes one relevant study by Kalichman (2001), which used a social-cognitive intervention, including motivational enhancement. After six months, findings showed that those who received the intervention significantly reduced unprotected sex and increased condom use with negative or unknown sero-status partners.

New evidence statement: There is tentative review-level evidence to conclude that theoretically based interventions can be effective in influencing the sexual risk behaviours of PWHIV.

7.1.1.3 African communities in the UK

Two reviews (Elwy et al., 2002; Robin et al., 2004) referred to 17 theory based interventions. A further nine relevant studies were referred to by Wilson and Miller (2003). However these authors failed to clearly specify the theory that underpinned each study.

Elwy et al. (2002) reported that overall, studies reported mixed results. Kalichman (1999) used motivational interviewing with inner-city black men and reported significant positive changes in sexual

^{*} Rotheram-Borus et al., (2001a) and (2001b) is also referred to in Woltiski et al., (2005).

communication (including discussions about AIDS), unprotected sex, and sexual negotiation. Kalichman (1997) used cognitive-behavioural skills in another intervention targeting black men. However, this study found no difference between the control and comparison groups. O'Donnell (1995) used theory of reasoned action and social learning theory in a video based intervention which resulted in those exposed to the video redeeming more condom vouchers than those not exposed. Stanton (1998) examined an intervention with Namibian adolescents based upon risk reduction. There was a significant increase in the intervention group for condom use, discussion of sexual history, risk-behaviour with sexual partner and higher self-efficacy for finding condoms when needed. St Lawrence (1995) reported that behavioural skills training for black adolescents resulted in reduced frequency of unprotected sex.

Robin et al. (2004) reported on 11 studies; five used multiple theories and produced positive results. Three studies produced null results and three produced negative results.

Robin et al. (2004) stated that, "many interventions were based on a combination of theories. Socialcognitive theories, the Health Belief Model, social learning theories and social influence theories were applied most often. No clear patterns emerged in study methods, year of publication, intervention content or duration with respect to the theory guiding interventions" (p.17).

New evidence statement: There is insufficient review-level evidence to either support or discount the conclusion that theoretically based interventions can be effective in influencing the sexual risk behaviours of BME groups.

7.1.1.4 Drug users

Three reviews (Elwy et al., 2002; Van Empelen et al., 2003; Woltiski et al., 2005) included 18 theory driven intervention studies aimed at drug users.

Elwy et al. (2002) reported on two RCTs. The first (Baker et al., 1994) was carried out in Australia and involved motivational interviewing with non-treatment seeking injecting drug users (IDU). This study had no effect on changing risk behaviour at either three or six month follow-up. The second study (Malow et al., 1994) incorporated culturally sensitive HIV risk-reduction interventions. The findings showed that rates of multiple partners fell and higher self-efficacy, sexual communication skills and condom use skills were reported in the intervention group. Mixed findings were reported from studies investigating both theory based and non-theory based interventions.

Van Empelen et al. (2003) reported on 15 theory driven interventions aimed at reducing the sexual risk of drug users. Of these studies, only five reported changes in behaviour (such as condom use and number of sexual partners), or condom use skills. However, in general, most studies reported a reduction in unsafe sexual behaviours for both intervention and comparison groups. Theories most

often used in these interventions were the Health Belief Model, theory of reasoned action, socialcognitive theory, or theoretical frameworks such as the transtheoretical model ('stages of change'). The authors concluded that, "the findings show that the programmes that are likely to be successful use multiple theories and methods" (p.1605).

Woltiski et al. (2005) reported the results of one theoretically underpinned intervention (Margolin et al., 2003). HIV positive IDUs were randomised to receive interventions including risk-reduction. Those included in the intervention group were less likely to engage in risky sexual and drug use behaviours at follow-up.

New evidence statement: There is tentative review-level evidence to conclude that multiple theory-based interventions can be effective in influencing the sexual risk behaviours of drug users.

7.1.1.5 Commercial sex workers (CSWs)

Herbst et al. (2005) reported on one theory-based intervention for CSWs (Miller et al., 1998). This intervention used popular opinion leaders to deliver HIV prevention messages to CSWs and potential clients in a community setting in conjunction with the diffusion of innovations theory. However, this intervention yielded a non-significant result.

Ross and Williams (2002) included two successful interventions based upon the diffusion of innovations (Hanenberg et al., 1994; Ngugi et al., 1996). Ross and Williams (2002) stated the importance of using a theoretical as well as a rigorous scientific methodology. They also emphasised the importance of changing social norms through the use of community opinion leaders, role models, and peer educators and intervention specialists.

New evidence statement: There is insufficient review-level evidence to either support or discount the conclusion that theoretically based interventions can be effective in influencing the sexual risk behaviours of CSWs.

7.1.1.6 Young people

Pedlow and Carey (2003) included 22 studies that were targeted at young people and incorporated theoretical frameworks. Almost all of the reviewed studies used more than one theory. The most frequently used theories were social cognitive theory, the theory of reasoned action, Health Belief Model, and the information-motivation-behavioural skills model. The authors stated that interventions guided by social cognitive theory provided the opportunity for skills acquisition. Interventions guided by the information-motivation-behavioural skills model provided behavioural skills components that addressed risk-reduction, behavioural self-management and cognitive strategies for implementing risk-reduction behaviour.

New evidence statement: There is insufficient review-level evidence to either support or discount the conclusion that theoretically based interventions can be effective in influencing the sexual risk behaviours of young people.

7.1.1.7 Other populations

Ross et al. (2002) included two studies with theory-based interventions. Santelli et al. (1995) reported on a social learning theory based community intervention aimed at preventing perinatal HIV transmission by preventing transmission to women. This study showed increased condom use in both the intervention and comparison communities however condom use was greater in the intervention community. The study reported on by the CDC (1999) entailed a social learning-theory based media intervention aimed at the general population that, supporting the diffusion of innovations theory, reached not only the intervention groups but also the wider communities.

New evidence statement: There is insufficient review-level evidence to either support or discount the conclusion that theoretically based interventions can be effective in influencing the sexual risk behaviours of general populations.

8 Discussion

- What works to reduce the sexual risk of HIV transmission among the priority populations in the UK? What works to change the modifying factors that influence sexual risk behaviours for HIV transmission?
- Are theory-based interventions more likely to be effective?
- Are multi-component interventions more likely to be effective?
- What works to reduce inequalities in sexual risk for HIV transmission?
- What interventions are cost-effective?

8.1 What works to reduce the sexual risk of HIV transmission among the priority populations? What works to change the modifying factors that influence sexual risk behaviour for HIV transmission?

The original Evidence Briefing found limited information regarding the effects of HIV prevention intervention approaches on the modifying factors that influence sexual risk behaviour. The literature identified for this update has included modifying factors such as 'health literacy' (e.g. knowledge, attitudes, skills, self-efficacy etc.) as evidence of effectiveness. Findings showed that six reviews reported effects on modifying factors (Elwy, 2002; Albarracin, 2005; Johnson, 2002; Johnson B.T., 2003; van Empelen, 2003; Rees et al, 2004), although one found ineffective results (van Empelen, 2003) and one found unclear results (Rees et al., 2004). The majority of the modifying factors that were evaluated as outcomes were in interventions aimed at MSM, heterosexual males and adolescent populations¹. However, overall a clear pattern of which interventions. It is important to note that reviews included in this update have excluded studies on the basis that they have not reported behavioural outcomes and although others have reported changes in modifying factors (e.g. increased knowledge or skills) in their outcomes they are primarily considered in conjunction with behavioural outcomes, such as condom use, and health outcomes, such as HIV incidence.

Some reviews have included intermediate health outcomes (e.g. behaviour) and health promotion/intervention impact measures and authors have provided differing views on personal modifying factors as outcomes (e.g. knowledge or skills). Rees et al. (2004) stated in their conclusions that modifying factors are not considered to be reliable due to the fact that sound evaluations of these types of interventions are lacking. Others have argued that future research ought to include modifying factors in conjunction with health outcomes (e.g. incidence of infection) in order to be clear on the

¹ They have been included in this evidence briefing update only where they overlap with the target populations under examination.

^{- 81 -} HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission, 2006.

effectiveness of the intervention (Elwy et al., 2002). However, calls to evaluate intervention impact at a psychosocial level remain (van Empelen et al., 2003; Robin et al., 2004).

8.2 The role of theory in intervention effectiveness

Section 7 addressed the studies that have used theoretical frameworks to underpin their interventions. The results show that diffusion of innovations (MSM and CSWs) and relapse prevention theories (MSM) were effective in reducing risky sexual behaviour. However, the majority of effective theory-based interventions tended to use more than one theory. Nevertheless, there was no clear pattern to the effectiveness of the methods used.

The original Evidence Briefing supported the use of theoretically based prevention interventions (Ellis et al., 2003) and this view has been heavily supported throughout the HIV prevention literature. However, weaknesses are clear to see and it was often the case that no rationale was provided for choosing a specific theory and in some cases interventions were based on a theoretical framework but did not clearly report it in the write-up. In addition to this, it has been noted that not only should interventions be based on theory but that they ought to be developed systematically, possibly with the use of Intervention Mapping² which may help with the implementation and evaluation of theory driven interventions (van Empelen et al., 2003).

8.3 Multi-component interventions

The evidence identified for this update found nothing to support or discount the view that multicomponent interventions are more effective at preventing HIV through sexual transmission than other, single component interventions.

Two studies by Johnson et al. (2002, 2003) call for more rigorous evaluations of HIV prevention efforts to ascertain with confidence the effects of specific intervention components. This was also the conclusion of the previous evidence briefing. Neumann et al. (2002) called for future studies to include component analysis, report all outcome data related to a study's hypotheses, avoid aggregate indices to measure sexual risk and report important descriptive variables. We support these views.

8.4 Inequalities

The evidence identified for this update found no interventions that addressed the issue of inequality relevant to the UK populations. Thus there has been no change from the original evidence briefing where a lack of interventions addressing relevant socio-economic inequalities was also found.

² A framework for health education intervention development.

^{- 82 -} HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission, 2006.

Rees et al. (2004) reported the need for future studies concerning interventions with MSM to focus upon the views of all groups of MSM, in particular young MSM, working class MSM, black minority ethnic MSM, disabled MSM, and other vulnerable MSM groups. However, we would maintain that future studies concerning inequalities ought to be considered for all of the UK priority populations.

8.5 Cost-effectiveness

This update includes five studies containing cost-effectiveness evidence. These studies addressed three out of the four priority populations and VCT. However, tentative level evidence was only found for one priority population (MSM) and insufficient level evidence was found for two others (BME, VCT).

From this evidence we can tentatively conclude that small-group, cognitive-behavioural interventions, skills building sessions, peer-education and HIV testing as well as risk-reduction counselling and community peer led interventions can be cost-effective and cost-saving when aimed at MSM. Further studies are needed to add to the existing evidence base for BME and VCT before conclusions can be made regarding the cost-effectiveness of workshops (BME) and VCT. One category 3 review also provided information on the cost-effectiveness of condom distribution schemes and health education counselling for CSWs. However, further Category 1 or 2 evidence are required before conclusions of effectiveness can be made.

Although progress has been made since the original evidence briefing it is clear that further reviews on cost-effectiveness relevant to the UK priority populations are required. Areas have been specifically highlighted in the literature for further study: Vidanapathirana et al. (2005) stated that "Additional research would identify the cost-effectiveness, type of mass media interventions, and characteristics of messages in mass media to promote HIV testing" (p10). Matthews et al. (2002) state that, "the costs and potential harms of partner notification need to be measured and compared" (p299).

These findings show that there are gaps in the evidence for not only the cost-effectiveness of intervention methods but also for the cost-effectiveness of interventions relative to each of the priority populations considered here.

8.6 Methodological issues

One of the main recommendations of the reviews included in this update was to improve the amount, design and methodological rigour of studies assessing interventions (Pedlow and Carey, 2003; Rees et al., 2004; Elwy et al., 2002; Johnson et al., 2003; Johnson et al., 2002; Logan et al., 2002; Neumann, 2002; Robin et al., 2004; Semann et al., 2002; van Empelen et al., 2003; Weinhardt, 2005) and most studies referred to the need for further randomised controlled trials. However, Elwy et al. (2002) recognised the contribution that study designs other than RCTs (e.g. qualitative design) can

make to the evidence base. Other published work has gone one step further and argued that RCTs are an inappropriate method for evaluating health promotion interventions (Kippax, 1998). However, Elwy et al. (2002) remain cautious regarding non-RCTs and states that "the effectiveness of these interventions requires further rigorous evaluation before widespread implementation" (p1829).

The main methodological limitations reported in the reviews concerned small sample sizes. Weinhardt (2005) reported that effect sizes could not be discerned for each intervention due to limited information. In their meta-analysis, Neuman et al. (2002) found that there were too few studies with which to determine the specific characteristics of the interventions that had produced significant protective results. Albarracin et al. (2005) reported inconsistent findings between their fixed and random effects meta-analyses. They recommended that future research should aim to provide a sufficiently large number of effect sizes to estimate the population variance more precisely, thereby reconciling the discrepancies between the fixed and random effect findings. Elwy et al. (2002) hold the view that "studies should have the statistical power to demonstrate effectiveness, and this should be calculated prior to research being funded and implemented" (p1829).

Following on from these points, Rees et al. (2004) emphasises not only the need for studies to be conducted rigorously, but also the need for research to be rigorously reported. Several reviews were limited in their analyses or conclusions because the primary studies assessed did not report all of the available evidence that was required for inclusion in the review or meta-analysis. Logan et al. (2002) called for future research to be more standardised in its reporting of intervention results and Neumann et al. (2002) hold the view that journals ought to allow more space for the results of reviews to be reported comprehensively. They also stated that information that cannot fit into a journal should be widely available on a website, the URL of which should be included in the journal article. Currently this is the practice of some journals (e.g. British Medical Journal) but not all.

8.7 Limitations, inconsistencies and gaps in the evidence

The findings of this update show many of the same limitations, inconsistencies and gaps as the original Evidence Briefing. The main issues we found with the current review level evidence are stated below. For implications of the limitations, inconsistencies and gaps in the evidence see the policy and practice recommendations section (9) and the research recommendations section (10).

There was a dearth of UK-based evidence on intervention effectiveness. Thus, one of the main limitation we found was the lack generalisability to the UK population as so few studies included in the reviews took place in the UK.

We found that much of the literature included in the reviews focused upon MSM and BME groups, with only very little review-level evidence concentrating on CSWs and people with HIV. However, there have been improvements in the evidence available for people with HIV since the first evidence

briefing, where no review-level evidence was found. Nevertheless, there is a distinct lack of secondary research focusing upon interventions with either male or female CSWs.

Following on from the previous point there was a lack of evidence reported for individual and group level interventions for CSWs and very little evidence reported at the community level for people with HIV. There were also a few studies that we were unable to categorise by intervention level because full details of the studies were not available in the reviews.

We found no evidence relating to the impact of interventions on inequalities or interventions addressing socio-political issues. In particular, there was a lack of evidence addressing the impact of interventions on socio-economic status and on vulnerable groups.

8.8 Men who have sex with men (MSM)

Based on the review level evidence identified, there were insufficient results to make conclusions regarding the effectiveness of community level interventions for MSM. Kegeles and Hart (1998; cited in Johnson et al., 2003) note that "community-based interventions have the capacity to reach people who would not participate in facility-based interventions, and who may be at higher risk than many who enrol in small group or individual level interventions" (p.2). Positive findings have been reported regarding peer-led community interventions, however, more evidence is needed on the effective components of community level interventions.

There is a lack of evidence, and hence the potential for future research to focus upon newly identified predictors of increased sexual risk behaviour among MSM relating to: HIV treatment optimism, social norms regarding intentional unsafe sexual practices, and use of the internet to solicit sexual partners (Herbst et al., 2005). Rees et al. (2004) noted that no review has synthesised the effects of interventions using the primary outcome of serodiscordant or unknown status unprotected anal intercourse.

We found no reviews that incorporated studies addressing inequalities or socio-political related interventions. Rees et al. (2004) reported the need for future studies concerning interventions with MSM to focus upon the views of all groups of MSM, in particular young MSM, working class MSM, black minority ethnic MSM, disabled MSM, and other vulnerable MSM groups.

Even though several Category 1 and 2 reviews incorporated studies of interventions focusing on MSM we found few studies that had been conducted in UK populations. Almost all of the reviews examined for MSM contained studies from outside the UK, with the exception of Rees et al. (2004). More studies of interventions targeted at UK-based populations of MSM are urgently required.

8.9 Commercial sex workers (CSWs)

There was tentative review level evidence to conclude that community level interventions can affect the sexual risk behaviours of CSWs and their patrons. However, insufficient evidence was found to support or discount the effectiveness of individual or group level interventions. Once again the studies reported were carried out in countries other than the UK (including Thailand) and as such their findings may be of limited relevance to UK-based populations of CSWs. As CSWs are considered a priority population and at a high-risk for sexual transmission of HIV, it is crucial that further research is undertaken to determine which interventions are effective in this population. In addition to the gaps reported in the original Evidence Briefing (Ellis et al., 2003), Logan et al. (2002) noted that future research in this population should include an understanding of coping styles, social support needs, problem-solving skills, and promotion of health practices, possibly through the use of cognitive-behavioural therapy.

Miller et al. (1998; description in Herbst et al., 2005) carried out a peer-led intervention with male CSWs, the findings showing that the intervention had no effect on the risk behaviour of patrons or CSWs. However, this intervention was carried out in the US, took place in a bar and may represent a significantly different culture to the UK's male CSW population and patron population. Rees et al. (2004) reported several qualitative studies with male sex workers that aimed to elicit sexual health needs (not reported in this update due to their low quality). The results showed that male CSWs were frequently involved in risky sexual behaviour, were involved with drugs, and felt a lack of control over their lives. Men interviewed suggested that outreach, counselling and befriending networks would be welcome. Based on these findings, research should be carried out in UK-based populations of male CSWs to determine if community-based interventions are effective.

Limited, moderate-quality evidence has been reported regarding the effectiveness of interventions in the workplace with heterosexual men who potentially frequent sex workers (e.g. truck drivers). None of these studies were relevant to UK populations. However, relevant and rigorous studies with the same focus ought to be carried out in the UK.

8.10 People with HIV (PWHIV)

The evidence identified led us to conclude that there was tentative review-level evidence to support the effectiveness of partner notification and small-group level interventions in influencing the sexual risk behaviours of people living with HIV. The findings of partner notification interventions have been positive. However, Mathews et al. (2002) raised the issue that the harms of partner notification have been poorly investigated. They also stated that "patient education and counselling and provider training are likely to improve partner notification and prevent domestic violence. However, there is no clear evidence to guide decisions about the provision of effective patient education and counselling, and provider training" (p.298). There was insufficient review-level evidence to support the effectiveness of community-level interventions in influencing the sexual risk behaviour of people living with HIV. One Category 3 review (Fogarty et al., 2001; described in Metsch et al., 2005) aimed at women with HIV included a community-based study involving peer-led condom and contraception promotion. This study (Fogarty et al., 2001) was deemed to be successful because "First, women reported a preference for interacting with HIV-positive peers. Second, there was a close collaboration with case managers and community referral agencies due to the high unmet needs of participants. Third, behaviour change messages were tailored to address current motivations, intentions, and partner characteristics. Finally, intervention messages for the peers were reinforced by medical care providers, and most participants were in care" (p.199). This study, carried out in the US, showed promising results, however, it was limited to African American HIV positive women. Therefore, more research is needed to determine the effectiveness of community-level interventions targeting people living with HIV.

One Category 3 review Padian et al. (1993; description from Kelly et al., 2002) reported on a successful group-level intervention aimed at serodiscordant couples using risk-reduction and condom use modelling. This study used a longitudinal cohort as opposed to a control group design. However, findings suggested positive effects for the intervention. However, no details were provided of the sexual orientation or gender of the population.

We found a lack of evidence related to inequality or that considered issues such as access to services, stigma and discrimination. In addition, none of the studies identified for this update were carried out in the UK, and therefore their transferability to UK-based populations living with HIV may be questionable.

8.11 African communities in the UK

Although our searches uncovered a wide variety of studies that included evidence relating to BME groups, none of the studies included in these reviews were carried out in the UK. We found that there was tentative review-level evidence to conclude that group-level interventions may be effective in influencing the sexual risk behaviour of adolescent BME and community-level interventions may be effective in influencing the sexual risk behaviour of female, adolescent and heterosexual male BME groups. The majority of studies were undertaken in the US and included black African American or Hispanic populations. As such these studies have limited transferability to UK populations of BME groups.

In addition to the gaps identified in the original Evidence Briefing (Ellis et al., 2003) the dearth of evidence to support or discount the effectiveness of individual-level interventions is notable.

Evidence suggests that racial, ethnic and cultural, as well as social norms, may hinder safe sex practices and act as barriers to HIV prevention interventions (Logan et al., 2002). Although studies

including culturally specific interventions have been carried out, none have taken place in the UK. The UK's BME communities differ to those in other countries and may require alternative approaches to those reported in the review-level evidence.

8.12 Voluntary counselling and testing (VCT)

Although we found additional VCT studies to add to the evidence base, there were no findings to compel us to change the evidence statements and conclusions from the original Evidence Briefing (Ellis et al., 2003). We can however, add to them that there is tentative evidence to support the view that mass media interventions are effective in influencing the uptake of VCT for MSM, women, and the general public. However, there was a lack of review-level evidence to support this view for CSWs.

8.13 Cost-effectiveness

This is the first time that cost-effectiveness reviews have been reported in an HIV evidence briefing. The evidence reported in this update led us to state that there was tentative review level evidence to conclude that interventions aimed at MSM populations are effective. The cost-effectiveness reviews included studies aimed at BME groups, CSWs and VCT. However, information on the primary studies was limited and all of the studies included were non-UK based. Nevertheless, all studies reported in this update (section 6) demonstrated that the interventions were cost-effective and may even be cost-saving to society. Based on these findings research should be undertaken on the cost-effectiveness of UK-based interventions.

8.14 Theory-based interventions

This update includes the first attempt to record and collate the theories that underpin interventions for each priority population. Many authors have recommended that future primary studies are designed within a theoretical framework. The evidence identified led to tentative conclusions to support the effectiveness of theory-based interventions for studies aimed at MSM and people with HIV, including injecting drug users. There is additional evidence to support the effectiveness of theory-based interventions, however, it is insufficient to draw conclusions on. It is likely that other primary studies used theories within their interventions. However, if this is the case it was not clearly reported by either the authors of the primary, or the secondary study.

8.15 Limitations of the Evidence Briefing

This update followed the same methodology as the original Evidence Briefing and therefore has all the same limitations reported in Ellis et al. (2003). In addition to these limitations we add that although all efforts were made to retrieve shortlisted documents we were unable to obtain 11% of the literature, which may have resulted in some relevant articles being missed.

Our most recent category one paper was published in 2005 (Johnson et al.), and the most recent primary studies presented within this level were published in 2004. This updates the existing evidence base of primary studies and reviews by six years. However, the majority of primary studies were published before this date with only 29% of published after 1998 and two studies unpublished*.

^{*} Of those primary studies included in Tables 6-12, Appendix F.

^{- 89 -} HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission, 2006.

9 Implications for policy and practice

The policy and practice recommendations reported below are based solely on the evidence statements of sufficient or tentative review-level evidence of effectiveness. As such, they consider only the evidence provided and retain the gaps and limitations that have been previously discussed. They ought not to be considered alone but in conjunction with other non-review evidence and other relevant source information. The recommendations provided here should not be considered alone but in addition to those of the original Evidence Briefing (Ellis et al., 2003).

9.1 Implications for policy and practice with men who have sex with men (MSM)

The evidence presented for MSM populations in this update supports the effectiveness of cognitivebehavioural individual-level interventions; cognitive-behavioural group work focusing upon riskreduction, sexual negotiation and communication skills; peer-related group level interventions; and community-level interventions involving peer and popular opinion leaders. Very few of the interventions reported were carried out within the UK and therefore not all findings may be transferable to the UK's MSM population.

The recommendations for MSM ought to be considered in conjunction with the MSM section of the Discussion (8.8) and the MSM evidence section (5.1). Our recommendations add to those of the original Evidence Briefing. We recommend that future research should:

- Place interventions in relevant theoretical frameworks
- Undertake individual-level cognitive-behavioural interventions, including self-justification, risk-reduction, education, and counselling
- Undertake group-level peer-led interventions, that address, for example, issues of intimacy, relationships, communication skills, coping skills, interpersonal skills, risk-reduction and relapse prevention.

9.2 Implications for policy and practice with commercial sex workers (CSWs)

Our findings show that although there was evidence supporting the effectiveness of peer-led community level interventions for female CSWs in the previous evidence briefing (Ellis et al., 2003), these same methods were not effective when used with male sex workers.

Two Category 3 studies demonstrated effectiveness for peer-led interventions and condom distribution schemes. However, these interventions were not carried out in the UK and may represent different CSW and patron populations to those in the UK. Also, several studies were reported (by Albarracin et al., 2005) but few details were provided and therefore we are unable to make any
practice recommendations. These conclusions should be considered with reference to the CSWs Discussion (8.9) and evidence (5.2) sections.

9.3 Implications for policy and practice with people with HIV (PWHIV)

Based on the evidence identified for people with HIV we made tentative review-level evidence statements supporting the effectiveness of partner notification and small-group level interventions. Very few of the interventions reported were carried out within the UK and therefore not all findings may be transferable to the UK's PWHIV population.

The recommendations for PWHIV should be considered in conjunction with the PWHIV section of the Discussion (8.10) and the PWHIV evidence section (5.3). Our recommendations add to those of the original Evidence Briefing. We recommend that future research should:

- Place interventions in relevant theoretical frameworks
- Undertake partner notification, including the choice between patient or provider referral
- Undertake small-group work, including stress management, HIV coping, and counselling.

9.4 Implications for policy and practice with African communities in the UK

The reviews identified in this update revealed tentative evidence to conclude that group-level interventions may have a modest effect on the sexual risk behaviours of BME heterosexual adults. Tentative review-level evidence was also available to conclude that community-level interventions may be effective for females, adolescents and heterosexual males. However, interventions reported in this update took place outside the UK, primarily in the US with African American, Hispanic and multi-ethnic populations. Given the cultural differences to the UK's African populations (see African section in the Discussion, 8.11) we suggest that these findings are not transferable to the UK's African population, therefore we are unable to make any specific practice recommendations.

These conclusions for African communities in the UK should be considered with reference to the African evidence section (5.4), the discussion and the African section (8.11) in the Discussion for further details.

9.5 Implications for policy and practice with voluntary counselling and testing (VCT)

The evidence identified for this update has made no further conclusions regarding VCT intervention effectiveness than those stated in the original Evidence Briefing (Ellis et al., 2003). However, we did conclude that there is tentative review level evidence to support the effectiveness of mass media interventions in influencing the uptake of HIV VCT in MSM, women, heterosexual men, and the

general public. As there is insufficient evidence regarding the harms of HIV VCT we suggest that VCT interventions and VCT promotion are aimed only at high-risk populations.

The recommendations for VCT should be considered in conjunction with the VCT section of the Discussion (8.12) and the VCT evidence section (5.5). Our recommendations add to those of the original Evidence Briefing.

• Future research should explore mass media promotion for VCT aimed at MSM and African populations.

10 Recommendations for research

The research recommendations below build on those previously stated in the original Evidence Briefing (Ellis et al., 2003) and aim to address the gaps that exist in the current intervention evidence.

10.1 General recommendations for secondary research/reviews

This update has found no evidence that dispute the general recommendations for primary and secondary research stated in the original Evidence Briefing. Although we have identified high quality, well-reported reviews, almost all of them originate from outside of the UK. As such, all original recommendations remain applicable. To the secondary research recommendations we would add that there is a need to:

- Conduct a review of UK-based primary studies with respect to cost-effectiveness of HIV prevention interventions
- Conduct a review of good quality UK-based qualitative research of HIV prevention interventions
- Ensure that future studies report on the details of the population included in the intervention, and state the proportion of HIV positive people included
- Ensure that future studies report the details of the intervention level (individual, group, community).

10.2 General recommendations for primary research

To the primary research recommendations we would add that there is a need to:

- Underpin primary research with relevant theory
- Include full details of the theory/theories used to be reported
- Standardise the reporting of primary research
- Design research studies that analyse findings in order to identify the component characteristics that result in sexual risk reduction
- Explore further research on migrant populations resident in the UK who come from countries with high HIV prevalence
- Include more rigorous qualitative research of HIV prevention at all levels and with all target groups (including migrant populations living in the UK)
- Explore more research at the socio-political level for HIV prevention with all priority populations
- Explore more research that considers the potential effects of criminal prosecutions for the sexual transmission of HIV on onward transmission of the virus.

10.3 Recommendations for research with men who have sex with men (MSM)

In addition to the recommendations above and those in the original Evidence Briefing, those below apply to MSM. There is a need for research to:

- Investigate HIV treatment optimism and its impact upon sexual risk-taking behaviour
- Consider interventions of sexual risk taking, including social norms regarding intentional unsafe sexual practices, and use of the internet to solicit sexual partners
- Explore interventions using the primary outcome of serodiscordant or unknown status unprotected anal intercourse
- Look at disabled MSM and equality
- Examine African MSM and explore issues such as cultural impediments to sexual health promotion interventions
- Look at the intervention effectiveness and cost-effectiveness of individual versus group level interventions underpinned by cognitive-behavioural techniques
- Examine the effectiveness interventions that include, for example, condom distribution, skills training, sexual communication, sexual health knowledge and HIV disclosure.

10.4 Recommendations for research with commercial sex workers (CSWs)

There is a need for research with CSWs to:

- Be carried out within a relevant theoretical framework
- Include an understanding of coping styles, social support needs, problem-solving skills and health promotion practices
- Explore the effectiveness of condom distribution schemes for both male and female CSWs.
- Consider workplace interventions focusing on men who potentially use CSWs, such as truck drivers
- Investigate interventions with male CSWs in the UK, including community-based outreach interventions.

10.5 Recommendations for research with African communities in the UK

- There is an urgent need for UK-based culturally sensitive research with African women, men and adolescents to be carried out at all intervention levels
- There is also a need for specific research to be conducted with African migrant populations living in the UK.

10.6 Recommendations for research with people with HIV

There is a need for research with people with HIV to:

• Investigate HIV treatment optimism and its impact upon sexual risk-taking behaviour

- Explore the potential harms of partner notification
- Include community-based peer-led condom and contraceptive promotion
- Be aimed at serodiscordant couples
- Address the issue of access to services for people with HIV
- Explore the effects of interventions including coping skills and disclosure on people with HIV
- Examine the effectiveness of community-based organisation interventions compared to clinicbased interventions.

10.7 Recommendations for research in relation to voluntary testing and counselling (VCT)

• There is a need for UK-based research on the effects of an HIV test resulting in a positive test on the sexual risk behaviour of serodiscordant couples.

11 References

Aggleton, P. (2004). Sexuality, HIV prevention, vulnerability and risk. *Journal of Psychology and Human Sexuality* 16 (1): 1-11.

Albarracin, D., Gillette, J. C., Earl, A. N., Glasman, L. R., Durantini, M. R., and Ho, M. H. (2005). A test of major assumptions about behavior change: A comprehensive look at the effects of passive and active HIV-prevention interventions since the beginning of the epidemic. *Psychological Bulletin* 131 (6): 856-897.

Albarracin, D., Johnson, B. T., Fishbein, M. et al. (2001). Theories of reasoned action and planned behavior as models of condom use: a meta-analysis. *Psychological Bulletin* 127 (1): 142-161.

Albarracin, D., Kumkale, G. T., and Johnson, B. T. (2004). Influences of social power and normative support on condom use decisions: a research synthesis. *Aids Care* 16 (6): 700-723.

Albarracin, D., McNatt, P. S., Klein, C. T. et al. (2003). Persuasive communications to change actions: an analysis of behavioral and cognitive impact in HIV prevention. *Health Psychology* 22 (2): 166-177.

American Academy of Pediatrics (2001). Adolescents and human immunodeficiency virus infection: the role of the pediatrician in prevention and intervention. *Pediatrics* 107 (1): 188-190.

American Academy of Pediatrics (2001). Condom use by adolescents. Pediatrics 107 (6): 1463-1469.

American Academy of Pediatrics (2001). Sexuality education for children and adolescents. *Pediatrics* 108 (2): 498-502.

Aral, S.O. and Peterman, T. A. (2002). STD diagnosis and treatment as an HIV prevention strategy. In: O'Leary, A. (ed) *Beyond Condoms: Alternative Approaches to HIV Prevention,* pp 77-90. New York, NY, US: Kluwer Academic / Plenum Publishers.

Arriola, K. R. J., Louden, T., Doldren, M. A. et al. (2005). A meta-Analysis of the relationship of child sexual abuse to HIV risk behavior among Women. *Child Abuse & Neglect* 29 (6): 725-746.

Askin, S. (2004). Review: HIV risk reduction interventions reduce some HIV risk behaviours in adolescents. *Evidence-Based Nursing* 7 (1): 11-

Bailey, R. C., Plummer, F. A., and Moses, S. (2001). Male circumcision and HIV prevention: current knowledge and future research directions. *The Lancet Infectious Diseases* 1 (4): 223-231.

Bandura A. (1977) Social Learning Theory. Prentice-Hall International, Inc., London.

Biddlecom, A. E. (2004). Trends in sexual behaviours and infections among young people in the United States. *Sexually Transmitted Infections.* 80 (SUPPL. 2): ii74-ii79.

Bok, M. (2002). A review of attitudes, social policy and educational materials on adolescent sexuality and HIV prevention. *Journal of HIV/AIDS Prevention & Education for Adolescents & Children* 5 (1-2): 45-60.

Bonell, C. (2002). The Politics of the Research-Policy Interface: Randomised Trials and the Commissioning of HIV Prevention Services. *Sociology of Health and Illness* 24 (4): 385-408.

Bonell, C. and Imrie, J. (2001). Behavioural interventions to prevent HIV infection: rapid evolution, increasing rigour, moderate success. *British Medical Bulletin* 58 : 155-170.

Bonner, K. (2001). Male Circumcision as an HIV Control Strategy: Not a 'Natural Condom'. *Reproductive Health Matters* 9 (18): 143-155.

Borgia, P., Marinacci, C., Schifano, P. et al. (2005). Is peer education the best approach for HIV prevention in schools? Findings from a randomized controlled trial. *Journal of Adolescent Health* 36 (6): 508-516.

Bradley-Springer, L. (2001). HIV prevention: what works? American Journal of Nursing 101 (6): 45-50.

Brown, P., (2000). Rate of HIV transmission among Africans in UK 'underestimated' BMJ 320: 735.

Burke, B. L., Arkowitz, H., and Menchola, M. (2003). The efficacy of motivational interviewing: a metaanalysis of controlled clinical trials. *Journal of Consulting & Clinical Psychology* 71 (5): 843-861.

Burke, B. L., Dunn, C. W., Atkins, D. C. et al. (2004). The Emerging Evidence Base for Motivational Interviewing: A Meta-Analytic and Qualitative Inquiry. *Journal of Cognitive Psychotherapy* 18 (4): 309-322.

Caceres, C. F. (2002). HIV among gay and other men who have sex with men in Latin America and the Caribbean: A hidden epidemic? *AIDS* 16 (Suppl 3): S23-S33.

Card, J., Benner, T., Shields, J. et al. (2001). The HIV/AIDS Prevention Program Archive (HAPPA): A collection of promising prevention programs in a box. *Journal of Acquired Immune Deficiency Syndrome* 13 (1): 1-28.

Cates, W., Jr. (2005). Review of non-hormonal contraception (condoms, intrauterine devices, nonoxynol-9 and combos) on HIV acquisition. *Journal of Acquired Immune Deficiency Syndromes: JAIDS* 38 (Suppl 1): S8-10.

Centers for Disease Control and Prevention (2001). US Centers for Disease Control and Prevention (CDC) report regarding selected public health topics affecting women's health. *MMWR*. *Recommendations and Reports* 50 (RR-6): 1-48.

Centers for Disease Control and Prevention (2002). Clinical prevention guidelines. Sexually transmitted diseases treatment guidelines 2002. *MMWR. Recommendations and Reports* 51(RR-6): 2-5.

Centers for Disease Control and Prevention, Health Resources and Services Administation, National Institutes of Health et al. (2003). Incorporating HIV prevention into the medical care of persons living with HIV. Recommendations of CDC, the Health Resources and Services Administration, the National Institutes of Health, and the HIV Medicine Association of the Infectious Diseases Society of America. *MMWR. Recommendations and Reports* 52 (RR-12):1-24.

Chippindale, S. and French, L. (2001). Clinical review. ABC of AIDS: HIV counselling and the psychosocial management of patients with HIV or AIDS. *BMJ* 2001 Jun 23; 322 (7301): 1533-1535.

Cohen DA, Wu SY, and Farley TA (2004). Comparing the cost-effectiveness of HIV prevention interventions. *Journal of Acquired Immune Deficiency Syndrome* 37 (3): 1404-1414.

Cook PA, Hargreaves S, Tocque K, Bellis MA (2004) Relationships between the use of hospital services and place of residence deprivation score among HIV positive individuals in the North West of England. Communicable Disease and Public Health 7: 319-321

Coon, D. W., Lipman, P. D., and Ory, M. G. (2003). Designing effective HIV/AIDS social and behavioral interventions for the population of those age 50 and older. *Journal of Acquired Immune Deficiency Syndromes: JAIDS* 33 (Suppl 2): S194-S205.

Cosgrove, P., Thomson, R. and Bellis, M. A. (2001a) Equitable strife. Health Service Journal, 4th January 2001:23.

Cosgrove, P., Lyons, M. and Bellis, M. A. (2001b) Economics of HIV and AIDS in the North West of England. Public Health Sector, Liverpool John Moores University.

Crepaz, N., Hart, T. A., and Marks, G. (2004). Highly active antiretroviral therapy and sexual risk behavior: A meta-analytic review. *JAMA* 14 (2): 224-236.

Demmer, C. (2003). HIV prevention in the era of new treatments. *Health Promotion Practice* 4 (4): 449-456.

Department of Health (2001). Better Prevention, Better Services, Better Sexual Health: the National strategy for sexual health and HIV. London: Department of Health.

Department of health (2002). *The National Strategy for Sexual Health and HIV: Implementation Action Plan.* London: Department of Health.

Department of Health (2003a). Effective commissioning of sexual health and HIV services: a sexual health and HIV commissioning toolkit for primary care trusts and local authorities. London: Department of Health (Available at www.doh.gov.uk)

Department of Health (2003b). *Effective sexual health promotion: a toolkit for primary care trusts and others working in the field of promoting good sexual health and HIV prevention*. London: Department of Health. (Available at www.doh.gov.uk)

Department of Health (2004) *Choosing Health: making health choices easier.* London: Department of Health (Available at www.doh.gov.uk)

Department of Health (2005a). *Delivering Choosing Health: making health choices easier*. London: Department of Health.

Department of Health (2005b). HIV and AIDS in African communities: A framework for better prevention and care. London: Department of Health. (Available at www.doh.gov.uk)

DiClemente, R. J., Crosby, R. A., and Wingood, G. M. (2002). Enhancing STD/HIV prevention among adolescents: the importance of parenteral monitoring. *Minerva Pediatrica* 54 (3): 171-177.

Dodds, C. (2002). Messages of Responsibility: HIV/AIDS Prevention Materials in England. *Health* 6 (2): 139-171.

Doncel, G. and Mauck, C. (2004). Vaginal microbicides: a novel approach to preventing sexual transmission of HIV. *Current HIV/AIDS Reports* 1 (1): 25-32.

Donenberg, G. R. and Pao, M. (2005). Youths and HIV/AIDS: psychiatry's role in a changing epidemic. *Journal of the American Academy of Child & Adolescent Psychiatry* 44 (8): 728-747.

Dore, H. (2005). HIV programs for teenagers work. *Perspectives on Sexual & Reproductive Health* 35 (4): 160.

Downing, J., Jones, L., Cook, P. A., Bellis, M. A. (Awaiting publication) Prevention of sexually transmitted infections (STIs): a review of reviews into the effectiveness of non-clinical interventions. Evidence briefing update. National Institute of Health and Clinical Excellence.

Dudgeon, W. D., Phillips, K. D., Bopp, C. M. et al. (2004). Physiological and Psychological Effects of Exercise Interventions in HIV Disease. *AIDS Patient Care & Stds* 18 (2): 81-98.

Dunn, C., DeRoo, L., and Rivara, F. P. (2001). The use of brief interventions adapted from motivational interviewing across behavioral domains: A systematic review. *Addiction* 96 (12): 1725-1742.

Elford, J., Bucher, H. C., Rawstorne, P. (2005). International perspectives. In: Kalichman, S. C. (ed) *Positive Prevention: Reducing Transmission Among People*, pp 245-274. New York, NY, US: Kluwer Academic / Plenum Publishers.

Elias, C. and Coggins, C. (2001). Acceptability research on female-controlled barrier methods to prevent heterosexual transmission of HIV: Where have we been? Where are we going? *Journal of Womens Health & Gender-Based Medicine* 10 (2): 163-173.

Elkavich, A., Rotheram-Borus, M. J., Goldstein, R., Flannery, D., and Jones, P. (2005). Young People Living with HIV. In: Kalichman, S. (ed) *Positive Prevention: Reducing Transmission Among People Living with AIDS/HIV*, pp 163-193. New York: Kluwer Academic/Plenum Publishers.

Ellis, S., Barnett-Page, E., Morgan, A., et al. (2002). HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission. London: Health Development Agency (now the National Institute for Health and Clinical Excellence)

Ellis, S. and Grey, A. (2004). Prevention of sexually transmitted infections (STIs): a review of reviews into the effectiveness of non-clinical interventions. London: Health Development Agency (now the National Institute for Health and Clinical Excellence)

Elwy, A. R., Hart, G. J., Hawkes, S., and Petticrew, M. (2002). Effectiveness of interventions to prevent sexually transmitted infections and human immunodeficiency virus in heterosexual men: a systematic review. *Archives of Internal Medicine* 162: 1818-1830.

EMCDDA (2002) Annual report on the state of the drugs problem in the European Union and Norway. Italy: European Monitoring Centre for Drugs and Drug Addiction.

Evans, B., McHenry, A. and Mortimer, J. (2001) HIV and AIDS in the UK. An Epidemiological Review: 2000. CDSC, London.

Farnham, P. G., Pinkerton, S. D., Holtgrave, D. R., and Johnson-Masotti, A. P. (2002). Costeffectiveness of counseling and testing to prevent sexual transmission of HIV in the United States. *AIDS and Behavior* 6 (1): 33-43.

Fenton, K. A. (2001). Strategies for improving sexual health in ethnic minorities. *Current Opinion in Infectious Diseases* 14 (1): 63-69.

Fenton, K.A., Chinouya, M., Davidson, O. and Copas, A. for the MAYISHA study team. (2002) HIV testing and high risk sexual behaviour among London's migrant African communities: a participatory research study. *Sex Transmitted Infections* 78: 241-245

Fisher J.D. and Fisher W.A. (2000). Theoretical approaches to individual level change in HIV risk behaviour. In: Peterson, J.L & DiClemente, R.J. (eds), *Handbook of HIV prevention*. New York: Kluwer Academic/Plenum Publishers

Flores, S. A. and Crepaz, N. (2004). Quality of study methods in individual- and group-level HIV intervention research: Critical reporting elements. *AIDS Education & Prevention* 16 (4): 341-352.

Fortenberry, J. D. (2002). Clinic-based service programs for increasing responsible sexual behavior. *Journal of Sex Research* 39 (1): 63-66.

Frankis, J. and Flowers, P. (2005). Men who have sex with men (MSM) in public sex environments (Pses): a systematic review of quantitative literature. *Aids Care* 17 (3): 273-288.

Freeman, E. E., Weiss, H. A., Glynn, J. R. et al. (2006). Herpes simplex virus 2 infection increases HIV acquisition in men and women: Systematic review and meta-analysis of longitudinal studies. *Aids* (1): 73-83.

Gallant, M. and Maticka, T. E. (2004). School-based HIV prevention programmes for African youth. *Social Science and Medicine* 58: 1337-1351.

Gazzard, B., Anderson, J., Ainsworth, J. and Wood, C. (2005) Treat with respect: HIV, public health and immigration. Expert medical panel independent report to the All Party Parliamentary Group on AIDS. (Available on www.ukcoalition.org/migration/HIV-Treat_With_Respect1.pdf)

Genuis, S. J. and Genuis, S. K. (2004). Managing the sexually transmitted disease pandemic: a time for reevaluation. *American Journal of Obstetrics & Gynecology* 191 (4): 1103-1112.

Genuis, S. J. and Genuis, S. K. (2005). Primary prevention of sexually transmitted disease: applying the ABC strategy. *Postgraduate Medical Journal* 81 (955): 299-301.

Ghaziani, A. and Cook, T. D. (2005). Reducing HIV infections at circuit parties: from description to explanation and principles of intervention design. *Journal of the International Association of Physicians in AIDS Care: JIAPAC* 4 (2): 32-46.

Gilleece, Y. and Sullivan, A. (2005). Management of sexually transmitted infections in HIV positive individuals. *Current Opinion in Infectious Diseases* 18 (1): 43-47.

Glatt, A. E. (2001). Review: preventive therapy reduces the incidence of active tuberculosis in the short to medium term in patients with HIV... commentary on Wilkinson, D. et al. (2000) Drugs for preventing tuberculosis in HIV infected persons. In: The Cochrane Database of Systematic Reviews, Issue 4. *ACP Journal Club* 134 (3): 102.

Glik, D., Nowak, G., Valente, T. et al. (2002). Youth performing arts entertainment-education for HIV/AIDS prevention and health promotion: practice and research. *Journal of Health Communication* 7 (1): 39-57.

Gordon, C. M., Forsyth, A. D., Stall, R. et al. (2005). Prevention interventions with persons living with HIV/AIDS: state of the science and future directions. *AIDS Education & Prevention* 17 (1 Suppl A): 6-20.

Gott, M. (2004). Are older people at risk of sexually transmitted infections? A new look at the evidence. *Reviews in Clinical Gerontology* 14 (1): 5-13.

Hader, S. L., Smith, D. K., Moore, J. S. et al. (2001). HIV infection in women in the United States: Status at the millennium. *JAMA* (9): 1186-1192.

Harper, G. W., Hosek, S. G., Contreras, R., and Doll, M. (2003). Psychosocial Factors Impacting Condom Use Among Adolescents: A Review and Theoretical Integration. *Journal of HIV/AIDS Prevention & Education for Adolescents & Children* 5 (3-4): 33-69.

Helms, J. L. and Hirbour, C. A. (2004). HIV and juvenile delinquency. *American Journal of Forensic Psychology* 22 (2): 57-81.

Herbst, J. H., Sherba, R. T., Crepaz, N., Deluca, J. B., Zohrabyan, L., Stall, R. D., and Lyles, C. M. (2005). A meta-analytic review of HIV behavioral interventions for reducing sexual risk behavior of men who have sex with men. *Journal of Acquired Immune Deficiency Syndromes* 39 (2): 228-241.

Heuveline, P. (2004). Impact of the HIV epidemic on population and household structure: The dynamics and evidence to date. *AIDS* 18 (Suppl 2): S45-S53.

Hickson, F., Nutland, W., Doyle, T., et al. (2000). *Making it Count: A Collaborative Planning Framework to Reduce the Incidence of HIV Infection During Sex Between Men.* London, Sigma Research.

Hickson, F., Weatherburn, P., Reid, D., Jessop, K., and Hammond, G. (2007) *Consuming Passions: Findings from the United Kingdom Gay Men's Sex Survey 2005.* London, Sigma Research.

Hilton, B. A., Thompson, R., Moore-Dempsey, L. et al. (2001). Harm reduction theories and strategies for control of human immunodeficiency virus: a review of the literature. *Journal of Advanced Nursing* 33 (3): 357-370.

Holmes, K. K., Levine, R., and Weaver, M. (2004). Effectiveness of condoms in preventing sexually transmitted infections. *Bulletin of the World Health Organization* 82 (6): 454-461.

Holtgrave, D. R. (2004). HIV prevention, cost-utility analysis, and race/ethnicity: methodological considerations and recommendations. *Medical Decision Making* 24 (2): 181-191.

Horowitz, S. M. (2003). Applying the Transtheoretical Model to Pregnancy and STD Prevention: A Review of the Literature. *American Journal of Health Promotion* 17 (5): 304-328.

HPA (2007) AIDS/HIV new diagnoses quarterly surveillance tables. Health Protection Agency Centre for Infections, Health Protection Scotland and the Institute of Child Health (London) No 74:07/1.

Infectious Diseases Society of America (2004). Primary care guidelines for the management of persons infected with human immunodeficiency virus: recommendations of the HIV Medicine Association of the Infectious Diseases Society of America.

Jemmott, J. B. I. and Fry, D. (2002). The abstinence strategy for reducing sexual risk behavior. In: O'Leary, A. (ed) *Beyond Condoms: Alternative Approaches to HIV Prevention,* pp 109-137 New York, NY, US: Kluwer Academic / Plenum Publishers.

Jenkins, R. A. and Kim, B. (2004). Cultural Norms and Risk: Lessons Learned From HIV in Thailand. *Journal of Primary Prevention* 25 (1): 17-40.

Johnson, A. M., Mercer, C.H., Erens, B. et al. (2001) Sexual behaviour in Britain: partnerships, practices and HIV risk behaviours. *Lancet* 358 (9296): 1835-42.

Johnson, B. T., Carey, M. P., Marsh, K. L., Levin, K. D., and Scott-Sheldon, L. A. (2003). Interventions to reduce sexual risk for the human immunodeficiency virus in adolescents, 1985-2000: a research synthesis. *Archives of Pediatrics & Adolescent Medicine* 157 (4): 381-388.

Johnson, W. D., Hedges, L. V., and Diaz, R. M. (2003). Interventions to modify sexual risk behaviors for preventing HIV infection in men who have sex with men. In: *The Cochrane Database of Systematic Reviews*, Issue 1. Chichester: John Wiley & Son, Ltd.

Johnson, W. D., Hedges, L. V., Ramirez, G., Semaan, S., Norman, L. R., Sogolow, E., Sweat, M. D., and Diaz, R. M. (2002). HIV prevention research for men who have sex with men: a systematic review and meta-analysis. *Journal of Acquired Immune Deficiency Syndromes* 30 (Suppl 1): S118-S129.

Johnson, W. D., Holtgrave, D. R., McClellan, W. M., Flanders, W. D., Hill, A. N., and Goodman, M. (2005) HIV intervention research for men who have sex with men: a 7-year update. *AIDS Education & Prevention* 17 (6): 568-589.

Johnson, W. D., Semaan, S., Hedges, L. V. et al. (2002). A protocol for the analytical aspects of a systematic review of HIV prevention research.[erratum appears in J Acquir Immune Defic Syndr 2002 Nov 1;31(3):368]. *Journal of Acquired Immune Deficiency Syndromes: JAIDS* 30 (Suppl 1): S62-S72.

Johnson-Masotti, A. P., Laud, P. W., Hoffmann, R. G. et al. (2001). Probabilistic cost-effectiveness analysis of HIV prevention. Comparing a Bayesian approach with traditional deterministic sensitivity analysis. *Evaluation Review* 25 (4): 474-502.

Johnson-Masotti, A.P., Weinhardt, L.S., Pinkerton, S.D., and Otto-Salaj, L.L. (2003). Efficacy and cost-effectiveness of the first generation of HIV prevention interventions for people with severe and persistent mental illness. *Journal of Mental Health Policy and Economics* 6:23-35.

Joint United Nations Programme on HIV/AIDS (UNAIDS) and World Health Organisation (WHO), (2005) AIDS Epidemic Update – December 2005.

Keithley, J. K. and Swanson, B. (2001). Oral nutritional supplements in human immunodeficiency virus disease: a review of the evidence. *Nutrition in Clinical Practice* 2001 Apr; 16 (2): 98-104.

Kelly, J. A. and Kalichman, S. C. (2002). Behavioral research in HIV/AIDS primary and secondary prevention: Recent advances and future directions. *Journal of Consulting & Clinical Psychology* 70 (3): 626-639.

Kelly, P. J., Lesser, J., and Smoots, A. (2005). Tailoring STI & HIV prevention programs for teens. *MCN, American Journal of Maternal Child Nursing* 30 (4): 237-242.

Kennedy, M. and Doll, L. S. (2001). Male bisexuality and HIV risk. *Journal of Bisexuality* 1 (2-3): 109-135.

Kiene, S. M., Fisher, J. D. and Fisher, W. A. (2005). Interventions in clinical settings. In: Kalichman, S.C. (ed) *Positive Prevention: Reducing Transmission Among People*, pp 219-238. New York, NY, US: Kluwer Academic / Plenum Publishers.

Kinniburgh, J. (2004). The Effect of Organisational Change on the Working and Private Lives of Gay Men Working in HIV Prevention Organisations. *Dissertation Abstracts International* 2002 (63): 680-68C.

Kippax, S. and Van De Ven, P. (1998) An epidemic of orthodoxy? Design and methodology in the evaluation of the effectiveness of HIV health promotion. *Critical Public Health* 8 (4): 371-386.

Kippax, S. (2002). Negotiated Safety agreements among gay men In: O'Leary, A. (ed) *Beyond Condoms: Alternative Approaches to HIV Prevention,* pp 1-15 New York, NY, US: Kluwer Academic / Plenum Publishers.

Kirby, D. (2002). Effective approaches to reducing adolescent unprotected sex, pregnancy, and childbearing. *The Journal of Sex Research* 39 (1): 51-57.

Kirby, D. (2002). The impact of schools and school programs upon adolescent sexual behavior. *Journal of Sex Research* 39 (1): 27-33.

Klein, C. (2002). Structural barriers and facilitators in HIV prevention: A review of international research In: O'Leary, A. (ed) *Beyond Condoms: Alternative Approaches to HIV Prevention,* pp 17-46 New York, NY, US: Kluwer Academic / Plenum Publishers.

Komiti, A., Judd, F., Grech, P. et al. (2001). Suicidal behaviour in people with HIV/AIDS: A review. *Australian & New Zealand Journal of Psychiatry* 35 (6): 747-757.

Logan, T. K., Cole, J., and Leukefeld, C. (2002). Women, sex, and HIV: Social and contextual factors, meta-analysis of published interventions, and implications for practice and research. *Psychological Bulletin* 128 (6): 851-885.

Lyttle, P. H. and Thompson, S. C. (2004). Maintaining sexual health in commercial sex workers in Australia: condom effectiveness, screening, and management after acquiring sexually transmissible infections. *Australian and New Zealand Journal of Public Health* Aug; 28 (4): 351-359.

Marks, G., Crepaz, N., Senterfitt, J. W. et al. (2005). Meta-analysis of high-risk sexual behavior in persons aware and unaware they are infected with HIV in the United States: Implications for HIV prevention programs. *Journal of Acquired Immune Deficiency Syndromes* 39 (4): 446-453.

Marlatt, G.A. & Gordon, J.R. Determinants of relapse: Implications of the maintenance of behavior change. In: Davidson, P.O., and Davidson, S.M., eds. *Behavioral Medicine: Changing Health Lifestyle.* New York: Brunner/Mazel, 1980. pp.410-452.

Marsh, K. L., Johnson, B. T., and Carey, M. P. (2001). Conducting meta-analyses of HIV prevention literatures from a theory-testing perspective. *Evaluation & the Health Professions* 24 (3): 255-276.

Martin, G. (2005). Reducing the risk of HIV in gay and bisexual men. Nursing Times 101 (34): 38-41.

Mathews, C., Coetzee, N., Zwarenstein, M. et al. (2004). Partner notification. [update of Clin Evid. 2003 Dec;(10):1863-70]. *Clinical Evidence* (11): 2113-2120.

Mathews, C., Coetzee, N., Zwarenstein, M., Lombard, C., Guttmacher, S., Oxman, A., and Schmid, G. (2002). A systematic review of strategies for partner notification for sexually transmitted diseases, including HIV/AIDS. *International Journal of STD & AIDS* 13: 285-300.

Mayaud, P. and McCormick, D. (2001). Interventions against sexually transmitted infections (STI) to prevent HIV infection. *British Medical Bulletin* 58: 129-153.

MAYISHA II Collaborative Group. (2005). Assessing the feasibility and acceptability of community based prevalence surveys of HIV among black Africans in England. London: Health Protection Agency Centre for Infections.

McKay, A. (2000). Prevention of sexually transmitted infections in different populations: a review of behaviourally effective and cost-effective interventions. *Canadian Journal of Human Sexuality* 9: 95-120.

MedFASH (2003). *Recommended Standards for NHS HIV Services.* The Medical Foundation for Sexual Health and HIV.

Merzel, C. and D'Afflitti, J. (2003). Reconsidering community-based health promotion: promise, performance, and potential. *American Journal of Public Health* 93 (4): 557-574.

Metsch, L. R., Gooden, L. K., and Purcell, D. W. (2005). Interventions in Community Settings. In: Kalichman, S. (ed) *Positive Prevention: Reducing Transmission Among People Living with AIDS/HIV*, pp 193-217. New York: Kluwer Academic/Plenum Publishers.

Mijajlovic, S., Jones, G. L., Minichiello, V. et al. (2003). A comprehensive review of evidence based complementary medicine use in HIV/AIDS. *Journal of the Australian Traditional-Medicine Society* 9 (1): 17-23.

Miller, W.R. and Rollnick, S. (2002) *Motivational Interviewing: Preparing people for change*, 2nd ed. The Guilford press, London.

Minnis, A. M. and Padian, N. S. (2005). Effectiveness of female controlled barrier methods in preventing sexually transmitted infections and HIV: current evidence and future research directions. *Sexually Transmitted Infections* 81 (3): 193-200.

Mize, S. J., Robinson, B. E., Bockting, W. O. et al. (2002). Meta-analysis of the effectiveness of HIV prevention interventions for women (Structured abstract). *Aids Care* 14: 163-180.

Moore, J.S. (2002) Female-controlled Prevention Technologies. In: O'Leary, A. (ed) *Beyond Condoms: Alternative Approaches to HIV Prevention,* pp 47-76 New York, NY, US: Kluwer Academic / Plenum Publishers.

Mullen P.D., Ramirez G, Strouse D, Hedges LV, and Sogolow E. (2002). Meta-analysis of the effects of behavioral interventions on the sexual risk behavior of sexually experienced adolescents in controlled studies in the United States. *Journal of Acquired Immune Deficiency Syndrome* 30 (Suppl 1): S94-S105.

Mutchler, M. G. (2002). Will It Be Silence and Violence or Safety and Prevention for the Next Generation of Gay Youth? *International Journal of Sexuality and Gender Studies* 7 (1): 23-38.

Neumann, M. S., Johnson, W. D., Semaan, S., Flores, S. A., Peersman, G., Hedges, L. V., and Sogolow, E. (2002). Review and meta-analysis of HIV prevention intervention research for heterosexual adult populations in the United States. *Journal of Acquired Immune Deficiency Syndromes* 30 (Suppl 1): S106-S117.

O'Brien, K., Nixon, S., Tynan, A. M. et al. (2004). Effectiveness of Aerobic Exercise in Adults Living with HIV/AIDS: Systematic Review. *Medicine and Science in Sports and Exercise* 36 (10): 1659-1666.

O'Leary, A. E. (2002). Beyond condoms: Alternative approaches to HIV prevention.

Orel, N. A., Spence, M., and Steele, J. (2005). Getting the message out to older adults: Effective HIV health education risk reduction publications. *Journal of Applied Gerontology* 24 (5): 490-508.

Parsons, J.T. (2005). HIV-positive gay and bisexual men. In: Kalichman, S. C. (ed) *Positive Prevention: Reducing Transmission Among People,* pp 99-129. New York, NY, US: Kluwer Academic / Plenum Publishers.

Paxton, L. (2002) HIV treatment advances as prevention. In: O'Leary, A. (ed) *Beyond Condoms: Alternative Approaches to HIV Prevention,* pp 91-108 New York, NY, US: Kluwer Academic / Plenum Publishers.

Pedlow, C. T. and Carey, M. P. (2003). HIV sexual risk-reduction interventions for youth: A review and methodological critique of randomized controlled trials. *Behavior Modification* 27 (2): 135-190.

Pinkerton, S.D., and Holtgrave, D.R. (2002c) Assessing the cost-effectiveness of alternative approaches to HIV prevention. In: O'Leary, A. (ed) *Beyond condoms: Alternative approaches to HIV prevention*, pp139-171. New York: Kluwer Academic/Plenum Publishers.

Pinkerton, S.D., Johnson-Masotti, A.P., Holtgrave, D.R., and Farnham, P.G. (2002a). A review of the cost-effectiveness of interventions to prevent sexual transmission of HIV in the United States. *AIDS and Behavior* 6 (1): 15-31.

Pinkerton SD, Johnson-Masotti AP, Otto-Salaj LL et al. (2006). Cost-effectiveness of an HIV prevention intervention for mentally ill adults. *Mental Health Services Research* 3 (1): 45-55.

Pinkerton, S.D., Kahn, J.G., and Holtgrave, D.R. (2002b). Cost-effectiveness of community-level approaches to HIV prevention: A review. *The Journal of Primary Prevention* 23 (2): 175-198.

Porche, D. J. and Swayzer, R., III (2003). HIV prevention: a review of interventions. *Journal of the Association of Nurses in AIDS Care* 14 (1): 79-81.

Power, R., Gore-Felton, C., Vosvick, M. et al. (2002). HIV: effectiveness of complementary and alternative medicine. *Primary Care: Clinics in Office Practice* 29 (2): 361-378.

Price, N. (2001). The performance of social marketing in reaching the poor and vulnerable in AIDS control programmes. *Health Policy & Planning* 16 (3): 231-239.

Rees, R., Kavanagh, J., Burchett, H., Shepherd, J., Brunton, G., Harden, A., Thomas, J., Oliver, S., and Oakley, A. (2004). HIV health promotion and men who have sex with men (MSM): a systematic review of research relevant to the development and implementation of effective and appropriate interventions. London: EPPI-Centre, Social Science Research Unit, Institute of Education, University of London.

Rhodes, T., Sarang, A., Bobrik, A. et al. (2004). HIV transmission and HIV prevention associated with injecting drug use in the Russian Federation. *International Journal of Drug Policy* 15 (1): 1-16.

Robin, L., Dittus, P., Whitaker, D., Crosby, R., Ethier, K., Mezoff, J., Miller, K., and Pappas-Deluca, K. (2004). Behavioral interventions to reduce incidence of HIV, STD, and pregnancy among adolescents: a decade in review. *Journal of Adolescent Health* 34 (1): 3-26.

Robinson, B. B., Bockting, W. O., Rosser, B. R. et al. (2002). The Sexual Health Model: application of a sexological approach to HIV prevention. *Health Education Research* 17 (1): 43-57.

Rodriguez, C. M., Marques, L. F., and Touze, G. (2002). HIV and injection drug use in Latin America. *AIDS* 16 (Suppl 3): S34-S41.

Ross, M. W. and Williams, M. L. (2001). Sexual behavior and illicit drug use. *Annual Review of Sex Research* 12 : 290-310.

Ross, M. W. and Williams, M. L. (2002). Effective targeted and community HIV/STD prevention programs. *Journal of Sex Research* 39 (1): 58-62.

Rothman, R. E., Ketlogetswe, K. S., Dolan, T. et al. (2003). Preventive care in the emergency department: should emergency departments conduct routine HIV screening? a systematic review. *Academic Emergency Medicine* 10 (3): 278-285.

Russak, S. M., Ortiz, D. J., Galvan, F. H. et al. (2005). Protecting our militaries: A systematic literature review of military human immunodeficiency virus/acquired immunodeficiency syndrome prevention programs worldwide. *Military Medicine* 170 (10): 886-897.

Sangani, P., Rutherford, G., and Wilkinson, D. (2004). Population-based interventions for reducing sexually transmitted infections, including HIV infection. In: *The Cochrane Database of Systematic Reviews*, Issue 3. Chichester, UK: John.Wiley.& Sons., Ltd.

Sarkar, N. N. (2001). Adolescents' and young adults' sexual behaviour, STD and AIDS. *International Medical Journal* 8 (3): 167-173.

Savasta, A. M. (2004). HIV: associated transmission risks in older adults--an integrative review of the literature. *Journal of the Association of Nurses in AIDS Care* 15 (1): 50-59.

Schmiedl, R. (2004). School-based condom availability programs. *Journal of School Nursing* 20 (1): 16-21.

Semaan S, Kay, L., Strouse D, Sogolow E, Mullen PD, Neumann MS, Flores SA, Peersman G, Johnson WD, Lipman, P. D., Eke, A., and Des-Jarlais, D. C. (2002). A profile of US-based trials of behavioral and social interventions for HIV risk reduction. *Journal of Acquired Immune Deficiency Syndrome* 30 (Suppl 1): S30-S50.

Semaan, S., Des-Jarlais, D. C., Sogolow, E., Johnson, W. D., Hedges, L., V, Ramirez, G., Flores, S. A., Norman, L., Sweat, M. D., and Needle, R. (2002). A meta-analysis of the effect of HIV prevention interventions on the sex behaviors of drug users in the United States. *Journal of Acquired Immune Deficiency Syndromes* 30 (Suppl 1): S73-S93.

Shoveller, J. A. and Pietersma, W-A. W. (2002). Preventing HIV/AIDS risk behavior among youth. *AIDS and Behavior* 6 (2): 123-129.

Siegfried, N., Muller, M., Deeks, J. et al. (2005). HIV and male circumcision - A systematic review with assessment of the quality of studies. *Lancet Infectious Diseases* 5 (3): 165-173.

Siegfried, N., Muller, M., Volmink, J. et al. (2003). Male circumcision for prevention of heterosexual acquisition of HIV in men. In: *The Cochrane Database of Systematic Reviews,* Issue 3. Chichester, UK: John.Wiley.& Sons., Ltd.

Simoni, J. M. and Pantalone, D. W. (2004). Secrets and safety in the age of AIDS: does HIV disclosure lead to safer sex? *Topics in HIV Medicine* 12 (4): 109-118.

Simoni, J. M. and Pantalone, D. W. (2005). HIV Disclosure and Safer Sex. In: Kalichman, S. C. (ed) *Positive Prevention: Reducing Transmission Among People,* pp 65-99. New York, NY, US: Kluwer Academic / Plenum Publishers.

Smith, M. D. (2001) HIV risk in adolescents with severe mental illness: Literature review. *Journal of Adolescent Health* 29 (5): 320-329.

Strathdee, S. A. and Patterson, T. L. (2005). HIV-Positive and HCV-Positive Drug Users. In: Kalichman, S. C. (ed) *Positive Prevention: Reducing Transmission Among People*, pp 135-163. New York, NY, US: Kluwer Academic / Plenum Publishers.

Suarez, T., O'Leary, A. (2002). Selective serotonin reuptake inhibitors as a treatment for sexual compulsivity. In: O'Leary, A. (ed) *Beyond Condoms: Alternative Approaches to HIV Prevention*, pp 199-220 New York, NY, US: Kluwer Academic / Plenum Publishers.

Swearingen, S. G. and Klausner, J. D. (2005). Sildenafil use, sexual risk behavior, and risk for sexually transmitted diseases, including HIV infection. *American Journal of Medicine* 118 (6): 571-577.

Updegrove, K. K. (2001). An evidence-based approach to male circumcision: what do we know? *Journal of Midwifery & Women's Health* 2001 Nov-Dec; 46 (6): 415-422.

UK Collaborative Group for HIV and STI Surveillance. (2006) A Complex Picture. HIV and other Sexually Transmitted Infections in the United Kingdom. London: Health Protection Agency Centre for Infections.

Uphold, C. R. and Mkanta, W. N. (2005). Use of Health Care Services Among Persons Living with HIV Infection: State of the Science and Future Directions. *AIDS Patient Care & Stds* 19 (8): 473-485.

van-Empelen, P., Kok, G., van-Kesteren, N. M. C. et al. (2003). Effective methods to change sex-risk among drug users: a review of psychosocial interventions. *Social Science & Medicine* 57 (9): 1593-1608.

Vidanapathirana, J., Abramson, M. J., Forbes, A., and Fairley, C. (2005). Mass media interventions for promoting HIV testing. In: *The Cochrane Database of Systematic Reviews,* Issue 3. Chichester: John Wiley & Son, Ltd.

Villa, D. P. and Demmer, C. (2005). Exploring The Link Between AIDS-Related Grief and Unsafe Sex. *Illness, Crisis & Loss* 13 (3): 219-233.

Ward, D. J., Rowe, B., Pattison, H. et al. (2004). Behavioural interventions to reduce the risk of sexually transmitted infections in genitourinary medicine clinic patients: a systematic review. Birmingham, UK: West Midlands Health Technology Assessment Collaboration, Department of Public Health and Epidemiology, University of Birmingham.

Weatherburn, P., Davies, P., Hickson, F. et al. (1999). A class apart: the social stratification of HIV infection among homosexually active men. London: Sigma Research.

Weatherburn P, Reid D, Hickson F, Hammond G, Stephens M., (2005). Risk and reflexion: findings from the UK gay men's sex survey 2004, Sigma Research (available at www.sigmaresearch.co.uk)

Weinhardt, L.S. (2005). HIV diagnosis and risk behavior. In: Kalichman, S. (ed) *Positive Prevention: Reducing Transmission Among People Living with AIDS/HIV*, pp 29-60. New York: Kluwer Academic/Plenum Publishers.

Wellings, K., Nanchahal, K., MacDowall, W. et al. (2001) Sexual behaviour in Britain: early heterosexual experience. *Lancet* 358 (9269): 1843-50.

Wilkinson, D., Ramjee, G., Tholandi, M. et al. (2002). Nonoxynol-9 for preventing vaginal acquisition of sexually transmitted infections by women from men. In: *The Cochrane Database of Systematic Reviews*, Issue 4. Chichester, UK: John Wiley & Sons, Ltd.

Wilkinson, D., Tholandi, M., Ramjee, G. et al. (2002). Nonoxynol-9 spermicide for prevention of vaginally acquired HIV and other sexually transmitted infections: Systematic review and metaanalysis of randomised controlled trials including more than 5000 women. *Lancet Infectious Diseases* 2 (10): 613-617.

Wilson, B. D. and Miller, R. L. (2003). Examining strategies for culturally grounded HIV prevention: a review. *AIDS Education & Prevention* 15 (2): 184-202.

Wolitski, R. J. and Branson, B. M. (2002). "Gray area behaviors" and partner selection strategies: Working toward a comprehensive approach to reducing the sexual transmission of HIV. In: O'Leary, A. (ed) *Beyond Condoms: Alternative Approaches to HIV Prevention,* pp 173-198. New York, NY, US: Kluwer Academic / Plenum Publishers.

Wolitski, R. J., Janssen, R. S., Onorato, I. M., Purcell, D. W., and Crepaz, N. (2005). An Overview of Prevention with People Living with HIV. In: Kalichman, S. C. (ed) *Positive Prevention: Reducing Transmission Among People Living with AIDS/HIV*, pp 1-29. New York: Kluwer Academic/Plenum Publishers.

Yang, H., Li, X., Stanton, B. et al. (2005). Heterosexual transmission of HIV in China: a systematic review of behavioral studies in the past two decades. *Sexually Transmitted Diseases* 32 (5): 270-280.

Yep, G. A., Merrigan, G. M., Martin, J. B., Lovaas, K. E., and Cetron, A. B. (2002-2003). HIV/AIDS in Asian and Pacific Islander communities in the United States: a follow-up review and analysis with recommendations for researchers and practitioners. *International Quarterly of Community Health Education* 21 (3): 199-227.

Appendix A. Theories of behaviour change

The summaries (based on Fisher & Fisher's chapter in Peterson and DiClemente's *Handbook of HIV Prevention*, 2000) for the Health Belief Model, AIDS Risk Reduction Model, Transtheoretical Model and the Information-Motivation-Behavioural Skills model have been taken directly from the original Evidence Briefing by Ellis et al. (2003) and added to here with summaries of Motivational Interviewing (Miller and Rollnick, 2002) and Social Learning Theory (Bandura, 1977).

The Health Belief Model (HBM)

(Source: Fisher & Fisher in Peterson, J.L & DiClemente, R.J. (eds) (2000) Handbook of HIV prevention. New York: Kluwer Academic/Plenum Publishers).

The 'grandparent of all health behaviour models' (Fisher and Fisher, p5), the Health Belief Model (HBM) (Rosenstock, 1974) is a model of conscious decision-making. Perceived susceptibility (to illness, damage etc) and perceived severity (of negative health outcomes) together lead to perceived vulnerability which in turn leads to readiness to act. Health behaviour options are evaluated by the individual in terms of their perceived benefits and costs. The effectiveness of the new behaviour is judged as one of the 'benefits' of behaviour change (i.e. whether it works in averting harm). The HBM recognises the importance of a 'cue stimulus' to kick-start the new health behaviour (e.g. symptoms, mass media messages, opportunistic interventions). In 1998, the concept of self-efficacy was added. The HBM historically has focused on simple rather than complex behaviours (e.g. getting an injection rather than negotiating safer sex) and therefore its relevance to HIV prevention is questionable. There is only equivocal support for the HBM outside of the HIV field. The model has been used more for its predictive power in terms of behaviour than for designing interventions, and 'is essentially a listing of constructs rather than a model *per se'* and merely 'suggests conditions that prompt one to seek health-relevant services' (Fisher and Fisher, p8).

AIDS Risk Reduction Model (ARRM)

(Source: Fisher & Fisher in Peterson, J.L & DiClemente, R.J. (eds) (2000) Handbook of HIV prevention. New York: Kluwer Academic/Plenum Publishers).

The AIDS Risk Reduction Model (ARRM) was developed by Catania et al. (1990), and is one of two stage models of behaviour change; the other is the Transtheoretical Model (see below). Both assume change is a process, with different factors intervening at each step. However, the change processes are not viewed as irreversible or unidirectional. The initial stage, the labelling of behaviour as 'problematic', is facilitated by accurate information. Then comes the decision to commit to change, which is affected by a range of personal and social factors. Weighing the costs of behaviour change is important, taking into account factors such as enjoyment etc. The final stage is taking action. Health education messages can be critical here. But, Fisher and Fisher (2000) argue, there has been little research on those variables connected with the enactment stage. Catania et al. outline few conditions

affecting enactment of behaviour change (and 'thus suggest little in the way of content for effective interventions', Fisher and Fisher, p12). Internal/external cues facilitate movement from one stage to another and health messages can be crafted to whichever stage an individual or population has reached in the change process. Fisher and Fisher point out that knowing which stage a population is at can be useful in effectively targeting interventions, but they acknowledge mixed support for the relationship between variables, stage attainment and outcomes, though many critical variables in the ARRM are shown in the literature to be linked with ultimate behavioural change. Importantly, little work has been done on how people move between stages. Also, there is little on incorporating the ARRM into interventions. Overall, empirical support is equivocal.

Transtheoretical Model (TM)

(Source: Fisher & Fisher in Peterson, J.L & DiClemente, R.J. (eds) (2000) Handbook of HIV prevention. New York: Kluwer Academic/Plenum Publishers).

The Transtheoretical Model (TM) outlines six stages of change:

- Precontemplation not interested in changing behaviour
- Contemplation ready for information on behaviour and how to change
- Preparation 'appropriate recruits for traditional 'action-oriented' interventions'
- Action
- Maintenance six months after action
- Termination beyond temptation to relapse.

In addition, the authors of the model, Prochaska and Velicia (1997), suggested 10 processes (for example, consciousness raising; environmental evaluation) that assist the individual in changing behaviour, reflecting critical common elements in the hundreds of extant models of change, and which have been validated with safer sex and condom use. Some of these processes are more suited to earlier in the cycle, some to later; earlier, there should be more emphasis on experiential processes; later, on behavioural processes. As with the ARRM, the intervention implications of the TM are that there is the need to know where the population is in terms of stage, then to deliver 'stage-matched' interventions. From other health areas, there is support for different change processes used at different stages. Research suggests the TM can be applied to HIV as to other areas, but it is not clear how the components of the model interact: Fisher and Fisher note that there has been a lack of multivariate work done on this. They also note that few longitudinal or experimental studies have been done on the TM, and – an important point – that it may be very difficult to design interventions based on the TM.

The Social Cognitive Theory (SCT)

The Social Cognitive Theory (SCT) has been effectively applied to behaviour change in a number of different health areas; Bandura (1994) has played a key role in making it applicable to HIV prevention.

The SCT emphasises skills and self-beliefs: in other words, self-efficacy, which 'affects whether people will attempt to change at all, how much effort they will exert and how much they will persist in a change attempt without giving up' (Fisher and Fisher, p24). They state that knowledge and skills count for nothing without self-efficacy. There are four components:

- Information (including convincing people that they can change their behaviour). Bandura emphasises the need to put stress on perseverance so setbacks will not throw the individual; information needs to be culturally competent, credible etc.
- Self-regulation/risk reduction skill development. Self-regulation skills consist of knowing one's risk triggers, rehearsing arguments in favour of the new behaviour, the ability to self-reinforce; cognitive self-guidance (suggested intervention methods are the use of role models and of cognitive rehearsal); risk reduction, which can involve technical skills (use of condoms) or social skills (negotiating safer sex) (suggested intervention methods include, again, the use of role models, who should be identifiable with through culture, race, ethnicity, gender etc.)
- Self-efficacy with regard to skills. There is the need to increase self-efficacy in relation to skills, by practising in progressively more difficult contexts, with constructive feedback
- Social support development/engagement. It is posited that the social support which is most proximate is most important.

As Fisher and Fisher point out, there is strong evidence for the role of self-efficacy in promoting behaviour change, though this is not universal (see O'Leary et al. for contradictory findings). They also note that there is empirical evidence for the importance of normative support. Kalichman's review (1996) supports the effectiveness of interventions based on or incorporating 'core' elements of the SCT. Overall, the SCT is supported by evidence across a number of health behaviours. However, it shares many features with other models and doesn't contain an explicit elicitation research component.

The Theory of Reasoned Action (TRA)

Fishbein and Ajzen's Theory of Reasoned Action posits the following equation of behaviour change:

B~BI=[Aact]w1+ [SN]w2

where B= behaviour, BI= behavioural intent, A act=attitude toward an HIV preventive act, SN= subjective norm or perception of referent support for performance of act, and W1+ W2 are regression weighting. Attitudes and norms are themselves functions of psychological factors, e.g. beliefs and evaluations of consequences of acts etc. According to the theory, attitudes toward an HIV preventive act are a function of beliefs about the consequences of performing the act (Bi) multiplied by evaluations of those consequences (ei).Subjective norms concerning HIV preventive acts are a function of perceptions of whether specific categories of 'referent other' (e.g. peers, partners) want the

individual to perform the act (NBj) multiplied by the individual's motivation to comply with these referents' wishes (MCj).

Elicitation research is seen as crucial to identify these beliefs. Personality, demographic and other variables are influential only via beliefs, attitudes and norms etc. The TRA states that it is necessary to strengthen intentions in order to increase preventive behaviour. Interventions can strengthen intentions by strengthening attitudes and norms, by changing psychological factors. It is not clear that all factors external to the TRA influence behaviour only via the components proposed by the model: feelings about sexuality, information and skills, vulnerability perception, gender and ethnicity need crucial examination. However, overall, Fisher and Fisher judge the TRA to be 'well-specified and well-tested' (p29), and that it has proved successful in predicting HIV preventive behaviour. Significantly, there is broad support for interventions based on the TRA.

Theory of Planned Behaviour (TPB)

The Theory of Reasoned Action (TRA) doesn't address the extent to which prevention does not fall within the individual's control. This is what Ajzen, the author of the Theory of Planned Behaviour (TPB), seeks to address. When an individual does not perceive that they have total control over their behaviour, behaviour is affected. Perceptions of control interact with attitudes, norms and intentions, such that perceived control should affect behaviour when attitudes norms and intentions are favourable to behaviour but not when they aren't. Where an individual perceives they have complete control over their behaviour, the TPB reverts to the TRA.

Perceived control and behavioural intention have been significantly correlated; there is inconsistent evidence on the direct relationship between perceived control and behaviour. Interventions have proved broadly supportive of the TPB, but the ability of the constructs to predict HIV preventive behaviour over the TRA seems negligible. Perceptions of control play a significant role in influencing intentions to practise HIV preventive behaviour. 'Research suggests that promoting perceptions of control is helpful in promoting HIV preventive behaviour, a fact that is consistent with the TPB' (Fisher and Fisher, p38).

Motivational Interviewing (MI)

(Source: Miller & Rollnick. Motivational Interviewing: Preparing people for change, 2nd ed. The Guilford press, London, 2002)

Motivational interviewing is defined as a client-centred, directive method for enhancing intrinsic motivation to change by exploring and resolving ambivalence and is heavily influenced by Carl Roger's humanistic approach to person-centred counselling. MI focuses upon an individual's current interests and concerns; it does not address past issues. The discrepancies that are explored and developed as part of MI are due to incongruities among aspects of the client's own values and

experiences. The aim of MI is to resolve any such ambivalence, usually in a specific direction of change.

MI is a method of communication as opposed to a set of techniques. It differs from other motivational strategies as it does not impose any external approaches (e.g. behavioural approaches) but instead focuses upon an intrinsic motivation to change. This method holds that unless a change is in the person's interest it will not occur, as change is inherent to a person's own values and concerns.

MI can be used in conjunction with other theories (e.g. TTM) and has shown positive effects when used as a consultation prior to the use of other services. There are four stages to MI:

- **Expressing empathy** acceptance facilitated change; skilful reflective listening is fundamental; ambivalence is normal.
- **Developing discrepancy** the client rather than the counsellor should present the arguments for change; change is motivated by a perceived discrepancy between present behaviour and important personal gains or values.
- **Rolling with resistance** avoid arguement for change; resistance is not directly opposed; new perspectives are invited but not imposed; the client is a primary resource in finding answers and solutions; resistance is a signal to respond differently.
- **Support self-efficacy** a person's belief in the possibility of change is an important motivator; the client, not the counsellor, is responsible for choosing and carrying out change; the counsellor's own belief in the person's ability to change becomes a self-fulfilling prophecy.

Social Learning Theory (SLT)

(Source: Bandura A. Social Learning Theory. Prentice-Hall International, Inc., London. 1977)

Social Learning Theory emphasises the importance of observing and modelling the behaviours, attitudes and emotional reactions of others. Bandura (1977) argues that most human behaviour is learned in this way and that from observing others one forms an idea of how new behaviours are performed. These ideas are subsequently coded and act as a future guide for action. The processes of observational learning are:

- Attentional processes: including modelled behaviour (distinctiveness, affective valence, complexity, prevalence, functional value) and observer characteristics (sensory capacities, arousal level, perceptual set, past reinforcement). Modelled conduct varies in effectiveness and some models are able to hold the attention of others for extended periods, as such, specific people (e.g. community leaders) are sought out as models.
- Retention process: including symbolic coding, cognitive organisation, symbolic rehearsal and motor rehearsal, as individuals cannot be influenced by observation of modelled behaviour if they cannot remember it.

- **Motor reproduction**: including physical capabilities, availability of component responses, self-observation of reproductions and accuracy feedback. This involves converting symbolic representations into appropriate actions.
- Motivational process: including external reinforcement, vicarious reinforcement and selfreinforcement, because individuals are more likely to adopt modelled behaviour if it results in outcomes that are perceived as valuable and rewarding.

Modelling plays a role in spreading ideas and social practices within a society. Diffusion of innovation is achieved by introducing new behaviour, through prominent examples, which are then adopted at a rapid rate and either stabilise or decline dependent upon their perceived value.

Relapse prevention

(Source: Marlatt, G.A. & Gordon, J.R. Determinants of relapse: Implications of the maintenance of behavior change. In: Davidson, P.O., and Davidson, S.M., eds. Behavioral Medicine: Changing Health Lifestyle. New York: Brunner/Mazel, 1980. pp.410-452).

Most of the work on the relapse prevention model (RPM) has been done in the context of substance abuse, where relapse is a significant factor in the success of interventions. While the addictive nature of some substances and their physiological effects are likely to play a part in relapse, the RPM posits that there are also environmental and social determinants, and suggest that aspects of their model could be applied to 'refraining from illicit sexual activities'. The RPM predicts response to the relapse in terms of cognitive-behavioural theoretical principles.

The basic assumptions are that the individual has entered into a voluntary agreement to follow a behavioural rule, and by complying has a perceived sense of personal control over the target behaviour. The perception of control continues until a high risk situation is encountered: in many cases this would be an event beyond the individual's control such a confrontation with another person, peer pressure or another social trigger, or an environmental trigger. Typically the situation gives feelings of hopelessness or powerlessness, and the ability to cope depends on adequate coping response (e.g. be assertive in responding to social pressure). Successful control leads to expectation of control over subsequent events (associated to Bandura's 1977 notion of self efficacy). As self efficacy decreases and relapse is immanent, individuals may recall good feelings/outcomes from engaging in the activity, rather than the negative ones. At this point the model becomes less specific to sexual behaviour, as for substance use, since substances themselves are often used as methods of coping with stress.

Once the behavioural rule is broken, there is no going back, and this is known as the 'abstinence violation effect' (AVE). To cope with the cognitive dissonance this causes, individuals may perceive that they may as well relapse again. If the failure is attributed to a personal weakness, the expectancy of failure in the future is reinforced.

The authors suggest that relapse training should be considered before giving up the behaviour. The settings in which behaviour takes place should investigated, and strategies to avoid situations or cope with them should be devised. Individuals should be forewarned of psychological consequences of failure (AVE) with the reassurance that just because the individual has failed, this does not mean that they would necessarily fail in the future.

Summary

It is important to note that the theories outlined above by no means constitute an exhaustive list. In addition, the literature makes reference to Empowerment theory, self-disclosure theory and Coleman's maturational model of coming out.

Appendix B. Example of a search strategy

Medline search strategy

- 1. meta-analysis/
- 2. review literature/
- 3. (meta-analy\$ or meta analy\$ or metaanaly\$).ti,ab.
- 4. (systematic\$ adj4 (review\$ or overview\$)).mp.
- 5. meta-analysis.pt.
- 6. review.pt.
- 7. review.ti.
- 8. review literature.pt.
- 9. 1 or 2 or 3 or 4 or 5 or 6 or 7 or 8
- 10. Case Report.mp
- 11. letter.pt.
- 12. historical article.pt.
- 13. review, reported cases.mp.
- 14. review, multicase.mp.
- 15. 10 or 11 or 13 or 14
- 16. 9 not 15
- 17. animal/
- 18. human/
- 19. 17 not (17 and 18)
- 20. 16 not 19
- 21. Acquired Immunodeficiency Syndrome/
- 22. exp HIV/
- 23. HIV Infections/
- 24. acquired immunodeficiency syndrome.ti,ab.
- 25. aids.ti,ab.
- 26. hiv.ti,ab.
- 27. human immunodeficiency virus.ti,ab.
- 28. acquired immun\$1 deficiency syndrome.ti,ab.
- 29. human immun\$1 deficiency virus.ti,ab.
- 30. 21 or 22 or 23 or 24 or 25 or 26 or 27 or 28 or 29
- 31. (prevent\$ or reduc\$ or minimis\$ or minimiz\$ or decreas\$).mp.
- 32. (interven\$ or promot\$ or educat\$ or influenc\$).mp.
- 33. health education/
- 34. Patient Education/
- 35. exp Health Promotion/
- 36. Sex Education/
- 37. Safe Sex/
- 38. safe\$ sex.ti,ab.
- 40. (condom\$ or barrier\$).ti,ab.
- 41. (behavior\$ or behaviour\$).ti,ab.
- 42. Health Behavior/
- 43. Sex Behavior/
- 44. Contraception Behavior/

- 45. Counseling/
- 46. counsel\$.ti,ab.
- 47. Behavior Therapy/
- 48. Psychotherapy, Group/
- 49. attitude\$ change\$.mp.
- 50. (lifestyle change\$ or life style change\$).mp.
- 51. 31 or 32 or 33 or 34 or 35 or 36 or 37 or 38 or 39 or 40 or 41 or 42 or 43 or 44 or 45 or 46 or 47 or 48 or 49 or 50
- 52. 30 and 51
- 53. Disease Transmission, Vertical/
- 54. (vertical transmission or (mother adj5 child\$) or pediatric or paediatric or perinatal).mp.
- 55. AIDS Vaccines/
- 56. Vaccin\$.mp.
- 57. exp Anti-HIV Agents/
- 58. exp Antiviral Agents/
- 59. (antiviral\$ or anti viral\$ or antivirus\$ or anti virus\$).mp.
- 60. (antiretroviral\$ or anti retroviral\$ or antiretrovirus\$ or anti retrovirus\$).mp.
- 61. (antihiv\$ or anti hiv\$).mp.
- 62. Occupational Exposure/
- 63. (occupational exposure or occupational safety or occupational hazard\$ or occupational risk\$).ti,ab.
- 64. Hearing Aids/
- 65. hearing aid\$.ti,ab.
- 66. exp Blood Transfusion/
- 67. exp Organ Transplantation/
- 68. (blood transfusion\$ or organ transplant\$).ti,ab.
- 69. Developing Countries/
- 70. exp Africa/
- 71. India/
- 72. (developing countr\$ or underdeveloped countr\$ or under developed countr\$ or third world or africa\$ or india\$).ti,ab.
- 73. 53 or 54 or 55 or 56 or 57 or 58 or 59 or 60 or 61 or 62 or 63 or 64 or 65 or 66 or 67 or 68 or 69 or 70 or 71 or 72
- 74. 52 not 73
- 75. 20 and 74
- 76. Limit 75 to english language
- 77. Limit 76 to yr=2001-2006
- 78. from 77 keep 1-1000
- 79. from 77 keep 1001-1324

Appendix C: CAT screening tool

The CAT is divided into two stages. The first stage assesses the strengths of the methods used to identify and select all of the available literature, since this is regarded as one of the most important factors in ensuring a balanced view of the evidence. If a paper passes this first stage, then the quality of its methodological analysis and the appropriateness of its conclusions are assessed.

Authors:			
Title:			
Source:			
Does this paper address your topic area? Yes	No	Unsure	2
Circle whether the paper is a:			
 Systematic review Synthesis Other review (please specify) 			
Does it address (circle as appropriate):			
 Effectiveness (interventions and treatments) Monitoring and surveillance trends Other (please specify) Causation 			
Does the paper have a clearly focused aim or research question? (1)	Yes	No	Unsure
Consider whether the following are discussed:			
The population studied	Yes	No	Unsure
The interventions given	Yes	No	Unsure
The outcomes considered	Yes	No	Unsure
Inequalities	Yes	No	Unsure
Do the reviewers try to identify all relevant English language studies?	Yes	No	Unsure
Consider whether details are given for:			
Databases searched (2a)	Yes	No	Unsure
References followed up (2b)	Yes	No	Unsure
Experts consulted (2b)	Yes	No	Unsure
Grey literature searched (2b)	Yes	No	Unsure
Years searched (2c)	Yes	No	Unsure
Search terms specified (2c)	Yes	No	Unsure
Indusion criteria described (2d)	Yes	No	Unsure
Is it worth continuing?	Yes	No	
Why / Why not?			

Do the authors address the quality (rigour) of the included studies? (3, 3d)	Yes	No	Unsure
Consider whether the following are used:	N.		
A rating system (3b)	Yes	NO	Unsure
More than one assessor (3c)	Yes	No	Unsure
If study results have been combined, was it reasonable to do so?	Yes	No	Unsure
Consider whether the following are true:			
Are the results of included studies clearly displayed? (4a)	Yes	No	Unsure
Are studies addressing similar research questions? (4b)	Yes	No	Unsure
Are the studies sufficiently similar in design? (4b)	Yes	No	Unsure
Are the results similar from study to study (test of heterogeneity)? (4b)	Yes	No	Unsure
Are the reasons for any variation in the results discussed?	Yes	No	Unsure

What is the overall finding of the review? Consider: How the results are expressed (numeric – relative risks, etc); whether the results could be due to chance (p-values and confidence intervals).

Are sufficient data from individual studies included to mediate between data and interpretation/conclusions? (5)		Yes	No	Unsure
Does this paper cover all appropriate interventions and approaches for this field (within the aims of the study)?		Yes	No	Unsure
If no, what?				
Relevance				
Can the results be applied/are generalisable				
to a UK population/population group?		Yes	No	Unsure
Are there cultural differences from the UK?		Yes	No	Unsure
Are there differences in healthcare provision with the UK? Is the paper focused on a particular target group (age, sex,		Yes	No	Unsure
population sub-group etc)?		Yes	No	Unsure
Accept for inclusion in the evidence briefing?	Yes	No	Refer to	
		1,2,3	4,5	third party
Use to inform the review of effectiveness?		Yes	No	
Use to inform the background discussion?		Yes	No	
Additional comments:				

Appendix D. List of reviews by Category 1-5

The appraised reviews were categorised as follows:

Category 1 reviews

- Elwy, A. R., Hart, G. J., Hawkes, S., and Petticrew, M. (2002). Effectiveness of interventions to prevent sexually transmitted infections and human immunodeficiency virus in heterosexual men: a systematic review. *Archives of Internal Medicine* 162: 1818-1830.
- Herbst, J. H., Sherba, R. T., Crepaz, N., Deluca, J. B., Zohrabyan, L., Stall, R. D., and Lyles, C. M. (2005). A meta-analytic review of HIV behavioral interventions for reducing sexual risk behavior of men who have sex with men. *Journal of Acquired Immune Deficiency Syndromes* 39 (2): 228-241.
- Mathews, C., Coetzee, N., Zwarenstein, M., Lombard, C., Guttmacher, S., Oxman, A., and Schmid, G. (2002). A systematic review of strategies for partner notification for sexually transmitted diseases, including HIV/AIDS. *International Journal of STD & AIDS* 13: 285-300.
- Pedlow, C. T. and Carey, M. P. (2003). HIV sexual risk-reduction interventions for youth: A review and methodological critique of randomized controlled trials. *Behavior Modification* 27 (2): 135-190.
- Rees, R., Kavanagh, J., Burchett, H., Shepherd, J., Brunton, G., Harden, A., Thomas, J., Oliver, S., and Oakley, A. (2004). HIV health promotion and men who have sex with men (MSM): a systematic review of research relevant to the development and implementation of effective and appropriate interventions. London: EPPI-Centre, Social Science Research Unit, Institute of Education, University of London.
- Vidanapathirana, J., Abramson, M. J., Forbes, A., and Fairley, C. (2005). Mass media interventions for promoting HIV testing. *The Cochrane Database of Systematic Reviews*, Issue 3.

Category 2 reviews

- Albarracin, D., Gillette, J. C., Earl, A. N., Glasman, L. R., Durantini, M. R., and Ho, M. H. (2005). A Test of Major Assumptions About Behavior Change: A Comprehensive Look at the Effects of Passive and Active HIV-Prevention Interventions Since the Beginning of the Epidemic. *Psychological Bulletin* 131 (6): 856-897.
- Farnham, P. G., Pinkerton, S. D., Holtgrave, D. R., and Johnson-Masotti, A. P. (2002). Costeffectiveness of counseling and testing to prevent sexual transmission of HIV in the United States. *AIDS and Behavior* 6 (1): 33-43.
- Johnson, B. T., Carey, M. P., Marsh, K. L., Levin, K. D., and Scott-Sheldon, L. A. (2003). Interventions to reduce sexual risk for the human immunodeficiency virus in adolescents, 1985-2000: a research synthesis. *Archives of Pediatrics & Adolescent Medicine* 157 (4): 381-388.

- Johnson, W. D., Hedges, L. V., and Diaz, R. M. (2003). Interventions to modify sexual risk behaviors for preventing HIV infection in men who have sex with men. *The Cochrane Database of Systematic Reviews*, Issue 1.
- Johnson, W. D., Hedges, L. V., Ramirez, G., Semaan, S., Norman, L. R., Sogolow, E., Sweat, M. D., and Diaz, R. M. (2002). HIV prevention research for men who have sex with men: a systematic review and meta-analysis. *Journal of Acquired Immune Deficiency Syndromes* 30 (Suppl 1): S118-S129.
- Johnson, W. D., Holtgrave, D. R., McClellan, W. M., Flanders, W. D., Hill, A. N., and Goodman, M. (2005) HIV intervention research for men who have sex with men: a 7-year update. *AIDS Education & Prevention* 17 (6): 568-589.
- Logan, T. K., Cole, J., and Leukefeld, C. (2002). Women, sex, and HIV: Social and contextual factors, meta-analysis of published interventions, and implications for practice and research. *Psychological Bulletin* 128 (6): 851-885.
- Merzel, C. and D'Afflitti, J. (2003). Reconsidering community-based health promotion: promise, performance, and potential. *American Journal of Public Health* 93 (4): 557-574.
- Mullen, P. D., Ramirez, G., Strouse, D., Hedges, L. V., and Sogolow, E. (2002). Meta-analysis of the effects of behavioral interventions on the sexual risk behavior of sexually experienced adolescents in controlled studies in the United States. *Journal of Acquired Immune Deficiency Syndrome* 30 (Suppl 1): S94-S105.
- Neumann, M. S., Johnson, W. D., Semaan, S., Flores, S. A., Peersman, G., Hedges, L. V., and Sogolow, E. (2002). Review and meta-analysis of HIV prevention intervention research for heterosexual adult populations in the United States. *Journal of Acquired Immune Deficiency Syndromes* 30 (Suppl 1): S106-S117.
- Pinkerton, S. D., Johnson-Masotti, A. P., Holtgrave, D. R., and Farnham, P. G. (2002). A review of the cost-effectiveness of interventions to prevent sexual transmission of HIV in the United States. *AIDS and Behavior* 6 (1): 15-31
- Robin, L., Dittus, P., Whitaker, D., Crosby, R., Ethier, K., Mezoff, J., Miller, K., and Pappas-Deluca, K. (2004). Behavioral interventions to reduce incidence of HIV, STD, and pregnancy among adolescents: a decade in review. *Journal of Adolescent Health* 34 (1): 3-26.
- Semaan, S., Des-Jarlais, D. C., Sogolow, E., Johnson, W. D., Hedges, L., V, Ramirez, G., Flores, S. A., Norman, L., Sweat, M. D., and Needle, R. (2002). A meta-analysis of the effect of HIV prevention interventions on the sex behaviors of drug users in the United States. *Journal of Acquired Immune Deficiency Syndromes* 30 (Suppl 1): S73-S93.
- Van-Empelen, P., Kok, G., van-Kesteren, N. M. C., van-den-Borne, B., Bos, A. E. R., and Schaalma, H. P. (2003). Effective methods to change sex-risk among drug users: a review of psychosocial interventions. *Social Science & Medicine* 57 (9): 1593-1608.

- Wilson, B. D. and Miller, R. L. (2003). Examining strategies for culturally grounded HIV prevention: a review. *AIDS Education & Prevention* 15 (2): 184-202.
- Weinhardt, L. S. (2005). HIV diagnosis and risk behavior. In: Kalichman, S. (ed) *Positive Prevention: Reducing Transmission Among People Living with AIDS/HIV*, pp 29-60. New York: Kluwer Academic/Plenum Publishers.

Category 3 reviews

- Elkavich, A., Rotheram-Borus, M. J., Goldstein, R., Flannery, D., and Jones, P. (2005). Young People Living with HIV. In: Kalichman, S. (ed) *Positive Prevention: Reducing Transmission Among People Living with AIDS/HIV*, pp 163-193. New York: Kluwer Academic/Plenum Publishers.
- Fortenberry, J. D. (2002). Clinic-based service programs for increasing responsible sexual behavior. *Journal of Sex Research* 39 (1): 63-66.
- Harper, G. W., Hosek, S. G., Contreras, R., and Doll, M. (2003). Psychosocial Factors Impacting Condom Use Among Adolescents: A Review and Theoretical Integration. *Journal of HIV/AIDS Prevention & Education for Adolescents & Children* 5 (3-4): 33-69.
- Johnson-Masotti, A.P., Weinhardt, L.S., Pinkerton, S.D., and Otto-Salaj, L.L. (2003). Efficacy and cost-effectiveness of the first generation of HIV prevention interventions for people with severe and persistent mental illness. *Journal of Mental Health Policy and Economics* 6:23-35.
- Kelly, J. A. and Kalichman, S. C. (2002). Behavioral research in HIV/AIDS primary and secondary prevention: Recent advances and future directions. *Journal of Consulting & Clinical Psychology* 70 (3): 626-639.
- Kirby, D. (2002). Effective approaches to reducing adolescent unprotected sex, pregnancy, and childbearing. *The Journal of Sex Research* 39 (1): 51-57.
- Metsch, L. R., Gooden, L. K., and Purcell, D. W. (2005). Interventions in Community Settings. In: Kalichman, S. (ed) *Positive Prevention: Reducing Transmission Among People Living with AIDS/HIV*, pp 193-217. New York: Kluwer Academic/Plenum Publishers.
- Pinkerton, S.D., Kahn, J.G., and Holtgrave, D.R. (2002b). Cost-effectiveness of community-level approaches to HIV prevention: A review. *The Journal of Primary Prevention* 23 (2): 175-198.
- Pinkerton, S.D., and Holtgrave, D.R. (2002) Assessing the cost-effectiveness of alternative approaches to HIV prevention. In: O'Leary, A. (ed) *Beyond condoms: Alternative approaches to HIV prevention*, pp139-171. New York: Kluwer Academic/Plenum Publishers.
- Ross, M. W. and Williams, M. L. (2002). Effective targeted and community HIV/STD prevention programs. *Journal of Sex Research* 39 (1): 58-62.

Wolitski, R. J., Janssen, R. S., Onorato, I. M., Purcell, D. W., and Crepaz, N. (2005). An Overview of Prevention with People Living with HIV. In: Kalichman, S. (ed) *Positive Prevention: Reducing Transmission Among People Living with AIDS/HIV*, pp 1-29. New York: Kluwer Academic/Plenum Publishers.

Category 4 - see References for full details

Author(s)	Year	Reason for not including in Category 1-3
DiClemente, R.J. et al.	2002	Not a review or synthesis, but identifies gaps in the evidence.
Elford J. et. al.	2002	Not a review or synthesis, but includes information about relevant communities and sexual practices in the UK.
Gallant, M. & Maticka, T.E.	2004	Study based on African population in Africa.
Genuis, S.J. & Genuis, S.K.	2005	Not a review or synthesis, but good overview of HIV prevention interventions.
Glik, D. et al.	2002	Not a review or synthesis, but useful information about performing arts education.
Gordon, C.M. et al.	2005	Not a review or synthesis, but has some useful information on published studies.
Kennedy, M. & Doll, L.S.	2001	Not a review or synthesis, but includes information on relevant prevention interventions.
Kirby, D.	2002b	Not a review or synthesis, but some useful information on the effect of school and school programmes on adolescent sexual behaviour.
Martin, G.	2005	Not a review or synthesis, but contains information on relevant prevention interventions.
Semaan, S. et al.	2002	Not about intervention effectiveness but includes useful information about US-based HIV prevention studies.
Shoveller, J. A. and Pietersma, W-A. W.	2002	Not about intervention effectiveness but includes useful information about the quality of US-based HIV preventions studies.
Simoni, J.M. & Pantalone, D.W.	2002	Not about prevention interventions but includes some useful research into HIV disclosure and safer sex.
Simoni, J.M. & Pantalone, D.W.	2004	Not about prevention interventions but includes some useful research into HIV disclosure and safer sex.
Smith M.D.	2001	Not about intervention effectiveness to relevant groups
Yep, G. A., Merrigan, G. M., et al	2002-03	Not a review or synthesis, but contains information on relevant prevention interventions.

Category 5 - see References for full details

Author(s)	Year	Reason for not including in Category 1-4
Aggleton, P.	2004	Not HIV prevention intervention
Albarracin, D. et al.	2001	Not about intervention effectiveness. Examines behaviour models.
Albarracia D et al	2003	Not about intervention effectiveness. Examines behavioural and
Albarracin, D. et al.	2005	cognitive models.
Albarracin, D. et al.	2003	Duplicate
Albarracin, D. et al.	2004	Not HIV prevention interventions
Albarracin, D. et al.	2004	Duplicate

American Academy of Pediatrics	2001b	Not a synthesis, review or meta-analysis.
American Academy of Pediatrics	2001	Not a review and about condom effectiveness
American Academy of Pediatrics.	2001a	Academic paper, not review or synthesis
Aral, S.O et al	2002	Treatment as HIV prevention
Arriola, K.R.J. et al.	2005	Not about intervention effectiveness.
Askin, S.	2004	Comment on another review
Bailey, R.C. et al.	2001	Male circumcision to prevent HIV transmission
Biddlecom, A.E.	2004	Not about intervention effectiveness
Bok, M.	2002	Description of HIV prevention interventions/reports/books etc. but no analyses of the effect of the programmes.
Bonell, C.	2002	Not about intervention effectiveness
Bonell, C. & Imrie, J.	2001	Not about intervention effectiveness
Bonner, K.	2001	Male circumcision.
Borgia, P. et al.	2005	Not a review
Bradley-Springer, L.	2001	Not about intervention effectiveness
Burke, B.L. et al.	2003	Not relevant to priority populations.
Burke, B.L. et al.	2003	Duplicate
Burke, B.L. et al.	2004	Not relevant to priority populations.
Caceres, C.F.	2002	Not about intervention effectiveness
		About the development of an archive of effectiveness programmes.
Card, J.J. et al.	2001	But doesn't give sufficient details about primary studies in terms of
		effectiveness.
Cates, W. Jr.	2005	Microbicides and condom effectiveness
Centers for Disease Control and Prevention	2001	Guidelines (mostly treatment).
Centers for Disease Control and Prevention	2002	STIs/HIV and treatment/care
Centers for Disease Control and Prevention,		
et al.	2003	Guidance to incorporate prevention into treatment
Chippindale, S. & French, L.	2001	Not a review of the literature. Not about intervention effectiveness.
Cohen, D.A. et al.	2004	Development of a tool to estimate the relative cost-effectiveness of HIV prevention interventions
Coon D.W. et al	2003	Not a synthesis of primary studies
	2000	
Ciepaz, N. et al.	2004	Not about intervention effectiveness although it does include
Demmer, C.	2003	implications for practice around HIV prevention
Dodds, C.	2002	Review HIV prevention literature.
Doncel, G. & Mauck, C.	2004	Vaginal microbicides
Donenberg, G. R. & Pao, M.	2005	Treatment and not prevention interventions
Dore, H.	2005	Comment on review – Johnson, B.T. et al. (2003)
Dudgeon, W.D. et al.	0004	
0	2004	Not about an intervention for HIV prevention.
Dunn, C. et al.	2004 2001	Not about an intervention for HIV prevention. Not relevant to priority populations.
Dunn, C. et al. Elias, C. & Coggins, C.	2004 2001 2001	Not about an intervention for HIV prevention. Not relevant to priority populations. About microbicides
Dunn, C. et al. Elias, C. & Coggins, C. Fenton, K.A.	2004 2001 2001 2001	Not about an intervention for HIV prevention. Not relevant to priority populations. About microbicides About STI prevention
Dunn, C. et al. Elias, C. & Coggins, C. Fenton, K.A. Flores, S.A. & Crepaz, N.	2004 2001 2001 2001 2004	Not about an intervention for HIV prevention. Not relevant to priority populations. About microbicides About STI prevention Not about intervention effectiveness. Focuses on quality markers for intervention research.
Dunn, C. et al. Elias, C. & Coggins, C. Fenton, K.A. Flores, S.A. & Crepaz, N. Frankis, J. & Flowers, P.	2004 2001 2001 2001 2004 2005	Not about an intervention for HIV prevention. Not relevant to priority populations. About microbicides About STI prevention Not about intervention effectiveness. Focuses on quality markers for intervention research. Sexual behaviour change not related to HIV prevention
Dunn, C. et al. Elias, C. & Coggins, C. Fenton, K.A. Flores, S.A. & Crepaz, N. Frankis, J. & Flowers, P. Freeman, E.E., et al.	2004 2001 2001 2001 2004 2005 2006	Not about an intervention for HIV prevention. Not relevant to priority populations. About microbicides About STI prevention Not about intervention effectiveness. Focuses on quality markers for intervention research. Sexual behaviour change not related to HIV prevention Doesn't cover intervention effectiveness.
Dunn, C. et al. Elias, C. & Coggins, C. Fenton, K.A. Flores, S.A. & Crepaz, N. Frankis, J. & Flowers, P. Freeman, E.E., et al. Genuis, S.J. & Genuis, S.K.	2004 2001 2001 2001 2004 2005 2006 2004	Not about an intervention for HIV prevention. Not relevant to priority populations. About microbicides About STI prevention Not about intervention effectiveness. Focuses on quality markers for intervention research. Sexual behaviour change not related to HIV prevention Doesn't cover intervention effectiveness. Not review etc. and STI focused not HIV

2005 Gilleece, Y. & Sullivan, A. Not review etc. and HIV/STI treatment not prevention 2001 Overview of a Cochrane review Glatt, A.E. Glik, D. et al. 2002 Duplicate 2005 Duplicate Gordon, C.M. et al. Gott, M. 2004 Not about HIV prevention interventions Hader, S.L. et al. 2001 Not about the effectiveness of an intervention Helms, J.L. & Hirbour, C.A. 2004 No - just description of interventions, but may be useful in write up. Heuveline, P. 2004 Not about intervention effectiveness HIV prevention in drug users focussing on harm reduction rather than Hilton, B.A. et al. 2001 sexual behaviour. Holmes, K.K. et al. 2004 Condom effectiveness Holtgrave, D.R. 2004 Methodology paper. US population only. Horowitz, S.M. 2003 STI prevention. 2004 Infectious Diseases Society of America Guidance on care Jemmott, J. B. et. al. 2002 Not relevant to priority populations Jenkins, R.A. & Kim, B. 2004 Not relevant to HIV prevention in the UK Johnson, W.D. et al. 2002 Not about intervention effectiveness 2001 Not based on synthesis of data. Johnson-Mascotti, A.P. et al. Keithley, J.K. & Swanson, B. 2001 HIV treatment Kelly, J.A. & Kalichman, S.C. 2002 Duplicate Kelly, P. J. et al. 2005 Not relevant Kiene, S.M. et.al. 2005 Not relevant review Kinniburgh J. 2004 Does not look at HIV prevention interventions. Kippax, S. 2002 Not a review, synthesis etc. Klein, C. et. al. 2002 Not about intervention effectiveness. Komiti, A. et al 2001 Not about intervention effectiveness Lyttle, P.H. & Thompson, S.C. 2004 Not HIV prevention intervention Marks, G. et al. 2005 Not about intervention effectiveness Methodological paper about meta-analysis with HIV prevention 2001 Marsh, K.L., et al. examples. Mathews, C. et al. 2004 Not a synthesis of primary studies. See Mathews, C et al. (2002) Mayaud, P. & McCormick, D. 2001 Treatment of STIs Mijajlovic, S. et al. 2003 HIV treatment Minnis, A.M. & Padian, N.S. 2005 Barrier method contraception and microbicides Mize, S.J. et al. 2002 Only deals with US studies of African-American women. Moore, J. S. et al 2002 Microbicides and female condoms Mullen, P.D. et al. 2002 Duplicate. Mutchler, M.G. 2002 Primary study O'Brien, K. et al. 2004 Not about a HIV prevention intervention O'Leary, A. et. al. 2002 Academic not a review Not about a prevention intervention, discusses 'effectiveness' of US Orel, N.A. et al. 2005 publications Parsons, J.T. 2005 Not a review Paxton, L. et al 2002 HIV treatment Pedlow, C.T. & Carey, M. P. 2003 Duplicate Pinkerton, S.D. et al. 2006 Not a review or synthesis. Cost-effectiveness of a single study. Porche, D.J. & Swayzer, R. III 2003 Lists types of interventions but doesn't comment on effectiveness Porche, D.J. & Swayzer, R. III 2003 Duplicate
Porche, D.J. & Swayzer, R., III	2003	Duplicate
Power, R. et al.	2002	Exclude - HIV treatment
Price N	2001	Not about intervention effectiveness. Mainly draws on developing
FICE, N.	2001	countries so not relevant to UK.
Rhodes, T. et al.	2004	Not relevant to UK population. Link between IDU and HIV transmission in Russia.
Robinson, B.B. et al.	2002	Not a review of intervention effectiveness
Rodriguez, C.M., et al.	2002	Interventions to prevent HIV via IDU. Not relevant to UK population.
Ross, M.W. & Williams, M.L.	2001	Discusses link between sex and drug use.
Rothman, R.E. et al.	2003	HIV testing not prevention interventions.
Russak, S.M. et al.	2005	Population covers military personnel only.
Sangani, P. et al.	2004	STI treatment for the prevention of HIV infection.
Sarkar, N.N.	2001	Not a review or synthesis
Savasta, A.M.	2004	Not about intervention effectiveness
Schmiedl, R.	2004	Not a review of HIV prevention intervention effectiveness.
Siegfried, N. et al.	2003	Circumcision to reduce HIV transmission.
Siegfried, N., et al.	2005	Male circumcision
Strathdee, S.A. & Patterson, T.L.	2005	Not a review etc.
Suarez, T.et. al.	2002	Not HIV prevention intervention
Swearingen, S.G. & Klausner, J. D.	2005	Not HIV prevention interventions
Updegrove, K.K.	2001	Male circumcision
Uphold, C.R. & Mkanta, W.N.	2005	Not about HIV prevention intervention effectiveness
Van Empelen P. et al.	2003	Duplicate
Villa, D.P. & Demmer, C.	2005	Not HIV prevention or relevant review
Ward D.J. et al.	2004	STI interventions not HIV
Weller, S. & Davis, K,	2001	About condom effectiveness
Wilkinson D. et al.	2002	Microbicides
Wilkinson, D. et al.	2002	Microbicides
Wilkinson, D., et al.	2002	Microbicides
Wolitski, R. J. et. al.	2002	Not HIV prevention intervention
Yang, H. et al.	2005	Behaviour and transmission not prevention interventions

Appendix E. In depth summary of Category 1 and 2 reviews

Category 1

• Elwy, A.R. (2002) Effectiveness of interventions to prevent sexually transmitted infections and human immunodeficiency virus in heterosexual men.

Data pool:	
Population	Black and ethnic minority groups and voluntary counselling and testing in heterosexual men
Setting(s)	Any
Interventions	Any
Searches	Databases were searched: AIDSline, ASSIA, BIDS (IBSS;EMBASE), British Nursing Index
	(BNI), CINAHL, Cochrane Database of Systematic Reviews, Current Research in Britain
	(CRIB), Database of Reviews of Effectiveness (DARE), ERIC, Health Promis (Health
	Development Agency, United Kingdom), HealthSTAR, Health CD, Helping Involve Men (HIM;
	John Hopkins), HMIC (King's Fund, United Kingdom), MEDLINE, National Research Register
	(United Kingdom), OCLC Papers First & Proceedings First, Popline, Psycinfo, SIGLE,
	Sociological Abstracts. Hand searches were also carried out in: AIDS, AIDS Care,
	International Journal of STD and AIDS, and Sexually Transmitted Diseases.
Selection/inclusion criteria	Studies were included if:
	(1) Intervention populations included heterosexual male subjects 15 years or older.
	(2) Data on heterosexual men's sexual behaviour were present and analysed separately
	from other groups included in the study or at least 80% of the study population consisted
	of heterosexual participants.
	(3) Study designs were coded by: (i) randomised or non-randomised controlled (ii)
	prospective observational, and (iii) retrospective observational.
	(4) The outcomes assessed in the study included at least one of the following: morbidity;
	behavioural outcomes and; social psychological outcomes. Studies were coded as being
	moderate to high quality.
Quality assessment	See selection/inclusion criteria
Types of studies	See selection/inclusion criteria
Number of studies	28 studies
Outcomes reported	Morbidity; behavioural outcomes (condom use, reduction in number of sex partners,
and/or used for	unprotected sex); social psychological outcomes (attitudes towards condoms or HIV, intention
assessing 'effectiveness'	to use condoms or change risky behaviour, knowledge of HIV and AIDS, self-efficacy of
	condom use, communication skills and quality of sexual relationships).
Review findings:	Drug users receiving treatment – interventions were highly successful at changing sexual
	behaviour; including decreases in the number of sexual partners. These studies varied in
	design, however, two studies used motivational and educational aspects, focused upon
	risk prevention, increasing levels of personal concern regarding HIV/AIDS and
	personalised the threat of HIV. These interventions also addressed condom use and drug
	and risky behaviour relapse prevention.
	Out of treatment injecting drug users – two RCTs aimed to increase condom use and
	decrease the number of sexual partners. They used stress management, drug
	awareness. AIDS education and personal risk awareness through peer education
	However, no changes in condom use of sexual partners was seen.
	Men in the workplace – Three studies targeted men in the workplace: truck drivers
	• went in the workplace – Three studies targeted men in the workplace. truck drivers

	targeted with VCT, skills in condom negotiation, condom use, demonstration and
	prevention and STI and HIV risk reduction showed a significant intervention effect on
	lowering STI incidence and decreasing the number of sexual partners; Brazilian port
	workers were targeted with face-to-face contact by peer outreach workers to discuss
	HIV/AIDS as well as free distribution of condoms. This resulted in significant reported
	increase in condom use and decrease in the numbers of sexual partners over two years.
	However, no change in attitude towards condoms or their knowledge of HIV/AIDS was
	reported. Truck workers in Senegal were targeted with peer-education intervention to
	increase condom use as well as skills training in condom use, condom negotiation, and
	communication and HIV/STI education. However, results were seen as unreliable and
	reports of decreased sexual partners were not corroborated by women.
	• Men attending clinics – nine studies (incorporating video-based education, motivation and
	skills based training, showed mixed results reported for attitudes towards condoms and
	knowledge of AIDS. Results were positive for improved communication skills with sex
	partner and condom use (two studies).
	Heterosexual males – six studies showed mixed results.
	Other men – four studies were included; one showed a significant reduction in STI
	incidence; one reported a positive effect on frequency of unprotected sex, condom use
	and number of sex partners; one showed no effect for condom use; changes were also
	reported in intention to engage in risky behaviour, and AIDS related knowledge.
Gaps and inconsistencies	Bias of mixed sex groups.
identified by the review:	Small sample sizes reported significant results, which may show publication bias.
	Most studies were conducted in the United States and involved black and minority ethnic
	groups and the findings for these studies may not be generalisable.
Research	• Research needs to focus on morbidity outcomes (e.g. incident infection) rather than only
recommendations:	behavioural or social psychological outcomes.
	 Interventions need to target heterosexual men or at least ensure that heterosexual men
	participate in single-sex intervention groups and then evaluations can then identify the
	approaches that are best suited to this population.
	 More research needs to be carried out in other parts of the world where rates of STIs and
	HIV are high among beterosexual men and where they are increasing
	Chudies offen then DOTe can identify many identify interventions the effective of the second se
	 Studies other than RC is can identify promising interventions; the effectiveness of these interventions approximate further riservery evolutions to first and the state of the
	Other the should have the statistical assess to be said to be fore widespread implementation.
	Studies should have the statistical power to demonstrate effectiveness, and this should
	be calculated prior to research being funded and implemented.

• Herbst, J.H. et al. (2005) A meta-analytic review of HIV behavioral interventions for reducing sexual risk behavior of men who have sex with men.

Data pool:			
Population	Men who have sex with men.		
Setting(s)	Any.		
Interventions	Any behavioural interventions.		
Searches	The Prevention Research Synthesis project database was updated through to 2003. Electronic databases were searched (AIDSLINE, Medline, PsycInfo, Embase, and the Web of Science, SocioFile and ERIC). In addition, searches were conducted of published abstracts from recent international AIDS conferences, the Cochrane Controlled Trials Register, the Current Controlled Trials Register, and the Computer Retrieval of Information on Scientific Projects database. The EPPI Centre were also requested to do a search of their database of effectiveness reviews. Forty journals published between June 2002 and July 2003 were also		
	hand searched and experts in the field were contacted.		
Selection/inclusion criteria	Studies had to meet the following criteria:		
	 Evaluate an HIV, AIDS, or STD behavioural intervention targeting or focusing on MSM (including homosexual, bisexual, or non–gay-identified men); Include only men, with 85% or more reporting same-sex behaviour; 		
	 Assess the same group prospectively over time or compare multiple groups receiving an intervention with a control or comparison group; 		
	 Assess outcomes before and after completion of the intervention; and 		
	 Report outcome data necessary to calculate effect size estimates for at least 1 sex behaviour or biologic outcome measure. (For reports that did not report sufficient information to calculate effect size estimates, the authors attempted to contact authors for clarification or additional information). 		
Quality assessment	The methodological quality of the included studies was assessed by coding the following study		
	design elements:		
	• Evaluation design (e.g., 1-group vs. multiple-group),		
	• Assignment method (random vs. non-random),		
	• Type of control group (wait list vs. treatment),		
	Time of follow-up assessments, and		
	Overall and differential retention rates		
Types of studies	Randomised controlled trials and non-randomised controlled trials (employed assignment		
	methods that allowed self selection or assignment based on capacity or convenience).		
Number of studies	33 studies were included.		
Outcomes reported	Unprotected anal intercourse, number of sex partners, condom use, and unprotected oral		
and/or used for	intercourse, STD diagnosis.		
assessing 'effectiveness'			
Review findings:	Interventions were associated with a significant decrease in unprotected anal intercourse (24		
	studies; OR 0.77, 95% CI: 0.65, 0.92), number of sexual partners (9 studies; OR 0.76; 95% CI:		
	0.61, 0.94) and with a significant increase in condom use during anal intercourse (9 studies;		
	OR 1.61, 95% CI: 1.16, 2.22).		
	Intervention characteristics associated with effectiveness were identified:		
	Theoretical models (based on diffusion of peer norms or relapse prevention)		
	 Interpersonal skills training (negotiation/communication of safer sex and assertiveness training) 		

	 ≥ 4 delivery methods (any of the following methods: counselling, group discussions, lectures, live demonstrations, and role play/practice) Exposure complexity (intensity/dose of intervention, >1 session, ≥4 hours exposure, ≥3 week time span)
Gaps and inconsistencies identified by the review:	 Lack of biologic endpoints. "Although the results of this review demonstrate that HIV behavioural interventions are efficacious in changing self-reported risk behaviours among MSM, it remains uncertain how large an effect would be required to lead to significant reductions in STD or HIV incidence". Also see <i>Research recommendations</i>
Research recommendations:	 The following recommendations were identified: Careful empiric tests that demonstrate these same effects across subgroups of MSM (e.g., non–gay-identified and substance-using MSM).
	• Evaluations of the effects of behavioural interventions designed specifically for subgroups of MSM and the effects of interventions adapted and tailored for these subgroups.
	• Future intervention efforts should consider newly identified predictors of increased sexual risk-taking among MSM, including HIV treatment optimism, social norms regarding intentional unsafe sex, and use of the Internet to solicit sex partners.
	 There is a crucial need for intervention research to meet the needs of emerging MSM communities throughout the world. An overwhelming need for proven models of HIV prevention for MSM in developing world settings is self-evident.
	• Future intervention studies should complement self-reported behavioural measures with more objective biologic outcomes in predictions of intervention efficacy and collect information on partner selection (e.g. "HIV serosorting").

• Mathews C. et al. (2002) A systematic review of strategies for partner notification for sexually transmitted diseases including HIV/AIDS.

Data pool:	
Population	STD/HIV patients (USA, Denmark, Lusaka, South Africa based study)
Setting(s)	Clinic setting
Interventions	Partner notification
Searches	Searches were conducted on five databases: MEDLINE, EMBASE, Psychological
	Abstracts, Sociological Abstracts, Cochrane Controlled Trials register.
	Hand-searches were carried out on the Proceedings of the International AIDS
	Conferences from 1994 to 2000 and the International Society for STD Research
	meetings (ISSTDR) from 1991 to 1999.
Selection/inclusion	Studies were included if they met the following criteria:
criteria	Published or unpublished randomised controlled trials (RCTs)
	• comparing two or more alternative partner notification strategies for people diagnosed
	with STDs which measured: partner elicited, located, notified, medically evaluated,
	harms to the index patient or partner, index patient re-infection rates, incidence of
	STDs, or changes in index patient's or partner's behaviour.
Quality assessment	Six core methodological criteria were used to identify study quality:
	(1) Randomisation designed and completed in an appropriate manner.
	(2) Participation rate greater than 80%.
	(3) Participants analysed in the groups to which they were assigned.
	(4) Outcome assessors blinded to the assignment status of the participants.
	(5) Groups similar at the start of the trial.
	(6) Groups treated equally in all aspects other than the intervention.
Types of studies	Published or unpublished randomised controlled trials
Number of studies	11 studies.
Outcomes reported and/or	Patient notification
used for assessing	
'effectiveness'	
Review findings:	In the first trial those patients that were offered a choice between provider and patient
	referral, compared with referral alone, resulted in more partners being notified with NNT=
	0.6; 95% CI=0.5 to 0.7). In the group with the choice most partners (70, 89.7%) were
	notified by the provider and only eight by the index patient. More partners in the group with
	the choice tested HIV-positive: 0.23 compared with 0.03 per index patient; NNT-5; 95%
	CI=2.7 to 25. There was no difference in the number of partners elicited. The participation
	in the first study was 46%, those who declined to participate, were ineligible or unavailable
	were more likely to be female, black and to have been tested confidentially rather than
	anonymously. The male participants were more likely to be homosexual or bisexual.
	The second study compared a choice between provider and patient referral with actient
	referral alone. Among the group of index nations randomly allocated to receive a choice
	hetween provider and patient referral provider referral was preferred with 82% of patients
	choosing provider referral with at least one partner
	Evidence suggests that giving index patients a choice between provider and patient
	referral may be more effective than nationt referral
Research recommendations:	
	One of the sime of these interventions needs to be long form behaviour shares and
	One of the alms of these interventions needs to be long-term behaviour change and appropriate support
	appropriate support.
	Another aim needs to be the availability of effective treatments.

٠	Evaluations of partner notification strategies need to address these issues, as well as
	potentially harmful effects, and cost.

• Pedlow C.T. and Carey M.P. (2003) HIV sexual risk-reduction interventions for youth: A review and methodological critique of randomized controlled trials.

Data pool:	
Population	Adolescents, ages 13 to 19 years.
Setting(s)	Any.
Interventions	Any intervention which aimed to reduce sexual risk behaviours.
Searches	Computerised literature searches of PsycInfo, AIDSLINE, Medline, and CINAHL, using the
	keywords and descriptors: HIV, AIDS, STD, prevention, education, risk reduction,
	intervention, adolescent(s), teen(s), and youth. In addition, the reference sections from
	articles and reviews were also searched.
Selection/inclusion	Studies were included if they met the following criteria: (1) primarily teenage samples
criteria	(mean ages ≤ 19 years old or, for studies that did not report the mean age, the age range
	was within 13 to 19 years); (2) randomized controlled trials (RCTs); (3) primary outcomes
	of reduction of sexual risk behaviors (e.g., frequency of unprotected sex, condom use);
	and (4) studies published in peer-reviewed journals prior to September 2000 were
	included.
	Interventions designed to increase rates of HIV testing were not included unless they also
	had specific sexual risk reduction outcomes.
Quality assessment	Only RCTs were included and the authors undertook a critique of the literature by
	examining the use of theory, intervention content and format, measurement, study design,
	and data analysis.
Types of studies	Randomised controlled trials.
Types of studies Number of studies	Randomised controlled trials. 22 RCTs.
Types of studies Number of studies Outcomes reported and/or	Randomised controlled trials. 22 RCTs. Sexual risk behaviours.
Types of studies Number of studies Outcomes reported and/or used for assessing	Randomised controlled trials. 22 RCTs. Sexual risk behaviours.
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness'	Randomised controlled trials. 22 RCTs. Sexual risk behaviours.
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide considerable evidence for the efficacy and effectiveness of HIV prevention programmes for
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings:	Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide considerable evidence for the efficacy and effectiveness of HIV prevention programmes for adolescents.
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings: Research recommendations:	Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide considerable evidence for the efficacy and effectiveness of HIV prevention programmes for adolescents. Theory
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings: Research recommendations:	Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide considerable evidence for the efficacy and effectiveness of HIV prevention programmes for adolescents. Theory • Research is needed to evaluate strategies to enhance readiness-to-change risk
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings: Research recommendations:	Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide considerable evidence for the efficacy and effectiveness of HIV prevention programmes for adolescents. Theory • Research is needed to evaluate strategies to enhance readiness-to-change risk behaviour, including motivational interviewing.
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings: Research recommendations:	Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide considerable evidence for the efficacy and effectiveness of HIV prevention programmes for adolescents. Theory • Research is needed to evaluate strategies to enhance readiness-to-change risk behaviour, including motivational interviewing. • Future studies can better apply theory to intervention design by determining the
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings: Research recommendations:	 Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide considerable evidence for the efficacy and effectiveness of HIV prevention programmes for adolescents. Theory Research is needed to evaluate strategies to enhance readiness-to-change risk behaviour, including motivational interviewing. Future studies can better apply theory to intervention design by determining the needs of the population including developmental factors, prior to implementing the
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings: Research recommendations:	 Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide considerable evidence for the efficacy and effectiveness of HIV prevention programmes for adolescents. Theory Research is needed to evaluate strategies to enhance readiness-to-change risk behaviour, including motivational interviewing. Future studies can better apply theory to intervention design by determining the needs of the population, including developmental factors, prior to implementing the intervention
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings: Research recommendations:	 Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide considerable evidence for the efficacy and effectiveness of HIV prevention programmes for adolescents. Theory Research is needed to evaluate strategies to enhance readiness-to-change risk behaviour, including motivational interviewing. Future studies can better apply theory to intervention design by determining the needs of the population, including developmental factors, prior to implementing the intervention.
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings: Research recommendations:	 Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide considerable evidence for the efficacy and effectiveness of HIV prevention programmes for adolescents. Theory Research is needed to evaluate strategies to enhance readiness-to-change risk behaviour, including motivational interviewing. Future studies can better apply theory to intervention design by determining the needs of the population, including developmental factors, prior to implementing the intervention. Intervention format and delivery
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings: Research recommendations:	 Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide considerable evidence for the efficacy and effectiveness of HIV prevention programmes for adolescents. Theory Research is needed to evaluate strategies to enhance readiness-to-change risk behaviour, including motivational interviewing. Future studies can better apply theory to intervention design by determining the needs of the population, including developmental factors, prior to implementing the intervention. Intervention format and delivery More research is needed to compare individual and group approaches and to develop offective brief evides intervention.
Types of studies Number of studies Outcomes reported and/or used for assessing 'effectiveness' Review findings: Research recommendations:	 Randomised controlled trials. 22 RCTs. Sexual risk behaviours. The authors found that, overall, 13 of the 23 interventions were shown to be effective in reducing sexual risk behaviour. The authors concluded that these studies provide considerable evidence for the efficacy and effectiveness of HIV prevention programmes for adolescents. Theory Research is needed to evaluate strategies to enhance readiness-to-change risk behaviour, including motivational interviewing. Future studies can better apply theory to intervention design by determining the needs of the population, including developmental factors, prior to implementing the intervention. Intervention format and delivery More research is needed to compare individual and group approaches and to develop effective brief, single-session, individualised interventions.

• Rees R et al. (2004) HIV health promotion and men who have sex with men (MSM): a systematic review of research relevant to the development and implementation of effective and appropriate interventions

Data pool:	
Population	Men who have sex with men.
Setting(s)	Any.
Interventions	HIV health promotion interventions or sexual risk reduction interventions in the context of HIV.
Searches	Searches were conducted in 13 databases/registers: Medline, Embase, CINAHL, ERIC, Social Science Citation Index, PsycInfo, British Education Index, EPPI Centre databases (BiblioMap, PrevRev), Database of Abstracts of Reviews of Effects (DARE), HealthPromis, Cochrane Controlled Trials Register, Cochrane Database of Systematic Reviews, Cochrane HIV/AIDS Group trials register, African Trials Register, Health Promotion Library Scotland catalogue, SIGLE, NGC, HIV/AIDS Prevention Research Synthesis (PRS) database. Hand searching was undertaken in Social Aspects of AIDS book series (eds 1993, 1995, 1997 and 2000). Bibliographies of systematic reviews were scanned.
Selection/inclusion	Studies were included if they met the following criteria:
criteria	Randomised controlled trial or controlled trial.
	 Intervention delivery completed after 1996 (reports were excluded if publication was prior to 1996 or if this data was unavailable).
Quality assessment	 Four core methodological criteria were used to identify three levels of study quality. 'Sound' outcome evaluations were those deemed to meet the four criteria of: Providing pre-intervention data for all individuals in each group; Providing post-intervention data for each group; Reporting findings for each outcome measure indicated in the aims of the study; and Employing a control/comparison group equivalent to the intervention group on socio- demographic and outcome variables.
Types of studies	Randomised controlled trials and controlled trials
Number of studies	 12 effectiveness reviews, 12 outcome evaluations of interventions delivered in or after 1996 were identified and 14 studies on MSM's views. 8 studies were considered methodologically sound and included in the effectiveness synthesis.
Outcomes reported and/or	Unprotected anal intercourse, other sexual practices, attitudes/motivations/intentions,
used for assessing	beliefs/perceptions, anal intercourse, and knowledge/awareness.
'effectiveness'	
Review findings:	 Findings from the eight studies indicated that: Interventions based on cognitive-behavioural techniques for MSM who identify themselves as at high risk appear to be effective in reducing the number of men reporting sero-discordant or unknown status unprotected anal intercourse. The effect on casual unprotected anal intercourse of information provision or information and counselling given within the context of participants' lifestyles is unclear. Due to limitations in the reporting of outcomes, the effect of various interventions on knowledge/ awareness and attitudes/beliefs were all deemed to be unclear. No evidence of effect was found for UK community interventions based upon peer-delivered HIV risk reduction messages for any of the review's priority outcomes. Peer oducators had mixed responses to those interventions.

	provision to discussion of behaviours or attitudes.		
	A key message of the report was "There is a relatively large evidence base for informing		
	policy and practice in the area of MSM and the barriers to and facilitators of HIV-related		
	sexual health. It is limited, however, in terms of its coverage of UK-based interventions, the		
	extent of use of unreliable evaluation designs and the study of selected vulnerable groups		
	of MSM".		
Research recommendations:	The following recommendations were reported:		
	Further rigorously conducted and reported primary and secondary research is		
	required on the views of all groups of MSM, in particular young MSM, working class		
	MSM, black and minority ethnic MSM, disabled MSM and other vulnerable groups of MSM.		
	Explore the comparative effectiveness of individual versus group level interventions		
	based on cognitive-behavioural techniques;		
	Explore the effectiveness of interventions which address the complexity of the		
	competing risks that MSM have to balance when making decisions about their HIV-		
	related sexual health: and		
	 Evaluate the offectiveness of interventions that sim to inform MCM of the verticus 		
	Explore the ellectiveness of interventions that aim to inform MSM of the values		
	markers that they of their sexual partners might be using to determine HIV status,		
	and the extent to which these can be relied upon.		
	Interventions that call for further evaluation include the following:		
	Interventions for young MSM to take into consideration the complicating factors		
	surrounding condom use and the impact of condoms on sexual pleasure;		
	Interventions specifically for young MSM to support inclusive conceptualisations of		
	MSM identity(s);		
	Interventions for HIV positive MSM to address the communication and strategic skills		
	needed to deal with situations HIV positive MSM find difficult (e.g. disclosure, condom		
	use);		
	Interventions targeting HIV positive MSM relating to the conflicts inherent in balancing		
	sexual intimacy and pleasure with condom use and communication about HIV: and		
	sexual mainacy and pleasure war condom use and communication about thy, and		
	 Interventions to help men deal with the psychological impact of HIV diagnosis and 		
	subsequent life as a sexual being.		
	Intervention areas that call for evaluation since they are lacking in sound evaluations and		
	yet match needs identified by particular groups of vulnerable MSM, include:		
	• Interventions aimed at young MSM to address gaps in their knowledge about the HIV		
	risks of oral sex and to support their testing decisions;		
	• Interventions aimed at all MSM that develop understanding of the way lives vary with		
	HIV status, understanding of the range of approaches men have to disclosing status,		
	shared responsibility for sexual safety between positive, negative and untested MSM,		
	and communication about HIV status;		
	 Interventions aimed at family and friends of MSM that enable them to support their 		
	HIV information and other support needs:		
	Interventions aimed at the gay community and at society in general to reduce the		
	stigma of HIV and attributions of blame for HIV;		
	Interventions aimed at society in general that enable development of an		
	understanding of the HIV sexual health needs of MSM, the means to address these		

	and to provide knowledge about where to go for further information on the HIV sexual health needs of MSM;
•	Interventions aimed at health professionals to provide training in HIV specific communication skills;
•	Support for drop-in centres to develop inclusive, relevant and non-judgemental services;
•	Support for specialist HIV services to provide personally relevant information and advice that is accessible and understandable and to ensure the necessary time is spent on HIV positive MSM's information needs;
•	Accessible provision of free condoms for MSM who sell sex; and
٠	Provision of resources for meeting places and befriending networks for young MSM.

• Vidanapathirana J. et al. (2005) Mass media interventions for promoting HIV testing.

Data pool:	
Population	Any intervention targeting the general public, as well as those targeting specific groups (including, commercial sex workers, drug users, MSM, bisexual people, pregnant women and adolescents).
Setting(s)	Any.
Interventions	Specific or general mass media campaigns, targeted at a population level or a specific target group that aimed to increase VCT for HIV. Relevant media: radio, television, print (newspapers, magazines, booklets, leaflets, posters, pamphlets), films, documentaries, billboards, street theatre or combinations of these.
Searches	Used the Cochrane HIV/AIDS search strategy. Following databases searched: Cochrane Central Register of Controlled Trials, Medline, Embase, NLM Gateway, CINAHL, AIDSearch, PsycInfo, Sociological Abstracts and Communication studies. All databases were searched until end of April 2004. The reference lists of related reviews and all articles obtained were also reviewed for additional citations. Experts in the field were contacted to identify ongoing research. Relevant websites of international agencies (e.g. UNAIDS, WHO, UNFPA) were also searched.
Selection/inclusion criteria	Studies had to meet the following criteria: (1) RCT, controlled clinical trial or interrupted time series analysis; (2) intervention targeting the general public or specific target groups; (3) specific or general mass media campaigns, targeted at population level or at specific target groups; and (4) outcome measure of direct impact of utilisation of HIV testing.
Quality assessment	For RCTs and controlled clinical trials: quality assessment included an evaluation of the following components for each included study, (a) similarity of both the intervention and control groups and that groups were treated equally expect for the intervention; (b) the method of generation of the randomised sequence; (c) the method of allocation concealment; (d) who was blinded/not blinded; (e) losses to follow-up and whether an intention to treat analysis was undertaken. Interrupted time series: quality assessment was considered according to seven criteria determined by the EPOC group, (a) intervention occurred at clearly defined point in time and there were at least three data points before and after the intervention; (b) interventions were unlikely to have affected data collection; (d) there was assessment of primary outcomes, or there were objective outcome variables; (e) completeness of the data set, which covered 80% or more of the total number of participants.
Types of studies	RCT, controlled clinical trial or interrupted time series analysis.
Number of studies	14 studies (2 RCTs, 3 non-randomised controlled studies and 9 interrupted time series)
Outcomes reported and/or used for assessing 'effectiveness'	Direct impact of intervention on the utilisation of HIV testing.
Review findings:	All individual studies concluded that mass media were effective, and this was confirmed by reanalysis by the authors of the interrupted time series studies which all had initial impact. Mass media interventions for the promotion of VCT showed significant immediate (random effect: estimated mean = 5.49, 95% CI: 2.37 to 8.61) and overall (random effect: estimated mean = 6.10, 95% CI: 1.81 to 10.38) effects. No long term effects were seen (random effect: estimated mean = 4.45, 95% CI: -0.19 to 9.08).
Research recommendations:	 Future research is required to identify the impact of mass media intervention for promotion of HIV testing on seropositivity status in high-risk groups and epidemic countries. Additional research would identify the cost-effectiveness, type of mass media

- 138 - HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission, 2006.

	interventions, and characteristics of messages in mass media to promote HIV testing.
•	More studies are required of new media strategies to maintain impacts in the long
	term.

Category 2

• Albarracin, D. et al. (2005) A test of major assumptions about behaviour change: A comprehensive look at the effects of passive and active HIV-prevention interventions since the beginning of the epidemic.

Data pool:	
Population	Any
Setting(s)	Any
Interventions	Theoretically underpinned behavioural interventions
Searches	• Five databases were searched, there were: MEDLINE, PsychINFO, ERIC, Social
	Science Citation Index and Dissertation Abstracts International.
	• Hand searches were also undertaken in AIDS, AIDS Education and Prevention, AIDS
	Research, American Behavioural Scientist, American Journal of Community
	Psychology, American Journal of Nursing, American Journal of Public Health, Basic
	and Applied Social Psychology, Communication Research, Communications, Health
	Communication, Health Education Quarterly, Health Education Research, Health
	Psychology, Journal of the American Medical Association, Journal of Applied
	communication Research, Journal of applied Social Psychology, Journal of
	Consulting and clinical Psychology, Journal of Personality and Social Psychology,
	journal of Sex Research, Medical Anthropology, Morbidity and Mortality Weekly
	Report, Qualitative Health Research, and Social Science and Medicine.
Selection/inclusion	Studies were included if they met the following criteria:
criteria	Outcome to promote the use of condoms
	Role-playing, practicing condom-use-related skills
	HIV counselling and testing
	Studies where effect sizes could be calculated
Quality assessment	Included studies met the inclusion criteria (no other details were provided)
Types of studies	Intervention Groups and control groups
Number of studies	191 studies
Outcomes reported and/or	Change in attitude; change in norms; change in perceived threat; change in control
used for assessing	perceptions; change in behavioural skills; change in knowledge.
'effectiveness'	
Review findings:	Findings from the HIV studies included were:
	• There was significant association at the point of pre-test and behaviour change based
	on the joint consideration of experimental and control groups, which suggests that
	HIV positive people generally increase their condom use.
	Samples including MSM changed most in response to interventions than other
	samples
	However, MSM were generally insensitive to the type of intervention strategy used
	with the exception of greater behaviour change in response to condom provision and
	less change in response to attitudinal arguments.
	• Those groups with high numbers of people with African backgrounds show more
	behaviour change in general and this change is attributable to behavioural skills
	arguments; self-management strategies and; HIV counselling and testing.
	Condom provision seems more effective for those with European backgrounds.
Gaps and inconsistencies	Limited data included on HIV positive individuals limited the analysis of the effectiveness of
identified by the review:	differing intervention components.

	There were discrepancies between the results of fixed- and random-effect analyses with
	the significant results reducing using the random-effects model.
Research recommendations:	Recommendations were that:
	Future research for MSM might concentrate on improving the efficacy of other techniques
	that are efficacious to other high-risk groups.
	Future research should try to provide a sufficiently large number of effect sizes to estimate
	the population variance more precisely and thus reconcile the discrepancies between the
	fixed and random-effect findings.

• Farnham, P. G., Pinkerton, S. D., Holtgrave, D. R. and Johnson-Masotti, A.P. (2002). Costeffectiveness of counseling and testing to prevent sexual transmission of HIV in the United States. *AIDS and Behavior* 6 (1): 33-43.

Data pool:	
Population	Any.
Setting(s)	Any.
Interventions	HIV counselling and testing.
Searches	Searches of electronic databases including AIDSLINE, Medline, PsycINFO and HealthStar
	to update searches for a previous review (Farnham, 1998).
Selection/inclusion	Studies had to meet the following inclusion criteria: (1) a clear identification of a prevention
criteria	or care and treatment programme; (2) a quantitative estimate of programme costs; (3) a
	quantitative estimate of programme outcomes, and (4) a quantitative comparison of costs
	and outcomes. Studies also had to report on sexual behaviour change following VCT.
Quality assessment	None reported.
Types of studies	Cost-effectiveness analyses and cost-benefit analyses.
Number of studies	11 cost studies.
Outcomes reported and/or	Benefit-cost ratio and cost per year of life saved.
used for assessing	
'effectiveness'	
Review findings:	The literature indicated that the interventions reviewed were generally cost-effective if
	targeted to groups of infected or high-risk individuals, although there is continuing debate
	over the efficacy of HIV counselling and testing in changing sexual behaviour. The authors
	conclude that HIV counselling and testing should be part of an overall prevention strategy
	which also included more intensive sexual risk reduction interventions such as individual,
	small-group, and community-level approaches.
Research recommendations:	None made.

• Johnson, B.T. et al. (2003) Interventions to reduce sexual risk for the human immunodeficiency virus in adolescents, 1985-2000: A research synthesis.

Data pool:	
Population	Adolescents.
Setting(s)	Any.
Interventions	Educational, psychosocial or behavioural interventions advocating sexual risk reduction for HIV
	prevention.
Searches	The following strategies were used to locate articles: (1) Six electronic databases (Medline,
	PsycInfo, AIDSLine, CINAHL, Dissertation Abstract Online and ERIC) from inception to 2000;
	(2) requests for papers were sent to researchers; (3) references of obtained articles; (4) manual
	searching of conference proceedings, and (5) manual searching of recent journals (including
	AIDS Education and Prevention, American Journal of Public Health, Archives of Pediatric and
	Adolescent Medicine, Health Psychology, Journal of Consulting and Clinical Psychology and
	JAMA).
Selection/inclusion	The following selection criteria were used:
criteria	• Evaluate an intensive educational, psychosocial or behavioural intervention(s) advocating
	sexual risk reduction for HIV prevention;
	• Use a randomised controlled trial or quasi-experimental design with rigorous controls;
	Have specific behavioural-dependent measures relevant to sexual risk;
	Have sample adolescents (pre-college); and
	Provide sufficient information to calculate effect sizes (ES).
	Interventions were excluded if they did not emphasis HIV content and extremely brief
	interventions for which message exposure was not ensured. In addition, interventions that used
	measures not specific to a particular sexual behaviour (e.g. general composite measures) were
	excluded.
Quality	No details reported.
assessment	
Types of studies	Randomised controlled trials and quasi-experimental studies using rigorous controls.
Number of studies	44 studies
Outcomes reported and/or	Sexual frequency, condom use, communications with sex partner, skills for condom use
used for assessing	negotiations, skills for condom use, acquired condoms or spermicide, diagnosed STDs (other
'effectiveness'	than HIV).
Review findings:	Relative to control, interventions significantly enhanced (1) participants' skills for sexual risk
	communications, (2) skills for condom use, (3) the quantity of sexual risk communications, and
	(4) participant's condom use. Interventions also reduced sexual frequency outcomes. Twelve
	studies examined behavioural markers; there was no constant intervention effect on acquisition
	of condoms or spermicide, being diagnosed with an STD other than HIV, or obtaining tests for
	HIV (one study).
Research	Future studies should be conducted across a wider range of geographical regions and risk
recommendations:	contexts.

• Johnson, W.D. et al. (2003) Interventions to modify sexual risk behaviours for preventing HIV infection in men who have sex with men

Data pool:	
Population	MSM of any age, race/ethnicity, sexual orientation, gender identity or nationality.
Setting(s)	Any.
Interventions	Interventions designed to promote sexual risk reduction in order to reduce transmission of
	HIV or other STDs.
Searches	Studies were sought from the following sources from 1988 to 1997:
	Nine electronic databases: AIDSLine, CINAHL, Dissertation Abstracts International,
	ERIC, HealthSTAR, Medline, PsycInfo, Sociofile, and Web of Science.
	Hand searches of relevant journals (including AIDS, AIDS and Behavior, AIDS
	Education and Prevention, American Journal of Public Health, International Journal of
	STD and AIDS, Journal of Acquired Immune Deficiency Syndromes), bibliographies of
	relevant articles, and other published reviews.
	Unpublished reports from researchers.
Selection/inclusion	Studies were considered eligible if they examined the effects of behavioural interventions to
criteria	reduce risk for HIV or STD transmission. Studies were reviewed against the following criteria:
	(1) outcome relevance (measurement of at least one of a list of behavioural or biologic
	outcomes, e.g., unprotected sex or incidence of HIV infections) and, (2) methodologic rigour
	(randomised controlled trials or certain strong quasi-experimental designs with comparison
	groups).
Quality	Studies were eligible based on study design; in addition allocation concealment was
assessment	assessed. Other details were not reported.
Types of studies	Controlled trials, randomised or quasi-experimental with a control group.
Number of studies	Thirteen eligible studies were identified.
Outcomes reported and/or	Intervention effects on unprotected sex.
used for assessing	
'effectiveness'	
Review findings:	A summary measure of intervention effects on reducing unprotected sex was favourable
	across all 12 studies (odds ratio [OR] 0.73; 95% CI: 0.60, 0.88). Effects were found to be
	homogenous across the studies. These results indicated that 23% fewer men reported
	unprotected anal sex after receiving an intervention than after the comparison condition.
Gaps and inconsistencies	Few studies focusing on MSM of colour, or HIV-seropositive men. Few behavioural outcome
identified by the review:	studies have examined individual-level interventions.
Research	Many more rigorous evaluations of HIV prevention efforts with MSM are needed to ascertain
recommendations:	with confidence the effects of specific intervention components, population characteristics,
	and methodologic features.

• Johnson, W.D. et al. (2005) HIV interventions research for men who have sex with men: a 7year update

Data pool:	
Population	Men who have sex with men.
Setting(s)	Any.
Interventions	Any behavioural intervention.
Searches	Resources included electronic databases (including Medline, PsycInfo, PubMed, AIDSLine,
	Web of Science), reviews and other studies in the HIV prevention literature, expert
	recommendations, hand searches of selected journals, and manuscripts and unpublished
	reports submitted by researchers in the field.
Selection/inclusion	Evaluated according to outcomes measured and study design. Eligible studies measured
criteria	intervention effects on behaviours understood to affect risk of HIV transmission (e.g.
	unprotected sex, condom use, number of sexual partners) and biological outcomes (including
	incidence of infection by HIV or other STD). Acceptable study designs were randomised
	controlled trials, and certain quasi-experimental designs (required to include independent
	comparison groups assigned without bias, and to include separate baseline data for
	intervention and control groups).
Quality	Studies were restricted by inclusion. No further details were reported.
assessment	
Types of studies	Randomised controlled trials, and certain quasi-experimental designs.
Number of studies	40 studies.
Outcomes reported and/or	Unprotected sex (defined as anal intercourse without a condom)
used for assessing	
'effectiveness'	
Review findings:	Versus minimal to no HIV prevention: 38 interventions reduced unprotected sex by 27%
	(95% CI: 15%, 37%). The proportion of participants reporting unprotected sex was reduced
	by 16% (95% CI: 10%, 21%). In subgroup analyses, significant favourable effects were
	observed for small group and community interventions. Effects among individual-level
	interventions were also favourable, but not statistically significant.
	Versus standard or other HIV prevention: 16 interventions reduced unprotected sex by
	17% beyond changes observed in standard or other HIV prevention interventions, and the
	proportion of participants reporting unprotected sex was reduced by 6%.
	Interventions for HIV-positive MSM: A 21% reduction in unprotected sex was observed
	across 15 studies which included HIV-positive MSM. Effects were more favourable across 7
	studies of small group interventions.
Gaps and inconsistencies	Some high-risk populations have been under researched: African American and Latino MSM,
identified by the review:	and MSM in countries where English is not the primary language.
Research	Further research on effective interventions for the above.
recommendations:	

• Johnson, W.D. et al. (2002) HIV prevention research for men who have sex with men: a systematic review and meta-analysis

Data pool:	
Population	Men who have sex with men (US-based interventions/studies)
Setting(s)	Any.
Interventions	Any. Most interventions were delivered in small group settings.
Searches	Online databases (only Medline, AIDSLine, PsychLit specified), reviews and other studies
	in the HIV prevention literature, expert recommendations, and manuscripts and
	unpublished reports submitted by researchers. The search was finished in June 1998.
Selection/inclusion	Studies were evaluated for inclusion according to the following criteria: (1) behavioural
criteria	outcome of interest including unprotected sex, use of male condoms, number of sex
	partners, and 17 other sexual, drug use and testing behaviours associated with risk for HIV
	transmission (not specified); (2) controlled trial, (either randomised or quasi-experimental
	with comparison group) that met the rigor criteria for the HIV/AIDS Prevention Research
	Synthesis (PRS) project.
Quality assessment	Included studies met the rigor criteria for the HIV/AIDS PRS project. No further details
	were reported.
Types of studies	See Selection/inclusion criteria.
Number of studies	Of 99 US studies, which met the inclusion criteria, ten studies were focused on MSM. Of
	the ten studies, one was focused on reduction in emotional distress after HIV testing and
	did not report behavioural outcomes. The analyses were therefore limited to nine
	controlled trials.
Outcomes reported and/or	The three most commonly reported outcomes across the studies were unprotected sex,
used for assessing	condom use, and number of sex partners.
'effectiveness'	
Review findings:	A summary effect measure of intervention effects on unprotected sex was found to be
	favourable compared to comparison conditions (odds ratio [OR], 0.69; 95% confidence
	interval [CI]: 0.56, 0.86). Favourable results were found for increasing condom use across
	4 studies, and for reducing number of sexual partners across 6 studies.
	Interventions that included interpersonal skills training had the most favourable effects on
	unprotected sex (unadjusted). Interventions were found to be most effective among
	studies that included younger participants and populations with a higher background
	prevalence of unprotected sex. Community studies had more favourable effects than small
	group interventions.
Gaps and inconsistencies	The following gaps were identified:
identified by the review:	Lack of studies examining MSM of ethnic minorities, HIV-seropositive men.
	 Lack of eligible studies with behavioural outcomes data focused on individual-level
	interventions for MSM in the US.
	Lich rick participants
	 Effects of small group interventions over extended period of time
	Community-level interventions at short follow-up
	Measurement at multiple follow-up times within studies.
Research recommendations:	More rigorous evaluations of HIV prevention efforts with MSM are needed to ascertain with
	confidence the effects of specific intervention components, population characteristics, and
	methodologic features.

• Logan T.K. et al. (2002) Women, sex, and HIV: Social and contextual factors, Meta-analysis of published interventions, and implications for practice and research.

Data pool:	
Population	Various populations including ethnic minorities
Setting(s)	Any
Interventions	Social and contextual factors
Searches	14 databases were searched: AIDSLINE, Social Science, Citation Index, Science Citation
	Index Expanded, ERIC, CINAHL, HealthSTAR, TROPAG & RURAL, Social Work Abstracts,
	SERFILE; Sociological Abstracts, Info Trac Web, PsychINFO, CC Search, and Medline.
	Hand searches were conducted for studies reported from 1990 to May 2000 in : AIDS, AIDS
	and Behaviour, AIDS Care, AIDS Education and Prevention, American Journal of Public
	Health, JAMA, Sexually Transmitted Diseases and The Lancet.
Selection/inclusion	Studies were included if they:
criteria	Targeted high-risk, heterosexual adult populations for HIV-prevention interventions;
	Included sexual behaviour outcome data or a proxy measure for condom use and/or
	number of sex partners;
	Had post-test outcomes for participants in an intervention group and a control group;
	• Targeted the same individuals for intervention and follow-up measures (which primarily
	excluded community level programs);
	• Examined the effects of an intervention (studies with HIV testing and counselling only
	or condom distribution only were not included);
	Included summary or inferential statistics sufficient for the calculation of effect sizes.
Quality assessment	See selection/inclusion criteria.
Types of studies	Randomised control trials
Number of studies	30 studies
Outcomes reported and/or	Condom use and number of partners
used for assessing	
'effectiveness'	
Review findings:	Gender was examined as a possible moderator, these analyses showed a
	nonsignificant negative effect size for men indicating that there is likely no difference in
	the outcome on number of partners for men in intervention and control groups.
	Focused comparison of effect sizes from continuous variables and those from the
	condom outcome variable revealed a negative and significant finding between days to
	follow-up and condom use (z = -3.52, p = .0002, one-tailed).
	Interventions that incorporated a culture – race component than those without it
	showed very little difference, for condom use (r = .085, p<.001 vs. r = .057, p<.001) and
	non-significance for number of partners (r = .090, p = .0192 vs. r = .060, p = .0307).
	• For both condom use and number of partners variables, the class of interventions with
	social factors had slightly larger weighted mean effect sizes that the class of
	interventions with no social factors (r = .082, p<.001 vs. r = .057, p<.001 [condom use]
	and r = .079, p = .0183 vs. r = .064, p = .0384 [partner, non-significant]).
	For condom use the greatest weighted mean effect size was found for those
	interventions that targeted low income women (r = .090, p<.001). The smallest
	weighted mean effect size was in studies targeting women from other clinics (r = .054,
	p = .0392)
	• For the partner dependent variable, the greatest weighted mean effect size found in the
	diffuse comparisons was found for the STD clinic class (r = .100, p<.01). This class only
	included two study effect sizes and must be viewed with caution. The other clinic class

	yielded the lowest weighted mean effect size (r = .029, p = .2912), which was not
	significant.
Gaps and inconsistencies	There were a small number of studies included in the meta-analysis.
identified by the review:	The included studies had such varied measures of condom use and number of sexual
	partners, varied periods of follow-up, and varied ways of estimating change that the effects
	may have been diluted.
	Coding the moderators could have been more precise had there been more information
	available in each primary study.
	(Also see review findings).
Research recommendations:	Recommendations for practice are to:
	Increase the comprehensiveness of HIV prevention interventions
	Advance female-controlled methods, and
	Change social and cultural norms regarding sexual behaviour.
	Future HIV prevention intervention research needs:
	More efforts to close the gap between research on HIV risk behaviour and intervention
	research;
	• To advance research for effective behavioural as well as biological interventions, and
	Better and more standardised reporting of intervention results.

• Merzel & D'Aflitti (2003) Reconsidering community-based health promotion: promise, performance, and potential.

Population Any Setting(s) Any community setting Interventions Community-based interventions Searches Searches were conducted on: Medline, PubMed, HealthComm-Key Selection/inclusion Studies were included if: criteria • They were based in the US since 1980. • Only articles published from 1980 to 2001. • Studies that targeted communities as opposed to select groups or limited settings. Quality Unclear Types of studies Unclear Number of studies 6 HIV prevention studies Outcomes reported and/or Outcomes of effectiveness were: condom use; carrying condoms; risky sexual behaviour.
Setting(s) Any community setting Interventions Community-based interventions Searches Searches were conducted on: Medline, PubMed, HealthComm-Key Selection/inclusion Studies were included if: criteria • They were based in the US since 1980. • Only articles published from 1980 to 2001. • Only articles published from 1980 to 2001. • Studies that targeted communities as opposed to select groups or limited settings. • Programmes that had some sort of published outcome. Quality Unclear Types of studies Unclear Number of studies 6 HIV prevention studies Outcomes reported and/or Outcomes of effectiveness were: condom use; carrying condoms; risky sexual behaviour.
Interventions Community-based interventions Searches Searches were conducted on: Medline, PubMed, HealthComm-Key Selection/inclusion Studies were included if: criteria • They were based in the US since 1980. • Only articles published from 1980 to 2001. • Studies that targeted communities as opposed to select groups or limited settings. • Programmes that had some sort of published outcome. Unclear Quality Unclear Types of studies Unclear Outcomes reported and/or Outcomes of effectiveness were: condom use; carrying condoms; risky sexual behaviour. used for assessing 'effectiveness'
Searches Searches were conducted on: Medline, PubMed, HealthComm-Key Selection/inclusion Studies were included if: criteria They were based in the US since 1980. Only articles published from 1980 to 2001. Only articles published from 1980 to 2001. Studies that targeted communities as opposed to select groups or limited settings. Programmes that had some sort of published outcome. Quality Unclear Types of studies Unclear Number of studies 6 HIV prevention studies Outcomes reported and/or Outcomes of effectiveness were: condom use; carrying condoms; risky sexual behaviour.
Selection/inclusion Studies were included if: criteria They were based in the US since 1980. Only articles published from 1980 to 2001. Only articles published from 1980 to 2001. Studies that targeted communities as opposed to select groups or limited settings. Programmes that had some sort of published outcome. Quality Unclear Types of studies Unclear Number of studies 6 HIV prevention studies Outcomes reported and/or Outcomes of effectiveness were: condom use; carrying condoms; risky sexual behaviour. used for assessing 'effectiveness'
criteria They were based in the US since 1980. Only articles published from 1980 to 2001. Studies that targeted communities as opposed to select groups or limited settings. Programmes that had some sort of published outcome. Quality Unclear assessment Unclear Types of studies Unclear Number of studies Outcomes reported and/or used for assessing Outcomes of effectiveness were: condom use; carrying condoms; risky sexual behaviour.
 Only articles published from 1980 to 2001. Studies that targeted communities as opposed to select groups or limited settings. Programmes that had some sort of published outcome. Quality Unclear Types of studies Unclear Unclear Number of studies 6 HIV prevention studies Outcomes reported and/or used for assessing 'effectiveness'
 Studies that targeted communities as opposed to select groups or limited settings. Programmes that had some sort of published outcome. Quality Unclear Types of studies Unclear Unclear Mumber of studies Outcomes reported and/or used for assessing 'effectiveness' Ei. ii. a law back targeted communities as opposed to select groups or limited settings. Programmes that had some sort of published outcome. Outcomes reported and/or Outcomes of effectiveness were: condom use; carrying condoms; risky sexual behaviour.
Programmes that had some sort of published outcome. Quality Unclear assessment Types of studies Unclear Number of studies Outcomes reported and/or used for assessing 'effectiveness' Determine the studies Determine the studies
Quality Unclear assessment Unclear Types of studies Unclear Number of studies 6 HIV prevention studies Outcomes reported and/or used for assessing 'effectiveness' Outcomes of effectiveness were: condom use; carrying condoms; risky sexual behaviour.
assessment Types of studies Types of studies Unclear Number of studies 6 HIV prevention studies Outcomes reported and/or used for assessing 'effectiveness' Outcomes of effectiveness were: condom use; carrying condoms; risky sexual behaviour.
Types of studies Unclear Number of studies 6 HIV prevention studies Outcomes reported and/or used for assessing 'effectiveness' Outcomes of effectiveness were: condom use; carrying condoms; risky sexual behaviour.
Number of studies 6 HIV prevention studies Outcomes reported and/or used for assessing 'effectiveness' Outcomes of effectiveness were: condom use; carrying condoms; risky sexual behaviour.
Outcomes reported and/or used for assessing Outcomes of effectiveness were: condom use; carrying condoms; risky sexual behaviour. 'effectiveness' Ein in the last of the formation of t
used for assessing 'effectiveness'
'effectiveness'
Review findings: Findings showed that:
HIV projects sought to modify social norms regarding risk behaviours and increasing the
social acceptability of risk avoidance. This was attempted through role modelling, self-
efficacy to engage in risk reducing behaviour, reinforcing educational messages.
These programmes were tailored to the specific communities.
The HIV interventions relied on informal methods of community involvement.
Gaps and inconsistencies • The studies were too limited to be able to calculate statistical significance.
identified by the review: • The interventions were also limited in that they primarily focused upon individuals, with
most people receiving mass education alone.
Many studies did not address normative and policy changes.
There was a propensity for HIV studies to demonstrate type II and type III errors.
Research • Community members, particularly peers should be closely involved in intervention design
recommendations: and delivery and messages should be tailored to target audiences through the use of real
role model stories of success.
There is a need to continually monitor norms and tailor interventions towards difference
populations.
• Future studies ought to be underpinned by theory.
Health promotion theories need to move beyond the individual level and incorporate the
social context in which behaviour occurs.
Future community programmes should include intensive educational and skills-building
interventions aimed at high-risk individuals and messages reaching across the entire
community.

• Mullen P.D. et al. (2002) Meta-analysis of the effects of behavioural HIV prevention interventions on the sexual risk behaviour of sexually experienced adolescents in controlled studies in the United States.

Data pool:		
Population	Adolescents (United States)	
Setting(s)	Schools and community settings	
Interventions	Behavioural and social interventions	
Searches	Published in Sogolow et al. (2002)	
Selection/inclusion	Included trials tested:	
criteria	Tested behavioural or social intervention programmes designed to prevent HIV infection	
	• Were conducted in the United States with adolescents of middle or high school age (13-19	
	years)	
	Were studied or reported from 1988 to 1998, and	
	Measured change in sexual risk behaviour or a biological indicator	
	 Used control or comparison groups with participants assigned to conditions either 	
	randomly or by other than random methods, such as matching that appeared to be	
	unbiased.	
	 Non-random tests met inclusion criteria if they included pre-test measures and either 	
	reported no baseline differences between study groups or controlled statistically for such	
	differences.	
Quality	See selection/inclusion criteria	
assessment		
Types of studies	Randomised control trials and guasi-experimental trials with control groups.	
Number of studies	20 studies	
Outcomes reported and/or	Condom use number of nartners, mixed behavioural index. Composite behavioural risk	
used for assessing	variable, incidence of STDs.	
'effectiveness'		
Review findings:	The studies that measured frequency of sex without condoms indicated that the	
	intervention group was significantly less likely than the control or comparison group to	
	have sex without condoms (IOR) 0.66 [95% CI] $0.55 - 0.79$, p<001).	
	The analysis for the second measure, number of partners, was not significant.	
	Both the studies that used a mixed behaviour index to measure sexual risk reported	
	significant positive effects (IOR) 0.66 [95% CI] 0.50 – 0.88 p< 01) and (IOR) 0.65 [95%	
	C[10, 50 - 0.85, p < 0.1)	
	 The composite behavioural risk variable (made up of one effect from each intervention) 	
	was comparable to the outcome for sex without condoms (IOR) 0.65, [95% CI] 0.50 –	
	0.85 p < 01)	
	 Neither of the two studies that measured incidence of STDs was significant from the null 	
	 Stratified analysis showed that of the five studies with 100% single ethnicity four consisted 	
	of African American, one was with Hispanic. None of these studies were based in	
	classrooms (other studies (mixed ethnicities) show no differences). These studies show	
	more protective outcomes that may be an indicator of the importance of a cultural fit with	
	the approach to this topic. It may also be possible that the discussions are more effective	
	with adolescents whose perspectives are more aligned, possibly due to ethnic similarities	
Gaps and inconsistencies	Although tests for publication bias were not significant, the funnel plot analysis suggests	
identified by the review	publication bias in some studies.	
Research	Euture research such to the to better understand the variable of acting	
NGOGAIGH		
recommendations	Future research ought to try to better understand the variable of setting.	

effects, this paper recommends that interventions with evidence of effectiveness be
implemented as designed, with minimal adaptation to meet local needs.

• Neuman M.S. (2002) Review and meta-analysis of HIV prevention intervention research for heterosexual adult populations in the United States.

Data pool:			
Population	Heterosexual adults (United States)		
Setting(s)	Any		
Interventions	Behavioural and social interventions		
Searches	Published in Sogolow et al. (2002)		
Selection/inclusion	Studies were included if they:		
criteria	• Tested HIV prevention interventions to reduce sex- and drug-related risk behaviours,		
	Reported positive, negative or null findings,		
	Were conducted anywhere in the world		
	Were reported from 1988 to present, and		
	Are published or unpublished		
	 Used a control or comparison group, collected at least post-intervention data (except for 		
	quasi-experimental studies that collected both pre- and post-intervention data), and have		
	either random assignment to study conditions, no apparent assignment bias, no statistical		
	difference between groups at baseline.		
Quality	See selection/inclusion criteria		
assessment			
Types of studies	See selection/inclusion criteria		
Number of studies	17 studies		
Outcomes reported and/or	Unprotected sex and condom use and STD incidence.		
used for assessing			
'effectiveness'			
Review findings:	The overall weighted average effect sizes show that risk-reduction interventions with		
-	heterosexual adults were associated with significant and modest reductions in unsafe		
	sexual behaviours.		
	• Stratification for participant grouping was significant (p = .030) which indicates that		
	interventions delivered to groups had stronger effects than those delivered to individuals.		
	 Individual level counselling was less likely to be a good primary prevention strategy. 		
	 More of the studies included in the biologic meta-analysis had interventions delivered in 		
	fewer than four sessions, and had the first follow-up conducted three months or more after		
	the intervention has ended. Fewer of the studies included the biologic meta-analysis		
	delivered interventions to individuals or focused exclusively on women.		
Gaps and inconsistencies	There are too few studies included to determine which intervention characteristics produce		
identified by the review:	significant protective effects.		
	 This study is limited because of the small number of included studies due to the fact that it 		
	focuses on only heterosexual adults.		
	The sample was too small to analyse for attrition.		
	There is a lack of institutional interventions; few community level interventions; few		
	interventions for men who do not inject drugs; a lack of interventions for behavioural		
	maintenance; a lack of interventions with an alternative to the use of the male condom:		
	and a need for component analyses to specify the components responsible for		
	effectiveness.		
	Also, few interventions conducted in residential, commercial or public access settings and		
	few interventions that teach men condom use skills and negotiation.		
Research	Further rigorous studies need to be conducted on the subject of interventions for		
recommendations:	heterosexual adults.		

•	It would be useful if researchers tried to establish consensus on behavioural outcomes
	and their measurement, use of intent to treat analysis, periodicity and duration of follow-up
	and length of recall periods for behaviours.
•	Included component analysis; report all outcome data related to a study's hypotheses;
	avoid an aggregate index to measure sexual risk and important descriptive variables.
•	Journals ought to allow the space for results to be reported comprehensively and
	information that cannot fit into a journal should be widely available on a website, the URL
	of which should be included in the journal article.

• Pinkerton et al. (2002b) A review of cost-effectiveness of interventions to prevent sexual transmission of HIV in the United States

Data pool:	
Population	Any (US).
Setting(s)	Any (US).
Interventions	Interventions to reduce the sexual transmission of HIV conducted in the US.
Searches	The authors conducted searches in electronic databases, including AIDSLINE, Medline,
	PsycInfo and HealthSTAR using combinations of the following keywords: costs, cost-
	effectiveness, cost-benefit, HIV prevention, AIDS prevention, and HIV infection.
Selection/inclusion	Articles had to be published in peer-reviewed journals.
criteria	
Quality	No details reported.
assessment	
Types of studies	Cost utility analyses, cost effectiveness analyses, 'effectiveness analyses' and cost analyses.
Number of studies	16 cost studies
Outcomes reported and/or	Quality adjusted life year (QALY), cost per HIV infection averted
used for assessing	
'effectiveness'	
Review findings:	Small-group, community-level, and outreach-based sexual risk reduction interventions, in
	particular, were shown to be cost saving and efficient strategies for the prevention of HIV
	transmission in moderate to high-risk populations.

• Robin L. et al. (2004) Behavioural interventions to reduce incidence of HIV, STD and pregnancy among adolescents: A decade in review. (Level 2)

Data pool:		
Population	Adolescents and ethnic minorities	
Setting(s)	Any	
Interventions	Behavioural interventions to reduce sexual risk-taking	
Searches	Six electronic databases were searched: Medline, Psychlit, Popline, ERIC, Sociofile and CHID.	
	Unspecified manual searches were also carried out.	
Selection/inclusion	Studies were included if they:	
criteria	Specified a theoretical basis for the intervention programme.	
	 Provided information about the intervention (e.g. duration, content, facilitators). 	
	Defined clear aims.	
	 Random assignment or matched control groups using a guasi-experimental design that 	
	matched units through stratification of risk behaviours and demographic variables.	
	 Studies were included if they had more than 16 participants per condition, followed 	
	participants for at least four weeks post-intervention, or had immediate pre- or post-test for	
	interventions lasting four months or longer.	
	 Had attrition rates of less than 40% at follow-up four weeks after the end of the 	
	intervention.	
	 Studies were also included if they measured sexual intention for those 13 years or 	
	vounger.	
Quality	See selection/inclusion criteria	
assessment		
Types of studies	Randomised control trials or quasi-experimental design with controls.	
Number of studies	24 studies	
Outcomes reported and/or	Abstinence: reduced sexual activity or number of sexual partners: less risky sexual behaviours:	
used for assessing	reduced number of pregnancy or repeat pregnancy; or reduced STD prevalence.	
'effectiveness'		
Review findings:	 Most interventions were based on multiple theories, however, no clear pattern of study 	
	method presented itself. The most commonly used theories were: social cognitive	
	theories; the Health Belief Model; social learning theories and; social influence theories.	
	 Among commonly measured behaviours (eight out of 12 studies), condom use was 	
	affected most consistently, and delayed initiation of sexual intercourse was least affected	
	(four of 11 studies).	
	Among measures less commonly used, pregnancy or partner impregnation was recorded.	
	Three studies recorded negative results; increased likelihood of males engaging in sexual	
	intercourse in the last month compared to the control group; increased reports of	
	pregnancy and STDs; less contraceptive use at most recent sexual encounter for females	
	who were sexually inexperienced at baseline or; less contraception efficiency among	
	females in the intervention group.	
	• Once studies were aggregated into the 17 programmes and variants, 10 programmes had	
	positive effects, one had mixed effects, four had null effects and two had negative effects.	
	• Studies with positive effects were published after 1995, included both males and females,	
	targeted African American youth and took place in schools.	
	 Programmes that produced positive effects used trained adult facilitators. 	
	Effective programmes included content that was specific to reducing sexual risk behaviour	
	(e.g. sexual refusal strategies and condom-use skills).	
	Effective programmes employed interactive and participatory educational strategies.	

	Programmes that emphasised skills to reduce specific behaviours, interventions more
	generally targeted toward increasing youth resiliency and competencies show promising
	approaches to reducing sexual risk behaviour.
Gaps and inconsistencies	Limitations included differences among groups that were reported but not controlled for in
identified by the review:	the analysis, and the unit of randomisation not matching the unit of analysis.
	Limits on the strength of effect sizes due to the differences in follow-up periods and
	possible limits on information available from the primary studies included.
Research	
recommendations:	Future programmes ought to focus upon appropriate skills, adapting programmes for length,
	being clear what constitutes a given programme and decide who ought to facilitate them
	Resiliency-based programmes ought to be further explored with regard to reducing the sexual
	risk behaviour of adolescents.
	Researchers should design studies that will clearly reveal which programme characteristics
	drive positive effects in sexual risk-reduction.

• Semaan, S. et al. (2002) A meta-analysis of the effect of HIV prevention interventions on the sex behaviors of drug users in the United States.

Data pool:	
Population	Drug users and condom use
Setting(s)	United States
Interventions	Reduce HIV drug and sexual risk behaviours and infection rates among drug users
Searches	Databases were searched: HIV/AIDS Prevention Research Synthesis (PRS), AIDSline,
	MEDLINE, HealthSTAR, PsycINFO, Sociofile, Educational Resource Information Centre, the
	Cumulative Index of Nursing and Allied Health, and Dissertation Abstracts. Manual searches
	were also carried out in 53 key journals, books, reports and reference lists.
Selection/inclusion	• Inclusion criteria were related to: scope, outcome measurement, methodological rigour,
criteria	and calculation of effect sizes.
	• Scope: focus on HIV/AIDS prevention, use of behavioural risk reduction as a means of
	prevention, and to a publication date as early as 1988.
	• Outcome measurement: evaluation data must have been collected on at least one of 20
	specified HIV-related behavioural outcomes, or 10 biological outcomes.
	• Methodological rigour: studies must be defined as experimental designs or certain
	quasi-experimental designs (experimental designs: individuals, groups or communities
	were randomly assigned and at least post-intervention data collected. Quasi-
	experimental: assignment was unbiased, not random, pre- and post-intervention data
	were collected and groups were not reported to be significantly different at baseline or
	that baseline differences had been controlled significantly.
	• Calculation of effect sizes: data had to be available to calculate effect sizes for
	increasing condom use and reducing unprotected sex.
Quality assessment	See selection/inclusion criteria
Types of studies	See selection/inclusion criteria
Number of studies	33 studies
Outcomes reported and/or	Behavioural outcomes for reducing the number of unprotected sex acts or increasing the use
used for assessing	of condoms.
'effectiveness'	
Review findings:	• The overall weighted effect size for 33 studies was protective and significant (OR, 0.86:
	95% Cl, 0.76-0.98).
	• The effects sizes were heterogeneous and studies with comparison groups that did not
	receive an intervention showed a significantly stronger effect (k = 3; OR, 0.60; 95% CI,
	0.43-0.85) than did studies with comparison groups that received a HIV prevention
	intervention (k = 30; OR, 0.91; 95% Cl, 0.81-1.03).
	• Four of the 33 studies in the meta-analysis had a significant positive effect. They had
	no distinguishing characteristics from the other studies.
Gaps and inconsistencies	Descriptive data comes from a single report
identified by the review:	Attrition was not examined
	• Did not include studies that were not published or reported due to non-significant
	results.
	Analysis was restricted to the first follow-up data.
Research recommendations:	Suggestions for future analysis (see gaps and inconsistencies identified by the review)

• van Empelen et al. (2003) Effective methods to change sex-risk among drug users: a review of psychosocial interventions

Data pool:	
Population	Drug users in or out of treatment
Setting(s)	Any
Interventions	Psychosocial or behavioural
Searches	Databases were searched: MEDLINE, PsycINFO, ERIC and Online Contents.
Selection/inclusion	Studies were included if they:
criteria	Were identifiable as psychosocial or behavioural in nature;
	• Aimed at a target group that was clearly identifiable as IDU, non-IDU or substance
	dependent.
	• Compared the intervention group to a comparison group (regardless of whether it was
	comparison or control).
Quality assessment	See selection/inclusion criteria
Types of studies	See selection/inclusion criteria
Number of studies	17 studies
Outcomes reported and/or	Psychosocial or behavioural (e.g. unsafe sex, number of sexual partners and condom use)
used for assessing	
'effectiveness'	
Review findings:	Only five out of 17 studies found changes in sexual behaviour or condom use skills.
	 Three studies found marginal trend with regards to condom use.
	• Most studies reported a reduction in unsafe behaviour for the entire population i.e.
	both experimental and comparison groups.
	Individual interventions were generally ineffective and were characterised by a lack of
	theoretical basis, short intervention period and limited number of sessions.
	 Group and community studies indicate that social norms and social support are important.
	• Observed behaviour by other IDUs resulted in a reduction in personal risk-taking by
	other IDUs.
	• Effective interventions were characterised by more sessions, longer sessions, or
	sustainability over time (e.g. community interventions).
	• Modelling (social learning theory) was used most often in the community
	interventions.
Gaps and inconsistencies	Most studies reported a reduction in unsafe behaviour for the entire population i.e. both
identified by the review:	experimental and comparison groups (both groups education in HIV prevention).
Research recommendations:	Future interventions should be theory driven as well as empirically based.
	Should use RCTs ideally with a comparison group receiving standardised treatment
	• Evaluations should be more comparable using standardised and specific behavioural
	outcomes and follow-up levels.
	Should evaluate the intervention impact at a psychosocial level.
	Intervention Mapping may help in the design of models.

• Weinhardt L.S. (2005) HIV diagnosis and risk behaviour

Data pool:	
Population	HIV-positive, HIV-negative, and untested participants, serodiscordant couples, and mixed samples
Setting(s)	Δηγ
	HIV diagnosis, voluntary courselling and testing (VCT)
Sograpon	Two detabases were searched: MEDLINE and DevelNEO. Eurther band searches were
Searches	Two databases were searched. MEDLINE and Psychine O. Fulliter Hand-searches were
	undertaken in AIDS, AIDS Care, AIDS Education and Prevention, American Journal of
	Public Health, Health Psychology, Journal of the American Medical Association, and
Selection/inclusion	Included studies provided:
criteria	Assessment of when, relative to data collection, at least some participants received
	an HIV-positive diagnosis or test result;
	Sexual behaviour outcome data or proxy measure (STD incidence)
	• Two or more assessments with the same participants, to allow examination of
	behaviour change over time;
	Data from a sample independent from earlier studies included in this review;
	• Summary or inferential statistics sufficient for the calculation of within-group effect
	sizes.
Quality assessment	See selection/inclusion criteria
Types of studies	Randomised control trials and quasi-experimental with controls
Number of studies	21 studies
Outcomes reported and/or	Number of partners, condom use, unprotected intercourse.
used for assessing	
'effectiveness'	
Review findings:	The findings show that:
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26)
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% CI, 0.37 to
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% CI, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% CI, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95%
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% CI, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% CI, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% CI, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% CI, 0.12 to 0.23; Ps< .001).
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% CI, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% CI, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% CI, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% CI, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15)
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% Cl, 1.14 to 1.48) were positive
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% Cl, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% Cl, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative participants (Ps< .005). Only the discordant couples had larger increases than untested
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% Cl, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative participants (Ps< .005). Only the discordant couples had larger increases than untested participants.
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% Cl, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative participants (Ps< .005). Only the discordant couples had larger increases than untested participants. Number of sexual partners (effect sizes n=14)
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% CI, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% CI, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% CI, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% CI, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% CI, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% CI, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative participants (Ps< .005). Only the discordant couples had larger increases than untested participants. Number of sexual partners (effect sizes n=14) The weighted mean effect size for the HIV-positive group was significantly positive (d+ =
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% Cl, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative participants (Ps< .005). Only the discordant couples had larger increases than untested participants. Number of sexual partners (effect sizes n=14) The weighted mean effect size for the HIV-positive group was significantly positive (d+ = 0.34; 95% Cl, 0.20 to 0.47), as was the HIV-negative effect size (d+ = 0.24; 95% Cl, 0.15
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% Cl, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative participants (Ps< .005). Only the discordant couples had larger increases than untested participants. Number of sexual partners (effect sizes n=14) The weighted mean effect size for the HIV-positive group was significantly positive (d+ = 0.34; 95% Cl, 0.20 to 0.47), as was the HIV-negative effect size (d+ = 0.24; 95% Cl, 0.15 to 0.34). Both groups showed greater change than the untested group (d+ = 0.07; 95% Cl,
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% Cl, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative participants (Ps< .005). Only the discordant couples had larger increases than untested participants. Number of sexual partners (effect sizes n=14) The weighted mean effect size for the HIV-negative effect size (d+ = 0.24; 95% Cl, 0.15 to 0.34). Both groups showed greater change than the untested group (d+ = 0.07; 95% Cl, 0.03 to 0.18). There was severe heterogeneity in the HIV-negative group. No data on
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps<.001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% Cl, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative participants (Ps< .005). Only the discordant couples had larger increases than untested participants. Number of sexual partners (effect sizes n=14) The weighted mean effect size for the HIV-positive group was significantly positive (d+ = 0.34; 95% Cl, 0.20 to 0.47), as was the HIV-negative effect size (d+ = 0.24; 95% Cl, 0.15 to 0.34). Both groups showed greater change than the untested group (d+ = 0.07; 95% Cl, 0.03 to 0.18). There was severe heterogeneity in the HIV-negative group. No data on number of sexual partners from serodiscordant couples studies.
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% Cl, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative participants (Ps< .005). Only the discordant couples had larger increases than untested participants. Number of sexual partners (effect sizes n=14) The weighted mean effect size for the HIV-positive group was significantly positive (d+ = 0.34; 95% Cl, 0.20 to 0.47), as was the HIV-negative effect size (d+ = 0.24; 95% Cl, 0.15 to 0.34). Both groups showed greater change than the untested group (d+ = 0.07; 95% Cl, 0.03 to 0.18). There was severe heterogeneity in the HIV-negative group. No data on number of sexual partners from serodiscordant couples studies. HIV and STD incidence (effect sizes n=6)
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% Cl, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative participants (Ps< .005). Only the discordant couples had larger increases than untested participants. Number of sexual partners (effect sizes n=14) The weighted mean effect size for the HIV-negative group was significantly positive (d+ = 0.34; 95% Cl, 0.20 to 0.47), as was the HIV-negative effect size (d+ = 0.24; 95% Cl, 0.15 to 0.34). Both groups showed greater change than the untested group (d+ = 0.07; 95% Cl, 0.03 to 0.18). There was severe heterogeneity in the HIV-negative group. No data on number of sexual partners from serodiscordant couples studies. HIV and STD incidence (effect sizes n=6) The weighted mean effect size among HIV-positive participants was significantly greater
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% CI, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% CI, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% CI, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% CI, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% CI, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% CI, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative participants (Ps< .005). Only the discordant couples had larger increases than untested participants. Number of sexual partners (effect sizes n=14) The weighted mean effect size for the HIV-positive group was significantly positive (d+ = 0.34; 95% CI, 0.20 to 0.47), as was the HIV-negative effect size (d+ = 0.24; 95% CI, 0.15 to 0.34). Both groups showed greater change than the untested group (d+ = 0.07; 95% CI, 0.03 to 0.18). There was severe heterogeneity in the HIV-negative group. No data on number of sexual partners (effect sizes n=6) The weighted mean effect size among HIV-positive participants was significantly greater than the HIV-negative and untested participants. These data indicated that the incidence
Review findings:	The findings show that: Unprotected intercourse (effect sizes n=26) The mean weighted effect sizes for the HIV-positive group (d+ = 0.44; 95% Cl, 0.37 to 0.51) and the serodiscordant couple group (d+ = 0.85; 95% Cl, 0.71 to 0.99) indicated significant risk reduction, and both were greater than the weighted mean effect size for the untested (d+ = 0.13; 95% Cl, 0.07 to 0.18) and HIV-negative participants (d+ = 0.18; 95% Cl, 0.12 to 0.23; Ps< .001). Condom use (effect sizes n=15) The mean weighted effect sizes for the HIV-positive group (d+ = 0.59; 95% Cl, 0.38 to 0.81) and the serodiscordant couple group (d+ = 1.31; 95% Cl, 1.14 to 1.48) were positive and significant and were greater than the mean weighted effect size for the HIV-negative participants (Ps< .005). Only the discordant couples had larger increases than untested participants. Number of sexual partners (effect sizes n=14) The weighted mean effect size for the HIV-positive group was significantly positive (d+ = 0.34; 95% Cl, 0.20 to 0.47), as was the HIV-negative effect size (d+ = 0.24; 95% Cl, 0.15 to 0.34). Both groups showed greater change than the untested group (d+ = 0.07; 95% Cl, 0.03 to 0.18). There was severe heterogeneity in the HIV-negative group. No data on number of sexual partners from serodiscordant couples studies. HIV and STD incidence (effect sizes n=6) The weighted mean effect size among HIV-positive participants was significantly greater than the HIV-negative and untested participants. These data indicated that the incidence of STD infection decreased among HIV-positive individuals (d+ = 0.18, 95% Cl, 0.08 to

	well as untested individuals (d+ = -0.05, 95% CI, -0.09 to -0.01).
	Key message 'This pattern of results is consistent with the idea that individuals testing HIV
	positive initially reduce their overall sexual activity compared to pre-diagnosis, and
	therefore do not increase the number of times they use condoms in the period immediately
	after being diagnosed. These results provide a positive out-look in terms of reduced
	transmission risk behaviour following an HIV diagnosis.'
Gaps and inconsistencies	• The characteristics of the counselling used in several studies were not described
identified by the review:	thoroughly, for example, the duration or specific procedures used (interventions
	varied from pre- and post-test counselling of specific duration to no details of
	counselling and counselling supplemented with other educational intervention such
	as peer-group discussion, videotaped presentations and partner counselling). As
	such effect sizes for types of counselling intervention provided could not be carried
	out.
Research recommendations:	Studies such as this help to inform how HIV tests are conducted, types of counselling
	provided, the perceived severity of an HIV infection, and maturation of cohorts affected by
	HIV. As these populations are continually changing it is necessary to carry out more
	studies on the effects of HIV diagnosis as an intervention for high risk sexual behaviour.

• Wilson, B. D. M. and Miller, R. M. (2003) Examining strategies for culturally grounded HIV prevention: A review.

Data pool:		
Population	Ethnic minority groups	
Setting(s)	Any	
Interventions	Culturally grounded interventions	
Searches	Four electronic databases were searched: PsychINFO, MEDLINE abstracts, AIDS Line and	
	ERIC.	
Selection/inclusion	Studies were included if they:	
criteria	Were published in peer-review journals,	
	Specifically aimed to prevent HIV transmission within a group or groups that reside within	
	the United States,	
	Evaluated and reported intervention results,	
	• Indicated that culture was a central or critical component of the intervention by including the	
	term within their title, abstract, or key words/subject headings,	
	• Were published between 1985 and 2001.	
Quality assessment	None provided	
Types of studies	Randomised control trials or comparison designs	
Number of studies	17 studies	
Outcomes reported	Aims to:	
and/or used for	Identify the populations studied	
assessing 'effectiveness'	Identify intervention methodologies used	
	Categorise the types of strategies used to make HIV prevention programmes culturally	
	specific and	
	Locate strategies for integrating culture in theoretical programme models of studies	
Review findings:	• Two types of strategies were in current use for integrating culture: attending to intervention	
	presentation and attending to intervention content.	
	Examples of meaningful methods of attending to culture were identified in the current	
	review. Nearly half the studies integrated ethnic-specific themes into the HIV programme	
	and two studies sought to respond to the unique cultural experiences of MSM from	
	racial/ethnic minority group. A couple of studies also sought to attend to sex and race.	
	Two culturally bound factors that have been identified as influential in HIV prevention	
	initiatives are societal preferences for heterosexual relationships over homosexual ones	
	and beliefs about gender roles that privilege men over women.	
	Future efforts to examine empirically the relationships between critical cultural	
	characteristics and HIV risk and protective behaviours and attitudes are an important	
	direction in the HIV field (Diaz's (1998) model of psychocultural self-regulation is a	
	promising step in this direction).	
Gaps and inconsistencies	There are very few efforts made to articulate how these strategies relate to or evolve from	
identified by the review:	the theoretical models guiding HIV prevention efforts and relatively few authors have	
	assessed whether culturally grounding programmes enhances their effects.	
	Searches were limited to abstracts, titles, and key words, as such studies referring to	
	culturally grounded interventions on in the body of the text may have been excluded.	
	Similarly, studies may have been excluded that targeted norms of a social group, such as	
Bassarah	gay men.	
Research	Defining authors	
recommendations:	Defining culture,	
	Developing models of best practice for intervention design, and	

٠	Developing theory that accommodates cultural concepts as part of the logic that underlies
	interventions.
Appendix F: Tables of primary studies within level one and two review papers

Table 6. Primary studies from review level evidence relevant to men who have sex with men (MSM).

							Review	ws						
Primary studies	Herbst J.H. et al. (2005)	Rees R et al (2004)	Johnson W.D. et al. (2002)	Johnson W.D. et al. (2003)	Johnson W.D. et al. (2005)	Semaan et al. (2002)	Albarracin et al. (2005)	Wilson & Miller (2003)	Merzel & D'Aflitti (2003)	Cohen et al. (2004)	Pinkerton et al (2002a)	Pinkerton et al (2002b)	Kegeles and Hart (1998)	Oakley et al (1996)
Individual														
Dilley et al (2002) (VCT)														
Explore et al. (2004)														
Gold and Rosenthal (1995)														
Gold and Rosenthal (1998)														
Imrie et al (2001)														
Patterson et al (2003)														
Piccano et al. (2001)														
Richardson et al (2004)														
Rosser (1990)														
Rotherham-Borus et al (2004) (substance using)														
Sorensen et al (2003) (substance abusers)														
Group														
Amirkhanian et al. (2003)														
Carballo-Dieguez et al (2004) (Latino)														
Choi et al. (1996)														
Coates et al. (1989) (PWHIV)														
Dockrell et al. (1999)														
Goldbaum et al. (1999) unpublished														
Harding et al (2004)														
Kalichman et al (2001)														
Kelly et al. (1989)														

							Revie	ws						
Primary studies	Herbst J.H. et al. (2005)	Rees R et al (2004)	Johnson W.D. et al. (2002)	Johnson W.D. et al. (2003)	Johnson W.D. et al. (2005)	Semaan et al. (2002)	Albarracin et al. (2005)	Wilson & Miller (2003)	Merzel & D'Aflitti (2003)	Cohen et al. (2004)	Pinkerton et al (2002a)	Pinkerton et al (2002b)	Kegeles and Hart (1998)	Oakley et al (1996)
Kelly et al. (1993)														
Kelly et al. (1996)														
Martin et al (2001)														
Miller (1995)														
Perry (1991)														
Peterson et al. (1996)														
Remafedi (1994)														
Roffman et al (1997)														
Roffman et al (1998)														
Rosser et al. (1990)														
Rosser et al. (2002)														
Rotherham-Borus et al (2001)														
Sampaio et al. (2002)														
Shoptaw et al. (in press) (meth-dependent)														
Stall et al (1999)														
Toro-Alfonso et al. (2002)														
Tudiver et al. (1992)														
Turner and Heywood (2000)														
Valdiserri et al. (1989)														
Holtgrave & Kelly (1997) *														
Pinkerton et al. (1997) *														
Community			•	•			•	•				•		
CDC ACDP (1999) (identified as non-gay MSM)														

							Review	ws						
Primary studies	Herbst J.H. et al. (2005)	Rees R et al (2004)	Johnson W.D. et al. (2002)	Johnson W.D. et al. (2003)	Johnson W.D. et al. (2005)	Semaan et al. (2002)	Albarracin et al. (2005)	Wilson & Miller (2003)	Merzel & D'Aflitti (2003)	Cohen et al. (2004)	Pinkerton et al (2002a)	Pinkerton et al (2002b)	Kegeles and Hart (1998)	Oakley et al (1996)
Elford et al. (2001)														
Flowers et al. (2002)														
Hoff et al (1997)														
Honnen and Kleinke (1990)														
Kegeles et al. (1996)														
Kegeles et al. (2002)														
Kelly et al. (1991)														
Kelly et al. (1992)														
Kelly et al. (1997)														
Miller et al. (1998)														
Rosser (1991)														
Shepherd et al. (1997)														
St Lawrence (1994)														
Zimmerman et al. (1997)														
Kahn & Haynes-Sanstad (1997) *														
Kahn et al. (2001) *														
Pinkerton et al. (1998) *														
Tao & Remafedi (1997) *														

Shaded areas denote reviews included in the original Evidence Briefing. * Cost-effectiveness primary studies.

Table 6 continued:

Primary studies	Intervention	Study design	Outcomes reported in review	Included in previous briefing?
Individual			I	
Dilley et al (2002) (VCT)	CT and self-justification sessions, diary intervention	RCT	Any UAI with non-primary partner of unknown status	No
Explore et al. (2004)	Counselling session	Non-RCT	None	No
Gold and Rosenthal (1995)	Self-justification group. Relapse prevention. Diary	RCT	Any UAI slip-ups during intervention period	Yes for MSM
Gold and Rosenthal (1998)	relapse prevention. Recall UAI slip-ups, AIDs prevention education, diary	RCT	Any UAI slip-ups during follow-up period	No
Imrie et al (2001)	Gay men project: counselling (CB)	RCT	UAI, STD incidence	No
Patterson et al (2003)	Social cognitive theory based intervention	Non-RCT	None	No
Piccano et al. (2001)	Telephone-based motivational counselling	Non-RCT	UAI, number of sexual partners, UOI, CU	No
Richardson et al (2004)	Prevention counseling	Non-RCT	None	No
Rosser (1990)	Individual HIV prevention counselling and safe sex education	RCT	Safe sex	No
Rotherham-Borus et al (2004) (substance using)	Individual HIV risk education	Non-RCT	None	No
Sorensen et al (2003) (substance abusers)	Intensive case management	Non-RCT	None	No
Group				
Amirkhanian et al. (2003)	Social network leader training programme: HIV prevention conversations	Non-RCT	CU during last intercourse, UAI	No
Carballo-Dieguez et al (2004) (Latino)	Sessions on various topics, discussion of diaries	Non-RCT	Unclear	Yes PWHIV
Choi et al. (1996)	API living well project: counselling, safe-sex education (culturally specific)	RCT	UAI, number of sexual partners	Yes for MSM
Coates et al. (1989) (PWHIV)	Stress reduction training with relaxation, health habit change, and stress management skills.	RCT	Number of sexual partners	Yes for PWHIV
Dockrell et al. (1999)	Small group work, work-books	Non-RCT	Unclear	No

Primary studies	Intervention	Study design	Outcomes reported in review	Included in previous briefing?
Goldbaum et al. (1999) unpublished	AIDS community demonstration: outreach distribution of risk reducing supplies	Non-RCT	stage of change for anal sex using condoms	No
Harding et al (2004)	Sessions on S&M scene	Non-RCT	Unclear	No
Kalichman et al (2001)	Support group	Non-RCT	Unclear	No
Kelly et al. (1989)	Project ARIES: risk reduction education, cognitive- behavioural self-management, sexual assertior training, development of relationship skills.	RCT	UAI, number of casual sexual partners, UOI, CU	Yes for MSM
Kelly et al. (1990)				
Kelly et al. (1993)	cognitive-behavioural group	RCT	Insertive and receptive UAI	No
Kelly et al. (1996)				
Martin et al (2001)	Effects on sexual risk-behaviour of being told that viral load is undetectable	Non-RCT	Unclear	No
Miller (1995)	Keep it up: relapse prevention workshop	Quasi- experimental	UAI	No
Perry (1991)	Counselling before and after HIV testing, stress training, interactive video, relaxation techniques	RCT	Unclear	No
Peterson et al. (1996)	AIDs risk education, self-management assertiveness training	RCT	Any UAI	Yes for MSM
Remafedi (1994)	University of Minnesota youth and AIDs project risk-assessment, counselling peer-education and referals	Non-RCT	UAI (insertive and receptive)	No
Roffman et al (1997)	Project ARIES: relapse prevention counselling relaxtion, coping skills and motivationa enhancement	RCT	UAI, number of sexual partners, UOI, CU	No
Roffman et al (1998)	Relapse prevention, Group counselling	Non-RCT	UAI, number of sexual partners, UOI, CU	No
Rosser et al. (1990)	Group discussion, workshops	RCT	Unclear	No

Primary studies	Intervention	Study design	Outcomes reported in review	Included in previous briefing?
Rosser et al. (2002)	Minnesota 500 men's study: risk factor seminar, risk-assessment and sex communication	RCT	UAI outside of serconcordant, long- term, monogomous relationship, CU	No
Rotherham-Borus et al (2001) HIV+	HIV risk education	Non-RCT	Unclear	No
Sampaio et al. (2002)	Project contact: workshop, skills training and discussion	RCT	UAI with any partner	No
Shoptaw et al. (in press) (meth-dependent)	Cognitive behavioural therapy and risk reduction intervention group	RCT	UAI	No
Stall et al (1999)	facilitator and counsellor for coping and interpersonal skills, substance misuse treatment	Non-RCT	UAI with non-monogomous partner	No
Toro-Alfonso et al. (2002)	Stop AIDs project: small group meeting, peer- education on sex, emotion, relationships, drug use.	Non-RCT	UAI	No
Tudiver et al. (1992)	Talking sex project: AIDS education and relationship, education and coping strategies, discussion, role-play	RCT	UAI, CU	Yes for MSM
Turner and Heywood (2000)	Work-shop sessions	Non-RCT	Unclear	No
Valdiserri et al. (1989)	AIDs prevention programme: group lecture, training skills	RCT	Number of sexual partners, CU (insertive and receptive)	Yes for MSM
Holtgrave & Kelly (1997) *	Cognitive-behavioural intervention	Cost utility analysis	QALY	No
Pinkerton et al. (1997) *	Safer sex skills building session plus educational session	Incremental CUA	QALY	No
Community				
CDC ACDP (1999) (identified as non-ga MSM)	Distribution and discussion of flyers and condoms	Non-RCT	Unclear	No
Dahl et al. (1997)	Community-wide, social marketing	Non-RCT	Unclear	No
Elford et al. (2001)	Peer-led, HIV prevention conversations	Non-RCT	A reduction in the rate of HIV transmission; a reduction in UAI with status-unknown partners; an increase in HIV testing and reduction in needle sharing	No

Primary studies	Intervention	Study design	Outcomes reported in review	Included in previous briefing?
Flowers et al. (2002)	Gay men's task force: peer-led	Non-RCT	UAI with casual partners	No
Hoff et al (1997)	Community mobilisation, social suppor education, outreach	t Non-RCT	Unclear	No
Honnen and Kleinke (1990)				
Kegeles et al. (1996)	Mpowerment project: Community-wide, peer-led	RCT	Any UAI	Yes for MSM
Kegeles et al. (2002)	Mpowerment replication of peer-led programme Outreach, group and publicity campaign	Non-RCT	UAI with non-primary partner	No
Kelly et al. (1991)	Community-wide, peer-led	RCT	UAI, multiple partners, CU	Yes for MSM
Kelly et al. (1992)	Community-wide, peer-led			
Kelly et al. (1997)	Community wide, peer-led group leaders distribution of literature	, RCT	UAI, number of sexual partners, CU	Yes for MSM
Miller et al. (1998)	Peer education in hustler bars	Non-RCT	Unclear	No
Rosser (1991)	Grim reaper public AIDs campaign: fear-based campaign	Non-RCT	UAI (including no AI and consistent CU)	No
Shepherd et al. (1997)	Peer-education	Non-RCT	Unclear	No
St Lawrence (1994)				
Zimmerman et al. (1997)	Empowerment training of AIDS prevention and health care: peer outreach	Non-RCT	Number of sexual partners	No
Kahn & Haynes-Sanstad (1997) *	Peer leader	Cost effectiveness analysis	Cost per HIV infection averted	No
Kahn et al. (2001) *	Community mobilisation	Cost effectiveness analysis	Cost per HIV infection averted	No
Pinkerton et al. (1998) *	Peer leader	Cost utility analysis	QALY	No
Tao & Remafedi (1997) *	Risk reduction counselling, peer education, HIV testing	Cost utility analysis	QALY	No

Shaded areas denote reviews included in the original Evidence Briefing. *Cost-effective primary studies

Table 7. Primary studies from review level evidence relevant to commercial sex workers

			Reviews						
Primary studies	Herbst J.H. et al. (2005)	Albarracin et al. (2005)	Exner et al. (1997)	Oakley et al. (1994)	Ickovics and Yoshikawa (1998)	Intervention	Study design	Outcomes reported in review	Included in previous briefing?
Individual		•					•		
Corby et al. (1990)		1							
Pickering et al. (1993)									
Group									
Ngugi et al. (1988)									
Community			•			•	·		
Asamoah-Adu et al (1994)						Community-wide, peer-led	Not RCT	Behavioural (not specified in review)	Yes
Bhave et al. (1995)						Community-wide, peer-led	Quasi-experimental	Behavioural (not specified in review)	Yes
Corby et al. (1993)									
Corby and Woltiski (1996)									
Dorfman et al. (1992)									
Ford, Wirwan, Fajans (1996)									
Fox et al. (1993)									
Fritz and Schaffer (1992)									
Ford et al. (2002)						Unclear from review	Unclear from review	Behavioural (not specified in review)	No
Ford et al. (1999)						Unclear from review	Unclear from review	Behavioural (not specified in review)	No
Hunt et al. (1992)									
Miller et al. (1998)						Hustler bar project: community, peer-led	non-RCT	UAI	No
Leonard et al. (2000)						Prevention among male clients of CSWs	Unclear from review	Behavioural (not specified in review)	No
Singh, Malavyla (1994)									

			Reviews						
Primary studies	Herbst J.H. et al. (2005)	Albarracin et al. (2005)	Exner et al. (1997)	Oakley et al. (1994)	Ickovics & Yoshikawa (1998)	Intervention	Study design	Outcomes reported in review	Included in previous briefing?
van Emeijden et al. (1994)									
Visrutaratna et al. (1995)									
Not Clear			•			•	•		
Papaevangelou et al. (1988)						Education intervention to Greek CSWs	Unclear from review	Behavioural (not specified in review)	No
Van Griensven et al. (1998)						Targeted intervention	Unclear from review	Behavioural (not specified in review)	No
Walden et al. (1999)						Measuring behaviour change in CSWs and their potential clients	Unclear from review	Behavioural (not specified in review)	No

Shaded areas denote reviews included in the original Evidence Briefing.

Table 8. Primary studies from review level evidence relevant to people with HIV

			Reviews						
Primary studies	Herbst et al. (2005)	Mathews et al. (2002)	Albarracin et al. (2005)	Johnson et al. (2005)	Cohen et al. (2004)	Intervention	Study design	Outcomes reported in review	Included in previous briefing?
Individual	•	•		•	•				
Landis (1992)						Patient referral, interview with counsellor to discuss the process of notification	Unclear	Unclear	Yes PWHIV
Levy 1998 (trial ongoing)						Partner notification	RCT	Unclear	No
Patterson et al., (2003)						Cognitive behavioural intervention	Unclear	Behavioural	No
Rotheram-Borus et al. (2004)	- ,					Improving health, issues of disclosure etc	RCT	Behavioural	No
Richardson et al., (2004)						Prevention counselling	Non-RCT	Behavioural	No
Sorensen et al., (2003)						Case management	Non-RCT	Behavioural	No
Toomey et al. (1998)					~	Partner notification	RCT	Condom use, number of sex partners, number of sexual encounters	No
Weinhardt et al. (1999)					~	Counselling and testing (standard model), counselling without testing, discordant couple counselling	Meta- analysis	Condom use, number of sex partners, number of sexual encounters	No
Wykoff et al. (1991)					~	Partner notification	Unclear	Condom use, number of sex partners, number of sexual encounters	No
Group		1							
Coates et al. (1989)						Stress reduction training with relaxation, health habit change, and stress management skills.	RCT	Number of sexual partners	Yes for PWHIV
Wolitski et al. (2005)	1					Peer-education	Non-RCT	Unclear	No
Cleary et al., (1995)						Counselling	Non-RCT	Unclear	No

Primary studies	Herbst et al. (2005)	Mathews et al. (2002)	Review Albarracin et al. (2005)	Johnson et al. (2005)	Cohen et al. (2004)	Intervention	Study design	Outcomes reported in review	Included in previous briefing?
Kelly et al., (1993)						cognitive behavioural skills or support group		Behavioural	No
Rotheram-Borus et al., (2001)						HIV coping, disclosure - small group	RCT	Behavioural	No
Kalichman et al., (2001)						support group	Unclear	Unclear	No
Community	•				-				
Fogarty et al. (2001)						peer-based condom and contraceptive promotion	Unclear	Unclear	No

Table 9. Primary studies from review level evidence relevant to African communities in the UK

						R	eviews						
Primary studies	Elwy (2002)	Albarracin (2005)	Wilson & Miller (2003)	Semaan et al. (2002)	Van Empelen et al. (2003)	Neumann (2002)	Mullen (2002)	Johnson B.T. (2003)	Pedlow & Carey (2003)	Kirby (2002)	Robin (2004)	Logan (2002)	lckovics & Yoshikawa (1998)
Individual											1		
Belcher et al. (1998)													
Birkel et al. (1993)													
Boekeloo et al. (1999)													
DeLamater et al. (2000)													
Gillmore et al. (1997)													
Grinstead et al. (1999)													
Mansfield et al. (1993)													
Orr et al. (1996)													
Wagstaff et al. (1999)													
Kamb et al. (1998)													
Group													
Branson et al. (1998)													
Carey et al. (1997)													
Choi et al. (1996)													
Cohen et al. (1991)													
Cohen et al. (1992)													
Cohen, MacKinnon et al. (1992)													
Cottler et al. (1998)													
Dancy et al. (2000)													
DiClemente & Wingood (1995)													
Dushay et al (2001)													
Herek et al (1998b)													
Herek et al. (1998)													

	Reviews													
Primary studies	Elwy (2002)	Albarracin (2005)	Wilson & Miller (2003)	Semaan et al. (2002)	Van Empelen et al. (2003)	Neumann (2002)	Mullen (2002)	Johnson B.T. (2003)	Pedlow & Carey (2003)	Kirby (2002)	Robin (2004)	Logan (2002)	Ickovics & Yoshikawa (1998)	
Hobfoll et al. (1994)														
Kalichman et al. (1996)														
Kalichman et al. (1997)														
Kalichman et al. (1999)														
Kelly et al. (1994)														
Malow et al. (1994)														
National Institute of Mental Health														
Nyamathi and Stein (1997)														
Nyamathi et al. (1994)														
Nyamathi et al. (1998)														
Nyamathi et al. (1998)														
O'Donnell et al. (1998)														
O'Leary et al. (1998)														
Schilling et al. (1991)														
Shain et al. (1999)														
Sikkema et al. (2000)														
Solomon and DeJong (1989)														
Adolescents					·					•	•			
Damond et al (1993)														
Fisher et al. (1998)														
Hewitt et al. (1998)														
Howard et al. (1990)														
Howard et al. (1992)														
Jemmott et al. (1992)														
Jemmott et al. (1998)			T	T				T						
Jemmott et al. (1999)														

	Reviews													
Primary studies	Elwy (2002)	Albarracin (2005)	Wilson & Miller (2003)	Semaan et al. (2002)	Van Empelen et al. (2003)	Neumann (2002)	Mullen (2002)	Johnson B.T. (2003)	Pedlow & Carey (2003)	Kirby (2002)	Robin (2004)	Logan (2002)	Ickovics & Yoshikawa (1998)	
Kirby et al. (1997)														
Levy et al. (1995)														
O'Donnell et al. (1999)														
Schinke et al. (1990)														
Rotherham-Borus et al. (1997)														
Rotherham-Borus et al. (1998)														
St Lawrence et al. (1995)														
St Lawrence et al. (1999)														
Stanton et al. (1996)														
Stanton et al. (1997)														
Stanton et al. (2000)														
Stevenson et al. (1995)														
Sturdevant et al. (1998)														
Walter & Vaughan (1993)														
Weeks et al. (1995)														
Weeks et al. (1997)														
Workman et al. (1996)														
Community	•	•		•		•		•		•	•			
Deren et al. (1995)														
Flaskerud and Nyamathi (1990)														
Lauby et al. (2000)														
O'Donnell et al. (1995)														
Sellers et al. (1994)							I			Ī				
Solomon and DeJong (1989)														

Shaded areas denote reviews included in the original Evidence Briefing.

Table 10. Primary studies from review level evidence relevant to voluntary counselling and testing

		Reviews*													
Primary studies	Weinhart (2005)	Elwy (2002)	Vidanapathirana et al. (2005)	Albarracin et al. (2005)	Neumann (2002)	Cohen et al. (2004)	Pinkerton et al. (2002a)	Pinkerton et al. (2002b)	Semaan et al. (2002)	van Empelen et al. (2003)	Included in previous briefing?				
MSM		1													
Coates, Morin, McKusick (1987)											Yes				
Colfax et al. (2002)											No				
Fox, Odaka, Brookmeyer and Polk (1987)											Yes				
Huggins et al. (1991)											Yes				
McOwan (2002)											Yes MSM				
Ostrow et al. (1989)											Yes				
Schechter et al. (1988)											Yes				
van Griensven et al. (1989)											Yes				
Zapka et al. (1991)											Yes				
Tao & Remafedi (1997) (CUA)											No				
Injecting drug users (IUDs)															
Casadonte et al. (1990)											Yes				
Cottler et al. (1998)											No				
Gibson et al. (1998)											No				
Kotranski et al. (1998)											No				
Magura et al. (1990)											No				
McCoy et al. (1998)											No				
Women and heterosexual couples															
Allen, Serufilia et al. (1992a)											Yes				
Allen, Tice et al. (1992b)											Yes				
Amaro (2001)											Ethnic minority				
Apanovitch (2003)											No				
Flaskerud et al. (1997)											No				
Kamenga et al. (1991)											Yes				
Padian et al. (1993)											Yes				

		Reviews*												
Primary studies	Weinhart (2005)	Elwy (2002)	Vidanapathirana et al. (2005)	Albarracin et al. (2005)	Neumann (2002)	Cohen et al. (2004)	Pinkerton et al. (2002a)	Pinkerton et al. (2002b)	Semaan et al. (2002)	t van Empelen et al. (2003)	Included in previous briefing?			
Roth et al. (2001)											No			
Simpson (1998)											No			
Simpson (1999)											No			
CSWs										<u>.</u>				
Pickering et al. (1993)											Yes			
Harcourt (1998)											No			
STI clinic attenders														
Bentley et al. (1978)											No			
George, Green and Murphy (1998)											No			
Otten et al. (1993)											Yes			
Mixed population														
Beck (1987)											No			
Beck (1990)											No			
Cleary et al. (1991)											Yes			
Dwyer (1988)											No			
Joshi (1988)											No			
Maayan (1989)											No			
Mills et al. (1986)											No			
Tesoriero (1992)											No			
Turner (1987)											No			
VCT group (2000)											No			
Wenger et al. (1991)											Yes			
Other														
Deren, Beardsley, Coyle and Singer (1998) (IDU & crack users)											No			
Jackson et al. (1997) (Mombasa truck workers)											No			
Poulin (1997)											No			
Robles et al. (1998) (out of treatment IDU)											No			

		Reviews*												
Primary studies	Weinhart (2005)	Elwy (2002)	Vidanapathirana et al. (2005)	Albarracin et al. (2005)	Neumann (2002)	Cohen et al. (2004)	Pinkerton et al. (2002a)	Pinkerton et al. (2002b)	Semaan et al. (2002)	van Empelen et al. (2003)	previous briefing?			
Wenger et al. (1992) (college students)											Yes			
Kamb et al. (1998)											No			
Weinhardt et al. (1999)											No			

* The VCT studies from the original Evidence Briefing were too numerous to include here. For details of the original VCT studies please refer to Ellis et al. (2003).

Table11. Primary studies from review level evidence included in cost-effectiveness reviews

			Review	vs				
Primary studies	Cohen et al. (2004)	Pinkerton et al. (2002a)	Pinkerton et al. (2002b)	Pinkerton & Holtgrave (2002)	Johnson- Mascotti et al. (2003)	Farnham et al. (2002)	Intervention	Outcomes reported
Individual							·	·
Kamb et al. (1998)	~						Counselling and testing (standard model)	
O'Donnell et al. (1998)	~						Videos in STD clinics	
Toomey et al. (1998)	~						Partner notification	Condom use, number of sex partners, number of sexual encounters
Weinhardt et al. (1999)	~						Couselling and testing (standard model), counseling without testing, discordant couple counselling	Condom use, number of sex partners, number of sexual encounters
Wykoff et al. (1991)	~						Partner notification	Condom use, number of sex partners, number of sexual encounters
Group							•	•
Coyle et al. (1999)	~						School-based education, multiple sessions	
Kalichman et al. (1995)					~		Risk education, skills training, self-management and problem solving training	Number of safe sex conversations with partner, unprotected sex, condom use
Kelly et al. (1994)	~						Group counselling (multiple sessions)	
Kelly et al. (1996)					~		Cognitive behavioural skills training	number of partners, unprotected sex
Otto-Salaj et al. (2001)					~		Cognitive behavioural skills training	Unprotected sex, condom use, number of sexual partners
Shain et al. (1999)	~						Group counselling (multiple sessions)	
St Lawrence et al. (1995)	✓						Group counseling for youth	

				Revie	ws				
Primary studies	Cohen (2004)	et a	I. Pinkerton et al (2002a)	. Pinkerton et al. (2002b)	Pinkerton & Holtgrave (2002)	Johnson- Mascotti e al. (2003)	Farnham e tal. (2002)	t Intervention	Outcomes reported
Susser et al. (1998)						~		Cognitive behavioural skills training	Vaginal Episode Equivalent sexual risk index
Wang et al. (2000)		/						School-based education, multiple sessions	
Weinhardt et al. (1998)						~		Sexual assertiveness skills training, HIV/AIDS information, HIV risk behaviour feedback	Behavioural intentions, HIV risk behaviour, HIV preventative behaviour
Community/social networ	k			1	L	1			
Cohen et al. (1999)		/			✓			Condom social marketing	
Kegeles et al. (1996)		/						Community mobilisation (Mpowerment Project)	
Kelly et al. (1992)		/						Opinion leader programme	
Lauby et al. (2000)		/						Community mobilisation	
Sikkema et al. (2000)		/						Street outreach	
Wendell et al. (2003)		/						Street outreach	
Structural									
Dubois-Arber et al. (1997)		/						Mass media campaign	
O'Donnell et al. (1999)		/						Youth supervision programme	
Group									
Holtgrave & Kelly (1996) *		/		✓				Cognitive-behavioural intervention	cost per QALY saved
Holtgrave & Kelly (1997) *				✓	~			Cognitive-behavioural intervention	cost per QALY saved
Johnson-Mascotti et al. (2000)				✓	~			Educational session plus peer leader training	cost per QALY saved
Pinkerton et al. (1997) *				✓	~			Safer sex skills building session plus educational session	cost per QALY saved
Pinkerton et al. (2000) *				~	~			Workshop focusing on knowledge, attitudes and behaviour	cost per QALY saved

			Review	ws				
Primary studies	Cohen et al. (2004)	Pinkerton et al. (2002a)	Pinkerton et al. (2002b)	Pinkerton & Holtgrave (2002)	Johnson- Mascotti et al. (2003)	Farnham et al. (2002)	Intervention	Outcomes reported
Pinkerton et al. (2001) *			~				Cognitive-behavioural intervention	cost per QALY saved
Pinkerton et al. (2002) *			~				7-session small group intervention	cost per QALY saved
Community/social networ	k				•	•		
Bedimo et al. (2002) *	✓	\checkmark	\checkmark	\checkmark			Condom distribution	cost per QALY saved
Kahn & Haynes-Sanstad (1997) *		~	~				Peer leader	Cost per infection averted
Kahn et al. (2001) *	~	✓		\checkmark			Community mobilisation	Cost per infection averted
Moses et al. (1991)				~			Free condoms and health education to promote condom use	Cost per infection averted
Norton et al. (1998) *			~				Basic outreach and intervention services	Annual cost
Pinkerton et al. (1998) *	✓	✓	\checkmark	\checkmark			Peer leader	cost per QALY saved
Pinkerton et al. (2000) *			~				Basic outreach and intervention services	HIV infections averted
Tao & Remafedi (1997) *		~	~	✓			Risk reduction counselling, peer education, HIV testing	cost per QALY saved
Counseling/Testing					•	•	•	
Holtgrave et al. (1993)						~	HIV counseling, testing, refferal and partner notification	Benefit-cost ratio
Varghese et al. (1999)						~	HIV counseling, testing, refferal and partner notification	Cost per infection averted
Gelles (1993)						~	Direct screening and assumed behaviour change of blood donors	Cost per infection averted
Kahn et al. (1992)						✓	HIV CT of IDUs.	Cost per infection averted
Owens et al. (1995)						✓	HIV CT of surgeons.	Cost per year of life saved
LaCroix & Russon (1996)		ľ				✓	HIV CT of hospital patients	Cost-benefit ratio
Owens et al. (1996)						✓	HIV CT of hospital patients	Cost per year of life saved

			Review	vs				
Primary studies	Cohen et al. (2004)	Pinkerton et al. (2002a)	Pinkerton et al. (2002b)	Pinkerton & Holtgrave (2002)	Johnson- Mascotti et al. (2003)	Farnham et al. (2002)	Intervention	Outcomes reported
Brandeau et al. (1993)						✓ HIV CT of women of age		Cost-benefit ratio
Brandeau and Owens (1994)						~	HIV CT of women of childbearing age plus relapse	Cost-benefit ratio
Cleary et al. (1987)						~	HIV CT for couples obtaining marriage licences	Cost per infection averted
McKay & Phillips (1991)						~	HIV CT for couples obtaining marriage licences	Cost per infection averted
Weinstein et al. (1989)						~	HIV CT for couples obtaining marriage licences	cost per QALY saved

Table 12: Theory-based interventions

Primary studies	Theoretical interventions													
	SLT	SAM	DI	СВТ	ARRM	RP	ET	НВМ	ME/I	PST	PM	SI	TRA/ SLT **	М
мѕм	1		1		1						1	1	1	1
Kelly et al. (1989)														•*0x
Kelly et al. (1991)			•*X∎											
Kelly et al. (1992)														
Kelly et al. (1997)			•*X∎											
Choi et al. (1996)														*X\$
Kegeles et al. (1996)			*∎											
Peterson et al. (1996)														•*0x\$
Roffman et al. (1996)						*								
Zimmerman et al. (1997)							*							
Roffman et al. (1998)														*
Goldbaum et al (1999) unpublished														*
Elford et al. (2001)			*X											
Flowers et al. (2002)			*											
Picciano et al. (2001)									*X					
Kegeles et al. (2002)			*X											
Gold & Rosenthal (1995)						*								
Gold & Rosenthal (1998)						*								
Stall et al. (1999) (MSM substance users)														*X
Imrie et al. (2001)														*
Dilley et al. (2002)														*
Shoptaw et al. (in press) (MSM meth users)						*								
Miller et al. (1995)						*0								
Miller et al. (1998)						Х								
Toro-Alfonso et al (2002)								*						
Amirkhanian et al. (2003)														*
PWHIV														
Kelly et al. (1993)				*x +										
Butler et al., (2003)														∎
Rorheram-Borus et al 2001a		•+												
Rorheram-Borus et al 2001b		•+												
Rorheram-Borus et al (in press)		∎												
Kalichman (2000)				•									<u> </u>	
Kalichman (2001)		1		+										
		1	1	1			1	1						
BMEG/Africans		•		•	•		•	•						
Herek et al. (1998)													\$	
DiClemente & Wingood (1995)													\$	
Howard et al. (1990)		1						1						

Primary studies	Theoretical interventions													
	SLT	SAM	DI	СВТ	ARRM	RP	ET	НВМ	ME/I	PST	PM	SI	TRA/ SLT **	М
Howard M. (1992)														
Jemmott et al. (1992) (young people)													\$�▲	
Jemmott et al. (1998) (young people)				\$▲										
Kalichman et al. (1996) (women)													\$	
Kalichman et al. (1997)				۲										
Kalichman et al. (1993) (women)													\$	
Kalichman et al. (1999)									۲					
Kirby et al. (1991) (young people)														A
Kirby et al. (1997) (young people)														A
Magura et al. (1994) (adolescents)														
O'Donnell et al (1995) (women)													\$�	
O'Donnell et al (1999) (young people)														
Shain et al. (1999) (women)													\$	
Stanton et al. (1996) (young people)														
Stanton et al. (1997) (young people)														
Stanton et al. (1998)					۲									
Stanton et al. (2000)													\$	
Stevenson et al. (1995)													\$	
St Lawrence et al. (1995) (young people)														⊗ ⊾
St Lawrence et al. (1999) (young people)														
Sturdevant et al. (1998)													\$	
Walter et al. (1993) (young people)														
Coyle et al. (1999) (young people)														
Weeks et al. (1995) (young people)														
Levy et al (1995) (young people)														
Weeks et al. (1997) (young people)														
Boyer et al. (1997) (young people)														♦▲
Eisen et al. (1990)														
Drug users														
Baker et al (1994)									۲					
Baker et al (1993)														0

- 185 - HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission, 2006.

Primary studies	Theoretical interventions															
	SLT	SAM	DI	СВТ	ARRM	RP	ET	НВМ	ME/I	PST	PM	SI	TRA/ SLT **	М		
Gibson et al., (1999a)										0						
Gibson et al., (1999b)										0						
O'Neill et al., (1996)														0		
Malow et al. (1994)					۲											
Margolin et al., (2003)					+											
El-bassel et al (1995)														0		
El-bassel et al (1997)																
Magura (1995)										0						
Marlow et al., (1994)					0											
McCoy et al., (1998)														0		
McCusker et al., (1992)														0		
Sorensen et al (1994a)														0		
Sorensen et al (1994b)														0		
Collins et al., (1999)														0		
Jamner et al., (1997)														0		
Rietmeijer et al., (1996)														0		
CSWs				•									· · · · ·			
Miller et al. (1998)			*													
Hanenberg et al., (1994)			Ħ													
Ngugi et al., (1996)			Ħ													
Other populations/mixed																
Santelli et al., (1995)	ж															
CDC (1999)														ж		

SLT - Social Learning Theory; SAM - Social Action Model; DI - Diffusion of innovations; CBT - Cognitive Behavioural Therapy; ARRM - AIDS risk-reduction Method; RP - Relapse Prevention; ET - Empowerment Theory; HBM - Health Belief Model; ME/I - Motivational Enhancement/Interviewing; PST - problem solving therapy; TRA - theory of reasoned action; PM - protection Motivation; SI - social influence; M - multiple therapies used.

Reviews

- * Herbst et al. (2005)
- 0 Johnson et al. (2002)
- x Johnson et al. (2005)
- \$ Wilson and Miller (2003)
- ▲ Robin et al. (2004)
- - Johnson et al. (2003)
- O Van Empelen et al. (2003)
- I Elkavich et al. (2005)
- ◆ Kelly et al. (2002)
- + Woltiski et al. (2005)
- # Ross et al. (2002)

Pedlow & Carey (2003) (21 of 22 studies underpinned by theory, almost all by more than one theory)



Jennifer Downing Lisa Jones Penny A. Cook Mark A. Bellis

Centre for Public Health Castle House North Street Liverpool John Moores University L3 2AY Tel: 0151 231 4447 Fax: 0151 231 4515 Email: j.downing1@ljmu.ac.uk

- 188 - HIV prevention: a review of reviews assessing the effectiveness of interventions to reduce the risk of sexual transmission, 2006.